HANDBOOK OF Arsenic Toxicology

Edited by S. J. S. Flora





Handbook of Arsenic Toxicology

In the memory of my mother Paramjit who always told me, 'do not be successful but be valuable'

Handbook of Arsenic Toxicology

Edited by S. J. S. Flora



AMSTERDAM • BOSTON • HEIDELBERG • LONDON • NEW YORK OXFORD • PARIS • SAN DIEGO • SAN FRANCISCO SINGAPORE • SYDNEY • TOKYO Academic Press is an imprint of Elsevier



Academic Press is an imprint of Elsevier 32 Jamestown Road, London NW1 7BY, UK 525 B Street, Suite 1800, San Diego, CA 92101-4495, USA 225 Wyman Street, Waltham, MA 02451, USA The Boulevard, Langford Lane, Kidlington, Oxford OX5 1GB, UK

Copyright © 2015 Elsevier Inc. All rights reserved.

No part of this publication may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or any information storage and retrieval system, without permission in writing from the publisher. Details on how to seek permission, further information about the Publisher's permissions policies and our arrangements with organizations such as the Copyright Clearance Center and the Copyright Licensing Agency, can be found at our website: www.elsevier.com/permissions.

This book and the individual contributions contained in it are protected under copyright by the Publisher (other than as may be noted herein).

Notices

Knowledge and best practice in this field are constantly changing. As new research and experience broaden our understanding, changes in research methods, professional practices, or medical treatment may become necessary.

Practitioners and researchers must always rely on their own experience and knowledge in evaluating and using any information, methods, compounds, or experiments described herein. In using such information or methods they should be mindful of their own safety and the safety of others, including parties for whom they have a professional responsibility.

To the fullest extent of the law, neither the Publisher nor the authors, contributors, or editors, assume any liability for any injury and/or damage to persons or property as a matter of products liability, negligence or otherwise, or from any use or operation of any methods, products, instructions, or ideas contained in the material herein.

ISBN: 978-0-12-418688-0

British Library Cataloguing-in-Publication Data

A catalogue record for this book is available from the British Library

Library of Congress Cataloging-in-Publication Data

A catalog record for this book is available from the Library of Congress

For Information on all Academic Press publications visit our website at http://store.elsevier.com/

Typeset by MPS Limited, Chennai, India www.adi-mps.com

Printed and bound in the United States of America





Contents

Foreword xvii Preface xix Acknowledgements xxiii List of Contributors xxv

1.	Arsenic: Chemistry, Occurrence, and Exposure Swaran J.S. Flora		1
	1.1	Introduction	1
	1.2	Chemistry of Arsenic	2
	1.3	Arsenic Minerals and Compounds	8
	1.4	Organoarsenicals	11
	1.5	Arsenic Mobilization in the Environment	13
	1.6	Sources of Arsenic in the Biosphere	16
	1.7	Arsenic in Hydrothermal and Geothermal Fluids	24
	1.8	Arsenic Release from Mining and Mineral Processing	29
	1.9	Global Occurrence of Arsenic in Ground Water	32
	1.10	Methods of Arsenic Removal from Water	37
	1.11	Conclusions	40
	Refe	erences	41
2.	Gro Effe Sk A	ound Water Arsenic Contamination and Its Health acts in Bangladesh Akhtar Ahmad and Manzurul Haque Khan	51
	2.1	Introduction	51
	2.2	Arsenic Contamination in Bangladesh Ground Water	52
	2.3	Extent of Arsenic Contamination in Bangladesh	54

	2.4	Arsenic in Different Environmental Media of Bangladesh	55
	2.5	Health Effects of Arsenic Toxicity in Bangladesh	59
	2.6	Epidemiology of Arsenicosis in Bangladesh	66
	2.7	Management of Arsenicosis Patients in Bangladesh	66
	2.8	Socio-Cultural Aspects of Arsenicosis in Bangladesh	67
	2.9	Conclusions	68
	Refe	erences	68
3.	Arse Wa ⁻ Shai	enic and Fluorescent Humic Substances in the Ground ter of Bangladesh: A Public Health Risk fi M. Tareq	73
	3.1	Introduction	73
	3.2	Materials and Methods	74
	3.3	Results and Discussion	77
	3.4	Conclusions	90
	Refe	erences	91
4.	Ars Sant	enic Risk Assessment thini Ramasamy and Janice S. Lee	95
	4.1	Introduction	95
	4.2	Arsenic Chemistry	97
	4.3	Arsenic Occurrence and Exposure	97
	4.4	Hazard Identification	98
	4.5	Arsenic Metabolism, Mode of Action, and Physiologically	
		Based Pharmacokinetic Modeling	103
	4.6	Potential Sources of Susceptibility	106
	4.7	Dose-Response Approaches	108
	4.8	Risk Characterization	108
	Disc	claimer	111
	Refe	prences	111

5.	Eva Ads Lucy Isaa	luation of Novel Modified Activated Alumina as sorbent for Arsenic Removal / Mar Camacho, Sundaravadivelnathan Ponnusamy, c Campos, Thomas A. Davis, and Shuguang Deng	121
	5.1	Introduction	121
	5.2	Materials and Methods	122
	5.3	Results and Discussion	125
	5.4	Conclusions	134
	Ack	nowledgments	135
	Refe	erences	135
C			407
6.	Hea	aith Effects Chronic Arsenic Toxicity . Guha Mazumder	137
	6.1	Introduction	137
	6.2	Dermatological Manifestations	138
	6.3	Epidemiological Study of Dermatological	
		Manifestations	144
	6.4	Pregnancy Outcome	162
	6.5	Arsenicosis and Cancer	163
	6.6	Diagnosis	165
	6.7	Treatment	168
	Refe	erences	171

7.	Changing Concept of Arsenic Toxicity with Development of Speciation Techniques Badal Kumar Mandal	179
	7.1 Introduction	179
	7.2 Conclusions	197
	References	197

8.	Med Jyot	hanism for Arsenic-Induced Toxic Effects irmoy Ghosh and Parames C. Sil	203
	8.1	Introduction	203
	8.2	Biological Consequences of Chronic Arsenic	
		Exposure in Humans	204
	8.3	Arsenic Metabolism	205
	8.4	Pathophysiology of Arsenic Toxicity	206
	8.5	Mechanism for the Toxic Effects of Arsenic	211
	8.6	Preventing Arsenic-Induced Toxic Effects by Antioxidants	214
	8.7	Conclusions and Future Directions	220
	Refe	rences	221
9.	Arse	enic-Induced Mutagenesis and Carcinogenesis:	
	A Po Sma	ossible Mechanism rajit Maiti	233
	9.1	Arsenic, a Potent Mutagen and Carcinogen	233
	9.2	Epidemiological Perspectives of Arsenic-Induced Human Cancers	234
	9.3	Arsenic-Associated Metabolism and Carcinogenesis in	
		Animal Models	235
	9.4	Human Arsenic Carcinogenesis	251
	9.5	Conclusions and Future Directions	264
	Ackı	nowledgments	265
	Refe	rences	265
10.	Arse Mar	enic Through the Gastrointestinal Tract ta Calatayud and José Moisés Laparra Llopis	281
	10.1	General Aspects	281
	10.2	Arsenic in Foods: Household Processing and	
		Toxicological Risk	282

	10.3 Role of the Gut in Arsenic Toxicity: Bioavailability and	
	Intestinal Health	283
	10.4 Arsenic-Induced Metabolic/Immune Toxicity	289
	10.5 Conclusions and Future Perspectives	291
	References	292
11.	Cutaneous Toxicology of Arsenic	301
	Katherine Marchiony Hunt, Ritesh Kumar Srivastava, and Mohammad Athar	
	11.1 Introduction	301
	11.2 Epidemiology	302
	11.3 Clinical Manifestations	303
	11.4 Histopathology	304
	11.5 Molecular Pathogenesis	304
	11.6 Treatment	310
	References	311
12.	Arsenic-Induced Liver Injury Amal Santra	315
	12.1 Introduction	315
	12.2 Source of Exposure	316
	12.3 Biotransformation and Elimination of Arsenic	317
	12.4 History of Arsenic-Related Liver Diseases	318
	12.5 Hepatoportal Sclerosis	319
	12.6 Arsenic-Related Hepatic Fibrosis and Cirrhosis	320
	12.7 Epidemiological Study to Assess Arsenic-Related Liver Dysfunction	321
	12.8 Animal Studies for Understanding the Pathogenesis	322
	12.9 Arsenic and Liver Cancer	329
	References	330

x Contents

13.	Arsenic and Respiratory Disease	335
	13.1 Introduction	335
	13.2 Chronic Arsenic Exposure and Respiratory Health	336
	13.3 Early Life Arsenic Exposure and Lung Health	338
	13.4 Mechanistic Data on Arsenic Exposure and the Lung	341
	13.5 Conclusions	343
	References	343
14.	Arsenical Kidney Toxicity Bruce A. Fowler	349
	14.1 Introduction	349
	14.2 Clinical Manifestations of Arsenical Toxicity in Humans	350
	14.3 Nephrotoxic Arsenical Compounds	351
	14.4 Mechanisms of Arsenical Toxicity	352
	14.5 Biomarkers of Nephrotoxicity	354
	14.6 Arsenical Interactions with Other Nephrotoxic Elements	356
	14.7 Summary and Future Research Needs	357
	References	357
15.	Arsenic-Induced Developmental Neurotoxicity Jiaohua Luo and Weiqun Shu	363
	15.1 Introduction	363
	15.2 Arsenic Exposure Impairs Intellectual Function in Children	364
	15.3 Developmental Neurobehavioral Toxicity in Animals	367
	15.4 Arsenic Distribution After Exposure in Early Life	368
	15.5 Mechanism of Developmental Neurotoxicity	369
	15.6 Neuroprotective Agents Against Arsenic Toxicity	376
	15.7 Conclusions	377
	References	380

16.	Developmental Arsenic Exposure Impacts Fetal Programming of the Nervous System Fengyuan Piao, Shuangyue Li, and Huai Guan	387
	16.1 Introduction	387
	16.2 Accumulation of Arsenic in Fetal Brain Tissue	388
	16.3 Effect of Arsenic on the Development of the	
	Nervous System	389
	16.4 Effect of Arsenic on Neurobehavior	391
	16.5 Mechanisms for the Effect of Arsenic on the	
	Nervous System During Development	395
	16.6 Conclusions and Future Directions	399
	References	399
17.	Health Effects of Prenatal and Early-Life	
	Exposure to Arsenic	405
	Kathryn A. Bailey and Rebecca C. Fry	
	17.1 Introduction	405
	17.2 Adverse Health Effects Associated with Chronic and	
	Early-Life Arsenic Exposure	406
	17.3 Mechanisms Implicated in Disease Development	
	Associated with Prenatal Exposure	412
	17.4 Conclusions and Future Directions	421
	References	421
18.	Arsenic, Kidney, and Urinary Bladder Disorders Ernesto Sabath	429
	18.1 Introduction	429
	18.2 Arsenic and Renal Disease	430
	18.3 Arsenic and Bladder Disease	436
	References	438

19.	Developmental Arsenic Exposure: Behavioral Dysfunctions and Neurochemical Perturbations Riyaz Basha and G. Rajarami Reddy	443
	19.1 Introduction	443
	19.2 Toxicity	444
	19.3 Developmental Toxicity	444
	19.4 Conclusions	449
	References	451
20.	Arsenic and the Cardiovascular System Ashish Mehta, Chrishan J.A. Ramachandra, and Winston Shim	459
	20.1 Introduction	459
	20.2 Cardiovascular System	461
	20.3 Arsenic Effects on Blood	462
	20.4 Arsenic Effects on the Vascular System	468
	20.5 Arsenic Effects on the Heart	474
	20.6 Human Pluripotent Stem Cells: Understanding	
	Arsenic Toxicity	481
	20.7 Conclusions	483
	References	483
21.	Immunotoxic Effects of Arsenic Exposure Laila N. Islam	493
	21.1 Introduction	494
	21.2 Influence of Nutritional Factors	496
	21.3 Effects on Blood Leukocytes	496
	21.4 Interruption of Energy Production	496
	21.5 Effects on ROS Production	497
	21.6 Genotoxic and Carcinogenic Potentials	497
	21.7 Hematological Effects on Experimental Animals	498
	21.8 Effects on Heme Synthesis	499

	21.9	Hepatic Effects and Lipid Peroxidation	499
	21.10	Effects on Immune Responses in Fish	500
	21.11	Effects on Immune Responses in Laboratory Animals	501
	21.12	Effects of Arsenic in Drinking Water on Human Health	502
	21.13	Immunotoxic Effects of Organic Arsenicals in Foods	503
	21.14	Medicinal Use of Arsenic and Its Mechanism of Action	504
	21.15	Effects of Arsenic Compounds on Human Cells in Culture	504
	21.16	Immunotoxic Effects on Murine and Human Monocytes/	
		Macrophages	506
	21.17	Immunotoxic Effects on Murine Mononuclear Cells	509
	21.18	Decreased Cytokine Production by Human T Cells	509
	21.19	Effects of In Utero Exposure on Infant Immune System	510
	21.20	Gender-Related Immunotoxic Effects in Human	510
	21.21	Effects of Chronic Exposure on Immune Response	511
	21.22	Association with Respiratory Complications and Impaired Immune Responses	512
	21.23	Effects of Chronic Exposure on Serum Complement	
		Function	513
	21.24	Conclusions	513
	Refere	ences	514
22.	Arsei	nic and Developmental Toxicity and Reproductive	
	Disor	ders	521
	Molly	L. Kile and Maitreyi Mazumdar	
	22.1 I	ntroduction	521
	22.2 I	Developmental Toxicity	523
	22.3 I	Reproductive Toxicity	525
	22.4 I	Early-Life Exposures and Delayed Health Effects	526
	22.5 I	Reproductive Disorders	527
	22.6 0	Conclusions	528
	Refere	ences	528

23.	Arsenic and Cancer	533
	Paul B. Tchounwou, Udensi K. Udensi, Raphael D. Isokpehi, Clement G. Yedjou, and Sanjay Kumar	
	23.1 Arsenic and Arsenic-Containing Compounds	533
	23.2 Sources of Arsenic and Potential for Human Exposure	535
	23.3 Molecular Mechanisms of Arsenic-Induced Carcinogenesis	538
	23.4 Health Effects Associated with Arsenic Exposure	543
	23.5 Conclusions	549
	Acknowledgments	549
	References	549
24.	The Association between Chronic Arsenic Exposure and Type 2 Diabetes: A Meta-Analysis Bayzidur Rahman, Tanvir Abir, AKM Akbar Hossain, Md. Rafiqul Islam, and Abul Hasnat Milton	557
	24.1 Introduction	557
	24.2 Methods and Materials	559
	24.3 Results	560
	24.4 Discussion	568
	References	570
25.	Arsenic Biosensors: Challenges and Opportunities for High-Throughput Detection <i>Minni Singh and Manel del Valle</i>	575
	25.1 Arsenic: The Toxic Metalloid	575
	25.2 Arsenic Biosensors	576
	25.3 Nanosensor Platforms—Towards High-Throughput	
	Detection	581
	25.4 Conclusions and Future Directions	585
	References	586

26.	Medical Countermeasures—Chelation Therapy Govinder Flora, Megha Mittal, and Swaran J.S. Flora	589
	26.1 Introduction	589
	26.2 Clinical Aspects of Arsenic	590
	26.3 Diagnosis	591
	26.4 Chelation Therapy	592
	26.5 Chelators in Clinical Use	598
	26.6 Analogues of DMSA as Potential New Arsenic Chelators	605
	26.7 Role of Antioxidants in Preventing Arsenic Toxicity	608
	26.8 Newer Strategies	614
	26.9 Concluding Remarks and Future Directions	618
	References	618
27.	Biochemical and Molecular Basis of Arsenic Toxicity and Tolerance in Microbes and Plants Sarita Pandey, Rashmi Rai, and Lal Chand Rai	627
	27.1 Introduction	627
	27.2 Arsenic Toxicity and Tolerance in Microbes	631
	27.3 Arsenic Toxicity and Tolerance in planta	641
	27.4 Mechanisms of Arsenic Tolerance and Detoxification	651
	27.5 As Hyperaccumulation and Phytoextraction	657
	27.6 Summary Points	658
	Acknowledgments	659
	References	659
28.	Arsenic Contents and Its Biotransformation in the Marine Environment Kiran Kalia and Devang B. Khambholja	675
	28.1 Introduction	675
	28.2 Arsenic Concentration in Sea Water	676

28.3 Arsenic Concentration in Marine Sediments	679
28.4 Arsenic Speciation in Marine Ecosystems	681
28.5 Arsenic Cycle in the Marine Environment	683
28.6 Role of Marine Biological Systems in Arsenic	
Biotransformation	686
28.7 Arsenic in Seafood and Its Toxicity	691
28.8 Future Directions	694
References	694
Index	701

Foreword

Handbook of Arsenic Toxicology is a significant addition to the literature dealing with the toxicology of this very important metalloid. The chapters in this book are relevant to arsenic exposure via drinking water, the usual route of exposure of an estimated 200 million people in the world – many in developing countries. In Bangladesh, arsenic exposure is also more common through ingestion of food. Populations with high intake of rice and vegetables may also be at risk.

Arsenic is a recognized human carcinogen and it can also cause cardiovascular and metabolic disease. It is a unique toxicant to the respiratory system – the only one following ingestion that is associated with malignant and non-malignant respiratory diseases.

Arsenic for many years was known as the King of Poisons, and the Poison of Kings. It was the poison of choice for homicides because no chemical test was available to identify and measure it. Arsenic trioxide, As₂O₃, was the highly favored poison form, because it is odorless, easily incorporated into food and drink, and before the development of the Marsh test, it was untraceable in the body. This chemical test was developed in about the 1830s after which time the number of homicides using arsenic compounds was greatly reduced.

The metabolism of inorganic arsenic has been carefully investigated during recent years. It was discovered surprisingly that inorganic arsenite was metabolized to the more toxic +3-methylated metabolites, methylarsonous acid (MMA^{III}) and dimethylarsinous acid (DMA^{III}). The latter compound is so toxic that it is wise to use a fume hood when working with it.

In his chapter on arsenical kidney toxicity, Bruce Fowler aptly points out that "the mitochondria are a major target organelle for arsenical toxicity with resultant inhibition of respiratory function, loss of ATP production, and generation of reactive oxygen species leading to altered cell signaling pathways, gene expression patterns, and induction of apoptosis."

Many, but not all, chapters of this book are written by the new generation of arsenic investigators. They bring to the forefront many newer thoughts and information about the toxicity of compounds of this toxic metalloid. One cannot ignore, however, the contributions by some of the well-known established authors. The first chapter by Dr. Flora on the chemistry, occurrence and exposure to arsenic compounds sets the tone as to what is to come. Dr. D.N. Guha Mazumder is another highly respected investigator with a lengthy list of significant publications dealing with the human clinical effects of exposure to arsenic compounds.

One also needs to remember other highly respected investigators like Dr. Mariano Cebrian. Although he is not an author of a chapter in this book, he was among the earliest investigators of such research. His first arsenic paper was published in 1983, entitled *Chronic Arsenic poisoning in the North of Mexico*. This early paper is very relevant because it dealt with a human population and answered some questions dealing with arsenic toxicity. The paper also presented

new questions concerning arsenic exposure and human health and stimulated the entrance of other investigators into this previously neglected area of scientific investigation. The chapters of the present book expand these questions and answers.

With the number of people at risk today, it is surprising that no useful treatment of chronic arsenic toxicity for large populations has become available. Prevention and/or limiting exposure remain the only approaches, and the information supplied in this book is a great step forward in this area.

H. Vasken Aposhian, Ph.D. Professor Emeritus of Molecular and Cellular Biology, University of Arizona, USA

Preface

I have been a chemical toxicologist for nearly 35 years. In this time, I have studied and evaluated the toxicities of toxic metals and the health effects produced by human exposure to metals.

Arsenic, a naturally occurring metalloid, is ubiquitously present in the environment. Arsenic is ranked first among toxicants posing a significant potential threat to human health based on known or suspected toxicity. This naturally occurring metalloid is a known poison, a co-carcinogen, and in lower concentrations has been shown to cause damage to almost all major organs including liver, lungs, bladder and brain. Currently, the permitted concentration of arsenic in water is $10 \,\mu$ g/L ($10 \,p$ pb). Yet, an estimated 100 million people worldwide are exposed to excessive amounts of arsenic via drinking water (in the ppm, not ppb, range). Many of these individuals obtain drinking water from unregulated sources (wells) or live in regions where arsenic levels are high, such as Bangladesh. Arsenic leaches from rock formations into water sources as the water table recedes, and hence exposure to high amounts of arsenic will continue to persist whilst the demand for clean water increases. This phenomenon particularly affects the Western region of the United States, where it is estimated that certain areas contain up to $3100 \,\mu$ g/L arsenic ($31 \,p$ pm) in drinking water, on par with levels reported in Taiwan, China, Bangladesh and India.

Although the largest number of people affected worldwide by the arsenic contamination of drinking water are in Bangladesh, the problem is not unique to that area. As early as 1960, scientists reported the link between various forms of cancer and arsenic in drinking water in Taiwan. Communities in North and South America, Europe, Asia and Australia also face the problem of arsenic-contaminated drinking water. The problem of arsenic-contaminated groundwater is found in communities throughout Canada and western USA that use groundwater as their source of drinking water. It is now almost certain that arsenic contamination is a worldwide problem; however, some of the most affected regions lie in the flood plains of the great rivers of Bangladesh, Nepal, and West Bengal, India. In Bangladesh alone, seventy million people are impacted. Problems associated with drinking groundwater were first noticed in Bangladesh by healthcare workers in the early 1990s. While the World Health Organization (WHO) and the Environmental Protection Agency (EPA) regulate water sources of arsenic, lack of strict regulations on food, beverages, and air quality can lead to increased arsenic exposure. Ingestion of arsenic activates metabolic pathways for excretion, resulting in a number of metabolites, some of which are more potent and toxic than the originally ingested inorganic form of arsenic.

Inorganic arsenic exposure of humans, by the inhalation route, has been shown to be strongly associated with lung cancer, while ingestion of inorganic arsenic by humans has been linked to a form of skin cancer and also to bladder, liver, and lung cancer. The EPA has classified inorganic arsenic as a human carcinogen. This explosion of information in the recent years reflects the vast increase in number of researchers studying about the mechanisms of action of arsenic. The specific knowledge of the chemistry, biochemistry, toxicology, and epidemiology of arsenic is far greater than that for any other environmentally-occurring chemical carcinogen.

This book really began over 20 years ago, when I was confronted with the first of scores of instances in Uttar Pradesh, Maharashtra, West Bengal and Bihar in India, where individuals exposed to arsenic subsequently developed symptoms and effects that could notably be explained by the known toxicological effects. In some instances the exposures led to effects far in excess of what would be expected. In others, effects were noted following exposures to extremely low levels of arsenic; in even more instances the body organs targeted were not those known to be impacted by arsenic. I carried out number of studies during these years; a few of them with my colleagues in West Bengal. These studies led to one serious concern: that we do not have a safe, specific and effective chelating drugs in this part of the world, and those drugs available in the developed countries are largely ineffective against arsenic toxicity.

As time progressed, I began to think that the task was too big and a solution remained elusive. The breakthrough came when I got a reprint from Prof. M.M. Jones, Vanderbilt University in which his group synthesized and evaluated the efficacy of number of di- and monoesters of meso 2,3-dimercaptosuccinic acid (DMSA) with limited success against cadmium intoxication. There was a small but an interesting note on the top of the reprints written in red ink, where he asked me to try these esters against arsenic. This led me to my interest in arsenic poisoning and in particular searching for a new chelating agent. A review of the literature and our research group's own studies highlighted the shortcomings with DMSA, DMPS and BAL, and it was then hypothesized that monoesters of DMSA might be a better option to treat cases of arsenicosis.

I have attempted to bring together as comprehensive a group of scientists as possible in assembling this book. Whilst at first glance, the literature on arsenic toxicology seems exhaustive and systematic, this is not the case. There is no comprehensive and in-depth analysis of its effects on major organs, preventive and therapeutic measures; additionally, there are a few new topics where not much work has been undertaken but could be of potential future interest. This book thus promises to provide a comprehensive coverage of arsenic and its toxic effects, including its toxicokinetics, mode(s) of action, effects on all major organs and medical countermeasures. To my knowledge, this book perhaps is the first in-depth analysis of data on toxicology, risk assessment, and management. Included in these 28 chapters are detailed reviews of the many important mechanistic aspects of arsenic.

Chapters1 and 2 provide an orientation and introduction to the subject of arsenic. The focus of these chapters is to provide an overview of various critical factors affecting arsenic chemistry, the natural and anthropogenic sources of exposure. The focus of Chapters 3 and 4 are risk assessment following arsenic exposure while Chapter 5 provides data for the removal of arsenic using activated alumina (AA) and modified AA adsorbents. Chapters 5 and 6 provide information on the general health effects of arsenic and the role of arsenic metabolites in

the toxic manifestation, respectively. Chapters 7–9 focus on various proposed modes of actions for arsenic, exposure pathways and toxicokinetics, various alterations in mediating genotoxic effects such as altered DNA repair, signal transduction, cellular proliferation, and altered DNA methylation. One of the major mechanisms of arsenic-induced toxic manifestation is oxidative stress. These chapters provide in-depth information regarding alterations at the biochemical level, detailed mechanisms of toxicity and oxidative injury, and the links between arsenic, oxidative stress and cancer. Chapter 10 discusses the gastrointestinal tract as one of the target organs of arsenic and a factor affecting its toxicity and the resultant risk assessments required. The authors suggest that arsenic species with higher toxicity degree than those ingested may appear in the intestinal lumen as a result of interactions with food components and from metabolism by enterocytes and micro biota. Also, the biotransformation may modulate arsenic intestinal absorption and therefore adverse effects.

Although arsenic impacts on the physiological cellular processes in numerous organ systems, the outcomes of its toxicity are usually first seen in the skin. The major focus of Chapter 11 is on skin manifestations from acute toxicity such as flushing, erythema, facial edema, acrodynia, urticarial, alopecia, loss of nails, and Mees lines visible on nails. The liver is the target organ of arsenic and many important various metabolizing reactions take place in liver, rendering it the most susceptible organ to any xenobiotic. Exposure to arsenic leads to various hepatic disorders, which has been discussed in Chapter 12. Arsenic, the only environmental toxicant has been linked to both malignant and non-malignant respiratory disease following ingestion, rather than inhalation, making arsenic a unique toxicant to the respiratory system. Chapter 13 suggests that chronic exposure to arsenic has been associated with the development of respiratory symptoms, impaired lung function and chronic lung disease. Chapter 14 provides an overview of information that arsenic disturbs various vital renal functions such as the excretion of nitrogenous waste products and maintenance of electrolyte balance, which leads to immediate effects on circulating blood and hence whole body. Chapter 15, 16, 17, and 19 make a strong argument for the potential role of arsenic in disrupting the normal functions of the central nervous system, thereby causing impairment of learning, concentration and short term memory. It also alters the release of various neurotransmitters. Since the brain is the most vital organ, it's important to fully understand the effect of arsenic intoxication and associated neuropathologies, which are discussed here.

Arsenic accumulates in the urinary bladder epithelium, causing activation of specific signaling pathways, leading to increased cell proliferation and increased incidence of urinary bladder tumors. Such manifestations are discussed in Chapter 18. Chronic arsenic exposure induces high oxidative stress which affect the structure and function of the cardiovascular system, resulting in cardiac anomalies such as arrhythmia and atherosclerosis. Moreover, newer technologies like the use of pluripotent stem cells could provide a better opportunity to study the effects of arsenic in a cell type-specific manner. Chapter 18 also briefly discusses emerging technologies that provide a new dimension to evaluate diseases.

Immunosuppression caused by arsenic is another major cause of cancer development in humans. Chronic exposure to arsenic has been shown to significantly compromise the host response to infection, thereby leading to development of fatal conditions. Chapter 21 provides comprehensive information about such immunosuppressive effects. There have been many reports exhibiting significant relationship between arsenic exposure and adverse reproductive health outcomes such as stillbirths, low birth-weight, childhood stunting, neural tube defects and under-weight children. Chapter 22 specifically deals with various adverse effects of arsenic on reproductive system and potential mechanisms. Arsenic is classified as a human carcinogen and has been linked with various malignancies such as skin, lungs and liver. Chapter 23 specifically explores the association between arsenic exposure and cancer development and current mechanisms that have been proposed to contribute to such neoplastic processes. An association between chronic arsenic exposure and Type 2 diabetes is still inconclusive. Chapter 24 deals with the analysis of 27 studies including 15 cross-sectional, 4 case-control and 3 cohort studies. Authors conclude that their meta-analysis suggests that chronic arsenic exposure is likely to increase the risk of Type 2 diabetes. Chapter 25 attempts to highlight the advancement made so far in the development of arsenic biosensors – analytical devices that have high sensitivity, portability, small sample requirements and are easily used for quantitative and qualitative monitoring of various analytes.

Both chemical and synthetic medical countermeasures have been used against chronic arsenic poisoning. Chelating agents such as Dimercaprol and DMSA were found to be quite effective; however they are compromised with a few shortcomings. Extensive research in future is required to establish an effective drug against chronic arsenicosis. Also development of specific biomarkers is an important strategy. Chapter 26 focuses on available arsenic anti-dotes, their clinical uses, their drawbacks, side effects and the new developments in the area. Chapter 27 deals with recent progress on the mechanisms of arsenic uptake, toxicity, and detoxification in microbes and in plants. Chapter 28 provides information about arsenic concentration and biogeochemical cycle in marine environment (marine animals, bacteria, phytoplankton, etc).

I intend that this book be useful for health scientists, including nutritionists and dietitians, pharmacologists, public health scientists, and colleges, epidemiologists, health workers and practitioners, agriculturists, botanists, healthcare professionals of various disciplines, policy-makers, and marketing and economic strategists. It is also designed for teachers and lecturers, undergraduates and graduates.

Finally, I am deeply indebted to all authors for their sincere and dedicated contributions to this book. I have tried to get all contributions written by specialists in the respective area; however, we have missed a few well-known names who expressed their inability due to various professional and personal commitments. The editor views these contributions as excellent summaries by some of the world's outstanding scientists/ researchers practitioners of arsenic toxicology and hope the readers will find them highly enlightening and useful. A special thanks to Molly McLaughlin, the project manager at Elsevier/Academic Press, who was a big support throughout the process and provided vital input in the preparation of this book.

Swaran J.S. Flora

Acknowledgements

I am grateful to all of the contributors for taking the time and effort to complete these chapters in a thorough yet timely manner. In addition to the individual contributions of each of our chapter authors, who are recognized leaders in their respective areas of specialization, many other scientists from around the world have contributed extensively to our understanding of arsenic biochemistry and toxicology. I would like to acknowledge the work of all these contributors. Their efforts made it much easier to do my job of compiling and editing this book.

Many of us who are involved in research about arsenic find that our work is not just fascinating and compelling, but are drawn to this metalloid for its public health significance in the developing world. The daily exposure to arsenic in different parts of the world are manifold. In some countries, human exposure can exceed 3000 ppb and in parts of the developed world, the exposure is much less than 10 ng per day. Given this situation it is hoped that the understanding of basic mechanisms of the action of arsenic will lead to the design of effective prevention strategies for both the developed and developing world.

In addition to the contributors, several other scientists have also provided scientific substance for this book and they should be congratulated for their efforts, I also thank Molly McLaughlin, Editorial Project Manager at Elsevier/Academic Press for her extensive and excellent administrative support and unfailing help in the development of this book from its inception. This book would not have been physically possible without her tireless work and extraordinary skills. I am also grateful to the many people at Elsevier/Academic Press for their assistance, patience, and understanding in working with me to complete this book in a timely manner. A heartfelt thanks is also extended to Melissa Read at Elsevier/Academic Press for the fine production work.

I would be remiss without specifically acknowledging the support provided by the Defence Research and Development Organization, Ministry of Defence, Government of India, New Delhi, India especially Dr. Avinash Chander, Scientific Advisor to Honourable Defence Minister and Dr. Manas K. Mandal, Distinguished Scientist and Director General (Life Sciences), for their support and fostering my interest in completing this book.

I dedicate this book to Gurpyari, my life-long companion, for her unconditional support in many projects. I also dedicate this book to my daughter Preeti, my son Ujjwal, and brother Govinder for providing me with reasons to fight for the protection of the environment and especially to my parents for their support during my studies and thereafter. I also dedicate this book to my mentor Dr S.K. Tandon (in memoriam) for his guidance, foresights and for opening my eyes to the science of toxicology.

Swaran J.S. Flora

This page intentionally left blank



List of Contributors

- Tanvir Abir Centre for Health Research, School of Medicine, University of Western Sydney, NSW, Australia
- Sk Akhtar Ahmad Bangladesh University of Health Sciences, Dhaka, Bangladesh
- **Mohammad Athar** Department of Dermatology and Skin Disease Research Center, University of Alabama at Birmingham, Birmingham, Alabama, USA
- Kathryn A. Bailey Department of Environmental Sciences and Engineering, Gillings School of Global Public Health, University of North Carolina at Chapel Hill, NC, USA
- **Riyaz Basha** Department of Pediatrics and Institute for Cancer Research, University of North Texas Health Science Center, Fort Worth, TX, USA
- **Marta Calatayud** Institute of Agrochemistry and Food Technology, High Research Council of the Spanish Government; Faculty of Health Sciences, Technical University of Ambato, Ecuador
- Lucy Mar Camacho Department of Environmental Engineering, Texas A&M University—Kingsville, Kingsville, TX, USA
- **Isaac Campos** Department of Civil Engineering, The University of Texas at El Paso, El Paso, TX, USA
- **Thomas A. Davis** Department of Civil Engineering, The University of Texas at El Paso, El Paso, TX, USA
- Manel del Valle Sensors and Biosensors Group, Department of Analytical Chemistry, Universitat Autonoma de Barcelona, Barcelona, Spain
- **Shuguang Deng** Department of Chemical Engineering, New Mexico State University, Las Cruces, NM, USA
- Govinder Flora SBBS Institute of Technology and Management, Jalandhar, India
- Swaran J.S. Flora Department of Pharmacology and Toxicology, Defence Research and Development Establishment, Gwalior, India; Division of Regulatory Toxicology, Defence Research and Development Establishment, Gwalior, India
- **Bruce A. Fowler** Emory University Rollins School of Public Health, Atlanta, GA, USA; and University of Alaska-Fairbanks, Fairbanks, AK, USA
- **Rebecca C. Fry** Department of Environmental Sciences and Engineering, Gillings School of Global Public Health, University of North Carolina at Chapel Hill, NC, USA

- Jyotirmoy Ghosh Natural Product Chemistry Division, CSIR-North East Institute of Science and Technology, Assam, India
- **Huai Guan** Department of Occupational and Environmental Health, College of Public Health, Dalian Medical University, Dalian, People's Republic of China
- **D.N. Guha Mazumder** DNGM Research Foundation, New Alipore, Kolkata, India; Department of Medicine & Gastroenterology, Institute of Post Graduate Medical Education & Research, Kolkata, India
- **AKM Akbar Hossain** Centre for Health Research, School of Medicine, University of Western Sydney, NSW, Australia
- Katherine Marchiony Hunt University of Alabama School of Medicine, University of Alabama at Birmingham, Birmingham, Alabama, USA
- Laila N. Islam Department of Biochemistry and Molecular Biology, University of Dhaka, DHAKA, Bangladesh
- **Md.Rafiqul Islam** Centre for Clinical Epidemiology and Biostatistics, School of Medicine and Public Health, University of Newcastle, NSW, Australia
- **Raphael D. Isokpehi** Molecular Toxicology Research Laboratory, NIH RCMI-Center for Environmental Health, College of Science, Engineering and Technology, Jackson State University, Jackson, Mississippi, USA
- Kiran Kalia Laboratory of Biochemistry, Bhanuben and Ratilal Doshi School of Biosciences, Sardar Patel Maidan, Sardar Patel University, Vallabh Vidyanagar, Gujarat, India
- **Devang B. Khambholja** Laboratory of Biochemistry, Bhanuben and Ratilal Doshi School of Biosciences, Sardar Patel Maidan, Sardar Patel University, Vallabh Vidyanagar, Gujarat, India
- Manzurul Haque Khan National Institute of Preventive and Social Medicine, Dhaka, Bangladesh
- Molly L. Kile College of Public Health and Human Sciences, Oregon State University, Corvallis, OR, USA
- Sanjay Kumar Molecular Toxicology Research Laboratory, NIH RCMI-Center for Environmental Health, College of Science, Engineering and Technology, Jackson State University, Jackson, Mississippi, USA
- José Moisés Laparra Llopis Institute für Translational Immunologie, University Medical Center of the Johannes Gutenberg-University, Mainz, Germany
- Janice S. Lee National Center for Environmental Assessment, Office of Research and Development, US Environmental Protection Agency, Research Triangle Park, North Carolina, USA
- **Shuangyue Li** Department of Occupational and Environmental Health, College of Public Health, Dalian Medical University, Dalian, People's Republic of China

- Jiaohua Luo Department of Environmental Hygiene, College of Preventive Medicine, Third Military Medical University, Chongqing, People's Republic of China
- **Smarajit Maiti** Post Graduate Department of Biochemistry & Biotechnology, Cell & Molecular Therapeutics Laboratory, Oriental Institute of Science and Technology, Midnapore, Vidyasagar University, Midnapore, India
- **Badal Kumar Mandal** Trace Elements Speciation Research Laboratory, Environmental and Analytical Chemistry Division, School of Advanced Sciences, VIT University, Vellore, Tamilnadu, India
- Maitreyi Mazumdar Boston Children's Hospital and Harvard Medical School, Boston, MA, USA
- Ashish Mehta National Heart Research Institute Singapore, National Heart Centre Singapore, Singapore
- Abul Hasnat Milton Centre for Clinical Epidemiology and Biostatistics, School of Medicine and Public Health, University of Newcastle, NSW, Australia
- **Megha Mittal** Department of Chemistry, Manav Rachna College of Engineering, Faridabad, India
- **Sarita Pandey** Molecular Biology Section, Laboratory of Algal Biology, Centre of Advanced Study in Botany, Banaras Hindu University, Varanasi, India
- **Fengyuan Piao** Department of Occupational and Environmental Health, College of Public Health, Dalian Medical University, Dalian, People's Republic of China
- **Sundaravadivelnathan Ponnusamy** Department of Chemical Engineering, New Mexico State University, Las Cruces, NM, USA
- **Bayzidur Rahman** School of Public Health and Community Medicine, University of New South Wales, NSW, Australia
- Lal Chand Rai Molecular Biology Section, Laboratory of Algal Biology, Centre of Advanced Study in Botany, Banaras Hindu University, Varanasi, India
- Rashmi Rai Laboratory of Morphogenesis, Centre of Advanced Study in Botany, Banaras Hindu University, Varanasi, India
- Chrishan J.A. Ramachandra National Heart Research Institute Singapore, National Heart Centre Singapore, Singapore
- **Santhini Ramasamy** Office of Science and Technology, Office of Water, US Environmental Protection Agency, Washington DC, USA
- Kathryn Ramsey Telethon Kids Institute, University of Western Australia, Subiaco, Western Australia, Australia
- G.Rajarami Reddy Vikrama Simhapuri University, Nellore, A.P., India
- **Ernesto Sabath** Universidad Autonoma de Queretaro, Queretaro, Mexico, and Renal Department, Hospital General de Queretaro, Queretaro, Mexico

- Amal Santra Centre for Liver Research, School of Digestive & Liver Diseases, Institute of Post Graduate Medical Education & Research, Kolkata, West Bengal, India
- Winston Shim National Heart Research Institute Singapore, National Heart Centre Singapore, Singapore
- **Weiqun Shu** Department of Environmental Hygiene, College of Preventive Medicine, Third Military Medical University, Chongqing, People's Republic of China
- Parames C. Sil Division of Molecular Medicine, Bose Institute, Kolkata, India
- Minni Singh Biosensor Development Group, Department of Biotechnology, Punjabi University, Patiala, India
- **Ritesh Kumar Srivastava** Department of Dermatology and Skin Disease Research Center, University of Alabama at Birmingham, Birmingham, Alabama, USA
- **Shafi M. Tareq** Department of Environmental Sciences, Jahangirnagar University, Dhaka, Bangladesh; School of Biosciences, The University of Nottingham Malaysia Campus, Jalan Broga, Selangor Darul Ehsan, Malaysia
- **Paul B. Tchounwou** Molecular Toxicology Research Laboratory, NIH RCMI-Center for Environmental Health, College of Science, Engineering and Technology, Jackson State University, Jackson, Mississippi, USA
- **Udensi K. Udensi** Molecular Toxicology Research Laboratory, NIH RCMI-Center for Environmental Health, College of Science, Engineering and Technology, Jackson State University, Jackson, Mississippi, USA
- **Clement G. Yedjou** Molecular Toxicology Research Laboratory, NIH RCMI-Center for Environmental Health, College of Science, Engineering and Technology, Jackson State University, Jackson, Mississippi, USA

Arsenic: Chemistry, Occurrence, and Exposure

Swaran J.S. Flora

DEPARTMENT OF PHARMACOLOGY AND TOXICOLOGY, DEFENCE RESEARCH AND DEVELOPMENT ESTABLISHMENT, GWALIOR, INDIA

CHAPTER OUTLINE

1.1	Introduction	1
1.2	Chemistry of Arsenic	2
1.3	Arsenic Minerals and Compounds	8
1.4	Organoarsenicals	. 11
1.5	Arsenic Mobilization in the Environment	. 13
1.6	Sources of Arsenic in Biosphere	. 16
1.7	Arsenic in Hydrothermal and Geothermal Fluids	. 24
1.8	Arsenic Release from Mining and Mineral Processing	. 29
1.9	Global Occurrence of Arsenic in Ground Water	. 32
1.10	Methods of Arsenic Removal from Water	. 37
1.11	Conclusions	. 40
Refe	rences	. 41

1.1 Introduction

Arsenic, the king of poisons, has influenced the human population more than any other element or toxic compound for thousands of years. Today, millions of people are being chronically exposed to elevated doses of arsenic from air, food, water, and soil. Throughout the history of human progress, arsenic has been seen as a bizarre and frightful element. Toxic effects of arsenic are highly prevalent in both developed and developing countries. Arsenic toxicity has become a principal concern owing to the escalating contamination of air, water, and soil. It has the ability to readily change its oxidation state and bonding configuration, thus showing diverse chemical behavior in the environment and forming large numbers of organic and inorganic compounds. Specific electronic configurations of valence shells with filled s orbitals and half-filled p orbitals enable arsenic to easily donate electrons and overlap in covalent bonds. Naturally, arsenic forms bonds with oxygen and sulfur and generates large

2 HANDBOOK OF ARSENIC TOXICOLOGY

numbers of oxides and sulfides. It is also capable of forming large numbers of bio-molecules as it forms stable bonding with the methyl group. The peculiar chemistry of arsenic is the basis for its dual action as a toxin and as a curative.

Drinking water and contaminated soils are the major means by which arsenic gains its entry into the food chain. Most of the arsenic compounds are readily soluble in water and so can easily enter water bodies such as rivers, lakes, and ponds, and by surface runoff. The main pathways of human exposure to arsenic are ingestion of drinking water, consumption of food, and inhalation of air. Among all sources, drinking water has been reported to be the main route of arsenic exposure around the globe [1,2]. Occupational exposure to arsenic is also very common in individuals working in wood preservation, desiccant, chemical warfare agent, pigment, drug and arsenic-based pesticide industries, and those involved in smelting and mining operations and residing in the vicinity of mining areas.

Prolonged exposure to arsenic leads to various dermatological, respiratory, neurological, and reproductive disorders and it is therefore referred to as a carcinogen and mutagenic agent [3,4]. The US Department of Health and Human Services in its 9th Report on Carcinogens listed arsenic compounds as human carcinogens. Arsenic exposure may cause severe health manifestations including cancers, melanosis (hyperpigmentation or hypopigmentation), hyperkeratosis (hardened skin), blackfoot disease (peripheral vascular disorder), gangrene, diabetes mellitus, hypertension, ischemic heart disease, etc. [5,6].

Due to increasing health concerns and adverse effects of arsenic on humans, US EPA in 2002 reduced the maximum permissible limit of arsenic in drinking water from 50 to $10 \mu g/L$ [7]. Despite a number of corrective and preventive measures, the increase of arsenic contamination in ground water continued to develop with the addition of new areas to the list of arsenic contaminated regions. In view of the global health issues associated with arsenic exposure, it has become essential to understand arsenic sources, geochemistry, interaction with water, and various mechanisms associated with arsenic release into the environment. This chapter deals with the occurrence of arsenic in air, water, and soil, and its chemistry in the entire medium along with its global distribution. The chapter also discusses a number of treatment methodologies for the effective removal of arsenic from ground water and to reduce the global arsenic burden.

1.2 Chemistry of Arsenic

1.2.1 Origin and History

Arsenic has been known since ancient times in its sulfide form. The Greek philosopher Theophrastus knew about two arsenic minerals: bright yellow orpiment (As_2S_3) and red colored realgar (As_4S_4) . Greek historian Olympiodorus of Thebes (5th century AD) was the first to obtain white arsenic (As_2O_3) by heating arsenic sulfide. The discovery of the element arsenic is attributed to Albertus Magnus, a German philosopher in the 1200s, who was the first to report the metallic behavior of arsenic. *De Mineralibus* described pure arsenic being obtained by the heating of orpiment with soap. Arsenic trioxide (As_2O_3) , a by-product of copper refining, when mixed with olive oil and heated gives arsenic metal. Chinese scientist Tsao Kan-Mu studied toxicity of arsenic compounds in the 1500s during the Ming dynasty and mentioned their use as pesticides in rice fields. The elemental name is believed to come from the Greek word *arsenikos* meaning potent. Arsenic ranks as the 20th most common element in Earth's crust, 14th in the sea, and 12th in the human body.

Humans have been using arsenic since ancient times both as a poison and a curative. It has also been used in pyrotechnics, metallurgy, warfare, and pigmentation, and for decoration. One of the most popular oxides of arsenic (arsenic trioxide) is a tasteless, odorless, white powder used in the past as a chemical warfare agent; however, green colored copper acetoarsenate was traditionally used in wallpapers as a pigment [8].

1.2.2 Atomic Structure and Bonding

Arsenic ranks 33rd in the periodic table, as part of the elements in Group 15, being a member of the nitrogen family. Its atomic number is 33 and its atomic weight is 74.921, placing it as heavier than iron, nickel, and manganese but lighter than silver, lead, or gold. The most stable and non-radioactive isotope of arsenic is arsenic-75 (75 As) with 33 protons and 42 neutrons inside the nucleus surrounded by 33 electrons in different energy shells. Other unstable and radioactive isotopes of arsenic are listed in Table 1–1. The isotopes are converted to stable elements by electron capture, electron emission, positron emission, neutron emission, and internal transition [9]. The electronic configuration of the stable form of arsenic, As(0), can be written as shown below to illustrate how the presence of five valence electrons allows arsenic to participate in chemical bonding, with an empty p orbital for electron occupation; this can also be illustrated by the electron dot model of arsenic (Figure 1–1).

$$1s^{2} 2s^{2} 2p^{6} 3s^{2} 3p^{6} 3d^{10} 4s^{2} 4p^{3}$$

The electronic configuration suggests that the first shell has two electrons, the second has eight electrons, the third has 18 electrons, while the fourth with five electrons is considered as

Icotono	Mass	Mode of Decay	Half life
isotope	IVIdSS	Wode of Decay	пан-ше
⁶⁸ As	67.9368	EC to ⁶⁸ Ge	2.53 min
⁶⁹ As	68.93228	EC to ⁶⁹ Ge	15.2 min
⁷⁰ As	69.93093	EC to ⁷⁰ Ge	52.6 min
⁷¹ As	70.927114	EC to ⁷¹ Ge	2.72 d
⁷² As	71.926753	EC to ⁷² Ge	26.0 h
⁷³ As	72.923825	EC to ⁷³ Ge	80.3 d
⁷⁴ As	73.923829	β^- to ⁷⁴ Se; EC to ⁷⁴ Ge	17.78 d
⁷⁶ As	75.922394	β^- to 76 Se	26.3 h
⁷⁷ As	76.920648	β^- to ⁷⁷ Se	38.8h
⁷⁸ As	77.92183	β^- to ⁷⁸ Se	1.512 h
⁷⁹ As	78.92095	β^- to ⁷⁹ Se	9.0 min

Table 1–1 Various Isotopes of Arsenic with their Decay Properties

Abbreviations used: min = minutes; d = days; h = hours; EC = electron capture; β^- = electron emission; As = arsenic; Ge = germanium; Se = selenium.



FIGURE 1–1 Electron dot model of arsenic (75 As) depicting number of protons (P) and neutrons (N) inside the nucleus and electrons (–) in various energy shells.

incomplete. In the fourth shell, the s orbital is completely filled and the p orbital is half filled, and therefore arsenic shows four common redox states: -3, 0, +3, and +5. Oxidation state -3 arises by the addition of three more electrons to the p orbital in order to complete the total of six electrons in this orbital. Elemental arsenic (0 As) equally shares its three electrons present in the 4p orbital with the other three surrounding arsenic atoms in a trigonal pyramidal structure and appears as a brittle gray metal. The +3 oxidation state forms when three electrons of the 4p orbital become more attracted towards a non-metal, usually oxygen or sulfur, whereas arsenic exhibits the +5 oxidation state when all the five electrons of the 4s- and 4p orbitals become more associated with the non-metal.

Table 1–2 lists selected physical properties of elemental arsenic, which suggest that the electronegativity of arsenic is greater than that of nitrogen and similar to that of phosphorus. Arsenic has a greater ability to lose electrons (oxidation potential) as compared to nitrogen and phosphorus, which increases the cationic character of arsenic and it thus can easily exhibit +3 and +5 oxidation states. Variable oxidation states of arsenic suggest that it can combine with many elements to form covalent compounds; however, in nature it most commonly bonds to oxygen and sulfur. Arsenic can occupy electrons in its bonding and anti-bonding orbitals and can exhibit the properties of ligands by sharing its valence electrons. Thus, it shows the ability to shift from an electropositive state (oxo-anions) to an electronegative state (metal arsenides). Arsenic in its pure form is a brittle, gray metal but in nature it is found with other metals, such as iron, copper, silver, and nickel as oxides and sulfides.

Arsenite is known to be more toxic and 25–60 times more mobile than arsenate [10]. In both oxidation states, it can combine with methyl groups to form organic species. Common organic species of arsenic are monomethylarsonic acid (MMA) and dimethylarsinic acid (DMA). However, their natural occurrence is low compared to inorganic forms. Inorganic arsenicals are more toxic than organic arsenicals and the trivalent oxidation state is more toxic than the pentavalent oxidation state. Arsenic toxicity is also related to the rate at which it is metabolized and the degree to which it accumulates in the tissues. Generally, the As toxicity pattern is $AsH_3 > As^{3+} > As^{5+} > RAs-X$.

Property	Value
Atomic number	33
Atomic mass	74.9216 g/mol
Appearance	Gray, brittle, non-metal flakes
Electronegativity according to Pauling	2.0
Density	5.7 g/cm ³ at 14°C
Molar volume	13.08 cm ³ /mole
Melting point	814°C (36atm)
Boiling point	615°C (sublimation)
Specific heat	0.33 J/g K
van der Waals radius	0.139nm
Ionic radius	0.222 nm (–2) 0.047 nm (+5) 0.058 (+3)
Isotopes	8
Electronic shell	[Ar] 3d ¹⁰ 4s ² 4p ³
Energy of first ionization	947 kJ/mol
Energy of second ionization	1798 kJ/mol
Energy of third ionization	2736 kJ/mol
Standard potential	-0.3V (As ³⁺ /As)
Enthalpy of atomization	301.3 kJ/mole @ 25°C
Enthalpy of fusion	24.44 kJ/mole
Enthalpy of vaporization	34.76 kJ/mole
Electrical conductivity	0.0345 10 ⁶ /cmΩ
Thermal conductivity	0.502 W/cmK

Table 1–2 Physical and Electronic Properties of Elemental Arsenic

1.2.3 Arsenic Oxidation and Reduction

Oxidation and reduction of arsenic mainly depend upon redox potential and pH conditions. In natural water bodies, arsenic predominantly exists as oxyanions of trivalent arsenite (As^{III}) or pentavalent arsenate (As^{V}). Under oxidizing conditions, the pentavalent form of arsenic predominates and the oxyanions' existence depends upon pH—at a pH less than 6.9, $H_2AsO_4^-$ predominates, while at a higher pH, $HAsO_4^{-2}$ becomes dominating. Under reducing conditions at a pH less than 9.2, the neutral trivalent arsenic species H_3AsO_3 exists, which dissociates to form anions under high pH conditions only. Most often, more trivalent arsenic than pentavalent arsenic is found in reducing groundwater conditions, whereas the pentavalent arsenic prevails under oxidizing groundwater conditions. The stabilities of various arsenic species under different pH conditions are shown in Figure 1–2, suggesting the significance of pH in the formation and occurrence of various forms of arsenic in natural water bodies. The dissociation reactions and corresponding equilibrium constants of H_3AsO_4 and H_3AsO_3 are as shown below:

Arsenite (As^{III})

$$H_3AsO_3 \leftrightarrow H_2AsO_3^- + H^+ \quad pKa: 2.24$$
 (1.1)



FIGURE 1–2 Pictorial depiction of (1) oxidation of arsenic under oxidizing and reducing conditions, (2) formation of acids by $As(^{+3})/As(^{III})$ and $As(^{+5})/As(^{V})$ under different pH conditions, and (3) dissociation of acids to oxyanions under various sets of pH conditions.

$$H_2AsO_3^- \leftrightarrow HAsO_3^{-2} + H^+ \text{ pKa: 6.69}$$
 (1.2)

$$\mathrm{HAsO}_{3}^{-2} \leftrightarrow \mathrm{AsO}_{3}^{-3} + \mathrm{H}^{+} \quad \mathrm{pKa: 11.5} \tag{1.3}$$

Arsenate (AsV)

$$H_3AsO_4 \leftrightarrow H_2AsO_4^- + H^+ \text{ pKa: 9.2}$$
 (1.4)

$$\mathrm{H}_{2}\mathrm{AsO}_{4}^{-} \leftrightarrow \mathrm{HAsO}_{4}^{-2} + \mathrm{H}^{+} \quad \mathrm{pKa: 12.1}$$

$$(1.5)$$

$$\mathrm{HAsO}_{4}^{-2} \leftrightarrow \mathrm{AsO}_{4}^{-3} + \mathrm{H}^{+} \quad \mathrm{pKa:} 13.4 \tag{1.6}$$

1.2.4 Arsenic Methylation

Methylation is a process of adding one or more methyl $(-CH_3)$ groups to the chemical species. Bio-methylation plays an important role in the metabolism of inorganic arsenic and the mechanism involves alternate steps of two-electron reduction followed by oxidative addition



FIGURE 1–3 Biotransformation of inorganic arsenic in the mammalian system. Abbreviations: SAM— S-adenosylmethionine; SAHC—S-adenosylhomocysteine; GSH—reduced glutathione; GSSG—glutathione disulfide; PNP—purine nucleoside phosphorylase.

of a methyl group known as oxidative methylation [11]. Arsenate reduction can also occur nonenzymatically under low oxygen concentration or at pH 2 or lower. S-Adenosylmethionine, a methyl donor, methylates inorganic arsenic into monomethylarsonic acid (MMA) and then to dimethylarsinic acid (DMA). This reaction is catalyzed by methyltransferases in the presence of glutathione, which acts as a co-factor. It has been reported that arsenic methylation does not depend upon arsenic concentration in water [12]; rather it is assumed to be a detoxification mechanism of arsenic as, in organic form, arsenic cannot ionize readily to interact with bio-molecules and becomes less available for chemical and bio-chemical reactions [13]. The methylation process of arsenic via alternate reduction and oxidative methylation is depicted in Figure 1–3.

1.2.5 Historical and Modern Applications of Arsenic

Burning of coal in thermal power plants and disposal of fly ash, long-term mining and smelting of the sulfide ores, runoff from mine tailings, and application of pesticides and herbicides release
huge amounts of arsenic in to the biosphere. Additionally, arsenic is also used in the production of semiconductors, lead-acid batteries, and pesticides and herbicides, in the glass industry and copper refining industry, and in the hardening of metal alloys. Use of arsenic in wood preservation is very common and has increased significantly in the last few decades [14]. Wood may deteriorate by the attack of insects, fungi, bacteria, and animals, but can be protected by impregnating with CCA with the composition CuO (18.5%), Cr₂O₃ (47.5%), and As₂O₃ (18.5%). At one time, arsenic compounds such as lead arsenate, calcium arsenate, and sodium arsenate were used as pesticides for debarking trees, to control ticks, fleas, and lice, and in aquatic weed control. However, these applications have been banned due to the toxic effects of arsenic and later public awareness about food safety and environmental contamination [15].

1.3 Arsenic Minerals and Compounds

A mineral is a naturally occurring crystalline and inorganic solid. Arsenic minerals primarily include elemental arsenic, arsenolamprite, para-arsenolamprite, and nearly 320 other inorganic compounds. Arsenic minerals can be classified under five categories: elemental arsenic, arsenides, arsenosulfides, arsenites, and arsenates. Arsenides and arsenosulfides are commonly found associated with anoxic hydrothermal ore deposits, and with metamorphic and igneous rocks. When these minerals come in contact with water and oxygen, they are rapidly converted into arsenites and arsenates. Examples of arsenic minerals falling under the five categories are shown in Table 1–3.

1.3.1 Arsenosulfides

Arsenopyrite (FeAsS), orpiment (As_2S_3), and realgar (AsS/As_4S_4) are the most common arsenic sulfide minerals, occurring primarily in hydrothermal and magmatic ore deposits. Arsenic is commonly found in sulfide-bearing mineral deposits; especially with gold mineralization. Orpiment, due to its golden color, was used in ancient times as a pigment and dye, while realgar was a common red pigment for paints and dyes. Realgar decomposes in air to a yellow-orange compound para-realgar; consequently, old unrestored paintings have a yellow-orange tinge over a red color. Arsenic can exist in sulfide minerals either as a dominant mineral-forming element or as an impurity. Arsenic release into nature is a slow process as a result of mineral weathering; however, physical forces such as grinding, crushing, and pulverization from mining activities greatly increase the release rate. Arsenosulfides also combine with transition metals such as Co, Ni, and Cu to form a variety of other sulfides and sulfosalts [16].

In arsenic sulfides, elemental arsenic and sulfur are covalently bonded with different arrangements of As-S and As-As units. In arsenopyrite, each Fe atom is octahedrally coordinated by three As and three S atoms through edges and corners. In dimeric form ([As-S)]⁻²) each As or S atom is tetrahedrally coordinated to three Fe atoms and one S-As atom. In the presence of water and oxygen, the arsenic present in arsenopyrite rapidly oxidizes to As^{+2} , As^{+3} , As^{+5} , and a precipitate in the form of scorodite (FeAsO₄·2H₂O) or amorphous Fe(III) arsenate [17].

Elemental arsenic Native arsenic As Arsenolamprite As Paraarsenolamprite As Arsenic sulfides Arsenopyrite Arsenic sulfides FeAsS Cobaltite CoAsS Orpiment As ₂ S ₃ Realgar AsS/As ₄ S ₄ Gersdorffite NiAsS Enargite Cu ₃ AsS ₄ Metal arsenides Domeykite CoAs ₂ Nickeline or niccolite Nickeline or niccolite NiAs Rammelsbergite NiAs Sperrylite PtAs ₂ Safflorite CoAs ₂ Arsenolite As ₂ O ₃ Claudetite As ₂ O ₃ Sodium arsenite NaAsO ₂ Leiteite Zn ₃ (AsO ₃) ₂ Gebhardite Pb ₆ (As ₂ O ₃) ₂ OCl ₆ Arsenate Johnbaumite Ca ₅ (AsO ₄) ₃ (OH) Mirmetite Db ₅ (AsO ₄) ₃ (OH) Miretite Ca ₅ (AsO ₄) ₃ (OH) Miretite Ca ₅ (AsO ₄) ₃ (OH) Miretite Ca ₅ (AsO ₄) ₃ (OH) <th>Group</th> <th>Mineral</th> <th>Formula</th>	Group	Mineral	Formula
Native arsenicAsArsenolampriteAsParaarsenolampriteAsArsenic sulfidesFeAsSCobaltiteCoAsSOrpimentAs ₂ S ₃ RealgarAsS/As ₄ S ₄ GersdorffiteNiAsSEnargiteCu ₃ AsSMetal arsenidesDomeykiteCu ₃ AsLollingiteFeAs2Nickeline or niccoliteNickeline or niccoliteNiAsRammelsbergiteNiAs2SaffloriteCoAs2ArsenoliteAs2O3ClaudetiteAs2O3ClaudetiteAs2O3GebharditePbg(As2O4)2OCI6ArsenateJohnbaumiteCazn(AsO4)3(CI)AustiniteCazn(AsO4)3(CI)AustiniteCazn(AsO4)3(CI)AustiniteCazn(AsO4)2H2OErythriteCoditeFeAsO4/3E/2PArsenateCazn(AsO4)2H2OErythriteCo3(AsO4)2H2OErythriteCo3(AsO4)2H2OErythriteCo3(AsO4)2H2OErythriteCo3(AsO4)2H2OErythriteCo3(AsO4)2H2OErythriteCo3(AsO4)2H2O	Elemental arsenic		
Arsenolamprite As Paraarsenolamprite As Arsenic sulfides Arsenopyrite FeAsS Cobaltite CoAsS Orpiment As_2S_3 Realgar AsS/As_4S_4 Gersdorffite NiAsS Enargite Cu_3AsS_4 Metal arsenides Domeykite Cu_3As Lollingite FeAs_2 Nickeline or niccolite NiAs Rammelsbergite NiAs Sperrylite PtAs_2 Safflorite CoAs_2 Coas_2 Arsenite Arsenolite As_2O_3 Claudetite As_2O_3 Claudetite Claudetite Zn_3(AsO_3)_2 Gebhardite Beinerite Zn_3(AsO_3)_2 Gebhardite Arsenate Johnbaumite Ca_a(AsO_4)_3(OH) Mimetite Pb_5(AsO_4)_3(CH) Austinite Scorodite FeAsO_4/2H_2O Erythrite		Native arsenic	As
Paraarsenolamprite As Arsenic sulfides Arsenopyrite FeAsS Cobaltite CoAsS Orpiment As2S3 Realgar As52S3 Realgar As52S3 Realgar As52S3 Realgar As52S3 Realgar As52S3 Realgar As52S3 Metal arsenides Cu3AS5 Metal arsenides Domeykite Cu3AS Cu3AS Lollingite FeAs2 Nickeline or niccolite NiAs Rammelsbergite NiAs2 Sperrylite PtAs2 Safflorite CoAs2 Arsenolite As2O3 Claudetite As2O3 Claudetite As2O3 Sodium arsenite NaAsO2 Leiteite Zn3(ASO3)2 Gebhardite Pb8(AS2O5)2OCI6 Arsenate Johnbaumite Johnbaumite Ca5(ASO4)3(OH) Mimetite Pb5(ASO4)3(CI) Austinite Ca2n(ASO4)2(OH) Scorodite FeAS04/2H2O Erythrite Co3(ASO4)2/3H2O		Arsenolamprite	As
Arsenic sulfides Arsenopyrite FeAsS Cobaltite CoAsS Orpiment As2s3, Realgar AsS/As4S4, Gersdorffite NiAsS Enargite Cu3AsS Metal arsenides Domeykite Cu3AsS, Cu3As Lollingite FeAs2 Nickeline or niccolite NiAs Rammelsbergite NiAs2 Sperrylite PtAs2 Safflorite CoAs2 Arsenite Arsenolite Arsenite As2O3 Claudetite As2O3 Sodium arsenite NaAsO2 Leiteite Zn3(ASO3)2 Gebhardite Pb8(As2O3)2Cl6 Arsenate Johnbaumite Johnbaumite Ca5(AsO4)3(OH) Miretite Pb5(AsO4)3(Cl) Austinite Ca7(AsO4)2(H) Scordite FeAsO4/2H2O Erythrite Co3(AsO4)2/3H2O		Paraarsenolamprite	As
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	Arsenic sulfides		
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$		Arsenopyrite	FeAsS
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$		Cobaltite	CoAsS
RealgarAsS/As4S4GersdorffiteNiAsSEnargiteCu3AsS4Metal arsenidesDomeykiteCu3AsLollingiteFeAs2Nickeline or niccoliteNiAsRammelsbergiteNiAs2SperrylitePtAs2SaffloriteCoAs2ArsenoliteAs2O3ClaudetiteAs2O3Sodium arseniteNaAsO2LeiteiteZnAs2O4ReineriteZnAs2O4ArsenateDomeykiteLeiteiteZnAs2O4KeineriteZn3(AsO3)2GebharditePb8(As2O3)2OCI6ArsenateJohnbaumiteCa5(AsO4)3(OH)MimetiteMimetitePb5(AsO4)3(CI)AustiniteCa7n(AsO4)(OH)ScoroditeFeAs04·2H2OErythriteCo3(AsO4)2·8H2O		Orpiment	As ₂ S ₃
Gersdorffite Enargite NiAsS Cu ₃ AsS ₄ Metal arsenides Domeykite Cu ₃ As Lollingite FeAs ₂ Nickeline or niccolite NiAs Rammelsbergite NiAs ₂ Sperrylite PtAs ₂ Safflorite CoAs ₂ Arsenite Arsenolite Arsenite NaAsO ₂ Leiteite ZnAsO ₃ Sodium arsenite NaAsO ₂ Leiteite ZnAsO ₃ Gebhardite Pb ₈ (As ₂ O ₃) ₂ Gebhardite Pb ₅ (AsO ₄) ₃ (Cl) Austinite CaZn(AsO ₄)(OH) Scorodite FeAsO ₄ ·2H ₂ O Erythrite Co ₃ (AsO ₄) ₂ ·8H ₂ O		Realgar	AsS/As ₄ S ₄
EnargiteCu ₃ AsS ₄ Metal arsenidesDomeykiteCu ₃ AsLollingiteFeAs2Nickeline or niccoliteNiAsRammelsbergiteNiAs2SperrylitePtAs2SaffloriteCoAs2ArseniteArsenoliteArsenoliteAs2O3ClaudetiteAs2O3Sodium arseniteNaAsO2LeiteiteZnAs2O4ReineriteZn3(AsO3)2GebharditePb8(As2O5)2OCI6ArsenateJohnbaumiteCa5(AsO4)3(CI)AustiniteCa2n(AsO4)3(CI)AustiniteScoroditeFeAsO4·2H2OErythriteCo3(AsO4)2·8H2O		Gersdorffite	NiAsS
Metal arsenides Domeykite Cu ₃ As Lollingite FeAs2 Nickeline or niccolite NiAs Rammelsbergite NiAs2 Sperrylite PtAs2 Safflorite CoAs2 Arsenolite As203 Claudetite As203 Sodium arsenite NaAs02 Leiteite ZnAs204 Reinerite Zn3(As03)2 Gebhardite Pb8(As205)20Cl6 Arsenate Johnbaumite Johnbaumite Ca5(As04)3(CH) Mimetite Pb5(As04)3(CH) Austinite CaZn(As04)(OH) Scorodite FeAs04:2H20 Erythrite Co3(As04)2:8H20		Enargite	Cu ₃ AsS ₄
$\begin{tabular}{l lllllllllllllllllllllllllllllllllll$	Metal arsenides		
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$		Domeykite	Cu ₃ As
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$		Lollingite	FeAs ₂
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$		Nickeline or niccolite	NiAs
Sperrylite Safflorite PtAs2 CoAs2 Arsenite Arsenolite As2O3 Claudetite Assonative As2O3 Sodium arsenite NaAsO2 Leiteite ZnAs2O4 Reinerite ZnAs2O4 Reinerite Znas2O4 Reinerite Znas2O4 Reinerite Johnbaumite Pb8(As2O5)2OCI6 Arsenate Johnbaumite Ca5(AsO4)3(OH) Mimetite Austinite CaZn(AsO4)3(OH) Scorodite FeAsO4·2H2O Erythrite Co3(AsO4)2·8H2O		Rammelsbergite	NiAs ₂
Safflorite CoAs2 Arsenite Arsenolite As203 Claudetite As203 Sodium arsenite NaAsO2 Leiteite ZnAs204 Reinerite Zn3(AsO3)2 Gebhardite Pb8(As2O5)2OCI6 Arsenate Johnbaumite Ca5(AsO4)3(OH) Mimetite Pb5(AsO4)3(CI) Austinite CaZn(AsO4)(OH) Scorodite FeAsO4·2H2O Erythrite Co3(AsO4)2·8H2O		Sperrylite	PtAs ₂
Arsenite Arsenolite As203 Claudetite As203 Sodium arsenite NaAs02 Leiteite ZnAs204 Reinerite Zn3(As03)2 Gebhardite Pb8(As205)20Cl6 Arsenate Johnbaumite Cazn(AsO4)3(CH) Mimetite Austinite CaZn(AsO4)3(CH) Scorodite FeAsO4·2H2O Erythrite Co3(AsO4)2·8H2O		Safflorite	CoAs ₂
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	Arsenite		
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$		Arsenolite	As ₂ O ₃
Sodium arsenite NaAsO2 Leiteite ZnAs2O4 Reinerite Zn3(AsO3)2 Gebhardite Pb8(As2O5)2OCI6 Arsenate Johnbaumite Ca5(AsO4)3(OH) Mimetite Pb5(AsO4)3(CI) Austinite CaZn(AsO4)(OH) Scorodite FeAsO4·2H2O Erythrite Co3(AsO4)2·8H2O		Claudetite	As ₂ O ₃
LeiteiteZnAs2O4ReineriteZn3(AsO3)2GebharditePb8(As2O5)2OCI6ArsenateDohnbaumiteJohnbaumiteCa5(AsO4)3(OH)MimetitePb5(AsO4)3(CI)AustiniteCaZn(AsO4)(OH)ScoroditeFeAsO4·2H2OErythriteCo3(AsO4)2·8H2O		Sodium arsenite	NaAsO ₂
Reinerite GebharditeZn ₃ (AsO ₃)2 Pb ₈ (As2O ₅)2OCI6ArsenateDohnbaumite MimetiteCa ₅ (AsO ₄) ₃ (OH) Pb ₅ (AsO ₄) ₃ (CI) AustiniteAustiniteCaZn(AsO ₄)(OH) ScoroditeFeAsO ₄ ·2H ₂ O ErythriteErythriteCo ₃ (AsO ₄)2·8H ₂ O		Leiteite	ZnAs ₂ O ₄
Gebhardite Pb ₈ (As ₂ O ₅) ₂ OCl ₆ Arsenate Johnbaumite Ca ₅ (AsO ₄) ₃ (OH) Mimetite Pb ₅ (AsO ₄) ₃ (Cl) Austinite CaZn(AsO ₄)(OH) Scorodite FeAsO ₄ ·2H ₂ O Erythrite Co ₃ (AsO ₄) ₂ ·8H ₂ O		Reinerite	Zn ₃ (AsO ₃) ₂
Arsenate Johnbaumite Ca ₅ (AsO ₄) ₃ (OH) Mimetite Pb ₅ (AsO ₄) ₃ (Cl) Austinite CaZn(AsO ₄)(OH) Scorodite FeAsO ₄ ·2H ₂ O Erythrite Co ₃ (AsO ₄) ₂ ·8H ₂ O		Gebhardite	Pb ₈ (As ₂ O ₅) ₂ OCl ₆
Johnbaumite Ca5(AsO4)3(OH) Mimetite Pb5(AsO4)3(Cl) Austinite CaZn(AsO4)(OH) Scorodite FeAsO4·2H2O Erythrite Co3(AsO4)2·8H2O	Arsenate		
Mimetite $Pb_5(AsO_4)_3(CI)$ Austinite $CaZn(AsO_4)(OH)$ Scorodite $FeAsO_4 \cdot 2H_2O$ Erythrite $Co_3(AsO_4)_2 \cdot 8H_2O$		Johnbaumite	Ca ₅ (AsO ₄) ₃ (OH)
AustiniteCaZn(AsO ₄)(OH)ScoroditeFeAsO ₄ ·2H2OErythriteCo3(AsO4)2·8H2O		Mimetite	Pb ₅ (AsO ₄) ₃ (Cl)
ScoroditeFeAsO4:2H2OErythriteCo3(AsO4)2·8H2O		Austinite	CaZn(AsO ₄)(OH)
Erythrite Co ₃ (AsO ₄) ₂ ·8H ₂ O		Scorodite	FeAsO ₄ ·2H ₂ O
		Erythrite	Co ₃ (AsO ₄) ₂ ·8H ₂ O

Table 1–3Examples of Arsenic Minerals Classified under theFive Categories

In realgar and its polymorphs As-As and S-S, dimers form a discrete molecular cage-like structure in which units are connected by van der Waals forces. Polymorphs of realgar, alacrinite (As₈S₉), and amorphous arsenic sulfide can also occur in hydrothermal deposits, volcanic emissions, intrusive igneous rocks, and hot springs. Realgar exists as high and low temperature polymorphs and at about 240°C α -realgar converts to β -realgar. α -Realgar in exposure to oxygen and sunlight converts to another polymorph, para-realgar.

In orpiment (As_2S_3) , molecular units are present in chain form and connected by bridging S atoms and cross-linked by van der Waals attraction forces. Both the minerals realgar and orpiment are stable over a wide range of temperature; however, orpiment possesses a greater stability range due to reduced sulfur fugacity [18].

Oxidation of arsenopyrite is the widespread mechanism for the distribution of arsenic into the environment. Arsenopyrite is formed under high temperature and a reductive environment, such as areas around buried plant roots or other nuclei of decomposing organic matter. Pyrite readily oxidizes in aerobic conditions with the formation of iron oxides and traces of arsenic. Arsenic is also found associated with phosphate minerals; however, the concentrations are less than those of oxide and sulfide minerals. By substituting Si⁴⁺, Al³⁺, Fe³⁺, and Ti⁴⁺ arsenic can also be found associated with many other minerals; however, the concentrations are comparatively less.

Other arsenosulfides of transition metals are also known and the most common are enargite (Cu₃AsS₄), cobaltite (CoAsS), and gersdoeffite (NiAsS). Enargite is orthorhombic thioarsenate where each arsenic atom is coordinated with four sulfur atoms as (AsS₄)⁻³ [19]. Below 300°C enargite is converted to its polymorph luzonite with tetragonal geometry.

1.3.2 Metal Arsenides

A large number of metal arsenides are known and generally can be considered as metal alloys with the formula MAs_n , where M is a metal and n = 1, 2, or 3. In these compounds arsenic exists in a negative oxidation state; however, bonding is strongly covalent and the compounds are semiconducting in nature. The well-known semiconductor is GaAs while other arsenides can be formed with Fe, Co, Ni, and Cu. Naturally occurring common metal arsenides are domeykite (Cu₃As), lollingite (FeAs₂), nickeline or niccolite (NiAs), nickel-skutterudite (Ni,Co) As₃, rammelsbergite (NiAs₂), safflorite (CoAs₂), and sperrylite (PtAs₂).

In skutterudite, each Co atom or any other divalent metal atom is surrounded by six arsenic atoms in octahedral mode, while in lollingite each Fe atom is coordinated with six arsenic atoms in orthorhombic geometry [20]. Another important metal arsenide is GaAs, which is synthetic in nature and each arsenic atom is coordinated by four Ga atoms [21]. It has various applications as a semiconductor in solar cells, light emitting diodes (LEDs), laser windows, etc. Other synthetic arsenides are indium (III) arsenide (InAs) and mixed indium gallium arsenide (InGaAs) with similar applications.

1.3.3 Arsenites

The most common arsenites are polymorphs of As(III) oxides, arsenolite (As_2O_3 , isometric), and claudetite (As_2O_3 , monoclinic), with similar thermodynamic stability, while claudetite is thought to be slightly more stable at standard conditions [18]. These oxides occur naturally as secondary weathering products of arsenic sulfides and industrially during combustion of arsenic-bearing minerals and coal in gaseous form, which upon condensation gives the powdered form of white As_2O_3 . The structure of arsenolite consists of $<As_4O_6>$ cages, while in claudetite each arsenic atom is surrounded by three oxygen atoms in a trigonal pyramidal structure. Arsenic trioxides are moderately soluble in water and have been used industrially as pigments

and pesticides for a long time. The solubility of arsenite salts is largely due to the type of cation in the compound. The most soluble are the alkali arsenites, whereas metal arsenites are the least soluble.

1.3.4 Arsenates

Arsenates are usually considered a subclass of the phosphate mineral group owing to the similarity in size and charge of the phosphate and arsenate anionic unit. In arsenate, the unit $(AsO_4)^{-3}$ is either tetrahedrally or octahedrally coordinated to transition metal or alkaline earth metal with anions $(OH^-, Cl^-, and F^-)$ for charge balancing. Similarly to phosphate minerals arsenate minerals also occur in a variety of soil and oxidized environments with a range of waters of hydration. During the period of the 1870s – 1970s arsenates of copper, lead, and calcium were extensively used as herbicides and insecticides and significantly increased the environmental burden of arsenic. As_2O_5 is rarely found in nature; however, it was synthesized and widely used at one time in the production of copper arsenate, a wood preservative. As_2O_5 decomposes at 315°C to As_4O_6 and O_2 . In water it can dissolve to form various species, H_3AsO_4 , $H_2AsO_4^-$, $HAsO_4^{-2}$, and AsO_4^{-3} , depending upon the pH of the solution.

1.4 Organoarsenicals

In organoarsenic compounds arsenic combines easily with carbon to form various organic compounds with one or more As-C bonds and is widely used in agriculture and plant protection agents. A wide variety of organoarsenic compounds are found naturally in the environment as a consequence of bio-methylation and other biosynthetic pathways. They can be grouped into aliphatic and aromatic organoarsenic compounds. The most common classes of natural organoarsenic compounds are methylated forms of arsenite and arsenate. These can be generated by replacing a hydroxyl (-OH) ligand by a methyl group (-CH₃). The structures of major organoarsenic compounds are depicted in Figure 1-4. By the replacement of one or more methyl or hydroxyl groups around arsenic by other organic moieties such as sugars, lipids, or cyclic groups, a wide variety of arsenic-bearing organic compounds are generated. Major aliphatic organoarsenic compounds methane-arsenic acid and dimethylarsenic acid were at one time used as fungicides, herbicides, and desiccants. On the other hand, aromatic organic arsenicals such as arsanilic acid have been used as corrosion inhibitors for iron and steel, and as additives for motor fuel, agricultural bactericides, herbicides, and fungicides. In organoarsenicals arsenic can exist in three oxidation states: As(I), As(III), and As(V). Compounds of oxidation state As(I) feature three bonds to As but only an As-As single bond and are used in antisyphylitic drugs such as Salvarsin and Neosalvarsan. Arsenic (V) compounds show the general formula $RAsO(OH)_2$ or $R_2AsO(OH)$ (R = alkyl or aryl). Monomethylarsonic acid (CH₃AsO(OH)₂) and dimethylarsinic acid, commonly known as cacodylic acid ($(CH_3)_2AsO_2H$), figure prominently throughout the chemistry of organoarsenic compounds. Arsenic (III) compounds are prepared by alkylation of AsCl₃ and its derivatives



FIGURE 1-4 Structure of major methylated organoarsenic compounds.

using organolithium and Grignard reagents [22], e.g., methylarsenic dichloride (CH_3AsCl_2), dimethylarsenic chloride ((CH_3)₂AsCl), and trimethylarsine ((CH_3)₃As). Reduction of these chloride derivatives produces hydrides, such as dimethylarsine ((CH_3)₂AsH) and methylarsine (CH_3AsH_2). Trimethylarsine and triphenylarsine are symmetrical organoarsenic (III) compounds, commonly used as ligands in coordination chemistry.

1.4.1 Organoarsenicals in the Food Chain

Inorganic arsenic and its compounds enter the food chain and metabolize through a methylation process known as bio-methylation in the presence of specific enzymes related to vitamin B_{12} . The organic compound arsenobetaine is found in some marine foods such as fish and algae, and is considered a non-toxic form of organoarsenic compound. Carbohydrates bound to arsenic known as arsenosugars are found especially in seaweeds, and if lipid is bound to arsenic an arsenolipid is formed [23].



FIGURE 1–5 Structure of three different lewisites illustrating the number of chlorine atoms in each that are responsible for its reaction with bio-molecules.

1.4.2 Organoarsenicals in Chemical Weapons

Lewisite (2-chloroethenylarsinous dichloride), an organoarsenic compound initially manufactured in the USA and Japan as a chemical weapon, acts as a blister agent and lung irritant. It is synthesized by the addition of arsenic trichloride to acetylene in the presence of a suitable catalyst. It is usually found as a mixture of 2-chlorovinylarsonous dichloride (lewisite 1) as well as *bis*(2-chloroethenyl) arsinous chloride (lewisite 2) and *tris*(2-chlorovinyl)arsine (lewisite 3) (Figure 1–5). It diminishes energy metabolism of the human body by inhibiting activity of pyruvate dehydrogenase necessary for the generation of ATP via converting pyruvate to acetyl-CoA. Sequentially, nervous pathology arises as the nervous system essentially depends on glucose as a catabolic fuel. It can easily penetrate ordinary clothing and rubber, and causes immediate pain and itching on skin with a rash and swelling, which convert to large, fluidfilled blisters after approximately 12 hours. Eye exposure causes stinging and strong irritation to blistering and scarring of the cornea. Sufficient absorption causes systemic poisoning leading to liver necrosis or death.

1.4.3 Methylation in Mammals

In mammals, methylation of inorganic arsenic compounds occurs in the liver, via the activity of methyltransferases, to $(CH_3)_2As^{III}OH$ (dimethylarsinous acid) and $(CH_3)_2As^V(O)OH$ (dimethylarsinic acid). The source of the methyl groups is methionine, in the form of S-adenosyl methionine in the process of detoxification of inorganic arsenicals [24]. When the liver's methylation capacity is inhibited by the excess exposure to inorganic arsenic compounds, an imbalance occurs, which has deleterious effects on various organs and leads to carcinogenesis.

1.5 Arsenic Mobilization in the Environment

Arsenic is mobilized in the environment via various natural processes such as weathering of rocks, hydrothermal and geothermal activities, biological activities, and a range of



FIGURE 1–6 Arsenic mobilization in air, water, and soil. Arsenic accumulation via weathering and mining, and oxidative dissolution of arsenic-bearing mineral releases to the soil. Arsenic undergoes reduction, methylation, demethylation, and precipitation reaction and finally returns to soil and forms arsenic-rich sediments. Industrial processes, burning of coal, and use of pesticides release arsenic into the atmosphere. None of the various species of arsenic escapes from the environment; rather they continue to cycle from one species to another and from one medium to another.

anthropogenic activities. Widespread anthropogenic activities include mining, combustion of fossil fuels, use of arsenical pesticides, herbicides and crop desiccants, and the use of arsenic as an additive to livestock feed, particularly for poultry (Figure 1–6). Arsenic may enter the human body through drinking water, inhalation, and diet, but above all drinking water probably possesses the greatest menace to human health. Drinking water contains arsenic at various concentrations depending on the source. However, ground water holds the highest concentration of arsenic owing to various natural and human activities.

1.5.1 Aqueous Chemistry of Arsenic

Arsenic is ubiquitously present in the aquatic environment with concentrations near $0.62 \mu g/L$ in river water and $0.5-3.0 \mu g/L$ in ocean water [25]. Based on redox conditions and pH, arsenic can exist in all possible oxidation states +5, +3, 0, and -3, among which As(III) and As(V) are the most dominating in aqueous systems [26]. Various forms of arsenic are distributed in the environment; the most common species are listed in Table 1–4.

Arsenic readily forms oxyanions in both oxidation states As(III) (H₃AsO₃) and As(V) (H₃AsO₄), and becomes mobile in the pH range of 6.5–8.5 usually found in ground water.

Name	Abbreviation	Chemical Formula
Arsenite, arsenous acid	As ^{III}	As(OH) ₃
Arsenate, arsenic acid	As ^V	AsO(OH) ₃
Monomethylarsonic acid	MMA ^V	CH ₃ AsO(OH) ₂
Monomethylarsonous acid	MMA ^{III}	$CH_3As(OH)_2$
Dimethylarsinic acid	DMA ^V	(CH ₃) ₂ AsO(OH)
Dimethylarsinous acid	DMA ^{III}	(CH ₃) ₂ AsOH
Trimethylarsine oxide	TMAO	(CH ₃) ₃ AsO
Trimethylarsine	TMA ^{III}	(CH ₃) ₃ As
Arsenobetaine	AsB	$(CH_3)_3As^+CH_2COO^-$
Arsenocholine	AsC	(CH ₃) ₃ As ⁺ CH ₂ CH ₂ OH
Tetramethylarsonium ion	Me ₄ As ⁺	(CHM ₃) ₄ As ⁺
Dimethylarsinoyl ethanol	DMAE	(CHM ₃) ₂ AsOCH ₂ CH ₂ OH

 Table 1–4
 List of Various Arsenic Species and their Chemical Formulae

At this pH range, most of the toxic trace metal ions remain in solution form and precipitate or co-precipitate with an increase in pH as an oxide, hydroxide, carbonate, or phosphate. In contrast, oxyanion-forming elements such as arsenic, chromium, and selenium also exist in solution at higher pH values. Among all oxyanion-forming elements arsenic is unique in terms of its relative mobility over a wide range of redox conditions. Other oxyanion-forming elements such as Se, Cr, U, Mo, and V remain mobile under oxidizing conditions only and become immobilized in a reducing environment due to stronger adsorption or reduction to metallic form. On the other hand, arsenic oxyanions remain mobile under both oxidizing and reducing conditions.

Redox potential (Eh) and pH are the most important factors controlling arsenic speciation in aqueous systems. The redox potential of arsenic oxyanions is such that arsenite becomes stable in aqueous form under moderately reducing conditions (+300 mV at pH 4 to - 200 mV at pH 9) while arsenate is stable in oxidized aqueous solutions [18]. Under oxidizing conditions and low pH (less than 6.9), arsenic exists as As(V) and H_2AsO_4 species dominate, while at higher pH due to ionization of one more proton, $HAsO_4^{-2}$ exists predominantly. In reducing conditions and pH less than 9.2, the neutral arsenite species H₃AsO₃ predominates. The structural configuration of arsenous acid (As^{3+}) reveals that arsenic has three pyramidal bonds with a lone pair of electrons occupying the fourth arm of a tetrahedron (Figure 1-4). It has been suggested spectroscopically that in solution form also, arsenite retains its pyramidal structure [27]. Arsenous acid showing a pKa of 9.23 dominates in natural aqueous solution, while deprotonated forms can exist under alkaline conditions. The geometry of arsenic acid suggests a regular tetrahedron, which can easily donate protons under alkaline conditions to form the stable arsenate $(AsO_4)^{-3}$ anion, which resembles the orthophosphate $(PO_4)^{-3}$ anion and shows competitive chemical behavior in natural water systems. The adsorption and desorption behavior of arsenic on mineral surfaces plays an important role in regulating its aqueous concentration in various water systems. In the presence of reduced sulfur, arsenic exists in solution form as sulfides, while under reducing conditions these sulfides (As₂S₃, AsS) become precipitated. Arsenic species are also affected by temperature, such as in hydrothermal fluids from shallow-water islands; the dominant species between pH 5 and 7 is trivalent $As(OH)_3$, whereas at higher temperature and pressure in deep-water systems the $H_2AsO_4^-$ arsenate species predominates. Moreover, at high temperature and high concentration of Fe and S, orpiment and claudetite preferentially precipitate. Thio-arsenic species are also possible in sulfide-rich hydrothermal systems; such thio-arsenates were reported in the geothermal waters of the Yellowstone National Park [28].

1.5.2 Redox-Dependent Mobilization of Arsenic

Redox reactions are of great significance for controlling the behavior of many chemical species in natural water systems, including arsenic. Thermodynamic equilibrium suggests that under strongly reducing conditions As(V) should dominate over As(III); however, this is not necessarily followed in all natural systems where other redox couples also influence arsenic oxidation/ reduction. In oxygenated water, the As(V)/As(III) ratios should be of the order of 10^{15} – 10^{26} , which remain stable for a period of days or weeks due to slow oxidation reactions. It has been suggested that the As(V)/As(III) ratio can be used as a reliable redox indicator in groundwater systems [29]. Experimental studies suggested that oxidation of As(III) becomes slow under acidic conditions (pH 5), while at pH 8–12.5 the reaction accelerates. The rate of As(III) oxidation increases in the presence of manganese oxide, which is evident by the fact that it can reduce the half-life of As(III) from 1–3 years to 10–20 minutes. Oxidation reaction is independent of the concentration of dissolved oxygen; however, it is controlled by surface reactions such as photochemical oxidation and reduction triggered by titanium-containing particles [30].

1.5.3 Microbial-Dependent Mobilization

Redox reaction of As(III)/As(V) is also catalyzed by microorganisms; thus sterile water samples are less susceptible to speciation changes. Arsenite in geothermal water in the southwest USA was found to oxidize rapidly with a rate constant of less than 0.3 hours. Arsenate reduced rapidly in Mono Lake, CA, in the presence of bacteria with rate constant ranging from 0.02 to 0.3 per day [31]. Methylated arsenic species also readily oxidized chemically and biologically.

1.6 Sources of Arsenic in the Biosphere

Arsenic is a natural constituent of Earth's crust and ranks 20th in abundance in relation to the other elements; it is found in approximately 320 mineral forms in rocks. Arsenic can enter terrestrial and aquatic environments via both natural geologic processes and anthropogenic activities. Natural occurrence of arsenic depends upon various processes such as weathering of rocks and sediments, hydrothermal and geothermal activities, dissolution of sulfide minerals, forest fire, and wind-blown dust [32]. Anthropogenic sources of arsenic include mineral processing, glass manufacturing, wood preservation, pesticide production and uses, leaching through landfill or waste, and production of coal and petroleum products and their processing. Anthropogenic sources of arsenic are responsible for its release into the atmosphere via emission due to copper smelting and coal combustion and into the hydrosphere and lithosphere through all types of waste [33]. Major natural and anthropogenic sources of arsenic



FIGURE 1–7 Major natural and anthropogenic sources of arsenic entry in to the environment.

are depicted in Figure 1–7. The detailed process of arsenic release from various sources is discussed in subsequent sections.

1.6.1 Arsenic in Rocks and Soils

Historically, the concentration of arsenic in Earth's crust was reported to be 2 mg/kg and subsequently it was estimated in various forms of rocks and soils. Indisputably, the arsenic burden has increased at a tremendous rate on planet Earth and varies according to the types of rocks, sediments, and soils. In igneous rocks, arsenic concentrations vary from 1.5 to 5.9 mg/kg with the major contributor being volcanic material generating high-arsenic water [34]. Metamorphic rocks generally show less than 5 mg/kg arsenic with the exception of pelitic rocks containing unusually high concentrations, 18 mg/kg. In sedimentary rocks, arsenic concentrations range between 5 and 10 mg/kg. Argillaceous deposits have a high arsenic concentration (13 mg/kg) probably owing to enhanced sulfide mineral, oxides, clays, and organic matter. Iron-rich rocks, phosphorites, coal, and bituminous deposits are enriched with exceptionally high arsenic concentration ranges of between 100 and 900 mg/kg [35].

Marine deposits due to higher sulfur content contain arsenic at greater concentrations compared to non-marine deposits. Historically, the average concentration of arsenic in deepsea sediments was estimated to be approximately 40 mg/kg. The most important arsenic minerals in the oceanic crust are cobaltite (CoAsS), loellingite-safflorite (FeS₂(Co,Fe)As₃), luzonite (Cu₂AsS₄), tennantite (Cu₁₀(Fe,Zn)₂As₄S₁₃), claudetite, and elemental arsenic, principally intergrown with elemental gold [36]. In oceanic sediments, arsenic is usually concentrated in finegrain sediments, particularly those rich in organic matter, sulfide minerals, phosphate, or iron oxides [37]. Elevated arsenic concentrations were found in river sediments in India, from the Ganges plain ranging between 1.2 and 2.6 mg/kg, from the Brahmaputra river (between 1.4 and 5.9 mg/kg), and from the Meghna river (between 1.3 and 5.6 mg/kg) [38].

Availability of metals to organisms is not only dependent upon the total amount of a metal present in the soil, but also on the nature of chemical bonding of the metal with the soil particles. Arsenic concentration in soil also varies between 5 and 10 mg/kg, with peats and bog soils having higher values (13 mg/kg) due to increased pervasiveness of sulfide minerals under reduced conditions. The presence of arsenic in soils primarily depends upon geological reactions. However, it is also greatly enhanced by smelting and fossil fuel combustion, mining activity, and use of pesticides and phosphate fertilizers. Previously in orchard soils, the concentration of arsenic was reported in the range of 366-732 mg/kg due to excessive application of arsenical pesticides. Arsenic occurs in soils mainly as arsenate, AsO₄³⁻, under oxic conditions. This form of arsenic is strongly adsorbed onto clays, oxides/hydroxides of Fe and Mn (AlAsO₄ and FeAsO₄), and organic matter. In alkaline and calcareous soils, the main form is $Ca_3(AsO_4)_2$. Arsenic leaching from surface soil becomes higher in sandy soil due to weak sorption. The amount of sorption depends upon arsenic concentration, contact time, and concentration of Fe and manganese in the soil. The mobility of As in soils is increased under reducing conditions such as flooded soils. An increase in pH also increases the mobility of the arsenic apparently by bringing about a change from Al and Fe arsenates to calcium arsenate. Bacteria are also important in increasing the mobility of arsenic as they increase the bio-methylation of arsenic, the compounds formed being comparatively mobile forms of arsenic. Various factors regarding the mobilization of arsenic in the soil are presented in Figure 1-8. The natural occurrence of arsenic-bearing minerals may significantly enrich the soil concentration of arsenic. Soils in the vicinity of gold mines as well as gold processing wastes often contain elevated concentrations of arsenic [39]. As mentioned above, arsenic accumulation in soil not only occurs



FIGURE 1-8 Various factors affecting arsenic mobilization in soil.

via natural sources but also from anthropogenic sources such as mining, airborne emissions, waste discharges from metal processing facilities, burning of fossil fuels, use of wood preservatives, and agricultural application of arsenical pesticides [40].

Since the beginning of the industrial revolution, the most common human-contributing sources leading to the introduction of trace metals into the soil have included disposal of industrial effluents, sewage sludge, mining, landfills, and use of agricultural chemicals [41]. Arsenic contamination of soils may cause loss of vegetation, groundwater pollution, and arsenic toxicity in plants, animals, and humans [42]. The chemistry and behavior of arsenic is determined by arsenic speciation rather than its total amount. Conversion of the arsenic forms is also affected by different soil factors, such as soil reaction, oxidation-reduction state, microbial activity, organic matter content, etc. Wood impregnation plant. To carry out the processing, chiefly arsenic pentoxide (As_2O_5) is used, which is easily dissolved in water (65.8 g/L). Another method of wood impregnation is by pressure treatment with chromated copper arsenate (CCA). In many countries use of CCA for wood preservation has already been banned.

The bio-availability of arsenic in soil depends upon physical and chemical properties of soil and water such as pH, redox conditions, presence of precipitating, complexing, and sorbing agents, and mineralogical composition of the soil and its grain size. In addition, climatic, hydrogeological, and geomorphological activities also influence the mobility and availability of arsenic. Arsenic leached in the pentavalent form is able to form anions, such as $H_2AsO_4^-$, of a strong acid, with pKa values of 2.24, 6.94, and 11.5, which adsorb most effectively at low pH. Consequently, its mobility is fairly low in acidic soils with high clay or oxide content.

1.6.2 Arsenic in the Atmosphere

In 1936 James Svoboda first described mass eradication of honeybee families (Tisin's disease) as a most sensitive ecological indicator of arsenic pollution in air [43]. The phenomenon was observed in the vicinity of various smelters and power plants that burned arsenic-containing coal. In the atmosphere, arsenic concentration increases from smelting, fossil-fuel combustion, and volcanic activity.

Arsenic emission depends upon its concentration in raw material, quality of fuel or waste, production procedure, temperature, pressure, gas composition, and gas cleaning equipment. Due to specific evaporation-condensation processes at high temperatures, arsenic partially or entirely evaporates into the atmosphere. The diameter of arsenic particles dispersed in the atmosphere was typically found to be $1 \mu m$, suggesting efficient movement with wind and air [44]. Combustion of fossil fuels releases arsenic to the atmosphere with typical levels of 0.1-80 mg/kg (dry) for coal and biomass fuels, and up to 500 mg/kg (dry) for sewage sludge.

Airborne arsenic is transferred to water bodies by wet or dry precipitation and increases aqueous concentration. Atmospheric arsenic arising from coal burning has been cited as a major cause of lung cancer in parts of China and India; exposure is known to be lethal from direct inhalation from domestic coal fires [45].

1.6.2.1 Arsenic Emission due to Burning of Coal

Coal is a natural source of arsenic and primarily responsible for the release of arsenic into the air, via smoke stacks during coal combustion, and into soil and water via combustion of products, mainly fly ash. India has the world's fourth largest reserves of coal and the third largest coal excavator, after the United States and China, serving the electricity demands of the nation. Electricity generation is mainly dependent upon coal-based thermal power plants. In eastern India, coal combustion is one of the major sources of anthropogenic arsenic emission into the biosphere. Coal contains both inorganic and organic forms of arsenic; however, its enrichment is related to sulfide mineralization, either syngenetic or epigenetic [46]. The arsenic-bearing sulfides are epigenetic and contain high concentrations of arsenic, whereas pyrites are syngenetic containing very low arsenic levels [47]. Release of arsenic during coal combustion depends upon various factors such as initial concentration in the parent coal, design and operating conditions of the furnace, and combustion temperature. During low temperature combustion (1000–1200°C) AsO and As₂O₃ escape in the gaseous phase, while at high temperature (1200–1600°C) only As₂O₃ is released.

1.6.2.2 Arsenic Dissipation from Fly Ash

Fly ash is produced in substantial amounts around the world and contributes as a potential and significant anthropogenic source of arsenic [48]. During combustion of coal in thermal power plants, the main part of the initial arsenic evaporates into the atmosphere in the gaseous phase. Most of the organic arsenic, pyrite, and some micro-minerals also escape into the atmosphere in the gaseous phase and only a minor part of arsenic remains in bottom ash. The escaping arsenic is captured by fly ash and collected by an electrostatic precipitator. This suggests that the atmospheric emission of arsenic in solid and gaseous phases is rather minor and most of the arsenic ends up with fly ash. Upon cooling, the combustion gases are adsorbed onto fly ash particles and their concentration becomes comparatively higher in particles less than 10 µm due to high surface area. Particles of less than 1 µm when inhaled may be deposited in the pulmonary tissue of the respiratory tract and enter the blood. Release of arsenic from fly ash produces air, soil, and water pollution. Excessive amounts of arsenic over permissible levels have been reported in the soil, tubewell, and surface water near thermal power plants. Arsenic associated with fly ash ends up in ground water through leaching and adversely affects water quality. Reports suggest that arsenic is highly leachable in fly ash as it occurs as a surface precipitate [49]. Arsenic concentration increases with time in ground water from acidic to alkaline fly ash leaching. However, under alkaline conditions, after attaining a maximum level concentrations start declining with time. Many factors such as pH, concentration of Ca/Mg, reducing or oxidizing conditions, leaching time, temperature, and anionic constituents (sulfate and phosphate) strongly influence leaching of arsenic from fly ash [50].

The extent of the arsenic in fly ash depends on both the mineralogy and particle size distribution of the coal and combustion temperature. Arsenic concentrations in fly ash generally range from 2 to 440 mg/kg and sometimes can be as high as 1000 mg/kg depending upon the concentration in the original coals, the combustion methods, and combustion temperature [49]. The toxicity of arsenic when present with fly ash depends upon its state, mobility, and

Name of Species	Formula
Arsenopyrite	FeAsS
Arsenic trioxide	As ₂ O ₃
Calcium orthoarsenate	Ca ₃ (AsO ₄) ₂
Scorodite	FeAsO ₄ ·2H ₂ O
Thallian arsenian pyrite	(Fe,Tl)(S,As) ₂

Table 1–5	Various	Forms o	f Arsenic
Present in	Fly Ash		

availability in the biosphere. Arsenic concentration was found to be higher in hollow spherical particles (low density) than in compact spherical particles (high density) [51]. The concentration of arsenic in fly ash also depends upon sulfur content of the coal and is found higher in those feed coals with high sulfur content. The pH of fly ash also influences concentration and species of condensed arsenic. In an acidic medium, fly ash contains more Si, thus occurring as $Fe_3[AsO_4]_2$, while in alkaline medium due to an increase in Ca concentration it is present as $Ca_3[AsO_4]_2$. Arsenic is captured by calcium-bearing minerals and hematite, and forms a stable complex with Ca or Fe in fly ash. It has been reported that arsenic is largely present in Ca-rich fly ash as $Ca_3(AsO_4)_2$ and $Ca_2As_2O_7$ because Ca can react with arsenic vapor and capture the metal in water-insoluble forms as As(V) [52]. Fly ash may contain various forms of arsenic depending upon pH condition, particle size, and presence of other minerals, as summarized in Table 1–5.

1.6.3 Arsenic in Various Water Resources

1.6.3.1 Rain Water

Arsenic enters the atmosphere in various ways including volcanic emissions, marine aerosols, burning of fossil fuels, and industrial pollution, and is then reversed in the form of precipitation. It is estimated that anthropogenic sources of atmospheric arsenic emission contribute to about 30% of global arsenic release. Concentrations of arsenic in rainfall were found to be very high: approximately $0.5 \mu g/L$ in the areas affected by smelting operation, power plants, soil dust, and burning of coal. In copper smelting areas, the average arsenic concentration was found to be $16 \mu g/L$. Nevertheless, rainfall contributes little arsenic to surface and groundwater bodies. In rain water the most dominant species of arsenic are As_2O_3 near smelters and coal burning areas, AsH_3 near landfills, and reducing soils and arsenate near marine aerosols. The reduced form of arsenic undergoes oxidation in the presence of atmospheric SO_2 and O_3 .

1.6.3.2 River and Lake Water

Arsenic concentration in river water depends upon bedrock lithology, base flow contribution, surface recharge, and industrial or sewage effluents. Increased concentrations of arsenic are found in areas where water flows into surface water from surrounding rocks and increases the pH and alkalinity of resultant water. A high concentration of arsenic was found in the Loa River Basin of northern Chile ranging from 190 to 21,800 μ g/L, which correlated well with high salinity of water [53]. Increased arsenic concentrations of 114 μ g/L was also reported in river waters

in central Argentina owing to increased pH conditions [54]. A higher concentration of arsenic $(1100 \,\mu\text{g/L})$ has been observed in Sugar Creek, South Carolina, downstream of an industrial complex. Arsenic concentration also increases in river water due to mining activities such as mine waste and mill tailing. A high arsenic concentration (200–300 μ g/L) was found in surface water affected by tin and gold mining activities [55].

Due to high evaporation rate, weathering of volcanic rocks, and geothermal activities, high arsenic concentrations are also found in lake water. Arsenic concentrations of 10-20 mg/L in Mono Lake in California, USA, was reported below the pH range of 9.5-10 [56]. Thermal stratification in lake water also affects arsenic concentration and is found to be increased at depths up to 10 m due to increasing ratio of As(III) to As(V) and influx of mining contaminated water. The concentration of As(III) becomes higher in summer due to depleted dissolved oxygen levels and increased biological productivity in the deeper lake water. In lake and river waters, As(V) is generally the dominant species, varying according to seasonal variation, redox conditions, temperature, and biological activity. During summer As(III) can be detected in water due to biological reduction of As(V) in the thermally stratified Mono Lake, under hypersaline and hyperalkaline conditions, As(V) was found in the upper oxic layer and As(III) in the lower reducing layer [31]. The ratio of organic arsenic species also increases in summer due to enhanced methylation reaction catalyzed by microbial activity, and the most common organic forms that exist in water are DMA^V and MMA^V [57].

1.6.3.3 Sea and Estuarine Water

In open sea water the average arsenic concentration varies around $1.5 \mu\mu g/l$; however, in estuaries it is more variable (0.1–4 µg/L) due to various river inputs, salinity, and redox gradient. Increased arsenic concentration was observed in the Krka Estuary in Croat with increased salinity from 0.13 to 1.8 µg/L dominated by As(V) [58]. In sea water arsenate (As(V)) dominates over arsenite (As(III)) and their ratio exists in the range of 10–100 µg/L. At seawater pH (7.5–8.2), As(V) exists as HAsO₄⁻² and H₂AsO₄⁻, while As(III) exists as the neutral species H₃AsO₃. Organic arsenical also exists in seawater as a result of methylation reactions by phytoplankton. In seasonally anoxic estuarine waters, variations in the relative proportions of As(III) and As(V) can be large. In marine and estuarine waters, organic forms are usually less abundant but are often detected and depend upon temperature and variation in aquatic biota [59].

1.6.3.4 Ground Water

Concentrations of arsenic in water bodies depends upon the source of arsenic and local geochemical conditions varying more than four orders of magnitude. Among all the sources of arsenic the greatest range and highest concentrations are found in groundwater due to considerable mineral water interactions and favorable conditions for arsenic mobilization and accumulation. Governments, external agencies, academic institutions, and private research and development centers are making determined efforts to identify increasing concentrations of arsenic worldwide. There are many countries in the world where the arsenic concentration in drinking water was found to be more than the World Health Organization (WHO) permissible value $(10 \mu g/L)$ [60]. Over the past two or three decades, the occurrence of high concentrations of arsenic in drinking water has been recognized as a major public health concern in various parts of the world; apparently, there are areas where the concentration still needs to be recognized. Early identification of arsenic-affected areas can serve as a milestone for the remediation program and to provide safe drinking water. In ground waters, the ratio of As(III)/As(V) varies to a great extent due to variation in redox conditions, sources of water, and environmental conditions. Arsenic mobility in surface and ground water is dependent upon five factors: (1) redox reactions; (2) adsorption and desorption; (3) competitive adsorption (ion exchange); (4) solid phase precipitation and dissolution; and (5) biological activity. These processes are regulated by redox potential, pH, chemical composition of the system, competing anions, aquifer minerals, and reaction kinetics [61].

1.6.4 Arsenic in Dietary Products

1.6.4.1 Plants and Crops

The presence of arsenic in soil adversely affects growth of plants and the effects depend upon the chemical form and availability of As. The toxicity of arsenic varies in the following order as per its oxidation state and form: $AsH_3 > As(III) > As(V) > organic As$. Arsenic availability to plants depends upon soil texture and is found to be high in coarse-textured soils with less colloidal material and little ion exchange capacity, while the concentration is low in fine-textured soils that are high in clay, organic material, iron, calcium, and phosphate. The average arsenic concentration naturally occurring in the soil worldwide is 10 mg/kg. In arsenic-contaminated soil, the uptake of arsenic by plants was found to significantly increase, particularly in vegetables and edible crops [62]. Uptake of arsenic follows a different mechanism for the two oxidation states, As(III) and As(V). As(V) is taken up via a phosphate uptake system in plants due to the similarity of As(V) to phosphate [63]. Therefore, phosphate supplementation has been suggested to reduce arsenic uptake because of competition between phosphate and As(V).

1.6.4.2 Rice and Other Food Items

It is reported that 40% of arsenic in the human body comes from the food chain [64]. Arsenic intake via food varies from 17 to 291 μ g/day in different countries. Seafood accounts for 60–96% of the total dietary intake of arsenic, mostly in the form of arsenobetaine and arsenosugars, relatively non-toxic forms of arsenic [65]. Other food sources are vegetables, mushrooms, grains, milk, chicken, and beef, which account for inorganic arsenic consumption [66]. In non-arsenic endemic regions, the principal sources of inorganic arsenic in the diet are rice and chicken, which results in the accumulation (55–97 ng/g) of methylated arsenic compounds [67]. Herbal medicines and seaweed are also important sources of arsenic, e.g., Hijiki seaweed has high levels of inorganic arsenic and organic arsenic [68].

Rice is the major crop in India and covers about 23% of the country's total irrigated land for its production. Nearly 42% of the total food grain production is rice in India and West Bengal is one of the major rice producing states covering an area of 5,900,000 ha. In West Bengal, a huge amount of the population is dependent upon water for irrigation purposes but it is found to be contaminated with arsenic. For the cultivation of rice during lean periods an enormous amount of ground water is used, which leads to oxidative decomposition of pyrite to form Fe^{+2}

Rice Variety	Origin	Total As (μg/g)
Basmati white	India	0.05
Long grain	America	0.26
Long grain	Bangladesh	0.13
Long grain	Europe	0.18

Table 1–6Concentration of Arsenic in DifferentRice Varieties

and Fe⁺³ sulfate and sulfuric acid, responsible for arsenic mobilization. In South East Asian countries, rice is grown mostly under waterlogged lowland conditions in which the physical and chemical condition of the soil affects the mobilization of arsenic and its uptake by rice [69].

In West Bengal, villagers living in arsenic-contaminated areas largely depend upon rice for their calorific intake [63]. The concentration of arsenic varies according to rice species and is summarized in Table 1–6. Due to the high arsenic levels found in paddy rice, such rice has been considered an important source of arsenic intake and risk is increased if it is cooked in arsenic-polluted water due to chelation of arsenic to the rice starch/bran [70]. According to WHO, the permissible limit of arsenic in rice and other food items has been reported as between 1.0–2.0 mg/kg and 2 mg/kg [71].

1.6.4.3 Tobacco

Cigarette smoking is a well-established risk factor for carcinogenesis. The chemical agents associated with cigarette smoke are classified as group I carcinogens and include benzene, polycyclic aromatic hydrocarbons (PAHs), cadmium, arsenic, nickel, chromium, 2-naphthyl-amine, vinyl chloride, 4-aminobiphenyl, and beryllium [72]. In the United States, arsenic levels in tobacco rose from 12 mg/g in the 1930s to over 50 mg/g in the 1950s. By the 1970s, in response to policy efforts, arsenic levels reduced to 1 mg/g of tobacco, which further declined in subsequent years [73]. Arsenic exposure due to tobacco smoking has been prevalent in Asian countries due to the smoking of handmade cigarettes, especially bidis [74].

1.7 Arsenic in Hydrothermal and Geothermal Fluids

1.7.1 Arsenic Occurrence in Hydrothermal Fluids

The concentration of arsenic in the world's oceans is usually high and governed by various factors such as riverine input, weathering and sedimentation on the sea floor, water exchange between atmosphere and sea, inputs of volcanic gases, and hydrothermal fluids. The dwelling time of arsenic in the oceans is assumed to be between 32,000 and 63,000 years with an average concentration of $1.7 \,\mu$ g/L.

In the past 25 years, active hydrothermal systems, polymetallic mineral deposits, and robust biological communities have been discovered including sediment-free and sediment-covered ocean ridges, the East Pacific rise, mid-Atlantic ridges, rifted continental margins,

Major Hydrothermal Locations	
• Montserrat	
• Tutum Bay	
• Milos	
 Bahía Concepción 	
 Champagne Hot Springs 	
• Menez Gwen	
• Lucky Strike	
• TAG	
Snake Pit	
Logatchev	
• Guaymas Basin	
• Lau Basin	
• Manus Basin	

FIGURE 1–9 Major hydrothermal locations distributed worldwide contributing significantly to the environmental arsenic burden.

Property	Shallow Water	Deep Water
Depth	Less than 200 m	More than 200 m
Photosynthesis rate	High due to benthos and phytoplanktons	None takes place
Water source	Meteoric and sea water	Sea water
Temperature	Up to 100°C	Above 400°C

Table 1–7 Comparison between Shallow and Deep Water Systems

volcanic arcs, back-arc basins, and seamounts. Major hydrothermal locations worldwide are listed in Figure 1–9. Chemical analyses of hydrothermal fluids and suspended particles from submarine systems reveal that arsenic and other metals are effectively mobilized from the oceanic crust and sedimentary, transported to the sea floor, and concentrated in surface mineral deposits. Hydrothermal fluids emerging at the sea floor are known to contain considerable amounts of dissolved transition metals and toxic elements due to water/rock interaction at elevated temperatures and potentially magmatic degassing. Arsenic concentrations in hydrothermal fluids are 2–500 times higher as compared to seawater concentrations.

The chemistry of hydrothermal fluids is chiefly governed by physicochemical variables, phase separation, depth and temperature, rock lithology, water/rock interaction, and heat gradient [75]. Cold and dense water percolates downward and is heated up to 400°C, and hot water rises upwards due to resilience. Hydrothermal systems can be divided into shallow and deep water based on their depth, and systems at a depth of 200 m are considered deep water systems [76]. Comparison of shallow and deep water systems is summarized in Table 1–7.

In shallow water systems, photosynthetic organisms can absorb arsenate from the surrounding sea water owing to its similarity to phosphate [77]. Arsenate is converted to arsenite

and then further to methylated species, which are again converted and detoxified to different arsenosugars. These arsenosugars are further converted to the less toxic arsenobetaine in higher marine organisms.

1.7.2 Arsenic Concentration in Shallow and Deep Hydrothermal Systems

Some shallow and deep water hydrothermal systems are listed in Figure 1–7. Montserrat Island in the Caribbean, located at the northern end of the inner arc consisting mainly of andesitic rocks, showed a large increase in shallow water hydrothermal discharge near its western coast after a volcanic eruption in July 1995 [78]. The springs showed alarmingly high concentrations of arsenic (3600 µg/L) after the eruption, followed by a steady decline.

Another shallow-water hot springs in Tutum Bay located 150 m offshore along the southwest part of Ambitle Island (Papua New Guinea), shows temperatures up to 98°C in 5–10 m of water. The sediments in this region consist mainly of feldspar, hornblende, pyroxene, magnetite, and weathering product coated with hydrous ferric oxide [79]. Arsenic concentrations in the water range from 750 to $1050 \mu g/L$, while in ferric oxide they reach up to 50,000 mg/kg.

Milos, another hydrothermal site situated in the central Hellenic volcanic arc, showed arsenic concentrations as high as $2900 \,\mu\text{g/L}$ in Paleochori Bay, whereas concentrations were even higher (up to $5850 \,\mu\text{g/L}$) in Spathi Bay, known to be the highest in a hydrothermal system. At Milos, hydrothermal fluids have been divided into two groups, one with high chlorine containing less arsenic and the other with low chlorine but supplemented with more arsenic [80].

Bahia Concepcion located in the Gulf of California, covered by a marine and continental sedimentary layer, contains hydrated ferric oxide enriched with high concentrations of arsenic up to 50,000 mg/kg [81]. Champagne hot springs located near the island of Dominica in the Caribbean consist mainly of medium-K calk-alkaline andesites and hydrothermal hydrated ferric oxide, augmented with arsenic concentrations up to 1880 mg/kg [82].

Various shallow water systems differ in terms of temperature, pH, H_2S content, redox state, fluid pressure, iron concentration, and gas fugacity, which seem to be responsible for leaching of arsenic from host rock and its transportation in hydrothermal fluid. Aquifers with high temperature, low pH, and high H_2S content show high concentrations of arsenic, perhaps due to higher leaching capacity and phase separation. Phase separation generates a low chlorine vapor phase in which arsenic partitioning is high pertaining to dominance of uncharged arsenic species and SH⁻ complexation [83].

The Mid-Atlantic Ridge (MAR) is a deep hydrothermal region occupied by two hydrothermal systems, Menez Gwen and Lucky Strike, which are both similar in rock composition and fluid geochemistry. The systems were found to be poor in chlorine and iron, while rich in gaseous components [84]. Menez Gwen hydrothermal discharge, located at a depth of 840–865 m with a temperature around 285°C, had an arsenic concentration found to be 18.5 μ g/L, while Lucky Strike at a temperature range of 170–324°C had an arsenic concentration of 4.3–24 μ g/L [85]. Another hydrothermal field in the MAR region is TAG, discovered in 1972, and, further towards the south, the Snake Pit hydrothermal system was identified. Both systems were found to be chlorine rich with TAG showing low arsenic $(0.8 \,\mu\text{g/L})$ and Snake Pit in the range of 3.2 to $14.2 \,\mu\text{g/L}$ [86].

East Pacific Rise spreads 9000 km in the Pacific Ocean with various hydrothermal regions studied in different sections, primarily Guaymas Basin located in the central Gulf of California characterized by 2000 m water depth. The basin is set apart with a temperature of up to 315° C, highly reducing conditions, and arsenic concentration in the range of $21.1-80.5 \,\mu$ g/L [87]. The area possesses a high sedimentation rate of organic and carbonate-rich material, which undergoes pyrolysis at high temperature to release NH⁴₄ ions in the fluid [88].

Deep-water hydrothermal springs also occur in back-arc basins with a high temperature gradient and a different chemical composition from those of mid-ocean ridges [89]. The fluid is characterized by very low pH (\approx 2), high amounts of dissolved metals (Mn, Zn, Cu, and Pb), and the presence of rocks like dacites and andesites [90]. The most common back-arc basin is Lau Basin located between the remnant Lau Ridge and the active Tofua volcanic arc with 1700–2000 m water depth [91]. At low pH and with high chlorine, arsenic concentration lies in the range of 450–825 µg/L.

Another hydrothermal system is Manus Basin located at the New Britain Trench, active in 1650 to 2500 m water depth [92]. Hydrothermal systems from back-arc basins show higher values than those from mid-ocean ridges. Higher arsenic concentration could be due to transportation of high amounts of dissolved iron leading to precipitation and incorporation of arsenic into iron sulfides.

1.7.3 Arsenic Occurrence in Geothermal Systems

High temperature geothermal systems occur throughout the world generally in one of three tectonic plates. The edge of the Pacific plate is defined by geothermal fields in New Zealand, Papua New Guinea, the Philippines, Indonesia, Japan, Kamchatka, Alaska, western USA, Mexico, Central America, and Chile. Other geothermal systems are Earth's "hot spots," such as in Hawaii, Yellowstone, and the Azores. The presence of arsenic in geothermal fluids has been known since the mid-19th century. Geothermal fluid arsenic concentrations are usually three orders of magnitude higher than those in uncontaminated surface and ground waters. Arsenic concentrations in the thermal fields of Yellowstone National Park range from 0.1 to 10 mg/kg, a range that is observed in most active geothermal fields. On the other hand, extremely high arsenic concentrations (more than 20 mg/kg) were found in geothermal well fluids. Arsenic is a ubiquitous component of geothermal systems and occurs together with other elements such as mercury, antimony, selenium, thallium, boron, lithium, and fluoride [93]. In geothermal areas, arsenic contamination occurs via two main processes: (1) natural contamination where the geothermal waters reach the surface as natural springs and then mix with surface water flows or shallow groundwater bodies used by humans for both irrigation and drinking water supply; and (2) human exploitation of geothermal waters as an energy resource causing mobilization of arsenic and other heavy metals contained in the geothermal waters to reach the surface and then contaminate surface and shallow groundwater bodies.

Arsenic concentrations in natural geothermal systems exceed permissible limits, and in some parts such as West Yellowstone, the concentration reached as high as $2000 \mu g/L$. The population in these areas shows symptoms of chronic arsenic poisoning such as skin lesions and malignant growth [94]. River systems receive geothermal fluids that accumulate arsenic in aquatic plants, a notable source of exposure.

Geothermally active areas came into existence due to the emergence of hot water and steam at the Earth's surface and development of high geothermal gradient. The elevation in temperature is related to volcanic or magmatic activity, metamorphism, faulting, and radioactivity. Deeply circulating hot fluids are of low density and rise through the host rock, while cold water from the water table moves downward to recharge ground water. The chemical composition of geothermal fluid is regulated by fluid temperature and host rock composition. At high temperature, water behaves as an excellent solvent and can dissolve various ionic crystalline minerals forming new hydrated materials. At high temperature due to enhanced water/ rock interaction, various elements such as Li, Rb, Cs, B, and Cl leach into solute. The concentration of dissolved gases such as CO₂ and H₂S are also a function of mineral solubility, especially carbonate and sulfide minerals. When geothermal fluid rises through the crust, pressure decreases and fluid separates into two phases, steam and water. Geothermally active areas can be divided into two groups: (1) hot water springs with neutral pH, rich in chloride and silica; and (2) steam vents of acidic pH, which are sulfate rich and susceptible to microbial oxidation of sulfuric acid. Geysers and carbonate-rich springs are also hot water discharges, but include steam heating and steam-phase mixing, respectively.

The development of geothermal fields for power generation increases the rate and volume of geothermal water reaching the surface. Geothermal sites used for power generation require substantial amounts of water to be extracted. During the separation of steam and water in geothermal power generation, arsenic is retained in the waste bore water and creates a number of problems in its disposal [95]. It becomes essential to re-instill thermal waste to its reservoir in order to prevent contamination of cold water. The geothermal springs are also used as swimming pools and hot spas, both for local residents and for tourists.

1.7.4 Arsenic Speciation and Deposition in Geothermal Fluids

According to water chemistry, geothermal waters can be classified into two main categories: (1) neutral $Cl-SO_4^{-2}$ water; and (2) acid- SO_4^{-2} water. The former system has a high concentration of chlorine (263–1337 mg/L) with arsenic concentration ranges from 1 to 890 µg/L while the latter has a low concentration of Cl (1.19 to 6.9 mg/L) and a low concentration of arsenic (0.8–660 µg/L) [96].

The behavior of arsenic in geothermal systems and surface water environments can be interpreted using thermodynamic data. In most hydrothermal fluids, arsenic is expected to be transported as arsenous acid (H_3AsO_3), which forms by the dissolution of orpiment at acidic to neutral pH and a wide range of temperatures [97]. Orpiment solubility increases with temperature, however, independent of pH under acidic or acid neutral conditions and then increases with pH.

In sulfide-rich hydrothermal fluids, arsenic exists as thioarsenite complexes, a group of covalently bonded arsenic-sulfide complexes that may or may not include oxygen. Orpiment solubility increases with pH and sulfide concentration but is less affected by temperature changes below 200°C. When rising geothermal fluid is exposed to atmospheric oxygen or mixed with another oxidizing fluid, such as a shallow ground water, arsenous acid (As^{+3}) oxidizes to arsenate ($H_2AsO_4^-$ or $HAsO_4^{-2}$). Oxidation to arsenate occurs very rapidly in hot spring overflows and receiving streams and rivers. Various geothermal waters such as found in Yellowstone National Park, on the island of Dominica, in the Valles Caldera (New Mexico), and Massive Central (France) are likely to contain mostly As(V), due to strong oxidizing conditions. Hot springs formed from reservoir fluids contain mainly As(III), whereas that with acid sulfate and bicarbonate is enriched with As(V).

1.7.5 Biogeochemical Fate of Arsenic in Hydro- and Geothermal Systems

The biogeochemical fate of arsenic at submarine hydrothermal sites is less well known. Average arsenic concentrations in soft tissues of mussels (40 ppm) from the Mid-Atlantic Ridge are higher than those of mussels (2–20 ppm) from coastal environments. In contrast, shrimp recovered from the Mid-Atlantic Ridge have a lower arsenic content (13 ppm) compared to shrimp (15 ppm) from surface waters and coastal zones [98]. The high levels of arsenic (up to 70 ppm) in gill tissues and digestive tracts of these organisms were found mostly in the form of arsenobetaine and arsenosugars, suggesting assimilation of arsenic through the activity of symbiotic bacteria and by filtration of Fe oxyhydroxide particles. The presence of arsenic-bearing sulfosalt minerals proustite (Ag₃AsS₃) and pearceite (Ag₁₆As₂S₁₁) in filamentous bacteria from northern Gorda Ridge located off the coast of Oregon and northern California is a further indication that microorganisms play a key role in the chemistry of arsenic in submarine hydrothermal environments. Thermophilic bacteria and cyanobacteria in hot spring bio-films also facilitate the formation of silica, silicates, carbonates, and oxide minerals in sinter deposits and indicate enrichment of arsenic along with other elements such as Sb, P, and Zn [99].

1.8 Arsenic Release from Mining and Mineral Processing

1.8.1 Oxidation of Arsenic Sulfide and Industrial Ore Leaching

High concentrations of arsenic and other heavy metals such as Cd, Fe, Pb, Ni, and Zn are commonly present in acid mine drainage (AMD) effluents chiefly due to oxidation of arsenopyrite. AMD is polluted water, and contains high levels of iron, aluminum, and sulfuric acid. Arsenic is one of the priority pollutants associated with AMD, especially from gold mining operations [100]. Mining activities accidentally accelerate physical weathering of pyrite by grinding up the ore and then oxidation of pyrite takes place. The presence of pyrite in sulfide-rich coals also plays a major role in the generation of acid rain and AMD.

$$FeAsS + 13Fe^{+3} + 8H_2O \leftrightarrow 14Fe^{+2} + SO_4^{-2} + 13H^+ + H_3AsO_4$$
 (1.7)

 Fe^{2+} and H^+ are released after oxidation of pyrite and move to the surface and ground water, where Fe^{2+} is oxidized to Fe^{3+} in the presence of dissolved oxygen in water. Fe^{3+} can further hydrolyze in water and precipitate as $Fe(OH)_3$, which releases more H^+ into the water. In addition, iron-(oxy)hydroxysulfates often precipitate in iron-rich, acid-sulfate waters, under acidic conditions [101]. Under such conditions, soluble Fe^{3+} reacts readily with more pyrite and oxidizes it due to the action of acidophilic chemolithotroph bacteria, particularly *Acidithiobacillus ferrooxidans* [102].

The dissolution of the most common arsenic sulfide minerals orpiment and arsenopyrite follows the same mechanism and largely depends upon pH. Orpiment dissolution increases linearly from acidic to alkaline conditions [103], whereas arsenopyrite dissolution is minimal at near neutral pH and increases towards acidic and alkaline conditions [104]. Orpiment dissolution is controlled by weak van der Waal forces. At alkaline pH, due to absorption of hydroxide ion (OH⁻) at the mineral surface, van der Waals forces becomes weak and two species can be formed, S-As-S and S-As-S-OH [105]. These species release arsenite and sulfide followed by recombination to thioarsenites and oxidation to thioarsenates in solution. In sulfide-deficient solutions, arsenite dominates at pH 2–8; however, in anoxic, sulfide-rich solutions, thioarsenite ($AsO_xS_x^{-3}$) has been proposed to pre dominate [106]. The formation of mono- (AsO_3S^{-3}), di- ($AsO_2S_2^{-3}$), and trithioarsenate ($AsOS_3^{-3}$) was proposed upon dissolution of orpiment at neutral pH [107].

Dissolution of arsenopyrite releases iron along with arsenic and sulfur, which also influence arsenic mobility and formation of thioarsenic species. Initially, Fe^{+2} , S^{-2} , and As^{-} species are released from the mineral surface and undergo oxidation at alkaline pH to form iron-arsenic oxide or hydroxide surface coating to prevent further release of arsenic [108,109]. The presence of iron and rapid formation of iron sulfides leave no or little free sulfide for thioarsenate formation.

Under acidic conditions, neither mineral forms thioarsenate because at low pH thioarsenic species transform quantitatively to arsenite or precipitate as amorphous arsenic sulfide minerals [110]. At highly alkaline conditions, thioarsenite transforms to arsenite due to competitive ligand exchange of SH⁻ vs. OH⁻ groups. Monothioarsenate is stable towards reduction; however, di- and trithioarsenate are reduced to arsenite (As(III)). It has been postulated that thioarsenates once released by mineral dissolution might have higher mobility compared to arsenites and arsenates and may increase arsenic burden in natural or mining-impacted environments.

Release of arsenic from arsenopyrite is also enhanced in the presence of hydrogen peroxide (H_2O_2), a common constituent of rain water [111]. Hydrogen peroxide is a stronger oxidizing agent compared to molecular oxygen. It can also generate more potential oxidizing agent hydroxyl radical via Fenton's reaction. Hydrogen peroxide can decompose readily and generate dissolved oxygen as per the following chemical equation:

$$2H_2O_2 + O_2 \to 2H_2O + 2O_2 \tag{1.8}$$

Release of arsenic from the mineral surfaces in the presence of oxygen can be illustrated as:

$$\text{FeAsS} + \frac{11}{4}\text{O}_2 + \frac{3}{2}\text{H}_2\text{O} \rightarrow 2\text{Fe}^{+2} + 2\text{H}_3\text{AsO}_3 + 2\text{SO}_4^{-2}$$
(1.9)

 $\rm H_3AsO_3$ can be further oxidized by molecular oxygen to produce $\rm As^{5+}$:

$$H_3AsO_3 + \frac{1}{2}O_2 \to H_2AsO_4^- + H^+$$
 (1.10)

Oxidative decomposition of arsenopyrite releases Fe^{2+} , and thus it may react with H_2O_2 to generate the hydroxyl radical, which can convert As^{3+} into As^{5+} :

$$Fe^{2+} + H_2O_2 \rightarrow Fe^{+3} + OH + OH^-$$
 (1.11)

$$As^{+3} + 2 OH \rightarrow As^{+5} + 2OH^{-}$$
 (1.12)

Arsenic concentrations were noted from 180 samples of mine drainage from the USA in the range of 1 to $340,000 \,\mu$ g/L, much higher when compared to arsenic concentrations in natural water bodies, 0.5–5000 μ g/L [112].

1.8.2 Chemistry of Arsenic Within Mining Wastes

Arsenate generally predominates under oxidizing conditions in mining waste water, while arsenite predominates when conditions become sufficiently reducing. It has been suggested by thermodynamic calculations and experimental results that at high redox levels and pH > 10, arsenate predominates, while under reduced conditions and pH < 8, arsenic is the dominating species. Arsenite oxidation is a slow process in distilled and demineralized water, whereas at the mineral surface oxidation occurs rapidly and increases the environmental burden of arsenic. Arsenite oxidation is also enhanced in the presence of manganese oxides in sediments, aquifer materials, river and lake water, and by biogenic manganese oxide. It has also been postulated that the oxidized species of couples with positive redox potential can oxidize the reduced species of couples with negative redox potential at specified pH. It has been suggested that manganese hydroxides are responsible for arsenite oxidation under near-neutral to alkaline conditions (pH 6–12), while iron hydroxide behaves as an oxidizing agent under acidic conditions in general.

Redox reactions also regulate stability of most arsenic-bearing sulfide and hydroxide minerals, which are sources and sinks of arsenic species. Sulfide minerals are generally stable under reducing conditions but oxidize under oxidizing conditions to release arsenic into the biosphere. These minerals are generally stable in the neutral to high pH range but are dissolved in acidic or highly alkaline solutions due to their amphoteric nature. Under highly reduced conditions, arsenic accumulates at redox boundaries in sediments and associates with sulfide [113].

1.8.3 Arsenic Release by Artisanal Mining

Artisanal mining is becoming progressively more common in many parts of the world, with more than 30 million active artisanal miners in more than 55 countries. Artisanal mining activities are strenuous in rural areas due to lack of proper employment and the individuals undertaking informal mining generally lack education, training, management skills, and essential equipment for safe mining practices. Arsenic, a constituent of gold ore, is a very common occupational pollutant resulting from pervasive environmental contamination. These mining sites create great menace as most of them are run without due consideration given to environmental, occupational, or community exposures. Moreover, environmental monitoring and waste management in artisanal gold mining areas are not appropriate. People living in proximity to artisanal mining areas are vulnerable to arsenic exposure, especially pregnant women, their fetuses, and young children. Women in these gold mining areas because of poor nutritional status and cultural acceptance of the practice [114]. Consumption of 50g of sikor (a molded soil sold in the local market, from an arsenic-contaminated area) per day is known to be equivalent to $\approx 370 \,\mu$ g of arsenic [115].

1.9 Global Occurrence of Arsenic in Ground Water

1.9.1 Arsenic Release and Mobility in Natural Water

Arsenic release from naturally-occurring sources, principally reductive dissolution of iron oxides and hydroxides, desorption of arsenic from mineral surfaces, and oxidation of arsenicbearing sulfide minerals, is the dominant cause of elevated arsenic concentrations in groundwater [16]. These processes usually occur simultaneously and can be abiotic and microbially mediated. Redox chemistry of iron greatly influences release and mobilization of arsenic in ground water. Under reductive conditions iron oxide is dissolved and releases adsorbed arsenic to subsurface waters [116]. The reduction reactions accelerate in anaerobic surroundings via consumption of oxygen and release of more CO_2 through the decomposition of organic matter in water bodies. Major reduction reactions are as shown below:

$$NO_3^- \to NO_2 + N_2O + N_2$$
 (1.13)

$$MnO_2 \rightarrow Mn^{2+}$$
 (1.14)

$$\operatorname{Fe}_2\operatorname{O}_3 \to \operatorname{Fe}^{2+}$$
 (1.15)

$$\mathrm{SO}_4^{-2} \to \mathrm{S}^{-2} \tag{1.16}$$

$$\mathrm{As}^{+5} \to \mathrm{As}^{+3} \tag{1.17}$$

Release and mobilization of arsenic in water is also influenced by oxygen concentration and growth of microorganisms. The US Department of Natural Resources (DNR) issued guidelines for maintenance of domestic wells in arsenic-sensitive areas suggesting disinfection with a low dose of chlorine solution as high doses of chlorine can increase arsenic concentration by maintaining oxidizing conditions. The release of arsenic from soils and sediments is associated with iron oxide dissolution. A speedy increase in pore water arsenic concentration (about $30 \mu g/L$) has been observed with the increase in dissolved iron [117]. Gotkowitz et al. [118] demonstrated effects of strongly oxidizing conditions on mineral transformations and arsenic release, and suggested that microbially induced reduction of arsenic-bearing iron oxides releases significant quantities of arsenic in to water.

The majority of the population in India and other developing countries depends upon ground water as a source of drinking water, which has been found to contain elevated levels of arsenic. In developed countries such as the USA however, treated water enters through urban drinking water distribution systems, which contain arsenic below its permissible limit, i.e., $10 \mu g/L$ [119]. Arsenic has been well reported in pipe scales and corrosion products [120] suggesting its pervasive prevalence within drinking water. Arsenic has been found in various types of pipe scales including iron corrosion products and lead scales reaching maximum levels up to 14 mg/g pipe scale [121].

1.9.2 Natural and Anthropogenic Occurrence Globally

Analysis of groundwater samples from various parts of the world, including Argentina, Australia, Bangladesh, Cambodia, Chile, China, Hungary, West Bengal (India), Inner Mongolia, Mexico, Nepal, New Zealand, the Philippines, Taiwan, Thailand, Vietnam, and the United States, has identified the presence of arsenic levels above $10 \mu g/L$, the WHO guideline value and new US maximum contaminant level. The situation in Bangladesh, West Bengal, and Vietnam is extremely bad with 40 million people estimated to be exposed to arsenic concentrations above $50 \mu g/L$. Dermal absorption during bathing and hand-washing is also an important exposure route. Most health studies investigate the effects of drinking arsenic-contaminated water. However, there is a scarcity of data regarding the differential health effects of arsenite and arsenate. Evidently, cultured human keratinocytes exhibit greater permeability to arsenite as low as $10 \mu g/L$ resulting in cytotoxicity and inhibition of DNA and protein synthesis [122]. *In vivo* studies with rhesus monkeys revealed dermal absorption at arsenic concentrations as low as $14 \mu g/L$ [123].

1.9.2.1 India, Bangladesh, and Nepal

In India, groundwater arsenic contamination has been identified in many states, especially West Bengal, Jharkhand, Bihar, Uttar Pradesh, Assam, Manipur, and Rajnandgaon village in Chhattisgarh, with concentrations above the permissible limit of $50 \mu g/L$. All the arsenic-affected river plains have their river routes originating from the Himalayan region. The arsenic problem in ground water from the alluvial and deltaic aquifers of Bangladesh and West Bengal represents the most serious calamity identified globally. Groundwater arsenic contamination

has exceeded 2000 µg/L in some areas of India and Bangladesh [124]. Arsenic occurs widely in ground water of the Bengal Basin in West Bengal and Bangladesh beneath the flood plains of the rivers Ganges, Brahmaputra, and Meghna [125]. Among Asian countries, the large population in West Bengal consumes arsenic-contaminated drinking water and many suffer from arsenicosis [126]. Ninety-five percent of the population in this region relies on ground water for drinking and domestic use, and for irrigation purposes. It has been estimated that out of a total population of 125 million people, 35 million people were exposed to arsenic levels in water above $50 \mu g/L$ and 57 million people to that in water above $10 \mu g/L$. The biggest arsenic calamity in the world has been reported from the seven districts of West Bengal, namely, Malda, Murshidabad, Burdwan, Nadia, Hooghly, and 24-Parganas (North and South), covering an area of 37,493 km² and having a population of about 34 million. It has been estimated that 27% of tubewells used for domestic water supply in this region contain arsenic above the Bangladesh national permissible limit of $50 \mu g/L$ and 47% exceed the WHO recommended limit of $10 \mu g/L$. It has been reported that the 8-9 million people of West Bengal are affected by arsenic toxicity due to use of arsenic-contaminated ground water for drinking purposes [42]. The average arsenic concentration in the drinking water is about $200 \,\mu g/L$ reaching as high as 3700 µg/L. Water samples from shallow and deep tubewells in Bangladesh were found to contain arsenic concentrations up to a maximum of $1660 \,\mu g/L$ [127].

A positive correlation has been found from the reports of Taiwan and Bangladesh between high arsenic exposure and several dermatologic and vascular diseases such as blackfoot disease, hypertension, and cerebrovascular disease. Studies have also suggested an association between arsenic in drinking water and diabetes, infant mortality, and reduction in birth weight. Arsenic groundwater contamination and its health effects in the Rajnandgaon district of Chhattisgarh have been reported recently [128]. In this area, wells located in the granitic terrain with pegmatitic intrusions were found to have arsenic concentrations above WHO [129] and BIS (Bureau of Indian Standards) standards, with the highest being found in a well with more than $250 \,\mu$ g/L of arsenic. Buragohain and Sarma [130] presented a comprehensive analytical and spatial analysis of arsenic distribution in ground water of five blocks of Dhemaji district, Assam. They reported that the majority of samples were found to contain arsenic concentrations exceeding WHO permissible limits while a few were approaching them. The most probable source of arsenic in ground water in this area was assumed to be heavy deposition of sediments due to surface erosion from surrounding hills.

The Department of Water Supply and Sanitation of Nepal in collaboration with the WHO surveyed three districts of eastern Terai for the first time in 1999 and found that arsenic concentrations exceeded the 10 ppb guideline recommended by WHO. Over 90% of the Terai population draws ground water from tubewells for drinking, household use, and irrigation. Of the 18,635 tubewells tested, 23.7% have arsenic contents above the WHO limits of 10 ppb, and 7.4% were above the Nepal interim standard of 50 ppb.

1.9.2.2 Latin America

The native population of Latin America may have suffered arsenicism 7000 years ago caused by high arsenic exposure due to consumption of arsenic-contaminated food and water [131].

At the beginning of the 21st century, high arsenic concentrations were found in water bodies from Latin American countries and continued to increase sharply. As of now, 15 of the 20 Latin American countries, such as El Salvador, Nicaragua, Brazil, Bolivia, Cuba, Ecuador, Honduras, Uruguay, Colombia, and Guatemala [132], show high arsenic concentrations in surface and ground water. Castro de Esparza [133] estimated that in Latin America 4.5 million people are exposed to drinking water with more than $50 \,\mu\text{g/L}$ of arsenic and 14 million people are exposed to $10 \,\mu\text{g/L}$ arsenic concentrations. Studies from Latin American countries revealed that smoking, gender, long latency period, exposure at early life, and incomplete metabolism of arsenic significantly increase the risk of cancer [134].

1.9.2.3 Argentina, Mexico, and Chile

From the beginning of the 19th century to the 1960s, a number of reports suggested arsenic contamination of water from Argentina, Chile, and Mexico. Argentina was the first country in Latin America where ground water was found contaminated with appreciable amounts of arsenic. The first cases of human arsenic poisoning were described from the locality of Bell Ville in the Chaco-Pampean plain in Argentina, related to arsenic-contaminated drinking water. During the years 1913–1917 initial adverse effects of arsenic on human health were reported [135] and further it was found that the effects were predominantly from the northern part of the Chaco-Pampean plain. This region is reported to be the largest identified area in the world with high arsenic concentrations in ground water [136]. The predominating species was found to be As(V) due to moderately reducing to oxidizing conditions.

Mexico is one of the world's major arsenic producers, generating 1946 t in the year 2002. In the Lagunera area, wells were found to contain arsenic concentrations up to 0.624 mg/L. The Lagunera region, located at the northern part of the country, was identified in 1958 as an arsenic-contaminated zone [137]. Thirteen out of 31 states in Mexico, namely Durango, Coahuila, Zacatecas, Morelos, Aguascalientes, Chihuahua, Sonora, Puebla, Nuevo Leon, Jalisco, Oaxaca, Guanajuato, and San Luis Potosi, receive arsenic in supplied drinking water. From these locations, approximately 450,000 people are reported to be exposed to arsenic concentrations of more than $50 \,\mu\text{g/L}$ [138]. About 400,000 people were determined as being exposed to arsenic concentrations greater than $50 \,\mu\text{g/L}$ in Comarca Lagunera, one of the important cattle-raising areas of Mexico. Sources of groundwater arsenic in these regions were assumed to be due to oxidation of arsenopyrite and dissolution of scorodite. Zimapan mining district was also reported to contain high arsenic content in soil due to smelting and tailings [139]. Arsenic concentrations were also determined in lake water and sediments of Chihuahua state with ranges from 9.11 to 16.62 mg/kg in sediments, while 6 to $170 \,\mu\text{g/L}$ were found in water samples [140].

The northern Chile region is characterized by mountainous landscape and a semi-arid climate with highly variable precipitations. The Elqui watershed covers an area of about 9800 km², and its main river (the Elqui) constitutes one of the few water resources of this hilly, semi-arid region of Chile. Epidemiological investigations from arsenic-endemic areas of Chile and Argentina have linked high levels of arsenic exposure with elevated risk of bladder and lung cancer mortality [141]. In northern Chile arsenic in river water is largely released from volcanic rocks and weathering of sulfide ore deposits at the Andean volcanic chain and

mobilized by snow melt and rain to the rivers and springs. Geothermal springs are also potential sources of arsenic in this region. Cases of first skin lesions were detected at the beginning of the 1960s, especially in children. During that period approximately 500,000 people were exposed to arsenic-contaminated drinking water. The population of Esquina consumes drinking water from small waterfalls (As $-12.2-74.0 \mu g/L$), whereas the population of Illapata consumes drinking water from the Camarones river (As $-48.7 \mu g/L$) and waterfalls (As $-1252 \mu g/L$). Further, arsenic concentrations in water, urine, hair, and nails were studied in people from the same regions; arsenic was found to be contained therein, in water localities [142]. In this study, the arsenic was found in drinking water (50 and 1090 $\mu g/L$), urine (0.20 and 1.10 $\mu g/L$), hair (0.30 and 3.8 mg/kg), and nails (3.2 and 11.2 mg/kg), respectively. Another study in northern Chile is from the region of the Elqui valley, a drain area with important hydrothermal alterations and epithermal ore deposits, including arsenic-rich copper channels. Arsenic in the sediments of the valley ranges from 55 to 485 mg/kg in river sediments and from 1.19 to 2.344 mg/kg in Holocene lake sediments, generally associated with iron oxides and iron oxyhydroxides [143].

1.9.2.4 Peru

This region contains arsenic in high concentrations predominantly of geogenic origin, which are released in the Andean region by natural weathering and mining activities. In 2000, it was estimated that around 250,000 people were drinking water with arsenic concentrations exceeding $50 \mu g/L$ [133]. The rural population in Peru that depends upon untreated water for drinking and irrigation is mostly exposed to high arsenic. The Rimac river basin situated east of Lima is also contaminated with arsenic predominantly from mining activities in the middle and upper parts of the basin and from natural leaching of Andean volcanic rocks and sulfidic ore deposits. The presence of arsenic adversely affects the quality of water, soils, and vegetables in the lower part of the basin, and is a severe threat to human health. In 1994, the first investigation was performed in 53 drinking water samples from rivers, wells, and springs and it was found that 84.9% of the samples exceeded the limit recommended by the WHO.

1.9.2.5 California

The presence of high concentrations of arsenic in ground waters of the Tulare Basin, California, USA, has been well documented. In this region, shallow groundwater quality is generally poor with high salinity, pesticide residues, and toxic elements such as arsenic, boron, molybdenum, selenium, and uranium present. In 1989, the US Geological Survey investigated the areal distribution of dissolved arsenic in shallow ground water in the southern San Joaquin Valley from areas near Hanford to Lost Hills and extending over to areas near Bakersfield, and reported arsenic concentrations in the range $1-2600 \,\mu g/L$ [144]. Elevated arsenic concentrations exceeding $50 \,\mu g/L$ were also reported to the south of the Tulare Lake Bed in the Kern Fan Element (KFE) west of Bakersfield.

1.9.2.6 Central America

High levels of arsenic were detected in surface and ground water, from several locations in Central America, used especially for drinking and irrigation purposes. The origin of arsenic in these regions is natural due to the tectonic setting of the Pacific coast, leaching from volcanic rocks and hydrothermal settings, geothermal fluids, and mining and metallurgical processing [145]. In June 2007, there were several announcements about the presence of arsenic in ground water in different areas of the municipality of Mixco in the department of Guatemala with average arsenic concentrations of $15 \mu g/L$ [146]. Goff et al. [147] studied the geochemistry of springs in seven geothermal sites in Honduras and detected arsenic in four of them, namely Azacualpa, Pavana, Platanares, and Vanes Caldera, with arsenic concentrations of 70, 110, 1260, and $1160 \,\mu g/L$, respectively. High concentrations of arsenic have also been detected in surface and ground waters in El Salvador. In this area, water samples from three lakes were studied for arsenic contamination and found to contain arsenic in variable amounts: Coatepeque Lake 90 to 3090 µg/L, Ilopango Lake 290 to 780 µg/L [148], and Olomega Lake 4210 µg/L, the highest arsenic concentration determined in surface waters in El Salvador. In Nicaragua, the contamination of ground water with arsenic was detected in May 1996 in water of an artesian well from the community of El Zapote in the Sebaco valley, containing arsenic at 1320 µg/L, which was further confirmed by different agencies and institutions [149–151]. The largest water quality study (1488 wells) was carried out by UNICEF and the Pan American Health Organization (PAHO) during 2004-2005, suggesting that 18 of the 46 studied municipalities' water supplies had arsenic concentrations higher than 10 µg/L and reaching up to 161 µg/L [151]. Cienfuegos Bay, also in Central America, has a long history of heavy metal contamination due to industrial development in the close vicinity. The contamination became severe in 2001 due to spillage of a high pH solution containing inorganic oxides of arsenic from a nitrogen fertilizer plant.

1.10 Methods of Arsenic Removal from Water

1.10.1 Removal via Precipitation

Arsenic from mining and metallurgical effluents is removed with other metals by precipitation or non-precipitation methods. In precipitation methods, lime is used to precipitate As(III) and As(V) both as calcium arsenite/arsenate, while Fe(II) is used to precipitate arsenic as basic ferric arsenates or crystalline scorodite [152]. Alum and iron precipitations are also common methods used in water treatment for arsenic removal; however, they are not generally believed to be the major arsenate removal mechanisms. Scorodite (FeAsO₄·2H₂O) is widespread in arsenic-bearing ore deposits and its solubility regulates the concentration of arsenate in natural waters with low pH [153]. As(III) does not precipitate in salt form with metal ions; however, it precipitates in the form of orpiment (As₂S₃) in reducing and sulfide-rich environments.

Along with direct precipitation as pure minerals, arsenic may also co-precipitate with oxides and hydroxides of iron, aluminum, and manganese. Co-precipitation of arsenate with iron and aluminum is assumed to be the chief arsenic removal mechanism in water treatment plants. It has been reported that if the Fe/As weight ratio exceeds 5:1 at elevated pH then two mechanisms, surface complexation and electrostatic attraction, lead to co-precipitation of arsenate. Mamtaz and Bache [154] reported Fe(III) to be more efficient than Fe(II) at removing arsenite from groundwater via co-precipitation due to the lower rate of oxidation of Fe(II). Meng et al. [155] reported that Fe/As weight ratios greater than 40 can be beneficial in reducing arsenic levels to less than $50 \,\mu$ g/L in Bangladesh well water due to the presence of elevated phosphate. Fuller et al. [156] performed kinetic studies of arsenate adsorption and co-precipitation and showed that the initial arsenate uptake by co-precipitation was significantly greater than by adsorption.

1.10.2 Removal via Adsorption

Arsenic adsorption on different minerals has been reported. Fe oxide minerals are considered to be most important because of their strong binding ability to arsenic. Therefore, the interactions of arsenic with various Fe-bearing minerals such as hydrous ferric oxide, goethite, ferrihydrite, Fe oxide-coated sand, and red mud have been studied. Hydroxides of iron are considered to be most commonly involved minerals in adsorption of arsenic under both acidic and alkaline conditions, while other adsorbents perform the function under acidic conditions only. Arsenic adsorption on aluminum oxides such as alumina, gibbsite, kaolinite, illite, and montmorillonite has also been studied extensively [157]. Adsorption of arsenic in two different oxidation states, As(III) and As(V), was not identical under the same conditions such that As(V) adsorption was high under acidic conditions, while As(III) adsorbs under basic conditions. Several solid materials such as clay minerals, fly ashes, metal oxide mixtures, and specially hydrated lime [Ca(OH)₂] have been tested as sorbents for retaining arsenic and other metals. Lopez-Anton et al. [158] reported that activated carbons are able to capture arsenic and other metals. Lopez-Anton et al. [158] reported that activated carbons are able to capture arsenic and other metals.

Natural products are also used as adsorbents because of their good performance and low cost for the removal of heavy metals including arsenic [159]. The most widely adsorbent used is activated carbon manufactured from bituminous coal, agricultural wastes, wood, and petroleum. Efficacy of the adsorbents can be significantly enhanced by physical or chemical activation. The three-pitcher filtration system has been employed in Bangladesh at the household level to remove arsenic from ground water by adding Fe filings for arsenic adsorption [160]. In this method, an upper pitcher is used to reduce arsenate to arsenite, a middle pitcher contains charcoal for adsorption of arsenic species, and a lower pitcher collects arsenic-free water. Natural red earth can adsorb both species of arsenic, and thus by using this adsorbent three pitcher systems can be reduced to two pitcher systems.

It has also been reported that significant arsenic adsorption occurs on the surfaces of metal sulfides [161]. Arsenite and arsenate primarily adsorb at the surface of Fe oxides due to their wide occurrence, high point of zero charge, and high affinity for arsenic. Two commonly occurring minerals of Fe, ferrihydrite and goethite, are considered important due to their large surface area. Major soil constituents such as phosphate, silicic acid, sulfate, and natural organic matter also strongly affect arsenic adsorption. Numerous studies have shown that organic acids such as humic substances also compete with arsenic for adsorption sites on oxide surfaces due to the anionic nature of organic acids and strong affinity of their carboxylic and phenolic groups to the oxide surface [162].

Arsenic removal efficiency can also be improved by pretreatment of oxidized As(III) to As(V). The oxidation methods used include ozone and oxygen, hydrogen peroxide, manganese

oxides, UV/Fe, and TiO₂/UV. It has been suggested that ferric ions assisted by UV/TiO₂ could significantly accelerate the oxidation process of As(III) in comparison to other oxidizing agents [163]. Ferrate [Fe(VI)] is one of the environmentally friendly reagents used in water and wastewater treatment as it can be used as both an oxidant and a coagulant.

Adsorption of arsenic is a complex phenomenon and depends upon properties of the solid surface, hydration, soil pH, changes in cation coordination, isomorphous substitution, crystallinity, concentration, and species of arsenic and competing ions. Adsorption of arsenite and arsenate on mineral surfaces can be of two types: non-specific and specific. In non-specific adsorption, electrostatic attraction lies between a charged surface and an oppositely charged ion in solution in which the adsorbed ion resides at a certain distance from the mineral surface. On the other hand, in specific adsorption a coordination complex is formed between the ion and the mineral surface, which is stronger as compared to non-specific adsorption. It has been suggested that arsenate adsorbs mainly through non-specific sorption (electrostatic attraction), while arsenite is adsorbed via specific sorption (inner-sphere complexation). Adsorption of arsenite and arsenate on various adsorbents under different pH ranges is depicted in Table 1-7. Decreasing pH leads to increased arsenic retention because of the shift in mineral surface charges from negative to more positive; consequently, arsenic species become less mobile. At a very low pH range (less than 1) when nearly all arsenic species exist in neutral form, arsenic is no longer sorbed and becomes mobile again. At low pH, oxide mineral is dissolved in water and arsenic is released into the aqueous phase, with a resultant very high arsenic input to downstream water bodies. Precipitation of Fe^{3+} ions as hydroxides effectively scavenges most of the arsenate from water through sorption. However, arsenite sorption causes comparatively less contamination of downstream water bodies. The sorbed arsenate is also released into the water bodies under acidic conditions due to dissolution of iron hydroxide and under basic conditions due to the change in the surface charge of iron hydroxide.

Anions that are present with similar or higher charge densities such as phosphate (PO_4^{-3}) and sulfate (SO_4^{-2}) may compete for the sorption site of arsenic on mineral surfaces. The priority sequence of various anion adsorptions over activated alumina under the pH range 5.5–8.5 follows the order [164]:

$$OH^- > H_2AsO_4^- > Si(OH)_3O^- > F^- > HSeO_3^- > SO_4^{-2} > CrO_4^{-2} > HCO_3^- > Cl^- > NO_3^- > Br^- > I^-$$

Phosphate, which has similar chemical properties and behavior to arsenate, strongly competes with arsenate for sorption sites on metal oxide surfaces. Significant reduction in arsenate and arsenite adsorption by phosphorus and sulfate on various adsorbents such as amorphous iron oxide goethite, gibbsite, kaolinite, montmorillonite, aluminum oxide, and quartz has been reported [165]. It has been suggested that arsenic pollution of ground water in the Bengal Basin, Bangladesh, is due to competitive exchange of phosphate anions from overusage of fertilizers with arsenic thereby releasing arsenic into water bodies [166].

Yean et al. [167] observed a considerable decrease of arsenic sorption to magnetite nanoparticles in the presence of natural organic matter. Redman et al. [168] suggested that natural organic matter affects arsenic redox and complexation speciation by delaying formation of sorption equilibrium.

1.10.3 Phyto-Remediation

In the main, extraction of arsenic from polluted soil involves use of physical and chemical techniques which are neither economical nor environment friendly. In addition, they cause further deterioration of the land in the locale and render it unusable. A promising technology for arsenic extraction is phyto-remediation, in which plants known as hyperaccumulators are grown in arsenic-polluted sites. Plants extract large amounts of pollutant from the soil via their roots and concentrate it in aboveground tissues [169]. A Chinese brake fern (*Pteris vittata*) shows excellent arsenic hyperaccumulation properties in contaminated soil. It concentrates arsenic in aboveground biomass, which is a primary requirement for efficient phyto-remediation. Several other ferns (*Pityrogramma calomelanos* L., *Pteris cretica, Pteris longifolia*, and *Pteris umbrosa*) have also been reported as arsenic hyperaccumulators [170]. Other plants such as Indian mustard and tomato have also been reported to accumulate arsenic, principally in roots. It has been reported that *P. vittata* stores arsenic in epidermis and mesophyll tissues, and the xylem strands of the pinnae ultimately reach the leaves via transpiration stream [171].

Detoxification of heavy metals by plants is based upon biochemical mechanisms specific for plant species and metals. In general, the mechanism includes formation of intracellular metal-chelating ligands such as amino acids, organic acids, and two classes of peptides: phyto-chelatins and metallothioneins. Organic acids such as malic and citric acid are known for chelating Ni and Zn, whereas proteins and phenolics detoxify Cu. The thiol-rich phyto-chelatins are specific for the chelation of arsenic via bonding to thiol groups [172]. Arsenic toxicity can also be reduced by applying sulfates of Zn, Fe, and Al to the soil. The arsenic reacts with these sulfates and forms their insoluble salts.

Arsenate +
$$FeSO_4/ZnSO_4 \rightarrow Fe/Zn$$
-arsenate
(insoluble and unavailable to the plant) (1.18)

1.11 Conclusions

This chapter has provided a global scenario of current status of research on arsenic in the biosphere. The mitigation of arsenic-related problems is a huge challenge worldwide. The foremost requirement to plan advanced treatment and removal technologies is to understand the chemistry behind this unique element. A brief overview has been presented in this chapter pertaining to various sources of arsenic exposure and its mobilization into the environment, which serves as a critical information resource to individuals involved in studying arsenic environmental chemistry and epidemiology and its removal from water bodies [173]. A arsenic for the most part releases naturally from the leaching of volcanic rocks, sulfidic ores, hydrothermal and geothermal alterations, and volcanic exhalations. In addition, mining activities, metallurgical processes, and coal-based thermal power plants contribute significantly to the release of arsenic into air, water, and soil. The presence of arsenic and its mobilization are highly dependent upon geographical conditions, environmental factors, redox conditions, and pH. Thus, the information gained from a consideration of all these factors will contribute to the exploration and demarcation of high and low risk areas suffering from arsenic contamination.

References

- Bhattacharya P, Frisbie SH, Smith E, Naidu R, Jacks G, Sarkar B. Arsenic in the environment, a global perspective. In: Sarkar B, editor. Handbook of heavy metals in the environment. New York: Marcell Dekker; 2002. pp. 147–215.
- [2] Bundschuh J, Garcia ME. Rural Latin America—a forgotten part of the global groundwater arsenic problem. In: Bhattacharya P, Ramanathan AL, Bundschuh J, Chandrasekharam D, Mukherjee AB, editors. Groundwater for sustainable development: problems, perspectives and challenges. Leiden, the Netherlands: A.A. Balkema Publishers; 2007.
- [3] Flora SJS, Mehta A, Mital M. Heavy metal induced oxidative stress and its possible reversal by chelation therapy. Ind J Med Res 2008;128:221–4.
- [4] Flora SJS. Arsenic-induced oxidative stress and its reversibility. Free Rad Biol Med 2011;51:257–81.
- [5] Guha Mazumder D, Purkayastha I, Ghose A, Mistry G, Saha C, Nandy AK, et al. Hypertension in chronic arsenic exposure: a case control study in West Bengal. J Environ Sci Health A Tox Hazard Subst Environ Eng 2012;47(11):1514–20.
- [6] Rahman M, Vahter M, Wahed MA, Sohel N, Yunus M, Streatfield PK, et al. Prevalence of arsenic exposure and skin lesions. A population based survey in Matlab, Bangladesh. J Epidemiol Community Health 2006;60(3):242–8.
- [7] USEPA (United States Environmental Protection Agency) Fed Regist 2001;66:6975–7066. Washington, DC.
- [8] Nriagu J. Arsenic poisoning through the ages. In: Frankenberger Jr. WT, editor. Environmental chemistry of arsenic. New York: Marcel Dekker; 2002. pp. 1–26.
- [9] Audi G, Bersillon O, Blachot J, Wapstra AH. The NUBASE evaluation of nuclear and decay properties. Nucl Phys A 2003;729(1):3–128.
- [10] Dutre V, Vanecasteele C. Solidification/stabilization of arsenic containing waste: leach tests and behavior of arsenic in the leachate. Waste Manage 1995;15(1):55–62.
- Healy SM, Wildfang E, Zakharyan RA, Aposhian HV. Diversity of inorganic arsenic biotransformation. Biol Trace Elem Res 1999;68:249–66.
- [12] Chiou HY, Hsueh YM, Hsieh LL, Hsu LI, Hsu YH, Hsieh FI, et al. Arsenic methylation capacity, body retention, and null genotypes of glutathione S-transferase M1 and T1 among current arsenic-exposed residents in Taiwan. Mutat Res 1997;386:197–207.
- [13] Nemati B, Gregus Z. Reduction of arsenate to arsenite in hepatic cytosol. Toxicol Sci 2002;70:4–12.
- [14] Nico PS, Fendorf SE, Lowney YW, Holm SE, Ruby MV. Chemical structure of arsenic and chromium in CCA-treated wood: implications of environmental weathering. Environ Sci Technol 2004;38(19):5253-60.
- [15] Smith E, Naidu R, Alston AM. Arsenic in the soil environment: a review. In: Sparks DL, editor. Advances in agronomy, vol. 64. New York, NY: Academic Press; 1998. pp. 149–95.
- [16] Nordstrom DK. Worldwide occurrences of arsenic in groundwater. Science 2002;296:2143-5.
- [17] Williams M. Arsenic in mine waters: an international study. Environ Geol 2001;40(3):267-78.
- [18] Nordstrom DK, Archer DG. Arsenic thermodynamic data and environmental geochemistry. In: Welch AH, Stollenwerk KG, editors. Arsenic in goundwater: geochemistry and occurrence. Boston: Kluwer Academic Publishers; 2003. pp. 1–25.
- [19] Castro SH, Baltierra L. Study of the surface properties of enargite as a function of pH. Int J Miner Proc 2005;77(2):1-7.

- [20] Nesbitt HW, Uhlig I, Szargan R. Surface reconstruction/As-polymerization, fractured surface. Am Mineral 2002;87:1000–14.
- [21] Klein C. 22nd edition of the manual of mineral science. New York: John Wiley & Sons, Inc.; 2002.
- [22] Hanusch KH, Grossman KA, Herbst B, Rose H, Wolf V. Arsenic and arsenic compounds. In: Gerhartz W, Yamamoto YS, Campbell FT, Pfefferkorn R, Rounsaville JF, editors. Ullman's encyclopedia of industrial chemistry, vol. A3. Federal Republic of Germany: VCH; 1985. pp. 113–41.
- [23] Rumpler A, Edmonds JS, Katsu M, Jensen KB, Goessler W, Raber G, et al. Arsenic-containing long-chain fatty acids in cod-liver oil: a result of biosynthetic infidelity. Angew Chem Int Ed 2008;47(14):2665–7.
- [24] Elke D, Kligerman AD, Diaz-Bone RA. Organoarsenicals uptake, metabolism, and toxicity. Royal Society of Chemistry. In: Sigel A, Sigel H, Sigel RKO, editors. Organometallics in environment and toxicology, vol. 7 of metal ions in life sciences. Cambridge, UK: The Royal Society of Chemistry; 2010.
- [25] Neff JM. Arsenic in the ocean. In: Bioaccumulation in marine organisms. Oxford: Elsevier; 2002. pp. 57-78.
- [26] Sharma VK, Sohn M. Aquatic arsenic: toxicity, speciation, transformations, and remediation. Environ Int 2009;35(4):743–59.
- [27] Ramirez-Solis A, Mukopadhyay R, Rosen BP, Stemmler TL. Experimental and theoretical characterization of arsenite in water: insights into the coordination environment of As-O. Inorg Chem 2004;43(9):2954–9.
- [28] Planer-Friedrich B, Suess E, Scheinost AC, Wallschlager D. Arsenic speciation in sulfidic waters: reconciling contradictory spectroscopic and chromatographic evidence. Anal Chem 2010;82(24):10228–10235.
- [29] Yan XP, Kerrich R, Hendry MJ. Distribution of arsenic(III), arsenic(V) and total inorganic arsenic in porewaters from a thick till and clay-rich aquitard sequence, Saskatchewan, Canada. Geochim Cosmochim Acta 2000;64:2637–48.
- [30] Scott MJ, Morgan JJ. Reaction at oxide surfaces: oxidation of As (III) by synthetic birnessite. Environ Sci Technol 1995;29:1898–905.
- [31] Oremland RS, Dowdle PR, Hoeft S, Sharp JO, Schaefer JK, Miller LG, et al. Bacterial dissimilatory reduction of arsenate and sulfate in meromictic Mono Lake, California. Geochim Cosmochim Acta 2000;64:3073–84.
- [32] NRC (National Research Council) Arsenic in drinking water. Washington, DC: National Academy Press; 1999.
- [33] Matschullat J. Arsenic in the geosphere—a review. Sci Total Environ 2000;249(1-3):297-312.
- [34] Riedel FN, Eikmann T. Natural occurrence of arsenic and its compounds in soils and rocks. Wiss. Umwelt 1986;3–4:108–17.
- [35] Belkin HE, Zheng B, Finkelman RB. Human health effects of domestic combustion of coal in rural China: a casual factor foe arsenic and fluorine poisoning. Second World Chinese Conference on Geological Sciences, extended abstracts. Stanford CA: Stanford University; 2000. pp. 522–524.
- [36] Mozgova NN, Borodaev YS, Gablina IF, Cherkashev GA, Stepanova TV. Mineral assemblages as indicators of the maturity of oceanic hydrothermal sulfide mounds. Lithol Miner Resour 2005;40(4):293–319.
- [37] Plant JA, Kinniburgh DG, Smedley PL, Fordyce FM, Klinck BA. Arsenic and selenium. In: Heinrich DH, Karl KT, editors. Treatise on geochemistry. Oxford: Pergamon; 2003. pp. 17–66.
- [38] Datta DK, Subramanian V. Texture and mineralogy of sediments from the Ganges-Brahmaputra-Meghna river system in the Bengal Basin, Bangladesh and their environmental implications. Environ Geol 1997;30:181–8.
- [39] Sproal R, Turoczy NJ, Stagnitti F. Chemical and physical speciation of arsenic in a small pond receiving gold mine waste effluent. Ecotoxicol Environ Saf 2002;53:370–5.
- [40] McLaren RG, Megharaj M, Naidu R. Fate of arsenic in the soil environment. In: Naidu R, Smith E, Owens G, Bhattacharya P, Nadebaum P, editors. Managing arsenic in the environment: from soil to human health. Collingwood, Victoria, Australia: CSIRO Publishing; 2006. pp. 157–82.
- [41] Lasat M. The use of plants for the removal of toxic metals from contaminated soils. USEPA. American Association for Advancement of Science Fellowship Program; 2000. 34pp.

- [42] Mukherjee AB, Bhattacharya P. Arsenic in groundwater in the Bengal Delta Plain: slow poisoning in Bangladesh. Environ Rev 2001;9:189–220.
- [43] Bencko V. Arsenic. In: Fishbein L, Furst A, Mehlman MA, editors. Genotoxic and Carcinogenic Metals: Environmental and Occupational Occurrence and Exposure. Advances in Modern Environmental Toxicology, Vol. 11. Princeton, NJ: Princeton Scientific Publishing; 1987.
- [44] Pacyna JM. Atmospheric emissions of heavy metals for Europe: improvements, updates, historical data and projections. A Report for the International Institute for Applied Systems Alalysis (IIASA), Laxenburg, Austria; 1996. 13 pp.
- [45] Finkelman RB, Palmer CA, Kolker A, Mroczkowski SJ. Quantifying the modes of occurrence of elements in coal. In: Li BQ, Liu ZY, editors. Prospects for coal science in the 21st century. Proceedings of the Tenth International Conference on Coal Science. Xianyang: Shanxi Science and Technology Press; 1999. pp. 21–4.
- [46] Abernathy C, Chakraborti D, Edmonds JS, Gibb H, Hoet P, Hopenhayn-Rich C. Environmental health criteria for arsenic and arsenic compounds. Environ Health Criteria 2001;224:1–521.
- [47] Yudovich YE, Ketris MP. Arsenic in coal: a review. Int J Coal Geol 2005;61:141-96.
- [48] Senior CL, Bool LE, Morency JR. Laboratory study of trace element vaporization from combustion of pulverized coal. Fuel Process Technol 2000;63:109–24.
- [49] Eary LE, Rai D, Mattigod SV, Anisworth CC. Geochemical factors controlling the mobilization of inorganic constituents from fossil fuel combustion residues. I: Review of the major elements. J Environ Qual 1990;19:202–14.
- [50] Praharaj T, Powell MA, Hart BR, Tripathy S. Leachability of elements from sub-bituminous coal fly ash from India. Environ Int 2002;27:609–15.
- [51] Kizilstein LY. Ecogeochemistry of trace elements in coals. Rostova-na Donu [Rostov on the Don River]: SK NZ VSh [North-Caucasus Sci. Center of High School]; 2002. 296pp.
- [52] Sterling RO, Helble JJ. Gas-solid reaction of arsenic metal vapors with fly ash. Chemosphere 2003;51(10):1111-19.
- [53] Caceres L, Gruttner E, Contreras R. Water recycling in arid regions—Chilean case. Ambio 1992;21: 138-44.
- [54] Lerda DE, Prosperi CH. Water mutagenicity and toxicology in Rio Tercero, Cordoba, Argentina. Water Res 1996;30:819–24.
- [55] Williams M, Fordyce F, Paijitprapapon A, Charoenchaisri P. Arsenic contamination in surface drainage and groundwater in part of the southeast Asian tin belt, Nakhon Si Thammarat Province, southern Thailand. Environ Geol 1996;27:16–33.
- [56] Maest AS, Pasilis SP, Miller LG, Nordstrom DK. Redox geochemistry of arsenic and iron in Mono Lake, California, USA. In: Kharaka YK, Maest AS, editors. Proceedings of the Seventh International Symposium on Water-Rock Interaction. Rotterdam: A.A. Balkema; 1992. pp. 507–11.
- [57] Hasegawa H, Matsui M, Okamura S, Hojo M, Iwasaki N, Sohrin Y. Arsenic speciation including "hidden" arsenic in natural wate. Appl Organometal Chem 1999;13:113–19.
- [58] Seyler P, Martin JM. Arsenic and selenium in a pristine river-estuarine system: the Krka, Yugoslavia. Mar Chem 1991;34:137–51.
- [59] Howard AG, Hunt LE, Salou C. Evidence supporting the presence of dissolved dimethylarsinate in the marine environment. Appl Organometal Chem 1999;13:39–46.
- [60] WHO Environmental Health Criteria 224: arsenic compounds, 2nd ed. Geneva: World Health Organization; 2001.
- [61] O'Day PA, Vlassopoulos D, Root R, Rivera N. The influence of sulfur and iron on dissolved arsenic concentrations in the shallow subsurface under changing redox conditions. Proc Natl Acad Sci USA 2004;101:13703-13708.
- [62] Larsen EH, Moscholm L, Nielsen MM. Atmospheric deposition of trace elements around point sources and human health risk assessment: II. Uptake of arsenic and chromium by vegetables grown near a wood preservation factory. Sci Total Environ 1992;126:263–75.
- [63] Meharg AA. Arsenic in rice—understanding a new disaster for South-East Asia. Trends Plant Sci 2004;9:415-17.
- [64] BIAM. In 4th International Conference on Arsenic Contamination of Groundwater in Bangladesh: Cause, Effect and Remedy; 2002.
- [65] Delgado-Andrade C, Navarro M, Lopez H. Determination of total arsenic levels by hydride generation atomic absorption spectrometry in foods from south-east Spain: estimation of daily dietary intake. Food Addit Contam 2003;20:923–32.
- [66] Lasky T, Sun W, Kadry A. Mean total arsenic concentrations in chicken 1989–2000 and estimated exposures for consumers of chicken. Environ Health Perspect 2004;112:18–21.
- [67] Schoof RA, Yost LJ, Eickhoff J. A market basket survey of inorganic arsenic in food. Food Chem Toxicol 1999;37:839–46.
- [68] Canadian Food Inspection Agency. Fact Sheet, Inorganic Arsenic and Hijiki Seaweed Consumption. Available at: http://www.inspection.gc.ca/english/corpaffr/foodfacts/arsenice.shtml ; 2001.
- [69] Das DK, Mandal LN. Effect of puddling and different times of organic matter application before puddling on the availability of applied iron, manganese, copper and zinc in rice soils. Agrochimica 1988;32:327–36.
- [70] Bae M, Watanabe C, Inaoka T, Sekiyama M, Sudo N, Bokul MH. Arsenic in cooked rice in Bangladesh. Lancet 2002;360:1839–40.
- [71] Rohman MA, Hasegawa H, Rahman MM, Rahman MA, Miah MAM. Accumulation of arsenic in tissues of rice plant (*Oryza sativa* L.) and its distribution in fraction of rice grain. Chemosphere 2007;69:942–8.
- [72] Smith CJ, Livingston SD, Doolittle DJ. An international literature survey of "IARC Group I Carcinogens" reported in mainstream cigarette smoke. Food Chem Toxicol 1997;35:1107–30.
- [73] Noel B. Buerger disease or arsenic intoxication. Arch Intern Med 2001;161:1016.
- [74] Rahman M, Chowdhury AS, Fukui T. Association of thromboangiitis obliterans with cigarette and bidi smoking in Bangladesh: a case-control study. Int J Epidemiol 2000;29:266–70.
- [75] German CR, Von Damm KL. Hydrothermal processes. In: Heinrich DH, Karl KT, editors. Treatise on geochemistry. Oxford: Pergamon; 2003. pp. 181–222.
- [76] Tarasov VG, Gebruk AV, Mironov AN, Moskalev LI. Deep-sea and shallow-water hydrothermal vent communities: two different phenomena. Chem Geol 2005;224(1–3):5–39.
- [77] Zhao FJ, Ma JF, Meharg AA, McGrath SP. Arsenic uptake and metabolism in plants. New Phytol 2009;181(4):777-94.
- [78] Le Friant A, Harford CL, Deplus C, Boudon G, Sparks RSJ, Herd RA, et al. Geomorphological evolution of Montserrat (West Indies): importance of flank collapse and erosional processes. J Geol Soc 2004;161(1):147-60.
- [79] Price RE, Pichler T. Distribution, speciation and bioavailability of arsenic in a shallow water submarine hydrothermal system, Tutum Bay, Ambitle Island, PNG. Chem Geol 2005;224(1-3):122-35.
- [80] Valsami-Jones E, Baltatzis E, Bailey EH, Boyce AJ, Alexander JL, Magganas A, et al. The geochemistry of fluids from an active shallow submarine hydrothermal system: Milos island, Hellenic Volcanic Arc. J Volcanol Geotherm Res 2005;148(1-2):130-51.
- [81] Berquo TS, Banerjee SK, Ford RG, Penn RL, Pichler T. High crystallinity Si-ferrihydrite: an insight into its Néel temperature and size dependence of magnetic properties. J Geophys Res 2007;112(B2):B02102.
- [82] McCarthy KT, Pichler T, Price RE. Geochemistry of Champagne Hot Springs shallow hydrothermal vent field and associated sediments, Dominica, Lesser Antilles. Chem Geol 2005;224(1-3):55-68.

- [83] Pokrovski GS, Roux J, Harrichoury JC. Fluid density control on vapor-liquid partitioning of metals in hydrothermal systems. Geology 2005;33(8):657–60.
- [84] Charlou JL, Donval JP, Douville E, Jean-Baptiste P, Radford-Knoery J, Fouquet Y, et al. Compared geochemical signatures and the evolution of Menez Gwen (37°50'N) and Lucky Strike (37°17'N) hydrothermal fluids, south of the Azores Triple Junction on the Mid-Atlantic Ridge. Chem Geol 2000;171(1-2):49–75.
- [85] Pester NJ, Reeves EP, Rough ME, Ding K, Seewald JS, Seyfried Jr. WE. Subseafloor phase equilibria in high-temperature hydrothermal fluids of the Lucky Strike Seamount (Mid-Atlantic Ridge, 37°17′N). Geochim Cosmochim Acta 2012;90:303–22.
- [86] Douville E, Bienvenu P, Charlou JL, Donval JP, Fouquet Y, Appriou P, et al. Yttrium and rare earth elements in fluids from various deep-sea hydrothermal systems. Geochim Cosmochim Acta 1999;63(5):627–43.
- [87] Soto LA. Stable carbon and nitrogen isotopic signatures of fauna associated with the deep-sea hydrothermal vent system of Guaymas Basin, Gulf of California. Deep Sea Res (Part 2.) Top Stud Oceanogr 2009;56(19-20):1675-82.
- [88] Sturz AA, Sturdivant AE, Leif RN, Simoneit BRT, Gieskes JM. Evidence for retrograde hydrothermal reactions in near surface sediments of Guaymas Basin, Gulf of California. Appl Geochem 1996;11(5):645–65.
- [89] Martinez F, Okino K, Ohara Y, Reysenbach AL, Goffredi SK. Back-arc basins. Oceanography 2007;20:116–27.
- [90] Taylor B, Martinez F. Back-arc basin basalt systematics. Earth Planet Sci Lett 2003;210(3-4):481-97.
- [91] Mottl MJ, Seewald JS, Wheat CG, Tivey MK, Michael PJ, Proskurowski G, et al. Chemistry of hot springs along the Eastern Lau Spreading Center. Geochim Cosmochim Acta 2011;75(4):1013–38.
- [92] Tregoning P. Plate kinematics in the western Pacific derived from geodetic observations. J Geophys Res 2002;107(B1):2020.
- [93] Webster-Brown JG, Lane V, Webster KS. Arsenic in the Waikato River an update. In: Proceedings of the 22nd Geothermal Workshop, New Zealand; 2000.
- [94] Chen M, Ma LQ, Harris WG. Baseline concentrations of 15 trace elements in Florida surface soils. J Environ Qual 1999;28:1173–81.
- [95] Webster JG, Nordstrom DK. Geothermal arsenic. In: Welch A, Stollenwerk KG, editors. Arsenic in groundwater. Elsevier; 2003. pp. 101–25.
- [96] Haeri A, Strelbitskaya S, Porkhial S, Ashayeri A. Distribution of arsenic in geothermal waters from Sabalan geothermal field, N-W Iran. Proceedings of the Thirty-sixth Workshop on Geothermal Reservoir Engineering. Stanford, CA: Stanford University; 2011. Jan 31–Feb 2, 2011. SGP-TR-191.
- [97] Pokrovski G, Gout R, Schott J, Zotov A, Harrichoury JC. Thermodynamic properties and stoichiometry of As(III) hydroxide complexes at hydrothermal conditions. Geochim Cosmochim Acta 1996;60(5):737–49.
- [98] Larsen EK, Quetel CR, Munoz R, Fiala-Medioni A, Donard OFX. Arsenic speciation in shrimp and mussel from the Mid-Atlantic hydrothermal vents. Mar Chem 1997;57:341–6.
- [99] Konhauser KO, Ferris FG. Diversity of iron and silica precipitation by microbial mats in hydrothermal waters, Iceland: implications for Precambrian iron formation. Geology 1996;24:323–6.
- [100] Savage KS, Tingle TN, O'Day PA, Waychunas GA, Bird DK. Arsenic speciation in pyrite and secondary weathering phases, Mother Lode Gold District, Tuolumne County, California. Appl Geochem 2000;15:219-44.
- [101] Zanker H, Moll H, Richter W, Brendler V, Hennig C, Reich T, et al. The colloid chemistry of acid rock drainage solution from an abandoned Zn-Pb-Ag mine. Appl Geochem 2002;17:633–48.
- [102] Madigan ZT, Martinko JM, Parker J. Brock biology of microorganisms, 10th ed. Upper Saddle River, NJ: Prentice Hall; 2002.
- [103] Lengke MF, Sanpawanitchakit C, Tempel RN. The oxidation and dissolution of arsenic-bearing sulfides. Can Mineral 2009;47:593–613.

- [104] Yu Y, Zhu Y, Williams-Jones AE, Gao Z, Li D. A kinetic study of the oxidation of arsenopyrite in acidic solutions: implications for the environment. Appl Geochem 2004;19:435–44.
- [105] Floroiu RM, Davis AP, Torrents A. Kinetics and mechanism of As2S3(am) dissolution under N2. Environ Sci Technol 2004;38:1031–7.
- [106] Helz GR, Tossell JA, Charnock JM, Pattrick RAD, Vaughan DJ, Garner DC. Oligomerization in As (III) sulfide solutions: theoretical constraints and spectroscopic evidence. Geochim Cosmochim Acta 1995;59:4591–604.
- [107] Stauder S, Raue B, Sacher F. Thioarsenates in sulfidic waters. Environ Sci Technol 2005;39:5933-9.
- [108] Asta MP, Cama J, Ayora C, Acero P, de Giudici G. Arsenopyrite dissolution rates in O₂-bearing solutions. Chem Geol 2010;273:272–85.
- [109] Corkhill CL, Wincott PL, Lloyd JR, Vaughan DJ. The oxidative dissolution of arsenopyrite (FeAsS) and enargite (Cu3AsS4) by *Leptospirillum ferrooxidans*. Geochim Cosmochim Acta 2008;72:5616–33.
- [110] Helz GR, Tossell JA. Thermodynamic model for arsenic speciation in sulfidic waters; a novel use of *ab initio* computations. Geochim Cosmochim Acta 2008;72:4457–68.
- [111] Willey JD, Kieber RJ, Lancaster RD. Coastal rainwater hydrogen peroxide: concentration and deposition. J Atmos Chem 1996;25:149–65.
- [112] Plumlee GS, Smith KS, Montour MR, Ficklin WH, Mosier EL. Geologic controls on the composition of natural waters and mine waters draining diverse mineral deposit types. In: Filipek LH, Plumlee GS, editors. Environmental geochemistry of mineral deposits. Part B: case studies and research topics. Rev. Econ. Geol., vol. 6B; 1999. pp. 373–432.
- [113] Harrington JM, Fendorf SE, Rosenzweig RF. Bioticgeneration of arsenic(III) in heavy metal contaminated lake sediments. Environ Sci Technol 1998;32:2425–30.
- [114] Al-Rmalli SW, Jenkins RO, Watts MJ, Haris PI. Risk of human exposure to arsenic and other toxic elements from geophagy: trace element analysis of baked clay using inductively coupled plasma mass spectrometry. Environ Health 2010;9:79.
- [115] Wickre JB, Folt CL, Sturup S, Karagas MR. Environmental exposure and fingernail analysis of arsenic and mercury in children and adults in a Nicaraguan gold mining community. New Hampshire: Center for Environmental Health Sciences, Dartmouth College Hanover; 2004. pp. 2–6.
- [116] Root T, Bahr JM, Gotkowitz MB. Controls on arsenic concentrations in groundwater near Lake Geneva, Wisconsin. In: O'Day PA, Vlassopoulos D, Meng X, Benning LG, editors. Advances in arsenic research. Washington, DC: American Chemical Society; 2005. pp. 161–74.
- [117] De Vitre R, Belzile N, Tessier A. Speciation and adsorption of arsenic on diagenetic iron oxyhydroxides. Limnolol Oceanogr 1991;36:1480–5.
- [118] Gotkowitz M, Ellickson K, Clary A, Bowman G, Standridge J, Sonzogni W. Effect of well disinfection on arsenic in groundwater. Ground Water Monit Remediat 2008;28:60–7.
- [119] US EPA (United States Environmental Protection Agency). Arsenic and your distribution system. Available at: http://www.epa.gov/ogwdw/arsenic/pdfs/fs_arsenic_dist_sys_factsheet_final.pdf; 2007.
- [120] Gerke TL, Scheckel KG, Schock MR. Identification and distribution of vanadinite (Pb₅(V⁵⁺ O₄)₃Cl) in lead pipe corrosion by-products. Environ Sci Technol 2009;43:4412–18.
- [121] Lytle DA, Sorg TJ, Muhlen C, Wang LL. Particulate arsenic release in a drinking water distribution system. J Am Water Works Assoc 2010;102:87–98.
- [122] Bernstam L, Lan CH, Lee J, Nriagu JO. Effects of arsenic on human keratinocytes: morphological, physiological and precursor incorporation studies. Environ Res 2002;89(3):220–35.
- [123] Wester RC, Maibach HI, Sedik L, Melendres J, Wade M. In vivo and in vitro percutaneous absorption and skin decontamination of arsenic from water and soil. Fundam Appl Toxicol 1993;20:336–40.
- [124] Hossain MF Arsenic contamination in Bangladesh—a review. Agric Ecosyst Environ 2006;113:1-16.

- [125] Ravenscroft P, Burgess WG, Ahmed KM, Burren M, Perrin J. Arsenic in groundwater of the Bengal Basin: distribution, field relations, and hydrogeological setting. Hydrogeol J 2005;13:727–51.
- [126] Chakravarty AK, Das DK. Arsenic pollution and its environmental significance. J Interacad 1997;1:262-76.
- [127] Johnston R, Hug SJ, Inauen J, Khan NI, Mosler HJ, Yang H. Enhancing arsenic mitigation in Bangladesh: findings from institutional, psychological, and technical investigations. Sci Total Environ 2014; Aug 1;477-83,488-9.
- [128] Shukla DP, Dubey CS, Singh NP, Tajbakhsh M, Chaudhry, M. Sources and controls of arsenic contamination in groundwater of Rajnandgaon and Kanker District, Chattisgarh Central India. J Hydrol 2010; 395(1–2):49–66.
- [129] WHO Guidelines for Drinking Water Quality addendum to volume 1: recommendations, 2nd ed. Geneva: World Health Organization; 1998.
- [130] Buragohain M, Sarma HP. A study on spatial distribution of arsenic in groundwater samples of Dhemaji district of Assam, India by using arc view GIS software. Sci Rev Chem Commun 2012;2(1):7-11.
- [131] Byrne S, Amarasiriwardena D, Bandak B, Bartkus L, Kane J, Jones J, et al. Were Chinchorros exposed to arsenic. Arsenic determination in Chinchorro mummies' hair by laser ablation inductively coupled plasma-mass spectrometry (LA-ICP-MS). Microchem J 2010;94:28–35.
- [132] Birkle P, Bhattacharya P, Matschullat J, Mukherjee AB. Geogenic arsenic in groundwater of Latin America. In: Bundschuh J., Bhattacharya P, editors. Arsenic in the environment, vol. 1. Leiden, The Netherlands: CRC Press/Balkema Publisher; 2009. p. 3–15.
- [133] Castro de Esparza ML. The presence of arsenic in drinking water in Latin America and its effect on public health. In: Bundschuh J, Armienta MA, Birkle P, Bhattacharya P, Matschullat J, Mukherjee AB, editors. Natural arsenic in groundwater of Latin America. Leiden, the Netherlands: CRC Press/Balkema; 2009. pp. 17–29.
- [134] Steinmaus C, Yuan Y, Kalman D, Rey OA, Skibola CF, Dauphine D, et al. Individual differences in arsenic metabolism and lung cancer in a case-control study in Cordoba, Argentina. Toxicol Appl Pharmacol 2010;247(2):138–45.
- [135] Biagini RE, Salvador MA, de Qüerio RS, Torres-Soruco CA, Biagini MM, Diez-Barrantes A. Hidroarsenicismo crónico. Comentario de casos diagnosticados en el period 1972–1993. Arch Arg Dermatol 1995;45:47–52.
- [136] Nicolli HB, Tineo A, García JW, Falcón CM, Merino MH. Trace-element quality problems in groundwater from Tucumán, Argentina. In: Cidu R, editor. Water-rock interaction 2. Lisse, the Netherlands: Balkema; 2001. pp. 993–6.
- [137] Armienta MA, Segovia N. Arsenic and fluoride in the groundwater of Mexico. Environ Geochem Health 2008;30:345–53.
- [138] Armienta MA, Amat PD, Larios T, López DL. América Central y México Bundschuh J, Carrera AP, Litter M, editors. Distribución del arsénico en las regiones Ibérica e Iberoamericana. Buenos Aires, Argentina: Editorial Programa Iberoamericano de Ciencia y Tecnologia para el Desarrollo; 2008. pp. 187–210.
- [139] Ongley LK, Sherman L, Armienta A, Concilio A, Ferguson-Salinas C. Arsenic in the soils of Zimapán, Mexico. Environ Pollut 2007;145:793–9.
- [140] Hernández García Y. Metales pesados y arsénico en agua y sedimento de embalses artificiales del estado de Chihuahua, México. Chihuahua, Chih., Mexico: MSc Thesis, Facultad de Zootecnia y Ecología. Universidad Autónoma de Chihuahua; 2007.
- [141] Marshall G, Ferreccio C, Yuan Y, Bates MN, Steinmaus C, Selvin S, et al. Fifty-year study of lung and bladder cancer mortality in Chile related to arsenic in drinking water. J Natl Cancer Inst 2007;99(12):920–8.
- [142] Cornejo L, Lienqueo H, Arenas M, Acarapi J, Yañez J, Mansilla H. In field arsenic removal from natural water by zero valent iron. Environ Pollut 2008;156(3):827–31.

- [143] Oyarzun R, Lillo J, Higueras P, Oyarzú J, Maturana H. Strong arsenic enrichment in sediments from the Elqui watershed, Northern Chile: industrial (gold mining at El Indio-Tambo district) vs. geologic processes. J Geochem Explor 2004;84:53-64.
- [144] Swain WC, Duell Jr LFW. Water-quality data for shallow wells in the western San Joaquin Valley, California, May to August, 1989. U.S. Geological Survey Open-File Report, 92-655, Sacramento, CA; 1993.
- [145] Birkle P, Bundschuh J. High and low geothermal resources and potentials Bundschuh J, Alvarado GE, editors. Central America—geology, resources and hazards. Leiden, the Netherlands: Balkema; 2007. pp. 705–76.
- [146] Cardoso S, Grajeda C, Argueta S, Garrido SE. Experiencia satisfactoria para la remoción de arsénico en Condado Naranjo, Mixco, Guatemala. In: Litter MI, Sancha AM, Ingallinella AM, editors. Tecnologías económicas para el abatimiento de arsénico en aguas. Buenos Aires, Argentina: Editorial Programa Iberoamericano de Ciencia y Tecnologia para el Desarrollo; 2010. pp. 211–22.
- [147] Goff F, Grisby CO, Shevenell LA, Gutierrez JW. Geochemistry at Honduras geothermal sites. Los Alamos Sci 1986;Fall:90–3.
- [148] López D, Ramson L, Monterrosa J, Soriano T, Barahona J, Bundschuh J. Volcanic arsenic and boron pollution of Ilopango lake, El Salvador. In: Bundschuh J, Armienta MA, Birkle P, Bhattacharya P, Matschullat J, Mukherjee AB, editors. Natural arsenic in groundwater of Latin America. Leiden, the Netherlands: CRC Press/Balkema Publisher; 2009. pp. 129–43.
- [149] INAA Informe de los resultados de muestreo de As en comunidades rurales del municipio de San Isidro. Managua, Nicaragua: Instituto Nicaragüense de Acueductos y Alcantarillados; 1996.
- [150] WHO Arsenic in drinking-water. Background document for preparation of WHO Guidelines for drinkingwater quality. Geneva, Switzerland: World Health Organization; 2003. [WHO/SDE/WSH/03.04/75].
- [151] PAHO/WHO/UNICEF. Evaluación rápida de la calidad del agua de bebida Nicaragua. Informe Final, Managua, Nicaragua. <www.bvsde.paho.org>; 2005.
- [152] Bhattacharya P, Mukherjee A, Bundschuh J, Zevenhoven R, Loeppert R. Arsenic in soil and groundwater environment. Trace Metals Contam Environ 2007;9:127–56.
- [153] Magalhaes M, Clara E. Arsenic: an environmental problem limited by solubility. Pure Appl Chem 2002;74:1843–50.
- [154] Mamtaz R, Bache DH. Reduction of arsenic in groundwater by co-precipitation with iron. J Water Supply Res Technol 2001;50:313–24.
- [155] Meng X, Korfiatis GP, Christodoulatos C, Bang S. Treatment of arsenic in Bangladesh well water using a household co-precipitation and filtration system. Water Res 2001;35:2805–19.
- [156] Fuller CC, Davis JA, Waychunas GA. Surface chemistry of ferrihydrite: part 2. Kinetics of arsenate adsorption and coprecipitation. Geochim Cosmochim Acta 1993;57:2271–82.
- [157] Weerasooriya R, Tobschall HJ, Wijesekara HKDK, Bandara A. On the mechanistic modelling of As(III) adsorption on gibbsite. Chemosphere 2003;51:1001–13.
- [158] Lopez-Anton MA, Diaz-Somoano M, Fierro JLG, Martinez-Tarazona MR. Retention of arsenic and selenium compounds present in coal combustion and gasification flue gases using activated carbons. Fuel Process Technol 2007;88:799–805.
- [159] Katsoyiannis IA, Zouboulis AI. Removal of As from contaminated water sources by sorption onto iron oxide coated polymeric materials. Water Res 2002;36:5141–55.
- [160] Chowdhury AMR. Arsenic crisis in Bangladesh. J Sci Am 2004;291(2):70-5.
- [161] Bostick BC, Fendorf S, Manning BA. Arsenite adsorption on galena (PbS) and sphalerite (ZnS). Geochim Cosmochim Acta 2003;67:895–907.
- [162] Simeoni MA, Batts BD, McRae C. Effect of groundwater fulvic acid on the adsorption of arsenate by ferrihydrite and gibbsite. Appl Geochem 2003;18:1507–15.

- [163] Kim M-J, Nriagu J. Oxidation of arsenite in groundwater using ozone and oxygen. Sci Total Environ 2000;247(1):71–9.
- [164] Clifford DA. Ion exchange and inorganic adsorption. In: Letterman A, editor. Water quality and treatment (5th ed.). New York: McGraw Hill; 1999. pp. 9.1–9.91.
- [165] Jain A, Loeppert RH. Effect of competing anions on the adsorption of arsenate and arsenite by ferrihydrite. J Environ Qual 2000;29:1422–30.
- [166] Ravenscroft P, McArthur JM, Hoque BA. Geochemical and palaeohydrological controls on pollution of groundwater by arsenic Chappell WR, Abernathy CO, Calderon RL, editors. Arsenic exposure and health effects IV. Oxford, UK: Elsevier; 2001. pp. 53–77.
- [167] Yean S, Cong L, Yavuz CT, Mayo JT, Yu WW, Kan AT, et al. Effect of magnetite particle size on adsorption and desorption of arsenite and arsenate. J Mater Res 2005;20:3255–64.
- [168] Redman AD, Macalady DL, Ahmann D. Natural organic matter affects arsenic speciation and sorption onto hematite. Environ Sci Technol 2002;36:2889–96.
- [169] Pollard AJ. Metal hyperaccumulation: a model system for coevolutionary studies. New Phytol 2000;146(2):179-81.
- [170] Zhao FJ, Wang JR, Barker JHA, Schat H, Bleeker PM, McGrath SP. The role of phytochelatins in arsenic tolerance in the hyperaccumulator *Pteris vittata*. New Phytol 2003;159:403–10.
- [171] Gan KS, Wong SC, Yong JWH, Farquhar GD. (18)O spatial patterns of vein xylem water, leaf water, and dry matter in cotton leaves. Plant Physiol 2002;130:1008–21.
- [172] Hartley-Whitaker J, Ainsworth G, Vooijs R, Ten Bookum W, Schat H, Meharg AA. Phytochelatins are involved in differential arsenate tolerance in *Holcus lanatus*. Plant Physiol 2001;126:299–306.
- [173] Jakariya M. The use of alternative safe water options to mitigate the arsenic problem in Bangladesh: a community perspective. Available at: <<u>http://bicn.com/acic/resources/infobank/jakariya/sec07.htm</u>>; 2000.

This page intentionally left blank

2

Ground Water Arsenic Contamination and Its Health Effects in Bangladesh

Sk Akhtar Ahmad¹ and Manzurul Haque Khan²

¹BANGLADESH UNIVERSITY OF HEALTH SCIENCES, DHAKA, BANGLADESH, ²NATIONAL INSTITUTE OF PREVENTIVE AND SOCIAL MEDICINE, DHAKA, BANGLADESH

CHAPTER OUTLINE

2.1 Introduction	51
2.2 Arsenic Contamination in Bangladesh Ground Water	
2.3 Extent of Arsenic Contamination in Bangladesh	54
2.4 Arsenic in Different Environmental Media of Bangladesh	
2.5 Health Effects of Arsenic Toxicity in Bangladesh	59
2.6 Epidemiology of Arsenicosis in Bangladesh	
2.7 Management of Arsenicosis Patients in Bangladesh	
2.8 Socio-Cultural Aspects of Arsenicosis in Bangladesh	67
2.9 Conclusions	
References	68

2.1 Introduction

Arsenic contamination of ground water in Bangladesh was first reported in 1993, followed by official reporting of arsenicosis patients in 1994 [1]. It is estimated that more than 95% of the people in Bangladesh drink tubewell water and more than 50 million people are estimated to be ingesting arsenic-contaminated water [1–4]. Approximately 27% of shallow (<150 meter deep) wells in Bangladesh contain more than $50 \mu g/L$ arsenic [2]. In Bangladesh, concentrations of arsenic in ground water as high as $4.0 \, \text{mg/L}$ have been reported from Chatkhil of Noakhali district [2,4]. The worst-affected area is in the southeast of Bangladesh, whereas in some districts more than 90% of tubewells are affected [2,5,6]. Arsenic contamination of ground water tapped through tubewells had been reported from 62 out of 64 districts of Bangladesh [3]. In surveys covering 57,482 villages of 2934 unions in 270 upazillas, 29.12% of the 4,946,933 tubewells were found to be arsenic contaminated. About 15% (8540) of the villages had more

than 80% of their tubewells contaminated and these villages are identified as "hot spots." Among 66,034,962 individuals residing in 12,001,665 households in these upazillas, a total of 38,430 individuals had been reported as arsenicosis patients [5]. Many of the health problems known to be related to arsenic toxicity have been reported in Bangladesh [1,7–19].

In the early 1970s, when people in Bangladesh mainly relied on surface water (ponds and rivers) and subsurface (dug-well) water sources, diarrheal diseases and cholera were widely prevalent. This had prompted the search for a microbially safe water source. This search had led to mass use of tubewells. With the help of a hand-pump and pipes with strainers sunk a few meters into the ground, tubewells yielded water that was reasonably safe from the point of microbial contamination that causes diarrheal diseases. Moreover, this means of obtaining water became relatively cheap and the water was easy to collect, and in subsequent years hundreds of thousands of tubewells were installed on the basis of personal, governmental, and NGO initiatives. The tube-well installation initiative to supply safe water in Bangladesh was almost a total success, and increased the percentage of people with access to safe water from 77.6% in 1991 to 91.3% in 1994. Though the tubewell installation initiative provided safe water, a new problem regarding groundwater contamination surfaced. Tube-wells were found to yield water containing arsenic at levels not acceptable for consumption even by the Bangladesh water quality standard. The maximum allowable concentration of arsenic in drinking water in Bangladesh is 0.05 mg/L, which is five times higher than the World Health Organization provisional guideline value (0.01 mg/L) for arsenic in drinking water [1,2,6,9,20].

Arsenic contamination of tubewell water in Bangladesh was reported in 1993 by the Department of Public Health Engineering; and in 1994 the Department of Occupational and Environmental Health, National Institute of Preventive Medicine (NIPSOM), identified eight arsenicosis patients in Bangladesh [1,3]. The contaminated tubewells and patients were detected in the western part of Bangladesh in the village of Chamagram of Nawabganj District. Surveys for detection of arsenic in tubewells throughout the country showed a widespread distribution of contaminated tubewells. Regarding identification of arsenicosis patients in Bangladesh, up to 1997, 1625 arsenicosis patients had been identified and in 2008 the number of identified arsenicosis patients increased to 24,389. In 2012, 56,758 arsenicosis patients were identified by the Director General of Health Services (DGHS), Bangladesh [1,3].

2.2 Arsenic Contamination in Bangladesh Ground Water

Ground water is available at very shallow depths all over Bangladesh where the major aquifers are the Holocene alluviums and fan deposits and Pliocene fluvio-deltaic (Dupi Tila) sediments. Mio-Pliocene Tipam sands form minor aquifers in the hilly areas. The aquifers are highly transmissive and generally multilayered. The aquifer conditions vary from unconfined to leaky confined in the shallow alluvial deposits (Holocene alluviums) and are confined in the Dupi Tila and in deeper alluvial deposits. In the Ganges delta area the thickness of the recent sediment is higher and the Dupi Tila sandstone lies at greater depths. In the southern part (often called coastal plain) the thickness of the alluvial deposit is highest compared to other parts of the country. The aquifer systems in Bangladesh are geologically controlled and depend on the sedimentary characteristics, depositional environments, and other related parameters. Arsenic contamination in ground water is reported to be greater in the Holocene alluviums and fan deposit areas of Bangladesh [2,9,21].

The Bengal Basin is one of the largest sedimentary basins in the world. Bangladesh occupies most of the present-day delta in the Bengal Basin. The Bengal Basin is bounded on the west and northwest by the Rajmahal Hills (Trap). The northeast is bounded by the Garo, Khasi, and Jayantia hills (west to east), which stretch for about 97km from north to south and 240km from east to west. In the far northeast, Shillong or Assam Plateau acts as a boundary. Generally, the Bengal Delta is often referred to as the "Ganges-Brahmaputra-Meghna Delta" (GBMD), which is still active. It is reported that sediments from the Himalayas, adjoining India and Burma, had contributed to the development of the Bengal Basin, and the Ganges-Brahmaputra-Meghna river system in addition to tectonic activity, climatic changes, and accompanying sea level changes had played a significant role. The Ganga-Brahmaputra river system had mainly contributed to the buildup of the Bengal Fan. This river system had carried enormous volumes of sediments from the Himalayan belt. The sediments are derived from the upland Himalayan catchments, the Indo-Burman ranges, and from basement complexes of the northern and western parts of West Bengal (Rajmahal Hills, Choto Nagpur Plateau, Shillong Plateau). Along with sediments from those areas many weathered minerals including arsenic had entered the basin and had been deposited in the delta over thousands of years. The high arsenic content areas are found in the catchments of the Ganges, Brahmaputra, and Meghna rivers [22–31].

A variety of anthropogenic sources have been proposed as the cause of particular occurrences of elevated arsenic concentration in ground water in the Bengal Basin, including industrial pollution and the use of agrochemicals and wood preservatives; only the mineralogical source within the sediments of the Bengal Basin is consistent with the full regional extent and magnitude of arsenic occurrence as observed. It is now widely accepted that arsenic in ground water of the region has a source within the sediments of the Bengal Basin originating possibly from the Himalayan region; it is of a natural origin (geogenic in nature) mainly from the GBM river system, especially in the Holocene period [29–35].

2.2.1 Reasons for Leaching Arsenic in the Ground Water of Bangladesh

The reasons behind leaching arsenic in the ground water in Bangladesh are not clear. Three mechanisms have been proposed to explain arsenic pollution of ground water in the GBMD [2,32,33,36–42]:

1. Pyrite oxidation: It has been proposed that arsenic is present as arsenical pyrite in the alluvial sediments. Due to heavy withdrawal of ground water through shallow and deep tubewells for irrigation and domestic purposes, a vacuum is created that leads to entry of atmospheric oxygen into the aquifer subsequent to aquifer drawdown, which in turn leads to oxidation of arsenical pyrite and as a result arsenic is released.

2. Anion (competitive) exchange of sorbed arsenic with phosphate from fertilizer:

According to this hypothesis arsenic anions sorbed to aquifer minerals are displaced into solution by competitive exchange of phosphate anions derived from overapplication of fertilizer to surface soils. Phosphate derived from excessive use of phosphate fertilizer, from latrines, and from the fermentation/decay of buried peat deposits and other natural organic materials may leach into the aquifer and cause displacement of arsenic from sorption sites on aquifer minerals as a result of competitive (anion) exchange, resulting in arsenic pollution in the aquifer. However, it has been suggested that competitive exchange with phosphate generated *in situ* may contribute to arsenic pollution but this contribution is believed to be small.

3. Reductive dissolution of FeOOH and release of sorbed arsenic to ground water: The most widely accepted hypothesis is that, under anoxic conditions, reduction of iron oxyhydroxides (FeOOH) takes place, which results in release of sorbed arsenic to solution. Reduction of FeOOH is driven by microbial metabolism of organic matter.

2.2.2 Reasons for Variation of Arsenic Concentrations Between the Aquifers

The reasons for the distinction between groundwater arsenic concentrations in the shallow and deep aquifers of the Bengal Basin are not yet well understood. Differences between the sediments at depth may occur in terms of absolute arsenic concentrations and in the oxidation states and binding properties of the arsenic to the sediments. However, it is also possible that the history of groundwater movement and aquifer flushing in the Bengal Basin has been important in generating the differences in dissolved arsenic concentrations between the shallow and deep aquifers. Older, deeper sediments have been subject to longer periods of groundwater flow, aided by greater hydraulic heads during the Pleistocene period when glacial sea levels around the Bangladesh landmass were up to 130 m lower than today. Flushing of the deeper older aquifers with ground water is therefore likely to have been much more extensive than in the Holocene sediments deposited during the last 5000–10,000 years. Hence, much of the arsenic in the deep sediments may have previously been flushed away [2,9,21,43].

2.3 Extent of Arsenic Contamination in Bangladesh

A nationwide sample survey of tubewells, mostly from the Department of Public Health Engineering (DPHE)-installed tubewells located in 433 upazillas (eight samples per upazilla) from 61 districts (excluding the three hill districts), found that the arsenic concentrations ranged from less than 0.00025 to 1.670 mg/L. The median and mean arsenic concentrations in the samples were 0.004 mg/L and 0.055 mg/L respectively. Among the surveyed "shallow" tube-wells (wells less than 150 m deep), 27% of them exceeded the Bangladesh standard for arsenic in drinking water (0.05 mg/L) and 1% "deep" wells (more than 150 m deep) exceeded the Bangladesh standard. About 9% of the tubewells exceeded 0.20 mg/L, 1.8% exceeded 0.50 mg/L, and 0.1% exceeded 1.0 mg/L [1,2,44].

Deep wells tapping depths of more than 150–200 m have, almost invariably, low arsenic concentrations, less than $5 \mu g/L$ and usually less than $0.5 \mu g/L$. Wells from the older Plio-Pleistocene sediments of the Barind and Madhupur Tracts have low arsenic concentrations. The worst-affected areas of Bangladesh are south and east of Dhaka where in some villages more than 90% of the wells have arsenic concentrations above $50 \mu g/L$. The ground waters are predominantly reducing as evidenced by geo-chemical analysis. Arsenic speciation studies have revealed a large range in the relative proportions of dissolved arsenate and arsenite. The modal proportion of arsenite appears to be between 50 and 60% of the total arsenic. Reducing arsenic-rich ground waters from Bangladesh have As^{III}/AsT ratios varying between 0.1 and 0.9 but are typically around 0.5–0.6 [2,4,45,46].

2.4 Arsenic in Different Environmental Media of Bangladesh

Different environmental media information relating to arsenic in the soil, rivers, and foods of Bangladesh are reasonably accessible. However, this is not the case for information on arsenic concentration(s) in air in Bangladesh, which was not able to be retrieved.

2.4.1 Soil

Significantly high concentrations of arsenic have been found in agricultural soil irrigated with arsenic-contaminated ground water. In Bangladesh the average concentration of arsenic in alluvial sand and mud/clay has been reported to be 2.9 mg/kg and 6.5 mg/kg, respectively. Arsenic concentrations varying from 1.5 to 19 mg/kg have been found in Samta village of Jessore, higher concentrations being found in the top layers of the soil. Arsenic concentrations as high as 51 mg/kg and 83 mg/kg have been reported in soils of Faridpur and Comilla districts, respectively [9,47]. Arsenic contents in different depths of soils of Bangladesh have been reported to be less than 10 mg/kg. Table 2–1 shows arsenic contents in soils collected from different areas of Bangladesh and at different depths [48].

2.4.2 River

Arsenic concentrations in water samples from seven rivers in Bangladesh have been reported in the range <0.5-2.7 µg/L but with one sample having an exceptionally high concentration of 29 µg/L. The arsenic content of surface water of these rivers is extremely low and ranges from negligible to 2.25 ppb. The average arsenic content in the sediment from the River Ganges was found to be 2.0 mg/kg (1.2-2.6 mg/kg), that from the Brahmaputra River 2.8 mg/kg (1.4–5.9 mg/kg), and that from the Meghna River 3.5 mg/kg (1.3–5.6 mg/kg). The total arsenic in the sediments of these rivers ranges from 1.021 to 3.525 mg/kg in the pre-flood period (Table 2–2) and from 4.067 to 5.466 mg/kg in the post-flood period (Table 2–3) indicating the transport of arsenic from upstream. Higher levels of arsenic were detected in clay (3.52–6.48 mg/kg) than in silt (1.86–3.94 mg/kg) or sand (1.00–2.98 mg/kg) [2,9,49].

Location	Water As (mg/kg)	Soil Depth (cm)	Soil As (mg/kg)
Sharsha	0.041	0–15	13.67
Sirjdikhan	0.544	15–30	10.66
Alamdanga	0.021	15–30	11.82
	0.021	0–15	16.65
	0,191	0–15	11.91
Meherepur	0.016	15–30	28.22
	0.163	0–15	33.91
Laksham	0.037	15–30	16.97
	0.658	15–30	39.11
	0.729	0–15	18.13
	0.261	0–15	28.0
Chandina	0.160	15–30	19.27
	0.380	15–30	19.27
Sonargaon	0.860	15–30	14.83
	0.860	15–30	13.67
	0.682	0–15	38.93
	0.860	0–15	22.87
Bancharampur	0.092	0–15	17.15
	0.115	0–15	11.32
Nagarkanda	0.064	15–30	26.56
	0.077	0–15	81.25

Table 2–1Arsenic in Water and CorrespondingArsenic in Soils of Different Depths

Source: Huq et al. (2003) [48].

Table 2–2Arsenic Concentrations of Sedimentsfrom Side and Middle of River Bed duringPre-flood Period

ocation	Arsenic Concentration (mg per kg)
nuapur side	3.53
iddle	2.67
neramara	3.16
iddle	1.02
ikshi	1.02
audkandi	3.38
iddle	3.34
ajaria	1.08
	inapur side iddle ieramara iddle kshi audkandi iddle ajaria

Source: Chowdhury et al. (2003) [49].

		Arsenic Concentration (mg per kg)					
River	Location	Suspended Sediment	Riverbed (0–5 cm)	Top Riverbed Core	Bottom Riverbed Core		
Jamuna	Bahadurabad	4.87	4.41	3.20	3.29		
	Jagannathganj	4.07	2.65	2.87	3.54		
	Bhuanpur	4.34	4.17	4.19	3.72		
	Nagarbari	5.09	3.05	2.92	2.68		
Padma	Rajshahi (Kutibari)	4.32	3.65	3.78	3.13		
	Pakshi	4.32	3.40	2.46	5.81		
	Aricha-Paturia	4.34	3.30	3.37	3.20		
	Mawa	4.69	3.91	2.71	4.64		
	Tongibari-Digirpar	5.47	3.01	2.88	4.68		
Meghna	Sohagpur	_	4.31	5.84	3.53		
	Zia fertilizer factory	_	4.73	5.08	2.04		
	Daudkandi	_	4.74	3.31	3.87		
	Chandpur	4.99	3.24	3.80	4.23		
	Heimchar	4.16	3.29	3.33	4.27		

Table 2–3	Arsenic Concentrations in Suspended Sediment and River Bed Sediments
of Padma,	Meghna, and Jamuna Rivers during Post-flood Period

Source: Chowdhury et al. (2003) [49].

2.4.3 Food Chain

Arsenic has been found in foodstuffs as a mixture of inorganic and organoarsenical species. High arsenic content in foodstuffs has been reported from many areas of Bangladesh but the arsenic species in most of the food stuffs are not known. High arsenic content in vegetables and crops was found in the areas receiving arsenic-contaminated irrigation and in areas with arsenic-contaminated soils (Table 2-4). Further, it has been found that the arsenic concentration in crops and vegetables that grow in the Gangetic alluvium is high compared to those growing in the Teesta alluvium [50–55]. Studies conducted in arsenic-contaminated areas found more organic arsenic compared to inorganic arsenic in rice and vegetables. The proportions of inorganic arsenic, monomethylarsonic acid (MMA), and dimethylarsinic acid (DMA) in raw rice were 33.6%, 25.2%, and 41.1%, respectively, and in vegetables 27.9%, 21.5%, and 50.6%, respectively. The average total arsenic concentration in raw rice was 882.2 µg/kg. There are wide variations of arsenic content in vegetables; the high arsenic content vegetables include arum lati (1143.4 μ g/kg), arum leaf (1134.4 μ g/kg), arum root (1130.9 μ g/kg), eggplant (893.3 μ g/kg), etc. and the low arsenic content vegetables include kalmi leaf (111.5 μ g/kg), potato (448.5 μ g/kg), amaranth leaf $(458.2 \mu g/kg)$, etc. (Table 2-5). In arsenic-affected areas of Bangladesh the average daily intake of arsenic through food and drink by an adult was 1017.9 µg of which 54.3% was from rice and vegetables. The remaining 45.7% of arsenic intake was from drinking water. Considering the total intake of inorganic arsenic, rice and vegetables contributed only 27.4% [55]. It is well known that in the human body arsenite is more toxic compared to arsenate [1,7].

		1	tent (mg/kg)			
	No. of	Contamina	ated	Non-contaminated		
Crops	Samples	Range	Mean	Range	Mean	
Tomato	3	0.016-0.049	0.030	0.001-0.025	0.011	
Potato	14	0.000-0.390	0.078	0.000-0.284	0.049	
Brinjal	3	0.042-0.063	0.049	0.028-0.063	0.045	
Lalshakl	4	0.132-0.606	0.321	0.072-0.240	0.163	
Amaranth	8	0.093-2.791	0.605	0.060-0.370	0.168	
Kataua data	4	0.060-0.333	0.226	0.092-0.163	0.120	
Cabbage	3	0.031-0.042	0.06	0.000-0.059	0.047	
Indian spinach	5	0.096-0.387	0.205	0.000-0.228	0.111	
Okra	2	0.034-0.046	0.040	0.016-0.031	0.031	

Table 2–4	Arsenic Contents in Different Crops of Contaminated and
Non-conta	ninated Areas of Bangladesh

Source: Farid et al. (2003) [52].

	Number of	Speciation of	Total Arsonic		
Foodstuffs	Samples	Inorganic	ММА	DMA	(μg/kg)
Raw rice	75	296.3	222.5	363.4	882.2
Amaranth stem	39	166.0	144.4	309.0	619.4
Arum stem	40	229.7	323.1	410.1	962.9
Dhundal	9	214.7	241.7	182.2	638.6
Egg plant	42	252.6	191.6	449.1	893.3
Lady's finger	42	211.6	155.9	318.0	685.5
Papaya (green)	45	177.9	220.5	282.5	680.9
Pumpkin	2	0	331.4	433.5	764.9
Ridge gourd	11	141.2	89.3	486.0	716.5
Snake gourd	22	270.3	121.6	294.7	686.6
Arum lati	34	377.6	232.3	533.5	1143.4
Arum root	16	274.4	149.8	706.7	1130.9
Amaranth leaf	34	39.1	134.3	284.8	458.2
Arum leaf	45	369.3	230.6	534.4	1134.3
Halancha leaf	22	196.0	130.2	348.1	674.3
Indian spinach	73	227.7	157.5	386.0	771.2
Jute leaf	10	212.1	98.4	321.8	632.3
Kalmi leaf	3	62.3	0	49.2	111.5
Pumpkin leaf	26	225.3	193.4	365.9	784.6

Table 2–5Arsenic in Rice and Vegetables from Arsenic-exposed Areas ofBangladesh

Source: Misbahuddin et al. (2007) [55].

2.5 Health Effects of Arsenic Toxicity in Bangladesh

Chronic exposure to arsenic has been linked to adverse health effects in human populations. Arsenic is a known carcinogen and has potentialities of producing cancers at multiple sites, notably in the skin, bladder, kidneys, prostate, and lungs [56–58]. Arsenic is known to have both cancer and non-cancer health effects [56–58]. Many of the health problems related to arsenic exposure, ranging from the classical dermatological signs such as melanosis (Figure 2–1), keratosis (Figures 2–2 and 2–3), and leukomelanosis (Figure 2–4) to respiratory problems, anemia, weakness, conjunctival congestion, diabetes mellitus, hypertension, hepatopathy, peripheral neuropathy, non-pitting edema of lower limbs (Figure 2–5), adverse reproductive outcomes, gangrene (Figure 2–6), Bowen's disease (Figure 2–7), cardiovascular and cerebrovascular diseases, peripheral vascular disease, and skin cancers (Figures 2–8 and 2–9), are already evident among the arsenic-exposed population in Bangladesh [9–19].



FIGURE 2-1 Melanosis.



FIGURE 2–2 Keratosis of palm.

60 HANDBOOK OF ARSENIC TOXICOLOGY



FIGURE 2–3 Keratosis of sole.



FIGURE 2-4 Leukomelanosis.



FIGURE 2–5 Bilateral non-pitting edema.



FIGURE 2–6 Gangrene of foot.



```
FIGURE 2-7 Bowen's disease.
```



FIGURE 2–8 Squamous cell carcinoma of palm.



FIGURE 2-9 Squamous cell carcinoma of scalp.

In Bangladesh, the disease due to chronic arsenic toxicity is known as arsenicosis. For identification and diagnosis of arsenicosis cases the WHO case diagnosis and management protocol is usually followed. The disease is usually first diagnosed on the basis of skin manifestations. According to WHO protocol [59], arsenicosis has been defined as a chronic health condition arising from prolonged ingestion of arsenic above the safe dose for at least 6 months, usually manifested by characteristic skin lesions of melanosis and keratosis, occurring alone or in combination, with or without the involvement of internal organs. Two major diagnostic criteria, namely: (1) the presence of pigmentary and/or keratotic skin lesions, and (2) evidence of exposure to elevated levels of arsenic established by history of intake of arsenic-contaminated water (red tubewell water), or by significant arsenic concentrations in urine, hair or nails, are followed during diagnosis. A reliable history of consuming drinking water with an elevated concentration of arsenic (red tubewell water) for at least 6 months is considered to establish arsenic exposure in humans. In the absence of adequate information regarding a subject's exposure history, the finding of elevated levels of arsenic in a subject's urine, hair or nails is taken as presumptive evidence of elevated arsenic exposure. In Bangladesh, the categorization of arsenicosis patients is carried out on the basis of the WHO algorithm (Figure 2-10). The diagnostic algorithm labels individuals as suspected case, probable case, clinically confirmed case, laboratory confirmed case, and non-arsenic case.

A "suspected case" is a subject who shows characteristic skin lesions or pigmentary changes or keratosis on first presentation and who has not undergone in-depth medical examination.



FIGURE 2-10 Arsenicosis case definition algorithm. Source: WHO (2005) [59].

The classification of "suspected case" is temporary. It is reclassified as "probable," "confirmed" or "non-arsenic" after further clinical examination and/or history of exposure to arsenic.

A "probable case" is a suspected case that has undergone further clinical examination and belongs to one of the two categories below:

- 1. Suspected case showing melanosis and bilateral keratosis involving palms and soles.
- **2.** Suspected case showing melanosis or keratosis after excluding other skin lesions mimicking arsenicosis.

A probable case whose history of exposure to arsenic is not available or in whom tests for arsenic in body tissue are found to be negative, maintains the status of probable case.

A "clinically confirmed case" is a "probable case" in which the presence of other arsenicosis-simulating skin lesions has been ruled out by differential in-depth skin examination by either a trained dermatologist or an arsenic expert.

A probable case classified on the basis of differential skin diagnosis becomes a "laboratory confirmed case" when subsequent laboratory tests for arsenic in water or human tissue prove to be positive.

A "non-arsenic case" is a "suspected" or "probable" case in which the medical specialist finds that the patient's skin condition is due to a cause other than arsenic exposure.

The present case definition algorithm for non-cancerous skin lesions shows acceptable sensitivity (>80%) and specificity (>80%) for prevalent arsenic-associated skin lesions.

In different studies (Table 2–6) it was reported that almost 100% of arsenicosis patients had melanosis. Keratosis was found in about 58.8 to 80% of the reported cases. Other common

Arsenicosis						
Manifestations	<i>n</i> = 110	<i>n</i> = 96	<i>n</i> = 363	<i>n</i> = 250	<i>n</i> = 116	Total
Melanosis	100.0	98.9	99.5	100.0	100.0	99.7
Keratosis	58.8	92.7	68.9	80.0	79.3	75.5
Conjunctival congestion		9.4	15.7	12.0	25.0	15.3
Chronic cough	33.8	25.8	23.7	20.0	26.7	21.8
Weakness	88.2			100.0	93.0	93.7
Pedal edema	4.4		2.8		6.9	4.7
Skin cancer	2.9	1.04	0.8	1.0		1.4

 Table 2–6
 Common Manifestations of Chronic Arsenic Toxicity Patients

Source: [7,14,15,60,61].

Table 2–7	Prevalence of Some Non-communicable Diseases (NCDs) Among the
Arsenic-Ex	posed Population

	Exposed to Arsenic			Unexposed			
NCDs	Yes	No	Total	Yes	No	Total	Statistical Test
Diabetes	21	142	163	25	829	854	CPR 4.4
	12.9%	87.1%	100.0	2.9%	97.1%	100.0	(95%, CI 2.5–7.7)
Hypertension	198	1283	1481	9	105	114	CPR 1.7
	13.4%	86.6%	100.0	7.9%	92.1%	100.0	(95%, Cl 0.8–3.3)
Respiratory	29	65	94	13	111	124	CPR 2.9
effect	30.8%	69.2%	100.0	10.5%	89.5%	100.0	(95%, CI 1.6–5.0)

Source: [7-19].

manifestations were chronic cough (20.0 to 33.8%), conjunctival congestion (9.4 to 25.0%), and pedal edema (2.5 to 6.9%). Pedal edema is a complication of arsenicosis, usually bilateral, and is mainly reported from Bangladesh and West Bengal, India. However, one of the major complications of the arsenicosis patient is weakness and in Bangladesh weakness was found among 88.2 to 100% of arsenicosis patients [1,10,14,15].

Occurrence of non-communicable diseases such as hypertension and diabetes mellitus is higher among the arsenic-exposed population compared to the non-exposed population. Studies conducted on a Bangladesh population had demonstrated increased occurrence of hypertension, diabetes mellitus, and chronic respiratory disorders in terms of chronic cough and asthma among arsenicosis patients compared to those having no exposure to arsenic [17–19,62–65]. Studies revealed (Table 2–7) that the crude prevalence ratio for diabetes mellitus among arsenicosis patients was 4.4 (95%, CI 2.5–7.7). The crude prevalence ratio for hypertension was 1.7 (95%, CI 0.8–3.3) and that for respiratory disorders was 2.9 (95%, CI 1.6–5.0).

A study conducted in Bangladesh compared ECG findings among the arsenicosis, non-arsenicosis, and non-exposed groups of respondents, and revealed that overall, abnormal ECG findings were high (58%) among arsenicosis cases and highly significant

Adverse Pregnancy		Arsenic	
Outcomes	Exposed	Non-exposed	<i>P</i> -value
Abortion	68.8	23.7	<i>z</i> = 2.65, <i>p</i> = 0.008
Still birth	53.1	23.7	<i>z</i> = 2.0, <i>p</i> = 0.046
Pre-term birth	68.8	27.1	z = 2.45, p = 0.014

Table 2–8Adverse Pregnancy Outcomes Among Exposed and
Non-exposed Population

Source: [11].

(p < 0.001). Prolonged corrected QT was found among 46%, 18%, and 8% of the arsenicosis, non-arsenicosis, and non-exposed groups of respondents, respectively, and was significantly different. Abnormal QRS complex was found among 14%, 8%, and 2% of the arsenicosis, non-arsenicosis, and non-exposed groups, respectively [66]. Another study revealed that mortality due to cardiovascular diseases among the arsenic-exposed population was high in Bangladesh: it was found that 43% of total arsenic-related deaths was due to such disease [65].

An extreme manifestation of lower extremity arterial disease (LEAD) is gangrene, which has been reported in a number of studies focusing on the health effects of arsenic in Bangladesh [1,14]. A study found that the prevalence and risk of LEAD was higher among the population whose drinking water source(s) contained arsenic in excess of 0.05 mg/L than among those whose drinking water source(s) did not contain excess arsenic (<0.05 mg/L). An a normal ankle brachial systolic pressure index (ABSPI) was found more frequently in the arsenic-exposed (13.3%) than in the non-exposed (2.5%) group. Further, the study found that the prevalence and risk of LEAD was higher if patients had developed signs of arsenicosis (melanosis \pm keratosis) [67].

For arsenic-exposed mothers, adverse pregnancy outcome has been reported from various parts of the world. In Bangladesh, adverse pregnancy outcome has also been evident in studies [11,68,69]. A study (Table 2–8) that compared pregnancy outcomes in terms of live birth, stillbirth, spontaneous abortion, and pre-term birth among arsenic-exposed and non-exposed women revealed that spontaneous abortion, stillbirth, and pre-term birth rates were significantly higher among those women who had been exposed to arsenic-contaminated water than among those who had not been exposed (p = 0.008, p = 0.046, and p = 0.018, respectively). The participants of the study had been matched for age, socio-economic status, education, and age at marriage [11]. In another study, the odds ratios for spontaneous abortion, stillbirth and neonatal deaths were found to be 2.5 (95% CI 1.5-4.3), 2.5 (1.3-4.9) and 1.8 (0.9-3.6) respectively in mothers whose drinking water arsenic content was greater than 50 µg/L compared to that in mothers whose drinking water arsenic level was 50 µg/L or less [69].

It is known that arsenic is neurotoxic; studies conducted in Bangladesh showed reduced mental development among children aged 1–5 years. In that study 18% of the children who were exposed to arsenic-contaminated water (arsenic levels $>50 \,\mu$ g/L) showed abnormal mental development and reduced intellectual function compared to the children who did not drink

arsenic-contaminated water [70]. Another study conducted among children aged 8–11 years revealed an inverse association of arsenic exposure through drinking water with lower developmental scores of the children [71].

Arsenic is a known carcinogen and has the potential to produce cancers at multiple sites, notably in the skin, bladder, kidneys, prostate, and lungs [56–58]. In Bangladesh, skin cancer, lung cancer, and renal cancer have been reported in different studies [12,72,73]. However, skin cancers were more common and the types of skin cancers are squamous cell cancer and basal cell cancer. Bowen's disease is a precancerous skin lesion and is evident among the arsenicosis patients of Bangladesh [7,12,74]. It is estimated that there will be a doubling of the potential lifetime mortality risk from cancer in Bangladesh due to arsenic in drinking water and it is indicated that the overall lifetime mortality risk due to cancer of the lung, liver, and bladder resulting from such exposure will be 229.6 per 100,000 population [72].

2.6 Epidemiology of Arsenicosis in Bangladesh

Arsenicosis is prevalent in the rural areas of Bangladesh, and is associated with age, sex, socio-economic conditions, and nutritional status of the host. In Bangladesh, most arsenicosis patients are between 20 and 40 years of age. People are being exposed to arsenic mainly through consuming arsenic-contaminated tubewell water. So far, no arsenicosis patient has been found among villagers who consume tubewell water having arsenic levels less than 0.082 mg/L. Arsenicosis has been found to be more common among males (53.7%) in comparison to females (46.3%). Female arsenicosis patients are found to be associated with low concentrations of arsenic in tubewell water. A significantly low dose of arsenic intake was found among females (1.321 mg) compared to males (1.734 mg). These findings indicate that females are more susceptible to the toxic effects of chronic arsenic exposure. Arsenicosis patients are mostly from low socio-economic backgrounds and suffering from malnutrition. The majority of these patients are in a mild or moderate stage, and severe cases or cases with complications are few. Mild and moderate cases are found to improve quickly on cessation of further intake of arsenic-contaminated water and by taking protein and vitamin A, E, C-rich food [7,12,14, 75,76].

2.7 Management of Arsenicosis Patients in Bangladesh

To date there is no specific treatment for chronic arsenic toxicity in humans available in Bangladesh. Consumption of safe water both for drinking and cooking purposes is key to arsenicosis patient management. In Bangladesh, the current practice of patient management continues to be stoppage of further intake of arsenic-contaminated water; increased intake of vitamin A (β -carotene), E, and C through food and medicinal supplement; increased intake of locally available proteins; and application of keratolytic ointment to remove keratotic lesions of the palm and sole [1,59,77]. These measures are helpful in the early recovery from

symptoms of arsenicosis patients. Mild and moderate arsenicosis cases are found to recover from illnesses using the above management regimen. However, severe arsenicosis cases or cases with complications take longer to recover [1,74,77]. Studies carried out on arsenicosis patient management with different treatment regimens such as spirulina, selenium, and folic acid in Bangladesh found that none of these regimens is very effective [61,77,78], However, the vitamin A, E, and C regimen is still commonly used for arsenicosis management in Bangladesh [74].

2.8 Socio-Cultural Aspects of Arsenicosis in Bangladesh

Arsenic contamination in ground water and thereby the occurrence of arsenicosis is not only a major public health problem but leads to many social problems and economic setbacks in rural Bangladesh. Studies found that poor people are more affected than others by arsenicosis. Skin lesions are higher among men compared to women, and women are more socially damaged than men. The majority of rural people believe that arsenicosis is a contagious disease and could spread from person to person through physical contact, or through contact with objects touched or used by individuals with arsenicosis [75,79,80].

In Bangladesh, women suffering from arsenicosis are ignored in society and may be divorced or abandoned, and unmarried women find difficulty getting married. Female arsenicosis patients are less likely to receive treatment and face further problems getting treatment from a healthcare facility. Social and cultural values make it difficult for them to attend to their own health care and travel to service providers. Other concerns for female patients need-ing treatment are long waiting times, discrimination in service delivery, and inadequate separate facilities for female patients [79,80].

Arsenic is a poison; in Bangla, poison is "Bish" and rural people know that ingestion of Bish may have many serious consequences, even death. But ingesting arsenic through drinking water does not cause such acute problems, which creates confusion and ignorance among rural people regarding drinking arsenic-contaminated tubewell water. In a study it was found that almost one-third of the respondents knowingly use arsenic-contaminated tubewell water [80].

Besides social problems, there are some economic consequences of arsenicosis to consider. Many arsenic victims who are ill are found to be too weak to work, lose their jobs, or are refused work because of fears of contagion. In a study, the majority (58.6%) of the respondents said they faced various economic setbacks [79,80].

Regarding access to a healthcare facility for arsenicosis treatment, about half (50.7%) of arsenicosis patients face difficulties getting treatment. On the other hand, the length of time needed to recover from symptoms of arsenicosis lead to a loss of faith in the efficacy of treatment, which extends to laxity in seeking health care. Study findings reveal that a significant proportion (79.9%) of arsenicosis patients are found to access alternative health care, which includes homeopathy, village doctors, and *kabiraj* [79].

2.9 Conclusions

Considering the nature of the problem, arsenic contamination in Bangladesh has been recognized as a major public health issue. Arsenic-contaminated tubewells are widely distributed throughout Bangladesh, except in three hilly districts. About 50 million people are at risk of arsenic toxicity.

Besides common skin manifestations, increased risks of non-communicable diseases (NCDs), adverse pregnancy outcomes, and skin cancers among arsenicosis patients are evident and could become a big public health problem.

The most important concern is increased incidence of cancers due to long-term exposure to low dose arsenic—which could overburden the economy and health system of Bangladesh.

The environmental health crisis stemming from arsenic contamination of ground water is currently one of the world's greatest environmental health crises. Millions of people continue to be exposed to arsenic through water and food. The situation is especially worse among rural populations with poor nutrition. Although the full spectrum of arsenic-related morbidity and mortalities in Bangladesh remains to revealed, lessons learnt from other areas, especially Taiwan, Argentina, and Chile, are ominous. The population in Bangladesh will be confronting the consequences of arsenic contamination in water supply, agriculture, and health sectors for generations to come. There is an urgent need to look for alternative sources of water for irrigation, drinking, and cooking. Surface water could be harnessed and used in agriculture. But to find a source of safe water for cooking and drinking both in terms of chemicals and pathogens would be challenging and needs to be given serious thought. Many arsenic removal technologies and systems to remove pathogens have been field tested in rural Bangladesh, but have yielded limited success. In many cases, they were not only expensive, but were either cumbersome or required regular sustained attention. As for health issues, there is a need to introduce programs to enhance the nutritional status of the rural population and surveillance programs for NCDs and cancers. All exposed individuals, not only those having dermatological signs of arsenicosis, should be brought under the protection of a surveillance program.

References

- Khan AW, Ahmad SA, Sayed MHSU, Hadi SA, Khan MH, Jalil MA, et al. Arsenic contamination in groundwater and its effect on human health with particular reference to Bangladesh. J Prev Soc Med 1997;16(1):65–73.
- [2] BGS. Arsenic contamination of ground water in Bangladesh. Final report vols. 1 & 2. In: Kinniburg DG, Smedley PL, editors. British Geological Survey Report WC/00/19; 2001.
- [3] DGHS. Health bulletin-2012, Chapter 11, non-communicable disease. Bangladesh, Mohakhali, Dhaka-1212: Directorate General of Health Services; 2012.
- [4] BGS MML. Groundwater studies for arsenic contamination in Bangladesh. Phase I: rapid investigation phase. UK: British Geological Survey and Mott MacDonald Ltd; 1999.
- [5] NAMIC. Upazilla wise summary results. Bangladesh arsenic mitigation and water supply project. National Arsenic Mitigation Information Centre (NAMIC); 2004. Available at: <<u>http://www.bamwsp.org</u>/>.
- [6] GOB. Arsenic mitigation in Bangladesh. International workshop report, Dhaka, 14–16 January 2002. Local Government Division, Ministry of LGRD and Co-operatives, Government of Bangladesh; 2002.

- [7] Ahmad SA, Bandaranayake D, Khan AW, Hadi SA, Uddin G, Halim MA. Arsenic contamination in ground water and arsenicosis in Bangladesh. Int J Environ Health Res 1997;7:271–6.
- [8] Dhar RK, Biswas BK, Samanta G, Mandal BK, Choudhury TR, Chanda CR, et al. Groundwater arsenic contamination and sufferings of people in Bangladesh may be the biggest arsenic calamity of the world. Proceedings of International Conference on Arsenic Pollution of Ground Water in Bangladesh: Causes, Effects and Remedies. Dhaka, Bangladesh. 8–12th February 1998.
- [9] Ahmed MF. Arsenic contamination of groundwater in Bangladesh. In: Ahmed MF, editor. Arsenic contamination: Bangladesh perspective. Dhaka: ITN-Bangladesh; 2003. 525pp.
- [10] Maidul AZM, Momin A, Akramullah SM, Ahmed Z, Afsar S, Salahuddin A, et al. Arsenical keratosis (chronic arsenism). Bangladesh J Dermatol Venerol Leprol 1996;13(1):1–4.
- [11] Ahmad SA, Sayed MHSU, Barua S, Khan MH, Faruquee MH, Jalil MA, et al. Arsenic in drinking water and pregnancy outcomes. Environ Health Perspect 2001;109(6):629–31.
- [12] Ahmad SA, Sayed MHSU, Khan MH, Faruquee MH, Jalil MA, Ahmed R. Arsenicosis: neoplastic manifestations of skin. J Prev Soc Med 1998;17(2):110–15.
- [13] Dhar RK, Biswas BK, Samanta G, Mandal BK, Chakraborti D, Roy S, et al. Groundwater arsenic calamity in Bangladesh. Curr Sci 1997;73(1):48–59.
- [14] Ahmad SA, Sayed MHSU, Hadi SA, Faruquee MH, Khan MH, Jalil MA, et al. Arsenicosis in a village in Bangladesh. Int J Environ Health Res 1999;9:187–95.
- [15] Sikder MS, Maidul AZM, Momin A. Clinical manifestation of chronic arsenic toxicity in Bangladesh—A 250 case study. Bangladesh J Dermatol Venerol Leprol 1999;16(1):6–8.
- [16] Tondel M, Rahman M, Magnuson A, Chowdhury IA, Faruquee MH, Ahmad SA. The relationship of arsenic levels in drinking water and the prevalence rate of skin lesions in Bangladesh. Environ Health Perspect 1999;107:727–9.
- [17] Rahman M, Tondel M, Ahmad SA, Chowdhury IA, Faruquee MH, Axelson O. Hypertension and arsenic exposure in Bangladesh. Hypertension 1998;33:74–8.
- [18] Rahman M, Tondel M, Ahmad SA, Alexon A. Diabetes mellitus associated with arsenic exposure in Bangladesh. Am J Epidemiol 1998;148:198–203.
- [19] Milton AH, Hasan Z, Rahman A, Rahman M. Chronic arsenic poisoning and respiratory effects. J Occup Health 2001;43:136–40.
- [20] WHO. Guidelines for drinking-water quality, 2nd ed. Health criteria and other supporting information, vol. 2. Geneva: World Health Organization; 1996.
- [21] Umitsu M. Late Quaternary sedimentary environments and landforms in the Ganges Delta. Sediment Geol 1993;83:177–86.
- [22] Uddin A, Lundberg N. A paleo-Brahmaputra? Subsurface lithofacies analysis of Miocene deltaic sediments in the Himalayan–Bengal system, Bangladesh. Sediment Geol 123:239–54.
- [23] Goodbred Jr. SL, Kuehl SA, Steckler MS, Sarker MH. Controls on facies distribution and stratographic preservation in the Ganges-Brahmaputra delta sequence. Sediment Geol 2003;155:301–16.
- [24] Goodbred Jr. SL, Kuehl SA. The significance of large sediment supply, active tectonism, and eustasy on margin sequence development: late Quaternary stratigraphy and evolution of the Ganges-Brahmaputra delta. Sediment Geol 2000;133:227-48.
- [25] Acharyya SK. Arsenic levels in groundwater from quaternary alluvium in the Ganga Plain and the Bengal Basin, Indian subcontinent: insights into influence of stratigraphy. Gondwana Res 2005;8(1):55–66.
- [26] Mukherjee A, Fryar AE, Thomas WA. Geologic, geomorphic and hydrologic framework and evolution of the Bengal basin, India and Bangladesh. J Asian Earth Sci 2009;34:227–44.
- [27] Alam M, Alam MM, Curray JR, Chowdhury MLR, Gani MR. An overview of the sedimentary geology of the Bengal Basin in relation to the regional tectonic framework and basin-fill history. Sediment Geol 2003;155:179–208.

- [28] Acharyya SK, Shah BA. Groundwater arsenic contamination affecting different geologic domains in India—a review: influence of geological setting, fluvial geomorphology and Quaternary stratigraphy. J Environ Sci Health (Part A) 2007;42:1795–805.
- [29] Ahmed KM, Bhattacharya P, Hasan MA, Akhter SH, Alam SMM, Bhuyian MAH, et al. Arsenic enrichment in groundwater of the alluvial aquifers in Bangladesh: an overview. Appl Geochem 2004;19:181–200.
- [30] Garzanti E, Vezzoli G, Ando S, France Lanord C, Singh SK, Foster G. Sand petrology and focused erosion in collision orogens the Brahmaputra case. Earth Planet Sci Lett 2004;220:157–74.
- [31] Uddin A, Lundberg N. Cenozoic history of the Himalayan-Bengal system: sand composition in the Bengal basin, Bangladesh. Geol Soc Am Bull 1998; April:497–511.
- [32] Acharyya SK, Chackraborty P, Lahiri S, Raymahashay BC, Guha S, Bhowmik A. Arsenic poisoning in the Ganges delta. Nature 1999;401:545–7.
- [33] Nickson R, McArthur J, Ravenscroft P, Burgess WG, Ahmed KM. Mechanism of arsenic poisoning of groundwater in Bangladesh and West Bengal. Appl Geochem 2000;15:403–13.
- [34] Lowry D. Arsenic and other drinking water quality issues, Muzaffargarh District, Pakistan. Appl Geochem 2005;20:55–68.
- [35] Clift PD. Controls on the erosion of Cenozoic Asia and the flux of clastic sediment to the ocean. Earth Planet Sci Lett 2006;241:571–80.
- [36] Mallick S, Rajagopal NR. Groundwater development in the arsenic-affected alluvial belt of West Bengal some questions. Curr Sci 1996;70(11):956–8.
- [37] Chowdhury TR, Basu GK, Mandal BK, Biswas BK, Samanta G, Chowdhury UK, et al. Comment on Nickson et al. 1998, Arsenic poisoning of Bangladesh groundwater. Nature 1999;401:545–6.
- [38] Das D, Chatterjee A, Samanta G, Mandal BK, Chowdhury TR, Chowdhury PP, et al. Arsenic in groundwater in six districts of West Bengal, India: the biggest arsenic calamity in the world. Analyst 1994;119:168–70.
- [39] Das D, Chatterjee A, Samanta G, Mandal BK, Chowdhury TR, Chanda CR, et al. Arsenic in groundwater in six districts of West Bengal, India. Environ Geochem Health 1996;18(1):5–15.
- [40] Nickson RT, McArthur JM, Burgess WG, Ravenscroft P, Ahmed KM, Rahmam M. Arsenic poisoning of Bangladesh groundwater. Nature 1998;395:338.
- [41] McArthur JM, Ravenscroft P, Safiullah S, Thirlwall MF. Arsenic in groundwater: testing pollution mechanisms for sedimentary aquifers in Bangladesh. Water Resour Res 2001;37(1):109–17.
- [42] Mukherjee AB, Bhattacharya P. Arsenic in groundwater in the Bengal Delta Plain: slow poisoning in Bangladesh. Environ Rev 2001;9:189–220.
- [43] Hussain MM, Abdullah SKM. Geological setting of the areas of arsenic safe aquifers Ground Water Task Force Interim Report No. 1. Dhaka: Ministry of Local Government, Rural Development, and Cooperatives Local Government Division. Govt. of Bangladesh; 2001.
- [44] Karim MM. Arsenic in groundwater and health problems in Bangladesh. Water Res 2000;34(1):304–10.
- [45] DPHE/BGS/MML Groundwater studies for arsenic contamination in Bangladesh. Phase I: rapid investigation phase. UK: British Geological Survey and Mott MacDonald Ltd; 1999.
- [46] Acharyya SK. Arsenic in groundwater—geological overview. In: Consultation on arsenic in drinking water and resulting arsenic toxicity in India and Bangladesh. Proceedings WHO Conference, New Delhi; 1997. 12pp.
- [47] Ali MA, Ahmed MF. Environmental chemistry of arsenic. In: Ahmed MF, editor. Arsenic contamination: Bangladesh perspective. Dhaka: ITN-Bangladesh; 2003. pp. 21–41.
- [48] Huq SMI, Rahman A, Sultana N, Naidu R. Extent and severity of arsenic contamination in soils of Bangladesh. In: Feroze AM, Ashraf AM, Adeel Z, editors. Fate of arsenic in the environment. Dhaka: Bangladesh University of Engineering and Technology; 2003. pp. 69–84.
- [49] Chowdhury MAI, Ahmed MF, Ali MA. Influence of upstream sediment on arsenic contamination of groundwater in Bangladesh. In: Feroze AM, Ashraf AM, Adeel Z, editors. Fate of arsenic in the environment. Dhaka: Bangladesh University of Engineering and Technology; 2003. pp. 21–35.

- [50] Huq SMI, Joardar JC, Parvin S, Correll R, Naidu R. Arsenic contamination in food-chain: transfer of arsenic into food materials through groundwater irrigation. J Health Popul Nutr 2006;24(3):305–16.
- [51] Hironaka H, Ahmad SA. Arsenic concentration of rice in Bangladesh. In: Feroze AM, Ashraf AM, Adeel Z, editors. Fate of arsenic in the environment. Dhaka: Bangladesh University of Engineering and Technology; 2003. pp. 123–30.
- [52] Farid ATM, Roy KC, Hossain KM, Sen R. A study of arsenic contaminated irrigation water and its carried over effect on vegetable. In: Feroze AM, Ashraf AM, Adeel Z, editors. Fate of arsenic in the environment. Dhaka: Bangladesh University of Engineering and Technology; 2003. pp. 113–21.
- [53] Williams PN, Price AH, Raab A, Hossain SA, Feldmann J, Meharg AA. Variation in arsenic speciation and concentration in paddy rice related to dietary exposure. Environ Sci Technol 2005;39:5531–40.
- [54] Islam MN, Das BK, Huque ME. Arsenic accumulation in common vegetables from irrigation. J Sci Res 2012;4(3):675–88.
- [55] Misbahuddin M, Anjumanara LAK, Khan MAR, Rahman MS, Khandker S. Speciation of arsenic in rice and vegetables from arsenic exposed areas in Bangladesh. In: Misbahuddin M, editor. Applied research on arsenic in Bangladesh. Dhaka, Bangladesh: Environmental Health Unit World Health Organization; 2007. pp. 43–52.
- [56] WHO. Arsenic and Arsenic Compounds, International Programme for Chemical Safety. Environmental Health Criteria 224, second ed. Geneva: World Health Organization; 2001.
- [57] ATSDR. Toxicological profile for Arsenic. Atlanta, Georgia, USA: U.S. Department of Health and Human Services. Public Health Service Agency for Toxic Substances and Disease Registry, Division of Toxicology and Environmental Medicine/Applied Toxicology Branch; 2005.
- [58] WHO. Arsenic. Environmental health criteria 18. International Programme on Chemical Safety (IPCS). Geneva: World Health Organization; 1981.
- [59] WHO. In: Caussy D, editor. A field guide for detection, management and surveillance of arsenicosis cases. New Delhi, Geneva: SEARO, World Health Organization; 2005.
- [60] Hoque MA, Amin AA, Alam MM, Taher A. Chronic arsenic poisoning in a village of Chapai Nawabganj District—a cross sectional study. Teach Assoc J 1996;9:9–41.
- [61] Sikder MS, Maidul AZM, Khan MAK, Haq MA, Choudhury SAR, Mishbahuddin M. Effect of spirulina in the treatment of chronic arsenicosis. Bangladesh J Dermatol Venereol Leprol 2000;17(1):9–13.
- [62] Islam MR, Khan I, Attia J, Hassan SMN, McEvoy M, D'Este C, et al. Association between hypertension and chronic arsenic exposure in drinking water: a cross-sectional study in Bangladesh. Int J Environ Res Public Health 2005;9:4522–36.
- [63] Nabi AH, Rahman MM, Islam LN. Evaluation of biochemical changes in chronic arsenic poisoning among Bangladeshi patients. Int J Environ Res Public Health 2005;2(3–4):385–93.
- [64] Parvez F, Chen Y, Rauf PWB, Slavkovich V, Islam T, Ahmed A, et al. A prospective study of respiratory symptoms associated with chronic arsenic exposure in Bangladesh: findings from the Health Effects of Arsenic Longitudinal Study (HEALS). Thorax 2010;65:528–33.
- [65] Chen Y, Graziano JH, Parvez F, Liu M, Slavkovich V, Kalra T, et al. Arsenic exposure from drinking water and mortality from cardiovascular disease in Bangladesh: prospective cohort study. BMJ 2011;342:d2431.
- [66] Ahmad SA, Khatun F, Sayed MHSU, Khan MH, Aziz RM, Hossain MZ, et al. ECG findings among the arsenic exposed persons through ground water. J Health Nutr Popul 2006;24(2):221–7.
- [67] Khan MH, Sarker S, Khan N, Sarwar AFM, Ahmad SA. Assessment of low ABSPI among arsenic exposed and non-exposed population: a pilot study. Med Res Counc Bull 2010;36:23–6.
- [68] Milton AH, Smith W, Rahman B, Hasan Z, Kulsum U, Dear K, et al. Chronic arsenic exposure and adverse pregnancy outcomes in Bangladesh. Epidemiology 2005;16(1):82–6.
- [69] Kwok RK, Kaufmann RB, Jakariya M. Arsenic in drinking-water and reproductive health outcomes: a study of participants in the Bangladesh Integrated Nutrition Programme. J Health Popul Nutr 2006;24(2):190–205.

72 HANDBOOK OF ARSENIC TOXICOLOGY

- [70] Akhtar N, Islam AZMM, Mannan MA, Misbahuddin M, Khandker S, Mahmud IA, et al. Evaluation of physical and mental development of children of arsenic exposed areas in Bangladesh: Applied research on arsenic in Bangladesh. Dhaka, Bangladesh: Environmental Health Unit, World Health Organization; 2007. pp. 15–30.
- [71] Parvez F, Wasserman GA, Litvak PF, Liu X, Slavkovich V, Siddique AB, et al. Arsenic exposure and motor function among children in Bangladesh. Environ Health Perspect 2011;119:1665–70.
- [72] Chen Y, Ahsan H. Cancer burden from arsenic in drinking water in Bangladesh. Am J Public Health 2004;94(5):741-4.
- [73] Mostafa MG, Cherry N. Arsenic in drinking water and renal cancers in rural Bangladesh. Occup Environ Med 2013;70(11):768–73.
- [74] Alam M, Hazari SKS, Alam ASMT. Prevalence of skin cancer in chronic arsenicosis in Chittagong Medical College Hospital. J Chitt Med Coll Teach Assoc 2010;21(1):23–9.
- [75] Sultana S, Hossian QZ, Pervin R. Socioeconomic condition and health status of chronic arsenicosis patients in Jessore, Bangladesh. Int J Adv Nutr Health Sci 2012;1(1):9–17.
- [76] Ahmad SA, Sayed MHSU, Faruquee MH, Khan MH, Jalil MA, Ahmed R, et al. Arsenicosis: sex differentials. J Prev Soc Med 1999;18(1):35–40.
- [77] Ahmad SA, Faruquee MH, Sayed MHSU, Khan MH, Jalil MA, Ahmed R, et al. Chronic arsenicosis: management by vitamin A, E, C regimen. J Prev Soc Med 1998;17(1):19–26.
- [78] Momin A, Ali SK, Haque MM. Randomized placebo-controlled trial to evaluate the effectiveness of selenium in the treatment of arsenicosis patients. In: Misbahuddin M, editor. Applied research on arsenic in Bangladesh. Dhaka, Bangladesh: Environmental Health Unit, World Health Organization; 2007. pp. 115–27.
- [79] Ahmad SA, Sayed MHSU, Khan MH, Karim MN, Haque MA, Bhuyian SA, et al. Sociocultural aspects of arsenicosis in Bangladesh: community perspective. J Environ Sci Health 2006;42(12):1942–58.
- [80] APSU. Selected papers on the social aspects of arsenic and arsenic mitigation in Bangladesh. Dhaka: Arsenic Policy Support Unit, Ministry of Local Government Rural Development, and Cooperatives, Government of Bangladesh; 2006.

3

Arsenic and Fluorescent Humic Substances in the Ground Water of Bangladesh: A Public Health Risk

Shafi M. Tareq

DEPARTMENT OF ENVIRONMENTAL SCIENCES, JAHANGIRNAGAR UNIVERSITY, DHAKA, BANGLADESH; SCHOOL OF BIOSCIENCES, THE UNIVERSITY OF NOTTINGHAM MALAYSIA CAMPUS, JALAN BROGA, SELANGOR DARUL EHSAN, MALAYSIA

CHAPTER OUTLINE

3.1	Introduction	73
3.2	Materials and Methods	74
3.3	Results and Discussion	77
3.4	Conclusion	90
Ref	erences	91

3.1 Introduction

Drinking water should be thoroughly purified to ensure the quality of the water and to protect the health of consumers. Consumption of surface water for drinking purposes has been a primary source of waterborne disease in developing countries like Bangladesh. Over the past few decades, ground water has become an important alternative source for the inhabitants of the Ganges-Brahmaputra-Meghna (GBM) floodplains. Unfortunately, much of the water extracted from the alluvial aquifers contains naturally occurring As (arsenic) from the alluvium sediments deposited in these floodplains [1–3]. Arsenic contamination in ground water of deltaic aquifers of Bangladesh represents the most serious public health concern in terms of global population density [3–5]. Resultant health problems were first identified in West Bengal in the 1980s but the first diagnosis in Bangladesh was not made until 1993 [5–7].

In ground water, arsenic is primarily found in its inorganic forms, either As(III) or As(V). Both inorganic forms are toxic for the human body, where in As(V) is reduced to As(III). The mechanisms of causing toxic effects are based on the inhibition of various mitochondrial enzymes by As(III) and the uncoupling of oxidative phosphorylation. The affinity of As(III) for sulfhydryl groups of enzymes and the chemical similarity of As and phosphorus, which allows PO_4^{3-} to be replaced by AsO_4^{3-} , lead to these toxic effects [8]. Several cases of arsenic-related skin lesions were observed in Bangladesh [9]. Since the daily use of groundwater for drinking purposes has become popular in Bangladesh only during the last 20–30 years, it is expected that there will be many more victims of chronic arsenic poisoning in the near future.

Regardless of source, ground water contains dissolved humic substances (HS). HS consists of a complex mixture of organic compounds originating from the degradation of terrestrial and aquatic organisms [10,11]. However, little is known about the actual chemical composition of this material as it consists of a vast number of compounds at very low concentrations, presenting a considerable analytical challenge. Fluorescent HS has been implicated as a contributory factor for blackfoot disease, which is an endemic peripheral vascular disease, occurring on the As-contaminated southwest coast of Taiwan [12–15]. The clinical symptoms include numbness or coldness of one or more of the extremities, resulting in black discoloration, ulceration, or gangrenous changes to the extremities [16–18]. An epidemiological study revealed a causal association between drinking well water and the occurrence of blackfoot disease [18]. In the 1980s, it was reported that the artesian well water in the area of Taiwan contained a high level of fluorescent compounds that induced blackening of the tail and feet of experimental mice [19,20].

Although extensive research has been carried out on arsenic contamination in ground water in Bangladesh by many national and international organizations (e.g., Department of Public Health Engineering (DPHE) Bangladesh; British Geological Survey (BGS), 2001) as well as individual researchers [2,6,9,21,22] during the last decade, the mobilization process of As in the GBM aquifers had still not been properly identified and characterized, and no systematic study had been carried out on fluorescent HS distribution in the ground water in Bangladesh. In addition, as the mobility and poisoning of As in the ground water were controlled by its association with humic substances through different biogeochemical processes [20], it is important to study the linkage between the humic substance properties of ground water and the state of As presence in order to elucidate the mobilization process and health risk. Therefore, the aim of this study was to provide a comprehensive overview of arsenic contamination of ground water and major food composites of Bangladesh, the fluorescence property of ground water, and the qualitative assessment of the health risk of arsenic and fluorescent HS.

3.2 Materials and Methods

3.2.1 Geologic and Demographic Overview

The Ganges, Brahmaputra, and Meghna river systems transport a huge amount of sediments and converge at the lower reaches to form the great delta complex—the GBM delta (Figure 3–1). Bangladesh, located at the head of the Bay of Bengal, occupies most of the Bengal Basin, one of the largest sedimentary basins in the world. The climate is monsoonal humid and tropical, with an average temperature of 27–30°C. The rainy season lasts from May to September. Ground water occurs at very shallow depths all over the country where the major aquifers are the Holocene alluviums and fan deposits and Pliocene fluvio-deltaic (Dupi Tila)



FIGURE 3–1 Geochemical map of arsenic concentration in ground water in Bangladesh. The input data are laboratory analyses from the regional survey by DPHE-BGS [21]. Also shown are dermatological effects of arsenic poisoning in drinking water and locations of three intensive survey areas.

sediments. Water levels lie within a few meters of the ground surface and fluctuate with the annual dry and wet season conditions. The aquifers are recharged during the monsoon season (July–September) when the area receives more than 80% of its annual precipitation (around 2000 mm). Huge amounts of annual flood water standing on a large part of the country for a considerable period also contribute to the recharge process. Annual fluctuations in ground water levels are controlled by the local hydrogeological conditions and withdrawal of ground water for irrigation. In general, the fluctuations are more pronounced in the northwestern part of the country as compared to the coastal plains.

Bangladesh, a developing country, is overburdened with one of the highest population densities, and one of the highest rates of water-related diseases, low per capita income, malnutrition problems, and natural disasters in the world. Bangladesh has 15.6 million people living in a 144,000 km² area. More than 80% of the population depends on agriculture for their livelihood. Historically, clean, uncontaminated water ran through forests and soils that filtered naturally to form clean ground water safe for human and animal use. Nowadays, 90% of rural peoples of Bangladesh depends on ground water for drinking purposes because much of the surface water of Bangladesh is microbially unsafe to drink. Before the 1970s, people usually used surface water (e.g., pond, lake, and river) for drinking and domestic purposes, but severe microbial diseases, especially cholera, were very common at that time. As a poor, developing country, Bangladesh has many health-related problems caused by poverty and insufficient distribution of services, e.g., safe drinking water. As the population grows 1.59% each year, health-related risks increase and the rural/urban gap widens with high health risk from drinking water. Health and social problems caused by environmental issues are heavily impacting on the demography and development.

3.2.2 Sampling

We selected three study areas, namely, Faridpur, Lakshmipur, and Nawabganj, for intensive survey along the side of the main rivers (Figure 3–1). In order to estimate the effect of arsenic on a few other worst-affected districts (Chandpur, Satkhira, Barisal, Pabna, Rajshahi, and Jessore), we have used the data from the DPHE-BGS [21]. Groundwater samples were collected from 100 tubewells in the Faridpur, Lakshmipur, and Nawabganj areas. At least three well bore volumes of ground water were pumped before sampling. Some physicochemical parameters, e.g., pH/Eh, total dissolved solids (TDS), and dissolved oxygen (DO) were also measured *in situ* using portable meters during sampling. Samples for As and other metals analysis were collected in prewashed polyethylene bottles. At each well, bottles were rinsed three times with the respective tubewell water just before sampling. Water was filtered through a Millipore GF/F filter (0.7μ m pore size) prior to acidifying with ultra-pure concentrated HNO₃ to bring the pH to 2. This was done to prevent precipitation of dissolved iron as well as adsorption of trace metal onto the container surface. Samples were stored at 4°C and immediately transported to the laboratory for analysis.

Samples collected for dissolved organic carbon (DOC), fluorescence, and high-performance size exclusion chromatography (HPSEC) analyses were filtered through a pre-combusted Millipore GF/F filter ($0.7 \mu m$ pore size). GF/F filters were combusted for 4 hr at 450°C to remove organic material prior to use. After filtration, samples were kept in the dark and transported in a cooler box and finally stored in a freezer to prevent microbial degradation and/or other changes before analysis. In addition, a number of vegetables as well as rice samples were collected from the study areas in both Aman and Boro seasons (the two main harvesting seasons in Bangladesh) for measuring total As content.

3.2.3 Analysis

Arsenic was determined by graphite furnace atomic absorption spectrometry (GF-AAS), with a subset of samples also analyzed by inductively coupled plasma mass spectrometry (ICP-MS).

Dissolved Fe and Mn were determined by ICP-MS. The instrument was linearly calibrated from 1.0 to $50.0 \mu g/L$ with custom multi-element standard (SPEX Industries) yielding a minimum r^2 of 0.999. The lower detection limit of the instrument was $1 \mu g/L$. All samples were diluted several times to adjust to the operation range and analyzed. Arsenic content in rice samples was measured after digestion. Cation and anion analysis was performed using the ion chromatography (Dionex) technique while HCO_3^- concentration was determined by titration method. DOC was measured by the high temperature catalytic oxidation method (HTCO) with a TOC 5000A (Shimadzu, Japan) using potassium hydrogen phthalate (KHP) as a standard. Samples and standards were acidified to pH 2 with 6 M HCl and sparged with carbon dioxide-free carrier gas for 10 min at a flow rate of 100 mL/min to remove inorganic carbon prior to injection. A non-dispersive infrared detector measured carbon dioxide gas from the combusted carbon. Each sample was injected 4–6 times. The average relative standard deviation for all the samples was 5%.

Fluorescence properties of samples were determined by a spectrofluorometer (F-4500; Hitachi, Tokyo, Japan). Bandwidth was 5 nm for excitation and emission, scanning speed was 1200 nm/min, and scanning range was 225–500 nm for excitation (Ex) and 240–600 nm for emission (Em). Fluorescence data were collected every 5 nm of excitation wavelength and every 2 nm of emission wavelength. The corrections of spectra were performed with rhodamine B solution for excitation wavelength and with a light diffuser for emission wavelength, according to the instrument operation manual. Fluorescence data were obtained in triplicate for each sample and averaged. EEM data were calibrated by normalization to water Raman scattering [23]. To calculate the Raman area of the samples, the Raman signals were corrected by the baseline and integrated over the entire Raman peak for each excitation wavelength. Then, the fluorescence intensities were divided by the Raman area for the corresponding excitation wavelength to obtain a Raman unit (RU, nm⁻¹). The advantages of Raman normalization are suitable for samples with wide ranges of concentration and different properties of DOM [24].

The HPSEC analysis followed the method of Yamada et al. [25] using an HPLC system (Shimadzu LC-10AD) equipped with a column of Superose 12 10/300 GL (Pharmacia Biotech, 10 mm ID-300 mm), and 0.01 M NaOH solution was used as an eluent at a flow rate of 0.40 mL/min. Separated humic molecules were detected by both UV (280 nm) and fluorescence (Fl, Ex: 340 nm, Em: 435 nm) detectors. Natural organic matter (NOM) purchased from the International Humic Substance Society (IHSS) was used as a standard to quantify humic substance carbon. Sodium polystyrene sulfonates (PSS) were used as apparent molecular weight (MW) calibration standards.

3.3 Results and Discussion

3.3.1 Groundwater Quality

The groundwater quality of Bangladesh has been studied extensively during recent years [9,21,22]. Groundwater pH is predominantly near neutral to slightly alkaline (pH 6.5-7.6) with low DO. The Eh values (+0.594 to -0.444V) suggest a mildly oxidizing to moderate/strongly

reducing character of the aquifers. Before the discovery of arsenic contamination, the quality of ground water of the Holocene floodplains was assumed to be generally good [26], although shallow ground water is vulnerable to contamination by bacteria [27]. Iron was known to be a widespread problem, and salinity is a constraint in the shallow aquifers of the coastal area. Subsequently, in addition to arsenic, the DPHE [21] has identified manganese and boron as common, naturally occurring constituents, present in places at concentrations above the World Health Organization (WHO) health-related guidelines for drinking water, 0.5 mg/L in both cases. Ground water is generally of the Ca-HCO₃ or Ca-Mg-HCO₃ type, although Ca-Na-HCO₃ type and Na-Cl type water are also found locally in several parts of the country, mostly near the coast [28]. The ion composition is dominated by HCO_{3}^{-} (169-848 mg/L) and appears to vary according to depth and lithology [28]. There are elevated bicarbonate levels, together with the high dissolved iron and other indications of reducing conditions. Concentrations of SO_4^{-} (<2.50 mg/L) and NO₃⁻ (<0.70 mg/L) are generally low, except for some local variations [21]. Dissolved iron is typically present at around 5-10mg/L. Manganese commonly exceeds 0.50 mg/L. Considerable variability is noted in the levels of total As (range from 0.02 to 2.40 mg/L), total Fe (range from 0.40 to 11.30 mg/L), and Mn (range from 0.01 to 1.86 mg/L) in the groundwater samples as a function of both depth and region. Although As(V) is prevalent in some wells, As(III) is the dominant species, representing about 50-90% of total As in the ground water [28,29].

High HCO_3^- concentrations correlate with the levels of DOC (1.15–14.7 mg/L) in groundwater. DOC levels in the analyzed groundwater indicate distinct trends of variation with both total As and Fe concentrations. Although DOC characteristics in Bangladesh's groundwater are not investigated in detail, organic matter in the Holocene sediments can be considered as an active source for DOC in groundwater [30]. Sulfate and NO_3^- have no significant correlation with As. Microbial degradation of the organic matter in the aquifers results in the reduction of both SO_4^{2-} and NO_3^- , thereby increasing the concentrations of sulfide (2 mg/L) and NH_4^+ (up to 13.2 mg/L) in groundwater [31]. Little is known about the concentrations of pesticides in the ground water in Bangladesh, although they are widely used for agriculture practices, and surface water has been found to contain chlorpyriphos, carbofuran, and carbaryl (commonly used pesticides) at concentrations ranging from 0.54 to 0.89, 0.95 to 1.67, and 0 to 0.19 µg/L, respectively [32].

3.3.2 Arsenic Poisoning in Ground Water

A nationwide survey by DPEH-BGS showed that As concentrations ranged from less than 0.01 mg/L to more than 1.60 mg/L. Figure 3–1 shows clear differences in As concentrations in different parts of Bangladesh with the greatest number of high As wells in the south and south-east of the country. However, superimposed on this regional pattern is considerable well-to-well variability over the scale of a few kilometers. The high As content region in the south and east of Bangladesh is clear from this map. The survey showed that 25% of all the tubewells sampled contain in excess of 0.05 mg/L As, the Bangladesh drinking-water standard. In addition, 9% of the tubewells had As levels exceeding 0.20 mg/L, 1.8% exceeding 0.50 mg/L, and



FIGURE 3–2 Percentage of contaminated wells at different levels of arsenic concentration in ground water in Faridpur, Lakshmipur, Nawabganj, and the whole of Bangladesh.

0.1% exceeding 1.00 mg/L. Few shallow ground waters from the south of the country were arsenic free. They are also typical of the deep aquifer and the water derived from aquifers in the older sediments of the Madhupur and Barind tracts. The mean concentration was about 0.05 mg/L. This value depends to some extent on the concentration of arsenic assumed to be in the large number of wells containing less than the detection limit.

Intensive survey of the three study areas, namely Faridpur, Lakshmipur, and Nawabganj, showed that the level of arsenic in the well with the highest percentage of contamination is in the range of 0.05–0.25 mg/L (Figure 3–2). A comparison of these results with the magnitude of pollution throughout the country by using the DPHE-BGS national database shows a similar pattern of pollution with different orders of magnitude, although the DPHE-BGS country-wide data were not built on intensive survey or based on severely contaminated zones. The majority (76%) of the tubewells yielded arsenic concentrations above the current WHO guideline of 0.01 mg/L and the concentration varied greatly (Faridpur: 0.02–2.25 with an average value of 0.55 mg/L, Lakshmipur: 0.02–2.05 with an average value of 0.35 mg/L, and Nawabganj: 0.06–2.40 mg/L with an average value of 0.62 mg/L) within the present study areas. Arsenic concentrations in 71% of the studied wells in the Faridpur area, 73% in the


FIGURE 3–3 Vertical distributions of arsenic (As) (A) and iron (Fe) (B) in ground water in Faridpur, Lakshmipur, and Nawabganj, Bangladesh.

Lakshmipur area, and 81% in the Nawabganj area exceeded the Bangladesh standard of 0.05 mg/L.

The deeper aquifer water of the study area appears to be less contaminated than the overlying alluvial aquifer with respect to arsenic (Figure 3–3A). Ground water from depths of more than 150 m in all the areas typically had low As concentrations. Water from very shallow handdug wells also had low As concentrations. Variation in Fe concentration with depth was neither systematic nor correlated with As concentration. Iron concentration varied in wide ranges of 2.22–10.3, 0.15–11.3, and 0.01–10.3 for Faridpur, Lakshmipur, and Nawabganj, respectively. The concentration of iron in small number of shallow wells (20–50 m) was very low while water of some of the deep wells nearby contained considerable amounts of Fe in both Lakshmipur and Nawabganj (Figure 3–3B), suggesting Fe occurrence depends on the fluvial-sedimentary history.

Figure 3–4 shows the spatial distribution of arsenic in groundwater in Faridpur, Lakshmipur, and Nawabganj. The spatial distribution of arsenic in the shallow aquifer forms no systematic pattern, although there is some tendency for acute arsenic enrichment within and immediately towards the meander of the rivers. In many areas adjacent to the Lower Meghna Estuary more than 80% of wells exceed the 0.05 mg/L concentration limit. For example, Jakariya and Bhattacharya [33] have reported that 93% of 12,000 wells in Hajiganj Upazila in southeast Bangladesh exceed the limit. The probability of encountering extreme As concentrations, above 0.25 mg/L, is also highest in the south and southeast (Figure 3–1 and Figure 3–4). High concentrations of arsenic are found in the lower catchments of all three



FIGURE 3–4 Spatial distributions of arsenic (As) in ground water in Faridpur, Lakshmipur, and Nawabganj, Bangladesh.

major rivers of the GBM system, indicating the existence of multiple source areas and the likelihood of related mechanisms of mobilization across the whole of Bangladesh. Arsenic speciation showed that the median percentage of As(III) was close to 50% but there was a wide range of As(III) to As(V) ratios and little relationship with other measured parameters, which confirms earlier experience in Bangladesh and West Bengal [2,3,6,7,23]. The more detailed chemical data from the DPHE-BGS report confirm that the waters are anoxic with high dissolved ammonium concentrations in Faridpur and Lakshmipur (but not Nawabganj), and low nitrate concentrations everywhere except where surface pollution was suspected.

3.3.3 Arsenic in Rice and Vegetables

The pattern of arsenic bioaccumulation and its transfer from one trophic level to another is important. The immediate and long-term impact of using As-contaminated water for irrigating paddy soils is a burning concern as As can transfer from water to soil and plants and several studies have proven this phenomenon. Meharg and Rahman [34] predicted that soil arsenic levels could be raised by 1.00 mg/kg per annum due to irrigation with As-contaminated water. Alam and Sattar [35] showed that As contained in soils was positively correlated with As content in water. In the unaffected areas, where irrigation water contained little As (<1 μ g/L), As concentrations of rice field soils ranged from 1.5 to 3.0 mg/kg and did not vary significantly with either depth or sampling time throughout the irrigation period. In the As affected areas where the irrigation water contained elevated As (0.08 to 0.44 g/l), As concentrations of rice field soils were much higher compared to those in the unaffected areas and varied significantly with both depth and sampling time [36].

Arsenic is not an essential element for plants and animals. Food crops such as rice and vegetables can become a route by which As may enter in the food chain, because they can reflect the levels of As that exist in the environment in which they are cultivated (soil and irrigation water). So, the accumulation of As in rice field soil and its introduction to the food chain through uptake by the rice plant and vegetables are of major concern (Figure 3–5).

There are two seasons for rice cultivation: aman and boro. The aman cultivation period is in the rainy season when no irrigation is required but the boro cultivation phase (in the dry season) is completely dependent on irrigation. About 86% of total ground water withdrawn in Bangladesh is utilized in the agricultural sector, especially in rice cultivation in the dry season. A total of 925,152 shallow and 24,718 deep tubewells were used for irrigation during the 2004 dry season [37] and groundwater irrigation covered about 75% of the total irrigated area. It is thought that surface soil of agricultural land accumulates arsenic from contaminated water due to its high affinity with metal oxides/hydroxides in the soil.

Concentrations of arsenic in rice of both amon and boro seasons collected from Faridpur, Lakshmipur, and Nawabganj in Bangladesh were investigated and compared with As concentrations in rice in other regions (Table 3–1). It was observed that the maximum concentrations of arsenic was observed in boro rice in Faridpur ($0.51 \pm 0.07 \text{ mg/kg}$), followed by Lakshmipur ($0.42 \pm 0.08 \text{ mg/kg}$), Satkhira ($0.38 \pm 0.03 \text{ mg/kg}$), and Nawabganj ($0.34 \pm 0.05 \text{ mg/kg}$). Boro season rice contained more As than was found in aman season rice in all regions (Table 3–1). Williams et al. [38] showed that rice obtained from districts with contaminated waters (>0.05 mg/L) had clearly more elevated As levels than those found in rice from less contaminated or uncontaminated (<0.05 mg/L) districts. A high level of As in rice grain (ranging



FIGURE 3–5 Possible routes of arsenic exposure to humans through the water-soil-vegetation-animal food chain in Bangladesh.

Location	Aman Season Rice	Boro Season Rice	As in Soil (mg/kg dry wt)	As in Ground
	(ing/kg dry wt.)			water (ing/L)
Faridpur	0.34 ± 0.06	0.51 ± 0.07	2.53-16.78	0.55
Lakshmipur	0.36 ± 0.04	0.42 ± 0.08	1.24–11.25	0.35
Nawabganj	0.29 <u>+</u> 0.03	0.34 ± 0.05	3.17-18.31	0.18
Barisal ^a	0.16 ± 0.01	0.25 ± 0.06	-	0.09
Chandpur ^a	0.22 ± 0.02	0.28 ± 0.09	-	0.37
Dhaka ^a	0.11 ± 0.02	0.12 ± 0.23	-	0.04
Khulna ^a	0.12 ± 0.01	0.17 ± 0.02	-	0.04
Satkhira ^a	0.36 ± 0.04	0.38 ± 0.03	-	0.13

Table 3–1Arsenic Content in Rice Grain and Corresponding Water and Soil inDifferent Districts of Bangladesh

Data are presented as mean \pm SD (n = 3).

^aData obtained from [37].

between 0.34 and 0.51 mg/kg) in Faridpur and Nawabganj has been found. Duxbury et al. [39] found that arsenic concentrations in rice could varied from 0.01 to 0.42 mg/kg in dry conditions. On the other hand, the study of Abedin et al. [40] showed that no samples of rice grain had arsenic concentrations more than the recommended limit of 1.0 mg/kg in different regions of Bangladesh. Arsenic accumulation in rice grain depends on the variety of rice [38]. Thus, the magnitude of As levels in rice grain is related to the magnitude of As levels in irrigated water and soil as well as variety of rice species. It appears that As present in irrigation water and soil results in higher levels of As in rice plant root, leaf, and stem. A very recent study of Khair [40] on the distribution of As in rice plant showed that the order of As accumulation in the rice plant was root > leaf > grain and they detected levels of As up to $248 \pm 65 \text{ mg/kg}$ in root tissue whereas 1.25 ± 0.23 mg/kg was detected in the grain. It can be predicted that As-contaminated irrigation water could easily increase the As level in rice grain, straw, and other parts of the rice plant. Arsenic contents in boro rice could be 1.3 times higher than for aman rice (Table 3-1). However, accumulation of As by rice largely depends on redox potential in plant and soil phosphate concentration, rhizosphere iron plaque formation, microbial activity, and rice variety [34]. The precise mechanisms controlling the translocation of As to grain are yet to be determined.

Several studies have indicated that the concentration of As in edible parts of most plants is generally low but long-term ingestion of As-enriched rice grain could be dangerous for human health. In Bangladesh, the majority of the residents depend on rice for their caloric intake (about 70% of total), suggesting that rice is an important dietary source of arsenic for the Bangladesh population (Figure 3–5). The average daily rice consumption by an adult in Bangladesh is between 400 and 650 g raw rice grain [39]. Therefore, intake of arsenic from rice and its potential impact on human exposure should not be ignored. Use of contaminated groundwater for drinking and cooking may exacerbate the overall situation.

Arsenic concentrations in 14 different vegetables collected from homestead gardens of some arsenic-prone villages in Bangladesh were investigated and the data are presented in Table 3–2. It was observed that the maximum concentrations of As was observed in string beans (*Vigna sesquipedalis*) (1.26 ± 0.06 and $0.88 \pm 0.0.04 \text{ mgkg}^{-1}$), followed by chilli (*Capsicum melongena*) ($0.87 \pm 0.48 \text{ mgkg}^{-1}$), mint (*Mentha viridis*) (0.56 ± 0.04 and $0.59 \pm 0.07 \text{ mgkg}^{-1}$), and beans (*Lablab niger*) (0.40 ± 0.21 and $0.44 \pm 0.02 \text{ mgkg}^{-1}$), in Lakshmipur and Faridpur areas, respectively; meanwhile, the As concentrations were below the detectable limit (BDL) in okra (*Abelmoschus esculentus*) and bottle gourd (*Lagenaria siceraria*). In light of the current legislation and health considerations, the vegetable products, except for string beans, are safe to consume because the average As concentrations in these vegetables are much lower than the acceptable limit in Bangladesh (1.0 mg/kg). As reported in the literature, the total As contents in vegetable products were <0.004 to 0.303 mg/kg fresh weight [41], which is within the range of values found in the samples in Bangladesh (Table 3–1).

The average arsenic concentration in the vegetables collected from some arsenic-prone areas of Bangladesh was 0.28 mg/kg fresh weight (ranging between 0.25 and 0.38 mg/kg fresh weight), which was higher than that of the United Kingdom, 0.003 mg/kg fresh weight [42]. A Bangladeshi individual, regardless of gender, consumes an average of 130–200 g of vegetables

		Arsenic Content (mg kg ⁻¹ fresh weight)	
Vegetables	Scientific Name	Lakshmipur Area	Faridpur Area
Chilli	Capsicum frutescens	0.22 ± 0.07	0.87 ± 0.48
Sweet gourd (pumpkin)	Cucurbita maxima	0.11 ± 0.01	0.12 ± 0.02
Okra	Abelmoschus esculentus	BDL	-
Red amaranth	Amaranthus gangeticus	_	0.16 ± 0.03
Tomato	Lycopersicon esculentum	0.08 ± 0.01	0.54 ± 0.31
Bottle gourd	Lagenaria siceraria	BDL	BDL
Bean	Lablab niger	0.44 ± 0.02	0.40 ± 0.21
Bitter gourd	Momordica charantia	_	0.37 ± 0.05
Brinjal	Solanum melongena	0.24 ± 0.01	0.26 ± 0.07
Green papaya	Carica papaya	0.08 ± 0.03	-
Mint	Mentha viridis	0.59 ± 0.07	0.56 ± 0.04
Potato	Solanum tuberosum	0.12 ± 0.07	-
Pumpkin leaf	Cucurbita maxima	_	0.41 ± 0.07
String bean	Vigna sesquipedalis	1.26 ± 0.06	0.88 ± 0.04

Table 3–2Arsenic Content in Common Vegetables Collected from Arsenic-proneAreas of Bangladesh

Data are presented as mean \pm SD (n = 3). BDL = below detection limit.

per day (leafy and non-leafy) [43]. Thus, the average dietary intake of total arsenic from vegetables by the inhabitants of arsenic-prone areas in Bangladesh was estimated to be 0.036– 0.056 mg/day. In another study, Rahman et al. [44] reported that the average dietary intake of arsenic from vegetables by the inhabitants of Bangladesh was 0.015 mg/day. However, the recommended daily dietary intake of vegetables is 200 g/person/day, though the availability of vegetables is only about one-fifth of the suggested intake in Bangladesh [43]. If we consider that every person is able to fulfill the recommended amount of vegetables in their daily diets, the estimated average daily dietary intake of arsenic from vegetables in Bangladesh would be 0.056 mg/day. The average intake of total arsenic from vegetables by an adult has been reported as 0.015 mg/day in the Netherlands [45], 0.0592 mg/day in Canada [46], and 0.160– 0.280 mg/day in Japan [47]. From a toxicological point of view, a daily intake of 2 µg of inorganic arsenic per kg body weight should not be exceeded to minimize the health risk [48].

3.3.4 Fluorescence Properties of DOC

All of the groundwater samples collected from three study areas exhibit high fluorescence, implying the presence of fluorescent materials in DOC of groundwater in Bangladesh. The three-dimensional excitation emission matrix (3DEEM) fluorescence spectra of groundwater samples generally exhibited two major fluorescence maxima: humic-like and protein-like fluorescence (Figure 3–6A). The humic-like fluorescence peak occurred at Ex/Em = 335–365/435–480 nm, and the protein-like fluorescence occurred at Ex/Em = 275–290/310–335 nm (Figure 3–6A). The major peaks of the groundwater samples in three study areas were found at Ex/Em = 335-350 nm/440-465 nm with shoulder peaks at Ex/Em = 325-340 nm/425-450 nm



FIGURE 3–6 A representative illustration of 3DEEM fluorescence spectra of ground water from arsenic contaminated (A) and uncontaminated (B) areas of Bangladesh. Fluorescence intensity is expressed in Raman units (nm⁻¹). RU, Raman unit; H, humic-like; P, protein-like.

and 355-365 nm/460-480 nm, respectively. These fluorescence EEMs are typical for humic substances [24]. The EEMs for the majority of the samples show a peak at Ex/Em = 335-350/440-465 nm, which indicates the presence of humic substances derived from soil. The fluorescence spectral patterns of groundwater samples from both As-contaminated and -uncontaminated areas are very similar (Figure 3-6A and B). However, ground water in the As-contaminated area contains both humic-like and protein-like peaks with the highest relative fluorescence intensity. In contrary, ground water from uncontaminated areas has only a humic-like peak with relatively low fluorescence intensity (Figure 3-6B).

Figure 3–7 shows variation in DOC concentration, humic-like peak, and protein-like peak fluorescence intensities of the three study areas. The DOC concentrations in ground water in the three study areas vary from 1.80 to 13.85 mg/L with an average concentration of 6.95 mg/L. All DOC concentrations, humic-like fluorescence intensities, and protein-like fluorescence intensities decrease with the depth of the tubewells. The decrease in DOC concentration during transport has been previously found in other groundwater sites in Bangladesh [49]. The simultaneous increasing DOC concentration and fluorescence intensity in the shallow depth (<50 m) might reflect the infiltration of DOC stored in top soils during the monsoon period. During the monsoon period, comparatively large amounts of less decomposed DOC were washed out from the top soils and infiltrated the shallow aquifer. In addition, temperature was higher in the dry summer event, which may consequently enhance photodegradation and sed-imentation of DOC, thus contributing to the increase in fluorescence intensity.

Humic-like fluorescence and DOC concentration were strongly correlated. It appears that the groundwater DOC was mainly allochthonous, and humic substances were the main



FIGURE 3–7 Vertical distributions of dissolved organic carbon (DOC) (A), humic-like peak (B), and protein-like peak (C) intensity in groundwater in Faridpur, Lakshmipur, and Nawabganj, Bangladesh. Fluorescence intensity is expressed in Raman units (nm⁻¹).

component of DOC. Consistent increases in fluorescence intensity, together with the occurrence of protein-like fluorescence, indicated that autochthonous sources of DOC also infiltrated the shallow aquifer. Protein-like fluorescence materials remained relatively small in the deep aquifer, but occasionally increased in variability (Figure 3–7). Larger variability and higher fluorescence intensities were found in the shallow aquifer. Protein-like fluorescence was often associated with biological activities such as plankton blooms, phytoplankton, and bacteria in aquatic environments [23,50]. The high protein-like fluorescence intensity in the shallow aquifer indicated specific heterotrophic activities in the top section of the groundwater table. The scattered patterns of protein-like fluorescence, as compared to DOC concentration and humic-like fluorescence, imply considerable environmental variability, and indicate that this fluorescing DOC persisted for a period of time in the ground water. However, more detailed sampling and biological investigations are warranted.

The decrease in DOC concentration with depth of well was accompanied by a series of simultaneous changes in DOC quality and biogeochemical process. This may derive from the different remineralization of large molecular weight organic matters (humic substances) from sediment providing the short chain small molecular organic compounds in ground water. The different concentrations of DOC may also be attributed to the precipitation and readsorption of DOC on reactive minerals, strong local variation, dilution, and dispersion of DOC concentration due to groundwater recharge and water flow depending on depth. The microbial

degradation and adsorption by minerals matrix within the aquifers may be responsible for the overall distribution of DOC in ground water.

3.3.5 Molecular Characteristics of Humic Substances

The representative HPSEC chromatograms of humic substances in ground water from both As-contaminated and -uncontaminated areas of Bangladesh, using a fluorescence detector at Ex/Em 350/450 nm and a UV detector at 280 nm, are shown in Figure 3–8. HPSEC separated groundwater humic substances into three completely resolved peaks of different molecular weights with particular retention times. The molecular weight of separated peaks with both the fluorescence and the UV detector were estimated to be 1.81 kDa for peak 1, 1.24 kDa for peak 2, and 0.65 kDa for peak 3 based on the molecular weight calibration of PSS (Figure 3–8). In most of the groundwater samples, 1.24 and 0.65 kDa molecules were predominant, with the presence of 1.81 kDa molecules. On the other hand, surface water in the study areas had 1.24 kDa and 0.65 kDa molecules [49]. One common characteristic of all ground water and surface water was that the largest fractions were almost undetectable, and the predominant fractions were the smallest molecular weight fractions [50].

Relatively high percentages of small molecules in groundwater and surface water DOC indicated that the largest molecular fractions of the DOC were either biologically degraded into smaller molecules or physicochemically adsorbed into soil particles during the recharges of ground water. Degradation of large molecules within the surface by photodegradation and/ or biological degradation during infiltration to ground water might be one alternative explanation for such observation. The small molecular fraction of DOC is high in ground water up to 50 m depth in three study areas, and suggested that this is an indication of infiltration of biologically reactive DOC from surface water.

Among three molecular sizes (1.81, 1.24, and 0.65 kDa), the small molecular weight fraction (0.65 kDa) of DOC was the only variable showing significant correlation with As in ground water. Similar results have already been observed in other As-contaminated areas [49]. Temporal changes in the concentration of As were generally related to variations in the availability of low molecular weight fractions (peak 3 of Figure 3–8) of DOC in ground water. This might indicate a potential binding of As with high fluorescence active low molecular weight DOC components, and low molecular weight fluorescent compounds of DOC in the ground water of Bangladesh may be a serious threat to public health.

3.3.6 Health Risks

In Bangladesh, the inhabitants of As-contaminated areas consume a large amount of groundwater (4–6 L/day) as drinking water. Considering that, on average, 5.0 L of water is consumed per day, people who are drinking water containing 0.20 mg/L of arsenic, i.e., 1.0 mg of arsenic per day for a few years, are vulnerable to multiple health hazards [48]. The DPHE-BGS [21] estimated that about 20 million people in Bangladesh are exposed to As concentrations >0.05 mg/L. The long-term consumption of water above that level is harmful and may cause multiple health hazards including damage to the kidneys, liver, and bladder. Anawar



FIGURE 3–8 Representative HPSEC of both As-contaminated (A) and -uncontaminated (B) ground water in Bangladesh for both the UV and the fluorescence detector.

et al. [9] estimated that about 20% of the total population are drinking As-contaminated water above a 0.2 mg/L level and nearly 3.58 million people are potentially affected and at high risk of health hazards. The present study shows that half of the total population of these three As-contaminated districts were drinking As-contaminated water in the ranges of 0.05–0.25 mg/L (Figure 3–2) and nearly 2.25 million people in these districts are at high risk of health hazards.

There are some instances of patients with skin lesions in Taiwan and Chile who are drinking water containing very low concentrations of As [5,18] but of high humic substances [5]. These results demonstrate that arsenic toxicity may be aggravated by humic substances. There is a compelling reason to suspect that the presence of fluorescent humic substances is related to the cause of skin lesions in Bangladesh because most of the As-contaminated samples have fluorescence characteristics (ranging from 0.20 to $5.2 \text{ RU} \text{ nm}^{-1}$ with an average value of $2.5 \text{ RU} \text{ nm}^{-1}$). It is possible that high As concentrations combined with other trace or minor elements in some cases may enhance the toxicity of these humic compounds to biological systems. High fluorescence intensities and elevated concentrations of fluorescent humic substances are characterized as phenolic and phenolic carboxylic polymer structures containing both -COOH and -OH as their main functional groups [20]. Both arsenic and fluorescent substances are found in high concentrations in Bangladesh ground water. Therefore, it is highly possible that HS-metal complexes in the tubewell water play an important role in the pathogenesis of arsenocosis [15,18].

The synergistic effects of arsenic and fluorescent humic substances in ground waters with respect to biological toxicity are not clear yet. The groundwater samples revealed high fluorescence intensities accompanied by elevated levels of DOC in water, which are possible compounding factors or cause synergistic effects. Animal model experiments found some significant relationships between the intensity of blue-green fluorescence and the occurrence of disease. The fluorescent compounds present in well water can induce some toxic effects and crippling, phlegmasia, ulceration, necrosis, and gangrene in the extremities in mice [20]. The complexes composed of humic substance and arsenic show enhanced inhibition of plasma activity as compared with either humic substance or arsenic alone [51]. Both arsenic and fluorescent substances are found in high concentrations in ground water in Bangladesh. Therefore, it is probable that humic substances and a few more governing factors may raise the toxicity of arsenic in ground water.

3.4 Conclusions

The results of the present study show that ground water from the three study areas is characterized by neutral to slightly alkaline pH, low DO, high DOC, and high concentrations of As and fluorescence humic substances. Arsenic concentrations in ground water are highly variable (0.02-2.40 mg/L), and mostly exceed the maximum permissible limits of the WHO (0.01 mg/L)and Bangladesh (0.05 mg/L) for drinking water. The most contaminated aquifers in the three study areas are confined to a depth of around 20 to 50 m. Elevated As concentrations are related to reducing conditions in a shallow aquifer environment with high DOC and HCO_3^- . Drinking As-contaminated water is the main cause of arsenic poisoning in the public health of Bangladesh. Arsenic might be enriched in rice and vegetables due to using As-contaminated ground water for irrigation. Food composites are a major source of arsenic exposure in Bangladesh and elsewhere in regions with subsistence rice and vegetable diets, which can affect human health through the water-soil-vegetation-animal food chain. The bioavailability of arsenic in rice and vegetables must be addressed to understand the importance of arsenic exposure from the food source.

In the past, only arsenic concentrations in ground water were considered as having a direct correlation with the epidemical degrees of arsenicosis including different types of dermatological diseases, but the results of this investigation indicate that fluorescent humic substances may also have some correlation. Groundwater samples in Bangladesh have relatively high fluorescence intensities of humic substances. Both fluorescent compounds and arsenic in well waters therefore have a direct correlation with the epidemical degree of arsenicosis; fluorescent compounds are a form of humic substance, i.e., a polymer with a multiplicity of anions, having a strong chelating ability, which may therefore bind many positively charged metal cations and other compounds. In the process of chelation, arsenic in well water can combine with fluorescent compounds to form a complex. The evidence of a strong correlation between fluorescence intensity of small molecular size fractions and arsenic in ground water supports the above inference [49]. Therefore, mitigation efforts must be undertaken to provide safe drinking water and these efforts should not be limited to arsenic, which is unquestionably the most significant health risk factor; they should also address fluorescent humic substances.

References

- Nickson R, McArthur JM, Burgress W, Ahmed KM, Ravenscroft P, Rahman M. Arsenic poisoning of Bangladesh groundwater. Nature 1998;395:338.
- [2] Safiullah S, Khannal DP, Tareq SM, Khan MMK. Arsenic mobilization in groundwater: analysis and leaching experiments on aquifer soils of Bangladesh & Nepal. In: Ittekot V, Subramanian V, Annadurai S, editors. Biogeochemistry of rivers in Tropical South and Southeast Asia. Germany: Hamburg University; 1999. pp. 125–30. UNEP/SCOPE Sonderband-Helf-82.
- [3] Acharyya SK, Chakraborty P, Lahiri S, Raymahashay BC, Guhu S, Bhowmik A. Arsenic poisoning in the Ganges delta. Nature 1999;401:545.
- [4] Smith AH, Lingas EO, Rahman M. Contamination of drinking-water by arsenic in Bangladesh: a public health emergency. Bull WHO 2000;78:1093–103.
- [5] Smith AH, Lopipero PA, Bates MN, Steinmaus CM. Arsenic epidemiology and drinking water standards. Science 2002;296:2145–6.
- [6] Chakraborti D, Sengupta MK, Rahman MM, Ahamed S, Chowdhury UK, Hossain MA, et al. Groundwater arsenic contamination and its health effects in the Ganga-Meghna-Brahmaputra plain. J Environ Monit 2004;6:75–83.
- [7] Ravenscroft P, Burgess WG, Ahmed KM, Burren M, Perrin J. Arsenic in groundwater of the Bengal Basin, Bangladesh: distribution, field relations, and hydrogeological setting. Hydrogeol J 2005;13:727–51.
- [8] Scott N, Hatlelid KM, Mackenzie NE, Carter DE. Reactions of arsenic(III) and arsenic(V) species with glutathione. Chem Res Toxicol 1993;6(1):102–6.
- [9] Anawar HM, Akai J, Mostofa KMG, Safiullah S, Tareq SM. Arsenic poisoning in groundwater health risk and geochemical sources in Bangladesh. Environ Int 2002;27:597–604.
- [10] Thurman EM, Malcolm RL. Preparative isolation of aquatic humic substances. Environ Sci Technol 1981;15:463-6.
- [11] Artinger R, Buckau G, Geyer S, Fritz P, Wolf M, Kim JI. Characterization of groundwater humic substances: influence of sedimentary organic carbon. Appl Geochem 2000;15:97–116.
- [12] Yang YW. The redox reaction of arsenic in the groundwater of the Blackfoot disease area in the Chia-Nan plain, southwestern Taiwan. MS thesis, Department of Earth Sciences, National Cheng Kung University, Taiwan; 2006.
- [13] Lu K-L, Liu C-W, Jang C-S. Using multivariate statistical methods to assess the groundwater quality in an arsenic-contaminated area of South-western Taiwan. Environ Monit Assess 2012;184:6071–85.
- [14] Tseng WP. Effects and dose-response relationships of skin cancer and blackfoot disease with arsenic. Environ Health Perspect 1977;19:109–19.
- [15] Yu HS, Sheu HM, Ko SS, Chiang LC, Chien CH, Lin SM, et al. Studies on blackfoot disease and chronic arsenism in southern Taiwan: with special reference to skin lesions and fluorescent substances. J Dermatol 1984;11:361–70.
- [16] Tseng CH. Blackfoot disease and arsenic: a never-ending story. J Environ Sci Health C 2005;23:55-74.
- [17] Tseng CH, Chong CK, Tseng CP, Centeno JA. Blackfoot disease in Taiwan: its link with inorganic arsenic exposure from drinking water. Ambio 2007;36:82–4.
- [18] Lu FJ. Review: fluorescent humic substance and Blackfoot disease in Taiwan. Appl Organomet Chem 1990;4:191–5.

- [19] Lu FJ, Yamamura Y, Yamauchi H. Studies on fluorescent compounds in water of a well in blackfoot disease endemic areas in Taiwan. J Formosan Med Assoc 1988;87:66–75.
- [20] Lu FJ, Huang TS, Lin YS, Pang VF, Lin SU. Peripheral vasculopathy in rats induced by humic acids. Appl Organomet Chem 1994;8:223–8.
- [21] DPHE Report. Groundwater studies for arsenic contamination in Bangladesh, Final Reports, Rapid investigation phase, Dhaka: Department of Public Health Engineering, Government of Bangladesh, Mott MacDonald and British Geological Survey; 1999.
- [22] Safiullah S. Arsenic pollution in the groundwater in Bangladesh: an overview. Asian J Water Environ Pollut 2006;4:47–59.
- [23] Determann S, Reuter R, Wagner P, Willkomm R. Fluorescent matter in the eastern Atlantic Ocean. Part 1: method of measurement and near-surface distribution. Deep Sea Res I 1994;41:659–75.
- [24] Yoshioka T, Mostofa KMG, Konohira E, Tanoue E, Hayakawa H, Takahashi M, et al. Distribution and characteristics of molecular size fractions of freshwater dissolved organic matter in watershed environments: its implication to degradation. Limnology 2007;8:29–44.
- [25] Yamada E, Doi K, Okano K, Fuse Y. Simultaneous determinations of the concentration and molecular weight of humic substances in environmental water by gel chromatography with a fluorescence detector. Anal Sci 2000;16:125–9.
- [26] Davies J. The hydrochemistry of alluvial aquifers in central Bangladesh. In: Nash H, McCall GJH, editors. Groundwater quality. London: Chapman and Hall; 1995. pp. 9–18.
- [27] Hoque BA. Biological contamination of tubewell water. Environmental Health Programme Report, International Centre for Diarrhoeal Disease Research, Bangladesh. Report for Department for International Development (UK); 1998.
- [28] Bhattacharya P, Jacks G, Ahmed KM, Khan AA, Routh J. Arsenic in groundwater of the Bengal Delta Plain aquifers in Bangladesh. Bull Environ Contam Toxicol 2002;69:538–45.
- [29] Tareq SM, Safiullah S, Anawar HM, Rahman MM, Ishizuka T. Arsenic pollution in groundwater: a selforganizing complex geochemical process in the deltaic sedimentary environment, Bangladesh. Sci Total Environ 2003;313:213–26.
- [30] Routh J, Grossman E, Murphy EM, Benner R. Characterization and origin of dissolved organic carbon in Yegua groundwater in Brazos County, Texas. Ground Water 2001;39:760–7.
- [31] McArthur JM, Ravencroft P, Safiullah S, Thirlwall MF. Arsenic in groundwater: testing pollution mechanism for sedimentary aquifers in Bangladesh. Water Resour Res 2001;37:109–17.
- [32] Chowdhury AZ, Jahan SA, Islam MN, Moniruzzaman M, Alam MK, Zaman MA, et al. Occurrence of organophosphorus and carbamate pesticide residues in surface water samples from the Rangpur district of Bangladesh. Bull Environ Contam Toxicol 2012;89:202–7.
- [33] Jakariya M, Bhattacharya P. Use of GIS in local level participatory plans to mitigate arsenic poisoning: a case study from Matlab upazila, Bangladesh. J Environ Sci Health Part A 2006;42:1933–44.
- [34] Meharg AA, Rahman M. Arsenic contamination of Bangladesh paddy field soils: implications for rice contribution to arsenic consumption. Environ Sci Technol 2003;37:229–34.
- [35] Alam MB, Sattar MA. Assessment of arsenic contamination in soils and waters in some areas of Bangladesh. Water Sci Technol 2000;42:185–93.
- [36] Saha GC, Ali MA. Dynamics of arsenic in agricultural soils irrigated with arsenic contaminated groundwater in Bangladesh. Sci Total Environ 2007;379:180–9.
- [37] BADC (Bangladesh Agricultural Development Corporation). Survey report on irrigation equipment and irrigated area in boro/2003 season. Survey and monitoring project for development of minor irrigation. Sech Bhaban, Dhaka, Bangladesh: BADC; 2003.

- [38] Williams PN, Islam MR, Adomokao EE, Raab A, Hossain SA, Zhu YG, et al. Increase in rice grain arsenic for regions of Bangladesh irrigating paddies with elevated arsenic in groundwater. Environ Sci Technol 2006;40:4903–8.
- [39] Duxbury JM, Mayer AB, Lauren JG, Hassan N. Arsenic content of rice in Bangladesh and impacts on rice productivity. Proceedings of 4th Annual Conference on Arsenic Contamination in Groundwater in Bangladesh, Dhaka, Bangladesh; 2002.
- [40] Abedin MJ, Malcolm S, Andy A, Cotter-Howells J. Arsenic accumulation and metabolism in rice (Oryza sativa). Environ Sci Technol 2002;36:962–8.
- [41] Schoof RA, Yost LJ, Eickhoff J, Crecelius EA, Cragin DW, Meacher DM, et al. A market basket survey of inorganic arsenic in food. Food Chem Toxicol 1999;37:839–46.
- [42] Ministry of Agriculture, Fisheries and Food. Total Diet Study: aluminium, arsenic, cadmium, chromium, copper, lead, mercury, nickel, selenium, tin and zinc. Food Surveillance Information Sheet, No. 191. HMSO, London, UK: Ministry of Agriculture, Fisheries and Food; 1997.
- [43] Khair A. Vegetable. In: Islam S, editor. Banglapedia: National Encyclopedia of Bangladesh. Dhaka: Asiatic Society of Bangladesh; 2006.
- [44] Rahman IMM, Uddin MN, Hasan MT. Assimilation of arsenic into edible plants grown in soil irrigated with contaminated groundwater. In: Bundschuh J, Armienta MA, Birkle P, Bhattacharya P, Matschullat J, Mukherjee AB, editors. Natural arsenic in groundwater of Latin America. London, UK: Taylor & Francis; 2008. pp. 351–8.
- [45] De Vos RH, Dokkum WV, Olthof PDA, Quirijns JK, Muys T, Van der Poll JM. Pesticides and other chemical residues in Dutch total diet samples (June 1976-July 1978). Food Chem Toxicol 1984;22:11–21.
- [46] Dabeka RW, McKenzie AD, Lacroix GM, Cleroux C, Bowe S, Graham RA, et al. Survey of arsenic in total diet food composites and estimation of the dietary intake of arsenic by Canadian adults and children. J AOAC Int 1993;76:14–25.
- [47] Tsuda T, Inoue T, Kojima M, Aoki S. Market basket and duplicate portion estimation of dietary intakes of cadmium, mercury, arsenic, copper, manganese, and zinc by Japanese adults. J AOAC Int 1995;78:1363–8.
- [48] WHO. Inorganic arsenic compounds other than arsine: health and safety guide. Health and Safety Guide; no. 70. Geneva: World Health Organization; 1992.
- [49] Tareq SM, Maruo M, Ohta K. Characteristics and role of groundwater dissolved organic matter in arsenic mobilization and poisoning in Bangladesh. Phys Chem Earth 2013;58–60:77–84.
- [50] Coble PG, Schultz CA, Mopper K. Fluorescence and contouring analysis of DOC intercalibration experiment samples: a comparison of techniques. Mar Chem 1993;41:173–8.
- [51] Hseu YC, Chang WC, Yang HL. Inhibition of human plasma activity using humic acids with arsenic. Sci Total Environ 2001;273:93–9.

This page intentionally left blank



Arsenic Risk Assessment

Santhini Ramasamy¹ and Janice S. Lee²

¹OFFICE OF SCIENCE AND TECHNOLOGY, OFFICE OF WATER, US ENVIRONMENTAL PROTECTION AGENCY, WASHINGTON DC, USA, ²NATIONAL CENTER FOR ENVIRONMENTAL ASSESSMENT, OFFICE OF RESEARCH AND DEVELOPMENT, US ENVIRONMENTAL PROTECTION AGENCY, RESEARCH TRIANGLE PARK, NORTH CAROLINA, USA

CHAPTER OUTLINE

4.1	Introduction	95
4.2	Arsenic Chemistry	97
4.3	Arsenic Occurrence and Exposure	97
4.4	Hazard Identification	98
4.5	Arsenic Metabolism, Mode of Action, and Physiologically Based Pharmacokinetic	
	Modeling	103
4.6	Potential Sources of Susceptibility	106
4.7	Dose–Response Approaches	108
4.8	Risk Characterization	108
Disc	laimer	111
Refe	rences	111

4.1 Introduction

Risk assessment is an important scientific tool for informing risk management policy decisions made by regulatory agencies such as the United States Environmental Protection Agency (USEPA), the Food and Drug Administration (FDA), the Consumer Products Safety Commission (CPSC), and the United States Department of Agriculture (USDA). Industries and other entities might also apply this tool when evaluating products to support their registration and approval before marketing and other regulatory decisions. Advisory groups and organizations such as the World Health Organization (WHO) might use this tool in guiding other countries in chemical safety. Below, we first present the general principles of risk assessment as background information before discussing the specifics of arsenic risk assessment. This chapter focuses mainly on human health assessment of arsenic compounds.

The scope and focus of a risk assessment depends on the specific needs of various organizations. Risk assessments are context specific. For example, a risk assessment might be used to inform emergency response needs, to derive safe drinking water goals and to set safe air exposure levels, to help establish food tolerance levels, or to provide guidance on chemical or microbial contaminant safety.

The risk assessment process consists of the following four components: hazard identification, dose-response assessment, exposure assessment, and risk characterization. The execution of these steps depends on the needs in scoping and problem formulation, of various groups to inform risk management decisions. In hazard identification, whether a particular chemical is causally inked to particular health effects is determined. The dose-response assessment determines the relationship between the magnitude of exposure and the probability that health effects will occur. The dose-response assessment can be based on an apical endpoint or a precursor event mediating the particular health outcome. This component typically culminates in the derivation of a toxicity value such as reference dose or cancer slope factor based on the hazard identification and characterization and on the mode of action conclusions. The associated mode of action in mediating different endpoints or health effects is supported by toxicokinetic information (e.g., absorption, metabolism, distribution, and excretion). In the exposure assessment, the exposure to humans from different sources (e.g., drinking water, food, air) and by different routes (oral, inhalation, dermal) is estimated. In recent years, biomonitoring data such as chemical or metabolite concentrations in urine or blood have been used because they enable chemical exposures to humans to be ascertained at the individual level and thus serve as biomarkers of exposure. In the final step of risk assessment, risks to humans from chemical exposures are characterized by combining reference values with estimates of exposure. In the risk characterization step, various assumptions used in estimating the human health risk are described. The influence of factors such as age, gender, life stage susceptibility (e.g., childhood, pregnancy), and genetic polymorphism, and the impact of lifestyle factors such as smoking and socioeconomic status, is examined. In the risk characterization, the pros and cons of various risk assessment options are provided to risk managers to inform their decision making. In addition gaps in research data are identified that could provide a rationale for development of new data to address uncertainties and to help understand the variability of risk estimates for human populations from exposure to chemical contaminants. The extent to which each of the four components of risk assessment is developed depends on the availability of resources (time, money, expertise) and the needs of the organization.

In this chapter, the authors do not conduct a risk assessment, Instead, they present arsenicrelated literature that is pertinent to each component of risk assessment. The chapter presents discussion on arsenic chemistry and metabolism, occurrence and exposure, hazard identification, mode of action risk characterization and susceptibility. The authors also do not present a dose-response assessment but rather review reference levels from various state, federal, and international regulatory and advisory entities. Additionally, this chapter focuses primarily on inorganic arsenic but, does present some information on other arsenicals such as (monomethyl arsenic (MMA) and dimethyl arsenic (DMA)) and arsine. Finally, the chapter highlights the oral route of exposure although reference values also are provided for inhalation exposures of arsenic compounds.

This chapter is not intended to be an extensive review of the literature on various health risk aspects posed by arsenic. References provided at the end of this chapter and the material presented in other chapters serve as useful resources for in-depth information on specific arsenic-mediated health outcomes. Unless otherwise specified, "arsenic" as used in this chapter refers to inorganic arsenic.

4.2 Arsenic Chemistry

Inorganic arsenic is a naturally occurring element. Arsenic exists in different oxidation states (-3, 0, +3, +5). Arsine (-3) exists as gas. In drinking water, arsenic exists as arsenite (+3) or arsenate (+5) depending upon the oxidation conditions.

4.3 Arsenic Occurrence and Exposure

Arsenic is widely distributed in Earth's crust, which contains about 3.4 ppm arsenic. Because arsenic is naturally occurring, low levels are found naturally in all environmental media. Inorganic arsenic and organic arsenic compounds also have been introduced into the environment through anthropogenic use (e.g., pesticides, pharmaceuticals, glassmaking, and semiconductors) [1–3]. Arsenic has been identified in at least 1149 of the 1684 hazardous waste sites and is included in the EPA National Priorities List (NPL). Mean total arsenic concentrations range from 0.02 to 4 ng/m^3 in remote and rural areas and from 3 to 200 ng/m^3 in urban areas. Near industrial sources such as metal smelters and coal-burning power plants, concentrations greater than 1000 ng/m^3 have been reported. The concentration of arsenic in ambient water sources is less than $10 \mu \text{g/L}$, although it can be as high as 5 mg/L near anthropogenic sources [2,4].

The primary route of arsenic exposure for the general human population is via the ingestion of contaminated water or food. When the arsenic exposure via drinking water is low, the contamination of arsenic from food sources (e.g., rice) becomes a significant contributor to the total arsenic exposure in humans. Exposure from air through inhalation also has been reported in occupational settings and could contribute to total arsenic exposure [1–3]. High levels of arsenic contamination in drinking water have been found in southwestern Taiwan, Argentina, and Chile (generally >100 μ g/L). Populations from Bangladesh, India/West Bengal, Mexico, northeastern Taiwan, and other countries have been exposed to arsenic in drinking water sources over a wide range of arsenic levels (up to several hundred μ g/L or ppb). In the United States, the average drinking water concentration is about 2 μ g/L, although higher levels have been reported in the west, midwest, and New England [2].

Arsenic in food can be inorganic and organic. Organic arsenic forms are less toxic than inorganic arsenic. For example, the arsenic in rice is primarily inorganic, and in seafood it is mainly organic (e.g., arsenobetaine and arsenocholine). Total arsenic concentrations can be substantially higher in certain seafood; however, approximately 10% of seafood arsenic is inorganic arsenic [5]. Inorganic arsenic is found in many foods, at concentrations that usually range from 20 to 140 µg/kg [2]. Analyzing a large number of food commodities and drinking water, the EFSA (European Food Safety Authority) estimated the range of average inorganic arsenic intake as to be 0.13 to $0.56 \mu g/kg$ bw/day and the 95th percentile intake as to be 0.37 to $1.22 \mu g/kg$ bw/day. Because 98% of occurrence data was reported as total arsenic, several assumptions were made in estimating lower and upper bound ranges for inorganic arsenic intake. High consumers of rice in Europe, such as certain ethnic groups, have a daily dietary intake of inorganic arsenic of about $1 \mu g/kg bw/day$. High consumers of algae-based products have a daily dietary intake of inorganic arsenic of about $4 \mu g/kg bw/day$. For children under 3 years of age, inorganic arsenic intake was reported as two- to three-fold higher compared to adults [6].

Arsenic exposure from drinking water are based on measured concentrations in the water and estimations of the amount of water consumed. Obtaining accurate estimates of human exposure due to this source is difficult because arsenic concentrations in drinking water can change over time and the individuals might not remember the actual amount of water consumed. Biological monitoring provides a better estimate of the absorbed dose in arsenicexposed individuals. Among the various exposure biomarkers, arsenic levels in the urine are commonly reported in epidemiological investigations and provide information on very recent arsenic exposure (i.e., in the past few days). Some studies report arsenic measurements in nail and hair, and although interpreting these levels poses some challenges due to external contamination; however, the presence of arsenic in these samples indicates longer exposure to arsenic-weeks or several months. Such biomarkers help ascertain individual exposures but, they do not distinguish exposure due to drinking water or food or other sources of exposure. Evidence from the United States National Health and Nutrition Examination Survey (NHANES) suggests that direct exposures in food (e.g., from pesticide residues) or from seafood arsenicals and their metabolites can confound measurements of urinary dimethylated arsenic [7]. Urinary monomethylated arsenic therefore might be a preferable marker of inorganic arsenic exposure via drinking water.

4.4 Hazard Identification

4.4.1 Arsenite and Arsenate

Acute or subacute arsenic exposure at high levels (greater than several milligrams of inorganic arsenic per day) might induce overt gastrointestinal disturbances ranging from mild abdominal cramping and diarrhea to severe life-threatening hemorrhagic gastroenteritis associated with shock. These symptoms are absent under chronic exposure conditions [5].

Chronic exposure to inorganic arsenic is related to many adverse health effects. The most commonly reported are skin lesions, cancers in multiple tissues, cardiovascular diseases, diabetes, pulmonary effects, neurological impairment, and developmental and reproductive toxicity. Other effects reported in cohorts exposed to arsenic include immune effects and hematological and hepatic effects.

4.4.1.1 Skin Lesions

Numerous epidemiological studies conducted in arsenic-contaminated areas around the globe have reported skin lesions. The skin lesions that arsenic exposure causes are one of the earliest detectable health effects in exposed populations and occur prior to other endpoints

such as cancers and cardiovascular diseases. These skin lesions are distinct and characterized by hyperpigmentation and hyperkeratosis; they are observed in individuals exposed to arsenic in drinking water or via inhalation, or in patients treated with Fowler's solution (potassium arsenite) [5]. The hyperpigmentation commonly appears in a finely freckled, "raindrop" pattern that is particularly pronounced on the trunk and extremities but that might also involve mucous membranes such as tongue or buccal mucosa [5]. Arsenic-induced hyperkeratosis is manifested predominantly on the palms and plantar aspects of the feet [5]. Skin lesions are also reported in young children as old as 2 years. Evidence suggests these skin symptoms could serve as markers of susceptibility to other disease outcomes such as cancer and cardiovascular diseases [5]. The dose-dependent association between low dose levels of arsenic and incidence of skin lesions was reported in prospective cohort subjects (HEALS, Health Effects of Arsenic Longitudinal Study) in Bangladesh [8,9]. Increased monomethylarsonic acid (MMA) concentrations in urine are associated with increased risk for skin lesions [10,11].

4.4.1.2 Cancer Effects

The International Agency for Research on Cancer (IARC) first reported in 1987 that there was sufficient evidence of arsenic carcinogenicity in humans 1973. In a recent evaluation, IARC concluded that the evidence in humans is sufficient to indicate that inorganic arsenic causes lung, bladder, and skin cancers. IARC found that the evidence of cancers in other tissues cannot rule out the possibility of chance or bias; the agency concluded, however, that there is a positive association between exposure to arsenic and cancer of kidney, liver, and prostate [12].

The carcinogenic role of arsenic compounds was first noted over 200 years ago by J. Hutchinson when patients treated with arsenicals (Fowler's solution) developed an unusual number of skin tumors. Later evidence from patients treated with arsenicals exhibited cancers in internal tissues (e.g., lung, bladder). Evidence from occupational exposure studies in workers exposed to arsenic in mining and smelting industries also indicated cancers in respiratory tissues. In the 1980s, evidence from several environmental exposure studies in which populations were exposed to arsenic from drinking water for many years revealed increased mortality from cancers in multiple tissue sites.

Many epidemiological studies (ecological, case–control, cohort, cross-sectional) evaluating the association between inorganic arsenic exposure and carcinogenic effects have been conducted in populations exposed to arsenic in drinking water. The earlier ecological studies conducted in southwestern Taiwan, an endemic area for arsenic, reported a prevalence of skin lesions and skin cancers in populations exposed to arsenic over a lifetime mostly at high levels (several hundred µg/L arsenic in drinking water) [13,14]. The skin cancers caused by inorganic arsenic exposure were identified as non-melanoma type (basal cell carcinoma and squamous cell carcinoma). Ecological and cohort studies of these populations subsequently reported increased mortality from internal tissue cancers (lung, liver, bladder, kidney, and other tissues) [15–18]. Similar observations were also made in populations exposed to inorganic arsenic from drinking water sources at high levels in Chile [19], Argentina [20,21], and Japan [22]. In a 12-year follow-up cohort study in northeastern Taiwan, Chen et al. [23,24] found increased risk for lung and urinary tract cancers in populations exposed to arsenic over a wide range of exposure levels including levels <100 µg/L.

Over the last 15 years, epidemiological studies have attempted to evaluate the low dose inorganic arsenic and the occurrence of internal tissue cancers in the United States [25–34], Europe [35–37], and other places such as Chile and Bangladesh [38,39]. The results from these studies are inconsistent. Small sample size, lack of adequate exposure period (a long latency period is needed for development of cancer), and exposure misclassification are some aspects that challenge the ability to detect risk at low arsenic exposures [40,41].

Some evidence suggests that chronic inorganic arsenic exposure produces lung tumors in mice [42,43] and evidence is growing that tumor formation occurs in mice upon short-term arsenic exposure during critical periods such as gestation. For example, *in utero* exposure to arsenic during gestation alone has been shown to produce tumors in liver, lung, adrenal gland, and other tissues of the offspring after 2 years [44]. The development of tumors in rodents gestationally exposed to inorganic arsenic informs findings in epidemiological studies. For example, using Chilean cohorts, Smith et al. [45] demonstrated that early-life arsenic exposure is associated with increased risk for lung cancers. The likelihood of susceptibility to inorganic arsenic has been shown to enhance the carcinogenic effects of other agents such as smoking in humans, UV light and nitrosamines in animal models [3].

4.4.1.3 Diabetes

The National Toxicology Program conducted a comprehensive review of epidemiological and animal evidence on arsenic and diabetes. The review found that the available human data provide limited to sufficient evidence for an association between arsenic and diabetes in populations with relatively high exposure levels in drinking water (\geq 150 ppb), but insufficient evidence at lower exposure levels in drinking water (<150 ppb). The findings from animal literature investigating the link between arsenic and diabetes are mostly inconclusive [46]. Nevertheless, some human studies that have more fully characterized the exposures and outcomes support an association between moderate arsenic exposure and diabetes [47–49].

4.4.1.4 Cardiovascular Effects

Acute exposure to inorganic arsenic in the range of milligram to grams per day has caused overt cardiovascular manifestations, including hypotension, congestive heart failure, and cardiac arrhythmias (prolonged QT interval).

Many epidemiological studies have demonstrated a dose-responsive association between chronic arsenic exposure in drinking water and various cardiovascular diseases. These cardiovascular disorders include carotid atherosclerosis, impaired microcirculation, prolonged QT intervals and increased QT dispersion in electrocardiography, and clinical outcomes such as hypertension, blackfoot disease, coronary artery disease, and cerebral infarction [50]. Peripheral vascular disease (thromboangiitis, and overt gangrene of the extremities) was noted in German vintners from consumption of arsenic-contaminated wine substitutes [5].

In a systematic review of 13 epidemiological studies (8 from Taiwan, 3 from the United States, and 1 each in Hungary and Spain), increased relative risk for coronary artery disease, stroke, and peripheral arterial disease was reported at high exposure levels [51]. In a subsequent systematic review of 18 additional studies conducted in Taiwan, the United States,

Spain, Japan, Bangladesh, Chile, Inner Mongolia, Pakistan, and Slovakia, the authors found conclusive evidence for a causal association between chronic arsenic exposure and clinical cardiovascular endpoints at high arsenic levels and inconclusive evidence at low-to-moderate arsenic exposures [52].

From meta-analysis of many epidemiological studies, the evidence relating environmental arsenic exposure to hypertension or increased blood pressure is positive in endemic areas and is limited in low exposure regions [53]. The evidence linking arsenic and hypertension in another meta-analysis was inconclusive [54]. The studies included in the two analyses and the methodologies used, however, differed.

4.4.1.5 Neurological Effects

Acute inorganic arsenic poisoning producing initial gastrointestinal or cardiovascular symptoms can be followed by both central nervous system and peripheral nervous system effects. Central nervous system effects occur within 1–5 days of acute poisoning and can range from headache and mild confusion to florid encephalopathy, seizures, and coma. Peripheral neuropathy effects emerge within 1–4 weeks. In subacute or chronic exposure, arsenic occasionally can result in peripheral neuropathy without gastrointestinal or cardiovascular symptoms. The occurrence of peripheral neuropathy is inconsistent in individuals chronically exposed to arsenic in drinking water [5]. The neurological effect of arsenic is characterized by symmetrical sensorimotor neuropathy are paresthesias, numbness, and pain. In addition, the affected individuals have reduced nerve conduction velocities, typical of those seen in axonal degeneration [55]. Auditory, visual, and somatosensory impairment have been reported in Mongolian farmers living in the Yellow River Valley, where drinking water is contaminated by arsenic. The arsenic-exposed subjects in Inner Mongolia exhibited reduced pinprick scores and vibration thresholds at arsenic exposure concentrations well below the 1 ppm drinking water level [56].

In addition to acute neurotoxicity and peripheral neuropathy, exposure to inorganic arsenic during brain development might have adverse health outcomes. For example, decreased IQ in children exposed to arsenic has been reported in studies conducted in Araihazar, Bangladesh, and West Bengal, India [57,58]. Hamadani et al. [59] found that pre- or postnatal arsenic exposure did not affect child development (problem solving and motor development) at 18 months in Matlab, Bangladesh. However, Hamadani et al. [60] demonstrated associations of early-life arsenic exposure and decreased verbal IQ and full scale IQ in preschool-aged children (Matlab). In a systematic review, Rodriguez-Barranco et al. [61] reported evidence for an association between increased arsenic levels in urine and decreased intelligence quotient (IQ) in children aged 5–15 years.

Neurobehavioral impairments are also reported in the animal studies at high dose levels [62]. Perinatal exposure at low dose levels, however, also impairs learning and behavior in mice [63].

4.4.1.6 Pulmonary Effects

Studies from India and Bangladesh report increased respiratory symptoms (e.g., cough, shortness of breath, bronchitis) upon exposure to arsenic [64,65]. Impaired pulmonary

function, chronic obstructive pulmonary disease (COPD), and tuberculosis were observed in populations from Bangladesh, West Bengal/India and Chile [45,64,66–70]. In Chile, Dauphine et al. [71] reported that early-life arsenic exposure was associated with pulmonary dysfunction (decreased forced expiratory volume, lower forced vital capacity, and increased breathlessness) in adult subjects but these effects were reported nearly four decades after the peak arsenic exposure. However, urinary arsenic exposure levels in NHANES subjects from the USA did not correlate with prevalence of respiratory diseases (asthma, chronic bronchitis, emphysema) [72].

4.4.1.7 Kidney Effects

Chronic exposure to arsenic in drinking water-from low to high exposure levels- has been shown to increase the risk for kidney effects (e.g., chronic glomerulonephritis, renal failure, decreased glomerular filtration rate) [73–76].

4.4.1.8 Immune Effects

Epidemiological, animal, and *in vitro* studies all support an association between arsenic exposure and suppression of the immune system [3]. Evidence suggests that maternal arsenic exposure impacts the thymus weight and increased the respiratory infection incidence in infants. A positive association between arsenic exposure and pro-inflammatory cytokine levels has been shown in humans. Increased levels of autoimmune markers (antinuclear antibodies, antidouble stranded DNA) in the serum of arsenic-exposed individuals in West Bengal/India suggest that arsenic at moderate levels may induce autoimmune diseases such as rheumatoid arthritis [77].

4.4.1.9 Other Effects

Increased levels of biomarkers of hepatic damage such as aminotransferases, aspartate transaminase, alanine transaminase, alkaline phosphatase, gamma-glutamyl transpeptidase, and bilirubin were reported in individuals exposed to arsenic. Increased liver enzyme levels such as those of serum alkaline phosphatase, aspartate transaminase, and alanine transaminase were associated with increased arsenic levels in drinking water in Bangladesh [78]. Das et al. [77] found a similar increase in liver enzymes and elevated bilirubin levels in people who lived in arsenic-contaminated areas compared to people in non-exposed areas in West Bengal/ India. Acute and chronic arsenic poisoning might result in anemia, leukopenia, and thrombocytopenia [5]. A linear relationship between total urinary porphyrin concentration and urinary arsenic has been observed in a limited number of human cohorts [5].

4.4.1.10 Developmental and Reproduction Effects

Inorganic arsenic and its methylated metabolites can pass through the placenta in humans [79,80] and mice [81]. Some studies have found that arsenic induces several effects in the fetus, some of which result in fetal loss or growth retardation [88,89]. A growing body of evidence

indicates that some of the changes induced in fetal or infant life lead to detectable adverse health effects later in childhood as well as in adult life [82].

Studies from arsenic endemic areas such as Bangladesh, West Bengal, Chile, and northeastern Taiwan have reported many reproductive and developmental effects including increased risks of spontaneous abortions, stillbirths, infant mortality, preterm birth, low birth weight, and growth restriction [83–92].

4.4.2 Hazard Identification—DMA and MMA

The acid and salt forms of pentavalent monomethylated arsenic (MMA^V) and pentavalent dimethylated arsenic (DMA^V) are used as herbicides. In the environment, the organoarsenicals can be converted to inorganic arsenic, which is more toxic to humans [93]. Further, ingested organoarsenicals are eliminated more rapidly than ingested inorganic arsenic. Evidence from studies of animals administered MMA^V suggests that the large intestine is the target organ. Diarrhea and vomiting were observed in dogs chronically exposed to MMA. No evidence of carcinogenicity following exposure to MMA^V has been observed in rodent bioassays. EPA's Office of Pesticide Programs considers MMA as "not likely" to be a human carcinogen [93]. The thyroid and bladder have been identified as the target organs for DMA. Thyroid lesions, primarily incidence of cuboidal to columnar epithelial cells lining thyroid follicles, were observed in rats upon subchronic and chronic exposures. Long-term exposure to DMA^V has resulted in urinary bladder tumors in male rats but not in female rats or mice [94]. Chronic exposure to MMA^V produced no tumor effects [95]. IARC [12] concluded that there is sufficient evidence of carcinogenicity of DMA^V in experimental animals. More recently, however, IARC classified DMA as "possibly carcinogenic to humans" (Group 2B) [172].

4.4.3 Hazard Identification—Arsine

The predominant effects of arsine poisoning shown in clinical studies are intravascular hemolysis and subsequent passage of dark or bloody urine and jaundice. This hemolytic anemia is accompanied by low hemoglobin values and the presence of reticulocytosis, red cell fragments, and ghost cells. Early death after acute arsine poisoning could result from myocardial failure accompanied by massive pulmonary edema that precedes the onset of renal failure. Animal studies show decreased red cell count and hemoglobin level correlated with increasing arsine concentrations.

4.5 Arsenic Metabolism, Mode of Action, and Physiologically Based Pharmacokinetic Modeling

After entering the body, pentavalent arsenic (As^V) can be reduced to trivalent arsenic (As^{III}) which can then proceed through a series of methylation and conjugation reactions, some of which involve reoxidation of arsenic to As^V . The traditional metabolic pathways proposed

for arsenic are shown in Figure 4–1. Compared to the pentavalent species, the trivalent species are more cytotoxic, more genotoxic, and more potent inhibitors of enzyme activity [96]. In this metabolic scheme less toxic species (i.e., As^V, MMA^V, and DMA^V) can be converted to the more toxic species (i.e., As^{III}, MMA^{III}, and DMA^{III}). The predominant metabolite excreted in humans and animals is DMA^V, but some animal species (e.g., rats) further metabolize DMA^V through DMA^{III} to trimethylarsine oxide (TMAO).

Methylation is an important factor affecting arsenic tissue distribution and excretion. Humans and most experimental animal models methylate inorganic arsenic to MMA and DMA, with the amounts differing across species, as determined by analysis of urinary metabolites. Historically, the methylated metabolites have been considered less acutely toxic, less reactive with tissue constituents, less cytotoxic, and more readily excreted in the urine than inorganic arsenic [98–105]. The trivalent species MMA^{III} and DMA^{III}, however, are more cytotoxic in a human liver cell line called Chang cells [106,107], Chinese hamster ovary cells [108], and cultured primary rat hepatocytes [109,110] than As^{III}, As^V, MMA^V, or DMA^V.

A small percent of DMA^{III} can be methylated further to TMAO in mice and hamsters (see [111] for a review). TMAO can be detected in urine following DMA exposure, but has not been detected in the blood or tissues of mice exposed intravenously to DMA [112] or in the urine of mammals exposed orally to inorganic arsenic. The only data available on human metabolism of DMA to TMAO are from a study with a single human volunteer who ingested DMA and excreted 3.5% of the dose as TMAO [111]. Minimal to no detection of TMAO in the urine of animal models or humans could be due to rapid clearance of DMA and MMA from cells [113]; most analytical methods, however, are not optimized for the detection of TMAO.



FIGURE 4-1 Traditional metabolic pathways for arsenic [97].

The major route of excretion for most arsenic compounds, including TMAO, by humans is via the urine [114–117].

The US EPA defines the term mode of action (MOA) as "a sequence of key events and processes, starting with the interaction of an agent with a cell, proceeding through operational and anatomical changes, and resulting in cancer formation or other adverse outcomes" [118]. MOA data can inform the human relevance of data in animal or other model systems, the identification of susceptible populations, and dose-response assessment. Arsenic has been observed to affect numerous key events, but no specific MOA has been identified for arsenicmediated health effects. Several key events could be common for both cancer and non-cancer endpoints, for example, gene expression changes, oxidative stress, cytotoxicity, signal transduction, interference with hormone function, apoptosis, cell cycle arrest, and enzyme activity inhibition.

Elevated levels of reactive oxygen species (ROS) have been measured as arsenic is metabolized and could be involved in its toxicity; however, the specific mechanism of ROS production is unknown [119]. Arsenic has been observed to produce the following reactive species: superoxide (O_2^{-}) , peroxyl radicals (ROO'), nitric oxide (NO), hydrogen peroxide (H₂O₂), dimethyl arsenic peroxyl radicals [(CH₃)₂AsOO'], and dimethylarsinic radical [(CH₃)₂As']. ROS have been linked to vascular diseases [120,121], diabetes [122,123], and neurological effects [124,125]. Changes in genes and proteins also have been identified as key events relevant in the development of skin lesions [126–128], vascular diseases [129–136], and reproductive effects [137]. Changes in cell proliferation and apoptosis have been linked to skin lesions and effects on the immune response [138–142].

PBPK models for inorganic arsenic are central to developing a biologically based doseresponse (BBDR) models which has proven challenging because inorganic arsenic appears to mediate its toxicity through a range of metabolites, the roles of which with regard to specific adverse effects are not clear [143]. Mann et al. [144] developed a PBPK model for exposure to inorganic arsenic (orally, intravenously, and intratracheally) was developed in hamsters and rabbits. The same group [145] then extended the PBPK model for use in humans by adjusting physiological parameters (organ weights, blood flows) and reestimating absorption and metabolic rate constants. Yu [146,147] also developed a PBPK model for arsenic in humans that includes tissue compartments for lung, skin, fat, muscle, combined kidney and richly perfused tissues, liver, intestine, gastrointestinal and stomach contents, and bile. Gentry et al. [148] adapted the model proposed by Mann et al. [144] to different mouse strains by adjusting physiological parameters (organ weights and perfusion rates).

In 2007, Clewell et al. [143] noted that these PBPK models did not incorporate the most recent available information on arsenic methylation kinetics and suggested several steps for improving them. El-Masri and Kenyon [149] developed a PBPK model incorporating some of the improvements that Clewell et al. [143] suggested (although not the simulation of changes in gene expression). The model predicts the levels of inorganic arsenic and its metabolites in human tissues and urine following oral exposure to As^V, As^{III}, and oral exposure to organo-arsenical pesticides. The model consists of interconnecting submodels for inorganic arsenic (As^{III} and As^V), MMA^V, and DMA^V. Reduction of MMA^V and DMA^V to their trivalent forms

also is modeled. The submodels address the gastrointestinal tract (lumen and tissue), lung, liver, kidney, muscle, skin, heart, and brain, with reduction of MMA^V and DMA^V to their trivalent forms modeled as occurring in the lung, liver, and kidney. The model also uses non-competitive inhibition to incorporate the inhibitory effects of As^{III} on the methylation of MMA^{III} to DMA and of MMA^{III} on the methylation of As^{III} to MMA. This model differs from the other models described above because it provides an updated description of metabolism using biochemical data on the mechanism of arsenic methylation. This model also is an improvement over previous models because it can quantitatively assess impacts of parameter variability arising from genetic polymorphism. It uses *in vitro* studies to estimate most of the model parameters (statistically optimizing those that are sensitive to urinary excretion levels to avoid problems with parameter identifiability) and it can predict the formation and excretion of trivalent methylated arsenicals. The partition coefficients estimated in the model are comparable to those Yu. [147] developed Model performance was tested against limited human data on urinary excretion; the model should be evaluated in large numbers of subjects for its ability to predict the tissue and urinary concentrations of arsenicals.

4.6 Potential Sources of Susceptibility

Studies have found that individuals with a slower secondary methylation capacity for arsenic (i.e., MMA to DMA), as measured by low urinary DMA concentration or high urinary MMA concentration, have an increased risk of disease [10,150–158]. Children are another subpopulation that is more susceptible to adverse health effects following arsenic exposures. Although children are generally exposed to arsenic through the same sources as adults, behavior and physiology of children might result in higher absorbed doses relative to body weight than adults for a given set of exposure conditions. Because children tend to eat less varied foods than adults, exposure to contaminated food, juice, or infant formula prepared with contaminated water could result in higher relative doses than in adults. Children are more likely to ingest arsenic-contaminated soil through hand-to-mouth activity. In addition, exposure occurs *in utero* likely at concentrations similar to those occurring in the mothers. Hall et al. [80] measured levels of MMA, DMA, and inorganic arsenic in maternal blood and cord blood and found the levels were similar, indicating that children are likely exposed at similar levels as adults during embryonic development.

Limited data are available on the relative efficiency of absorption of arsenic from the gastrointestinal tract of children compared to adults, but measurement of urinary arsenic levels in children indicates that absorption does occur. Data are equivocal on the capacity of children to metabolize arsenic compared to adults.

Sex might present another susceptibility factor for arsenic. Specifically, differences in methylation patterns have been noted between men and women in several studies. Higher MMA:DMA ratios have been observed in men compared to women in a variety of populations tested, including in the USA [20,153,159,160], Taiwan [154], and Bangladesh [150]. Loffredo et al. [161] observed that gender differences (ages were not specified, but subjects were presumably adults) in arsenic methylation varied across populations studied in Mexico, China, and Chile, and sometimes by exposure level.

Age and reproductive status also could affect the male-female differences in arsenic methylation patterns. Concha et al. [162] demonstrated that pregnant women in their third trimester excrete approximately 90% of arsenic as DMA. Engström et al. [163] also observed pregnant women to have an increased proportion of DMA in their urine compared to nonpregnant women in the same population, with increases occurring with gestational age. Lindberg et al. [164] also observed possible hormonal effects on arsenic methylation, noting that women less than 60 years of age, likely premenopausal, generally were more efficient in methylating arsenic than men of the same age, while the difference narrowed considerably in men and women older than 60 years. Lindberg et al. [165] observed that although females of all ages generally methylated arsenic at a higher rate than males, the greatest disparity between the sexes occurred between the ages of 20 and 55 years. Lindberg et al. [164] also observed that selenium, body mass index, and AS3MT (arsenic methyl transferase) gene polymorphism affected the observed proportions of methylated urinary arsenic metabolites in men only. The pattern of arsenic methylation was also altered in men with mutations in one allele of the methylenetetrahydrofolate reductase (MTHFR) gene, but in women, variants in both alleles were required [164].

Although evidence suggests that age and sex play a role in differences in arsenic methylation capacity, data from Bangladesh analyzed by Lindberg et al. [165] suggest that genetic polymorphism is the most important factor affecting the methylation of inorganic arsenic. According to Lindberg et al. [165] only 30% of variation in methylation patterns is attributable to the level of arsenic exposure, gender, and age. Polymorphisms G12390C, C14215T, and A35991G in the AS3MT gene were associated with an increase in urinary DMA. Polymorphisms TT/AA in the MTHFR gene caused an increase in urinary levels of inorganic arsenic and MMA and increased the risk for skin lesions. Polymorphisms in GST (glutathione S-transferase) were associated with different methylation patterns. GST polymorphisms were not associated with risk for skin lesions, but were associated with changes in risk for carotid atherosclerosis. Data in *Drosophila* by Ortiz et al. [166] suggest that genetic variations in the glutathione synthetase gene also might be related to interindividual variability to arsenic.

Nutritional deficiencies could also be related to changes in arsenic methylation and in some cases to increased risk of specific diseases. The risks associated with nutritional deficiencies are not as strong as other risk factors such as gender or genetic polymorphism. Therefore, even though nutritional deficiencies might add to the risk of developing arsenic-induced diseases, their role is minimal compared to the arsenic exposure.

Cigarette smokers, both current and former, have been observed to have a significantly decreased secondary methylation capacity, resulting in increased urinary MMA (p < 0.01) and decreased DMA concentrations (p < 0.05) [151]. Steinmaus et al. [159] observed that in a population in Argentina, the proportion of excreted MMA was associated with bladder cancer risk in former smokers, but not in individuals who had never smoked. Subjects who had smoked and had proportions of MMA in the upper tertile had a two-fold elevated risk (i.e., OR = 2.13; 95% CI = 1.03-4.40) of bladder cancer compared to subjects with proportions of MMA in the lower two tertiles. Therefore, the conclusion was that individuals who smoke had an increased susceptibility to arsenic toxicity. Lindberg et al. [167] also observed a relationship between impaired arsenic methylation in male smokers compared to non-smokers and increased risk

of skin lesions (population included all individuals older than 4 years, and one category examined effects in individuals 18 years old or younger).

Lindberg et al. [168] reported that the higher prevalence of arsenic-related skin lesions in men was the result of less efficient methylation in men. Melkonian et al. [169] looked at whether an interaction between cigarette smoking and arsenic exposure occurred in the risk for skin lesions in a population in Bangladesh as part of HEALS. These researchers [169] found an additive effect between smoking status and level of arsenic exposure.

4.7 Dose–Response Approaches

A dose-response assessment examines the mathematical relationship between exposure and effects. The outcome of the dose-response assessment can help in the derivation of reference and guideline values. As noted in the introduction, a dose-response assessment is outside the scope of this chapter, but regulatory levels and the reference toxicity values determined for inorganic and organoarsenical compounds by various government agencies and organizations are listed in a subsequent section.

4.8 Risk Characterization

The goal of risk characterization is to summarize and integrate information from the preceding steps of the risk assessment to synthesize an overall conclusion about risk including information on uncertainty and variability. To account for uncertainty and variability, reference values are calculated by applying uncertainty factors to the point of departure. Areas of variability or uncertainty include human variation, animal-to-human extrapolation, adverse-effect level to no-observed-adverse-effect level, subchronic to chronic exposure, and incomplete databases.

4.8.1 Reference Values and Regulatory Standards for Arsenical Compounds

4.8.1.1 US EPA

4.8.1.1.1 INORGANIC ARSENIC

The maximum contaminant level (MCL) is the highest level of a contaminant that is allowed in drinking water based on cost benefit analysis and is enforceable. The maximum contaminant level goal (MCLG) is non-enforceable level at which no known or anticipated adverse effects on the health of persons occur and which allows adequate margin of safety. The US EPA published an MCL of 0.01 mg/L and an MCLG of zero for arsenic on January 22, 2001 [170]. The MCL for arsenic was set at the level at which lung and bladder cancer reduction benefits were maximized at a cost that was justified by the benefits.

The reference dose (RfD) is an estimate of daily oral exposure to the human population that is likely to be without an appreciable risk of deleterious effects during a lifetime. Similarly, the reference concentration (RfC) assesses inhalation risks. The human population addressed in these toxicity values includes sensitive groups, such as asthmatics, or life stages, such as children or the elderly. The RfD for inorganic arsenic for oral exposure is 0.0003 mg/kg/day, and is based on hyperpigmentation, keratosis, and possible vascular complications in numerous southwestern Taiwanese subjects from the blackfoot disease endemic area for arsenic [171]. A chronic reference (RfC) concentration for arsenic exposure via inhalation was not provided.

The EPA concluded that inorganic arsenic from oral and inhalation exposures is a human carcinogen. Cancer slope factors (CSF) are used to estimate the risk of cancer associated with exposure to a carcinogenic substance. A slope factor is an upper bound, with 95% confidence limit, on the increased cancer risk from a lifetime exposure to an agent by ingestion or inhalation. An oral cancer slope factor for inorganic arsenic of 1.5 per mg/kg/day was derived in 1993 based on skin cancers observed in a large Taiwanese cohort exposed to arsenic via drinking water [171]. This cancer risk estimate corresponds to the drinking water arsenic concentration of $0.02 \,\mu$ g/L for a cancer risk of 1 in 1,000,000. Inhalation unit risk is the upper-bound excess lifetime cancer risk estimated from continuous exposure to an agent at a concentration of $1 \,\mu$ g/m³ in air for a lifetime. The inhalation unit risk was determined as 0.0043 per μ g/m³ based on respiratory cancer mortality observed in smelter workers. This cancer risk estimate corresponds to the air arsenic concentration of 0.02 μ g/L in the inhalation of 0.002 μ g/m³ for 1 in 1,000,000 cancer risk.

4.8.1.1.2 ORGANOARSENICALS

The US EPA Office of Pesticide Programs [95] derived a chronic RfD of 0.03 mg/kg/day for MMA based on decreased body weights, diarrhea, body weight gains, food consumption, histopathology of gastrointestinal tract and thyroid in rats, and 0.014 mg/kg/day for DMA based on regenerative proliferation of the bladder epithelial tissue in rats. For carcinogenicity effects, the US EPA Office of Pesticide Programs determined there was "no evidence for carcinogenicity" for MMA and that DMA was "not carcinogenic at doses that do not result in enhanced cell proliferation." More recently, in 2009 IARC classified MMA and DMA as "possibly carcinogenic to humans" (Group 2B) [172].

4.8.1.1.3 ARSINE

The 1994 US EPA Integrated Risk Information System (IRIS) assessment indicates the chronic inhalation reference concentration of $0.05 \,\mu\text{g/m}^3$ based on increased hemolytic effects and spleen changes in animal models [173].

4.8.1.2 ATSDR

4.8.1.2.1 INORGANIC ARSENIC

ATSDR did not evaluate inhalation risk levels for inorganic or organic arsenical compounds. In 2007 the agency [2], however, determined oral toxicity values for potential non-cancer effects associated with exposure to inorganic and organic arsenical compounds. Minimal risk levels (MRLs) are estimates of the daily human exposure to a chemical that is likely to be without appreciable risk of adverse non-cancer health effects over a specified duration of exposure. The acute oral MRL for inorganic arsenic was determined as 0.005 mg/kg/day based on edema of the face, and gastrointestinal effects (nausea, vomiting, diarrhea) observed in humans exposed to arsenic-contaminated soy sauce in Japan. The chronic oral

Arsine Exposure Time	AEGL 1 (ppm) (Discomfort)	AEGL 2 (ppm) (Impaired escape)	AEGL 3 (ppm) (Life Threatening/Death)
10 minutes	NR	0.3	0.91
30 minutes	NR	0.21	0.63
1 hour	NR	0.17	0.5
4 hours	NR	0.04	0.13
8 hours	NR	0.02	0.06

 Table 4–1
 Acute Exposure Guideline Levels (AEGLs) for Arsine [175]

NR: not reported.

MRL of 0.0003 mg/kg/day was based on skin lesions (hyperpigmentation and hyperkeratosis) observed in southwestern Taiwanese populations exposed to arsenic in drinking water.

4.8.1.2.2 ORGANOARSENICALS

In 2007 ATSDR [2] derived an intermediate oral MRL for MMA of 0.1 mg/kg/day based on diarrhea observed in rats and a chronic oral MRL of 0.01 mg/kg/day based on progressive glomerular nephropathy observed in mice. The chronic oral MRL for DMA was derived as 0.02 mg/kg/day based on vacuolization of the urothelium in the urinary bladder in female mice.

4.8.1.3 RIVM National Institute for Public Health and the Environment

Tolerable daily intake (TDI) refers to the amount of a chemical that has been assessed as safe on a long-term basis, usually whole lifetime. Baars et al. [174] determined the TDI of $1.0 \,\mu\text{g/kg/day}$ for chronic oral exposures and the tolerable concentration in air as $1.0 \,\mu\text{g/m}^3$ for chronic inhalation exposures for inorganic arsenic.

4.8.1.4 Other Standards, Regulations, and Guidelines

As a collaborative effort with public and private sectors in handling emergency situations, the US EPA [175] has derived the Acute Exposure Guideline Levels (AEGLs) for arsine in 2010 presented in Table 4–1. AEGLs are threshold exposure limits for the public and are applicable to emergency exposure periods ranging from 10 minutes to 8 hours. The recommended exposure levels are also applicable to infants and children, and other individuals who might be susceptible.

4.8.1.5 World Health Organization (WHO)

WHO guidance criteria for arsine was derived as $0.05 \,\mu\text{g/m}^3$ based on hematological effects observed in animal models [176]. As of 2011 WHO recommends a provisional guideline for arsenic in drinking water of 10 ppb because of scientific uncertainties in the assessment [177].

4.8.1.6 California Environmental Protection Agency

In 2008 the California Environmental Protection Agency (Cal EPA) determined the acute inhalation reference exposure levels as $0.2\,\mu g/m^3$ and the chronic inhalation reference exposure

levels as $0.015 \,\mu\text{g/m}^3$ for inorganic arsenic compounds including arsine [178]. The inhalation unit risk for inorganic arsenic is determined as 0.0033 per $\mu\text{g/m}^3$. The inhalation slope factor derived in 2009 is 12.0 per mg/kg/day based on lung tumors observed in smelter workers [179].

Disclaimer

This work is not a product of the United States Government or the United States Environmental Protection Agency, and the authors are not doing this work in any governmental capacity. The views expressed are those of the authors only and do not necessarily represent those of the United States or the US EPA.

References

- WHO. Evaluation of certain contaminants in food. Seventy-second report of the Joint FAO/WHO Expert Committee on Food Additives. Geneva: World Health Organization; 2011. WHO Technical Report Series, no. 959.
- [2] ATSDR. Arsenic. Toxicological Profile (Update), Atlanta, GA: Agency for Toxic Substances and Disease Registry, Public Health Service, US Department of Health and Human Services; 2007.
- [3] NRC. Critical Aspects of EPA's IRIS Assessment of Inorganic Arsenic: Interim Report. Washington, DC: The National Academies Press; 2014.
- [4] WHO. Arsenic and arsenic compounds, Environmental Health, Criteria No. 224. Geneva: World Health Organization; 2001.
- [5] NRC. Arsenic in drinking water. Washington, DC: The National Academies Press; National Research Council: 1999.
- [6] EFSA. Scientific opinion on Arsenic in food. EFSA J 2009;7(10):1351.
- [7] Aylward LL, Ramasamy S, Hays SM, Schoeny R, Kirman CR. Evaluation of urinary speciated arsenic in NHANES: issues in interpretation in the context of potential inorganic arsenic exposure. Regul Toxicol Pharmacol 2014;69:49–54.
- [8] Ahsan H, Chen Y, Parvez F, Zablotska L, Argos M, Hussain I, et al. Arsenic exposure from drinking water and risk of premalignant skin lesions in Bangladesh: baseline results from the Health Effects of Arsenic Longitudinal Study. Am J Epidemiol 2006;163:1138–48.
- [9] Argos M, Kalra T, Pierce BL, Chen Y, Parvez F, Islam T, et al. A prospective study of arsenic exposure from drinking water and incidence of skin lesions in Bangladesh. Am J Epidemiol 2011;174:185–94.
- [10] Yu RC, Hsu KH, Chen CJ, Froines JR. Arsenic methylation capacity and skin cancer. Cancer Epidemiol Biomarkers Prev 2000;9:1259–62.
- [11] Kile ML, Hoffman E, Rodrigues EG, Breton CV, Quamruzzaman Q, Rahman M, et al. A pathway-based analysis of urinary arsenic metabolites and skin lesions. Am J Epidemiol 2011;173:778–86.
- [12] IARC. IARC Monographs on the Evaluation of Carcinogenic Risks to Humans. A Review of Human Carcinogens: Arsenic, Metals, Fibres, and Dusts, vol. 100C. Geneva: World Health Organization; 2012.
- [13] Tseng WP. Effects and dose-response relationships of skin cancer and blackfoot disease with arsenic. Environ Health Perspect 1977;19:109–19.
- [14] Tseng WP, Chu HM, How SW, Fong JM, Lin CS, Yeh S. Prevalence of skin cancer in an endemic area of chronic arsenicism in Taiwan. J Natl Cancer Inst 1968;40:453–63.
- [15] Chen CJ, Kuo TL, Wu MM. Arsenic and cancers. Lancet 1988;1:414–15.

- [16] Chen CJ, Wu MM, Lee SS, Wang JD, Cheng SH, Wu HY. Atherogenicity and carcinogenicity of higharsenic artesian well water. Multiple risk factors and related malignant neoplasms of blackfoot disease. Arteriosclerosis 1988;8:452–60.
- [17] Chen CJ, Chuang YC, Lin TM, Wu HY. Malignant neoplasms among residents of a blackfoot diseaseendemic area in Taiwan: high-arsenic artesian well water and cancers. Cancer Res 1985;45:5895–9.
- [18] Wu MM, Kuo TL, Hwang YH, Chen CJ. Dose-response relation between arsenic concentration in well water and mortality from cancers and vascular diseases. Am J Epidemiol 1989;130:1123–32.
- [19] Smith AH, Goycolea M, Haque R, Biggs ML. Marked increase in bladder and lung cancer mortality in a region of Northern Chile due to arsenic in drinking water. Am J Epidemiol 1998;147:660–9.
- [20] Hopenhayn-Rich C, Biggs ML, Fuchs A, Bergoglio R, Tello EE, Nicolli H, et al. Bladder cancer mortality associated with arsenic in drinking water in Argentina. Epidemiology 1996;7:117–24.
- [21] Hopenhayn-Rich C, Biggs ML, Smith AH. Lung and kidney cancer mortality associated with arsenic in drinking water in Cordoba, Argentina. Int J Epidemiol 1998;27:561–9.
- [22] Tsuda T, Babazono A, Yamamoto E, Kurumatani N, Mino Y, Ogawa T, et al. Ingested arsenic and internal cancer: a historical cohort study followed for 33 years. Am J Epidemiol 1995;141:198–209.
- [23] Chen CL, Chiou HY, Hsu LI, Hsueh YM, Wu MM, Wang YH, et al. Arsenic in drinking water and risk of urinary tract cancer: a follow-up study from northeastern Taiwan. Cancer Epidemiol Biomarkers Prev 2010;19:101–10.
- [24] Chen CL, Chiou HY, Hsu LI, Hsueh YM, Wu MM, Chen CJ. Ingested arsenic, characteristics of well water consumption and risk of different histological types of lung cancer in northeastern Taiwan. Environ Res 2010;110:455–62.
- [25] Bates MN, Smith AH, Cantor KP. Case-control study of bladder cancer and arsenic in drinking water. Am J Epidemiol 1995;141:523–30.
- [26] Lewis DR, Southwick JW, Ouellet-Hellstrom R, Rench J, Calderon RL. Drinking water arsenic in Utah: a cohort mortality study. Environ Health Perspect 1999;107:359–65.
- [27] Steinmaus C, Yuan Y, Bates MN, Smith AH. Case-control study of bladder cancer and drinking water arsenic in the western United States. Am J Epidemiol 2003;158:1193–201.
- [28] Lamm SH, Engel A, Kruse MB, Feinleib M, Byrd DM, Lai S, et al. Arsenic in drinking water and bladder cancer mortality in the United States: an analysis based on 133 U.S. counties and 30 years of observation. J Occup Environ Med 2004;46:298–306.
- [29] Karagas MR, Stukel TA, Tosteson TD. Assessment of cancer risk and environmental levels of arsenic in New Hampshire. Int J Hyg Environ Health 2002;205:85–94.
- [30] Karagas MR, Tosteson TD, Morris JS, Demidenko E, Mott LA, Heaney J, et al. Incidence of transitional cell carcinoma of the bladder and arsenic exposure in New Hampshire. Cancer Causes Control 2004;15:465–72.
- [31] Heck JE, Andrew AS, Onega T, Rigas JR, Jackson BP, Karagas MR, et al. Lung cancer in a U.S. population with low to moderate arsenic exposure. Environ Health Perspect 2009;117:1718–23.
- [32] Meliker JR, Wahl RL, Cameron LL, Nriagu JO. Arsenic in drinking water and cerebrovascular disease, diabetes mellitus, and kidney disease in Michigan: a standardized mortality ratio analysis. Environ Health 2007;6:4.
- [33] Meliker JR, Slotnick MJ, AvRuskin GA, Schottenfeld D, Jacquez GM, Wilson ML, et al. Lifetime exposure to arsenic in drinking water and bladder cancer: a population-based case-control study in Michigan, USA. Cancer Causes Control 2010;21:745–57.
- [34] Dauphine DC, Smith AH, Yuan Y, Balmes JR, Bates MN, Steinmaus C. Case-control study of arsenic in drinking water and lung cancer in California and Nevada. Int J Environ Res Public Health 2013;10:3310-24.
- [35] Kurttio P, Pukkala E, Kahelin H, Auvinen A, Pekkanen J. Arsenic concentrations in well water and risk of bladder and kidney cancer in Finland. Environ Health Perspect 1999;107:705–10.

- [36] Michaud DS, Wright ME, Cantor KP, Taylor PR, Virtamo J, Albanes D. Arsenic concentrations in prediagnostic toenails and the risk of bladder cancer in a cohort study of male smokers. Am J Epidemiol 2004;160:853–9.
- [37] Baastrup R, Sorensen M, Balstrom T, Frederiksen K, Larsen CL, Tjonneland A, et al. Arsenic in drinkingwater and risk for cancer in Denmark. Environ Health Perspect 2008;116:231–7.
- [38] Ferreccio C, Gonzalez C, Milosavjlevic V, Marshall G, Sancha AM, Smith AH. Lung cancer and arsenic concentrations in drinking water in Chile. Epidemiology 2000;11:673–9.
- [39] Mostafa MG, McDonald JC, Cherry NM. Lung cancer and exposure to arsenic in rural Bangladesh. Occup Environ Med 2008;65:765–8.
- [40] Cantor KP, Lubin JH. Arsenic, internal cancers, and issues in inference from studies of low-level exposures in human populations. Toxicol Appl Pharmacol 2007;222:252–7.
- [41] Gibb H, Haver C, Gaylor D, Ramasamy S, Lee JS, Lobdell D, et al. Utility of recent studies to assess the National Research Council 2001 estimates of cancer risk from ingested arsenic. Environ Health Perspect 2011;119:284–90.
- [42] Cui X, Wakai T, Shirai Y, Hatakeyama K, Hirano S. Chronic oral exposure to inorganic arsenate interferes with methylation status of p16INK4a and RASSF1A and induces lung cancer in A/J mice. Toxicol Sci 2006;91:372–81.
- [43] Soffritti M, Belpoggi F, Degli Esposti D, Lambertini L. Results of a long-term carcinogenicity bioassay on Sprague-Dawley rats exposed to sodium arsenite administered in drinking water. Ann NY Acad Sci 2006;1076:578–91.
- [44] Waalkes MP, Ward JM, Liu J, Diwan BA. Transplacental carcinogenicity of inorganic arsenic in the drinking water: induction of hepatic, ovarian, pulmonary, and adrenal tumors in mice. Toxicol Appl Pharmacol 2003;186:7–17.
- [45] Smith AH, Marshall G, Yuan Y, Ferreccio C, Liaw J, von Ehrenstein O, et al. Increased mortality from lung cancer and bronchiectasis in young adults after exposure to arsenic in utero and in early childhood. Environ Health Perspect 2006;114:1293–6.
- [46] Maull EA, Ahsan H, Edwards J, Longnecker MP, Navas-Acien A, Pi J, et al. Evaluation of the association between arsenic and diabetes: a National Toxicology Program workshop review. Environ Health Perspect 2012;120:1658–70.
- [47] Coronado-Gonzalez JA, Del Razo LM, Garcia-Vargas G, Sanmiguel-Salazar F, Escobedo-de la Pena J. Inorganic arsenic exposure and type 2 diabetes mellitus in Mexico. Environ Res 2007;104:383–9.
- [48] Ettinger AS, Zota AR, Amarasiriwardena CJ, Hopkins MR, Schwartz J, Hu H, et al. Maternal arsenic exposure and impaired glucose tolerance during pregnancy. Environ Health Perspect 2009;117: 1059–64.
- [49] Del Razo LM, Garcia-Vargas GG, Valenzuela OL, Castellanos EH, Sanchez-Pena LC, Currier JM, et al. Exposure to arsenic in drinking water is associated with increased prevalence of diabetes: a cross-sectional study in the Zimapan and Lagunera regions in Mexico. Environ Health 2011;10:73.
- [50] Wang CH, Hsiao CK, Chen CL, Hsu LI, Chiou HY, Chen SY, et al. A review of the epidemiologic literature on the role of environmental arsenic exposure and cardiovascular diseases. Toxicol Appl Pharmacol 2007;222:315–26.
- [51] Navas-Acien A, Sharrett AR, Silbergeld EK, Schwartz BS, Nachman KE, Burke TA, et al. Arsenic exposure and cardiovascular disease: a systematic review of the epidemiologic evidence. Am J Epidemiol 2005;162:1037–49.
- [52] Moon K, Guallar E, Navas-Acien A. Arsenic exposure and cardiovascular disease: an updated systematic review. Curr Atheroscler Rep 2012;14:542–55.
- [53] Abhyankar LN, Jones MR, Guallar E, Navas-Acien A. Arsenic exposure and hypertension: a systematic review. Environ Health Perspect 2012;120:494–500.

- [54] Abir T, Rahman B, D'Este C, Farooq A, Milton AH. The association between chronic arsenic exposure and hypertension: a meta-analysis. J Toxicol 2012:198793.
- [55] Vahidnia A, van der Voet GB, de Wolff FA. Arsenic neurotoxicity—a review. Hum Exp Toxicol 2007;26:823-32.
- [56] Otto D, Xia Y, Li Y, Wu K, He L, Telech J, et al. Neurosensory effects of chronic human exposure to arsenic associated with body burden and environmental measures. Hum Exp Toxicol 2007;26:169–77.
- [57] Wasserman GA, Liu X, Parvez F, Ahsan H, Factor-Litvak P, van Geen A, et al. Water arsenic exposure and children's intellectual function in Araihazar, Bangladesh. Environ Health Perspect 2004;112:1329–33.
- [58] von Ehrenstein OS, Poddar S, Yuan Y, Mazumder DG, Eskenazi B, Basu A, et al. Children's intellectual function in relation to arsenic exposure. Epidemiology 2007;18:44–51.
- [59] Hamadani JD, Grantham-McGregor SM, Tofail F, Nermell B, Fangstrom B, Huda SN, et al. Pre- and postnatal arsenic exposure and child development at 18 months of age: a cohort study in rural Bangladesh. Int J Epidemiol 2010;39:1206–16.
- [60] Hamadani JD, Tofail F, Nermell B, Gardner R, Shiraji S, Bottai M, et al. Critical windows of exposure for arsenic-associated impairment of cognitive function in pre-school girls and boys: a population-based cohort study. Int J Epidemiol 2011;40:1593–604.
- [61] Rodriguez-Barranco M, Lacasana M, Aguilar-Garduno C, Alguacil J, Gil F, Gonzalez-Alzaga B, et al. Association of arsenic, cadmium and manganese exposure with neurodevelopment and behavioural disorders in children: a systematic review and meta-analysis. Sci Total Environ 2013;454–455:562–77.
- [62] Rodriguez VM, Carrizales L, Mendoza MS, Fajardo OR, Giordano M. Effects of sodium arsenite exposure on development and behavior in the rat. Neurotoxicol Teratol 2002;24:743–50.
- [63] Martinez-Finley EJ, Ali AM, Allan AM. Learning deficits in C57BL/6J mice following perinatal arsenic exposure: consequence of lower corticosterone receptor levels? Pharmacol Biochem Behav 2009;94:271–7.
- [64] Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborti D, et al. Arsenic in drinking water and the prevalence of respiratory effects in West Bengal, India. Int J Epidemiol 2000;29:1047–52.
- [65] Milton AH, Rahman M. Respiratory effects and arsenic contaminated well water in Bangladesh. Int J Environ Health Res 2002;12:175–9.
- [66] Mazumder DN, Steinmaus C, Bhattacharya P, von Ehrenstein OS, Ghosh N, Gotway M, et al. Bronchiectasis in persons with skin lesions resulting from arsenic in drinking water. Epidemiology 2005;16:760-5.
- [67] Milton AH, Hasan Z, Rahman A, Rahman M. Non-cancer effects of chronic arsenicosis in Bangladesh: preliminary results. J Environ Sci Health A Tox Hazard Subst Environ Eng 2003;38:301–5.
- [68] Smith AH, Marshall G, Yuan Y, Liaw J, Ferreccio C, Steinmaus C. Evidence from Chile that arsenic in drinking water may increase mortality from pulmonary tuberculosis. Am J Epidemiol 2011;173:414–20.
- [69] Parvez F, Chen Y, Brandt-Rauf PW, Bernard A, Dumont X, Slavkovich V, et al. Nonmalignant respiratory effects of chronic arsenic exposure from drinking water among never-smokers in Bangladesh. Environ Health Perspect 2008;116:190–5.
- [70] Rahman A, Vahter M, Ekstrom EC, Persson LA. Arsenic exposure in pregnancy increases the risk of lower respiratory tract infection and diarrhea during infancy in Bangladesh. Environ Health Perspect 2011;119:719–24.
- [71] Dauphine DC, Ferreccio C, Guntur S, Yuan Y, Hammond SK, Balmes J, et al. Lung function in adults following in utero and childhood exposure to arsenic in drinking water: preliminary findings. Int Arch Occup Environ Health 2011;84:591–600.
- [72] Amster ED, Cho JI, Christiani D. Urine arsenic concentration and obstructive pulmonary disease in the U.S. population. J Toxicol Environ Health A 2011;74:716–27.

- [73] Wang SL, Chiou JM, Chen CJ, Tseng CH, Chou WL, Wang CC, et al. Prevalence of non-insulin-dependent diabetes mellitus and related vascular diseases in southwestern arseniasis-endemic and nonendemic areas in Taiwan. Environ Health Perspect 2003;111:155–9.
- [74] Chen Y, Parvez F, Liu M, Pesola GR, Gamble MV, Slavkovich V, et al. Association between arsenic exposure from drinking water and proteinuria: results from the Health Effects of Arsenic Longitudinal Study. Int J Epidemiol 2011;40:828–35.
- [75] Chen JW, Chen HY, Li WF, Liou SH, Chen CJ, Wu JH, et al. The association between total urinary arsenic concentration and renal dysfunction in a community-based population from central Taiwan. Chemosphere 2011;84:17–24.
- [76] Zheng LY, Umans JG, Tellez-Plaza M, Yeh F, Francesconi KA, Goessler W, et al. Urine arsenic and prevalent albuminuria: evidence from a population-based study. Am J Kidney Dis 2013;61:385–94.
- [77] Das N, Paul S, Chatterjee D, Banerjee N, Majumder NS, Sarma N, et al. Arsenic exposure through drinking water increases the risk of liver and cardiovascular diseases in the population of West Bengal, India. BMC Public Health 2012;12:639.
- [78] Islam K, Haque A, Karim R, Fajol A, Hossain E, Salam KA, et al. Dose-response relationship between arsenic exposure and the serum enzymes for liver function tests in the individuals exposed to arsenic: a cross sectional study in Bangladesh. Environ Health 2011;10:64.
- [79] Concha G, Vogler G, Lezcano D, Nermell B, Vahter M. Exposure to inorganic arsenic metabolites during early human development. Toxicol Sci 1998;44:185–90.
- [80] Hall M, Gamble M, Slavkovich V, Liu X, Levy D, Cheng Z, et al. Determinants of arsenic metabolism: blood arsenic metabolites, plasma folate, cobalamin, and homocysteine concentrations in maternalnewborn pairs. Environ Health Perspect 2007;115:1503–9.
- [81] Jin Y, Xi S, Li X, Lu C, Li G, Xu Y, et al. Arsenic speciation transported through the placenta from mother mice to their newborn pups. Environ Res 2006;101:349–55.
- [82] Vahter M. Health effects of early life exposure to arsenic. Basic Clin Pharmacol Toxicol 2008;102:204–11.
- [83] Ahmad SA, Sayed MH, Barua S, Khan MH, Faruquee MH, Jalil A, et al. Arsenic in drinking water and pregnancy outcomes. Environ Health Perspect 2001;109:629–31.
- [84] Hopenhayn C, Ferreccio C, Browning SR, Huang B, Peralta C, Gibb H, et al. Arsenic exposure from drinking water and birth weight. Epidemiology 2003;14:593–602.
- [85] Yang CY, Chang CC, Tsai SS, Chuang HY, Ho CK, Wu TN. Arsenic in drinking water and adverse pregnancy outcome in an arseniasis-endemic area in northeastern Taiwan. Environ Res 2003;91:29–34.
- [86] Huyck KL, Kile ML, Mahiuddin G, Quamruzzaman Q, Rahman M, Breton CV, et al. Maternal arsenic exposure associated with low birth weight in Bangladesh. J Occup Environ Med 2007;49:1097–104.
- [87] Milton AH, Smith W, Rahman B, Hasan Z, Kulsum U, Dear K, et al. Chronic arsenic exposure and adverse pregnancy outcomes in Bangladesh. Epidemiology 2005;16:82–6.
- [88] Rahman A, Persson LA, Nermell B, El Arifeen S, Ekstrom EC, Smith AH, et al. Arsenic exposure and risk of spontaneous abortion, stillbirth, and infant mortality. Epidemiology 2010;21:797–804.
- [89] Rahman A, Vahter M, Smith AH, Nermell B, Yunus M, El Arifeen S, et al. Arsenic exposure during pregnancy and size at birth: a prospective cohort study in Bangladesh. Am J Epidemiol 2009;169:304–12.
- [90] Rahman A, Vahter M, Ekstrom EC, Rahman M, Golam Mustafa AH, Wahed MA, et al. Association of arsenic exposure during pregnancy with fetal loss and infant death: a cohort study in Bangladesh. Am J Epidemiol 2007;165:1389–96.
- [91] Vahter M. Effects of arsenic on maternal and fetal health. Annu Rev Nutr 2009;29:381-99.
- [92] von Ehrenstein OS, Guha Mazumder DN, Hira-Smith M, Ghosh N, Yuan Y, Windham G, et al. Pregnancy outcomes, infant mortality, and arsenic in drinking water in West Bengal, India. Am J Epidemiol 2006;163:662–9.
- [93] US EPA. Revised Reregistration Eligibility Decision for MSMA, DSMA, CAMA, and cacodylic acid. The Office of Pesticide Programs. USEPA, Washington, DC; 2006.
- [94] Arnold LL, Eldan M, Nyska A, van Gemert M, Cohen SM. Dimethylarsinic acid: results of chronic toxicity/ oncogenicity studies in F344 rats and in B6C3F1 mice. Toxicology 2006;223:82–100.
- [95] Arnold LL, Eldan M, van Gemert M, Capen CC, Cohen SM. Chronic studies evaluating the carcinogenicity of monomethylarsonic acid in rats and mice. Toxicology 2003;190:197–219.
- [96] Thomas DJ, Styblo M, Lin S. The cellular metabolism and systemic toxicity of arsenic. Toxicol Appl Pharmacol 2001;176:127-44.
- [97] Sams II R, Wolf DC, Ramasamy S, Ohanian E, Chen J, Lowit A. Workshop overview: arsenic research and risk assessment. Toxicol Appl Pharmacol 2007;222:245–51.
- [98] Vahter M, Marafante E. Intracellular interaction and metabolic fate of arsenite and arsenate in mice and rabbits. Chem Biol Interact 1983;47:29–44.
- [99] Vahter M, Marafante E, Dencker L. Tissue distribution and retention of 74As-dimethylarsinic acid in mice and rats. Arch Environ Contam Toxicol 1984;13:259–64.
- [100] Yamauchi H, Yamamura Y. Metabolism and excretion of orally administered dimethylarsinic acid in the hamster. Toxicol Appl Pharmacol 1984;74:134–40.
- [101] Marafante E, Vahter M, Norin H, Envall J, Sandstrom M, Christakopoulos A, et al. Biotransformation of dimethylarsinic acid in mouse, hamster and man. J Appl Toxicol 1987;7:111–17.
- [102] Moore MM, Harrington-Brock K, Doerr CL. Relative genotoxic potency of arsenic and its methylated metabolites. Mutat Res 1997;386:279–90.
- [103] Rasmussen RE, Menzel DB. Variation in arsenic-induced sister chromatid exchange in human lymphocytes and lymphoblastoid cell lines. Mutat Res 1997;386:299–306.
- [104] Hughes MF, Kenyon EM. Dose-dependent effects on the disposition of monomethylarsonic acid and dimethylarsinic acid in the mouse after intravenous administration. J Toxicol Environ Health A 1998;53:95-112.
- [105] Sakurai T, Kaise T, Matsubara C. Inorganic and methylated arsenic compounds induce cell death in murine macrophages via different mechanisms. Chem Res Toxicol 1998;11:273–83.
- [106] Petrick JS, Ayala-Fierro F, Cullen WR, Carter DE, Vasken Aposhian H. Monomethylarsonous acid (MMA(III)) is more toxic than arsenite in Chang human hepatocytes. Toxicol Appl Pharmacol 2000;163:203–7.
- [107] Petrick JS, Jagadish B, Mash EA, Aposhian HV. Monomethylarsonous acid (MMA(III)) and arsenite: LD(50) in hamsters and in vitro inhibition of pyruvate dehydrogenase. Chem Res Toxicol 2001;14:651–6.
- [108] Dopp E, Hartmann LM, Florea AM, von Recklinghausen U, Pieper R, Shokouhi B, et al. Uptake of inorganic and organic derivatives of arsenic associated with induced cytotoxic and genotoxic effects in Chinese hamster ovary (CHO) cells. Toxicol Appl Pharmacol 2004;201:156–65.
- [109] Styblo M, Del Razo LM, LeCluyse EL, Hamilton GA, Wang C, Cullen WR, et al. Metabolism of arsenic in primary cultures of human and rat hepatocytes. Chem Res Toxicol 1999;12:560–5.
- [110] Styblo M, Del Razo LM, Vega L, Germolec DR, LeCluyse EL, Hamilton GA, et al. Comparative toxicity of trivalent and pentavalent inorganic and methylated arsenicals in rat and human cells. Arch Toxicol 2000;74:289–99.
- [111] Kenyon EM, Hughes MF. A concise review of the toxicity and carcinogenicity of dimethylarsinic acid. Toxicology 2001;160:227-36.
- [112] Hughes MF, Del Razo LM, Kenyon EM. Dose-dependent effects on tissue distribution and metabolism of dimethylarsinic acid in the mouse after intravenous administration. Toxicology 2000;143:155–66.
- [113] Styblo MV, Vega L, Germolec DR, Luster MI, Del Razo LM, Wang C, et al. Metabolism and toxicity of arsenicals in cultured cells. In: Chappell WA, Abernathy CO, Calderon RL, editors. Arsenic exposure and health effects. Oxford: Elsevier; 1999. pp. 311–23.

- [114] Yamauchi H, Yamamura Y. Dynamic change of inorganic arsenic and methylarsenic compounds in human urine after oral intake as arsenic trioxide. Ind Health 1979;17:79–84.
- [115] Tam GK, Charbonneau SM, Bryce F, Pomroy C, Sandi E. Metabolism of inorganic arsenic (74As) in humans following oral ingestion. Toxicol Appl Pharmacol 1979;50:319–22.
- [116] Pomroy C, Charbonneau SM, McCullough RS, Tam GK. Human retention studies with 74As. Toxicol Appl Pharmacol 1980;53:550–6.
- [117] Buchet JP, Lauwerys R, Roels H. Comparison of the urinary excretion of arsenic metabolites after a single oral dose of sodium arsenite, monomethylarsonate, or dimethylarsinate in man. Int Arch Occup Environ Health 1981;48:71–9.
- [118] US EPA, Risk Assessment Forum. Guidelines for carcinogen risk assessment. US Environmental Protection Agency. EPA/630/P-03/001F; 2005.
- [119] Flora SJ, Mittal M, Mehta A. Heavy metal induced oxidative stress, and its possible reversal by chelation therapy. Indian J Med Res 2008;128:501–23.
- [120] Lee MY, Jung BJ, Chung SM, Bae ON, Lee JY, Park JD, et al. Arsenic-induced dysfunction in relaxation of blood vessels. Environ Health Perspect 2003;111:513–17.
- [121] Simeonova PP, Hulderman T, Harki D, Luster MI. Arsenic exposure accelerates atherogenesis in apolipoprotein E(-/-) mice. Environ Health Perspect 2003;111:1744–8.
- [122] Mukherjee S, Das D, Mukherjee M, Das AS, Mitra C. Synergistic effect of folic acid and vitamin B12 in ameliorating arsenic-induced oxidative damage in pancreatic tissue of rat. J Nutr Biochem 2006;17:319–27.
- [123] Izquierdo-Vega JA, Soto CA, Sanchez-Pena LC, De Vizcaya-Ruiz A, Del Razo LM. Diabetogenic effects and pancreatic oxidative damage in rats subchronically exposed to arsenite. Toxicol Lett 2006;160:135–42.
- [124] Garcia-Chavez E, Jimenez I, Segura B, Del Razo LM. Lipid oxidative damage and distribution of inorganic arsenic and its metabolites in the rat nervous system after arsenite exposure: influence of alpha tocopherol supplementation. Neurotoxicology 2006;27:1024–31.
- [125] Zarazua S, Perez-Severiano F, Delgado JM, Martinez LM, Ortiz-Perez D, Jimenez-Capdeville ME. Decreased nitric oxide production in the rat brain after chronic arsenic exposure. Neurochem Res 2006;31:1069–77.
- [126] Li Y, Chen Y, Slavkovic V, Ahsan H, Parvez F, Graziano JH, et al. Serum levels of the extracellular domain of the epidermal growth factor receptor in individuals exposed to arsenic in drinking water in Bangladesh. Biomarkers 2007;12:256–65.
- [127] Argos M, Kibriya MG, Parvez F, Jasmine F, Rakibuz-Zaman M, Ahsan H. Gene expression profiles in peripheral lymphocytes by arsenic exposure and skin lesion status in a Bangladeshi population. Cancer Epidemiol Biomarkers Prev 2006;15:1367–75.
- [128] Lee CH, Chen JS, Sun YL, Liao WT, Zheng YW, Chai CZ, et al. Defective beta1-integrins expression in arsenical keratosis and arsenic-treated cultured human keratinocytes. J Cutan Pathol 2006;33:129–38.
- [129] Bunderson M, Brooks DM, Walker DL, Rosenfeld ME, Coffin JD, Beall HD. Arsenic exposure exacerbates atherosclerotic plaque formation and increases nitrotyrosine and leukotriene biosynthesis. Toxicol Appl Pharmacol 2004;201:32–9.
- [130] Kamat CD, Green DE, Curilla S, Warnke L, Hamilton JW, Sturup S, et al. Role of HIF signaling on tumorigenesis in response to chronic low-dose arsenic administration. Toxicol Sci 2005;86:248–57.
- [131] Lee PC, Ho IC, Lee TC. Oxidative stress mediates sodium arsenite-induced expression of heme oxygenase-1, monocyte chemoattractant protein-1, and interleukin-6 in vascular smooth muscle cells. Toxicol Sci 2005;85:541–50.
- [132] Nuntharatanapong N, Chen K, Sinhaseni P, Keaney Jr. JF. EGF receptor-dependent JNK activation is involved in arsenite-induced p21Cip1/Waf1 upregulation and endothelial apoptosis. Am J Physiol Heart Circ Physiol 2005;289:H99–H107.

- [133] Soucy NV, Mayka D, Klei LR, Nemec AA, Bauer JA, Barchowsky A. Neovascularization and angiogenic gene expression following chronic arsenic exposure in mice. Cardiovasc Toxicol 2005;5:29–41.
- [134] Tsou TC, Tsai FY, Hsieh YW, Li LA, Yeh SC, Chang LW. Arsenite induces endothelial cytotoxicity by downregulation of vascular endothelial nitric oxide synthase. Toxicol Appl Pharmacol 2005;208:277–84.
- [135] Tsou TC, Yeh SC, Tsai EM, Tsai FY, Chao HR, Chang LW. Arsenite enhances tumor necrosis factor-alphainduced expression of vascular cell adhesion molecule-1. Toxicol Appl Pharmacol 2005;209:10–18.
- [136] Pereira FE, Coffin JD, Beall HD. Activation of protein kinase C and disruption of endothelial monolayer integrity by sodium arsenite—potential mechanism in the development of atherosclerosis. Toxicol Appl Pharmacol 2007;220:164–77.
- [137] Jana K, Jana S, Samanta PK. Effects of chronic exposure to sodium arsenite on hypothalamo-pituitary-testicular activities in adult rats: possible an estrogenic mode of action. Reprod Biol Endocrinol 2006;4:9.
- [138] Patterson TJ, Rice RH. Arsenite and insulin exhibit opposing effects on epidermal growth factor receptor and keratinocyte proliferative potential. Toxicol Appl Pharmacol 2007;221:119–28.
- [139] Hwang BJ, Utti C, Steinberg M. Induction of cyclin D1 by submicromolar concentrations of arsenite in human epidermal keratinocytes. Toxicol Appl Pharmacol 2006;217:161–7.
- [140] Mudipalli A, Owen RD, Preston RJ. The effect of arsenicals on ultraviolet-radiation-induced growth arrest and related signaling events in human keratinocytes. Int J Oncol 2005;27:769–78.
- [141] Liao WT, Chang KL, Yu CL, Chen GS, Chang LW, Yu HS. Arsenic induces human keratinocyte apoptosis by the FAS/FAS ligand pathway, which correlates with alterations in nuclear factor-kappa B and activator protein-1 activity. J Invest Dermatol 2004;122:125–9.
- [142] Zhang TC, Schmitt MT, Mumford JL. Effects of arsenic on telomerase and telomeres in relation to cell proliferation and apoptosis in human keratinocytes and leukemia cells in vitro. Carcinogenesis 2003;24:1811–17.
- [143] Clewell HJ, Thomas RS, Gentry PR, Crump KS, Kenyon EM, El-Masri HA, et al. Research toward the development of a biologically based dose response assessment for inorganic arsenic carcinogenicity: a progress report. Toxicol Appl Pharmacol 2007;222:388–98.
- [144] Mann S, Droz PO, Vahter M. A physiologically based pharmacokinetic model for arsenic exposure. I. Development in hamsters and rabbits. Toxicol Appl Pharmacol 1996;137:8–22.
- [145] Mann S, Droz PO, Vahter M. A physiologically based pharmacokinetic model for arsenic exposure. II. Validation and application in humans. Toxicol Appl Pharmacol 1996;140:471–86.
- [146] Yu D. A pharmacokinetic modeling of inorganic arsenic: a short-term oral exposure model for humans. Chemosphere 1999;39:2737-47.
- [147] Yu D. A physiologically based pharmacokinetic model of inorganic arsenic. Regul Toxicol Pharmacol 1999;29:128–41.
- [148] Gentry PR, Covington TR, Mann S, Shipp AM, Yager JW, Clewell III HJ. Physiologically based pharmacokinetic modeling of arsenic in the mouse. J Toxicol Environ Health A 2004;67:43–71.
- [149] El-Masri HA, Kenyon EM. Development of a human physiologically based pharmacokinetic (PBPK) model for inorganic arsenic and its mono- and di-methylated metabolites. J Pharmacokinet Pharmacodyn 2008;35:31-68.
- [150] Ahsan H, Chen Y, Kibriya MG, Slavkovich V, Parvez F, Jasmine F, et al. Arsenic metabolism, genetic susceptibility, and risk of premalignant skin lesions in Bangladesh. Cancer Epidemiol Biomarkers Prev 2007;16:1270–8.
- [151] Huang YK, Tseng CH, Huang YL, Yang MH, Chen CJ, Hsueh YM. Arsenic methylation capability and hypertension risk in subjects living in arseniasis-hyperendemic areas in southwestern Taiwan. Toxicol Appl Pharmacol 2007;218:135–42.

- [152] McCarty KM, Chen YC, Quamruzzaman Q, Rahman M, Mahiuddin G, Hsueh YM, et al. Arsenic methylation, GSTT1, GSTM1, GSTP1 polymorphisms, and skin lesions. Environ Health Perspect 2007;115:341–5.
- [153] Steinmaus C, Carrigan K, Kalman D, Atallah R, Yuan Y, Smith AH. Dietary intake and arsenic methylation in a U.S. population. Environ Health Perspect 2005;113:1153–9.
- [154] Tseng CH, Huang YK, Huang YL, Chung CJ, Yang MH, Chen CJ, et al. Arsenic exposure, urinary arsenic speciation, and peripheral vascular disease in blackfoot disease-hyperendemic villages in Taiwan. Toxicol Appl Pharmacol 2005;206:299–308.
- [155] Valenzuela OL, Borja-Aburto VH, Garcia-Vargas GG, Cruz-Gonzalez MB, Garcia-Montalvo EA, Calderon-Aranda ES, et al. Urinary trivalent methylated arsenic species in a population chronically exposed to inorganic arsenic. Environ Health Perspect 2005;113:250–4.
- [156] Chen YC, Su HJ, Guo YL, Houseman EA, Christiani DC. Interaction between environmental tobacco smoke and arsenic methylation ability on the risk of bladder cancer. Cancer Causes Control 2005;16:75–81.
- [157] Huang YL, Hsueh YM, Huang YK, Yip PK, Yang MH, Chen CJ. Urinary arsenic methylation capability and carotid atherosclerosis risk in subjects living in arsenicosis-hyperendemic areas in southwestern Taiwan. Sci Total Environ 2009;407:2608–14.
- [158] Li X, Li B, Xu Y, Wang Y, Jin Y, Itoh T, et al. Arsenic methylation capacity and its correlation with skin lesions induced by contaminated drinking water consumption in residents of chronic arsenicosis area. Environ Toxicol 2011;26:118–23.
- [159] Steinmaus C, Bates MN, Yuan Y, Kalman D, Atallah R, Rey OA, et al. Arsenic methylation and bladder cancer risk in case-control studies in Argentina and the United States. J Occup Environ Med 2006;48:478–88.
- [160] Steinmaus C, Moore LE, Shipp M, Kalman D, Rey OA, Biggs ML, et al. Genetic polymorphisms in MTHFR 677 and 1298, GSTM1 and T1, and metabolism of arsenic. J Toxicol Environ Health A 2007;70:159–70.
- [161] Loffredo CA, Aposhian HV, Cebrian ME, Yamauchi H, Silbergeld EK. Variability in human metabolism of arsenic. Environ Res 2003;92:85–91.
- [162] Concha G, Nermell B, Vahter MV. Metabolism of inorganic arsenic in children with chronic high arsenic exposure in northern Argentina. Environ Health Perspect 1998;106:355–9.
- [163] Schlawicke Engstrom K, Broberg K, Concha G, Nermell B, Warholm M, Vahter M. Genetic polymorphisms influencing arsenic metabolism: evidence from Argentina. Environ Health Perspect 2007;115:599–605.
- [164] Lindberg AL, Kumar R, Goessler W, Thirumaran R, Gurzau E, Koppova K, et al. Metabolism of low-dose inorganic arsenic in a central European population: influence of sex and genetic polymorphisms. Environ Health Perspect 2007;115:1081–6.
- [165] Lindberg AL, Ekstrom EC, Nermell B, Rahman M, Lonnerdal B, Persson LA, et al. Gender and age differences in the metabolism of inorganic arsenic in a highly exposed population in Bangladesh. Environ Res 2008;106:110–20.
- [166] Ortiz JG, Opoka R, Kane D, Cartwright IL. Investigating arsenic susceptibility from a genetic perspective in Drosophila reveals a key role for glutathione synthetase. Toxicol Sci 2009;107:416–26.
- [167] Lindberg AL, Sohel N, Rahman M, Persson LA, Vahter M. Impact of smoking and chewing tobacco on arsenic-induced skin lesions. Environ Health Perspect 2010;118:533–8.
- [168] Lindberg AL, Rahman M, Persson LA, Vahter M. The risk of arsenic induced skin lesions in Bangladeshi men and women is affected by arsenic metabolism and the age at first exposure. Toxicol Appl Pharmacol 2008;230:9–16.
- [169] Melkonian S, Argos M, Pierce BL, Chen Y, Islam T, Ahmed A, et al. A prospective study of the synergistic effects of arsenic exposure and smoking, sun exposure, fertilizer use, and pesticide use on risk of premalignant skin lesions in Bangladeshi men. Am J Epidemiol 2011;173:183–91.

120 HANDBOOK OF ARSENIC TOXICOLOGY

- [170] US EPA. National primary drinking water regulations: arsenic and clarifications to compliance and new source contaminants monitoring. Federal Register 2001;66:14.
- [171] US EPA. Toxicological review of inorganic arsenic. EPA IRIS. Washington, DC; 1993.
- [172] Straif K, Benbrahim-Tallaa L, Baan R, Grosse Y, Secretan B, El Ghissassi F, et al. WHO International Agency for Research on Cancer Monograph Working Group. A review of human carcinogens—part C: metals, arsenic, dusts, and fibres. Lancet Oncol 2009;10:453–4.
- [173] US EPA. Toxicological review of arsine. EPA IRIS. Washington, DC; 1994.
- [174] Baars A. Re-evaluation of human-toxicological maximum permissible risk levels. (The National Institute of Public Health and the Environment), Bilthoven, the Netherlands; 2001. pp. 25–9.
- [175] US EPA. Acute exposure guideline levels (AEGLs), Washington, DC; 2013.
- [176] WHO. Arsine: human health aspects. CICAD, doc. no. 47; 2002.
- [177] WHO. Arsenic in drinking-water. WHO/SDE/WSH/03.04/75/Rev/1, WHO Guidelines for Drinking-water Quality, Geneva; World Health Organization: 2011.
- [178] Cal EPA. Inorganic arsenic reference exposure levels Sacramento: State of California Office of Environmental Health Hazard Assessment; 2008.
- [179] Cal EPA. Adoption of the revised air toxics hot spots program technical support document for cancer potency factors Sacramento: State of California Office of Environmental Health Hazard Assessment; 2009.

Evaluation of Novel Modified Activated Alumina as Adsorbent for Arsenic Removal

Lucy Mar Camacho¹, Sundaravadivelnathan Ponnusamy², Isaac Campos³, Thomas A. Davis³, and Shuguang Deng²

¹DEPARTMENT OF ENVIRONMENTAL ENGINEERING, TEXAS A&M UNIVERSITY—KINGSVILLE, KINGSVILLE, TX, USA, ²DEPARTMENT OF CHEMICAL ENGINEERING, NEW MEXICO STATE UNIVERSITY, LAS CRUCES, NM, USA, ³DEPARTMENT OF CIVIL ENGINEERING, THE UNIVERSITY OF TEXAS AT EL PASO, EL PASO, TX, USA

CHAPTER OUTLINE

5.1 Introduction	121
5.2 Materials and Methods	122
5.3 Results and Discussion	125
5.4 Conclusions	
Acknowledgments	135
References	135

5.1 Introduction

Arsenic (As) is a persistent contaminant and highly toxic in nature. As is a known carcinogen that can affect the liver, lungs, bladder, and skin. It also causes cardiovascular diseases and hypertension. Further, As is known to interfere with some of the functions performed by enzymes in the body as well as replace phosphate in some processes [1]. Arsenic is found in food and environmental media in various forms; however, the most common route of exposure is via consumption of arsenic-contaminated drinking water [2,3]. Predicting the mobility of As in drinking water sources, including groundwater and surface water, becomes difficult since As mobility depends on the redox properties of the water. Redox conditions in turn determine the chemical properties of As. As(V) is commonly present in surface water while As(III) is present in anaerobic groundwater conditions. As(V) is less toxic and easier to remove than As(III) [4].

The presence of As in drinking water has become a major health concern in the world [5]. The permissible limit of arsenic in drinking water has been recommended as $10 \mu g/L$ by the

World Health Organization (WHO) [6,7]. The limit suggested by the WHO is being enforced in countries such as the United States, New Zealand, Taiwan, and Vietnam [8]. Other countries such as Mexico, Argentina, Bangladesh, China, and Nepal have set their limit at $50 \mu g/L$ [8]. Regions like Bangladesh, India, and eastern Croatia are well known for having high levels of As contamination in water. Other regions reported with high probabilities of As contamination include Zambia, Central Australia, and New Zealand. Unhealthy As levels are present in Mexico and in the Latin American countries of Argentina, Chile, Peru, and Nicaragua [4].

In view of the toxic effects of As to human health, it becomes important to remove it from the environment. Some of the most common technologies for arsenic removal include oxidation, coagulation/precipitation, adsorption, lime treatment, ion exchange, electrocoagulation, and membranes [9,10]. Among these removal methods, adsorption has proven to be the best method because of its versatility and affordability [11]. Additionally, adsorbent materials can be regenerated and reused and in this way the amount of generated waste material is considerably reduced [8]. The affordability of adsorbents makes them ideal to implement in thirdworld countries [12] and in small municipalities seeking to comply with standards.

Some of the most widely used adsorbents for arsenic removal include activated alumina (AA), natural zeolites [13], and granular ferric oxide [14]. Activated alumina was approved by the US Environmental Protection Agency (EPA) as the best available treatment for arsenic removal. Few studies have been reported to remove As using alumina adsorbents [15,16]. Deng and Lin [16] developed a sol-gel method for preparing AA beads that enhanced the surface area in the mesopore structure of the adsorbent material. Substantially higher capacity for arsenic was reported for this material, which was attributed to the unconventional pore structure of the novel sol-gel material [17]. The purpose of this research study was to evaluate the performance of sol-gel activated alumina and to determine the effect of the presence of nanoparticles in the adsorption process.

5.2 Materials and Methods

5.2.1 Preparation of Sol-Gel Activated Alumina

Granular AA adsorbent beads were prepared following the Yoldas [18] sol-gel process. The first stage of this process was synthesis of a stable boehmite sol (γ -ALOOH) by dissolving alumina-trisecondary butoxide in deionized water at 75°C. The obtained solution was peptized with 1.0 M HNO₃ and refluxed at 90–100°C for 10 hours to obtain a stable boehmite sol. Granular boehmite beads with a median diameter of about 2 mm were generated through a glass pipette and a thin layer of paraffin oil. Ammonia solution was used to convert the partially gelated spherical wetgel beads to firm wet-gel spheres. The wet-gel spheres were then washed, dried, and calcined at 450°C. Reagent grade chemicals and paraffin oil were obtained from Fischer Scientific.

5.2.2 Surface Modifications of Sol-Gel Activated Alumina

Three different modifications were made for preparing the sol-gel AA adsorbent. These modifications included: (1) addition of calcium oxide to the already prepared AA beads; (2) addition of zinc oxide nanoparticle solution during the gelation process; and (3) use of supercritical CO_2 drying instead of calcination. The purpose of the modifications was to further improve the adsorption properties of the AA adsorbent. Adsorption enhancement is expected to occur due to the increase of pore spaces inside the AA. Supercritical CO_2 will increase the void spaces inside the AA; zinc oxide nanoparticles will enhance the affinity for As to the AA and will provide additional nanopores for adsorption; and calcium oxide will increase the ion-exchange capacity of the AA.

- 1. CaO-modified AA: The calcium oxide-modified AA (CaO-AA) was obtained by mixing 5 g of prepared Solgel AA beads with 100 mL of a 3M analytical grade CaCl₂ solution. The obtained solution was shaken for 24 hours and filtered. The coated CaO-AA beads were then dried in a conventional oven at 100°C for 24 hours and calcined at 450°C for about 12 hours. Afterwards, the coated adsorbent was cooled at room temperature.
- **2.** Nanoparticle-modified AA: The ZnO nanoparticle-modified AA (Nanogel AA) was obtained by suspending 30 mg of ZnO nanoparticles in 200 mL of water in an ultrasonic bath. The resulting suspension was added to 106 mL of 2 M alumina-tri-secondary butoxide during the hydrolysis step of the process. The obtained solution was peptized with 1.0 M HNO₃ and refluxed at 90–100°C for 10 hours. The nano-AA beads were obtained with the help of a glass pipette and a thin layer of paraffin oil as established by the Yoldas method. The wetgel spheres were then washed, dried, and calcined at 450°C.
- **3.** Sol-gel AA dried with supercritical CO_2 (Aerogel AA): The supercritical- CO_2 -dried AA (Aerogel AA) was obtained by drying the wet-gel AA spheres in a supercritical CO_2 reactor at 1200°C and 80 psi for 2 hours. The gas was released from the reactor after 2 hours to send the evaporated water out of the system, and the beads were dried for a second time in the reactor at 1000°C and 90 psi for 1 hour. The beads were then removed from the chamber and calcined at 450°C for 3 hours.

5.2.2.1 Characterization of Adsorbent Materials

Novel sol-gel AA and sol-gel-modified AA adsorbent materials were analyzed for chemical composition and pore textural properties using scanning electron microscopy with energy dispersive spectroscopy (SEM/EDS) (Hitachi, S-3400N) and N₂ adsorption and desorption (Micromeritics[®] ASAP 2020) analysis. The pore textural properties including BET-specific surface area, pore volume, and pore size distribution were obtained from the built-in software in the ASAP-2020 instrument.

5.2.3 Batch Adsorption Experiments

Batch adsorption experiments were conducted to determine the As adsorption equilibrium properties of the adsorbent materials. These experiments included pH versus time, pH versus concentration, adsorbent amount versus concentration (adsorption isotherms), and break-through column experiments. A total of four adsorbent materials was analyzed, namely, Solgel AA, CaO-AA, Nanogel AA, and Aerogel AA.

1. Equilibrium experiments—determination of time required to achieve equilibrium: In order to determine the time required to reach equilibrium, a 100 mL solution and two replicates

were prepared to a concentration of $100 \mu g/L$ of arsenic (V). An arsenic standard solution from High-Purity Standards was used as the bases for the preparation of the As solutions. Each solution was adjusted to have a pH between 6.5 and 7.0 by adding drop-wise 0.1 N HCl or 0.1 N NaOH. Then, 1.0 g of the adsorbent was added to each of the bottles containing the solutions. The bottles were placed in a shaker at 100 rpm. The pH of each of the solutions was monitored as a function of time until equilibrium was reached. Equilibrium was reached when the pH did not change significantly. The pH of the solutions as well as the As concentrations were measured at the beginning and end of the equilibrium experiments.

- **2.** Effect of pH on As adsorption: To determine the effect of pH on the adsorption removal of As, solutions of 100 μg/L of arsenic (V) were prepared and adjusted with 0.1 N HCl or 0.1 N NaOH to an initial pH of 2, 4, 5, 6, 7, 8, and 10. Each solution was prepared in a 200-mL plastic bottle and to each solution 1.0 g of AA adsorbent was added. Once the adsorbent was added to the As solution, the bottle was covered and placed in the automatic shaker at 100 rpm until the solution reached equilibrium. For each pH an original and a replicate were prepared. Both pH and As concentration were measured at the beginning and end of the experiment.
- **3.** Effect of As concentration on adsorbed quantity: To determine how the initial As concentration affects the amount of As adsorbed, the two best adsorbents from the previous pH vs. concentration experiment were selected and tested in an equilibrium experiment. The experiment was conducted by preparing 100 mL solutions with As concentrations of 5, 10, 25, 50, 75, 100, and $200 \mu g/L$. The pH of the solutions was adjusted to the selected pH using 0.1 N HCl or 0.1 N NaOH. One gram of the adsorbent was added to the solution, and the solution was shaken for a specific period of time. The pH and As concentration of the solutions were measured at the beginning and end of the experiment. The results obtained from these experiments were modeled using Freundlich and Langmuir isotherms.

5.2.3.1 Breakthrough Column Experiments

Breakthrough (BT) column experiments were conducted to determine the time needed by the two selected adsorbents to reach saturation. Breakthrough experiments were performed on a 12" PVC column with an outside diameter of 3/8" (1.72 cm) and an inside diameter of 0.460" (1.17 cm). A glass-wool plug was placed at the bottom of the column to prevent loss of the adsorbent. The adsorbent was carefully added to the column, and the entrapped air was removed by fluidizing the column with deionized (DI) water. The system was run in upflow mode by gravity. Arsenic (As) solutions with an initial concentration of $500 \,\mu$ g/L of arsenic (5 +) were used to run these experiments. The pH of the solutions was adjusted with 0.1 N HCl or 0.1 N NaOH to a pH between 7.0 and 7.5. The amount of adsorbent used depended on the height of the column with which each experiment was to be conducted. Two different column heights or empty bed volume (EBV) were selected. The As solution was fed to the column continuously without recirculation, and sample solutions were taken periodically at the exit of the column to measure the changes in the As concentration. An inductively coupled plasma optical-emission spectroscopy (ICP-OES) instrument was used to measure the As concentrations. Release of As at the exit of the column meant that the saturation of the adsorbent material had already started.

5.2.3.2 Adsorbent Regeneration

Regeneration of selected adsorbent materials was made by passing a 2% NaOH (0.5 M) solution (Reagent Chemical) through the BT column. The column contained the adsorbent material already exhausted or saturated with As. The 2% NaOH solution was passed by gravity in down-flow mode through the column. Solution samples were collected at the exit of the column and were analyzed for As until the As concentration was minimal. Once the regeneration process was finished, the regenerated adsorbent material was washed with DI water and was ready for reuse.

5.3 Results and Discussion

5.3.1 Adsorbents Properties

Figure 5-1 shows the nitrogen adsorption and desorption isotherms of Solgel AA and modified AA adsorbents. The mesoporous nature of the adsorbent can be observed by the shape of the adsorption/desorption hysteresis. Solgel AA and Nanogel AA hysteresis suggests a wide range of pore sizes in the surface. The shape of Aerogel AA hysteresis suggests that drying the wet-gel spheres with supercritical CO_2 causes the pores inside of the surface to become larger. Pores in CaO-AA tend to be of much smaller diameter. This is confirmed by the BJH and H-K average pore size measured by the ASAP instrument (Table 5-1). The values of BET-specific



FIGURE 5-1 Nitrogen adsorption/desorption isotherms on Solgel AA and modified AA adsorbents.

surface area and pore volume for the adsorbents are also presented in Table 5–1. Based on the BET-specific surface area the adsorbent materials follow the order Solgel AA > Aerogel AA > Nanogel AA > CaO-AA. Even though the BET surface area of Nanogel AA and Xerogel AA are smaller than the surface area of their precursor (Solgel AA), their pore size and pore volume are bigger. This behavior may play an important role in the adsorption capability of the adsorbents. Based on the H-K pore size the adsorbents follow the order Aerogel AA > Nanogel > Solgel AA > CaO-AA.

SEM images for Solgel AA and modified AA adsorbents are presented in Figure 5–2. Small amorphous clusters can be seen in the SEM image of Solgel AA. The main components of these adsorbents are alumina and oxide.

Table 5–1	Surface Properties of	Adsorbent Materials	Produced in the Laborate	ory
-----------	-----------------------	---------------------	--------------------------	-----

Solgel AA	CaO-AA	Nanogel AA	Aerogel AA
308.01	92.77	278.11	297.19
48.70	48.09	53.67	63.14
0.474	0.129	0.463	0.589
59.84	55.59	68.26	87.90
0.468	0.129	0.459	0.589
	Solgel AA 308.01 48.70 0.474 59.84 0.468	Solgel AA CaO-AA 308.01 92.77 48.70 48.09 0.474 0.129 59.84 55.59 0.468 0.129	Solgel AACaO-AANanogel AA308.0192.77278.1148.7048.0953.670.4740.1290.46359.8455.5968.260.4680.1290.459



FIGURE 5–2 SEM image of (A) Solgel AA, (B) CaO-AA, (C) Nanogel AA, (D) Aerogel AA. (Scale bar = 5μ m.)

SEM images of modified Solgel AA adsorbents reveal smaller and less heterogeneous clusters than the clusters in Solgel AA. This observation is particularly notorious in CaO-AA adsorbent. Zinc oxide crystals are randomly distributed in Nanogel AA. Homogeneous distribution of clusters is evident in Aerogel AA. This is possible because air was completely extracted from the pores and void spaces of the adsorbent by drying it with supercritical CO_2 . The slope of the Aerogel AA's hysteresis in Figure 5–1 is characteristic of the presence of mesopores in the adsorbent.

5.3.2 Arsenic Adsorption Equilibrium

5.3.2.1 Solution pH as a Function of Adsorption Time

Analysis of pH as a function of adsorption time was conducted to determine the minimum time required for obtaining equilibrium adsorption. Figure 5–3 displays the experimental data for arsenic solutions with an initial concentration of $100 \mu g/L$ adsorbed onto one gram of Solgel AA and modified AA adsorbents. Solgel AA showed an initial pH of 6.68. Even though the Solgel AA's pH seemed to stabilize during the first 10 hours, it completely stabilized after about 50 hours. Observed behavior demonstrated that this adsorbent may have high alkalinity or internal buffer capacity, which helped to maintain the small pH changes during the first hours. After 72 hours the final pH of Solgel AA was 6.58.

During the first 10 hours, the initial pH of CaO-AA adsorbent decreased from 6.7 to 6.5 (Figure 5–3). It then increased continuous and steadily throughout the experiment. After 50 hours CaO-AA had a pH of about 10.0. After 72 hours the pH for this adsorbent was about 10.6 and had not completely stabilized.

Variations in pH for Nanogel AA and Aerogel AA (Figure 5–3) were minimal with respect to their precursor Solgel AA (Figure 5–3). The initial pH for these two adsorbents was about 6.8; the final pH after 72 hours was slightly higher than 6.0; at this time they were completely stabilized. From the four adsorbents produced in the laboratory, Solgel AA, Nanogel AA, and



FIGURE 5-3 Time to reach adsorption equilibrium in Solgel AA and modified AA adsorbents.

Aerogel AA showed that they were more stable materials in the presence of As solution. This may be an important factor to consider when treating drinking water, because the pH of the treated water would not need to be adjusted after treatment. Based on the time to reach equilibrium, 60, 72, and 72 hours were considered the appropriate number of hours for Solgel AA, Nanogel, and Aerogel, respectively, to conduct the next set of adsorption experiments.

5.3.2.2 As Adsorbent Amount as a Function of pH

Experiments on the adsorption equilibrium of As as a function of pH were conducted on Solgel AA and modified AA adsorbents to determine the optimum pH at which the highest As adsorption occurs. An initial As concentration of 50 or $100 \,\mu$ g/L was used for all the experimental runs. The initial and final pH was monitored for each of the arsenic solution concentrations tested. The As adsorbent amount (μ g As/g adsorbent) of the adsorbent materials was determined by mass balance using the equation:

$$q_{\rm e} = (C_{\rm o} - C_{\rm e}) * V/m \tag{5.1}$$

where C_0 is the initial As concentration (μ g/L), C_e is the concentration at equilibrium (μ g/L), V is the volume of As solution (mL), and m is the mass of adsorbent (g).

The variation of the quantity of As adsorbed with final pH is illustrated in Figure 5–4. In all cases covering pH from 2.0 to 10.0, the equilibrium concentrations were lower than the initial solution concentrations of 50 or $100 \,\mu$ g/L, suggesting that removal of As by the adsorbent materials occurred.



FIGURE 5-4 Effect of pH on quantity of As adsorbed by AA adsorbents.

The maximum As adsorption capacity obtained by precursor Solgel AA was $5.4 \mu g/g$ at pH 2.0. The minimum adsorption capacity of $0.01 \mu g/g$ was obtained by CaO-AA at pH 2.0. Nanogel showed an average As adsorption capacity of $9.0 \mu g/g$ at any of the pH values tested, except at pH 2.0. Its adsorption capacity was more than two times that of Solgel AA. In general, even though the adsorption capacity of most of the adsorbent materials was pH independent, Solgel AA and Nanogel AA showed superior performance at almost any pH value. The buffer capacity of these two adsorbents may be a key component that reflects in their performance. Based on the observed adsorption capacity, the adsorbent materials followed the order Nanogel AA > Solgel AA > Aerogel > CaO-AA. The poor performance observed in CaO-AA may be explained by the high pH at which it equilibrates (Figure 5–4). Based on the observed pH, it may be assumed that CaO-AA becomes a good adsorbent material at pH higher than 10.0. However, most of the existing environmental conditions are at a pH between 5.0 and 8.0. Therefore using CaO-AA as adsorbent will require the addition of chemicals to increase the pH of the system to at least 10.0.

The observed results demonstrated that As removal capacity depends mostly on the pH properties of the adsorbent materials and less on the surface area properties of the material. This can be explained by the amphoteric properties of As, the solubility of which depends mainly on the protonated species that are present at the specific pH of the solution. At a pH of 5.0-8.0, which is typical of natural waters, arsenate, As(V), is predominantly present in the form of $H_2AsO_4^-$ and $H_2AsO_4^{2-}$ species, while arsenite, As(III), remains fully protonated and is present as the HAsO₂ neutral species. At a pH of 2.0, As(V) is present in the form of H₃AsO₄ neutral species and $H_2AsO_4^-$ species and As(III) is present in the form of HAsO₂ species. At a pH of 10.0, As(V) is present in the form of $H_2AsO_4^{2-}$ species and As(III) is present in the form of $H_2AsO_3^-$ and $HAsO_3^{-2}$ species [4,13]. Results indicate that the addition of zinc oxide nanoparticles during the hydrolysis step in the Yoldas process enhances the adsorption properties of the resulting adsorbent. This is turn contributes to increase the adsorption capacity of the new material. Even though chemical modification of already produced Solgel AA helps to increase its surface area, an improvement of the Yoldas process by itself, i.e., by incorporating nanoparticles during the process of producing the adsorbent, may be even more convenient to help increase the adsorption capacity of this material.

Considering a typical sample of drinking water with neutral or slightly basic pH and the advantage of having an adsorbent material that is pH independent, the three adsorbents selected for showing the best properties for maximum adsorption capacity were Nanogel AA > Solgel AA > Aerogel AA. The fact that Nanogel AA adsorbed more As than Solgel AA suggests that the novel sol-gel method to produce AA is important and successful. The high As adsorption capacity of Nanogel AA can be attributed to the enhanced pore textural and surface properties but mostly to its relatively high pH independency.

5.3.2.3 Effect of Initial Concentration on As Removal Efficiency

The effect of initial concentration on the adsorption of As by Solgel AA and modified adsorbents was analyzed by calculating the percentage of As removal at different initial concentrations. The removal efficiency was calculated based on the difference of initial and final As



FIGURE 5-5 Removal efficiency of As for produced adsorbents.

concentration at the end of the equilibrium process. Figure 5–5 displays the As removal efficiency as a function of initial As solution concentration.

For all the adsorbents, As removal efficiency increased with the increase of initial As concentration and reached a maximum at high initial As concentrations. Nanogel AA removed almost 100% As from solutions with high As concentrations; it removed 95% As from an As solution of $10 \mu g/L$. Performance of Nanogel AA may be improved by increasing the concentration of ZnO nanoparticles during the beads' formation process. Solgel AA removed up to 98% and 97% from a solution containing an initial As concentration of $50 \mu g/L$ and $10 \mu g/L$, respectively. This corresponded to an adsorption capacity of $4.5 \mu g/g$ and $1.0 \mu g/g$, respectively (Figure 5–5). Only 3% As remained in the solution containing $10 \mu g/L$ As, which is the MCL for As in drinking water. Aerogel removed 90% As from a $10 \mu g/L$ As solution. These equilibrium experimental results demonstrated that the Solgel AA and modified Solgel AA adsorbents are promising adsorbent materials for As removal.

5.3.2.4 Adsorption Equilibrium

The As adsorption equilibrium data for selected adsorbents are presented in Figure 5–6. Results were obtained at selected pH for solutions with initial As concentrations varying from 5 to $200 \mu g/L$. Freundlich and Langmuir adsorption isotherm models were used to correlate the experimental data. The Freundlich and Langmuir isotherms are two-parameter models that provide useful information to predict the adsorption capacity of the adsorbent materials at given feed concentrations. Correlation of the equilibrium data provide important parametric information that can be used to optimize the design of the adsorption system for the removal of As.

Table 5–2 presents the Freundlich and Langmuir equations as well as the estimated adsorption isotherm parameters for the three AA adsorbents produced in the lab. In the two models, $q_{\rm m}$ (mg/g) is the amount of arsenic adsorbed per weight of adsorbent at equilibrium, and $C_{\rm e}$ (mg/L) is the equilibrium concentration of the adsorbate in the solution. In the Freundlich model, *K* is the Freundlich constant representing the relative adsorption affinity of the adsorbent toward the adsorbate molecules, and *n* represents the heterogeneity of the adsorbent (Table 5–2). In the Langmuir model, *b* (L/mg) is the Langmuir constant that characterizes the adsorbent–adsorbate interaction, and *a* (mg/g) represents the monolayer adsorption capacity (Table 5–2).

The Freundlich adsorption isotherm fitted well the As adsorption on the three AA adsorbents. The *R*-squared values for Solgel AA, Nanogel AA, and Aerogel AA are 0.9582, 0.9717, and 0.9731, respectively. Figure 5–6 shows the fitted model for each of the adsorbents tested.



FIGURE 5–6 Effect of concentration on quantity of As adsorbed for the three adsorbents studied (dashed lines represent the Freundlich model of the adsorption process).

	Freundlich Isotherm $q_e = k * C_e^{1/n}$			Langmu	o*b*C _e /	
Adsorbent	k	n	R ²	<i>a</i> (mg/g)	<i>b</i> (L/mg)	R ²
Solgel AA	4.13505	0.52815	0.9582	2.6702	0.56088	0.8763
Nanogel AA	15.9809	0.37127	0.9717	1.0086	0.84911	0.8812
Aerogel AA	1.26921	0.83381	0.9731	14.144	12.9165	0.9591

 Table 5–2
 Freundlich and Langmuir Isotherm Constants for As Adsorption on Tested

 Adsorbents

Although the Langmuir isotherm model has been reported as a good model to represent the adsorption of As with a variety of adsorbent materials, it failed to correlate the data for As adsorption by the AA adsorbents tested. This may be attributed to changes that occurred to the Solgel AA adsorbent during the different modifications made to the Yoldas process. It has to be taken into consideration that the correlation coefficient accounts only for the linearity of the adsorption process.

5.3.3 Breakthrough Column Analysis

Breakthrough (BT) column experiments were conducted using Solgel AA adsorbent. Behavior of As concentration in the BT column was analyzed as a function of the empty bed volume (EBV) in the column. The value of one EBV was obtained by calculating the volume of the column. Table 5–3 shows the conditions under which the experiments were run. Each experiment was run until the As concentration at the exit of the column was approximately the same as the initial As concentration. It was assumed that at this point the column is saturated and therefore not able to adsorb more As. The capacity of the column was obtained by calculating the total bed volumes (BVs) of water treated before the concentration of As exceeded 10 μ g/L. The cumulative number of BVs was obtained by adding the volume fed to the column at time "t" to the volume that has been already fed at a specific flow rate and by dividing this accumulative sum by one EBV as shown by the equation:

Cumulative BV =
$$(V_t + (Q^* \Delta t))/EBV$$
 (5.2)

where V_t is the volume of solution at time t (mL), Q is the flow rate (mL/min), Δt is the interval of time (min), and EBV is the volume of the empty column at the particular height in the column (mL).

During the elapsed time of the BT column experiments, the pH of the effluent water for the two samples increased slowly until it reached a typical pH of drinking water (pH = 7.3). The average pH of the solution fed to the column was approximately 6.7. The total duration of the BT experiment was 208 hours and 178 hours for Solgel 1 and 2, respectively. The column had a small diameter and was not large, and therefore gravity was sufficient to feed the As solution to the column.

The As concentration changes at the effluent of the BT column for Solgel AA are presented in Figure 5–7. The EBV for Solgel 1 was 9; for Solgel 2 it was 19, two times the EBV of Solgel 1.

Parameter	Solgel 1	Solgel 2
Initial As concentration (µg/L)	514	504
Mass of adsorbent in column (g)	6.8	15.2
EBV (mL)	9	19
Average flow rate (mL/min)	30.4	37.1
Initial solution pH	6.55	6.8

Table 5–3Experimental Conditions for BT ColumnExperiments on Solgel AA Adsorbent



FIGURE 5-7 BK for As removal at different column heights for Solgel AA (dashed line represents MCL of 10µg/L for As).

This means that Solgel 2 would need two times the BV used by Solgel 1 to reach the same As effluent concentration. Due to the small EBV in the Solgel 1 column, the adsorbent started saturating very rapidly and the breakthrough at $10 \mu g/L$ (MCL for As in drinking water, dashed line in Figure 5–7) happened before the first concentration reading was made at the exit of the column (elapsed time less than 5 minutes). The first water sample taken at the effluent already measured an As concentration of $193 \mu g/L$ (19 times the MCL for As). Solgel 1 reached saturation, i.e., As effluent concentration equal to As influent concentration of $514 \mu g/L$, after treating about 40,000 BV of water (368 liters of water).

The EBV of 19 in Solgel 2 column was still small and did not allow observing breakthrough at a concentration of As of 10 µg/L. The first As reading made at the effluent of the column was already 93 µg/L (9.3 times the MCL for As) (elapsed time less than 5 minutes). Interestingly, the first As concentration reading in the Solgel 2 column (193 µg/L) was about two times the first As concentration reading in the Solgel 2 column (93 µg/L), as expected. Based on the 1:2 ratio in both EBV and As concentration at the effluent, it is expected that Solgel 2 will treat a total of about 80,000 BV (736 liters of water) before reaching column saturation (As concentration of 514 µg/L). In order to be able to observe column breakthrough at 10 µg/L of As, under the given process and parametric conditions, the EBV of the Solgel 1 column would have to be two times the EBV of the Solgel 2 column (EBV \approx 38).

Additional BT column experiments for Solgel AA with prehydration (to improve removal capacity), different flow rate, EBV, column length, particle size, and adsorbate concentration are needed to determine the optimal process conditions that will extend the good performance of this novel and promising adsorbent material prior to saturation. A combined experimental design will allow combining process and parametric conditions in the laboratory to reduce the number of experiments needed as well as cost. Development of a mathematical model specific for this particular process could also help to optimize the BT column performance for further pilot test-scale experiments.

5.3.4 Adsorbent Regeneration

Regeneration of the adsorbent material is important in relation to the cost-effectiveness and user-friendliness of the process. After adsorbing arsenic for a certain time in the BT column, the Solgel AA adsorbent material became exhausted. The exhausted adsorbent must either be replaced or regenerated to restore its adsorption capacity. The regeneration of Solgel 1 and Solgel 2 was effected by passing a strong base solution (2% NaOH) through the column to wash the adsorbent until no arsenic was detected at the exit of the column. In the regeneration process, desorption of arsenic from the adsorbent took place by increasing the hydroxyl ions in contact with the adsorbent in the column. The high pH of NaOH (pH > 10.5) caused arsenic in the form of the arsenate ion ($HAsO_4^-$) to lose adsorptive attraction to the alumina and become repelled, following the reaction:

$$Alumina.H_2AsO_4^- + 2NaOH \rightarrow Alumina.NaOH + NaHAsO_4 + H_2O$$
(5.3)

Reuse of the AA adsorbent material was possible after regeneration with 2% NaOH solution. Equilibrium experiments using the regenerated material were conducted following the same procedure. A 100 mL solution with a given As concentration was prepared. The pH of the solution was adjusted to the selected pH. Then 1 g of the adsorbent was added to the solution and the solution was shaken for a specific period of time. The pH and As concentration of the solution were measured at the beginning and end of the experiment. The adsorption capacity of regenerated Solgel AA was not calculated, because the As concentration of the solution after reaching equilibrium was higher than the initial concentration. This may be explained by As from the previous run being still contained in the adsorbent material. This might be possible if As was strongly bonded to Solgel AA. In this case, additional NaOH 2% or NaOH with a higher concentration may be needed to completely wash out the As.

5.4 Conclusions

A novel Solgel AA adsorbent produced in the laboratory and its modifications using CaO (CaO-AA) ZnO nanoparticles (Nanogel AA) and supercritical CO_2 drying (Aerogel AA) were studied to explore the feasibility of applying these adsorbents for arsenic removal from drinking water. Equilibrium and BT column experiments were conducted to determine the removal efficiency and saturation of the adsorbent material. The latter was recovered for reuse.

The following conclusions were drawn from this experimental study:

- The BET specific surface area of the adsorbent tested followed the order Solgel AA > Aerogel AA > Nanogel AA > CaO-AA.
- From the adsorbents tested, Solgel AA, Nanogel AA, and Aerogel AA appeared to be more stable materials in the presence of As solution. This is an important factor to consider when treating drinking water.
- The adsorbents that showed the best properties for maximum adsorption capacity were Nanogel AA > Solgel AA > Aerogel AA. The high As adsorption capacity of Nanogel AA

is attributed to the enhanced pore textural and surface properties as well as to the pH independency of the adsorbent.

- For all the adsorbents produced in the laboratory, As removal efficiency increased with the increase of initial As concentration.
- The Freundlich adsorption isotherm fitted well the As adsorption on the AA adsorbents tested.
- Additional BT column experiments for Solgel AA with different flow rates, EBVs, column lengths, particle sizes, and adsorbate concentrations are needed to determine the optimal process conditions that will extend the performance of the adsorbent prior to saturation.

Acknowledgments

The financial support of this project provided by the Anita Mochen Loya Innovation Fund in Engineering at the University of Texas at El Paso is gratefully acknowledged. The in-kind contribution provided by the Chemical Engineering Department at New Mexico State University is also gratefully acknowledged.

References

- [1] Duker AA, Carranza EJM, Hale M. Arsenic geochemistry and health. Environ Int 2005;31:631-41.
- [2] Ghosh P, Banerjee M, Giri AK, Ray K. Toxicogenomics of arsenic: classical ideas and recent advances. Mutat Res Rev Mutat Res 2008;659:293–301.
- [3] Rosado JL, Ronquillo D, Kordas K, Rojas O, Alatorre J, Lopez P, et al. Arsenic exposure and cognitive performance in Mexican school children. Environ Health Perspect 2007;115:1371–5.
- [4] Camacho LM, Gutiérrez M, Alarcon-Herrera MT, Villalba MD, Deng S. Occurrence and treatment of arsenic in groundwater and soil in northern Mexico and southwestern USA. Chemosphere 2011;83:211–25.
- [5] Feng H, Gao Y, Zhao L, Wei Y, Li Y, Wei W, et al. Biomarkers of renal toxicity caused by exposure to arsenic in drinking water. Environ Toxicol Pharmacol 2013;35:495–501.
- [6] World Health Organization (WHO) 3rd ed. Guidelines for drinking water quality. Recommendations, vol. 1 Geneva: World Health Organization; 2004. pp. 306–8.
- [7] Hao J, Han MJ, Meng XG. Preparation and characterization of thiol-functionalized activated alumina for arsenite removal from water. J Hazard Mater 2009;167:1215–21.
- [8] Mohan D, Pittman Jr. CU. Arsenic removal from water/wastewater using adsorbents—a critical review. J Hazard Mater 2007;142:1-53.
- [9] Litter MI, Morgada ME, Bundschuh J. Possible treatments for arsenic removal in Latin American waters for human consumption. Environ Pollut 2010;158(5):1105–18.
- [10] Kobya M, Gebologlu U, Ulu F, Oncel S, Demirbas E. Removal of arsenic from drinking water by electrocoagulation using Fe and Al electrodes. Electrochim Acta 2011;56:5060–70.
- [11] Tsair-Fuh L, Jun-Kun W. Adsorption of arsenite and arsenate within activated alumina grains: equilibrium and kinetics. Water Res 2001;35:2049–57.
- [12] Manna B, Ghosh UC. Adsorption of arsenic from aqueous solution on synthetic hydrous stannic oxide. J Hazard Mater 2007;144:522–31.
- [13] Camacho LM, Parra RR, Deng S. Arsenic removal from groundwater by MnO₂-modified natural clinoptilolite zeolite: effect of pH and initial feed concentration. J Hazard Mater 2011;189:286–93.
- [14] Giles DE, Mohapatra M, Issa TB, Anand S, Singh P. Iron and aluminium based adsorption strategies for removing arsenic from water. J Environ Manage 2011;92(12):3011–22.

- [15] Singh TS, Pant KK. Equilibrium, kinetics and thermodynamic studies for adsorption of As(III) on activated alumina. Sep Purif Technol 2004;36(2):139–47.
- [16] Deng S, Lin YS. Sol-gel preparation and properties of alumina adsorbents for gas separation. AIChE J 1995;41(3):559–70.
- [17] Deng S, Viswanathan V, Candelaria D. Sol-gel derived mesoporous alumina for fluoride and arsenic removal from drinking water. N M J Sci 2006;44:183–202.
- [18] Yoldas BE. Alumina gels that form porous transparent Al2O3. J Mater Sci 1975;10:1856-60.

6

Health Effects Chronic Arsenic Toxicity

D.N. Guha Mazumder

DNGM RESEARCH FOUNDATION, NEW ALIPORE, KOLKATA, INDIA; DEPARTMENT OF MEDICINE & GASTROENTEROLOGY, INSTITUTE OF POST GRADUATE MEDICAL EDUCATION & RESEARCH, KOLKATA, INDIA

CHAPTER OUTLINE

6.1	Introduction	137
6.2	Dermatological Manifestations	138
6.3	Epidemiological Study of Dermatological Manifestations	144
6.4	Pregnancy Outcome	162
6.5	Arsenicosis and Cancer	163
6.6	Diagnosis	165
6.7	Treatment	168
Ref	erences	171

6.1 Introduction

Arsenic, a metalloid, occurs naturally, being the twentieth most abundant element in Earth's crust and is a component of more than 245 minerals. The inorganic forms are toxic to human health and consist mostly of arsenite and arsenate compounds. Humans are exposed to arsenic primarily from air, food, and water. Drinking water may be contaminated with arsenic from arsenical pesticide, natural mineral deposits, or improperly disposed arsenical chemicals. Arsenic in drinking water has been recognized as a major public health problem in several regions of the world. Reports of arsenic contamination in ground water are emerging from more than 20 countries as and when cases of chronic arsenic toxicity are reported. Major countries affected are Bangladesh, India, Taiwan, China, Mexico, Argentina, Chile, and the USA.

Human health effects of chronic arsenic toxicity are designated by the term arsenicosis, which was first coined by our group [1] and later used by the World Health Organization (WHO) [2] to imply a chronic disease caused by prolonged exposure of arsenic in humans. Previously, the

condition was described as arseniasis, arsenism, arsenicism, etc. Most of the reports of chronic arsenic exposure in humans focus attention on skin manifestations because of their diagnostic specificity. However, data derived from population-based studies, clinical case series, and reports relating to intake of inorganic arsenic in drinking water, medications or occupational and environmental exposure show that chronic arsenic exposure adversely affects multi-organ systems of the human body. The symptoms of chronic arsenic toxicity (arsenicosis) are insidious in onset and are dependent on the magnitude of the dose and duration of its exposure. The duration of the patient's arsenic exposure with the date of onset of symptoms does not follow a particular timeframe. It is even the case that not all members of an affected family show clinical effects. The reason for such variation of disease expression is an enigma. Further, there is a wide variation in the incidence of chronic arsenicosis in an affected population. It needs to be mentioned that a few epidemiological studies in the USA reported that none of the population exposed to environmental arsenic showed any clinical manifestation of chronic arsenic toxicity [3,4].

Over and above dermatological manifestations, chronic arsenic exposure in humans produces various non-specific symptoms owing to the fact that arsenic affects multi-organ systems in the body. However, most of the reports of chronic arsenic exposure in humans focus attention on skin manifestations because of their diagnostic specificity. In a field guide for detection, management, and surveillance relating to arsenicosis, the condition has been defined by WHO as "a chronic health condition arising from prolonged ingestion of arsenic above the safe dose for at least six months, usually manifested by characteristic skin lesions of melanosis (hyperpigmentation) and keratosis, occurring alone or in combination with or without involvement of internal organs" [2]. The appearance of lesions usually follows temporal progression, beginning with hyperpigmentation, and then progressing to palmar-plantar keratosis, which is dependent on the magnitude of the dose and the duration of exposure.

6.2 Dermatological Manifestations

The specific skin lesions of chronic arsenic toxicity are characterized by hyperpigmentation, depigmentation, keratosis, and Bowen's disease.

6.2.1 Hyperpigmentation

The hyperpigmentation of chronic arsenic poisoning commonly manifests as a finely freckled, "raindrop" pattern of pigmentation or depigmentation that is particularly pronounced on the trunk and extremities and has a bilateral symmetrical distribution. This occurs anywhere in the body, and is particularly marked in non-exposed parts of the body, such as trunk, buttocks, and thighs. Pigmentation may sometimes be blotchy and involve mucous membranes such as the undersurface of the tongue or buccal mucosa [1,5–11]. The raindrop appearance results from the presence of numerous rounded hyperpigmented or hypopigmented macules (typically 2–4 mm in diameter) widely dispersed against a tan-to-brown hyperpigmented background [7]. Hyperpigmentation is often present combined with hypopigmented spots [5,12]. Other patterns include diffuse hyperpigmentation (melanosis) [5,9], and localized or patchy pigmentation, particularly affecting skin folds [7,13,14], although less commonly seen. Pigmentation is not histopathologically related to arsenical hyperkeratosis, nor is it a direct precursor of cancer [15].

6.2.2 Hypopigmentation

Hypopigmentation or so-called leukoderma or leukomelanosis [9,10] is found in chronic arsenic exposed people in which the hypopigmented macules take a spotty, white appearance. This follows the same distribution as hyperpigmentation. These spots may be present in the absence of hyperpigmentation [8]. Hypopigmentation, which was described as part of the raindrop-like appearance [16] and as a separate entity named leukoderma [17], was considered the earliest cutaneous sign of chronic arsenic toxicity [8]. However, others suggest that leukomelanosis appears to occur in an arsenicosis patient following stoppage of drinking arseniccontaminated water for some duration [9,15]. (See Figures 6–1–6–5.)



FIGURE 6-1 Melanosis (diffuse pigmentation) in the back.



FIGURE 6-2 Pigmentation (spotty, mild).



FIGURE 6–3 Pigmentation (moderate).



FIGURE 6-4 Pigmentation (severe).



```
FIGURE 6-5 Leukomelanosis.
```

6.2.3 Hyperkeratosis

Arsenical hyperkeratosis appears predominantly on the palms and the plantar aspect of the feet, although involvements of the dorsum of the extremities and the trunk have also been described. They are found most frequently on the thenar eminence and the lateral borders of palms and fingers, and on the soles and heels and toes. In the early stages, the involved skin might have an indurated, grit-like character that can be best appreciated by palpation; how-ever, the lesions usually advance to form raised, punctate, 2–4 mm wart-like keratoses that are readily visible [7]. Occasional lesions might be larger (0.5 to 1 cm) and have a nodular or horny appearance occurring in the palm or dorsum of the feet. In severe cases, the hands and soles present with diffuse verrucous lesions. Cracks and fissures may be severe in the soles [1,5–9,17–22]. A scoring system has been developed for assessing the severity of skin lesions of pigmentation and keratosis (Table 6–1) [23]. (See also Figures 6–6–6–9.)

Table 6–1	Dermatological Criteria and Gradation of Skin Lesions of Chronic Arsenic
Toxicity (So	coring System) [23]

Pigmentation (Score)					
Mild (1)	Moderate (2)	Severe (3)			
Diffuse melanosis, mild spotty pigmentation, leukomelanosis	Moderate spotty pigmentation	Blotchy pigmentation, pigmentation of undersurface of tongue, buccal mucosa			
Keratosis (score)					
Slight thickening or minute papules (<2 mm) in palm and soles	Multiple raised keratosis papules (2 to 5 mm) on palm and soles with diffuse thickening	Diffuse severe thickening, large discrete or confluent keratotic elevations (>5 mm) on palm and soles (also dorsum of extremity and trunk)			

Maximum total skin score = 6.



FIGURE 6-6 Mild keratosis.



FIGURE 6-7 Moderate keratosis.



FIGURE 6-8 Severe keratosis.



FIGURE 6–9 Severe verrucous keratosis.



FIGURE 6-10 Bowen's disease in the back.

Histological examination of skin biopsy material of a keratotic skin lesion shows extensive hyperkeratosis and parakeratosis. Parakeratosis is more prominent along either side of the sweat duct orifice. The epidermis is markedly acanthotic with irregular downward prolongation of enlargement of the rete ridges [5,7]. Varying degrees of squamous cell hyperplasia, without any basophilic degeneration of the dermis, are usually seen. In some cases, there might be evidence of cellular atypia, mitotic figures, in large vacuolated epidermal cells [24–26]. Yeh [6] classified two types of arsenical keratosis, a benign type A, further subdivided into two groups, one with no cell atypia and the other with mild cell atypia, and a malignant type B [15].

6.2.4 Bowen's Disease

The skin lesion of Bowen's disease appears as sharply demarcated round plaques or has an irregular polycyclic lenticular configuration. The lesions are usually erythematous, pigmented, crustated, fissured, and keratotic. Some may be nodular, ulcerated, or eroded. The diameter of the lesions may vary from 0.8 to 3.5 cm. The lesions are firm and indurated and rough on palpation [6]. Clinically, arsenic-induced Bowen's disease can be distinguished from non-arsenical Bowen's disease by its multiple and recrudescent lesions and predominant occurrence on the sun-protected areas of the skin [27,28]. The disease is generally believed to be an intradermal carcinoma from its onset [29,30]. Histologically, Bowen's disease can be divided into hypertrophic and atrophic types. Common features in both varieties are windblown appearance of the epithelial cells, with multinucleated giant cells, vacuolated cells, and dyskeratosis; abnormal mitotic figures may be present at all levels of the epidermis. In classical Bowen's disease, the epidermis is composed of atypical cells haphazardly arranged at all levels. In the hypertrophic forms, papillomatosis and acanthosis are marked. In atrophic forms, areas of keratosis are usually thin and dense: papillomatosis, although present, is not prominent, and sometimes even acanthosis is not significant [6]. (See Figure 6–10.)

6.3 Epidemiological Study of Dermatological Manifestations

The first large epidemiological study was carried out to ascertain the prevalence of arsenical skin lesions in a population of 40,421 out of a total population of 103,154 in 37 villages of southwest Taiwan where people had been drinking arsenic-contaminated artesian well water for more than 45 years. In the villages surveyed, the arsenic content of the well water examined ranged from 10 to 1820 μ g/L. The overall prevalence rates for keratosis and hyperpigmentation were 7.10 and 18.35 per 100, respectively. The male to female ratio for both conditions was 1.1:1. The prevalence rates for hyperpigmentation increased steadily with age for males but for females a peak appeared at age group 50–59, followed by a gradual downturn in rates. The prevalence rates for keratosis for both sex increased up to age 70 and then declined. However, arsenic levels in drinking water were reported by villages in Taiwan. The youngest patient with hyperpigmentation in this study was aged 3 years and with keratosis, 4 years [19].

Another epidemiological study was carried out in the Región Lagunera, Mexico, where ground water in wells was found to be contaminated with arsenic ranging from 240 to $1000 \mu g/L$. Arsenic was tested by use of chemical methodology. For this study, 296 exposed and 318 unexposed participants were selected from two towns having well water arsenic concentrations of $410 \mu g/L$ and $5 \mu g/L$ respectively. Every member of each family of both exposed and unexposed households of the respective towns were selected by systematic sampling beginning at a random point, every third house being surveyed. In the exposed population, 64 out of 296 (21.6%), while in the unexposed control population 7 out of 318 (2.2%), showed at least one of the signs of arsenical skin lesioning. The discovery of such lesioning in the control population did not match with observations of two earlier studies in Chile and Taiwan where there were no affected people in the control populations chosen. The authors did not find any sex difference in regard to the prevalence of either hyperpigmentation or keratosis. The shortest time of exposure, after which lesions were detected, was 8 years for hypopigmentation, 12 years for hyperpigmentation and palmoplantar keratosis, and 25 years for popular keratosis [8].

The first population-based survey with individual arsenic exposure data from drinking water sources was carried out in the district of South 24 Parganas, West Bengal, India, on 7683 participants (4093 female and 3590 male) out of a total population of 150,457 to assess the dose-response relationship and arsenical skin lesioning [12]. Water samples were collected from all recruited households of the participants and tested for the presence of arsenic by flow-injection hydride generation atomic absorption spectrophotometry. The tubewell arsenic concentration ranged from non-detectable to $3200 \,\mu$ g/L. The overall prevalence rates for keratosis and hyperpigmentation were 3.64 and 8.82 per 100, respectively. Prevalence of keratosis and hyperpigmentation was examined by water arsenic level. A clear relationship was apparent between water levels of arsenic and keratosis and hyperpigmentation. Age-adjusted prevalence of keratosis and hyperpigmentation was strongly related to water arsenic levels, rising from zero and 0.3 in the lowest exposure level (<50 μ g/L), to 8.3 and 11.5 per 100 respectively for females drinking water containing >800 μ g/L arsenic, and increasing from 0.2 and 0.4 per 100 in the lowest exposure category to 10.7 and 22.7 per 100 respectively for males in the highest exposure level (>800 μ g/L). The prevalence of skin lesions was also examined by daily dose per body weight (μ g/kg/day). This showed that men had roughly two to three times the prevalence of keratosis and pigmentation compared to women apparently ingesting the same dose of arsenic from drinking water. Subjects who were below 80% of the standard body weight for their age and sex had a 1.6-fold increase in the prevalence of keratosis, suggesting that malnutrition might play some role in increasing susceptibility. A nested case-control study was further carried out using detailed lifetime (at least 20 years) exposure assessment among the above-mentioned study population [31]. Lifetime water arsenic exposure data could be obtained in a total of 192 cases having definite evidence of arsenical skin lesions and 213 age-and sex-matched controls. The lowest peak level of arsenic ingested by a confirmed case was 115 µg/L. Strong dose-response gradients with both peak and average arsenic water concentrations were also observed [31].

The arsenic crisis in Bangladesh is another example of a severe case of environmental toxicity. Fifty out of 65 districts, involving nearly 85% of the total land area of Bangladesh, have arsenic present in ground water. It is suspected that about 25 million people are exposed to arsenic-contaminated ground water [32]. A cross-sectional study was conducted in Bangladesh in which arsenic-related skin lesions were assessed in 1481 subjects (903 males and 578 females) in four rural villages with arsenic exposure data (BDL below the detectable limit] to 2040 µg/L) related to drinking water in take by the individuals [33]. A total of 430 subjects had skin lesions (keratosis, hyperpigmentation, or hypopigmentation). Individual exposure assessment could only be estimated by present levels and in terms of a dose index, i.e., arsenic levels divided by individual body weight. The crude overall prevalence rate for skin lesions was 29/100. After age adjustment to the world population, the prevalence rate was 30.1/100 for males and 26.5/100 for females. In the males in the lowest exposure category $(<150 \,\mu g/L)$ the age-adjusted prevalence of any arsenic-associated skin lesion was 18.6/100 and increased to 37.0/100 in the highest exposure category (>1000 µg/L) (chisquare dose for trend p < 0.001). The corresponding rates for females were 17.9/100 in the lowest exposure category and 24.9 in the highest exposure category (test for trend, p < 0.02). There was a significant trend for the prevalence rate both in relation to exposure levels and to dose index (p < 0.05), regardless of sex. This study shows a higher prevalence rate of arsenic skin lesions in males than females, with a clear dose-response relationship [33]. To ascertain the degree of severity of skin lesions in the arsenic-affected population a crosssectional study was carried out in an arsenic-affected district of West Bengal (Nadia). Out of 10,469 participants examined, 1616 (15.43%) people showed clinical features of arsenical skin lesions. Mean arsenic concentration in drinking water among the cases was 103.46 [standard deviation (SD) \pm 153.28] µg/L while among the controls (participants without skin lesions) it was 73.18 (SD \pm 115.10) μ g/L. Duration of arsenic exposure of the former was 12.47 (SD \pm 7.39) years while that of the later was 12.43 (SD \pm 6.59) years. A scoring system was adopted to classify the degrees of severity of skin manifestations (see Table 6-1). Skin score was mild in 87.5%, while moderate in 11.5%, and severe in 0.87% of the people with arsenical skin disease. Out of a population of 0.84 million people suspected to be exposed to

Symptoms	No. of Cases	%	Signs	No. of Cases	%
Weakness	110	70.5	Pigmentation	156	100.0
Headache	32	20.5	Keratosis	96	61.5
Burning of the eyes	69	44.2	Anemia	74	47.4
Nausea	17	10.9	Hepatomegaly	120	76.9
Pain in the abdomen	60	38.4	Splenomegaly	49	31.4
– epigastric	39	25.0	Ascites	5	3.0
– paraumbilical	21	13.4	Pedal edema	18	11.5
Diarrhea	51	32.6	Sign of lung disease	45	28.8
Cough	89	57.0	Sign of polyneuropathy	21	13.4
 with expectoration 	53	33.9			
 without expectoration 	36	23.1			
Hemoptysis	8	5.1			
Dyspnea	37	23.7			
Paresthesia	74	47.4			

 Table 6–2
 Clinical features of 156 Cases of Chronic Arsenicosis—Hospital-Based Study

 in West Bengal, India [11]

arsenic-contaminated water in the district, 0.14 million people were estimated to be suffering from arsenical skin lesions [23].

6.3.1 Systemic Manifestations

Prolonged ingestion of arsenic-contaminated water is reported to produce not only skin manifestations (pigmentation and keratosis), but also many systemic features such as weakness, anaemia, chronic lung disease, conjunctival congestion, hepatomegaly and non-cirrhotic portal fibrosis, polyneuropathy, chronic diarrhea, dyspepsia, solid edema of limbs, and gangrene of toes. This will be evident from the report of a hospital-based study of the clinical features in 156 cases chronically drinking arsenic-contaminated water in West Bengal, India (Table 6–2) [11]. Further, it is to be noted that the frequency of occurrence of various systemic disease manifestations was found to be significantly higher in people exposed to arsenic but having no skin lesions when compared to those not exposed to arsenic [34].

The global health impact due to arsenic exposure was considered in an epidemiological study carried out in the district of Nadia, one of the severely arsenic-affected districts of West Bengal, India. Out of 10,469 arsenic-exposed participants examined, 1616 patients showed clinical features of arsenical skin lesions and various systemic disease while 8853 participants had features of systemic disease but no skin lesions. Chronic lung disease was found in 207 (12.81%) subjects with skin lesions and in 69 (0.78%) without skin lesions (p < 0.001). Peripheral neuropathy and abdominal pain were found in 257 (15.9%) and 67 (4.15%) subjects in the former group while in 136 (1.5%) and 67 (0.87%) subjects in the latter group, respectively (p < 0.001). Other systemic features found in significantly higher numbers of cases with skin

	Skin Les (<i>n</i> = 161	ions Present 6)	Skin Lesi (n = 885	ons Absent 3)	<i>p</i> -value
Mean peak tubewell As concentration (µg/L)	103.4 ±	153.3	73.2 ±	73.2 ± 115.1	
Duration of exposure to peak concentration (years)	12.5 ±	7.4	12.4 <u>+</u>	6.6	<0.001
Age (mean + SD)	53.36 ±	15.60	33.74. <u>+</u> 15.99		
	n	%	n	%	
Sex					
Male	934	57.80	3213	36.29	<0.001
Female	682	42.20	5640	63.71	NS
Arsenicosis skin score					
Mild (1–2 score)	1415	87.56			
Moderate (3–4 score)	187	11.57			
Severe (>4 score)	14	0.87			
Disease symptoms					
Lung disease	207	12.81	69	0.78	<0.001
Chronic cough	127	7.86	52	0.59	<0.001
Dyspnea	146	9.03	39	0.44	<0.001
Neuropathy (limb pain/tingling, numbness)	257	15.90	136	1.54	<0.001
Pain in abdomen	67	4.15	77	0.87	<0.001
Chronic diarrhea	19	1.18	15	0.17	<0.001
Liver—palpable	5	0.31	0	0.00	<0.001
Ascites	3	0.19	1	0.01	<0.05
Pallor (anemia)	1	0.06	7	0.08	>0.05
Non-pitting edema of limbs	4	0.25	2	0.02	<0.01

Table 6–3Characteristics of Systemic Disease Manifestation in an Arsenic-exposedPopulation with and Without Skin Lesions Nadia District, West Bengal, India [23]

lesions compared to those without such lesions are chronic diarrhea, hepatomegaly, ascites, and non-pitting edema of the limbs (Table 6–3). Thus, systemic disease manifestations occur in arsenic-exposed people in a higher number of cases in whom skin lesions are present as opposed to those who do not have such lesions [23].

Individual disease manifestation associated with chronic arsenic exposure is presented below.

6.3.2 Respiratory Disease

Symptoms of lung disease were reported from Kolkata, West Bengal, in a study of 156 cases of chronic arsenic toxicity with arsenical skin lesions associated with drinking of arsenic-contaminated ground water. Symptoms of chronic cough with and without expectoration and chest signs of lung disease (crepitations and rhonchi) were reported in 89 (57%) and 45 (28.5%) cases, respectively. Lung function tests carried out on 17 patients showed features of

restrictive lung disease in nine (53%) and combined obstructive and restrictive lung disease in seven (41%) cases [11]. In another hospital-based study carried out in Kolkata on 107 arsenic-exposed (arsenic level in drinking water, mean \pm SD: 550 \pm 470 µg/L) and 52 control subjects (arsenic exposure <50 µg/L), 33 cases (30.8%), and four controls (7.6%) had symptoms of lung disease (p < 0.01). Lung function study showed obstructive, restrictive, and mixed obstructive and restrictive patterns in 20 (68.9%), 1 (3.7%), and 8 (27.6%) cases, respectively. Bronchiectasis was diagnosed in three cases on the basis of chest X-ray and HRCT (done on four cases) [35]. Pulmonary disease characteristic of bronchitis was also reported in 86 (23.7%) out of 360 people who were found to have arsenical skin lesions in a village in Bangladesh having an average level of arsenic in drinking water of 240 µg/L [22].

6.3.2.1 Epidemiological Study

Earlier, epidemiological studies carried out in Antafagosta, Chile, from survey data collected between 1968 and 1972 reported that the prevalence of cough among 398 children correlated with mean drinking-water arsenic concentration. In addition, the prevalence of reported cough declined from 38% to 7% following the installation of an arsenic removal plant in Antafagosta (p < 0.001) [36]. Further, the prevalence of bronchiectasis was found to be greater among children with arsenic-affected skin lesions living in Antafagosta compared to children living in the rest of Chile [36]. Autopsies were conducted on five children, with evidence of arsenical skin lesions on the body, who had died between 1968 and 1969 in Antafagosta. Lung tissue was examined in four of these children, with abnormalities being found in all four. Interstitial fibrosis was detected in two of the cases [37]. A cross-sectional survey was also carried out in Antafagosta in which 144 school children with arsenic-induced skin lesions were examined. Bronchopulmonary disease occurred 2.5 times more often in these children (15.9%) compared with children with normal skin (6.9%) [20].

A large cross-sectional epidemiological survey was carried out during 1998-1999 on 6864 participants (all non-smokers) in the district of South 24 Parganas, West Bengal, India, to ascertain the relationship between chronic arsenic exposure and occurrence of lung disease. The participants were clinically examined and interviewed, and the arsenic content of their current drinking water source was measured (arsenic concentration ranged from nondetectable to 3200 µg/L). Among males and females, the prevalence of cough, shortness of breath, and chest sounds (crepitation and/or rhonchi) in the lungs rose with increasing arsenic concentration in drinking water. These respiratory effects were most pronounced in individuals exposed to high levels of arsenic in drinking water who also had skin lesions. Prevalence odds ratio (POR) estimates were markedly increased for participants with arsenic-induced skin lesions who had high levels of arsenic in their drinking water (\geq 500 µg/L) compared with individuals who had normal skin and who were exposed to low levels of drinking-water arsenic $(<50 \mu g/L)$. In participants with skin lesions, the age-adjusted POR estimates for cough were 7.8 for females (95% confidence interval (CI): 3.1-19.5) and 5.0 for males (95% CI: 2.6-9.9); for chest sounds, POR for females was 9.6 (95% CI: 4.0-22.9) and for males 6.9 (95% CI: 3.1-15.0). The POR for shortness of breath in females was 23.2 (95% CI: 5.8-92.8) and in males 3.7 (95%

CI: 1.3–10.6). These results add to the evidence that long-term ingestion of inorganic arsenic can have respiratory effects [38].

A prevalence comparison study of respiratory effects among subjects with and without arsenic exposure through drinking water was conducted in Bangladesh. Exposed participants were recruited through a health awareness campaign program. Unexposed participants were randomly selected, where tubewells were not contaminated with arsenic. A total of 218 individuals participated (94 exposed individuals exhibiting skin lesions; 124 unexposed individuals without skin lesions). The arsenic concentrations ranged from 136 to $1000 \mu g/L$. Only non-smokers without any history of asthma or tuberculosis were recruited and data on respiratory symptoms and signs were collected. The crude prevalence ratios for chronic bronchitis were found to be 1.6 (95% CI: 0.8–3.1) and 10.3 (95% CI: 2.4–43.1) for males and females, respectively. These results also showed respiratory effects on long-term arsenic exposure [39].

A case-control study was carried out during 2000–2003 to ascertain the relationship between respiratory symptoms, lung function, and exposure to arsenic in drinking water in West Bengal among a cohort of 287 participants selected among a source population of 7683 subjects surveyed for arsenic-related skin lesions during 1995–1996. Cases were selected on the basis of their having skin lesions, and primary drinking water sources contained arsenic at levels of $50-500 \mu g/L$, while controls were selected who did not have arsenic-related skin lesions and whose main tubewell water source contained an arsenic concentration of $<50 \mu g/L$. For each case, one control matched on age (within 5 years) and sex was randomly identified from all eligible non-cases. The study was confined to those participants of at least 20 years of age who completed the pulmonary function testing and for whom information on smoking was available. In this study 132 and 155 subjects fulfilled the criteria of cases and controls, respectively. Respiratory symptoms were increased in men with arsenic-caused skin lesions (versus those without lesions), particularly "shortness of breath at night" [odds ratio (OR) = 2.8, 95% CI: 1.1, 7.6] and "morning cough" (OR = 2.8, 95% CI: 1.2, 6.6) in smokers and "shortness of breath ever" (OR = 3.8, 95% CI: 0.7, 20.6) in non-smokers [40].

Pronounced decrements in lung function were observed in males with skin lesions (both non-smokers and smokers) as compared to those without skin lesions. Male smokers had, on average, lower mean residual values for spirometric parameters than male non-smokers. The decreases in FEV₁ and forced vital capacity (FVC) in male non-smokers with skin lesions as compared with non-smoking men without skin lesions were 157.3 mL (95% CI: -24.7, 339.2) for FEV₁ and 188.5 mL (95% CI: 0.6, 376.3) for FVC. In male smokers, the decreases were 271.1 mL (95% CI: 158.0, 384.2) for FEV₁ and 304.1 mL (95% CI: 180.1, 428.1) for FVC. Among women, the respective reductions were 63.2 mL (95% CI: -31.8, 158.2) for FEV₁ and 101.5 mL (95% CI: -8.8, 211.8) for FVC. Decreases in FEV₁ and FVC related to increased water arsenic concentration were observed among men; reduction in mean values from low exposure (arsenic level <100 µg/L) to high exposure (arsenic level ≥400 µg/L) were 194.7 mL (95% CI: 35.5, 353.9) for FEV₁ and 83.8 mL (95% CI: -93.8, 261.5) for FVC in non-smokers and 226.1 mL (95% CI: 45.2, 407.0) for FEV₁ and 247.6 mL (95% CI: 58.3, 436.9) for FVC in smokers. Among women,

the respective reductions were 28.5 mL (95% CI: -71.3, 128.2) for FEV₁ and 7.5 mL (95% CI: -122.4, 137.5) for FCV [40].

In the multivariate linear regression analyses stratified by sex and adjusted for age, height, and smoking, lung function was significantly decreased for signs of arsenic-related skin lesions among men, with a reduction in FEV₁ of 256.2 mL (95% CI: 113.9, 398.4, p value: <0.001) and in FVC of 287.8 mL (95% CI: 134.9, 440.8, p value: <0.001). To further investigate the effects of ingested arsenic on flow, the FEV₁/FVC ratio and the forced expiratory flow between 25 and 75% of forced vital capacity (FEF₂₅₋₇₅) were investigated and significant reductions related to the presence of skin lesions consistent with the findings for FEV_1 and FVC in men were found. Reductions were also observed related to smoking in FEV₁ (156.4 mL; 95% CI: -3.2, 316.0, p value: 0.055) and FVC (119.7 mL: -52.0, 291.4, p value: 0.2) but the effect size was smaller than for presence of skin lesions. Using arsenic levels in water as a measure of exposure instead of skin lesions, significant decreases were found in FEV₁ of 45.0 mL (95% CI: 6.2, 83.9, *p* value: 0.02) and in FVC of 41.4 mL (95% CI: -0.7, 83.5, *p* value: 0.054) in men per 0.1 mg/L increase of arsenic. Potential confounders such as weight, type of house, education, and occupation were assessed in the multivariate models but did not change the estimates for skin lesions or arsenic in water. Interestingly, in women estimates for skin lesions or arsenic in water did not indicate a strong relation with lung function. In this study, consumption of arsenic-contaminated water was found to be associated with respiratory symptoms and reduced lung function in men, particularly among those with skin lesions [40].

To ascertain the incidence of bronchiectasis in the population, the data on 108 subjects with arsenic- caused skin lesions and 150 subjects without skin lesions from a population survey of over 7000 people of an endemic region of West Bengal carried out during 1995–1996 were analyzed. The median highest level of arsenic in drinking water was $330 \,\mu\text{g/L}$ [standard deviation (SD) $\pm 881 \,\mu\text{g/L}$] in subjects with skin lesions compared with $28 \,\mu\text{g/L}$ [(SD) $\pm 147 \,\mu\text{g/L}$] in those without such lesions.

Subsets of both the groups who reported chronic cough (more than 3 months per year for at least 2 years) were referred to a tertiary referral center in Kolkata for high resolution computed tomography (HRCT). The severity of bronchiectasis in each lobe was ranked on a scale of 0 to 4. Briefly, a five-point grading system was employed: 0 = no bronchiectasis; 1 = mild bronchiectasis (non-tapering cylindrical internal bronchial diameter 1.5–3 times the diameter of the accompanying artery); 2 = moderate bronchiectasis (non-tapering cylindrical internal bronchiectasis) (non-tapering cylindrical internal bronchiectasis); and 4 = cystic bronchiectasis. The lungs were divided into six lobes (considering the lingual as a separate lobe) and a score was assigned to each lobe. A single bronchiectasis severity score was assigned to each subject by summing the bronchiectasis scores from each lobe of each lung.

Thirty-three (31%) subjects with skin lesions and 18 (12%) subjects without lesions had reported chronic cough for more than 2 years (OR = 3.2; CI = 1.7-6.1). Of these, 27 subjects with skin lesions and 11 without lesions agreed to travel to Kolkata for HRCT. Overall, the participation rate was 82% in subjects with skin lesions and 61% in subjects without skin lesions. For those subjects who underwent HRCT, the average bronchiectasis severity score was 3.4 (SD \pm 3.6) in 27 subjects with skin lesions and 0.9 (SD \pm 1.6) in 11 subjects without lesions.

In subjects who reported chronic cough, CT evidence of bronchiectasis was found in 18 (67%) participants with skin lesions and 3 (27%) subjects without skin lesions. Only one (9%) of the 11 subjects without skin lesions had a bronchiectasis severity score greater than 2, while 14 (52%) of the 27 skin lesion subjects had bronchiectasis severity scores greater than 2.

The unadjusted odds ratio for bronchiectasis was 10 (95% CI = 2.9-35) in subjects with arsenic-caused skin lesions compared with subjects having no lesions. The corresponding adjusted odds ratio was 10 (2.7-37). The adjusted odds ratio was 13 (2.6-62) in men and 6.1 (0.6-62) in women [41].

This study was the first investigation of HRCT findings in a population-based study of people exposed to high levels of arsenic in drinking water. In this study, the authors found that persons with arsenic-caused skin lesions have a 10-fold higher rate of bronchiectasis on HRCT. Given the large magnitude of the relative risk estimate they identified, this association is not likely to be due to chance. The highly characteristic skin lesions diagnosed in this study are known to result from the consumption of arsenic-contaminated drinking water. The findings therefore provide evidence that long-term ingestion of arsenic results in increased risks of non-malignant pulmonary disease, in particular, bronchiectasis [41].

6.3.2.2 Mortality Studies

In a blackfoot disease-prone area of Taiwan (arsenic in drinking water: median = $780 \mu g/L$) an ecological study of mortality showed that the standardized mortality ratio (SMR) for "bronchitis" increased significantly relative to a nearby reference population (SMR = 1.53; 95% CI = 1.30-1.80) during the period from 1971 to 1994. The authors observed that it was unlikely that the differences in the rate of smoking account for the increased bronchitis mortality [42].

Further, a retrospective study on mortality due to bronchiectasis, carried out (between 1988 and 2000) in Antafagosta, Chile, suggested that exposure of arsenic in drinking water during early childhood or *in utero* has a pronounced pulmonary effect, greatly increasing subsequent mortality in young adults from the condition. For the birth cohort born just before the high exposure period (1950–1957) and exposed in early childhood, the SMR for bronchiectasis was 12.4 (95% CI: 3.3–31.7; p < 0.001) while for those born during the high exposure period (1958–1970) with probable exposure *in utero* and early childhood, the corresponding SMR was 46.2 (95% CI: 21.1–87.7; p < 0.001) [43].

6.3.3 Gastrointestinal Disease

Symptoms of dyspepsia were observed in 60 out of 156 (38.4%) cases of chronic arsenic toxicity studied in West Bengal [11]. However, in an epidemiological study carried out in the affected population there was no difference in the incidence of abdominal pain among people drinking arsenic-contaminated water and the control population (27.84% vs. 31.81%) [44]. Gastroenteritis was reported in a study of 1447 cases of chronic arsenicosis caused by drinking arsenic-contaminated water (0.05–1.8 mg/L) in the Inner Mongolian autonomous region of China [45]. Many investigators variously reported symptoms like nausea, diarrhea, anorexia, and abdominal pain in cases of chronic arsenic toxicity [1,8,17,20,21].
6.3.4 Liver Disease

Exposure of inorganic arsenic compounds has been found to be associated with the development of chronic pathological changes in the liver. Several authors reported cases of liver damage following treatment of patients with arsenic as Fowler's solution [46,47]. All these patients developed features of portal hypertension with signs of liver fibrosis. Typical cutaneous signs of long-term arsenic exposure were also observed in some of the patients. There have also been case reports on liver cirrhosis following medication with inorganic arsenic compounds [37,48].

Portal hypertension associated with periportal fibrosis was reported in nine patients who were found to have high arsenic levels in their liver in Chandigarh, India. Two of those patients had been found to be drinking arsenic-contaminated water (549 and $360 \,\mu g/L$) [49]. From a population-based study in West Bengal, hepatomegaly was reported in 62 out of 67 members of families who drank arsenic-contaminated water ($200-2000 \,\mu g/L$), while only in six out of 96 people who took safe water ($<50 \mu g/L$) from the same area. Thirteen arsenicexposed patients who had hepatomegaly were further investigated in a hospital in Kolkata. All showed various degrees of portal zone expansion and fibrosis on liver histology. Four of the five patients who had splenomegaly showed evidence of increased intrasplenic pressure (30-36 cm saline) suggesting portal hypertension. Splenoportography done in those cases showed evidence of intrahepatic portal vein obstruction. Although routine liver function tests were normal in all these cases, the bromsulphthalein retention test done in three patients was abnormal. The arsenic level in liver tissue (estimated by neutron activation analysis) was found to be elevated in 10 out of those 13 cases (As levels: Cases 0.5 to 6 mg/kg; control 0.10 \pm 0.04 mg/kg) [1]. In another study from West Bengal, hepatomegaly was found in 190 out of 248 cases of chronic arsenicosis investigated in the same hospital. Arsenic levels in drinking water varied from 50 to $3200 \,\mu g/L$. Evidence of portal zone fibrosis on liver histology was found in 63 out of 69 cases of hepatomegaly. Liver functions tests carried out on 93 such patients showed evidence of elevation of ALT, AST, and ALP in 25.8%, 6.3%, and 29% of cases, respectively. Serum globulin was found to be high (>3.5 gm/dL) in 19 (20.7%) cases [50].

Liver function tests were studied in arsenic-exposed subjects in three towns of Leguneara, Mexico. Significant elevation of bilirubin and alkaline phosphatase levels was observed among people living in the town having the highest $(239 \pm 88 \,\mu\text{g/L})$ arsenic exposure category compared to those living in the lowest $(14 \pm 3.1 \,\mu\text{g/L})$ exposure group. Serum transaminases and albumin level were, however, normal in all the groups [51].

A cross-sectional epidemiological study was carried out on 7683 people in an arsenicaffected district of West Bengal. Out of these, 3467 and 4216 people consumed water contaminated with arsenic below and above $50 \mu g/L$, respectively. Prevalence of hepatomegaly was significantly higher in arsenic-exposed people (10.2%) compared to controls (2.99%, p < 0.001). The incidence of hepatomegaly was found to have a linear relationship proportional to increasing exposure of arsenic in drinking water in both sexes (p < 0.001) [44].

Liver enlargement was also reported by other workers in people drinking arseniccontaminated water in, respectively, Bangladesh, Inner Mongolia, and West Bengal [22,45,52].

6.3.4.1 Experimental Model for Liver Fibrosis

In an experimental study, BALB/C mice were given water contaminated with arsenic (3.2 mg/L) *ad libitum* for 15 months, the animals being sacrificed at 3-month intervals. In the experimental animals, progressive reduction of hepatic glutathione and enzymes of an antioxidative defense system were found to be associated with lipid peroxidation. Liver histology showed fatty infiltration at 12 months and hepatic fibrosis at 15 months [53]. In another experimental study, increasing the dose and duration of arsenic exposure in mice was found to cause progressive increase of oxy-stress associated with elevation of cytokines, interleukin (IL)-6, and tumor necrosis factor (TNF)- α with increasing collagen content and arsenic levels in liver [54].

All these studies show that prolonged drinking of arsenic-contaminated water is associated with hepatomegaly. The predominant lesion of hepatic fibrosis appears to be caused by arsenic-induced oxy-stress.

6.3.5 Cardiovascular Disease

Blackfoot disease (BFD), a form of peripheral vascular disease, has been reported to be one of the important complications of chronic arsenic toxicity in Taiwan. It is a unique peripheral arterial disease characterized by the severe systemic arteriosclerosis as well as dry gangrene and spontaneous amputations of affected extremities at end stages. Histologically, BFD can be divided into two reaction groups: arteriosclerosis obliterans and thromboangiitis obliterans, particularly affecting small vessels [55]. Clinically, the disease begins with patients' subjective complaints of coldness or numbness in the extremities (usually in the feet) and intermittent claudication, progressing over the course of several years to ulceration, gangrene, and spontaneous amputation [56]. The prevalence of BFD has been reported to be 8.9 per 1000 among 40,421 inhabitants studied in Taiwan [19]. Comparable peripheral vascular disorders with varying degrees of severity including Raynaud's syndrome and acrocyanosis have also been reported by other researchers, among people drinking arsenic-contaminated water [11,20,22,37,45,57]. It needs to be emphasized that there are differences in the prevalence of peripheral vascular disease causing gangrene and amputation among different populations exposed to arsenic, the incidence being high in Taiwan, while low in Chile, India, and Bangladesh while there is no report from Mexico and Argentina [58].

An epidemiological study reported an increased prevalence of hypertension among residents in the endemic area of BFD and a dose-response relationship between ingested inorganic arsenic and prevalence of hypertension [59]. The investigators studied a total of 382 men and 516 women residing in arsenic hyperendemic areas in Taiwan. They observed 1.5-fold increases in age- and sex-adjusted prevalence of hypertension compared with residents in non-endemic areas. The higher the cumulative arsenic exposure, the higher was the prevalence of hypertension. The dose-response relationship remained significant after adjustment for age, sex, diabetes mellitus, proteinuria, body mass index, and serum triglyceride level. Increased prevalence of hypertension was also reported in 6.2% patients affected with arsenic-induced skin lesions compared to none in those without skin lesions in Antafagesta, Chile [20]. Association of cumulative arsenic exposure

in drinking water was also found to be associated with increased risk of hypertension in a study of 1595 people in Bangladesh [60]. Further study also showed evidence of increased association of hypertension in individuals resident in arsenic endemic regions compared to those from a non-endemic region in West Bengal, India. There were increased odds ratios for hypertension [adjusted OR, 2.87 (95%, CI = 1.26-4.83)] in participants with arsenic exposure compared to participants living in arsenic non-endemic regions. There was a dose–effect relationship seen with increasing cumulative arsenic exposure, with higher arsenic levels in hair and an increased incidence of hypertension in participants living in arsenic endemic regions [61].

Mortality rates from ischemic heart disease (IHD) with endemic arsenicosis (from 1973 through 1986) were correlated with arsenic levels in drinking water among residents of 60 villages in Taiwan [62]. Based on 1,355,915 person-years and 217 IHD deaths, the cumulative IHD mortalities from birth to age 79 years were 3.4%, 3.5%, 4.7%, and 6.6%, respectively, for residents who lived in villages in which the median arsenic concentrations in drinking water were <0.1, 0.1 to 0.34, 0.35 to 0.59, and \geq 0.6 mg/L. A cohort of 263 patients affected with BFD and 2293 non-BFD residents in the endemic area of arsenicosis were recruited and followed up for an average period of 5.0 years. There was a monotonous biological gradient relationship between cumulative arsenic exposure through drinking artesian well water and IHD mortality. The relative risks were 2.5, 4.0, and 6.5, respectively, for those who had a cumulative arsenic exposure of 0.1 to 9.9, 10.0 to 19.9, and \geq 20.0 mg/L-years compared to those without the arsenic exposure after adjustment for age, sex, cigarette smoking, body mass index, serum cholesterol and triglyceride levels, and disease status for hypertension and diabetes through proportional hazards regression analysis [63].

A significant biological gradient between the risk of IHD and the duration of consuming high arsenic artesian well water was reported from Taiwan. Further, a community-based health survey in an arsenic endemic region was carried out to determine whether there was any association between arsenic-related IHD and serum antioxidant micronutrient levels. A total of 74 patients affected with IHD, who were diagnosed through both electrocardiography and Rose questionnaire interview, and 193 age- and sex-matched healthy controls, were selected for the examination of serum levels of micronutrients by high performance liquid chromatography (HPLC). A significant reverse dose-response relationship with arsenic-related IHD was observed for serum levels of alpha- and beta-carotene, but not for serum levels of retinol, lycopene, and alpha-tocopherol. Multivariate analysis showed a synergistic interaction on arsenic-related IHD between duration of consumption of artesian well water and low serum carotene levels. An increased risk of arsenic-related IHD was also associated with hypertension and elevated body mass index, but not with serum lipid profile, cigarette smoking, and alcohol drinking. The authors suggested that arsenic-related IHD has a pathogenic mechanism, which is at least partially different from that of IHD unrelated to long-term exposure to arsenic [64].

Whether long-term arsenic exposure could be associated with IHD was ascertained in a community-based study in the blackfoot disease-endemic villages of Taiwan. A total of 462 subjects characterized by long-term arsenic exposure from drinking artesian well water were studied. IHD was diagnosed by coding the resting electrocardiograms with the Minnesota code. History of arsenic exposure was estimated through information obtained from a

personal interview according to a structured questionnaire and the arsenic content in artesian well water of the villages. Cumulative arsenic exposure (CAE) was calculated as the sum of the products multiplying the arsenic concentration in artesian well water (mg/L) by the duration of drinking the water (years) in consecutive periods of living in the different villages. Among the subjects, 78 cases (16.9%) were diagnosed as having IHD. The prevalence rates of IHD for the age groups of 30–39, 40–49, 50–59, and \geq 60 years were 4.9%, 7.5%, 16.8%, and 30.7%, respectively (p < 0.001). For those with a CAE of 0, 0.1–14.9, and \geq 15 mg/L-years, the prevalence rates of IHD were 5.2%, 10.9%, and 24.1%, respectively (p < 0.001). The odds ratios (95% confidence intervals) for IHD were 1.60 (0.48, 5.34) and 3.60 (1.11, 11.65), respectively, for those with a CAE of 0.1–14.9 and \geq 15.0 mg/L-years, when compared with those lacking drinking water exposure to arsenic after multivariate adjustment. The authors concluded that IHD is associated with long-term arsenic exposure. However, the limitation of the paper was that cumulative arsenic exposure was measured on the basis of group level by village. Further prospective studies with individual arsenic data and in other geographic regions are needed to establish the cause-effect relationship [65].

Increased prevalence of microvascular diseases, including neurological and renal disorders, were reported to be associated with arsenic ingestion. Microvascular diseases in relation to arsenic exposure level have been studied in a total of 28,499 subjects living in the study area of Taiwan, from information obtained through their medical records from the National Health Insurance database in 1999-2000. The arsenic concentrations of artesian well water in the villages of the study area were utilized as indices of previous ingestion level. Both stratified analysis and unconditional logistic regression were used to examine mainly neurological and renal disease in relation to arsenic exposure taking into account diabetes status. The age-adjusted and gender-adjusted prevalence of microvascular diseases was 7.51% (95% CI: (5.50-7.51) for an arsenic level of $100 \,\mu\text{g/L}$, and then increased from (5.59%) ((5.59-6.60) for arsenic concentrations of 100-290 µg/L to 8.02% (8.02-8.03) and 11.82% (11.81-11.83) for those of $300-590 \,\mu\text{g/Lmg/L}$ and $\geq 600 \,\mu\text{g/Lmg/L}$ in non-diabetic subjects. For diabetic patients, the prevalence was 16.41% (95% CI: 16.37-16.45), 15.85% (15.8-15.9), 21.69% (21.6-21.8), and 28.31% (28.2-28.4) for arsenic levels of <100, 100-0.290, 300-590, and \geq 600 µg/L, respectively. The prevalence of microvascular diseases increased significantly with arsenic exposure, especially at higher levels, and the relationship is stronger in diabetics than in non-diabetic subjects. The results for neurological disease are very similar, and the patterns are the same for renal disease [66]. In the report, the authors acknowledged that the precise level of arsenic exposure at the individual level, as a consequence of well-water consumption, was rather difficult to assess without individual lifetime histories. The consequences of exposure misclassification might have some limitation in the observed relationship of arsenic exposure and microvascular disease.

6.3.6 Diseases of the Nervous System

There are many reports of the occurrence of peripheral neuropathy due to chronic exposure of arsenic through drinking water [8,9,22,45,67,68]. Peripheral neuritis characterized by paresthesia (tingling, numbness, limb weakness, etc.) was present in 74 (47.4%) out of 156 patients with chronic arsenicosis due to drinking of arsenic-contaminated water (0.5–14.2 mg/L) in West Bengal, India. Objective evaluation of neuronal involvement, done in 29 patients, showed abnormal electromyography (EMG) in 10 (30.8%) and altered nerve conduction velocity and EMG in 11 (38%) cases [69]. Abnormal EMG findings, suggestive mostly of sensory neuropathy, were reported in 10 out of 32 subjects drinking arsenic-contaminated well water (range 0.06–1.4 mg/L) in Canada [70]. In another electrophysiological study carried out on 88 patients of arsenicosis in West Bengal, sensory neuropathy was found in 24 (27.3%), motor neuropathy in 13 (14.7%), and abnormal EMG in 5 (5.7%) cases [71].

A cross-sectional study of nerve conduction velocity was carried out in Taiwan on 130 students aged 12–24 years exposed to arsenic from drinking arsenic-contaminated water. After adjustment of gender and height, a significant odds ratio of 2.9 (95% CI: 1.1–7.5) was observed for development of slow conduction velocity of sural sensory action potential (SAP) among the study subjects with cumulative arsenic exposure of >100 mg. The authors concluded that chronic arsenic exposure might induce peripheral neuropathy [72].

Increased prevalence of peripheral neuritis was reported in a cross-sectional study from an arsenic endemic district (Nadia) of West Bengal. Out of a population of 10,469 arsenicexposed participants selected from all 17 affected blocks of the district by multistage sampling, 1616 (15.43%) cases had arsenical skin lesions while 8853 control subjects had no skin lesions. Mean arsenic content in drinking water among the cases was 103.46 (SD \pm 153.28) µg/L while among the controls it was 73.18 (SD \pm 115.10) µg/L. Peripheral neuropathy characterized by tingling, numbness, and limb pain was found in 257 (15.90%) cases and in 136 (1.54%) controls (p < 0.001) [23].

Peripheral neuritis, sleep disturbances, weakness, and cognitive and memory impairment have been reported in residents of Byan College Station, Texas, exposed to arsenic trioxide in air and water used to produce defoliants from an Atochem plant [68]. Headache has been reported to occur in people drinking arsenic-contaminated water in Mexico [8] and in West Bengal [11]. Irritability, lack of concentration, depression, sleep disorders, headache, and vertigo were reported in people with arsenicosis showing features of neuropathy in West Bengal [71].

Increased prevalence of cerebrovascular disease and ingestion of inorganic arsenic in drinking water was reported in a cross-sectional study in Taiwan. A total of 8102 men and women from 3901 households were recruited in this study. The status of cerebrovascular disease of study subjects was identified through home visit, personal interviews, and by review of hospital medical records according to WHO criteria. Information on consumption of well water, sociodemographic characteristics, cigarette smoking, and alcohol consumption habits, as well as personal and family history of disease, was also obtained. A significant dose-response relationship was observed between arsenic concentration in well water and prevalence of cerebrovascular disease after adjustment for age, sex, hypertension, diabetes mellitus, cigarette smoking, and alcohol consumption. The biological gradient was more prominent for cerebral infarction, showing multivariate-adjusted odds ratios of 1.0, 3.4, 4.5,

and 6.9, respectively, for those who consumed well water with an arsenic content of 0, 0.1 to 50.0, 50.1 to 299.9, and $>300 \,\mu$ g/L [73]. However, there were limitations in the methodology described in the study. Subjects were recruited from a registry of all adult residents within 18 villages, but criteria for village selection, methods of recruitment, and percentage participation were not stated.

Other studies from the region had not produced similar findings. Increased mortality was not observed due to cerebrovascular accidents when mortality and population data were obtained from the local household registration offices and Taiwan Provincial Department of Health during 1973–1986 and correlated with arsenic levels in well water determined in 1964–1966 in 42 study villages in Taiwan [74].

However, a recent study from Taiwan again showed that exposure to arsenic in drinking water was associated with a higher risk of cerebrovascular disease (CVD). A study of (CVD) mortality was carried out with the national death registry data from two areas in Taiwan from 1971 to 2005. The arsenic levels in the drinking water in the BFD area were generally higher than those in the unexposed area. SMRs adjusted for gender and age were calculated using the whole population in Taiwan and the population in the unexposed area as the reference population, respectively. The SMR of CVD in Taiwan decreased from 2.46/100 in 1971 to 0.63/100 in 2005. The exposed group had higher SMRs of CVD in comparison with the reference populations, with SMRs from 1.06 to 1.09 in men and 1.12 to 1.14 in women. The BFD endemic area had higher mortality rates of CVD than the unexposed area, with SMR = 1.05 in men and SMR = 1.04 in women [75].

Recently, an increased mortality was reported due to CVD with low dose range of arsenic in ground water in the USA. An SMR analysis was conducted in a contiguous six county study area of southeastern Michigan to investigate the relationship between moderate arsenic levels and several selected disease outcomes. Arsenic data were compiled from 9251 well water samples tested by the Michigan Department of Environmental Quality from 1983 through 2002. The six county study area had a population-weighted mean arsenic concentration of $11.00 \,\mu$ g/L and a population-weighted median of $7.58 \,\mu$ g/L. Michigan Resident Death Files data were amassed for 1979 through 1997 and sex-specific SMR analyses were conducted with indirect adjustment for age and race; 99% CI were reported. Elevated mortality rates were observed for both males (M) and females (F) for cerebrovascular disease (M SMR, 1.19; CI: 1.14-1.25; F SMR, 1.19; CI: 1.15-1.23) [76].

6.3.7 Hematological Effects

Hematological abnormalities have been reported in acute and chronic arsenic poisoning [15]. A characteristic pattern of anemia, leukopenia, and thrombocytopenia was found in 55 individuals exposed to arsenic in drinking water in Niigata Prefecture in Japan for approximately 5 years, half of the subjects having arsenical skin lesions [77]. In one study in West Bengal, anemia was reported in all the 13 people exposed to arsenic-contaminated groundwater (0.2-2 mg/L) [1]. A further study in West Bengal on 156 people exposed to arsenic-contaminated water (0.05-14.2 mg/L) showed the incidence of anemia in 47.4% of cases [11]. However, no association of anemia was found in people drinking well water (mean 0.22 mg/L) in Alaska [3] and in two towns in Utah, USA (arsenic exposure 0.18 and 0.27 mg/L) [78].

6.3.8 Diabetes

The association of diabetes mellitus with chronic arsenic exposure has been variably reported. A dose-response relation between cumulative arsenic exposure and prevalence of diabetes mellitus was observed in Taiwan following a study on 891 persons living in arsenic endemic areas (arsenic levels in artesian well water, 700 to $930 \mu g/L$). Cumulative lifetime ingestion of arsenic in drinking water could be determined for 718 subjects aged over 30 years. The status of diabetes mellitus was determined by an oral glucose tolerance test and a history of diabetes regularly treated with sulfonylurea or insulin. The researchers observed a dose-response relation between cumulative arsenic exposure and prevalence of diabetes mellitus. The relation remained significant after adjustment for age, sex, body mass index, and activity level at work using a multiple logistic regression analysis giving a multivariate adjusted odds ratio of 6.61 (95% CI: 0.86–51.0) and 10.05 (95% CI: 1.30–77.9), respectively, for those who had a cumulative arsenic exposure of 0.1–15.0 mg/L-year and greater than 15.0 mg/L-year compared with those who were unexposed. However, the study had a few limitations, one being the prevalence of diabetes observed in 108 unexposed subjects as only 0.9%, much lower than the 5.1% background prevalence cited by the authors [79].

From Bangladesh a significantly increased prevalence of diabetes mellitus was reported due to drinking arsenic-contaminated water among 163 subjects with arsenical keratosis living in seven villages having groundwater contamination when compared with the prevalence of diabetes mellitus among 854 subjects recruited from an arsenic-uncontaminated region. Diagnosis of diabetes mellitus was made from history of symptoms, previously diagnosed disease, glycosuria, and blood glucose after oral glucose challenge. Using control population's age- and sex-adjusted prevalence as a reference (1.0), the prevalence ratios for diabetes mellitus were calculated to be 2.6, 3.9, and 8.8 for time-weighted average exposure categories of less than $500 \,\mu g/L$, $500-1000 \,\mu g/L$, and greater than $1000 \,\mu g/L$ respectively [80]. The lack of a comprehensive, systematic, long-term sampling of the water supplies in the study area is a limitation of the study because directly measured individual exposure data over time would have been desirable. However, these results suggest that chronic arsenic exposure may induce diabetes mellitus in humans.

High mortality attributed primarily to diabetes mellitus was reported from arsenic-contaminated water (arsenic levels in artesian well water, $250-1400 \mu g/L$) in four townships in Taiwan from the early 1900s until the mid- to late 1970s. Observed mortality between 1971 and 1994 was compared with age- and sex-specific expected mortality based on data from (1) a local reference group similar to a study group derived from nearby counties and (2) all of Taiwan [42].

A prospective cohort follow-up study was carried out on 446 subjects who agreed to participate in investigation for detection of diabetes mellitus by estimating fasting blood sugar and glucose tolerance test in 1991 and 1993. The participants were drawn from 632 non-diabetic subjects, enrolled in 1970 from three villages in Taiwan having a history of arsenic exposure $(700-930 \,\mu g/L)$ from artesian well water. The incidence of diabetes mellitus in the study population was calculated as the total number of incident cases divided by the sum of follow-up person-time in all subjects. Data on each subject included age, sex, body mass index, and index of lifetime CAE. CAE (in units of milligrams per liter-years) was calculated as the product of median arsenic concentration of the well water in every village that a subject inhabited at some point in his or her life multiplied by the length of time they consumed well water in that village. Incidences for diabetes mellitus in the study population were compared with those reported for a demographically similar population that was studied contemporaneously. During the followup period, which included 1499.5 person-years, 41 out of 446 subjects developed diabetes mellitus (all non-insulin-dependent diabetes mellitus). The incidence for new cases was particularly increased in subjects 55 years of age or older (50.8 per 1000 person-years). The relative risk for developing diabetes mellitus among those more than 17 mg/L-years CAE compared with those with less than 17 mg/L CAE was 2.1(995% CI: 1.1-4.2), adjusted for age, sex, and body mass index in a multivariate Cox proportional hazard model. When considered as a continuous variable, the CAE was associated with an adjusted relative risk of developing diabetes mellitus of 1.03 for every 1 mg/L-year of exposure (p < 0.05) [81].

Though earlier reports from Taiwan and Bangladesh showed an association between the development of diabetes mellitus and exposure to relatively high concentrations ($>500 \mu g/L$) of arsenic in drinking water, the risk of diabetes following exposure to lower levels of As $(<300 \mu g/L)$ has not been conclusively demonstrated. A population-based cross-sectional study was done on 11,319 participants using baseline data in the Health Effects of Arsenic Longitudinal Study in Araihazar, Bangladesh, to evaluate the associations of well water arsenic and total urinary arsenic concentration and the prevalence of diabetes mellitus and glycosuria. The investigators also assessed the concentrations of well water arsenic, total urinary arsenic, and urinary arsenic metabolites in relation to blood glycosylated hemoglobin (HbA1c) levels in subsets of the study population. More than 90% of the cohort members were exposed to $<300 \,\mu g/L$ of arsenic in drinking water. The authors reported no association between arsenic exposure and the prevalence of diabetes and no association between arsenic exposure and prevalence of glycosuria. The adjusted odds ratios for diabetes were 1.00 (referent), 1.35 (95% CI: 0.90-2.02), 1.24 (0.82-1.87), 0.96 (0.62-1.49), and 1.11 (0.73-1.69) in relation to quintiles of time-weighted water arsenic concentrations of 0.1-8, 8-41, 41-91, 92-176, and \geq 177 µg/L, respectively, and 1.00 (referent), 1.29 (0.87-1.91), 1.05 (0.69-1.59), 0.94 (0.61-1.44), and 0.93 (0.59-1.45) in relation to quintiles of urinary arsenic concentrations of 1-36, 37-66, 67-114, 115–204, and \geq 205µg/L, respectively. No association between arsenic exposure and prevalence of glycosuria and no evidence of an association between well water arsenic, total urinary arsenic, or the composition of urinary arsenic metabolites and HbA1c level were observed in the study. The study population in this report was generally lean and was of low socioeconomic and nutritional status. The findings, therefore, may not be generalized to other study populations, given the possible different distribution of risk factors for diabetes that may influence the effect of arsenic exposure [82].

6.3.9 Eye Disease

6.3.9.1 Conjunctivitis

Symptoms of eye disease are not infrequently found to be associated with chronic arsenic exposure. In a hospital-based study in West Bengal, eye symptoms characterized by burning of the eve and conjunctival congestion were found in 69 (42.2%) out of 156 patients with chronic arsenicosis associated with arsenical skin lesions and a history of prolonged drinking of arseniccontaminated water [11]. Conjunctivitis associated with chronic arsenic ingestion had also been reported from Bangladesh in 57 (15.7%) out of 363 arsenic-exposed subjects [22]. Increased incidence of conjunctivitis was reported from a study of 1482 arsenicosis patients living in six of 496 upazilas (subdistricts) of Bangladesh, who were identified through household screening. The investigators observed increased duration of arsenic-exposure symptoms and older age to be significantly associated with increased occurrence of conjunctivitis [83]. Papillary conjunctivitis was reported in two members of a family with dermatological manifestations and a history of drinking arsenic-contaminated water for 15 years in Bangladesh. Levels of arsenic in the nails and hair of both patients were very high. Histopathological examination of conjunctival tissue confirmed the inflammatory response of a papillary type; however, an arsenic estimation in conjunctival tissue was not possible. There were no inclusion bodies in conjunctival smears stained with hematoxylin and eosin (H&E) stain. There was no response to the usual treatment for papillary conjunctivitis, which only subsided along with a regression of dermatological changes, when patients were treated with the chelating agent dimercaprol, and multivitamin preparations, as well as no longer drinking the contaminated water [84].

6.3.9.2 Pterygium

Chronic exposure to arsenic in drinking water was reported to be associated with the occurrence of pterygium in Taiwan. Pterygium is a fibrovascular growth of the bulbar conjunctiva and underlying subconjunctival tissue that may cause blindness. Eye examination and a questionnaire interview of 223 participants from three exposure villages and 160 participants from four comparison villages were carried out and photographs taken. Subsequent grading of pterygium status was done by an ophthalmologist. After adjusting for age, sex, working under sunlight, and working in sandy environments, the authors found that cumulative arsenic exposure of 0.1-15.0 mg/L-year and $\geq 15.1 \text{ mg/L-year}$ was associated with increased risks of developing pterygium. The adjusted odds ratios were 2.04 (95% CI: 1.04–3.99) and 2.88 (95% CI: 1.42–5.83), respectively [85].

6.3.9.3 Cataract

Increasing prevalence of posterior subcapsular cataract with an increase in exposure to ingested arsenic was reported from Taiwan. The study was carried out on a total of 349 residents living in arseniasis hyperendemic villages of southwestern Taiwan with recording of cumulative arsenic exposure and determination of different types of lental opacity. The cataract surgery prevalence was 10% for the age group of 50 or more years. Cortical opacity was most common (35%), while nuclear and posterior subcapsular opacities were observed in 24%

and 22% of subjects, respectively. Diabetes mellitus was a significant risk factor for all types of cataract. Occupational sunlight exposure was associated with cortical and posterior subcapsular opacities in a dose-response relationship. The cumulative exposure to arsenic from artesian well water and the duration of consuming artesian well water were associated with an increased risk of all types of lens opacity. But statistically significant dose-response relations with the cumulative arsenic exposure and the duration of consuming artesian well water were observed only for posterior subcapsular opacity (p = 0.014 and p = 0.023, respectively) after adjustment for age, sex, diabetes status, and occupational sunlight exposure [86].

6.3.10 Non-Pitting Edema of Limbs

A study carried out in Mexico showed that edema of the legs occurred at a higher frequency (18 out of 296; 6.1% of arsenic-exposure cases through drinking water (>50 µg/L)) compared to unexposed controls (four out of 318; 1.3%, p < 0.01) [8]. However, whether the edema was pitting or not was not mentioned in the paper. Non-pitting edema of the legs was first reported in 23 (9.3%) cases out of 248 patients with chronic arsenicosis associated with arsenical skin lesions and a history of prolonged drinking of arsenic-contaminated water (50–14,200 µg/L) in West Bengal, India, in a hospital-based study [50,87]. Non-pitting edema of the limbs has also been reported in 10 (2.5%) cases out of 363 cases of arsenicosis in Bangladesh with a history of drinking arsenic-contaminated water (82–1371 µg/L) [22]. Further, an increased incidence of non-pitting edema of the limbs (hand and/or leg) was found in a population-based cross-sectional study carried out in all the arsenic-affected 17 blocks of the Nadia district of West Bengal. Non-pitting edema of the limbs was found in four out of 1616 (0.25%) cases (arsenic level in drinking water: mean, 103.46; SD \pm 153.28 µg/L) with arsenical skin lesions while in two out of 8853 (0.02%) control subjects (arsenic level in drinking water: mean, 73.18; SD \pm 115.10 µg/L) without skin lesions (p < 0.01) [23].

6.3.11 Miscellaneous

6.3.11.1 Weakness

Many reports are available where weakness and fatigue were described as occurring in people exposed to arsenic from the chronic drinking of arsenic-contaminated water [1,11,17,26,88]. Prevalence of weakness was studied in a large cross-sectional survey on 6864 arsenic-exposed (arsenic in drinking water, 0.3–3400 µg/L) participants in West Bengal, India. The age-adjusted prevalence of weakness increased strongly with arsenic concentrations (lowest, <50 µg/L and highest, ≥800 µg/L) in both sexes (from 1.7 to 11.9 per 100 among women, p < 0.0001, and from 0.9 to 9.5 per 100 among men, p < 0.0001) [38].

6.3.11.2 Erectile Dysfunction

A report from Taiwan suggested that chronic arsenic exposure has a negative impact on erectile function. A study had been conducted on 177 males \geq 50 years of age through health examinations conducted in three hospitals in Taiwan through assessment of risk factors associated with erectile dysfunction (ED), such as aging, sex hormone levels, hypertension, cardiovascular diseases, and diabetes mellitus. The investigators used a questionnaire (International Index of Erectile Function-5) to measure the level of erectile function and determined sex hormones, including total testosterone and sex hormone-binding globulin radioimmunoassay. Another standardized questionnaire was used to collect background and behavioral information (e.g., cigarette smoking; alcohol, tea, or coffee drinking; and physical activity). The prevalence of ED was found to be greater in the arsenic endemic area (83.3%) than in the non-arsenic endemic area (66.7%). Subjects with arsenic exposure >50 µg/L had a significantly higher risk of developing ED than those with exposure $\leq 50 \mu g/L$, after adjusting for age, cigarette smoking, diabetes mellitus, hypertension, and cardiovascular disease (OR = 3.4). Results also showed that the risk of developing severe ED was drastically enhanced by arsenic exposure (OR = 7.5), after adjusting for free testosterone and traditional risk factors of ED [89].

6.3.11.3 Proteinuria

Adverse effects of arsenic exposure from drinking water on the risk of proteinuria have been reported from a study in Bangladesh, the effects being modifiable by recent changes in arsenic exposure. Proteinuria was detected by urinary dipstick tests at baseline and at 2-year intervals on 11,122 participants in the Health Effects of Arsenic Longitudinal Study (HEALS). At baseline, well arsenic was positively related to prevalence of proteinuria; PORs for proteinuria in increasing quintiles of well arsenic (\leq 7, 8–39, 40–91, 92–179, and 180–864 µg/L) were 1.00 (ref), POR 0.99 (95% CI: 0.77–1.27), POR 1.23 (95% CI: 0.97–1.57), POR 1.50 (95% CI: 1.18–1.89), and POR 1.59 (95% CI: 1.26–2.00) (*p* for trend <0.01). Hazard ratios for incidence of proteinuria were POR 0.83 (95% CI: 0.67–1.03) and POR 0.91 (95% CI: 0.74–1.12) for participants with a decreasing level of >70 and 17–70µg/L in urinary arsenic over time, respectively, and were POR 1.17 (95% CI: 0.97–1.42) and POR 1.42 (95% CI: 1.16–1.73) for participants with an increasing level of 16–68 and >68µg/L in urinary arsenic over time, respectively, compared with the group with relatively little change in urinary arsenic as the reference group (urinary arsenic 15µg/L) [90].

6.4 Pregnancy Outcome

No conclusive information on pregnancy outcome and infant mortality in relation to arsenic levels in drinking water is available in the literature as few studies have included individual assessment of arsenic concentrations in all water sources used during each pregnancy. In an ecological study carried out in Chile, stillbirths (rate ratio 1.7; 95% CI: 1.5, 1.9), neonatal and post-neonatal infant mortality rates were found to be increased in the high arsenic exposure city of Antofagasta as compared with the low exposure city of Valparaiso [91]. A study conducted in Bangladesh showed an increased risk for stillbirth for women with current arsenic levels of $\geq 100 \,\mu$ g/L, although the risk estimates were smaller (OR = 2.5; 95% CI: 1.5, 5.9). The authors further reported increased effects on spontaneous abortions (OR = 2.5; 95% CI: 1.5, 4.4) [92]. However, no information was available on arsenic exposure during pregnancy, and high exposure levels of 200 μ g/L and more were not considered separately in this study. One

earlier cross-sectional study from Bangladesh compared rates of spontaneous abortions, stillbirths, and preterm delivery between 96 women in one village who were exposed to $\geq 100 \,\mu\text{g/L}$ arsenic to rates in 96 women in another village who were exposed to less than $20 \,\mu\text{g/L}$, and showed two to three times higher rates among exposed women [93]. Both Bangladesh studies reported a relation to overall duration of women's exposure without taking into account exposure during the actual time period of pregnancy.

A retrospective study of pregnancy outcomes and infant mortality was conducted in West Bengal, India, among 202 married women selected from a source population of 7683 between 2001 and 2003. Reproductive histories were ascertained by structured interviews. Arsenic exposure during each pregnancy was assessed based on all water sources used, involving measurements from 409 wells. Odds ratios for spontaneous abortions, stillbirth, and neonatal and infant mortality were estimated with logistic regressions based on the method of generalized estimating equations. High concentrations of arsenic $\geq 200 \,\mu g/L$ during pregnancy were associated with a six-fold increased risk for stillbirth after adjusting for potential confounders (OR = 6.25; 95% CI: 1.59, 24.6, p = 0.009). Arsenic-related skin lesions were found in 12 women who had a substantially increased risk of stillbirth (OR = 13.1, 95% CI: 3.17, 54.0, p = 0.002). The odds ratio for neonatal death was 2.03 (95% CI: 0.57, 7.24). No association was found between arsenic exposure and spontaneous abortion (OR = 0.90; 95% CI: 0.36, 2.26) or overall infant mortality (OR = 1.18, 95% CI: 0.38, 3.64). This study adds to the limited evidence that exposure to high concentrations of arsenic during pregnancy increases the risk of stillbirth. However, there was no indication of increased rates of spontaneous abortion and overall infant mortality [94].

6.5 Arsenicosis and Cancer

The evidence of carcinogenicity in humans from exposure to arsenic is based on epidemiological studies of cancer in relation to arsenic in drinking water. The working group of the International Agency for Research on Cancer (IARC) [95] evaluated data from ecological studies, cohort studies, and case-control studies from many countries.

6.5.1 Skin Cancer

Skin cancer is a commonly observed malignancy related to drinking of arsenic-contaminated water. The working group of the IARC [95] evaluated ecological studies from Taiwan, Mexico, Chile, and the USA, cohort studies from Taiwan, and a case-control study from the USA. Numerous cases of skin cancer have been documented from communities with arsenic-contaminated drinking water. A prevalence study was conducted on a population of 40,421 in an arsenic endemic area of Taiwan where 238 cases of skin cancer have been clinically detected. Prevalence rates for inhabitants residing in low ($<300 \,\mu g/L$), medium ($300-600 \,\mu g/L$), and high ($>600 \,\mu g/L$) arsenic-contaminated areas represented over an eight-fold difference in occurrence of skin cancer from the highest to the lowest category of arsenic exposure [19]. In an ecological study, sex- and site-specific cancer mortality for 1989–1993 in Region II of Chile

with national mortality rates was compared. The SMR for skin cancer was 7.7 (95% CI: 4.7-11.9) among men and 3.2 (95% CI, 1.3-6.6) among women [95].

Skin cancer due to chronic arsenic exposure occurs as Bowen's disease (intraepithelial carcinoma, or carcinoma *in situ*), basal cell carcinoma, and squamous cell carcinoma. Skin cancer might arise in the hyperkeratotic areas or might appear on non-keratotic areas of the trunk, extremities, or hand [6,18]. Features of Bowen's disease have been described earlier. Arsenicrelated basal cell carcinoma appears to be deep ulcerative or superficial type as those of ordinary type of basal cell carcinoma. Histologically, the cells have scanty and ill-defined cytoplasm. Nuclear atypy and giant cells are not ordinarily found. The size and gross appearance of epidermoid carcinoma varied greatly, some forming fungating masses measuring up to 5 by 5 cm and some causing large crater-like ulcers with elevated margins measuring up to 6 cm in diameter. Histologically, the grade of differentiation of epidermoid carcinoma varies [6].

6.5.2 Urinary Bladder Cancer

The working group of the IARC [95] evaluated ecological studies in Taiwan, Chile, Argentina, and Australia, cohort studies from Taiwan, Japan, and the USA, and case-control studies in Taiwan, the USA, and Finland and found evidence of increased risk for urinary bladder cancer associated with arsenic in drinking water. The first report of bladder cancer associated with drinking arsenic-contaminated water was published from the province of Cordoba in Argentina where 11% of cancer deaths were caused by this cancer [96]. An ecological mortality study for bladder cancer was conducted in Chile in Region II (arsenic endemic region) and in Region VIII (non-arsenic endemic area) for the period 1950–1992 and the SMR was found to be 10.2 (95% CI: 8.6–12.2) [97]. In Taiwan, the evidence of increased occurrence of bladder cancer due to arsenic was supported by case-control and cohort studies within the exposed communities that demonstrated evidence of a dose-response relationship with levels of arsenic in drinking water. There was also evidence of increased risks of bladder cancer from a small cohort study in Japan of persons drinking from wells that had been highly contaminated with arsenic wastes from a factory. The findings of epidemiological studies are consistent with a strong association of arsenicosis with bladder cancer [95].

6.5.3 Lung Cancer

On the basis of ecological studies using mortality data in Taiwan, Chile, Argentina, and Australia, cohort studies in Taiwan, Japan, and the USA, and case-control studies in Taiwan and Chile, a strong association of lung cancer has been observed in populations with high arsenic exposure [95]. In an ecological study, increased mortality from lung cancer was observed in men and women in 1968–1982 in an area endemic for blackfoot disease in Taiwan [98]. There was an exposure-response relationship between the SMR of lung cancer and the prevalence of blackfoot disease. Elevated SMRs (about three) were observed for lung cancer for both sexes in Region II of Chile using national rate as a standard [97].

6.5.4 Other Cancers

Increased risk of liver cancer has been reported in several studies from Taiwan, Japan, and Chile identified from death certificates. There is limitation of interpretation of these findings because of questionable accuracy of the diagnosis of liver cancer on death certificates and potential confounding or modifying effects of hepatitis virus infection or other factors. Ecological studies in Taiwan, Chile, Argentina, and Australia and cohort studies from Taiwan and the USA involving populations with high long-term exposure to arsenic found increased risks for kidney cancer. Relative risk estimates for kidney cancer were generally lower than those for urinary bladder cancer, and no studies have reported dose-response relationships on the basis of individual exposure data. Excess mortality from prostate cancer was found in southwest Taiwan. Inconsistent findings were reported for other cancers [95].

6.6 Diagnosis

The quantity of arsenic dose and the duration of arsenic exposure necessary to cause arsenical skin lesions have been reported by several investigators. The mean arsenic dose in Antofagasta was estimated to be approximately $60 \mu g$ per day for subgroups of children aged 3.13 ± 3.33 years but was approximately $20 \,\mu g$ per day for subgroups in their teens and twenties and $6 \,\mu g$ per day for a subgroup in their sixties, indicating an inverse relationship between daily arsenic dose rate/kg body weight and age [37]. The lowest peak arsenic exposure determined by using detailed lifetime (at least 20 years) exposure assessment by a confirmed case of arsenicosis with skin lesions was reported as 115 µg in West Bengal [31]. Prolonged exposure to a non-lethal dose of 5 to $90 \mu g/kg$ -body weight/day has been suggested to cause arsenical skin lesions [2]. In a retrospective study of 262 adults treated with Fowler's solution, it was reported that the minimal latency period for hyperkeratosis was 2.5 years following ingestion of approximately 2.2g of arsenite [99]. Rattner and Dorne reported the development of hyperpigmentation within 6-12 months of the start of treatment with arsenic at a dose of $4750 \,\mu\text{g}/\text{day}$ [100]. Hyperkeratosis appeared after approximately 3 years. The shortest time of exposure associated with hypopigmentation, hyperpigmentation, and keratosis was 8 and 12 years, respectively, in Mexico with arsenic exposure of $400 \,\mu g$ in drinking water [8]. Arsenical skin lesions were reported to occur in West Bengal, India, after drinking arsenic-contaminated water (50 to 14,200 ppb) for 1 to 15 years [69]. A history of chronic arsenic exposure for more than 6 months is recommended as one of the required diagnostic criteria for the clinical diagnosis of arsenicosis [2].

Although chronic arsenic toxicity produces varied systemic manifestations as well as cancer of skin and different internal organs, dermal manifestations such as pigmentation and keratosis are diagnostic of chronic arsenicosis. For this reason, the field guide of the diagnostic algorithm of arsenicosis of the WHO [2] is based on the presence or absence of characteristic dermatological manifestations of chronic arsenic toxicity. A clinically confirmed case of arsenicosis is a "probable case with pigmentation (diffuse, spotty or blotchy) and/or keratosis (thickening or nodularity or verrucousity) of the skin, distributed bilaterally [and]

Category	Major Conditions for Consideration		
Diffused melanosis	Actinic dermatosis		
	Melasma		
	Ashy dermatosis		
Spotted melanosis	Pityriasis versicolor		
	Freckle		
	Lichen planus		
Leukomelanosis	Idiopathic guttate hypomelanosis		
	Pityriasis versicolor		
	Pityriasis lichenoides chronica		
	Leprosy		
Diffuse keratosis	Psoriasis (palms and soles)		
	Eczema		
	Occupational keratosis		
	Tinea pedis		
	Pitted keratolysis		
Nodular keratosis	Occupational keratosis		
	Verruca vulgaris corns/calluses		
	Seborrheic keratosis		

Table 6-4Common Conditions to be Considered forDifferential Diagnosis of Non-cancer Skin Lesions

symmetrically in the body," in whom the presence of other arsenicosis simulating skin lesions has been ruled out. A "clinically and laboratory confirmed case" is a "clinically confirmed case" in whom the arsenic test is also positive by the recommended laboratory criteria. Laboratory criteria for establishing exposure history of arsenicosis cases are: (1) consumption of drinking water with an arsenic concentration in excess of prevailing national standards for at least 6 months (country standard in the Asia Pacific region is $50 \,\mu\text{g/L}$ while the WHO standard is $10 \,\mu\text{g/L}$) and (2) an elevated concentration of arsenic in hair (>1 mg/kg of hair) or in nail clippings (>1.5 mg/kg of nails) [2].

Clinical features of hypopigmentation, hyperpigmentation, and keratosis are quite distinctive for the suspicion of diagnosis of arsenicosis, and a history of arsenic exposure through ingestion is helpful in corroborating a diagnosis. However, manifestations such as diffuse melanosis cannot be differentiated from the normal dark complexioning of farmers in the tropics who work in the field bare-bodied under direct sunlight. Similarly, manual laborers and farmers, who work with bare hands and bare feet, might have thickening of the palms and soles. Further, many other dermatological diseases simulate the features of arsenical skin manifestations. Hence, in a probable case of arsenicosis, diagnosis of a clinically confirmed case could be done by excluding other dermatological diseases simulating arsenical skin lesion by differential in-depth skin examination. Common conditions to be considered for differential diagnosis of arsenical skin lesions are given in Table 6–4. The presence of the following features will exclude the diagnosis of arsenical skin lesions: presence since birth, scaling, and erythema surrounding the lesion.

6.6.1 Biomarkers with Special Focus on Diagnosis

The most important biomarkers of internal exposure to arsenic are the urinary excretion of the element and its concentration in hair and nails. Blood concentrations are too low for use in the diagnosis, though it has a place in metabolic and pharmacy-kinetic study. Arsenic measurements in hair and nails are not used as indices of currently absorbed dose. Efforts are needed to develop a standardized procedure to solve the problem of external contamination of samples. Taking advantage of the relationship between arsenic intake through water and urinary excretion of inorganic arsenic and its metabolites MMA and DMA appears to offer a better means of estimation.

The concentration of total arsenic in urine has often been used as an indicator of recent arsenic exposure, because urine is the main route of excretion of most arsenic species [101]. As a parameter of oral intake of arsenic via drinking water in steady-state conditions, the urinary level of arsenic (seafood arsenic excluded) has been reported to be diagnostic by several authors from different countries. Despite possible ethnic and environmental differences, reported results display a quite satisfactory consistency. Most strikingly, an increased excretion rate is observed when the arsenic concentration of the drinking water reaches $100-200 \,\mu g/L$ [102]. The half-life of inorganic arsenic in humans is about 4 days. Average background concentrations of arsenic in urine are generally below $10 \,\mu g/L$. A urine sample showing more than 50 µg/L may be taken as evidence of recent exposure provided the subject has not consumed seafood during the previous 4 days [2]. Although high arsenic excretion in urine is indicative of continued arsenic exposure, this is not always diagnostic of chronic arsenic toxicity. In West Bengal, India, a significant number of people (nine out of 17) who were drinking arseniccontaminated water and had high urinary arsenic excretion did not show cutaneous manifestations of chronic arsenic toxicity. On the other hand, many (33 out of 40) of the chronically arsenic-exposed people showing arsenical skin lesions did not have high urinary arsenic excretion [103]. These results might be explained by the fact that all those who are drinking arsenic-contaminated water at a particular point in time may not be showing clinical features of chronic arsenic toxicity, whereas others who might have consumed arsenic-contaminated over a prolonged period in the past and have had skin lesions but have stopped drinking arsenic-contaminated water, might currently have low arsenic levels in urine.

A report from Bangladesh highlighted that urinary arsenic might be a stronger predictor of skin lesions than arsenic in drinking water used by the population. Arsenic levels in drinking water sources and urinary arsenic values had been correlated with the incidence of arsenical skin lesions in residents of three villages in Bangladesh. Arsenic levels in current drinking water source were found to be $<50 \,\mu$ g/L in 13 (36.1%) out of 36 subjects with arsenical skin lesions among 167 participants studied. The risk of skin lesions in relation to the exposure estimates based on urinary arsenic was elevated more than three-fold, with the odds ratios for the highest versus the lowest quartiles 3.6 (95% CI: 1.2–12.1) for urinary total arsenic. The risks for skin lesions in relation to exposure estimates based on arsenic in drinking water with the odds ratios for the highest versus lowest quartiles of exposure were 1.7 (95% CI: 0.6–5.1) for current drinking water arsenic and 2.3 (95% CI: 0.7–7.6) for cumulative arsenic index [104].

Arsenic is normally found in higher concentrations in human hair and nails than in other parts of the body. This has been explained by the high content of keratin in these tissues [105]. In people with no known exposure to arsenic, the concentration of arsenic in hair is generally 0.02-0.2 mg/kg [106]. In one study in West Bengal, arsenic content in hair (analyzed by neutron activation analysis) in patients drinking arsenic-contaminated water (220–2000 µg/L) was found to vary from 1.4 to 20 mg/kg [1]. External contamination of the hair by arsenic must be excluded in order to use hair arsenic concentrations to assess toxicity.

Normal arsenic values in nails appear to range from 0.02 to 0.5 mg/kg [107,108]. Arsenic levels in nails (estimated by neutron activation analysis) in arsenic- (220–2000 µg/L) exposed people in West Bengal were found to vary between 16 and 66 mg/kg [1].

There is no correlation between arsenic levels in hair and nails and clinical features of chronic arsenic toxicity. In a village in West Bengal, all the 17 people drinking arsenic-contaminated water had raised hair and nail arsenic levels, but only eight had cutaneous lesions [69]. Further, out of 40 people with arsenical skin lesions in another village of West Bengal with a history of drinking arsenic-contaminated water, normal levels of arsenic in hair and nails were found in 31 and 26 cases, respectively [103].

6.7 Treatment

Chronic arsenicosis leads to irreversible damage to several vital organs and organ systems; arsenic is also an established carcinogen. Despite the magnitude of this potentially fatal toxicity, there is no effective therapy for this disease; once affected, patients may not recover, even after remediation of the arsenic-contaminated water. However, various modalities are used for the management of dermatological manifestations of chronic arsenic toxicity.

6.7.1 Ceasing Drinking of Arsenic-Contaminated Water

Changes in severity of skin lesions were reported among an affected cohort of arsenicosis patients in southern Thailand where interventions to reduce arsenic-contaminated water had been implemented. Over a 10-year period, both regression and progression of lesions occurred, though the majority of the subjects followed up remained the same. Drinking predominantly arsenic-free water increased the probability of regression in subjects with mild stage lesions but not in those with more advanced stage lesions. By contrast, high arsenic content in the household well water, even though it was not used for drinking, decreased the probability of lesion regression among the subjects in more advanced stages but not among milder stage cases. Irrespective of initial stage, a period of absence from the affected area increased the likelihood of lesion regression [109]. Another cohort follow-up study was carried out on 1074 people (arsenic exposed people 623, control population 451) in 2000, 5 years after the original clinical examination done on the same population at South 24 Parganas, West Bengal. Out of 199 people with skin lesions among the arsenic-exposed population who were consuming safe water during the previous 5 years, the skin lesions cleared or decreased in 49.7% of

people. However, out of 306 people who did not have such lesions previously, new skin lesions appeared in 32 (10.5%) [110]. Skin lesions were reported to improve to some extent in cases of arsenicosis in Inner Mongolia, China, after the drinking of low arsenic containing water for 1 year. However, a 5-year follow-up study showed no further significant improvement of skin lesions, while the potential risk of arsenic-induced cancers after cutting off high arsenic exposure was still uncertain and indefinite [111].

From the results of the few studies described above it becomes apparent that significant improvement of mild and moderate dermatological manifestations occurs in many cases of arsenicosis after continuous drinking of arsenic-free water. However, symptoms of severe pigmentation and keratosis may persist in spite of stoppage of consumption of arseniccontaminated water.

6.7.2 Specific Therapy

Chelation therapy for chronic arsenic toxicity is thought to be the specific therapy for relief of clinical manifestations and reduction of arsenic stores in the body. Chelating agents BAL and d-penicillamine had been recommended for reduction of arsenic levels in the body [112,113]. Piamphongsant reported the efficacy of penicillamine in the management of chronic arsenic toxicity. The dosage used in adults was 100 mg/day for 2–4 months. Follow-up cases after 10 years of therapy showed that a few cases out of several hundred who received the drug still developed Bowen's disease, whereas those who did not receive the agent developed more extensive lesions of Bowen's disease than those who received treatment. However, raindrop pigmentation and white macules remained unchanged in spite of therapy [114].

Specific chelation therapy with DMSA (dimercaptosuccinic acid), a chelating agent related to BAL, in patients suffering from chronic arsenic toxicity has not yielded better results than those seen in control subjects treated with placebo who were studied in a placebo-controlled trial in West Bengal. Twenty-one consecutive patients with chronic arsenicosis were randomized into two groups. Eleven patients (10 males, ages 25.5 ± 8.0 years) received DMSA at 1400 mg/day (1000 mg/m^2) in four divided doses orally in the first week, and then 1050 mg/ day (750 mg/m^2) in three divided doses during the next 2 weeks. The same was repeated after 3 weeks during which no drug was administered. The other 10 patients (all males, ages 32.2 ± 9.7 years) were given placebo capsules (resembling DMSA) in the same schedule. The patients were blinded about the nature of treatment being given. The symptoms and signs of dermatological manifestations of patients were evaluated by a scoring system before and after treatment. Therapy with DMSA did not cause any significant clinical improvement of dermatological and systemic score compared with patients treated with placebo. No patient developed any therapy-related side effects. The histological abnormalities in skin biopsy did not show any difference in patients treated with DMSA and placebo before and after therapy [26].

The efficacy of treatment with DMPS (dimercaptopropane sulfonate), another chelating agent, was reported in a single-blind placebo-controlled trial in patients suffering from chronic arsenic toxicity in West Bengal, India. The trial design was similar to that carried out in the DMSA trial described above. DMPS was given via a dosage of 100-mg capsules four times a day

Clinical Features	Drug	Before	After	<i>p</i> -value
 Pigmentation	DMPS	1.45 ± 0.52	0.90 ± 0.54	0.02
	Placebo	1.60 ± 0.84	1.10 ± 0.87	0.20
Keratosis	DMPS	1.54 ± 0.68	1.09 ± 0.70	0.14
	Placebo	1.40 <u>+</u> 0.96	1.11 ± 0.87	0.47

 Table 6–5
 Clinical Scores of Patients Before and After Therapy

for a course of 7 days for four courses, with a 1-week drug-free period between each course. Eleven patients (nine males and two females) received the drug, whereas 10 patients (five males, five females) received placebo capsules. The results of this study indicate that DMPS is more effective than a placebo in improving clinical features of chronic arsenic toxicity including skin score of pigmentation (Table 6–5) [87].

The beneficial effect of oral supplementation with vitamin A (retinol) in the treatment of cutaneous arsenicosis was described as early as 1946 [115]. In that report, oral vitamin A, 150,000 USP units per day for 3 months, resulted in a partial regression of hyperpigmentation and hyperkeratosis of the palms of a 39-year-old male who had taken Fowler's solution (potassium arsenite) for treatment of childhood chorea. The effectiveness of management of chronic arsenicosis in Bangladesh by administering a vitamin A, E, C regimen was further evaluated. Forty-three patients with chronic arsenicosis had been given the following regimen for 6 months: (1) withdrawal of exposure, (2) vitamin A (50,000 IU on alternate days), (3) vitamin E (200 mg daily), (4) vitamin C (250 mg twice daily), and topical application of a keratolytic agent (for arsenical keratosis only). Improvement of melanosis and keratosis was observed in 90.9% and 86.4% of patients, respectively, from among 22 patients who had used safe water and had taken the regimen regularly [116]. However, the characteristics of skin lesions for evaluation of the severity of arsenicosis were not described in the methodology.

A case series of nine patients with cutaneous arsenicosis were treated with oral etretinate, a synthetic aromatic retinoid, for 2–7 months. Clinical and histopathological improvement was noted in arsenical hyperkeratosis, but not in hyperpigmentation [117]. Regression of arsenical keratosis with etretinate treatment was also published [118]. Piamphongsant reported that moderate and severe degrees of palmoplantar keratosis due to arsenicosis improved with treatment with etretinate at a dose of 25 mg twice daily for 3–4 months and then 25 mg/day for 2 months. Keratotic papules and keratoderma usually slough off in 3–4 months. Many of the patients complained of sticky palms due to mucoid sweat [115]. It is noteworthy that etretinate and other retinoids have been reported to have antikeratinizing effects in other disorders of keratinization, such as hereditary palmoplantar keratoderma, pityriasis rubra pilaris, and certain ichthyoses [119].

Surgical excision is usually performed for the management of Bowen's disease. The combination of 5-aminolevulinic acid with photodynamic therapy and 5-fluororaccil iontophoresis may be an alternative therapy for the condition [114,120]. Supportive treatment could help in reducing many symptoms of these patients. Treatment in hospital with a good nutritious diet has been found to reduce symptoms in many patients and to reduce symptom score in subsets of placebo-treated arsenicosis patients during the course of the DMSA and DMPS trial [26,87]. Presently, the most prevailing practice of symptomatic treatment of keratosis is to apply locally 5–10% of salicylic acid and 10–20% urea-based ointment on keratotic skin lesions [2]. A higher dose (20% salicylic acid) may be used in severe cases of arsenical keratosis.

References

- Guha Mazumder DN, Chakraborty AK, Ghosh A, Das Gupta. J, Chakraborty DP, Dey SB, et al. Chronic arsenic toxicity from drinking tube-well water in rural West Bengal. Bull World Health Org 1988;66:499–506.
- [2] World Health Organization (WHO) A field guide for detection, management and surveillance of arsenicosis cases. Caussy D, editor. Technical Publication No. 30. New Delhi: WHO, SEARO; 2005. p. 5–18.
- [3] Harrington JM, Middaugh JP, Morse DL, Housworth J. A survey of a population exposed to high concentrations of arsenic in well water in Fairbanks, Alaska. Am J Epidemiol 1978;108:377–85.
- [4] Valentine JL, Reisbord LS, Kang HK, Schluchter MD. Arsenic effects on population health histories. In: Mills CF, Bremner I, Chester JK, editors. Trace elements in man and animals—TEMA 5. Proceedings of the Fifth International Symposium on Trace Elements in Man and Animals. Slough, UK: Commonwealth Agricultural Bureau; 1985.
- [5] Black MD. Prolonged ingestion of arsenic. Pharm J 1967;9:593-7.
- [6] Yeh S. Skin cancer in chronic arsenicism. Hum Pathol 1973;4:469–85.
- [7] Tay CH. Cutaneous manifestations of arsenic poisoning due to certain Chinese herbal medicine. Australas J Dermatol 1974;15:121–31.
- [8] Cebrian ME, Albores A, Aguilar M, Blakely E. Chronic arsenic poisoning in the north of Mexico. Hum Toxicol 1983;2:121–33.
- [9] Saha KC. Melanokeratosis from arsenic contaminated tubewell water. Ind J Dermat 1984;29:37-46.
- [10] Saha KC. Chronic arsenical dermatoses from tube-well water in West Bengal during 1983-87. Ind J Dermatol 1995;40:1–12.
- [11] Guha Mazumder DN, Das Gupta J, Santra A, Pal A, Ghosh A, Sarmar S. Chronic arsenic toxicity in West Bengal—the worst calamity in the world. J Indian Med Assoc 1998;96(4–7):18.
- [12] Guha Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborty D, et al. Arsenic levels in drinking water and the prevalence of skin lesions in West Bengal, India. Int J Epidemiol 1998;27:871–7.
- [13] Luchtrath H. The consequences of chronic arsenic poisoning among Moselle wine growers. Pathoanatomical investigations of post-mortem examinations performed between 1960 and 1977. J Cancer Res Clin Oncol 1983;105:173–82.
- [14] Szuler IM, Williams CN, Hindmarsh JT, Park-Dinesoy H. Massive variceal hemorrhage secondary to presinusoidal portal hypertension due to arsenic poisoning. Can Med Assoc J 1979;120:168–71.
- [15] National Research Council. Arsenic in drinking water. Washington, DC: National Academy Press; 1999.
- [16] Neubauer O. Arsenical cancer; a review. Br J Cancer 1947;1:192-251.
- [17] Zaldivar R. Arsenic contamination of drinking water and food-stuffs causing endemic chronic poisoning. Beitr Pathol 1974;151:384–400.
- [18] Sommers SG, Manus MC. Multiple arsenical cancers of skin and internal organs. Cancer 1953;6:347-59.

- [19] Tseng WP, Chu HM, How SW, Fong JM, Lin CS, Yeh S. Prevalence of skin cancer in an endemic area of chronic arsenicism in Taiwan. J Natl Cancer Inst 1968;40:453–63.
- [20] Borgono JM, Vicent P, Venturino H, Infante A. Arsenic in the drinking water of the city of Antofagasta: epidemiological and clinical study before and after the installation of the treatment plant. Environ Health Perspect 1977;19:103–5.
- [21] Ahmad SA, Bandaranayake D, Khan AW, Hadi SA, Uddein G, Halim MA. Arsenic contamination in ground water and arsenicosis in Bangladesh. Int J Environ Health Res 1997;7:271–6.
- [22] Ahmad SA, Sayed MHSU, Hadi SA, Faruquee MH, Jalil MA, Ahmed R, et al. Arsenicosis in a village in Bangladesh. Int J Environ Health Res 1999;9:187–95.
- [23] Guha Mazumder DN, Ghosh A, Majumdar KK, Ghosh N, Saha C, Guha Mazumder RN. Arsenic contamination of ground water and its health impact on population of district of Nadia, West Bengal, India. Indian J Community Med 2010;35:331–8.
- [24] Alain G, Tousignant J, Rozenfarb E. Chronic arsenic toxicity. Int J Dermatol 1993;32:899-901.
- [25] Ratnam KV, Espy MJ, Muller SA, Smith TF, Su WP. Clinicopathologic study of arsenic-induced skin lesions: no definite association with human papillomavirus. J Am Acad Dermatol 1992;27:120–2.
- [26] Guha Mazumder DN, Ghoshal UC, Saha J, Santra A, De BK, Chatterjee A, et al. Randomized placebocontrolled trial of 2,3-dimercaptosuccinic acid in therapy of chronic arsenicosis due to drinking arseniccontaminated subsoil water. Clin Toxicol 1998;36:683–90.
- [27] Tseng WP. Effects and dose-response relationships of skin cancer and blackfoot disease with arsenic. Environ Health Perspect 1977;19:109–19.
- [28] Liao WT, Yu CH, Lam CE, Lu CH, Chang CH, Chang LW, et al. Differential effects of arsenic on cutaneous and systemic abnormality: focusing on CD4+ cell apoptosis in patients with arsenic induced Bowen's disease. Carcinogenesis 2009;30:1064–72.
- [29] Lund HZ. Tumors of the skin. Atlas of tumor pathology. Washington, DC: Armed Forces Institute of Pathology; 1957. Sec. 1, Fasc. 2, p. 192–200, 258–264.
- [30] Belisario JC. Cancer of the skin., 1957. London: Butterworth & Co. (Publishers) Ltd; 1957. p. 26, 97.
- [31] Haque R, Guha Mazumder DN, Samanta S, Ghosh N, Kalman D, Smith MM, et al. Arsenic in drinking water and skin lesions: dose-response data from West Bengal, India. Epidemiology 2003;14:174–82.
- [32] Chakraborti D, Rahman MM, Paul K, Sengupta MK, Chowdhury UK, Lodh D, et al. Arsenic calamity in the Indian subcontinent—what lessons have been learned? Talanta 2002;58:3–22.
- [33] Tondel M, Rahman M, Magnuson A, Chowdhury OA, Faruquee MH, Ahmad SA. The relationship of arsenic levels in drinking water and the prevalence rate of skin lesions in Bangladesh. Environ Health Perspect 1999;107:727–9.
- [34] Majumder KK, Guha Mazumder DN, Ghosh N, Ghosh A, Lahiri S. Systemic manifestations in chronic arsenic toxicity in absence of skin lesions in West Bengal. Indian J Med Res 2009;129:75–8.
- [35] De BK, Majumdar D, Sen S, Guru S, Kundu S. Pulmonary involvement in chronic arsenic poisoning from drinking contaminated ground-water. J Assoc Physicians India 2004;52:395–400.
- [36] Zaldivar R, Ghai GL. Mathematical model of mean age, mean arsenic dietary dose and age-specific prevalence rate from endemic chronic arsenic poisoning: a human toxicology study. Zentralbl Bakteriol B 1980;170:402–8.
- [37] Rosenberg HG. Systemic arterial disease and chronic arsenicism in infants. Arch Pathol 1974;97:360–5.
- [38] Guha Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborti D, et al. Arsenic in drinking water and the prevalence of respiratory effects in West Bengal, India. Int J Epidemiol 2000;29:1047–52.

- [39] Milton AH, Hasan Z, Rahman A, Rahman M. Chronic arsenic poisoning and respiratory effects in Bangladesh. J Occup Health 2001;43:136–40.
- [40] von Ehrenstein OS, Guha Mazumder DN, Yuan Y, Samanta S, Balmes J, Sil A, et al. Decrements in lung function related to arsenic in drinking water in West Bengal, India. Am J Epidemiol 2005;162:533–41.
- [41] Guha Mazumder DN, Craig S, Bhattacharya P, von Ehrenstein OS, Ghosh N, Gotway M, et al. Bronchiectasis in persons with skin lesions resulting from arsenic in drinking water. Epidemiology 2005;16:760–5.
- [42] Tsai S-M, Wang T-N, Ko Y-C. Mortality for certain diseases in areas with high levels of arsenic in drinking water. Arch Environ Health 1999;54:186–93.
- [43] Smith AH, Marshall G, Yuan Y, Ferreccio C, Liaw J, Von Ehrenstein OS, et al. Increased mortality from lung cancer and bronchiectasis in young adults after exposure to arsenic in utero and in early childhood. Environ Health Perspect 2006;114:1293–6.
- [44] Guha Mazumder DN, Ghosh N, De BK, Santra A, Das S, Lahiri S, et al. Epidemiological study on various non-carcinomatous manifestations of chronic arsenic toxicity in a district of West Bengal. In: Abernathy C.O.Calderon RL, Chappell WR, editors. Arsenic exposure and health effects, vol. IV. Oxford, UK: Elsevier Science; 2001. p. 153–64.
- [45] Ma HZ, Xia YJ, Wu KG, Wu KG, Sun TZ, Mumford JL. Human exposure to arsenic and health effects in Bayingnormen, Inner Mongolia. In: Chappell WR, Abernathy CO, Calderon RL, editors. Arsenic exposure and health effects. Amsterdam: Elsevier Science; 1999. p. 127–31.
- [46] Morris JS, Schmid M, Newman S, Scheuer PJ, Sherlock S. Arsenic and non-cirrhotic portal hypertension. Gastroenterology 1974;66:86–94.
- [47] Nevens F, Van Steenbergen W, Sciot R, Desmet V, Groote JD. Arsenic and non-cirrhotic portal hypertension—a report of eight cases. J Hepatol 1990;11:80–5.
- [48] Franklin M, Bean W, Harden RC. Fowler's solution as an etiologic agent in cirrhosis. Am J Med Sci 1950;219:589–96.
- [49] Datta DV, Mitra SK, Chhuttani PN, Chakravarti RN. Chronic oral arsenic intoxication as a possible etiological factor in idiopathic portal hypertension (non-cirrhotic portal fibrosis) in India. Gut 1979;20:378-84.
- [50] Santra A, Das Gupta J, De BK, Roy B, Guha Mazumder DN. Hepatic manifestations in chronic arsenic toxicity. Ind J Gastroenterology 1999;18:152–5.
- [51] Hernandez-Zevala A, Del Razo LM, Aguilar C, Garcia-Vargas GG, Borja VH, Cebrián ME. Alteration in bilirubin excretion in individuals chronically exposed to arsenic in Mexico. Toxicol Lett 1998;99:79–84.
- [52] Chakraborty AK, Saha KC. Arsenical dermatosis from tubewell water in West Bengal. Indian J Med Res 1987;85:326–34.
- [53] Santra A, Maiti A, Das S, Lahiri S, Charkaborty SK, Guha Mazumder DN. Hepatic damage caused by chronic arsenic toxicity in experimental animals. Clin Toxicol 2000;38:395–405.
- [54] Das S, Santra A, Lahiri S, Guha Mazumder DN. Implications of oxidative stress and hepatic cytokine (TNF alpha and IL-6) response in the pathogenesis of hepatic collagenesis in chronic arsenic toxicity. Toxicol Appl Pharmacol 2005;204:18–26.
- [55] Yu HS, Sheu HM, Ko SS, Chiang LC, Chien CH, Lin SM, et al. Studies on blackfoot disease and chronic arsenism in southern Taiwan: with special reference to skin lesions and fluorescent substances. J Dermatol 1984;11:361–70.
- [56] Tseng WP, Chen WY, Sung JL, Chen JS. A clinical study of blackfoot disease in Taiwan: an endemic peripheral vascular disease. Memoirs Coll Med Natl Taiwan Univ 1961;7:1–17.

- [57] Tseng C-H, Chong C-K, Chen C-J, Tai T-Y. Dose-response relationship between peripheral vascular disease and ingested inorganic arsenic among residents in blackfoot disease endemic villages in Taiwan. Atherosclerosis 1996;120:125–33.
- [58] Engel RR, Smith AH. Arsenic in drinking water and mortality from vascular disease: an ecological analysis in 30 counties in the United States. Arch Environ Health 1994;49:418–27.
- [59] Chen CJ, Hsueh YM, Lai MS, Shyu MP, Chen SY, Wu MM, et al. Increased prevalence of hypertension and long-term arsenic exposure. Hypertension 1995;25:53–60.
- [60] Rahman M, Tondel M, Ahmad SA, Chowdhury IA, Faruquee MH, Axelson O. Hypertension and arsenic exposure in Bangladesh. Hypertension 1999;33:74–8.
- [61] Guha Mazumder DN, Kar-Purkayastha I, Ghose A, Mistry G, Saha C, Nandy AK, et al. Hypertension in chronic arsenic exposure: a case control study in West Bengal. J Environ Sci Health Part A 2012;47:1514–20.
- [62] Chen C-J, Chiou HY, Chiang M-H, Lin L-J, Tai T-Y. Dose-response relationship between ischemic heart disease mortality and long-term arsenic exposure. Arterioscler Thromb Vasc Biol 1996;16:504–10.
- [63] Chen C-J, Chiou H-Y, Huang W-I, Chen S-Y, Hsueh Y-M, Tseng C-H, et al. Systemic non-carcinogenic effects and developmental toxicity of inorganic arsenic. In: Abernathy RL, Calderon RL, Chappell. WR, editors. Arsenic exposure and health effects. London: Chapman & Hall; 1997. p. 124–34.
- [64] Hsueh YM, Wu WL, Huang YL, Chiou HY, Tseng CH, Chen CJ. Low serum carotene level and increased risk of ischemic heart disease related to long-term arsenic exposure. Atherosclerosis 1998;141:249–57.
- [65] Tseng CH, Chong CK, Tseng CP, Hsueh YM, Chiou HY, Tseng CC, et al. Long-term arsenic exposure and ischemic heart disease in arseniasis-hyperendemic villages in Taiwan. Toxicol Lett 2003;137:15–21.
- [66] Chiou JM, Wang SL, Chen CJ, Deng C-R, Lin W, Tai TY. Arsenic ingestion and increased microvascular disease risk: observations from the south-western arseniasis-endemic area in Taiwan. Int J Epidemiol 2005;34:93.
- [67] Hotta N. Clinical aspects of chronic arsenic poisoning due to environmental and occupational pollution in and around a small refining spot [in Japanese]. Nippon Taishitsugaku Zasshi [Jpn J Const Med] 1989;53:49–70.
- [68] Kilburn KH. Neurobehavioral impairment from long-term residential arsenic exposure. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic exposure and health effects. London: Chapman & Hall; 1997. p. 159–77.
- [69] Guha Mazumder DN, Das Gupta J, Santra A, Pal A, Ghose A, Sarkar S, et al. Non-cancer effects of chronic arsenicosis with special reference to liver damage. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic exposure and health effect. London: Chapman & Hall; 1997. p. 112–23.
- [70] Hindmarsh JT, McLetchine OR, Heffernan LPM, Hayne OA, Ellenberger HAA, McCurdy RR, et al. Electromyographic abnormalities in chronic environmental arsenicalism. J Anal Toxicol 1977;11:270–6.
- [71] Mukherjee SC, Rahman MM, Chowdhury UK, Sengupta MK, Lodh D, Chanda CR, et al. Neuropathy in arsenic toxicity from groundwater arsenic contamination in West Bengal, India. J Environ Sci Health A 2003;38:165–83.
- [72] Tseng HP, Wang YH, Wu MM, The HW, Chiou HY, Chen CJ. Association between chronic exposure to arsenic and slow nerve conduction velocity among adolescents in Taiwan. J Health Popul Nutr 2006;24:182–9.
- [73] Chiou HY, Hsueh YM, Hsieh LI, Chang SF, Hsu YH, Chen CJ. Arsenic methylation capacity, body retention, and null genotypes of glutathione S-transferase M1 and T1 among current arsenic exposed residents in Taiwan. Mutat Res 1997;386:197–207.
- [74] Wu MM, Kuo TL, Hwang YH, Chen CJ. Dose-response relation between arsenic concentration in well water and mortality from cancers and vascular diseases. Am J Epidemiol 1989;130:1123–32.
- [75] Cheng TJ, Ke DS, Wang YJ, Guo HR. The association between arsenic exposure and cerebrovascular disease—a study in the blackfoot disease endemic area in Taiwan. Epidemiology 2009;20:S59.

- [76] Meliker JR, Robert LW, Lorraine LC, Nriagu JO. Arsenic in drinking water and cerebrovascular disease, diabetes mellitus, and kidney disease in Michigan: a standardized mortality ratio analysis. Environ Health 2007;6:4.
- [77] Terada H, Sasagawa T, Saito H, Shirata H, Sekiya T. Chronic arsenical poisoning and hematopoietic organs. Acta Med Biol 1962;9:279–92.
- [78] Southwick JW, Western AE, Beck MM. An epidemiological study of arsenic in drinking water in Millard County, Utah. In: Lederer W, Fensterheim R, editors. Arsenic: industrial, biomedical, environmental perspectives. New York: Van Nostrand Reinhold; 1983. p. 210–25.
- [79] Lai MS, Hsueh YM, Chen CJ, Shyu MP, Chen SY, Kuo TL, et al. Ingested inorganic arsenic and prevalence of diabetes mellitus. Am J Epidemiol 1994;139:484–92.
- [80] Rahman M, Tondel M, Ahmad SA, Axelson O. Diabetes mellitus associated with arsenic exposure in Bangladesh. Am J Epidemiol 1998;148:198–203.
- [81] Tseng CH, Tai TY, Chong CK, Tseng CP, Lai MS, Lin BJ, et al. Long-term arsenic exposure and incidence of non-insulin-dependent diabetes mellitus: a cohort study in arseniasis-hyperendemic villages in Taiwan. Environ Health Perspect 2000;108:847–51.
- [82] Chen Y, Ahsan H, Slavkovich V, Peltier GL, Gluskin RT, Parvez F, et al. No association between arsenic exposure from drinking water and diabetes mellitus: a cross-sectional study in Bangladesh. Environ Health Perspect 2010;118:1299–305.
- [83] Hossain MK, Khan MM, Alam MA, Chowdhury AK, Delwar V, Feroze AM, et al. Manifestation of arsenicosis patients and factors determining the duration of arsenic symptoms in Bangladesh. Toxicol Appl Pharmacol 2005;208:78–86.
- [84] Baidya K, Raj A, Mondal V, Bhaduri G, Todani A. Persistent conjunctivitis associated with drinking arseniccontaminated water. J Ocul Pharmacol Ther 2006;22:208–11.
- [85] Lin W, Wang SL, Wu HJ, Chang KH, Yeh P, Chen CJ, et al. Associations between arsenic in drinking water and pterygium in southwestern Taiwan. Environ Health Perspect 2008;116:952–55.
- [86] See LC, Chiou HY, Lee JS, Hsueh YM, Lin SM, Tu MC, et al. Dose-response relationship between ingested arsenic and cataracts among residents in southwestern Taiwan. J Environ Sci Health A Tox Hazard Subst Environ Eng 2007;42:1843–5.
- [87] Guha Mazumder DN, De BK, Santra A, Ghosh N, Das S, Lahiri S, et al. Randomized placebo-controlled trial of 2,3-dimercapto-1-propane-sulfonate (DMPS) in therapy of chronic arsenicosis due to drinking arsenic contaminated water. J Toxicol Clin Toxicol 2001;39:665–74.
- [88] Guha Mazumder DN, Das Gupta J, Chakraborty AK, Chatterjee A, Das D, Chakraborti D. Environmental pollution and chronic arsenicosis in south Calcutta. Bull World Health Org 1992;70:481–5.
- [89] Hsieh FI, Hwang TS, Hsieh V, Lo HC, Su CT, Hsu HS, et al. Risk of erectile dysfunction induced by arsenic exposure through well water consumption in Taiwan. Environ Health Perspect 2008;116:532–6.
- [90] Chen Y, Parvez F, Liu M, Pesola GR, Gamble MV, Slavkovich V, et al. Association between arsenic exposure from drinking water and proteinuria: results from the Health Effects of Arsenic Longitudinal Study. Int J Epidemiol 2011;40:828–35.
- [91] Hopenhyne-Rich C, Browning S, Hertz-Picciotto I, Ferreccio C, Peralta C, Gibb H. Chronic arsenic exposure and risk of infant mortality in two areas in Chile. Environ. Health Perspect. 2000;108:667–73.
- [92] Milton AH, Smith W, Rahman B, Hasan Z, Kulsum U, Dear K, et al. Chronic arsenic exposure and adverse pregnancy outcomes in Bangladesh. Epidemiology 2005;16:82–6.
- [93] Ahmad SA, Sayed MHSU, Barua S, Khan MH, Faruquee M, Jalil A, et al. Arsenic in drinking water and pregnancy outcomes. Environ Health Perspect 2001;109:629–31.
- [94] von Ehrenstein OS, Guha Mazumder DN, Smith MH, Ghosh N, Yuan Y, Windham G, et al. Pregnancy outcomes, infant mortality and arsenic in drinking water in West Bengal, India. Am J Epidemiol 2006;163:662–9.

- [95] International Agency of Research on Cancer (IARC). Some drinking-water disinfectants and contaminants, including arsenic monographs on the evaluation of carcinogenic risks to humans, vol. 84. Lyon, France: WHO; 2004. p. 61–96.
- [96] Bergoglio RM. Cancer mortality in zones of arsenical waters of the province of Cordoba, Argentine Republic. Contribution to the Regional Pathology of Cancer. Prensa Med Argent 1964;51:994-8. [In Spanish.]
- [97] Smith AH, Goycolea M, Haque R, Biggs ML. Marked increase in bladder and lung cancer mortality in a region of northern Chile due to arsenic in drinking water. Am J Epidemiol 1998;147:660–9.
- [98] Chen CJ, Chuang YC, Lin TM, Wu HY. Malignant neoplasms among residents of a blackfoot diseaseendemic area in Taiwan: high-arsenic artesian well water and cancers. Cancer Res 1985;45:5895–9.
- [99] Fierz U. Catamnestic research into the side effects of inorganic arsenotherapy in skin disease. Dermatologica 1965;131:41–58. [in German.]
- [100] Rattner H, Dorne M. Arsenical pigmentation and keratoses. Arch Dermatol Syphilol 1943;48:458-60.
- [101] Vahter M. Species differences in the metabolism of arsenic compounds. Appl Organomet Chem 1994;8:175–82.
- [102] Buchet JP, Hoet P. Consistency among biomarkers for the assessment of exposure to inorganic arsenic: a literature review. Third International Conference on "As Exposure and Health Effects," July 12–15, San Diego, CA; 1998.
- [103] Chowdhury TR, Mandal BK, Samanta G, Basu GK, Chowdhury PP, Chanda CR, et al. Arsenic in groundwater in six districts of West Bengal, India: the biggest arsenic calamity in the world: the status report up to August, 1995. In: Abernathy CO, Calderon RL, Chappell. WR, editors. Arsenic exposure and health effects. London: Chapman & Hall; 1997. p. 93–111.
- [104] Ahsan H, Perrin M, Rahman A, Parvez F, Stute M, Zeng Y, et al. Associations between drinking water and urinary arsenic levels and skin lesions in Bangladesh. J Occup Environ Med 2000;42:1195–201.
- [105] Shapiro HA. Arsenic content of human hair and nails. Its interpretation. J Forensic Med 1967;14:65-71.
- [106] Valentine JL, Kang HK, Spivey G. Arsenic levels in human blood, urine, and hair in response to exposure via drinking water. Environ Res 1979;20:24–32.
- [107] Narang APS, Chawls LS, Khurana SB. Levels of arsenic in Indian opium eaters. Drug Alcohol Depend 1987;20:149–53.
- [108] Takagi Y. Survey of trace elements in human nails: an international comparison. Bull Environ Contam Toxicol 1988;41:690–5.
- [109] Oshikawa S, Geater A, Chongsuvivatwong V, Piampongsan T, Chakraborti D, Samanta G, et al. Long-term changes in severity of arsenical skin lesions following intervention to reduce arsenic exposure. Environ Sci 2001;85:435–48.
- [110] Guha Mazumder DN, Ghosh N, Mazumder K, Santra A, Lahiri S, Das S, et al. Natural history following arsenic exposure: a study in an arsenic endemic area of West Bengal, India. In: Chappell WR, Abernathy CO, Calderon RL, Thomas DJ, editors. Arsenic exposure and health effects V. Oxford, UK: Elsevier Science; 2003. p. 381–9.
- [111] Sun G, Li X, Pi J, Sun Y, Li B, Jin Y, et al. Current research problems of chronic arsenicosis in China. J Health Popul Nutr 2006;24:176–81.
- [112] Tannapai S, Piamphongsant T, Nakjang Y. Chronic environmental arsenism. Bull Dept Med Serv 1989;14:307-14.
- [113] Piamphongsant T, Erpaibool P, Sivaroroskul P. Treatment of chronic arsenic poisoning: comparative study between penicillamine and BAL. Bull Dept Med Serv 1989;18:88–91.
- [114] Piamphongsant T. Chronic environmental arsenic poisoning. Int J Dermatol 1999;38:401-10.

- [115] Hall AF. Arsenical keratosis disappearing with vitamin A therapy. Arch Dermatol Syphilol 1946;53:154.
- [116] Ahmad SA, Faruquee MH, Sayed MHSU, Khan MH, Jalil MA, Ahmed R, et al. Chronic arsenicosis: management by vitamin A, E, C regimen. J Prev Soc Med 1998;17:19–26.
- [117] Thiaprasit M. Chronic cutaneous arsenism treated with aromatic retinoid. J Med Assoc Thai 1984;67:93–100.
- [118] Sass U, Grosshans E, Simonart JM. Chronic arsenicism: criminal poisoning or drug intoxication? Report of two cases. Dermatology 1993;186:303–5.
- [119] Fritsch PO. Retinoids in psoriasis and disorders of keratinization. J Am Acad Dermatol 1992;27:S8-S14.
- [120] Welch ML, Grabski WJ, McCollough ML. 5-Fluorouracil iontophoretic therapy for Bowen's disease. J Am Acad Dermatol. 1997;36:956–8.

This page intentionally left blank



Changing Concept of Arsenic Toxicity with Development of Speciation Techniques

Badal Kumar Mandal

TRACE ELEMENTS SPECIATION RESEARCH LABORATORY, ENVIRONMENTAL AND ANALYTICAL CHEMISTRY DIVISION, SCHOOL OF ADVANCED SCIENCES, VIT UNIVERSITY, VELLORE, TAMILNADU, INDIA

CHAPTER OUTLINE

7.1 Introduction	179
7.2 Conclusions	197
References	197

7.1 Introduction

Arsenic (As), a metalloid, is a ubiquitous element that ranks 20th in abundance in Earth's crust, 14th in sea water, and 12th in the human body [1,2]. Since its isolation (as a gray, shiny, metallic-looking element with atomic number of 33 and atomic weight of 74.92) in AD 1250 by Albertus Magnus, this element has been the center of controversy in human history. It is found as a component of more than 245 minerals. These are mostly ores containing sulfide, along with copper, nickel, lead, cobalt, or other metals. Being a group VB element in the periodic table along with N, P, Sb, and Bi, it can exist in different valency states such as –III, 0, +III, and +V. Although it has a metallic allotropic form like the two higher group VA elements, antimony and bismuth, arsenic's chemical behavior is quite similar to that of phosphorus.

Based on different chemical forms and oxidation states, many As compounds exist in nature as well as in biological systems (Table 7–1). Chemical speciation is the determination of the individual concentrations of the various forms of an element that together make up the total concentration of that element. Still, 1970 arsenic species in natural waters were considered to be inorganic forms, such as arsenate [As(V)] and arsenite [As(III)]. Although the Guitzeit method was used initially for qualitative testing of As in aqueous solutions, this method was later improved for a quantification of lower amounts of As in water and other beverages [4,5] over the last 100 years. Until recently the modified Guitzeit technique [6] was used

Name of Species	Abbreviation	Structure of Species	
Most often determined			
Inorganic compounds			
Arsenous acid (arsenite)	As(III)	As(OH) ₃	
Arsenic acid (arsenate)	As(V)	AsO(OH) ₃	
Organic compounds			
Monomethylarsonic acid	MMA(V)	CH ₃ AsO(OH) ₂	
Dimethylarsinic acid	DMA(V)	$(CH_3)_2AsO(OH)$	
Arsenobetaine	AsB	(CH ₃) ₃ As ⁺ CH ₂ COOH	
Arsenocholine	AsC	(CH ₃) ₃ As ⁺ CH ₂ CH ₂ OH	
Rarely determined			
Trimethyl arsine oxide	TMAO	(CH ₃) ₃ AsO	
<i>p</i> -Arsanilic acid	acid p-ASA		
Trimethylarsoniopropionate	ТМАР	(CH ₃) ₃ AS ⁺ CH ₂ CH ₂ COO ⁻	
Dimethyldithioarsinic acid	DMDTA(V)	$(CH_3)_2As(S)(SH)$	
Dimethylmonothioarsinic acid	DMMTA(V)	$(CH_3)_2As(S)(OH)$	
Thio-arsenosugar glycerol	Thio-Gly	C ₇ H ₁₄ AsO ₃ (S)OCH ₂ CH(OH)CH ₂ OH	
Thio-arsenosugar sulfate	Thio-SO ₄	C ₁₀ H ₁₆ AsO ₅ (S)OSO ₃ H	
Thio-arsenosugar phosphate	Thio-PO ₄	C ₇ H ₁₄ AsO ₃ (S)OCH ₂ CH(OH)CH ₂ OPO ₂ – OCH ₂ CH(OH)CH ₂ OH	
Trimethylarsine sulfide	TMAS	$(CH_3)_3As(S)$	
Thio-arsenosugar sulfonate	Thio-SO₃	$C_{10}H_{16}AsO_5(S)SO_3H$	
Thio-dimethylarsenoacetate	Thio-DMAA	(CH ₃) ₂ As(S)CH ₂ COOH	
Thio-dimethylarsinate	Thio-DMA	$(CH_3)_2As(S)(OH)$	
Thio-dimethylarsenoethanol	Thio-DMAE	(CH ₃) ₂ As(S)CH ₂ COOH	
Phenylarsonic acid	PhAs, PAA	C ₆ H ₅ AsO(OH) ₂	
4-Hydroxy-3-nitrobenzenearsonic acid	Roxarson, Roxarsone	C ₆ H ₅ As(O)(OH) ₂	
Phenylarsine oxide	PhAsO, PAO	C ₆ H ₅ AsO	
Dimethylarsinous acid	DMA(III) (CH ₃) ₂ As(OH)		
Monomethylarsonous acid	nous acid MMA(III) (CH ₃) ₃ As(OH) ₂		
Dimethylarsinoylacetic acid	/larsinoylacetic acid DMAA (CH ₃) ₂ As(O)CH ₂ COOH		
Dimethylarsinoylethanol	oylethanol DMAE (CH ₃) ₂ As(O)CH ₂ CH ₂ OH		
Tetramethylarsonium ion	TMAs, Tetra, TeMA, TMA	$(CH_3)_4As^+$	
Diphenylarsinic acid	DPAA	$(C_6H_5)_2AsO(OH)$	

Table 7–1 Arsenic Compounds Found in Water Matrix

Reprinted from [3] with permission from Elsevier.

to quantify total inorganic As (iAs) in water samples. But this method was not reliable for water samples that contained less than $40 \mu g/L$ arsenic [7]. In addition, it was also used to quantify As(III) and As(V) in water samples [6]. A recent adaptation of this method is the analysis of the silver diethyldithiocarbamate (SDDC) complex using graphite tube furnace atomic absorption spectrometry (AAS), which gives an improved detection limit of about 10 ng/L [8].

Thereafter the methylated forms, i.e., mono methylarsonic acid [MMA(V)] and dimethylarsinic acid [DMA(V)], have been reported to be the minor species in water [9–12]. In the late

Techniques	LOD (µm/L)	Sample Size (mL)	Major Equipment Cost (US\$)	Analytical Throughput	Comments	Selected Standards
HGAAS	0.05	50	20–100,000	30–60	Single element	ISO 11969 SM 3114
GFAAS	1–5	1–2	40-100,000	50–100	Single element	ISO/CD 15586 SM 3113
ICP-AES	35–50	10–20	60–100,000	50–100	Single element	ISO/CD 11885 SM 3120
ICP-MS	0.02-1	10–20	150–400,000	20–100	Single element	SM 3125 USEPA 1638
HGAFS	0.01	40–50	20–25,000	30–60	Single element	CEN/TC/230/WG1/ TG 12 N 3
ASV	0.1	25–50	10–20,000	25–30	Only free dissolve arsenic	USEPA 7063
SDDC	1–10	100	2–10,000	20–30	Limited to water samples	SM 3500 ISO 6595

 Table 7–2
 Summary of Analytical Methods

1980s and early 1990s several laboratory-based analytical methods were well described for routine analyses of aqueous samples such as hydride generation atomic absorption spectrometry (HGAAS), graphite furnace atomic absorption spectrometry (GFAAS), atomic fluorescence spectrometry with hydride system (HGAFS), inductively coupled plasma atomic emission spectrometry (ICP-AES or ICP-OES), inductively coupled plasma mass spectrometry (ICP-MS), anodic stripping voltammetry (ASV), and the silver diethyldithiocarbamate (SDDC)-based spectrometric method using Guitzeit apparatus. The last-mentioned method has been widely used due to its simplicity regarding instrumentation (spectrometry) but it has serious limitations of accuracy and toxicity due to handling of carcinogenic organic solvent chloroform and the highly toxic arsine gas. Also, at least a 100 mL water sample containing a high concentration of As (\approx 50 µg/L) is needed for a reproducible and accurate result. The detection limits for As range between less than 0.1 and 50 µg/L depending upon the technique employed [13,14]. The limits of detection of the above analytical techniques is summarized in Table 7-2 [15-21].

Arsenic is a known human carcinogen that depends on its chemical forms [12,22,23] as well as oxidation states [24]. Humans are exposed to iAs via air, water, and food. Due to lack of hyphenated techniques for speciation of different arsenicals with respect to the methylation and oxidation states of As, researchers studied only the metabolic pattern of iAs in humans and rodents [25–31]. Rodents and humans can metabolize iAs by consecutive reduction and oxidative methylations after ingestion via drinking water or eating food. Based on the studies of iAs metabolites in rodents and humans [25–37], it was found that approximately 60–75% of the dose was excreted in urine within a few days and the distribution of arsenic species in urine was 10–15% inorganic arsenic, 10–15% MMA(V), and 60–80% DMA(V).

Most mammals methylate iAs to MMA and DMA [38]. Normally, iAs is more reactive to tissue constituents compared to methylated arsenic metabolites, is more toxic, and is less readily excreted in the urine [25–38]. Inorganic arsenic, especially As(III), is the main form of arsenic interacting with tissues [26,32,39]. Due to lower methylation capacity iAs is associated with higher tissue concentrations [27,28] and leads to a higher risk of toxic effects. Based on these results, the methylation of arsenic was considered a detoxification mechanism in mammalian species and humans.

There are major species differences with respect to arsenic methylation [25–38]. In brief, humans excrete significant amounts of MMA (10–20% of the total metabolite concentration in the urine) following exposure to iAs, whereas mammalian species methylate arsenic more efficiently than humans and the percentage of DMA is greater in urinary arsenic compared to MMA. The lack of ability to methylate iAs in mammalian species such as chimpanzees, marmoset monkeys, squirrel monkeys, tamarins, and guinea pigs [29,30,40–42] results in increased tissue concentrations of arsenic. This lower rate of excretion compared to animal species creates complexity with these species, which ultimately leads to toxicity.

Recently, some investigations have reported different methylation patterns of iAs [30,31,43,44] in arsenic-affected populations. The native Andean people, the Mestizo population in northern Argentina, and the population from West Bengal, India, excrete just a few percent of MMA in urine [30,31,44], whereas a Taiwan study indicates an unusually high percentage of MMA in urine [43]. It was assumed that an inhibition of the enzymes (methyl-transferases) involved in the methylation of arsenic may be responsible for this variation in methylation capacity among these populations [45].

Vahter (2000) has reported the biotransformation of iAs and its toxicity among different As-affected populations after comparing results of different studies reported in the literature from the As-affected regions as well as after a single dose of As(V) in different mammalian species [46]. The author attempted to find out the reason for the variation in methylation capacity among the mammalian species (Figure 7–1) and in the As-affected populations



FIGURE 7–1 Species variations in the excretion of methylated arsenic metabolites following a single dose of arsenate. *Reprinted from* [47] with permission from Elsevier.

(Figures 7–2–7–4). It is very clear from Figure 7–1 that the marmoset monkey and the chimpanzee do not methylate As(V). Figures 7–2–7–4 show the variation in arsenic metabolites (MMA, DMA, and iAs) in people exposed to arsenic via drinking water. Polymorphism of the genes regulating the expression of arsenic methyltransferases may be responsible for this variation.

Also, the study on the excretion pattern of ingested arsenicals suggests that blood As comes down to a minimum level within a few hours of ingestion, is mostly excreted via urine within 3–4 days, and the remaining part accumulates in keratin-rich biological derivatives of ectoderm such as hair, scales, and nails of the chronically As-exposed population [32,44,45,51,52]. In the past, most investigators measured total As in human nails or hair due to lack of techniques for efficient separation and sensitive detection of particular species [6,52–56].



FIGURE 7–2 Inter-individual variation in the fraction of MMA in urine of people exposed to arsenic via drinking water [30,31,42,48–50]. Reprinted from [47] with permission from Elsevier.



FIGURE 7–3 Inter-individual variation in the fraction of DMA in urine of people exposed to arsenic via drinking water [30,31,42,48–50]. Reprinted from [47] with permission from Elsevier.



FIGURE 7–4 Inter-individual variation in the fraction of inorganic arsenic in urine of people exposed to arsenic via drinking water [30,31,42,48–50]. Reprinted from [47] with permission from Elsevier.

Not only do keratin-rich tissue arsenicals alter the activity of enzymes, which generates cellular energy in the citric acid cycle, but inactivation of pyruvate dehydrogenase by complexation with As(III) is reported, which prevents the generation of adenosine-5-triphosphate (ATP) in the cells. In the presence of As(III), lipoic acid of the protein inside the enzyme replaces two hydrogen atoms from the thiol groups, attaches with a sulfur molecule forming a dihydrolipoylarsenite chelate complex, and prevents the reoxidation of the dihydrolipoyl group (Scheme 7–1). As a result this pivotal enzyme step is blocked and the amount of pyruvate in the blood increases, energy production is reduced, and finally the cell is slowly damaged [57,58]. The strong affinity of As(III) to sulfur is reported when trivalent arsenic forms a stable ringed structure with vicinal dithilols of keratin in hair [59]. Arsenic inhibits enzymes, such as pyruvate oxidase, *S*-amino acid oxidase, choline oxidase, and transaminase. In addition, As(V) can be disruptive by competing with phosphate uncoupling oxidative phosphorylation [60] where reduced nicotinamide adenine dinucleotide (NADPH) is oxidized and reduces ATP production.

$$3ADP + 3H_3PO_4 \rightarrow 3ATP + 3H_2O_4$$

$$NADPH + H^{+} + \frac{1}{2}O_{2} \rightarrow NADP^{+} + H_{2}O_{2}$$

Subsequently, arsenate produces unstable arsenate ester of ADP, which undergoes hydrolysis non-enzymatically and disrupts the production of ATP. This process is termed arsenolysis [61].

Glucose-6-arsenate is produced rather than glucose-6-phosphate, which inhibits the energy metabolism. The clastogenicity of arsenic appears because an arsenodiester bond is weaker than the normal phosphodiester bond after replacement of phosphorus in DNA by



SCHEME 7-1 Interaction of iAs with proteins and blockage of ATP production.

As(V) [62] and inhibits the DNA repair mechanism. More detailed research is needed to establish this concept of incorporation of As(V) into DNA. Due to this affinity to protein, the onset of skin effects (including pigmentation changes, hyperkerotosis, and skin cancers) was linked to the consumption of arsenic in medicines and drinking water in the latter part of the 19th century [63]. Hence, chronic exposure to iAs may give rise to several health issues including effects on the gastrointestinal tract, respiratory tract, skin, liver, cardiovascular system, hematopoietic system, nervous system, etc. [2].

In the first decade of the 21st century several hyphenated analytical techniques were published in different journals which suddenly changed the concept of the metabolism pattern of arsenicals in humans and mammalian species and hence the concept of arsenic toxicity in the arsenic-affected population [64–71]. These techniques separated different chemical species of As with high resolution at lower detection limits. Also, different arsenicals were separated based on their oxidation states of As, which brought about a breakthrough in As metabolism and its toxicity towards humans. Procedures for As speciation in different matrices are schematically presented in Figure 7–5 [72].

Identification and quantification of toxic methylated trivalent arsenicals were carried out by using HPLC hyphenated with HGAAS/HGAFS/HGAFD/ICP-MS/ICP-AFS, and outcomes of simultaneous toxicity and metabolite studies on animals and cell lines created a big question mark over the traditional detoxification concept of biomethylation [52–56,64–80].



FIGURE 7–5 Procedures for arsenic speciation in different matrixes. UV-vis. = spectrophotometry; EQ = electrochemical methods; AAS = atomic absorption spectrometry; ICP-AES = inductively coupled plasma atomic emission spectrometry; ICP-MS = inductively coupled plasma mass spectrometry; HG = hydride generation; FAAS = flame atomic absorption spectrometry; CT = cryogenic trapping; CC = gas chromatography; HPLC = high performance liquid chromatography; IEC = ion exchange chromatography; ETAAS = electrothermal atomic absorption spectrometry; NAA = neutron activation analysis; DCP = direct current plasma; AFS = atomic fluorescence spectrometry; MIP = microwave induced plasma; and CE = capillary electrophoresis). *Reprinted from* [72] with permission from Elsevier.

A representative chromatogram is presented in Figure 7–6 after development of a hyphenated technique coupling HPLC-ICP-MS [64]. At a single run all eight arsenicals, i.e., As(III), As(V), MMA(III), MMA(V), DMA(III), DMA(V), AsC, and AsB, are separated. Using this technique biological samples such as urine (Figure 7–7), nails, and hair samples (Figure 7–8) collected from the As-affected areas of West Bengal were analyzed and quantified. Contents of different arsenicals quantified using the above technique are summarized in Table 7–3 for urine, Table 7–4 for finger nails, Table 7–5 for hair, and Table 7–6 for blood plasma and for red blood cell (RBC) samples collected from the As-affected population of West Bengal. Both DMA(III) and MMA(III) have been detected for the first time directly in urine of some humans exposed to iAs through their drinking water. Of 428 subjects, MMA(III) and DMA(III) were found in 48% and 72%, respectively. MMA(III) was 2–5% and DMA(III) was 4–21% of the total urinary arsenic. These results clearly show that this technique is capable of quantifying submicro levels of arsenicals in different biological samples. Also, Aposhian et al. [70] detected MMA(III) in human urine collected from the arsenic-affected areas of Romania.



FIGURE 7–6 HPLC/ICP MS chromatogram of a mixture ($20 \mu g$ of As/L each) of eight authentic arsenic compounds using an anion exchange HPLC/ICP MS system. Separation was carried out on an anion exchange ES-502N 7C column (7.6 mm × 100 mm) with a mobile phase containing 15 mM citric acid. The isocratic pH of the mobile phase was adjusted to 2.0 with 10% HNO₃ at 25°C. The flow rate of the mobile phase was 1 mL/min throughout the study. The injection volume of the sample was 20 µL. The ion intensities at m/z 75 and 77 were recorded with timeresolved analysis software. *Reprinted from* [64] with permission from American Chemical Society.

Styblo et al. [81] tried to separate MMA(III) and MMA(V) using thin-layer chromatography, but this is not applicable to biological fluids. Del Razo et al. (2001) used HPLC/hydride generation atomic absorption spectrometry (HPLC-HGAAS) for the separation and quantification of different arsenicals in urine (Table 7–7), metabolites in HepG2 cell lysate after exposure to As(III)-containing arsenicals (Table 7–8), and stability data of As(III)-containing arsenicals in donor urine (Table 7–9) [68]. Thus, analytical techniques based on HG-AAS can also separate and quantify different arsenicals in different matrices. Le et al. used a hydride generation atomic fluorescence detector (HGAFD) coupled with HPLC for the speciation of different arsenicals at submicro levels in different arsenicals is their stability (Table 7–9). However, there is evidence that As(III) undergoes time-dependent oxidation in these matrices [68]. Among the three trivalent species tested [As(III), MMA(III), DMA(III)], As(III) was the most stable species and MMA(III) was more stable than DMA(III). The same authors have tested the stability of these arsenicals [As(III), MMA(III)O, and DMA(III)GS] after spiking in human


FIGURE 7–7 Elution profile of arsenicals obtained from HPLC/ICP MS. The urine samples collected from groups A–D were diluted four-fold. Their chromatograms represent the peaks of both DMA(III) and MMA(III). Experimental parameters were the same as those described in the legend of Figure 7–6. *Reprinted from* [64] *with permission from American Chemical Society.*

urine and found considerable inter-individual variation in the stability of As(III)-containing species [68]. They speculated that this inter-individual variation may arise due to differences in amounts of oxidants or antioxidants in urine coming from dietary sources. More data on the stability of methylated trivalent arsenicals are urgently needed for *in vitro* and *in vivo* toxicological evaluation of these arsenicals.

Several reports show more toxicity of methylated trivalent arsenicals in different human cell types compared to the pentavalent form [69,73,75]. Methylarsine oxide [MMA(III)O] was more toxic and iododimethylarsine [DMA(III)I] was at least as cytotoxic as As(III) for most of the cell types examined (human normal hepatocytes, epidermal keratinocytes, bronchial epithelial, and urinary bladder UROtsa cells) [69]. Also, several studies have been carried out on human cell lines (such as human hepatocellular carcinoma [HepG2] cells, human bladder transient carcinoma [T24] cells, human acute promyelocytic leukemia [NB4] cells, human monoblastoid [U937] cells, human osteosarcoma [HOS] cells, and human neuroblastoma [SK-N-SH] cells) in assessing cytotoxicity of methylated trivalent arsenicals [74]. It was found



FIGURE 7–8 HPLC-ICP MS chromatograms of As in water extract of nail and hair. (A) Fingernails collected from the As-exposed individual and incubated in water at 90°C for 0.5 h. (B) Hair collected from the As-exposed individual and incubated in water at 90°C for 6 h. Experimental parameters were the same as those described in the legend of Figure 7–6. *Reprinted from* [52] *with permission from Elsevier.*

		Urinary As (μg/L [%]), <i>n</i> = 41						
	AsB	iAs ^{III}	iAs ^v	MMA ^{III}	MMA ^v	DMA ^{III}	DMA ^v	tAs ^a
Mean	1.08 [1.0]	17.5 [11.3]	15.2 [10.1]	9.39 [6.6]	17.6 [10.5]	21.9 [13.0]	93.9 [47.5]	177
SD	1.02	13.8	9.45	4.09	11.6	30.1	77.6	97.4
Minimum	0.29	4.53	2.36	5.31	2.52	1.24	9.83	50.1
Maximum	3.45	64.3	46.7	22.2	59.0	150	267	390
Water As (µg/	Ľ)			Urinary As	σ (μg/L [SD])			
29.0 (8) ^b	1.49 (0.9)	9.27 (4.0)	8.58 (3.0)	6.49 (0.9)	8.63 (4.3)	6.31 (3.1)	25.8 (14)	66.5 (12)
55.0 (8) ^c	1.82 (1.7)	11.2 (8.9)	8.94 (3.5)	7.44 (2.5)	10.3 (3.8)	16.8 (7.5)	31.5 (11)	87.9 (13)
130 (12) ^d	1.37 (0.6)	19.9 (15)	17.7 (11)	9.37 (2.9)	14.9 (5.2)	23.1 (9.6)	80.8 (45)	167 (33)
163 (13) ^e	1.18 (0.7)	22.9 (16)	19.3 (9.5)	11.9 (5.2)	28.5 (13)	(32.2) (11)	171 (71)	287 (56)

 Table 7–3
 Arsenic Concentrations in Human Urine and Distribution of Urinary

 Arsenicals according to Arsenic Concentrations in Individuals' Drinking Water Relating to the Study Subjects

 $^{a}tAs = concentration of total As after speciation.$

^bNo. of subjects drinking water containing 29 µg As/L.

 $^c\text{No.}$ of subjects drinking water containing 55 μg As/L.

^dNo. of subjects drinking water containing 130 µg As/L.

 $^e\text{No.}$ of subjects drinking water containing 163 μg As/L.

Reprinted from [43] with permission from Elsevier.

Table 7-4Arsenic Concentrations in Human Fingernails and Distribution of FingernailArsenicals According to Arsenic Concentrations in Individuals' Drinking Water ofthe Study Subjects (All Fingernail Samples were Incubated in Milli-Q Water at 90°Cfor 0.5 h)

		Fingernail As (μ g/g [%]), $n =$ 41						
	iAs ^{III}	iAs ^v	MMA ^v	DMA ^{III}	DMA ^v	t ₁ As ^a	tAs ^b	
Mean	1.81 [62.4]	0.59 [20.2]	0.16 [5.7]	0.26 [8.9]	0.08 [2.8]	2.89	4.57	
SD	0.82	0.29	0.09	0.13	0.03	1.26	2.12	
Minimum	0.76	0.19	0.06	0.09	0.06	1.36	1.95	
Maximum	3.95	1.31	0.42	0.63	0.16	5.67	7.83	
Water As (µg	ı/L)		Finge	ernail As (μg/g	g [SD])			
29.0 (8) ^c	0.95 (0.06)	0.27 (0.03)	0.09 (0.02)	0.11 (0.02)	0.04 (0.02)	1.47 (0.06)	2.24 (0.16)	
55.0 (8) ^d	1.27 (0.08)	0.41 (0.07)	0.11 (0.02)	0.19 (0.03)	0.07 (0.02)	2.05 (0.15)	3.26 (0.29)	
130 (12) ^e	1.71 (0.39)	0.58 (0.15)	0.18 (0.08)	0.27 (0.09)	0.08 (0.02)	2.82 (0.52)	4.52 (0.74)	
163 (13) ^f	2.76 (0.65)	0.92 (0.22)	0.21 (0.13)	0.38 (0.12)	0.09 (0.03)	4.36 (0.93)	7.39 (1.77)	

SD = standard deviation.

 $at_1As =$ concentration of total As after speciation.

 ${}^{b}tAs = concentration of total As after H_2O_2/HNO_3 digestion.$

^cNo. of subjects drinking water containing 29µg As/L.

^dNo. of subjects drinking water containing 55 µg As/L.

 $^e\text{No.}$ of subjects drinking water containing 130 μg As/L.

 $^{\textit{f}}\text{No.}$ of subjects drinking water containing 163 μg As/L.

Reprinted from [43] with permission from Elsevier.

	Hair As (µg/g [%]), <i>n</i> = 41						
	iAs ^{III}	iAs ^v	MMA ^v	DMA ^v	t ₁ As ^a	tAs ^b	
Mean	0.84 [58.9]	0.46 [34.8]	0.04 [2.9]	0.04 [3.4]	1.43	2.45	
SD	0.57	0.30	0.02	0.03	0.83	1.11	
Minimum	0.21	0.08	0.02	0.02	0.05	0.07	
Maximum	2.64	1.42	0.20	0.13	3.80	4.61	
Water As (µg/L	.)		Hair As (μg/g [SD])			
29.0 (8) ^c	0.29 (0.05)	0.26 (0.04)	0.03 (0.02)	0.03 (0.02)	0.59 (0.07)	1.01 (0.24)	
55.0 (8) ^d	0.56 (0.23)	0.35 (0.11)	0.04 (0.01)	0.04 (0.02)	0.95 (0.34)	1.77 (0.35)	
130 (12) ^e	0.73 (0.21)	0.41 (0.11)	0.04 (0.01)	0.04 (0.02)	1.21 (0.23)	2.11 (0.54)	
163 (13) ^f	1.35 (0.63)	0.65 (0.43)	0.05 (0.02)	0.07 (0.02)	2.12 (0.89)	3.53 (0.76)	

Table 7–5 Arsenic Concentrations in Human Hair and Distribution of Hair Arsenicals according to Arsenic Concentrations in Individuals' Drinking Water of the Study Subjects (All Hair Samples were Incubated in Milli-Q Water at 90°C for 6 h)

SD = standard deviation.

 ${}^{a}t_{1}As = concentration of total As after speciation.$

 ${}^{b}tAs = concentration of total As after H_2O_2/HNO_3 digestion.$

 $^cNo.$ of subjects drinking water containing 29 μg As/L.

 $^{\textit{d}}\text{No.}$ of subjects drinking water containing 55 μg As/L.

 $^e\text{No.}$ of subjects drinking water containing 130 μg As/L.

 $^{\textit{f}}\text{No.}$ of subjects drinking water containing 163 μg As/L.

Reprinted from [43] with permission from Elsevier.

Table 7–6	Arsenic Concentrations in Human Blood Plasma and RBCs of the Study
Subjects W	ho Stopped Drinking Arsenic-Contaminated Water 2 years Before Their
Sample Co	llection

	As in Blood Plasma (μ g/L [%]), $n = 25$							
	AsB	iAs ¹¹¹	MMA ^V	DMA ^v	t ₁ As ^a	tAs ^b		
Mean	1.28 [16.7]	1.37 [21.1]	1.95 [27.1]	2.56 [35.1]	7.16	7.48		
SD	1.64	0.51	1.29	1.45	3.59	5.05		
Minimum	0.59	0.81	0.93	0.95	3.56	2.42		
Maximum	8.66	2.59	5.32	5.97	19.5	23.3		

As in Blood (μ g/L [%]), n = 25

AsB	DMA ^v	t ₁ As ^a	tAs ^b	Plasma ^b	RBCs ^b	Blood ^b
5.13 [22.5]	25.1[77.5]	30.2	207	7.48 [28]	18.7 [75]	26.3
7.51	51.8	54.5	165	5.05	10.1	13.8
2.12	4.45	7.12	84.1	2.42	7.41	10.3
37.8	262	276	887	23.3	45.4	65.1
	AsB 5.13 [22.5] 7.51 2.12 37.8	AsB DMA ^V 5.13 [22.5] 25.1[77.5] 7.51 51.8 2.12 4.45 37.8 262	AsB DMA ^V t ₁ As ^a 5.13 [22.5] 25.1[77.5] 30.2 7.51 51.8 54.5 2.12 4.45 7.12 37.8 262 276	AsB DMA ^V t ₁ As ^a tAs ^b 5.13 [22.5] 25.1[77.5] 30.2 207 7.51 51.8 54.5 165 2.12 4.45 7.12 84.1 37.8 262 276 887	AsB DMA ^V t ₁ As ^a tAs ^b Plasma ^b 5.13 [22.5] 25.1[77.5] 30.2 207 7.48 [28] 7.51 51.8 54.5 165 5.05 2.12 4.45 7.12 84.1 2.42 37.8 262 276 887 23.3	AsB DMA ^V t ₁ As ^a tAs ^b Plasma ^b RBCs ^b 5.13 [22.5] 25.1[77.5] 30.2 207 7.48 [28] 18.7 [75] 7.51 51.8 54.5 165 5.05 10.1 2.12 4.45 7.12 84.1 2.42 7.41 37.8 262 276 887 23.3 45.4

SD = standard deviation.

 $at_1As = concentration of total As after speciation.$

 b tAs = concentration of total As after H₂O₂/HNO₃ digestion.

Reprinted from [43] with permission from Elsevier.

Subject	iAs ^{III}	iAs ^v	MAs ^{III}	MAs ^v	DMAs ^{II}	DMAs ^v	Sum	Total As
1	8.1	4.5	ND ^b	7.1	ND	18.6	38.3	41.8
2	11.2	12.2	1.1	11.2	18.3	52.7	106.6	122.2
3	25.2	16.8	2.3	37.7	18.2	40.1	140.3	169.3
4	14.7	10.1	1.2	19.3	5.7	90.5	141.3	147.4
5	59.4	72.1	5.2	119.8	59.8	408.2	724.5	734.6
6	104.2	122.8	12.3	276.7	114.2	467.2	1097.5	1211.1

Table 7–7 Speciated Arsenicals in Urines from Individuals Chronically Exposed to Inorganic Arsenic in Drinking Water^a

^aFirst morning void urine samples were provided by three males and three females (11 to 39 years old) residing in Zimapan. As species were determined by a pH-specific HG-AAS; total As was analyzed by HG-AAS after complete digestion of urine samples. Results are the average concentration (ng As per ml) based on duplicate analyses. The variation of values determined for duplicates did not exceed \pm 6.5% of the mean.

^bND, not detected.

Reprinted from [68] with permission from Elsevier.

Table 7–8	Trivalent and Pentavalent Arsenic Metabolites in HepG2 Cells Exposed
to As(III)-Co	ontaining Arsenicals ^a

Treatment		Sample	Arser	Sum of As				
Arsenical (µM)	ng As/well		iAs ^{III}	MAs ^{III}	MAs ^v	DMAs ^{III}	DMAs ^v	in Culture
iAs ^{III} (0.1)	15	Cells	2.9	ND ^c	ND	ND	2.5	30.5
		Medium	10.6	ND	2.6	11.9	ND	
iAs ^{III} (1.0)	150	Cells	6.6	6.2	ND	ND	3.1	159.0
		Medium	110.6	20.9	ND	9.3	2.3	
iAs ^Ⅲ (10.0)	1500	Cells	67.6	10.4	ND	40	ND	1374.3
		Medium	1079.0	67.8	ND	ND	109.5	
MAs ^{III} (1.0)	150	Cells	2.3	24.9	ND	ND	ND	175.3
		Medium	6.6	129.6	2.8	ND	9.1	
DMAs ^{III} GS (1.0)	150	Cells	2.4	ND	1.4	28.3	ND	165.0
		Medium	9.3	ND	6.1	117.5	ND	
Untreated culture		Cells	2.4	ND	ND	ND	ND	10.0
		Medium	7.6	ND	ND	ND	ND	

^aHepG2 cells (6×10^6 cells and 2 mL medium per well) were incubated in the presence of inorganic arsenite (iAs^{III}) for 24 h or in the presence of methyloxoarsine (MAs^{III}) or of dimethylarsenic – glutathione complex (DMAs^{III}–GS) for 30 min. At the end of exposure, media and cell were collected for As speciation.

^bResults are the average concentration (ng As per ml) based on duplicate analyses. The maximal variation among duplicates (±13.6% of the mean) was found.

^cND, not detected.

Reprinted from [68] with permission from Elsevier.

that toxicity of these arsenicals was independent of cell line type and MMA(III)O was the most potent cytotoxin and equivalent to DMA(IIII), but dimethylated trivalent arsenicals are more cytotoxic than As(III) in most cell types [74]. Other studies investigated interaction of MMA(III) O and DMA(IIII) with DNA in intact human peripheral lymphocytes and found that MAA(III) O and DMA(IIII) damaged DNA at a rate 77 times greater and 386 times greater, respectively,

Donors	Arsenical Added	As ^{III} Oxidized During a Given Time Interval [*]				
		1 Day (%)	1 Week (%)	2 Months (%)		
a	iAs ^{III}	<1	20	32		
	MAs ^{III}	14	33	30		
	DMAs ^{III} GS	27	47	100		
b	iAs ^{III}	8	11	21		
	MAs ^{III}	36	43	100		
	DMAs ^{III} GS	79	86	100		
С	iAs ^{III}	2	20	30		
	MAs ^{III}	26	24	43		
	DMAs ^{III} GS	17	33	42		

 Table 7–9
 Stability of As(III)-Containing Arsenicals in Urine from Three Donors

*Percentages calculated from HG-AAS profiles shown in Figure 7–4. Experimental conditions as described in the legend for Figure 7–4.

Reprinted from [68] with permission from Elsevier.

than did As(III) [78]. Hence, MMA(III) and DMA(III) are considered to be direct-acting forms of arsenic that are genotoxic. Even exposures to very low concentrations $(0.001-0.01\,\mu\text{M})$ of these arsenicals induced cell proliferation and production of growth promoting cytokines in normal human epidermal keratinocytes [80]. Similarly, studies on human UROtsa, T24, or HepG2 cells and primary human hepatocytes suggest higher potency of MMA(III)O and DMA(III) than As(III) in inducing activator protein-1 (AP-1), DNA binding activity, or AP-1-dependent gene transcription (nuclear phospho-c-Jun) [74]. Also, MMA(III) is up to 26 times more toxic than As(III) in Chang human hepatocytes [82]. Therefore, formation of MMA(III) appears to represent toxification of both As(V) and As(III).

Although natural water contains mostly As(III) and As(V), a few investigators reported the presence of small quantities DMA(V) and MMA(V) [71,83–85]. Interestingly, MMA(III) and DMA(III) were also identified in lake water for the first time in 1999 [82]. Both MMA(III) and DMA(III) have been identified in environmental and biological samples such as human urine [64,68,76,77], cultured human hepatocytes [68], and hamster liver tissues [79], whereas only MMA(III) was detected in human urine [64,70] and rat bile [86].

Some recent reports have identified thioarsenicals in biological samples [52,87–93]. Appearance of thioarsenicals in such samples warrants thorough study of arsenical metabolites therein as well as the mechanism of their metabolism in humans. Normally, iAs undergoes reduction followed by oxidative methylation (Figure 7–9) to form DMA(III). In this process, iAs is enzymatically biotransformed to methylated arsenicals including MMA and DMA; these are the end metabolites and the biomarkers of chronic arsenic exposure [64,75,94]. First, glutathione reduces iAs(V) to iAs(III) acting as reducing agent and then the methyl group is transferred to iAs(III) from S-adenosyl methionine to form MMA(V). Next, an intermediate metabolite MMA(III) is produced from MMA(V) during the methylation process followed by a second methylation where MMA(III) is oxidized to DMA(V) [64,75,77]. Glutathione and



FIGURE 7–9 Putative pathway for biotransformation of inorganic arsenic *in vivo* forming both dimethylarsinous acid [DMA(III)] and methylarsonous acid [MMA(III)]. SAM, S-adenosylmethionine; GSH, glutathione; SAHC, S-adenosylhomocysteine; DMA^V, dimethylarsinic acid; MMA^V, monomethylarsonic acid. *Reprinted from* [64] *with permission from American Chemical Society.*



FIGURE 7–10 Proposed reaction pathway for MeAs(V) (S1) and DMeAs(V) (S2) under sulfide-reducing environment. Reprinted from [87] with permission from Elsevier.

S-adenosyl methionine act as co-substrate [95]. The activity of the first methylation step is monitored by calculating the ratio of iAs/MMA and MMA/DMA. The high ratio value of iAs/MMA indicates poor methylation whereas a low ratio value of MMA/DMA indicates good methylation [47,48]. Children are poor methylators and good excretors in comparison to adults. Thus, children are less susceptible to arsenicism [96].

However, in a sulfide-reducing environment iAs is metabolized to both methylated and thiomethylated metabolites (Figure 7–10) [87]. Figure 7–11 shows HPLC-ICP-MS chromatograms of



FIGURE 7–11 HPLC-ICP MS chromatograms of reduced standard MeAs^V and DMeAs^V. In **panel (L):** (L1) a sample (20 μ g As L⁻¹) of MeAsIII-I₂ by an anion exchange ES-502N 7C column with a mobile phase containing 15 mM citric acid at pH 2.0 with 10% HNO₃ at 25°C at a flow rate of 1 mLmin⁻¹; (L2) aqueous DMeAsIII-I sample; (L3) MeAsIII-Cys; (L4) DMeAsIII-Cys; (L5) reduced std. MeAs^V by Reay and Asher; (L6) reduced std. DMeAs^V by Reay and Asher; (L7) reduced std. MeAs^V by Na₂S and H₂SO₄; (L8) reduced std. DMeAs^V by Na₂S and H₂SO₄. In **panel (R):** (R1) an aqueous sample (20 μ g As L⁻¹) of MeAsIII-I₂ analyzed by HPLC-ICP MS using a size exclusion GS-220 HQ with 50 mM ammonium acetate buffer (pH 6.5 at 25°C) at a flow rate of 0.6 mL min⁻¹; (R2) aqueous DMeAs^{III}-I sample; (R3) MeAs^{III}-Cys; (R4) DMeAs^{III}-Cys; (R5) reduced std. MeAs^V by Reay and Asher; (R6) reduced std. DMeAs^V by Na₂S and H₂SO₄. The injection volume of the sample was 20 μ L. The ion intensities at *m/z* 75 and 77 were recorded with time-resolved analysis software. *Reprinted from* [87] *with permission from Elsevier.*

reduced MMA(V) and DMA(V) under a sulfide-reducing environment [87]. Using this hyphenated technique it is possible to identify and quantify different thioarsenicals in biological samples.

With the development of hyphenated analytical techniques and subsequent toxicity studies of iAs, tri-, and pentavalent methylated arsenicals the toxicity scenario has been changed

Standard	Countries
Standard lower than 0.01 mg/L	Australia (0.007 mg/L, 1996)
Standard of 0.01 mg/L	European Union (1998), Japan (1993), Jordan (1991), Laos (1999),
	Laos, Mongolia (1998), Namibia, Syria (1994)
Standard lower than 0.05 mg/L but higher than	Canada (0.025 mg/L, 1999)
0.01 mg/L	
Standard under consideration for lowering from	United States (1986*), Mexico (1994)
0.05 mg/L	
Standard of 0.05 mg/L	Bahrain, Bangladesh (unknown), Bolivia (1997), China (unknown),
	Egypt (1995), India (unknown), Indonesia (1990), Oman, Philippines (1978), Saudi Arabia, Sri Lanka (1983), Vietnam (1989), Zimbabwe

Table 7–10The Currently Accepted National Standards for Arsenic in DrinkingWater [103]

() shows the year standard was established; *new standard value 0.005 mg/L is being proposed and 0.002 mg/L was accepted in 2001.

around the globe. Based on toxicity data from human and mammalian studies, the World Health Organization (WHO) has had a public position on arsenic in drinking water since 1958. Although 0.20 mg/L was considered as an allowable concentration [97], the updated standards in 1963 kept arsenic in the same category and established a lower allowable concentration of 0.05 mg/L without citing any specific reason for this reduction [98]. WHO was updating the progress in research and advancement of arsenicals and an update in 1971 retained arsenic in the toxic substances category reaffirming the allowable value of 0.05 mg/L. Its explanatory notes referred to the fact that figures higher than that quoted are found in a number of Latin American countries and levels up to 0.2 mg/L were not known to have caused difficulties in drinking water [99]. From epidemiological studies in the arsenic-affected areas around the world, especially in Chile and China (province of Taiwan) and data on toxicological studies of different arsenicals, WHO (1984) recommended 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a concentration of 0.05 mg/L as a guideline value since a

After development of various hyphenated instrumental analytical techniques the reproducibility and accuracy of determining different arsenicals in water and biological samples were established. Actual data from toxicological studies of arsenicals based on chemical and oxidation states of arsenic were released, which helped and guided WHO to rethink the provisional value of arsenic in drinking water. The last edition of WHO Guidelines for Drinking-water Quality (GDWQ) (1993) established 0.01 mg/L as a provisional guideline value for arsenic in drinking water, because lower levels preferred for health protection were not reliably measurable at that time. In a number of countries, WHO provisional guidelines of 0.01 mg/L have been adopted as the standard [101]. However, many countries have kept 0.05 mg/L as the national standard or as an interim target before tackling populations exposed to lower but still significant concentrations of arsenic in the range of 0.01–0.05 mg/L. Under the 1996 Safe Drinking Water Act (SDWA) amendments [102], the US Environmental Protection Agency (USEPA) proposed a new standard (an MCL) for arsenic in drinking water in June 2000 of 0.005 mg/L (Table 7–10) [103]. This chronological lowering of GDWQ values between 1958 and 2000 is possible due to the outcome of different hyphenated techniques, which have lowered the limit of detection to submicro levels accurately and reproducibly. Also, several toxicological studies have benefited from these techniques to separate, identify, and quantify different chemical species of arsenic in *in vitro* and *in vivo* studies in humans and mammalian species. Consequently, this has helped the policy makers to redraw and reestablish an updated guideline in drinking water, foods, and air.

7.2 Conclusions

The stability of methylated trivalent arsenicals is still a challenge to chemists. Lack of experimental data of methylated trivalent arsenicals on toxicology warrants reliable and authentic reference materials of these arsenicals. Due to the unavailability of reference standard materials of methylated trivalent arsenicals, the generated toxicological data need validation using authentic standard reference materials. Hence, a low cost hyphenated technique with a lower detection limit is sorely needed to quantify arsenicals in different matrices collected from those arsenicaffected populations who consume As-contaminated drinking water at concentrations of less than 0.05 mg/L. In addition, further in-depth research on thioarsenicals is indispensable for the elucidation of metabolic pathways as well as toxicity in humans and other mammalian species.

References

- [1] Woolson EA. Arsenical pesticides. ACS Ser 1975;7:vii.
- [2] Mandal BK, Suzuki KT. Arsenic round the world: a review. Talanta 2002;58:201-35.
- [3] Komorowicz I, Barałkiewicz D. Arsenic and its speciation in water samples by high performance liquid chromatography inductively coupled plasma mass spectrometry—last decade review. Talanta 2011;84:247-61.
- [4] Gutzeit H. The quantitative determination of arsenic. Pharmaz Zeitung 1891;36:748-756.
- [5] Sanger CB, Black OF. The quantitative estimation of arsenic by the Gutzeit method. J Soc Chem Ind 1907;26:1115–23.
- [6] Chatterjee A, Das D, Mandal BK, Chowdhury TR, Samanta G, Chakraborti D. Arsenic in groundwater in six districts of West Bengal, India: the biggest arsenic calamity in the world. Part I: arsenic species in drinking water and urine of the affected people. Analyst 1995;120:643–50.
- [7] Greschonig H, Kuehneit D, Irgolic KJ. The mercuric-bromide-stain method and the Natelson method for the determination of arsenic: implications for assessment of risks from exposure to arsenic in Taiwan. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic exposure and health effects. London: Chapman and Hall; 1997. p. 17–32.
- [8] Shaikh AU, Tallman DE. Determination of sub-microgram per liter quantities of arsenic in water by arsine generation followed by graphite furnace atomic absorption spectrometry. Anal Chem 1977;49:1093–6.
- [9] Andreae MO. Distribution and speciation of arsenic in natural waters and some marine algae. Deep Sea Res 1978;25:391–402.
- [10] Andreae MO. Arsenic speciation in seawaters: the influence of biological-chemical interactions on the chemistry of a trace element. Limnol Oceanogr 1979;24:440–52.
- [11] Andreae MO. Organoarsenic compounds in the environment. In: Craig PJ, editor. Organometallic compounds in the environment: principles and reactions. New York: Longman; 1986. p. 198–228.

198 HANDBOOK OF ARSENIC TOXICOLOGY

- [12] Cullen WR, Reimer KJ. Arsenic speciation in the environment. Chem Rev 1989;89:713–64.
- [13] World Health Organization (WHO). Guidelines for drinking water quality, 2nd ed. Geneva: WHO; 1993.
- [14] Eaton A, Wang HC, Northington J. Analytical chemistry of arsenic in drinking water. AWWA, Research Foundation and American Water Works Association. Denver, CO: AWWA; 1998.
- [15] ISO 11969. Water quality—Determination of arsenic—Atom absorption spectrometric method (hydride technique); 1996.
- [16] ISO/CD 15586. Water quality—Determination of trace elements by atomic absorption spectrometry with graphite furnace; 2000.
- [17] ISO 11885. Water quality—Determination of 33 elements by inductively coupled plasma atomic emission spectrometry; 1996.
- [18] SM 31251(999). Standard methods for the examination of water and wastewater. Metals by inductively coupled plasma/mass spectrometry.
- [19] CEN/TC/230/WG1/TG 12 N 3. Water quality—Determination of arsenic in drinking, surface, ground, saline and industrial and domestic waste waters. 1st working draft, 29 March; 1999.
- [20] USEPA 7063. Arsenic in aqueous samples and extracts by anodic stripping voltammetry (AVS), method 7063. US Environmental Protection Agency; 1996.
- [21] SM 3500-As B. Standard methods for the examination of water and wastewater, silver diethyldithiocarbamate method; 1999.
- [22] National Research Council (NRC) Arsenic in drinking water. Washington, DC: National Academy Press; 1999. 330 pp.
- [23] National Research Council (NRC) Arsenic in drinking water: 2001 update. Washington, DC: National Academy Press; 2001. 244 pp.
- [24] World Health Organization (WHO) Arsenic and arsenic compounds Environmental health criteria no. 224, 2nd ed. Geneva: World Health Organization; 2001. 521 pp.
- [25] Tam GKH, Charbonneau SM, Bryce F, Pomroy C, Sandi E. Metabolism of inorganic arsenic (74As) in humans following oral ingestion. Toxicol Appl Pharmacol 1979;50:319–22.
- [26] Vahter M, Marafante E. Intracellular interaction and metabolic fate of arsenite and arsenate in mice and rabbits. Chem Biol Interact 1983;47:29–44.
- [27] Marafante E, Vahter M. The effect of methyltransferase inhibition on the metabolism of (74As) arsenite in mice and rabbits. Chem Biol Interact 1984;50:49–57.
- [28] Marafante E, Vahter M, Envall J. The role of the methylation in the detoxication of arsenate in the rabbit. Chem Biol Interact 1985;56:225–38.
- [29] Vahter M, Marafante E. Reduction and binding of arsenate in marmoset monkeys. Arch Toxicol 1985;57:119-24.
- [30] Vahter M, Couch R, Nermell B, Nilsson R. Lack of methylation of inorganic arsenic in the chimpanzee. Toxicol Appl Pharmacol 1995;33:262–8.
- [31] Concha G, Vogler G, Nermell B, Vahter M. Metabolism of inorganic arsenic in children with chronic high arsenic exposure in northern Argentina. Environ Health Perspect 1998;106:355–9.
- [32] Buchet JP, Lauwerys R, Roels H. Comparison of the urinary excretion of arsenic metabolites after a single oral dose of sodium arsenite, monomethylarsenate, or dimethylarsenate in man. Int Arch Occup Environ Health 1981;48:71–9.
- [33] Yamauchi H, Takahashi K, Mashiko M, Yamamura Y. Biological monitoring of arsenic-exposed gallium arsenide- and inorganic arsenic-exposed workers by determination of inorganic arsenic and its metabolites in urine and hair. Am Ind Hyg Assoc J 1989;50:606–12.
- [34] US Environmental Protection Agency. Special report on ingested inorganic arsenic skin cancer; nutritional essentiality, Technical Report EPA/625/8-87/013, Washington, DC; 1988.

- [35] Vahter M. Species differences in the metabolism of arsenic compounds. Appl Organomet Chem 1994;8:175-82.
- [36] Aposhian HV. Enzymatic methylation of arsenic species and other new approaches to arsenic toxicity. Annu Rev Pharmacol Toxicol 1997;37:397–419.
- [37] Foa V, Colombi A, Maroni M, Buratti M, Calzaferri G. The speciation of the chemical forms of arsenic in the biological monitoring of exposure to inorganic arsenic. Sci Total Environ 1984;34:241–59.
- [38] Bogdan GM, Sampayo-Reyes A, Aposhian HV. Arsenic binding proteins of mammalian systems: I. Isolation of three arsenite-binding proteins of rabbit liver. Toxicology 1994;93:175–93.
- [39] Zakharyan RA, Wildfang E, Aposhian HV. Enzymatic methylation of arsenic compounds: III. The marmoset and tamarin, but not the rhesus, monkey are deficient in methyltransferases that methylate inorganic arsenic. Toxicol Appl Pharmacol 1996;140:77–84.
- [40] Healy SM, Zakharyan RA, Aposhian HV. Enzymatic methylation of arsenic compounds: IV. In vitro and in vivo deficiency of the methylation of arsenite and monomethylarsonic acid in the guinea pig. Mutat Res 1997;386:229–39.
- [41] Vahter M, Marafante E, Lindgren A, Dencker L. Tissue distribution and subcellular binding of arsenic in marmoset monkeys after injection of 74As-arsenite. Arch Toxicol 1982;51:65–77.
- [42] Chiou H-Y, Hsueh YM, Hsieh L-L, Hsu L-I, Hsu Y-H, Hsieh F-I, et al. Arsenic methylation capacity, body retention, and null genotypes of glutathione S-transferase M1 and T1 among current arsenic-exposed residents in Taiwan. Mutat Res 1997;386:197–207.
- [43] Mandal BK, Ogra Y, Anzai K, Suzuki KT. Speciation of arsenic in biological samples. Toxicol Appl Pharmacol 2004;198:307–18.
- [44] Buchet JP, Lauwerys R. Role of thiols in the in-vitro methylation of inorganic arsenic by rat liver cytosol. Biochem Pharmacol 1988;37:3149–53.
- [45] Bencko V, Dobisova A, Macaj M. Arsenic in the hair of a non-occupationally exposed population. Atmos Environ 1971;5:275–9.
- [46] Vahter M. Genetic polymorphism in the biotransformation of inorganic arsenic and its role in toxicity. Toxicol Lett 2000;112-113:209-17.
- [47] Vahter M. Methylation of inorganic arsenic in different mammalian species and population groups. Sci Prog 1999;82:69–88.
- [48] Del Rezo LM, Garcia-Vargas GG, Vargas H, Albores A, Gonsebatt M, Montero R, et al. Altered profile of urinary arsenic metabolites in adults with chronic arsenicism: a pilot study. Arch Toxicol 1997;71:211–17.
- [49] Hopenhayn-Rich C, Smith AH, Goeden HM. Human studies do not support the methylation threshold hypothesis for the toxicity of inorganic arsenic. Environ Res 1993;60:161–77.
- [50] Hopenhayn-Rich C, Biggs ML, Smith AH, Kalman DA, Moore LE. Methylation study of a population environmentally exposed to arsenic in drinking water. Environ Health Perspect 1996;104:620–8.
- [51] Mandal BK, Roy Chowdhury T, Samanta G, Basu GK, Chowdhury PP, Chanda CR, et al. Arsenic in ground water in seven districts of West Bengal, India: the biggest calamity in the world. Curr Sci 1996;70:976–86.
- [52] Mandal BK, Ogra Y, Suzuki KT. Speciation of arsenic in human nail and hair from arsenic-affected area by HPLC-inductively coupled argon plasma mass spectrometry. Toxicol Appl Pharmacol 2003;189:73–83.
- [53] Karagas MR, Morris JS, Weiss JE, Spate V, Baskett C, Greenberg ER. Toenail samples as an indicator of drinking water arsenic exposure. Cancer Epidemiol Biomarkers Prev 1996;5:849–52.
- [54] Karagas MR, Tosteson TD, Blum J, Klaue B, Weiss JE, Stannard V, et al. Measurement of low levels of arsenic exposure: a comparison of water and toenail concentrations. Am J Epidemiol 2000;152:84–90.
- [55] Morton J, Carolan VA, Gardiner PHE. Removal of exogenously bound elements from human hair by various washing procedures and determination by inductively coupled plasma mass spectrometry. Anal Chim Acta 2002;21759:1–12.
- [56] Sera K, Futasugawa S, Murao S. Quantitative analysis of untreated-hair for monitoring human exposure to heavy metals. Nucl Instr Methods Phys Res B 2002;189:174–9.

- [57] Belton JC, Benson NC, Hanna ML, Taylor RT. Growth inhibitory and cytotoxic effects of three arsenic compounds on cultured Chinese hamster ovary cells. J Environ Sci Health 1985;20A:37–72.
- [58] Wolochow H, Putman EM, Domdoroff M, Hassid WZ, Barker HA. Preparation of sucrose labeled with C14 in the glucose or fructose component. J Biol Chem 1949;180:1237–42.
- [59] Peters RA. The study of enzymes in relation to selective toxicity in animal tissues. Symp Soc Exp Biol 1949;3: 36–59.
- [60] Dixon HBF. The biochemical action of arsonic acids especially as phosphate analogues. Adv Inorg Chem 1997;44:191–227.
- [61] Aposhian HV. Biochemical toxicology of arsenic. Rev Biochem Toxicol 1989;10:265–99.
- [62] Elder, J, and Davies, TT. Draft drinking water criteria. Document on Arsenic, US EPA, Under EPA Contract No. 68-C8-0033, Washington DC, 20460, VIII-4; 1993. p. 1.
- [63] WHO. Arsenic compounds. Environmental health criteria 224, 2nd ed. Geneva: World Health Organization; 2001.
- [64] Mandal BK, Ogra Y, Suzuki KT. Identification of dimethylarsinous and monomethylarsonous acids in human urine of the arsenic affected areas in West Bengal, India. Chem Res Toxicol 2001;14:371–8.
- [65] Suzuki KT, Mandal BK, Katagiri A, Sakuma. Y, Kawakami A, Ogra Y, et al. Dimethylthioarsenicals as arsenic metabolites and their chemical preparations. Chem Res Toxicol 2004;17:914–21.
- [66] Gong Z, Jiang G, Cullen WR, Aposhian HV, Le XC. Determination of arsenic metabolic complex excreted in human urine after administration of sodium 2,3-dimercapto-1-propane sulfonate. Chem Res Toxicol 2002;15:1318–23.
- [67] Gong Z, Lu X, Cullen WR, Aposhian HV, Le XC. Unstable trivalent arsenic metabolites, monomethylarsonous acid and dimethylarsinous acid. J Anal At Spectrom 2001;16:1409–13.
- [68] Del Razo LM, Styblo M, Cullen WR, Thomas DJ. Determination of trivalent methylated arsenicals in biological matrices. Toxicol Appl Pharmacol 2001;174:282–93.
- [69] Styblo M, Del Razo LM, Vega L, Germolec DR, LeCluyse EL, Hamilton GA, et al. Comparative toxicity of trivalent and pentavalent inorganic and methylated arsenicals in human cells. Arch Toxicol 2000;74:289–99.
- [70] Aposhian HV, Gurzau ES, Le XR, Gurzau A, Healy SM, Lu X, et al. Occurrence of monomethylarsonous acid in urine of humans exposed to inorganic arsenic. Chem Res Toxicol 2000;13:693–7.
- [71] Hasegawa H, Sohrin Y, Matsui M, Hojo M, Kawashima M. Speciation of arsenic in natural waters by solvent extraction and hydride generation atomic absorption spectrometry. Anal Chem 1994;66:3247–52.
- [72] Burguera M, Burguera JL. Analytical methodology for speciation of arsenic in environmental and biological samples. Talanta 1997;44:1581–604.
- [73] Petrick JS, Ayala-Fierro F, Cullen WR, Carter DE, Aposhian HV. Monomethylarsonous acid (MMAIII) is more toxic than arsenite in Chang human hepatocytes. Toxicol Appl Pharmacol 2000;163:203–7.
- [74] Styblo M, Drobna Z, Jaspers I, Lin S, Thomas DJ. The role of biomethylation in toxicity and carcinogenicity of arsenic: a research update. Environ Health Perspect 2002;110(Suppl. 5):767–71.
- [75] Thomas DJ, Styblo M, Lin S. The cellular metabolism and systemic toxicity of arsenic. Toxicol Appl Pharmacol 2001;176:127–44.
- [76] Le XC, Lu X, Ma M, Cullen WR, Aposhian HV, Zheng B. Speciation of key arsenic metabolic intermediates in human urine. Anal Chem 2000;72:5172–7.
- [77] Le CX, Ma M, Lu X, Cullen WR, Aposhian HV, Zheng B. Determination of monomethylarsonous acid, a key arsenic methylation intermediate, in human urine. Environ Health Perspect 2000;108:1015–18.
- [78] Mass MJ, Tennant A, Roop BC, Cullen WR, Styblo M, Thomas DJ, et al. Methylated trivalent arsenic species are genotoxic. Chem Res Toxicol 2001;14:355–61.
- [79] Sampayo-Reyes A, Zakharyan RA, Healy SM, Aposhian HV. Monomethylarsonic acid reductase and monomethylarsonous acid in hamster tissue. Chem Res Toxicol 2000;13:1181–6.
- [80] Vega L, Styblo M, Patterson R, Cullen WR, Wang C, Germolec D, et al. Differential effects of trivalent and pentavalent arsenicals on cell proliferation and cytokine secretion in normal human epidermal keratinocytes. Toxicol Appl Pharmacol 2001;172:225–32.

- [81] Styblo M, Yamauchi H, Thomas DJ. Comparative in vitro methylation of trivalent and pentavalent arsenicals. Toxicol Appl Pharmacol 1995;135:172–8.
- [82] Lin S, Cullen WR, Thomas DJ. Methylarsenicals and arsinothiols are potent inhibitors of mouse liver thioredoxin reductase. Chem Res Toxicol 1999;12:924–30.
- [83] Lin TH, Huang YL, Wang MY. Arsenic species in drinking water, hair, fingernails, and urine of patients with blackfoot disease. J Toxicol Environ Health A 1998;53:85–93.
- [84] Shraim A, Chandra Sekaran N, Anuradha CD, Hirano S. Speciation of arsenic in tube-well water samples collected from West Bengal, India, by high-performance liquid chromatography-inductively coupled plasma mass spectrometry. Appl Organomet Chem 2002;16:202–9.
- [85] Mandal BK, Suzuki KT, Anzai K. Impact of food-arsenic on the people living in the arsenic-affected areas of West Bengal, India. J Environ Sci Health Part A 2007;42:1741–52.
- [86] Suzuki KT, Tomita T, Ogra Y, Ohimichi M. Glutathione-conjugated arsenics in the potential hepatoenteric circulation in rats. Chem Res Toxicol 2001;14:1604–11.
- [87] Mandal BK, Suzuki KT, Anzai K, Yamaguchi K, Sei Y. A SEC-HPLC-ICP MS hyphenated technique for identification of sulfur-containing arsenic metabolites in biological samples. J Chrom B 2008;874:64–76.
- [88] Raml R, Goessler W, Traar P, Ochi T, Francesconi KA. Novel thioarsenic metabolites in human urine after ingestion of an arsenosugar, 2',3'-dihydroxypropyl 5-deoxy-5-dimethylarsinoyl-α-D-riboside. Chem Res Toxicol 2005;18:1444–50.
- [89] Suzuki KT, Iwata K, Naranmadura H, Suzuki N. Metabolic differences between two dimethylthioarsenicals in rats. Toxicol Appl Pharmacol 2007;218:166–173.
- [90] Naranmandura H, Suzuki K. Formation of dimethylthioarsenicals in red blood cells. Toxicol Appl Pharmacol 2008;227:390–9.
- [91] Suzuki S, Arnold LL, Pennington KL, Chen B, Naranmandura H, Le XC, et al. Dietary administration of sodium arsenite to rats: relations between dose and urinary concentrations of methylated and thiometabolites and effects on the rat urinary bladder epithelium. Toxicol Appl Pharmacol 2010;244:99–105.
- [92] Rehman K, Naranmandura H. Arsenic metabolism and thioarsenicals. Metallomics 2012;4:881–92.
- [93] Contreras-Acuña M, García-Barrera T, García-Sevillano MA, Gómez-Ariza JL. Arsenic metabolites in human serum and urine after seafood (*Anemonia sulcata*) consumption and bioaccessibility assessment using liquid chromatography coupled to inorganic and organic mass spectrometry. Microchem J 2014;112:56-64.
- [94] Biggs ML, Kalman DA, Moore LE, Hopenhayn Rich C, Smith MT, Smith AH. Relationship of urinary arsenic to intake estimates and a biomarker of effect, bladder cell micronuclei. Mutat Res 1995;386:185–95.
- [95] Styblo M, Thomas DJ. In vitro inhibition of glutathione reductase by arsenotriglutathione. Biochem Pharmacol 1995;49:971–4.
- [96] Concha G, Nermelli B, Vahter M. Metabolism of inorganic arsenic in children with high arsenic exposure in northern Argentina. Environ Health Perspect 1998;107:9–15.
- [97] WHO. International standards for drinking-water. Geneva: World Health Organization; 1958.
- [98] WHO. International standards for drinking-water, 2nd ed. Geneva: World Health Organization; 1963.
- [99] WHO. International standards for drinking-water, 3rd ed. Geneva: World Health Organization; 1971.
- [100] WHO. Guidelines for drinking-water quality, vol. 1. Geneva: World Health Organization; 1984.
- [101] WHO. Guidelines for drinking-water quality, 2nd ed., vol. 1. Geneva: World Health Organization; 1993.
- [102] WHO. Guidelines for drinking-water quality, 2nd ed., vol. 2. Geneva: World Health Organization; 1996.
- [103] USEPA, Proposed revision to arsenic drinking water standard, Technical fact sheet: proposed rule for arsenic in drinking water and clarifications to compliance and new source contaminants monitoring, http://www.epa.gov/safewater/ars/prop_techfs.html; May 2000.

This page intentionally left blank

Mechanism for Arsenic-Induced Toxic Effects

Jyotirmoy Ghosh¹ and Parames C. Sil²

¹NATURAL PRODUCT CHEMISTRY DIVISION, CSIR-NORTH EAST INSTITUTE OF SCIENCE AND TECHNOLOGY, ASSAM, INDIA, ²DIVISION OF MOLECULAR MEDICINE, BOSE INSTITUTE, KOLKATA, INDIA

CHAPTER OUTLINE

8.1	Introduction	203
8.2	Biological Consequences of Chronic Arsenic Exposure in Humans	204
8.3	Arsenic Metabolism	205
8.4	Pathophysiology of Arsenic Toxicity	206
8.5	Mechanism for the Toxic Effects of Arsenic	211
8.6	Preventing Arsenic-Induced Toxic Effects by Antioxidants	214
8.7	Conclusions and Future Directions	220
Refe	erences	221

8.1 Introduction

Arsenic (As), one of the most widely studied, notoriously poisonous metalloids, is ubiquitous in the environment. It is found in natural and anthropogenic sources, in some abundance in Earth's crust, and in small quantities in rock, soil, water, and air. Arsenic in the environment occurs in both organic and inorganic forms. The major inorganic forms of arsenic include trivalent meta-arsenite (As^{3+}) and pentavalent arsenate (As^{5+}). Among the two states the most toxicologically potent arsenic compounds are in the trivalent oxidation state. This is due to their interaction of trivalent arsenicals with sulfur in proteins and the ability of arsenic to generate oxidative stress. However, humans are exposed to both trivalent and pentavalent arsenicals [1]. Humans are chronically exposed to low levels of arsenic via inhalation in the ambient air and ingestion of food; but the unavoidable source of this poisoning is drinking of arsenic-contaminated groundwater especially in India, Bangladesh, China, and some Central and South American countries [2]. Absorbed arsenic, irrespective of its form, is widely distributed throughout the entire body. Flora [2] recently reported that arsenic concentrations in drinking water in Argentina (200 ppb) [3,4], Mexico (400 ppb) [5,6], Taiwan (50–1980 ppb) [7], and the Indo-Bangladesh region (800 ppb) are well above the WHO guidelines' maximum permissible value (10 ppb) [8]. A considerable percentage of the total populations of the world are suffering from chronic arsenicosis due to drinking arsenic-contaminated water [2,9,10]. Understanding the chemistry behind arsenic toxicity and its mode of action depends on several factors and is quite complicated. The major determinants for arsenic to cause toxicity are valence state (trivalent/pentavalent), charge at physiological pH, extent of methylation and electrostatic attraction and repulsion to active sites on important macromolecules, and several pharmacokinetic factors like absorption, distribution, metabolism, protein binding, and excretion [2]. With recent advances in technology and the development of animal experimental models for arsenic toxicity, the understanding of the toxicology of arsenic will continue to improve.

This chapter briefly summarizes the existing literature on toxicology and environmental aspects of arsenicals and describes the molecular mechanisms of arsenic-induced organ pathophysiology and its preventive strategies using antioxidants.

8.2 Biological Consequences of Chronic Arsenic Exposure in Humans

Previous studies suggest that arsenic exposure for a long period of time (chronic exposure) could induce various degrees of toxicities, which might ultimately cause serious organ dysfunctions. Prenatal exposure to arsenic usually results in short- and long-term organ toxicities; both the arsenic and its methylated metabolites can cross the placenta. This exposure can result in impaired fetal growth, fetal loss during pregnancy, or even increased post-birth infant mortality [11]. Besides the development of various organ dysfunctions and illness occurring as a result of arsenic exposure in early life, evidence also shows serious late effects like the development of certain malignancies and other syndromes and diseases [12]. Arsenic is known to be a potent carcinogen; various types of malignancies and solid tumors, including lung, prostate, bladder, cancers, etc. have been reported to develop due to exposure to it [13-17]. Some other reports, however, suggest that malignancies do not develop during low level exposure in some parts of the world (Denmark); it may, instead, reduce the incidence of non-melanoma skin cancer [18]. Combining all this evidence suggests that arsenic-associated carcinogenesis largely depends on the levels of exposure, genetic factors, and geographical locations. Here, it should also be mentioned that malignancies are not the only serious problem resulting from long-term arsenic exposure; other chronic illnesses like hypertension, cardiovascular disease, diabetes, etc. have also been found to be associated with arsenic exposure [11,19]. Further, long-term memory loss and modification of hormonal regulation have also been reported because of chronic exposure to arsenic [11]. Thus, a variety of different chronic illnesses and malignancies can be induced by arsenic exposure, underscoring the diversity of its cellular targets and its ability to deregulate important and diverse cell functions (Figure 8-1).



FIGURE 8-1 Biochemical outcomes resulting from arsenic toxicity.

8.3 Arsenic Metabolism

The nutritional requirement for arsenic is not known; however like other toxic metals, the metabolism of arsenic is believed to be associated with the conversion of its most potentially toxic form to the less toxic one, which could either be accumulated in or excreted from the cell, although it does not appear to be biomagnified through the food chain as is found with many toxic metals for example mercury and cadmium [20].

8.3.1 Biomethylation

Since inorganic arsenic-containing molecules are highly reactive and potentially more toxic to humans and other animals, the primary detoxification mechanism of arsenic is considered to be its biomethylation. This idea is also supported by the fact that transformation of inorganic arsenic compounds to the respective methyl derivatives has been observed in the diverse species of yeast, fungi, algae, plants, and animals [21,22]. The organoarsenic species (mostly of methyl derivatives), on the other hand, are widespread in aquatic organisms and it is highly unlikely that consumption of these compounds could constitute any hazard, although there are some exceptions in which certain seafood ingestion has been shown to elevate the urinary arsenic content beyond the expected maximum contaminant level [20,23–25].

Inorganic arsenic is methylated to monomethylarsonic acid (MMA) and finally to dimethylarsinic acid (DMA) by the methyl donor S-adenosylmethionine (SAM) in higher organisms [26]. The transformations are catalyzed by methyltransferases in the presence of glutathione (GSH) and it is found that the trivalent arsenicals are preferred substrates for methylation reactions. Reduction of pentavalent arsenic to trivalent arsenic might, therefore, be an important step in controlling the rate of arsenic metabolism [27]. It is known that after exposure, methylation of inorganic arsenic reaches a certain level and then begins to decline and results in an increased toxic effect in the species concerned. Evidence shows that DMA excretion reaches its highest level after a few days and at that time the excretion of the inorganic form is substantially increased (while that of MMA is still elevated), suggesting the existence of two successive methylating enzyme activities in this process. In general, arsenic is converted to the less toxic methylated form in the liver and is excreted via urine at 28 h, 59 h, and 9 h (half-lives ranging between 27 and 86 h for different species) following the sequence: $As^{5+} < MMA < As^{3+} < DMA$ [20].

8.4 Pathophysiology of Arsenic Toxicity

8.4.1 Cardiovascular Dysfunction

Cardiac dysfunction such as atherosclerosis, hypertension, ischemic heart diseases, ventricular arrhythmias, peripheral arterial disease, impaired microcirculation, and coronary heart disease are associated with long-term arsenic exposure [28-31]. Arsenic induces cytotoxic effects in cardiomyocytes by reactive oxygen species (ROS) (such as superoxides and hydrogen peroxide) produced by the stimulation of nicotinamide adenine dinucleotide phosphate (NADPH) oxidase present in the plasma membrane of vascular endothelial cells and vascular smooth muscle cells (VSMCs) [28,32,33]. ROS thus produced cause loss of cardiac actin, reduces size, and damage the nuclei, which coordinates well with disruption of the vascular extracellular matrix and leads to apoptosis via involvement of mitochondria-dependent caspase-3 signaling. It is also reported that arsenic-induced cardiomyocyte death is concentration dependent, with an EC₅₀ of 1 mM [2,34,35]. NAD(P)H oxidase-generating ROS couple with nitric oxide (NO) and forms strong oxidant peroxynitrite, which plays a role in the upregulation of inflammatory mediators such as cyclooxygenase-2 [36]. ROS so produced increase the expression of atherosclerosis-related genes such as heme oxygenase-1 (HO-1), monocyte chemo-attractant protein (MCP-1), and interleukin-6 (IL-6) and thus promote the attachment, penetration, and migration of monocytes in VSMC [37]. This also alters VSMCs' focal adhesion proteins, which leads to their proliferation and migration [38]. Singh et al. [28] recently reported that arsenite exposure decreases the activity of endothelial nitric oxide synthase (eNOS) and Akt/protein kinase B, which subsequently decreases nitric oxide (NO) bioavailability, which in turn leads to vascular endothelial dysfunction and associated cardiovascular complications [39,40].

Arsenic exposure causes vasoconstriction of the blood vessels by phosphorylating myosin light chain kinase (MLCK) and increases calcium sensitization, which leads to hypertension [41]. At the same time, it also alters the release of vasoactive mediators in blood vessels leading to elevation of blood pressure [28,42]. Lee et al. [37] showed that sodium arsenite could induce the expression of MCP-1 and IL-6 in vascular smooth muscle cells. Similar results of increased expression of MCP-1 and IL-6 in vascular lesions have also been shown in an *in vivo* model (As-exposed mice). These findings are consistent with the results of some earlier studies in

which Wu et al. [43] showed the increased expression of circulating lymphocyte MCP-1 mRNA and plasma MCP-1 in humans exposed to arsenic. More recent studies by Hsieh et al. [44] suggest that humans exposed to drinking water containing > $10 \mu g/L$ of arsenic and carrying two risk genotypes of apolipoprotein E (ApoE) and MCP1 have a > 10-fold risk of carotid atherosclerosis [45]. Moreover, in vascular endothelial cells, arsenic has been shown to increase the synthesis of inflammatory mediators (like leukotriene E (LTE), prostacyclin, tumor necrosis factor-alpha, and nuclear factor kappa B), which induce the pathogenic process of atherosclerosis [28,46,47]. Combining all these observations suggests that As-induced inflammation could act as an important risk factor for atherosclerosis.

Arsenic exposure could also cause enhanced hypertension [2], although this observation is the result of only a few dose-response studies between arsenic ingestion and hypertension [29,48]. Additionally, some previous studies reported increased hypertension prevalence among patients with arsenic-affected skin lesions [49,50]. Signal transduction mechanism studies suggest a close resemblance between arsenic-induced hypertension and that for atherosclerosis, in which vascular redox signaling is suggested to play the major role. Reduced NO bioavailability could also be a major factor in arsenic-induced hypertension [2]. Arsenic trioxide, in addition, could add to the development of ventricular arrhythmia by inducing prolonged Q-T interval and action potential duration [28]. In summary, it may be said that arsenic has the potential to induce cardiovascular complications via oxidative stress accompanied by reduced eNOS activation and enhanced MLCK phosphorylation.

8.4.2 Neurological Disorders

Acute As intoxication is associated with neurological dysfunctions and ranges from neurobehavioral disturbances to memory and cognitive impairments, visual or auditory sensory defects, Parkinson's disease, Guillain–Barré-like neuropathy, verbal miscomprehension, encephalopathy, and peripheral neuropathy [2,28,51–56]. Although the cellular and molecular mechanisms underlying arsenic-induced neurological defects are not clear, involvement of oxidative stress with increased ROS, lipid peroxides along with decreased superoxide dismutase activity, and reduced glutathione levels are considered to be the major reasons behind this [28,57]. After chronic arsenic exposure, numbness of distal extremities (especially the legs), decreased sensibility, ataxia, pain and paresis with nausea, vomiting, etc. are the most common reported symptoms. All these are indicative of systemic peripheral neuropathies [58]. The mechanism for this pathophysiology is, however, not clearly known yet and remains to be elucidated [59–62].

The longer axons of sensory neurons are found to be more affected than motor neurons in arsenic-induced neuropathy. Arsenic exposure $(50 \,\mu g \, arsenic/m^3)$ of smelter workers demonstrated that either demyelination or cytoskeletal defects in neuronal axons are the operative pathologies for the peripheral neuropathy that usually hampers the conduction velocity in peripheral nerves [2]. In addition, arsenic exposure has been shown to alter metabolism of various neurotransmitters (such as monoamines, acetylcholine, gamma amino butyric acid, and glutamate) [63].

Chronic arsenic exposure has been recently reported to cause a significant reduction in monoamines (adrenaline, noradrenaline, dopamine, etc.) in the corpus striatum, frontal cortex, and hippocampal areas of the brain [28]. Signal transduction studies showed that arsenitemediated neurotoxicity involves the activation of p38 mitogen-activated protein kinase (p38MAPK) and JNK3 pathways and could ultimately induce apoptosis in the cerebral neurons [64]. Moreover, destabilization and disruption of cytoskeletal framework is another adverse effect of arsenic exposure that induces neurotoxicity and eventually leads to axonal degeneration [65]. Thiamine (vitamin B1) deficiency is well known to induce neuronal complications and arsenic could cause this deficiency, which in turn results the inhibition of pyruvate decarboxylase, elevation of blood pyruvate, and ultimately causing encephalopathy [54]. Arsenicinduced oxidative stress causes oxidative DNA damage and subsequent cell death in the brain and leads to the degeneration of dopaminergic neurons resulting in Parkinson-like symptoms [52,53]. Acute exposure of arsenic decreases acetyl cholinesterase activity and causes a cholinergic crisis-like situation with altered mental status and weakness. This can be associated with peripheral neuropathy, neuropsychiatric abnormalities, and extrapyramidal disorders [66]. Moreover, the peripheral nervous system is also affected by disrupting the neuroskeletal integrity because of arsenic toxicity; the nerve conduction velocity in the peripheral nerves is markedly diminished and thus causes peripheral neuropathy [28].

Arsenic and its metabolites (monomethylarsonic acid and monomethylarsonous acid) could suppress the NMDA receptors in the hippocampus, leading to neurobehavioral disorders and cognitive dysfunction as these receptors play a pivotal role in synaptic plasticity, learning, and memory [67,68]. The chronic exposure of arsenic causes morphological changes in axons and nerve fibers of the striatum and thus disturbs central structural organization [28,69]. Arsenic-induced brain encephalopathy involves the disruption of the central nervous system (CNS) [70]. Duration of arsenic exposure is an important factor determining the severity and reversibility of arsenic-induced CNS manifestations and in some cases these may disappear with the cessation of exposure; however, the effects may remain irreversible in some cases [2,71,72].

Calderon et al. [73] reported the adverse effects of arsenic toxicity (such as visual and auditory sensory impairment; memory and cognitive defects) in clinical studies. Many *in vivo* and *in vitro* studies suggest that oxidative stress, induction of thiamine deficiency, and inhibition of pyruvate decarboxylase and acetyl cholinesterase as well as reduction in biogenic monoamines lead to arsenic-induced neurological defects. Also, there are some inconsistent reports on neurotoxicity using animal models and these discrepancies are thought to be due to the varying doses, duration, and route of arsenic exposure used [2,28].

Taken together, these findings act as a summary of the pathophysiological mechanisms involved in arsenic-induced neurological disorders.

8.4.3 Hepatic and Renal Toxicity

Liver is the prime site for arsenic metabolism and considered to be the main target for arsenic to cause toxicity. The highest level of arsenic is detected in the liver [74]. Arsenic-induced

hepatotoxicity is manifested by increased levels of alanine aminotransferase and aspartate aminotransferase. Another primary target organ of arsenic is the kidneys. Arsenite exposure increases ROS production, which consequently enhances the level of lipid peroxidation, other oxidative stress-related biomarkers, and cellular damage in both hepatic and renal tissue [28,75]. Several groups of researchers showed that arsenic-induced hepatotoxicity follows multiple mechanisms, including ROS-mediated oxidative stress, inflammatory response, or metabolic hindrance. This arsenic toxicity depends on the dose; as in acute high-dose arsenic exposure, a number of oxidative stress-related genes (heme oxygenase 1 and metallothionein) are up-regulated [76], although this is not observed after low-dose, chronic arsenic exposure [77]. Moreover, subacute and chronic exposure of arsenic has been reported to enhance ROS generation, and decrease the GSH/GSSG ratio and antioxidant enzyme activities in association with increased lipid peroxidation in hepatic tissues in vivo [2,78-81]. Continuous arsenic exposure for long periods of time causes increased hepatic collagen accumulation dose dependently that can be correlated with elevated IL-6 concentrations [82]. An important enzyme of glucose metabolism, pyruvate dehydrogenase (PDH), is susceptible to arsenic-induced ROS generation [83] and its inhibition could cause carbohydrate metabolism disorder resulting in diabetes [84]. An increased level of the oxidative DNA damage biomarker, 8-OHdG, has been found to be associated with arsenic-induced hepatocarcinogenesis [2]. Returning to renal toxicity, arsenic has been reported to cause acute renal failure and chronic renal dysfunction; elevated creatinine levels in the serum have been reported in acute and subchronic arsenic exposure in some cases [85]. The mechanisms of these adverse effects are not very clear. However, because of the direct involvement of kidney in arsenic excretion, hypotensive shock, hemoglobinuric or myoglobinuric tubular injury, or the direct effect of arsenic on tubule cells are thought to be associated with these processes. Studies suggest that sublethal arsenic poisoning is the cause of renal insufficiency and necrosis [86]. Loss of capillary integrity and increased glomerular capillary permeability could be the fundamental lesions in arsenic toxicity resulting in proteinuria. Other injuries in kidney could be due to the presence of hematouria, leukocyturia, and glycouria [2]. In summary, arsenite-induced nephrotoxicity and hepatotoxicity could be thought of as the outcome of oxidative stress, apoptosis, and upregulation of transcription factors such as AP-1, ATF-2, and Elk-1.

8.4.4 Testicular Toxicity

The effect of sodium arsenite on the male reproductive system is poorly defined. There are some existing reports in which arsenite intoxication is associated with spermatotoxicity [87–89], inhibition of testicular androgenesis, and reduction of the weight of the testes and accessory sex organs [90] in experimental animals. But the actual mechanism causing testicular toxicity from arsenic exposure remains unclear. Mice, continuously exposed to arsenic for 35 days, showed decreased sperm counts and motility [91,92]. Similarly, in rats, exposure to sodium arsenite suppressed spermatogenesis and testosterone release, inhibited testicular enzyme function, and reduced the weight of the male sex organs [90,93]. Arsenic exposure decreased epididymidal sperm count and testicular 17β -HSD and 3β -HSD enzyme activities,

reduced GSH levels, and increased protein carbonyl content in mouse testes. These oxidative stress-induced changes were prevented by ascorbic acid [91]. Waalkes et al. [94] showed that repeated arsenic exposure induced proliferative pre-neoplastic lesions of the testis. Testis proliferative lesions have also been found to be the outcome of estrogen treatment. In addition, estrogen also disrupts spermatogenesis and inhibits androgen production in association with suppressed gonadotrophin secretion from the pituitary [95,96]. Later, Jana et al. [87] investigated the mechanism of the arsenic-induced estrogenic mode of action on the hypothalamopituitary-testicular axis and concluded that the adverse effect of arsenite on male reproductive functions occurs via oxidative stress.

8.4.5 Arsenic Carcinogenicity

Existing literature and experimental evidence suggest that arsenic is a potent carcinogen. However, unsolved questions still remain regarding the permissible dose and exposure time that lead to variation in results of different types of cancer [2,15,97-101]. The literature provides sufficient evidence to classify arsenic as a human carcinogen [102]. Its role in laboratory animals is, however, not very clear, although several studies on the carcinogenicity of inorganic arsenic in hamsters were carried out and provided positive results [103-106]. In addition, DMA(V), the predominant metabolite of inorganic arsenic, has been reported as a rat bladder carcinogen [28,102]. Arsenic possesses tumor-promoting properties as is evidenced from its efficacy in activating transcription factors and changing gene expression for cell growth, proliferation, and malignant transformation; the exact molecular mechanism of its action in carcinogenicity, however, needs more study. Recent reports suggest that transcription factors (such as AP-1 and NF-KB) are activated via the arsenic-induced MAPK signaling pathway. These factors, in turn, can alter the gene expression responsible for arsenic-associated carcinogenicity [107]. Also, focal adhesion kinase, Src, Rho, Grb2, EGFR, ERK, cadherins, etc. are reported to be activated by arsenic exposure and remain involved in various cellular functions including carcinogenesis and tumor cell necrosis [28]. In addition, DMA(V) and TMAO(V) could generate oxidative stress and enhance the level of the oxidative DNA damage marker 8-hydroxydeoxyguanosine. As an outcome, cell proliferation is stimulated and carcinogenicity is induced [108,109]. Literature also suggests that arsenic could provoke bladder epithelial cell proliferation, up-regulate c-fos, c-jun, and EGR-1 proto-oncogenes expression, and collectively contribute to bladder cancer [110]. Smoking is another factor that potentiates the adverse effect of arsenic on bladder and lung cancer. These two factors, in combination, play a negative role in healthy subjects and possibly cause DNA damage [111,112]. Similarly, enhanced risk of arsenic-induced skin cancer can be thought of as a synergistic effect of its combination with sunlight. This additive effect can block DNA repair, stimulate angiogenesis, and alter DNA methylation patterns, which might deregulate cell cycle control and block physiological apoptosis [28]. Summarizing, we can say that the carcinogenic action of arsenic in various possible modes is due to enhanced oxidative stress, direct genotoxicity, altered DNA repair, and altered growth factor expressions.

8.5 Mechanism for the Toxic Effects of Arsenic

8.5.1 Oxidative Stress

For many reasons oxidative stress is considered to play a major role in arsenic-induced toxicity. Arsenic usually creates oxidative stress both in vitro and in vivo via the induction of heat shock or stress proteins [113,114]. GSH depletion could also alter the redox status and create a stressful and toxic situation in the cell [102]. This adverse situation normally arises when excess ROS are generated and react with cellular constituents. In fact, Liu et al. [115] reported that ROS could be detected in human-hamster hybrid cells within 5 min after exposure to arsenite. More evidence in this regard came from a number of experiments in which the generation of O_2^{-} and H_2O_2 after arsenic exposure was shown in some cell lines such as human vascular smooth muscle cells [116], human-hamster hybrid cells [115], and vascular endothelial cells [32]; whereas induction of only H₂O₂ was detected in HEL30 [117], NB4 [118], and CHOK1 [119] cells. In addition, generation of 'OH has been detected in rat striatum [120]. Arsenic-induced free radical formation was first demonstrated by Yamanaka and colleagues [121,122]. They were able to detect a DMA peroxyl radical *in vitro*. Further, the group reported oxidative damage in experimental mouse lungs after exposure of the mice to DMA(V). Studies suggest that oxygen in molecular form reacts with dimethylarsine, forming dimethylarsinic radical and superoxide anion. The second molecule of oxygen then combines and results in a dimethylarsinic peroxyl radical. These arsenic radicals are reported to be detrimental to cells [122]. Arsenic-induced free radical generation in mouse livers has also been reported [2]. Some other groups reported that the primary mechanism of arsenical-induced cell injury, cell death, and cancer involves the inhibition of mitochondrial respiration supported by NAD-linked substrates, the cofactor lipoic acid being used for the pyruvate dehydrogenase complex [123,124]. Specifically, inorganic arsenicals induce oxidative stress by inhibiting mitochondrial respiration and enhanced ROS generation that in turn might cause DNA mutations and ultimately contribute to the development of cancer [123].

The oxidant's metabolizing enzymes, catalase (CAT) and superoxide dismutase (SOD), have been reported to reduce arsenite-induced micronuclei in CHO cells [125]. Besides, antioxidants (vitamin E, methylamine, benzyl alcohol, etc.) could protect human fibroblasts from arsenite toxicity [102,126]. Trivalent organic arsenicals have been reported to inhibit GSH and thioredoxin reductase and thus could decrease the ability of cells to protect against oxidant-induced toxicity [127,128]. Thioredoxin has also been reported to affect gene expression because of its role in the regulation of the binding of transcription factors to DNA [129,130]. DNA damaging ROS are generated from iron released from ferritin *in vitro*, which is significantly enhanced by DMA(III) and DMA(V) in presence of ascorbic acid [131]. In this pathway, free iron plays a central role in generating harmful oxygen species by enhancing the conversion of highly reactive 'OH radical from O_2^- and H_2O_2 via the Haber-Weiss reaction [102]. Only DMA(III), however, remains active for iron release in the absence of ascorbic acid. Here, it should be noted that arsenate, arsenite, MMA(V), and MMA(III) remain inactive in iron release from ferritin.



FIGURE 8-2 Different targets and signaling pathways of arsenic toxicity.

8.5.2 Signaling Mechanism

The mechanisms underlying arsenic-induced organ pathophysiology have been studied for a long period of time and some excellent recent reviews and reports have revealed the probable signal transduction pathways involved in this toxic response. Once arsenic enters the human body it can alter signal transduction pathways, which leads to either activation or inhibition of different transcription factors and regulatory proteins that bind to DNA and regulate gene transcription (Figure 8-2) [1,132-135]. Recently, Ghosh et al. [136] showed the involvement of MAPKs in As intoxicated hearts and cardiomyocytes. Data suggest a marked increase in protein content of phosphorylated p38 MAPK and p-JNK in the heart tissue as well as in cardiomyocytes without any change in total protein content of these kinases. In contrast, a less marked increment of pERK was only noted in both the heart tissue and myocytes. Liu et al. [115] also reported that arsenic exposure *in vitro* strongly activated JNK and p38 but to a lesser extent ERK. Das et al. [137] investigated the involvement of p38 and JNK MAPKs in As-induced apoptosis in hepatocytes. When hepatocytes were pretreated with SB203580 and SP600125 separately, the effect of As on cell viability and caspase-3 activation then being determined, it was shown that JNK inhibition significantly increased cell viability and reduced caspase-3 activation in NaAsO₂-exposed hepatocytes, whereas p38 inhibition had no effect on these events, indicating that the p38-MAPK pathway is not involved in the prevention of As-induced apoptosis. Das et al. also determined the role of major PKC isoforms in As-induced hepatic pathophysiology and observed that arsenic exposure significantly increased the expression of PKC δ both *in vivo* and *in vitro*, whereas the expression of the other two major isoforms of PKC (PKC α and PKC ζ) remained unchanged [137]. All these activation processes appeared to involve generation of oxidative stress, as the free radical scavenger N-acetylcysteine inhibited the activation of the kinase pathways [1,138]. In different cell

lines (such as BEAS-2B cells, HaCaT cells, mouse epidermal JB6 cells, PC12 cells, Rat1 cells, and UROtsa cells), arsenic exposure activates the MAPK signaling pathway through the EGFR/MEK, EGFR/Ras/MEK, or Src/EGFR cascade [2]. It has been reported that arsenic-induced activation of the ERK and p38 pathway is mediated by Ras/Raf/MEK pathways and the JNK pathway is modulated by Rac, Rho, and MEKK3-4 [139]. At the same time in various cell types, EGFR-independent and MAPK-dependent activation of transcription factors by arsenic exposure has been described [2,125,134,140,141].

Arsenic exposure may cause type II diabetes. Recently Singh et al. [28] reviewed the effect of arsenic in diabetes mellitus. Prolonged exposure to it could decrease PPAR- γ expression and might reduce the sensitivity of insulin responsible for the induction of type II diabetes [142]. Replacement of a phosphate group from adenosine triphosphate (ATP) by arsenites slows down glucose metabolism, interrupts energy production, and interferes with ATP-dependent insulin secretion [143]. Because of its high affinity for sulfhydryl groups, arsenite forms covalent bonds with this group in insulin, with its receptors, as well as with the other molecules and enzymes involved in glucose metabolism [143]. In addition, a significant decrease in glucose-6-phosphatase activity due to chronic exposure to arsenic may cause hypoglycemia in both liver and kidneys [144]. Inorganic arsenic could increase oxidative stress and in turn leads to overexpression of various stress mediators such as NF-KB, JNK/SAPK, and hexosamine, causing insulin resistance and dysfunction of beta cells of the islets of Langerhans [145]. Experimental data suggest that trivalent arsenicals (like inorganic arsenic [iAs(III)], dimethylarsinous acid [DMA(III)], and monomethylarsonous acid [MMA(III)]) efficiently suppress Akt/protein kinase B phosphorylation by inhibiting 3-phosphoinositide-dependent kinase-I (PDK-1) activity and could cause significant inhibition of insulin-dependent glucose uptake. These sequences of steps result in hyperglycemia [28,146]. In summary, the results of all these studies suggest that the induction of arsenic-associated diabetes and its modulation is the outcome of decreased PPAR- γ expression, interference with ATP-dependent insulin secretion, altered glucocorticoid receptor mediated transcription, and PDK-1 inhibition.

8.5.3 Role of Transcription Factor

MAPKs are generally considered to be the mediators of apoptotic cell death under various pathophysiological conditions [147]. It has been reported that intracellular oxidative stress stimuli can activate both NF- κ B and MAP kinase modules [148]. Further, signal transduction via the MAPK pathway also activates the transcription factor activator protein-1 (AP-1) [1,132,134]. Literature suggests that arsenite activates JNK and p38 in HeLa cells following the stimulation of AP-1 transcriptional activity. This process leads to increased expression of the proto-oncogenes c-Jun and c-Fos [149].

Luster and Simeonova [150] correlated the activation of the MAPK pathway with hyperproliferation of bladder epithelium and suggest that the increased activation of AP-1 followed by the expression of AP-1-related genes plays some role in cell proliferation. Simeonova et al. [151] showed that the pathway for this activation probably involves c-Src and epidermal growth factor receptor (EGFR) signaling cascades.

Barchowsky et al. [152] showed that c-Src is activated at low arsenite concentrations $(<5 \mu M)$ and is also sufficient to induce cell proliferation and to increase H₂O₂ and superoxide accumulation, H₂O₂-dependent tyrosine phosphorylation, and NF- κ B-dependent transcription. The protein tyrosine phosphatases (PTPs) are known to act as negative regulators of EGFR. These molecules contain a preserved thiol group, which is oxidized by ROS and ultimately results in their inactivation. This process in turn leads to the transactivation of EGFRs without EGF binding [1,134]. No direct evidence has been found so far describing the interaction of arsenic with EGFR. But some reports proposed that arsenic could interact directly with the -SH group of EGFR or indirectly via ROS and this interaction might cause structural changes or dimerization [2].

Another report suggests that arsenic increases levels of EGFR ligand and heparin-binding EGF, and activates EGFR phosphorylation in human bronchial epithelial cells (downstream of EGFR effects include increased pERK and cell cycle promoter cyclin D1 levels) [153]. In addition, phosphorylated EGFR levels were found to be higher in those specimens (in human lung tumor biopsies) obtained from subjects with elevated toe nail arsenic content compared with subjects with lower exposures [1]. Arsenic also activates p53 in a dose-, cell type-, oxidative state-, and chemical species-specific manner; e.g., arsenite (As^{3+}) has the greatest ability to activate p53, although MMA shows no effect [154]. The role of p53 is considered to be important as its activation initiates apoptosis through DNA damage [2,155–157].

The transcription factor nuclear factor erythroid-2-related factor 2 (Nrf2) is also affected by arsenic exposure. Nrf2 tightly regulates the transcriptional machinery and gene expression, and controls the cellular defense system in response to exogenous oxidative insult. In normal its physiological state, Nrf2 remains bound to Keap1 as an inactive complex. Upon a stimulus via ROS, Keap1 renders Nrf2 free for its translocation to the nucleus by some chemical or electrophilic modification. After translocation to nuclease, Nrf2 activates the antioxidant-responsive element as well as the electrophilic-responsive element and stimulates the expression of relevant transcriptional proteins [2]. Hughes et al. [1] reported that tert-butylhydroquinone (tBHQ) and sulforaphane (SF) can protect a cell from trivalent arsenic-induced toxicity by activation of Nrf2 *in vitro*. With a knockout animal model it was shown that Nrf2 null mice are more prone to arsenic-induced hepatic and bladder toxicity compared to Nrf2 homozygous mice [158]. Wang et al. [159] also suggested that Nrf2 activation by arsenite or MMA(III) is different from that by tBHQ or SF.

8.6 Preventing Arsenic-Induced Toxic Effects by Antioxidants

Among the most important factors governing the toxic effects of arsenic are the induction of ROS and depletion of antioxidant defenses in a system. It is, therefore, evident that some device that can increase the antioxidant capacity of the cells can be a therapeutic strategy in arsenic poisoning and this could be accomplished by either reducing the possibility of the metal interacting with critical biomolecules of the cells or by supplementation with antioxidant molecules from appropriate external sources. Some of the important antioxidant molecules that have been explored for their role in reversing arsenic-induced oxidative stress and related organ dysfunctions have been discussed in brief.

8.6.1 Role of Taurine

Taurine (2-aminoethanesulfonic acid) is a derivative of the sulfur-containing amino acid cysteine and is present in many tissues of mammals in high concentrations. It has been reported that the conditionally essential amino acid taurine accounts for 25 to 50% of the amino acid pool in myocardium, possesses antioxidant properties, and has the ability to reduce arsenicinduced oxidative burden of cardiac damage [136]. It is neither metabolized nor incorporated into cellular proteins, suggesting that it may play an important role in cellular cytosolic functioning [136].

8.6.1.1 Taurine in Prevention of Arsenic-Induced Cardiac Disorder

Arsenic exposure reduced viability of cardiomyocytes, increased ROS production and intracellular calcium release, and induced apoptosis by mitochondrial-dependent caspase-3 activation and poly-ADP ribose polymerase (PARP) cleavage. In arsenic-induced cardiomyocytes death, the transcription factor NF-KB plays an important role and it has been observed that all these changes are associated with increased IKK and NF- κ B (p65) phosphorylation. Results by Ghosh et al. [136] suggest that arsenic induces a marked increase in the activity of p38 and JNK MAPKs, but not ERK to that extent in heart tissue as well as in cardiomyocytes. Arsenic administration also down-regulated the anti-apoptotic (Bcl-2, Bcl-xL) and up-regulated the proapoptotic (Bax, Bad) Bcl-2 family proteins in cardiomyocytes. Taurine treatment, on the other hand, mitigated all these apoptotic events and maintained the balance, suggesting its protective role in arsenic-induced cardiomyocyte apoptosis is mediated via the attenuation of p38 and JNK MAPK signaling pathways. Also, arsenic administration altered a number of cardiac oxidative stress biomarkers and other apoptotic indices in vivo and taurine supplementation reduced the activation of this pathway and blocked the apoptotic signaling cascade. Results suggest that taurine supplementation prevented this arsenic-induced myocardial pathophysiology, attenuated NF-KB activation via IKK, p38, and JNK MAPK signaling pathways, and represents a promising approach for the protection of heart tissue against As-induced cardiovascular burden [136].

8.6.1.2 Taurine against As-Induced Hepatic and Testicular Apoptosis

The hepatotoxic effect of arsenic has been investigated extensively but its exact mechanism is still not clear. The main cause for arsenic-induced hepatic disorder is oxidative stress and the exhaustion of GSH in the liver [160–162]. Arsenic exposure significantly increased the level of all the markers related to hepatic oxidative stress. Arsenic administration also caused mito-chondrial injury and disturbed the balance of pro-apoptotic and anti-apoptotic members of the Bcl-2 family proteins, which leads to increased level of Apaf-1, activation of caspase 9/3, cleavage of PARP protein, and ultimately apoptosis. Investigating the detailed cell signaling pathway reveals that arsenic exposure markedly increased JNK and p38 phosphorylation. It also up-regulated PKCδ and pre-exposure with its specific inhibitor (rottlerin) blocked the



FIGURE 8–3 Schematic diagram showing the protective action of taurine against As-induced hepatotoxicity [137,160].

activation of JNK suggesting that PKCδ is upstream for As-induced JNK activation and mitochondrial-dependent apoptosis [137,160]. On the other hand, taurine treatment both pre- and post-arsenic exposure and/or incubation of the hepatocytes with taurine was found to be very effective in mitigating As-induced oxidative stress and apoptosis [137,160]. Combining all, the results suggest that taurine treatment confers its hepatic protection via blocking the activation of PKCδ and JNK, as well as the generation and accumulation of ROS (Figure 8–3).

Many reports from the literature suggest that arsenic also has a severe toxic effect on testicular tissue of experimental animals [87,163,164]. Das et al. [163] showed that arsenic exposure causes testicular apoptosis through the mitochondrial-dependent pathway via mitogenactivated protein kinases (MAPKs) and Akt as well as NF- κ B (p65) in testicular tissue. As administration significantly decreased testicular steroidogenic enzymes, Δ^5 -3 β -HSD and 17 β -HSD activities reduced the plasma testosterone level, testicular sperm count, and sperm motility [87,163]. In addition, As exposure increased the oxidative stress, ROS production, and serum TNF- α level, and decreased the activities of the antioxidant enzymes in the testicular tissue [165]. However, taurine supplementation was found to be effective in counteracting As-induced oxidative stress through its antioxidative properties and amelioration of apoptosis in testicular tissue by down-regulating the activation of phospho-ERK1/2, phospho-p38, NF- κ B and other mitochondrial-dependent signaling molecules, as well as by up-regulating the expression of Bcl-2 and other molecules of the phospho-Akt signaling pathways.

8.6.1.3 Taurine against As-Induced Oxidative Cerebral and Renal Disorders

In addition to heart, liver, and testis, arsenic intoxication has also been detected in the kidneys and brain [166,167]. Increased productions of ROS and RNS as well as enhanced lipid peroxidation and decreased intracellular antioxidant functions have been reported in the kidney tissue of As-exposed experimental animals. Intracellular mechanisms suggest that oxidative stressmediated renal dysfunction is associated with mitogen-activated protein kinases (MAPKs) and NF- κ B, which ultimately induce mitochondria-dependent apoptotic cell death [166]. In the brain, both chronic and acute As toxicity have also been found to be mediated via oxidative stress, leading to enhanced ROS formation, lipid peroxidation, etc. and decreased activities of the antioxidant and membrane-bound enzyme acetylcholine esterase and levels of reduced glutathione [167]. Treatment with taurine prior to arsenic administration, however, effectively ameliorated As-induced oxidative renal and neuronal dysfunctions and apoptotic cell death [167]. These results suggest that taurine possesses the ability to ameliorate arsenic-induced oxidative insult and renal damage, probably because of its antioxidant power and functioning via MAPKs/NF- κ B and mitochondria-dependent pathways. This unique molecule could, therefore, be considered as a beneficial agent in renal and cerebral oxidative stress [160].

8.6.2 Role of N-Acetyl Cysteine

N-Acetyl cysteine (NAC), the thiol-based antioxidant and derivative of cysteine and precursor of reduced glutathione, plays an important role in the protection of cellular constituents against oxidative damage and in the detoxification of many electrophiles. The protective mechanism of NAC originates from its ability to stimulate and sustain the intracellular levels of reduced glutathione levels, thereby maintaining intracellular GSH levels and detoxifying ROS [168]. The other advantage of NAC is that it possesses metal chelating properties. Recent studies by Modi et al. [169] suggested that co-supplementation of NAC along with zinc reduces the arsenic-induced oxidative burden in rat liver and kidneys. Administration of NAC and succimer in combination, post-arsenic exposure, led to a significant removal of arsenic from soft organs [170]. More recently, protective effects of N-acetylcysteine against arsenic-induced oxidative stress and reprotoxicity have been reported [168]. Moreover, Flora [2] also presented evidence of a novel therapeutic combination of NAC with meso-2,3-dimercaptosuccinic acid (DMSA), which provides greater effectiveness during chelation of arsenic in rats by co-administration.

8.6.3 Role of Melatonin

Melatonin or N-acetyl-5-methoxytryptamine, the main secretory hormonal product of the pineal gland, participates in many physiological functions, such as control of reproductive functions, modulation of immune system activity, limitation of tumorogenesis, and effective inhibition of oxidative stress due to its efficacy as a free radical scavenger and indirect antioxidant [171–174]. Due to its small size and lipophilicity, melatonin can cross biological membranes easily and reach all the compartments of the cellular system. Melatonin acts as an antagonist for a number of endogenous and exogenous free radicals during various cellular processes. Moreover, it has been shown to be an efficient protector of DNA, protein, and lipids in cellular membranes [171,175]. It can stimulate GPx in the brain and provide an indirect

protection against free radical attack. Melatonin has been shown to prevent the induction of free radical damage in various ways, such as that caused by ingestion of toxins, ionizing radiation, ischemia, reperfusion, and excessive exercise [144]. Mahabady et al. [171] recently showed that melatonin and vitamin E protect against sodium arsenite-induced skeletal malformations in rats and they suggested that melatonin has more prophylactic effect than vitamin E on the incidence of sodium arsenite-induced cleft palate and spina bifida, but this difference was not significant. Pal and Chatterjee [144] investigated the protective role of melatonin against arsenic-induced metabolic toxicity in Wistar rats and suggested that melatonin opposes the carbohydrate depleting effect of arsenic in liver. This carbohydrate depletion may in turn contribute to the accumulation of glycolytic intermediates in liver, and thus help in increasing glucose 6-phosphatase activity to provide more glucose to blood to replenish the loss of blood glucose due to arsenic treatment. Combining all these facts, melatonin is found to reduce the arsenic-induced hypoglycemic condition with associated glycosuria and increased glycogeno-lytic and decreased glycolytic activities of the liver. Accordingly, it may be concluded that melatonin can serve as a prophylactic agent against arsenic-induced metabolic toxicity [144].

8.6.4 Role of α-Lipoic Acid

 α -Lipoic acid (LA) is a low molecular weight dithiol antioxidant and is an important co-factor in several multienzyme complexes in the mitochondria [176]. LA, and its reduced form dihydrolipoic acid (DHLA), has two free sulfhydryl groups and the two forms LA/DHLA possess a high antioxidant potential capable of quenching reactive oxygen and nitrogen species such as hydroxyl radicals, peroxyl radicals, superoxide, and hypochlorous acid [177]. In addition, LA also has metal chelating activity. Hatch et al. showed that lipoic acid acts as an antidote in arsenic-induced toxicity [178] and as it is effective in a micromolar range (millimolar NAC is needed for a similar effect), it has an advantage over NAC in protecting against GSH loss [179]. Its coadministration with DMSA has been shown to be of significant value in experimental lead intoxication and could also be expected to have the same possible beneficial effect in arsenic poisoning [170]. Samuel et al. [180] reported that lipoic acid treatment mitigates arsenic-induced protein oxidative damage of rats, because of its antioxidant nature consisting of free radical scavenging and metal chelating properties. A closer look at all these reports suggests that the protective effects of LA against arsenic-induced and oxidative stress-mediated toxicity could be attributed to its ability to be reduced by NADH to dihydrolipoic acid (DHLA), which, being a stronger antioxidant than LA, could scavenge excess oxidants and recycle other antioxidants such as vitamin E, vitamin C, and glutathione; it could form chelate with arsenic, thus preventing free radical generation and diminishing oxidant attacks on biomacromolecules. Also, LA could promote antioxidant defense by elevating the levels of antioxidant molecules (like GSH) via the induction of phase II enzymes.

8.6.5 Role of Silymarin and Quercetin

Silymarin and quercetin are polyphenolic antioxidant flavonoids which possess anticancer and cytoprotective effects and are widely available in vegetable sources [181]. Bioflavonoids,

ingested through a well-balanced diet, could function with diversified pharmacological properties and prevent a wide variety of human illnesses [182]. Bongiovanni et al. [183] investigated the effects of the plant flavonoids silymarin and quercetin on arsenite-induced oxidative stress in CHO-K1 cells and reported the biomedical potential of these flavonoids in arsenic-induced cytotoxicity in the CHO-K1 cell line. Soria et al. [182] showed the differential effects of these flavonoids on arsenite-induced cytotoxicity in two human breast adenocarcinoma cell lines. Mishra and Flora [184] showed the efficacy of quercetin and a thiol chelating agent, monoisoamyl 2,3-dimercaptosuccinic acid (MiADMSA), either individually or in combination against arsenic-induced oxidative stress and mobilization of metal in mice. They suggested that combination therapy with an antioxidant and a thiol chelator might be a better treatment option than monotherapy to treat chronic cases of arsenic poisoning. Ghosh et al. [185] demonstrated that a combination of QC and DMSA in a nanocapsulated drug delivery system provided better therapeutic efficacy than its bulk form and it protected liver from arsenic-induced fibrosis, safeguarded cells from ROS-induced damage, restored mitochondrial integrity, and down-regulated ROS-induced signaling pathways that led the cells toward p53-dependent apoptosis [186].

8.6.6 Herbal Antioxidant

Arjunolic acid (AA: 2,3,23-trihydroxyolean-12-en-28-oic acid, Figure 8-4A), a natural chiral triterpenoid saponin (isolated from the bark of *Terminalia arjuna*), is well known for various biological functions, including cytoprotective [187], anti-fungal [188], anti-bacterial [189], anti-diabetic [190], and insect growth inhibitor activity [191]. Several researchers demonstrated the role of arjunolic acid against arsenic-induced organ pathophysiology [192,193]. Sodium arsenite administration at a dose of 10 mg/kg body weight for 2 days significantly induced severe oxidative stress, altered the prooxidant-antioxidant balance in the hepatic tissue as evidenced by significant reduction in the antioxidant enzymes' activities, and depleted cellular reduced glutathione and total thiols level. In addition, the same exposure enhanced leakage of hepatic serum marker enzymes (alanine transaminase and alkaline phosphatase) indicating significant hepatic damage. It also enhanced levels of lipid peroxidation and protein carbonylation, which are considered to be the two important parameters of oxidative stressinduced organ pathophysiology. As-induced cell death is primarily necrotic in nature as confirmed by DNA fragmentation analysis. Arjunolic acid supplementation at a dose of 20 mg/kg body weight for 4 days prior to arsenic exposure prevented all the alterations and protected the cells from necrotic death. Histological findings also confirmed the disruption of liver architecture around the central vein associated with extensive centrilobular necrosis and bile duct proliferation. Arjunolic acid pretreatment prevented such changes and maintained the normalcy of the organ. Arjunolic acid treatment prevented arsenic-induced ROS production. This protective action arises because arjunolic acid itself has a radical scavenging property as was evidenced from its superoxide, hydroxyl, and nitric oxide radical quenching ability in a cellfree system [187,194]. In addition, arjunolic acid also contains one carboxylic hydrogen atom, which can easily be abstracted by any free radical (Figure 8-4B), thus making arjunolic acid a potent radical scavenger and antioxidant. Therefore, it is likely that part of the beneficial effects



FIGURE 8–4 (A): Structure of arjunolic acid. (B): Possible mechanism of free radical scavenging activity by arjunolic acid. (C): Possible site of chelation of arjunolic acid with arsenic [187, 193].

of arjunolic acid result from the detoxification of ROS produced during arsenic intoxication. On the other hand, arjunolic acid also contains polyhydroxyl groups, two of which are vicinal; these two hydroxyl groups may form a five-membered chelate complex with arsenic (Figure 8–4C) and consequently prevent arsenic from exhibiting its toxicity [187,194]. Arjunolic acid can, therefore, be used as a reliable chelator against arsenic-induced toxicity.

In addition to the liver, similar results on the beneficial role of arjunolic acid have been reported for arsenic-induced organ dysfunction in the kidneys, brain, testes, and heart tissue [136,137,163,195].

8.7 Conclusions and Future Directions

Arsenic (As)-induced organ dysfunction and apoptosis have received considerable attention in the past, which continues up to the present time. New relevant information on arsenic-induced apoptosis has been provided recently. Collective data suggest that arsenic, after entering a cell, primarily disturbs the balance between prooxidants and endogenous antioxidants, which in turn triggers a number of pathways and ultimately determines the fate of the cell either to survive or to undergo apoptosis [123]. Although the mechanism of arsenic-induced toxicity is still not clear, findings from mechanistic studies indicate that ROS produced by arsenic play the most crucial role in exerting its toxic effects in various organ pathophysiologies. Mitochondria are considered to be one of the major targets for the generation of reactive species. The downstream cascade including mitochondrial depolarization and caspase-3 activation could be influenced by ROS produced earlier in the system. Inorganic arsenic and its methylated derivatives also play a key role in disturbing intracellular calcium while inducing apoptosis [196]. The unique molecule 8-OHdG is considered to be an excellent marker of arsenic-induced oxidative stress damage on DNA as its increased levels have been reported in arsenic-exposed animals [123]. The contribution of MAPK pathways to cell growth regulation and death during arsenic exposure has also been reported. In fact, data suggest that some important cellular functions such as proliferation, differentiation, and apoptosis might be controlled by various arsenic-induced signaling pathways. Based on the published reports, the following points can be presented as key features in arsenic-induced cellular responses leading to apoptosis. These are: generation of ROS and RNS; increased levels of 8-OHdG; intracellular GSH depletion, Ca disturbances, and PKC activation; Bax/Bak-dependent signaling; and activation of c-Jun–N-terminal kinase (JNK) as well as transcription factors, AP-1/ NF- κ B and DNA strand breaks [123]. Future studies are needed for a better insight into dissecting the exact mechanisms of action as well as identifying the target genes and modern techniques like microarray-based gene expression analysis and others that would be very helpful in this particular direction of research [2]. From the above discussion, it is evident that the role of oxidative stress and associated biochemical events in apoptosis deserves special attention.

It is also known that various antioxidants could protect cells from oxidative injury. Since arsenic causes toxicity by damaging the body's oxidative defense mechanism, supplementation of suitable antioxidants via chelation therapy with a proper chelator (like a thiol chelator) can be used as a possible solution. A lipophilic thiol chelator (like MiADMSA) may be logically preferred for this purpose. Here, it should be kept in mind that this approach would play a major role for the identification of a safe, suitable, and practical effective treatment for arsenic poisoning in the future. Some groups believe that chelator therapy alone may not be beneficial for better clinical recoveries and a combination therapy will be more effective [2,170]. In support of this opinion it can be said that ROS and RNS play the major role in arsenic poisoning via the direct binding with SH groups. Combination therapy using antioxidants and a thiol-chelating agent could, therefore, be a major future strategy for the treatment of chronic arsenic poisoning.

References

- Hughes MF, Beck BD, Chen Y, Lewis AS, Thomas DJ. Arsenic exposure and toxicology: a historical perspective. Toxicol Sci 2011;123:305–32.
- [2] Flora SJS. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51:257–81.
- [3] Concha G, Nermell B, Vahter M. Metabolism of inorganic arsenic in children with chronic high arsenic exposure in northern Argentina. Environ Health Perspect 1998;106:355–9.

- [4] Concha G, Vogler G, Nermell B, Vahter M. Low-level arsenic excretion in breast milk of native Andean women exposed to high levels of arsenic in the drinking water. Int Arch Occup Environ Health 1998;71:42–6.
- [5] Garciavargas GG, Garciarangel A, Aguilarromo M, Garciasalcedo J, Delrazo LM, Ostroskywegman P, et al. A pilot-study on the urinary-excretion of porphyrins in human populations chronically exposed to arsenic in Mexico. Hum Exp Toxicol 1991;10:189–93.
- [6] Garciavargas GG, Delrazo LM, Cebrian ME, Albores A, Ostroskywegman P, Montero R, et al. Altered urinary porphyrin excretion in a human-population chronically exposed to arsenic in Mexico. Hum Exp Toxicol 1994;13:839–47.
- [7] Yen CC, Lu FJ, Huang CF, Chen WK, Liu SH, Lin-Shiau SY. The diabetogenic effects of the combination of humic acid and arsenic: in vitro and in vivo studies. Toxicol Lett 2007;172:91–105.
- [8] British Geological Survey: Arsenic Contamination of Groundwater in Bangladesh, vol. 2, Final Report. BGS Technical Report WC/00/19. British Geological Survey, Keyworth; 2001.
- [9] Mazumder DNG, Haque R, Ghosh N, De BK, Santra A, Chakraborty D, et al. Arsenic levels in drinking water and the prevalence of skin lesions in West Bengal, India. Int J Epidemiol 1998;27:871–7.
- [10] Chowdhury UK, Biswas BK, Chowdhury TR, Samanta G, Mandal BK, Basu GC, et al. Groundwater arsenic contamination in Bangladesh and West Bengal, India. Environ Health Perspect 2000;108:393–7.
- [11] Platanias LC. Biological responses to arsenic compounds. J Biol Chem 2009;284:18583–18587.
- [12] Vahter M. Health effects of early life exposure to arsenic. Basic Clin Pharmacol Toxicol 2008;102:204–11.
- [13] Smith AH, Hopenhayn-Rich C, Bates MN, Goeden HM, Hertz-Picciotto I, Duggan HM, et al. Cancer risks from arsenic in drinking water. Environ Health Perspect 1992;97:259–67.
- [14] Chiou HY, Hsueh YM, Liaw KF, Horng SF, Chiang MH, Pu YS, et al. Incidence of internal cancers and ingested inorganic arsenic: a seven-year follow-up study in Taiwan. Cancer Res 1995;55:1296–300.
- [15] Celik I, Gallicchio L, Boyd K, Lam TK, Matanoski G, Tao X, et al. Arsenic in drinking water and lung cancer: a systematic review. Environ Res 2008;108:48–55.
- [16] Yang CY, Chang CC, Chiu HF. Does arsenic exposure increase the risk for prostate cancer? J Toxicol Environ Health Part A 2008;71:1559–63.
- [17] Liu J, Waalkes MP. Liver is a target of arsenic carcinogenesis. Toxicol Sci 2008;105:24–32.
- [18] Baastrup R, Sørensen M, Balstrøm T, Frederiksen K, Larsen CL, Tjønneland A, et al. Arsenic in drinkingwater and risk for cancer in Denmark. Environ Health Perspect 2008;116:231–7.
- [19] Chen CJ, Wang SL, Chiou JM, Tseng CH, Chiou HY, Hsueh YM, et al. Arsenic and diabetes and hypertension in human populations: a review. Toxicol Appl Pharmacol 2007;222:298–304.
- [20] Roy P, Saha A. Metabolism and toxicity of arsenic: a human carcinogen. Curr Sci 2002;80:38-45.
- [21] Cullen WR, Reimer KJ. Arsenic speciation in the environment. Chem Rev 1989;89:713–64.
- [22] Aposhian HV. Enzymatic methylation of arsenic species and other new approaches to arsenic toxicity. Annu Rev Pharmacol Toxicol 1997;37:397–419.
- [23] Edmonds JS, Francesconi KA. Arseno-sugars from brown kelp (*Ecklonia radiata*) as intermediates in cycling of arsenic in a marine ecosystem. Nature 1981;289:602–4.
- [24] Edmonds JS, Francesconi KA. Transformations of arsenic in the marine environment. Experimentia 1987;43:553–7.
- [25] Buchet JP, Pauwels J, Lauwerys R. Assessment of exposure to inorganic arsenic following ingestion of marine organisms by volunteers. Environ Res 1994;66:44–51.
- [26] Buchet JP, Lauwerys R. Study of inorganic arsenic methylation by rat liver in vitro: relevance for the interpretation of observations in man. Arch Toxicol 1985;57:125–9.
- [27] Styblo M, Yamauchi H, Thomas DJ. Comparative in vitro methylation of trivalent and pentavalent arsenicals. Toxicol Appl Pharmacol 1995;135:172–8.

- [28] Singh AP, Goel RK, Tajpreet K. Mechanisms pertaining to arsenic toxicity. Toxicol Int 2011;18:87–93.
- [29] Chen CJ, Hsueh YM, Lai MS, Shyu MP, Chen SY, Wu MM, et al. Increased prevalence of hypertension and long-term arsenic exposure. Hypertension 1995;25:53–60.
- [30] Chen CJ, Chiou HY, Chiang MH, Lin LJ, Tai TY. Dose-response relationship between ischemic heart disease mortality and long-term arsenic exposure. Arterioscler Thromb Vasc Biol 1996;16:504–10.
- [31] Ohnishi K, Yoshida H, Shigeno K, Nakamura S, Fujisawa S, Naito K, et al. Prolongation of the QT interval and ventricular tachycardia in patients treated with arsenic trioxide for acute promyelocytic leukemia. Ann Intern Med 2000;133:881–5.
- [32] Barchowsky A, Klei LR, Dudek EJ, Swartz HM, James PE. Stimulation of reactive oxygen, but not reactive nitrogen species, in vascular endothelial cells exposed to low levels of arsenite. Free Radic Biol Med 1999;27:1405–12.
- [33] Smith KR, Klei LR, Barchowsky A. Arsenite stimulates plasma membrane NADPH oxidase in vascular endothelial cells. Am J Physiol Lung Cell Mol Physiol 2001;280:L442–9.
- [34] Luong MW, Rabkin SW. Verapamil but not calpain or creatine alters arsenate induced cardiac cell death. Toxicol Ind Health 2009;25:169–76.
- [35] Hays AM, Lantz RC, Rodgers LS, Sollome JJ, Vaillancourt RR, Andrew AS, et al. Arsenic-induced decreases in the vascular matrix. Toxicol Pathol 2008;36:805–17.
- [36] Bunderson M, Coffin JD, Beall HD. Arsenic induces peroxynitrite generation and cyclooxygenase-2 protein expression in aortic endothelial cells: possible role in atherosclerosis. Toxicol Appl Pharmacol 2002;184:11–18.
- [37] Lee PC, Ho IC, Lee TC. Oxidative stress mediates sodium arsenite-induced expression of heme oxygenase-1, monocyte chemoattractant protein-1, and interleukin-6 in vascular smooth muscle cells. Toxicol Sci 2005;85:541–50.
- [38] Pysher MD, Chen QM, Vaillancourt RR. Arsenic alters vascular smooth muscle cell focal adhesion complexes leading to activation of FAK-src mediated pathways. Toxicol Appl Pharmacol 2008;231:135–41.
- [39] Tsou TC, Tsai FY, Hsieh YW, Li LA, Yeh SC, Chang LW. Arsenite induces endothelial cytotoxicity by downregulation of vascular endothelial nitric oxide synthase. Toxicol Appl Pharmacol 2005;208:277–84.
- [40] Balakumar P, Kaur J. Arsenic exposure and cardiovascular disorders: an overview. Cardiovasc Toxicol 2009;9:169–76.
- [41] Lee MY, Lee YH, Lim KM, Chung SM, Bae ON, Kim H, et al. Inorganic arsenite potentiates vasoconstriction through calcium sensitization in vascular smooth muscle. Environ Health Perspect 2005;113:1330–5.
- [42] Cifuentes F, Bravo J, Norambuena M, Stegen S, Ayavire A, Palacios J. Chronic exposure to arsenic in tap water reduces acetylcholine-induced relaxation in the aorta and increases oxidative stress in female rats. Int J Toxicol 2009;28:534–41.
- [43] Wu MM, Chiou HY, Ho IC, Chen CJ, Lee TC. Gene expression of inflammatory molecules in circulating lymphocytes from arsenic-exposed human subjects. Environ Health Perspect 2003;111:1429–38.
- [44] Hsieh YC, Hsieh FI, Lien LM, Chou YL, Chiou HY, Chen CJ. Risk of carotid atherosclerosis associated with genetic polymorphisms of apolipoprotein E and inflammatory genes among arsenic exposed residents in Taiwan. Toxicol Appl Pharmacol 2008;227:1–7.
- [45] States JS, Srivastava S, Chen Y, Barchowsky A. Arsenic and cardiovascular disease. Toxicol Sci 2009;107:312–23.
- [46] Bunderson M, Brooks DM, Walker DL, Rosenfeld ME, Coffin JD, Beall HD. Arsenic exposure exacerbates atherosclerotic plaque formation and increases nitrotyrosine and leukotriene biosynthesis. Toxicol Appl Pharmacol 2004;201:32–9.
- [47] Tsai SH, Hsieh MS, Chen L, Liang YC, Lin JK, Lin SY. Suppression of Fas ligand expression on endothelial cells by arsenite through reactive oxygen species. Toxicol Lett 2001;123:11–19.
- [48] Tseng CH, Huang YK, Huang YL, Chung CJ, Yang MH, Chen CJ, et al. Arsenic exposure, urinary arsenic speciation, and peripheral vascular disease in blackfoot disease-hyperendemic villages in Taiwan. Toxicol Appl Pharmacol 2005;206:299–308.
- [49] Borgono JM, Vicent P, Venturino H, Infante A. Arsenic in the drinking water of the city of Antofagasta: epidemiological and clinical study before and after the installation of a treatment plant. Environ Health Perspect 1977;19:103–5.
- [50] Zaldívar RA. Morbid condition involving cardio-vascular, broncho-pulmonary, digestive and neural lesions in children and young adults after dietary arsenic exposure. Zentralbl Bakteriol B 1980;170:44–56.
- [51] Piao F, Ma N, Hiraku Y, Murata M, Oikawa S, Cheng F, et al. Oxidative DNA damage in relation to neurotoxicity in the brain of mice exposed to arsenic at environmentally relevant levels. J Occup Health 2005;47:445–9.
- [52] Felix K, Manna SK, Wise K, Barr J, Ramesh GT. Low levels of arsenite activates nuclear factor kappaB and activator protein-1 in immortalized mesencephalic cells. J Biochem Mol Toxicol 2005;19:67–77.
- [53] Yip SF, Yeung YM, Tsui EY. Severe neurotoxicity following arsenic therapy for acute promyelocytic leukemia: potentiation by thiamine deficiency. Blood 2002;99:3481–2.
- [54] Gopalkrishnan A, Rao MV. Amelioration by vitamin A upon arsenic induced metabolic and neurotoxic effects. J Health Sci 2006;52:568–77.
- [55] Vahidnia A, Romijn F, Tiller M, van der Voet GB, de Wolff FA. Arsenic-induced toxicity: effect on protein composition in sciatic nerve. Hum Exp Toxicol 2006;25:667–74.
- [56] Bardullas U, Limón-Pacheco JH, Giordano M, Carrizales L, Mendoza-Trejo MS, Rodríguez VM. Chronic low-level arsenic exposure causes gender-specific alterations in locomotor activity, dopaminergic systems, and thioredoxin expression in mice. Toxicol Appl Pharmacol 2009;239:169–77.
- [57] Dwivedi N, Flora SJ. Concomitant exposure to arsenic and organophosphates on tissue oxidative stress in rats. Food Chem Toxicol 2011;49:1152–9.
- [58] Klaassen CD. Heavy metals and heavy-metal antagonist. In: Gilman AG, Rall TW, Nies AS, Taylor P, editors. The pharmacological basis of therapeutics. New York: McGraw-Hill; 1996. pp. 1592–614.
- [59] Otto D, Xia Y, Li Y, Wu K, He L, Telech J, et al. Neurosensory effects of chronic human exposure to arsenic associated with body burden and environmental measures. Hum Exp Toxicol 2007;26:169–77.
- [60] Tseng HP, Wang YH, Wu MM, The HW, Chiou HY, Chen CJ. Association between chronic exposure to arsenic and slow nerve conduction velocity among adolescents in Taiwan. J Health Popul Nutr 2006;24:182–9.
- [61] Fujino Y, Guo X, Shirane K, Liu J, Wu K, Miyatake M, et al. Japan Inner Mongolia Arsenic Pollution Study Group. Arsenic in drinking water and peripheral nerve conduction velocity among residents of a chronically arsenic-affected area in Inner Mongolia. J Epidemiol 2006;16:207–13.
- [62] Vahidnia A, van der Voet GB, de Wolff FA. Arsenic neurotoxicity—a review. Hum Exp Toxicol 2007;26:823-32.
- [63] Rodríguez VM, Carrizales L, Mendoza MS, Fajardo OR, Giordano M. Effects of sodium arsenite exposure on development and behavior in the rat. Neurotoxicol Teratol 2002;24:743–50.
- [64] Namgung U, Xia Z. Arsenic induces apoptosis in rat cerebellar neurons via activation of JNK3 and p38 MAP kinases. Toxicol Appl Pharmacol 2001;174:130–8.
- [65] Vahidnia A, Romijn F, van der Voet GB, de Wolff FA. Arsenic-induced neurotoxicity in relation to toxicokinetics: effects on sciatic nerve proteins. Chem Biol Interact 2008;176:188–95.
- [66] Patlolla AK, Tchounwou PB. Serum acetyl cholinesterase as a biomarker of arsenic induced neurotoxicity in Sprague-Dawley rats. Int J Environ Res Public Health 2005;2:80–3.
- [67] Luo JH, Qiu ZQ, Shu WQ, Zhang YY, Zhang L, Chen JA. Effects of arsenic exposure from drinking water on spatial memory, ultra-structures and NMDAR gene expression of hippocampus in rats. Toxicol Lett 2009;184:121–5.

- [68] Krüger K, Straub H, Hirner AV, Hippler J, Binding N, Musshoff U. Effects of monomethylarsonic and monomethylarsonous acid on evoked synaptic potentials in hippocampal slices of adult and young rats. Toxicol Appl Pharmacol 2009;236:115–23.
- [69] Ríos R, Zarazúa S, Santoyo ME, Sepúlveda-Saavedra J, Romero-Díaz V, Jiménez V, et al. Decreased nitric oxide markers and morphological changes in the brain of arsenic-exposed rats. Toxicology 2009;261:68–75.
- [70] Lin AM, Chao PL, Fang SF, Chi CW, Yang CH. Endoplasmic reticulum stress is involved in arsenite induced oxidative injury in rat brain. Toxicol Appl Pharmacol 2007;224:138–46.
- [71] Sinczuk-Walczak H. Nervous system disorders induced by occupational exposure to arsenic and its inorganic compounds: a literature review. Med Pr 2009;60:519–22.
- [72] Hall AH. Chronic arsenic poisoning. Toxicol Lett 2002;128:69–72.
- [73] Calderon J, Navarro ME, Jimenez-Capdeville ME, Santos-Diaz MA, Golden A, Rodriguez-Leyva I, et al. Exposure to arsenic and lead and neuropsychological development in Mexican children. Environ Res 2001;85:69–76.
- [74] Nandi D, Patra RC, Swarup D. Oxidative stress indices and plasma biochemical parameters during oral exposure to arsenic in rats. Food Chem Toxicol 2006;44:1579–84.
- [75] Kokilavani V, Devi MA, Sivarajan K, Panneerselvam C. Combined efficacies of DL-alpha-lipoic acid and meso 2,3-dimercaptosuccinic acid against arsenic induced toxicity in antioxidant systems of rats. Toxicol Lett 2005;160:1–7.
- [76] Flora SJS, Tripathi N. Hepatic and renal metallothionein induction following single oral administration of gallium arsenide in rats. Biochem Mol Biol Int 1998;45:1121–7.
- [77] Xie YX, Liu J, Liu YP, Klaassen CD, Waalkes MP. Toxicokinetic and genomic analysis of chronic arsenic exposure in multidrug-resistance mdr1a/1b (-/-) double knockout mice. Mol Cell Biochem 2004;255:11–18.
- [78] Flora SJS, Dube SN, Kannan GM, Arora U, Malhotra PR. Therapeutic potential of meso 2,3-dimercaptosuccinic acid and 2,3-dimercaptopropane 1-sulfonate against chronic arsenic poisoning in rats. Biometals 1995;8:111–17.
- [79] Flora SJS, Kannan GM, Kumar P. Selenium effects on gallium arsenide induced biochemical and immunological changes in male rats. Chem Biol Interact 1999;122:1–13.
- [80] Flora SJS, Bhadauria S, Dhaked RK, Pant SC. Arsenic induced blood and brain oxidative stress and its response to some thiol chelators in male rats. Life Sci 2005;77:2324–37.
- [81] Mishra D, Flora SJS. Differential oxidative stress and DNA damage in rat brain regions and blood following chronic arsenic exposure. Toxicol Ind Health 2008;24:247–56.
- [82] Das S, Santra A, Lahiri S, Mazumder DNG. Implications of oxidative stress and hepatic cytokine (TNF-α and IL-6) response in the pathogenesis of hepatic collagenesis in chronic arsenic toxicity. Toxicol Appl Pharmacol 2005;204:18–26.
- [83] Aposhian HV, Aposhian MM. Arsenic toxicology: five questions. Chem Res Toxicol 2006;19:1–15.
- [84] Paul DS, Hernandez ZA, Felecia SW, Adair BM, Dedina J, Matousek T, et al. Examination of the effects of arsenic on glucose homeostasis in cell culture and animal studies: development of a mouse model for arsenic-induced diabetes. Toxicol Appl Pharmacol 2007;222:305–14.
- [85] Gamble MV, Liu XH. Urinary creatinine and arsenic metabolism. Environ Health Perspect 2005;113:A442-4.
- [86] Gerhardt RE, Hudson JB, Rao RN, Sobel RE. Chronic renal-insufficiency from cortical necrosis induced by arsenic poisoning. Arch Intern Med 1978;138:1267–9.
- [87] Jana K, Jana S, Samanta PK. Effects of chronic exposure to sodium arsenite on hypothalamo-pituitarytesticular activities in adult rats: possible an estrogenic mode of action. Reprod Biol Endocrinol 2006;4:9.
- [88] Waalkes MP, Ward JM, Liu J, Diwan BA. Transplacental carcinogenicity of inorganic arsenic in the drinking water: induction of hepatic, ovarian, pulmonary, and adrenal tumors in mice. Toxicol Appl Pharmacol 2003;186:7–17.

- [89] Pant N, Murty RC, Srivastava SP. Male reproductive toxicity of sodium arsenite in mice. Human Exp Toxicol 2004;23:399–403.
- [90] Sarkar M, Ray Chaudhuri G, Chattopadhyay A, Biswas NM. Effect of sodium arsenite on spermatogenesis, plasma gonadotrophins and testosterone in rats. Asian J Androl 2003;5:27–31.
- [91] Chang SI, Jin B, Youn P, Park C, Park JD, Ryu DY. Arsenic-induced toxicity and the protective role of ascorbic acid in mouse testis. Toxicol Appl Pharmacol 2007;218:196–203.
- [92] Pant N, Kumar R, Murthy RC, Srivastava SP. Male reproductive effect of arsenic in mice. Biometals 2001;14:113–17.
- [93] Sarkar M, Biswas NM, Ghosh D. Effect of sodium arsenite on testicular and 3beta and 17beta hydroxysteroid dehydrogenase activation in albino rats: dose and duration dependent response. Med Sci Res 1991;19:789–93.
- [94] Waalkes MP, Keefer LK, Diwan BA. Induction of proliferate lesions of the uterus, testes, and liver in Swiss mice given repeated injections of sodium arsenate: possible estrogenic mode of action. Toxicol Appl Pharmacol 2000;166:24–35.
- [95] Sarkar R, Mohanakumar KP, Chowdhury M. Effects of an organophosphate pesticide, quinalphos, on the hypothalamogonadal axis in adult male rats. J Reprod Fertil 2000;118:29–38.
- [96] Blanco-Rodriguez J, Martinez-Garcia C. Apoptosis pattern by oestradiol treatment of the seminiferous epithelium of the adult rat. J Reprod Fertil 1997;110:61–70.
- [97] National Research Council Board on Environmental Studies and Toxicology. Arsenic in drinking water. Washington, DC: Natl Acad Press; 1999. Committee on Toxicology, Sub-committee on Arsenic in Drinking Water.
- [98] Morales KH, Ryan L, Kuo TL, Wu MM, Chen CJ. Risk of internal cancers from arsenic in drinking water. Environ Health Perspect 2000;108:655–61.
- [99] Alonso FT, Garmendia ML, Bogado ME. Increased skin cancer mortality in Chile beyond the effect of ageing: temporal analysis 1990 to 2005. Acta Dermatol Venereol 2010;90:141–6.
- [100] Meliker JR, Slotnick MJ, Avruskin GA, Schottenfeld D, Jacquez GM, Wilson ML, et al. Lifetime exposure to arsenic in drinking water and bladder cancer: a population-based case control study in Michigan, USA. Cancer Causes Control 2010;21:745–57.
- [101] Chen CL, Chiou HY, Hsu LI, Hsueh YM, Wu MM, Wang YH, et al. Arsenic in drinking water and risk of urinary tract cancer: a follow-up study from northeastern Taiwan. Cancer Epidemiol Biomarkers Prev 2010;19:101–10.
- [102] Hughes MF. Arsenic toxicity and potential mechanisms of action. Toxicol Lett 2002;133:1-16.
- [103] Ishinishi N, Yamamoto A, Hisanaga A, Inamasu T. Tumorigenicity of arsenic trioxide to the lung in Syrian golden hamsters by intermittent instillations. Cancer Lett 1983;21:141–7.
- [104] Pershagen G, Nordberg G, Bjorklund NE. Carcinomas of the respiratory tract in hamsters given arsenic trioxide and/or benzo(a)pyrene by the pulmonary route. Environ Res 1984;34:227–41.
- [105] Pershagen G, Bjorklund NE. On the pulmonary tumorigenicity of arsenic trisulfide and calcium arsenate in hamsters. Cancer Lett 1985;27:99–104.
- [106] Yamamoto A, Hisanaga A, Ishinishi N. Tumorigenicity of inorganic arsenic compounds following intratracheal instillations to the lungs of hamsters. Int J Cancer 1987;40:220–3.
- [107] Yang C, Frenkel K. Arsenic-mediated cellular signal transduction, transcription factor activation, and aberrant gene expression: implications in carcinogenesis. J Environ Pathol Toxicol Oncol 2002;21:331–42.
- [108] Kinoshita A, Wanibuchi H, Wei M, Yunoki T, Fukushima S. Elevation of 8-hydroxydeoxyguanosine and cell proliferation via generation of oxidative stress by organic arsenicals contributes to their carcinogenicity in the rat liver and bladder. Toxicol Appl Pharmacol 2007;221:295–305.

- [109] Suzuki S, Arnold LL, Pennington KL, Kakiuchi-Kiyota S, Cohen SM. Effects of co-administration of dietary sodium arsenite and an NADPH oxidase inhibitor on the rat bladder epithelium. Toxicology 2009;261:41–6.
- [110] Simeonova PP, Wang S, Toriuma W, Kommineni V, Matheson J, Unimye N, et al. Arsenic mediates cell proliferation and gene expression in the bladder epithelium: association with activating protein-1 transactivation. Cancer Res 2000;60:3445–53.
- [111] Bates MN, Smith AH, Cantor KP. Case-control study of bladder cancer and arsenic in drinking water. Am J Epidemiol 1995;141:523–30.
- [112] Hays AM, Srinivasan D, Witten ML, Carter DE, Lantz RC. Arsenic and cigarette smoke synergistically increase DNA oxidation in the lung. Toxicol Pathol 2006;34:396–404.
- [113] Keyse SM, Tyrell RM. Heme oxygenase is the major 32-kDa stress protein induced in human skin fibroblasts by UVA radiation, hydrogen peroxide and sodium arsenite. Proc Natl Acad Sci USA 1989;86:99–103.
- [114] Brown IR, Rush SJ. Induction of a "stress" protein in intact mammalian organs after the intravenous administration of sodium arsenite. Biochem Biophys Res Commun 1984;120:150–5.
- [115] Liu SX, Athar M, Lippai I, Waldren C, Hei TK. Induction of oxygen radicals by arsenic: implications for mechanism of genotoxicity. Proc Natl Acad Sci USA 2001;98:1643–8.
- [116] Lynn S, Gurr JR, Lai HT, Jan KY. NADH oxidase activation is involved in arsenite-induced oxidative DNA damage in human vascular smooth muscle cells. Circ Res 2000;86:514–19.
- [117] Corsini E, Asti L, Viviani B, Marinovich M, Galli CL. Sodium arsenate induces overproduction of interleukin-1 alpha in murine keratinocytes: role of mitochondria. J Invest Dermatol 1999;113:760–5.
- [118] Jing YK, Dai J, Chalmers-Redman RME, Tatton WG, Waxman S. Arsenic trioxide selectively induces acute promyelocytic leukemia cell apoptosis via a hydrogen peroxide-dependent pathway. Blood 1999;94:2102–11.
- [119] Wang TS, Kuo CF, Jan KY, Huang HM. Arsenite induces apoptosis in Chinese hamster ovary cells by generation of reactive oxygen species. J Cell Physiol 1996;169:256–68.
- [120] Garcia-Chavez E, Santamaria A, Diaz-Barriga F, Mandeville P, Bertha I, Juarez M, et al. Arsenite-induced formation of hydroxyl radical in the striatum of awake rats. Brain Res 2003;976:82–9.
- [121] Yamanaka K, Ohba H, Hasegawa A, Sawamura R, Okada S. Mutagenicity of dimethylated metabolites of inorganic arsenics. Chem Pharm Bull 1989;37:2753–6.
- [122] Yamanaka K, Hoshino M, Okamoto M, Sawamura R, Hasegawa A, Okada S. Induction of DNA damage by dimethylarsine, a metabolite of inorganic arsenics, is for the major part likely due to its peroxyl radical. Biochem Biophys Res Commun 1990;168:58–64.
- [123] Rana SVS. Metals and apoptosis: recent developments. J Trace Elem Med Biol 2008;22:262-84.
- [124] Liu SX, Davidson MM, Tang X, Walker WF, Athar M, Ivanov V, et al. Mitochondrial damage mediates genotoxicity of arsenic in mammalian cells. Cancer Res 2005;65:3236–42.
- [125] Wang TS, Huang H. Active oxygen species are involved in the induction of micronuclei by arsenite in XRS-5 cells. Mutagenesis 1994;9:253–7.
- [126] Lee TC, Ho IC. Differential cytotoxic effects of arsenic on human and animal cells. Environ Health Perspect 1994;3:101-5.
- [127] Styblo M, Serves SV, Cullen WR, Thomas DJ. Comparative inhibition of yeast glutathione reductase by arsenicals and arsenothiols. Chem Res Toxicol 1997;10:27–33.
- [128] Lin S, Cullen WR, Thomas DJ. Methylarsenicals and arsinothiols are potent inhibitors of mouse liver thioredoxin reductase. Chem Res Toxicol 1999;12:924–30.
- [129] Arrigo AP. Gene expression and the thiol redox state. Free Radic Biol Med 1999;27:936-44.
- [130] Powis G, Mustacich D, Coon A. The role of the redox protein thioredoxin in cell growth and cancer. Free Radic Biol Med 2000;29:312–22.

- [131] Ahmad S, Kitchin KT, Cullen WR. Arsenic species that cause release of iron from ferritin and generation of activated oxygen. Arch Biochem Biophys 2000;382:195–202.
- [132] Bode AM, Dong Z. The paradox of arsenic: molecular mechanisms of cell transformation and chemotherapeutic effects. Crit Rev Oncol Hematol 2002;24:5–24.
- [133] Druwe IL, Vaillancourt RR. Influence of arsenate and arsenite on signal transduction pathways: an update. Arch Toxicol 2010;84:585–96.
- [134] Kumagi Y, Sumi D. Arsenic: signal transduction, transcription factor and biotransformation involved in cellular response and toxicity. Annu Rev Pharmacol Toxicol 2007;47:243–62.
- [135] Leonard SS, Harris GK, Shi X. Metal-induced oxidative stress and signal transduction. Free Radic Biol Med 2004;37:1921–42.
- [136] Ghosh J, Das J, Manna P, Sil PC. Taurine prevents arsenic-induced cardiac oxidative stress and apoptotic damage: role of NF-kappaB, p38 and JNK MAPK pathway. Toxicol Appl Pharmacol 2009;240:73–87.
- [137] Das J, Ghosh J, Manna P, Sil PC. Protective role of taurine against arsenic-induced mitochondria-dependent hepatic apoptosis via the inhibition of PKCδ-JNK pathway. PLOS ONE 2010;5:e12602.
- [138] Liu Y, Guyton KZ, Gorospe M, Xu Q, Lee JC, Holbrook NJ. Differential activation of ERK, JNK/SAPK and p38/CSBP/RK MAP kinase family members during the cellular response to arsenite. Free Radic Biol Med 1996;21:771–81.
- [139] Wu W, Jaspers I, Zhang W, Graves LM, Samet JM. Role of Ras in metal induced EGF receptor signaling and NF-kappaB activation in human airway epithelial cells. Am J Physiol Lung Cell Mol Physiol 2002;282:L1040-L1048.
- [140] Yamanaka K, Hayashi H, Tachikawa M, Kato K, Hasegawa A, Oku N, et al. Metabolic methylation is a possible genotoxicity-enhancing process of inorganic arsenics. Mutat Res 1997;394:95–101.
- [141] Yamanaka K, Takabayashi F, Mizoi M, An Y, Hasegawa A, Okada S. Oral exposure of dimethylarsinic acid, a main metabolite of inorganic arsenics, in mice leads to an increase in 8-oxo-2'-deoxyguanosine level, specifically in the target organs for arsenic carcinogenesis. Biochem Biophys Res Commun 2001;287:66-70.
- [142] Wauson EM, Langan AS, Vorce RL. Sodium arsenite inhibits and reverses expression of adipogenic and fat cell-specific genes during in vitro adipogenesis. Toxicol Sci 2002;65:211–19.
- [143] Tseng CH. The potential biological mechanisms of arsenic-induced diabetes mellitus. Toxicol Appl Pharmacol 2004;197:67–83.
- [144] Pal S, Chatterjee AK. Prospective protective role of melatonin against arsenic-induced metabolic toxicity in Wistar rats. Toxicology 2005;208:25–33.
- [145] Izquierdo-Vega JA, Soto CA, Sanchez-Peña LC, De Vizcaya-Ruiz A, Del Razo LM. Diabetogenic effects and pancreatic oxidative damage in rats subchronically exposed to arsenite. Toxicol Lett 2006;160:135–42.
- [146] Paul DS, Harmon AW, Devesa V, Thomas DJ, Stýblo M. Molecular mechanisms of the diabetogenic effects of arsenic: inhibition of insulin signaling by arsenite and methylarsonous acid. Environ Health Perspect 2007;115:734–42.
- [147] Feuerstein GZ, Young PR. Apoptosis in cardiac diseases: stress- and mitogen-activated signaling pathways. Cardiovasc Res 2000;45:560–9.
- [148] Schulze-Osthoff K, Ferrari D, Riehemann K, Wesselborg S. Regulation of NF-κB activation by MAP kinase cascades. Immunobiology 1997;198:35–49.
- [149] Cavigelli M, Li WW, Lin A, Su B, Yoshioka K, Karin M. The tumor promoter arsenite stimulates AP-1 activity by inhibiting a JNK phosphatase. EMBO J 1996;15:6269–79.
- [150] Luster MI, Simeonova PP. Arsenic and urinary bladder cell proliferation. Toxicol Appl Pharmacol 2004;198:419–23.

- [151] Simeonova PP, Wang S, Hulderman T, Luster MI. c-Src-dependent activation of the epidermal growth factor receptor and mitogen-activated protein kinase pathway by arsenic. J Biol Chem 2002;277:2945–50.
- [152] Barchowsky A, Roussel RR, Klei LR, James PE, Ganju N, Smith KR, et al. Low levels of arsenic trioxide stimulate proliferative signals in primary vascular cells without activating stress effector pathways. Toxicol Appl Pharmacol 1999;159:65–75.
- [153] Andrew AS, Mason RA, Memoli V, Duell EJ. Arsenic activates EGFR pathway signaling in the lung. Toxicol Sci 2009;109:350–7.
- [154] Filippova M, Duerksen-Hughes PJ. Inorganic and dimethylated arsenic species induce cellular p53. Chem Res Toxicol 2003;16:423–31.
- [155] Zhu J, Chen Z, Lallemand-Breitenbach V, de The H. How acute promyelocytic leukemia revived arsenic. Nat Rev Cancer 2002;2:705–14.
- [156] Bachleitner-Hofmann T, Kees M, Gisslinger H. Arsenic trioxide: acute promyelocytic leukemia and beyond. Leuk Lymphoma 2002;43:1535–40.
- [157] Hyun Park W, Hee Cho Y, Won Jung C, Oh Park J, Kim K, Hyuck Im Y, et al. Arsenic trioxide inhibits the growth of A498 renal cell carcinoma cells via cell cycle arrest or apoptosis. Biochem Biophys Res Commun 2003;300:230–5.
- [158] Jiang T, Huang Z, Chan JY, Zhang DD. Nrf2 protects against As(III)-induced damage in mouse liver and bladder. Toxicol Appl Pharmacol 2009;250:8–14.
- [159] Wang XJ, Sun Z, Chen W, Li Y, Villeneuve NF, Zhang DD. Activation of Nrf2 by arsenite and monomethylarsonous acid is independent of Keap1-C151: enhanced Keap1-Cul3 interaction. Toxicol Appl Pharmacol 2008;230:383–9.
- [160] Das J, Roy A, Sil PC. Mechanism of the protective action of taurine in toxin and drug induced organ pathophysiology and diabetic complications: a review. Food Funct 2012;3:1251–64.
- [161] Schaffer S, Azuma J, Takahashi K, Mozaffari M. Why is taurine cytoprotective? Adv Exp Med Biol 2003;526:307–21.
- [162] Hagar HH. The protective effect of taurine against cyclosporine A-induced oxidative stress and hepatotoxicity in rats. Toxicol Lett 2004;151:335–43.
- [163] Das J, Ghosh J, Manna P, Sinha M, Sil PC. Taurine protects rat testes against NaAsO2-induced oxidative stress and apoptosis via mitochondrial dependent and independent pathways. Toxicol Lett 2009;187:201–10.
- [164] Nava-Hernandez MP, Hauad-Marroquin LA, Bassol-Mayagoitia S, García-Arenas G, Mercado-Hernandez R, Echavarri-Guzman MA, et al. Lead-, cadmium-, and arsenic-induced DNA damage in rat germinal cells. DNA Cell Biol 2009;28:241–8.
- [165] Fouad AA, Al-Sultan AI, Yacoubi MT. Coenzyme Q10 counteracts testicular injury induced by sodium arsenite in rats. Eur J Pharmacol 2011;655:91–8.
- [166] Roy A, Manna P, Sil PC. Prophylactic role of taurine on arsenic mediated oxidative renal dysfunction via MAPKs/NF-kappaB and mitochondria dependent pathways. Free Radic Res 2009;43:995–1007.
- [167] Das J, Ghosh J, Manna P, Sinha M, Sil PC. Arsenic-induced oxidative cerebral disorders: protection by taurine. Drug Chem Toxicol 2009;32:93–102.
- [168] Reddy PS, Rani GP, Sainath SB, Meena R, Supriya Ch. Protective effects of N-acetylcysteine against arsenic-induced oxidative stress and reprotoxicity in male mice. J Trace Elem Med Biol 2011;25:247–53.
- [169] Modi M, Kaul RK, Gurusamy M, Kannan GM, Flora SJS. Co-administration of zinc and n-acetylcysteine prevents arsenic-induced tissue oxidative stress in male rats. J Trace Elem Med Biol 2006;20:197–204.
- [170] Flora SJS, Bhadauria S, Kannan GM, Singh N. Arsenic induced oxidative stress and the role of antioxidant supplementation during chelation: a review. J Environ Biol 2007;28:333–47.

- [171] Mahabady MK, Varzi HN, Ranjbar R, Mehrzadi S. Melatonin and vitamin E protects against sodium arsenite-induced skeletal malformations in rats. American-Eurasian J Toxicol Sci 2011;3:184–9.
- [172] Tan DX, Manchester LC, Reiter RJ, Qi WB, Karbownik M, Calro JR. Significance of melatonin in antioxidative defense system: reactions and products. Biol Signals Rep 2000;9:137–59.
- [173] Reiter RJ, Tan DX. Melatonin: a novel protective agent against oxidative injury of the ischemic/reperfused heart. Cardiovasc Res 2003;58:10–19.
- [174] Hardeland R, Reiter RJ, Poeggeler B, Tan DX. The significance of the metabolism of the neurohormone melatonin: antioxidative protection and formation of bioactive substances. Neurosci Biobehav Rev 1993;17:347-57.
- [175] Cuzzocrea S, Reiter RJ. Pharmacological action of melatonin in acute and chronic inflammation. Curr Top Med Chem 2002;2:153–65.
- [176] Reed LJ. Multienzyme complexes. Acc Chem Res 1974;7:40-6.
- [177] Carreau JP. Biosynthesis of lipoic acid via unsaturated fatty acids. Methods Enzymol 1979;62:152-8.
- [178] Hatch RC, Clark JD, Jain AV. Use of thiols and thiosulphate for treatment of experimentally induced acute arsenite toxicosis in cattle. Am J Vet Res 1978;39:1411–14.
- [179] Packer L, Witt EH, Tritschler HJ. Alpha lipoic acid as a biological antioxidant. Free Radic Biol Med 1995;19:227-50.
- [180] Samuel S, Kathirvel R, Jayavelu T, Chinnakkannu P. Protein oxidative damage in arsenic induced rat brain: influence of dl-α-lipoic acid. Toxicol Lett 2005;155:27–34.
- [181] Volate SR, Davenport DM, Muga SJ, Wargovich MJ. Modulation of aberrant crypt foci and apoptosis by dietary herbal supplements (quercetin, curcumin, silymarin, ginseng and rutin). Carcinogenesis 2005;26:1450–6.
- [182] Soria EA, Eynard AR, Quiroga PL, Bongiovanni GA. Differential effects of quercetin and silymarin on arsenite-induced cytotoxicity in two human breast adenocarcinoma cell lines. Life Sci 2007;81:1397–402.
- [183] Bongiovanni GA, Soria EA, Eynard AR. Effects of the plant flavonoids silymarin and quercetin on arsenite-induced oxidative stress in CHO-K1 cells. Food Chem Toxicol 2007;45:971–6.
- [184] Mishra D, Flora SJS. Quercetin administration during chelation therapy protects arsenic-induced oxidative stress in mice. Biol Trace Elem Res 2008;122:137–47.
- [185] Ghosh S, Dungdung SR, Chowdhury ST, Mandal AK, Sarkar S, Ghosh D, et al. Encapsulation of the flavonoid quercetin with an arsenic chelator into nanocapsules enables the simultaneous delivery of hydrophobic and hydrophilic drugs with a synergistic effect against chronic arsenic accumulation and oxidative stress. Free Radic Biol Med 2011;51:1893–902.
- [186] Ghosh D, Ghosh S, Sarkar S, Ghosh A, Das N, Das Saha K, et al. Quercetin in vesicular delivery systems: Evaluation in combating arsenic-induced acute liver toxicity associated gene expression in rat model. Chem Biol Interact 2010;186:61–71.
- [187] Manna P, Sinha M, Sil PC. Arjunolic acid, a triterpenoid saponin, ameliorates arsenic-induced cytotoxicity in hepatocytes. Chem Biol Interact 2007;170:187–200.
- [188] Masoko P, Mdee LK, Mampuru LJ, Eloff JN. Biological activity of two related triterpenes isolated from *Combretum nelsonii* (Combretaceae) leaves. Nat Prod Res 2008;22:1074–84.
- [189] Djoukeng JD, Abou-Mansour E, Tabacchi R, Tapondjou AL, Bouda H, Lontsi D. Antibacterial triterpenes from *Syzygium guineense* (Myrtaceae). J Ethnopharmacol 2005;101:283–6.
- [190] Manna P, Sinha M, Sil PC. Protective role of arjunolic acid in response to streptozotocin-induced type-I diabetes via the mitochondrial dependent and independent pathways. Toxicology 2009;257:53–63.
- [191] Bhakuni RS, Shukla YN, Tripathi AK, Prajapati V, Kumar S. Insect growth inhibitor activity of arjunolic acid isolated from *Cornus capitata*. Phytother Res 2002;16:68–70.

- [192] Ghosh J, Das J, Manna P, Sil PC. Arjunolic acid, a triterpenoid saponin, prevents acetaminophen (APAP) induced liver and hepatocytes injury via the inhibition of APAP bioactivation and JNK-mediated mito-chondrial protection. Free Radic Biol Med 2010;48:535–53.
- [193] Ghosh J, Sil PC. Arjunolic acid: a new multifunctional therapeutic promise of alternative medicine. Biochimie 2013;95:1098–109.
- [194] Manna P, Sinha M, Sil PC. Protection of arsenic-induced hepatic disorder by arjunolic Acid. Basic Clin Pharmacol Toxicol 2007;101:333–8.
- [195] Sumitra M, Manikandan P, Kumar DA, Arutselvan N, Balakrishna K, Manohar BM, et al. Experimental myocardial necrosis in rats: role of arjunolic acid on platelet aggregation, coagulation and antioxidant status. Mol Cell Biochem 2001;224:135–42.
- [196] Florea AM, Yamoah EN, Dopp E. Intracellular calcium disturbances induced by arsenic and its methylated derivatives in relation to genomic damage and apoptosis induction. Environ Health Perspect 2005;113:659–64.

This page intentionally left blank

9

Arsenic-Induced Mutagenesis and Carcinogenesis: A Possible Mechanism

Smarajit Maiti

POST GRADUATE DEPARTMENT OF BIOCHEMISTRY & BIOTECHNOLOGY, CELL & MOLECULAR THERAPEUTICS LABORATORY, ORIENTAL INSTITUTE OF SCIENCE AND TECHNOLOGY, MIDNAPORE, VIDYASAGAR UNIVERSITY, MIDNAPORE, INDIA

CHAPTER OUTLINE

9.1 Arsenic, a Potent Mutagen and Carcinogen	
9.2 Epidemiological Perspectives of Arsenic-Induced Human Cancers	
9.3 Arsenic-Associated Metabolism and Carcinogenesis in Animal Models	
9.4 Human Arsenic Carcinogenesis	
9.5 Conclusions and Future Directions	
Acknowledgments	
References	

9.1 Arsenic, a Potent Mutagen and Carcinogen

Several international agencies and government organizations recognize arsenic as one of most toxic environmental pollutants. The database of the Integrated Risk Assessment System (IRIS) of the Environmental Protection Agency (EPA) suggests that chronic and oral exposure of arsenic may result in hyperpigmentation, keratosis, and finally cancers of several organs in the body [1]. The EPA and the disease registry division pointed out from the epidemiological data that arsenic manifests carcinogenic growth in the liver, lungs, bladder, skin, and several other organs in affected individuals [2]. In line with this finding, the American Cancer Society (ACS) indicated that chronic environmental or occupational arsenic exposure may initiate free radical-associated genotoxicity, oxidative tissue degeneration, and cancerous growth [3]. The Agency for Toxic Substances and Disease Registry (ATSDR) of the Department of Health and Human Services (DHHS) has categorized inorganic arsenic as a potent human carcinogen [4]. Recently, environmental arsenic exposure has been a great concern in a number of countries. The International Agency for Research on Cancer (IARC) also suggests the potent carcinogenic

nature of inorganic arsenic due to the consumption of arsenic-contaminated drinking water by a large population in certain areas worldwide [5,6].

The International Groundwater Resources Assessment Centre (IGRAC) suggests that exposure to inorganic arsenic (iAs), a potent human carcinogen and toxicant, occurs mainly via drinking water and industrial emissions. Long-term exposure to arsenic may result in cancerous growth in several organs [7]. The World Health Organization (WHO) has stated that arsenic is widely distributed throughout Earth's crust, and mixes with groundwaters, which are consumed by a vast population in several countries [8]. In addition to arsenic, other metals are recognized to exert a toxic impact on different metabolic processes. The European Confederation of Iron and Steel Industries and the International Council of Mining and Metals suggested that environmental and/or occupational exposure to arsenic and some other metals interferes with several genetic events, and manifests tissue/organ carcinogenesis in humans [9,10].

Moreover, the epidemiological data highlight that the adverse effects of arsenic in reproductive, pregnancy, and gestational outcomes of women are also evident with long-term health anomalies in their children [11]. Animal models to study iAs-induced cancer in skin, a major site for effects in humans, have recently been suggested by a few investigators [12–14] and this is basically in agreement with the report of the National Research Council (NRC) [15]. Gestational arsenic exposure in rodents is known to be associated with tumors of the liver, ovary, lung, and with adrenal and significant pre-neoplastic lesions in the uterus and oviduct of the offspring [16,17].

9.2 Epidemiological Perspectives of Arsenic-Induced Human Cancers

Arsenic, a human carcinogen, potentially affects ≈ 160 million people worldwide via exposure to it through drinking contaminated water, industrial processes, and the generation of power from coal. It is also widely used in agriculture and was formerly used extensively in medicine [18]. Exposure to arsenic may also occur through the ingestion of foodstuffs containing inorganic and organic arsenicals. It is widely established that the liver and lungs are the main target organs for arsenic-related carcinogenesis [19,20]. Arsenic induces tumors in several organs, especially the lung, which may exhibit features like squamous cell-type specificity and high tumor incidence even among non-smokers [21].

A study conducted in Michigan, USA, revealed an association between arsenic levels and several diseases outcomes, which include different forms of cancer, diseases of the circulatory and respiratory system, diabetes mellitus, and kidney and liver diseases. Skin or bladder cancers have been widely reported among arsenic-exposed persons [22–24]. Lung cancer is observed in smelter workers [25], while males are considerably more susceptible than females, and low body weight, presumably a result of poor nutrition, is predisposing [26].

A strong association between chronic arsenic exposure and various adverse health effects, including cardiovascular diseases, neurological defects, and cancers is now widely known [15,24,27,28]. Though the final disease outcomes are well defined, the processes involved in arsenic carcinogenesis remain an enigma. A variety of mechanisms, both genotoxic and

non-genotoxic, have been proposed to explain the carcinogenicity of arsenic at the cellular and molecular levels [29,30]. During the last decade, a number of mechanisms have been proposed for arsenic-induced carcinogenesis but the elucidation of a definite mechanism is still awaited. To protect the population from long-term arsenic exposure, the EPA lowered the maximum concentration limit (MCL) for arsenic from 50 to $10 \mu g/L$. The acceptable level for maximum arsenic concentrations in safe drinking water was also reduced to $10 \mu g/L$ [31].

9.3 Arsenic-Associated Metabolism and Carcinogenesis in Animal Models

Arsenic metabolism is diversified in primates, rodents, and humans. Certain primates do not methylate inorganic arsenic [32]. Sometimes, a dissimilar metabolic profile of iAs is shown in different variants of even the same animal species. The polymorphic nature of the methyltransferase gene in primates [33] and humans [34] is one of the determining factors of the differential metabolism of inorganic arsenicals (Figure 9–1). This indicates that the same pattern of arsenic exposure may have varied cytotoxic effects in sex- and age-matched persons with similar socioeconomic/demographic and nutritional profiles, only because of their interindividual genetic variations [35] (Figure 9–2). Inorganic arsenic and its several methylated compounds are described as cytotoxic and genotoxic to different degrees [36]. Among those compounds, in terms of severity, dimethylarsinic acid (DMA) has been shown to exhibit a wide range of cytotoxicity, mitotic arrest, and tetraploid induction [37]. Thus, DMA has been used in a number of animal models to study arsenic-induced tumorigenesis and carcinogenesis. Arsenical compounds show less efficacy towards mutagenicity in animal experiments alone [38,39]. Monomethylarsonic acid (MMA) has been detected in the urine in human cases of carcinogenesis [40]. Arsenical metabolites produced during arsenic metabolism are shown in Figure 9–1.



FIGURE 9–1 Predominant arsenical species found in rats and humans. Inorganic arsenic is metabolized differentially in rats and humans. The metalloid arsenic is dominant in 3+ and 5+ forms in organic and inorganic forms. Oxidation state is one of the determinants of the toxic potential of As in biological systems.



Factors associated with the qualitative and quantitative differential nature of arsenic toxicity/mutagenesis/carcinogenesis

FIGURE 9–2 Outline of the differential toxicity manifestations of arsenical compounds. A large number of metabolic pathways are affected by arsenic. Almost all important metabolic regulations and rate limiting steps are influenced by arsenic. Various factors related to arsenic exposure are the determinants of this differential manifestation.

9.3.1 Arsenic Metabolism and Carcinogenesis

A differential degree of arsenic toxicity is manifested by organic and inorganic arsenicals of different valency states. DMA has been shown, with promotional activity in a rat liver carcinogenesis model, to increase oxidative DNA damage (8-hydroxy-2'-deoxyguanosine (8-OHdG)). It induces placental glutathione S-transferase (GST- ρ isoforms +) and increases the activity of ornithine decarboxylase (ODC), a rate limiting enzyme of polyamine biosynthesis. These markers are indicative of cell proliferation [41–43]. GST- ρ is regarded as a reliable tumor marker for experimental rat hepatocarcinogenesis. GST- ρ overexpression in several hepatic foci in experimental animals suggests that the essential enhancer element GPE1 is the responsible agent in the *GST-\rho* gene. Further binding of nuclear factor-erythroid 2-related factor 2 (Nrf2/MafK) heterodimer on the GPE1 element activates gene expression in preneoplastic lesions and in hepatomas. Arsenic-induced and redox-regulated activation of Nrf2 has also been reported [44].

The enzyme ODC catalyzes the formation of putrescine. It is the rate limiting step of polyamine synthesis [45]. Polyamines have been associated with cell growth and cancer. Specific oncogenes and tumor-suppressor genes regulate polyamine metabolism. Inhibition of polyamine synthesis is shown in implementing chemopreventive strategies against different types of cancers [46]. It may also be exploited against arsenic-associated carcinogenesis. All methylated metabolites of inorganic arsenicals like MMA, DMA, and trimethylarsine oxide (TMAO, which are predominant in rats) are able to induce hydroxyl-radical productions and oxidative stress-related tissue degeneration. Oral or intratracheal DMA ingestion was not able to promote lung carcinogenesis [47,48]. But the iAs with the concomitant exposure of UV light was able to promote skin tumor [49,50]. The co-promotional activity of arsenic may also be applicable in humans. Methylated species of arsenic result in increased efficiency in terms of tumorigenesis in comparison to iAs. Studies suggest that exposure to DMA may promote tumors in the urinary, bladder, liver, skin, and kidney along with other potent organic carcinogens or UV irradiation [51-55]. DMA has been shown to have a variety of carcinogenic effects in F344 male rats [56,57]. DMA in relation to long-term exposure may act as a primary carcinogen (rather than a co-carcinogen) in rats, inducing carcinoma and papilloma in the urinary bladder [58,59], and lymphoma and sarcoma in mice [60]. The inner lining of the epithelium of several organs like intestines, lungs, breasts, and skin may generate adenocarcinoma, squamous cell carcinoma, and anaplastic carcinoma. Sarcoma develops in the connective tissue such as tendons, nerve tissues, and blood vessels. Examples include alveolar soft part sarcoma and fibrosarcoma, while the lymphoma develops in lymphatic organs and vessels in the body. This suggests that a certain level of arsenic metabolites may act as an independent carcinogen.

Discrepancies arise in the case of humans where exposure to methylated arsenic species is unlikely since arsenic exposure takes place mostly through drinking water, which largely contains iAs. Thus, it is likely that the pattern of toxicity, tumerogensis, and carcinogenesis in humans may not be similar to that of an experimental animal model (Figure 9–2). Making things more complex, iAs predominantly converts to organic arsenic in certain types of cells. The qualitative and quantitative features of arsenic methylation are not ubiquitous. Although it is generally accepted that methylated arsenicals have a deleterious impact on biological systems [61]. For example, human epidermal keratinocytes respond differentially against different inorganic and methylated arsenicals of varied valency state [61]. It is demonstrated that arsenite and UV irradiation together increase the incidence of carcinomas whereas only UV irradiation of cells produce papillomas in mouse skin. This increase may be due to the inhibition of DNA repair and/or distortion of tumor suppression gene *p*53 functions and up-regulations of cyclin D1 action [62–64].

Chronic exposure to drinking water mixed with DMA in a male rat led to a dose-related development of tumors in the bladder, but not in other organs [65]. Administration of DMA to F344 rats and 36C3F1 mice of both sexes for 2 years led to no tumor formation in mice but rats developed urinary bladder tumors. Female rats were more sensitive to these effects. On the other hand, MMA exposure to the same strains of animals resulted in no tumor formation [65].

These observations suggest the more toxic nature of DMA over MMA, which is also diverse in different animal models.

9.3.2 Arsenic Metabolism and Free Radical Influences

The genetic basis of arsenic-induced mutagenesis and tumorigenesis may be attributed to direct free radical attack on DNA strands resulting in damage. Carcinogenic effects of DMAs are the result of metabolism of DMAs(V) to dimethyl arsinous acid DMAs(III) which induces oxidative stress in the cells followed by cytotoxicity [66]. Impairment of catalase and super-oxide dismutase (SOD)-associated DNA breakage and apoptotic cell death were noticed in intestinal epithelial cells of rats chronically exposed to iAs [67]. Hydrogen peroxide (H₂O₂)-associated free radicals are implicated in the high rate of DNA fragmentation in rodent lymphocyte and liver tissues of catalase knockout mice. Electron spin resonance with the spin-trap agent detects the generation of the hydroxyl radical (\cdot OH) via MMA(III) when H₂O₂ is present and it might thus be participating in the DNA damage mechanism [68]. These findings clearly indicate the importance of catalase in cellular protection from reactive oxygen species (ROS)-related damage (Figure 9–3).

Chronic arsenic exposure in rats also caused significant single-strand DNA damage in neuronal cell and lymphocytes as depicted by single cell comet assay. It is accelerated by an



FIGURE 9–3 Interactive roles of genetic and epigenetic changes by arsenic alone or after initiating oxidative responses may influence several important modes of gene expression and signal transduction processes at nuclear and/or cellular levels. These mechanisms finally lead to carcinogenesis.

increased production of ROS [69]. These results are of significance in suggesting that chronic arsenic exposure in humans may result in a significant DNA degradation pattern in a mixed white blood cell (WBC) population with significant decrease of several serum antioxidant components including uric acid [70]. Chronic arsenic exposure also leads to ROS-mediated, mitochondrial-driven, and caspase-dependent apoptosis in hepatic cells with a significant increase in glutathione disulfide (GSSG) levels and a decrease in glutathione reductase activity [71]. This suggests that oxidative stress-related mitochondrial instability is an important factor for arsenic-related apoptotic tissue degeneration. Arsenic-induced increase in 8-OHdG, a modified DNA base as reported in several studies, suggests that oxidative DNA damage is a major cause of arsenic-related mutagenesis [72]. MMA has also been reported to induce the preneoplastic lesions in rats via the production of ROS [73].

Arsenic-associated lung damage and carcinogenesis are predominant in arsenic-exposed individuals. A study describing synergism between tobacco smoke and arsenic exposures has demonstrated more pronounced cytotoxic effects, emphysema-like lesions, and inflammatory events in mice [74]. Arsenic and chromium in drinking water promote tumorigenesis in a mouse colitis-associated colorectal cancer model via a ROS-mediated Wnt/ β -catenin signaling pathway. Arsenic exposure increased nicotinaminde adenine dinucleotide phosphate-reduced, NADPH oxidase1 (NOX1) and the level of 8-OHdG suggesting ROS-related DNA damage [75]. The depletion of antioxidant enzymes, such as SOD and catalase, has been shown and linked to arsenic-related DNA and tissue damage. In addition, in vitro supplementation of SOD and catalase inhibited β -catenin expression and activity [75]. The study suggests that exposure to As(III) may promote colorectal cancer and tumorigenesis partially through the ROS-mediated β -catenin signaling pathway [75]. The β -catenin acts as an intracellular signaling molecule that acts through the Wnt pathway. It regulates the coordination of cell-cell adhesion and gene transcription. The study further revealed that mutations and/or overexpressions of β -catenin may result in multiple types of tumor and cancerous growth in different organs [76].

Arsenic significantly reduces mRNA expression of superoxide dismutase 2 (*SOD2*) genes possibly by decreasing the nullifications of superoxide anion radicals, leading to increased oxidative stress [77]. Dimethylarsinous acid (DMA(III)) has been reported for its urothelial toxicity and regenerative hyperplasia in arsenite methyltransferase (As3mt) knockout C57BL/6 mice [78]. DMA(V), on the other hand, does not induce hyperplasia in mice, suggesting urinary concentration of DMA(III) does not reach cytotoxic levels in DMA(V)-treated mice [79]. This further supports the important role of methyltransferase gene function in arsenic metabolism and tumorigenesis. The discovery of arsenite (+3 oxidation state) methyltransferase (As3MT) to be the enzyme responsible for arsenic methylation, genetic polymorphism, and the alternative splicing in relation to this gene regulation also contributes to the differential arsenic metabolism and toxicity [80]. Nevertheless, studies report inorganic arsenic to be carcinogenic in humans and mice. Adult offspring mice from an inorganic arsenic-exposed mother developed tumors of the lung, liver, adrenal, uterus, and ovary [81]. Male offspring had unusual testicular lesions, including rete testis (tubular meshwork in the hilum of the testicle that carries sperm), carcinomas, adenomas, and interstitial cell tumors. Overall, maternal consumption of MMA3+ during pregnancy in CD1 mice produced similar proliferative lesions in adult offspring arising from gestationally applied inorganic arsenic [81]. Chronic exposure to arsenite in rats resulted in a high rate of DNA fragmentation with postmitotic apoptotic cell death and a significant increase in free radical-associated oxidative stress and DNA strand breakage. A study relating to arsenic therapeutics suggested that DNA and tissue were protected following supplementation of vitamin B12 and folic acid [82]. This study demonstrated that antioxidative protection and the restoration of the methyl pool were helpful in protecting DNA from arsenic-induced damage. Arsenic(III) is also known to cause developmental toxicity in human embryonic stem cell-derived embryoid bodies and in *in vivo* mouse models [83]. Developmental cytotoxicities have been observed in human embryoid bodies (EBs). *In vivo* studies suggested that arsenic exposure may lead to a significant down-regulation of gene expression in all the three germ layers, which could be correlated with high mortality and visceral and skeletal defects in pups [83]. Overall, free radicals are extremely deleterious to biomolecules. Various arsenicals exert differential potency to generate free radicals. As explained, damage to the DNA base may result in aberrant gene expression and cancer.

9.3.3 Arsenic May Influence Drug Metabolizing Enzymes

Regulation of drug metabolizing enzymes is of significance as they modify several endogenous and exogenous xenobiotic compounds. Biotransformation/activation or excretion of any compound predicts its ability to be either a toxic or a non-toxic initiator or promoter (if found carcinogenic or co-carcinogenic) in nature. In this regard, phase I and phase II drug metabolizing enzymes are of relevant concern. An arsenic-induced significant increase in phase I drug metabolizing enzyme cytochrome P450 reductase (Cyt-P450) expression has been reported [84]. Arsenic may significantly impair the metabolism of different endogenous and exogenous compounds in rat by interfering with several drug-metabolizing enzymes like Cyt-P450 reductase and Cyt b5 groups, uridine diphosphate glucuronosyltransferase (UGT), aniline p-hydroxylase (APH), and aminopyrine N-demethylase (ANDM). As a result possible bioactivation and/ or excretion of several compounds may play a role in the carcinogenesis process [85].

Arsenic(III) can differentially modulate the catalytic efficacy of isoforms *Cyp1a1* and *Cyp1a2* (cytochrome P450 genes). These gene expressions are regulated by the modulation of aryl hydrocarbon receptor (AhR) sensitivity, which may also be influenced by As(III) compounds [86]. The altered P450 may regulate the metabolism of several exogenous or endogenous compounds by possible bioactivation and/or excretion [86]. Recent reports suggest that oncogenic transformations are a crucial factor in converting certain types of renal stem cells into those with cancerous phenotypic characteristic following chronic arsenite exposure [87]. Colony formation, cellular proliferation, and invasiveness are found to be associated with the altered matrix metalloprotease (MMP) activities. These results indicate multipotent stem cells being one of the targets for arsenic, which might lead to renal carcinogenesis [87]. Chronic arsenic exposure initiates an inflammatory response in the liver and also down-regulates the nuclear retinoid xenobiotic receptor α (RXR α), which is a heterodimeric pregnane xenobiotic receptor (PXR) and mediates the inactivation of the drug metabolizing enzyme *CYP3A* gene.

As a result, the lower expression in mRNA and corresponding amount of protein may impair the metabolism of several endobiotics and xenobiotics [88]. The alteration of expressions and activities of drug metabolizing enzymes by arsenic may influence the metabolism of endogenous or exogenously consumed substances. These features may influence the regular pattern of bioactivation, biotransformation, and/or excretion of several molecules. Bioactivation may sometimes result in procarcinogenic activation to form carcinogen.

9.3.4 Gut Microflora Influences Arsenic Toxicity and Tumorigenesis/Carcinogenesis

Gut microflora are known to be affected by arsenic metabolism when As is ingested through drinking water. Metagenomics studies of the gut microbiome in arsenic-intoxicated C57BL/6 mice and arsenic metabolites in their urine suggest that the gut microorganisms significantly interfere with intestinal arsenic metabolism. This is reflected in the urinary excretion of methylated and thiolated arsenical metabolites of those mice [89]. This suggests that infectious diseases and environmental exposure may be a new risk factor for individual susceptibility to carcinogenesis from environmental chemicals [89]. Under aerobic or anaerobic conditions, cecum tissue homogenates produce a certain amount of iAs MMA or DMA. These data suggest that cecum microflora utilize a high capacity methylation system that might contribute significantly to methylation-associated tumorigenesis and carcinogenesis of iAs in intact animals [90]. It is noticed that the contents of the rat small intestine or cecum reduce sodium arsenate to arsenite. In the pre-reduced media containing thioglycollate or cysteine in the absence of intestinal bacteria, sodium arsenate was reduced to arsenite. The rate of reduction of arsenate by the gut flora was increased by the presence of hydrogen sulfide or bile acids [91]. Besides the formation of monomethylarsonic acid (MMA(V)), highly toxic monomethylarsonous acid (MMA(III)) is also produced. This is an important finding of microbial thiolation leading to formation of monomethylmonothioarsonic acid (MMMTA(V)) [92].

Methylenetetrahydrofolate reductase (Mthfr) is an important enzyme that plays a key role in folate metabolism and methylation reactions. Transgenic mouse *Mthfr+* gene loci have been reported to promote more pronounced urinary excretion of arsenic than do the Mthfr– loci, suggesting the role of folate in arsenic metabolism [93]. In contrast, the Mthfr(-/-) mouse excretes significantly lower amounts of pentavalent arsenic when fed on a folate-deficient diet compared to a control diet. This suggests that active Mthfr and folate are important not only for arsenic detoxification, but also for its increased urinary elimination [93]. Adequate folate pool and S-adenosyl methionine (SAM) are very important determinants for the arsenic methylation process [94]. Supplementation of vitamin B12/folate-related restoration of the cellular methyl pool was effective in protecting hepatic tissue/DNA from arsenic-related toxicity [82]. Chronic arsenic exposure has been demonstrated to significantly decrease SAM and phosphatidylcholine (PC) with a concomitant increase in choline. This suggests that choline has a role as a methyl donor in arsenic-intoxicated conditions [95]. Hepatic impairment due to the combined effect of arsenic and other environmental factors may come into play in the case of abnormal metabolism of micronutrients like folate, SAM, and choline [95]. Inorganic arsenic causes the depletion of S-adenosylmethionine, the main cellular methyl donor, and represses the expression of the DNA methyltransferase genes *DNMT1* and *DNMT3A*, possibly as a consequence of these two complementary mechanisms. Long-term exposure to arsenic results in DNA hypomethylation leading to aberrant gene expression and carcinogenesis [96] (Figure 9–3). Intestinal epithelial cells are the first pass site of any ingested substance. Hence, the metabolism of ingested substances by intestinal indigenous metabolic machinery or by the gut microflora gives rise to additional factors in the manifestation of toxicity of arsenical compounds ingested.

9.3.5 Arsenic-Induced Carcinogenesis in Animal Models: Role of Signal Transduction

Tumorigenesis and carcinogenesis are multifactorial and multistep processes. They deal with the transformations of normal cells into malignant cells. Transformation may occur due to the induction of multiple cancer-specific heritable phenotypes triggered by genetic (natural mutational or DNA strand breakages and other damage) or epigenetic (non-inheritable DNA-associated modifications) events [97]. The outcome of the interactive nature of genetic and epigenetic factors on exposure to environmental and endogenous components may determine the extent of manifestations of cytotoxicity and tissue carcinogenesis [98]. For example, the toxicity outcome may alter the intracellular redox state. Imbalance in the cellular redox state and related complications may arise from impaired toxicant excretion and/or DNA damage/modifications and/or disrupted signal transduction processes [99]. One or more of these factors may result in tumorigenesis and/or carcinogenesis. The arsenic-associated animal tumorigenesis/carcinogenesis process may be mechanistically implicated via genetic, epigenetic, and signal transduction processes. There are numerous reports available in the literature suggesting that inorganic arsenic (iAs) can induce human carcinogenesis; a limited data supports the carcinogenic manifestation in animal experimental models [100]. However, mostly negative results for carcinogenicity were reported in mice, rat, hamster, rabbit, and certain other animals [101–104]; frequent tumorigenesis and distinct manifestations resembling carcinogenic features were demonstrated in arsenic-intoxicated rodent models. These investigations mostly confirmed co-mutagenic/co-carcinogenic effects rather than carcinogenicity of inorganic arsenical compounds. In most of these cases, arsenic exposure was associated with cell proliferation, differentiation, and signal transduction-related carcinogenesis, which were mostly demonstrated using cultured cell-based assay.

9.3.6 Arsenic Induces Cellular Transformation Signaling

Chronic arsenic exposure of rat-cultured liver epithelial cells induced malignant transformation with an increased expression of c-Myc and H-Ras products. The regulator gene *c-Myc* codes for a transcription factor. The mutated Myc is found in various cancers [105]. The arsenical compound phenylarsine oxide (PAO) has been shown to bind at a specific location in H-Ras and suggests mechanistically how this binding influences the pattern of H-Ras activation in cells [106]. Such findings help in the elucidation of the involvement of the Ras ligand in arsenic-associated carcinogenesis. Ras protein is a small GTPase family member from the Ras superfamily and is involved in intracellular signal transduction in cell growth and differentiation. A mutation in *ras* genes and auto-activated/generated Ras protein signaling leads to cancer. Out of three types of human *ras* genes (*H-Ras, N-Ras,* and *K-Ras* with subtypes generated from alternative splicing [107–109]), H-ras has been mostly shown to be involved in arsenic-associated carcinogenesis in animal models. Arsenic exposure in v-Ha-ras transgenic mice (an animal model for skin cancer) caused an increase in the number of papillomas as well as overexpression of growth factors, suggesting that they participate in arsenic-induced skin papilloma development [110].

Alteration in Ras expression has been reported with a concomitant suppression of MAPK/ ERK gene expression in perinatally arsenic-exposed mice, indicating the relation between Ras ligand signaling and the activities of different kinases [111]. Gestational arsenic exposure induces oxidative stress (as observed by the induction of the stress inducible gene heme oxygenase-1, HO-1) and overexpression of certain retrotransposon genes resulting in tumor formation in the offspring of certain (C3H) *Ha-Ras* mutant mice [112]. The results of this study demonstrated that gestational arsenic exposure particularly increased hepatic tumors with a C61A *Ha-Ras* mutation [112]. Though H-Ras has been linked increasingly with arsenicassociated cancers, some evidence also suggests the involvement of K-Ras in arsenic-related carcinogenesis in human-cultured prostate cells. The inorganic arsenic-induced transformation in these cells is associated with genomic DNA hypomethylation and K-Ras overexpression. Notwithstanding, this overexpression occurred without mutations and through a mechanism other than promoter region hypomethylation [113]. The mutated cells become very sensitive and susceptible to the promoter compounds, resulting in proliferation, which is noticed in several arsenic-associated animal carcinogenesis models. Transgene (mutated) mRNA expression with increased aberrant H-ras protein levels is associated with the DNA hypomethylation of specific *Msp1-HpaII* sites within the transgene [114]. MspI and HpaII are the restriction enzymes which recognize the same DNA sequence with differential sensitivities to methylated DNA sites. This serves as the locus-specific discrimination between methylated and unmethylated DNA sequences. Thus, the epigenetic modification is involved in the arsenic-induced aberrant signaling process and carcinogenesis mechanism (Figure 9-3) [115,116].

The possible involvement of several factors, i.e., H-ras, K-ras, or β -catenin in arsenicinduced cancers has been described in the previous section. Aurora A, the mitotic kinase, has been designated as an evolutionary conserved oncoprotein and is frequently overexpressed in various human cancers. Reports reveal that Aurora A overexpression associates with Ha-ras codon-12 mutation and blackfoot disease in arsenic-affected populations manifesting with bladder cancer. This report explains the causative factors of aberrant H-Ras expression and cancer occurrence [117]. β -Catenin, a subunit of the cadherin protein family, serves as an intracellular signaling molecule, which acts via the Wnt signaling pathway [118]. It regulates the coordination of cell-cell adhesion and gene transcription. The report reveals that mutations and/or overexpressions of β -catenin may result in multiple types of tumor and cancerous growth in different organs [118,119]. The repression in Wnt/ β -catenin signaling at the time of early embryogenesis results in a significant decrease in the expression of several transcription factors. Arsenic exposure to P19 mouse embryonic stem cells significantly increased the expression of a transcription factor Nanog, which maintains the pluripotency of stem cells [76]. It directs a particular cell type to proceed towards the proliferation state. The decrease in Wnt/ β -catenin signaling by arsenic can perturb the embryonic differentiation process. This suppression of differentiation may promote cellular proliferation and towards transformation [76]. A recent report suggests that transcriptional coregulator Mastermind-like 1 (MAML1) proteins act with p53, nuclear factor $\kappa\beta$ (NF- $\kappa\beta$), and β -catenin and functions in various signaling pathways. Most of these pathways are associated with the process of cell proliferation. This result is confirmed from the finding that overexpression of MAML1 increased the proliferation of human embryonic kidney (HEK) 293 cells, while MAML1 down-regulation by siRNA decreased cell proliferation [120]. In several animal experimental studies, the impairment of the differentiation/proliferation balance has been regarded as the basis of cellular tumorigenesis/carcinogenesis. The signaling machinery for cell division/differentiation is somehow compromised by arsenic, and results in cancer (Figure 9–3).

9.3.7 Arsenic and Epigenetic Modification

At the protein level, tumor necrosis factor p53 has been found to be involved (6%) in transitional cell carcinoma (TCC). A major fraction of lesions in TCC and papillomas were noticed to be involved with decreased numbers of p27kip.1 protein molecules. Cyclin D1 and cycloxygenase 2 (Cox-2) overexpression were also involved with a major number of lesions, i.e., papillary or nodular (PN) hyperplasia and different types of TCCs [21]. p27kip.1 is a cell cycle regulatory protein and interacts with the cyclin-CDK class of molecules. It may act via transforming growth factor β (TGF- β) induction and cell cycle arrest at the G1 phase [121]. CDK-dependent phosphorylation initiates the degeneration of this molecule, which sensitizes cells to proceed through the proliferation stage [121]. Overexpression of Cox-2 may lead to several inflammatory responses and may also increase matrix metaloprotease MMP-2 and MMP-9 expression after chronic arsenic exposure in rat kidney stem cells. This expression was shown to transform these cells into non-adherent spheroid cell bodies with cancerous properties [87]. Arsenic, especially DMA(V), has been shown to induce the expression of cyclin D1 and COX-2 mRNA in the bladder epithelium of rats [122]. In addition to this, there was an increase in TGF- β 1 and a decrease in tumor necrosis factor-alpha (TNF- α) levels in the urine of these treated rats. All of these findings have been correlated with a significant increase in urothelial cellular proliferation, suggesting that chronic inflammation and proliferation might be the basis of the arsenic toxicity [122]. Some investigations also suggest arsenic to be the promoter, or an initiator or an inducer of different degrees of carcinogenecity via involvement in Cox-2 regulation [123–125].

Genetic/epigenetic or signal transduction alone may not be able to result in carcinogenesis. In fact, in several cases it has an overlapping or shared activity. Epigenetic changes may influence the signal transduction-associated regulation of different gene expression modalities in a qualitative and/or quantitative manner. This series of changes may result in cellular transformation via impaired differentiation/proliferation balance. Apart from the regulations master gene p53, the impaired expressions of other genes and altered methylation (DNA) in tumor related several gene loci have been demonstrated in arsenic-induced carcinogenesis. Epigenetic modifications such as DNA methylation and histone modifications were screened in relation to the tumor promotion in C57B1/6 mice [126]. In addition, other tumor-related genes, namely P16 (INK4a), RASSF1A, H-Ras, and $ER-\alpha$, were also tested [126]. A de novo promoter hypermethylation of important tumor suppressor genes such as P16INK4a and RASSF1A results in the inactivation of these genes, promoting several types of organ cancer [127]. Long-term exposure to arsenic has been reported to decrease the expression of P16(INK4a) with an increased level of dimethylated histone H3 lysine-9 (H3K9). Influence in the promoter region by histone modifications results in the transcriptional suppression of certain genes. An increase was noticed in the recruitment of H3K9 histone methyltransferase G9a to the promoter after arsenic exposure [126]. The lysine 9 methylation of histone H3 is an epigenetic event that determines the inductance of pluripotent stem cells. H3K9 methyltransferase and its corresponding demethylase act reciprocally by regulating H3K9 methylation status. This histone methylation is related to the transcriptional repression of several important genes [128,129].

Estrogen receptor α (ER α) is a marker determinant of response of breast cancers. About 30% of breast cancers, however, are hormone independent because of lack of ER α expression [130]. It is reported that arsenic trioxide (2 mg/kg b.w. to mice) induces partial demethylation of the ER α promoter, which is protected by the universal methyl donor SAM. It reduces the degree of arsenic trioxide-induced re-expression of ER α and demethylation [130]. This suggests the importance of the cellular methyl pool and restoration of DNA methylation. Thus, arsenic trioxide exhibits the induction of re-expression ER α in ER-negative breast cancer cells through demethylation of the ER α promoter [130]. The INK4a/ARF locus produces (transcribes from an **a**lternate **r**eading frame) ARF tumor suppressor protein in response to continuous mitogenic stimuli, which may be generated by MYC and Ras ligand [131,132]. As described in this section, it may be concluded that epigenetic modifications are as important as genetic impairment; however, sometimes epigenetic dysregulations are more relevant in cellular transformation mechanisms.

9.3.8 Arsenic and Tumor/Heat Shock-Associated Proteins

Heat shock protein 70 (HSP70) has been implicated in the cellular stress initiation process and is associated with cellular transformation [133]. Abnormal methylation in the pertinent gene promoter and/or in the functional gene region has been implicated as an epigenetic phenomenon that regulates gene functions. This may result in cellular toxicity and even carcinogenesis. Perinatal arsenic exposure results in increased methylation in the transcribed region of the *HSP70* gene in apolipoprotein E (ApoE) knockout mice. And this altered induction of Hsp70 can cause hepatic stress and its degeneration [134]. ApoE is a class of apolipoproteins, which is primarily produced by the liver and macrophages. In the absence of this factor, such as in knockout mice [134], cholesterol metabolism is impaired, which might be an added factor for hepatic stress [135,136]. The HSP70 family is induced by a wide range of intrinsic or extrinsic

stresses and keeps proteins in a properly folded state [137]. They also protect nascent translating proteins, promote the cellular or organellar transport of proteins, and serve a general housekeeping role in maintaining protein homeostasis. Several human tumors overexpress the HSP70 family members, which result in a poor prognosis for the disease outcome [137]. Tumor suppressor p53 protein and the related gene regulation preserve the genetic stability by preventing genome-wide mutations [138]. In relation to p53 regulation, several indications demonstrate the involvement of p53 aberrant expression by iAs, which is related to both rodent and human carcinogenesis. The ubiquitin-conjugating enzymes (Ube2d) promote p53 ubiquitination and proteasomal degradation. A recent report suggests that inorganic arsenic may induce p53-dependent apoptotic pathways through down-regulation of gene expression of the Ube2d family in rat proximal tubular cells [17,50,139,140]. p53 has been shown to play a very important role in regulating the cellular proliferation/differentiation balance which is differentially maintained in different cultured cells [141]. Arsenic severely impairs this equilibrium [141].

Arsenite and UV radiation together have been reported to increase the incidence of carcinomas whereas UV cell irradiation alone produces papillomas in mouse skin [50,142]. These benign epithelial tumors grow externally. On the other hand, malignant skin tumors or cancers derived from epidermal cells are non-melanomas, while melanomas are derived from melanocytes. The non-melanomas include squamous cell carcinoma and basal cell carcinoma derived from all layers of the epidermis and the basal layer of the epidermis, respectively [143]. The increase in severity of tumorigenesis by arsenic has been demonstrated from a mechanistic point of view. It may be due to the inhibition of DNA repair or distortion of p53 functions and up-regulation of cyclin D1 activity [50]. The *p53* gene is mutated early in skin carcinogenesis and the resulting mutations are often present on the sites of dipyrimidine [144]. In general, the carcinogenic effects are not implicated by arsenical compounds alone [145]. Arsenite prevents terminal differentiation and helps pre-adipocytes to remain in an undifferentiated state by blocking the up-regulation of $c/EBP\alpha$ and p21. This process helps cells to move through a stage of cellular proliferation [146]. HSP70 is a general protector against cellular stress and p53 is the master regulator against tumor promotion. As described, arsenic can impair the functions of both molecules, which are lethal to living systems (Figure 9-4).

9.3.9 Arsenic and Apoptotic Signaling via Transcription Regulation

Premature ubiquitinization of proteins is noted with low doses of arsenite, which increases the cellular senescence of protein molecules [147]. Ubiquitin-like protein, which is an oncogenic product, may block arsenic-induced ubiquitination. Arsenic associated with modulation of the glucocorticoid receptor blocks this receptor's role in tumor suppression, which may be partially related to skin cancer [148]. The experimental results suggested that DMA-induced rat urinary bladder carcinogenesis might be caused by ROS generation, which might act as the initiator for further progression of the disease. This has also been shown in arsenic-exposed humans. In humans, different inorganic and organic arsenicals have been shown with varied ability in ROS production and p53 expression [58,59,149,150]. The comutagenic activity of iAs with TPA, discussed earlier, results in more invasive cellular carcinoma. The studies reveal that



FIGURE 9–4 Description of different regulatory steps and main participants in the arsenic-related mutagenesis and carcinogenesis processes.

impaired stem cell (SC) signaling and tumor growth are also interrelated [151]. Developmental arsenic exposure affects the stem cells and it is assumed that arsenic-related carcinogenesis correlates with a few stem cell-related diseases. Arsenic impacts human SC population dynamics *in vitro* by blocking exit into differentiation pathways thereby creating further critical objectives transformation [152]. These findings may be corroborated by earlier results on the role of arsenical compounds to promote cellular proliferation after blocking of the potential for differentiation [124,125]. Pancreatic tissue degeneration and a decrease in the number of cells in islets were noticed in male rats following exposure to iAs. These alterations can be attributed to oxidative stress and inflammatory responses with high-level release of TNF- α and interleukin-6 (IL-6) [153].

The nature of arsenical compounds and the dosage are reported to have differential effects in relation to steroid receptor binding. The methylated arsenicals were found to be more potent than iAs in disrupting steroid receptor binding to DNA response elements, thus impairing the hormone-associated gene expression pattern [154]. Testicular apoptosis is initiated by arsenic in the rat by the activation of caspase-3 and degenerative signaling from Bcl-2/ Bad with an increase of cytosolic cytochrome c. This suggests the mitochondrial involvement in this apoptotic process is accomplished through degeneration via the ROS-regulated pathway. Arsenite is found to induce mitogen-activated protein kinases (MAPKs), protein kinase B (AKT), and NF- $\kappa\beta$ in testicular tissue with an increase in inflammatory response [155]. The



FIGURE 9–5 Arsenic may induce free radical production and cellular and organelle membrane damage. Mitochondrial membrane damage results in the release of cytochrome-c. This initiates free radical cascades and activates several pro-caspases to caspases. An apoptotic pathway is introduced under the regulation of several others proteins, namely Bcl-2, BAD, and Bcl-xL. The cumulative effects of apoptotic signals and differentiation/ proliferation balance regulate cellular transformation ability. These are the important determinants of the processes of tumorigenesis and carcinogenesis.

carcinogenic potential of arsenic may be attributed to the activation of redox-sensitive transcription factors and other signaling pathways involving NF- $\kappa\beta$, activator protein-1, and p53. Impairment of cellular thiols is shown to induce oxidative stress-related injuries [156] and result in impaired redox regulation of not only transcriptional but also other factors.

Bcl-2-associated death promoter (*BAD*) is a pro-apoptotic gene family that initiates apoptosis. The Bax/BaK complex may also initiate apoptosis by promoting the cytoplasmic release of cytochrome c after forming mitochondrial membrane pores. Cellular redox imbalance may destabilize the mitochondrial membrane and cause release of cytochrome c (Figure 9–5). Such release is inhibited by the anti-apoptotic Bcl-2 and Bcl-xL complex. This process promotes inactivation of the cytoplasmic caspase pathway initiated by cytochrome c [157,158]. Chronic arsenic exposure increases ROS and nitric oxide (NO) (which is regarded as a strong signaling molecule) in neuronal cells of rodents [159]. The calcium influx-associated disruption of mitochondrial membrane potential has also been seen [159]. Mitochondrial degeneration, capase-3 activation, and bax/bcl2 signaling have been reported to be associated with neuronal apoptosis [159]. The reversal of some stepwise processes such as mitochondrial oxidative stress, and alteration of cytochrome c, caspase-3, and bax/bcl 2 ratio by the thiol chelator monoisoamyl dimercaptosuccinic acid (MiADMSA), strongly support that mitochondrial instability is the prime cause of arsenic-induced apoptotic cell death [160].

Angiotensin II receptor signaling has been implicated in arsenic-induced liver toxicity. The apoptotic and cancerous death of hepatocytes has been shown to regulate the nitric oxide synthase (NOS) and Cox-2 related pathway. As explained earlier, NF- $\kappa\beta$, inflammatory modulator TNF- α , and caspase-3 have been shown to be involved in arsenic toxicity and the carcinogenesis mechanism (Figure 9-3) [161]. Besides direct NF- $\kappa\beta$ signaling, transcriptional regulation of several important genes by TF is involved in cellular responses to stimuli such as stress, cytokines, and free radicals [162]. Mitogen-activated protein kinases (MAPKs) and extracellular signal-regulated kinases (ERKs) are groups of cell surface receptor proteins, which, upon activation, control transcriptional regulations to produce cell division promoting proteins. Phosphorylation-based activation of these kinases is also associated with the earlier discussed Ras-RAF pathways [163,164]. Arsenic-induced impairment of cellular cyclin D1 gene expression is shown to be mediated via miR-2909. Induction of splice switching of the tumor suppressor CYLD (cylindromatosis) gene as well as modulation of specificity protein 1 (SP1) has also been linked to this process. Thus, the important epigenomic event may play a role in arsenic-related carcinogenesis processes [165]. CYLD is a de-ubiquitinating enzyme acting via NF- $\kappa\beta$, c-Jun N-terminal kinase (JNK) activity, TNF receptor-associated factor 2 (TRAF2), and processes such as ubiquitination. Current data suggest that CYLD limits inflammation and tumorigenesis by regulating ubiquitination *in vivo* [166]. Impairment of immune functions by cellular and humoral levels of arsenic is a potential risk factor, which augments infections and chronic diseases, including various cancers [167].

Recently, it has been demonstrated that a signaling pathway, Salvador/Warts/Hippo (SWH), may explain the regulation of animal organ size, which is controlled by cell proliferation and apoptosis balance. Protein kinase Hpo (phosphorylates Wts and LATS1/2 in mammals) and its mutation is shown to be involved in tissue overgrowth. In other words, less phosphorylation of LAST1/2 results in impaired gene expression signaling and induction of tissue overgrowth. These kinases are the regulator of cell cycle progression and development. Important proteins like Sav and MOB activate Wts as tumor suppressors [168,169]. The report suggests that Hippo signaling plays a crucial role in organ formation during the embryonic development and its malfunctioning contributes to the pathogenesis of epithelial neoplasms. Arsenic treatment enhanced phosphorylation-dependent activation of LATS1 kinase and other Hippo signaling regulatory proteins Sav1 and MOB1. Arsenic-induced Phospho-LATS kinase inactivates the transcriptional co-activator Yap, which is usually inactivated without arsenic. Yap is indirectly related to the transcriptional regulations of several target genes. It is also important in regulating tight and adherence junctions through its binding to α -catenin. Arsenic-induced alterations of these pathways result in neoplastic lesions in mouse epidermis [170]. Arsenic-induced up-regulations of angiotensin II type I receptor (AT1R) subtypes may be caused by the ROS-mediated phosphorylations/activation of c-Jun N-terminal kinases (JNK), activated protein 1 (AP-1), and the JNK signaling pathway in the mouse aortic endothelial cell line. This step results in an aberrant signaling pathway and disease pathogenesis [171]. Arsenic trioxide can induce apoptotic morphogenesis, caspase-3 and TGF- β 1 protein

expression, Bax/Bcl-2 ratio, and ERK1/2 phosphorylation in guinea pigs. This suggests that TGF-β is functionally linked to ERK1/2 and its aberrant signaling is responsible for arsenic trioxide-induced cellular apoptosis [172]. The pigment epithelium-derived factor (PEDF) plays a role in sodium arsenite-induced cell apoptosis of liver and brain in rat and the results also demonstrate that the ratio of Bax/Bcl-2 might be important in the action of PEDF. The ratio of Bax/Bcl-2 in the higher dose of the arsenic group may suggest the possible induction of a cellular apoptosis pathway [173]. Arsenic trioxide (As₂O₃)-induced apoptosis in head and neck squamous carcinoma cells has also been reported [174]. The mechanism suggests that Bax was up-regulated without a change in Bcl-2 with an ultimate consequence of Bax/Bcl-2 increase and apoptotic death signal (Figure 9–5). The results also showed that the apoptosis of PCI-1 cells by As₂O₃ was induced by the activation of the caspase-3 via cytochrome c, caspase-9, and the Apaf-1 complex [174].

Combinatorial interactions of certain transcription factors regulate the expressions of some important genes. Hepatocyte nuclear factor 1α and 4α (HNF1 α and HNF4 α) are key members of a transcriptional network essential for normal liver architecture [175]. Changes in HNF1 α and HNF4 α expression are clearly associated with the development of liver malignancies and diabetes in humans. Inorganic arsenic has been shown to down-regulate both $HNF1\alpha$ and HNF4 α in human hepatocarcinoma (HepG2) cells and in liver of golden Syrian hamster cells. This process has been linked to arsenic-associated liver cancer [175]. HNF1 α and HNF4 α control pancreatic β -cell function and growth, and mutation (in this gene) may result in certain forms of diabetes [176]. Thus, it is important to address how these genetic programs control cellular functions and disease processes. Epistasis analysis with transcriptomes of single- and double-mutant islets revealed that Hnf1 α and Hnf4 α regulate common targets synergistically, suggesting that the mechanisms for synergistic regulation are gene specific. Hnf1 α and Hnf4 α control a common islet-cell regulatory program that is defective in human monogenic diabetes [176]. It is further demonstrated that the transition program between epithelial-to-mesenchymal cells lineages and reverse is controlled by HNF4 α , in cooperation with its target HNF1 α . This process determines the possibility of the cell lineage to move towards a terminal differentiated state or a regular step of its development, tissue repair, and tumor progression [177].

Imbalance in the cellular redox state and its associated complications has been closely associated with the regulation of nuclear factor-erythroid 2-related factor 2 (Nrf2). It is a basic-leucine zipper transcription factor that activates the antioxidant responsive element and the electrophilic responsive element. This process up-regulates the expression of a variety of downstream genes [99,178]. Exposure to a low dose of arsenic trioxide caused transformation of BALB/c 3T3 cells and, in a xenograft mouse model via induction of polycomb (PcG) proteins, BMI1 and SUZ12 functions. This protein function is related to histone H3K27 methylation (similar to H3K9 methylation as discussed earlier) after arsenic exposure. In BMI1- or SUZ12-knockdown BALB/c 3T3 cells, the expression of p16 (INK4a) and p19 (ARF) is restored compared to corresponding wild-type cells. This suggests the involvement of p16 (INK4a) and p19 (ARF) proteins in the carcinogenesis process, as mentioned in the previous section [179].

Arrigo [180] first reported that exposure to arsenic in *Drosophila* cells led to a complete abolition of methylation of histones H3 and H4 [180]. Little is known about histone phosphorylation and arsenic-induced carcinogenesis. Histone phosphorylation may also contribute to arsenic-induced carcinogenesis. The best-studied histone phosphorylation event is that of H₂AX, a form of H₂A that represents up to 25% of the total H₂A pool in mammals. Evidence suggests that arsenic trioxide induces apoptosis by the up-regulation of phosphorylated H₂AX [181]. Studies have suggested that H3 phosphorylation induced by arsenic exposure might be responsible for the up-regulation of the oncogenes c-*fos* and c-*jun* [182] and induction of a protoapoptotic factor, caspase-10. Interestingly, As(III) exposure has also been shown to induce elevated histone acetylation, which was reportedly responsible for the upregulation of genes involved in apoptosis or the response to cell stress after exposure to arsenic [182]. As(III) has been shown to inhibit histone deacetylase *HDAC* genes that correlate with increased global histone acetylation [183]. A small class of non-protein-coding RNAs, called microRNAs (miRNAs), participate in diverse biological regulatory events and are transcribed mainly from non-protein-coding regions of the genome [184]. These have been found to be involved in arsenic-associated carcinogenesis.

In an earlier section, ligand activated Ras signaling was described during arsenic carcinogenesis in rodent models. It is shown that the *RAS* oncogene is regulated by the let–7 miRNA family [185]. Arsenic exposure has been shown to alter methylation levels of both global DNA and gene promoters. Besides this, it influences histone acetylation, methylation, phosphorylation, and miRNA expression [186]. Thus, a comprehensive epigenomic approach may elucidate the mechanisms of arsenic-induced carcinogenesis. Such an approach would also facilitate our search for the biomarkers of arsenic exposure and early effects, associated diseases and disease progression, and factors that confer susceptibility. This comprehensive discussion on the transcriptional regulation of several important genes, including cell cycle regulators, growth factors or oncogenic materials or histone-modifying enzyme genes, is highly relevant in the mechanism of arsenic-related carcinogenesis (Figure 9–6).

9.4 Human Arsenic Carcinogenesis

9.4.1 Genetic/Epigenetic Changes in Arsenic-Associated Human Carcinogenesis

The possible mechanistic layout was described in an earlier section to explain the carcinogenesis processes in animal models. One or more of these pathways may also be implicated in arsenic-associated tumorigenesis and/or carcinogenesis in humans. The data from several *in vitro* studies further add important information to the toxicity processes. The metabolic pattern, however, varies significantly between laboratory-exposed rodent models and environmentally-exposed human individuals. This is not only because of the differences in arsenic exposure pattern, but is also due to a major variability in gene mutations, single nucleotide polymorphisms (SNPs), genetic polymorphism, and other genetic and epigenetic events that exist in rodents and humans. It indicates that arsenic is highly mutagenic and carcinogenic to humans, but is not as potent to laboratory animal models. Several *in vitro* studies further help in delineating the mutagenesis/carcinogenesis pathway, but their definite applicability



FIGURE 9–6 Arsenic can induce genetic and epigenetic changes. Alteration in gene expression regulation and impaired signal transduction at the cytosolic and the nuclear level result in differentiation/proliferation imbalance and cellular transformation. Different growth factors and oncogenic mutation/activation are the key players in the activation of several mitogen-driven kinases. These in turn modulate the function of transcription factors. Reactive oxygen species are one of the potent inducers for all these phenomena.

in whole animal models or in humans is largely inconclusive. A suitable experimental model is still a matter of debate between investigators; it is equally difficult to characterize mechanisms of arsenic-induced oncogenesis/carcinogenesis based on *in vitro* studies. However, rodent models where a xenotoxic substance causes the induction of tumors can be invaluable because they are clearly pertinent to a carcinogenic process. To explain the mechanisms behind arsenic-associated human carcinogenesis, the epidemiological data may be of immense assistance. It is agreed that there is a limitation in animal experiments due to their differential sensitivities in toxicological pathogenesis processes. With this background, the best possible and most reasonable integrative features of different mechanistic pathways resolved from animal experiments, cell culture studies, and human reports and the overlapping regions



FIGURE 9–7 Different important overlapping pathways of arsenic-associated toxicity mechanism that may finally result in mutagenesis and/or carcinogenesis. In general, the fewer the number of factors involved, the less chance of disease occurrence/severity. The overlapping region denotes the interactive nature of several pathways, which usually manifest adverse cumulative impact. The central overlapping region (*) may be represented as the most severe disease status. The important unique molecules/steps at the overlapping regions might have been a good therapeutic target.

of these pathways should be taken into account in defining the arsenic-associated human carcinogenesis process (Figure 9–7).

9.4.2 Arsenic Affects Cultured Human Cells and Environmentally Exposed Human Populations

Inorganic arsenic is undoubtedly carcinogenic in humans. In the preceding section the mechanism for arsenic-associated carcinogenesis at the molecular level was discussed. The results are mostly analyzed from human-cultured cell experiments or from environmentally exposed human sample data. Arsenic can induce genotoxicities by introducing large deletion mutations in human/hamster hybrid cells and at the thymidine kinase (TK) locus of mouse lymphoma cells [187]. Genotoxic damage has also been implicated by chromosomal abnormalities, sister chromatid exchange, micronuclei induction, and abnormal DNA synthesis. DNA/protein crosslinking has also been demonstrated to occur in human cells exposed to arsenic [187]. A micronucleus is an extranuclear body containing some DNA damage and random mutations in a daughter cell [188]. DNA/protein crosslinks occur when various exogenous or endogenous agents react with two different positions in the DNA. DNA replication is blocked by crosslinks, which causes replication arrest and cell death if the crosslink is not repaired [189]. As described in the previous section, arsenic is highly capable of influencing these molecular events. TK is a phosphotransferase (a kinase): 2'-deoxythymidine kinase, ATP-thymidine 5'-phosphotransferase. TKs have a central role in the synthesis of DNA and thereby in cell division, as they participate in the reaction chain to incorporate deoxythymidine in DNA [190].

Low dose and long-term exposure ($\leq 0.1 \,\mu$ M for 8 weeks) to arsenite in human osteosarcoma cells cause transformation and mutations in another important enzyme hypoxanthine phosphoribosyltransferase (HPRT) gene locus [191]. Its product is hypoxanthine-guanine phosphoribosyltransferase (HGPRT) and the gene is located on the X chromosome. HGPRT catalyzes conversion of hypoxanthine to inosine monophosphate and guanine to guanosine monophosphate. This reaction transfers the 5-phosphoribosyl group from 5-phosphoribosyl 1-pyrophosphate (PRPP) to the purine [192]. Arsenic binds with sulfhydryl groups of DNA repair enzymes and initiates genotoxic and carcinogenic effects. This toxic metalloid also impairs DNA ligase and poly (ADP-ribose) polymerase (PARP) activities and results in mutations in the presence of UV irradiation [191]. PARP is a family of nucleoproteins that repair single-strand DNA breaks (SSB) and control programmed cell death. PARP is activated in response to physical or chemical stress-induced DNA SSB. It repairs SSB after binding to the DNA, and begins the synthesis of a poly (ADP-ribose) chain (PAR). It acts as a signal for the other DNA-repairing enzymes such as DNA polymerase beta ($pol\beta$), DNA ligase III (LigIII), and some important scaffolding. The PAR chains are degraded via poly (ADP-ribose) glycohydrolase (PARG) after the repair process [193]. As mentioned, arsenic-associated DNA strand breakage is also repaired by these enzyme systems.

The reports from an arsenic-exposed Bangladesh population suggest that the DNA repair gene *XPD* (also known as *ERCC2*) is associated with premalignant skin lesions and hyperkeratosis. The interindividual variability in arsenical skin lesions was due to different SNPs in the *XPD* gene. XPD is a nucleotide excision repair (NER) enzyme [194], which also participates in the basal level of transcription by interacting with transcription factor TFIIH [195,196]. The important SNP in the *XPD* gene is present on codon 751 in exon 23 ($A \rightarrow C$, Lys \rightarrow Gln), which has been shown to be associated with the risk of skin cancer. Arsenic impairs DNA repair by inhibiting DNA ligase 1 [197], a required enzyme for NER mechanism. In addition, arsenic also generates oxidative DNA damage, which can be corrected by NER. This may indicate that the XPD genotype is a potential modifier of arsenic effects [198]. In brief, various genes related to DNA and nucleic acid metabolism are the prime target in arsenic-associated human carcinogenesis. Impairment of the role of some transcription factors is also a notable concern in this regard (Figure 9–6).

9.4.3 Arsenic Induces Telomere Instability

The telomeric regions of the chromosome and the telomerase enzyme have been considered as among the significant determinants of genome stability. The condensed nucleoprotein structures at chromosome termini are known as the telomere. It is composed of simple, repetitive G/C-rich DNA complexes with different histone proteins [199,200]. The investigators showed that arsenic inhibits telomerase transcription and results in the chromosomal

end lesions in human cells *in vitro*. Accordingly, arsenic may interfere with chromosome stability, which leads to clastogenicity in human cells. Telomeres play a critical role in maintaining chromosome stability by protecting chromosome ends from degradation and fusion [201]. The human telomeric DNA sequence is (TTAGGG)n and the terminal restriction fragment (TRF) contains the (TTAGGG)n sequence [200]. The DNA synthesis machinery is incapable of replicating three primed termini of the chromosome replication; thus there is a gradual loss of telomeric sequence under normal conditions during cell division. The eventual consequences of this process are senescence or apoptosis. Telomerase, a special ribonucleoprotein reverse transcriptase, is important in maintaining telomere length. It is necessary for the sustained growth of most human tumors and plays an important role in tumorigenesis [199,202]. The expression of human telomerase reverse transcriptase (hTERT) is the rate-limiting step for telomerase activity. Telomerase activity has been detected in different types of tumor cells [201–203].

The telomerase RNA template (TR) interacts with telomerase reverse transcriptase and this process regulates telomere elongation during cell division. It is reported that telomere shortening, chromosome end-to-end fusion, and reactivation of telomerase occur excessively, which promotes carcinogenesis in p53 mutant status [204,205]. Human cells are more sensitive to arsenite than are the cells of rodent origin [206], and one possible reason for this difference could be attributed to the short telomeres found in human cells [207]. Differences between mouse and human telomeres restrict the laboratory mouse as a reliable model to critically test the role of telomere dysfunction and telomerase in human malignancy. However, telomerase knockout mice show progressive telomere shortening and chromosomal instability with increasing generations. There are defects in highly proliferative tissues, affecting fetal development, growth, immune function, and carcinogenesis [208,209]. ROS are found to damage telomeres and lead to genomic instability [210]. Arsenic-induced ROS production is very much evident in all types of experimental model and in environmentally exposed human individuals. Sustained damage to telomeres is regarded as promoting the aging process. Arsenicassociated ROS production and telomere instability may mimic senescence-related features. The arsenic-associated carcinogenesis mechanism is summarized in the Figure 9-4.

9.4.4 Arsenic-Associated DNA Damage and Genome Dysfunction in Humans

Environmental conditions and addiction habits have been linked with the severity of arsenic pathogenesis. It has been reported that tobacco use may increase the risk of bladder cancer and its pathogenesis in arsenic-exposed individuals in comparison to their non-smoker controls [211]. Studies suggest that peripheral blood cells may be affected by long-term arsenic exposure. Water, blood, and urinary As have been positively correlated with global peripheral blood mononuclear cell (PBMC) DNA methylation [212]. This suggests that such epigenetic events and the gene function efficacy may be impaired by arsenic exposure in mixed white blood cell (WBC) preparations. In an epidemiological study involving a large number of arsenic-exposed populations from West Bengal in India, it has been reported that long-term

consumption of arsenic-enriched water caused skin lesions and other dermatological and multi-organ health problems resulting in cancer and death. In this study, an association between arsenicism and exonic single nucleotide polymorphisms (SNPs) in NAcht Leucinerich repeat Protein 2, (NALP2 gene, also known as NLRP2 and PAN1), an important component of the inflammasome complex, was suggested [213]. Among several important SNPs found in the NALP2 gene, the A1052E polymorphism is the most significant. These findings suggest that the NALP2 A1052E SNP plays an important role in the development of arsenicinduced skin lesions and chromosomal damage [213]. NALP2 is a cytosolic, 121 kDa member of the NLRP family of proteins. It is expressed in macrophages and exhibits divergent effects on inflammation. Through the involvement of NF- $\kappa\beta$, it suppresses TNF- α production in response to lipopolysaccharide (LPS), and via procaspase-1 it promotes proIL-1 β cleavage and release. Human NALP2 is 1062 amino acids in length [214,215]. An association has been shown between arsenic exposures levels being measured in toenails, and blood DNA methylation in Alu and Long interspersed nucleotide element-1 (LINE-1) repetitive elements in elderly men environmentally exposed to low levels of arsenic [216]. This study also revealed that higher arsenic contents are associated with increasing Alu and decreasing LINE-1 DNA methylation. The link between plasma folate, cobalamin (vitamin B12), and arsenic-associated methylation of these DNA regions suggests that the metabolism of the methyl group is an important factor [216]. The study also revealed that women had lower levels of LINE-1 methylation than did men [217]. Different arsenic species have been shown to induce strand breakages differentially. A discussion of DNA damage events which are associated with arsenic-induced carcinogenesis is given in the following paragraph.

9.4.5 Arsenic and Epigenetic DNA Modification

As(III) induces DNA strand breaks in human fetal lung fibroblast (2BS) cells [217], while DMA(V) induces DNA strand breaks in human alveolar epithelial type II (L-132) cells [218]. As(III) induces single-strand DNA breaks, DNA/protein adducts, and sister chromatid exchanges in human fibroblast cell lines [219]. Formation of apurinic/apyrimidinic sites has been shown in the human alveolar epithelial cell line L-132 as caused by DMA [220]. As(III) has been shown to increase 8-OHdG levels in human breast cancer (MCF-7 adenocarcinoma epithelial cells) [221]. Increases in 8-OHdG levels through $(CH_{3})_{2}AsOO DMA(V)$ have also been demonstrated [55,222,223]. Both As(III) and As(V) may result in the alteration of methylation in the p53 promoter of the A549 cell line [224]. As(III) led to a marked increase in the activity in chromosome H3K9 dimethylation and decreased H3K27 trimethylation (gene silencing) events. It also increased the H3K4 trimethylation (gene-activating) process, and histone methyltransferase G9a protein levels in the human A549 cell line [225]. Changes to histone H3 acetylation and DNA promoter methylation, and decreases in expression of DBC1, FAM83A, ZSCAN12, and C1QTNF6 genes in human non-tumorigenic cell lines have been observed after arsenic exposure [226]. Arsenic(III)-associated global hypomethylation has been shown in different cell lines of human sources like prostate epithelial cells RWPE-1 and keratinocytes HaCaT [186,227]. Hypomethylation of specific gene locus has also been noticed by several investigators. As(III) and MMA(III) influence *DBC1*, *FAM83A*, *ZSCAN12*, and *C1QTNF6* genes in some cultured cells [97,226]. As(III) may influence *DAPK*, *P16*, and *P53* genes in uroepithelial SV-HUC-1 cells, myeloma cell line U266, and lung adenocarcinoma A549 cells, respectively [228].

A recent study found the occurrence of 22 CpG sites with increased methylation in WBC DNA collected in arsenic-exposed populations [229]. Some genes annotated by these 22 CpG sites are known to be involved in As-associated diseases [230]. A significant increase in 8-OHdG levels in the As(III)-exposed mouse liver was confirmed in a study that also showed the presence of severe oxidative DNA damage. These results demonstrate that arsenite is mutagenic *in vivo* and suggest that arsenite induces G:C to T:A transversions through oxidative stress induced by 8-OHdG formation [231]. The alteration in regulations of various miRNAs has been noted after iAs treatment in Jurkat cells. Several miRNAs show specific dysregulation, and particularly a panel of eight miRNAs is demonstrated to be involved in arsenic-response pathways. This identification is invaluable and can serve as a potential biomarker of arsenic toxicity with useful diagnostic value [232]. Differential dysregulation of important genes provides variability in different humans. Overall, these findings provide evidence that interindividual differences in arsenic metabolism may be an important risk factor for arsenic-related lung cancer, and may play a role in cancer risks among people exposed to relatively low arsenic levels [233].

Melanocytes are more resistant to arsenite and UVR in the stimulation of superoxide production than are keratinocytes. However, the same concentration of arsenite may promote UVR-induced DNA damage in both types of cell by inhibiting poly(ADP-ribose)polymerase. These findings suggest that arsenic, on chronic exposure, could promote retention of unrepaired DNA damage in melanocytes, acting as a co-carcinogen in melanoma [234]. Few earlier studies have demonstrated that hypermethylation of the promoter region of *p53* and *p16* genes is occurring in persons exposed to arsenic [235]. Until now no genomic hotspot has been decisively identified that is frequently hypermethylated or hypomethylated in persons chronically exposed to environmental arsenic [235]. FBJ murine osteosarcoma viral oncogene homolog B is known as FosB protein. *Fosb*, an important oncogene, has been shown to undergo alteration in DNA methylation via gestational arsenic exposure [236]. It was shown that Fosb expression significantly increased corresponding to the DNA methylation of the gene in the arsenic-exposed group. These findings suggest that DNA methylation status can be used to identify tumors resulting from gestational arsenic exposure [236].

Arsenic species patterns in urine are associated with risk of cancer and cardiovascular diseases. The organic anion transporter coded by the gene *SLCO1B1* may transport arsenic species depending on this genetic status (presence/absence or variants) [237]. Hypermethylation was found in the promoters of both *DAPK* and *p16* genes in individuals with arsenic-induced skin lesions in comparison to the individuals without skin lesions (positive controls) [238]. A significant decrease was noted in the expression of death-associated protein kinase and in the gene expression of p16 in induced cases compared to controls [238]. The lowest expression was observed in cancerous tissues. The hypermethylation of the corresponding promoter of specific genes was also associated with higher risk of developing arsenic-induced skin lesions, peripheral neuropathy, and ocular and respiratory diseases [238]. Not only arsenic, but nickel, cadmium, and chromium (VI) are known to alter the DNA methylation profile by inducing both hyper- and hypomethylation events. Recent developments in finding a suitable therapy are based on the cancer methylome [239]. Differential DNA methylation was noted in the umbilical cord blood of infants exposed to low levels of arsenic in utero. It was also suggested that in utero exposure to low levels of arsenic may affect the epigenome [11,240]. Chronic exposure to arsenic is known to generate reactive oxygen species, which may ultimately lead to the induction of certain proinflammatory cytokines. Polymorphism of inflammatory genes and arsenic methvlation capacity are associated with urothelial carcinoma [241]. Here, specific polymorphism in TNF- α -308 G/A, IL-6 -174 G/C, IL-8 -251 T/A, and urinary arsenic profiles were found to be related to urothelial carcinoma (UC) risk. The study evaluated 300 pathologically confirmed cases of UC and 594 cancer-free controls. A significant decrease in UC risk was found in subjects who carried the IL-8 -251 T/T genotype [241]. This study strongly suggests that not only metabolism associated genes, but immune system associated genes too showed impaired methylation with cancer manifestations. Arsenic-contaminated drinking water and the risk of various human malignancies were verified and the study revealed that the FHIT and WWOX genes are active in common fragile sites FRA3B and FRA16D, respectively [242]. Reduced expression of FHIT or WWOX is known to be an early indicator of carcinogen-induced cancers. An interesting study recently suggested that arsenite decreased the expressions of ATR, WWOX, and FHIT via ERK1/2 activation in SV-HUC-1 cells. These findings also confirmed that dysregulations of these markers might also be contributing factors to arsenite-induced carcinogenesis [242]. As a whole, a large number of genetic materials may be interfered with by arsenic at their regulatory steps. Several of these interferences may result in obvious onset of tumorigenesis and/or carcinogenesis (Figure 9-6).

9.4.6 Arsenic-Associated Signal Transduction Processes and Human Carcinogenesis

Arsenic-associated mutagenesis/carcinogenesis processes may be associated with the impairment of signal transduction processes. In animal models several of these pathways have been explored. Most of these pathways are also applicable to human-associated carcinogenesis to different degrees. The interacting molecules may have a varied extent of association in terms of quality and quantity.

The regulation of p53 tumor suppressor gene product is very important in cellular transformation. Normal p53 protein is involved in cell cycle regulation, in maintenance of genomic stability, and in controlled cell apoptosis. Studies revealed involvement of arsenic-associated p53 mutation in the manifestations of bladder and skin tumors, and the induction of carcinogenesis through inhibition of DNA repair. All these processes significantly resulted in oxidative damage to DNA [243]. It is reported that the bladder tumors from people chronically exposed to arsenic showed mutations in the *TP53* gene at codon 175 and transitions at points 9 and 10 [244]. iAs(III) treatment (10μ M) has been reported to induce an accumulation of p53 protein in different cell lines having a wild or mutated gene, with a greater effect in cells having normal *TP53* than in those with mutated *TP53* [245]. The p53 protein is a transcription factor that accumulates in response to DNA damage and other stress stimuli. This increases the expression of cyclin-dependent kinase inhibitor p21 (CKI p21) protein, which is able to silence cyclin-dependent kinases (CDKs) essential for S-phase entry (via cyclins E and A) leading to the deactivation of cell cycle progression [246]. The p21 protein has been characterized as a cyclin kinase inhibitor that induces growth arrest by preventing phosphorylation of retinoblastoma pRb in G1/S transition. pRb regulates G1/S transition depending on its phosphorylation status, which is modified by CDKs/cyclin complexes [245,247].

Arsenite binds with the sulfhydryl (SH) group of enzymes, leading their inhibition [248]. Arsenite induces production of reactive oxygen species, which affects multiple targets in the cell such as inhibition of I κ B kinase (IKK)1 by arsenite [249]. This results in the suppression of NF- κ B transcription factor activation followed by a marked change in the anti-apoptotic functions of the cell [250]. Simultaneously, oxidative stress induces kinase pathways (MKK6-MAPK p38, MKK4/7- JNK, and MEK-ERK1/2), up-regulates the AP-1 (Jun-Fos, Jun- ATF2) transcription factors, and induces AP-1-dependent gene expression and the regulation of the STAT family [251,252]. In general, different cancerous growths result from the involvement of high levels of cytokines and growth factors (i.e., TNF- α , transforming growth factor- β , and TRAIL) supporting their autonomous growth [253]. Moreover, the important transcriptional regulator NF- κ B pathway and two other signaling pathways, Ras-Raf-MEK-ERK and PI3K-AKT, are shown to be critically involved in the regulation of cell survival including the scenario of arsenic-associated carcinogenesis [250,254,255].

Inorganic arsenic transforms the RWPE-1 human prostate epithelial line to CASE-PE cells, and a derivative normal stem cell (SC) line, WPE-stem, to the arsenic-cancer SC (As-CSC) line. miRNAs are non-coding but exert negative control of expression by degradation or translational repression of target miRNAs. Aberrant miRNA expression is important in carcinogenesis [256]. In both transformants, decreased miRNAs targeting KRAS and RAS superfamily members impair signal transduction processes. Reduced miR-134, miR-373, miR-155, miR-138, miR-205, miR-181d, miR-181c, and let-7 in CASE-PE cells was correlated with increased target of RAS oncogenes, RAN, RAB27A, RAB22A mRNAs, and KRAS protein. Reduced miR-143, miR-34c-5p, and miR-205 in As-CSC was correlated with increased target RAN miRNA, and KRAS, NRAS, and RRAS proteins. The RAS/ERK and PI3K/PTEN/AKT pathways control cell survival, differentiation, and proliferation, and when dysregulated promote a cancer phenotype. iAs transformation increased expression of activated ERK kinase in both transformants and altered components of the PI3K/PTEN/AKT pathway, including decreased PTEN, and increased Bcl-2, Bcl-xL, and vascular endothelial growth factor (VEGF) in the absence of AKT activation. Thus, dysregulated miRNA expression may be linked to RAS activation in both transformants [256,257].

As mentioned earlier, IkBs are a family of related proteins that have an N-terminal regulatory domain, followed by six or more ankyrin repeats and a PEST domain near their C terminus. Although the IkB family consists of IkB α , IkB β , IkB ϵ , and Bcl-3, the best-studied and major IkB protein is IkB α . Activation of NF-kB is initiated by the signal-induced degradation of IkB proteins. This occurs primarily via activation of a kinase called the IkB kinase (IKK). IKK is
composed of a heterodimer of the catalytic IKK α and IKK β subunits and a "master" regulatory protein termed the NF- κ B essential modulator IKKy [258]. The regulation of I κ B protein is very important because arsenic-associated tumorigenesis and carcinogenesis have been shown to be influenced by NF- κ B signaling [258]. The activator protein 1 (AP-1) is a transcription factor that is a heterodimeric protein composed of proteins belonging to the c-Fos, c-Jun, ATF, and JDP families, which are also shown to participate in arsenic-related cancerous growth. It regulates gene expression in response to a variety of stimuli, including cytokines, growth factors, stress, metal-associated carcinogenesis including arsenic, and bacterial and viral infections. AP-1 in turn controls a number of cellular processes, including differentiation, proliferation, and apoptosis [259]. Arsenic has been shown to promote tumor formation by stimulating AP-1 activity and inhibiting a JNK phosphatase [251]. The signal transducer and activator of transcription 3, also known as STAT3, is a transcription factor protein, which in humans is encoded by the STAT3 gene, which has been shown to influence oncogenic activation upon arsenical threat [260,261]. In response to cytokines and growth factors, STAT family members are phosphorylated by receptor-associated kinases and then form homo- or heterodimers that translocate to the cell nucleus, where they act as transcription activators [260]. Cytokine mediation in STAT regulation in arsenic carcinogenesis is evident [260].

A very recently characterized gene, *mdig* (mineral dust-induced gene), is a newly identified oncogene linked to occupational lung diseases and lung cancer. It is unclear whether *mdig* is also involved in arsenic-induced malignant transformation of lung cells. By using human bronchial epithelial cells and human lung cancer cell lines, researchers demonstrated that arsenic is able to induce expression of *mdig* [261]. This group further demonstrated that *mdig* induction by arsenic was partially dependent on the JNK and STAT3 signaling pathways. Disruption of JNK or STAT3 by their corresponding siRNAs or some chemical inhibitors diminished arsenic-induced accumulation of *mdig* mRNA and protein. The group also showed that microRNA-21 (miR-21) and Akt were downstream effectors of JNK and STAT3 signaling pathways in arsenic-induced mdig expression. Collectively, these data suggest that *mdig* may play an important role in the pathogenesis of arsenic-induced lung cancer. Further, JNK and STAT3 signaling pathways are essential in mediating arsenic-induced *mdig* expression [261]. There is also evidence that miR-21 is involved in arsenite-induced malignant transformation and associated signaling pathways. During arsenite-induced transformation of human embryo lung fibroblast (HELF) cells, miR-21 is up-regulated, and the ERK/NF- κ B signaling pathway is activated [262]. Moreover, superoxide dismutase (SOD, a scavenger of superoxide) and catalase (a scavenger of hydroperoxides) block the arsenite-induced effects in HELF cells and mouse embryonic fibroblasts. Overexpression of miR-21 and the activation of ERK and NF-κB promote the malignancy of HELF cells exposed to arsenite. But knockdown of miR-21 and a feedback blockage of ERK and NF-KB activation decrease anchorage-independent growth of arsenite-transformed cells. Thus, the transformation of HELF cells by chronic arsenic exposure is mediated by increased miR-21 expression. This overexpression is mediated by the ROS activation of the ERK/NF-kB pathway [21,262]. MicroRNA-21 is also known as hsa-mir-21 or miRNA21, a mammalian microRNA, which is encoded by the MIR21 gene. MIRNA21 was one of the first mammalian microRNAs identified. The mature miR-21 sequence is strongly conserved throughout evolution. The human microRNA-21 gene is located on the plus strand of chromosome 17q23.2 (55273409–55273480) within a coding gene *TMEM49* (also called vacuole membrane protein) [263].

Arsenic, due to its ability to regulate genes that link cell cycle control with apoptosis, has been widely recognized to play a crucial role in oncogenomics. miR-2909 regulates *CCND1* (*cyclin D1*) gene expression within these cells by inducing the splice-switching of tumor suppressor *CYLD* (cylindromatosis) gene as well as by modulation of specificity protein 1 (SP1) activity through the repression of Kruppel-like factor4 (KLF4) expression at the translational level [264].

Inorganic arsenic is a well-known human skin carcinogen. Chronic arsenic exposure results in various types of human skin lesions, including squamous cell carcinoma (SCC). To investigate whether mutant stem cells participate in arsenic-associated carcinogenesis, HaCaT cell lines were repeatedly exposed to an environmentally relevant level of arsenic (0.05 ppm) *in vitro* for 18 weeks. Following sodium arsenite administration, cell cycle, colony-forming efficiency (CFE), cell tumorigenicity, and expression of CD44v6, NF- κ B, and p53 were analyzed. The expression level of cleaved caspase-3, which is a molecular indicator of cell apoptosis, was remarkably elevated after long-term arsenic exposure [265]. An arsenic-induced tumorigenesis and carcinogenesis mechanism is summarized in Figure 9–4.

Investigators also observed that chronic exposure of HaCaT cells to a low level of arsenic led to a cancer stem-like phenotype. Further, arsenic-treated HaCaT cells also became tumorigenic in nude mice and their growth cycle was predominantly in the G2/M and S phases. Arsenic was capable of increasing cell proliferation and the sprouting of the cancer stem-like phenotype. Immunohistochemical analysis demonstrated that CD44v6 expression was upregulated in HaCaT cells exposed to a low level of arsenic. Likewise, the expressions of activating transcription factor NF-kB and p53 genes in the arsenic-treated HaCaT cells were also significantly higher than in non-treated cells. Higher expressions of CD44v6, NF-κB, and p53 were also observed in tumor tissues isolated from Balb/c nude mice. Arsenic promotes malignant transformation in human skin lesions through an NF-kB signaling pathway-stimulated expression of CD44v6 [266]. The chronic arsenic-exposed breast epithelial (CABE) cells showed increases in secreted matrix metalloproteinase (MMP) activity, colony formation, invasion, and proliferation rate, indicating an acquired cancer cell phenotype. These CABE cells presented basal-like breast cancer characteristics, including ER- α , HER-2, and progesterone receptor negativity, and overexpression of K5 and p63. Putative CD44(+)/CD24(-/low) breast SCs were increased to 80% over the control in CABE cells [267]. However, the role of epithelial to mesenchymal transition (EMT) in arsenic carcinogenesis is not well understood, although previous studies have shown that short-term exposure of endothelial cells to arsenic stimulates angiogenesis. It still remains to be determined whether cells that were malignantly transformed by long-term arsenic exposure have a pro-angiogenic effect. It was found that the conditioned medium from arsenic-transformed cells strongly stimulated tube formation by human umbilical vein endothelial cells (HUVECs). Moreover, enhanced angiogenesis was detected in mouse xenograft tumor tissues resulting from inoculation of arsenic-transformed cells. Mechanistic studies revealed that β -catenin was activated in arsenic-transformed cells,

up-regulating its target gene expression, including angiogenic-stimulating vascular endothelial growth factor (VEGF). Reverse transcriptase-PCR analysis revealed that the mRNA levels of canonical Wnt ligands were not increased in arsenic-transformed cells. These findings suggest that EMT in arsenic-transformed cells promotes angiogenesis through activating the β -catenin-VEGF pathway [268].

The Nrf2-Keap1 signaling pathway is a protective mechanism promoting cell survival. Activation of the Nrf2 pathway by natural compounds has been proven to be an effective strategy for chemoprevention. Interestingly, a cancer-promoting function of Nrf2 has recently been observed in many types of tumors due to deregulation of the Nrf2-Keap1 axis, which leads to constitutive activation of Nrf2 [269]. Recently, a novel mechanism of Nrf2 activation by arsenic has been postulated that is distinct from that of chemopreventive compounds. Arsenic deregulates the autophagic pathway through blockage of autophagic flux, resulting in accumulation of autophagosomes and sequestration of p62, Keap1, and LC3. Thus, arsenic activates Nrf2 through a non-canonical mechanism (p62 dependent), leading to a chronic, sustained activation of Nrf2. In contrast, activation of Nrf2 by sulforaphane (SF) and tert-butylhydroquinone (tBHO) depends upon Keap1-C151 and not p62 (the canonical mechanism). More importantly, SF and tBHQ do not have any effect on autophagy. In fact, SF and tBHQ alleviate arsenic-mediated deregulation of autophagy. Collectively, these findings provide evidence that arsenic causes prolonged activation of Nrf2 through autophagy dysfunction, possibly providing a scenario similar to that of constitutive activation of Nrf2 found in certain human cancers. This may represent a previously unrecognized mechanism underlying arsenic toxicity and carcinogenicity in humans [270].

The major signaling pathways altered include NRF2-mediated stress response, interferon, p53, cell cycle regulation, and lipid peroxidation. These results showed a similar process qualitatively and quantitatively for all three cell types from human sources, urothelial (1T1), keratinocyte (HEK001), and bronchial epithelial (HBE) cells, and support a mode of action involving cytotoxicity and regenerative proliferation [271]. TGF- β is a multifunctional peptide related to arsenic-associated carcinogenesis and controls proliferation, differentiation, and other functions in many cell types [272]. Another important protein TNF-related apoptosis-inducing ligand (TRAIL) has been shown to induce apoptosis and is linked to arsenic-related apoptotic tissue death [273]. Arsenic is known to influence mitochondrial permeability, transition pore opening, and loss of the mitochondrial trans-membrane potential, which down-regulates the Bcl-2 protein and activates caspase-3 activity (Figure 9-5). The disruption of mitosis was shown to be due to interference with tubulin polymerization and disruption of mitotic spindles, inactivation of centrosomes, formation of aberrant spindles, and blockage of chromosome segregation [274–277]. Another study assessed the role of Akt in the cell death induced by As_2O_3 . As_2O_3 caused dose-dependent apoptotic cell death [278]. As_2O_3 activated caspase-3 and -9 and PARP cleavage in a dose-dependent manner. Altered mitochondrial membrane potential and an increased protein level of Bax indicated involvement of mitochondria [278] (Figure 9-5). As₂O₃ decreased the levels of p-Akt (Ser473), p-Akt (Thr308), and p-GSK- 3β (Ser9), suggesting that As_2O_3 inactivated Akt kinase. These results demonstrated that inhibition of PI3K/Akt signaling was involved in As₂O₃-induced apoptosis of gastric cancer SGC-7901 cells [278]. The arsenic-associated carcinogenesis mechanism is summarized in Figure 9-4.

Arsenic induces the expression of angiogenesis-related factors through PI3K and MAPK pathways in SV-HUC-1 human uroepithelial cells [279]. Several important angiogenesis related-factors, including cyclooxygenase-2 (COX-2), vascular endothelial growth factor (VEGF), and hypoxia-inducible factor- 1α (HIF- 1α) were up-regulated and PI3K/AKT and MAPK signal pathways were activated in the human uroepithelial cell line (SV-HUC-1) [279]. In conclusion, arsenite-induced COX-2, VEGF, and HIF-1 α expressions, mediated partially by ROS, were regulated by MAPK and PI3K/AKT signaling pathways in human uroepithelial cells [279,280]. This study also evaluated the changes in the expression and release of cytokines in Caco-2 cells, widely used as an intestinal epithelial model. Differentiated cells were exposed to 1μ M of As(III), 0.1μ M of monomethylarsonous acid (MMA(III)), and 1μ M of dimethylarsinous acid (DMA(III)) for 2, 4, 6, and 24h. Additionally, the effect of arsenic co-exposure with lipopolysaccharide (LPS, 10 ng/mL) was evaluated [281]. The results of arsenic exposure of human carcinoma Caco-2 cells, widely used as an intestinal epithelial model, showed that arsenic increases the expression and release of the pro-inflammatory cytokines TNF- α , IL-6, and IL-8 [281]. Arsenite also caused malignant transformation of cells. Another study has also revealed the molecular mechanisms underlying inflammation during neoplastic transformation induced in HBE cells. The results showed that, on acute or chronic exposure to arsenite, HBE cells overexpressed the pro-inflammatory cytokines IL-6, IL-8, and IL-1β. The data also indicated that HIF-2α was involved in arsenite-induced inflammation. Moreover, IL-6 and IL-8 were essential for the malignant progression of arsenite-transformed HBE cells. The results provided a link between the inflammatory response and the acquisition of a malignant transformed phenotype by cells chronically exposed to arsenite [282]. A further study aimed to evaluate global miRNA and mRNA expression changes induced by a metal mixture containing NaAsO₂, CdCl₂, Pb($C_2H_3O_2$)₂·3H₂O and to predict possible metal-associated disease development under these conditions [283]. Results showed that this metal mixture led to an miRNA expression profile that may be responsible for the mRNA expression changes observed under experimental conditions. The proteins coded by these mRNAs are shown to be involved in metal-associated responses to apoptotic cell death, cellular proliferation, and cancer [283]. Different species of inorganic and organic arsenical compounds are shown to induce epigenetic and genetic events in human cultured cells, which may be implicated in the carcinogenesis processes (Figure 9-6). As(III) can also alter the expression of has-miR-210, -22, -34a, -221, and -222 in human lymphoblastoid cells [284]. As(III) exposure also resulted in the reduction of miR-200 in immortalized p53-knocked down human bronchial epithelial cells (HBECs) [285].

The mir group (miRNA) of sequences is microRNAs, which are also known as hsa-mir. This mammalian microRNA is encoded by the respective *MIR* gene. MIRN21 was one of the first mammalian microRNAs identified. A recent interesting study suggested that As(III) induces MAPK activation, phosphorylation of ATF-2 and c-Jun, and elevated IL-8 release in the human BEAS 2B cell line [286]. Induction of p53-independent expression of GADD45 protein (a G2/M cell cycle checkpoint protein) by As(III) has been shown in the human BEAS 2B line [287]. Stabilization of GADD45 alpha mRNA As(III) through nucleoli is demonstrated in the human BEAS 2B line [288]. As(III) significantly enhanced centrosome amplification in p53-compromised cells. Resistance to arsenite-induced G2/M cell cycle arrest and arsenite-induced apoptosis in p53-compromised cells has been observed [289]. A higher reduction in

cell viability in BEAS-2B cells (with functional p53) vs p53 compromised cells (H1355 or p53 inhibited BEAS-2B are noticed, after a low dose (0.1–5 μ M) arsenic exposure. There was a significant reduction in arsenic-induced apoptosis in p53 compromised cells. A reduction in Gadd45a expressions and increased centrosomal abnormalities were observed in arsenite-treated p53 compromised cells [289]. As(III) (\leq 0.5 μ M) increased cell viability and resulted in the down-regulation of APE1 and Pol β mRNA (above 1 μ M) in GM847-immortalized human lung fibroblast [290]. As(III) also produces a series of molecular and signaling phenomena. It increased plating efficiency (cell growth advantage), micronuclei incidence (marker of chromosomal instability), gene amplification (PALA resistance), invasive capabilities, anchorage-independent growth (oncogenic transformation), loss of β 4 integrin expression, up-regulation of phosphorylation of Rb and ERK, and decreased expression of p53 protein in h-TERT-immortalized human small airway epithelial cells [291].

9.5 Conclusions and Future Directions

The inevitable scarcity of ground water and the unavoidable situation of increased use of deeper earth water is gradually increasing the possibility of unaffected areas and their associated populations becoming exposed to arsenic in the future. Inorganic arsenic and its organic species exert their toxicities in different ways at the cellular and molecular levels. These include modulation of several genetic and epigenetic factors, which are directly or indirectly associated with the cellular transformation process. Arsenicals are highly potent in initiating oxidative stress. Reactive oxygen species or nitrogen species can modulate cellular signaling pathways at nuclear and cellular levels. The consequence is an impaired differentiation/ proliferation balance, resulting in cellular transformation. This chapter discusses several pathways explored by a number of researchers and scientists using cell culture or animal models or human epidemiological data. Nonetheless, it cannot be concluded that all possible mechanisms are occurring at the same time in the human carcinogenesis process in a particular location and with specific nutritional and environmental conditions. Rather, it is reasonable to believe that few or even one of those mechanisms are sufficient to initiate carcinogenesis pathways. But it is unequivocal that the greater the number of pathways involved, the greater the chances of disease pathogenesis with higher degrees of severity within a shorter period. Better investigation and a comprehensive understanding the disease mechanism will enable us to gain a better handle on a reliable therapeutic approach. It would of course be extremely beneficial for the millions of affected people around the world.

Significant new and emerging research constitute the present and future directions in the determination of the mechanisms of arsenic toxicity. Mechanistically, elucidation of the overlapping zone of the epigenetic, genetic, and cellular signal transduction pathways will pave the way in the search for important molecules that unequivocally result in arsenic-related carcinogenesis (Figure 9–7). Discovery of the molecules participating in tumorigenesis/carcinogenesis mechanisms will enable scientists to focus on obvious therapeutic targets. Recent advancements in neutragenomic and neutraceutical studies also might be a great help to counteract arsenic toxicity at the therapeutic and/or prevention level(s). Future studies on sociodemographic profiles, nutritional status, and, above all, genetic susceptibility/predisposition, will also help to evaluate the possible progression of the toxicity mechanism in affected individuals. All the above would consolidate a better therapeutic approach.

Acknowledgments

The author sincerely wishes health and long life for arsenic-exposed populations. The help from PhD students Nirmallya Acharyya (institutional fellow), Sk. Sajed Ali (University Grants Commission, New Delhi; UGC-MANF fellow), and Bimal Dev, who is working on arsenic toxicology, is gratefully acknowledged. A small number of representative studies/references have been explored here, but it is sincerely acknowledged that the conceptual basis of this discussion is built on and highly motivated by many additional significant contributions from renowned investigators/scientists around the world.

References

- Environmental Protection Agency. Integrated Risk Information System (IRIS): arsenic, inorganic. http://www.epa.gov/iris/subst/0278.htm; 1998.
- [2] US Environmental Protection Agency Health assessment document for inorganic arsenic. EPA/540/1-86/020. Washington, DC: Environmental Criteria and Assessment Office, Office of Health and Environmental Assessment, Office of Research and Development; 1984.
- [3] Fontham ET, Thun MJ, Ward E, Portier KM, Balch AJ, Delancey JO, et al. ACS Cancer and the Environment Subcommittee. American Cancer Society perspectives on environmental factors and cancer. CA Cancer J Clin 2009;59:343–51.
- [4] Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for arsenic. Atlanta, GA: US Department of Health and Human Services, Public Health Services; 2007.
- [5] Evaluation of carcinogenic risks to humans: overall evaluations of carcinogenicity: an updating of International Agency for Research on Cancer (IARC) Monographs, vols 1 to 42. Suppl. 7, pp. 32–42; 1987.
- [6] A review of human carcinogens: arsenic, metals, fibres, and dusts. IARC monograph on the evaluation of carcinogenic risks to humans 2012, vol. 100(Part C):11–465.
- [7] Netherlands National Committee of the IAH. In: Appelo T, editor. Arsenic in groundwater: a world problem. Seminar. Utrecht: International Groundwater Resources Assessment Center; 2008. 29 November 2006.
- [8] World Health Organization. Guidelines for drinking water. Available at: http://www.who.int/water_sani-tation_health/dwq/arsenicsum.pdf; 2003.
- [9] Health Risk Assessment Guidance for Metals. Available at: <www.herag.net>.
- [10] Health Risk Assessment Guidance for Metals (HERAG). Fact sheet: mutagenicity. Available from: http://www.icmm.com/document/265; 2007.
- [11] Koestler DC, Avissar-Whiting M, Houseman EA, Karagas MR, Marsit CJ. Differential DNA methylation in umbilical cord blood of infants exposed to low levels of arsenic in utero. Environ Health Perspect 2013;121:971–7.
- [12] Germolec DR, Spalding J, Boorman GA, Wilmer JL, Yoshida T, Simeonova PP, et al. Arsenic can mediate skin neoplasia by chronic stimulation of keratinocyte-derived growth factors. Mutat Res 1997;386:209–18.
- [13] Germolec DR, Spalding J, Yu HS, Chen GS, Simeonova PP, Humble MC, et al. Arsenic enhancement of skin neoplasia by chronic stimulation of growth factors. Am J Pathol 1998;153:1775–85.
- [14] Rossman TG, Uddin AN, Burns FJ, Bosland MC. Arsenite is a cocarcinogen with solar ultraviolet radiation for mouse skin: an animal model for arsenic carcinogenesis. Toxicol Appl Pharmacol 2001;176:64–71.

- [15] National Research Council. Arsenic in drinking water. National Research Council, National Academy Press, Washington, DC.
- [16] Waalkes MP, Ward JM, Liu J, Diwan BA. Transplacental carcinogenicity of inorganic arsenic in the drinking water: induction of hepatic, ovarian, pulmonary, and adrenal tumors in mice. Toxicol Appl Pharmacol 2003;186:7-17.
- [17] Waalkes MP, Ward JM, Diwan BA. Induction of tumors of the liver, lung, ovary and adrenal in adult mice after brief maternal gestational exposure to inorganic arsenic: promotional effects of postnatal phorbol ester exposure on hepatic and pulmonary, but not dermal cancers. Carcinogenesis 2004;25:133–41.
- [18] Léonard A, Lauwerys RR. Carcinogenicity, teratogenicity and mutagenicity of arsenic. Mutat Res 1980;75:49-62.
- [19] Bencko V, Symon K. Health aspects of burning coal with a high arsenic content. I. Arsenic in hair, urine, and blood in children residing in a polluted area. Environ Res 1977;13:378–85.
- [20] Bencko V, Rames J, Fabiánová E, Pesek J, Jakubis M. Ecological and human health risk aspects of burning arsenic-rich coal. Environ Geochem Health 2009;31:239–43.
- [21] Hubaux R, Becker-Santos DD, Enfield KS, Rowbotham D, Lam S, Lam WL, et al. Molecular features in arsenic-induced lung tumors. Mol Cancer 2013;12:20.
- [22] Brown KG, Boyle KE, Chen CW, Gibb HJ. A dose-response analysis of skin cancer from inorganic arsenic in drinking water. Risk Anal 1989;9:519–28.
- [23] Brown KG, Guo HR, Kuo TL, Greene HL. Skin cancer and inorganic arsenic: uncertainty-status of risk. Risk Anal 1997;17:37-42.
- [24] Chiou HY, Chiou ST, Hsu YH, Chou YL, Tseng CH, Wei ML, et al. Incidence of transitional cell carcinoma and arsenic in drinking water: a follow-up study of 8,102 residents in an arseniasis-endemic area in northeastern Taiwan. Am J Epidemiol 2001;153:411–18.
- [25] Viren J, Silvers A. Nonlinearity in the lung cancer dose-response for airborne arsenic: apparent confounding by year of hire in evaluating lung cancer risks from arsenic exposure in Tacoma smelter workers. Regul Toxicol Pharmacol 1999;30:117–29.
- [26] Guha Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborty D, et al. Arsenic levels in drinking water and the prevalence of skin lesions in West Bengal, India. Int J Epidemiol 1998;27:871–7.
- [27] Calderón J, Navarro ME, Jimenez-Capdeville ME, Santos-Diaz MA, Golden A, Rodriguez-Leyva I, et al. Exposure to arsenic and lead and neuropsychological development in Mexican children. Environ Res 2001;85:69–76.
- [28] Chen CJ, Hsueh YM, Lai MS, Shyu MP, Chen SY, Wu MM, et al. Increased prevalence of hypertension and long-term arsenic exposure. Hypertension 1995;25:53–60.
- [29] Kitchin KT. Recent advances in arsenic carcinogenesis: modes of action, animal model systems, and methylated arsenic metabolites. Toxicol Appl Pharmacol 2001;172:249–61.
- [30] Rossman TG. Mechanism of arsenic carcinogenesis: an integrated approach. Mutat Res 2003;533:37-65.
- [31] Yamamura, S. Drinking water guidelines and standards—World Health Organization. Available at: <www.who.int/water_sanitation_health/dwq/arsenicun5.pdf>; n.d.
- [32] Vahter M. Species differences in the metabolism of arsenic compounds. Appl Organomet Chem 1994;8:141–5.
- [33] Vahter M, Marafante E. Reduction and binding of arsenic in marmoset monkeys. Arch Toxicol 1985;57:119–24.
- [34] Chiou HY, Hsueh YM, Hsieh LI, Hsu YH, Hsieh FI, Wei ML, et al. Arsenic methylation capacity, body retention and null genotypes of glutathione S-transferase M1 and T1 among current arsenic-exposed residents in Taiwan. Mutat Res 1997;386:197–207.
- [35] Bhattacharjee P, Chatterjee D, Singh KK, Giri AK. Systems biology approaches to evaluate arsenic toxicity and carcinogenicity: an overview. Int J Hyg Environ Health 2013;216:574–86.

- [36] Sumi D, Himeno S. Role of arsenic (+3 oxidation state) methyltransferase in arsenic metabolism and toxicity. Biol Pharm Bull 2012;35:1870–5.
- [37] Nishikawa T, Wanibuchi H, Ogawa M, Kinoshita A, Morimura K, Hiroi T, et al. Promoting effects of monomethylarsonic acid, dimethylarsinic acid and trimethylarsine oxide on induction of rat liver preneoplastic glutathione S-transferase placental form positive foci: a possible reactive oxygen species mechanism. Int J Cancer 2002;100:136–9.
- [38] Eguchi N, Kuroda K, Okamoto A, Horiguchi S. Metabolites of arsenic induced tetraploids and mitotic arrest in cultured cells. Arch Environ Contam Toxicol 1997;32:141–5.
- [39] Chen PH, Lan CC, Chiou MH, Hsieh MC, Chen GS. Effects of arsenic and UVB on normal human cultured keratinocytes: impact on apoptosis and implication on photocarcinogenesis. Chem Res Toxicol 2005;18:139–44.
- [40] Pellizzari ED, Clayton CA. Assessing the measurement precision of various arsenic forms and arsenic exposure in the National Human Exposure Assessment Survey. Environ Health Perspect 2006;114:220–7.
- [41] Shen J, Wanibuchi H, Salim EI, Wei M, Kinoshita A, Yoshida K, et al. Liver tumorigenicity of trimethylarsine oxide in male Fischer 344 rats—association with oxidative DNA damage and enhanced cell proliferation. Carcinogenesis 2003;24:1827–35.
- [42] Wanibuchi H, Salim EI, Kinoshita A, Shen J, Wei M, Morimura K, et al. Understanding arsenic carcinogenicity by the use of animal models. Toxicol Appl Pharmacol 2004;198:366–76.
- [43] Morikawa T, Wanibuchi H, Morimura K, Ogawa M, Fukushima S. Promotion of skin carcinogenesis by dimethylarsinic acid in keratin (K6)/ODC transgenic mice. Jpn J Cancer Res 2000;91:579–81.
- [44] Sakai M, Muramatsu M. Regulation of GST-P gene expression during hepatocarcinogenesis. Methods Enzymol 2005;401:42–61.
- [45] Kern AD, Oliveira MA, Coffino P, Hackert ML. Structure of mammalian ornithine decarboxylase at 1.6 A resolution: stereochemical implications of PLP-dependent amino acid decarboxylases. Structure 1999;7:567–81.
- [46] Gerner EW, Meyskens Jr. FL. Polyamines and cancer: old molecules, new understanding. Nat Rev Cancer 2004;4:781–92.
- [47] Seike N, Wanibuchi H, Morimura K, Nishikawa T, Kishida H, Nakae D, et al. Lack of promoting effect due to oral administration of dimethylarsinic acid on rat lung carcinogenesis initiated with N-bis(2hydroxypropyl)nitrosamine. Cancer Lett 2002;175:113–19.
- [48] Wang JP, Qi L, Moore MR, Ng JC. A review of animal models for the study of arsenic carcinogenesis. Toxicol Lett 2002;133:17–31.
- [49] Sorlini S, Gialdini F, Stefan M. UV/H2O2 oxidation of arsenic and terbuthylazine in drinking water. Environ Monit Assess 2014;186:1311–16.
- [50] Rossman TG, Uddin AN, Burns FJ, Bosland MC. Arsenite cocarcinogenesis: an animal model derived from genetic toxicology studies. Environ Health Perspect 2002;110:749–52.
- [51] Burns FJ, Uddin AN, Wu F, Nádas A, Rossman TG. Arsenic-induced enhancement of ultraviolet radiation carcinogenesis in mouse skin: a dose-response study. Environ Health Perspect 2004;112:599–603.
- [52] Wanibuchi H, Yamamoto S, Chen H, Yoshida K, Endo G, Hori T, et al. Promoting effects of dimethylarsinic acid on N-butyl-N-(4-hydroxybutyl)nitrosamine-induced urinary bladder carcinogenesis in rats. Carcinogenesis 1996;17:2435–9.
- [53] Wei M, Wanibuchi H, Salim EI, Yamamoto S, Yoshida K, Endo G, et al. Promotion of NCI-Black-Reiter male rat bladder carcinogenesis by dimethylarsinic acid an organic arsenic compound. Cancer Lett 1998;134:29–36.
- [54] Yamamoto S, Konishi Y, Matsuda T, Murai T, Shibata MA, Matsui-Yuasa I, et al. Cancer induction by an organic arsenic compound, dimethylarsinic acid (cacodylic acid), in F344/DuCrj rats after pretreatment with five carcinogens. Cancer Res 1995;55:1271–6.

- [55] Yamanaka K, Katsumata K, Ikuma K, Hasegawa A, Nakano M, Okada S. The role of orally administered dimethylarsinic acid, a main metabolite of inorganic arsenics, in the promotion and progression of UVBinduced skin tumorigenesis in hairless mice. Cancer Lett 2000;152:79–85.
- [56] Vahter M. Biotransformation of trivalent and pentavalent inorganic arsenic mice and rats. Environ Res 1981;25:286–93.
- [57] Suzuki S, Arnold LL, Pennington KL, Kakiuchi-Kiyota S, Chen B, Lu X, et al. Effects of co-administration of dietary sodium arsenate and 2,3-dimercaptopropane-1-sulfonic acid (DMPS) on the rat bladder epithelium. Toxicology 2012;299:155–9.
- [58] Wei M, Wanibuchi H, Yamamoto S, Li W, Fukushima S. Urinary bladder carcinogenicity of dimethylarsinic acid in male F344 rats. Carcinogenesis 1999;20:1873–6.
- [59] Wei M, Wanibuchi H, Morimura K, Iwai S, Yoshida K, Endo G, et al. Carcinogenicity of dimethylarsinic acid in male F344 rats and genetic alterations in induced urinary bladder tumors. Carcinogenesis 2002;23:1387–97.
- [60] Salim EI, Wanibuchi H, Morimura K, Wei M, Mitsuhashi M, Yoshida K, et al. Carcinogenicity of dimethylarsinic acid in p53 heterozygous knockout and wild-type C57BL/6J mice. Carcinogenesis 2003;24:335–42.
- [61] Vega L, Styblo M, Patterson R, Cullen W, Wang C, Germolec D. Differential effects of trivalent and pentavalent arsenicals on cell proliferation and cytokine secretion in normal human epidermal keratinocytes. Toxicol Appl Pharmacol 2001;172:225–32.
- [62] Komissarova EV, Rossman TG. Arsenite induced poly(ADP-ribosyl)ation of tumor suppressor P53 in human skin keratinocytes as a possible mechanism for carcinogenesis associated with arsenic exposure. Toxicol Appl Pharmacol 2010;243:399–404.
- [63] Huang Y, Zhang J, McHenry KT, Kim MM, Zeng W, Lopez-Pajares V, et al. Induction of cytoplasmic accumulation of p53: a mechanism for low levels of arsenic exposure to predispose cells for malignant transformation. Cancer Res 2008;68:9131–6.
- [64] Yedjou CG, Tchounwou PB. Modulation of p53, c-fos, RARE, cyclin A, and cyclin D1 expression in human leukemia (HL-60) cells exposed to arsenic trioxide. Mol Cell Biochem 2009;331:207–14.
- [65] Arnold LL, Eldan M, van Gemert M, Capca C, Cohen SM. Chronic studies evaluating the carcinogenicity of monomethylarsonic acid in rats and mice. Toxicology 2003;190:197–219.
- [66] Cohen SM, Arnold LL, Eldan M, Lewis AS, Beck BD. Methylated arsenicals: the implications of metabolism and carcinogenicity studies in rodents to human risk assessment. Crit Rev Toxicol 2006;36:99–133.
- [67] Acharyya N, Ali SS, Deb B, Chattopadhyay S, Maiti S. Green tea (*Camellia sinensis*) alleviates arsenicinduced damages to DNA and intestinal tissues in rat and *in situ* intestinal-loop by reinforcing antioxidant system. Environ Toxicol 2014 doi: 10.1002/tox.21977. [Epub ahead of print].
- [68] Muñiz Ortiz JG, Wallace KA, Leinisch F, Kadiiska MB. Catalase has a key role in protecting cells from the genotoxic effects of monomethylarsonous acid: a highly active metabolite of arsenic. Environ Mol Mutagen 2013;54:317–26.
- [69] Mishra D, Flora SJ. Differential oxidative stress and DNA damage in rat brain regions and blood following chronic arsenic exposure. Toxicol Ind Health 2008;24:247–56.
- [70] Maiti S, Chattopadhyay S, Deb B, Samanta T, Maji G, Pan B, et al. Antioxidant and metabolic impairment result in DNA damage in arsenic-exposed individuals with severe dermatological manifestations in Eastern India. Environ Toxicol 2012;27:342–50.
- [71] Mishra D, Mehta A, Flora SJ. Reversal of arsenic-induced hepatic apoptosis with combined administration of DMSA and its analogues in guinea pigs: role of glutathione and linked enzymes. Chem Res Toxicol 2008;21:400–7.
- [72] Pei Q, Ma N, Zhang J, Xu W, Li Y, Ma Z, et al. Oxidative DNA damage of peripheral blood polymorphonuclear leukocytes, selectively induced by chronic arsenic exposure, is associated with extent of arsenicrelated skin lesions. Toxicol Appl Pharmacol 2013;266:143–9.

- [73] Matsui M, Nishigori C, Toyokuni S, Takada J, Akaboshi M, Ishikawa M, et al. The role of oxidative DNA damage in human arsenic carcinogenesis: detection of 8-hydroxy-2'-deoxyguanosine in arsenic-related Bowen's disease. J Invest Dermatol 1999;113:26–31.
- [74] Wang CK, Lee HL, Chang H, Tsai MH, Kuo YC, Lin P. Enhancement between environmental tobacco smoke and arsenic on emphysema-like lesions in mice. J Hazard Mater 2012;221–222:256–63.
- [75] Wang X, Mandal AK, Saito H, Pulliam JF, Lee EY, Ke ZJ, et al. Arsenic and chromium in drinking water promote tumorigenesis in a mouse colitis-associated colorectal cancer model and the potential mechanism is ROS-mediated Wnt/β-catenin signaling pathway. Toxicol Appl Pharmacol 2012;262:11–21.
- [76] Hong GM, Bain LJ. Arsenic exposure inhibits myogenesis and neurogenesis in P19 stem cells through repression of the β-catenin signaling pathway. Toxicol Sci 2012;129:146–56.
- [77] Rana T, Bera AK, Das S, Bhattacharya D, Pan D, Das SK. Metabolic adaptations to arsenic-induced oxidative stress in male wistar rats. J Biochem Mol Toxicol 2012;26:109–16.
- [78] Arnold LL, Suzuki S, Yokohira M, Kakiuchi-Kiyota S, Pennington KL, Cohen SM. Time course of urothelial changes in rats and mice orally administered arsenite. Toxicol Pathol 2013;42:855–62.
- [79] Dodmane PR, Arnold LL, Pennington KL, Thomas DJ, Cohen SM. Effect of dietary treatment with dimethylarsinous acid (DMA(III)) on the urinary bladder epithelium of arsenic (+3 oxidation state) methyltransferase (As3mt) knockout and C57BL/6 wild type female mice. Toxicology 2013;305:130–5.
- [80] Agusa T, Fujihara J, Takeshita H, Iwata H. Individual variations in inorganic arsenic metabolism associated with AS3MT genetic polymorphisms. Int J Mol Sci 2011;12:2351–82.
- [81] Tokar EJ, Diwan BA, Thomas DJ, Waalkes MP. Tumors and proliferative lesions in adult offspring after maternal exposure to methylarsonous acid during gestation in CD1 mice. Arch Toxicol 2012;86:975–82.
- [82] Chattopadhyay S, Deb B, Maiti S. Hepatoprotective role of vitamin B(12) and folic acid in arsenic intoxicated rats. Drug Chem Toxicol 2012;35:81–8.
- [83] Flora SJ, Mehta A. Monoisoamyl dimercaptosuccinic acid abrogates arsenic-induced developmental toxicity in human embryonic stem cell-derived embryoid bodies: comparison with in vivo studies. Biochem Pharmacol 2009;78:1340–9.
- [84] Noreault-Conti TL, Fellows A, Jacobs JM, Trask HW, Strom SC, Evans RM, et al. Arsenic decreases RXRα-dependent transcription of CYP3A and suppresses immune regulators in hepatocytes. Int Immunopharmacol 2012;12:651-6.
- [85] Naraharisetti SB, Aggarwal M, Sarkar SN, Malik JK. Concurrent sub acute exposure to arsenic through drinking water and malathion via diet in male rats: effects on hepatic drug-metabolizing enzymes. Arch Toxicol 2008;82:543–51.
- [86] Anwar-Mohamed A, Abdelhamid G, Amara IE, El-Kadi AO. Differential modulation of aryl hydrocarbon receptor regulated enzymes by arsenite in the kidney, lung, and heart of C57BL/6 mice. Arch Toxicol 2012;86:897–910.
- [87] Tokar EJ, Person RJ, Sun Y, Perantoni AO, Waalkes MP. Chronic exposure of renal stem cells to inorganic arsenic induces a cancer phenotype. Chem Res Toxicol 2013;26:96–105.
- [88] Medina-Díaz IM, Estrada-Muñiz E, Reyes-Hernández OD, Ramírez P, Vega L, Elizondo G. Arsenite and its metabolites, MMA(III) and DMA(III), modify CYP3A4, PXR and RXR alpha expression in the small intestine of CYP3A4 transgenic mice. Toxicol Appl Pharmacol 2009;239:162–8.
- [89] Lu K, Cable PH, Abo RP, Ru H, Graffam ME, Schlieper KA, et al. Gut microbiome perturbations induced by bacterial infection affect arsenic biotransformation. Chem Res Toxicol 2013;26:1893–903.
- [90] Hall LL, George SE, Kohan MJ, Styblo M, Thomas DJ. In vitro methylation of inorganic arsenic in mouse intestinal cecum. Toxicol Appl Pharmacol 1997;147:101–9.
- [91] Rowland IR, Davies MJ. In vitro metabolism of inorganic arsenic by the gastro-intestinal microflora of the rat. J Appl Toxicol 1981;1:278–83.

- [92] Van de Wiele T, Gallawa CM, Kubachka KM, Creed JT, Basta N, Dayton EA, et al. Arsenic metabolism by human gut microbiota upon in vitro digestion of contaminated soils. Environ Health Perspect 2010;118:1004–9.
- [93] Wlodarczyk B, Spiegelstein O, Hill D, Le XC, Finnell RH. Arsenic urinary speciation in Mthfr deficient mice injected with sodium arsenate. Toxicol Lett 2012;215:214–18.
- [94] Ajees AA, Marapakala K, Packianathan C, Sankaran B, Rosen BP. Structure of an As(III) S-adenosylmethionine methyltransferase: insights into the mechanism of arsenic biotransformation. Biochemistry 2012;51:5476–85.
- [95] Ríos R, Santoyo ME, Cruz D, Delgado JM, Zarazúa S, Jiménez-Capdeville ME. Methyl group balance in brain and liver: role of choline on increased S-adenosyl methionine (SAM) demand by chronic arsenic exposure. Toxicol Lett 2012;215:110–18.
- [96] Reichard JF, Schnekenburger M, Puga A. Long term low-dose arsenic exposure induces loss of DNA methylation. Biochem Biophys Res Commun 2007;352:188–92.
- [97] Pogribny IP, Rusyn I. Environmental toxicants, epigenetics, and cancer. Adv Exp Med Biol 2013;754:215–32.
- [98] Herceg Z, Vaissière T. Epigenetic mechanisms and cancer: an interface between the environment and the genome. Epigenetics 2011;6:804–19.
- [99] Sinha D, Biswas J, Bishayee A. Nrf2-mediated redox signaling in arsenic carcinogenesis: a review. Arch Toxicol 2013;87:383–96.
- [100] Basu A, Mahata J, Gupta S, Giri AK. Genetic toxicology of a paradoxical human carcinogen, arsenic: a review. Mutat Res 2001;488:171–94.
- [101] Chan P, Huff J. Arsenic carcinogenesis in animals and in humans: mechanistic, experimental, and epidemiological evidence. Environ Carcinog Ecotoxicol Rev 1997;C15:83–122.
- [102] Huff J, Chan P, Nyska A. Is the human carcinogen arsenic carcinogenic to laboratory animals? Toxicol Sci 2000;55:17–23.
- [103] Tokar EJ, Benbrahim-Tallaa L, Ward JM, Lunn R, Sams II RL, Waalkes MP. Cancer in experimental animals exposed to arsenic and arsenic compounds. Crit Rev Toxicol 2010;40:912–27.
- [104] NRC. National Research Council: arsenic in the drinking water. Washington: National Academy Press; 1999. 310 pp.
- [105] Liu J, Benbrahim-Tallaa L, Qian X, Yu L, Xie Y, Boos J, et al. Further studies on aberrant gene expression associated with arsenic-induced malignant transformation in rat liver TRL1215 cells. Toxicol Appl Pharmacol 2006;216:407–15.
- [106] Oeste CL, Díez-Dacal B, Bray F, García de Lacoba M, de la Torre BG, Andreu D, et al. The C-terminus of H-Ras as a target for the covalent binding of reactive compounds modulating Ras-dependent pathways. PLOS ONE 2011;6:e15866.
- [107] Goodsell DS. The molecular perspective: the ras oncogene. Oncologist 1999;4:263-4.
- [108] Malumbres M, Barbacid M. RAS oncogenes: the first 30 years. Nat Rev Cancer 2003;3:459-65.
- [109] Solus JF, Kraft S. Ras, Raf, and MAP kinase in melanoma. Adv Anat Pathol 2013;20:217–26.
- [110] Luster MI, Wilmer JL, Germolec DR, Spalding J, Yoshida T, Gaido K, et al. Role of keratinocyte-derived cytokines in chemical toxicity. Toxicol Lett 1995;471:82–3.
- [111] Martinez-Finley EJ, Goggin SL, Labrecque MT, Allan AM. Reduced expression of MAPK/ERK genes in perinatal arsenic-exposed offspring induced by glucocorticoid receptor deficits. Neurotoxicol Teratol 2011;33:530–7.
- [112] Nohara K, Tateishi Y, Suzuki T, Okamura K, Murai H, Takumi S, et al. Late-onset increases in oxidative stress and other tumorigenic activities and tumors with a Ha-ras mutation in the liver of adult male C3H mice gestationally exposed to arsenic. Toxicol Sci 2012;129:293–304.

- [113] Benbrahim-Tallaa L, Waterland RA, Styblo M, Achanzar WE, Webber MM, Waalkes MP. Molecular events associated with arsenic-induced malignant transformation of human prostatic epithelial cells: aberrant genomic DNA methylation and K-ras oncogene activation. Toxicol Appl Pharmacol 2005;206:288–98.
- [114] Cannon RE, Spalding JW, Virgil KM, Faircloth RS, Humble MC, Lacks GD, et al. Induction of transgene expression in Tg.AC(v-Ha-ras) transgenic mice concomitant with DNA hypomethylation. Mol Carcinog 1998;21:244–50.
- [115] Takamiya T, Hosobuchi S, Asai K, Nakamura E, Tomioka K, Kawase M, et al. Restriction landmark genome scanning method using isoschizomers (MspI/HpaII) for DNA methylation analysis. Electrophoresis 2006;27:2846–56.
- [116] Oda M, Glass JL, Thompson RF, Mo Y, Olivier EN, Figueroa ME, et al. High-resolution genome-wide cytosine methylation profiling with simultaneous copy number analysis and optimization for limited cell numbers. Nucleic Acids Res 2009;37:3829–39.
- [117] Tseng YS, Tzeng CC, Huang CY, Chen PH, Chiu AW, Hsu PY, et al. Aurora-A overexpression associates with Ha-ras codon-12 mutation and blackfoot disease endemic area in bladder cancer. Cancer Lett 2006;241:93–101.
- [118] Kraus C, Liehr T, Hülsken J, Behrens J, Birchmeier W, Grzeschik KH, et al. Localization of the human betacatenin gene (CTNNB1) to 3p21: a region implicated in tumor development. Genomics 1994;23:272–4.
- [119] MacDonald BT, Tamai K, He X. Wnt/beta-catenin signaling: components, mechanisms, and diseases. Dev Cell 2009;17:9–26.
- [120] Pournara A, Holmlund T, Lu Y, Ceder R, Putnik M, Grafström R, et al. Arsenic-induced suppression of kidney cell proliferation and the transcriptional coregulator MAML1. Metallomics 2014;6:498–504.
- [121] Gold LI. The role for transforming growth factor-beta (TGF-beta) in human cancer. Crit Rev Oncog 1999;10:303–60.
- [122] Lin Z, Shengnan L, Fei W, Yingli S, Qingshan S, Wei S, et al. Dimethylarsinic acid (DMA(v)) changed the expressions of proliferative related factors and secretion of inflammatory cytokines in rat bladder. J Appl Toxicol 2014. http://dx.doi.org/doi:10.1002/jat.3001>.
- [123] He J, Wang M, Jiang Y, Chen Q, Xu S, Xu Q, et al. Chronic arsenic exposure and angiogenesis in human bronchial epithelial cells via the ROS/miR-199a-5p/HIF-1α/COX-2 pathway. Environ Health Perspect 2014;122:255–61.
- [124] Zuo Z, Ouyang W, Li J, Costa M, Huang C. Cyclooxygenase-2 (COX-2) mediates arsenite inhibition of UVB-induced cellular apoptosis in mouse epidermal Cl41 cells. Curr Cancer Drug Targets 2012;12:607-16.
- [125] Liu Y, Hock JM, Sullivan C, Fang G, Cox AJ, Davis KT, et al. Activation of the p38 MAPK/Akt/ERK1/2 signal pathways is required for the protein stabilization of CDC6 and cyclin D1 in low-dose arsenite-induced cell proliferation. J Cell Biochem 2010;111:1546–55.
- [126] Suzuki T, Nohara K. Long-term arsenic exposure induces histone H3 Lys9 dimethylation without altering DNA methylation in the promoter region of p16(INK4a) and down-regulates its expression in the liver of mice. J Appl Toxicol 2013;33:951–8.
- [127] Wang J, Lee JJ, Wang L, Liu DD, Lu C, Fan YH, et al. Value of p16INK4a and RASSF1A promoter hypermethylation in prognosis of patients with resectable non-small cell lung cancer. Clin Cancer Res 2004;10:6119–25.
- [128] Chen J, Liu H, Liu J, Qi J, Wei B, Yang J, et al. H3K9 methylation is a barrier during somatic cell reprogramming into iPSCs. Nat Genet 2013;45:34–42.
- [129] Rosenfeld JA, Wang Z, Schones DE, Zhao K, DeSalle R, Zhang MQ. Determination of enriched histone modifications in non-genic portions of the human genome. BMC Genomics 2009;10:143.
- [130] Du J, Zhou N, Liu H, Jiang F, Wang Y, Hu C, et al. Arsenic induces functional re-expression of estrogen receptor α by demethylation of DNA in estrogen receptor-negative human breast cancer. PLOS ONE 2012;7:e35957.

272 HANDBOOK OF ARSENIC TOXICOLOGY

- [131] Sherr CJ. Divorcing ARF and p53: an unsettled case. Nat Rev Cancer 2006;6:663-73.
- [132] Abida WM, Gu W. p53-Dependent and p53-independent activation of autophagy by ARF. Cancer Res 2008;68:352-7.
- [133] Liontos M, Pateras IS, Evangelou K, Gorgoulis VG. The tumor suppressor gene ARF as a sensor of oxidative stress. Curr Mol Med 2012;12:704–15.
- [134] Ngalame NN, Micciche AF, Feil ME, States JC. Delayed temporal increase of hepatic Hsp70 in ApoE knockout mice after prenatal arsenic exposure. Toxicol Sci 2013;131:225–33.
- [135] Singh PP, Singh M, Mastana SS. Genetic variation of apolipoproteins in North Indians. Hum Biol 2002;74:673-82.
- [136] van den Elzen P, Garg S, León L, Brigl M, Leadbetter EA, Gumperz JE, et al. Apolipoprotein-mediated pathways of lipid antigen presentation. Nature 2005;437:906–10.
- [137] Murphy ME. The HSP70 family and cancer. Carcinogenesis 2013;34:1181-8.
- [138] Jiang P, Du W, Yang X. p53 and regulation of tumor metabolism. J Carcinog 2013;12:21.
- [139] ATSDR. Agency for toxic substances and disease registry, toxicological profiles for arsenic. Atlanta: USDHHS; 2000.
- [140] Tokumoto M, Lee JY, Fujiwara Y, Uchiyama M, Satoh M. Inorganic arsenic induces apoptosis through downregulation of Ube2d genes and p53 accumulation in rat proximal tubular cells. J Toxicol Sci 2013;38:815–20.
- [141] Sandoval M, Morales M, Tapia R, del Carmen Alarcón L, Sordo M, et al. p53 response to arsenic exposure in epithelial cells: protein kinase B/Akt involvement. Toxicol Sci 2007;99:126–40.
- [142] Sun Y, Kojima C, Chignell C, Mason R, Waalkes MP. Arsenic transformation predisposes human skin keratinocytes to UV-induced DNA damage yet enhances their survival apparently by diminishing oxidant response. Toxicol Appl Pharmacol 2011;255:242–50.
- [143] de Martel C, Franceschi S. Infections and cancer: established associations and new hypotheses. Crit Rev Oncol Hematol 2009;70:183–94.
- [144] Brash DE, Rudolph JA, Somon JA, Lin A, McKenna GJ, Baden HP, et al. A role for sunlight in skin cancer: UV induced p53 mutations in squamous cell carcinoma. Proc Natl Acad Sci USA 1991;88:10124–8.
- [145] Popovicova J, Moser GJ, Goldsworthy TL, Tice RR. Carcinogenicity and co-carcinogenicity of sodium arsenite in p53 +/- male mice. Toxicologist 2000;54:134.
- [146] Wauson EM, Langan AS, Vorce RL. Sodium arsenite inhibits and reverses expression of adipogenic and fat cell-specific genes during in vitro adipogenesis. Toxicol Sci 2002;65:211–19.
- [147] Rossman TG, Wang Z. Expression cloning for arsenite-resistance resulted in isolation of tumorsuppressor *fau* cDNA: possible involvement of the ubiquitin system in arsenic carcinogenesis. Carcinogenesis 1999;20:311–16.
- [148] Budunova IV, Carbajal S, Kang H, Viaje A, Slaga TJ. Altered glucocorticoid receptor expression and function during mouse skin carcinogenesis. Mol Carcinog 1997;18:177–85.
- [149] Naranmandura H, Carew MW, Xu S, Lee J, Leslie EM, Weinfel M, et al. Comparative toxicity of arsenic metabolites in human bladder cancer EJ-1 cells. Chem Res Toxicol 2011;24:1586–96.
- [150] Cohen SM, Ohnishi T, Arnold LL, Le XC. Arsenic-induced bladder cancer in an animal model. Toxicol Appl Pharmacol 2007;222:258–63.
- [151] Waalkes MP, Liu J, Germolec DR, Trempus CS, Cannon RE, Tokar EJ, et al. Arsenic exposure in utero exacerbates skin cancer response in adulthood with contemporaneous distortion of tumor stem cell dynamics. Cancer Res 2008;68:8278–85.
- [152] Tokar EJ, Qu W, Waalkes MP. Arsenic, stem cells, and the developmental basis of adult cancer. Toxicol Sci 2011;120:S192–203.

- [153] Mukherjee S, Das D, Mukherjee M, Das AS, Mitra C. Synergistic effect of folic acid and vitamin B12 in ameliorating arsenic-induced oxidative damage in pancreatic tissue of rat. J Nutr Biochem 2006;17:319–27.
- [154] Gosse JA, Taylor VF, Jackson BP, Hamilton JW, Bodwell JE. Monomethylated trivalent arsenic species disrupt steroid receptor interactions with their DNA response elements at non-cytotoxic cellular concentrations. J Appl Toxicol 2013;34:498–505.
- [155] Das J, Ghosh J, Manna P, Sinha M, Sil PC. Taurine protects rat testes against NaAsO(2)-induced oxidative stress and apoptosis via mitochondrial dependent and independent pathways. Toxicol Lett 2009;187:201–10.
- [156] Flora SJ. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51:257–81.
- [157] Adachi M, Imai K. The proapoptotic BH3-only protein BAD transduces cell death signals independently of its interaction with Bcl-2. Cell Death Differ 2002;9:1240–7.
- [158] Philchenkov A. Caspases: potential targets for regulating cell death. J Cell Mol Med 2004;8:432-44.
- [159] Pachauri V, Mehta A, Mishra D, Flora SJ. Arsenic induced neuronal apoptosis in guinea pigs is Ca2+ dependent and abrogated by chelation therapy: role of voltage gated calcium channels. Neurotoxicology 2013;35:137-45.
- [160] Dwivedi N, Mehta A, Yadav A, Binukumar BK, Gill KD, Flora SJ. MiADMSA reverses impaired mitochondrial energy metabolism and neuronal apoptotic cell death after arsenic exposure in rats. Toxicol Appl Pharmacol 2011;256:241–8.
- [161] Fouad AA, Al-Mulhim AS, Jresat I. Telmisartan treatment attenuates arsenic-induced hepatotoxicity in mice. Toxicology 2012;300:149–57.
- [162] Perkins ND. Integrating cell-signalling pathways with NF-kappaB and IKK function. Nat Rev Mol Cell Biol 2007;8:49–62.
- [163] Avruch J, Khokhlatchev A, Kyriakis JM, Luo Z, Tzivion G, Vavvas D, et al. Ras activation of the Raf kinase: tyrosine kinase recruitment of the MAP kinase cascade. Recent Prog Horm Res 2001;56:127–55.
- [164] Hilger RA, Scheulen ME, Strumberg D. The Ras-Raf-MEK-ERK pathway in the treatment of cancer. Onkologie 2002;25:511–18.
- [165] Sharma M, Sharma S, Arora M, Kaul D. Regulation of cellular Cyclin D1 gene by arsenic is mediated through miR-2909. Gene 2013;522:60–4.
- [166] Zhang J, Stirling B, Temmerman ST, Ma CA, Fuss IJ, Derry JM, et al. Impaired regulation of NF-kappaB and increased susceptibility to colitis-associated tumorigenesis in CYLD-deficient mice. J Clin Invest 2006;116:3042–9.
- [167] Dangleben NL, Skibola CF, Smith MT. Arsenic immunotoxicity: a review. Environ Health 2013;12:73.
- [168] Saucedo LJ, Edgar BA. Filling out the Hippo pathway. Nat Rev Mol Cell Biol 2007;8:613–21.
- [169] Wu S, Huang J, Dong J, Pan D. hippo encodes a Ste-20 family protein kinase that restricts cell proliferation and promotes apoptosis in conjunction with salvador and warts. Cell 2003;114:445–56.
- [170] Li C, Srivastava RK, Elmets CA, Afaq F, Athar M. Arsenic-induced cutaneous hyperplastic lesions are associated with the dysregulation of Yap, a Hippo signaling-related protein. Biochem Biophys Res Commun 2013;438:607–12.
- [171] Hossain E, Ota A, Takahashi M, Karnan S, Damdindorj L, Konishi Y, et al. Arsenic upregulates the expression of angiotensin II Type I receptor in mouse aortic endothelial cells. Toxicol Lett 2013;220:70–5.
- [172] Li C, Qu X, Xu W, Qu N, Mei L, Liu Y, et al. Arsenic trioxide induces cardiac fibroblast apoptosis in vitro and in vivo by up-regulating TGF-β1 expression. Toxicol Lett 2013;219:223–30.
- [173] Zhang W, Feng H, Gao Y, Sun L, Wang J, Li Y, et al. Role of pigment epithelium-derived factor (PEDF) in arsenic-induced cell apoptosis of liver and brain in a rat model. Biol Trace Elem Res 2013;151:269–76.

- [174] Seol JG, Park WH, Kim ES, Jung CW, Hyun JM, Lee YY, et al. Potential role of caspase-3 and -9 in arsenic trioxide-mediated apoptosis in PCI-1 head and neck cancer cells. Int J Oncol 2001;18:249–55.
- [175] Pastoret A, Marcos R, Sampayo-Reyes A, Saucedo-Cardenas O, Lozano-Garza GH, Hernandez A. Inhibition of hepatocyte nuclear factor 1 and 4 alpha (HNF1α and HNF4α) as a mechanism of arsenic carcinogenesis. Arch Toxicol 2013;87:1001–12.
- [176] Boj SF, Petrov D, Ferrer J. Epistasis of transcriptomes reveals synergism between transcriptional activators Hnflalpha and Hnf4alpha. PLoS Genet 2010;6:e1000970.
- [177] Santangelo L, Marchetti A, Cicchini C, Conigliaro A, Conti B, Mancone C, et al. The stable repression of mesenchymal program is required for hepatocyte identity: a novel role for hepatocyte nuclear factor 4α. Hepatology 2011;53:2063–74.
- [178] Moi P, Chan K, Asunis I, Cao A, Kan YW. Isolation of NF-E2-related factor 2 (Nrf2), a NF-E2-like basic leucine zipper transcriptional activator that binds to the tandem NF-E2/AP1 repeat of the beta-globin locus control region. Proc Natl Acad Sci USA 1994;91:9926–30.
- [179] Kim HG, Kim DJ, Li S, Lee KY, Li X, Bode AM, et al. Polycomb (PcG) proteins, BMI1 and SUZ12, regulate arsenic-induced cell transformation. J Biol Chem 2012;287:31920–8.
- [180] Arrigo AP. Acetylation and methylation patterns of core histones are modified after heat or arsenite treatment of Drosophila tissue culture cells. Nucleic Acids Res 1983;11:1389–404.
- [181] Zykova TA, Zhu F, Lu C, Higgins L, Tatsumi Y, Abe Y, et al. Lymphokine-activated killer T-cell-originated protein kinase phosphorylation of histone H2AX prevents arsenite-induced apoptosis in RPMI7951 melanoma cells. Clin Cancer Res 2006;12:6884–93.
- [182] Li J, Gorospe M, Barnes J, Liu Y. Tumor promoter arsenite stimulates histone H3 phosphoacetylation of proto-oncogenes c-fos and c-jun chromatin in human diploid fibroblasts. J Biol Chem 2003;278:13183–91.
- [183] Ramirez T, Brocher J, Stopper H, Hock R. Sodium arsenite modulates histone acetylation, histone deacetylase activity and HMGN protein dynamics in human cells. Chromosoma 2008;117:147–57.
- [184] Bartel DP. MicroRNAs: genomics, biogenesis, mechanism, and function. Cell 2004;116:281–97.
- [185] Johnson SM, Grosshans H, Shingara J, Byron M, Jarvis R, Cheng A, et al. Ras is regulated by the let-7 microRNA family. Cell 2005;120:635–47.
- [186] Ren X, McHale CM, Skibola CF, Smith AH, Smith MT, Zhang L. An emerging role for epigenetic dysregulation in arsenic toxicity and carcinogenesis. Environ Health Perspect 2011;119:11–19.
- [187] Danaee H, Nelson HH, Liber H, Little JB, Kelsey KT. Low dose exposure to sodium arsenite synergistically interacts with UV radiation to induce mutations and alter DNA repair in human cells. Mutagenesis 2004;19:143–8.
- [188] Luzhna L, Kathiria P, Kovalchuk O. Micronuclei in genotoxicity assessment: from genetics to epigenetics and beyond. Front Genet 2013;4:131.
- [189] Wood RD. Mammalian nucleotide excision repair proteins and interstrand crosslink repair. Environ Mol Mutagen 2010;51:520–6.
- [190] Kit S. Thymidine kinase. Microbiol Sci 1985;2:369–75.
- [191] Mure K, Uddin AN, Lopez LC, Styblo M, Rossman TG. Arsenite induces delayed mutagenesis and transformation in human osteosarcoma cells at extremely low concentrations. Environ Mol Mutagen 2003;41:322–31.
- [192] Sculley DG, Dawson PA, Emmerson BT, Gordon RB. A review of the molecular basis of hypoxanthineguanine phosphoribosyltransferase (HPRT) deficiency. Hum Genet 1992;90:195–207.
- [193] Isabelle M, Moreel X, Gagné JP, Rouleau M, Ethier C, Gagné P, et al. Investigation of PARP-1, PARP-2, and PARG interactomes by affinity-purification mass spectrometry. Proteome Sci 2010;8:22.
- [194] Cleaver JE. Common pathways for ultraviolet skin carcinogenesis in the repair and replication defective groups of xeroderma pigmentosum. J Dermatol Sci 2000;23:1–11.

- [195] Drapkin R, Reardon JT, Ansari A, Huang JC, Zawel L, Ahn K, et al. Dual role of TFIIH in DNA excision repair and in transcription by RNA polymerase II. Nature 1994;368:769–72.
- [196] Drapkin R, Sancar A, Reinberg D. Where transcription meets repair. Cell 1994;77:9–12.
- [197] Andrew AS, Karagas MR, Hamilton JW. Decreased DNA repair gene expression among individuals exposed to arsenic in United States drinking water. Int J Cancer 2003;104:263–8.
- [198] Ahsan H, Chen Y, Wang Q, Slavkovich V, Graziano JH, Santella RM. DNA repair gene XPD and susceptibility to arsenic-induced hyperkeratosis. Toxicol Lett 2003;143:123–31.
- [199] McEachern MJ, Krauskopf A, Blackburn EH. Telomeres and their control. Annu Rev Genet 2000;34:331–58.
- [200] Blackburn EH. The end of the DNA line. Nat Struct Biol 2000;7:847–50.
- [201] Zhang TC, Schmitt MT, Mumford JL. Effects of arsenic on telomerase and telomeres in relation to cell proliferation and apoptosis in human keratinocytes and leukemia cells in vitro. Carcinogenesis 2003;24:1811–17.
- [202] Shay JW, Wright WE. Telomerase activity in human cancer. Curr Opin Oncol 1996;8:66–71.
- [203] Cong YS, Wright WE, Shay JW. Human telomerase and its regulation. Microbiol Mol Biol Rev 2002;66:407-25.
- [204] Chin L, Artandi SE, Shen Q, Tam A, Lee SL, Gottlieb GJ, et al. p53 deficiency rescues the adverse effects of telomere loss and cooperates with telomere dysfunction to accelerate carcinogenesis. Cell 1999;97:527–38.
- [205] González-Suárez E, Samper E, Flores JM, Blasco MA. Telomerase-deficient mice with short telomeres are resistant to skin tumorigenesis. Nat Genet 2000;26:114–17.
- [206] Lee TC, Wei ML, Chang WJ, Ho IC, Lo JF, Jan KY, et al. Elevation of glutathione levels and glutathione S-transferase activity in arsenic-resistant Chinese hamster ovary cells. In Vitro Cell Dev Biol 1989;25:442–8.
- [207] Harley CB, Futcher AB, Greider CW. Telomeres shorten during ageing of human fibroblasts. Nature 1990;345:458–60.
- [208] Blasco MA, Lee HW, Hande MP, Samper E, Lansdorp PM, DePinho RA, et al. Telomere shortening and tumor formation by mouse cells lacking telomerase RNA. Cell 1997;91:25–34.
- [209] Lee HW, Blasco MA, Gottlieb GJ, Horner II JW, Greider CW, DePinho RA. Essential role of mouse telomerase in highly proliferative organs. Nature 1998;392:569–74.
- [210] Liu L, Trimarchi JR, Navarro P, Blasco MA, Keefe DL. Oxidative stress contributes to arsenic-induced telomere attrition, chromosome instability, and apoptosis. J Biol Chem 2003;278:31998–2004.
- [211] Tsuji JS, Alexander DD, Perez V, Mink PJ. Arsenic exposure and bladder cancer: quantitative assessment of studies in human populations to detect risks at low doses. Toxicology 2014;317C:17–30.
- [212] Niedzwiecki MM, Hall MN, Liu X, Oka J, Harper KN, Slavkovich V, et al. A dose-response study of arsenic exposure and global methylation of peripheral blood mononuclear cell DNA in Bangladeshi adults. Environ Health Perspect 2013;121:1306–12.
- [213] Bhattacharjee P, Das N, Chatterjee D, Banerjee A, Das JK, Basu S, et al. Association of NALP2 polymorphism with arsenic induced skin lesions and other health effects. Mutat Res 2013;755:1–5.
- [214] Tschopp J, Martinon F, Burns K. NALPs: a novel protein family involved in inflammation. Nat Rev Mol Cell Biol 2003;4:95–104.
- [215] Bertin J, DiStefano PS. The PYRIN domain: a novel motif found in apoptosis and inflammation proteins. Cell Death Differ 2000;7:1273–4.
- [216] Lambrou A, Baccarelli A, Wright RO, Weisskopf M, Bollati V, Amarasiriwardena C, et al. Arsenic exposure and DNA methylation among elderly men. Epidemiology 2012;23:668–76.
- [217] Tajuddin SM, Amaral AF, Fernández AF, Rodríguez-Rodero S, Rodríguez RM, Moore LE, et al. Genetic and non-genetic predictors of LINE-1 methylation in leukocyte DNA. Environ Health Perspect 2013;121:650–6.

- [218] Yamanaka K, Hayashi H, Tachikawa M, Kato K, Hasegawa A, Oku N, et al. Metabolic methylation is a possible genotoxicity-enhancing process of inorganic arsenics. Mutat Res 1997;394:95–101.
- [219] Mourón SA, Grillo CA, Dulout FN, Golijow CD. Induction of DNA strand breaks, DNA-protein crosslinks and sister chromatid exchanges by arsenite in a human lung cell line. Toxicol In Vitro 2006;20:279–85.
- [220] Yamanaka K, Hayashi H, Kato K, Hasegawa A, Okada S. Involvement of preferential formation of apurinic/apyrimidinic sites in dimethylarsenic-induced DNA strand breaks and DNA-protein crosslinks in cultured alveolar epithelial cells. Biochem Biophys Res Commun 1995;207:244–9.
- [221] Ruiz-Ramos R, Lopez-Carrillo L, Rios-Perez AD, De Vizcaya-Ruíz A, Cebrian ME. Sodium arsenite induces ROS generation, DNA oxidative damage, HO-1 and c-Myc proteins, NF-kappaB activation and cell proliferation in human breast cancer MCF-7 cells. Mutat Res 2009;674:109–15.
- [222] Kitchin KT, Ahmad S. Oxidative stress as a possible mode of action for arsenic carcinogenesis. Toxicol Lett 2003;137:3–13.
- [223] Yamanaka K, Mizol M, Kato K, Hasegawa A, Nakano M, Okada S. Oral administration of dimethylarsinic acid, a main metabolite of inorganic arsenic, in mice promotes skin tumorigenesis initiated by dimethylbenz(a)anthracene with or without ultraviolet B as a promoter. Biol Pharm Bull 2001;24:510–14.
- [224] Mass MJ, Wang L. Arsenic alters cytosine methylation patterns of the promoter of the tumor suppressor gene p53 in human lung cells: a model for a mechanism of carcinogenesis. Mutat Res 1997;386:263–77.
- [225] Zhou X, Sun H, Ellen TP, Chen H, Costa M. Arsenite alters global histone H3 methylation. Carcinogenesis 2008;29:1831–6.
- [226] Jensen TJ, Novak P, Eblin KE, Gandolfi AJ, Futscher BW. Epigenetic remodeling during arsenical-induced malignant transformation. Carcinogenesis 2008;29:1500–8.
- [227] Hou L, Zhang X, Wang D, Baccarelli A. Environmental chemical exposures and human epigenetics. Int J Epidemiol 2012;41:79–105.
- [228] Reichard JF, Puga A. Effects of arsenic exposure on DNA methylation and epigenetic gene regulation. Epigenomics 2010;2:87–104.
- [229] Pilsner JR, Hall MN, Liu X, Ilievski V, Slavkovich V, Levy D, et al. Influence of prenatal arsenic exposure and newborn sex on global methylation of cord blood DNA. PLOS ONE 2012;7:e37147.
- [230] Liu X, Zheng Y, Zhang W, Zhang X, Lloyd-Jones DM, Baccarelli AA, et al. Blood methylomics in response to arsenic exposure in a low-exposed US population. J Expo Sci Environ Epidemiol 2014;24:145–9.
- [231] Takumi S, Aoki Y, Sano T, Suzuki T, Nohmi T, Nohara K. In vivo mutagenicity of arsenite in the livers of gpt delta transgenic mice. Mutat Res 2014;760:42–7.
- [232] Sturchio E, Colombo T, Boccia P, Carucci N, Meconi C, Minoia C, et al. Arsenic exposure triggers a shift in microRNA expression. Sci Total Environ 2014;472:672–80.
- [233] Melak D, Ferreccio C, Kalman D, Parra R, Acevedo J, Pérez L, et al. Arsenic methylation and lung and bladder cancer in a case-control study in northern Chile. Toxicol Appl Pharmacol 2014;274:225–31.
- [234] Cooper KL, Yager JW, Hudson LG. Melanocytes and keratinocytes have distinct and shared responses to ultraviolet radiation and arsenic. Toxicol Lett 2014;224:407–15.
- [235] Chanda S, Dasgupta UB, Guhamazumder D, Gupta M, Chaudhuri U, Lahiri S, et al. DNA hypermethylation of promoter of gene p53 and p16 in arsenic-exposed people with and without malignancy. Toxicol Sci 2006;89:431–7.
- [236] Suzuki T, Yamashita S, Ushijima T, Takumi S, Sano T, Michikawa T, et al. Genome-wide analysis of DNA methylation changes induced by gestational arsenic exposure in liver tumors. Cancer Sci 2013;104:1575-85.
- [237] Gribble MO, Voruganti VS, Cropp CD, Francesconi KA, Goessler W, Umans JG, et al. SLCO1B1 variants and urine arsenic metabolites in the Strong Heart Family Study. Toxicol Sci 2013;136:19–25.

- [238] Banerjee N, Paul S, Sau TJ, Das JK, Bandyopadhyay A, Banerjee S, et al. Epigenetic modifications of DAPK and p16 genes contribute to arsenic-induced skin lesions and nondermatological health effects. Toxicol Sci 2013;135:300–8.
- [239] Brocato J, Costa M. Basic mechanics of DNA methylation and the unique landscape of the DNA methylome in metal-induced carcinogenesis. Crit Rev Toxicol 2013;43:493–514.
- [240] Ray PD, Yosim A, Fry RC. Incorporating epigenetic data into the risk assessment process for the toxic metals arsenic, cadmium, chromium, lead, and mercury: strategies and challenges. Front Genet 2014;10:3389.
- [241] Wu CC, Huang YK, Chung CJ, Huang CY, Pu YS, Shiue HS, et al. Polymorphism of inflammatory genes and arsenic methylation capacity are associated with urothelial carcinoma. Toxicol Appl Pharmacol 2013;272:30–6.
- [242] Huang YC, Hung WC, Chen WT, Yu HS, Chai CY. Expression of WWOX and FHIT is downregulated by exposure to arsenite in human uroepithelial cells. Toxicol Lett 2013;220:118–25.
- [243] Moore LE, Smith AH, Eng C, DeVries S, Kalman D, Bhargava V, et al. P53 alterations in bladder tumors from arsenic and tobacco exposed patients. Carcinogenesis 2003;24:1785–91.
- [244] Shibata A, Ohneseit PF, Tsai YC, Spruck 3rd CH, Nichols PW, Chiang HS, et al. Mutational spectrum in the p53 gene in bladder tumors from the endemic area of black foot disease in Taiwan. Carcinogenesis 1994;15:1085–7.
- [245] Hernández-Zavala A, Córdova E, Del Razo LM, Cebrián ME, Garrido E. Effects of arsenite on cell cycle progression in a human bladder cancer cell line. Toxicology 2005;207:49–57.
- [246] Sherr CJ, Roberts JM. Inhibitors of mammalian G1 cyclin-dependent kinases. Genes Dev 1995;9:1149–63.
- [247] Bartek J, Lukas J. Pathways governing G1/S transition and their response to DNA damage. FEBS Lett 2001;490:117–22.
- [248] Snow ET. Metal carcinogenesis: mechanistic implications. Pharmacol Ther 1992;53:31-65.
- [249] Kapahi P, Takahashi T, Natoli G, Adams SR, Chen Y, Tsien RY, et al. Inhibition of NF-kappa B activation by arsenite through reaction with a critical cysteine in the activation loop of Ikappa B kinase. J Biol Chem 2000;275:36062–6.
- [250] Karin M, Lin A. NF-kappaB at the crossroads of life and death. Nat Immunol 2002;3:221-7.
- [251] Cavigelli M, Li WW, Lin A, Su B, Yoshioka K, Karin M. The tumor promoter arsenite stimulates AP-1 activity by inhibiting a JNK phosphatase. EMBO J 1996;15:6269–79.
- [252] Levy DE, Darnell Jr. JE. Stats: transcriptional control and biological impact. Nat Rev Mol Cell Biol 2002;3:651-62.
- [253] Hussein MR, Haemel AK, Wood GS. Apoptosis and melanoma: molecular mechanisms. J Pathol 2003;199:275-88.
- [254] Shaulian E, Karin M. AP-1 as a regulator of cell life and death. Nat Cell Biol 2002;4:E131-6.
- [255] Ivanov VN, Hei TK. Arsenite sensitizes human melanomas to apoptosis via tumor necrosis factor alphamediated pathway. J Biol Chem 2004;279:22747–58.
- [256] Ngalame NN, Tokar EJ, Person RJ, Xu Y, Waalkes MP. Aberrant microRNA expression likely controls RAS oncogene activation during malignant transformation of human prostate epithelial and stem cells by arsenic. Toxicol Sci 2014;138:268–77.
- [257] Kaul D, Sharma S, Sharma M, Arora M. Arsenic programmes cellular genomic-immunity through miR-2909 RNomics. Gene 2014;536:326–31.
- [258] Israël A. The IKK complex, a central regulator of NF-kappaB activation. Cold Spring Harb Perspect Biol 2010;2:a000158.
- [259] Hess J, Angel P, Schorpp-Kistner M. AP-1 subunits: quarrel and harmony among siblings. J Cell Sci 2004;117:5965-73.

- [260] Akira S, Nishio Y, Inoue M, Wang XJ, Wei S, Matsusaka T, et al. Molecular cloning of APRF, a novel IFN-stimulated gene factor 3 p91-related transcription factor involved in the gp130-mediated signaling pathway. Cell 1994;77:63–71.
- [261] Sun J, Yu M, Lu Y, Thakur C, Chen B, Qiu P, et al. Carcinogenic metalloid arsenic induces expression of mdig oncogene through JNK and STAT3 activation. Cancer Lett 2014;346:257–63.
- [262] Ling M, Li Y, Xu Y, Pang Y, Shen L, Jiang R, et al. Regulation of miRNA-21 by reactive oxygen speciesactivated ERK/NF-κB in arsenite-induced cell transformation. Free Radic Biol Med 2012;52:1508–18.
- [263] Shen L, Ling M, Li Y, Xu Y, Zhou Y, Ye J, et al. Feedback regulations of miR-21 and MAPKs via Pdcd4 and Spry1 are involved in arsenite-induced cell malignant transformation. PLOS ONE 2013;8:e57652.
- [264] Sharma M, Sharma S, Arora M, Kaul D. Regulation of cellular cyclin D1 gene by arsenic is mediated through miR-2909. Gene 2013;522:60–4.
- [265] Qian C, Wang Y, Chen Y, Zeng L, Zhang Q, Shuai X, et al. Suppression of pancreatic tumor growth by targeted arsenic delivery with anti-CD44v6 single chain antibody conjugated nanoparticles. Biomaterials 2013;34:6175–84.
- [266] Huang S, Guo S, Guo F, Yang Q, Xiao X, Murata M, et al. CD44v6 expression in human skin keratinocytes as a possible mechanism for carcinogenesis associated with chronic arsenic exposure. Eur J Histochem 2013;57:e1.
- [267] Xu Y, Tokar EJ, Waalkes MP. Arsenic-induced cancer cell phenotype in human breast epithelia is estrogen receptor-independent but involves aromatase activation. Arch Toxicol 2014;88:263–74.
- [268] Wang Z, Humphries B, Xiao H, Jiang Y, Yang C. Epithelial to mesenchymal transition in arsenictransformed cells promotes angiogenesis through activating β-catenin-vascular endothelial growth factor pathway. Toxicol Appl Pharmacol 2013;271:20–9.
- [269] Lau A, Zheng Y, Tao S, Wang H, Whitman SA, White E, et al. Arsenic inhibits autophagic flux, activating the Nrf2-Keap1 pathway in a p62-dependent manner. Mol Cell Biol 2013;33:2436–46.
- [270] Yue W, Santen RJ, Wang JP, Li Y, Verderame MF, Bocchinfuso WP, et al. Genotoxic metabolites of estradiol in breast: potential mechanism of estradiol induced carcinogenesis. J Steroid Biochem Mol Biol 2003;86:477–86.
- [271] Dodmane PR, Arnold LL, Kakiuchi-Kiyota S, Qiu F, Liu X, Rennard SI, et al. Cytotoxicity and gene expression changes induced by inorganic and organic trivalent arsenicals in human cells. Toxicology 2013;312:18–29.
- [272] Itoh K, Tong KI, Yamamoto M. Molecular mechanism activating Nrf2-Keap1 pathway in regulation of adaptive response to electrophiles. Free Radic Biol Med 2004;36:1208–13.
- [273] Lau A, Wang XJ, Zhao F, Villeneuve NF, Wu T, Jiang T, et al. A noncanonical mechanism of Nrf2 activation by autophagy deficiency: direct interaction between Keap1 and p62. Mol Cell Biol 2010;30:3275–85.
- [274] Li YM, Broome JD. Arsenic targets tubulins to induce apoptosis in myeloid leukemia cells. Cancer Res 1999;59:776–80.
- [275] Huang SC, Lee TC. Arsenite inhibits mitotic division and perturbs spindle dynamics in HeLa S3 cells. Carcinogenesis 1998;19:889–96.
- [276] Hut HM, Lemstra W, Blaauw EH, Van Cappellen GW, Kampinga HH, Sibon OC. Centrosomes split in the presence of impaired DNA integrity during mitosis. Mol Biol Cell 2003;14:1993–2004.
- [277] Yih LH, Hsueh SW, Luu WS, Chiu TH, Lee TC. Arsenite induces prominent mitotic arrest via inhibition of G2 checkpoint activation in CGL-2 cells. Carcinogenesis 2005;26:53–63.
- [278] Gao YH, Zhang HP, Yang SM, Yang Y, Ma YY, Zhang XY, et al. Inactivation of Akt by arsenic trioxide induces cell death via mitochondrial-mediated apoptotic signaling in SGC-7901 human gastric cancer cells. Oncol Rep 2014;31:1645–52.
- [279] Wang F, Liu S, Xi S, Yan L, Wang H, Song Y, et al. Arsenic induces the expressions of angiogenesisrelated factors through PI3K and MAPK pathways in SV-HUC-1 human uroepithelial cells. Toxicol Lett 2013;222:303–11.

- [280] Gao N, Shen L, Zhang Z, Leonard SS, He H, Zhang XG, et al. Arsenite induces HIF-1alpha and VEGF through PI3K, Akt and reactive oxygen species in DU145 human prostate carcinoma cells. Mol Cell Biochem 2004;255:33–45.
- [281] Calatayud M, Gimeno-Alcañiz JV, Vélez D, Devesa V. Trivalent arsenic species induce changes in expression and levels of proinflammatory cytokines in intestinal epithelial cells. Toxicol Lett 2014;224:40–6.
- [282] Xu Y, Zhao Y, Xu W, Luo F, Wang B, Li Y, et al. Involvement of HIF-2α-mediated inflammation in arseniteinduced transformation of human bronchial epithelial cells. Toxicol Appl Pharmacol 2013;272:542–50.
- [283] Martínez-Pacheco M, Hidalgo-Miranda A, Romero-Córdoba S, Valverde M, Rojas E. MRNA and miRNA expression patterns associated to pathways linked to metal mixture health effects. Gene 2014;533:508–14.
- [284] Marsit CJ, Eddy K, Kelsey KT. MicroRNA responses to cellular stress. Cancer Res 2006;66:10843-8.
- [285] Wang Z, Zhao Y, Smith E, Goodall GJ, Drew PA, Brabletz T, et al. Reversal and prevention of arsenicinduced human bronchial epithelial cell malignant transformation by microRNA-200b. Toxicol Sci 2011;121:110–22.
- [286] Samet JM, Graves LM, Quay J, Dailey LA, Devlin RB, Ghio AJ, et al. Activation of MAPKs in human bronchial epithelial cells exposed to metals. Am J Physiol 1998;275:L551–8.
- [287] Chen F, Lu Y, Zhang Z, Vallyathan V, Ding M, Castranova V, et al. Opposite effect of NF-kappa B and c-Jun N-terminal kinase on p53-independent GADD45 induction by arsenite. J Biol Chem 2001;276:11414–19.
- [288] Zhang Y, Bhatia D, Xia H, Castranova V, Shi X, Chen F. Nucleolin links to arsenic-induced stabilization of GADD45alpha mRNA. Nucleic Acids Res 2006;34:485–95.
- [289] Liao WT, Lin P, Cheng TS, Yu HS, Chang LW. Arsenic promotes centrosome abnormalities and cell colony formation in p53 compromised human lung cells. Toxicol Appl Pharmacol 2007;225:162–70.
- [290] Sykora P, Snow ET. Modulation of DNA polymerase beta-dependent base excision repair in cultured human cells after low dose exposure to arsenite. Toxicol Appl Pharmacol 2008;228:385–94.
- [291] Wen G, Calaf GM, Partridge MA, Echiburú-Chau C, Zhao Y, Huang S, et al. Neoplastic transformation of human small airway epithelial cells induced by arsenic. Mol Med 2008;14:2–10.

This page intentionally left blank

10

Arsenic Through the Gastrointestinal Tract

Marta Calatayud^{1,2} and José Moisés Laparra Llopis³

¹INSTITUTE OF AGROCHEMISTRY AND FOOD TECHNOLOGY, HIGH RESEARCH COUNCIL OF THE SPANISH GOVERNMENT, ²FACULTY OF HEALTH SCIENCES, TECHNICAL UNIVERSITY OF AMBATO, ECUADOR, ³INSTITUTE FÜR TRANSLATIONAL IMMUNOLOGIE, UNIVERSITY MEDICAL CENTER OF THE JOHANNES GUTENBERG-UNIVERSITY, MAINZ, GERMANY

CHAPTER OUTLINE

10.1	General Aspects	281
10.2	Arsenic in Foods: Household Processing and Toxicological Risk	282
10.3	Role of the Gut in Arsenic Toxicity: Bioavailability and Intestinal Health	283
10.4	Arsenic-Induced Metabolic/Immune Toxicity	289
10.5	Conclusions and Future Perspectives	291
Refe	rences	292

10.1 General Aspects

Arsenic (As) is widely distributed throughout Earth's crust where it can be found in several different chemical species, inorganic and organic, with a significant and variable degree of toxicity. Inorganic arsenic (iAs) is one of the most toxic species found in water and food, the main sources of As exposure for the general population [1]. The World Health Organization (WHO) estimates that >200 million persons worldwide might be chronically exposed to As in drinking water at concentrations above the WHO safety standard of $10 \,\mu$ g/L [2]. This estimation of the affected population would be increased if the intake of As through food were considered.

The term iAs includes both tri- (As^{III}) and pentavalent (As^V) oxidation states, As^V representing the main arsenical species in waters. The biotransformation of iAs occurs mainly in the liver and implies successive reduction and oxidative methylation steps producing different species with an increasing degree of methylation (mono-, di-, trimethylarsenic) and valency (trivalent or pentavalent), two critical features largely determining the toxicity of As [3]. Inorganic species are more toxic than methylated ones and trivalent species are more toxic than pentavalent ones. The biomethylation process was considered a detoxification process; however, there is still an open discussion ongoing on this matter because it has been demonstrated that trivalent mono- and dimethylated compounds [4] and thiolated intermediary metabolites have higher toxicity than iAs [5–8]. *In vitro* studies have also evidenced the existence of a presystemic biomethylation of iAs by human intestinal epithelial cells [9] and human gut microbiota [10].

Adverse health effects have been widely described after iAs intake in chronic and acute studies, with the conclusion that As is a highly potent toxicant and carcinogen [11], even at low doses. iAs is classified by the International Agency for Research on Cancer (IARC) as a carcinogen to humans (Group I) associated also with increasing prevalence of skin lesions, cardiovascular and cerebrovascular pathologies, type 2 diabetes, respiratory illnesses, and neurobehavioral and developmental disorders [1].

10.2 Arsenic in Foods: Household Processing and Toxicological Risk

Food and dietary habits constitute the most important source of As exposure for persons with limited exposure via drinking water [1]. Particular attention needs to be focused on infants during early stages of life. Moreover, persons suffering specific metabolic diseases that could affect As accumulation and metabolism, causing a potential major susceptibility to toxicity, deserve special consideration.

Studies conducted in different countries to evaluate total dietary As have shown that in areas where there is no endemic As contamination, intakes between 8 and $345\,\mu$ g/day can be reached [12]. Fish and derived foods constitute the group of major contributors to As intake. In these foods As contents are usually greater than $1\,\text{mg/g}$, exceeding $10\,\text{mg/g}$ in flatfish and crustaceans [13]. Arsenobetaine represents between 50 and 80% of total As content, although other species such as DMA (dimethylarsinic acid), MMA (monomethylarsonic acid), TMAO (trimethylarsine oxide), AC (arsenocholine), TMA+ (tetramethylarsonium ion) [13], and arsenosugars [14] have been detected. The iAs normally has been found in concentrations up to 0.1 mg/g [15], although unusual high levels of iAs have been described in crustaceans and bivalves (0.001–4.5 mg/kg) [16]. From a toxicological point of view, edible algae have received special attention because As content can reach concentrations between <0.04 and 141 mg/kg [17–19]. The iAs represents the major arsenical species in the alga *Hizikia fusiforme* [20], whereas dimethylated arsenosugars are the main species in the remaining seaweed [21].

With respect to terrestrial animal foods (milk, liver, and derived products), As^{III} and As^{V} contents have been described below 0.05 mg/g [22,23]. The presence of DMA and AB found in chicken meat [24] could be a consequence of the biotransformation of As^{V} [25]. Otherwise, in land-based plant foods, the concentration of As is generally lower than 0.1 mg/kg [26,27]; how-ever, this content can be higher in areas with water and soil As contamination (e.g., *Ipomoea aquatica*, 0.367 mg/kg wet weight) [28]. Importantly, the iAs content in rice—the staple diet for a large proportion of the world population—and rice products varies between 0.02 and 0.56 mg/kg, accounting for 11 to 100% of the total As content [26,29]. It is alarming that As concentrations up to 1.8 mg/kg have been reported in rice samples or rice products [30,31].

Moreover, the presence of MMA (0.015 mg/kg) and DMA (0.486–0.539 mg/kg) has also been reported [32,33].

In the estimation of the toxicological risk, exposure assessment must take into account the frequency of consumption as well as the amount of the toxicant and its chemical form in the considered food. Since diet is the exposure path for humans, assessing the extent of exposure should also consider the influence of cooking on the concentration of the toxicant and its oral bioavailability. Most existing studies of As in the literature do not consider the effects of cooking or bioavailability, but are limited to a determination of the concentrations of the toxicant in the raw products.

Several studies have showed that contaminated waters used for cooking purposes increase the toxicological risk associated with food consumption [34-36]. While the use of water with the least amount of As is the best choice [37] this is not always possible in endemic areas of As contamination. However, there can be recommended cooking procedures that will reduce the amount of As absorbed by the food during cooking. For example, the influence that several different cooking processes used by Asian consumers may have on the retention of As in rice has been studied. If rice is boiled with an excess of water, which later on is discarded, the values for As retention (43%) are lower than those calculated when rice is cooked with a limited amount of water (72–99%) [38]. Moreover, the use of parboiled rice did not imply large variations in the concentration of the toxicant retained by the rice [39]. Similarly, around 90% of total As can be removed from the edible alga Hizikia fusiforme by household processes such as soaking and boiling [40]. Besides, after cooking vegetables in water containing inorganic, mono- and dimethylated pentavalent As species there have been reported changes (reduction and thiolation) in these species [41]. The production of TMA⁺ has also been detected during baking, frying or grilling of seafood products, possibly because heating facilitates decarboxylation of arsenobetaine to TMA⁺ with higher toxic potential [42,43].

10.3 Role of the Gut in Arsenic Toxicity: Bioavailability and Intestinal Health

Several approaches have been used to estimate relative As bioaccessibility (maximum fraction of the toxic solubilized in the gastrointestinal tract and available for absorption) and bioavailability (fraction of the toxicant that becomes distributed through blood circulation to target tissues and organs).

The limited existing data from human studies concerning As absorption indicate a high uptake rate (>50%) for the different As species calculated from the rate of urinary excretion [44]. However, this approach can underestimate As absorption because of the bioaccumulation of As in nails [45], hair, and skin [27], and in different organs. The *in vivo* measurement constitutes the only tool to estimate bioavailability, but due to the ethical issues affecting toxicants, animal models and *in vitro* approaches have also been developed as a useful alternative for screening and relative estimation of As bioavailability. Several different animal models (primates, pigs, dogs, rabbits, and rodents) have been used [46–50], but results are not

extrapolable to humans. In this sense, there can be identified marked differences for methylation capability with respect to humans; for example, the marmoset monkey and chimpanzee have been shown not to methylate iAs [51], and dog or rat hepatocytes have a higher methylation capability than those in humans [52]. *In vitro* methods have also been developed as an alternative to the use of experimental animals [53,54]. In this sense, it is worth noting the inclusion of gut microorganisms in the *in vitro* system [10] because the human gut microbiota (fecal inocula) possess a substantial metabolic potential for As [10,55].

Most current research concerning As bioavailability has been focused on contaminated soils [56]. However, few studies have considered food matrices such as algae, rice, and seafood. Total diet studies conducted in countries without As contamination in water show a wide variability in intake of As, with values ranging from traces up to 345 mg/day for the Japanese population [57], for example. Recently, iAs consumption through staple foods such as rice has emerged as a worldwide health concern, especially for groups of populations with a high consumption frequency of this cereal and derived products (infants, coeliacs, vegans, different ethnic groups) [58,59]. These studies demonstrated increased (by 14.2%) urinary As concentrations in children and infants consuming rice derivatives-based foods, which resulted in higher than maximum exposures predicted for adults drinking contaminated waters. In the case of a staple food such as rice, the juvenile swine model revealed large differences in As bioavailability (33 vs. 100%) depending on the DMA or As^V [60] concentration in the considered sample.

Arsenic intestinal absorption constitutes a complex process that implies several routes and underlying molecular mechanisms, depending on the chemical form in which the toxicant is presented. Overall, a more effective absorption for trivalent species, has been described despite their greater degree of methylation, than their pentavalent counterparts. Absorption rates have been calculated for pentavalent methylated species between 17% for MMA^V and 33–50% DMA^V [60]. Using the rat intestine model it was shown that DMA^V is absorbed at a rate directly proportional to their concentration over a 100-fold range [61]. Data concerning absorption rates for trivalent methylated species have only been obtained with intestinal epithelial cells showing increasing absorption rates according to the degree of methylation (DMA^{III}>MMA^{III}>As^{III}), contrary to the pentavalent ones (As^V>MMA^V>DMA^V) [62]. From a molecular point of view, As^V, As^{III}, and MMA^{III} showed mixed components of trans- and paracellular transport (Figure 10–1). Otherwise, DMA^{III} is mostly transported transcellularly involving energy-dependent processes.

The mucus layer represents a barrier to the entry of these highly toxic species to the bloodstream. It is possible that a thinner mucus layer by different factors such as antibiotic treatment [63] or pathological conditions such as inflammatory bowel disease [64] could negatively affect trivalent As absorption and enterocytes uptake, and therefore worsen the toxicological effects. Moreover, it would also be necessary to evaluate the effect of As on the amount and composition of glycoproteins present in the mucus, as there are studies demonstrating the ability to dissolve mucus by compounds interacting with -SH groups [65]. Moreover, the intestinal oxidative stress is associated with loss of mucosal redox balance [66], altering its protective effect.

Due to their high reactivity, trivalent As forms are highly unstable in aqueous solutions and oxidize rapidly resulting in DMA^V and MMA^V. Thus, it is considered that their presence



FIGURE 10–1 Schematic representation of routes for As transport across the intestinal epithelium. Arsenic intestinal uptake occurs through the paracellular and transcellular pathway. Paracellular route is relevant for intestinal absorption of pentavalent arsenic species, while for the trivalent ones, transcellular path is involved. As^{III} comes into intestinal (Caco-2) cells by various carriers, including organic anion-transporting polypeptide B (OATPB), glucose transporter 5 (GLUT5), and aquaporin 10 (AQP10). As^V enterocyte uptake is made through the sodium-dependent phosphate transporter NaPillb. Inside the cell iAs is metabolized to AsIII and MMA^{III}, which reach the bloodstream or are eliminated into the intestinal lumen by the same proportion. Thiolated metabolytes are absorbed in a lesser extent than iAs and only thio-DMA^V is observed inside the cell. Exposure to iAs modifies the mRNA expression of apical domain transporters (GLUT5, AQP10, OATPB and NaPillb) as well AQP3, AQP4 and GLUT2, transporters located on the basolateral membrane, potentially facilitating iAs intestinal uptake.

after food intake is minimal. Only Yahtvakilla et al. [67] have shown MMA^{III} in carrot samples grown in soil contaminated with As. On the other hand, after the progress made in recent years regarding the metabolism of iAs in intestinal epithelium and gut microbiota, it can be concluded that trivalent forms may be present in the intestinal lumen or produced within enterocytes and later reach the bloodstream and/or exert a toxic effect at intestinal level. In addition, As triglutathione and methyl As diglutathione have been evidenced to be present in bile from rats administered with iAs [68,69], and therefore could reach the intestinal lumen by the common hepatic duct, entering the enterohepatic circulation. More recently, the formation of monomethyl monothio arsonic acid (MMMTA^V) and dimethyl monothio arsinic acid (DMMTA^V) in rats after orally administered As^{III} has been suggested to be dependent on enterohepatic circulation, possibly with the involvement of the intestinal microbiota [70].

Over the last few years, thiolated arsenical species have gained in importance. They are shown to be metabolites of As, but their toxicokinetic and toxicodynamic behavior is still unknown. Some of these As metabolites are directly present in food, such as thio-DMA^V in rice samples [71] or may occur via biotransformation of arsenosugars by gut microbiota [10,72,73], intestinal epithelial cells [74], and/or due to enterohepatic circulation [70]. More than 20 structures of arsenosugars have been described in foods, varying the R group attached to the aglycone [75]. Accessing new technologies of analytical chemistry will offer as yet unknown As species, increasing the complexity related to As exposure risk assessment. Currently, studies of *in vitro* bioavailability of arsenosugars (DMA^V-sugar-glycerol and DMA^V-sugar-sulfate) have indicated low absorption percentages (1.7–2.8%) [74]. Hata et al. observed a 30% excretion of the ingested As, mainly as arsenosugar AsSug328, through wakame seaweed in human volunteers [76]. These data differ from those previously presented by Francesconi et al. [77] in a volunteer that ingested the same synthesized arsenosugar, in which 80% of the administered As was excreted. *In vitro* and *in vivo* assays differ considerably, highlighting the need for further studies related to the metabolism and absorption of arsenosugars and improving *in vitro* systems to assess As bioavailability.

Due to the metabolic activity of the gut microbiota, ingested arsenosugars can be rapidly converted to sulfur analogs [72]. The bioavailability of some of these metabolites is variable depending on the chemical form, with similar values to As^{III} for thio-DMA^V and thio-DMAE^V, whereas oxo-DMAA^V, thio-DMAA^V, and oxo-DMAE^V passed the *in vitro* intestinal barrier in small amounts [74]. The most toxic compound, thio-DMA^V, crosses the intestinal barrier by passive diffusion and facilitated transport. However, in vitro assays have evidenced that As absorption processes can be influenced by significant interactions between As and food components exhibiting biological activity such as antioxidants [78] or chelating properties (water spinach) [79]. These effects are supported by spinach water-mediated reduction of As concentration in internal organs of rats [80]. Similarly, there was a reduced As-induced toxicity when animals were fed with curcumin [81]. The effect of biologically active compounds such as vitamin E, vitamin C, or selenium against As toxicity has been evidenced in in vitro and in vivo models [82-85], and although this was not statistically proven when evaluated through human trials [86], epidemiological studies indicate a protective effect of these compounds against As adverse effects [87,88]. Other factors affecting gut physiology such as the low pH, reduced peristaltism, or favored bile acids production could positively increase As absorption [89,90].

10.3.1 Gastrointestinal Health

The gastrointestinal tract is the largest organ in contact with exogenous compounds and constitutes the first physiological barrier to exogenous toxicants. The intestinal epithelium displays a complex architecture where epithelial, goblet, and immune cells are involved. Moreover, the intestine supplies the body with nutrients and other metabolites, but also with signals that stem from the intestinal mucosa and immune system, forming the so-called gut-liver axis. Notably, the liver is a central organ for the turnover and transformation of As and metabolites that pass through it from the intestine. In this context, the development and function of the host cells responsible for the barrier function of the intestinal surface (e.g., M cells, Paneth cells, goblet cells, and columnar epithelial cells) are strictly regulated through both

positive and negative stimulation by the luminal microbiota. Signaling by damage-associated molecular patterns (DAMPs) and commensal bacteria-derived microbe-associated molecular patterns (MAMPs) mainly through Toll-like receptors is involved in maintaining the integrity of the intestinal epithelium. Moreover, gut-associated immune cells also play critical roles in regulating both the mucosal barrier and the relative composition of the luminal microbiota. Imbalance in these components or disruption of mucosal homeostasis causes intestinal inflammation and loss of physiological functionality of the intestinal epithelium.

In vitro studies indicated that As exposure results in cytotoxic effects on intestinal epithelial (Caco-2) cells through several different underlying mechanisms similar to those identified in As-targeted tissues: oxidative stress, lipoperoxidation, and alteration in cellular antioxidant compounds and enzymes [91–93] (Figure 10–2). As a consequence of alterations in the intracellular redox status, As^{III} causes disorganization and redistribution of cytosolic actin [94,95], a protein involved in membrane vesicle and apoptotic bodies arrangement, marginalization



FIGURE 10–2 Hypothetic representation of As-induced immunotoxicity at intestinal level. The presence of trivalent arsenical species (especially As^{III} and DMA^{III}), with or without lipopolysaccharide (LPS) derived from *Salmonella enterica* serovar *typhimurium*, elicits oxidative stress and inflammatory cytokine secretion. This leads to an increase into cytokine levels in the interstitial space and the apical medium. This could attract peripheral blood mononuclear cells (PBMCs), intraepithelial lymphocytes, and dendritic cells, leading to an intestinal inflammatory reaction. All of these cells in response to inflammatory stimulus can generate more cytokines and positive feedback that maintain the inflammatory condition. It is not known whether damage to the tight junctions could affect epithelial barrier function and increase basal lamina contact with luminal antigenic substances.

of nuclear chromatin, and the maintenance of cell structure and signaling. This effect could be the result of activation of multiple cellular signaling pathways caused by oxidative stress [96–99]. *In situ* studies with rodents also evidenced a decreased activity of enzymes present in the plasmatic membrane of intestinal cells as well as injuries to cell membrane after As^{III} exposure [100]. Similarly to some other tissues, intestinal epithelial cells appear more susceptible to trivalent As species than their pentavalent counterparts. As^{III} and DMA^{III} induce inflammatory responses associated with increased interleukin (IL)-8 and tumor necrosis factor (TNF) α production by intestinal epithelial cells has also been reported [101]. Moreover, the coexposure of As and lipopolysaccharide (LPS) from pathogenic *Salmonella enterica* exacerbates the inflammatory response [101]. Otherwise, the intestine does not constitute an organ where a significant amount of As is accumulated in comparison to kidneys, lungs, or liver. All the aforementioned events contribute to alterations in intestinal architecture impairing gut integrity, thereby favoring local and/or systemic inflammatory reactions.

Although As is classified as carcinogenic and the drinking of water and the eating of foods are the main sources of exposure to humans, only a relatively few studies have described carcinogenic effects in the gut [102,103]. This association has shown gender (male)-specific differences in the various processes related to As toxicity. However, clinical signs of gastrointestinal irritation have been reported in cases of short-term, high-dose exposures to iAs and DMA^V [104,105]. Data from animal models demonstrated more severe symptoms with acute inflammation and hemorrhage of the small intestine in rhesus monkeys chronically exposed to high doses of As trioxide [106]. Also, long-term exposure of rodents to MMA^V led to a thickened wall, edema, and hemorrhagic, and necrotic, ulcerated or perforated mucosa in the large intestine [107]. These observations were accompanied by a significant increase in the incidence of squamous metaplasia of the epithelial columnar absorptive cells in the colon and rectum. As a function of the dose of As administered, there have been signs of stomach and intestinal inflammation (hemorrhage in stomach and intestine, rose-red inflammation, severe enteritis) in rats [108] and intestinal dysfunction related to wall movement and disrupted absorption capabilities in mice [109]. Taken together these studies demonstrate the marked impact of As on intestinal barrier, thereby potentially causing significant alterations in nutrient absorption processes and inflammatory and immune signals that can finally affect liver function.

However, not only do high-dose exposure levels of As cause intestinal physiological dysfunction. In a recent prospective population-based cohort study [110], an increased risk of diarrheal diseases has been observed in relation to prenatal low-moderate As exposure. These gastrointestinal diseases constitute the most important cause of illness and the second leading cause of death among children under 2 years [111]. For example, in Bangladesh, one of the most affected areas by As contamination in ground water, children have been identified as a sensitive population group to chronic As toxicity due to drinking As-contaminated water [112,113], which causes significant morbidity in children resulting in skin lesions, lung disease, and defects in intellectual function.

The human intestine harbors a complex, diverse, and vast microbial community, in which the genome (microbiome) plays a profound role in metabolic, immune, and cognitive development, and epithelial homeostasis. However, colonization of the intestine is a dynamic process, which depends on both environmental and genetic factors that can be affected by As exposure. Data from rodent models clearly demonstrated that As exposure not only alters the gut microbiome at the abundance level, but also has deleterious effects on its function. These alterations have been suggested as a potential aggravating factor of human diseases [114,115]. Removal of toxic metals and toxins using probiotics (live microorganisms that when administered in an adequate dose have beneficial effects on host health) could be an inexpensive, new, promising method on top of conventional methods to eliminate, inactivate, or reduce bioavailability of As in foods [116]. However, it is important to consider that some microbes can favor As biotransformation, sometimes increasing its bioaccessibility or toxicity [117]; however, only inferential studies have addressed these aspects.

10.4 Arsenic-Induced Metabolic/Immune Toxicity

Arsenic toxicity is complex and multifaceted, not only in terms of As being a well-established agent causing skin, lung, and bladder cancers and cardiovascular diseases, but also in relation to the adversely effects on the metabolic and immune functions where its specific effects are particularly poorly understood.

As needs to cross from the intestinal lumen into the lamina propria where lymphocytes reside in order to affect their function and activity. In contrast to other tissues, intestinal mucosa has a continuous low-grade inflammation where regulatory systems participate to maintain a proper immune homeostasis and a healthy gut, but also inducing a protective immune response against pathogens and exogenous toxicants. In this sense, microbial colonization of the intestine in early life is particularly important as it is the main environmental stimulus for immune system maturation. This is a dynamic process, which depends on both environmental and genetic factors. The abnormal host-microbe interplay is largely genetically determined, which should also be considered because of As-mediated toxicity. In addition, recent animal studies demonstrated the influence of high concentrations $(10 \,\mu\text{g/mL})$ of As in the drinking water on gut microbiota [118]. The effects appear not only related to abundance level, but also to function, with several different microbiota-related metabolites being perturbed in multiple biological matrices.

In vitro and *in vivo* animal and human epidemiological data clearly demonstrate As immunomodulatory effects by increased apoptotic rates in immunocompetent cells, including macrophages and neutrophils [119–121]. Immune deficiencies are particularly relevant at early stages of life because of their contribution to adequate antigen tolerance or to increasing overreactions and disease risk. In this sense, the scarce existing data concerning the physiological consequences of gestational exposure to As reveal the marked negative impact of the metalloid on both branches of immune system in early stages of life, which correlate to increased fetal loss and infant mortality [122,123]. Reports on immune-related effects of prenatal exposure in newborns also showed a negative correlation between maternal urinary As concentrations and child thymic index and breast milk trophic factors such as IL-7 and lactoferrin, but a positive correlation between maternal urinary As and infant morbidity (fever, diarrhea, and respiratory infections) [124]. Recently, epigenetic modifications in microRNAs associated with innate and adaptive immune signaling in newborn cord blood as a consequence of prenatal As exposure, have also been reported [125]. Follow-up studies indicate that these findings appear to be associated with elevated oxidative stress and inflammation in chronically-exposed adults [126,127]. Overall, the existing data increase the evidence that As-associated adverse immune-related outcomes from early life will have long-term negative consequences into adulthood.

Reduced immune surveillance caused by As could potentially play a significant role not only in carcinogenesis, but also in aggravating the severity of inflammatory and metabolic diseases such as type 2 diabetes (T2D) and non-alcoholic fatty liver disease (NAFLD) [128,129]. T2D accounts for more than 90–95% of all diabetes with unknown specific etiology, although genetic factors, aging, and obesity are important risk factors [130]. In areas with endemic As contamination such as Bangladesh a positive association of modest levels of As in drinking water ($\leq 1.7 \,\mu g/L$) with increased risk of T2D has been made [131]. Being overweight or smoking are aggravating factors of the risk. However, this relation between long-term iAs exposure and risk of T2D did not show up so clearly from a more extended meta-analysis of articles published in English or Chinese (1990-2013) [132]. Several different mechanisms for As-induced T2D have also been proposed including phosphate substitution by As^V in the formation of adenosine triphosphate (ATP), As^{III} high affinity for sulfhydryl groups in the molecules of insulin, insulin receptors, glucose transporters (GLUTs), and enzymes involved in glucose metabolism, stimulation of glucose transport, induction of oxidative stress, and interference in signal transduction or gene expression [133]. Although the molecular mechanism remains elusive, a common conclusion from the existing literature is that insulin resistance and β -cell dysfunction can be induced by chronic As exposure.

The liver is a central organ for the transformation of nutrients and metabolites from the intestine, and long-term As immune suppression can impair the ability of liver to detoxify exogenous toxicants. The prevalence of NAFLD among Asian Indians has been reported as comparable to that in Western populations (15–30%), and its more severe steatosis with inflammation (NASH) may be present in approximately 20% of these patients [134]. Notably, about 85% of patients with NASH exhibit hallmarks of the metabolic syndrome, mainly insulin resistance, T2D, and hyperlipidemia. The interaction and effects of As exposure in NAFLD animal models have been poorly studied and the scarce existing data show an increased transport rather than biotransformation of iAs [129]. This effect may have significant implications for the overall toxicity associated with As in patients suffering from NAFLD/NASH. The exact mechanisms responsible for the pathological progression from hepatic steatosis to NASH are not entirely understood, but there are several factors involved such as mitochondrial dysfunction, oxidative stress, and pro-inflammatory cytokine production, similar to the effects resulting from As exposure.

The aryl hydrocarbon receptor (AhR) is a member of the basic-helix-loop-helix (bHLH)/ Per-ARNT-Sim family of transcription proteins, the activated nuclear complex of which is known as xenobiotic responsive element regulating phase I and II enzymes [135]. Previous studies report that As^{III} exerts modulatory effects on the expression of several phase I and II AhR-regulated genes in *in vitro* systems [136]. This observation is supported by a few existing *in vivo* studies, which showed a differential modulation of AhR-regulated enzymes by As^{III} in different organs such as kidneys, lungs, and heart of C57BL/6 mice [137]. Humans are exposed to several polyphenols, for example quercetin, resveratrol, and curcumin [138], through the diet and nutritional supplements. These compounds have been shown to activate the AhR in spite of the fact that they bind to the receptor with low affinity, differentially modulating inflammatory responses of human keratinocytes by interfering also with other activation of transcription factors such as NF- κ B and the EGFR-ERK pathway [139].

Arsenic can also induce bone-marrow toxicity [140]. Mesenchymal stem cells (MSCs) are non-hematopoietic components of bone marrow involved in immune-suppressive, immunemodulatory, and cytoprotective responses. It has been shown that alterations in cellular homeostasis, such as increased oxidative stress, increases MSC senescence, poor differentiation, and cell death. These effects have been associated with the As-induced down-regulation of the peroxisome proliferator-activated receptor (PPAR)- γ by using C3H 10T1/2 cells [141]. The PPARs belong to the nuclear hormone receptor superfamily, the dysregulation of which has an important role in immune disease development.

Combining the data stated above, it is easy to conclude that there is a deleterious immunosuppressive effect due to As exposure. However, studies can also be found showing beneficial As-mediated effects derived from its synergistic action with molecules such as all-transretinoic acid inhibiting breast cancer and prostate cell lines [142]. Moreover, As-induced antitumor activity has also been reported in Beige/Nude/X-Linked immune-deficient mice associated with activation of PPAR- γ by the agonist RWJ-241947 (MCC-555) [143] and As reduction of lung allergic inflammatory responses in a murine model of asthma by a direct inhibitory effect on the production of eotaxin by pulmonary cells [144].

All the aforementioned aspects clearly show complex and multifaceted As-mediated immunoregulatory activity that appears conditioned, positive or negatively, by certain other environmental factors and physiological conditions, as well as As concentration. The related effects are unavoidable and might impair metabolic processes masking As toxicity in exposed populations.

10.5 Conclusions and Future Perspectives

Arsenic is widely distributed in Earth's crust and is able to penetrate to all levels of the food chain. Eating food and drinking water are the major sources for As exposure either in developing and developed countries with estimated daily intakes around $300 \mu g/day$. Despite the progress made in As remediation from drinking water in recent years, there are currently many geographic areas with levels of As above the values recommended by the WHO. Minimizing As exposure is the best solution to reduce health problems resulting from its ingestion, but in several areas it is economically or technically not feasible to decrease As in water to levels lower than $10 \mu g/L$. Moreover, because of the lack of knowledge and/or evidence on which the maximum dose of iAs is safe for humans, it is vital to address all the variables that could impact As toxicity, including food interactions, gut microbiota, and previous diseases.

Bioaccumulation and/or transformation of the several different As species are of particular toxicological interest because these processes can promote reduction or worsening of As toxicity. The toxicological risk evaluation concerning As exposure from food must take into account not only the concentration of the toxicant, but also interaction with other food components and its resulting bioavailability. These processes can limit the dose of the toxicant that reaches target tissues as well as its toxicity.

Moreover, several different scientific studies have revealed different physiological alterations affecting health status derived from As exposure even to low-moderate levels. These alterations are of special importance because they affect nutrient absorption processes and immunological function, thereby affecting adequate growth and development during early stages of life. In this sense, particular attention should be paid to rice-derived products widely used for infant nutrition and replacement of wheat and other cereals for susceptible groups of the population such as celiac patients. Therefore, the risk derived from As intake should be evaluated independently in specific population groups with increased consumption of products with high content of As and/or increased susceptibility to its effects.

Although there have been major advances over the last few years concerning As effects on human health, only a comparatively small number of studies have evaluated the influence and potential contribution of this toxicant to diseases still unrelated to As exposure such as pandemic liver-related metabolic disorders in Western populations.

Finally, despite evidence that populations from non-As endemic areas are exposed to As through food, we should not forget that currently the biggest problem of As contamination continues in As endemic areas such as India, Bangladesh, Argentina, and Taiwan. Thus, it should be a priority to develop available and open technology to reduce the levels of As in drinking water in these areas, to develop economically accessible food strategies to reduce As exposure, and to implement citizenship education policies for diminishing As intake by processing or food-chelating techniques. These last issues can offer alternatives with a broad economic and social impact in areas of chronic exposure to As, until the objective of safe food and drinking water is achieved.

References

- [1] European Food Safety Authority Scientific opinion on arsenic in food. EFSA J 2009;7(10):1351.
- [2] World Health Organization Guidelines for drinking-water quality recommendations. Geneva: WHO Press; 2008.
- [3] Vasken Aposhian H, Zakharyan RA, Avram MD, Sampayo-Reyes A, Wollenberg ML. A review of the enzymology of arsenic metabolism and a new potential role of hydrogen peroxide in the detoxication of the trivalent arsenic species. Toxicol Appl Pharmacol 2004;198(3):327–35.
- [4] Styblo M, Del Razo LM, Vega L, Germolec DR, LeCluyse EL, Hamilton GA, et al. Comparative toxicity of trivalent and pentavalent inorganic and methylated arsenicals in rat and human cells. Arch Toxicol 2000;74(6):289–99.
- [5] Naranmandura H, Carew MW, Xu S, Lee J, Leslie EM, Weinfeld M, et al. Comparative toxicity of arsenic metabolites in human bladder cancer EJ-1 cells. Chem Res Toxicol 2011;24(9):1586–96.
- [6] Leffers L, Ebert F, Taleshi MS, Francesconi KA, Schwerdtle T. *In vitro* toxicological characterization of two arsenosugars and their metabolites. Mol Nutr Food Res 2013;57(7):1270–82.
- [7] Leffers L, Unterberg M, Bartel M, Hoppe C, Pieper I, Stertmann J, et al. *In vitro* toxicological characterisation of the S-containing arsenic metabolites thio-dimethylarsinic acid and dimethylarsinic glutathione. Toxicology 2013;305(0):109–19.

- [8] Naranmandura H, Xu S, Koike S, Pan LQ, Chen B, Wang YW, et al. The endoplasmic reticulum is a target organelle for trivalent dimethylarsinic acid (DMAIII)-induced cytotoxicity. Toxicol Appl Pharmacol 2012;260(3):241–9.
- [9] Calatayud M, Velez D, Devesa V. Metabolism of inorganic arsenic in intestinal epithelial cell lines. Chem Res Toxicol 2012;25(11):2402–11.
- [10] Van de Wiele T, Gallawa CM, Kubachka KM, Creed JT, Basta N, Dayton EA, et al. Arsenic metabolism by human gut microbiota upon *in vitro* digestion of contaminated soils. Environ Health Perspect 2010;118(7):1004–9.
- [11] International Agency for Research on Cancer. IARC monographs on the evaluation of carcinogenic risks to humans. Lyon, France: IARC; 2004. 39 pp.
- [12] Llobet J, Falco G, Casas C, Teixido A, Domingo J. Concentrations of arsenic, cadmium, mercury, and lead in common foods and estimated daily intake by children, adolescents, adults, and seniors of Catalonia, Spain. J Agric Food Chem 2003;51(3):838–42.
- [13] Súñer MA, Devesa V, Clemente MJ, Vélez D, Montoro R, Urieta I, et al. Organoarsenical species contents in fresh and processed seafood products. J Agric Food Chem 2002;50(4):924–32.
- [14] Francesconi KA, Edmonds JS. Arsenic species in marine samples. Croatica Chemica Acta 1998;71:343–59.
- [15] Muñoz O, Devesa V, Suñer MA, Vélez D, Montoro R, Urieta I, et al. Total and inorganic arsenic in fresh and processed fish products. J Agric Food Chem 2000;48(9):4369–76.
- [16] Sloth JJ, Julshamn K. Survey of total and inorganic arsenic content in blue mussels (*Mytilus edulis* L.) from Norwegian fiords: revelation of unusual high levels of inorganic arsenic. J Agric Food Chem 2008;56(4):1269–73.
- [17] Almela C, Algora S, Benito V, Clemente M, Devesa V, Suner M, et al. Heavy metal, total arsenic, and inorganic arsenic contents of algae food products. J Agric Food Chem 2002;50(4):918–23.
- [18] García-Salgado S, Quijano MA, Bonilla MM. Arsenic speciation in edible alga samples by microwaveassisted extraction and high performance liquid chromatography coupled to atomic fluorescence spectrometry. Anal Chim Acta 2012;714(0):38–46.
- [19] Almela C, Jesús Clemente M, Vélez D, Montoro R. Total arsenic, inorganic arsenic, lead and cadmium contents in edible seaweed sold in Spain. Food Chem Toxicol 2006;44(11):1901–8.
- [20] Laparra J, Velez D, Montoro R, Barbera R, Farre R. Bioaccessibility of inorganic arsenic species in raw and cooked *Hizikia fusiforme* seaweed. Appl Organometallic Chem 2004;18(12):662–9.
- [21] Gallagher PA, Shoemaker JA, Wei X, Brockhoff-Schwegel CA, Creed JT. Extraction and detection of arsenicals in seaweed via accelerated solvent extraction with ion chromatographic separation and ICP-MS detection. Fresenius J Anal Chem 2001;369(1):71–80.
- [22] Zbinden P, Andrey D, Blake C. A routine ion chromatography ICP-MS method for the analysis of arsenic species applicable in the food industry. At Spectrosc Norwalk Conn 2000;21(6):205–15.
- [23] Cava-Montesinos P, de la Guardia A, Teutsch C, Luisa Cervera M, de la Guardia M. Non-chromatographic speciation analysis of arsenic and antimony in milk hydride generation atomic fluorescence spectrometry. Anal Chim Acta 2003;493(2):195–203.
- [24] Polatajko A, Szpunar J. Speciation of arsenic in chicken meat by anion-exchange liquid chromatography with inductively coupled plasma-mass spectrometry. J AOAC Int 2004;87(1):233–7.
- [25] Pizarro I, Gómez M, Fodor P, Palacios M, Cámara C. Distribution and biotransformation of arsenic species in chicken cardiac and muscle tissues. Biol Trace Elem Res 2004;99(1-3):129-43.
- [26] Schoof R, Yost L, Eickhoff J, Crecelius E, Cragin D, Meacher D, et al. A market basket survey of inorganic arsenic in food. Food Chem Toxicol 1999;37(8):839–46.
- [27] Roychowdhury T, Tokunaga H, Ando M. Survey of arsenic and other heavy metals in food composites and drinking water and estimation of dietary intake by the villagers from an arsenic-affected area of West Bengal, India. Sci Total Environ 2003;308(1):15–35.

- [28] Wang H-S, Sthiannopkao S, Chen Z-J, Man Y-B, Du J, Xing G-H, et al. Arsenic concentration in rice, fish, meat and vegetables in Cambodia: a preliminary risk assessment. Environ Geochem Health 2013;35(6):745-55.
- [29] Heitkemper DT, Vela NP, Stewart KR, Westphal CS. Determination of total and speciated arsenic in rice by ion chromatography and inductively coupled plasma mass spectrometry. J Anal At Spectrom 2001;16(4):299–306.
- [30] Meharg AA, Rahman MM. Arsenic contamination of Bangladesh paddy field soils: implications for rice contribution to arsenic consumption. Environ Sci Technol 2003;37(2):229–34.
- [31] Sun G-X, Williams PN, Carey A-M, Zhu Y-G, Deacon C, Raab A, et al. Inorganic arsenic in rice bran and its products are an order of magnitude higher than in bulk grain. Environ Sci Technol 2008;42(19):7542–6.
- [32] Sanz E, Muñoz-Olivas R, Cámara C. A rapid and novel alternative to conventional sample treatment for arsenic speciation in rice using enzymatic ultrasonic probe. Anal Chim Acta 2005;535(1-2):227-35.
- [33] Alava P, Tack F, Laing GD, Wiele TV. HPLC-ICP-MS method development to monitor arsenic speciation changes by human gut microbiota. Biomed Chromatogr 2012;26(4):524–33.
- [34] Bae M, Watanabe C, Inaoka T, Sekiyama M, Sudo N, Bokul MH, et al. Arsenic in cooked rice in Bangladesh. Lancet 2002;360(9348):1839-40.
- [35] Rahman MA, Hasegawa H. High levels of inorganic arsenic in rice in areas where arsenic-contaminated water is used for irrigation and cooking. Sci Total Environ 2011;409(22):4645–55.
- [36] Torres-Escribano S, Leal M, Vélez D, Montoro R. Total and inorganic arsenic concentrations in rice sold in Spain, effect of cooking, and risk assessments. Environ Sci Technol 2008;42(10):3867–72.
- [37] Ohno K, Yanase T, Matsuo Y, Kimura T, Hamidur Rahman M, Magara Y, et al. Arsenic intake via water and food by a population living in an arsenic-affected area of Bangladesh. Sci Total Environ 2007;381(1):68–76.
- [38] Sengupta M, Hossain M, Mukherjee A, Ahamed S, Das B, Nayak B, et al. Arsenic burden of cooked rice: traditional and modern methods. Food Chem Toxicol 2006;44(11):1823–9.
- [39] Rahman MA, Hasegawa H, Rahman MA, Rahman MM, Miah M. Influence of cooking method on arsenic retention in cooked rice related to dietary exposure. Sci Total Environ 2006;370(1):51–60.
- [40] Ichikawa S, Kamoshida M, Hanaoka K, Hamano M, Maitani T, Kaise T. Decrease of arsenic in edible brown algae Hijikia fusiforme by the cooking process. Appl Organometallic Chem 2006;20(9):585–90.
- [41] Calatayud M, Bralatei E, Feldmann J, Devesa V, Vélez D. Transformation of arsenic species during *in vitro* gastrointestinal digestion of vegetables. J Agric Food Chem 2013;61(49):12164–170.
- [42] Devesa V, Martínez A, Súner M, Vélez D, Almela C, Montoro R. Effect of cooking temperatures on chemical changes in species of organic arsenic in seafood. J Agric Food Chem 2001;49(5):2272–6.
- [43] Devesa V, Vélez D, Montoro R. Effect of thermal treatments on arsenic species contents in food. Food Chem Toxicol 2008;46(1):1–8.
- [44] Koch I, Steven S, Lai VWM, Owen A, Reimer KJ, Cullen WR. Bioaccessibility and excretion of arsenic in Niu Huang Jie Du Pian pills. Toxicol Appl Pharmacol 2007;222(3):357–64.
- [45] Karagas MR, Tosteson TD, Blum J, Klaue B, Weiss JE, Stannard V, et al. Measurement of low levels of arsenic exposure: a comparison of water and toenail concentrations. Am J Epidemiol 2000;152(1):84–90.
- [46] Freeman GB, Schoof RA, Ruby MV, Davis AO, Dill JA, Liao SC, et al. Bioavailability of arsenic in soil and house dust impacted by smelter activities following oral administration in Cynomolgus monkeys. Fundam Appl Toxicol 1995;28(2):215–22.
- [47] Juhasz AL, Smith E, Weber J, Rees M, Rofe A, Kuchel T, et al. Application of an *in vivo* swine model for the determination of arsenic bioavailability in contaminated vegetables. Chemosphere 2008;71(10):1963–9.
- [48] Ng J, Johnson D, Moore MR, Imray P, Chiswell B. Speciation of arsenic metabolites in the urine of occupational workers and experimental rats using an optimised hydride cold-trapping method[+]. Analyst 1998;123:929–33.

- [49] Groen K, Vaessen HAMG, Kliest JJG, Deboer JLM, Vanooik T, Timmerman A, et al. Bioavailability of inorganic arsenic from bog ore-containing soil in the dog. Environ Health Perspect 1994;102(2):182–4.
- [50] Freeman G, Johnson J, Killinger J, Liao S, Davis A, Ruby M, et al. Bioavailability of arsenic in soil impacted by smelter activities following oral administration in rabbits. Fundam Appl Toxicol 1993;21(1):83–8.
- [51] Vahter M. Methylation of inorganic arsenic in different mammalian species and population groups. Sci Prog 1999;82(Pt 1):69–88.
- [52] Drobná Z, Walton FS, Harmon AW, Thomas DJ, Stýblo M. Interspecies differences in metabolism of arsenic by cultured primary hepatocytes. Toxicol Appl Pharmacol 2010;245(1):47–56.
- [53] Ruby MV, Davis A, Schoof R, Eberle S, Sellstone CM. Estimation of lead and arsenic bioavailability using a physiologically based extraction test. Environ Sci Technol 1996;30(2):422–30.
- [54] Laparra JM, Velez D, Barbera R, Montoro R, Farre R. An approach to As(III) and As(V) bioavailability studies with Caco-2 cells. Toxicol In Vitro 2005;19(8):1071–8.
- [55] Sun GX, Van de Wiele T, Alava P, Tack F, Du Laing G. Arsenic in cooked rice: effect of chemical, enzymatic and microbial processes on bioaccessibility and speciation in the human gastrointestinal tract. Environ Pollut 2012;162:241–6.
- [56] Environmental Protection Agency (US EPA). Compilation and review of data for relative bioavailability of arsenic in soil and recommendations for default value for relative bioavailability. Washington, DC: Environmental Protection Agency; 2012.
- [57] Ikebe K. Daily intake of metals by females in Osaka, Japan. Bull Environ Contam Toxicol 1992;49(4):509–13.
- [58] Meharg AA, Sun G, Williams PN, Adomako E, Deacon C, Zhu Y-G, et al. Inorganic arsenic levels in baby rice are of concern. Environ Pollut 2008;152(3):746–9.
- [59] Davis MA, Mackenzie TA, Cottingham KL, Gilbert-Diamond D, Punshon T, Karagas MR. Rice consumption and urinary arsenic concentrations in US children. Environ Health Perspect 2012;120(10):1418–24.
- [60] Juhasz AL, Smith E, Weber J, Rees M, Rofe A, Kuchel T, et al. *In vivo* assessment of arsenic bioavailability in rice and its significance for human health risk assessment. Environ Health Perspect 2006;114(12):1826–31.
- [61] Hwang SW, Schanker LS. Absorption of organic arsenical compounds from the rat small intestine. Xenobiotica 1973;3(6):351–5.
- [62] Calatayud M, Gimeno J, Vélez D, Devesa V, Montoro R. Characterization of the intestinal absorption of arsenate, monomethylarsonic acid, and dimethylarsinic acid using the Caco-2 cell line. Chem Res Toxicol 2010;23(3):547-56.
- [63] Wlodarska M, Willing B, Keeney K, Menendez A, Bergstrom K, Gill N, et al. Antibiotic treatment alters the colonic mucus layer and predisposes the host to exacerbated *Citrobacter rodentium*-induced colitis. Infect Immun 2011;79(4):1536–45.
- [64] Johansson ME, Gustafsson JK, Holmén-Larsson J, Jabbar KS, Xia L, Xu H, et al. Bacteria penetrate the normally impenetrable inner colon mucus layer in both murine colitis models and patients with ulcerative colitis. Gut 2014;63(2):281–91.
- [65] Allen A, Flemström G. Gastroduodenal mucus bicarbonate barrier: protection against acid and pepsin. Am J Physiol Cell Physiol 2005;288(1):C1–C19.
- [66] Aw TY. Molecular and cellular responses to oxidative stress and changes in oxidation-reduction imbalance in the intestine. Am J Clin Nutr 1999;70(4):557–65.
- [67] Yathavakilla SKV, Fricke M, Creed PA, Heitkemper DT, Shockey NV, Schwegel C, et al. Arsenic speciation and identification of monomethylarsonous acid and monomethylthioarsonic acid in a complex matrix. Anal Chem 2008;80(3):775–82.
- [68] Gregus Z, Gyurasics Á, Csanaky I. Biliary and urinary excretion of inorganic arsenic: monomethylarsonous acid as a major biliary metabolite in rats. Toxicol Sci 2000;56(1):18–25.
- [69] Kala SV, Neely MW, Kala G, Prater CI, Atwood DW, Rice JS, et al. The MRP2/cMOAT transporter and arsenic-glutathione complex formation are required for biliary excretion of arsenic. J Biol Chem 2000;275(43):33404–8.
- [70] Bu N, Wang HY, Hao WH, Liu X, Xu S, Wu B, et al. Generation of thioarsenicals is dependent on the enterohepatic circulation in rats. Metallomics 2011;3(10):1064–73.
- [71] Ackerman AH, Creed PA, Parks AN, Fricke MW, Schwegel CA, Creed JT, et al. Comparison of a chemical and enzymatic extraction of arsenic from rice and an assessment of the arsenic absorption from contaminated water by cooked rice. Environ Sci Technol 2005;39(14):5241–6.
- [72] Conklin SD, Ackerman AH, Fricke MW, Creed PA, Creed JT, Kohan MC, et al. *In vitro* biotransformation of an arsenosugar by mouse anaerobic cecal microflora and cecal tissue as examined using IC-ICP-MS and LC-ESI-MS/MS. Analyst 2006;131(5):648–55.
- [73] Kubachka KM, Kohan MC, Conklin SD, Herbin-Davis K, Creed JT, Thomas DJ. In vitro biotransformation of dimethylarsinic acid and trimethylarsine oxide by anaerobic microflora of mouse cecum analyzed by HPLC-ICP-MS and HPLC-ESI-MS. J Anal At Spectrom 2009;24(8):1062–8.
- [74] Leffers L, Wehe CA, Hüwel S, Bartel M, Ebert F, Taleshi MS, et al. *In vitro* intestinal bioavailability of arsenosugar metabolites and presystemic metabolism of thio-dimethylarsinic acid in Caco-2 cells. Metallomics 2013;5(8):1031–42.
- [75] Francesconi KA, Kuehnelt D. Determination of arsenic species: a critical review of methods and applications, 2000–2003. Analyst 2004;129(5):373–95.
- [76] Hata A, Yamanaka K, Endo G, Yamano Y, Haba R, Fujitani N, et al. Arsenic metabolites in humans after ingestion of wakame seaweed. E3S Web of Conferences 2013;1:26006.
- [77] Francesconi KA, Tanggaar R, McKenzie CJ, Goessler W. Arsenic metabolites in human urine after ingestion of an arsenosugar. Clin Chem 2002;48(1):92–101.
- [78] Calatayud M, Devesa V, Montoro R, Velez D. *In vitro* study of intestinal transport of arsenite, monomethylarsonous acid, and dimethylarsinous acid by Caco-2 cell line. Toxicol Lett 2011;204(2):127–33.
- [79] Yang U-J, Yoon S-R, Chung J-H, Kim YJ, Park K-H, Park T-S, et al. Water spinach (*Ipomoea aquatic a* Forsk.) reduced the absorption of heavy metals in an *in vitro* bio-mimicking model system. Food Chem Toxicol 2012;50(10):3862–6.
- [80] Islam M, Awal M, Mostofa M, Ghosh A, Khair A. Effect of spinach against arsenic toxicity in rats. Bangladesh J Vet Med 2009;7(2):358–63.
- [81] Gao S, Duan X, Wang X, Dong D, Liu D, Li X, et al. Curcumin attenuates arsenic-induced hepatic injuries and oxidative stress in experimental mice through activation of Nrf2 pathway, promotion of arsenic methylation and urinary excretion. Food Chem Toxicol 2013;59:739–47.
- [82] Karasavvas N, Cárcamo JM, Stratis G, Golde DW. Vitamin C protects HL60 and U266 cells from arsenic toxicity. Blood 2005;105(10):4004–12.
- [83] Chattopadhyay S, Bhaumik S, Purkayastha M, Basu S, Nag Chaudhuri A, Das Gupta S. Apoptosis and necrosis in developing brain cells due to arsenic toxicity and protection with antioxidants. Toxicol Lett 2002;136(1):65–76.
- [84] Mittal M, Flora S. Vitamin E supplementation protects oxidative stress during arsenic and fluoride antagonism in male mice. Drug Chem Toxicol 2007;30(3):263–81.
- [85] Messarah M, Klibet F, Boumendjel A, Abdennour C, Bouzerna N, Boulakoud MS, et al. Hepatoprotective role and antioxidant capacity of selenium on arsenic-induced liver injury in rats. Exp Toxicol Pathol 2012;64(3):167–74.
- [86] Verret WJ, Chen Y, Ahmed A, Islam T, Parvez F, Kibriya MG, et al. A randomized, double-blind placebocontrolled trial evaluating the effects of vitamin E and selenium on arsenic-induced skin lesions in Bangladesh. J Occup Environ Med 2005;47(10):1026–35.

- [87] Chen Y, Hall M, Graziano JH, Slavkovich V, van Geen A, Parvez F, et al. A prospective study of blood selenium levels and the risk of arsenic-related premalignant skin lesions. Cancer Epidemiol Biomarkers Prev 2007;16(2):207–13.
- [88] Khandker S, Dey RK, Islam AM, Ahmad SA, Al-Mahmud I. Arsenic-safe drinking water and antioxidants for the management of arsenicosis patients. Bangladesh J Pharmacol 2006;1(2):42–50.
- [89] Charman WN, Porter CJ, Mithani S, Dressman JB. Physicochemical and physiological mechanisms for the effects of food on drug absorption: the role of lipids and pH. J Pharm Sci 1997;86(3):269–82.
- [90] Alava P, Du Laing G, Odhiambo M, Verliefde A, Tack F, Van de Wiele TR. Arsenic bioaccessibility upon gastrointestinal digestion is highly determined by its speciation and lipid-bile salt interactions. J Environ Sci Health Part A 2013;48(6):656–65.
- [91] Laparra JM, Vélez D, Barberá R, Granero L, Polache A, Montoro R, et al. Cytotoxic effect of As (III) in Caco-2 cells and evaluation of its human intestinal permeability. Toxicol In Vitro 2006;20(5):658–63.
- [92] Laparra J, Vélez D, Barberá R, Farré R, Montoro R. As2O3-induced oxidative stress and cycle progression in a human intestinal epithelial cell line (Caco-2). Toxicol In Vitro 2008;22(2):444–9.
- [93] Calatayud M, Devesa V, Vélez D. Differential toxicity and gene expression in Caco-2 cells exposed to arsenic species. Toxicol Lett 2013;28(1):70–80.
- [94] Izdebska M, Grzanka A, Ostrowski M, Żuryń A, Grzanka D. Effect of arsenic trioxide (Trisenox) on actin organization in K-562 erythroleukemia cells. Folia Histochem Cytobiol 2009;47(3):453–9.
- [95] Grzanka A, Grzanka D, Orlikowska M. Cytoskeletal reorganization during process of apoptosis induced by cytostatic drugs in K-562 and HL-60 leukemia cell lines. Biochem Pharmacol 2003;66(8):1611–17.
- [96] Qian Y, Liu KJ, Chen Y, Flynn DC, Castranova V, Shi X. Cdc42 regulates arsenic-induced NADPH oxidase activation and cell migration through actin filament reorganization. J Biol Chem 2005;280(5):3875–84.
- [97] Huot J, Houle F, Marceau F, Landry J. Oxidative stress-induced actin reorganization mediated by the p38 mitogen-activated protein kinase/heat shock protein 27 pathway in vascular endothelial cells. Circ Res 1997;80(3):383–92.
- [98] Huot J, Houle F, Rousseau S, Deschesnes RG, Shah GM, Landry J. SAPK2/p38-dependent F-actin reorganization regulates early membrane blebbing during stress-induced apoptosis. J Cell Biol 1998;143(5):1361–73.
- [99] Pérez LM, Milkiewicz P, Ahmed-Choudhury J, Elias E, Ochoa JE, Sánchez Pozzi EJ, et al. Oxidative stress induces actin-cytoskeletal and tight-junctional alterations in hepatocytes by a Ca2+-dependent, PKCmediated mechanism: protective effect of PKA. Free Radic Biol Med 2006;40(11):2005–17.
- [100] Upreti RK, Kannan A, Pant A. Experimental exposure of arsenic in cultured rat intestinal epithelial cells and cell line: toxicological consequences. Toxicol In Vitro 2007;21(1):32–40.
- [101] Calatayud M, Gimeno-Alcañiz J, Vélez D, Devesa V. Trivalent arsenic species induce changes in expression and levels of proinflammatory cytokines in intestinal epithelial cells. Toxicol Lett 2014;224(1):40–6.
- [102] Yang C-Y, Chang C-C, Ho S-C, Chiu H-F. Is colon cancer mortality related to arsenic exposure? J Toxicol Environ Health Part A 2008;71(8):533–8.
- [103] Acharya S, Chaudhuri S, Chatterjee S, Kumar P, Begum Z, Dasgupta S, et al. Immunological profile of arsenic toxicity: a hint towards arsenic-induced carcinogenesis. Asian Pac J Cancer Prev 2010;11:479–90.
- [104] Tchounwou PB, Patlolla AK, Centeno JA. Invited reviews: carcinogenic and systemic health effects associated with arsenic exposure—a critical review. Toxicol Pathol 2003;31(6):575–88.
- [105] Uede K, Furukawa F. Skin manifestations in acute arsenic poisoning from the Wakayama curry-poisoning incident. Br J Dermatol 2003;149(4):757–62.
- [106] Heywood R, Sortwell R. Arsenic intoxication in the rhesus monkey. Toxicol Lett 1979;3(3):137–44.
- [107] Arnold LL, Eldan M, van Gemert M, Capen CC, Cohen SM. Chronic studies evaluating the carcinogenicity of monomethylarsonic acid in rats and mice. Toxicology 2003;190(3):197–219.

- [108] Islam M, Awal M, Kabir S, Islam M, Islam M. Effects of selenium, iron and zinc supplementation on arsenic concentration in tissues with gross pathology in Long Evans rats. Bangladesh J Vet Med 2012;3(1):83–5.
- [109] Ghosh M, Paul G. Intestinal dysfunction and alteration of various systemic and morphometric characters in albino rats (Charles Foster) under stress of inorganic arsenic (iAs) compounds: a pilot study. Int J Pharm Bio Sci 2013;4(2):(B)1008–(B)1016.
- [110] Rahman A, Vahter M, Ekström E-C, Persson L-Å. Arsenic exposure in pregnancy increases the risk of lower respiratory tract infection and diarrhea during infancy in Bangladesh. Environ Health Perspect 2011;119(5):719–24.
- [111] Black RE, Cousens S, Johnson HL, Lawn JE, Rudan I, Bassani DG, et al. Global, regional, and national causes of child mortality in 2008: a systematic analysis. Lancet 2010;375(9730):1969–87.
- [112] Deb D, Biswas A, Ghose A, Das A, Majumdar KK, Guha Mazumder DN. Nutritional deficiency and arsenical manifestations: a perspective study in an arsenic-endemic region of West Bengal, India. Publ Health Nutr 2013;16(9):1644–55.
- [113] Majumdar KK. Effect of drinking arsenic-contaminated water in children. Indian J Public Health 2012;56(3):223-6.
- [114] Choudhry ZK, Misbahuddin M, Hossain AM, Saleh AA. Inhibitory effect of arsenic on aerobic gut flora in rat. Bangladesh Med Res Counc Bull 2010;35(3):79–83.
- [115] Lu K, Mahbub R, Cable P, Ru H, Parry N, Bodnar W, et al. Gut microbiome phenotypes driven by host genetics affect arsenic metabolism. Chem Res Toxicol 2014;27(2):172-4.
- [116] Zoghi A, Khosravi-Darani K, Sohrabvandi S. Surface binding of toxins and heavy metals by probiotics. Mini Rev Med Chem 2014;14(1):84–98.
- [117] Michalke K, Schmidt A, Huber B, Meyer J, Sulkowski M, Hirner AV, et al. Role of intestinal microbiota in transformation of bismuth and other metals and metalloids into volatile methyl and hydride derivatives in humans and mice. Appl Environ Microbiol 2008;74(10):3069–75.
- [118] Lu K, Abo RP, Schlieper KA, Graffam ME, Levine S, Wishnok JS, et al. Arsenic exposure perturbs the gut microbiome and its metabolic profile in mice: an integrated metagenomics and metabolomics analysis. Environ Health Perspect 2014;122(3):284–91.
- [119] Portales PD, Baranda L, Díaz-Barriga F, Saavedra-Alanis V, Layseca E, Gonzalez-Amaro R. Effect of arsenic, cadmium and lead on the induction of apoptosis of normal human mononuclear cells. Clin Exp Immunol 2002;129(1):69–77.
- [120] Biswas R, Ghosh P, Banerjee N, Das J, Sau T, Banerjee A, et al. Analysis of T-cell proliferation and cytokine secretion in the individuals exposed to arsenic. Hum Exp Toxicol 2008;27(5):381–6.
- [121] Binet F, Antoine F, Girard D. Interaction between arsenic trioxide and human primary cells: emphasis on human cells of myeloid origin. Inflamm Allergy Drug Targets 2009;8(1):21–7.
- [122] Rahman A, Vahter M, Ekström E-C, Rahman M, Mustafa AHMG, Wahed MA, et al. Association of arsenic exposure during pregnancy with fetal loss and infant death: a cohort study in Bangladesh. Am J Epidemiol 2007;165(12):1389–96.
- [123] Rahman A, Persson L-Å, Nermell B, El Arifeen S, Ekström E-C, Smith AH, et al. Arsenic exposure and risk of spontaneous abortion, stillbirth, and infant mortality. Epidemiology 2010;21(6):797–804.
- [124] Raqib R, Ahmed S, Sultana R, Wagatsuma Y, Mondal D, Hoque AMW, et al. Effects of in utero arsenic exposure on child immunity and morbidity in rural Bangladesh. Toxicol Lett 2009;185(3):197–202.
- [125] Rager JE, Bailey KA, Smeester L, Miller SK, Parker JS, Laine JE, et al. Prenatal arsenic exposure and the epigenome: altered microRNAs associated with innate and adaptive immune signaling in newborn cord blood. Environ Mol Mutagen 2013;55(3):196–208.
- [126] Ahmed S, Kippler M, Wagatsuma Y, Al-Arifeen S, Raqib R, Vahter M. Prenatal arsenic exposure is associated with impaired thymic function in newborns. Toxicol Lett 2012;211:S133.

- [127] Ahmed S, Ahsan KB, Kippler M, Mily A, Wagatsuma Y, Hoque AW, et al. In utero arsenic exposure is associated with impaired thymic function in newborns possibly via oxidative stress and apoptosis. Toxicol Sci 2012;129(2):305–14.
- [128] Díaz-Villaseñor A, Burns AL, Hiriart M, Cebrián ME, Ostrosky-Wegman P. Arsenic-induced alteration in the expression of genes related to type 2 diabetes mellitus. Toxicol Appl Pharmacol 2007;225(2):123–33.
- [129] Canet MJ, Hardwick RN, Lake AD, Kopplin MJ, Scheffer GL, Klimecki WT, et al. Altered arsenic disposition in experimental nonalcoholic fatty liver disease. Drug Metab Dispos 2012;40(9):1817–24.
- [130] American Diabetes Association (AD). Standards of medical care in diabetes. Diabetes Care 2004;27:S15–35.
- [131] Pan W-C, Seow WJ, Kile ML, Hoffman EB, Quamruzzaman Q, Rahman M, et al. Association of low to moderate levels of arsenic exposure with risk of type 2 diabetes in Bangladesh. Am J Epidemiol 2013;178(10):1563-70.
- [132] Wang W, Xie Z, Lin Y, Zhang D. Association of inorganic arsenic exposure with type 2 diabetes mellitus: a meta-analysis. J Epidemiol Community Health 2014;68(2):176–84.
- [133] Tseng C-H. The potential biological mechanisms of arsenic-induced diabetes mellitus. Toxicol Appl Pharmacol 2004;197(2):67–83.
- [134] Farrell GC. Non-alcoholic steatohepatitis: what is it, and why is it important in the Asia–Pacific region? J Gastroenterol Hepatol 2003;18(2):124–38.
- [135] Rivera SP, Saarikoski ST, Hankinson O. Identification of a novel dioxin-inducible cytochrome P450. Mol Pharmacol 2002;61(2):255–9.
- [136] Anwar-Mohamed A, Elbekai RH, El-Kadi AO. Regulation of CYP1A1 by heavy metals and consequences for drug metabolism; *Expert Opinion on Drug Metabolism & Toxicology*, 2009;5(5):501–21.
- [137] Anwar-Mohamed A, Abdelhamid G, Amara IE, El-Kadi AO. Differential modulation of aryl hydrocarbon receptor regulated enzymes by arsenite in the kidney, lung, and heart of C57BL/6 mice. Arch Toxicol 2012;86(6):897–910.
- [138] Mohammadi-Bardbori A, Bengtsson J, Rannug U, Rannug A, Wincent E. Quercetin, resveratrol, and curcumin are indirect activators of the aryl hydrocarbon receptor (AHR). Chem Res Toxicol 2012;25(9):1878-84.
- [139] Potapovich AI, Lulli D, Fidanza P, Kostyuk VA, De Luca C, Pastore S, et al. Plant polyphenols differentially modulate inflammatory responses of human keratinocytes by interfering with activation of transcription factors NFκB and AhR and EGFR-ERK pathway. Toxicol Appl Pharmacol 2011;255(2):138–49.
- [140] Yadav S, Anbalagan M, Shi Y, Wang F, Wang H. Arsenic inhibits the adipogenic differentiation of mesenchymal stem cells by down-regulating peroxisome proliferator-activated receptor gamma and CCAAT enhancer-binding proteins. Toxicol In Vitro 2013;27(1):211–19.
- [141] Wauson EM, Langan AS, Vorce RL. Sodium arsenite inhibits and reverses expression of adipogenic and fat cell-specific genes during *in vitro* adipogenesis. Toxicol Sci 2002;65(2):211–19.
- [142] Koshiuka K, Elstner E, Williamson E, Said J, Tada Y, Koeffler H. Novel therapeutic approach: organic arsenical (melarsoprol) alone or with all-trans-retinoic acid markedly inhibit growth of human breast and prostate cancer cells *in vitro* and *in vivo*. Br J Cancer 2000;82(2):452–8.
- [143] Kumagai T, Ikezoe T, Gui D, O'Kelly J, Tong X-J, Cohen FJ, et al. RWJ-241947 (MCC-555), a unique peroxisome proliferator-activated receptor-γ ligand with antitumor activity against human prostate cancer *in vitro* and in beige/nude/X-linked immunodeficient mice and enhancement of apoptosis in myeloma cells induced by arsenic trioxide. Clin Cancer Res 2004;10(4):1508–20.
- [144] Chu K-H, Lee C-C, Hsin S-C, Cai B-C, Wang J-H, Chiang B-L. Arsenic trioxide alleviates airway hyperresponsiveness and eosinophilia in a murine model of asthma. Cell Mol Immunol 2010;7(5):375–80.

This page intentionally left blank

11

Cutaneous Toxicology of Arsenic

Katherine Marchiony Hunt¹, Ritesh Kumar Srivastava², and Mohammad Athar²

¹UNIVERSITY OF ALABAMA SCHOOL OF MEDICINE, UNIVERSITY OF ALABAMA AT BIRMINGHAM, BIRMINGHAM, ALABAMA, USA, ²DEPARTMENT OF DERMATOLOGY AND SKIN DISEASE RESEARCH CENTER, UNIVERSITY OF ALABAMA AT BIRMINGHAM, BIRMINGHAM, ALABAMA, USA

CHAPTER OUTLINE

11.1	Introduction	301
11.2	Epidemiology	302
11.3	Clinical Manifestations	303
11.4	Histopathology	304
11.5	Molecular Pathogenesis	304
11.6	Treatment	310
Refe	leferences	

11.1 Introduction

Arsenic is a well-known toxin and human carcinogen known to impact cellular processes in numerous organ systems in a time- and dose-dependent manner [1–5]. However, its manifestations typically first appear in the skin. These early cutaneous manifestations act as a surrogate marker for internal diseases including hypertension, ischemic heart disease, atherosclerosis [3], peripheral vascular disease, diabetes mellitus, and peripheral neuropathy [1–3,5]. Arsenic is a metalloid found in great abundance within Earth's crust. In nature, inorganic arsenic is often complexed with the mineral pyrite and easily dissociates to enter ground water with subtle changes in soil pH, temperature, oxidation state, and solution composition [3]. Human activities have also contributed to rising groundwater arsenic concentrations. Contamination has been found to occur as a consequence of mining minerals such as silver, gold, tin, copper, lead, tungsten, zinc, and cobalt [6–9]. The rainwater-induced leaching of arsenic from chromate copper arsenate (CCA)-treated wood is an additional possible source of groundwater arsenic. Though CCA-treated wood was phased out of residential use in 2004, it remains very popular among commercial builders given its service life of up to 40 years [10]. Chronic arsenic

ingestion is the most common cause of arsenicosis, which may occur through environmental, occupational, and accidental exposure [4,5]. Ingestion of contaminated foods such as fish, cereal, algae [4], rice [11], and nutritional supplements [12] have also been associated with its toxicity.

Historically, arsenic was used as a medicine for various maladies, particularly those of the skin. It was used in ancient Greece to treat diseases such as eczema, furuncles, leprosy, lichen rubor, lupus, molluscum contagiosum, pemphigus, psoriasis, syphilis, urticaria, and warts [13]. Compounds known as Fowler's solution (liquor potassii arsenites), arsenic bromidum, arsenic iodidum, Pearson's solution (liquor sodii arseniatis), arsenicum hydrogenisatum, Donovan's solution (liquor arseni et hydrargyri iodide), and Asiatic pills (arsenic with opium or pepper) were historically used to treat disease [8,13]. Asiatic pills were one of the most commonly prescribed arsenicals and are still available for purchase in the Far East with possible continued use in traditional Chinese medicine [6,7]. Arsenic was first listed in the *United States Pharmacopoeia* in 1850 [14] and the Western world continued using Fowler's solution to treat asthma as late as the 1960s [6,7]. Interestingly, arsenic compounds are still used today in the treatment of tropical diseases such as African trypanosomiasis [14] and hematologic malignancies including promyelocytic leukemia [6,7]. Arsenic's ability to cause disease was documented as early as the late 19th century when it was noted to cause palmoplantar keratoses, papulosquamous eruptions, and pruritus [8,13].

11.2 Epidemiology

Abundant epidemiological data from countries such as Taiwan, Bangladesh, Chile, India, and Argentina have helped to show that arsenic is a serious and widespread threat to public health [11]. Current estimates suggest that as many as 45 million individuals in Bangladesh alone are at risk of arsenic exposure exceeding a level of $50 \mu g/L$ [15]. The World Health Organization (WHO) recommends a maximum allowable concentration of $10 \mu g/L$ of arsenic in drinking water, which suggests that arsenicosis in Bangladesh may be an even greater health concern than already known [16]. It is estimated that 200 million people worldwide are at risk of toxic exposure [11]. Evidence suggests a lifetime dose of 0.5 to 1 gram is capable of promoting arsenical keratoses and skin cancers [8,9]. Estimates from endemic areas of China have shown that approximately 10,000 individuals are currently suffering from arsenicosis [17]. In the United States, an estimated 1.5 million individuals were exposed to arsenic in the workplace in 1973 [18]. A retrospective analysis conducted by Focazio et al. predicted that approximately 11% of all United States public water systems had arsenic concentrations exceeding 10µg/L with the greatest exposures occurring in the Southwest, Midwest, and Northeast [19]. The largest groundwater burden was also found to occur in water sources serving populations of less than 1000 people [19]. Arsenic's ubiquitous nature and toxic potential have led the United States Agency for Toxic Substances and Disease Registry (ATSDR) to name arsenic as the number one threat to public health. In addition, the International Agency for Research on Cancer (IARC) categorized this metalloid as a class I human carcinogen [20]. Also, other factors such as environmental UV exposure, immunosuppression, and certain dietary factors may augment various cutaneous manifestations in chronically arsenic-exposed human populations [21]. In this regard, *in utero* exposure to arsenic results in impaired newborn and childhood immunity as assessed in chronically exposed human populations of Bangladesh [22,23].

11.3 Clinical Manifestations

Although arsenic impacts physiological cellular processes in numerous organ systems, the outcomes of its toxicity are usually first seen in the skin. Acute exposure, the significantly rarer entity, typically occurs as a result of medicinal, accidental, homicidal, or suicidal attempts. Skin manifestations of acute toxicity include flushing, erythema, facial edema, acrodynia, urticaria, alopecia, loss of nails, and Mees lines visible on nails approximately 8 weeks after exposure [6–9,24].

Chronic skin manifestations include hyperpigmentation with superimposed guttate hypopigmentation described as "raindrops on a dusty road," punctate hyperkeratoses on the palms and soles [25,26], diffuse alopecia, blackfoot vascular occlusive disease, Bowen's disease (BD, also called squamous cell carcinoma *in situ*), squamous cell carcinoma (SCC), and basal cell carcinoma (BCC) [5–9,12,24,27]. Blackfoot disease typically affects the lower extremities and often results in gangrene if left untreated [28]. An association between arsenic exposure and Merkel cell carcinoma, a neoplasm with substantially less common occurrence, has also been identified [4]. The latency period for the development of arsenical skin disease ranges from 30 to 50 years [6]. The development of skin cancer in sun-protected areas is especially concerning for the existence of chronic arsenic exposure. These cutaneous manifestations serve as a surrogate marker for associated diseases such as hypertension, ischemic heart disease, atherosclerosis [3], peripheral vascular disease, diabetes mellitus, peripheral neuropathy [2], bone marrow hypoplasia, and hepatic fibrosis [1,3,5–9,24].

Arsenical keratoses are precancerous lesions that begin as yellow pinpoint hyperkeratotic papules ranging in size from 0 to 2 mm, most commonly on pressure points of the palms and soles such as the lateral borders of the hands [12]. They eventually progress to larger corn-like papules and may eventually coalesce to form plaques [6–9,24]. These lesions are more commonly reported in patients with chronic medicinal exposure rather than that from the workplace [12].

Arsenical BD lesions typically occur in multiples on affected individuals and appear as small skin-colored to pink papules covered by a hyperkeratotic crust or horny layer. Removal of this crust often reveals an erythematous, papillomatous, oozing base [6–9,24]. In contrast to arsenical keratoses, BD lesions continually increase in size and may eventually form nodules. These lesions progress to invasive SCC in approximately 5 to 20% of cases [6–9,24].

Arsenic-induced SCC is more aggressive than its chronic UV-induced counterpart. In fact, 33% of untreated arsenical SCCs have been found to demonstrate metastatic behavior. [6-9,24,29]. These lesions may arise *de novo*, from existing benign lesions (arsenical keratosis), or may be the result of BD progression. It is believed that about half of all arsenic-induced SCCs ariose *de novo* [12]. Arsenical keratoses at risk for progression to SCC commonly present with pain, bleeding, fissuring, and eventually ulceration [28]. Early-life exposure to arsenic, multiple lesions, and acral sites are risk factors for the development of SCC [12]. Although not proven

experimentally, it is believed that pressure points on the palms and soles more commonly undergo malignant transformation due to ongoing irritation and trauma in these areas [28].

11.4 Histopathology

Arsenical keratoses display mild to moderate keratinocyte atypia, severe hyperkeratosis, and scattered parakeratosis. Adnexal structures are typically spared and basophilic degeneration of dermal connective tissue may or may not occur [30]. There is no reliable method to distinguish arsenical keratoses from actinic keratoses. However, some cases of arsenical keratoses demonstrate unique vacuolization of epithelial cells, keratin horn formation, absent solar elastosis, and a chronic dermal lymphocytic infiltrate [28]. Arsenic-induced BD, SCC, and BCC demonstrate the same histopathologic features as found in their non-arsenic-associated counterparts [6,8]. BD lesions exhibit full-thickness epidermal atypia over a broad zone with prominent pleomorphism and frequent mitoses observed in keratinocytes [30]. Well-differentiated SCCs exhibit histological features of keratoses with the addition of glassy, brightly eosinophilic keratinocytes forming detached aggregations and downward proliferating lobules. These cells display varying degrees of pleomorphism, mitoses, and inflammatory infiltrate. Keratinocytes aggregate at varying levels of the dermis and often exhibit prominent nucleoli. Poorly differentiated SCCs lack overt keratinization, may display spindle cell morphology, and more commonly exhibit perineural invasion. BCCs all display aggregations of basaloid keratinocytes within a variably fibromyxoid stroma. The cells typically exhibit large nuclei with scant cytoplasm. Lesions may demonstrate a focal connection to the epidermis, indistinct cellular borders, and retraction of the fibromyxoid stroma, which creates characteristic clefts around islands of tumor cells [30].

It is important to note that the clinical features of cutaneous arsenicosis may be similar to those of other diseases. Inherited skin diseases such as punctate palmoplantar keratoderma or Darier disease may appear clinically similar to chronic arsenicosis. However, individuals with heritable diseases typically present earlier in life and report a positive family history of the disease. Unlike arsenical keratoses, the lesions of palmoplantar keratoderma typically demonstrate crater-like pits following removal of keratin plugs. Common conditions such as corns and verruca vulgaris may be mistaken for arsenical keratoses. However, verruca exhibits characteristic thrombosed capillaries following removal of the hyperkeratotic wart surface. Corns occur in lower numbers than arsenical keratoses and rarely present on the hands [28,30].

11.5 Molecular Pathogenesis

11.5.1 Oxidative Stress

A large body of evidence has described arsenic's ability to cause oxidative and nitrosative stress by producing reactive oxygen species (ROS) and reactive nitrogen species (RNS) respectively [3,31–33]. The indicators of arsenic-induced oxidative stress include high levels of 8-hydroxydeoxyguanosine, lipid peroxides, glutathione, heme oxygenase-1, A170, and peroxiredoxin 1 [3,34]. In this regard, mitochondria-mediated production of hydroxyl and hydrogen peroxide plays a key role in the reduction of intracellular thiols, particularly glutathione [31,32]. Arsenic increases the expression and activity of NAD(P)H oxidase (NADPHO), an enzyme that produces the superoxide anion [3,35] through up-regulation of its subunit p22phox [36], the translocation of NADPHO-activating Rac1, and the activation of NADPHOenhancing CDC42 [3].

Chronic arsenic exposure also depletes stores of the antioxidant nitric oxide (NO) through its reaction with the superoxide anion and as a consequence of NO synthase (NOS) uncoupling. Arsenic has been found to uncouple the NOS enzyme by reducing levels of its cofactor, tetrahydrobiopterin (BH4). Unchanged concentrations of L-arginine in the setting of decreased BH4 shift NOS's production from NO to superoxide instead [3]. Arsenic has also been found to decrease NO and increase intracellular superoxide levels by increasing the abundance of Big endothelin 1 (Big ET1), a biological precursor to endothelin 1 (ET1). ET1 increases oxidative stress by both simulating NAPHO and uncoupling NOS [37].

11.5.2 Genotoxicity

Arsenic destabilizes the genome by disrupting chromosomes, damaging DNA strands, and weakening cellular DNA repair mechanisms [4,5,38]. A higher incidence of chromosomal aberrations and inappropriate sister chromatid exchanges in the DNA of individuals subjected to chronic arsenic exposure has been identified [38]. Tri- and pentavalent arsenic species readily accept methyl groups from S-adenosylmethionine and subsequently methylate DNA strands in an unregulated manner [4]. Arsenic's predilection for sulfhydryl (SH) groups enables the metalloid to bind SH moieties on various DNA repair enzymes such as I-III DNA polymerase β , O6-methyl-guanine-DNA methyltransferase, poly ADP ribose polymerase, and excision repair cross-complementing rodent repair, complementation group 2 (ERCC2), thus rendering them inactive [5]. Consequently, DNA repair processes such as nucleotide excision repair, base excision repair, mismatch repair [37], and transcriptional proofreading [39] are negatively impacted. Arsenic acts synergistically with other carcinogens to enhance mutations and destabilize the genome with its combination of direct genotoxicity and hampered DNA repair [4,5]. These effects may ultimately be important in unraveling the molecular mechanism of arsenic's carcinogenicity.

11.5.3 Disrupted Signal Transduction Pathways

Arsenic promotes many of its deleterious effects by aberrantly activating various signal transduction pathways. This is predominantly accomplished through As binding directly and/ or by ROS-mediated depletion of SH moieties [3,40]. Recent studies show that arsenic promotes growth and proliferation by altering the canonical Hippo signaling pathway (as shown in Figure 11–1) via up-regulating STE20-like kinase 1/2, Salvador homolog 1, large tumor suppressor kinase 1/2, and Mps one binder kinase activator-like 1A [41]. Independent of canonical Hippo signaling, it also promotes epidermal inflammation by promoting phosphorylation of



FIGURE 11–1 Flow diagram showing arsenic-mediated modulation of Hippo signaling and Yap activity in murine skin. Exposure to arsenic leads to up-regulation of core components of the Hippo signaling pathway. It is known that Mst1/2 kinases and Sav1 form a complex to phosphorylate and activate LATS1/2. LATS1/2 kinases in turn phosphorylate and inhibit the transcription co-activators YAP and TAZ, the two major downstream effectors of the Hippo signal transduction pathway. When dephosphorylated, YAP/TAZ translocate into the nucleus and interact with TEAD, other transcription factors, and genes that promote cell proliferation and inhibit apoptosis. Arsenic modulates α -catenin expression in the murine epidermis; α -catenin, together with p-YAP, is known to be a component of adherens junctions. Arsenic also activates Yap independent of its upstream regulation. Upon migration to the nucleus, Yap induces transcription of its target genes including *Cry61*, *Gli2*, *Ankrd1*, and *Ctgf*. YAP/TAZ is retained in the cytoplasm by sequestration via the 14-3-3 protein. Free and phosphorylated-Yap is targeted for β TRCP-dependent proteasomal degradation. Red arrows indicate upregulation by arsenic.

Yes-associated protein (Yap), a Hippo signaling protein that regulates tight junctions of epithelium when phosphorylated [41].

Although the Nrf2-Keap1 pathway (Nuclear factor 2-Kelch-like ECH associated protein 1) has traditionally been known for its role in promoting an antioxidant stress response, it has recently been implicated in arsenic-induced carcinogenesis. Nrf2 is a transcription factor that is constitutively inhibited by Keap1 until cell stressors inactivate Keap1. If this occurs, Nrf2 is allowed to accumulate, translocate to the nucleus, and activate the transcription of antioxidant machinery such as heme oxygenase-1 (HO-1) and thioredoxin reductase-1 (TrxR1) [42-44]. However, Nrf2 is constitutively activated in some cancer cells and may, in fact, play a role in chemoresistance. Evidence suggests that deactivation of the aberrantly overactivated Nrf2-Keap1 pathway in some cancer cells actually promotes sensitivity to chemotherapy [45].

Arsenic has also been found to promote cell proliferation and growth by stimulating constitutive activation of nuclear factor kappa-light-chain-enhancer of activated B cells (NF- κ B), a transcription factor that has been implicated in chronic inflammation and various cancers [46]. This is accomplished, at least in part, through direct binding of arsenic with a free SH moiety on inhibitory κ B α kinase (I κ K), which renders it active. Activated I κ K then phosphorylates inhibitory κ B α (I κ B α), targeting it for ubiquitin-dependent proteasomal degradation. Since I κ B α is normally responsible for inhibition of NF- κ B, this leads to constitutive activation of NF- κ B and resultant cell survival [47–49]. Arsenic also acts to promote cell survival by hypermethylating and subsequently silencing the promoter of the pro-apoptotic death-associated protein kinase (*DAPK*) and *p16* genes [2]. It decreases expression of cell cycle control molecules such as cyclin-dependent kinase inhibitor 1B (p27Kip1), an inhibitor of the cyclin D1/CDK4 complex, resulting in uncontrolled cell proliferation [50]. Arsenic also up-regulates *cyclin D1* by altering its epigenetic modification through microRNA 2909 [51].

The metalloid also activates oncogenes such as the mineral dust-induced gene (*MDIG*), a gene that has been linked to lung [52], breast, colon, and gastric cancers, esophageal squamous cell, hepatocellular, and renal cell carcinomas, as well as some forms of lymphoma. Enhancement in the JNK and STAT3 pathways is required for this *MDIG* activation [53]. It promotes epithelial to mesenchymal transition (EMT), invasion, and migration by disrupting the Akt signaling pathway [54] and up-regulating the β -catenin-vascular endothelial growth factor pathway [55].

It is well established that BCCs are driven by the aberrant activation of the Sonic hedgehog signaling pathway (Shh) [56]. Arsenic has been shown to activate this signaling pathway by decreasing the stability of the repressor form of GLI3, one of the transcription factors that ultimately regulates Shh activity. High levels of arsenic exposure were associated with high levels of Shh activity in a cohort of bladder cancer patients [57]. It is possible that this pathway is likewise up-regulated in skin cancers, although it is yet to be demonstrated experimentally (see Figure 11–2). In addition, arsenicals have been found to antagonize the Shh pathway largely by targeting Gli transcriptional effectors in cancer cells in the short-term exposure protocols. In this regard, arsenic was shown to block Shh-induced ciliary accumulation of Gli2, the primary activator of Shh-dependent transcription [58,59]. These differences may be dependent on various contexts such as cell type (normal versus cancer cell), total dose, and duration of exposure. However, involvement of other mechanisms that may also play important roles in the antitumor activities of arsenicals cannot be ruled out at this stage [33].

11.5.4 Immune Dysfunction and Inflammatory Responses

Arsenic impairs innate and cell-mediated immune function largely by activating the unfolded protein response (UPR) homeostatic mechanism [60]. Dysregulation of the UPR has been implicated in carcinogenesis [61] as well as immune dysfunction [60]. It is activated in response to a large number of unfolded proteins present in the endoplasmic reticulum (ER) lumen. Three distinctive pathways, named after their unfolded protein sensors, work to activate the UPR (see Figure 11–3). Inositol-requiring enzyme-1 (IRE1), protein kinase RNA-like



FIGURE 11–2 Arsenic-induced activation of the Hh signaling pathway. Binding of Hh ligands to PTCH de-represses SMO thereby allowing its signal for the transcriptional activation of Gli1/2. Generally, it is known that activation and nuclear translocation of Gli1/2 involves dissociation of Gli1/2 from its endogenous inhibitor SuFu. Kif7 cooperates with SuFu in catalyzing dissociation of the Sufu–Gli complex. Chronic arsenic exposure at low levels favors the nuclear translocation of Glis by an unknown mechanism. Nuclear import of Glis induces transcription of its target genes including *Gli1*, *Ptch1*, and *cyclin D1* that ultimately results in uncontrolled proliferation of cells.

endoplasmic reticulum kinase (PERK), and activating transcription factor 6 (ATF6) act as unfolded protein sensors. IRE1 and PERK activate signaling cascades that inhibit translation while ATF6 promotes the transcription of additional protein folding chaperones [62]. Evidence suggests that chronic arsenic exposure activates all three signaling pathways of the UPR by upregulating IRE1, PERK, and ATF6 [25]. Arsenic has been found to disrupt macrophage functions such as bacterial engulfment, cytokine release, and bacterial digestion through activation of PERK signaling. Reactive oxygen species have been causally implicated in UPR activation because N-acetylcysteine (NAC) was found to block the detrimental effects observed in macrophages [60].

Arsenic has also been found to exacerbate lipopolysaccharide-induced inflammation, mediated by macrophages and monocytes, by ROS stimulation of the Src kinase pathway. Src kinase is a non-receptor tyrosine kinase that has been implicated in the progression of numerous solid tumors. This increases the abundance of p38 kinase, which causes monocytic-lineage cells to mount an exaggerated and damaging immune response resulting in overproduction of inflammatory cytokines [63]. Arsenic has also been found to damage cell-mediated immunity. One study found that arsenic exposure enhanced HIV's ability to infect human



FIGURE 11–3 Flow diagram showing modulation of UPR signaling by arsenic. Under normal conditions in cells, the ER chaperone GRP78 is known to bind with the domains of the ER stress sensor proteins PERK, IRE1 α and ATF6 α , which is important for sustaining these proteins in an inactive form. Under the conditions of ER stress such as chronic arsenic exposure, GRP78 favorably binds to misfolded proteins and leads to activation of these sensors. Release of GRP78 from ER sensor proteins results in the activation of PERK, through PERK homodimerization and trans-auto-phosphorylation. Phosphorylated PERK then phosphorylates the translation initiation factor eIF2 and reduces the overall frequency of mRNA translation initiation. However, designated mRNAs, such as ATF4 mRNA, are preferentially translated in the presence of p-eIF2. ATF4 activates the transcription of UPR target genes. The release of GRP78 also allows IRE1 α dimerization and activates its protein kinase activity through autophosphorylation. IRE1 α then removes a 26-base intron from XBP-1u mRNA (u-unspliced). The spliced XBP1s (s-spliced) mRNA, which is a potent transcription factor that translocates into the nucleus and activates expression of UPR target genes. Release of GRP78 from ATF6 α allows ATF6 α migration to the Golgi apparatus, where it is cleaved by the proteases S1P and S2P to the ATF6 p50 fragment; following translocation to the nucleus it become transcriptionally active. Abbreviations: S1P, site-1 protease; S2P, site-2 protease; XBP-1, X-box-binding protein 1; ATF4, activation transcription factor 4. Red arrows indicate proteins upregulated by arsenic.

dendritic cells by dampening post-viral-entry cellular defenses [64]. Another investigation identified a disproportionate number of apoptotic CD4⁺ lymphocytes in arsenical BD lesions. Additionally, the peripheral blood of individuals with these lesions demonstrated fewer CD4⁺ T lymphocytes versus controls [65].

Arsenic promotes inflammation by activating the p38 MAPK pathway. This occurs through oxidation of an SH group on MAPK phosphatase 1, a protein tyrosine phosphatase responsible for the inhibition of an upstream MAPK effector, epidermal growth factor receptor (EGFR). Thus, oxidation of this SH group transiently inactivates the MAPK phosphatase 1 inhibitor, allowing increased pro-inflammatory MAPK signaling [3]. Furthermore, up-regulation of MAPK signaling has been causally associated with increased signaling through activated protein 1 (AP-1), a transcription factor that is known to regulate responses to cytokines, growth factors, stress, infection, and apoptosis [3,5,66]. In summary, arsenic disrupts signal transduction and effectively promotes carcinogenesis by activating oncogenes, inhibiting tumor suppressors, and up-regulating pathways that may invoke cutaneous inflammatory signaling [2,3,41,46].

11.6 Treatment

Acute arsenic toxicity is best treated by chelation therapy with agents such as dimercaprol (used most commonly), dimercaptosuccinic acid, and dimercaptopropane sulfonic acid [6,8]. There are no standard guidelines for the treatment of chronic arsenicosis or for the screening of underlying disease. Some authors recommend that patients with arsenicosis undergo a detailed history and physical as well as total skin examination every 6 months [12]. The history should focus on an individual's occupation, living conditions, environmental exposure, and medicinal exposure spanning the previous 10 to 40 years. A biopsy is indicated for any lesions that are changing, erythematous, or ulcerated to rule out malignancy. Laboratory tests such as a complete blood count, renal function tests, and liver function tests have been proposed. Hair and nail sampling may be indicated to quantify arsenic exposure [28]. Annual chest radiographs, nerve conduction studies, and other indicated clinical tests have been proposed [28] given the common coexistence of other underlying diseases such as ischemic heart disease, atherosclerosis [3], peripheral vascular disease, diabetes mellitus, peripheral neuropathy [2], and Raynaud's phenomenon [1,3,5] in the setting of arsenical skin disease. Though the treatment of arsenical keratoses is not required, these lesions are often treated to reduce patient discomfort caused by the lesions [28]. The most common treatment options include surgical excision, curettage, cryosurgery [6,8], and carbon dioxide laser ablation [6]. Although topical chemotherapy with 5-fluorouracil has also been employed, this option is less effective for the treatment of arsenical keratoses as opposed to actinic keratoses [6]. Topical keratolytics, photodynamic therapy, oral retinoids [67,68], and topical 5% imiquimod [69] have been used with some success, though these treatments are anecdotal or based upon small-scale investigations [6,8]. Evolving evidence describing the molecular pathogenesis of these cutaneous lesions may be helpful in developing novel therapeutic interventions for skin diseases and disorders among arsenic-exposed human populations.

References

- [1] Ahsan H, Steinmaus C. Invited commentary: use of arsenical skin lesions to predict risk of internal cancer: implications for prevention and future research. Am J Epidemiol 2013;177(3):213–16.
- [2] Banerjee N, Paul S, Sau TJ, Das JK, Bandyopadhyay A, Banerjee S, et al. Epigenetic modifications of DAPK and p16 genes contribute to arsenic-induced skin lesions and nondermatological health effects. Toxicol Sci 2013;135(2):300–8.
- [3] Kumagai Y, Sumi D. Arsenic: signal transduction, transcription factor, and biotransformation involved in cellular response and toxicity. Annu Rev Pharmacol Toxicol 2007;47:243–62.
- [4] Martinez VD, Vucic EA, Becker-Santos DD, Gil L, Lam WL. Arsenic exposure and the induction of human cancers. J Toxicol 2011 2011:431287.
- [5] Yu HS, Liao WT, Chai CY. Arsenic carcinogenesis in the skin. J Biomed Sci 2006;13(5):657-66. [Epub 2006 Jun 29].
- [6] Schwartz RA. Arsenic and the skin. Int J Dermatol 1977;36(4):241-50.
- [7] Ratnaike RN. Acute and chronic arsenic toxicity. Postgrad Med J 2003;79:391-6.
- [8] Mazumder DNG. Chronic arsenic toxicity and human health. Indian J Med Res 2008;128:436-47.
- [9] Rahman MM, Ng JC, Naidu R. Chronic exposure of arsenic via drinking water and its adverse health impacts on humans. Environ Geochem Health 2009;31:189–200.
- [10] Hasan AR, Hu L, Solo-Gabriele HM, Fieber L, Cai Y, Townsend TG. Field-scale leaching of arsenic, chromium and copper from weathered treated wood. Environ Pollut 2010;158(5):1479–86.
- [11] Naujokas MF, Anderson B, Ahsan H, Aposhian HV, Graziano JH, Thompson C, et al. The broad scope of health effects from chronic arsenic exposure: update on a worldwide public health problem. Environ Health Perspect 2013;121(3):295–302.
- [12] Wong SS, Tan KC, Goh CL. Cutaneous manifestations of chronic arsenicism: review of seventeen cases. J Am Acad Dermatol 1998;38(2 Pt 1):179–85.
- [13] Piffard HG. A Treatise on the materia medica and therapeutics of the skin. New York: William Wood and Co; 1881. 236 pp.
- [14] Amadori S, Fenaux P, Ludwig H, O'Dwyer M, Sanz M. Use of arsenic trioxide in haematological malignancies: insight into the clinical development of a novel agent. Curr Med Res Opin 2005;21:403–11.
- [15] Flanagan SV, Johnston RB, Zheng Y. Arsenic in tube well water in Bangladesh: health and economic impacts and implications for arsenic mitigation. Bull World Health Org 2012;90(11):839–46.
- [16] World Health Organization. Arsenic in drinking-water. Background document for development of WHO guidelines for drinking-water quality. Geneva, Switzerland: World Health Organization; 2011.
- [17] Yu G, Sun D, Zheng Y. Health effects of exposure to natural arsenic in groundwater and coal in China: an overview of occurrence. Environ Health Perspect 2007;115(4):636–42. [Epub 2007 Jan 9].
- [18] Blejer HP, Wagner W. Case study 4: Inorganic arsenic—ambient level approach to the control of occupational carcinogenic exposures. Ann NY Acad Sci 1976;271:179–86.
- [19] Focazio MJ, Welch AH, Watkins SA, Helsel DR, Horn MA. A retrospective analysis on the occurrence of Arsenic in ground-water resources of the United States and limitations in drinking-water-supply characterizations. Water-Resources Investigations Report 99-4279. Reston, VA: US Geological Survey; 2000.
- [20] International Agency for Research on Cancer (IARC). A review of human carcinogens: arsenic, metals, fibres, and dusts IARC monographs on the evaluation of carcinogenic risks to humans, vol. 100C. Lyon, France: International Agency for Research on Cancer, World Health Organization; 2012.
- [21] Bhattacharjee P, Banerjee M, Giri AK. Role of genomic instability in arsenic-induced carcinogenicity. A review. Environ Int 2013;53:29–40.
- [22] Raqib R, Ahmed S, Sultana R, Wagatsuma Y, Mondal D, Hoque AM, et al. Effects of in utero arsenic exposure on child immunity and morbidity in rural Bangladesh. Toxicol Lett 2009;185:197–202.

- [23] Ahmed S, Ahsan KB, Kippler M, Mily A, Wagatsuma Y, Hoque AM, et al. In utero arsenic exposure is associated with impaired thymic function in newborns possibly via oxidative stress and apoptosis. Toxicol Sci 2012;129(2):305–14.
- [24] Vahter M, Akesson A, Lidén C, Ceccatelli S, Berglund M. Gender differences in the disposition and toxicity of metals. Environ Res 2007;104(1):85–95.
- [25] Li C, Xu J, Li F, Chaudhary SC, Weng Z, Wen J, et al. Unfolded protein response signaling and MAP kinase pathways underlie pathogenesis of arsenic-induced cutaneous inflammation. Cancer Prev Res (Phila) 2011;4(12):2101–9.
- [26] Melkonian S, Argos M, Chen Y, Parvez F, Pierce B, Ahmed A, et al. Intakes of several nutrients are associated with incidence of arsenic-related keratotic skin lesions in Bangladesh. J Nutr 2012;142(12):2128–34.
- [27] Ghosh SK, Bandyopadhyay D, Bandyopadhyay SK, Debbarma K. Cutaneous malignant and premalignant conditions caused by chronic arsenicosis from contaminated ground water consumption: a profile of patients from eastern India. Skinmed 2013;11(4):211–16.
- [28] Duncan KO, Geisse JK, Leffell DJ. Epithelial precancerous lesions. In: Goldsmith LA, Katz SI, Gilchrest BA, Paller AS, Leffell DJ, Wolff K, editors. Fitzpatrick's dermatology in general medicine (8th ed). New York: McGraw-Hill; 2012. pp. 661–70.
- [29] Graham JH. Selected precancerous skin and mucocutaneous lesions. In: Susan B. Freitag, Diane L. Culhane, editors. Neoplasms of the skin and malignant melanoma: a collection of papers presented at the Twentieth Annual Clinical Conference on Cancer, 1975, at the University of Texas System Cancer Center, M.D. Anderson Hospital and Tumor Institute, Houston, Texas.
- [30] Bolognia JL, Jorizzo JL, Schaffer JV, editors. Dermatology, 3rd ed. Philadelphia, PA: Elsevier Limited; 2012. Chapter 88 (Environmental and sports-related skin diseases):1487–510; Chapter 108 (Actinic keratosis, basal cell carcinoma and squamous cell carcinoma):1773–93.
- [31] Liu SX, Davidson MM, Tang X, Walker WF, Athar M, Ivanov V, et al. Mitochondrial damage mediates genotoxicity of arsenic in mammalian cells. Cancer Res 2005;65(8):3236–42.
- [32] Liu SX, Athar M, Lippai I, Waldren C, Hei TK. Induction of oxyradicals by arsenic: implication for mechanism of genotoxicity. Proc Natl Acad Sci USA 2001;98(4):1643–8.
- [33] Thomas-Schoemann A, Batteux F, Mongaret C, Nicco C, Chéreau C, Annereau M, et al. Arsenic trioxide exerts antitumor activity through regulatory T cell depletion mediated by oxidative stress in a murine model of colon cancer. J Immunol 2012;189(11):5171–7.
- [34] Niedzwiecki MM, Hall MN, Liu X, Oka J, Harper KN, Slavkovich V, et al. A dose-response study of arsenic exposure and global methylation of peripheral blood mononuclear cell DNA in Bangladeshi adults. Environ Health Perspect 2013;121(11-12):1306-12.
- [35] Hughes MF, Beck BD, Chen Y, Lewis AS, Thomas DJ. Arsenic exposure and toxicology: a historical perspective. Toxicol Sci 2011;123(2):305–32.
- [36] Lynn S, Gurr JR, Lai HT, Jan KY. NADH oxidase activation is involved in arsenite-induced oxidative DNA damage in human vascular smooth muscle cells. Circ Res 2000;86(5):514–19.
- [37] Hossain E, Islam K, Yeasmin F, Karim MR, Rahman M, Agarwal S, et al. Elevated levels of plasma Big endothelin-1 and its relation to hypertension and skin lesions in individuals exposed to arsenic. Toxicol Appl Pharmacol 2012;259(2):187–94.
- [38] Mahata J, Basu A, Ghoshal S, Sarkar JN, Roy AK, Poddar G, et al. Chromosomal aberrations and sister chromatid exchanges in individuals exposed to arsenic through drinking water in West Bengal, India. Mutat Res 2003;534(1-2):133-43.
- [39] Paul S, Banerjee N, Chatterjee A, Sau TJ, Das JK, Mishra PK, et al. Arsenic-induced promoter hypomethylation and over-expression of ERCC2 reduces DNA repair capacity in humans by non-disjunction of the ERCC2-Cdk7 complex. Metallomics 2014;6:864–73.

- [40] Abernathy CO, Liu YP, Longfellow D, Aposhian HV, Beck B, Fowler B, et al. Arsenic: health effects, mechanisms of actions, and research issues. Environ Health Perspect 1999;107:593–7.
- [41] Li C, Srivastava RK, Elmets CA, Afaq F, Athar M. Arsenic-induced cutaneous hyperplastic lesions are associated with the dysregulation of Yap, a Hippo signaling-related protein. Biochem Biophys Res Commun 2013;438(4):607–12.
- [42] Anwar-Mohamed A, Elshenawy OH, Soshilov AA, Denison MS, Chris Le X, Klotz LO, et al. Methylated pentavalent arsenic metabolites are bifunctional inducers, as they induce cytochrome P450 1A1 and NAD(P)H:quinone oxidoreductase through AhR- and Nrf2-dependent mechanisms. Free Radic Biol Med 2014;67:171–87.
- [43] Lau A, Zheng Y, Tao S, Wang H, Whitman SA, White E, et al. Arsenic inhibits autophagic flux, activating the Nrf2-Keap1 pathway in a p62-dependent manner. Mol Cell Biol 2013;33(12):2436–46.
- [44] Lau A, Whitman SA, Jaramillo MC, Zhang DD. Arsenic-mediated activation of the Nrf2-Keap1 antioxidant pathway. J Biochem Mol Toxicol 2013;27(2):99–105.
- [45] Zhan L, Zhang H, Zhang Q, Woods CG, Chen Y, Xue P, et al. Regulatory role of KEAP1 and NRF2 in PPARγ expression and chemoresistance in human non-small-cell lung carcinoma cells. Free Radic Biol Med 2012;53(4):758–68.
- [46] Gilmore TD. Introduction to NF-κB: players, pathways, perspectives. Oncogene 2006;25(51):6680-4.
- [47] Felix K, Manna SK, Wise K, Barr J, Ramesh GT. Low levels of arsenite activates nuclear factor-kappaB and activator protein-1 in immortalized mesencephalic cells. J Biochem Mol Toxicol 2005;19:67–77.
- [48] Hu Y, Jin X, Snow ET. Effect of arsenic on transcription factor AP-1 and NF-kappaB DNA binding activity and related gene expression. Toxicol Lett 2002;133:33–45.
- [49] Wijeweera JB, Gandolfi AJ, Parrish A, Lantz RC. Sodium arsenite enhances AP-1 and NFkappaB DNA binding and induces stress protein expression in precision-cut rat lung slices. Toxicol Sci 2001;61:283–94.
- [50] Trouba KJ, Wauson EM, Vorce RL. Sodium arsenite-induced dysregulation of proteins involved in proliferative signaling. Toxicol Appl Pharmacol 2000;164(2):161–70.
- [51] Sharma M, Sharma S, Arora M, Kaul D. Regulation of cellular Cyclin D1 gene by arsenic is mediated through miR-2909. Gene 2013;522(1):60–4.
- [52] Sun J, Yu M, Lu Y, Thakur C, Chen B, Qiu P, et al. Carcinogenic metalloid arsenic induces expression of mdig oncogene through JNK and STAT3 activation. Cancer Lett 2014;346(2):257–63.
- [53] Lu Y, Chang Q, Zhang Y, Beezhold K, Rojanasakul Y, Zhao H, et al. Lung cancer-associated JmjC domain protein mdig suppresses formation of tri-methyl lysine 9 of histone H3. Cell Cycle 2009;8(13):2101–9.
- [54] Wang Z, Yang J, Fisher T, Xiao H, Jiang Y, Yang C. Akt activation is responsible for enhanced migratory and invasive behavior of arsenic-transformed human bronchial epithelial cells. Environ Health Perspect 2012;120(1):92–7.
- [55] Wang Z, Humphries B, Xiao H, Jiang Y, Yang C. Epithelial to mesenchymal transition in arsenic-transformed cells promotes angiogenesis through activating β-catenin-vascular endothelial growth factor pathway. Toxicol Appl Pharmacol 2013;271(1):20–9.
- [56] Athar M, Tang X, Lee JL, Kopelovich L, Kim AL. Hedgehog signalling in skin development and cancer. Exp Dermatol 2006;15(9):667–77.
- [57] Fei DL, Li H, Kozul CD, Black KE, Singh S, Gosse JA, et al. Activation of Hedgehog signaling by the environmental toxicant arsenic may contribute to the etiology of arsenic-induced tumors. Cancer Res 2010;70(5):1981–8.
- [58] Kim J, Lee JJ, Kim J, Gardner D, Beachy PA. Arsenic antagonizes the Hedgehog pathway by preventing ciliary accumulation and reducing stability of the Gli2 transcriptional effector. Proc Natl Acad Sci USA 2010;107(30):13432-7.
- [59] Beauchamp EM, Ringer L, Bulut G, Sajwan KP, Hall MD, Lee YC, et al. Arsenic trioxide inhibits human cancer cell growth and tumor development in mice by blocking Hedgehog/GLI pathway. J Clin Invest 2011;121(1):148–60.

- [60] Srivastava RK, Li C, Chaudhary SC, Ballestas ME, Elmets CA, Robbins DJ, et al. Unfolded protein response (UPR) signaling regulates arsenic trioxide-mediated macrophage innate immune function disruption. Toxicol Appl Pharmacol 2013;272(3):879–87.
- [61] Grivennikov SI, Greten FR, Karin M. Immunity, inflammation, and cancer. Cell 2010;140:883–99.
- [62] Walter P, Ron D. The unfolded protein response: from stress pathway to homeostatic regulation. Science 2011;334(6059):1081-6.
- [63] Wheeler DL, Iida M, Dunn EF. The role of Src in solid tumors. Oncologist 2009;14(7):667–78.
- [64] Pion M, Stalder R, Correa R, Mangeat B, Towers GJ, Piguet V. Identification of an arsenic-sensitive block to primate lentiviral infection of human dendritic cells. J Virol 2007;81:12086–90.
- [65] Liao WT, Yu CL, Lan CC, Lee CH, Chang CH, Chang LW, et al. Differential effects of arsenic on cutaneous and systemic immunity: focusing on CD4+ cell apoptosis in patients with arsenic-induced Bowen's disease. Carcinogenesis 2009;30(6):1064–72.
- [66] Ameyar M, Wisniewska M, Weitzman JB. A role for AP-1 in apoptosis: the case for and against. Biochimie 2003;85(8):747–52.
- [67] Son SB, Song HJ, Son SW. Successful treatment of palmoplantar arsenical keratosis with a combination of keratolytics and low-dose acitretin. Clin Exp Dermatol 2008;33(2):202.
- [68] Yerebkan O, Ermis O, Yilmaz E, Basaran E. Treatment of arsenical keratosis and Bowen's disease with acitretin. Int J Dermatol 2002;41(2):84.
- [69] Boonchai W. Treatment of precancerous and cancerous lesions of chronic arsenicism with 5% imiquimod cream. Arch Dermatol 2006;142(4):531.

12

Arsenic-Induced Liver Injury

Amal Santra

CENTRE FOR LIVER RESEARCH, SCHOOL OF DIGESTIVE & LIVER DISEASES, INSTITUTE OF POST GRADUATE MEDICAL EDUCATION & RESEARCH, KOLKATA, WEST BENGAL, INDIA

CHAPTER OUTLINE

12.1 Introduction	
12.2 Source of Exposure	
12.3 Biotransformation and Elimination of Arsenic	
12.4 History of Arsenic-Related Liver Diseases	
12.5 Hepatoportal Sclerosis	
12.6 Arsenic-Related Hepatic Fibrosis and Cirrhosis	320
12.7 Epidemiological Study to Assess Arsenic-Related Liver Dysfunction	321
12.8 Animal Studies for Understanding the Pathogenesis	322
12.9 Arsenic and Liver Cancer	329
References	330

12.1 Introduction

Arsenic is a toxic metalloid that constitutes 0.0001% of Earth's crust. Evidence of therapeutic use of this metalloid is available from the ancient Greeks and Romans. Historically, arsenic was also used as human poison. It is believed that Napoleon died due to arsenic poisoning. Human exposure to arsenic is inevitable. Exposure can occur via three principal routes, that is, through the inhalation of air, through the ingestion of arsenic-contaminated food and water, and via dermal absorption. Worldwide, the degree of non-occupational exposure to arsenic varies greatly, being dependent on local geochemistry and the level and proximity of anthropogenic activity.

Exposure to inorganic arsenic through drinking water is a major public health problem in both developing and developed countries [1,2]. In the United States of America, over 13 million people live in areas where arsenic concentration in the drinking water exceeds the US Environmental Protection Agency's safe cut-off value of $10 \mu g/L$ [3]. But the greatest catastrophe in the history of arsenic poisoning of a population is attributed to the densely populated deltas of the great rivers flowing through South and South-East Asia, where more than 100 million people are being exposed to unsafe levels of arsenic by drinking contaminated groundwater [2]. In parts of West Bengal, India, and Bangladesh, arsenic levels in drinking water have been reported to be as high as 1.5 to 3.4 mg/L [4–7]. Over the last three to four decades, consumption of arsenic-contaminated drinking water in public supply systems has posed a serious health hazard among inhabitants of at least nine districts in West Bengal and Bangladesh [8–11].

Arsenic causes a spectrum of adverse responses in many species. Acute exposure to inorganic arsenic in humans results in cardiac failure, peripheral neuropathy, anemia, leukopenia, and death, while chronic arsenic exposure can cause a range of cancers as well as liver injury, neuropathy, cardiovascular lesions, and other diseases. The toxicity of inorganic arsenic usually occurs after a considerable latent period. It is responsible for the genesis of a wide range of health problems involving most of the systems of the human body and is also implicated as an etiological agent of cancers of the different organs of the body. Emerging evidence indicates that ingested arsenic favors the development of hepatic disorders. The magnitude of the public health problem posed by chronic arsenic toxicity is fairly large with newer affected areas being identified in West Bengal and some other states in India and Bangladesh day by day, and there is no treatment for established chronic arsenic toxicity with proven efficacy. This makes it essential to improve our understanding of the pathophysiology of arsenic toxicity, which is difficult to discern in humans and has not been well studied in experimental animals. The present chapter deals with understanding arsenic-induced liver injury and its possible pathways, in the context of current available knowledge.

12.2 Source of Exposure

Environmental arsenic originates from both geochemical and anthropogenic activities. The anthropogenic sources of arsenic are soil, water, mining, metallurgical activities, and use of herbicides and pesticides. The sources of arsenic exposure vary in different parts of the world, i.e., from burning of arsenic-rich coal in China and mining activities particularly in Malaysia and Japan, to the ingestion of arsenic-contaminated drinking water in many countries around the world. Due to geological factors, some coals contain high concentrations of arsenic [12]. People of Guizhou province in southwest China are exposed to arsenic due to burning coals that have a high concentration of arsenic. In some areas of Guizhou province, coals contain 100–9000 ppm arsenic [12–14]. In these areas, chronic arsenic intoxication occurs due to burning of high arsenic containing coal in unventilated stoves, which is a routine practice in the drying of various foods. Arsenic, which is a constituent of some ores, is a common occupational exposure and frequently results in widespread environmental contamination. Other sources of contamination are the manufacture and use of arsenical fertilizers, pesticides, wood preservatives, and insecticides. Non-occupational exposure occurs primarily through the ingestion of food and water, with the inhalation pathway playing only a minor role. Food is more commonly the main contributor to total intake but in areas where drinking waters contain relatively high levels of arsenic, such water may be the most important source of arsenic intake. In Latin America, South Asia, Taiwan, and Japan, arsenic levels in drinking water have been reported as high as 1.5-3.4 mg/L [4,6,7]. Groundwater arsenic contamination in



FIGURE 12-1 Pathways of inorganic arsenic metabolism within hepatocytes of the liver.

Bangladesh and the West Bengal Delta of India has received the greatest international attention due to the large number of people potentially exposed and the high prevalence of arsenicinduced diseases. Residential areas of Behala, Kolkata, also reported groundwater arsenic contamination due to industrial pollution [15].

12.3 Biotransformation and Elimination of Arsenic

Following absorption, arsenic is distributed widely in the body, with high concentrations appearing in the skin, liver, kidney, lung, and spleen. Liver is the primary organ for biotransformation of toxic inorganic arsenic into less toxic metabolites for elimination from the body. In most mammals, inorganic arsenic is metabolized to methylarsonic acid (MMA) and dimethylarsenic acid (DMA) by stepwise transfer of the methyl group from S-adenosylmethionine. It is known that after ingestion, the pentavalent form of inorganic arsenic (As^{V+}) is reduced to trivalent inorganic arsenic (AS^{III+}) and subsequently methylated in the liver to the less toxic readily excreted metabolites MMA and DMA (Figure 12–1). The methylated metabolites have a shorter half-life *in vivo* than inorganic arsenic and are less reactive with tissue components, less toxic, and more rapidly eliminated from the body through urine than inorganic arsenic [16,17]. In general, little is known about factors influencing arsenic toxicity and metabolism in humans. Arsenic methylation patterning is influenced by age as well as exposure level. Urinary DMA

content in urine is lower in children compared to adults, indicating that children retain more arsenic in their tissues and they are more sensitive to arsenic exposure. The toxicity of arsenicals is highly dependent on the chemical forms. Trivalent inorganic arsenic (As^{III}) is the main form of arsenic interacting with tissues [16,17]. It is observed that a lower capacity to methylate arsenic is associated with higher tissue concentrations [18], and possibly also higher risk for toxic effects. Thus, the methylation of arsenic may be considered a detoxification mechanism.

The mechanism involved in arsenic metabolism is not well understood; it has been shown that glutathione (GSH) plays an important role in both reduction and methylation of arsenate [19]. Experimental studies have shown that GSH elevation is a natural reaction to arsenic insult, presumably as a protective mechanism [20], and that GSH depletion prior to treatment with arsenic leads to interference with arsenic metabolism, including inhibition of methylation in the liver and decrease in the elimination rate of arsenic metabolites [21]. Since GSH is involved in arsenic metabolism, it is possible that the individual variation in methylation capacity relates to glutathione *S*-transferase M1 and glutathione *S*-transferase T1 genotypes. Reduced levels of glutathione *S*-transferase in those with null genotypes may increase susceptibility by decreasing methylation efficiency, leading to exposure of the liver to more toxic forms of arsenic. Arsenic has been shown to produce toxic effects common to oxidative stress [22–24] and these effects are thought to be mediated by the binding of arsenic metabolites (e.g., As^{III}) to critical sulfhydryl groups within proteins [25]. Glutathione *S*-transferase T1 deficiency could therefore directly exacerbate the oxidant effects of arsenic exposure.

Inorganic arsenic and its metabolites are rapidly cleared from hepatocytes through bile and ultimately eliminated from the body through feces. The biliary transport of arsenic is dependent on the arsenic–GSH complex formation that transports it out of the hepatocytes into bile. High expression of multidrug resistance-associated protein 2 (MRP2) is observed in the liver during arsenic exposure. This MRP2 actively involves the biliary transport of arsenic [26]. Any defect of arsenic–GSH complex formation, lack of intracellular GSH store, or defect of MRP2 protein formation leads to accumulation of arsenic within the hepatocytes.

12.4 History of Arsenic-Related Liver Diseases

The liver is vulnerable to prolonged exposure to small amounts of arsenic; however, the precise degree of sensitivity to arsenic in any one person cannot be predicted. Chronic arsenic exposure has been typically associated with dermatological disorders. The relationship between arsenic exposure and liver disease is not clear. However, sporadic reports of liver disorders are available in the literature. One of the earliest descriptions of arsenic-induced liver disease dates back to the late 18th century, when Bang [27] first reported development of ascites due to prolonged use of liquid arsenic in therapeutic doses. Subsequently, Sir Jonathan Hutchinson [28] described a young man with psoriasis who developed ascites after treatment with Fowler's solution (arsenic trioxide). Early investigators also described an outbreak of liver disease in beer drinkers in the north of England, where arsenopyrite had been used to produce beer from starch [29,30]. The disease was directly related to the amount of arsenic in the beer. In 1921 Stockman [31] reported the development of ascites and liver cirrhosis in a patient who had

consumed arsenic in a bromide mixture. In 1936 Cannon [32] described a young woman with an enlarged liver who had ingested fruits and vegetables sprayed with an arsenical pesticide. In 1945 Maccallum described liver damage due to organic arsenicals used therapeutically [33]. Thus, the hepatotoxic effects of arsenic when used as a therapeutic agent have long been recognized. Scanty reports of liver diseases related to chronic exposure to arsenic-contaminated water are available in the literature. Worldwide, environmental exposures remain a significant problem. Chronic liver diseases, which can be ascribed to arsenic toxicity, include steatosis, cirrhosis, hepatoportal sclerosis, angiosarcoma, and perhaps hepatocellular carcinoma [34].

12.5 Hepatoportal Sclerosis

Chronic arsenic exposure may be associated with hepatoportal sclerosis (also known as noncirrhotic portal hypertension), which is characterized by portal hypertension without evidence of liver cirrhosis. Portal hypertension is a clinical syndrome defined by a portal venous pressure gradient between the portal vein and inferior vena cava exceeding 5 mmHg [35]. It is a disorder of unknown etiology. The disease has been reported from different parts of the world and predominantly from the developing countries. According to the consensus statement of the Asia Pacific Association for the Study of the Liver (APASL) on NCPF, the disease accounts for approximately 10-30% of all cases of variceal bleed in several parts of the world including India [36]. Clinically, this disease is characterized by features of portal hypertension, and moderate to massive splenomegaly, with or without hypersplenism with preserved liver functions and patent hepatic and portal veins [37]. Histologically, it is characterized by thickening and sclerosis of the wall of large portal vein branches. Increased perivascular fibrosis is seen in the portal tracts, which may show sclerosis of portal vein branches. The portal vein lumen is reduced and organized thrombi with recanalization may be seen. Several reports have documented the involvement of chronic arsenic exposure in the development of this disease. Zeegen and coworkers [38] described a series of 44 patients with portal hypertension, four of whom had a history of arsenic exposure. Viallet et al. [39] described the case of a patient who had consumed arsenic for the treatment of psoriasis for 12 years before development of portal hypertension in the absence of liver cirrhosis. Morris et al. [40] described two cases of non-cirrhotic portal hypertension due to treatment of Fowler's solution for 3 and 22 years respectively. Wedged hepatic vein pressures were normal, while intrasplenic pressure was high in these two patients. This indicated obstruction to portal flow in the portal tracts. Morris et al. [40] also suggested that arsenic caused damage of the intrahepatic portal veins. Huet et al. [41] reported the development of non-cirrhotic portal hypertension after intake of arsenicals. In addition, a case of massive variceal bleeding secondary to presinusoidal portal hypertension due to arsenic poisoning was reported from North America [42]. Nevens et al. [43] also described non-cirrhotic portal hypertension development in eight patients who had been treated with Fowler's solution for psoriasis. Total arsenic consumption was varied from 4 to 16g, with an interval between treatment and onset of symptoms of 2 to 16 years. From India, Dutta et al. [44] first reported nine patients with hepatoportal sclerosis who had consumed high levels of arsenic from contaminated drinking water, adulterated opium, and indigenous medicine. Estimated hepatic blood flow of those patients with hepatoportal sclerosis due to consumption of high amounts of arsenic was within the normal range and there was no correlation between hepatic arsenic and the hemodynamic findings. Hepatoportal sclerosis was also reported from patients consuming high levels of arseniccontaminated drinking water [10,45]. Santra et al. reported the presence of esophageal varices only in few cases in their series [45]. Thus, the majority of the patients had no definite evidence of portal hypertension though portal fibrosis was observed in liver biopsy in most of the cases. Hepatoportal sclerosis is often a result of damage to the local vasculature [43,46].

12.6 Arsenic-Related Hepatic Fibrosis and Cirrhosis

The hepatotoxic action of arsenic, when used as a therapeutic agent, has long been recognized. Development of ascites has been reported in patients taking liquor arsenic in therapeutic doses for prolonged periods [27-28]. Cases of cirrhosis of the liver resulting from continued use of Fowler's solution containing potassium arsenite have also been described [47]. Early investigators also described a few cases of chronic liver disease due to drinking arsenic-contaminated beer [30,31]. Baldridge [48] described cirrhosis in patients being treated with arsenic for syphilis. Luchtrath [49] reported that cirrhosis among German vintners had decreased since the banning of arsenical pesticides. However, data on liver involvement following chronic exposure to arsenic-contaminated water are scanty. Guha Mazumder et al. highlighted that non-cirrhotic portal fibrosis occurred in chronic arsenicosis due to drinking arsenic-contaminated water [10,50]. Subsequently, Santra et al. reported the largest series showing significant involvement of the liver in the form of non-cirrhotic portal fibrosis among people drinking arsenic-contaminated water [45]. In hospitalized arsenicosis patients who had been exposed to arsenic-contaminated water for up to 15 years and had hepatomegaly, from West Bengal, India, liver fibrosis is very common (Figure 12-2). A few of these patients



FIGURE 12–2 (A) Representative liver histology of a patient who had hepatomegaly due to drinking high arseniccontaminated water for a long period. Evidence of expansion of the portal zone is also observed in the section. (H & E staining; original magnification ×20.) (B) Grades of liver fibrosis in patients suffering from chronic arsenicosis due to drinking high levels of arsenic-contaminated water. Most of the patients had mild fibrosis (grades I and II) and few had moderate to severe fibrosis (grades III and IV). The data are based on 69 hospitalized arsenicosis patients [45].

had liver cirrhosis. The portal fibrosis was characterized by expansion of portal zones of varying degrees (Figure 12–2). Fibrous extension from portal tracts into liver lobules producing septa was observed in some cases. The fibrosis in the liver was mild in most cases, moderate to severe fibrosis being observed in few cases (Figure 12–2). At some regions the expanded portal zone contained a leash of vessels replacing the portal vein branches. The features of periportal fibrosis and multiple vascular channels in expanded portal zones reported by Santra et al. [45] are similar to those observed in non-cirrhotic portal fibrosis/idiopathic portal hypertension. These multiple portal venous vascular channels are most likely the end results of recanalization of thrombosed intrahepatic portal veins and have been observed in patients who develop non-cirrhotic portal hypertension [51]. This neovascularization appears to be an attempt to reestablish portal venous blood flow in previously injured or obstructed vessels.

Chronic arsenic exposure has been associated with peripheral vascular diseases. One such disorder, known as "blackfoot disease," seems to result from an early destruction of vascular endothelial cells [52]. In an editorial review, Boyer [53] raised an important issue—whether non-cirrhotic portal hypertension results from primary injury to the endothelial cells within the portal vein radical. Long-term follow-up of patients with arsenic-induced hepatic fibrosis is helpful to define the relationship between chronic arsenic toxicity and development of non-cirrhotic portal hypertension.

Most of the authors described liver profiles to be normal or near normal, but Santra et al. described elevated levels of serum ALT and AST in a significant number of cases with hepatomegaly [45], and arsenic in the liver tissues was detected in 23 of 29 samples tested. The maximum arsenic content in the liver tissue was found to be 6mg/kg. However, the arsenic content of the liver tissue did not correlate with the degree of hepatic fibrosis or with the arsenic content of the water consumed. While the nature of the toxicological mechanisms involved in hepatic damage due to inorganic arsenic is not clear, some reports have demonstrated that arsenic can accumulate in the liver [10,44,45]. Figueroa and colleagues examined mummies hundreds of years old that were found in region II of Chile, an area that has had arsenic levels in drinking water. Kidney, liver, nail, and lung tissues had some of the highest concentrations of total arsenic followed by skin, intestine, hair, and ribs, in that order [54]. High arsenic content in the liver was also demonstrated in patients suffering from non-cirrhotic portal fibrosis associated with chronic arsenic toxicity [10,44,45].

12.7 Epidemiological Study to Assess Arsenic-Related Liver Dysfunction

A few epidemiological studies related to liver dysfunctions due to chronic arsenic exposure are available. High prevalence of hepatomegaly was reported among inhabitants of six villages of three arsenic-affected districts in West Bengal [55]. Subsequently, Guha Mazumder et al. [10] also reported high prevalence of hepatomegaly among subjects in a hamlet drinking high arsenic-contaminated water (>50 μ g/L). Ma et al. reported occurrence of hepatomegaly and abnormal liver functions in a study population of 1447 cases of chronic arsenicosis due to drinking arsenic-contaminated water in Inner Mongolia, People's Republic of China [56].

Reports of liver dysfunction in chronic arsenicosis are also available from Bangladesh [57]. However, most of the reports described above have some limitations in selecting the subjects from the viewpoint of an epidemiological study.

Population-based studies of liver injuries related to chronic arsenic exposure are necessary for accurate information on the risk of liver diseases. One such example is a populationbased study reported from Kolkata. This study was designed to assess the prevalence of liver involvement and various health effects associated with chronic arsenic exposure through drinking high arsenic-contaminated water. This cross-sectional epidemiological study was conducted in one of the most affected districts of West Bengal [6]. This district was found to be most suitable for the epidemiological study because of heterogeneity in arsenic exposure. The drinking-water arsenic content in this district ranged from non-detectable to 3400 µg/L. In this study two particular areas of the district were selected. The first one had had high levels of arsenic detected in a number of shallow tubewells, but not all. The other area selected was from another part of the district where people used shallow tubewell water that did not contain increased levels of arsenic. A total of 7683 participants were included in this study. Convenience sampling was adopted in both high and low exposure zones. Enlargement of the liver was observed in 2.99% of participants drinking water containing arsenic at <0.05 mg/L (low exposure group), and while in 10.21% of participants who were drinking water containing arsenic at >0.05 mg/L (high exposure group). Hepatomegaly in this population was found to have a linear relationship proportional to increasing exposure of arsenic in drinking water in both the sexes (p < 0.001). The prevalence was greater among males than females. The prevalence of hepatomegaly among females was 3.46% in the lower exposure category ($<50 \mu g/L$) while its incidence was 17.83% in the highest exposure category ($\geq 800 \,\mu g/L$). A stronger trend appeared also in males; the prevalence increased from 2.62% in the lowest exposure category to 28.81% in the highest exposure category.

This epidemiological study is thought to be the first large-scale population study assessing the prevalence of hepatic enlargement due to chronic arsenic ingestion in an arsenic-affected area in a structured population survey. A major advantage of this study was the participation of subjects from all the different age groups of the community, and thus the data generated from this study truly reflect the actual prevalence of hepatic enlargement due to arsenic exposure.

12.8 Animal Studies for Understanding the Pathogenesis

The pathogenesis of liver injury in arsenicosis is not clear; however, some experimental studies on animals are available. Based on various experimental studies, it is now clear that arsenic induces liver injury from an early phase of exposure followed by a phase of adaptation, and if the exposure is continued long term, there is evidence of chronic liver injury. Evidence of oxidative stress in the liver following arsenic exposure is available. Cellular events at an early phase of high arsenic exposure that initiates liver injury in mice have been described [58]. The pathological lesions in liver include fatty degeneration and to a lesser extent necrosis and



FIGURE 12–3 Correlation between oxidative stress with arsenic content in liver and urinary excretion. (A) Data showing a negative correlation between hepatic GSH level and arsenic deposition in the liver in control and arsenic-exposed mice (r = 0.851; p < 0.001). (B) Decrease in urinary excretion of arsenic with increase in GSSG levels (r = 0.863; p < 0.001).

inflammation due to high-dose arsenic exposure. Hepatic steatosis is macrovesicular in nature and apoptosis is the predominant mode of hepatocyte death due to arsenic exposure. Further, apoptosis of the hepatocytes in an early phase of arsenic exposure is dose dependent. Urinary arsenic is generally used as a marker of current arsenic exposure. Animal studies showed increased deposition of arsenic in the liver despite an increase in urinary arsenic excretion, thus indicating the inability of the liver to eliminate the increased arsenic that the animals were exposed to. The deposition of arsenic in the liver was found to be related to the increasing doses of arsenic exposure. Deposition of arsenic in the liver inversely correlated with depletion of hepatic GSH content (Figure 12–3). Moreover, the urinary arsenic elimination was also found to be inversely correlated with the accumulation of GSSG in liver, a marker of oxidative stress, as a result of arsenic exposure (Figure 12–3).

Many endogenous or xenobiotic lipophilic substances are eliminated from the cells by the sequence of oxidation and conjugation to an anionic group (glutathione, glucuronate, or sulfate) and transport across the plasma membrane. The latter step is mediated by integral membrane glycoprotein belonging to the superfamily of ATP-binding cassette (ABC) transporters, most commonly known as the multidrug resistance-associated proteins (MRPs). MRP-2, a subfamily of MRPs, also contributes to the control of the intracellular glutathione disulfide (GSSG) levels. Although this protein has a low affinity towards GSSG, it can play an essential role in response to oxidative stress. The elimination of arsenic is one of the important criteria of adaptation, and this elimination depends on the function of MRP-2. Ghatak et al. had studied the role of MRP-2 protein as it has been well documented that arsenite appears to be a stronger inducer of MRP-2. Interestingly, Ghatak et al. [59] have found that the expression of MRP-2 is directly related to the level of cellular GSH.

One potential central mechanism of hepatic injury at the early phase of arsenic exposure is the development of oxidative stress due to generation of reactive oxygen species (ROS) from



FIGURE 12–4 Effect of arsenic exposure on hepatic oxidative stress. Mice were treated with $6 \mu g/g$ body weight p.o. of arsenic and sacrificed at different time intervals. The ratio of GSSG:GSH was plotted graphically using varied hours of arsenic exposure. Results were expressed as mean \pm SD of 10 mice per group (*p < 0.05; **p < 0.001). The red trace indicates the arsenic-treated groups while the blue trace indicates control.

mitochondrial as well as from non-mitochondrial pathways [58,60]. The ratio of GSSG to GSH is often regarded as the marker of oxidative stress. Animal studies indicated that from an early phase of arsenic exposure, there is evidence of increased GSSG to GSH ratio in the liver, suggesting development of oxidative stress in the liver of arsenic-exposed animals (Figure 12-4). Hepatocytes are vulnerable to increased oxidative stress due to arsenic exposure. Generation of oxidative stress in the liver and its impact on cellular integrity and function are critical early events in arsenic-induced liver injury. It has been observed that high levels of reduced glutathione (GSH) are associated with cellular resistance to arsenic [61-63] and decreasing intracellular GSH concentration causes increased sensitivity to arsenic toxicity [62]. Thus, the hepatotoxic effects of arsenic are attributed to the development of oxidative stress and the consequent depletion of GSH from the liver. GSH has been suggested to be important for arsenic methylation as well as transport [64,65] and thus helps in the removal of arsenic from the body. Depletion of hepatic GSH facilitates accumulation of arsenic in the liver and thus causes oxidative stress particularly at higher dose. However, at comparatively low dose of arsenic exposure, evidence of oxidative stress, especially in the mitochondrial fraction, is not significantly increased because at lower dose there is an adaptive response to re-establish the normal cellular redox status [58].

Arsenic exposure in animal models showed increased oxidative stress in the liver mitochondria, which causes mitochondrial functional alteration. A potentially deleterious effect of ROS production in mitochondria is facilitation of Ca²⁺-dependent mitochondrial permeability transition (MPT), which plays a key role in certain modes of cell death. Arsenic causes progressive induction of MPT pore opening of the liver mitochondria in an early phase of exposure (Figure 12–5) and decreased fatty acid oxidation leading to accumulation of triglycerides within the



FIGURE 12–5 Effect of arsenic on the functional activity of mitochondria. Arsenic-treated mice were scarified and the mitochondria were isolated. 1 mg/mL of mitochondria suspension was prepared and loaded with Ca²⁺ (20 nmol/mg protein) and the mitochondrial swelling was monitored by the O.D. at 540 nm.

hepatocytes. At present, although the detailed mechanisms regulating MPT are still largely elusive, it is known that the oxidative stress markedly sensitizes mitochondria toward MPT induction [66,67]. It is well known that onset of MPT always leads to a loss of mitochondrial membrane potential (MMP), though not all MMP losses are caused by MPT [66]. The onset of MPT results in alterations in the equilibration of ions between the mitochondrial matrix and the inner membrane space, leading to the loss of MMP [68]. Mitochondrial GSH plays an essential role in maintaining healthy mitochondria, and its depletion may be a key event in sensitization of hepatocytes to oxidant injury. Santra et al. [58] reported altered MPT associated with translocation of cytochrome c in the cytosol and activation of caspase 3 and caspase 9 activities in the early event of arsenic exposure in mice. Translocation of cytochrome c into cytosol is a primary event that leads to the formation of apoptosomes and activation of the caspase cascade. Higher doses of arsenic cause an increased DNA fragmentation, suggesting increased apoptosis. Moreover, a significant increase in caspase 3 and caspase 9 activities in arsenic-exposed mice suggests that apoptosis is caspase dependent [58]. Caspase, which is cysteine protease, cleaves different intracellular target molecules, resulting in cell shrinkage, chromatic condensation, and DNA fragmentation [69]. Caspase 3 is activated via either an extrinsic pathway (death receptor mediated) or an intrinsic pathway (mitochondria-dependent pathway [70]. The activation of caspase 9 in arsenic-induced mice strongly suggests the involvement of a mitochondrial pathway [58].

Treatment with an antioxidant such as N-acetyl cysteine (NAC) in animal models could replete cellular stores of the tripeptide GSH and is an effective intervention against oxidative stress developed due to arsenic exposure. One explanation of the observed hepatoprotection provided by NAC could be that the liver of NAC pretreated animals effectively detoxifies or removes arsenic via a GSH-dependent pathway. This is probably due to an enhanced ability to maintain GSH homeostasis during exposure to toxic electrophiles generated by arsenic as well as its rapid elimination/excretion from the body.



FIGURE 12–6 Liver histology of arsenic-exposed mice. (A) Hematoxylin and eosin staining of arsenic-exposed mice for 12 months (magnification: \times 40). The infiltrations of inflammatory cells are shown in the section. (B) Histology with Sirius red collagen staining of liver section of arsenic-exposed mice for 12 months (magnification: \times 20). There is evidence of marked portal fibrosis and pericellular fibrosis accumulated near the portal tract.

Evidence for adaptation against oxidative assault in the liver of experimental models due to short-term arsenic exposure is available in the literature [71]. Arsenic exposure for short periods initiates activation of a liver antioxidant defense system by increasing hepatic GSH content to counteract and protect the hepatocytes against free radical-mediated oxidative damage. This antioxidant defense system becomes less effective subsequently, due to its overreactivity and this might initiate hepatic injury.

The histological abnormalities observed in the liver of the experiment model in long-term arsenic exposure include hepatocellular degeneration with focal mononuclear cell collection, fatty degeneration (macrovesicular steatosis), Kupffer cell proliferation, and hepatic fibrosis (Figure 12–6). The hepatic fibrosis is not associated with nodular regeneration.

Exposure to arsenic-contaminated drinking water for a prolonged period is associated with pathological changes in the liver as evidenced by clinical and experimental studies. Providing arsenic-contaminated water to mice for 15 months and sacrificing those mice at different intervals revealed hepatocellular steatosis associated with significant reduction in hepatic antioxidant defense system at 9 months and evidence of hepatic fibrosis associated with further impairment of the antioxidant defense system at 12 months [72]. Oxidative stress is thought to underlie several chronic liver diseases that are associated with fibrosis. Oxidative stress is a complex process that can result in the peroxidative damage of the major cellular components including amino acids, proteins, lipids, carbohydrate, and nucleic acids [73]. Damage of the membrane lipids and associated alteration of the bulk properties of the membranes are considered to be the primary basis of chemical-induced hepatocellular injury. The increased lipid peroxidation and plasma membrane damage as evidenced by progressive reduction of Na/K ATPase activity in mice drinking arsenic-contaminated water parallel GSH depletion of the liver. Peroxidation of the lipids in the cell membrane, damage to DNA and macromolecules, recruitment of inflammatory cells, and activation of hepatic stellate cells can occur as a result of oxidative stress in sequence or in parallel and may contribute to the

progression of liver injury that may proceed the development of hepatic fibrosis. It is known that the lipid peroxidation products in the liver parenchymal cells are needed to initiate the fibrogenic process. Increased hepatic fibrogenesis with minimal hepatocellular injury due to high arsenic exposure in animal models was also observed by Sarin et al. [74]. In this model, hepatocellular necrosis and inflammation were either insignificant or mild. There was neither any feature of non-cirrhotic portal hypertension nor significant splenomegaly observed in this animal model. Das et al. showed elevated tissue necrosis factor- α (TNF- α) levels in the liver of mice exposed to arsenic for prolonged periods [75]. TNF- α plays an important role in activation of stellate cells [76], the predominant cells involved in the production of type I collagen.

The mechanism underlying the arsenic-induced oxidative stress response is not clearly understood. One potential candidate for the generation of ROS in chronic arsenic exposure may be NADPH oxidase (NOX), which is a potential non-mitochondrial source of ROS. Progressive increase of 4-HNE adduct, a marker of oxidative stress in the tissue, in the liver associated with increased NOX activity parallel to the duration of arsenic exposure was reported [77]. The role of NOX in arsenic-induced liver injury and subsequent development of fibrosis is suggested, since it is well established that ROS activate hepatic stellate cells (HSCs) and stimulate fibrogenesis [78]. Arsenic-induced oxidative stress in liver may up-regulate CD14 expression and thus increase the susceptibility of Kupffer cells to arsenic or its metabolites. It seems likely that the Kupffer cells, in turn, up-regulate TNF- α production, which plays an important role in the activation of HSCs, the predominant cells involved in the production of type I collagen [76,79,80]. Increased hepatic TNF- α production associated with up-regulation of CD-14 and TLR-4 mRNA transcript was reported after long-term arsenic exposure to mice [77]. However, the mechanism of Kupffer cell activation by arsenic remains unknown. Kupffer cell activation seems to be mediated by CD14 and TLR-4 [81].

In experimental studies, several lines of evidence suggest that arsenic activates HSCs as documented by immunohistochemical staining as well as mRNA levels for α -smooth muscle actin (α -SMA), a hallmark for stellate cell activation as demonstrated in the liver of mice exposed to arsenic for 1 year [77]. Further, the growth factors transforming growth factor- β (TGF- β) and platelet-derived growth factor receptor β (PDGF-R β) were reported to be increased in the liver tissues after arsenic exposure. Both proteins are important for the induction of HSC activation and proliferation and the development of fibrosis [82].

Evidence for progressive development of hepatic fibrosis after long-term exposure to high levels of arsenic comes from experimental works [72,77] as documented by increased accumulation of collagen in the liver associated with increased hepatic levels of tissue inhibitor of matrix metalloproteinases-1 (TIMP-1) mRNA, matrix metalloproteinase-2 (MMP-2), and matrix metalloproteinase-9 (MMP-9). Heightened expression of MMPs probably contributes to the pathogenesis of arsenic-induced liver injury, as several studies also report that MMP-2 and MMP-9 were up-regulated in the fibrotic liver [83]. The normal extracellular matrix (ECM) is essential for maintaining the homeostasis of all resident liver cells. Therefore, it is likely that degradation of the ECM by MMPs alters cell matrix as well as cell-cell interactions and enhances hepatocyte susceptibility to necrosis and/or apoptosis caused by prolonged arsenic exposure. The ECM also serves as a binding reservoir for several key cytokines such as TNF- α

and growth factors such as TGF- β and PDGF [84,85]. MMPs release soluble bioactive factors through ECM degradation and regulate macrophage chemoattractants and leukocyte infiltration during injury [86]. Collectively, these data strongly suggest that arsenic-mediated liver injury is associated with HSC activation and hepatic fibrogenesis.

In vitro experiments using co-culture models of HSCs with hepatocytes provided evidence of the involvement of oxidative stress in the hepatocytes due to arsenic treatment and were responsible for activation of HSCs that may transdifferentiate into myofibroblasts [77]. This model provided several lines of evidence that prove the role of oxidative stress in the hepatocytes in arsenic toxicity. First, the authors identified excessive ROS generation in both hepatocytes as well as HSCs. ROS are known to play a key role in HSC activation [87]. Second, they detected H_2O_2 and thiobarbituric acid reactive substance (TBARS) accumulation, derived from hepatocytes, in the incubating medium of HSCs. Third, they found are higher levels of α -SMA, PDGF-R β , and Pro-(α) collagen I expression in HSCs. H_2O_2 derived from hepatocytes may induce collagen transcription in HSCs [87]. Further, pretreatment of hepatocytes with the antioxidant NAC abrogates arsenic-induced generation of ROS in the cells, H_2O_2 and TBARS accumulation in the medium, and expression of α -SMA along with decreased expression of PDGF-R β and type I collagen. This study clearly supports a role of oxidative stress in the hepatocytes due to arsenic exposure in the activation of HSCs.

The results reported by Ghatak et al. from India differ from another study from the USA that used a comparatively low to moderate exposure to arsenic (250 ppb in drinking water) for a shorter duration (5 weeks) [88]. Such exposure caused increased sinusoidal endothelial cell (SEC) capillarization, vascularization of the peribiliary vascular plexus, and constriction of hepatic arterioles in mice. However, this lower level of arsenic exposure, equivalent to a human dose of $32 \mu g/day$ for 3.75 years, did not cause arsenic deposition in the liver or result in portal fibrosis, an important feature in liver histological lesion in chronic arsenic-induced hepatotoxicity of humans [10,40-45]. The rationale for the relatively high dose of arsenic used in the study of Ghatak et al. [77] comes from epidemiological studies of arsenicosis in this region of the Indian subcontinent [89]. These studies suggest that clinical manifestations of arsenicinduced liver injury develop only after a long latency with the intake of large volumes of arsenic-contaminated water, usually since childhood [10,44,45,89]. The dose used by Ghatak et al. [77] in their study was simulated to be equivalent to a human exposure of 6 mg of arsenic/day for more than 20 years, a time over which liver toxicity might be expected to develop. However, the precise relationship between duration of arsenic exposure and the onset of liver lesions is uncertain and remains to be clarified.

Non-cirrhotic portal hypertension associated with portal fibrosis has also been a major consequence of chronic arsenic toxicity found in human clinic-pathological studies from India and the West [10,41–45,89]. The pathophysiological mechanisms that contribute to angiogenesis and vascular remodeling in response to arsenic exposure are not very clear. Liver SECs are unique endothelial cells in respect of their architecture and function. In response to oxidizing agents [90], SECs differentiate into regular endothelium in a process termed capillarization. The hallmarks of capillarization are SEC defenestration, development of laminin-rich basement membrane, and junctional expression of platelet endothelial cell adhesion molecules (PECAMs). Capillarization precedes the development of many liver diseases, which include portal hypertension and liver fibrosis. Interestingly, low levels of arsenic exposure cause SEC injury that includes SEC defenestration, capillarization, and decreased liver clearance in experimental animal models [88]. Arsenic stimulates capillarization of SECs by disrupting the SEC signaling that maintains fenestrations and suppresses cell spreading, which may be the initial events in the pathogenic changes in the liver. Progressive SEC capillarization is mediated through activation of p47phox containing NADPH oxidase [91]. However, these animal models did not develop features of portal fibrosis, which is an important feature of arsenic-related hepatotoxicity in humans.

Activation of hepatic fibrogenesis is therefore a likely consequence of this arsenic-induced liver injury. Sustained arsenic-induced oxidative stress in the liver leads to increased apoptosis of hepatocytes and liberation of some mediators, which contribute to the activation of HSCs, the predominant fibrogenic cell type producing collagen type I in the liver [80,92]. HSCs also produce tissue inhibitor of matrix metalloproteinases (TIMP), which inhibits collagendegrading MMPs and shifts the balance between ECM synthesis and degradation towards fibrogenesis [79,93]. A high fat diet also enhances chronic arsenic-induced hepatic fibrogenesis [94].

12.9 Arsenic and Liver Cancer

Arsenic and arsenic-containing compounds are human carcinogens. Two types of liver cancer have been associated with chronic arsenic exposure: hepatocellular carcinoma (HCC) and hepatic angiosarcoma (HAS). HCC has been identified as a tumor type associated with arsenic exposure in humans [46,95-97]. HCC, non-alcoholic liver cirrhosis, and ascites are leading causes of mortality in the arsenic endemic area of Guizhou, China [97,98]. HAS is a malignant neoplasm of the liver constituting only 2% of all primary tumors in Western countries. A causal association between ingested inorganic arsenic and HAS is supported by a series of case reports of arsenic-poisoned winegrowers in Germany [99] and several studies in southwestern Taiwan where drinking-water has a naturally arsenic content [100]. HAS is often associated with exposure to vinyl chloride. Roth [101] reported three cases of HAS in a series of 27 autopsies among arsenic-poisoned German vintners. A study in Chile found that among a group of 16 male cancer patients exposed to high arsenic levels through the water supply (200–2000 μ g/L), 15 had skin carcinomas and only one had an HAS in addition to chronic arsenical dermatosis [102]. Falk et al. [103] identified 168 cases of HAS in the USA during the period 1964-1974; seven of these cases had used Fowler's solution for 6 to 17 years. Case reports of liver angiocarcinoma associated with medicinal ingestion of arsenic are also available [104–107]. Several epidemiological studies based on data from an area of southwestern Taiwan, having high levels of inorganic arsenic in the artesian well water supply, have found elevated rates of liver cancer death [107,108]. Childhood cancer was studied by Liaw et al. [109] in northern Chile, where the arsenic content in drinking water was as high as $860 \mu g/L$. No increases were detected for all cancers combined. However, childhood liver cancer mortality under age 20, which is normally extremely rare, was markedly increased for those who were young children when they would have experienced high arsenic concentrations in water.

Liver is a target organ for arsenic-induced carcinogenesis. Liu and Waalkes [98] reviewed the mechanisms of arsenic-related liver cancer and the authors described the potential mechanisms of arsenic-induced hepatic carcinogenesis. The potential mechanisms include oxidative DNA damage, impaired DNA damage repair, acquired apoptotic tolerance, hyperproliferation, altered DNA methylation, and aberrant estrogen signaling. Some of these mechanisms may be liver specific.

References

- [1] Harvey CF. Poisoned waters traced to source. Nature 2008;454:415-16.
- [2] Fendorf S, Michael HA, Van Geen A. Spatial and temporal variations of groundwater arsenic in south and Southeast Asia. Science 2010;328:1123–7.
- [3] Environmental Protection Agency (US EPA). National primary drinking water regulations; arsenic and clarifications to compliance and new source contaminants monitoring; final rule. Fed Register 2001;66:6976–7066.
- [4] World Health Organization. Arsenic in drinking water. Fact Sheet No. 210. Geneva: WHO Press; 1999. 1–6 pp.
- [5] National Research Council. Arsenic in drinking water. Washington, DC: National Academy Press; 1999.
- [6] Guha Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborty D, et al. Arsenic level in drinking water and the prevalence of skin lesions in West Bengal, India. Int J Epidemiol 1998;27:871–7.
- [7] Bagla P, Kaiser J. India's spreading health crisis draws global arsenic experts. Science 1996;274:174-5.
- [8] Garai R, Chakraborty AK, Dey SB, Saha KC. Chronic arsenic poisoning from tube well water. J Indian Med Assoc 1984;82:34–5.
- [9] Saha KC. Melanokeratosis from arsenic contaminated tubewell water. Indian J Dermatol 1984;29:37-46.
- [10] Guha Mazumder DN, Chakraborty AK, Ghosh A, Dasgupta J, Chakraborty DP, Dey SB, et al. Chronic arsenic toxicity from drinking tubewell water in rural West Bengal. Bull World Health Org 1988;66:499–506.
- [11] Dhar RK, Biswas BK, Samanta G, Mandal BK, Chakraborty D, Roy S, et al. Ground water arsenic calamity in Bangladesh. Curr Sci 1997;73:48–59.
- [12] Finkelman RB, Belkin HE, Zheng B. Health impacts of domestic coal use in China. Proc Natl Acad Sci USA 1999;96:3427–31.
- [13] Belkin HE, Zheng BS, Finkelman RB. Geochemistry of coal and endemic arsenism in Southwest Guizhou, China. U.S. Geological survey open file report No. 97–496; 1997.
- [14] Zheng BS, Ding ZH, Huang RG, Zhu JM, Yu XY, Wang AM, et al. Issues of health and disease relating to coal use in southwestern China. Int J Coal Geol 1999;40:119–32.
- [15] Chatterjee A, Das D, Chakraborti D. A study of groundwater contamination by arsenic in the residential area of Behala, Calcutta, due to industrial pollution. Environ Pollut 1993;80:57–65.
- [16] Buchet JP, Lauwerys R, Roels H. Comparison of the urinary excretion of arsenic metabolites after a single dose of sodium arsenite, monomethylarsonate or dimethylarsinate in man. Int Arch Occup Environ Health 1981;48:71–9.
- [17] Vahter M, Marafante E. Intracellular interaction and metabolic fate of arsenite and arsenate in mice and rabbits. Chem Biol Interact 1983;47:29–44.
- [18] Marafante E, Vahter M. The effect of methyltransferase inhibition on the metabolism of (74As) arsenite in mice and rabbits. Chem Biol Interact 1984;50:49–57.
- [19] Thompson DJ. A chemical hypothesis for arsenic methylation in mammals. Chem Biol Interact 1993;88:89–114.

- [20] Li JH, Rossman TG. Comutagenesis of sodium arsenite with ultraviolet radiation in Chinese Hamster V79 cells. Biol Metals 1991;4:197–200.
- [21] Hirata M, Tanaka A, Hisanaga A, Ishinishi N. Effects of glutathione depletion on the acute nephrotoxic potential of arsenite and on arsenic metabolism in hamster. Toxicol Appl Pharmacol 1990;106:469–81.
- [22] Keyse SM, Applegate LA, Tromvoukis Y, Tyrrell RM. Oxidant stress leads to transcriptional activation of the human heme oxygenase gene in cultured skin fibroblasts. Mol Cell Biol 1990;10:4967–9.
- [23] Lee TC, Ho IC. Expression of heme oxygenase in arsenic resistant human lung adenocarcinoma cells. Cancer Res 1994;54:1660-4.
- [24] Taketani S, Kohno H, Tokunaga R, Ishii T, Bannai S. Selenium antagonizes the induction of human heme oxygenase by arsenic cadmium ions. Biochem Int 1991;343:86–7.
- [25] Abernathy CO, Ohanian EV. Non carcinogenic effects of inorganic arsenic. Environ Geochem Health 1992;14:35–41.
- [26] Kala SV, Neely MW, Kala G, Prater CI, Atwood DW, Rice JS, et al. The MRP2/cMOAT transporter and arsenic-glutathione complex formation are required for biliary excretion of arsenic. J Biol Chem 2000;275:33404-33408.
- [27] Bang FL. De hydrope ex ingesto arsenico observattio. Soc Med Havn Collect 1774;1:307.
- [28] Hutchinson J. Diet and therapeutics. Arch Surg 1895;6:389–95.
- [29] Kelynack TN, Kirby W, Delepine S, Tattersall CH. Arsenical poisoning from beer drinking. Lancet 1900;156:1600-3.
- [30] Reynolds ES. An account of the epidemic outbreak of arsenical poisoning occurring in beer drinking in the north of England and in the Midland countries in 1900. Med Chir Trans 1901;84:409–52.
- [31] Stockman R. Chronic arsenic poisoning. Edinburgh Med J 1921;27:1.
- [32] Cannon AB. Chronic arsenical poisoning: symptoms and sources. N Y State J Med 1936;36:219.
- [33] Maccallum FO. Transmission of arsenotherapy jaundice by blood: failure with faeces and naso-pharyngeal washings. Lancet 1945;245:342.
- [34] Zimmerman H. Hepatotoxicity: the adverse effects of drugs and other chemicals on the liver, 2nd ed. Philadelphia, Pennsylvania: Lippincott Williams and Wilkins; 1999. pp. 365–90.
- [35] Sanyal AJ, Bosch J, Blei A, Arroyo V. Portal hypertension and its complications. Gastroenterology 2008;134:1715–28.
- [36] Sarin SK, Kumar A, Chawla YK, Baijal SS, Dhiman RK, Jafri W, et al. Noncirrhotic portal fibrosis/ idiopathic portal hypertension: APASL recommendations for diagnosis and treatment. Hepatol Int 2007;1:398–413.
- [37] Khanna R, Sarin SK. Non-cirrhotic portal hypertension—diagnosis and management. J Hepatol 2014;60:421-41.
- [38] Zeegen R, Stansfeld AG, Dawson AM, Hunt AH. Prolonged survival after portal decompression of patients with non-cirrhotic intrahepatic portal hypertension. Gut 1970;11:610–17.
- [39] Viallet A, Guillaume E, Corte J, Legare A, Lavoie P. Presinusoidal portal hypertension following chronic arsenic ingestion [Abstr]. Gastroenterology 1972;62:177.
- [40] Morris JS, Schmid M, Newman S, Schever PJ, Sherlock S. Arsenic and noncirrhotic portal hypertension. Gastroenterology 1974;66:86–94.
- [41] Huet PM, Guillaime E, Gote J, Legare A, Lavoie P, Viallet A. Non cirrhotic presinusoidal portal hypertension association with chronic arsenical intoxication. Gastroenterology 1975;68:1270–7.
- [42] Szuler IM, Williams CN, Hindmarsh JT, Park Dincsoy H. Massive variceal hemorrhage secondary to presinusoidal portal hypertension due to arsenic poisoning. Can Med Assoc J 1979;120:168–71.
- [43] Nevens F, Fevery J, Steenbergen WV, Sciot R, Desmet V, De Groote J. Arsenic and non cirrhotic portal hypertension. J Hepatol 1990;11:80–5.
- [44] Dutta DV, Mitra SK, Chhuttani PN, Chakrsvarti RN. Chronic oral arsenic intoxication as a possible aetiological factor in idiopathic portal hypertension (non-cirrhotic portal fibrosis) in India. Gut 1979;20:378–84.
- [45] Santra A, Das Gupta J, De BK, Roy B, Guha Mazumder DN. Hepatic manifestation in chronic arsenic toxicity. Indian J Gastroenterol 1999;18:152–5.
- [46] Centeno JA, Mullick FG, Martinez L, Page NP, Gibb H, Longfellow D, et al. Pathology related to chronic arsenic exposure. Environ Health Perspect 2002;110(Suppl. 5):883–6.
- [47] Franklin M, Bean WB, Hardin RC. Fowler's solution as an etiological agent in cirrhosis. Am J Med Sci 1950;219:589–96.
- [48] Baldridge CW. Relationship between antisyphilitic treatment and toxic cirrhosis. Am J Med Sci 1934;188:685–9.
- [49] Luchtrath H. Liver cirrhosis in chronic arsenic intoxication of vintagers. Dtsch Med Wochenschr 1972;97:21–2.
- [50] Guha Mazumder DN, Das Gupta J, Santra A, Pal A, Ghosh A, Sarkar S. Chronic arsenic toxicity in West Bengal—the worst calamity in the world. J Indian Med Assoc 1998;96:4–7.
- [51] Boyer JL, Sen Gupta KP, Biswas SK, Pal NC, Basu Mallick KC, Iber FL, et al. Idiopathic portal hypertension—comparison with portal hypertension of cirrhosis and extrahepatic portal vein obstruction. Ann Intern Med 1967;66:41–68.
- [52] Yu HS, Chang KL, Kao YH, Yu CL, Chen GSCCH, Fang KT. In vitro toxicity of IgG antibodies on vascular endothelial cells from patients with endemic peripheral vascular disease in Taiwan. Atherosclerosis 1998;137:141-7.
- [53] Boyer JL. Arsenic: an endothelial cell toxin in the liver? Indian J Gastroenterol 1999;18:141-2.
- [54] Figueroa L, Razmilic B, Gonzalex M. Corporal distribution of arsenic in mummied bodies owned to an arsenical habitat. In: Sancha FAM, editor. International seminar proceedings. Arsenic in the environment and its incidence on health. Santiago, Chile: Universidad de Chile, Facultad de Ciencias Fisicas y Matematicas; 1992. pp. 77–82.
- [55] Chakraborty AK, Saha KC. Arsenical dermatosis from tubewell water in West Bengal. Indian J Med Res 1987;85:326–34.
- [56] Ma HZ, Xia YJ, Wu KG, Wu KG, Sun TZ, Mumford JL. Human exposure to arsenic and health effects in Bayingnormen, Inner Mongolia. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic exposure and health effects. London: Elsevier Science; 1999. pp. 127–31.
- [57] Ahmad SA, Sayed MHSU, Hadi SA, Faruquee MH, Jalil MA, Ahmed R, et al. Arsenicosis in a village in Bangladesh. Int J Environ Health Res 1999;9:187–95.
- [58] Santra A, Chowdhury A, Ghatak S, Biswas A, Dhali GK. Arsenic induced apoptosis in mouse liver is mitochondria dependent and is abrogated by N-acetylcysteine. Toxicol Appl Pharmacol 2007;220:146–55.
- [59] Ghatak S. Studies on the mechanism of liver injury in arsenic induced toxicity. Ph.D. thesis under the West Bengal University of Health Sciences, Kolkata; 2012.
- [60] Flora SJ. Arsenic induced oxidative stress and its reversibility following combined administration of N-acetylcysteine and meso 2,3-dimercaptosuccinic acid in rats. Clin Exp Pharmacol Physiol 1999;26:865–9.
- [61] Cai X, Shen LY, Zhu Q, Jia PM, Yu Y, Zhou L, et al. Arsenic trioxide induced apoptosis and differentiation are associated respectively with mitochondrial transmembrane potential collapse and retinoic acid signalling pathways in acute promyelocytic leukemia. Leukemia 2000;14:262–70.
- [62] Dai J, Weinberg RS, Waxman S, Jing Y. Malignant cells can be sensitized to undergo growth inhibition and apoptosis by arsenic trioxide through modulation of the glutathione redox system. Blood 1999;93:268–77.
- [63] Lee TC, Wei ML, Chang WJ, et al. Elevation of glutathione levels and glutathione S-transferase activity in arsenic resistant Chinese hamster ovary cells. In Vitro Cell Dev Biol 1989;25:442–8.

- [64] Aposhian HV. Enzymatic methylation of arsenic species and other new approaches to arsenic toxicity. Annu Rev Pharmacol Toxicol 1997;37:397–419.
- [65] Carter DE, Aposhian HV, Gandolfi AJ. The metabolism of inorganic arsenic oxides, gallium arsenide and arsine: a toxicochemical review. Toxicol Appl Pharmacol 2003;193:309–34.
- [66] Lemasters JJ, Nieminen AL, Qian T, Trost LC, Elmore SP, Nishimura Y, et al. The mitochondrial permeability transition in cell death: a common mechanism in necrosis, apoptosis and autophagy. Biochim Biophys Acta 1998;1366:177–96.
- [67] Crompton M. The mitochondrial permeability transition pore and its role in cell death. Biochem J 1999;341:233–49.
- [68] Ichas F, Mazat JP. From calcium signalling to cell death: two conformations for the mitochondrial permeability transition pore. Switching from low- to high-conductance state. Biochim Biophys Acta 1998;1366:33–50.
- [69] Thornberry NA, Lazebnik Y. Caspases: enemies within. Science 1998;281:1312-16.
- [70] Verhagen A, Ekert PG, Pakusch M, Silke J, Connoly LM, Reid GE, et al. Identification of DIABLO, a mammalian protein that promotes apoptosis by binding and antagonizing IAP proteins. Cell 2000;102:43–53.
- [71] Santra A, Maiti A, Chowdhury A, Guha Mazumder DN. Oxidative stress in liver of mice exposed to arsenic contaminated water. Indian J Gastroenterol 2000;19:112–15.
- [72] Santra A, Maiti A, Das S, Lahiri S, Chakraborty SK, Guha Mazumder DN. Hepatic damage caused by arsenic (As) toxicity in experimental animals. Clin Toxicol 2000;38:395–405.
- [73] Bellomo G, Orrenius S. Altered thiol and calcium homeostasis in oxidative hepatocellular injury. Hepatology 1985;5:876–82.
- [74] Sarin SK, Sharma G, Banerjee S, Kathayat R, Malhotra V. Hepatic fibrogenesis using chronic arsenic ingestion: studies in a murine model. Indian J Exp Biol 1999;37:147–51.
- [75] Das A, Santra A, Lahiri S, Guha Mazumder DN. Implications of oxidative stress and hepatic cytokines (TNF-α and IL-6) response in the pathogenesis of hepatic collagenesis in chronic arsenic toxicity. Toxicol Appl Pharmacol 2005;204:18–26.
- [76] Bachem MG, Shell KM, Melchior R, Kroof J, Gressner AM. TNF-alpha and TGF-beta1 stimulate fibronectin synthesis and the transdifferentiation of fat storing cells in the rat liver into myofibroblast. Virchows Arch B Cell Pathol 1993;63:123–30.
- [77] Ghatak S, Biswas A, Dhali GK, Chowdhury A, Boyer JL, Santra A. Oxidative stress and hepatic stellate cell activation are key events in arsenic induced liver fibrosis in mice. Toxicol Appl Pharmacol 2011;251:59-69.
- [78] Nieto N, Friedman SL, Greenwel P, Cederbaum AI. CYP2E1 mediated oxidative stress induces collagen type I expression in rat hepatic stellate cells. Hepatology 1999;30:987–95.
- [79] Arthur MJ. Fibrogenesis II (metalloproteinases and their inhibitors in liver fibrosis). Am J Physiol Gastrointest Liver Physiology 2000;279:245–9.
- [80] Friedman SL. Mechanism of hepatic fibrogenesis. Gastroenterology 2008;134:1655–69.
- [81] Su GL. Lipopolysaccharides in liver injury: molecular mechanisms of Kupffer cell activation. Am J Physiol Gastrointest Liver Physiol 2002;283:256–65.
- [82] Bissell DM, Roulot D, George J. Transforming growth factor beta and the liver. Hepatology 2001;34:859-67.
- [83] Bansal MB, Kovalovich K, Gupta R, Li W, Agarwal A, Radbill B, et al. Interleukin-6 protects hepatocytes from CCl4-mediated necrosis and apoptosis in mice by reducing MMP-2 expression. J Hepatol 2005;42:548–56.
- [84] Taipale J, Keski-Oja J. Growth factors in the extracellular matrix. FASEB J 1997;11:51–9.
- [85] Davis G, Bayless KJ, Davis M, Meininger GA. Regulation of tissue injury responses by the exposure of matricryptic sites within extracellular matrix molecules. Am J Pathol 2000;156:1489–98.

- [86] Uchinami H, Seki E, Brenner DA, D'Armiento J. Loss of MMP 13 attenuates murine hepatic injury and fibrosis during cholestasis. Hepatology 2006;44:420–9.
- [87] Nieto N, Friedman SL, Cederbaum AI. Stimulation and proliferation of primary rat hepatic stellate cells by cytochrome p450 2e1-derived reactive oxygen species. Hepatology 2002;35:62–73.
- [88] Straub AC, Stolz DB, Ross MA, Hernández-Zavala A, Soucy NV, Klei LR, et al. Arsenic stimulates sinusoidal endothelial cell capillarization and vessel remodeling in mouse liver. Hepatology 2007;45:205–12.
- [89] Mazumder DN. Effect of chronic intake of arsenic-contaminated water on liver. Toxicol Appl Pharmacol 2005;206:169–75.
- [90] Cogger VC, Muller M, Fraser R, McLean AJ, Khan J, Le Couteur DG. The effects of oxidative stress on the liver sieve. J Hepatol 2004;41:370–6.
- [91] Straub AC, Clark KA, Ross MA, Chandra AG, Li S, Gao X, et al. Arsenic-stimulated liver sinusoidal capillarization in mice requires NADPH oxidase-generated superoxide. J Clin Invest 2008;118:3980–9.
- [92] Friedman SL, Roll FJ, Boyles J, Bissell DM. Hepatic lipocytes: the principal collagen-producing cells of normal rat liver. Proc Natl Acad Sci USA 1985;82:8681–5.
- [93] Elsharkawy AM, Oakley F, Mann DA. The role and regulation of hepatic stellate cell apoptosis in reversal of liver fibrosis. Apoptosis 2005;10:927–39.
- [94] Wu J, Liu J, Waalkes MP, Cheng ML, Li L, Li CX, et al. High dietary fat exacerbates arsenic induced liver fibrosis in mice. Exp Biol Med 2008;233:377–84.
- [95] Chen CJ, Yu MW, Liaw YF. Epidemiological characteristics and risk factors of hepatocellular carcinoma. J Gastroenterol Hepatol 1997;12:S294–308.
- [96] Chen Y, Ahsan H. Cancer burden from arsenic in drinking water in Bangladesh. Am J Public Health 2004;94:741-4.
- [97] Zhou YS, Du H, Cheng ML, Liu J, Zhang XJ, Xu L. The investigation of death from diseases caused by coalburning type of arsenic poisoning. Chin J Endemiol 2002;21:484–6.
- [98] Liu J, Zheng B, Aposhian HV, Zhou Y, Chen ML, Zhang A, et al. Chronic arsenic poisoning from burning high-arsenic-containing coal in Guizhou, China. Environ Health Perspect 2002;110:119–22.
- [99] Luchtrath H. The consequences of chronic arsenic poisoning among Moselle wine growers. Pathoanatomical investigations of post-mortem examinations performed between 1960 and 1977. J Cancer Res Clin Oncol 1983;105:173–82.
- [100] Ho SY, Tsai CC, Trai YC, Guo HR. Hepatic angiosarcoma presenting as hepatic rupture in a patient with long-term ingestion of arsenic. J Formos Med Assoc 2004;103:374–9.
- [101] Roth F. The sequelae of chronic arsenic poisoning in Moselle vintners. J Med Mon 1957;2:172-5.
- [102] Zaldivar R, Prumes L, Ghai GL. Arsenic dose in patients with cutaneous carcinomata and hepatic haemangioendothelioma after environmental and occupational exposure. Arch Tocicol 1981;47:145–54.
- [103] Falk H, Caldwell CG, Ishak KG, Thomas LB, Popper H. Arsenic related hepatic angiosarcoma. Am J Ind Med 1981;2:43–50.
- [104] Roat JW, Wald A, Mendelow H, Pataki KI. Hepatic angiosarcoma associated with short term arsenic ingestion. Am J Med 1982;73:933-6.
- [105] Lander JJ, Stanley RJ, Sumner HW, Boswell DC, Aach RD. Angiosarcoma of the liver associated with Fowler's solution (potassium arsenite). Environ Mutagen 1975;68:1582–6.
- [106] Kasper ML, Schoenfield L, Strom RL, Theologides A. Hepatic angiosarcoma and bronchioloalveolar carcinoma induced by Fowler's solution. JAMA 1984;252:3407–8.
- [107] Chen CJ, Kuo TL, Wu MM. Arsenic and cancer [letter]. Lancet 1988;I:414-15.
- [108] Wu MM, Kuo TL, Hwang YH, Chen CJ. Dose response relation between arsenic well water and mortality from cancer. Am J Epidemiol 1989;130:1123–32.
- [109] Liaw J, Marshall G, Yuan Y, Ferreccio C, Steinmaus C, Smith AH. Increased childhood liver cancer mortality and arsenic in drinking water in northern Chile. Cancer Epidemiol Biomarkers Prev 2008;17:1982–7.

13

Arsenic and Respiratory Disease

Kathryn Ramsey

TELETHON KIDS INSTITUTE, UNIVERSITY OF WESTERN AUSTRALIA, SUBIACO, WESTERN AUSTRALIA, AUSTRALIA

CHAPTER OUTLINE

13.1 Introduction	
13.2 Chronic Arsenic Exposure and Respiratory Health	
13.3 Early Life Arsenic Exposure and Lung Health	
13.4 Mechanistic Data on Arsenic Exposure and the Lung	
13.5 Conclusions	
References	

13.1 Introduction

The lung is frequently the target of airborne and systemic environmental exposures. An adult at rest breathes in more than 7000 liters of air daily into a series of branching airways, which become shorter, narrower, and more numerous with every division. Gas exchange occurs across the extremely thin blood–gas interface of the alveoli and pulmonary capillary network [1]. This makes the lung uniquely exposed to airborne toxins. In addition, the lung receives the entire cardiac output from the right side of the heart, placing it at risk of systemic toxins. As a result, the lung is highly susceptible to the development of respiratory disease caused by exposure to environmental toxins.

Arsenic is the only environmental agent that has been linked to both malignant and nonmalignant respiratory disease following ingestion, rather than inhalation, making arsenic a unique toxicant to the respiratory system. Chronic arsenic exposure via drinking water has been shown to increase the risk of developing lung cancer along with other malignant diseases [2–5]. The increased risk of developing lung cancer is similar whether the arsenic has been ingested or inhaled [6]. Chronic arsenic exposure has been associated with a variety of respiratory symptoms such as chronic cough and shortness of breath, as well as the development of non-malignant respiratory diseases such as bronchiectasis and chronic obstructive pulmonary disease (COPD). The lung is highly sensitive to environmental toxins during stages of development both *in utero* and in early postnatal life, and the risks of developing non-malignant respiratory disease appear highest when the exposure occurs in early life [7,8].

13.1.1 Lung Development

Lung development begins within 3 weeks of fertilization and continues after birth. The lung is highly susceptible to environmental insults during both the *in utero* and postnatal stages of development. Within the first weeks of gestation the lung bud (ventral bud of primitive foregut) forms into trachea and primary and secondary bronchi [1]. This is followed by differentiation of the airway epithelial lining and formation of cartilage, bronchial smooth muscle, and pulmonary blood vessels by week 17. Respiratory bronchioles, alveolar ducts, and primitive alveoli develop by week 27. Further development and differentiation of the alveolar cell types occurs between weeks 28 and 36. The final alveolar stage of development occurs from week 36 of gestation to early childhood and involves the septation and multiplication of alveoli and growth of the airways [1]. The switch at birth from the fluid-filled intrauterine environment to breathing air is dramatic and requires absorption of the fetal lung lining within the first few breaths. The lungs increase in volume from approximately 80 mL at birth to 6000 mL in adulthood, and therefore the lung remains susceptible to external insults throughout postnatal life [9].

13.2 Chronic Arsenic Exposure and Respiratory Health

13.2.1 Chronic Arsenic Exposure and Respiratory Symptoms

A high prevalence of respiratory symptoms has been reported among adults chronically exposed to arsenic [10]. The prevalence of cough, shortness of breath, and chest sounds correlated positively with arsenic concentrations in drinking water, and were most pronounced in individuals with skin lesions in West Bengal, India [11]. In the same population, arsenicexposed men with skin lesions had a higher prevalence of respiratory symptoms compared with arsenic-exposed men without skin lesions, in particular morning cough (odds ratio (OR) = 2.8, 95% confidence interval (CI) 1.2, 6.6) and shortness of breath (OR = 3.8, 95\% CI 0.7, 20.6) [12]. In arsenic-exposed men in Pakistan, a positive correlation between arsenic in drinking water and arsenic in scalp hair was found, but was stronger in those with respiratory disease symptoms (coughing, chest sounds, and shortness of breath) (r = 0.785, p < 0.001) than in those without respiratory disease symptoms (r = 0.511, p < 0.001) and the severity of respiratory symptoms was increased in those men with skin lesions [13]. Data from the Health Effects of Arsenic Exposure Longitudinal Study (HEALS), a large prospective longitudinal cohort study in Bangladesh, found a positive dose-response relationship between arsenic exposure and respiratory symptoms (chronic cough, breathing problems, or blood in sputum) [14] and the odds of having dyspnea in the previous 6 months was 1.32 times greater (95% CI 1.15, 1.52) in adults exposed to arsenic at high concentrations ($>50 \mu g/L$) compared with those exposed to arsenic at low concentrations ($<7 \mu g/L$, p < 0.01) [15].

13.2.2 Chronic Arsenic Exposure and Lung Function

Chronic arsenic exposure throughout life has also been linked to impaired lung function. There have been a small number of studies that have examined lung function in arsenic-exposed

populations of South Asia. These studies have reported low forced expiratory volume in 1 second (FEV_1) and forced vital capacity (FVC) among individuals with skins lesions or exposed to high levels of arsenic (>250 μ g/L); however, the type of lung function impairment varies among studies, with some studies recording predominantly obstructive lung disease [16], and others predominantly restrictive or mixed obstructive/restrictive lung disease [12,17]. The degree of lung function impairment is clinically significant, with 256.2 mL lower FEV₁ (95% CI 113.9, 398.4) and 287.8 mL lower FVC (95% CI 134.9, 440.8) in arsenic-exposed men with skin lesions in West Bengal [12]. The HEALS study published the largest lung function study from their populationbased cohort in 2013. They observed significant associations between both water and urinary arsenic levels and decreases in FEV_1 and FVC [18]. These data indicate that arsenic exposure can lead to both restrictive (lower FVC) and obstructive (lower FEV_1) patterns of lung function impairment and the balance may shift depending on a multitude of confounding factors, such as arsenic dose, speciation, timing, susceptibility, and additional insults. Arsenic exposure may alter lung function by impairing lung development or promoting lung remodeling. As lifetime exposure to arsenic is difficult to accurately quantify, it is unclear whether the reduction in lung function is due to recent or past exposure to arsenic.

13.2.3 Chronic Arsenic Exposure and Immunosuppression

Arsenic has been found to be immunosuppressive in humans [19–21]. A microarray study in the USA showed that arsenic exposure altered the expression of genes involved in defense responses, immune function, cell growth, apoptosis, cell cycle regulation, and T-cell receptor signaling in lymphocytes [22]. Arsenic exposure has also been shown to modify the expression of genes, cytokines, and growth factors involved in inflammation in lymphocytes [21], alter T-cell proliferation [19], and impair T-cell function [20]. In children in Mexico exposed to arsenic in drinking water, urinary arsenic levels were associated with reduced lymphocyte proliferation and interleukin-2 secretion [23]. Arsenic exposure in adults in West Bengal was associated with loss of cell adhesion capacity and impaired phagocytic capacity of macrophages [24]. A study examining humoral immune responses following chronic arsenic exposure found that patients with arsenicosis had elevated serum levels of immunoglobulins (IgG, IgE, and IgA) and subjects with arsenic-associated respiratory complications (chest sounds, asthma, bronchitis, and cough) had higher mean IgE levels than exposed subjects without respiratory complications [25]. An inverse association was also found between urinary arsenic concentration and serum Clara cell protein 16 in arsenic-exposed subjects with skin lesions, indicating potential damage to respiratory Clara cells [26]. Therefore, arsenic-induced immunosuppression in adults and children may contribute to increased susceptibility to respiratory infections and the development of chronic respiratory disease.

13.2.4 Chronic Arsenic Exposure and Non-Malignant Respiratory Disease

There is strong evidence that chronic exposure to arsenic increases the susceptibility to developing respiratory disease. The prevalence of chronic bronchitis and chronic cough was found to be significantly higher in arsenic-exposed residents of Bangladesh compared with unexposed residents and increased with increasing concentrations of arsenic [27,28]. A study of 108 arsenic-exposed subjects in West Bengal found an increase in bronchiectasis in patients with arsenic-related skin lesions (adjusted OR = 10, 95% CI 2.7, 37) compared to exposed subjects without skin lesions [29]. In Taiwan, the risks of developing bronchitis were higher in men (standardized mortality ratio (SMR) = 1.48, 95% CI 1.25, 1.73) and women (SMR = 1.53, 95% CI 1.3, 1.80) when arsenic-associated blackfoot disease was prevalent following the consumption of well water containing 780 µg/L arsenic [30]. After the peak arsenic exposure event occurred in Antofagasta, Chile (1958), mortality rate ratios for tuberculosis in males started increasing reaching a rate ratio of 2.1 (95% CI 1.7, 2.6) 20 years after the peak exposure event [31]. While the SMR for COPD over all ages was not significantly increased in Antofagasta, there was a very high relative risk for COPD deaths for men and women aged between 30 and 39 years. The individuals who died of COPD in this age group would have been children during the peak exposure event and this thus highlights the significance of early life arsenic exposure on the development of respiratory disease [3].

13.3 Early Life Arsenic Exposure and Lung Health

13.3.1 Early Life Arsenic Exposure, Birth Outcomes, and Growth

Exposure to arsenic during pregnancy has been shown to increase the risks of spontaneous abortions (OR range 1.14–6.07, 95% CI range 1.04, 24.0) [32–36], still births (OR range 1.23–2.5, 95% CI range 0.87, 4.9) [33,37], neonatal death (OR range 1.8–2.81, 95% CI range 0.9, 10.8) [32,33,35,38], and post-neonatal death (relative risk (RR) range 1.17–1.26, 95% CI range 1.03, 1.32) [32,34,36]. Exposure to arsenic during pregnancy is also associated with decreased infant birth weight [39,40]. The mechanism for this is not understood but may be a result of arsenic-induced oxidative stress resulting in placental insufficiency or disruption of the endocrine control of glucose homeostasis and cellular growth [41,42]. Fetal growth restriction is associated with impaired lung function and greater respiratory morbidity in early childhood [43–46] and adulthood [47,48]. This is significant as children who are small for gestational age are more likely to be hospitalized for respiratory tract infections [49,50]. This effect on somatic growth is just one of the potential mechanisms that may explain the link between *in utero* arsenic exposure and the development of chronic lung disease. However, as discussed below, there is also evidence that arsenic can have a direct effect on the lung over and above an effect that can be explained by impaired somatic growth.

13.3.2 Early Life Arsenic Exposure and Immune Development

Arsenic exposure during pregnancy has been linked with altered immune development and increased susceptibility to respiratory infections in infancy [36,51]. Arsenic exposure during pregnancy can increase oxidative stress and inflammation in the placenta, reduce placental T cells and alter the expression of cord blood cytokines (IL-1 β , IL-8, IFN- γ , TNF- α) [51]. Infants exposed to arsenic *in utero* had evidence of impaired thymic development and higher levels of

fever, diarrhea, and acute respiratory infections [52]. In a community-based prospective study of 1552 live-born infants in Bangladesh, exposure to arsenic at concentrations above $250 \mu g/L$ *in utero* increased the risk of developing lower respiratory tract infections by 69% (RR = 1.69, 95% CI 1.36, 2.09), severe lower respiratory tract infections by 54% (RR = 1.54, 95% CI 1.21, 1.97), and diarrhea by 20% (RR = 1.20, 95% CI 1.01, 1.43) compared to those exposed to arsenic at concentrations below $39 \mu g/L$ [7]. The New Hampshire Birth Cohort Study in the USA also reported increased upper and lower respiratory tract infections, as well as increased respiratory symptoms in 4-month-old infants exposed to relatively low level (0.45–58.3 $\mu g/L$) arsenic *in utero* [53]. Recurrent lower respiratory tract infections in early childhood and impaired pulmonary immune defenses are important risk factors for respiratory morbidity and mortality in childhood, and the development of chronic respiratory diseases in adulthood [46,54].

13.3.3 Early Life Arsenic Exposure and Non-Malignant Respiratory Disease in Children

The unique arsenic exposure event in Antofagasta, Chile, revealed the significance of early life arsenic exposure in the development of respiratory morbidity and mortality. In the 1950s the establishment of a new drinking water source resulted in the residents of Antofagasta consuming water containing high concentrations of arsenic (90–860 μ g/L) for more than 20 years [55]. Two years after the peak arsenic exposure in Antofagasta, children presented with arsenicosis associated with respiratory disease, including diffuse and segmentary bronchiectasis [56]. Autopsies performed on five children with evidence of arsenic poisoning who died between 1968 and 1969 in Antofagasta revealed lung abnormalities in four of the five children including evidence of pulmonary interstitial fibrosis and mild bronchiectasis [57]. In a study of schoolaged children in Antofagasta, 38.8% of children with abnormal skin pigmentation had chronic cough and 15.9% had bronchopulmonary disease (n = 144), compared with 3.1% and 6.2% of children with normal skin (n = 36), respectively [58]. The presence of cough and dyspnea correlated with the mean drinking water arsenic concentrations in 398 children [59] and after the installation of a water treatment plant, the prevalence of cough and dyspnea dropped from 38 to 7% [58]. Following a series of postmortem examinations, Zaldivar [60] described the pathology of those who died from chronic arsenic poisoning in Antofagasta, which included a combination of cardiovascular, pulmonary, digestive, and neurological abnormalities. Among the children who were examined, the prevalence of bronchiectasis and bronchopneumonia was 23 and 3.4 times higher in those with chronic arsenic dermatosis than the estimated prevalence in the general population of Chilean children [60]. It was postulated that the combination of malnutrition and dietary arsenic exposure was resulting in the high prevalence of bronchiectasis and bronchopneumonia in the children of Antofagasta, and arsenic was acting as an immunosuppressant in infants and children. In addition, children in Bangladesh exposed to high (>500 µg/L) arsenic *in utero* and early life had a marked increase in chronic respiratory symptoms such as wheezing and shortness of breath when walking or climbing [61]. These studies clearly demonstrated the importance of ingested inorganic arsenic in determining respiratory health early in life.

13.3.4 Early Life Arsenic Exposure and Non-Malignant Respiratory Disease in Adults

It was not until a study by Smith and colleagues that the long-term impacts of early life arsenic exposure, and particularly *in utero* exposure, were realized [8]. They found that the SMRs for lung cancer and bronchiectasis in those born just prior to the peak exposure event in Antofagasta (1950-1957) and exposed to arsenic in early childhood were 7.0 (95% CI 5.4, 8.9) and 12.4 (95% CI 3.3, 31.7), respectively. Those born during the peak exposure period (1958–1970), who were exposed to arsenic in utero and early childhood, had corresponding SMRs of 6.1 (95% CI 3.5, 9.9) for lung cancer and 46.2 (95% CI 21.1, 87.7) for bronchiectasis [8] (Figure 13-1). This groundbreaking study was the first to show that *in utero* and early life exposure to arsenic had pronounced effects on the lungs and the long-term risk of developing non-malignant lung disease, which is consistent with the literature on cigarette smoke [62]. In a subsequent study Dauphiné and colleagues [63] showed that a small group of residents of Antofagasta, exposed to $>800 \,\mu g/L$ arsenic before the age of 10 years (n = 32), had long-term deficits in lung function as adults, with 11.5% lower FEV₁ (p = 0.004), 12.2% lower FVC (p = 0.04), and increased breathlessness (OR = 5.94, 95% CI 1.36, 26.0) compared to unexposed adults. Surprisingly, the magnitude of lung function deficits as a result of exposure to arsenic in early life was equivalent to the effects of smoking cigarettes throughout adulthood. These studies provide clear associations between early life exposure to arsenic and long-term impairment in respiratory health, but do not provide crucial mechanistic data to explain these associations.



FIGURE 13–1 Increased mortality from bronchiectasis, COPD, and lung cancer following exposure to arsenic in early life. *Reproduced with permission from Smith et al.* [8].

13.4 Mechanistic Data on Arsenic Exposure and the Lung

Despite many epidemiological studies reporting links between chronic arsenic exposure and non-malignant lung disease, these investigations are constrained by a range of limitations including low sample sizes, unreliable measures of arsenic exposure (when mean/median levels in a community are used), heterogeneity of drinking water sources, migration, uncertainty of response latency, varied dose-response relationships, and population/genetic differences [64]. Mechanistic studies are therefore essential to show causality and unravel the mechanisms by which the ingestion of arsenic leads to respiratory disease [65].

13.4.1 Mechanistic Data on Early Life Arsenic Exposure

Experimental animal models have been critical to our understanding of the impact of arsenic on the lung. Some of the earliest animal studies using arsenic exposure were able to demonstrate that arsenic can pass through the placenta, the fetus thereby being exposed to the toxicant [66–68]. Additional studies have shown that methylated arsenic products (DMA and MMA) are present in fetal blood, lung, and liver tissue following *in utero* arsenic exposure [69]. As described earlier, arsenic exposure *in utero* is associated with impaired somatic growth in humans [40], which may have downstream effects on respiratory morbidity and mortality. Importantly, a recent mouse study was able to replicate this finding whereby mice exposed to arsenic *in utero* had significant deficits in growth in early life compared with unexposed offspring [70]. These observations highlight the validity of mouse models for exploring the mechanisms of arsenic-induced lung disease.

13.4.2 Mechanistic Data for Arsenic Exposure and Immune Function

Various studies have investigated the effect that arsenic has on the immune system. Mouse models of *in vivo* arsenic exposure have shown that arsenic alters the expression of key innate immunity genes in the lungs [71,72]. Exposing mice to arsenic suppresses antibody formation, inhibiting T-cell proliferation and macrophage activity, and altering cytokine expression [73–78]. Inhalation of arsenic trioxide in mice increased mortality from a pulmonary bacterial infection and decreased bactericidal activity [79]. Zebrafish exposed to arsenic at concentrations of 2 and $10 \,\mu$ g/L had drastically reduced ability to clear viral and bacterial infections [80]. Exposure to drinking water containing $100 \,\mu$ g/L arsenic in adult mice compromised the lung's immune response to influenza (H1N1) infection, resulting in greater viral load and higher mortality [81]. *In utero* and postnatal exposure to $100 \,\mu$ g/L arsenic in mice increased viral titer and exacerbated the inflammatory response to early life influenza A infection [82].

In human cell cultures, exposing macrophages to arsenic results in loss of adhesion, altered morphology, and impaired function [83]. Recent studies have found that arsenic increases ubiquitination and degradation of cystic fibrosis transmembrane conductance regulator (CFTR) chloride channels in the gills of killifish [84,85] and human airway epithelial cells [86].

The CFTR protein, a chloride channel in the apical membrane of airway epithelial cells, regulates the volume of airway surface liquid (ASL) by secreting chloride into the periciliary space, which drives the secretion of sodium and fluid into the ASL. When the CFTR protein is degraded or defective, as in cystic fibrosis lung disease, excess sodium is transported back into the cell and the ASL becomes dehydrated. Dehydrated ASL impairs the mucociliary clearance of pathogens and increases susceptibility to chronic respiratory infections [87]. Therefore along with impairing the morphology and function of immune cells, arsenic exposure can alter the distribution of cell types in the airway epithelium resulting in impaired mucociliary clearance, a significant component of innate immunity.

13.4.3 Mechanistic Data for Arsenic Exposure and Lung Structure and Function

Exposure to arsenic *in utero* has been shown to have profound effects on the lungs of experimental animals. Arsenic was shown to be a complete transplacental carcinogen in a series of studies involving exposure of mice to arsenic in their drinking water at concentrations of 0, 42.5, and 85 mg/L between days 8 and 18 of gestation [85,86]. There were dose-dependent increases in hepatocellular carcinomas, liver, and adrenal tumors in male offspring, and ovarian tumors and lung carcinomas in female offspring. This was the first study to show that ingestion of arsenic in drinking water during gestation could induce tumors in rodents [88,89]. *In utero* exposure to arsenic has also been shown to reduce lung size and alter gene expression in the lungs at birth—particularly genes involved in lung development and cell trafficking [90].

In utero and early postnatal arsenic exposure has been shown to irreversibly alter lung structure and lung function in mice [91]. Mice exposed to arsenic ($\leq 100 \,\mu$ g/L) *in utero* and early postnatal life via drinking water had hyper-responsive airways, increased airway smooth muscle, and altered airway matrix proteins compared to controls. These changes were not reversible and were only observed in mice exposed to arsenic in adulthood [91]. A study of *in utero* exposure to arsenic on three strains of mice found that the response to arsenic in the lungs of mice is genetically determined and the strain most susceptible to oxidative stress, the C57BL/6 strain, was also the most susceptible to arsenic exposure [92]. *In utero* exposure to arsenic impaired somatic growth, lung volume, and parenchymal lung mechanics in C57BL/6 mice, and altered the expression of genes that function in mucociliary clearance, innate immunity, and lung growth in the offspring [92,93]. Arsenic induced mucous cell metaplasia and increased expression of a calcium-activated chloride channel (CLCA3) in the offspring, known to regulate mucous production and secretion [92].

Arsenic exposure and infection with influenza in early life resulted in long-term deficits in lung mechanics, airway hyper-responsiveness, and increased airway smooth muscle and numbers of mucus producing cells [82]. These data demonstrate how exposure to arsenic in early life can alter the response to influenza infection resulting in both acute and long-term effects on respiratory health. Altered lung growth, lung structure, and increased mucous secretion may together impair the ability of the airways to clear pathogens, increasing the risk of chronic localized infections and chronic lung diseases, such as bronchiectasis.

13.5 Conclusions

Further investigation into the effects of arsenic exposure on lung structure and function will be critical to understanding arsenic-induced lung disease. The epidemiological data discussed suggests that exposure to arsenic, particularly *in utero* and early life, can result in long-term impairments to lung function and increased risk of developing non-malignant lung disease such as bronchiectasis. It is clear that arsenic has the potential to modulate lung growth and development and alter immune function. Impaired development of the lung may also alter the susceptibility and response to additional respiratory insults and pathogens. The development of chronic respiratory symptoms and progressive respiratory diseases in arsenic-exposed populations can have debilitating effects on the quality of life for individuals and the overall productivity and wellness of communities.

References

- [1] Joshi S, Kotecha S. Lung growth and development. Early Hum Dev 2007;83:789-94.
- [2] Smith A, Hopenhayn-Rich C, Bates M, Goeden HM, Hertz-Picciotto I, Duggan H, et al. Cancer risks from arsenic in drinking water. Environ Health Perspect 1992;97:259–67.
- [3] Smith AH, Goycolea M, Haque R, Biggs ML. Marked increase in bladder and lung cancer mortality in a region of northern Chile due to arsenic in drinking water. Am J Epidemiol 1998;147:660–9.
- [4] Ferreccio C, Gonzalez C, Milosavjlevic V, Marshall G, Sancha A, Smith A. Lung cancer and arsenic concentrations in drinking water in Chile. Epidemiology 2000;11:673–9.
- [5] Haynes R. The geographical distribution of mortality by cause in Chile. Soc Sci Med 1983;17:355–64.
- [6] Smith AH, Ercumen A, Yuan Y, Steinmaus C. Increased lung cancer risks are similar whether arsenic is ingested or inhaled. J Expo Sci Environ Epidemiol 2009;19:343–8.
- [7] Rahman A, Vahter M, Ekström EC, Persson LÂ. Arsenic exposure in pregnancy increases the risk of lower respiratory tract infection and diarrhea during infancy in Bangladesh. Environ Health Perspect 2011;119:719–24.
- [8] Smith AH, Marshall G, Yan Y, Ferreccio C, Liaw J, von Ehrenstein O, et al. Increased mortality from lung cancer and bronchiectasis in young adults after exposure to arsenic in utero and in early childhood. Environ Health Perspect 2006;114:1293–6.
- [9] Merkus P. Growth of lungs and airways in asthma. Leiden: Department of Physiology, University of Leiden; 1993.
- [10] Guha Mazumder DN. Arsenic and non-malignant lung disease. J Environ Sci Health A Tox Hazard Subst Environ Eng 2007;42:1859–67.
- [11] Guha Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborti D, et al. Arsenic in drinking water and the prevalence of respiratory effects in West Bengal, India. Int J Epidemiol 2000;29:1047–52.
- [12] von Ehrenstein OS, Mazumder DNG, Yuan Y, Samanta S, Balmes J, Sil A, et al. Decrements in lung function related to arsenic in drinking water in West Bengal, India. Am J Epidemiol 2005;162:533–41.
- [13] Arain MB, Kazi TG, Baig JA, Jamali MK, Afridi HI, Jalbani N, et al. Respiratory effects in people exposed to arsenic via the drinking water and tobacco smoking in southern part of Pakistan. Sci Total Environ 2009;407:5524–30.
- [14] Parvez F, Chen Y, Brandt-Rauf PW, Slavkovich V, Islam T, Ahmed A, et al. A prospective study of respiratory symptoms associated with chronic arsenic exposure in Bangladesh: findings from the Health Effects of Arsenic Longitudinal Study (HEALS). Thorax 2010;65:528–33.

- [15] Pesola GR, Parvez F, Chen Y, Ahmed A, Hasan R, Ahsan H. Arsenic exposure from drinking water and dyspnoea risk in Araihazar, Bangladesh: a population-based study. Eur Respir J 2012;39:1076–83.
- [16] De B, Majumdar D, Sen S, Guru S, Kunda S. Pulmonary involvement in chronic arsenic poisoning from drinking contaminated ground-water. J Assoc Physicians India 2004;52:395–400.
- [17] Guha Mazumder DN. Chronic arsenic toxicity: clinical features, epidemiology, and treatment: experience in West Bengal. J Environ Sci Health A Tox Hazard Subst Environ Eng 2003;38:141–63.
- [18] Parvez F, Chen Y, Yunus M, Olopade C, Segers S, Slavkovich V, et al. Arsenic exposure and impaired lung function. Findings from a large population-based prospective cohort study. Am J Respir Crit Care Med 2013;188:813–19.
- [19] Gonsebatt ME, Vega L, Montero R, Garcia-Vargas G, Del Razo LM, Albores A, et al. Lymphocyte replicating ability in individuals exposed to arsenic via drinking water. Mutat Res 1994;313:293–9.
- [20] Hernandez-Castro B, Doniz-Padilla LM, Salgado-Bustamante M, Rocha D, Ortiz-Perez MD, Jimenez-Capdeville ME, et al. Effect of arsenic on regulatory T cells. J Clin Immunol 2009;29:461–9.
- [21] Wu MM, Chiou HY, Ho IC, Chen CJ, Lee TC. Gene expression of inflammatory molecules in circulating lymphocytes from arsenic-exposed human subjects. Environ Health Perspect 2003;111:1429–38.
- [22] Andrew AS, Jewell D, Mason R, Whitfield M, Moore J, Karagas MR. Drinking-water arsenic exposure modulates gene expression in human lymphocytes from a US population. Environ Health Perspect 2008;116:524–31.
- [23] Soto-Pena GA, Luna AL, Acosta-Saavedra L, Conde P, Lopez-Carrillo L, Cebrian ME, et al. Assessment of lymphocyte subpopulations and cytokine secretion in children exposed to arsenic. FASEB J 2006;20:779–81.
- [24] Banerjee N, Banerjee S, Sen R, Bandyopadhyay A, Sarma N, Majumder P, et al. Chronic arsenic exposure impairs macrophage functions in the exposed individuals. J Clin Immunol 2009;29:582–94.
- [25] Islam LN, Nabi AHMN, Rahman MM, Shamim M, Zahid H. Association of respiratory complications and elevated serum immunoglobulins with drinking water arsenic toxicity in human. J Environ Sci Health A Tox Hazard Subst Environ Eng 2007;42:1807–14.
- [26] Parvez F, Chen Y, Brandt-Rauf PW, Bernard A, Dumont X, Slavkovich V, et al. Nonmalignant respiratory effects of chronic arsenic exposure from drinking water among never-smokers in Bangladesh. Environ Health Perspect 2008;116:190–5.
- [27] Milton A, Hasan Z, Rahman A, Rahman M. Chronic arsenic poisoning and respiratory effects in Bangladesh. J Occup Health 2001;43:136–40.
- [28] Milton AH, Rahman M. Respiratory effects and arsenic contaminated well water in Bangladesh. Int J Environ Health Res 2002;12:175–9.
- [29] Guha Mazumder DN, Steinmaus C, Bhattacharya P, von Ehrenstein O, Ghosh N, Gotway M, et al. Bronchiectasis in persons with skin lesions resulting from arsenic in drinking water. Epidemiology 2005;16:760-5.
- [30] Tsai S-M, Wang T-N, Ko Y-C. Mortality for certain diseases in areas with high levels of arsenic in drinking water. Arch Environ Health 1999;54:186–93.
- [31] Smith AH, Marshall G, Yuan Y, Liaw J, Ferreccio C, Steinmaus C. Evidence from Chile that arsenic in drinking water may increase mortality from pulmonary tuberculosis. Am J Epidemiol 2011;173:414–20.
- [32] Hopenhayn-Rich C, Browning SR, Hertz-Picciotto I, Ferreccio C, Peralta C, Gibb H. Chronic arsenic exposure and risk of infant mortality in two areas of Chile. Environ Health Perspect 2000;108:667–73.
- [33] Milton A, Smith W, Rahman B, Hasan Z, Kulsum U, Dear K, et al. Chronic arsenic exposure and adverse pregnancy outcomes in Bangladesh. Epidemiology 2005;16:82–6.
- [34] Rahman A, Vahter M, Ekstrom EC, Rahman M, Golam Mustafa AHM, Wahed MA, et al. Association of arsenic exposure during pregnancy with fetal loss and infant death: a cohort study in Bangladesh. Am J Epidemiol 2007;165:1389–96.

- [35] von Ehrenstein OS, Guha Mazumder DN, Hira-Smith M, Ghosh N, Yuan Y, Windham G, et al. Pregnancy outcomes, infant mortality, and arsenic in drinking water in West Bengal, India. Am J Epidemiol 2006;163:662–9.
- [36] Rahman A, Persson LA, Nermell B, El Arifeen S, Ekstrom EC, Smith AH, et al. Arsenic exposure and risk of spontaneous abortion, stillbirth, and infant mortality. Epidemiology 2010;21:797–804.
- [37] Cherry N, Shaikh K, McDonald C, Chowdhury Z. Stillbirth in rural Bangladesh: arsenic exposure and other etiological factors: a report from Gonoshasthaya Kendra. Bull World Health Org 2008;86:172–7.
- [38] Myers SL, Lobdell DT, Liu Z, Xia Y, Ren H, Li Y, et al. Maternal drinking water arsenic exposure and perinatal outcomes in Inner Mongolia, China. J Epidemiol Community Health 2010;64:325–9.
- [39] Huyck KL, Kile ML, Mahiuddin G, Quamruzzaman Q, Rahman M, Breton CV, et al. Maternal arsenic exposure associated with low birth weight in Bangladesh. J Occup Environ Med 2007;49:1097–104.
- [40] Rahman A, Vahter M, Smith AH, Nermell B, Yunus M, El Arifeen S, et al. Arsenic exposure during pregnancy and size at birth: a prospective cohort study in Bangladesh. Am J Epidemiol 2009;169:304–12.
- [41] Bodwell JE, Kingsley LA, Hamilton JW. Arsenic at very low concentrations alters glucocorticoid receptor (GR)-mediated gene activation but not GR-mediated gene repression: complex dose response effects are closely correlated with levels of activated GR and require a functional GR DNA binding domain. Chem Res Toxicol 2004;17:1064–76.
- [42] Davey JC, Nomikos AP, Wungjiranirun M, Sherman JR, Ingram L, Batki C, et al. Arsenic as an endocrine disruptor: arsenic disrupts retinoic acid receptor- and thyroid hormone receptor-mediated gene regulation and thyroid hormone-mediated amphibian tail metamorphosis. Environ Health Perspect 2008;116:165–72.
- [43] Chan KN, Noble-Jamieson CM, Elliman A, Bryan EM, Silverman M. Lung function in children of low birth weight. Arch Dis Child 1989;64:1284–93.
- [44] Rona RJ, Gulliford MC, Chinn S. Effects of prematurity and intrauterine growth on respiratory health and lung function in childhood. BMJ 1993;306:817–20.
- [45] Shaheen S, Barker DJ. Early lung growth and chronic airflow obstruction. Thorax 1994;49:533-6.
- [46] Barker DJ, Godfrey KM, Fall C, Osmond C, Winter PD, Shaheen SO. Relation of birth weight and childhood respiratory infection to adult lung function and death from chronic obstructive airways disease. BMJ 1991;303:671–5.
- [47] Stein CE, Kumaran K, Fall CH, Shaheen SO, Osmond C, Barker DJ. Relation of fetal growth to adult lung function in south India. Thorax 1997;52:895–9.
- [48] Stick S. Pediatric origins of adult lung disease. 1. The contribution of airway development to paediatric and adult lung disease. Thorax 2000;55:587–94.
- [49] Taylor B, Wadsworth J. Maternal smoking during pregnancy and lower respiratory tract illness in early life. Arch Dis Child 1987;62:786–91.
- [50] Vik T, Vatten L, Markestad T, Ahlsten G, Jacobsen G, Bakketeig LS. Morbidity during the first year of life in small for gestational age infants. Arch Dis Child Fetal Neonatal Ed 1996;75:F33–7.
- [51] Ahmed S, Mahabbate Khoda S, Rekha RS, Gardner RM, Ameer SS, Moore S, et al. Arsenic-associated oxidative stress, inflammation, and immune disruption in human placenta and cord blood. Environ Health Perspect 2011;119:258–64.
- [52] Raqib R, Ahmed S, Sultana R, Wagatsuma Y, Mondal D, Hoque AW, et al. Effects of in utero arsenic exposure on child immunity and morbidity in rural Bangladesh. Toxicol Lett 2009;185:197–202.
- [53] Farzan SF, Korrick S, Li Z, Enelow R, Gandolfi AJ, Madan J, et al. In utero arsenic exposure and infant infection in a United States cohort: a prospective study. Environ Res 2013;126:24–30.
- [54] King P, Holdsworth S, Freezer N, Holmes P. Bronchiectasis. Intern Med J 2006;36:729-37.
- [55] Ferreccio C, Sancha AM. Arsenic exposure and its impact on health in Chile. J Health Popul Nutr 2006;24:164–75.

- [56] Borgono JM, Greiber R. Epidemiologic study of arsenic poisoning in the city of Antofagasta [in Spanish]. Rev Med Chil 1971;99:702–7.
- [57] Rosenberg HG. Systemic arterial disease and chronic arsenicism in infants. Arch Pathol 1974;97:360-5.
- [58] Borgono J, Vicent P, Venturino HN, Infante A. Arsenic in the drinking water of the city of Antofagasta: epidemiological and clinical study before and after the installation of a treatment plant. Environ Health Perspect 1977;19:103–5.
- [59] Zaldivar R, Ghai G. Clinical epidemiological studies on endemic chronic arsenic poisoning in children and adults, including observations on children with high- and low-intake of dietary arsenic. Zentralbl Bakteriol B 1980;170:409–21.
- [60] Zaldivar R. A morbid condition involving cardio-vascular, bronchopulmonary, digestive and neural lesions in children and young adults after dietary arsenic exposure. Zentralbl Bakteriol B 1980;170:44–56.
- [61] Smith AH, Yunus M, Khan AF, Ercumen A, Yuan Y, Smith MH, et al. Chronic respiratory symptoms in children following in utero and early life exposure to arsenic in drinking water in Bangladesh. Int J Epidemiol 2013;42:1077–86.
- [62] DiFranza JR, Aligne CA, Weitzman M. Prenatal and postnatal environmental tobacco smoke exposure and children's health. Pediatrics 2004;113:1007–15.
- [63] Dauphiné D, Ferreccio C, Guntur S, Yuan Y, Hammond S, Balmes J, et al. Lung function in adults following in utero and childhood exposure to arsenic in drinking water: preliminary findings. Int Arch Occup Environ Health 2011;84:591–600.
- [64] Chen Y, Parvez F, Gamble M, Islam T, Ahmed A, Argos M, et al. Arsenic exposure at low-to-moderate levels and skin lesions, arsenic metabolism, neurological functions, and biomarkers for respiratory and cardiovascular diseases: review of recent findings from the Health Effects of Arsenic Longitudinal Study (HEALS) in Bangladesh. Toxicol Appl Pharmacol 2009;239:184–92.
- [65] Ramsey K, Foong R, Zosky G. Emerging early life environmental exposures and lung development. J Immunotoxicol 2013;1:118–27.
- [66] Hanlon DP, Ferm VH. Placental permeability of arsenate ion during early embryogenesis in the hamster. Experientia 1977;33:1221–2.
- [67] Lindgren A, Danielsson BR, Dencker L, Vahter M. Embryotoxicity of arsenite and arsenate: distribution in pregnant mice and monkeys and effects on embryonic cells in vitro. Acta Pharmacol Toxicol (Copenh) 1984;54:311–20.
- [68] Willhite CC, Ferm VH. Prenatal and developmental toxicology of arsenicals. Adv Exp Med Biol 1984;177:205–28.
- [69] Devesa V, Adair BM, Liu J, Waalkes MP, Diwan BA, Styblo M, et al. Arsenicals in maternal and fetal mouse tissues after gestational exposure to arsenite. Toxicology 2006;224:147–55.
- [70] Kozul-Horvath CD, Zandbergen F, Jackson BP, Enelow RI, Hamilton JW. Effects of low-dose drinking water arsenic on mouse fetal and postnatal growth and development. PLOS ONE 2012;7:e38249.
- [71] Kozul CD, Hampton TH, Davey JC, Gosse JA, Nomikos AP, Eisenhauer PL, et al. chronic exposure to arsenic in the drinking water alters the expression of immune response genes in mouse lung. Environ Health Perspect 2009;117:1108–15.
- [72] Andrew AS, Bernardo V, Warnke LA, Davey JC, Hampton T, Mason RA, et al. Exposure to arsenic at levels found in U.S. drinking water modifies expression in the mouse lung. Toxicol Sci 2007;100:75–87.
- [73] Corsini E, Asti L, Viviani B, Marinovich M, Galli CL. Sodium arsenate induces overproduction of interleukin-1alpha in murine keratinocytes: role of mitochondria. J Invest Dermatol 1999;113:760–5.
- [74] Sikorski EE, McCay JA, White Jr. KL, Bradley SG, Munson AE. Immunotoxicity of the semiconductor gallium arsenide in female B6C3F1 mice. Fundam Appl Toxicol 1989;13:843–58.

- [75] Soto-Pena GA, Vega L. Arsenic interferes with the signaling transduction pathway of T cell receptor activation by increasing basal and induced phosphorylation of Lck and Fyn in spleen cells. Toxicol Appl Pharmacol 2008;230:216–26.
- [76] Vega L, Styblo M, Patterson R, Cullen W, Wang C, Germolec D. Differential effects of trivalent and pentavalent arsenicals on cell proliferation and cytokine secretion in normal human epidermal keratinocytes. Toxicol Appl Pharmacol 2001;172:225–32.
- [77] Lantz RC, Parliman G, Chen GJ, Carter DE. Effect of arsenic exposure on alveolar macrophage function. I. Effect of soluble as(III) and as(V). Environ Res 1994;67:183–95.
- [78] Burns LA, Sikorski EE, Saady JJ, Munson AE. Evidence for arsenic as the immunosuppressive component of gallium arsenide. Toxicol Appl Pharmacol 1991;110:157–69.
- [79] Aranyi C, Bradof JN, O'Shea WJ, Graham JA, Miller F. Effects of arsenic trioxide inhalation exposure on pulmonary antibacterial defenses in mice. J Toxicol Environ Health 1985;15:163–72.
- [80] Nayak AS, Lage CR, Kim CH. Effects of low concentrations of arsenic on the innate immune system of the zebrafish (*Danio rerio*). Toxicol Sci 2007;98:118–24.
- [81] Kozul CD, Ely KH, Enelow RI, Hamilton JW. Low-dose arsenic compromises the immune response to influenza A infection *in vivo*. Environ Health Perspect 2009;117:1441–7.
- [82] Ramsey KA, Foong RE, Sly PD, Larcombe AN, Zosky GR. Early life arsenic exposure and acute and longterm responses to influenza A infection in mice. Environ Health Perspect 2013;121:1187–93.
- [83] Lemarie A, Morzadec C, Bourdonnay E, Fardel O, Vernhet L. Human macrophages constitute targets for immunotoxic inorganic arsenic. J Immunol 2006;177:3019–27.
- [84] Shaw JR, Bomberger JM, VanderHeide J, LaCasse T, Stanton S, Coutermarsh B, et al. Arsenic inhibits SGK1 activation of CFTR Cl⁻ channels in the gill of killifish, *Fundulus heteroclitus*. Aquat Toxicol 2010;98:157–64.
- [85] Shaw JR, Gabor K, Hand E, Lankowski A, Durant L, Thibodeau R, et al. Role of glucocorticoid receptor in acclimation of killifish (*Fundulus heteroclitus*) to seawater and effects of arsenic. Am J Physiol Regul Integr Comp Physiol 2007;292:R1052–60.
- [86] Bomberger JM, Coutermarsh BA, Barnaby RL, Stanton BA. Arsenic promotes ubiquitinylation and lysosomal degradation of cystic fibrosis transmembrane conductance regulator (CFTR) chloride channels in human airway epithelial cells. J Biol Chem 2012;287:17130–9.
- [87] Tarran R, Button B, Boucher RC. Regulation of normal and cystic fibrosis airway surface liquid volume by phasic shear stress. Annu Rev Physiol 2006;68:543–61.
- [88] Waalkes MP, Ward JM, Diwan BA. Induction of tumors of the liver, lung, ovary and adrenal in adult mice after brief maternal gestational exposure to inorganic arsenic: promotional effects of postnatal phorbol ester exposure on hepatic and pulmonary, but not dermal cancers. Carcinogenesis 2004;25:133–41.
- [89] Waalkes MP, Ward JM, Liu J, Diwan BA. Transplacental carcinogenicity of inorganic arsenic in the drinking water: induction of hepatic, ovarian, pulmonary, and adrenal tumors in mice. Toxicol Appl Pharmacol 2003;186:7–17.
- [90] Petrick J, Blachere F, Selmin O, Lantz R. Inorganic arsenic as a developmental toxicant: *in utero* exposure and alterations in the developing rat lungs. Mol Nutr Food Res 2009;53:583–91.
- [91] Lantz RC, Chau B, Sarihan P, Witten ML, Pivniouk VI, Chen GJ. In utero and postnatal exposure to arsenic alters pulmonary structure and function. Toxicol Appl Pharmacol 2009;235:105–13.
- [92] Ramsey K, Bosco A, McKenna K, Carter K, Elliot J, Berry L, et al. *In utero* exposure to arsenic alters lung development and genes related to immune and mucociliary function in mice. Environ Health Perspect 2013;121:244–50.
- [93] Ramsey K, Larcombe A, Sly P, Zosky G. *In utero* exposure to low dose arsenic via drinking water impairs early life lung mechanics in mice. BMC Pharm Tox 2013;14:13.

This page intentionally left blank

14

Arsenical Kidney Toxicity

Bruce A. Fowler

EMORY UNIVERSITY ROLLINS SCHOOL OF PUBLIC HEALTH, ATLANTA, GA, USA; UNIVERSITY OF ALASKA-FAIRBANKS, FAIRBANKS, AK, USA

CHAPTER OUTLINE

14.1 I	Introduction	349
14.2 (Clinical Manifestations of Arsenical Toxicity in Humans	350
14.3 ľ	Nephrotoxic Arsenical Compounds	351
14.4 ľ	Mechanisms of Arsenical Toxicity	352
14.5 E	Biomarkers of Nephrotoxicity	354
14.6	Arsenical Interactions with Other Nephrotoxic Elements	356
14.7 9	Summary and Future Research Needs	357
Refere	ences	357

14.1 Introduction

Arsenic, which is atomic number 33 in the Periodic Table of Elements, may exist in the -3, +3, and +5 oxidation states, all of which have been shown to produce cellular toxicity to varying degrees. Humans may be exposed to arsenicals from air, food, and water under a variety of environmental, occupational, and medical therapy conditions [1,2]. Arsenic is metabolized into several methylated species *in vivo*, which also vary in their toxic potential. There is genetic variation in the enzymes responsible for the methylation of arsenicals *in vivo* [2]. Urinary excretion is the major route for elimination of arsenicals from the body and the kidney concentrates these during this process. Hence, this physiological concentration process likely plays a major role in making the kidney a major target organ for arsenical toxicity. The present chapter will focus on the renal handling of the various arsenical species, target kidney cell populations, and the known mechanisms of arsenical cellular toxicity. A general summary of reported clinical manifestations of toxicity produced by exposure to arsenicals is presented in Table 14–1.

Acute Renal Failure	Arsenical	Reference #	# Measured Endpoint(s)	
	Arsine gas	[1-4]	Increased BUN, hemoglobinuria	
Chronic renal damage				
	Arsenate, arsenite	[1,5,6]	Proteinuria, glucosuria	
Renal cancer				
	Arsenate, Arsenite	[7]	Cancer cells	

Table 14–1Reported Clinical Manifestations of Arsenical Nephrotoxicity fromAcute or Chronic Arsenical Exposures

14.2 Clinical Manifestations of Arsenical Toxicity in Humans

14.2.1 Epidemiological Studies

In recent years, there have been a number of studies of human populations exposed to arsenic in drinking water in India [8], Taiwan [9], Bangladesh [10], Chile [11–13], and Argentina [14] as a result of geological contamination of water passing through arsenic-containing rock formations. The clinical manifestations of such arsenic exposures on the kidney are consistent across studies and may involve effects on the blood vasculature [5,6,10,15] leading to axotemia and/ or tubular toxicity, and in some studies renal carcinogenesis. Arsenic-induced diabetes [16,17] will further exacerbate the effects of this element on the kidney. The overall conclusion to be derived is that arsenic is a broad-spectrum nephrotoxin which is also capable of playing a role in the development of renal cancer in some individuals [7]. The chapter sections that follow will attempt to provide more specific information on the mechanisms underlying these clinical outcome measures and suggest ways in which this mechanistic information may be used to develop biomarkers for early detection of arsenical-induced nephrotoxicity or carcinogenicity prior to the onset of clinical disease.

14.2.2 Mechanisms of Arsenical Toxicity to Specific Renal Cell Populations

14.2.2.1 Renal Blood Vasculature

As noted above, the kidney is a highly vascular organ, which has a number of arteries, arteriololes, capillaries, and veins lined with endothelial cells. Recent studies [18] have demonstrated that endothelial cells and renal podocytes [19] are highly sensitive to arsenicals ostensibly as a result of arsenical inhibition of cellular respiration with attendant increases in formation of reactive oxygen species (ROS). ROS are highly toxic [20,21] and the endothelial cells are most likely highly sensitive as a result of a limited capacity to produce antioxidants such as glutathione. Endothelial cell death from arsenical exposures will lead to inflammation followed by scarring via a complex set of processes [22]. It should be noted that arsenical disruptions of cellular signaling pathways [23–28] have been demonstrated to lead to a number of effects related to altered regulation of normal cell death and replacement processes [29].

14.2.2.2 Proximal Tubules

The renal proximal tubules are a major metabolic component of the kidney, which is responsible for reabsorption of proteins, glucose, and electrolytes from the urinary filtrate [30]. Renal proximal convoluted tubule cells (PCTCs) have high energy requirements in the form of ATP in order to meet their metabolic functions for the kidney and for this reason contain many mitochondria [31] that are highly sensitive to arsenical inhibition of respiration [32]. The high metabolic activity of the renal PCTCs coupled with their high mitochondrial content combine to make these cells and this portion of the renal nephrons major targets for arsenical toxicity.

14.3 Nephrotoxic Arsenical Compounds

As noted above, arsenicals may exist in a number of oxidation states and as both inorganic and organic chemical species, which may undergo a number of metabolic interconversions *in vivo*. All of these chemical species of arsenic may exert nephrotoxic effects if given at sufficient dose levels or levels of exposure. A more detailed summary discussion of arsenical metabolism and the putative importance of methylated arsenical species in arsenic-induced toxicity and cancer has recently been presented elsewhere [2]. The following sections will focus on the nephrotoxicity of the major forms of inorganic arsenicals since these are of greatest public health interest.

14.3.1 Arsine

Arsine (AsH_3) exists as a gas at room temperature and has been known for many years as a potent hemolytic agent [1–4], which produces renal failure following acute high dose exposure. The mechanisms underlying these effects involve occlusion of the renal tubules by hemoglobin with resulting loss of urinary output. In addition, the direct action of arsine on respiration of renal tissue slices has also been reported [33] suggesting a more complex mechanism of nephrotoxicity.

14.3.2 Arsenate

Arsenate (As^{5+}) is a common form of arsenic found in water supplies, and is generally less acutely toxic than arsenite (As^{3+}) , which is the other common oxidation state discussed below. It is worth noting that the relative concentrations of these two oxidation states may vary depending on local redox conditions and hence it is not uncommon for both forms to be reported as present in water systems [34]. Both forms may also undergo biological methylation reactions *in vivo* as a result of tissue methylation processes or of microbial actions by the microbiome in the gastrointestinal tract following ingestion [35].

14.3.3 Arsenite

Arsenite (As^{3+}) is another commonly encountered form of inorganic arsenic, and is more acutely toxic than arsenate [1,2] ostensibly as a result of complexation with the dihydrolipoic acid cofactor required for mitochondrial metabolism of NAD-linked substrate-supported mitochondrial respiration. Studies in recent years have demonstrated that the mechanisms underlying arsenical inhibition of mitochondrial respiration are complicated by *in vivo* interconversions between oxidation states and methylation reactions, as discussed below.

14.4 Mechanisms of Arsenical Toxicity

Arsenical exposures produce a broad spectrum of clinical renal effects that may result from either direct toxic and carcinogenic effects on specific cell types of this multifunctional organ system or related secondary effects such as diabetes that also have negative consequences for the kidney. The following sections will examine some of the known molecular mechanisms of arsenical nephrotoxicity to specific biological pathways/organelle systems and possible roles these perturbations may play in development of renal cancer and interactions with diabetes.

14.4.1 Mitochondrial Effects

14.4.1.1 Mitochondrial Respiratory Function

The inhibitory effects of arsenicals on cellular respiration have been known for many decades [1,2]. The mitochondrion plays a major role in cellular respiration, and arsenical exposures have been shown to produce concomitant alterations in mitochondrial respiration and high-amplitude swelling of renal mitochondria *in situ* [31]. Other studies [36,37] have shown that this swelling is related to inhibition of energy-linked structural transformations in this organelle, which are essential to mitochondrial respiration and may be disrupted by arsenical exposure [38]. These data suggest that arsenical inhibition of mitochondrial respiratory function is a more complex process than originally thought, as noted above, with regard to arsenicals binding to the vicinal thiols of the dihydrolipoic acid cofactor of the pyruvate dehydrogenase complex. In addition, it should be noted that inhibition of mitochondrial respiration will also produce ROS, which play a central role in a number of toxic processes [39] such as altered cell signaling pathways, apoptosis, and proteotoxicity from oxidation of labile protein functional (e.g., SH) groups. The bottom line is that these interrelated effects may act in concert to exacerbate arsenical nephrotoxicity at the level of cellular respiration.

14.4.1.2 Heme Biosynthetic Pathway

Arsenical exposure has also been demonstrated to produce a characteristic porphyrinuria with greater concentrations of uroporphyrins relative to coproporphyrins in both animals [40-42] and humans [43] exposed to inorganic arsenic in drinking water. This uroporphyrindominated porphyrinuria pattern has also been observed in experimental animals exposed to arsine gas via inhalation [44]. The value of this information rests with the potential of this porphyrinuria pattern as a putative biomarker of arsenical toxicity since porphyrins are also toxic molecules capable of catalyzing formation of ROS, thus exacerbating arsenic-induced oxidative stress in target organs such as the kidney. The elimination of these toxic compounds via the urine indicates that exposure of kidney cells must occur and suggests that arsenicalinduced porphyrinuria may be an exacerbating factor in mediating arsenical nephrotoxicity.

14.4.1.3 Arsenic and Renal Cancer

As noted above, some epidemiological studies of human populations exposed to arsenic in drinking water have reported mild increases in renal cancer [7]. Such findings are reasonable given the increased formation of ROS discussed above with the attendant genetic and epigenetic cellular responses known to result from increased ROS formation. The relatively modest magnitude of renal cancer outcomes is most likely related to the robust nature of renal tubule cells with regard to antioxidant defense systems such as reduced glutathione [45], and induction of glutathione synthetase [46], superoxide dismutase [47], and stress proteins [48–50] such as heme oxygenase (HSP32). The combined impact of these antioxidant defense systems would be to attenuate the carcinogenic response in the kidney unless other factors such as genetic predisposition factors and nutritional deficiencies [39,45] are also operating. The bottom line is that the multiple factors are most likely operating to delineate subpopulations at special risk for both renal toxicity and carcinogenic outcomes.

14.4.1.4 Membrane Transport Systems

In addition to porphyrinurias, arsenical exposures in humans has been shown to produce alterations in the renal handling of a number of substances including glucose [51], phosphate [52], amino acids, and low molecular weight proteins [53], which are normally reabsorbed by membrane transport proteins and/or energy-dependent pinocytosis. Arsenate, which is chemically similar to phosphate, has been shown to be actively taken up by renal tubules via a phosphate transport system [54]. The overall results of these studies indicate that arsenic may accumulate in the kidney via an active transport system and also interfere with the membrane transport of a number of other metabolically important substances by competitive inhibition. Arsenical-induced decrements in mitochondrial energy production could compromise ATP-requiring transporters of a number of metabolically important substances. In either case, the cellular membrane of renal tubule cells should also be regarded as a target organelle system for arsenical toxicity.

14.4.1.5 Altered Cellular Signaling Pathways

As noted above, arsenical-induced ROS will have an impact on normal regulation of a number of cellular processes in the kidney and other organs via alteration of cellular signaling pathways that regulate a variety of renal processes including transport of metabolic products, cell death, and replacement processes including initiation of renal cancer. Table 14–2 is intended to summarize some of the reported effects of arsenicals on a number of these cellular signaling pathways in renal cells following either *in vivo* or *in vitro* exposures.

Pathway	Exposure Route	Impact	Reference #
Apoptosis	In vitro	+	[19,23,25,28]
Cell replication	In vitro	-	[24]
Glucose transport	In vitro	+ or –	[55]

Table 14–2Effects of Arsenical Exposures on CellularSignaling Pathways in the Kidney

14.5 Biomarkers of Nephrotoxicity

The field of biomarkers has rapidly expanded over the past 30 years and biomarkers related to chemical-induced renal toxicity are particularly prominent due to the release of putative biomarker substances into the urine and the reserve capacity of the kidney, which limits the ability to detect early ongoing signs of toxicity until the kidney is approaching overt renal failure and end stage renal disease (ESRD). Epidemiological studies of human populations exposed to arsenic in drinking water have demonstrated elevated rates of renal damage [56,57] and cancer [58], so development of renal biomarkers for early detection of arsenical-induced renal damage is of particular importance. The following section is intended to review some of the more current renal biomarkers for arsenic-induced toxicity and suggests needed areas of future research.

14.5.1 Omic Biomarkers

In recent years, the need for sensitive and specific biomarkers for delineating early cellular manifestations of arsenic toxicity has encompassed the field of "omics," which includes genomics, proteomics, and metabolomics/metabonomics, which are measurable cellular responses to toxic cell injury. In the case of arsenic, these responses are on an interconnected biological continuum, and the oxidative stresses produced by arsenic exposure described above are a common generic trigger for a number of specific "omic" responses. As also noted above, the kidney is a primary target for arsenical-induced toxicity since arsenic is concentrated by this organ and urinary excretion of arsenical metabolites is the major route of elimination for this element from the body. This section will review some of the published works on arsenic-induced alterations of omic biomarkers with particular emphasis on the kidney.

14.5.2 Altered Gene Expression Patterns (Genomics)

There are a number of arsenic-induced alterations in renal gene expression patterns that have been reported *in vivo* in intact animals [41] and *in vitro* cell systems [48–50]. Generally, these genomic responses usually demonstrate both up- and down-regulation of a number of genes. The extent to which this occurs is mediated by a number of factors including dose, duration of exposure, gender, and test system being evaluated [59]. Not all up-regulated genes are actually expressed. The mitigating effects of epigenetic responses and interconnecting signaling pathways will also influence measurable outcomes.

14.5.3 Altered Protein Synthesis Patterns (Proteomics)

Proteomic responses to arsenic exposure have the advantage that one is measuring an actual expressed gene product produced in response to arsenical exposure. The question of the duration of expressed proteomic responses *in vivo* is an important issue with regard to interpretation of biological significance since the degree and nature of the response will vary with time. Given the complexity of proteomic responses to arsenical compound exposures, computerized analysis of digitized 2D gel images [59] is essential to interpretation of generated data. For example, the intensity of proteomic responses of hamster renal tubule cells following acute *in vivo* exposures to gallium arsenide or indium arsenide particles at 10 days or 30 days post-exposure (Figure 14–1) have been shown to be both time and gender dependent [59].

14.5.4 Metabolomics

As noted above, arsenical exposure produces alterations to a number of metabolic pathways including those in the kidney [31]. These include those involved in intermediary metabolism [60] and heme biosynthesis [40–43].



FIGURE 14–1 A standard digitized image of a 2D gel-³⁵S methioine autoradiogram of kidney tubule cells from a male hamster treated with GaAs particles on day 10 post-exposure. The image shows computer assigned numbers for each identified protein spot on the gel. The computer then ratios the density of the spots to those from control hamster kidney cells loaded with an equal amount of ³⁵S radioactivity. See [58] for details and further discussion.

14.5.5 Arsenic-Induced Posttranslational Alterations of Proteins

Arsenic-induced alterations in posttranslational modifications of histones have been reported in both humans [61] and experimental systems [62,63] exposed to inorganic arsenic. These findings have potentially important risk assessment implications for both renal toxicity and carcinogenesis in human populations exposed to arsenicals.

14.5.6 Altered Heme Biosynthesis/Porphyrinuria Patterns

Arsenic exposure also results in inhibition of enzymes of the heme biosynthetic pathway resulting in a characteristic porphyrinuria dominated by uroporphyrins with lesser amounts of coproporphyin [40–43] and induction of heme oxygenase (HSP32), which is the rate limiting enzyme in the heme degradative pathway [59]. These metabolic disturbances in this essential biological pathway have been used successfully as biomarker endpoints for studies in both experimental animals and humans exposed to arsenic in drinking water [40,43].

14.5.7 Proteinuria Patterns

Increased excretion of proteins into the urine has also been reported in humans [56,57] following chronic exposure to arsenic in drinking water. Depending on the dose level, the proteins may be either low molecular weight at lower dose levels or both low and higher molecular weight at higher dose levels [2,56,57].

14.6 Arsenical Interactions with Other Nephrotoxic Elements

It is also common for arsenic exposures to occur in combination with other nephrotoxic metals with resultant additive or synergistic increases in toxicity. The most common co-exposures occur with lead and cadmium under environmental conditions, and gallium and indium in the semiconductor industry. Arsine gas is used in the fabrication of gallium arsenide and indium arsenide chips as the source of arsenic. The following short review is intended as an overview of the published literature on these co-exposures and known mechanisms of renal cell toxicity stemming from interactions among these elements.

14.6.1 Lead

Combined exposures to lead and arsenic are common in both occupational and environmental situations [64,65]. Since both these elements are nephrotoxic, additive or synergistic manifestations of renal cell injury are frequently reported depending on dose or exposure level. Toxic effects on renal tubule cells from combined lead and arsenic exposures have been reported following both moderate and low level dosing of experimental animals *in vivo* [66,67].

14.6.2 Cadmium

Nephrotoxic effects of combined cadmium arsenic exposures have been reported in human populations [68], experimental animals [66,67], and *in vitro* exposure situations [50]. As with lead, both additive and synergistic toxicities may be observed depending on dose or exposure level [66,67].

14.6.3 Gallium

Interaction studies between gallium and arsenic stem from the use of both elements in the semiconductor industry for the production of gallium arsenide from computer chips and light emitting diodes (LEDs). Worker exposures to arsenic may occur in the grinding and polishing of GaAs wafers and maintenance of fabrication equipment [69].

14.6.4 Indium

Indium arsenide (InAs) is a similar semiconductor to GaAs whose use has expanded due to the greater electronic speeds of InAs chips over those of GaAs chips. Both GaAs and InAs are toxic to the kidney but InAs appears to exert greater toxicity on kidney cells relative to GaAs equivalent exposure conditions [59].

14.7 Summary and Future Research Needs

Based on the above short review, it is clear that each of the various forms of arsenic discussed is capable of producing renal toxicity in both humans and experimental animals either on the basis of arsenic alone or in combination with other nephrotoxic metals. The basic toxicological issues stemming from these general observations serve as the basis for future research needs including, in particular, the need for sensitive and specific biomarkers of arsenical nephrotoxicity on an individual or multi-element mixture basis particularly at low dose exposures.

References

- [1] Agency for Toxic Substances and Disease Registry. Toxicological profile for arsenic (updated). Atlanta, GA: 2007. p. 41–228.
- [2] Fowler BA, Chou S-HS, Jones RL, Sullivan Jr. DW, Chen C-J. Arsenic. In: Nordberg GF, Fowler BA, Nordberg M, editors. Handbook on the toxicology of metals (4th ed.). Amsterdam: Elsevier Publishers; 2014. (In press.)
- [3] Fowler BA, Weissberg JB. Arsine poisoning. N Engl J Med 1974;291:1171-4.
- [4] Blair PC, Thompson MB, Morrissey RE, Moorman MP, Sloane RA, Fowler BA. Comparative toxicity of arsine gas in B6C3F mice, Fischer 344 rats, and Syrian golden hamsters: system organ studies and comparison of clinical indices of exposure. Fundam Appl Toxicol 1990;14:776–87.
- [5] Chen C-J. Arseniasis in the world: from endemic to pandemic. In: Chen C-J, Chiou H-Y, editors. Health hazards of environmental arsenic poisoning: from epidemic to pandemic. Singapore: World Scientific Publishing; 2011. pp. 1–20.

- [6] Chen C-J. Epidemic of arseniasis in the world. In: Chen C-J, Chiou H-Y, editors. Health Hazards of Environmental Arsenic Poisoning: From Epidemic to Pandemic. Singapore: World Scientific Publishing; 2011. pp. 33–49.
- [7] Chen C-J, Chen C-W, Wu M-M, Kuo TL. Cancer potential in liver, lung, bladder and kidney due to ingested inorganic arsenic in drinking water. Br J Cancer 1992;66:888–92.
- [8] Basu A, Mitra S, Chung J, Guha Mazumder DN, Ghosh N, Kalman D, et al. Creatinine, diet, micronutrients, and arsenic methylation in West Bengal, India. Environ Health Perspect 2011;119:1308–13.
- [9] Yang C-Y, Chiu H-F, Wu TN, Chang H-I, Ho S-C. Reduction in kidney cancer mortality following installation of a tap water supply system in an arsenic-endemic area of Taiwan. Arch Environ Health 2014;59:484–8.
- [10] Rahman MM, Chowdhury UK, Mukherjee SC, Mondal BK, Paul K, Lodh D, et al. Chronic arsenic toxicity in Bangladesh and West Bengal, India—a review and commentary. J Toxicol Clin Toxicol 2001;39:683–700.
- [11] Ferreccio C, Smith AH, Duran V, Balaro T, Benitez H, Valdes R, et al. Case-control study of arsenic in drinking water and kidney cancer in uniquely exposed northern Chile. Am J Epidemiol 2013;178:813–18.
- [12] Yuan Y, Marshall G, Ferreccio C, Steinmanus C, Liaw J, Bates M, et al. Kidney cancer mortality. Fifty year latency patterns related to arsenic exposure. Epidemiology 2010;21:103–8.
- [13] Smith AH, Marshall G, Liaw J, Yuan Y, Ferreccio C, Steinmaus C. Mortality in young adults following in utero and childhood exposure to arsenic in drinking water. Environ Health Perspect 2012;120:1527–31.
- [14] Hopenhayn-Rich C, Biggs ML, Smith AH. Lung and kidney cancer mortality associated with arsenic drinking water in Cordoba, Argentina. Int J Epidemiol 1998;27:561–659.
- [15] States JC, Srivastava S, Chen Y, Barchowsky A. Arsenic and cardiovascular disease. Toxicol Sci 2009;107:312–23.
- [16] Wang JP, Wang SL, Lin Q, Zhang L, Huang D, Ng JC. Association of arsenic and kidney dysfunction in people with diabetes and validation of its effects in rats. Environ Int 2009;35:507–11.
- [17] James KA, Marshall JA, Hokansson JE, Meliker JR, Zerbe GO, Byers TE. A case cohort study examining lifetime exposures to inorganic arsenic in drinking water and diabetes mellitus. Environ Res 2013;123:33–8.
- [18] Prozialeck WC, Edwards JR, Nebert DW, Woods JM, Barchowsky A, Atchison WD. The vascular system as a target of metal toxicity. Toxicol Sci 2008;102:207–18.
- [19] Eichler T, Ma Q, Kelly C, Mishra J, Parikh S, Ransom RF, et al. Single and combination toxic metal exposures induce apoptosis in cultured murine podocytes exclusively via the extrinsic caspase 8 pathway. Toxicol Sci 2006;90:392–9.
- [20] Jomova K, Jenisova Z, Feszterova M, Baros S, Liska J, Hudecova D, et al. Arsenic toxicity, oxidative stress and human disease. J Appl Toxicol 2011;31:95–107.
- [21] Flora SJS. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51:257–81.
- [22] Chen SC, Tsai MH, Wang HJ, Yu HS, Chang LW. Involvement of substance P and neurogenic inflammation in arsenic-induced early vascular dysfunction. Toxicol Sci 2007;95:82–8.
- [23] Bannerjee C, Singh A, Raman R, Mazumder S. Calmodulin-CaMKII mediated alteration of oxidative stress: interplay of the cAMP/PKA-ERK1/2-NFkB-NO axis on arsenic-induced head kidney macrophage apoptosis. Toxicol Res 2013;2:413–26.
- [24] Pournara A, Holmlund T, Lu Y, Ceder R, Grafstrom R, Vahter M, et al. Arsenic-induced suppression of kidney cell proliferation and transcriptional coregulator MAML1. Metabolomics 2014;6:498–504.
- [25] Lu Y, Yuan H, Deng S, Wei Q, Guo C, Yi J, et al. Arsanilic acid causes apoptosis and oxidative stress in rat kidney epithelial cells (NRK-52e cells) by activation of the caspase-9 and -3 signaling pathway. Drug Chem Toxicol 2014;37:55–62.
- [26] Roy A, Manna P, Sil PC. Prophylactic role NF-kB and mitochondrial dependent pathways. Free Radic Res 2009;43:995–1007.

- [27] Kimura A, Ishida Y, Hayashi T, Wada T, Tokoyama H, Sugaya T, et al. Interferon gamma plays protective roles in sodium arsenite-induced renal injury by up-regulating intrarenal multidrug resistance-associated protein1 expression. Am J Pathol 2006;169:1118–28.
- [28] Tokumoto M, Lee J-Y, Fujiwara Y, Uchiyama M, Satoh M. Inorganic arsenic induces apoptosis through downregulation of ube2d genes and p52 accumulation in rat proximal tubular cells. J Toxicol Sci 2013;38:815–20.
- [29] Catto JWF, Alcaraz AB, Jartel AS, White RDV, Evans CP, Fussel S, et al. MicroRNA in in prostate, bladder, and kidney cancer: a systematic review. Eur Urol 2011;59:671–81.
- [30] Sirac C, Bridoux F, Essig M, Devuyst O, Touchard G, Cogné M. Toward understanding renal Fanconi syndrome: step by step advances through experimental models. Contrib Nephrol 2011;169:247–61.
- [31] Brown MM, Rhyne BC, Goyer RA, Fowler BA. The intracellular effects of chronic arsenic administration on renal proximal tubule cells. J Toxicol Environ Health 1976;1:507–16.
- [32] Zamora PI, Rockenbauer A, Villamena FA. Radical model of arsenic (III) toxicity: theoretical and EPR spin trapping studies. Chem Res Toxicol 2014;27:765–74.
- [33] Huges W, Levvy GA. The study of arsine solutions for tissue slices. Biochem J 1947;41:8–11.
- [34] Song TJ, Chen AS, Wang L. Arsenic species in drinking water wells in the USA with high arsenic concentrations. Water Res 2014;48:156–69.
- [35] Lu K, Abo P, Schlieper KA, Graffam ME, Levine S, Wishnok JS, et al. Arsenic exposure perturbs the gut microbiome and its metabolic profile in mice: an integrated metagenomics and metabolomics analysis. Environ Health Perspect 2014;122:284–91.
- [36] Hackenbrock CR. Ultrastructural bases for metabolically linked mechanical activity in mitochondria. II. Electron transport-linked ultrastructural transformations in mitochondria. J Cell Biol 1968;37:345–69.
- [37] Hackenbrock CR. Ultrastructural bases for metabolically linked mechanical activity in mitochondria. I. Electron transport-linked ultrastructural transformations in mitochondria. J Cell Biol 1966;30:269–97.
- [38] Fowler BA, Woods JS, Schiller CM. Studies of hepatic mitochondrial structure and function: morphometric and biochemical evaluation of in vivo perturbation by arsenate. Lab Invest 1979;41:313–20.
- [39] Rizwan S, Naqshbandi A, Farooqui Z, Khan AA, Khan F. Protective effect of dietary flaxseed oil on arsenicinduced nephrotoxicity and oxidative damage in rat kidney. Food Chem Toxicol 2014;68:99–107.
- [40] Woods JS, Fowler BA. Altered regulation of mammalian hepatic heme biosynthesis and urinary porphyrin excretion during prolonged exposure to sodium arsenate. Toxicol Appl Pharmacol 1978;43:361–71.
- [41] Conner EA, Yamauchi H, Fowler BA, Akkerman M. Biological indicators for monitoring exposure/toxicity from III-V semiconductors. J Expo Anal Environ Epidemiol 1993;3:431–40.
- [42] Conner EA, Yamauchi H, Fowler BA. Alterations in the heme biosynthetic pathway from III-V semiconductor metal indium arsenide (InAs). Chem Biol Interact 1995;96:273–85.
- [43] Garcia-Vargas GG, Del Razo LM, Cebrián ME, Albores A, Ostrosky-Wegman P, Montero R, et al. Altered urinary porphyrin excretion in a human population chronically exposed to arsenic in Mexico. Hum Exp Toxicol 1994;13:839–47.
- [44] Fowler BA, Moorman MP, Adkins B, Bakewell WE, Blair PC, Thompson MB. Arsine: toxicity data from acute and short-term inhalation exposures. Proceedings of Conference on Hazard Assessment and Control Technology in Semiconductor Manufacturing. Am Conf Govt Indus Hygiene. Chelsea, MI: Lewis Publishers; 1989. pp. 85–93.
- [45] Maiti S, Chatterjee AK. Effects on levels of glutathione and some related enzymes in tissues after an acute arsenic exposure in rats and their relationship to dietary protein deficiency. Arch Toxicol 2001;75:531–7.
- [46] Shadid F, Rizwan S, Khan MW, Khan SA, Naqshbandi A, Yusufi AN. Studies on the effect of sodium arsenate on the enzymes of carbohydrate metabolism, brush border membrane, and oxidative stress in the rat kidney. Environ Toxicol Pharmacol 2014;37:592–9.

- [47] Mershiba SD, Dassprakash MV, Saraswathy SD. Protective effect of naringenin on hepatic and renal dysfunction and oxidative stress in arsenic intoxicated rats. Mol Biol Rep 2013;40:368–91.
- [48] Aoki Y, Lipsky MM, Fowler BA. Alteration of protein synthesis in primary cultures of rat kidney epithelial cells by exposure to gallium indium and arsenite. Toxicol Appl Pharmacol 1990;106:462–8.
- [49] Del Razo LM, Quintanilla-Vega B, Brambila-Colombres E, Calderón-Aranda ES, Manno M, Albores A. Stress proteins inducted by arsenic. Toxicol Appl Pharmacol 2001;177:132–48.
- [50] Madden EF, Akkerman M, Fowler BA. A comparison of 60 kDa,70 kDa and 90 kDa stress protein expression in normal rat NRK-52 and human HK-2 kidney cell lines following in vitro exposure to arsenite and cadmium alone or in combination. J Biochem Mol Toxicol 2002;16:24–32.
- [51] Brazy PC, Balaban RS, Gullans SR, Mandel JJ, Dennis VW. Inhibition of renal metabolism. Relative effects of arsenate on sodium, phosphate, and glucose transport by the rabbit proximal tubule. J Clin Invest 1980;66:1211–21.
- [52] Jobst-Schwan T, Knaup KX, Nielsen R, Hackenbeck T, Buettner-Herold M, Lechler P, et al. Renal uptake of the antiapoptotic protein survivin is mediated by megalin at the apical membrane of the proximal tubule. Am J Physiol Renal Physiol 2013;305:F734–44.
- [53] Robles-Osorlo ML, Pérez-Maldonado IN, Martín del Campo D, Montero-Perea D, Avilés-Romo I, Sabath-Silva E, et al. Urinary arsenic levels and risk of renal injury in a cross-sectional study in open population. Rev Invest Clin 2012;64:609–14.
- [54] Ginsburg JM, Lotspeich WD. Interrelations of arsenate and phosphate transport in the dog kidney. Am J Physiol 1963;205:707–14.
- [55] Tseng CH. The potential biological mechanisms of arsenic-induced diabetes mellitus. Toxicol Appl Pharmacol 2004;197:67–83.
- [56] Chen JW, Chen HY, Li WF, Liou SH, Chen CJ, Wu JH, et al. The association between total urinary arsenic concentration and renal dysfunction in a community-based population from central Taiwan. Chemosphere 2011;84:17–24.
- [57] Feng H, Gao Y, Zhao L, Wei Y, Li Y, Wei W, et al. Biomarkers of renal toxicity caused by exposure to arsenic in drinking water. Environ Toxicol Pharmacol 2013;35:495–501.
- [58] Huang C-Y, Chu JS, Pu YS, Yang HY, Wu CC, Chung CJ, et al. Effect of urinary total arsenic level and estimated glomerular filtration rate on risk of renal cell carcinoma in a low arsenic exposure area. J Urol 2011;185:2040–4.
- [59] Fowler BA, Conner EA, Yamauchi H. Metabolomic and proteomic biomarkers for III-V semiconductors: chemical-specific porphyrinurias and proteinurias. Toxicol Appl Pharmacol 2005;206:121–30.
- [60] Chen B, Burt CT, Goering PL, Fowler BA, London RE. In vivo 31P nuclear magnetic resonance studies of arsenite-induced changes in hepatic phosphate levels. Biochem Biophys Res Commun 1986;139:228–34.
- [61] Chervona Y, Hall MN, Arita A, Wu F, Sun H, Tseng HC, et al. Associations between arsenic exposure and global posttranslational histone modifications among adults in Bangladesh. Cancer Epidemiol Biomarkers Prev 2012;21:2252–60.
- [62] Chu F, Ren X, Chasse A, Hickman T, Zhang L, Yuh J, et al. Quantitative mass spectrometry reveals the epigenome as a target of arsenic. Chem Biol 2011;192:113–7.
- [63] Chervona Y, Costa M. The control of histone methylation and gene expression by oxidative stress, hypoxia, and metals. Free Radic Biol Med 2012;53:1041–7.
- [64] Bacigalupo C, Hale B. Human health risks of Pb and As exposure via consumption of home garden vegetables and incidental soil and dust ingestion: a probabilistic screening tool. Sci Total Environ 2012;423:27–38.
- [65] Coelho P, Costa S, Silva S, Walter A, Ranville J, Sousa AC, et al. Metal(loid) levels in biological matrices from human populations exposed to mining contamination—Panasqueira Mine (Portugal). J Toxicol Environ Health A 2012;75:893–908.

- [66] Mahaffey KR, Capar SG, Gladen BC, Fowler BA. Concurrent exposure to lead cadmium and arsenic: effects on toxicity and tissue metal concentrations in the rat. J Lab Clin Med 1981;98:463–81.
- [67] Whittaker M, Wang G, Chen X-Q, Lipsky M, Smith D, Gwiazda R, et al. Effects of trace element mixtures and the production of oxidative stress precursors: 30-day, 90-day, and 180-day drinking water studies in rats. Toxicol Appl Pharmacol 2011;254:154–66.
- [68] Huang ML, Choi SJ, Kim DW, Kim NY, Park CH, Yu SD, et al. Risk assessment of low-level cadmium and arsenic on the kidney. J Toxicol Environ Health A 2009;72:1493–8.
- [69] Fowler BA, Silbergeld EK. Occupational diseases—new workforces, new workplaces. Ann N Y Acad Sci 1989;572:46-60.

This page intentionally left blank

Arsenic-Induced Developmental Neurotoxicity

Jiaohua Luo and Weiqun Shu

15

DEPARTMENT OF ENVIRONMENTAL HYGIENE, COLLEGE OF PREVENTIVE MEDICINE, THIRD MILITARY MEDICAL UNIVERSITY, CHONGQING, PEOPLE'S REPUBLIC OF CHINA

CHAPTER OUTLINE

15.1 Introduction	363
15.2 Arsenic Exposure Impairs Intellectual Function in Children	364
15.3 Developmental Neurobehavioral Toxicity in Animals	367
15.4 Arsenic Distribution After Exposure in Early Life	368
15.5 Mechanism of Developmental Neurotoxicity	369
15.6 Neuroprotective Agents Against Arsenic Toxicity	376
15.7 Conclusions	377
References	380

15.1 Introduction

Arsenic toxicity has become a global health problem affecting millions of people. Aside from causing gastrointestinal, respiratory, cardiovascular, genitourinary, endocrine, hematopoietic system and skin diseases [1], acute and chronic arsenic exposure is associated with neurologic consequences in adults [2]. Arsenic is a known neurotoxicant that affects the peripheral nervous system, with effects which may last for several years or even a lifetime, causing a symmetrical peripheral neuropathy characterized by sensory nerves being more sensitive than motor nerves to arsenic effects, and neurons with large axons being more affected than neurons with short axons [3–5]. Acute arsenic exposure was reported to cause central nervous system (CNS) alterations [6–9], including learning, memory, and concentration.

When exposure occurs in the early developmental stages, the CNS is more susceptible to toxic agents. Children may be particularly susceptible to neurotoxic substances as suggested by findings from studies on the effects of lead [10], methylmercury [11], and PCBs [10]. Gestation in humans and rodents is a period of high sensitivity for offspring to chemical toxicity [12]. Recently, results of some epidemiological, experimental animal, and *in vitro* studies have suggested a possible association between arsenic exposure and neurodevelopment. Although neurotoxic effects of arsenic in adults have been well documented [2,13], there is no review addressing the effects of arsenic on neurodevelopment. The purpose of this review is to integrate the reports on possible interactions of arsenicals with neurodevelopment and to promote interest in exploring the mechanisms of arsenic developmental neurotoxicity.

15.2 Arsenic Exposure Impairs Intellectual Function in Children

Arsenic poisoning related to occupational exposure causes CNS alterations, including impairments of recent memory, learning, and concentration [14]. When exposure occurs in the early developmental stages, the CNS is more susceptible to toxic agents. Therefore, many studies were designed to investigate the possible association between arsenic exposure and neurodevelopment of children.

Calderon et al. [15] reported a cross-sectional study in Mexico to address the neurological effects on children of chronic environmental exposure to arsenic. In this study, two populations chronically exposed to either high (41 children) or low (39 children) levels of As and Pb were analyzed using the Wechsler Intelligence Scale for Children, Revised Version, for Mexico (WISC-RM). Mean urinary arsenic (UAs) values were 62.9 ± 0.03 (µg As/g creatinine) for the exposed group and 40.2 ± 0.03 (µg As/g creatinine) for the reference group. After controlling for significant potential variables, verbal intelligence quotient (IQ) decreased with increasing concentrations of urinary arsenic (UAs). Higher levels of UAs were significantly correlated to poorer performance on WISC-RM factors examining long-term memory and linguistic abstraction [15]. Another cross-sectional study compared 49 junior school students drinking arsenic-containing well water and 60 controls in Taiwan. The former was divided into two groups, exposed to arsenical levels of $184.99 \pm 225.29 \,\mu\text{g/L}$ (high dose group, n=29) and $131.19 \pm 343.70 \,\mu\text{g/L}$ (low dose group, n=20) over periods of 11.28 ± 2.58 and 8.10 ± 6.07 years, respectively. After adjusting for education and sex, pattern memory and switching attention were found to be significantly affected by long-term cumulative exposure to arsenic [16]. A study conducted in Shanyin County, China, involved 720 children between 8 and 12 years of age. The children were exposed to arsenic at the concentrations of $190 \pm 183 \,\mu\text{g/L}$ (high-As group), $142 \pm 106 \,\mu$ g/L (medium-As group), and $2 \pm 3 \,\mu$ g/L (low-As group, control). A modified Raven's test was used to determine the effects of these exposures on children's intelligence. The mean IQ scores for the high-As group $(95 \pm 17) (p < 0.01)$ and medium-As group (101 ± 16) (p < 0.05) were significantly lower than that of the control group (105 ± 15) [17]. A study among 351 children aged 5 to 15 years in West Bengal tried to address the possible impairment of children's intellectual function in relation to arsenic exposure *in utero* and during childhood. Authors found associations between arsenic and reductions in the adjusted scores of the vocabulary test, the object assembly test, and the picture completion test. However, exposure to arsenic during pregnancy or childhood was not associated with these test results. UAs, not water arsenic (WAs), were associated with small decrements in intellectual testing in schoolaged children in West Bengal [14].

Wasserman et al. conducted two separate cross-sectional investigations of intellectual function, in 201 10-year-olds and 301 6-year-olds in Bangladesh, respectively [18,19]. In the 10-year-olds study, drinking water As exposure was associated with reduced intellectual function after adjustment for sociodemographic covariates and water Mn. WAs was associated with reduced intellectual function in a dose-response manner, such that children exposed to WAs levels $>50 \mu g/L$ achieved significantly lower performance and full-scale scores than did children exposed to WAs levels $<5.5 \mu g/L$. The association was generally stronger for WAs than for UAs [18]. In the 6-year-olds study, with covariate adjustment, WAs remained significantly negatively associated with both performance and processing speed raw scores, but the associations were less strong than in 10-year-olds. Neither study found an association between As exposure and aspects of verbal intelligence [18,19]. It seems that children with less lengthy exposure to As may have diminished adverse consequences for intellectual function [19].

A consecutive investigation on the IQ of 408 children (two age groups: 9 and 10 years; 4 and 5 years) in Bangladesh was reported [20]. The level of UAs was associated with reduced IQ in a dose-response manner. The association between reduced IQ and UAs was stronger than that between reduced IQ and WAs. However, there was no association between verbal IQ scores and UAs of children in early childhood (aged 4 and 5 years). Based on these results, the authors concluded that UAs were negatively associated with the IQ of the children tested, and that this adverse effect of As may also gradually accumulate over time among the poor.

In their study, Rosado et al. [21] recruited 602 children 6–8 years old living within 3.5 km of a metallurgic smelter complex in Mexico. The mean level of UAs was $58.1 \pm 33.2 \,\mu$ g/L. After adjustment for hemoglobin concentration, blood lead concentration, and sociodemographic confounders, the results showed a significant inverse association between UAs and visualspatial abilities with figure design, the Peabody Picture Vocabulary Test, the WISC-RM Digit Span subscale, visual search, and letter sequencing tests (p < 0.05). Boys excreted significantly more UAs (p < 0.05) and were affected in different cognitive areas than were girls. Impairment of children's cognitive development caused by As is independent of any effect of lead [21]. McDermott et al. [22] conducted a retrospective cohort study, which involved 3988 mother/ child pairs who had lived during pregnancy and early childhood in South Carolina, to describe the complex association between soil concentrations of As combined with Pb and the probability of intellectual disability (ID) in children. The probability ID was increased with increasing concentrations of As and Pb in the soil. For normal weight for gestational age children, when As = 22 mg/kg and Pb = 200 mg/kg, the risk for ID was 11% and when As = 22 mg/kg and Pb = 400 mg/kg, the probability of ID was 65%. The probability of ID was significantly associated with the interaction between Pb and As for normal weight for gestational age infants [22].

Wright et al. [23] conducted a small (n=32) pilot study to explore the potential associations between hair metal levels and the neuropsychological function and behavior of 11–13-yearold children. The mean hair Mn and As levels were 471.5 and 17.8 ng/g. Children's general intelligence (particularly verbal IQ) scores and scores on assessment of verbal learning and memory were significantly negatively related to hair Mn and As levels. In some cases, a significant Mn-by-As interaction was found. No significant association between children's hair As level and parent or teacher ratings of children's behavior was found [23]. In another study, Wasserman et al. [24] recruited a new sample of 299 children (8–11 years old) to examine possible interactive effects of exposure to both As and Mn on neurobehavioral function. After adjustment for each other, both As and Mn (BAs, BMn) in whole blood were found to be significantly negatively related to most WISC-IV subscale scores. With further adjustment for sociodemographic features and ferritin, association between BAs and verbal comprehension was significantly negative. UAs (per gram creatinine) was significantly negatively associated with verbal comprehension scores, even after adjustment for BMn and other contributors. Mn by As interactions were not significant in either adjusted or unadjusted models [24].

The nervous system of an embryo is vulnerable to adverse impacts from pollutants because it goes through a long developmental process. This developmental process requires precise coordination of cell growth and movement, which may be disrupted by even shortterm exposures to environmental contaminants [25]. A study with 4436 pregnant women in Matlab, Bangladesh, was conducted to assess the effects of *in utero* arsenic exposure on infants' problem-solving ability and motor development. Urine specimens at 8 and 30 weeks of gestation were collected to measure arsenic concentration. Infants' problem-solving ability and motor development were assessed at 7 months of age. With covariate adjustment, there was no significant effect of urinary arsenic concentration on children's motor and PST scores and behavior ratings. It is possible that some other effects are as yet unmeasured or that effects will become apparent at a later age [26]. A hospital-based birth cohort study with 100 pregnant women in Nepal was conducted to investigate the association between in utero toxic (Pb and As) and essential element Zn levels and neurodevelopmental indicators after birth. Infants' neurodevelopmental indicators at 1 day after birth were tested using the Brazelton Neonatal Behavioral Assessment Scale, third edition (NBAS III). The cord blood levels of Pb and As, but not Zn, showed significant inverse association with the neurodevelopment of newborns [25].

Trying to find information on critical windows of arsenic exposure, Hamadani et al. [27] conducted a population-based longitudinal study, which recruited 1700 children at 5 years of age in rural Bangladesh. Arsenic exposure was measured by arsenic concentrations in mothers' urine collected in early (gestational week 8) and late (gestational week 30) pregnancy and in children's urine at 1.5 and 5 years. After controlling for all potential confounders and loss to follow-up, it was found that verbal IQ and full-scale IQ (FSIQ) were negatively associated with (log) UAs in girls. It was calculated that $100 \mu g/L$ UAs was associated with a decrement of 1–3 points in both verbal IQ and FSIQ in girls. However, UAs showed consistently low and non-significant associations with all IQ measures in boys [27].

To explore whether prenatal exposure to toxic heavy metals is associated with neonatal development, a stratified multi-stage cluster sampling methodology was utilized to investigate 1652 mother/infant pairs from 2008 through 2009 in Shanghai. The median concentrations of blood Pb, Hg, Cd, As, and Tl were all in the level considered safe. Increasing exposure to Cd, Hg, As, and Tl during pregnancy was associated with decreasing neonatal behavioral neuro-logical assessment (NBNA) scores. Those exposed to medium-level As (which is considered safe) had lower NBNA scores than those exposed to low-level As, which implied that the considered safe level of As was not safe to the newborns' development [28].

Some epidemiologic studies have focused on the possible associations between impaired motor function and arsenic exposure in children. Parvez et al. [29] conducted a study to investigate the associations of drinking water arsenic and Mn with motor function in 304 children in Bangladesh, 8-11 years of age. The motor function was assessed with the Bruininks-Oseretsky test, version 2, in four subscales. After adjustment of blood Mn, lead, selenium, and other covariates, blood As was associated with decreases in total motor composite (TMC), fine manual control (FMC), and body coordination (BC). Other As exposures (WAs, UAs, and toenail As) were also inversely associated with motor function scores, particularly TMC and BC. In addition, the study found blood selenium was positively associated with TMC, FMC, and MC in the unadjusted models. Mn exposure was not significantly associated with motor function [29]. Another study involved 526 children aged 6-7 years living near a metal foundry in Torreón, Mexico. Children's behavior was assessed by parents and teachers with Conners Behavior Rating Scales. The mean behavior scores from parent and teacher ratings were within the clinically normal range (T < 65). There was no significant association between any measure of UAs and parent ratings of behavior. However, there existed an association between higher total UAs and urinary dimethylarsinic acid (DMA) with higher ratings on the oppositional, cognitive problems and attention deficit hyperactivity disorder (ADHD) subscales of the teacher ratings. But after adjustment for the Peabody Picture Vocabulary Test scores, the associations between total UAs and behavior became statistically non-significant, suggesting that the harmful effects of arsenic on behavior may be secondary to arsenic-induced cognitive deficits [30].

In a cross-sectional study, 130 junior high school students aged 12–14 years were recruited and examined for the motor and sensory nerve conduction velocity of peripheral nerves in their right upper and lower limbs. After adjustment for gender and height, the development of slow nerve conduction velocity of the sural sensory action potential (SAP) was found among those subjects with a cumulative arsenic dosage of >100.0 mg. Thus chronic exposure to arsenic might induce peripheral neuropathy. The slowing of the nerve conduction velocity of sural SAP might be an early marker of chronic arsenic neuropathy [31].

15.3 Developmental Neurobehavioral Toxicity in Animals

Studies in animals suggest that arsenic exposure can impair cognitive development and neurobehavioral development. Sprague-Dawley (SD) rats were exposed to arsenite (36.70 mg/L in drinking water) from gestation day 15 (GD 15) or postnatal day 1 (PND 1) until approximately 4 months old. Rat offspring exposed from GD 15 showed increased spontaneous locomotor activity and both exposed groups showed increased numbers of errors in a delayed alternation task in comparison to the control group [32]. Xi et al. [33] exposed offspring rats to sodium arsenite in drinking water from GD 6 until PND 42. Exposure to 100 mg/L arsenite induced a significant decrease in the number of incidences in tail hung, auditory startle, and visual placing in rat pups. A dose-related increase in the trained numbers of rats and latency time were
observed among As-exposed groups in the square water maze test [33]. Pregnant C57BL6/J mice consumed drinking water containing sodium arsenite from GD 4 until birthing. Arsenic produced a range of behavioral impairments in male and female offspring at each of the test ages. The most striking effects of arsenic were on the development of gait and other motor responses including acoustic startle, righting reflexes, and forelimb grip [34]. Compared to control, exposure to arsenic during gestation and lactation significantly prolonged the time of completing reflex response of surface righting, negative geotaxis, and cliff avoidance of rat offspring in a 13.6 mg/L As-exposed group [35]. Moderate perinatal arsenic (50 ppb) exposure disrupted the regulatory interactions between the hypothalamic-pituitary-adrenal axis and the serotonergic system, and increased learned helplessness and measures of immobility in a forced swim task in adult mouse offspring [36]. Perinatal arsenic exposure significantly reduced corticosterone receptor (CR) levels in the hippocampus and impaired the performance of mouse offspring in an object recognition task and in the eight-way radial arm maze [37]. However, in the gestational administration of arsenic (0, 1.5, 3.0, and 4.5 mg/kg/day/po) from GD 8 till parturition, rat offspring showed no significant changes in the day of appearance of eye opening, startle reflex, negative geotaxis, and spontaneous alteration performance in comparison to the control group [38]. The different results might be due to different exposure duration and dose.

Weaned SD rats exposed to arsenic for 3 months showed poor performance in Morris water maze and Y maze tasks [39,40]. Another study found arsenic exposure impaired the performance, in Morris water maze, of Wistar rats [41]. Twenty-one-day-old male SD rats were exposed to iAs in drinking water for 1 year. There were significant decreases in the total-distance number of movements, and movement time during the dark phase of the dark-light cycle in the group treated with 50 mg iAs/L in comparison to the control group [42]. Zebrafish embryos exposed to arsenic between 4 and 120h post-fertilization (hpf) resulted in alterations in neural development including weak tactile responses to light (2.0–5.0 mM, 30 hpf), malformation of the spinal cord, and disordered motor axon projections (2.0 mM, 48 hpf) [43].

15.4 Arsenic Distribution After Exposure in Early Life

When exposure occurs in the early developmental stages, the CNS is more susceptible to toxic agents. Gestation in humans and rodents is a period of high sensitivity for offspring to chemical toxicity [12]. During gestation, both inorganic arsenic and its methylated metabolites (MMAs and DMAs) can readily cross the animal and human placenta and enter the fetal system. Concha et al. [44] found the concentration of arsenic in cord blood (median, $9 \mu g/L$) was almost as high as in maternal blood (median, $11 \mu g/L$), and these two were strongly correlated. Hall et al. [45] also found the same correlation. Thus, at least in late gestation, arsenic is easily transferred to the fetus. The main speciated arsenic in newborns and their mothers was DMA, which may indicate that DMA is the major form of arsenic transferred to the fetus [44]. Animal

studies also found arsenic can readily cross the placenta. When mice were exposed to arsenic during gestation and lactation, concentrations of iAs, MMA, and DMA in the liver and brain of newborn mice increased with the concentration of arsenic in drinking water [33,46–48]. Compared to those in the brain of mother mice, summation of arsenic species in the brain or liver of newborn mice was higher [46].

While arsenic concentrations in maternal blood and urine were high, averaging 10 and $320 \mu g/L$ respectively, the mean arsenic level in human milk was $2.3 \mu g/kg$ [49]. Fängström et al. [50] also found arsenic levels in human milk were much lower than in blood. An animal study confirmed the low level of arsenic in mothers' milk [48]. Therefore, breast-fed infants might be protected from arsenic exposure.

Animal studies found that, when exposed to arsenic in early life, arsenic can cross the immature blood-brain barrier (BBB) easily. When mice were exposed to arsenic during gestation and lactation, arsenic concentrations in the brain of newborn mice increased with the concentration of arsenic in drinking water [33,35,46–48,51]. Compared to those in the liver of newborn mice, contents of iAs in the brain were almost the same (1:1), and contents of DMA in brain were much higher (1:2). So, as an organic metabolite, DMA might accumulate preferentially in the brain [46].

The brain is particularly vulnerable during gestation and early life, and the damage induced by toxic agents is often permanent [47]. In arsenic-contaminated areas, fetuses and infants may be exposed to iAs and its metabolites via the placenta. The BBB is formed by blood vessel endothelial cells, astrocytes, pericytes, and a basement membrane [52], which can separate the brain from the circulatory system and protect the central nervous system (CNS) from potentially harmful chemicals. However, as the barrier is immature at early developmental stages, the transfer of toxic agents from blood into brain might not be limited efficiently. In addition, arsenic exposure could impair the structure and function of the BBB, and have deleterious effects such as the formation of impaired astrocytes [53], hippocampal neurons and vascular endothelial cells, irregular capillary lumina, and obvious edema around capillaries [35].

15.5 Mechanism of Developmental Neurotoxicity

15.5.1 Effects of Arsenic on Neurotransmitter Systems

Efficient communication among large numbers of brain cells (i.e., neurons) is necessary for the normal function of the nervous system. A central mechanism of neuronal communication involves the release of neurotransmitters that bind to specialized receptors on the target cell, changing its activity [54]. Any alteration in neurotransmitter systems could be reflected in the neurobehavioral deficits seen in humans and animals. Arsenic alters cholinergic, glutamatergic, and monoaminergic systems in adult rodents, with the dopaminergic system being the most affected [2]. Rodríguez et al. [2] have described the effects of arsenic on neurotransmitter systems in an excellent review, but we will not go further on this issue here.

15.5.2 Arsenic and Neurite Growth, Astrocyte, and Myelin

During prenatal development each neuron establishes neurites that develop into a unique and complex pattern of dendrites, which function as the sites of input for all connecting neurons, and an axon that carries neuronal output. The production and elaboration of dendrites and axons during fetal development is essential for neuron function and connectivity [55]. The establishment of the correct patterning of dendrite elaboration and axon outgrowth is highly regulated during fetal development [56]. Many studies indicate that developmental exposure to arsenic can alter neurite growth and complexity.

A study was designed to examine the effects of arsenic on brain explants (human fetal brain and newborn rat brain) and cells grown and maintained in a tissue culture system. Arsenic exposure resulted in changes in brain cell membrane function, reduced brain explants' growth, and caused loss of cells, inhibition of intracellular networking, and the loss of ground matrix [57]. Exposure of rats to arsenic in drinking water (3 ppm) from gestation to the age of 4 months resulted in moderate to severe alterations of thickness, organization, surrounding space, and shape of fiber tracts and axons, while cell bodies remained healthy. These anomalies were accompanied by significant decrease of all nitrergic markers in striatal tissue [58].

Hong and Bain [59] used P19 mouse embryonic stem cells to examine whether arsenic exposure could alter their differentiation into skeletal muscles and neurons. Although embryoid body formation was not altered, arsenic suppressed their differentiation into muscles and neurons. The altered differentiation was due to the repression of muscle- and neuron-specific transcription factors (Pax3, Myf5, MyoD, myogenin, neurogenin 1, neurogenin 2, and NeuroD) caused by repressed Wnt/ β -catenin signaling pathways in early embryogenesis.

Frankel et al. [55] investigated the effects of sodium arsenite on the early stages of neurite production and growth. The results showed that arsenic has concentration- and timedependent effects on initial neurite outgrowth *in vitro*. Exposure to low levels of arsenic for 5 days results in reduced neurite production, outgrowth, and complexity in newly differentiating PC12 cells. Another study found that exposure to low levels of arsenic ($\leq 5 \mu$ M) during differentiation reduced the neurite outgrowth in N2a cells in the absence of cell death [60]. Reduced neurite outgrowth induced by arsenic resulted from deficient activation of AMPK as a consequence of a lack of activation of LKB1. Oxidative stress induced by arsenic, especially excessive superoxide, played a critical role in blocking the LKB1-AMPK signaling pathway [60]. Primary cultured neurons obtained from the cerebral cortex of mouse embryos exposed to sodium arsenite (1 and 2 μ M) showed significant inhibition of total neurite length and expression of GluA1, a specific subunit of AMPA receptor. Compensatory overexpression of GluA1 protected against the deleterious effects of arsenic on neurite outgrowth [61].

A study was designed to determine whether sodium arsenite $(0, 1, 5, \text{ or } 10 \mu\text{M})$ affects both neurite outgrowth and/or induces apoptosis. The results showed that the area ratio of neurite to cell body in SCAT3-expressing cells was significantly reduced by 5 and $10 \mu\text{M}$ sodium arsenite. Apoptosis was significantly induced when arsenic exposure continued after the significant effects on neurite outgrowth were found. Arsenic treatment increased the mRNA levels of the neurofilament light (NF-L) and neurofilament medium (NF-M) and decreased the mRNA levels of tau and tubulin in a dose-dependent manner [62]. Male Wistar rats exposed to single doses of arsenite (15 and 20 mg/kg) showed reduced expression of NF and fibroblast proteins in sciatic nerve. Some fibroblast protein bands had disappeared in the 20-mg/ kg dose group. The analyzed NF-M and NF-L proteins decreased dose dependency over time [63]. Subchronically exposed to arsenic reduced expression of NF-L protein and increased expression of mu- and m-calpain protein in the sciatic nerve of rats, both in a dose/time pattern. Furthermore, NF-H protein was hypo-phosphorylated, while NF-L and microtubuleassociated protein tau (MAP-tau) proteins were hyper-phosphorylated [64]. Arsenic treatment of differentiated NB2/d1 cells and cultured dorsal root ganglion neurons decreased NF transport into axonal neurites and increased perikaryal phospho-NF immunoreactivity. Both of these effects were mediated by increasing activity of JNK, while GSK-3β activity may selectively interfere with NF transport following arsenic exposure [65]. Dubey et al. [66] also found arsenic exposure inhibited translocation of NFs into axonal neurites in culture and increased perikarval NF phosphoepitopes.

Astrocytes are the most numerous cell type within the CNS and perform a variety of tasks, from axon guidance and synaptic support, to the control of the BBB and blood flow [67]. Some studies showed that arsenic exposure can impair astrocytes. Exposure of primary cultured astrocytes to 0-30 µM arsenite for 24 h resulted in damage to astrocytes in a concentrationdependent manner, and inhibited glutamate metabolism reflected as lower activities and protein expression of glutamine synthetase and glutamate transporter (GLAST and GLT-1) [53]. Incubation of astrocytes for 2h with arsenate in concentrations of up to 10mM caused a time- and concentration-dependent loss of GSH from viable astrocytes, which was accompanied by a matching increase in the extracellular GSH content. The results suggested that arsenate exposure disturbs astrocytic GSH metabolism [68]. Primary neuron cultures were divided into four groups, which were neurons without astrocyte-conditioned medium (ACM) exposure (group I) and neurons exposed to ACM from 0, 5 or 10 µM arsenite-treated astrocytes (groups II-IV). Synaptic formation and protein expression of NR2A, NR2B, CaMKII, and AC in groups III and IV were significantly suppressed when compared to results in group II. Moreover, synaptic formation and protein expression of CaMKII and AC in group II were significantly enhanced when compared with group I. The results suggested that arsenic in astrocytes might impair synaptic formation through disturbing astrocytic effects on neuronal signal transduction [69].

Arsenic exposure also affects the myelin of nerve fibers. Adult female Wistar rats were exposed to arsenic (3 and 36 ppm) from gestation until 1, 2, 3, and 4 months of age. Arsenic intake resulted in myelin damage reflected as empty spaces in fiber tracts, and significant lower levels of dimethyl arginine in exposed animals, as compared with the controls [70]. Male SD rats weighing 40–60 g exposed to sodium arsenite through drinking water for 3 months showed abnormal structural changes in the myelin sheaths of nerve fibers and decreases in the terminals of mossy fibers [39].

15.5.3 Arsenic and Neuron Apoptosis

Studies have shown apoptosis to be involved in arsenic-induced neurotoxicity. After exposure to arsenic, human fetal and rat neonatal brain explants cells showed disturbance in lipid peroxidation, generation of nitric oxide (NO) and reactive oxygen species (ROS), apoptosis, and necrosis. The administration of vitamins C and E and DMSA partially reversed these effects [71]. Milton et al. [72] found apoptosis in a neuronal cell line peaked between 10 and 20 μ M of arsenic trioxide, while higher concentrations of arsenic caused increasing cell death. Zinc significantly decreased arsenic-induced apoptosis in a dose-dependent manner at 50 and 75 μ M zinc. Embryonic primary rat midbrain (GD 12) neuroepithelial cells exposed to sodium arsenite (0, 0.5, 1, 2, and 4 μ M) did not show apparent apoptosis until 4 μ M at 24 h [73].

Kaler et al. [74] studied the effects of arsenic exposure during the rapid brain growth period (RBGP) (postnatal days 4–11) on pyramidal neurons and granule cells of rat hippocampus. Microscopic study found decreased numbers and isolation of pyramidal neurons in superficial layers, misalignments of pyramidal cells in stratum pyramidale of CA1 and CA3 in rats exposed to 1.5 and 2.0 mg/kg sodium arsenite, and the presence of polymorphic cells in the subgranular zone of the ectal limb of the dentate gyrus. Morphometric assessments quantified and confirmed the microscopic findings. Increased apoptosis was found in the dentate gyrus of rats exposed to 2.0 mg/kg sodium arsenite. These results confirmed the vulnerability of pyramidal and granule cells of the hippocampus during RBGP [74]. Another study found arsenic exposure (5 and $10 \,\mu$ M) for 24 h caused obvious apoptosis in primary cultured rat cerebellar granule neurons. Neuroglobin protein and mRNA expression were significantly down-regulated shortly after arsenic exposure and then up-regulated after a longer time of exposure. Exposure to arsenic for 16 weeks induced higher neuroglobin expression in rat cerebellum [75].

Exposure to sodium arsenite (5, 10, or 15μ M) or dimethylarsinic acid (1 or 5 mM) induced apoptosis in cerebellar neurons, with the inorganic form being more toxic. Arsenite-induced cerebellar neuron apoptosis requires new gene expression and caspase activation. Sodium arsenite selectively activated p38 and JNK3 (not JNK1 or JNK2) in cerebellar neurons. Blocking the p38 or JNK signaling pathways protected cerebellar neurons against arsenite-induced apoptosis [76]. Arsenic exposure also induced cortical neuron apoptosis through activating JNK and p38 MAP Kinases [77]. Rat cerebellar granule neurons (CGNs) exposed to sodium arsenite (from 0 to 50 μ M) showed obvious apoptosis. Meanwhile, arsenite treatment increased phosphorylation expression of p38 and extracellular signal-regulated kinases (ERK1/2), but not of c-Jun N-terminal kinases. Furthermore, SB203580 (an inhibitor of p38) decreased the percentage of apoptotic cells whereas U0126 (an inhibitor of ERK1/2) enhanced arsenitestimulated toxicity. These data suggest that p38 may contribute to arsenite-induced apoptosis of rat CGNs, but that ERK1/2 is involved in cell growth and survival [78].

Arsenic-induced free radicals are known to cause cellular apoptosis through the mitochondrial-driven pathway. Dwivedi et al. [79] investigated the effect of arsenic interactions with various complexes of the electron transport chain and evaluated if there was a complex preference for arsenic that could trigger apoptosis. They found that arsenic exposure induced free radical generation in rat neuronal cells, which reduced mitochondrial potential and enzyme activities of all the complexes of the electron transport chain. These early events along with reduced ATP levels could be correlated with the later events of cytosolic migration of cytochrome c, altered bax/bcl2 ratio, and increased caspase-3 activity. Monoisoamyl dimercaptosuccinic acid could reverse most of these arsenic-induced events except DNA damage [79]. One study investigated the toxic effects and possible mechanisms of arsenic exposure (0.5 and 5 mg/L As₂O₃ in drinking water for 6 weeks) in the cerebrum of mice. Arsenic exposure induced oxidative stress, apoptosis, and activation of caspase-3, up-regulation of Bax and Bak, and down-regulation of Mcl-1 in the cerebral cortex. Exposure to arsenic also triggered the expression of ER stress-related genes, including GRP78, GRP94, and CHOP. Meanwhile, increased phosphorylation of p38 and decreased phosphorylation of ERK1/2 were shown in the cerebral cortex as a result of arsenic exposure of mice. These results suggested that arsenicinduced oxidative stress causes cellular apoptosis in the cerebrum and that signaling of p38 and ERK1/2, and ER stress, may be involved in iAs-induced cerebral toxicity [80]. Recently, other research has found that iAs exposure induced apoptosis in N2a cells, increased oxidative stress damage, and induced several features of mitochondria-dependent apoptotic signals, including: mitochondrial dysfunction, the activation of PARP and caspase cascades, and the increase in caspase-3 activity. A further study found that JNK/ERK1/2, but not p38-MAPK, mediated mitochondria-dependent apoptosis in N2a cells. In addition, GRP 78/CHOP pathways were also involved in the apoptosis [81].

15.5.4 Arsenic and Methylation

There is currently an intensive focus on epigenetic regulation of phenotypes to identify chronic enhancement of disease risk resulting from arsenic exposure during discrete developmental windows [82]. Arsenic exposure can cause changes in all three epigenetic markers, namely, DNA methylation, histone modifications, and expression of non-coding RNAs [83]. After its absorption, arsenic undergoes hepatic biomethylation to form monomethylarsenoc acid (MMA) and dimethylarsenic acid (DMA). S-adenosylmethionine (SAM) is the principal methyl donor in the organism and is present in every cell participating in almost all methylation reactions, from small molecules to DNA, RNA, and proteins [70]. It has been proposed that a high demand of S-adenosylmethionine (SAM) for arsenic methylation as well as failure in the regeneration of SAM would compromise the availability of the methyl group in the organism leading to the characteristic toxic effects of arsenic, namely, nervous system damage, vascular disease, hepatic and renal damage as well as different types of cancer [84,85].

Methylation has an important role in the synthesis of myelin basic protein (MBP), an essential component that confers compactness on myelin. Arsenic intake resulted in myelin damage accompanied by significant reduced levels of dimethyl arginine in rats [70]. In another study, Wistar rats were exposed to arsenic in drinking water (3 and 36 ppm) from gestation until 4 months of age. The results showed DNA hypomethylation in the brain cortex and significant increase in the non-methylated form of PP1 gene promoter either in cortex or hippocampus. Furthermore, the rats showed progressive and dose-dependent deficits of fear memory [86].

As one of the most important methyl donors is folate, folate deficiency contributes to a variety of age-related neurological and psychological disorders. One study examined whether or not folate deprivation potentiated the impact of arsenic on neurofilament dynamics. The study found that folate deprivation potentiated the toxicity of arsenic. Supplementation with SAM attenuated the impact of folate deprivation on arsenic neurotoxicity [66]. Another study found that in adult rats with life-long exposure to arsenic through drinking water (3 ppm), there were no changes of SAM, choline, and PC concentrations found in the brain, but SAM and phosphatidylcholine were severely decreased in liver accompanied by a significant increase of choline. The results indicated that choline plays an important role as methyl donor in arsenic exposure [85].

Recently, one study applied chromatin immunoprecipitation followed by high-throughput massive parallel sequencing (ChIP-seq) to evaluate the H3K9 acetylation pattern in the offspring of arsenic-exposed and control mice. The results showed that arsenic exposure during embryonic life caused global hypo-acetylation at H3K9 and changes in functional annotation with highly significant representation of Krüppel-associated box (KRAB) transcription factors in brain samples from exposed pups. The authors also found that arsenic exposure impaired spatial and episodic memory, as well as fear conditioning performance in adult mice [82].

15.5.5 Arsenic and Nitric Oxide

In the brain, nitric oxide is involved in long-term potential (LTP) in cortex and hippocampus [87], neuronal transmission [88], and potent neuroprotectant [89], and its production depends mainly on calcium entry through NMDA receptors for glutamate [90]. Several studies have found that arsenic exposure disturbs NO production.

Female Wistar rats exposed to arsenite in drinking water (4–5 mg/kg/day) from gestation until 4 months of age showed a significantly lower response to NMDA receptor stimulation, reduction of nitric oxide synthase (NOS) activity, and decreased levels of nitrites and nitrates in striatum. These markers of NO function were accompanied by significantly higher levels of LPx and ROS production [91]. Furthermore, on exposure to a lower concentration of arsenic in drinking water (0.4 mg/kg/day) from gestation until 4 months age, rats showed brain structural changes and significant decreases of all nitrergic markers (except the expression of nNOS-mRNA). These anomalies were not accompanied by ROS and/or LPx increases at low exposure levels [58]. In another study, rats exposed to arsenic showed significantly decreased levels of NOS and acetylcholinesterase—however, accompanied by increased levels of NO generation both *in vitro* and *in vivo* [41].

15.5.6 Arsenic and Gene Expression

Arsenite, at low and subcytoxic concentrations, appears to induce oxidative stress leading to activation of early transcription factors (NF- κ B and activator protein-1), whereas addition of antioxidant inhibited the activation of these factors [92]. Subchronic exposure to As increased 8-OHdG remarkably, and down-regulated expression of succinate dehydrogenase subunit A (Sdha) and SDH activity in brain tissue of mice [93].

In an attempt to investigate whether acute arsenate treatment could alter the expression of several cell cycle genes during murine neurulation, pregnant LM/Bc dams were injected

i.p. on GD 7:12 (day:hour) and 8:12 with 40 mg/kg of arsenate. Following arsenate treatment, the gene expression of Bc1-2 and p53 was significantly up-regulated at GD 9:0, compared to control. The results indicated that arsenic inhibits cell proliferation, rather than inducing apoptosis, which delayed neural tube closure [94]. In order to understand the mechanism of arsenate-induced neural tube defects (NTDs), highly sensitive Folr2 nullizygous mice were injected i.p. with sodium arsenate at the beginning of the neural tube formation process. The investigators identified several candidate genes as well as important ontology groups that may be responsible for arsenic's teratogenicity. These genes included: engrailed 1, plateletderived growth factor receptor alpha, and ephrin A7. The gene ontology groups included: morphogenesis, oxidative phosphorylation, redox response, and regulation of IKK/NF-κB cascade [95]. One study sought to evaluate proliferation and differentiation of adult neural progenitor cells (NPC) in the dentate gyrus after $50 \,\mu g/L$ As exposure throughout the perinatal period of development in mice. Compared to controls, proliferation of the NPC was decreased by 13% (not significantly); however, the number of differentiated cells was significantly decreased by 41% in As-exposed mice. Brief, daily exposure to environmental enrichment significantly increased proliferation and differentiation in both control and As-exposed animals. Expression levels of 31% of neurogenesis-related genes were altered after As exposure and restored after enrichment [96].

In a further study, mice were exposed to 1 and 4 ppm As₂O₃ through drinking water for 60 days. Arsenic exposure induced learning and memory impairment and down-regulated Camk4 expression, a very important regulator in the LTD pathway [97]. Weaning rats exposed to arsenic for 3 months had impaired neurobehavioral performance, reduced expression of NR2A, a subunit of NMDA receptor [40], and altered expression of postsynaptic signaling proteins (CaMKII, PSD-95, SynGAP, and ERK1/2) in rat hippocampus [98]. Polysialylation of NCAM (PSA-NCAM) is a critical functional feature of NCAM-mediated cell interactions and functions. Maternal and early life arsenite exposure up-regulated the expression of NCAM, PSA-NCAM, and polysialyltransferases (STX and PST) in hippocampus of rat offspring on PND 21 and PND 120 [35].

Perinatal arsenic exposure increased basal plasma corticosterone levels, decreased hippocampal glucocorticoid receptor (GR) levels in the hippocampus, and impaired learning and memory performance [99]. Reduced expression of MAPK/ERK genes was found in perinatal As-exposed offspring, induced by GR deficits [99]. One study found that perinatal As-exposed offspring showed an increase in hypothalamic corticotrophin-releasing factor, altered corticosterone secretion both at baseline and in response to a stressor, decreased hippocampal 11β -HSD 1, and altered subcellular GR distribution in the hypothalamus [100].

Exposure of SN56.B5.G4 cells expo to both sodium arsenite and DMA resulted in enhanced levels of APP and sAPP in the membrane and cytosolic fractions, respectively. Exposure of primary neuronal cells overexpressing the Swedish mutation of human APP to sodium arsenite resulted in an enhanced level of membrane-bound APP, a slightly higher level of sAPP β , and a reduced level of A β peptides in the culture medium. In contrast, exposure of neuronal cells to DMA considerably increased formation of A β and sAPP β , accompanied by an enhanced membrane APP level. The results indicated that sodium arsenite and DMA can affect processing of APP *in vitro* [101].

15.6 Neuroprotective Agents Against Arsenic Toxicity

Plant extracts and pharmacological agents have been used to investigate their neuroprotective efficacy against arsenic-induced neurotoxicity [102]. Curcumin, a non-toxic polyphenolic nonflavanoid compound, has been found effective in the treatment of neurodegenerative disorders [103] and chemical-induced neurotoxicity including lead and cadmium [104]. The efficacy of curcumin against arsenic-induced neurotoxicity has been investigated in rats. In a comparison of rats treated with arsenic as against those co-treated with arsenic (sodium arsenite, 20 mg/kg b.w.) and curcumin (100 mg/kg b.w.) orally for 28 days, co-treatment increased the neurobehavioral performance, binding of striatal dopamine receptors, and TH expression while it decreased arsenic levels and oxidative stress in brain regions [102]. Decrease of dopamine (DA), norepinephrine (NE), epinephrine (EPN), serotonin (5-HT), 3,4-dihydroxyphenylacetic acid (DOPAC), and homovanillic acid (HVA) in corpus striatum and hippocampus observed in arsenic rats showed a trend of recovery in rats co-treated with arsenic and curcumin. Furthermore, increased levels of NO in different brain regions in arsenic-treated rats were found decreased in rats co-treated with arsenic and curcumin [105]. Co-treatment with arsenic and curcumin increased learning and memory performance associated with increased binding of 3H-QNB and activity of acetylcholinesterase in brain as compared to arsenic-treated rats. Increase in the expression of ChAT protein, immunoreactivity of ChAT, and staining of Nissl bodies in the hippocampal region was also observed in rats co-treated with arsenic and curcumin as compared to those treated with arsenic alone [106].

Taurine, an end product of L-cysteine metabolism, is the most abundant free amino acid in many tissues. It protects many of the body's organs against toxicity and oxidative stress caused by various toxic substances [107]. Oral administration of taurine (100 mg/kg b.w. for 5 days) was found to be very effective in the prevention of As-induced oxidative impairment in the brain tissue of the experimental rats [108]. In another study, As exposure resulted in swellings, evident vacuolar degeneration in cytoplasm, karyorrhexis and karyolysis, and intensive expression of 8-nitroguanine in the brain of mice. Weak expression of 8-nitroguanine and relatively light pathological changes were observed in brain of the mice co-administered arsenic and taurine [109]. The protective effect of taurine on the disturbed biogenic amine neurotransmitter levels in the mouse brain induced by arsenic was examined in a recent study. The brain of rats exposed to As alone showed significantly decreased concentrations of NE, DA, and 5-HT, number of synaptic vesicles, and gene expressions of TH, TPH, and DBH. Administration of taurine significantly alleviated these toxic effects [110].

The neuroprotective effects of Zn, vitamin C, and vitamin E were investigated. Perinatal arsenic exposure resulted in less body weight gain, and retardation in development of morphology and sensory motor reflexes of the pups. Significant decreases in motor behavior along with significant decreases in GSH levels in the serum were observed in young adults. On the other hand, serum γ -GT and TBARS were significantly increased due to arsenic treatment. However, zinc in animals co-exposed to arsenic with zinc showed a remarkable ameliorating effect on all observed teratological and biochemical arsenic toxicities in male offspring [111]. Zinc also protected against arsenic-induced apoptosis in a neuronal cell line in a concentration-dependent manner [72]. One study was designed to investigate the protective effect of the antioxidants Zn, vitamin C, and vitamin E against toxic effects of arsenic exposure on the brain during pregnancy and lactation. Arsenic exposure enhanced alkaline phosphatase (ALP) activity, acetylcholinesterase activity, TBARS, and catalase activity. Co-exposure of arsenic and antioxidants (a mixture of Zn, vitamin C, and vitamin E) reversed the increase of ALP and TBARS [112]. Arsenic exposure caused disturbance in lipid peroxidation, and generation of nitric oxide (NO), ROS, apoptosis, and necrosis in human fetal brain explants. The administration of vitamins C, E, and DMSA showed partial reversal of the effects, indicating possible protection from arsenic toxicity [57].

Some studies found that exogenous glutathione and methionine can ameliorate the effects of arsenicals on NO metabolism in the brain and reduce arsenic burden in both blood and brain. Mice were exposed to sodium arsenite (50 mg/L in drinking water) for 4 weeks and treated with saline solution and glutathione (200, 400, or 800 mg/kg b.w.) at the 4th week. It was found that exogenous glutathione could promote both primary and secondary arsenic methylation capacity in the liver, which might facilitate excretion of arsenicals, and consequently reduce arsenic burden in both blood and brain. Administration of glutathione could significantly increase NOS activities and NO contents [113]. Exogenous methionine had almost the same effect as exogenous glutathione on arsenic burden and NO metabolism in brain of mice exposed to arsenite [114].

One study reported that arjunolic acid, a triterpenoid saponin, possessed the ability to ameliorate arsenic-induced oxidative insult in murine brain and is probably due to its antioxidant activity [115]. In another study, therapeutic efficacy of some thiol chelators was compared in terms of reducing arsenic burden, as well as recovery in the altered biochemical variables particularly suggestive of oxidative stress. Among these thiol chelators, monoisoamyl DMSA (MiADMSA) was the most effective chelator in reducing ROS and in counteracting arsenic-induced inhibition of ALAD, SOD, and GPx activity in the brain [116].

15.7 Conclusions

Epidemiologic (Table 15–1) and animal studies (Table 15–2) have demonstrated that arsenic exposure impairs neurodevelopment, and this adverse effect of arsenic may gradually accumulate over time. During gestation and lactation, both inorganic arsenic and its methylated metabolites (MMAs and DMAs) can readily cross the animal and human placenta and enter the fetal system. Animal studies found that, when exposed to arsenic in early life, arsenic can cross the immature BBB easily. Furthermore, arsenic exposure could impair the structure and function of the BBB. Our current knowledge indicates that arsenic may impair neurotransmitter systems, and impair neuronal signal transduction through impaired astrocyte and neurite growth, and myelin formation. In addition, arsenic can cause neuron apoptosis, and alter methylation status, level of nitric oxide, and expression of some important genes in the brain. Some plant extracts, vitamins, Zn, and pharmacological agents have shown their neuroprotective efficacy against arsenic-induced neurotoxicity. In the future, more studies are needed to uncover the molecular mechanisms of arsenic-induced neurodevelopmental toxicity and find more effective neuroprotective agents to counter toxicity.

Population Studied				
Number (male; female)	Age (years)	Concentration of As (mean±SD)	Neurodevelopmental Effects	Reference
Exposed: 41 (18; 23)	7.6±0.8	UAs: 62.9±0.03 (μg As/g Cr)	Decreased verbal IQ, poor performance on long-term	[15]
Control: 39 (23;16)	7.4 ± 1.0	UAs: 40.2±0.03 (µg As/g Cr)	memory and linguistic abstraction	
High As: 29 (16; 13)	13.5±0.6	WAs: 184.99 ± 225.29 μg/L	Alterations in pattern memory and switching attention	[16]
Low As: 20 (10; 10)	13.4±0.5	WAs: 131.19±343.70 μg/L		
Control: 60 (27; 33)	13.4±0.6	NR		
351 (190; 161)	5–15	Peak lifetime water: $147 \pm 322 \mu g/L$ Average lifetime water: $59 \pm 133 \mu g/L$ Pregnancy water: $110 \pm 243 \mu g/L$ Current child urine: $78 \pm 61 \mu g/L$	Decrements in intellectual testing: (1) vocabulary test; (2) the object assembly test; (3) the picture completion test	[14]
201 (98; 103)	10 ± 0.4	Mean (range) (µg/L) WAs: 177.8 (0.094–790)	Reduced intellectual function: lower performance and full-scale scores	[18]
301 (150; 151)	6±0.18	Mean (range) (µg/L) WAs: 120.1 (0.1–864)	Reduced intellectual function: lower performance and processing speed raw scores	[19]
High As: 91 (37; 54) M-As: 243 (135; 118) H-F: 180 (92; 88) Control: 196 (112; 84)	8–12	WAs: 190±183 μg/L WAs: 142±106 μg/L WAs: 3±3 μg/L; F: 8.3±1.9 mg/L WAs: 2±3 μg/L; F: 0.5±0.2 mg/L	Reduced intellectual function	[17]
9–10 years: 232	9–10	UAs mean (range) 181.9 (0.6–2336) µg/L	Reduced intellectual function in 9–10 group	[20]
4–5 years: 176	4–5	UAs mean (range) 137 (5.0–1021) µg/L		
591 (319; 272)	6–8	UAs: 58.1±33.2μg/L BPb: 11.5±6.3μg/dL	Reduced cognitive development Different effects on boys and girls Independent of any effect of lead	[21]
299 (151; 148)	9.64±0.77	BAs: 4.81±3.22μg/L UAs: 78.09±72.16μg/L BMn: 14.78±3.72μg/L	BAs negatively related to most WISC-IV subscale scores UAs negatively related to verbal comprehension scores	[24]

Table 15–1 Neurodevelopmental Effects of Arsenic Exposure in Children

1652 (894; 758)	Mother's age: 27.7±4.1	Median (µg/L) BAs: 0.86; BPb: 41 BHg: 1.88: BCd: 0.03	Decreased neonatal behavioral neurological assessments (DNBN) score; considered safe level of As related to lower DNBN scores	[28]
3988 (1861; 2127)	6.5–11.5	In soil: Mean (range) (mg/kg) As: 2.6 (0–42.1); Pb: 35.4 (0.9–1800)	Increased probability of intellectual disability (ID) Interaction between Pb and As on probability of ID	[22]
1700 (884; 816)	1.5 years old 5 years old	Median (10th and 90th percentiles) Maternal UAs (mg/L) GW 8: 81 (24, 380) GW 30: 84 (26, 415) Children UAs (mg/L) At 1.5 years: 34 (12, 155) At 5 years: 51 (20, 238)	Reduced verbal IQ (VIQ) and full scale IQ (FSIQ) in girls	[27]
100	Mother's age: 22.9±3.7 GW: 38.9±1.4	Mean (range) (μ g/L) As (n = 94): 1.46 (0.51–9.58) Pb (n = 79): 31.7 (6.83–220.8) Zn (n = 94): 2286 (1299–6430)	Cord blood levels of Pb and As inversely related with neurodevelopment of newborns	[25]
1799	7 months	Maternal UAs: Median (IQR) (µg/L) GW8: 81 (37–207); GW30: 84 (42–230)	No significant effects on children's motor and PST scores and behavior ratings	[26]
117 (62; 55)	12–14	WAs (µg/L): undetectable–3590	Slow nerve conduction velocity	[31]
303 (151; 152)	9.6±0.7	BAs (μg/L): 4.8±3.2 WAs (μg/L): 43.3±73.6 UAs (μg/L): 78.0±72.1 UAs (μg/q Cr): 246.5±183.9	Low levels of As in drinking water inversely associated with motor function scores; Blood selenium was positively associated with motor function scores	[29]
526 (292; 234)	6.9±0.4	Median (IQR) (μg/L) Total UAs: 55.2 (39.7); InAs: 7.5 (6.6) MMA: 6.7 (5.9); DMA: 39.3 (28.5)	Moderate association between TAs and DMA in urine and teacher ratings of oppositional, cognitive problems and ADHD-like behaviors among the children	[30]

TAs, total arsenic; InAs, inorganic arsenic; UAs, urinary arsenic; WAs, water arsenic; BAs, blood arsenic; M-As, medium-arsenic; MMA, monomethylarsonic acid; DMA, dimethylarsinic acid; BMn, blood manganese; BHg, blood mercury; BPb, blood lead; Cr, creatinine; H-F, high fluoride; IQ, intelligence quotient; IQR, interquartile range; GW, gestational week.

Animal	Exposure Duration	Arsenic Dose	Neurobehavioral Toxicity	Reference
Male SD rat	GD15 or PND1 to 4 months	36.70 mg/L in drinking water	Increased spontaneous locomotor activity Increased number of errors in a delayed alternation task	[32]
Wistar rat	GD 6 to PND 42	0, 10, 50, 100 mg/L NaAsO ₂ in drinking water	100 mg/L group: Decreased number of incidences in tail hung, auditory startle and visual placing; Poor performance in water maze test	[33]
C57BL6/J mice	GD4 to birth	0, 8, 25 or 80ppm NaAsO ₂ in drinking water	Impairment in development of gait, acoustic startle, righting reflexes, and forelimb grip	[34]
Male SD rat	GD 1 to PND 21 or PND 120	0, 2.62, 13.6 mg/L NaAsO ₂ in drinking water	Impaired performance in surface righting, negative geotaxis, and cliff avoidance	[35]
C57BL6/J mice	GD 1 to PND 23	50ppb NaAsO ₂ in drinking water	Increased learned helplessness and measures of immobility in a forced swim task in adult mouse offspring	[36]
C57BL6/J mice	GD 1 to PND 23	50 ppb NaAsO ₂ in drinking water	Reduced level of corticosterone receptor Impaired performance in object recognition task	[37]
Wistar albino rat	GD 8 to parturition	0, 1.5, 3.0, and 4.5 mg As/kg/day/p.o.	No significant changes in neurobehavioral developmental tests	[38]
Male weaning SD rat	3 months	0, 2.62, 13.6, and 68 mg/L NaAsO ₂ in drinking water	68 mg/L group: impaired performance in Morris water maze test	[40]
Male weaning SD rat	3 months	10 and 15 mg/kg NaAsO ₂ in drinking water	15 mg/kg group: Impaired performance in the Morris water maze and Y maze tasks	[39]
Male Wistar rats	6 months	1 mg/kg NaAsO ₂ or 10 mg/kg GaAs, oral	Significant deficit in hidden platform acquisition	[41]
Weaning male SD rat	1 year	0.05, 0.5, and 50 mg As/L in drinking water	Decrease in total-distance number of movements, and movement time during the dark phase	[42]

Table 15–2 Neurobehavioral Toxicity of Arsenic Exposure in Animals

SD, Sprague-Dawley.

References

- [1] Duker AA, Carranza EJM, Hale M. Arsenic geochemistry and health. Environ Int 2005;31:631-41.
- [2] Rodríguez VM, Jiménez-Capdeville ME, Giordano M. The effects of arsenic exposure on the nervous system. Toxicol Lett 2003;145(1):1-18.
- [3] Chhuttani PN, Chopra JS. Arsenic poisoning. In: Vinken PJ, Bruyn GW, editors. Handbook of clinical neurology. New York: North Holland; 1979. pp. 199–216.

- [4] Heymann A, Pfeiffer JB, Willet RW, Haywood M, Taylor HM. Peripheral neuropathy caused by arsenical intoxication: a study of 41 cases with observations on the effects of BAL (2,3-dimercapto-propanol). N Engl J Med 1956;254:401–9.
- [5] Schoolmeester WL, White DR. Arsenic poisoning. South Med J 1980;73:198-208.
- [6] Beckett WS, Moore JL, Keogh JP, Bleecker ML. Acute encephalopathy due to occupational exposure to arsenic. Br J Ind Med 1986;43:66–7.
- [7] Bolla-Wilson K, Bleecker ML. Neuropsychological impairment following inorganic arsenic exposure. J Occup Med 1987;29:500–3.
- [8] Frank G. Neurological and psychiatric disorders following acute arsine poisoning. J Neurol 1976;213:59–70.
- [9] Morton WE, Caron GA. Encephalopathy: an uncommon manifestation of workplace arsenic poisoning? Am J Ind Med 1989;15:1–5.
- [10] Winneke G. Developmental aspects of environmental neurotoxicology: lessons from lead and polychlorinated biphenyls. J Neurol Sci 2011;308(1-2):9–15.
- [11] Patel E, Reynolds M. Methylmercury impairs motor function in early development and induces oxidative stress in cerebellar granule cells. Toxicol Lett 2013;222(3):265–72.
- [12] Anderson LM, Diwan BA, Fear NT, Roman E. Critical windows of exposure for children's health: cancer in human epidemiological studies and neoplasm in experimental animal models. Environ Health Perspect 2000;108(Suppl. 3):573–94.
- [13] Vahidnia A, van der Voet GB, de Wolff FA. Arsenic neurotoxicity—a review. Hum Exp Toxicol 2007;26(10):823-32.
- [14] von Ehrenstein OS, Poddar S, Yuan Y, Mazumder DG, Eskenazi B, Basu A, et al. Children's intellectual function in relation to arsenic exposure. Epidemiology 2007;18:44–51.
- [15] Calderon J, Navarro ME, Jimenez-Capdeville ME, Santos-Diaz MA, Golden A, Rodriguez-Leyva I, et al. Exposure to arsenic and lead and neuropsychological development in Mexican children. Environ Res 2001;85:69–76.
- [16] Tsai SY, Chou HY, The HW, Chen CM, Chen CJ. The effects of chronic arsenic exposure from drinking water on the neurobehavioral development in adolescence. Neurotoxicology 2003;24:747–53.
- [17] Wang SX, Wang ZH, Cheng XT, Li J, Sang ZP, Zhang XD, et al. Arsenic and fluoride exposure in drinking water: children's IQ and growth in Shanyin County, Shanxi Province, China. Environ Health Perspect 2006;115:643–7.
- [18] Wasserman GA, Liu XH, Parvez F, Ahsan H, Factor-Litvak P, van Geen A, et al. Water arsenic exposure and children's intellectual function in Araihazar, Bangladesh. Environ Health Perspect 2004;112:1329–33.
- [19] Wasserman GA, Liu X, Parvez F, Ahsan H, Factor-Litvak P, Kline J, et al. Water arsenic exposure and intellectual function in 6-year-old children in Araihazar, Bangladesh. Environ Health Perspect 2007;115(2):285–9.
- [20] Nahar MN, Inaoka T, Fujimura M. A consecutive study on arsenic exposure and intelligence quotient (IQ) of children in Bangladesh. Environ Health Prev Med 2014;19(3):194–9.
- [21] Rosado JL, Ronquillo D, Kordas K, Rojas O, Alatorre J, Lopez P, et al. Arsenic exposure and cognitive performance in Mexican schoolchildren. Environ Health Perspect 2007;115(9):1371–5.
- [22] McDermott S, Wu JL, Cai B, Lawson A, Aelion CM. Probability of intellectual disability is associated with soil concentrations of arsenic and lead. Chemosphere 2011;84:31–8.
- [23] Wright RO, Amarasiriwardena C, Woolf AD, Jim R, Bellinger DC. Neuropsychological correlates of hair arsenic, manganese, and cadmium levels in school-age children residing near a hazardous waste site. Neurotoxicology 2006;27:210–16.
- [24] Wasserman GA, Liu X, Parvez F, Factor-Litvak P, Ahsan H, Levy D, et al. Arsenic and manganese exposure and children's intellectual function. Neurotoxicology 2011;32:450–7.

- [25] Parajuli RP, Fujiwara T, Umezaki M, Watanabe C. Association of cord blood levels of lead, arsenic, and zinc with neurodevelopmental indicators in newborns: a birth cohort study in Chitwan Valley, Nepal. Environ Res 2013;121:45–51.
- [26] Tofail F, Vahter M, Hamadani JD, Nermell B, Huda SN, Yunus M, et al. Effect of arsenic exposure during pregnancy on infant development at 7 months in rural Matlab, Bangladesh. Environ Health Perspect 2009;117(2):288–93.
- [27] Hamadani JD, Tofail F, Nermell B, Gardner R, Shiraji S, Bottai M, et al. Critical windows of exposure for arsenic-associated impairment of cognitive function in pre-school girls and boys: a population-based cohort study. Int J Epidemiol 2011;40(6):1593–604.
- [28] Yu XD, Yan CH, Shen XM, Tian Y, Cao LL, Yu XG, et al. Prenatal exposure to multiple toxic heavy metals and neonatal neurobehavioral development in Shanghai, China. Neurotoxicol Teratol 2011;33:437–43.
- [29] Parvez F, Wasserman GA, Factor-Litvak P, Liu X, Slavkovich V, Siddique AB, et al. Arsenic exposure and motor function among children in Bangladesh. Environ Health Perspect 2011;119:1665–70.
- [30] Roy A, Kordas K, Lopez P, Rosado JL, Cebrian ME, Vargas GG, et al. Association between arsenic exposure and behavior among first-graders from Torreón, Mexico. Environ Res 2011;111:670–6.
- [31] Tseng HP, Wang YH, Wu MM, The HW, Chiou HY, Chen CJ. Association between chronic exposure to arsenic and slow nerve conduction velocity among adolescents in Taiwan. J Health Popul Nutr 2006;24(2):182-9.
- [32] Rodríguez VM, Carrizales L, Mendoza MS, Fajardo OR, Giordano M. Effects of sodium arsenite exposure on development and behavior in the rat. Neurotoxicol Teratol 2002;24(6):743–50.
- [33] Xi S, Sun W, Wang F, Jin Y, Sun G. Transplacental and early life exposure to inorganic arsenic affected development and behavior in offspring rats. Arch Toxicol 2009;83(6):549–56.
- [34] Markowski VP, Reeve EA, Onos K, Assadollahzadeh M, McKay N. Effects of prenatal exposure to sodium arsenite on motor and food-motivated behaviors from birth to adulthood in C57BL6J mice. Neurotoxicol Teratol 2012;34(2):221–31.
- [35] Luo JH, Qiu ZQ, Chen JA, Zhang L, Liu WY, Tan Y, et al. Maternal and early life arsenite exposure impairs neurodevelopment and increases the expression of PSA-NCAM in hippocampus of rat offspring. Toxicology 2013;311(3):99–106.
- [36] Martinez EJ, Kolb BL, Bell A, Savage DD, Allan AM. Moderate perinatal arsenic exposure alters neuroendocrine markers associated with depression and increases depressive-like behaviors in adult mouse offspring. Neurotoxicology 2008;29(4):647–55.
- [37] Martinez-Finley EJ, Ali AM, Allan AM. Learning deficits in C57BL/6J mice following perinatal arsenic exposure: consequence of lower corticosterone receptor levels? Pharmacol Biochem Behav 2009;94(2):271–7.
- [38] Gandhi DN, Panchal GM, Patel KG. Developmental and neurobehavioural toxicity study of arsenic on rats following gestational exposure. Indian J Exp Biol 2012;50(2):147–55.
- [39] Jing J, Zheng G, Liu M, Shen X, Zhao F, Wang J, et al. Changes in the synaptic structure of hippocampal neurons and impairment of spatial memory in a rat model caused by chronic arsenite exposure. Neurotoxicology 2012;33(5):1230–8.
- [40] Luo JH, Qiu ZQ, Shu WQ, Zhang YY, Zhang L, Chen JA. Effects of arsenic exposure from drinking water on spatial memory, ultra-structures and NMDAR gene expression of hippocampus in rats. Toxicol Lett 2009;184(2):121–5.
- [41] Flora SJS, Bhatt K, Mehta A. Arsenic moiety in gallium arsenide is responsible for neuronal apoptosis and behavioral alterations in rats. Toxicol Appl Pharmacol 2009;240:236–44.
- [42] Rodríguez VM, Limón-Pacheco JH, Carrizales L, Mendoza-Trejo MS, Giordano M. Chronic exposure to low levels of inorganic arsenic causes alterations in locomotor activity and in the expression of dopaminergic and antioxidant systems in the albino rat. Neurotoxicol Teratol 2010;32(6):640–7.
- [43] Li D, Lu C, Wang J, Hu W, Cao Z, Sun D, et al. Developmental mechanisms of arsenite toxicity in zebrafish (Danio rerio) embryos. Aquat Toxicol 2009;91(3):229–37.

- [44] Concha G, Vogler G, Lezcano D, Nermell B, Vahter M. Exposure to inorganic arsenic metabolites during early human development. Toxicol Sci 1998;44(2):185–90.
- [45] Hall M, Gamble M, Slavkovich V, Liu X, Levy D, Cheng Z, et al. Determinants of arsenic metabolism: blood arsenic metabolites, plasma folate, cobalamin, and homocysteine concentrations in maternal-newborn pairs. Environ Health Perspect 2007;115(10):1503–9.
- [46] Jin Y, Xi S, Li X, Lu C, Li G, Xu Y, et al. Arsenic speciation transported through the placenta from mother mice to their newborn pups. Environ Res 2006;101(3):349–55.
- [47] Jin Y, Wang G, Zhao F, Liao Y, Sun D, Zhong Y, et al. Distribution of speciated arsenicals in mice exposed to arsenite at the early life. Ecotoxicol Environ Saf 2010;73(6):1323–6.
- [48] Xi S, Jin Y, Lv X, Sun G. Distribution and speciation of arsenic by transplacental and early life exposure to inorganic arsenic in offspring rats. Biol Trace Elem Res 2010;134(1):84–97.
- [49] Concha G, Vogler G, Nermell B, Vahter M. Low-level arsenic excretion in breast milk of native Andean women exposed to high levels of arsenic in the drinking water. Int Arch Occup Environ Health 1998;71(1):42-6.
- [50] Fängström B, Moore S, Nermell B, Kuenstl L, Goessler W, Grandér M, et al. Breast-feeding protects against arsenic exposure in Bangladeshi infants. Environ Health Perspect 2008;116(7):963–9.
- [51] Lu C, Zhao F, Sun D, Zhong Y, Yu X, Li G, et al. Comparison of speciated arsenic levels in the liver and brain of mice between arsenate and arsenite exposure at the early life. Environ Toxicol 2014;29(7):797–803.
- [52] Palmer AM. The role of the blood-CNS barrier in CNS disorders and their treatment. Neurobiol Dis 2010;37:3-12.
- [53] Zhao FH, Liao YJ, Jin YP, Li GX, Lv XQ, Sun GF. Effects of arsenite on glutamate metabolism in primary cultured astrocytes. Toxicol In Vitro 2012;26(1):24–31.
- [54] Valenzuela CF, Puglia MP, Zucca S. Focus on: neurotransmitter systems. Alcohol Res Health 2011;34(1):106–20.
- [55] Frankel S, Concannon J, Brusky K, Pietrowicz E, Giorgianni S, Thompson WD, et al. Arsenic exposure disrupts neurite growth and complexity in vitro. Neurotoxicology 2009;30(4):529–37.
- [56] McAllister AK. Cellular and molecular mechanisms of dendrite growth. Cereb Cortex 2000;10(10):963-73.
- [57] Chattopadhyay S, Bhaumik S, Nag Chaudhury A, Das Gupta S. Arsenic induced changes in growth development and apoptosis in neonatal and adult brain cells in vivo and in tissue culture. Toxicol Lett 2002;128(1-3):73-84.
- [58] Ríos R, Zarazúa S, Santoyo ME, Sepúlveda-Saavedra J, Romero-Díaz V, Jiménez V, et al. Decreased nitric oxide markers and morphological changes in the brain of arsenic-exposed rats. Toxicology 2009;261(1-2):68–75.
- [59] Hong GM, Bain LJ. Arsenic exposure inhibits myogenesis and neurogenesis in P19 stem cells through repression of the β -catenin signaling pathway. Toxicol Sci 2012;129(1):146–56.
- [60] Wang X, Meng D, Chang Q, Pan J, Zhang Z, Chen G, et al. Arsenic inhibits neurite outgrowth by inhibiting the LKB1-AMPK signaling pathway. Environ Health Perspect 2010;118(5):627–34.
- [61] Maekawa F, Tsuboi T, Oya M, Aung KH, Tsukahara S, Pellerin L, et al. Effects of sodium arsenite on neurite outgrowth and glutamate AMPA receptor expression in mouse cortical neurons. Neurotoxicology 2013;37:197–206.
- [62] Aung KH, Kurihara R, Nakashima S, Maekawa F, Nohara K, Kobayashi T, et al. Inhibition of neurite outgrowth and alteration of cytoskeletal gene expression by sodium arsenite. Neurotoxicology 2013;34:226–35.
- [63] Vahidnia A, Romijn F, Tiller M, van der Voet GB, de Wolff FA. Arsenic-induced toxicity: effect on protein composition in sciatic nerve. Hum Exp Toxicol 2006;25(11):667–74.
- [64] Vahidnia A, Romijn F, van der Voet GB, de Wolff FA. Arsenic-induced neurotoxicity in relation to toxicokinetics: effects on sciatic nerve proteins. Chem Biol Interact 2008;176(2–3):188–95.

- [65] DeFuria J, Shea TB. Arsenic inhibits neurofilament transport and induces perikaryal accumulation of phosphorylated neurofilaments: roles of JNK and GSK-3beta. Brain Res 2007;1181:74–82.
- [66] Dubey M, Shea TB. Potentiation of arsenic neurotoxicity by folate deprivation: protective role of S-adenosyl methionine. Nutr Neurosci 2007;10(5–6):199–204.
- [67] Blackburn D, Sargsyan S, Monk PN, Shaw PJ. Astrocyte function and role in motor neuron disease: a future therapeutic target? Glia 2009;57(12):1251–64.
- [68] Meyer N, Koehler Y, Tulpule K, Dringen R. Arsenate accumulation and arsenate-induced glutathione export in astrocyte-rich primary cultures. Neurochem Int 2013;62(7):1012–19.
- [69] Wang Y, Zhao F, Liao Y, Jin Y, Sun G. Effects of arsenite in astrocytes on neuronal signaling transduction. Toxicology 2013;303:43–53.
- [70] Zarazúa S, Ríos R, Delgado JM, Santoyo ME, Ortiz-Pérez D, Jiménez-Capdeville ME. Decreased arginine methylation and myelin alterations in arsenic exposed rats. Neurotoxicology 2010;31(1):94–100.
- [71] Chattopadhyay S, Bhaumik S, Purkayastha M, Basu S, Nag Chaudhuri A, Das Gupta S. Apoptosis and necrosis in developing brain cells due to arsenic toxicity and protection with antioxidants. Toxicol Lett 2002;136(1):65–76.
- [72] Milton AG, Zalewski PD, Ratnaike RN. Zinc protects against arsenic-induced apoptosis in a neuronal cell line, measured by DEVD-caspase activity. Biometals 2004;17(6):707–13.
- [73] Sidhu JS, Ponce RA, Vredevoogd MA, Yu X, Gribble E, Hong SW, et al. Cell cycle inhibition by sodium arsenite in primary embryonic rat midbrain neuroepithelial cells. Toxicol Sci 2006;89(2):475–84.
- [74] Kaler S, Dhar P, Bhattacharya A, Mehra RD. Preliminary morphological and immunohistochemical changes in rat hippocampus following postnatal exposure to sodium arsenite. Toxicol Int 2013;20(2):160–9.
- [75] Liu X, Gao Y, Yao H, Zhou L, Sun D, Wang J. Neuroglobin involvement in the course of arsenic toxicity in rat cerebellar granule neurons. Biol Trace Elem Res 2013;155(3):439–46.
- [76] Namgung U, Xia Z. Arsenic induces apoptosis in rat cerebellar neurons via activation of JNK3 and p38 MAP Kinases. Toxicol Appl Pharmacol 2001;174(2):130–8.
- [77] Namgung U, Xia Z. Arsenite-induced apoptosis in cortical neurons is mediated by c-Jun N-terminal protein kinase 3 and p38 mitogen-activated protein kinase. J Neurosci 2000;20(17):6442–51.
- [78] Liu X, Gao Y, Yao H, Zhou L, Pei J, Sun L, et al. p38 and extracellular signal-regulated kinases activations have opposite effects on primary-cultured rat cerebellar granule neurons exposed to sodium arsenite. J Biochem Mol Toxicol 2014;28(4):143–8.
- [79] Dwivedi N, Mehta A, Yadav A, Binukumar BK, Gill KD, Flora SJ. MiADMSA reverses impaired mitochondrial energy metabolism and neuronal apoptotic cell death after arsenic exposure in rats. Toxicol Appl Pharmacol 2011;256(3):241–8.
- [80] Yen CC, Ho TJ, Wu CC, Chang CF, Su CC, Chen YW, et al. Inorganic arsenic causes cell apoptosis in mouse cerebrum through an oxidative stress-regulated signaling pathway. Arch Toxicol 2011;85(6):565–75.
- [81] Lu TH, Tseng TJ, Su CC, Tang FC, Yen CC, Liu YY, et al. Arsenic induces reactive oxygen species-caused neuronal cell apoptosis through JNK/ERK-mediated mitochondria-dependent and GRP 78/CHOPregulated pathways. Toxicol Lett 2014;224(1):130–40.
- [82] Cronican AA, Fitz NF, Carter A, Saleem M, Shiva S, Barchowsky A, et al. Genome-wide alteration of histone H3K9 acetylation pattern in mouse offspring prenatally exposed to arsenic. PLOS ONE 2013;8(2):e53478.
- [83] Cheng TF, Choudhuri S, Muldoon-Jacobs K. Epigenetic targets of some toxicologically relevant metals: a review of the literature. J Appl Toxicol 2012;32(9):643–53.
- [84] Agusa T, Kunito T, Kubota R, Inoue S, Fujihara J, Minh TB, et al. Exposure, metabolism, and health effects of arsenic in residents from arsenic-contaminated groundwater areas of Vietnam and Cambodia: a review. Rev Environ Health 2010;25:193–220.

- [85] Ríos R, Santoyo ME, Cruz D, Delgado JM, Zarazúa S, Jiménez-Capdeville ME. Methyl group balance in brain and liver role of choline on increased S-adenosyl methionine (SAM) demand by chronic arsenic exposure. Toxicol Lett 2012;215(2):110–18.
- [86] Martinez L, Jimenez V, Garcia-Sepulveda C, Ceballos F, Delgado JM, Nino-Moreno P, et al. Impact of early developmental arsenic exposure on promotor CpG-island methylation of genes involved in neuronal plasticity. Neurochem Int 2011;58:574–81.
- [87] Garthwaite J, Boulton CL. Nitric oxide signaling in the central nervous system. Ann Rev Physiol 1995;57:682–706.
- [88] Gally JA, Montague PR, Reeke Jr. GN, Edelman GM. The NO hypothesis: possible effects of a short-lived, rapidly diffusible signal in the development and function of the nervous system. Proc Natl Acad Sci USA 1990;87:3547–51.
- [89] Calabrese V, Mancuso C, Calvani M, Rizzarelli E, Butterfield DA, Stell AMG. Nitric oxide in the central nervous system: neuroprotection versus neurotoxicity. Nat Rev Neurosci 2007;8:766–75.
- [90] Bredt DS, Snyder SH. Isolation of nitric oxide synthase, a calmodulin-requiring enzyme. Proc Natl Acad Sci USA 1990;87:682–5.
- [91] Zarazúa S, Pérez-Severiano F, Delgado JM, Martínez LM, Ortiz-Pérez D, Jiménez-Capdeville ME. Decreased nitric oxide production in the rat brain after chronic arsenic exposure. Neurochem Res 2006;31(8):1069–77.
- [92] Felix K, Manna SK, Wise K, Barr J, Ramesh GT. Low levels of arsenite activates nuclear factor-kappaB and activator protein-1 in immortalized mesencephalic cells. J Biochem Mol Toxicol 2005;19(2):67–77.
- [93] Hong Y, Piao F, Zhao Y, Li S, Wang Y, Liu P. Subchronic exposure to arsenic decreased Sdha expression in the brain of mice. Neurotoxicology 2009;30(4):538–43.
- [94] Wlodarczyk BJ, Bennett GD, Calvin JA, Finnell RH. Arsenic-induced neural tube defects in mice: alterations in cell cycle gene expression. Reprod Toxicol 1996;10(6):447–54.
- [95] Wlodarczyk BJ, Cabrera RM, Hill DS, Bozinov D, Zhu H, Finnell RH. Arsenic-induced gene expression changes in the neural tube of folate transport defective mouse embryos. Neurotoxicology 2006;27(4):547–57.
- [96] Tyler CR, Allan AM. Adult hippocampal neurogenesis and mRNA expression are altered by perinatal arsenic exposure in mice and restored by brief exposure to enrichment. PLOS ONE 2013;8(9):e73720.
- [97] Wang Y, Li S, Piao F, Hong Y, Liu P, Zhao Y. Arsenic down-regulates the expression of Camk4, an important gene related to cerebellar LTD in mice. Neurotoxicol Teratol 2009;31(5):318–22.
- [98] Luo JH, Qiu ZQ, Zhang L, Shu WQ. Arsenite exposure altered the expression of NMDA receptor and postsynaptic signaling proteins in rat hippocampus. Toxicol Lett 2012;211(1):39–44.
- [99] Martinez-Finley EJ, Goggin SL, Labrecque MT, Allan AM. Reduced expression of MAPK/ERK genes in perinatal arsenic-exposed offspring induced by glucocorticoid receptor deficits. Neurotoxicol Teratol 2011;33(5):530–7.
- [100] Goggin SL, Labrecque MT, Allan AM. Perinatal exposure to 50 ppb sodium arsenate induces hypothalamic-pituitary-adrenal axis dysregulation in male C57BL/6 mice. Neurotoxicology 2012;33(5):1338–45.
- [101] Zarazúa S, Bürger S, Delgado JM, Jiménez-Capdeville ME, Schliebs R. Arsenic affects expression and processing of amyloid precursor protein (APP) in primary neuronal cells overexpressing the Swedish mutation of human APP. Int J Dev Neurosci 2011;29(4):389–96.
- [102] Yadav RS, Sankhwar ML, Shukla RK, Chandra R, Pant AB, Islam F, et al. Attenuation of arsenic neurotoxicity by curcumin in rats. Toxicol Appl Pharmacol 2009;240(3):367–76.
- [103] Jagatha B, Mythri RB, Vali S, Bharath MM. Curcumin treatment alleviates the effects of glutathione depletion in vitro and in vivo: therapeutic implications for Parkinson's disease explained via in silico studies. Free Radic Biol Med 2008;44(5):907–17.

- [104] Daniel S, Limpson JL, Dairam A, Watkins GM, Daya S. Through metal binding, curcumin protects against lead and cadmium induced lipid peroxidation in rat brain homogenates and against lead induced tissue damage in rat brain. J Inorg Chem 2004;98:266–75.
- [105] Yadav RS, Shukla RK, Sankhwar ML, Patel DK, Ansari RW, Pant AB, et al. Neuroprotective effect of curcumin in arsenic-induced neurotoxicity in rats. Neurotoxicology 2010;31(5):533–9.
- [106] Yadav RS, Chandravanshi LP, Shukla RK, Sankhwar ML, Ansari RW, Shukla PK, et al. Neuroprotective efficacy of curcumin in arsenic induced cholinergic dysfunctions in rats. Neurotoxicology 2011;32(6):760–8.
- [107] Dogru-Abbasoglu S, Kanbagli O, Balkan J, Cevikbas U, Aykac-Toker G, Uysal M. The protective effect of taurine against thioacetamide hepatotoxicity of rats. Hum Exp Toxicol 2001;20:23–7.
- [108] Das J, Ghosh J, Manna P, Sinha M, Sil PC. Arsenic-induced oxidative cerebral disorders: protection by taurine. Drug Chem Toxicol 2009;32(2):93–102.
- [109] Ma N, Sasoh M, Kawanishi S, Sugiura H, Piao F. Protection effect of taurine on nitrosative stress in the mice brain with chronic exposure to arsenic. J Biomed Sci 2010;17(Suppl. 1):S7.
- [110] Liu X, Piao F, Li Y. Protective effect of taurine on the decreased biogenic amine neurotransmitter levels in the brain of mice exposed to arsenic. Adv Exp Med Biol 2013;776:277–87.
- [111] Ahmad M, Wadaa MA, Farooq M, Daghestani MH, Sami AS. Effectiveness of zinc in modulating perinatal effects of arsenic on the teratological effects in mice offspring. Biol Res 2013;46(2):131–8.
- [112] Herrera A, Pineda J, Antonio MT. Toxic effects of perinatal arsenic exposure on the brain of developing rats and the beneficial role of natural antioxidants. Environ Toxicol Pharmacol 2013;36(1):73–9.
- [113] Wang Y, Zhao F, Jin Y, Zhong Y, Yu X, Li G, et al. Effects of exogenous glutathione on arsenic burden and NO metabolism in brain of mice exposed to arsenite through drinking water. Arch Toxicol 2011;85(3):177-84.
- [114] Zhao F, Wang Y, Jin Y, Zhong Y, Yu X, Li G, et al. Effects of exogenous methionine on arsenic burden and NO metabolism in brain of mice exposed to arsenite through drinking water. Environ Toxicol 2012;27(12):700–6.
- [115] Sinha M, Manna P, Sil PC. Protective effect of arjunolic acid against arsenic-induced oxidative stress in mouse brain. J Biochem Mol Toxicol 2008;22(1):15–26.
- [116] Flora SJ, Bhadauria S, Pant SC, Dhaked RK. Arsenic induced blood and brain oxidative stress and its response to some thiol chelators in rats. Life Sci 2005;77(18):2324–37.

16

Developmental Arsenic Exposure Impacts Fetal Programming of the Nervous System

Fengyuan Piao, Shuangyue Li, and Huai Guan

DEPARTMENT OF OCCUPATIONAL AND ENVIRONMENTAL HEALTH, COLLEGE OF PUBLIC HEALTH, DALIAN MEDICAL UNIVERSITY, DALIAN, PEOPLE'S REPUBLIC OF CHINA

CHAPTER OUTLINE

16.1	Introduction	387
16.2	Accumulation of Arsenic in Fetal Brain Tissue	388
16.3	Effect of Arsenic on the Development of the Nervous System	389
16.4	Effect of Arsenic on Neurobehavior	391
16.5	Mechanisms for the Effect of Arsenic on the Nervous System During Development	395
16.6	Conclusions and Future Directions	399
Refe	rences	399

16.1 Introduction

Developmental processes (i.e., gestation and early childhood) in the nervous system are vulnerable to disruption by some chemicals at doses that may not be toxic to mature systems [1,2]. The vulnerability of the brain originates from the combination of immaturity and ongoing development [3]. The fetal and infant environment is known to be critical for the development of adverse health effects later in life [4]. Fetal and early postnatal development constitutes the most vulnerable stages with regard to adverse effects of some toxicants [5]. It has been reported that one in every six children has a developmental disability and in most cases these disabilities affect the nervous system [6]. The most common neurodevelopmental disorders include learning disabilities, sensory deficits, developmental delays, and cerebral palsy [6]. Treatment of these disorders is difficult and the disabilities may be permanent [3]. There is increasing evidence for adverse effects of early-life exposure to toxic chemicals on organ function late in life [4]. Thus, it is plausible that arsenic (As) and its metabolites, which easily pass to the fetuses, may contribute to such effects. Moreover, arsenic has been recognized as a cause of neurodevelopmental disorders [7,8].

There is a growing body of evidence that the intrauterine or early childhood exposure to arsenic also induces changes that will become apparent much later in life [5,8]. Epidemiological studies indicated that exposure to arsenic during early childhood or *in utero* was associated with an increased morbidity from nervous system disease. Several cross-sectional studies have shown the links between arsenic exposure and neurobehavioral deficits in school children [3]. For example, the incident of the Morinaga dried milk poisoning, where children eating As-contaminated milk powder revealed neurologic diseases, neurobehavioral dysfunction, and decreased cognitive skills [3]. Similar results were also obtained in children with arsenic exposure from a smelter [1]. Such evidence supports the notion that arsenic is a developmental neurotoxicant [3]. The developmental neurotoxic effects of arsenic are dependent on dose, route, and the day of gestation when exposure occurs [4]. It has been reported that a significantly lower dose of arsenic was sufficient to induce apoptosis in neuronal cells as compared to nonneuronal cells, illustrating the sensitivity of the nervous system to this metal [9]. It was reported that rats exposed to arsenic during development presented deficits in neurological function, suggesting neurodevelopmental risk with low concentrations [4]. Research has indicated that prenatal arsenic exposure increases the risk of adverse effects during early childhood [5] and produces developmental toxicity, including malformations, decreased prenatal rate of growth, and increased mortality and neural tube defects [3]. Some studies in neuronal cells exposed to arsenic showed decreased cell viability, altered cell growth, and apoptosis [10,11] reflecting mechanisms by which low doses of arsenic may lead to neurodevelopmental toxicity.

In this chapter, the accumulation of arsenic in fetal brain tissue, its effects on neurodevelopment and neurobehavior, and the mechanisms of As-induced developmental neurotoxicity are stated. Moreover, effects of developmental arsenic exposure on fetal programming of the nervous system are also discussed.

16.2 Accumulation of Arsenic in Fetal Brain Tissue

Although the placenta serves as a natural barrier between mother and fetus, it has been fully demonstrated that arsenic readily passes the placenta in humans [12–14] and in other mammals [15,16]. Concha et al. reported that the median maternal and cord blood arsenic levels in northern Argentina were $11 \mu g/L$ and $9 \mu g/L$, respectively [13]. Hall and colleagues reported that the mean maternal and cord blood arsenic levels in Matlab, Bangladesh, were $11.9 \mu g/L$ and $15.7 \mu g/L$, respectively [14]. These investigations suggested that the level of exposure of the fetus to arsenic is similar to that in the mother. Research by Concha et al. indicated that in contrast to the free passage of arsenic over the placenta to the fetus, the passage over the mammary gland is limited, and little arsenic is excreted in breast milk [13]. Working in Argentina, these researchers found a decrease in arsenic concentrations in the urine of the infants from about $80 \mu g/L$ during the first 2 days of life to less than $30 \mu g/L$ at 4 months of age [13]. This research has showed that the main route of arsenic into the developing organism is transplacental exposure *in utero*.

Once in the developing body, both arsenic and its methylated metabolites (MMA, DMA) cross the immature blood-brain barrier (BBB) easily and arrive at the brain, which has been

proved in animal experiments. Jin et al. treated maternal mice with arsenic-containing drinking water at concentrations of 0, 10, and 30 ppm for the entire pregnancy [17]. It was detected that the levels of total arsenic in the brains of newborn mice were <0.01 µg As/g in controls, and 0.07 µg/g in the 10 ppm and 0.22 µg/g in the 30 ppm groups, respectively. In terms of content, the level of total arsenic in the brain of newborn mice was similar to that in their own liver, and was far higher than that in the brain of their mothers. Xi et al. administered arsenic to rats from day 6 of gestation (GD 6) until postnatal day 42 (PND 42) at concentrations of 0, 10, 50, and 100 mg/L. It was shown that total arsenic content in brain tissue of pups on PND 42 was significantly increased in all three As-exposed groups compared with control [18]. A dose-related trend was found between total brain arsenic content and arsenic concentrations in drinking water. This indicated that the protection of the immature BBB against arsenic passing from blood to brain is very limited. Together with imperfect eliminating function of developing kidney, all these factors result in relatively high levels of arsenic in the brains of early-life organisms.

16.3 Effect of Arsenic on the Development of the Nervous System

The normal developmental processes of the nervous system are an outcome of extremely precise, interdependent cellular and molecular events such as protein synthesis, membrane development associated with axonal extension, dendritic outgrowth, synaptogenesis, and response to growth factors. Any type of alteration could affect the final outcome of this complex coordination [19]. It has been demonstrated that arsenic exposure during the developmental period leads to structural alterations in neuronal circuitry, and these provide the morphological evidence of structural alterations and functional deficits encountered in the later period of life.

Neurodevelopmental toxicity is shown in Table 16-1. Wlodarczyk et al. reported that after pregnant LM/Bc dams were injected intraperitoneally on GD 7 and GD 8 with 40 mg/kg of arsenate, the exencephaly was shown in 90 to 100% of the exposed fetuses [20]. Hill et al. found that, after maternal intraperitoneal exposure to sodium arsenate at 9.6 mg/kg in LM/Bc/Fnn mice on GD 7.5 and 8.5, neural tube defects were induced in all mice pups [21]. Similar results were also reported by Wlodarczyk et al. [22]. Nagaraja and Desiraju observed reduced body and brain weight in rats following arsenic administration from days 2 to 60 after birth and drew an association between decreased body and brain weight during the developmental period of the exposed animals [23]. Tyler et al. found that exposure to a low concentration of 50 ppb arsenic over the first three trimester equivalents significantly decreased the number of differentiated neural progenitor cells in adult mice. Additionally, prenatal arsenic exposure altered 31% of target neurogenesis-related genes as well, including several involved in growth and differentiation in adult animals [24]. Dhar et al. reported that exposure to arsenic in the stage of rapid brain growth in rats was associated with defects in migration, delayed maturation, and alteration in the nuclear area of the Purkinje cells of the cerebellum. Developmental arsenic exposure alters cerebellar morphology in the brain [19].

First Author and Year	Species	Treatments	Reported Effects
Single-or multip	le-day administratio	n during embryogenesis	
Finnell et al. (1996) [20]	mice (LM/BC)	Arsenate, single i.p. injection, GD 7:12 (day: hour) and 8:12, 40 mg/kg	40 mg/kg: increased % malformed fetuses (90–100%), especially neural-tube defect
Hill et al. (2009) [21]	mice (LM/BC)	Sodium arsenate, single i.p. injection, GD 7.5 and 8.5, 9.6 mg/kg (fetuses assessed GD 9.0)	100% NTDs
Wlodarczyka et al. (2000) [22]	mice (STOCK- Folr2tm Fnn null)	Arsenate, single i.p. injection, 9.6 mg/kg, E 7:12 and 8:12 (assessed 3 h, 12 h, 24 h later)	Developmentally delayed embryos; most with closed neural tube only at closure site I; significantly fewer somites
Nagaraja and Desiraju (1994) [23]	Rat (Wistar)	Arsenate, 5 mg/kg/day, PND 2 to 60	Decreased body and brain weights
Tyler and Allan, (2013) [24]	mice (C57BL/6J)	Arsenate, 50 ppb, drinking water, GD 0 to PND 23–25 (fetuses assessed on GD 34, 35, 63, 70)	41% decrease in number of differentiated neural progenitor cells, 31% alteration of target neurogenesis- related genes
Short- or long-te	erm arsenic exposure	e in vitro	
Frankel et al. (2009) [25]	PC12 cells	Arsenite, 0, 0.5, 1, 2.5, 5, 10 mM cells (assessed 48, 72, and 120 hours later)	Reduced neurite production, outgrowth and complexity in newly differentiating PC12 cells; inhibited further neurite development in more mature PC12 cells
Chattopadhyay et al. (2002) [11]	human fetal brain explants ^a rat neonatal brain explants ^b	Arsenite 0.3 mg/L, 24 h ^a ; arsenite, 0.3 mg/L, drinking water, GD 0 to PND 1 ^b	^a Cytoplasm vacuoles, frothing and nuclear condensation with the appearance of intact membrane ^b Intact membrane, large vacuoles and loss of cell–cell junction
Namgung and Xia, (2000) [26]	Primary cortical neurons from newborn SD rats	Arsenite 0, 2, 5, 7, 10μM; 12, 24, 48h	Apoptosis; As concentration and time dependence
Sidhu et al. (2005) [27]	Primary embryonic rat midbrain (GD 12) neuroepithelial cells	Arsenite 0, 0.5, 1, 2, and 4μM; 48h	Cell viability reduction and cell cycle progression, As concentration and time dependence, 4 µM 24h: apoptosis

Table 16–1	Experimental	Studies on	Arsenic Neuro	developmenta	l Toxicity
------------	--------------	------------	---------------	--------------	------------

GD: gestation day; PND: postnatal day: NTDs: neural tube defects.

A series of morphological changes were also found in developing neurons subsequent to arsenic exposure *in vitro*. Observation on human fetal brain explants exposed to arsenic showed cytoplasm vacuoles, frothing, and nuclear condensation with the appearance of intact membrane on day 12. Exposure to arsenic for 18 days resulted in signs of cellular degeneration, appearance of large vacuoles, and loss of cell-cell junction [11]. Arsenic exposure induced reduction in the viability of embryonic primary rat midbrain neuroepithelial cells with concentration- and time-dependent effects *in vitro* [27]. Namgung and Xia [28] and Chattopadhyay et al. [10] reported that primary neonatal rat neural cells exposed to arsenic showed a reduction of neuronal viability and morphological changes. Frankel et al. reported that arsenic has concentration- and time-dependent effects on initial neurite outgrowth *in vitro*. Exposure to low micromolar levels of arsenite for 5 days resulted in reduced neurite production and outgrowth and complexity in newly differentiating PC12 cells. Furthermore, it was found that exposure of more mature PC12 cells to arsenite could inhibit further neurite development. These results indicated that exposure to arsenic disrupts early stages of neuron differentiation [25].

Neurite outgrowth is an indispensable process for proper development of the nervous system. Abnormal formation of neurites during development is correlated with developmental disabilities and impaired behavioral functions in rodents and primates, including humans. Wang et al. demonstrated that arsenic inhibits neurite outgrowth by inhibiting the LKB1-AMPK signaling pathway. Oxidative stress induced by As, especially excessive super-oxide, plays a critical role in blocking the LKB1-AMPK pathway [29]. Neurite outgrowth is mainly orchestrated by three cytoskeletal components: microtubules, microfilaments, and neurofilaments. The coordinated rearrangement of cytoskeletal components, especially microtubules and microfilaments, plays a key role in the transition of neurons from an undifferentiated state to neurite-bearing morphology [30]. NaAsO₂ increased the mRNA levels of the light and medium subunits of neurofilaments and decreased the mRNA levels of tau and tubulin in a dose-dependent manner in SCAT3-expressing cells. The area ratio of neurite to cell body was significantly reduced by NaAsO₂. The changes in cytoskeletal gene expression are likely responsible for the inhibitory effects of NaAsO₂ on neurite outgrowth [30].

16.4 Effect of Arsenic on Neurobehavior

It was reported that arsenic exposure induced neural and behavioral changes and the alterations caused were correlated with the exposure dose, the duration of exposure, and the experimental conditions. Moreover, the animals exposed to arsenic had taken longer to acquire the learned behavior and to extinguish the operant [31]. Epidemiological studies on adverse effects of exposure to arsenic on neurobehavior are included in Table 16-2. Epidemiological studies indicated that arsenic exposure correlated with various learning and cognitive deficits in children in Bangladesh [32–35], West Bengal [45], Thailand [46], India [17], China [37,38], and Mexico [39-41] as well as in adults in Texas, USA [42] (Table 16-2). The intellectual function of 529 children aged 6-9 years in the As-exposed region of Thailand was assessed by using the Wechsler Intelligence Scale for Children (WISC). The percentage of children in the average intelligence quotient (IO) group decreased remarkably from 56.8 to 40.0 as the arsenic level increased. Moreover, a statistically significant relationship with arsenic levels could explain 14% of variance in children's IQ was observed after adjusting for confounders [46]. Meta-analysis suggested that a 50% increase of arsenic levels in urine would be associated with a 0.4 decrease in the IQ of children aged 5–15 years [47]. Rocha-Amador et al. observed a decrease in full scale IQ in children aged 6-10 years when total arsenic levels were increased in drinking water [41].

Location	Age	Sample Size	Study Design	Confounders Accounted for	Exposure Measure
Bangladesh	10	201	Cross-	Maternal education, maternal intelligence, house	Water,
India	5–15 years	351	Cross- sectional	Age, sex, maternal and paternal education, father's occupation, number of rooms in the house, type of house building material, BMI, mother's age	Water, Urine
Bangladesh	5 years	2260	Longitudinal cohort	Age, HOME, father's education, mother's BMI and IQ, assets, housing, number of children in the household, gestational age, birth length, concurrent HAZ and testers	Maternal urine at pregnancy Child urine
Bangladesh	8–11 years	299	Cross– sectional	Maternal intelligence, maternal age, school months, head circumference, plasma ferritin and blood manganese	Blood
India	5–15 years	351	Cross- sectional	Age, sex, maternal and paternal education, father's occupation, number of rooms in the house, type of house building material, BMI, and mother's age	Water, Urine
China	8–12 years	720	Cross- sectional	None	Water, Urine
China	4.5 years	106	Prospective cohort	Cord blood Pb, maternal age, height, weight, gestational weeks, maternal education, method of delivery, breastfeeding, nursery school age, tobacco exposure and family income	Placenta, maternal blood, cord blood
Mexico	6–9 years	41	Cross- sectional	Sex, age, socioeconomic status and parent's education	Urine
Mexico	6–8 years	602	Cross- sectional	Age, sex, mother's school level, Hb, Pb and for Pb \times AsU interaction	Urine
Mexico	6–10 years	132	Cross- sectional	Blood Pb, socioeconomic status, mother's education, height-for-age z-score and transferrin saturation	Water, Urine
USA	40–96 years	434	Community- based participatory research	Age, sex, education, ethnicity, APOE4 genotype, residence	Water
USA	11–13 years	31	Cross- sectional	Sex, maternal education	Hair
Taiwan	13–14 years	109	Cross- sectional	Sex, education	Water

Table 16–2Summary of Epidemiological Studies Assessing NeurobehavioralDisorders from Arsenic Exposure

^aMedian.

^bInterquartile range.

^cGeometric mean.

^d10th and 90th percentiles; AsH: arsenic in hair; AsN: arsenic in nails; AsS: arsenic in soil; AsU: arsenic in urine; AsW: arsenic in drinking water; BMI: body mass index; DMA: dimethylarsinic acid; HAZ: height-for-age z-score; Hb: hemoglobin;

In-As: inorganic arsenic; IQ: intelligence quotient; MMA: monomethylarsonic acid; MQ: methodological quality; Pb: lead; T-As: total arsenic; WHZ: weight-for-height z-score; CBPR: community-based participatory research; DPAA: diphenylarsinic acid.

Type of Arsenic	Mean \pm SD (range)	Psychological Test	Observed Effect	MQ	Reference
T-As	118 ± 145 μg/L water 116.6 ± 148.8 μg/L urine	WISC-II	AsW↓ Full-scale and performance IQ	High	[32]
T-As	147 ± 322 μg/L water (1–2480) 78 ± 61 μg/L urine (2–375)	WISC-III CRT SBIS	AsU↓ Vocabulary test, object assembly test and picture completion test	High	[33]
In-As T-As	84μ g/L maternal urine ^{a,b} (26–415) ^d 51 μ g/L Child urine ^a (20–238) ^d	WPPSI-III	AsU↓ Full-scale and verbal IQ in girls, but not in boys	High	[34]
T-As	4.81 ± 3.22 μg/L	WISC-IV	AsB↓ Full-scale score, verbal comprehension and working memory	High	[35]
T-As	$147 \pm 322 \mu$ g/L water (1–2480) 78 ± 61 μ g/L urine (2–375)	WISC-III CRT SBIS	AsU↓ Vocabulary test, object assembly test and picture completion test	High	[36]
T-As	$190 \pm 183 \mu$ g/L water 73 ± 3 μ g/L urine	CRT-RC2	AsU↓ Full-scale IQ	High	[37]
T-As	0.15 μg/g placenta ^a 1.80 μg/L maternal blood ^a 0.60 μg/L cord blood ^a	WPPSI-R	CdCBJ Full-scale and performance IQ	High	[38]
T-As	62.9 ± 0.03 μg/g crea (27.5–186.2)	WISC-RM	AsU1 Full-scale and verbal IQ	Medium	[39]
In-AMMA, DMA	58.1 ± 33.2 µg/L ^c	WISC-RM NLS CAT	AsU↓ Digit span subscale, letter sequencing, visual search	High	[40]
T-As	$194 \pm 1.3 \mu$ g/L water ^c 116 $\pm 2.2 \mu$ g/g crea ^c	WISC-RM	AsW↓ Full-scale, performance and verbal IQ AsU↓ Full-scale IQ	High	[41]
T-As	2.19–15.26 µg/L water	CBPR	AsW1 Language, visuospatial skills, executive functioning, global cognition, processing speed, and immediate memory	High	[42]
T-As	0.018 ± 0.014 µg/g (0.001-0.055)	WASI CVLT-C WRAML	AsH↓ Full-scale and verbal IQ and memory test	Low	[43]
T-As	184.99 ± 225.29 μg/L	NES2-T	AsW1 Pattern memory and switching attention	High	[44]

Wang et al. also obtained a similar result when arsenic levels were increased in drinking water, in 720 children aged 8-15 years in rural villages in Shanyin county, Shanxi province, China. Wasserman et al. also documented the same association with total arsenic levels in water in relation to 6- and 10-year-old children's intellectual function [32,33]. In the study of 201 children in Araihazar, Bangladesh, it was found that exposure to arsenic from drinking water was associated with reduced intellectual function in WISC III for children, in a dose-response manner after adjustment for sociodemographic covariates and water Mn. Children with water arsenic levels >50 mg/L achieved significantly lower performance and full-scale scores than did children with water arsenic levels <50 mg/L. The association was generally stronger for well water arsenic than for urinary arsenic [32]. In a later study conducted in children aged 8-11 years, a decrease in full scale IQ score, verbal comprehension and working memory was observed, associated with increased levels of total arsenic in blood [35]. Wright et al. also observed a negative effect on fullscale and verbal IQ when they studied total arsenic levels in hair among children aged 11-13 years [43]. Developmental and continuous arsenic exposure induced significant deficits in verbal IQ and long-term memory in children aged 6–9 years, as measured by the WISC [39]. A crosssectional study was conducted for examining cognitive function in 49 adolescents exposed to high and in 60 controls exposed to low levels of arsenic in drinking water in Taiwan. The results showed that memory and switching attention were significantly affected by long-term cumulative exposure to arsenic after adjusting for education and sex [44]. Von Ehrenstein et al. conducted a cross-sectional study among 351 children age 5 to 15 years in West Bengal, India. They found associations between arsenic concentration in the children's urine and intellectual function assessed using six subtests from the WISC as well as using the Total Sentence Recall test, the Colored Progressive Matrices test, and a pegboard test. Moreover, there were also significant relations between urinary arsenic and reductions in the adjusted scores of the vocabulary test, the object assembly test, and the picture completion test [36]. Neurobehavioral outcomes are influenced by the age at examination and many other covariates, such as nutrition. The exposure to arsenic and factors influencing the susceptibility at the time of the study may be very different from that occurring prenatally or in early childhood. Therefore, longitudinal studies are warranted for evaluation of late effects of early-life exposure. No such studies have yet been reported [5]. But it was found that infants who survived the Morinaga milk arsenic poisoning incident had highly raised risks of neurological disease during adult life [48].

Transplacental and early-life exposure to inorganic arsenic affected learning and memory functions and neuromotor reflex in offspring rats [49]. After administration of 100 mg/L diphenylarsinic acid (DPAA) in drinking water for 14–21 days, young rats showed behavioral abnormalities, such as gait disturbance and deficient acclimatization to a novel condition in the open field test [50]. Markowski et al. assessed the effects of prenatal exposure to arsenite on motor and food-motivated behaviors from birth to adulthood in C57BL6/J mice. The results showed that developmental arsenic exposure can produce other behavioral impairments in children in addition to cognitive impairment [51]. Rats exposed to arsenite at concentrations of 37 mg/L drinking water from GD 15 until 4 months of age showed increased spontaneous locomotor activity and alterations in a spatial learning tasks compared to control rats [4]. Rodriguez et al. also reported that spontaneous locomotor activity and alterations in a spatial learning task were deficits among the rats exposed to arsenic during development [52]. Developmental subchronic exposure to diphenylarsinic acid induced transient increased exploratory behavior, impaired learning behavior, and decreased cerebellar glutathione concentration in rats [53]. Exposure of young rats to 5 mg/kg arsenate resulted in increased time to acquire operant learning. At non-maternally toxic levels, arsenic given to pregnant dams via drinking water affected fetal brain development and postnatal behaviors [54]. However, the underlying mechanisms responsible for these behavioral outcomes in adulthood after developmental exposure to arsenic have not yet been elucidated [24].

16.5 Mechanisms for the Effect of Arsenic on the Nervous System During Development

The mechanisms involved in arsenic-induced developmental neurotoxicity likely include inductions of oxidative stress, apoptosis, and oxidative DNA damage, effects on neurotransmitters and the neuroendocrine and cell cycle, perturbation of one-carbon metabolism, epigenetic effects, etc. The major mechanisms are discussed as follows.

16.5.1 Induction of Oxidative Stress

Oxidative stress is defined as a disturbance in the balance between the production of reactive oxygen species (ROS) and antioxidant defenses. Oxidative stress is believed to cause a number of diseases. The oxidative stress induced by chronic exposure to arsenic is related to cytotoxic and genotoxic effects in the cells, playing a crucial role in the pathogenesis of nervous system disorders [55]. One of the principal mechanisms of arsenic toxicity is the induction of a strong oxidative stress with production of ROS in cells [56]. Several studies have shown that cell populations chronically exposed to arsenic have significant oxidative stress that, in turn, induces DNA damage [57,58], as well as lipid peroxidation and decreased glutathione levels [59,60]. There is increasing evidence that the induction of ROS plays a crucial role in arsenic neurotoxicity. Arsenic exposure was found to induce an increase in the production of ROS in human fetal brain explants [11]. Chattopadhyay et al. also reported that the human fetal brain cells that were exposed to arsenic in culture and neonatal rat brain cells derived from arsenic-treated mothers showed an increase in NO and ROS [10].

Arsenic may induce the production of ROS by impacting on antioxidant enzymes. It was reported that arsenic exposure resulted in a significant decrease in the activity levels of antioxidant enzymes such as manganese-superoxide dismutase (Mn-SOD), Cu/Zn superoxide dismutase (Cu/Zn-SOD), catalase (CAT), and glutathione peroxidase (GPx), while malondialdehyde (MDA) levels were significantly increased in the cerebral cortex, hippocampus, and cerebellum of developing rats [61]. Kadeyala et al. demonstrated that arsenic exposure significantly decreased the activities of SOD isoforms, CAT, GPx, and glutathione reductase (GR) with an increase in glutathione-S-transferase (GST), while lipid peroxidation (LPx) and arsenic levels were significantly increased in different brain regions. The induced alterations in these parameters were more pronounced in the cerebral cortex [62]. Gupta and Flora reported that the rats exposed to arsenic showed significant increases in oxidized glutathione (GSSG) levels in the brain [63]. However, the activities of brain SOD and CAT decreased marginally on arsenic exposure [63]. Developmental subchronic exposure to DPAA also induced reduction in the concentration of cerebellar GSH, which might be a primary cause of oxidative stress [53]. Comprehensive gene expression analyses in primary cultured rat cerebellar cells exposed to arsenic revealed significant alterations in the mRNA expression of genes encoding antioxidative stress proteins (heme oxygenase 1 and heat shock protein 72) [50]. Zhang et al. demonstrated that high-level arsenic induces severe redox imbalance by decreasing the levels of GSH and increasing the levels of ROS through the oxidative stress adaptor p66^{Shc} [65]. Knockdown of p66^{shc} improves the developmental competence of As-exposed embryos *in vitro* by increasing the resistance to oxidative stress. It is suggested that p66Shc-linked redox imbalance contributes to As-induced developmental retardation in mouse preimplantation embryos [64]. Zhang et al. showed that p66Shc-linked redox imbalance and metabolic abnormality of amino acids promote developmental delay in arsenite-exposed mouse preimplantation embryo. An antioxidant, N-acetyl-L-cysteine, improves the development of arsenite-exposed embryos by reducing intracellular ROS and adjusting amino acid metabolism. These results support further that redox imbalance is involved in arsenite-induced embryonic toxicity [65].

16.5.2 Induction of Apoptosis

It is well known that apoptosis is a form of cell death, also known as programmed cell death. Apoptosis can be triggered by exogenous and endogenous stimuli such as oxidative stress and genotoxic chemicals. It is well known that Bcl-2 family proteins and the caspase protein family play important roles in regulating the mitochondrial (intrinsic) apoptotic pathway [66,67]. Following an intrinsic death signal, proapoptotic Bcl-2 members, such as Bax, increase, and antiapoptotic Bcl-2 members, such as Bcl-2, decrease, which subsequently increases mitochondrial permeability transitions, cytochrome c release, and caspase 3 activation. The mitochondrial apoptotic pathway indicators, including mitochondrial membrane potential, cytochrome c protein, and caspase 3 activation, were all significantly changed correlating to the apoptosis trigger. A growing body of evidence indicates that abnormal apoptosis may be involved in the neurotoxicities induced by some neurotoxicants. Experiments on human fetal brain explants on exposure to arsenic in culture showed disturbance in lipid peroxidation, ROS, and apoptosis [11]. Namgung and Xia reported that apoptosis was induced in primary cerebellar and cortical neurons which were exposed to 5, 10 and 15μ M sodium arsenite [28]. In vitro studies showed that arsenite-induced apoptosis in cortical neurons prepared from newborn mice [68] and C17.2 cell lines originally derived from the developing mouse cerebellum neuronal cells [69], suggesting that arsenate exhibits neurotoxicity by inducing apoptosis. Arsenite leads to an increase in intracellular Ca levels and generation of ROS, which may cause a decrease in mitochondrial transmembrane potential, release of cytochrome c, and consequent activation of caspases. A slight activation of calpain also takes place, which might favor activation of the mitochondrial pathway or might activate other pathways [69]. Arsenic exposure resulted in an increase in mRNA expression levels of caspase-3 and caspase-9 in brain regions (cerebral cortex, hippocampus) and cerebellum of developing rats [61]. Namgung and Xia reported that arsenic-induced apoptosis in cortical neurons was mediated by c-Jun N-terminal protein kinase 3 and p38 mitogen-activated protein kinase [26]. These investigators also reported that sodium arsenite selectively activated p38 and JNK3, but not JNK1 or JNK2, to induce apoptosis in cerebellar neurons from 6 to 8-day-old rats [28]. Blocking the p38 or JNK signaling pathways using the inhibitors SB203580 or CEP-1347 protected cerebellar neurons against arsenite-induced apoptosis. These data suggest that arsenite neurotoxicity may be at least partly due to apoptosis caused by activation of p38 and JNK3 MAP kinases [28].

16.5.3 Effect of Arsenic on the Cell Cycle

Arsenic exposure disrupts cell cycle dynamics of neuroepithelial cells *in vitro* [19,27]. It was reported that arsenic interferes with cell regulatory control via the induction of p53 and Bcl-2 [20]. p53 is a critical regulator of the cell cycle and has been shown to undergo a dose- and time-dependent induction in response to arsenic [70,71]. Induction of p53 in response to As_2O_3 exposure was associated with subsequent cell cycle arrest at gap 2 phase (G2)/mitosis (M) [72–74]. It has been shown that arsenic induces p21, a critical cell cycle regulatory protein. Moreover, it was reported that the induced p21 by arsenic blocked cell cycle progression at G2/M [75] as well as G1/S (synthesis phase) [76]. Cell cycle arrest at G2/M induced by arsenic [73,77] has been associated with subsequent apoptosis, indicating the latter may occur due to As-mediated cell cycle inhibition. These studies highlight the impact of arsenic on cell cycle progression and its potential link to arsenic-associated neurodevelopmental effects. Sidhu et al. reported a dose-responsive delay in cell cycle progression in primary embryonic neuro-epithelial cells exposed to arsenic [27]. They also suggested that cell cycle arrest may contribute to the neurodevelopmental toxicity induced by this metal.

16.5.4 Effect on Central Neurotransmitters/Neuroendocrine

It has been reported that the neurotransmitter disrupting effects of arsenic are likely to occur at very low concentrations and to have long-term consequences, particularly if these effects are induced early in life. Moreover, it was demonstrated that developmental exposure to arsenic alters components of the hypothalamus-pituitary-adrenal (HPA) axis including increased hypothalamic corticotrophin releasing hormone (CRH), altered corticosterone (CORT) secretion, decreased hippocampal 11b-HSD 1, and altered subcellular glucocorticoid receptor (GR) distribution in the hypothalamus [78]. Arsenic has been shown to perturb components of the HPA stress axis, such as glucocorticoid receptors, and to alter gene expression of inducible genes. Perturbation of the HPA stress axis has been implicated in both cognitive damage and the promotion of carcinogenesis [79]. Mice given water with $50 \,\mu$ g/L arsenic during the prenatal period showed impaired regulatory interactions between the HPA axis and the serotonergic system in the dorsal hippocampus at adult age, indicating predisposition for depressive-like behavior, which was also indicated in behavioral studies [80]. There was also aberrant expression of genes related to insulin growth factor-signaling pathways [81].

It is reported that after weaning mice were exposed to arsenic over a period of 9 weeks, dopamine was significantly decreased in the hypothalamus [82]. Huo et al. reported that immature rats exposed to realgar showed significant changes of amino acid neurotransmitters, including aspartate, glutamate, glutamine, homocysteine, serine, glycine, γ-aminobutyric acid, and taurine, in the brain. The results indicated that the neurotoxicity induced by realgar may be associated with its effects on amino acid neurotransmitters [83]. Rat pups were exposed to arsenic from the second day after birth up to 60 days of age. The results showed that glutamic acid decarboxylase (GAD) activity and gamma-aminobutyric acid (GABA) levels were reduced in the cerebellum, hypothalamus, and brainstem [84]. Chandravanshi et al. investigated the changes in brain cholinergic receptors and acetylcholinesterase activity in male rats exposed to arsenic from PD 22 to PD 59 [45]. They found the decrease in the binding of muscariniccholinergic receptors (CHRM2) in frontal cortex and hippocampus was associated with reduced CHRM2 mRNA levels, acetylcholinesterase (AChE) activity, and expression of choline acetyltransferase (ChAT) and protein kinase C β -1. Moreover, the cholinergic alterations and AChE modifications were found to be linked with increased arsenic levels in the frontal cortex and hippocampus [45]. Nagaraja and Desiraju reported that exposure of young rats to arsenate resulted in increased time to acquire operant learning and decreased AChE activity in the brain [23]. It was reported that arsenic induced aberrant estrogen receptor signaling in pregnant mice, which may affect early-life genetic programming [5].

16.5.5 Other Mechanisms

Epigenetic mechanisms are crucial to regulate the expression of different genes required for neuronal plasticity. Neurotoxic substances such as As, which induces cognitive deficits in exposed children before any other manifestation of toxicity, could interfere with the epigenetic modulation of neuronal gene expression required for learning and memory. There is evidence that arsenic interferes with DNA methylation [85,86] and global histone acetylation [87] at very low exposure levels. The effects of arsenic may be implicated in long-term fetal programming. Martínez et al. assessed the effects of arsenic on DNA methylation patterns in the hippocampus and frontal cortex of treated rats from gestation until 4 months of age. Their results demonstrated alterations of the methylation pattern of genes involved in neuronal plasticity in an animal model of memory deficit associated with arsenic exposure [88]. Cronican et al. exposed C57Bl6/J mice to 100 mg/L arsenic in the drinking water starting 1 week before conception until birth. They found that arsenic exposure during embryonic life caused global hypoacety-lation at H3K9 in brain samples from exposed pups [89].

Neural cell adhesion molecules (NCAMs) play critical roles during the development of the nervous system. Polysialylation of NCAM (PSA-NCAM) is a critical functional feature of NCAM-mediated cell interactions and functions. Luo et al. investigated the effects of maternal and early-life arsenite exposure on NCAM and PSA-NCAM in rat offspring. The results showed that maternal arsenite exposure increased the expression of PSA-NCAM, NCAM, and polysialyltransferases in the hippocampus of rat offspring on PND 21 and PND 120, which indicated that the increased expression of these molecules might contribute to impaired neurodevelopment following arsenite exposure [2].

16.6 Conclusions and Future Directions

Taken together, epidemiological investigations and animal experiments indicate that chronic arsenic exposure can induce developmental neurotoxicity. Although the molecular mechanisms are still poorly understood, As-induced neurodevelopmental disorders may be associated with these mechanisms as discussed above. These data suggest that developmental arsenic exposure can have long-lasting effects on the nervous system well into adolescence, consistent with the fetal basis of adult disease hypothesis [90], which postulates that many adult diseases have a fetal origin. Further studies detailing the molecular effects of arsenic on the central nervous system during development are essential to explain further the mechanism of As-induced neurodevelopmental disorders including learning and memory deficits.

References

- [1] Dakeishi M, Murata K, Grandjean P. Long-term consequences of arsenic poisoning during infancy due to contaminated milkpowder. Environ Health 2006;5:31–7.
- [2] Luo J, Qiu Z, Chen J, Zhang L, Liu W, Tan Y, et al. Maternal and early life arsenite exposure impairs neurodevelopment and increases the expression of PSA-NCAM in hippocampus of rat offspring. Toxicology 2013;311:99–106.
- [3] Grandjean P, Murata K. Developmental arsenic neurotoxicity in retrospect. Epidemiology 2007;18:25-6.
- [4] Golub MS, Macintosh MS, Baumrind N. Developmental and reproductive toxicity of inorganic arsenic: animal studies and human concerns. J Toxicol Environ Health B Crit Rev 1998;1:199–241.
- [5] Vahter M. Health effects of early life exposure to arsenic. Basic Clin Pharmacol Toxicol 2008;102:204-11.
- [6] Boyle CA, Decouflé P, Yeargin-Allsopp M. Prevalence and health impact of developmental disabilities in US children. Pediatrics 1994;93:399–403.
- [7] Grandjean P, Landrigan PJ. Developmental neurotoxicity of industrial chemicals. Lancet 2006;368:2167-78.
- [8] Grandjean P, Landrigan PJ. Neurobehavioural effects of developmental toxicity. Lancet Neurol 2014;13:330–8.
- [9] Wang TS, Kuo CF, Jan KY, Huang H. Arsenite induces apoptosis in Chinese hamster ovary cells by generation of reactive oxygen species. J Cell Physiol 1996;169:256–68.
- [10] Chattopadhyay S, Bhaumik S, Nag Chaudhury A, Das Gupta S. Arsenic induced changes in growth development and apoptosis in neonatal and adult brain cells in vivo and in tissue culture. Toxicol Lett 2002;128:73–84.
- [11] Chattopadhyay S, Bhaumik S, Purkayastha M, Basu S, Nag Chaudhuri A, Das Gupta S. Apoptosis and necrosis in developing brain cells due to arsenic toxicity and protection with antioxidants. Toxicol Lett 2002;136:65–76.
- [12] Guan H, Piao F, Zhang X, Li X, Li Q, Xu L, et al. Prenatal exposure to arsenic and its effects on fetal development in the general population of Dalian. Biol Trace Elem Res 2012;149:10–15.
- [13] Concha G, Vogler G, Lezcano D, Nermell B, Vahter M. Exposure to inorganic arsenic metabolites during early human development. Toxicol Sci 1998;44:185–90.

- [14] Pilsner JR, Hall MN, Liu X, Ilievski V. Influence of prenatal arsenic exposure and newborn sex on global methylation of cord blood DNA. PLOS ONE 2012:e37147.
- [15] Lindgren A, Danielsson BR, Dencker L, Vahter M. Embryotoxicity of arsenite and arsenate: distribution in pregnant mice and monkeys and effects on embryonic cells in vitro. Acta Pharmacol Toxicol (Copenh) 1984;54:311–20.
- [16] Willhite CC, Ferm VH. Prenatal and developmental toxicology of arsenicals. Adv Exp Med Biol 1984;177:205–28.
- [17] Jin Y, Xi S, Li X, Lu C, Li G, Xu Y, et al. Arsenic speciation transported through the placenta from mother mice to their newborn pups. Environ Res 2006;101:349–55.
- [18] Xi S, Jin Y, Lv X, Sun G. Distribution and speciation of arsenic by transplacental and early life exposure to inorganic arsenic in offspring rats. Biol Trace Elem Res 2010;134:84–97.
- [19] Dhar P, Mohari N, Mehra RD. Preliminary morphological and morphometric study of rat cerebellum following sodium arsenite exposure during rapid brain growth (RBG) period. Toxicology 2007;234:10–20.
- [20] Wlodarczyk BJ, Bennett GD, Calvin JA, Finnell RH. Arsenic-induced neural tube defects in mice: alterations in cell cycle gene expression. Reprod Toxicol 1996;10:447–54.
- [21] Hill DS, Wlodarczyk BJ, Mitchell LE, Finnell RH. Arsenate-induced maternal glucose intolerance and neural tube defects in a mouse model. Toxicol Appl Pharmacol 2009;239:29–36.
- [22] Wlodarczyk BJ, Cabrera RM, Hill DS, Bozinov D, Zhu H, Finnell RH. Arsenic-induced gene expression changes in the neural tube of folate transport defective mouse embryos. Neurotoxicology 2006;27:547–57.
- [23] Nagaraja TN, Desiraju T. Effects on operant learning and brain acetylcholine esterase activity in rats following chronic inorganic arsenic intake. Hum Exp Toxicol 1994;13:353–6.
- [24] Tyler CR, Allan AM. Adult hippocampal neurogenesis and mRNA expression are altered by prenatal arsenic exposure in mice and restored by brief exposure to enrichment. PLOS ONE 2013;8:e73720.
- [25] Frankel S, Concannon J, Brusky K, Pietrowicz E, Giorgianni S, Thompson WD, et al. Arsenic exposure disrupts neurite growth and complexity in vitro. Neurotoxicology 2009;30:529–37.
- [26] Namgung U, Xia Z. Arsenite-induced apoptosis in cortical neurons is mediated by c-Jun N-terminal protein kinase 3 and p38 mitogen-activated protein kinase. J Neurosci 2000;20:6442–51.
- [27] Sidhu JS, Ponce RA, Vredevoogd MA, Yu X, Gribble E, Hong SW, et al. Cell cycle inhibition by sodium arsenite in primary embryonic rat midbrain neuroepithelial cells. Toxicol Sci 2006;89:475–84.
- [28] Namgung U, Xia Z. Arsenic induces apoptosis in rat cerebellar neurons via activation of JNK3 and p38 MAP kinases. Toxicol Appl Pharmacol 2001;174:130–8.
- [29] Wang X, Meng D, Chang Q, Pan J, Zhang Z, Chen G, et al. Arsenic inhibits neurite outgrowth by inhibiting the LKB1-AMPK signaling pathway. Environ Health Perspect 2010;118:627–34.
- [30] Aung KH, Kurihara R, Nakashima S, Maekawa F, Nohara K, Kobayashi T, et al. Inhibition of neurite outgrowth and alteration of cytoskeletal gene expression by sodium arsenite. Neurotoxicology 2013;34:226–35.
- [31] Rodríguez VM, Carrizales L, Jiménez-Capdeville ME, Dufour L, Giordano M. The effects of sodium arsenite exposure on behavioral parameters in the rat. Brain Res Bull 2001;55:301–8.
- [32] Wasserman GA, Liu X, Parvez F, Ahsan H, Factor-Litvak P, van Geen A, et al. Water arsenic exposure and children's intellectual function in Araihazar, Bangladesh. Environ Health Perspect 2004;112:1329–33.
- [33] Wasserman GA, Liu X, Parvez F, Ahsan H, Factor-Litvak P, Kline J, et al. Water arsenic exposure and intellectual function in 6-year-old children in Araihazar, Bangladesh. Environ Health Perspect 2007;115:285–9.
- [34] Hamadani JD, Tofail F, Nermell B, Gardner R, Shiraji S, Bottai M, et al. Critical windows of exposure for arsenic-associated impairment of cognitive function in pre-school girls and boys: a population-based cohort study. Int J Epidemiol 2011;40:1593–604.
- [35] Wasserman GA, Liu X, Parvez F, Factor-Litvak P, Ahsan H, Levy D, et al. Arsenic and manganese exposure and children's intellectual function. Neurotoxicology 2011;32:450–7.

- [36] von Ehrenstein OS, Poddar S, Yuan Y, Mazumder DG, Eskenazi B, Basu A, et al. Children's intellectual function in relation to arsenic exposure. Epidemiology 2007;18:44–51.
- [37] Wang SX, Wang ZH, Cheng XT, Li J, Sang ZP, Zhang XD, et al. Arsenic and fluoride exposure in drinking water: children's IQ and growth in Shanyin county, Shanxi province, China. Environ Health Perspect 2007;115:643–7.
- [38] Tian LL, Zhao YC, Wang XC, Gu JL, Sun ZJ, Zhang YL, et al. Effects of gestational cadmium exposure on pregnancy outcome and development in the offspring at age 4.5 years. Biol Trace Elem Res 2009;132:51–9.
- [39] Calderón J, Navarro ME, Jimenez-Capdeville ME, Santos-Diaz MA, Golden A, Rodriguez-Leyva I, et al. Exposure to arsenic and lead and neuropsychological development in Mexican children. Environ Res 2001;85:69–76.
- [40] Rosado JL, Ronquillo D, Kordas K, Rojas O, Alatorre J, Lopez P, et al. Arsenic exposure and cognitive performance in Mexican schoolchildren. Environ Health Perspect 2007;115:1371–5.
- [41] Rocha-Amador D, Navarro ME, Carrizales L, Morales R, Calderón J. Decreased intelligence in children and exposure to fluoride and arsenic in drinking water. Cad Saude Publica 2007(Suppl. 4):S579–87.
- [42] O'Bryant SE, Edwards M, Menon CV, Gong G, Barber R. Long-term low-level arsenic exposure is associated with poorer neuropsychological functioning: a Project Frontier study. Int J Environ Res Public Health 2011;8:861–74.
- [43] Wright RO, Amarasiriwardena C, Woolf AD, Jim R, Bellinger DC. Neuropsychological correlates of hair arsenic, manganese, and cadmium levels in school-age children residing near a hazardous waste site. Neurotoxicology 2006;27:210–16.
- [44] Tsai SY, Chou HY, The HW, Chen CM, Chen CJ. The effects of chronic arsenic exposure from drinking water on the neurobehavioral development in adolescence. Neurotoxicology 2003;24:747–53.
- [45] Chandravanshi LP, Yadav RS, Shukla RK, Singh A, Sultana S, Pant AB, et al. Reversibility of changes in brain cholinergic receptors and acetylcholinesterase activity in rats following early life arsenic exposure. Int J Dev Neurosci 2014;34C:60–75.
- [46] Siripitayakunkait U, Vishudhiphan P, Pradipasen M, Vorapongsathron T. Association between chronic arsenic exposure and children's intelligence in Thailand. In: Chappell WR, Abernathy CO, Calderon RL, editors. Arsenic exposure and health effects. Oxford, UK: Elsevier Science Ltd; 1999. pp. 141–9.
- [47] Rodríguez-Barranco M, Lacasaña M, Aguilar-Garduño C, Alguacil J, Gil F, González-Alzaga B, et al. Association of arsenic, cadmium and manganese exposure with neurodevelopment and behavioural disorders in children: a systematic review and meta-analysis. Sci Total Environ 2013;454–455:562–77.
- [48] Tanaka H, Tsukuma H, Oshima A. Long-term prospective study of 6104 survivors of arsenic poisoning during infancy due to contaminated milk powder in 1955. J Epidemiol 2010;20:430–45.
- [49] Xi S, Sun W, Wang F, Jin Y, Sun G. Transplacental and early life exposure to inorganic arsenic affected development and behavior in offspring rats. Arch Toxicol 2009;83:549–56.
- [50] Negishi T, Takahashi M, Matsunaga Y, Hirano S, Tashiro T. Diphenylarsinic acid increased the synthesis and release of neuroactive and vasoactive peptides in rat cerebellar astrocytes. J Neuropathol Exp Neurol 2012;71:468–79.
- [51] Markowski VP, Reeve EA, Onos K, Assadollahzadeh M, McKay N. Effects of prenatal exposure to sodium arsenite on motor and food-motivated behaviors from birth to adulthood in C57BL6/J mice. Neurotoxicol Teratol 2012;34:221–31.
- [52] Rodríguez VM, Carrizales L, Mendoza MS, Fajardo OR, Giordano M. Effects of sodium arsenite exposure on development and behavior in the rat. Neurotoxicol Teratol 2002;24:743–50.
- [53] Negishi T, Matsunaga Y, Kobayashi Y, Hirano S, Tashiro T. Developmental subchronic exposure to diphenylarsinic acid induced increased exploratory behavior, impaired learning behavior, and decreased cerebellar glutathione concentration in rats. Toxicol Sci 2013;136:478–86.

- [54] Wang A, Holladay SD, Wolf DC, Ahmed SA, Robertson JL. Reproductive and developmental toxicity of arsenic in rodents: a review. Int J Toxicol 2006;25:319–31.
- [55] Faita F, Cori L, Bianchi F, Andreassi MG. Arsenic-induced genotoxicity and genetic susceptibility to arsenic-related pathologies. Int J Environ Res Public Health 2013;10:1527–46.
- [56] Flora SJ. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51:257–81.
- [57] Méndez-Gómez J, García-Vargas GG, López-Carrillo L, Calderón-Aranda ES, Gómez A, Vera E, et al. Genotoxic effects of environmental exposure to arsenic and lead on children in region Lagunera, Mexico. Ann N Y Acad Sci 2008;1140:358–67.
- [58] Sampayo-Reyes A, Hernández A, El-Yamani N, López-Campos C, Mayet-Machado E, Rincón-Castañeda CB, et al. Arsenic induces DNA damage in environmentally exposed Mexican children and adults. Influence of GSTO1 and AS3MT polymorphisms. Toxicol Sci 2010;117:63–71.
- [59] Schoen A, Beck B, Sharma R, Dubé E. Arsenic toxicity at low doses: epidemiological and mode of action considerations. Toxicol Appl Pharmacol 2004;198:253–67.
- [60] Wu MM, Chiou HY, Wang TW, Hsueh YM, Wang IH, Chen CJ, et al. Association of blood arsenic levels with increased reactive oxidants and decreased antioxidant capacity in a human population of northeastern Taiwan. Environ Health Perspect 2001;109:1011–17.
- [61] Sannadi S, Kadeyala PK, Gottipolu RR. Reversal effect of monoisoamyl dimercaptosuccinic acid (MiADMSA) for arsenic and lead induced perturbations in apoptosis and antioxidant enzymes in developing rat brain. Int J Dev Neurosci 2013;31:586–97.
- [62] Kadeyala PK, Sannadi S, Gottipolu RR. Alterations in apoptotic caspases and antioxidant enzymes in arsenic exposed rat brain regions: reversal effect of essential metals and a chelating agent. Environ Toxicol Pharmacol 2013;36:1150–66.
- [63] Gupta R, Flora SJ. Effect of *Centella asiatica* on arsenic induced oxidative stress and metal distribution in rats. J Appl Toxicol 2006;26:213–22.
- [64] Ren K, Li X, Yan J, Huang G, Zhou S, Yang B, et al. Knockdown of p66Shc by siRNA injection rescues arsenite-induced developmental retardation in mouse preimplantation embryos. Reprod Toxicol 2014;43:8–18.
- [65] Zhang C, Liu C, Li D, Yao N, Yuan X, Yu A, et al. Intracellular redox imbalance and extracellular amino acid metabolic abnormality contribute to arsenic-induced developmental retardation in mouse preimplantation embryos. J Cell Physiol 2010;222:444–55.
- [66] Martinou JC, Youle RJ. Mitochondria in apoptosis: Bcl-2 family members and mitochondrial dynamics. Dev Cell 2011;21:92-101.
- [67] Volkmann N, Marassi FM, Newmeyer DD, Hanein D. The rheostat in the membrane: BCL-2 family proteins and apoptosis. Cell Death Differ 2014;21:206–15.
- [68] Wong HK, Fricker M, Wyttenbach A, Villunger A, Michalak EM, Strasser A, et al. Mutually exclusive subsets of BH3-only proteins are activated by the p53 and c-Jun N-terminal kinase/c-Jun signaling pathways during cortical neuron apoptosis induced by arsenite. Mol Cell Biol 2005;25:8732–47.
- [69] Rocha RA, Gimeno-Alcañiz JV, Martín-Ibañez R, Canals JM, Vélez D, Devesa V. Arsenic and fluoride induce neural progenitor cell apoptosis. Toxicol Lett 2011;203:237–44.
- [70] Filippova M, Duerksen-Hughes PJ. Inorganic and dimethylated arsenic species induce cellular p53. Chem Res Toxicol 2003;16:23–31.
- [71] Yeh JY, Cheng LC, Liang YC, Ou BR. Modulation of the arsenic effects on cytotoxicity, viability, and cell cycle in porcine endothelial cells by selenium. Endothelium 2003;10:127–39.
- [72] Zhao S, Tsuchida T, Kawakami K, Shi C, Kawamoto K. Effect of As2O3 on cell cycle progression and cyclins D1 and B1 expression in two glioblastoma cell lines differing in p53 status. Int J Oncol 2002;21:49–55.
- [73] Li X, Ding X, Adrian TE. Arsenic trioxide induces apoptosis in pancreatic cancer cells via changes in cell cycle, caspase activation, and GADD expression. Pancreas 2003;27:174–9.

- [74] Huang HS, Chang WC, Chen CJ. Involvement of reactive oxygen species in arsenite-induced down regulation of phospholipid hydroperoxide glutathione peroxidase in human epidermoid carcinoma A431 cells. Free Radic Biol Med 2002;33:864–73.
- [75] Chou RH, Huang H. Restoration of p53 tumor suppressor pathway in human cervical carcinoma cells by sodium arsenite. Biochem Biophys Res Commun 2002;293:298–306.
- [76] Park WH, Seol JG, Kim ES, Hyun JM, Jung CW, Lee CC, et al. Arsenic trioxide-mediated growth inhibition in MC/CAR myeloma cells via cell cycle arrestin association with induction of cyclin-dependent kinase inhibitor, p21, and apoptosis. Cancer Res 2000;60:3065–71.
- [77] Ma DC, Sun YH, Chang KZ, Ma XF, Huang SL, Bai YH, et al. Selective induction of apoptosis of NB4 cells from G2+M phase by sodium arsenite at lower doses. Eur J Haematol 1998;61:27–35.
- [78] Goggin SL, Labrecque MT, Allan AM. Perinatal exposure to 50 ppb sodium arsenate induces hypothalamic-pituitary-adrenal axis dysregulation in male C57BL/6 mice. Neurotoxicology 2012;33:1338–45.
- [79] Martinez-Finley EJ, Ali AM, Allan AM. Learning deficits in C57BL/6J mice following perinatal arsenic exposure: consequence of lower corticosterone receptor levels? Pharmacol Biochem Behav 2009;94:271–7.
- [80] Martinez EJ, Kolb BL, Bell A, Savage DD, Allan AM. Moderate perinatal arsenic exposure alters neuroendocrine markers associated with depression and increases depressive-like behaviors in adult mouse offspring. Neurotoxicology 2008;9:47–55.
- [81] Zierler S, Theodore M, Cohen A, Rothman KJ. Chemical quality of maternal drinking water and congenital heart disease. Int J Epidemiol 1988;17:589–94.
- [82] Delgado JM, Dufour L, Grimaldo JI, Carrizales L, Rodríguez VM, Jiménez-Capdeville ME. Effects of arsenite on central monoamines and plasmatic levels of adrenocorticotropic hormone (ACTH) in mice. Toxicol Lett 2000;117:61–7.
- [83] Huo T, Chang B, Zhang Y, Chen Z, Li W, Jiang H. Alteration of amino acid neurotransmitters in brain tissues of immature rats treated with realgar. J Pharm Biomed Anal 2012;57:120–4.
- [84] Nagaraja TN, Desiraju T. Effect of As2O3 on cell cycle progression and cyclins D1 and B1 expression in two glioblastoma cell lines differing in p53 status. Int J Oncol 2002;21:49–55.
- [85] Reichard JF, Schnekenburger M, Puga A. Long term low-dose arsenic exposure induces loss of DNA methylation. Biochem Biophys Res Commun 2007;352:188–92.
- [86] Davis CD, Uthus EO, Finley JW. Dietary selenium and arsenic affect DNA methylation in vitro in Caco-2 cells and in vivo in rat liver and colon. J Nutr 2000;130:2903–9.
- [87] Ramirez T, Brocher J, Stopper H, Hock R. Sodium arsenite modulates histone acetylation, histone deacetylase activity and HMGN protein dynamics in human cells. Chromosoma 2008;117:147–57.
- [88] Martínez L, Jiménez V, García-Sepúlveda C, Ceballos F, Delgado JM, Niño-Moreno P, et al. Impact of early developmental arsenic exposure on promotor CpG-island methylation of genes involved in neuronal plasticity. Neurochem Int 2011;58:574–81.
- [89] Cronican AA, Fitz NF, Carter A, Saleem M, Shiva S, Barchowsky A, et al. Genome-wide alteration of histone H3K9 acetylation pattern in mouse offspring prenatally exposed to arsenic. PLOS ONE 2013;8:e53478.
- [90] White LD, Cory-Slecta DA, Gilbert ME, Tiffany-Castiglioni E, Zawia NH, Virgolini M, et al. New and evolving concepts in the neurotoxicology of lead. Toxicol Appl Pharmacol 2007;225:1–27.
This page intentionally left blank

17

Health Effects of Prenatal and Early-Life Exposure to Arsenic

Kathryn A. Bailey and Rebecca C. Fry

DEPARTMENT OF ENVIRONMENTAL SCIENCES AND ENGINEERING, GILLINGS SCHOOL OF GLOBAL PUBLIC HEALTH, UNIVERSITY OF NORTH CAROLINA AT CHAPEL HILL, NC, USA

CHAPTER OUTLINE

17.1	Introduction	405
17.2	Adverse Health Effects Associated with Chronic and Early-Life Arsenic Exposure	406
17.3	Mechanisms Implicated in Disease Development Associated with Prenatal Exposure	412
17.4	Conclusions and Future Directions	421
Refe	rences	421

17.1 Introduction

The presence of inorganic arsenic (iAs) in drinking water presents a major threat to human health around the globe. While the World Health Organization (WHO) has recommended that the levels of inorganic arsenic (iAs) in drinking water should not exceed $10 \mu g/L$ [1], it is currently estimated that up to 100 million people worldwide are exposed to drinking water containing detrimental levels of iAs [2]. The presence of elevated levels of arsenic in drinking water predominantly stems from the use of naturally contaminated groundwater, but groundwater sources can become contaminated due to anthropogenic activities such as mining as well [3]. This contamination of groundwater sources is particularly problematic in rural areas and developing countries as these areas must often rely on ground water as the predominant source of drinking water. The detrimental impact of iAs-contaminated drinking water has been particularly evident in Southeast Asian countries such as Bangladesh and India [4,5]. Bangladesh is a severely affected country, as estimates suggest 35–77 million of the 125 million inhabitants are exposed to levels of iAs in drinking water that exceed the WHO standard [6-8]. For Bangladeshis exposed to iAs levels in their drinking water at levels that exceed 50 µg/L, there are yearly estimates of over 9100 deaths and 174,000 disability-adjusted life years (DALYs) attributed to iAs-associated disease [9]. While the problem of iAs in drinking water may be most evident in Southeast Asia, iAs levels in drinking water that exceed the

WHO standard have been detected in numerous areas around the world including Mexico and the United States of America (USA), emphasizing the worldwide nature of the problem [10].

As discussed in more detail below, numerous adverse health effects have been associated with arsenic exposure in human populations, including increased susceptibility to infections and increased risk of developing cancers and other non-communicable diseases (NCDs) later in life. The impact of NCDs on human health is of major concern as they were among the top causes of death worldwide in 2011 and were attributable for up to 67% of premature deaths, i.e., those occurring before age 60, in developing countries [11]. The susceptibility to NCDs is known to be dependent on both genetic and environmental components, and mounting evidence suggests that for most NCDs, environmental factors generally have a greater impact than genetic factors [12]. In developed countries, environmental factors such as physical inactivity and high caloric intake are believed to contribute significantly to NCD development. However, it is becoming increasingly clear that exposure to environmental toxicants, and particularly exposures that occur during critical developmental windows, is also linked to the development of NCDs such as diabetes, obesity, cardiovascular disease, and respiratory disease later in life [13]. Of great concern is the capacity of developmental toxicant exposure to not only affect the health of individuals throughout their lifetimes but also adversely affect the health of subsequent generations. For example, in utero exposure to maternal cigarette smoke is associated with increased risk of asthma [14] and lung function deficits [15] in children, and this risk is further increased if both the mother and maternal grandmother smoked during pregnancy [13,16]. These results underscore the urgent need for the identification of vulnerable populations such as pregnant women and the implementation of preventive measures.

In the following sections, we provide an overview of the health impacts associated with arsenic exposure and focus on health effects associated with prenatal/early-life exposure. We detail how defined time periods of arsenic exposure in human populations and animal models have helped to increase the understanding of the diversity and latencies of the observed effects, differences between susceptible populations, and the impacts of factors such as sex and co-exposures. Lastly, we discuss the molecular mechanisms implicated in the development of these latent effects and identify needs in current research trends to help better understand these mechanisms.

17.2 Adverse Health Effects Associated with Chronic and Early-Life Arsenic Exposure

17.2.1 Health Effects Associated with Chronic Exposure to iAs

Some of the best-characterized health effects associated with arsenic exposure are those observed in chronically exposed adults. The first clinical symptoms of chronic arsenic poisoning, or arsenicosis, are often specific lesions in the skin that include areas of aberrant pigmentation (both hypopigmentation and hyperpigmentation) and hyperkeratosis [17]. More prolonged exposure periods are associated with the development of numerous cancers,

especially non-melanoma skin cancers and cancers of the lung and urinary bladder, but cancers in the liver, prostate, and kidney have also been observed [1,18]. Based on epidemiological data, it has been estimated that drinking water containing $50 \mu g/L$ of iAs has a combined cancer risk of approximately 1 in 100, which is least 100 times greater than any other drinking water contaminant with a defined maximum contaminant level (MCL) [18]. Chronic arsenic exposure is also associated with non-malignant adverse health effects, including the development of NCDs such as diabetes mellitus, cardiovascular disease, and peripheral vascular disease as well as gastrointestinal effects and neurological effects [19]. The nature and severity of clinical symptoms/diseases observed in iAs-exposed individuals are likely multi-factorial and include exposure dose and duration, the genetic background and nutritional status of the population, and co-exposure to factors such as other environmental toxicants [20]. Human population studies have indicated that both cancerous and non-cancerous health effects associated with iAs exposure can be a significant cause of mortality. In a Bangladeshi population, there was an increased risk of death due to all non-accidental causes [hazard ratio (HR) = 1.16; 95% confidence interval (CI): 1.06–1.26)] at even relatively low levels of arsenic in drinking water ranging from 10 to $49 \,\mu g/L$ [21]. At higher levels of exposure (50–149 $\mu g/L$), there were dosedependent relationships between arsenic exposure and deaths from several causes including cancers (HR = 1.44; 95% CI: 1.0-1.95), cardiovascular disease (HR = 1.16; 95% CI: 0.96-1.40), and infectious diseases (HR = 1.30; 95% CI: 1.13–1.49). Another report identified an increase in hazard ratios associated with all-cause mortality and chronic disease mortality in Bangladeshi adults [22]. The investigators reported that approximately 21.4% of deaths and 23.5% of deaths associated with chronic diseases could be attributed to arsenic levels in drinking water that exceeded 10 µg/L [22]. In summary, experimental evidence in human populations underscores the numerous diseases associated with chronic arsenic exposure and that arsenic exposure can be a significant cause of death in exposed populations.

17.2.2 Health Effects Associated with iAs Exposure during Gestation and Infancy

Arsenic has been shown to readily cross the placenta in humans and experimental animals and therefore the developing fetus is not protected from arsenic exposure [23]. Pregnant women and their unborn children are particularly vulnerable to the adverse effects of arsenic exposure, which include increased risks of poor pregnancy/birth outcomes such as spontaneous abortion, preterm birth, stillbirth, and decreased birth weight [24–29]. While prenatal exposure to arsenic has been shown to be associated with some developmental abnormalities in mice including dose-dependent trends in neural tube defects and vertebral and calvarial malformations [30], data that support the association of developmental defects such as these in human populations are generally lacking [29]. In newborns from Chitwan Valley, Nepal, cord blood levels of arsenic were shown to be inversely associated with neurodevelopmental indicators [31]. Health effects in infants exposed to iAs *in utero* include reduced thymic index, higher rates of morbidity (e.g., diarrhea and pneumonia), and higher rates of mortality [32–34]. For example infants in Bangladesh born to mothers exposed to iAs levels in their drinking water that exceeded $50 \mu g/L$ had an increased risk of death, a response that showed a dose-dependent effect [26]. It is important to note that infants exposed prenatally to arsenic are likely also exposed throughout their childhood and into adulthood, so adverse health effects that are observed throughout the lifetime of an individual cannot be discerned as specifically associated with prenatal exposure or postnatal exposure. As a result there is a need for controlled toxicological experiments to clearly define the specific windows of developmental susceptibility to these exposures.

17.2.3 Health Effects of iAs Exposure Observed in Childhood

Children who are chronically exposed to iAs may exhibit skin lesions similar to those observed in chronically exposed adults [35]. However, unlike adults, epidemiological evidence in general does not support an association between chronic iAs exposure and an increase in cancer incidence and cancer mortalities in children [36]. As discussed further below, this may be due to the long latency times that are commonly observed for cancers associated with chronic and early-life exposure to arsenic. A study based in Chile did report an association between very high iAs exposure in early childhood (approximately $860 \mu g/L$ in drinking water) and an increase in mortality individuals <20 years of age from liver cancer, which is generally a rare cancer observed in children [37].

Due to the sensitivity of the developing brain to the effects of toxicants, the impact of arsenic exposure on the neurological development and intellectual capacity of children has been a major endpoint of interest. There is evidence of neurotoxic effects of arsenic associated with chronic exposure in rodents, including the impairment of spatial memory in rats [38] and genderdependent alterations in brain neurochemical markers and spontaneous locomotor activity in mice [39]. In rats, prenatal and early postnatal life exposure to arsenic is also associated with poor reflex response and ultrastructural changes and altered expression of neuronal cell adhesion molecules in the hippocampus [40]. Aside from neurological impairments observed in prenatally exposed newborns [31], several studies have linked iAs exposure with cognitive impairments in children or adolescents [41–46]. For instance, a dose-dependent inverse relationship between iAs exposure and child intelligence in both 10-year-old [42] and 6-year-old children [41] was observed in a Bangladeshi population. A negative association between iAs exposure and motor function in children 8–10 years of age was also observed in Bangladesh [47]. Reduced intellectual function was observed in 14-year-old children who were exposed to arsenic-contaminated milk powder early in life (as reviewed in [48]).

Some sex-specific neurological effects have been observed. For instance, considering Bangladeshi children aged 1.5 and 5 years of age, the concentration of total arsenic in urine (U-tAs) was inversely associated with verbal IQ (VIQ) and full-scale IQ (FSIQ) in girls but not boys [45]. In a cohort from the smelter community of Torreón, Mexico, iAs exposure was associated with cognitive deficiencies in both girls and boys aged 6–8 years [44]. Overall, while there was a greater impact in boys compared to girls, the particular effects observed differed between boys and girls and were dependent on levels of iAs exposure as well, with some

effects only evident in lower exposure groups (U-tAs $<50 \mu g/L$). There is also evidence that adverse effects may be associated with co-exposures to other toxicants. For example, a study of 11–13-year-old children in the United States indicated that high levels of either arsenic or manganese were associated with lower indicators of intelligence, with the most impact observed when there was co-exposure to both metals [49].

17.2.4 Delayed Health Effects Associated with Chronic and Prenatal Arsenic Exposure

Studies of a Chilean population that had a defined period of high arsenic levels in drinking water have provided increased insight into the diverse effects of arsenic exposure on human health, the latent nature of these health effects, and the impact of early-life exposure on adverse health effects apparent in adulthood. These studies focused on Region II of northern Chile, which had a period of very high arsenic concentrations in municipally supplied drinking water between the years of 1958 and 1970 [50]. Almost all who live in this region rely on municipally supplied water as their source of drinking water, which is obtained from a few river sources that originate in the Andes Mountains. Prior to 1958, Antofagasta, the main city of Region II, had approximately 90 µg/L arsenic in its municipally supplied drinking water [51]. Starting in 1958, the drinking water supply of this region was supplemented with water from two additional rivers with very high arsenic content. This initiated of a 13-year period (1958-1970) in which more than 250,000 people in the city of Antofagasta were exposed to average levels of 860 µg/L arsenic in drinking water, levels that are 86 times the current recommended limit of $10 \,\mu g/L$ [52]. After the drinking water treatment plant was installed in 1970, the levels of arsenic in the drinking water of Antofagasta quickly declined and are less than $10 \,\mu$ g/L today [52,53]. Thus, this defined period of high exposure to iAs provides a rare opportunity to examine the impacts of arsenic on human health, including exposure limited to early life and childhood.

Following the high exposure period, mortality rates of various causes from arsenicexposed Region II compared to those from unexposed regions of Chile revealed increases from numerous cancers such as those of the skin, lung, urinary bladder, and kidney [50,51,54,55]. Increased mortalities from non-cancerous causes such as pulmonary tuberculosis and acute mycocardial infarction (AMI) were also observed in Region II [56,57].

These studies not only provided insight into the diverse health effects associated with iAs exposure, but also revealed differential latencies associated with these diseases. Specifically, increases in death from most causes, including cancers and pulmonary tuberculosis, started around 1968, approximately 10 years after the start of the high exposure period [54,55,57]. The exception to this was deaths due to AMI, which had high highest rates of mortality during the high exposure period [56]. Comparisons of mortality rate ratios for urinary bladder cancer and lung cancer of the high exposure area (Region II) to the lower exposure area (Region V) between the years 1950 and 2000 demonstrated an increase in both urinary bladder cancer and lung cancer in the iAs-elevated area that began approximately 10 years after the start of the high exposure area of the high exposure area (Region II) to the lower exposure and lung cancer and lung cancer and lung cancer in the iAs-elevated area that began approximately 10 years after the start of the high exposure area of the high exposure area (Region II) to years after the start of the high exposure area in the iAs-elevated area that began approximately 10 years after the start of the high exposure area of the high exposure area (Region II) to years after the start of the high exposure area in the iAs-elevated area that began approximately 10 years after the start of the high exposure area of the high exposure period and peaked in 1986–1997 [54]. A similar comparison based upon the 1950–2000 death records in Regions II and V showed an increase in deaths due to AMI in Region II versus

Region V during the high exposure period of 1958–1970 [56]. At that time, the AMI mortality rate ratios were 1.48 for men (95% CI: 1.37–1.59; p < 0.001) and 1.26 for women (95% CI: 1.14–1.40; p < 0.001). A comparison of the mortality data for bladder cancer, lung cancer, and AMI showed that excess deaths in Region II during the high exposure period were predominantly attributable to AMI. This was in contrast to the data representing the time period 10 years after the high exposure period where the excess AMI-associated deaths were reduced, while those associated with urinary bladder cancer and lung cancer predominated [56]. These results highlight the differences in latency periods, and that the latency periods can be very long. Long latencies were particularly observed for various cancers, as the deaths from lung, urinary bladder, and kidney cancers were high at least 25 years after the end of the high exposure period [54,55].

The data also highlight the differences in the health outcomes of arsenic exposure between men and women. For instance, the increase in mortality from lung cancer, urinary bladder cancer, and AMI combined peaked in 1991–1995, with an estimated 10.9% of all deaths in men, yet 4.0% of all deaths in women were attributable to arsenic exposure during this period [56]. The peak kidney cancer mortality rate ratio was 3.4 (95% CI: 2.2–5.1) for men in 1981–1985, which later declined to 1.6 (95% CI: 1.2–2.1) in 1996–2000 [55]. Between 1981 and 1985, the mortality rate ratios were 2.9 (95% CI: 1.8–4.7) for women and were therefore lower than those observed for men, but further increased to 4.4 (CI: 3.0–6.4) in 1991–1995 and therefore remained high longer than for men [55].

Studies of the high exposure period in Region II of Chile also highlight the delayed impact of early-life exposure to arsenic. These analyses focused on individuals born during the peak exposure period of 1958-1970 and those born just before this period, i.e., between 1950 and 1957 and were "young adults" aged 30-49 at the time of their deaths. Comparison of mortality rates of young adults in Region II to those from the rest of Chile (excluding Region II) revealed an association between arsenic exposure in early life/childhood and mortality from non-malignant lung disease (i.e., bronchiectasis) and lung cancer [58]. In this study, individuals born just prior to the high exposure period had a standardized mortality ratio (SMR) for lung cancer of 7.0 (95% CI: 5.4–8.9; p < 0.001) and an SMR for bronchiectasis of 12.4 (95% CI: 3.3–31.7; p < 0.001). For those born during the high exposure period that were likely exposed prenatally and during early childhood, the SMR for lung cancer was 6.1 (95% CI: 3.5–9.9; p < 0.001) and the SMR for bronchiectasis was 46.2 (95% CI: 21.1–87.7; p < 0.001) [58]. Other effects on the lung have also been observed, in which adults living in Antofagasta younger than 10 years of age during the high exposure period demonstrated reduced lung function nearly 40 years after the high exposure period ended compared to adults who were unexposed as children [59]. These adults with early-life exposure had 11.5% lower forced expiratory volume in 1s (FEV₁; p = 0.04), 12.2% lower forced vital capacity (FVC; p = 0.04), and increased breathlessness (prevalence odds ratio = 5.94, 95% CI: 1.36-26.0) compared to those who did not experience the exposure. In addition, an exposure-response relationship was observed between exposure concentrations in early life and adult FEV₁ and FVC. The authors estimated that the decrease in lung function was similar to that observed for 45 pack-years of cigarette smoking [59].

Further examination of health effects in addition to those on the lung indicated that prenatal or early childhood arsenic exposure was associated with increased mortality from several causes, including cancers of the urinary bladder (SMR = 18.1; 95% CI: 11.3, 27.4), larynx (SMR = 8.1; 95% CI: 3.5, 16.0), and liver (SMR = 2.5; 95% CI: 1.6, 3.7), as well as chronic renal disease (SMR = 2.0; 95% CI: 1.5, 2.8) [60]. Similar effects were observed with mortality rates from kidney cancer in young adults aged 30–39 who were born in two Region II cities, Antofagasta and Mejillones, during or just prior to the high exposure period compared to the rest of Chile (mortality rate ratio = 7.1; CI = 3.1–14) [55]. In addition, the highest mortality rate ratios for AMI observed in high exposure Region II versus the low exposure Region V were those observed in young adult men aged 30–49 who were born during the high exposure period (rate ratio = 3.23, 95% CI: 2.79, 3.75; p < 0.001). Taken together, the data from the Antofagasta region indicate that prenatal/early-life exposure to arsenic is associated with the greatest increase in mortality in adults <50 years of age compared to any other prenatal toxicant identified to date [60].

Consistent with the effects observed in Chile, exposure of Japanese infants to arsenic in milk powder for a defined period of time also indicates early-life exposure to arsenic is associated with latent impacts on human health. A mass poisoning of bottle-fed infants occurred in 1955 in Japan due to arsenic-contaminated milk powder (as reviewed in [48,61]). Okayama Prefecture was the most severely affected area, with acute poisoning cases that resulted in 24 deaths, 2005 surviving cases, and 84 suspected cases in the time period between August 1955 and April 1956 [48]. Follow-up studies of poisoning victims from Kyoto and Hiroshima Prefecture victims at age 14 demonstrated increased rates of neurological symptoms/disorders and lower intelligence quotient (IQ) including mental retardation compared to unexposed children (as reviewed in [48]). A comparison of cancer mortality in Okayama between the years of 1970 and 2006 in individuals 15-54 years of age was performed. These analyses focused on comparisons of mortality of cohorts of individuals who were less than 5 years old at the time of the poisoning and therefore potentially exposed to contaminated milk powder and that in control groups of unexposed individuals. The results revealed an increase in cancer mortality risk in the exposed group from pancreatic cancers in all age groups and leukemias in the 30–39 age group [61]. A follow-up study that focused on individuals who were <1 year old at the time of the poisoning and unexposed individuals born approximately 1 year later revealed evidence of increased mortality from all cancers (aged 48-53 years; MR = 1.36, 95% CI: 1.03-1.78) and liver cancer (all ages; MR = 1.72, 95% CI: 0.99-2.97) in the exposed cohort. Unlike the results observed in Chile that identified health effects associated with chronic and prenatal exposure, there was no evidence of increased risk of lung, urinary bladder, or kidney cancers. The authors suggested that early-life exposure may result in different cancers than those observed due to lifetime exposures or chronic exposures that started later in life [62].

Taken together, these studies of defined exposure periods were informative in several ways. They have indicated that both chronic and early-life arsenic exposure is associated with deaths from a wide variety of causes, and that the latency periods between exposure periods and the development of adverse health effects differ by each disease. In addition, these studies high-light the important finding that there are sex-specific trends in the outcomes between men and women. They have also indicated that early-life exposure to arsenic is sufficient to increase risk of death/premature death due to various diseases later in life. The link between arsenic exposure and increased death due to pulmonary tuberculosis highlights that arsenic may increase

the risk of death due to infectious diseases, and suggests that the full impact of arsenic on morbidity and mortality due to various causes is likely currently underestimated. In addition, these studies also indicate that there may be differences in diseases that develop in those exposed to arsenic in early life compared to those exposed in adulthood or throughout their lifetimes—a finding that requires study in animal models. There is also some indication that characteristics of exposed populations, such as genetic background and nutritional status, may influence disease development. Thus far, studies on these populations have not indicated the transgenerational potential of health risks associated with prenatal exposure, nor have molecular alterations of study subjects been studied. In the following section, we summarize research on animal models and currently exposed human populations that have provided insight into the molecular mechanisms that drive disease development associated with early-life exposure.

17.3 Mechanisms Implicated in Disease Development Associated with Prenatal Exposure

Considering the multitude of adverse health effects associated with early-life exposure to arsenic, complex biological mechanisms are likely an underlying feature. Much remains to be learned about the precise molecular mechanisms that link prenatal exposure to altered disease risk, especially those associated with latent health effects.

Arsenic-associated cancers have been the most intensely studied disease associated with arsenic exposure in which several molecular events are believed to play a role. These include the generation of reactive oxygen species (ROS)/oxidative stress, alterations in key cellular signaling cascades, enzyme inhibition, interference with DNA repair, epigenetic alterations, induction of iAs biotransformation, immunomodulatory effects, endocrine disruption, and the formation of genetic damage such as chromosomal aberrations [63,64]. No single mechanism has emerged as a key event, and it is likely that several mechanisms are involved. For example, ROS generated by arsenic exposure can cause macromolecule damage as well as activate various cell signaling pathways implicated in cancer development [65,66]. In terms of prenatal arsenic exposure, a major research focus has been to understand the molecular alterations that may lead to delayed health effects and, in particular, cancers. As discussed below, studies in animal models and examinations of human populations have identified several factors that may be involved in the development of one or more effects associated with prenatal exposure including altered gene expression/hormone signaling, epigenetic reprogramming, the emergence of cancer stem cells (CSCs), and perturbed immune function.

17.3.1 Epigenetic Reprogramming as a Potential Key Event in Latent Disease Development

It has been proposed that permanent changes regulated by the epigenome are a plausible link between prenatal toxicant exposure and disease development, particularly for NCDs that develop later in life. The epigenome, which means "above the genome," refers to potentially heritable biological information contained outside the DNA sequence that impacts gene expression [67-69]. The epigenome is widely accepted to include three components, namely DNA methylation, histone post-translational modifications (PTMs), and microRNAs (miRNAs), which control gene expression at either the transcriptional level (DNA methylation/histone PTMs) or at the post-transcriptional level (miRNAs) [68,69]. Histone PTMs and especially DNA methylation patterns are the most extensively studied and best-characterized components of the epigenome. In mammals, DNA is methylated almost exclusively at the cytosine residue of CpG dinucleotides to generate 5'-meCpG-3' [70]. CpGs are clustered in two main areas of the genome: (1) \approx 15% of CpGs are found in regions of euchromatin known as CpG islands and CpG island shores and (2) \approx 85% of CpGs are found in repetitive, interspersed regions of DNA, e.g., transposable elements, in heterochromatin [71–73]. The association of transcriptional competency with particular epigenetic "marks," i.e., particular histone PTMs and/or DNA methylation patterns, has been established, especially in the context of gene promoters (see [74–76] for reviews). In general, there is a negative relationship between promoter CpG island/ CpG island shore methylation and gene expression [77], and CpGs are heavily methylated in the transcriptionally silenced regions of heterochromatin [73]. However, it should be noted that this relationship is complex and there are numerous instances in the literature in which no relationship or positive relationships between promoter DNA methylation and transcriptional competency have been observed [77,78]. This is important to consider in the context of functional implications of arsenic-associated DNA methylation.

The integrity of the epigenome is critical to cellular homeostasis and genetic stability as it has been shown to regulate and fine-tune gene expression changes associated with virtually every cellular process, including cellular differentiation during organismal development as well as response to environmental stimuli [79–81]. The epigenome is considered an especially sensitive target of the effects of perturbations to the fetal environment as the DNA methylome, for instance, undergoes many changes during early development [67]. Changes to the epigenome that occur during early developmental periods are particularly concerning as they may be metastable and therefore persist throughout the lifetime of an individual as well as be maintained in subsequent generations [82,83]. Such "reprogramming" of the epigenome is believed to be a mechanism that supports the fetal origins of adult disease through the altered, persistent expression of key genes involved in disease development such as the overexpression of growth factors or the silencing of tumor suppressors [80,84].

The DNA methylome is the most extensively studied epigenetic component in terms of examining the effects of toxicant-induced perturbations in the fetal environment and laterlife health effects. In animal models, evidence that the reprogramming of the DNA methylome may play a role in altering disease risk later in life is supported by the link between transient toxicant exposure during particular developmental periods, perturbed DNA methylation and gene expression patterns, and altered disease risk in adulthood [85]. For instance, neonatal male rats exposed to bisphenol A (BPA) or estradiol demonstrated permanent alterations in the DNA methylation patterns of several cell-signaling and apoptosis genes and an increased risk of developing prostate cancer later in life [85]. One of these modulated genes, namely phosphodiesterase type 4 variant 4 (*Pde4d4*), becomes transcriptionally silenced in the normal prostate during aging via hypermethylation of certain CpG sites within the *Pde4d4* promoter. However, the prostate tissue of the BPA- and estradiol-exposed rats, these CpG sites within the *Pde4d4* promoter remain hypomethylated with sustained *Pde4d4* expression during aging. In addition, *in vitro* evidence revealed that these CpG sites are hypomethylated with increased *Pde4d4* expression in the prostate tumor-cell line AIT compared to the normal prostate epithelial cell line NbE-1, suggesting this gene may play an important role in latent prostate cancer development [85]. Considering early-life exposure to arsenic is also associated with altered, delayed risk of the development of NCDs such as cancers, pulmonary disease, and cardiovascular disease in humans, it is plausible that DNA methylation changes provide a link between prenatal iAs exposure and altered disease risks in adulthood [86,87]. Thus, alterations in gene expression profiles and DNA methylation patterns have been a major focus of investigations of molecular impacts associated with early-life exposure to arsenic.

17.3.1.1 Epigenetic and Genomic Alterations in Experimental Animals

The transplacental exposure models developed by the Waalkes group have been valuable tools in investigations of the molecular events that may link prenatal arsenic exposure and cancer development later in life. While experimental evidence generally does not support that iAs acts as a complete carcinogen in any organ in any experimental animal when administered in adulthood [88], iAs is a complete transplacental carcinogen in multiple organs in several strains of mice [89]. This effect was first observed in C3H mice, in which exposure of offspring born to pregnant females to sodium arsenite [NaAs₂O₃; iAs(III)] in drinking water [42.5 or 85 ppm arsenic during gestational days (GDs) 8–18] resulted in increased incidence and/or multiplicity of tumors in multiple organs in adult offspring [89,90]. Sex-specific effects were observed in the rodent model. Specifically, female offspring had dose-related increases in lung adenocarcinoma, benign and malignant ovarian tumors, and preneoplasias in the uterus and oviduct. In male offspring, there was a dose-dependent incidence of liver adenoma, hepatocellular carcinoma (HCC), liver adenoma/carcinoma, and adrenal cortical adenoma. A very pronounced effect was the strong dose response in HCC incidence and multiplicity in male C3H offspring, a form of cancer that is epidemiologically linked to chronic arsenic exposure in humans [37,90,91].

Prenatal exposure to a carcinogenic dose of arsenic was also found to be associated with the development of cancers, tumors, and/or proliferative lesions in some organs may have initiating effects in other organs after postnatal application of tumor promoting agents such as diethylstilbestrol (DES) and tamoxifen (TAM) in CD1 mice [92,93] and 12-O-tetradecanoyl phorbol-13-acetate (TPA) in C3H mice and Tg.AC mice [94–96]. For instance, prenatal arsenic exposure alone had no effect in Tg.AC mice, but postnatal TPA application resulted in the development of papillomas and squamous cell carcinomas (SCCs) [96]. Mice that received prenatal arsenic and postnatal TPA had increased the multiplicity of SCCs compared to TPA exposure alone, and the SCCs were much more aggressive [96]. In terms of internal cancers, postnatal TPA alone did not result in any statistically significant changes compared to controls [95]. In male offspring, arsenic exposure was associated with the development of adrenal cortical adenomas (ACAs) independently of the administration of TPA, and urinary bladder (UB) hyperplasia was observed in the arsenic + TPA group. In females, prenatal arsenic exposure

alone was associated with the development of ACAs, UB hyperplasia, uterine hyperplasia, and to a lesser extent uterine tumors and UB papillomas, with higher incidence of ACAs in the arsenic + TPA group. In both males and females, no dose response was observed [95].

The Waalkes group also developed a whole-life arsenic exposure model in CD1 mice, in which mice were exposed to 0, 12, and 24 ppm iAs(III) starting in both males and females 2 weeks before breeding, in females during pregnancy and lactation, and in the resulting off-spring until 2 years of age [97]. These exposures resulted in dose-dependent increases in tumors in male and female mice in similar target sites as observed after *in utero* arsenic exposure only, except the tumors associated with whole-life exposure were more aggressive, which may reflect the cumulative effects of prenatal and postnatal exposure [97].

Importantly, the carcinogenic effects observed in these transplacentally exposed mice are comparable in many ways to those observed in human populations exposed to arsenic in early life, namely: (1) prenatal/early-life exposure is a particularly sensitive period to the effects of arsenic, (2) exposure during this period is sufficient to cause increased risk of cancer development later in life, and (3) cancers develop in multiple organs in a sex-dependent manner. These transplacental models also illustrate that prenatal exposure to arsenic can have an initiating effect as well as enhance the tumorigenic effects of other chemicals.

Molecular examinations of tumors and normal tissue from transplacentally exposed mice have supported the hypothesis that the persistent, altered expression of key genes is a plausible early event involved in cancer development after prenatal iAs exposure. Experimental evidence also suggests epigenetic reprogramming may underlie these altered gene expression profiles. For instance, compared to unexposed mice, the fetal livers of C3H male mice exposed to hepatocarcinogenic doses of arsenic in utero have alterations in the DNA methylation levels of GC-rich regions throughout the genome and altered expression of numerous genes, including several implicated in liver tumorigenesis and liver disease such as cytokeratins and insulin-like growth factors [89,98]. Similar gene expression profiles were observed in both HCCs and peritumor liver tissue from transplacentally exposed male C3H mice [99]. Together, these results indicate that alterations in gene expression profiles were long-lasting and their detection in prenatal and peritumorous tissue suggests they may be early events in the carcinogenic process. Of particular interest is the increased expression of estrogen receptor alpha (*Er-\alpha*) and the estrogen-responsive gene cyclin D1 (Cd1) as well as the feminized expression of P450 genes in normal-appearing tissue and HCCs of prenatally exposed male mice [99,100]. Concurrent with increased *Er*- α expression, several DNA regions within the *Er*- α promoter were found to be hypomethylated in the livers of adult male mice that had developed HCCs, which may be significant as loss of promoter methylation is often associated with increased transcriptional activity [100]. Increased $Er - \alpha$ expression and hypomethylation of the $Er - \alpha$ promoter were also observed in the livers of mice chronically exposed to arsenic [101], and both ER- α and CD1 are implicated in the development of liver cancers [102,103]. Similar to findings in the liver, the aberrant expression of $Er-\alpha$ was also implicated in the development of lung cancers in prenatally exposed C3H mice. The fetal lung had aberrant levels of $Er-\alpha$ at both the mRNA and protein levels in addition to the modulation of several estrogen-regulated genes and genes associated with lung cancers [104]. In addition, lung adenomas and adenocarcinomas that developed in the adult lung had intense ER- α staining, which was not observed in normal adult lung or lung adenocarcinomas associated with diethylnitrosamine (DEN) exposure. Interestingly, postnatal exposure to estrogens was shown to increase the carcinogenic response after prenatal exposure to arsenic in male and female CD1 mice, as there was high expression of ER- α protein in UB lesions and HCCs in males after postnatal DES/TAM exposure and uter-ine and urinary bladder tumors were present in females after DES exposure [92,93].

Together, these results are informative as they indicate that estrogen-responsive pathways may play a role in arsenic-mediated carcinogenesis, and that aberrant promoter DNA methylation of key target genes such as Er- α are associated with altered gene expression and cancer development associated with prenatal arsenic exposure. Given the similarities between early-life arsenic exposure and cancer development in adulthood between these animal models and human populations, it is plausible that animal models such as these provide insight into the molecular mechanisms that underlie the development of latent diseases associated with prenatal exposure in humans. However, it must be emphasized that a correlative, and not causal, relationship has been identified between altered promoter DNA methylation (e.g., Er- α), aberrant gene expression, and disease development in these models. Assuming that altered DNA methylation is an initial event in the alteration of the expression of a key gene such as Er- α , the exact mechanism by which prenatal arsenic exposure changes the DNA methylome is unknown, as well as the role that other epigenetic modifications, i.e., histone PTMs and miRNAs, may play in this process.

It is important to note that prenatal iAs exposure is also associated with accelerated/exacerbated atherosclerosis in apolipoprotein A-knockout (Apo $E^{-/-}$) mice, a mouse model often used to study atherosclerotic disease. Transplacental exposure of pregnant Apo $E^{-/-}$ females to 85 ppm sodium arsenite from GD 8 to birth (GD 20) is associated with evidence of increased atherosclerotic disease in male offspring 10 and 16 weeks after birth compared to controls [105]. In addition, prenatal exposure to 49 ppm sodium arsenite from GD 8 to birth was shown to cause changes in the mRNA and miRNA profiles in the liver of male offspring that were evident at postnatal day (PND) 1 and PND 70 [106]. Genes altered at PND 1 included those associated with gluconeogenesis/glycolysis whereas genes altered at PND 70 were associated with diverse processes such as protein export and antigen processing/presentation. These altered, sustained gene expression profiles are consistent with arsenic reprogramming gene expression signaling within the fetal liver and are of particular interest as liver stress/injury may be a contributing factor of atherosclerosis development. The authors suggest that sustained stress and pro-inflammatory response in the liver may play an important role in the development of early-onset atherosclerosis in prenatally exposed $ApoE^{-/-}$ mice, but the molecular mechanisms associated with this reprogramming have not been explored [106].

Taken together, the animal models of arsenic-induced disease support that fetal reprogramming may be a common and important link between prenatal/early-life exposure to iAs and the delayed development of both cancerous and non-cancerous NCDs.

17.3.1.2 Epigenetic and Genomic Alterations in Human Populations

Studies that examine perturbed gene expression and epigenetic profiles associated with prenatal arsenic exposure in human populations represent a growing area of interest. The first study of

gene expression changes associated with prenatal arsenic exposure examined the gene expression profiles of cord blood from newborns from the Ron Pibul District, Nakhon Sri Thammarat, Thailand [107]. This analysis identified 447 iAs-associated genes in cord blood leukocytes, of which $\approx 90\%$ were up-regulated in the exposed group. These genes were enriched in functions associated with immune and inflammatory response, stress response, cell signaling, apoptosis, and cancer/tumorigenesis. In particular, this study identified the nuclear factor kappa-lightchain-enhancer of activated B cells (NF- κ B) signaling cascade as a key mediator of the genomic response to prenatal arsenic exposure in newborn cord blood, which is of particular interest given the role of NF- κ B in inflammation-associated cancer development [108]. A set of 11 genes that exhibited a strong dose-response relationship with prenatal arsenic exposure was identified. These genes demonstrated a high level of accuracy (83%) in predicting prenatal arsenic exposure within this cohort and therefore were suggested to be potential biomarkers of prenatal exposure [107]. The effect of prenatal iAs exposure was also examined at the protein level in the cord blood plasma of Bangladeshi newborns [109]. Analysis of the association of 18 cord blood cytokines with maternal U-tAs at gestational week 30 (GW 30) revealed that the level of several pro-inflammatory cytokines, namely, interleukin 1 beta (IL-1 β), interleukin 8 (IL-8), tumor necrosis factor (TNF), and interferon gamma (INF- γ), had U-shaped curves across quartiles of maternal U-tAs with significant differences between quartiles 2 and 8 for IL-8 and TNF [109]. Although these studies are limited in number, results at both the mRNA and protein level are consistent with the up-regulation of pro-inflammatory genes in arsenic-exposed newborns [107,109].

To date, studies that have examined epigenetic effects associated with prenatal arsenic exposure have primarily focused on changes in DNA methylation. Some studies have examined methylation changes at the gene level, which may be significant as they may be correlated with the aberrant expression of disease-promoting genes. Another area of interest includes examining changes in the overall methylation status of the genome as vast changes may be an important factor in disease development. For instance, DNA hypomethylation within heterochromatic regions may lead to genetic instability, which is implicated as a predisposing event in the development of some cancers [110,111].

In Thailand, examination of DNA isolated from newborn cord blood revealed a positive relationship between arsenic exposure *in utero* and promoter DNA methylation of the tumor suppressor, tumor protein 53 (*TP53*), although no relationship was identified between prenatal exposure and the methylation of Long Interspersed Element 1 (LINE-1) elements, which were used as indicators of global DNA methylation levels [112]. In iAs-exposed mother/child pairs in Bangladesh exposed to drinking water levels of $<1-230 \mu g As/L$, the relationships between iAs exposure and DNA methylation patterns of LINE-1 and the promoters of tumor suppressors *TP53* and cyclin-dependent kinase inhibitor 2A (*CDKN2A/p16*) were examined in maternal and umbilical cord blood leukocytes [113]. The DNA methylation trends were similar between mothers and newborns, in which a positive association was identified between maternal U-tAs and LINE-1 methylation and some CpG sites within the *p16* promoter. The greatest effects were observed in the middle tertiles of exposure, therefore revealing evidence of a non-linear dose-response relationship and highlighting that epigenetic effects may be observed at relatively moderate levels of exposure. Changes in DNA methylation have also been observed in leukocytes derived from the cord blood of newborns in the USA exposed to relatively low

levels of iAs [114]. This genome-wide analysis of cord blood isolated from newborns from New Hampshire indicated there was no overall change in DNA methylation levels in relationship to maternal U-tAs. However, several differentially methylated areas were identified, most of which occurred in CpG islands [114]. Another genome-wide analysis of cord blood of prenatally exposed Bangladeshi newborns identified differential methylation within CpG regions of cord blood leukocytes in relation to prenatal arsenic exposure [115]. Importantly, this analysis identified a positive association of DNA methylation of CpG regions within 71 genes and levels of iAs in maternal drinking water after adjusting for the shifts in white blood cell populations that were observed in the blood of prenatally exposed newborns [115]. Another study identified sex-dependent DNA methylation profiles in prenatally exposed newborns in Bangladesh [116]. Using several different methods to analyze global DNA methylation status in newborn cord blood, data trends indicated the relationship between global DNA methylation levels and maternal U-tAs was positive for newborn males and negative for newborn females, which is interesting in light of various sex-dependent effects associated with iAs exposure [116].

Recently, we reported the first analysis of miRNA perturbations of newborns born to iAsexposed mothers from Gómez Palacio, Mexico [117]. This analysis identified 12 miRNAs and 334 mRNAs in newborn cord blood that were associated with maternal U-tAs. Functions associated with the modulated miRNAs included diseases and cellular responses with known links to iAs exposure including cancer, diabetes mellitus, respiratory diseases, and inflammatory responses. MiRNAs act as post-transcriptional regulators of gene expression by binding to their mRNA target(s), and nine of the U-tAs-associated miRNAs were predicted to regulate the expression levels of \approx 20% of the U-tAs-associated mRNAs. Major functions associated with the modulated mRNAs included an enrichment of those associated with innate or adaptive immune response signaling. These results highlight that multiple forms of epigenetic regulators may play a role in mediating the cellular response to iAs exposure.

Taken together, these results demonstrate that arsenic exposure *in utero* has been associated with both genome-wide and gene-specific changes in DNA methylation levels, including changes in genes such as *TP53* that are implicated in the development of arsenic-associated diseases. However, many of these changes have been examined in leukocyte-derived DNA and it is unknown whether these changes have functional consequences in terms of altered expression levels of these genes at the mRNA level and subsequently the protein level. It is also unknown whether these epigenetic alterations are persistent over time, how postnatal exposure to arsenic or other toxicants impacts the epigenetic changes observed in prenatally exposed individuals, or whether the relationship between these epigenetic alterations and disease susceptibility later in life.

17.3.2 Role of Cancer Stem Cells

Evidence from *in vitro* and *in vivo* studies has implicated cancer stem cells (CSCs) as potential origins of cancers associated with prenatal arsenic exposure. CSCs are believed to arise in the fetal environment from normal stem cells (SCs) or slightly differentiated SC progeny as a result of prenatal exposure [118]. These CSCs, which have altered SC functions, remain quiescent in

target organs until ultimately transforming into malignant cells [118]. The triggers that end the quiescent state to produce cancers are unknown but may include the effects of continued exposure to arsenic during the postnatal period or postnatal exposure to other chemicals as observed in the transplacental mouse exposure models described above [92–96]. Evidence that supports CSCs as potentially important factors in the development of arsenic-related cancers includes their high numbers in arsenic-transformed cell lines and malignant tumors associated with arsenic exposure [118]. For instance, the malignant transformation of the prostate epithelial cell line RWPE-1 by arsenic exposure in vitro results in an overabundance of CSCs compared to RWPE-1s malignantly transformed by cadmium or *N*-methyl-*N*-nitrosurea [119]. The malignant transformation of prostate stem cells (WPE-stem) by arsenic exposure in vitro resulted in the production of highly aggressive CSCs. Here the CSC phenotype was accompanied by the loss of the expression of normal SC markers and reduced expression of the tumor suppressor phosphatase and tensin homolog (PTEN), thus indicating the capacity of SCs to serve as progenitors of a CSC population [120]. An overabundance of CSCs is observed in numerous tumors associated with prenatal arsenic exposure in mice, such as the lung and liver cancers in CD1 mice after whole-life exposure to arsenic [97] and in the highly aggressive SCCs that developed in Tg.AC mice that were exposed prenatally to arsenic and postnatally to TPA [96].

Evidence for CSCs developing in the fetal environment and the potential for aberrant gene expression as a key, early event in CSC development is also observed in the fetal skin (GD 19) of Tg.AC mice prenatally exposed to arsenic. This is characterized by the increased expression of several genes such as the oncogene *v*-*Ha*-*ras*, *CD34*, a cell-surface marker for keratinocyte stem SCs and cancer skin cells, and ras-related C3 botulinum toxin substrate 1 (*Rac1*), a gene involved in keratinocyte SC renewal, compared to unexposed mice [96]. *v*-*Ha*-*ras* activation is considered a potential key, early event in Tg.AC skin cancer development, and while *v*-*Ha*-*ras* expression has been shown to be dependent on promoter DNA methylation status [121], no alteration in the DNA methylation status of the *v*-*Ha*-*ras* promoter was observed after prenatal arsenic exposure [96]. These results suggest that not all arsenic-associated aberrant gene expression observed in the fetal environment is associated with changes in promoter DNA methylation status of that gene. This finding suggests that other transcriptional regulators play a role in the control of *v*-*Ha*-*ras* expression after prenatal iAs exposure.

Similar to results observed in fetal skin, an examination of the highly aggressive SCCs that developed in Tg.AC mice after prenatal arsenic exposure and postnatal TPA exposure also revealed high expression of *v*-Ha-ras and the CSC markers *CD34* and *Rac* compared to tumors that arose after TPA treatment alone. A comparison of the tumors arising from these two treatment groups showed increased expression levels of the pro-growth gene *Cd1* and decreased expression of tumor suppressor *p16* in the aggressive SCCs in the iAs arsenic/TPA group. These data illustrate how subsequent alterations in genomic signaling may contribute to the acquisition of the malignant phenotype and potentially end the quiescence period in these prenatally exposed mice [96].

Taken together these data support that tumors associated with prenatal arsenic exposure have high numbers of CSCs and have possibly acquired aggressive characteristics through the altered expression of pro-growth genes and tumor suppressors. In human populations, it is unknown if tumors associated with prenatal arsenic exposure have an overabundance of CSCs as observed in the aforementioned animal models. In addition, the stimuli and molecular events that end the quiescent state associated with prenatal arsenic exposure to produce malignancies have yet to be elucidated.

17.3.3 Immunomodulatory Effects

Arsenic has been shown to be a potent immunomodulatory agent *in vitro, in vivo,* and in human populations [122]. Analyses of perturbed mRNAs and miRNAs in the cord blood of iAs-exposed newborns indicated that the immune response is a major altered function associated with prenatal iAs exposure [107,117]. In chronically and prenatally exposed human populations, there is evidence that arsenic can cause both inflammation and immunosuppression, which is likely of major significance as each effect has been linked to adverse health outcomes.

As a potent inducer of ROS, arsenic can cause inflammation via the activation of ROSsensitive pro-inflammatory signaling pathways such as those that involve mitogen-activated protein kinases (MAPKs) and the transcription factor NF- κ B [123]. There is evidence that prenatal iAs exposure is associated with inflammation as there was increased expression of pro-inflammatory genes at the mRNA level and protein level in newborn cord blood, oxidative DNA damage in the placenta, and induction of pro-inflammatory proteins in the placenta such as IL-1 β , TNF, and IFN- γ [107,109]. One of the adverse effects of a prolonged inflammatory state is the inflammation-induced generation of oxidative stress that can cause tissue/macromolecule damage, and inflammation during gestation has been associated with adverse pregnancy outcomes such as preterm birth, preeclampsia, and fetal growth restriction [124,125]. In addition to effects in early life, sustained inflammation is implicated in the development of chronic diseases that occur later in life through the activation of disease-promoting proteins such as growth factors and angiogenic factors [123,126]. This pro-inflammatory state is believed to be a major mechanism by which ROS-producing metalloids elicit carcinogenic effects [66]. Furthermore, inflammation is believed to play a role in the development of other iAs-associated diseases such as diabetes mellitus, atherosclerosis, and liver fibrosis [127–129].

Arsenic exposure has also been associated with immunosuppression, which has been demonstrated in several *in vivo* systems including zebrafish [130] and mice [131]. Arsenic decreases the functionality of several types of cells involved in the host defense system including peripheral blood mononuclear cells [132], macrophages [131], and T helper cells [133]. Arsenic exposure is associated with insufficient immune response to infection in mice and markers of immunosuppression in chronically exposed children and adults [132,134–137]. Prenatal exposure is associated with alterations of T cell populations in the cord blood of exposed newborns [115] and increased morbidity and mortality and reduced thymic index in infants [33,34,138]. The immunosuppressive effects *in utero* may not only be linked to the increased morbidity and mortality that is associated with iAs exposure in early life [33,34]; if this immunosuppressive effect is maintained, it likely also increases the susceptibility to infections and chronic diseases later in life [139,140]. Interestingly, it has been shown that the immunosuppressive effects of arsenic can lead to complex adverse health outcomes. For instance, arsenic-exposed C57BL/6J mice have been shown to exhibit a biphasic response in response to influenza A (H1N1) challenge in the lungs [134]. Whereas arsenic-exposed mice

initially have an inadequate immune response to influenza challenge compared to arsenicunexposed mice, this response is followed by an excessive inflammatory response in the lung resulting in extensive tissue damage that is not observed in arsenic-unexposed, H1N1challenged mice [134]. Since markers of both immunosuppression and inflammation have been observed simultaneously in arsenic-exposed populations as well [135], it is likely that these two seemingly divergent effects are both involved in the development of the diverse adverse health effects associated with prenatal arsenic exposure.

17.4 Conclusions and Future Directions

In summary, it is clear that there are diverse and delayed adverse health effects associated with arsenic exposure in early life. Defined periods of exposure in human populations have proven invaluable in understanding the diversity of the health effects involved and the different latency periods associated with these effects, and how various factors may impact the nature of the health effects that develop. Both human studies and animal models have demonstrated that the prenatal/early-life period is a particularly vulnerable time in relation to these toxic effects and that exposure during this period is sufficient to significantly alter susceptibility to disease. Due to the latent nature of many of these effects, the true health impacts of such exposure are likely vastly underestimated. These results underscore the dire need to identify susceptible populations and critical windows of susceptibility. This information will be critical to successfully implement preventive measures to protect the health of currently exposed populations and potentially the health of future generations particularly, if the health effects observed are transgenerational in nature. Therefore, there is also a need to determine the potential of altered disease risk to be heritable in the germ line of both males and females.

Lastly, while many of the current research trends have identified several molecular mechanisms that may be involved in latent disease development, much remains to be learned about these mechanisms, particularly for non-cancer diseases. For instance, while studies of prenatally exposed individuals have identified alterations in DNA methylation profiles, it is unknown what, if any, biological relevance these alterations have in terms of altered gene expression or disease development. It is also unknown which persistent molecular effects contribute to disease development, which molecular effects are disease specific, and what factors (e.g., postnatal environment) contribute to the length of latency periods. Importantly, emerging technologies, which allow for rapid and increasingly comprehensive and affordable analyses, will likely greatly increase our knowledge of the molecular events associated with early life exposure to toxicants and their relationship with human disease development.

References

- World Health Organization. Guidelines for drinking water quality, vol. 1. Geneva, Switzerland: WHO Press; 2006. First addendum to 3rd addition.
- [2] Uddin R, Huda NH. Arsenic poisoning in Bangladesh. Oman Med J 2011;26:207.
- [3] Garelick H, Jones H, Dybowska A, Valsami-Jones E. Arsenic pollution sources. Rev Environ Contam Toxicol 2008;197:17–60.

- [4] Mukherjee A, Sengupta MK, Hossain MA, Ahamed S, Das B, Nayak B, et al. Arsenic contamination in groundwater: a global perspective with emphasis on the Asian scenario. J Health Popul Nutr 2006;24:142-63.
- [5] Sen P, Biswas T. Arsenic: the largest mass poisoning of a population in history. BMJ 2013;346:f3625.
- [6] Fewtrell L, Fuge R, Kay D. An estimation of the global burden of disease due to skin lesions caused by arsenic in drinking water. J Water Health 2005;3:101–7.
- [7] Smith AH, Lingas EO, Rahman M. Contamination of drinking-water by arsenic in Bangladesh: a public health emergency. Bull World Health Org 2000;78:1093–103.
- [8] Karagas MR. Arsenic-related mortality in Bangladesh. Lancet 2010;376:213-14.
- [9] Lokuge KM, Smith W, Caldwell B, Dear K, Milton AH. The effect of arsenic mitigation interventions on disease burden in Bangladesh. Environ Health Perspect 2004;112:1172–7.
- [10] Nordstrom DK. Worldwide occurrences of arsenic in ground water. Science 2002; 296:2143-5.
- [11] World Health Organization. Noncommunicable diseases country profiles 2011; 2011. 1–207 pp.
- [12] Rappaport SM, Smith MT. Epidemiology. Environment and disease risks. Science 2010;330:460-1.
- [13] Jardim MJ. microRNAs: implications for air pollution research. Mutat Res 2011;717:38-45.
- [14] Gilliland FD, Li YF, Peters JM. Effects of maternal smoking during pregnancy and environmental tobacco smoke on asthma and wheezing in children. Am J Respir Crit Care Med 2001;163:429–36.
- [15] Gilliland FD, Berhane K, Li YF, Rappaport EB, Peters JM. Effects of early onset asthma and in utero exposure to maternal smoking on childhood lung function. Am J Respir Crit Care Med 2003;167:917–24.
- [16] Li YF, Gilliland FD, Berhane K, McConnell R, Gauderman WJ, Rappaport EB, et al. Effects of in utero and environmental tobacco smoke exposure on lung function in boys and girls with and without asthma. Am J Respir Crit Care Med 2000;162:2097–104.
- [17] Centeno JA, Mullick FG, Martinez L, Page NP, Gibb H, Longfellow D, et al. Pathology related to chronic arsenic exposure. Environ Health Perspect 2002;110(Suppl. 5):883–6.
- [18] Smith AH, Lopipero PA, Bates MN, Steinmaus CM. Public health. Arsenic epidemiology and drinking water standards. Science 2002;296:2145–6.
- [19] Naujokas MF, Anderson B, Ahsan H, Aposhian HV, Graziano JH, Thompson C, et al. The broad scope of health effects from chronic arsenic exposure: update on a worldwide public health problem. Environ Health Perspect 2013;121:295–302.
- [20] Rahman MM, Ng JC, Naidu R. Chronic exposure of arsenic via drinking water and its adverse health impacts on humans. Environ Geochem Health 2009;31(Suppl. 1):189–200.
- [21] Sohel N, Persson LA, Rahman M, Streatfield PK, Yunus M, Ekstrom EC, et al. Arsenic in drinking water and adult mortality: a population-based cohort study in rural Bangladesh. Epidemiology 2009;20:824–30.
- [22] Argos M, Kalra T, Rathouz PJ, Chen Y, Pierce B, Parvez F, et al. Arsenic exposure from drinking water, and all-cause and chronic-disease mortalities in Bangladesh (HEALS): a prospective cohort study. Lancet 2010;376:252–8.
- [23] Hall M, Gamble M, Slavkovich V, Liu X, Levy D, Cheng Z, et al. Determinants of arsenic metabolism: blood arsenic metabolites, plasma folate, cobalamin, and homocysteine concentrations in maternal-newborn pairs. Environ Health Perspect 2007;115:1503–9.
- [24] Hopenhayn C, Ferreccio C, Browning SR, Huang B, Peralta C, Gibb H, et al. Arsenic exposure from drinking water and birth weight. Epidemiology 2003;14:593–602.
- [25] Huyck KL, Kile ML, Mahiuddin G, Quamruzzaman Q, Rahman M, Breton CV, et al. Maternal arsenic exposure associated with low birth weight in Bangladesh. J Occup Environ Med 2007;49:1097–104.
- [26] Rahman A, Vahter M, Ekstrom EC, Rahman M, Golam Mustafa AH, Wahed MA, et al. Association of arsenic exposure during pregnancy with fetal loss and infant death: a cohort study in Bangladesh. Am J Epidemiol 2007;165:1389–96.

- [27] Rahman A, Vahter M, Smith AH, Nermell B, Yunus M, El Arifeen S, et al. Arsenic exposure during pregnancy and size at birth: a prospective cohort study in Bangladesh. Am J Epidemiol 2009;169:304–12.
- [28] Ahmad SA, Sayed MH, Barua S, Khan MH, Faruquee MH, Jalil A, et al. Arsenic in drinking water and pregnancy outcomes. Environ Health Perspect 2001;109:629–31.
- [29] Vahter M. Effects of arsenic on maternal and fetal health. Annu Rev Nutr 2009;29:381–99.
- [30] Hill DS, Wlodarczyk BJ, Finnell RH. Reproductive consequences of oral arsenate exposure during pregnancy in a mouse model. Birth Defects Res B Dev Reprod Toxicol 2008;83:40–7.
- [31] Parajuli RP, Fujiwara T, Umezaki M, Watanabe C. Association of cord blood levels of lead, arsenic, and zinc with neurodevelopmental indicators in newborns: a birth cohort study in Chitwan Valley, Nepal. Environ Res 2013;121:45–51.
- [32] Rahman A, Vahter M, Ekstrom EC, Persson LA. Arsenic exposure in pregnancy increases the risk of lower respiratory tract infection and diarrhea during infancy in Bangladesh. Environ Health Perspect 2011;119:719–24.
- [33] Rahman A, Persson LA, Nermell B, El Arifeen S, Ekstrom EC, Smith AH, et al. Arsenic exposure and risk of spontaneous abortion, stillbirth, and infant mortality. Epidemiology 2010;21:797–804.
- [34] Raqib R, Ahmed S, Sultana R, Wagatsuma Y, Mondal D, Hoque AM, et al. Effects of in utero arsenic exposure on child immunity and morbidity in rural Bangladesh. Toxicol Lett 2009;185:197–202.
- [35] Mazumder DN. Effect of drinking arsenic contaminated water in children. Indian Pediatr 2007;44:925-7.
- [36] Smith AH, Steinmaus CM. Health effects of arsenic and chromium in drinking water: recent human findings. Annu Rev Public Health 2009;30:107–22.
- [37] Liaw J, Marshall G, Yuan Y, Ferreccio C, Steinmaus C, Smith AH. Increased childhood liver cancer mortality and arsenic in drinking water in northern Chile. Cancer Epidemiol Biomarkers Prev 2008;17:1982–7.
- [38] Jing J, Zheng G, Liu M, Shen X, Zhao F, Wang J, et al. Changes in the synaptic structure of hippocampal neurons and impairment of spatial memory in a rat model caused by chronic arsenite exposure. Neurotoxicology 2012;33:1230–8.
- [39] Bardullas U, Limon-Pacheco JH, Giordano M, Carrizales L, Mendoza-Trejo MS, Rodriguez VM. Chronic low-level arsenic exposure causes gender-specific alterations in locomotor activity, dopaminergic systems, and thioredoxin expression in mice. Toxicol Appl Pharmacol 2009;239:169–77.
- [40] Luo J, Qiu Z, Chen J, Zhang L, Liu W, Tan Y, et al. Maternal and early life arsenite exposure impairs neurodevelopment and increases the expression of PSA-NCAM in hippocampus of rat offspring. Toxicology 2013;311:99–106.
- [41] Wasserman GA, Liu X, Parvez F, Ahsan H, Factor-Litvak P, Kline J, et al. Water arsenic exposure and intellectual function in 6-year-old children in Araihazar, Bangladesh. Environ Health Perspect 2007;115:285–9.
- [42] Wasserman GA, Liu X, Parvez F, Ahsan H, Factor-Litvak P, van Geen A, et al. Water arsenic exposure and children's intellectual function in Araihazar, Bangladesh. Environ Health Perspect 2004;112:1329–33.
- [43] von Ehrenstein OS, Poddar S, Yuan Y, Mazumder DG, Eskenazi B, Basu A, et al. Children's intellectual function in relation to arsenic exposure. Epidemiology 2007;18:44–51.
- [44] Rosado JL, Ronquillo D, Kordas K, Rojas O, Alatorre J, Lopez P, et al. Arsenic exposure and cognitive performance in Mexican schoolchildren. Environ Health Perspect 2007;115:1371–5.
- [45] Hamadani JD, Tofail F, Nermell B, Gardner R, Shiraji S, Bottai M, et al. Critical windows of exposure for arsenic-associated impairment of cognitive function in pre-school girls and boys: a population-based cohort study. Int J Epidemiol 2011;40:1593–604.
- [46] Tsai SY, Chou HY, The HW, Chen CM, Chen CJ. The effects of chronic arsenic exposure from drinking water on the neurobehavioral development in adolescence. Neurotoxicology 2003;24:747–53.
- [47] Parvez F, Wasserman GA, Factor-Litvak P, Liu X, Slavkovich V, Siddique AB, et al. Arsenic exposure and motor function among children in Bangladesh. Environ Health Perspect 2011;119:1665–70.

- [48] Dakeishi M, Murata K, Grandjean P. Long-term consequences of arsenic poisoning during infancy due to contaminated milk powder. Environ Health 2006;5:31.
- [49] Wright RO, Amarasiriwardena C, Woolf AD, Jim R, Bellinger DC. Neuropsychological correlates of hair arsenic, manganese, and cadmium levels in school-age children residing near a hazardous waste site. Neurotoxicology 2006;27:210–16.
- [50] Smith AH, Goycolea M, Haque R, Biggs ML. Marked increase in bladder and lung cancer mortality in a region of Northern Chile due to arsenic in drinking water. Am J Epidemiol 1998;147:660–9.
- [51] Ferreccio C, Smith AH, Duran V, Barlaro T, Benitez H, Valdes R, et al. Case-control study of arsenic in drinking water and kidney cancer in uniquely exposed Northern Chile. Am J Epidemiol 2013;178:813–18.
- [52] Steinmaus CM, Ferreccio C, Acevedo Romo J, Yuan Y, Cortes S, Marshall G, et al. Drinking water arsenic in northern Chile: high cancer risks 40 years after exposure cessation. Cancer Epidemiol Biomarkers Prev 2013;22:623–30.
- [53] Fraser B. Cancer cluster in Chile linked to arsenic contamination. Lancet 2012;379:603.
- [54] Marshall G, Ferreccio C, Yuan Y, Bates MN, Steinmaus C, Selvin S, et al. Fifty-year study of lung and bladder cancer mortality in Chile related to arsenic in drinking water. J Natl Cancer Inst 2007;99:920–8.
- [55] Yuan Y, Marshall G, Ferreccio C, Steinmaus C, Liaw J, Bates M, et al. Kidney cancer mortality: fifty-year latency patterns related to arsenic exposure. Epidemiology 2010;21:103–8.
- [56] Yuan Y, Marshall G, Ferreccio C, Steinmaus C, Selvin S, Liaw J, et al. Acute myocardial infarction mortality in comparison with lung and bladder cancer mortality in arsenic-exposed region II of Chile from 1950 to 2000. Am J Epidemiol 2007;166:1381–91.
- [57] Smith AH, Marshall G, Yuan Y, Liaw J, Ferreccio C, Steinmaus C. Evidence from Chile that arsenic in drinking water may increase mortality from pulmonary tuberculosis. Am J Epidemiol 2011;173:414–20.
- [58] Smith AH, Marshall G, Yuan Y, Ferreccio C, Liaw J, von Ehrenstein O, et al. Increased mortality from lung cancer and bronchiectasis in young adults after exposure to arsenic in utero and in early childhood. Environ Health Perspect 2006;114:1293–6.
- [59] Dauphine DC, Ferreccio C, Guntur S, Yuan Y, Hammond SK, Balmes J, et al. Lung function in adults following in utero and childhood exposure to arsenic in drinking water: preliminary findings. Int Arch Occup Environ Health 2011;84:591–600.
- [60] Smith AH, Marshall G, Liaw J, Yuan Y, Ferreccio C, Steinmaus C. Mortality in young adults following in utero and childhood exposure to arsenic in drinking water. Environ Health Perspect 2012;120:1527–31.
- [61] Yorifuji T, Tsuda T, Grandjean P. Unusual cancer excess after neonatal arsenic exposure from contaminated milk powder. J Natl Cancer Inst 2010;102:360–1.
- [62] Yorifuji T, Tsuda T, Doi H, Grandjean P. Cancer excess after arsenic exposure from contaminated milk powder. Environ Health Prev Med 2011;16:164–70.
- [63] Huang C, Ke Q, Costa M, Shi X. Molecular mechanisms of arsenic carcinogenesis. Mol Cell Biochem 2004;255:57-66.
- [64] Tapio S, Grosche B. Arsenic in the aetiology of cancer. Mutat Res 2006;612:215-46.
- [65] Shi H, Shi X, Liu KJ. Oxidative mechanism of arsenic toxicity and carcinogenesis. Mol Cell Biochem 2004;255:67–78.
- [66] Valko M, Rhodes CJ, Moncol J, Izakovic M, Mazur M. Free radicals, metals and antioxidants in oxidative stress-induced cancer. Chem Biol Interact 2006;160:1–40.
- [67] Dolinoy DC, Jirtle RL. Environmental epigenomics in human health and disease. Environ Mol Mutagen 2008;49:4–8.
- [68] Inbar-Feigenberg M, Choufani S, Butcher DT, Roifman M, Weksberg R. Basic concepts of epigenetics. Fertil Steril 2013;99:607–15.

- [69] Barros SP, Offenbacher S. Epigenetics: connecting environment and genotype to phenotype and disease. J Dent Res 2009;88:400–8.
- [70] Bird A. The essentials of DNA methylation. Cell 1992;70:5-8.
- [71] Irizarry RA, Ladd-Acosta C, Wen B, Wu Z, Montano C, Onyango P, et al. The human colon cancer methylome shows similar hypo- and hypermethylation at conserved tissue-specific CpG island shores. Nat Genet 2009;41:178–86.
- [72] Gardiner-Garden M, Frommer M. CpG islands in vertebrate genomes. J Mol Biol 1987;196:261-82.
- [73] Haluskova J. Epigenetic studies in human diseases. Folia Biologica 2010;56:83-96.
- [74] Fuks F. DNA methylation and histone modifications: teaming up to silence genes. Curr Opin Genet Dev 2005;15:490–5.
- [75] Li X, Wang X, He K, Ma Y, Su N, He H, et al. High-resolution mapping of epigenetic modifications of the rice genome uncovers interplay between DNA methylation, histone methylation, and gene expression. Plant Cell 2008;20:259–76.
- [76] Cedar H, Bergman Y. Linking DNA methylation and histone modification: patterns and paradigms. Nat Rev Genet 2009;10:295–304.
- [77] van Eijk KR, de Jong S, Boks MP, Langeveld T, Colas F, Veldink JH, et al. Genetic analysis of DNA methylation and gene expression levels in whole blood of healthy human subjects. BMC Genomics 2012;13:636.
- [78] Boellmann F, Zhang L, Clewell HJ, Schroth GP, Kenyon EM, Andersen ME, et al. Genome-wide analysis of DNA methylation and gene expression changes in the mouse lung following subchronic arsenate exposure. Toxicol Sci 2010;117:404–17.
- [79] Portela A, Esteller M. Epigenetic modifications and human disease. Nat Biotechnol 2010;28:1057–68.
- [80] Bollati V, Baccarelli A. Environmental epigenetics. Heredity (Edinb) 2010;105-105-12.
- [81] Baccarelli A, Bollati V. Epigenetics and environmental chemicals. Curr Opin Pediatr 2009;21:243–51.
- [82] Dolinoy DC, Weidman JR, Jirtle RL. Epigenetic gene regulation: linking early developmental environment to adult disease. Reprod Toxicol 2007;23:297–307.
- [83] Cropley JE, Suter CM, Beckman KB, Martin DI. Germ-line epigenetic modification of the murine A^{vy} allele by nutritional supplementation. Proc Natl Acad Sci USA 2006;103:17308–12.
- [84] Jirtle RL, Skinner MK. Environmental epigenomics and disease susceptibility. Nat Rev Genet 2007;8:253–62.
- [85] Ho SM, Tang WY, Belmonte de Frausto J, Prins GS. Developmental exposure to estradiol and bisphenol A increases susceptibility to prostate carcinogenesis and epigenetically regulates phosphodiesterase type 4 variant 4. Cancer Res 2006;66:5624–32.
- [86] Lillycrop KA, Burdge GC. Epigenetic mechanisms linking early nutrition to long term health. Best Pract Res Clin Endocrinol Metab 2012;26:667–76.
- [87] Vahter M. Health effects of early life exposure to arsenic. Basic Clin Pharmacol Toxicol 2008;102:204-11.
- [88] Tokar EJ, Benbrahim-Tallaa L, Ward JM, Lunn R, Sams II RL, Waalkes MP. Cancer in experimental animals exposed to arsenic and arsenic compounds. Crit Rev Toxicol 2010;40:912–27.
- [89] Waalkes MP, Liu J, Ward JM, Diwan BA. Animal models for arsenic carcinogenesis: inorganic arsenic is a transplacental carcinogen in mice. Toxicol Appl Pharmacol 2004;198:377–84.
- [90] Waalkes MP, Ward JM, Liu J, Diwan BA. Transplacental carcinogenicity of inorganic arsenic in the drinking water: induction of hepatic, ovarian, pulmonary, and adrenal tumors in mice. Toxicol Appl Pharmacol 2003;186:7–17.
- [91] Lin HJ, Sung TI, Chen CY, Guo HR. Arsenic levels in drinking water and mortality of liver cancer in Taiwan. J Hazard Mater 2013;262:1132–8.

- [92] Waalkes MP, Liu J, Ward JM, Powell DA, Diwan BA. Urogenital carcinogenesis in female CD1 mice induced by in utero arsenic exposure is exacerbated by postnatal diethylstilbestrol treatment. Cancer Res 2006;66:1337–45.
- [93] Waalkes MP, Liu J, Ward JM, Diwan BA. Enhanced urinary bladder and liver carcinogenesis in male CD1 mice exposed to transplacental inorganic arsenic and postnatal diethylstilbestrol or tamoxifen. Toxicol Appl Pharmacol 2006;215:295–305.
- [94] Waalkes MP, Ward JM, Diwan BA. Induction of tumors of the liver, lung, ovary and adrenal in adult mice after brief maternal gestational exposure to inorganic arsenic: promotional effects of postnatal phorbol ester exposure on hepatic and pulmonary, but not dermal cancers. Carcinogenesis 2004;25:133–41.
- [95] Tokar EJ, Diwan BA, Waalkes MP. Arsenic exposure in utero and nonepidermal proliferative response in adulthood in Tg.AC mice. Int J Toxicol 2010;29:291–6.
- [96] Waalkes MP, Liu J, Germolec DR, Trempus CS, Cannon RE, Tokar EJ, et al. Arsenic exposure in utero exacerbates skin cancer response in adulthood with contemporaneous distortion of tumor stem cell dynamics. Cancer Res 2008;68:8278–85.
- [97] Tokar EJ, Diwan BA, Ward JM, Delker DA, Waalkes MP. Carcinogenic effects of "whole-life" exposure to inorganic arsenic in CD1 mice. Toxicol Sci 2011;119:73–83.
- [98] Xie Y, Liu J, Benbrahim-Tallaa L, Ward JM, Logsdon D, Diwan BA, et al. Aberrant DNA methylation and gene expression in livers of newborn mice transplacentally exposed to a hepatocarcinogenic dose of inorganic arsenic. Toxicology 2007;236:7–15.
- [99] Liu J, Xie Y, Ward JM, Diwan BA, Waalkes MP. Toxicogenomic analysis of aberrant gene expression in liver tumors and nontumorous livers of adult mice exposed in utero to inorganic arsenic. Toxicol Sci 2004;77:249–57.
- [100] Waalkes MP, Liu J, Chen H, Xie Y, Achanzar WE, Zhou YS, et al. Estrogen signaling in livers of male mice with hepatocellular carcinoma induced by exposure to arsenic in utero. J Natl Cancer Inst 2004;96:466–74.
- [101] Chen H, Li S, Liu J, Diwan BA, Barrett JC, Waalkes MP. Chronic inorganic arsenic exposure induces hepatic global and individual gene hypomethylation: implications for arsenic hepatocarcinogenesis. Carcinogenesis 2004;25:1779–86.
- [102] Yager JD, Liehr JG. Molecular mechanisms of estrogen carcinogenesis. Annu Rev Pharmacol Toxicol 1996;36:203–32.
- [103] Deane NG, Parker MA, Aramandla R, Diehl L, Lee WJ, Washington MK, et al. Hepatocellular carcinoma results from chronic cyclin D1 overexpression in transgenic mice. Cancer Res 2001;61:5389–95.
- [104] Shen J, Liu J, Xie Y, Diwan BA, Waalkes MP. Fetal onset of aberrant gene expression relevant to pulmonary carcinogenesis in lung adenocarcinoma development induced by in utero arsenic exposure. Toxicol Sci 2007;95:313–20.
- [105] Srivastava S, D'Souza SE, Sen U, States JC. In utero arsenic exposure induces early onset of atherosclerosis in ApoE^{-/-} mice. Reprod Toxicol 2007;23:449–56.
- [106] States JC, Singh AV, Knudsen TB, Rouchka EC, Ngalame NO, Arteel GE, et al. Prenatal arsenic exposure alters gene expression in the adult liver to a proinflammatory state contributing to accelerated atherosclerosis. PLOS ONE 2012;7:e38713.
- [107] Fry RC, Navasumrit P, Valiathan C, Svensson JP, Hogan BJ, Luo M, et al. Activation of inflammation/ NF-kappaB signaling in infants born to arsenic-exposed mothers. PLoS Genet 2007;3:e207.
- [108] Hoesel B, Schmid JA. The complexity of NF-κB signaling in inflammation and cancer. Mol Cancer 2013;12:86.
- [109] Ahmed S, Mahabbat-e Khoda S, Rekha RS, Gardner RM, Ameer SS, Moore S, et al. Arsenic-associated oxidative stress, inflammation, and immune disruption in human placenta and cord blood. Environ Health Perspect 2011;119:258–64.

- [110] Eden A, Gaudet F, Waghmare A, Jaenisch R. Chromosomal instability and tumors promoted by DNA hypomethylation. Science 2003;300:455.
- [111] Wilson AS, Power BE, Molloy PL. DNA hypomethylation and human diseases. Biochim Biophys Acta 2007;1775:138-62.
- [112] Intarasunanont P, Navasumrit P, Waraprasit S, Chaisatra K, Suk WA, Mahidol C, et al. Effects of arsenic exposure on DNA methylation in cord blood samples from newborn babies and in a human lymphoblast cell line. Environ Health 2012;11:31.
- [113] Kile ML, Baccarelli A, Hoffman E, Tarantini L, Quamruzzaman Q, Rahman M, et al. Prenatal arsenic exposure and DNA methylation in maternal and umbilical cord blood leukocytes. Environ Health Perspect 2012;120:1061–6.
- [114] Koestler DC, Avissar-Whiting M, Houseman EA, Karagas MR, Marsit CJ. Differential DNA methylation in umbilical cord blood of infants exposed to low levels of arsenic in utero. Environ Health Perspect 2013;121:971–7.
- [115] Kile ML, Houseman EA, Baccarelli A, Quamruzzaman Q, Rahman M, Mostofa G, et al. Effect of prenatal arsenic exposure on DNA methylation and leukocyte subpopulations in cord blood. Epigenetics 2014;9:774–82.
- [116] Pilsner JR, Hall MN, Liu X, Ilievski V, Slavkovich V, Levy D, et al. Influence of prenatal arsenic exposure and newborn sex on global methylation of cord blood DNA. PLOS ONE 2012;7:e37147.
- [117] Rager JE, Bailey KA, Smeester L, Miller SK, Parker JS, Laine JE, et al. Prenatal arsenic exposure and the epigenome: altered microRNAs associated with innate and adaptive immune signaling in newborn cord blood. Environ Mol Mutagen 2014;55(3):196–208.
- [118] Tokar EJ, Qu W, Waalkes MP. Arsenic, stem cells, and the developmental basis of adult cancer. Toxicol Sci 2011;120(Suppl. 1):S192-203.
- [119] Tokar EJ, Qu W, Liu J, Liu W, Webber MM, Phang JM, et al. Arsenic-specific stem cell selection during malignant transformation. J Natl Cancer Inst 2010;102:638–49.
- [120] Tokar EJ, Diwan BA, Waalkes MP. Arsenic exposure transforms human epithelial stem/progenitor cells into a cancer stem-like phenotype. Environ Health Perspect 2010;118:108–15.
- [121] Cannon RE, Spalding JW, Virgil KM, Faircloth RS, Humble MC, Lacks GD, et al. Induction of transgene expression in Tg.AC(v-Ha-ras) transgenic mice concomitant with DNA hypomethylation. Mol Carcinog 1998;21:244–50.
- [122] Dangleben NL, Skibola CF, Smith MT. Arsenic immunotoxicity: a review. Environ Health 2013;12:73.
- [123] Sesti F, Tsitsilonis OE, Kotsinas A, Trougakos IP. Oxidative stress-mediated biomolecular damage and inflammation in tumorigenesis. In Vivo 2012;26:395–402.
- [124] Vahter ME. Interactions between arsenic-induced toxicity and nutrition in early life. J Nutr 2007;137:2798–804.
- [125] Challis JR, Lockwood CJ, Myatt L, Norman JE, Strauss III JF, Petraglia F. Inflammation and pregnancy. Reprod Sci 2009;16:206–15.
- [126] Reuter S, Gupta SC, Chaturvedi MM, Aggarwal BB. Oxidative stress, inflammation, and cancer: how are they linked? Free Radic Biol Med 2010;49:1603–16.
- [127] Iredale JP. Models of liver fibrosis: exploring the dynamic nature of inflammation and repair in a solid organ. J Clin Invest 2007;117:539–48.
- [128] Libby P, Ridker PM, Maseri A. Inflammation and atherosclerosis. Circulation 2002;105:1135–43.
- [129] Cosentino F, Egidy Assenza G. Diabetes and inflammation. Herz 2004;29:749-59.
- [130] Nayak AS, Lage CR, Kim CH. Effects of low concentrations of arsenic on the innate immune system of the zebrafish (*Danio rerio*). Toxicol Sci 2007;98:118–24.

- [131] Lewis TA, Hartmann CB, McCoy KL. Gallium arsenide modulates proteolytic cathepsin activities and antigen processing by macrophages. J Immunol 1998;161:2151–7.
- [132] Gonsebatt ME, Vega L, Montero R, Garcia-Vargas G, Del Razo LM, Albores A, et al. Lymphocyte replicating ability in individuals exposed to arsenic via drinking water. Mutat Res 1994;313:293–9.
- [133] Soto-Pena GA, Vega L. Arsenic interferes with the signaling transduction pathway of T cell receptor activation by increasing basal and induced phosphorylation of Lck and Fyn in spleen cells. Toxicol Appl Pharmacol 2008;230:216–26.
- [134] Kozul CD, Ely KH, Enelow RI, Hamilton JW. Low-dose arsenic compromises the immune response to influenza A infection in vivo. Environ Health Perspect 2009;117:1441–7.
- [135] Soto-Pena GA, Luna AL, Acosta-Saavedra L, Conde P, Lopez-Carrillo L, Cebrian ME, et al. Assessment of lymphocyte subpopulations and cytokine secretion in children exposed to arsenic. FASEB J 2006;20:779–81.
- [136] Biswas R, Ghosh P, Banerjee N, Das JK, Sau T, Banerjee A, et al. Analysis of T-cell proliferation and cytokine secretion in the individuals exposed to arsenic. Hum Exp Toxicol 2008;27:381–6.
- [137] Andrew AS, Jewell DA, Mason RA, Whitfield ML, Moore JH, Karagas MR. Drinking-water arsenic exposure modulates gene expression in human lymphocytes from a U.S. population. Environ Health Perspect 2008;116:524–31.
- [138] Ahmed S, Ahsan KB, Kippler M, Mily A, Wagatsuma Y, Hoque AM, et al. In utero arsenic exposure is associated with impaired thymic function in newborns possibly via oxidative stress and apoptosis. Toxicol Sci 2012;129:305–14.
- [139] DeWitt JC, Peden-Adams MM, Keil DE, Dietert RR. Current status of developmental immunotoxicity: early-life patterns and testing. Toxicol Pathol 2012;40:230–6.
- [140] Vega L, Soto G, Luna A, Acosta L, Conde P, Cebrián M, et al. Early signs of immunodepression induced by arsenic in children. Natural Arsenic in Groundwaters of Latin America 1; 2009. p. 435.

18

Arsenic, Kidney, and Urinary Bladder Disorders

Ernesto Sabath

UNIVERSIDAD AUTONOMA DE QUERETARO, QUERETARO, MEXICO AND RENAL DEPARTMENT, HOSPITAL GENERAL DE QUERETARO, QUERETARO, MEXICO

CHAPTER OUTLINE

18.1 Introduction	429
18.2 Arsenic and Renal Disease	430
18.3 Arsenic and Bladder Disease	436
References	438

18.1 Introduction

Both the incidence and prevalence of type 2 diabetes mellitus, hypertension, chronic renal failure, and cancer have risen constantly over the last three decades, and these diseases are now a growing public health problem, so the identification of risk factors associated with them is essential in order to take preventive measures.

Recently, the study of the toxic effects of arsenic (As) on the human body has become especially important. This is because, in the last 100 years, arsenic began to be used as a pesticide, a chemotherapeutic agent, and a constituent of consumer products; however, was arsenic concentration in drinking water and foods the most important concern for public health?

Arsenic is one of the most common heavy metals found, and the causal association between As and the formation of tumors in the skin, lungs, bladder, liver, and kidneys has been exhaustively described. Some epidemiological studies have shown an association between exposure to high levels of As and increased risk of cardiovascular disease, hypertension, and diabetes mellitus; however, the literature on the effects of As on the renal function of the general population is still scarce.

The aim of this chapter is to analyze the epidemiology, physiopathology, and clinical manifestations of nephrotoxicity and bladder complications associated with arsenic exposure.

18.2 Arsenic and Renal Disease

Chronic kidney disease (CKD) is an important public health problem that is characterized by poor health outcomes and very high healthcare costs; CKD is a devastating disease that represents a high socioeconomic burden to family and society.

CKD is defined as a reduced glomerular filtration rate, increased urinary albumin excretion, or both for a period longer than 3 months [1]. The incidence and prevalence of end-stage kidney disease differ substantially across countries and regions, and prevalence is estimated to be 8–16% worldwide [2]. Complications include increased all-cause and cardiovascular mortality [3], acute kidney injury [4], cognitive decline [5], anemia [6], fractures, and other mineral disorders [7]. Identification and staging of CKD rely on measurement of glomerular filtration rate and albuminuria [8].

Worldwide, diabetes mellitus is the most common cause of CKD, but in some regions other causes, e.g., herbal and environmental toxins such as long-term exposure to heavy metals and pesticides, are more common [9–11](Table 18–1). Tobacco smoke is another risk factor for kidney disease and there is some evidence that its effects on kidney function begin in childhood [12].

The poorest populations are at the highest risk and national screening and intervention programs can prevent CKD, and where management strategies have been implemented the incidence of end-stage kidney disease has been reduced.

18.2.1 Studies in Animals

Few experimental studies have evaluated the renal effects of arsenic exposure in animal models. In dogs, low-dose administration of sodium arsenate (0.73 mg/kg) results in mild degeneration and vacuolation of the ascending thick portion of the nephron; at this dosage level there was no change in glomerular filtration rate, fractional reabsorption of sodium, potassium, and chloride. Higher doses (14.66 mg/kg) result in moderate glomerular sclerosis and severe acute tubular necrosis involving all the nephron segments [13].

In mice, arsenic exposure through drinking water (22.5 mg/L) increases urine *N*-acetyl- β -D-glucosaminidase (NAG) but not urine albumin (UAlb) excretion. In that model, mice given both arsenic in drinking water and cadmium (Cd) in food exhibit increases in urine protein and NAG excretion that are markedly higher compared with those in mice given cadmium or arsenic alone [14].

In Chinese Dragon-Li cats exposed to arsenic trioxide to induce renal injury, resveratrol was used to evaluate its effects on arsenic nephrotoxicity. Resveratrol significantly attenuates the accumulation of arsenic in renal tissues, decreases markers of oxidative stress, and shows less morphologic injury and tubular necrosis [15]. Other compounds such as naringenin [16], a naturally occurring citrus flavanone; silibinin [17], a naturally occurring plant bioflavonoid found in the milk thistle of *Silybum marianum; Pleurotus florida* lectin [18]; green tea extract [19]; *Curcuma aromatica* leaf extract [20]; and taurine [21], have shown promising effects protecting the kidney against arsenic nephrotoxicity (Box 18–1).

Environmental Toxin	Type of Renal Disease		
Lead	Chronic tubulo-interstitial nephropathy		
	Fanconi syndrome		
	Hyperuricemia		
Cadmium	Chronic tubulo-interstitial nephropathy		
	Fanconi syndrome		
Arsenic	Glomerulosclerosis		
	Chronic tubulo-interstitial nephropathy		
Aristolochic acid	Chronic tubulo-interstitial nephropathy		
	Acute kidney injury		
	Tubular dysfunction with unchanged glomerular filtration rate		
Organochlorine pesticides	Decrease GFR		
Smoke (tobacco exposure)	Proteinuria		
	Decrease GFR		
Unknown	Mesoamerican nephropathy		

 Table 18–1
 Environmental Toxins Associated with Chronic Renal Disease

GFR: glomerular filtration rate.

BOX 18–1 NUTRITIONAL PRODUCTS THAT HAVE SHOWN PREVENTION OF ARSENIC NEPHROTOXICITY IN ANIMALS

- Selenium
- Naringenin
- Silibinin
- Pleurotus florida lectin
- Green tea extract
- Curcuma aromatica
- Taurine
- Antioxidants

18.2.2 Epidemiological Studies in Humans

Arsenic was first used in ancient Greece as a medicinal agent but renal toxicity associated with this metal was not described until recently. In 1915, Pearce and Brown [22,23] did histological analysis of kidneys from dogs with acute poisoning with arsenic compounds and they detected hemorrhages in the cortex, hyaline thrombus in the glomeruli, and acute tubular necrosis. In the 1960s it was evident that acute poisoning by arsenic was an important industrial hazard associated with hemolysis and acute renal failure [24,25].

Serial renal biopsies done in humans with acute arsenic poisoning showed acute tubular necrosis followed by rapid regeneration of tubular cells; however, biopsies done 6 months later in the same patients showed evolution to diffuse interstitial fibrosis [26]; subsequent reports suggested that acute arsenic poisoning may originate as renal cortical necrosis and chronic kidney disease [27].

Author	Year	Country	As Concentration (water)	As Urinary Excretion	Outcome
Lewis et al. [28]	1999	USA	14–166 µg/L	NA	Increased mortality from kidney disease
Meliker et al. [29]	2007	USA	10–100 μg/L (mean 11 μg/L)	NA	Increased mortality from kidney disease
Hsueh et al. [30]	2009	Taiwan	NA	Controls 20.7 ± 1.1 µg/g CKD 31.95 ± 2.59 µg/g	High urinary total arsenic associated positively with CKD
Chen et al. [31]	2011	Bangladesh	<7–869 µg/L	1 – >206µg/g	Adverse effects of As exposure on higher risk for proteinuria; these effects are modifiable by changes in As exposure
Robles-Osorio et al. [33]	2012	Mexico	NA	0.56 to 89µg/g	α1MG urinary excretion increased with the U-As concentrations
Zheng et al. [32]	2013	USA	NA	Median 12.7 µg/g	Increasing urine arsenic concentrations were cross- sectionally associated with increased albuminuria

Table 18–2 Summary of Epidemiological Studies on Renal Injury Mediated by Arsenic

CKD: chronic kidney disease α 1MG: α 1-microglobulin.

More recently, the assessment of health risks associated with exposure to low to moderately elevated levels of arsenic in drinking water has become the subject of considerable interest as some studies found a direct association between arsenic concentrations in drinking water and mortality (Table 18–2).

In a cohort-mortality study in Utah, USA, Lewis et al. [28] found an increase in the mortality ratio for renal disease (nephritis and nephrosis); this finding was significant only in men but not in women. Meliker et al. [29] found that arsenic in drinking water at levels in excess of $200-300 \mu g/L$ was associated with higher mortality rates for diabetes mellitus, cerebrovascular diseases, and kidney disease in both males and females.

There are few epidemiological studies linking chronic arsenic exposure and renal disease (Table 18–2). Hsuch et al. [30], in Taiwan, studied 125 people with GFR ≤ 60 mL/min and 229 people with normal renal function and found a weak association between urinary levels of As and decreased renal function ($r^2 = 0.04$, $p \leq .001$). They found that high urinary total arsenic or low plasma lycopene level was associated positively with CKD. The results also suggest that the capacity for arsenic methylation may be associated with CKD in individuals who ingest arsenic at low levels and lycopene at low plasma levels.

Chen et al. [31] in Bangladesh used urinary dipsticks test to detect proteinuria in 11,122 men and women, the results showing a positive association between long-term As exposure and the prevalence of proteinuria. During follow-up, increasing levels of urinary As were

associated with an increased risk of proteinuria, and decreasing levels of urinary As were related to a reduced risk of proteinuria.

Zheng et al. [32] evaluated the association between inorganic arsenic and albuminuria in American Indian adults living in rural areas of the United States and low to moderate exposure to As. This population has a high burden for diabetes (49.7%), obesity and albuminuria (30%), and after statistical adjustments for diabetes, urinary Cd excretion, hypertensive medication, and systolic blood pressure they found a higher prevalence ratio of albuminuria by urine arsenic concentration in subjects with higher As urinary excretion.

Robles-Osorio et al. [33], in central Mexico, in an open non-diabetic population found an increase in the α 1-microglobulin urinary excretion, but not albuminuria associated with higher urinary arsenic excretion.

There is not enough information about the effects of arsenic exposure and renal injury in children, and in the only epidemiological study to date there was found to be no association with renal disease [34].

Some animal data has shown that combined exposure to inorganic As and Cd gives rise to more pronounced renal toxicity than exposure to each of the agents alone [14]. This finding has been corroborated in some epidemiological studies done in humans [35].

In China, Hong et al. [36] studied 245 subjects co-exposed to arsenic and cadmium, 122 in the arsenic-cadmium polluted area, 123 in the non-polluted area. Arsenic exposure biomarkers β_2 -microglobulin (U β_2 MG), albumin (UAlb), and *N*-acetyl- β_2 -glucosaminidase (UNAG) in urine were used; the levels of U β_2 MG, UAlb, and UNAG in the polluted area were significantly higher than those in the non-polluted area.

Nordberg et al. [37] also studied renal dysfunction in 619 persons residing in two metalcontaminated areas in China; measurements of UB2MG, UNAG, URBP (urinary retinolbinding protein), and UAlb were used as markers of renal dysfunction. An interaction effect between As and Cd was demonstrated at higher levels of a combined exposure to both elements as an increased prevalence of B2MG-uria, NAG-uria, and ALB-uria was found in relation to Cd combined with As.

In conclusion, the evidence linking arsenic concentration in urine or drinking water and CKD is still scarce and limited to some populations; more studies are needed in open populations to determine the effects of low to moderate arsenic levels on renal disease.

Huang et al. [38] evaluated the effect of co-exposure to environmental low-level Cd and As on urinary biomarkers and oxidative stress. Urinary excretion of NAG and the oxidative stress indices urinary malondialdehyde and 8-hydroxy-2-deoxyguanosine were positively correlated with both Cd and As in urine and these effects were more pronounced with co-exposure to Cd and As than for exposure to each metal alone.

18.2.3 Early Biomarkers of Arsenic Exposure and Nephrotoxicity

One of the major challenges in the fields of As toxicology has involved the monitoring of risk populations for early signs of exposure and toxicity even though almost no attention has been focused on the development and use of sensitive biomarkers of As nephrotoxicity.

The United States National Institutes of Health have defined the term "biomarker" as a "characteristic that is objectively measured and evaluated as an indicator of normal biological processes, pathogenic processes, or pharmacologic responses to a therapeutic intervention" [39].

Biomarkers are by definition objective and quantifiable characteristics of biological processes, and could only serve as true replacements for clinical relevant endpoints if we understood the normal physiology of a biological process, the pathophysiology of that process in the disease state, and effects of an intervention on these processes [39].

A useful biomarker of arsenic nephrotoxicity has to be associated with a statistically significant increase in the relative risk (odds ratio or hazard ratio) of developing CKD; also, levels of the biomarker must be significantly different in individuals who will go on to develop CKD than in those who do not [40].

Unfortunately, arsenic has not been established as an important risk factor for CKD, and there are no studies that report progression to CKD according to As urinary excretion, so it is a poor predictor of the level of risk, as it is neither sensitive nor specific for development of CKD.

The measurement of GFR, levels of proteinuria and albuminuria, NAG, β_2 -microglobulin, α_1 -microglobulin, and retinol-binding protein has been reported in the studies on arsenic nephrotoxicity.

Microalbuminuria may be defined as the urinary excretion of 30–300 mg/d of albumin in a timed urine collection in adults, and macroalbuminuria as a urinary excretion higher than 300 mg/d. Albuminuria has long been recognized as a marker for renal disease and is a well-known predictor of poor renal and cardiovascular outcomes in patients with type 2 diabetes and essential hypertension [41]; albuminuria indicates glomerular and endothelial injury [42], and some studies have used this marker as clinical surrogate of kidney disease related to arsenic exposure [32]; how-ever, there is some controversy about the role of microalbuminuria in CKD progression [43].

Urinary markers characteristic of tubular damage essentially consist of urinary enzymes or LMW plasma proteins that are normally freely filtered through the glomerulus. The impaired reabsorption of these proteins through the proximal tubule results in an increased excretion in urine and tubular or low-molecular-weight proteinuria reflecting tubular injury and tubular dysfunction. Proteins with a molecular weight lower than 40 kDa, such as β_2 -microglobulin, α_1 -microglobulin, and retinol-binding protein, are freely filtered through the glomerular membrane, and when tubular dysfunction is present their reabsorption in this segment is reduced [44–46].

NAG is a high-molecular-weight lysosomal enzyme found in many tissues of the body, and is not filtered through the glomerular membrane due to its high molecular weight. However, this enzyme shows high activity in renal proximal tubular cells, and leaks into the tubular fluid when proximal tubular cells are injured as in arsenic exposure. Its urine level increases and thus is used as a reflection of proximal tubular cell necrosis; however, whether this protein has a role as a useful marker of risk for progression of renal disease is still unclear [47].

18.2.4 Physiopathology

Arsenic is absorbed by the intestine and lungs (inhalation), and, to a lesser extent, through the skin. Inorganic arsenic ingested in drinking water is rapidly and almost totally (80–90%)



AQ: Aquaporin, MRP: Multidrug Resistant Protein, GSH: Glutathione FIGURE 18–1 Mechanisms of arsenic-related nephrotoxicity.

absorbed by humans and animals [48]. Once it has been absorbed, it is transported to all tissues in the body. The intake of selenium and vitamin B decreases intestinal absorption of As in the diet. In addition, genetic variables influence arsenic methylation, so arsenic toxicity may be greater among those with poorer diets [49]. Arsenic is methylated in the liver in a glutathione (GSH)-mediated process, which decreases its toxicity and facilitates its biliary and urinary excretion [50], and then enters the intracellular medium through the aqua-glyceroporins AQ3 and AQ9. Studies in cell cultures have shown that the increase in AQ3 and AQ9 cellular expression increases intracellular accumulation of As. In the liver, AQ9 is important for biliary excretion of As [51]. Arsenic induces synthesis of metallothioneins, small, cysteine-rich, metalbinding proteins expressed in all eukaryotes and some prokaryotes; these proteins are highly inducible by heavy metals and represent a major means of metal detoxification in the body to maintain trace elements within a physiological range or to protect the body from damage by metal overload [52].

Another group of As transport proteins includes MRP-1 and -2 (ATP binding cassette-multidrug resistance protein), which were first described in the liver, where they transport As bound to GSH to the bile. The MRP-2 transporter is also located in proximal tubule cells, which favors entry of As into these cells [53]. Arsenic toxicity in PCT cells is due to GSH depletion and an increase in oxidative activity by free radicals. As also increases the expression of interleukin-6 and interleukin-8 expression, and it seems that it activates other pathways to increase the synthesis of reactive oxygen species (ROS) [54] (Figure 18–1). Other mechanisms related to tubular toxicity are apoptosis through activation of the p53 apoptotic pathway [55] and activation of the caspase-9 and -3 signaling pathway [56].

Arsenic uncouples oxidative phosphorylation, and this graded uncoupling of oxidative metabolism causes graded reductions in the net transport of sodium, phosphate, and glucose transport; this type of tubular injury is manifested clinically as Fanconi syndrome (phosphaturia, glucosuria, and low-molecular-weight proteinuria) [57].

Widespread vascular endothelial dysfunction or chronic low-grade inflammation also may be underlying mechanisms for albuminuria. Arsenic has been associated with vascular cell adhesion molecule 1 (VCAM-1), and up-regulation in the expression of the angiotensin type I receptor [58].

18.2.5 Clinical Manifestations of Arsenic-induced Renal Disease

The literature does not currently offer sufficient information on clinical manifestations of As nephrotoxicity, but it is likely to manifest as tubular damage, such as low-molecular-weight proteinuria, aminoaciduria, glycosuria, and phosphaturia, as well as progressive deterioration of renal function.

Acute arsenic poisoning is manifested as acute tubular necrosis with oliguria and azotemia, and some patients even require advanced support with dialysis and plasma exchange. Acute tubulointerstitial nephritis has been described as a clinical manifestation of acute As poisoning [59].

18.3 Arsenic and Bladder Disease

Inorganic arsenic is a known human carcinogen, and chronic exposure leads to an increase in the incidence of skin, lung, liver, kidney, bladder, and prostate tumors.

Bladder cancer is the most common malignancy involving the urinary system and is the 10th most common cancer worldwide; the incidence increases with age, arising most commonly in the seventh decade of life and rarely occurring before the age of 50. Histologically, most cases of bladder cancer are transitional cell carcinomas (90%); 70% of these are superficial and papillary subtypes [60]. Hematuria, frequent urination, and pain during urination are the most common symptoms of bladder cancer.

Cigarette smoking is the primary risk factor for bladder cancer. Current cigarette smokers have a three-fold higher bladder cancer risk than non-smokers, and ex-smokers experience a two-fold increased risk; about half of male bladder cancer and one-fifth of female bladder cancer cases were attributable to cigarette smoking [61,62].

Epidemiologic studies based on high arsenic exposure in Taiwan [63], Argentina [64], Japan, [65] and Chile [66] suggest an increased risk of cancer of the urinary tract, and the evidence for a link between bladder cancer and exposure to arsenic in drinking water at concentrations exceeding $300-500 \,\mu$ g/L is strong; however, the risk with As concentrations below $200 \,\mu$ g/L, except for the smokers, is controversial.

Few epidemiological studies have evaluated the association between low arsenic exposure levels and bladder cancer, and more studies are needed to confirm this association. In the United States [67], a case–control study in which arsenic concentrations in drinking water ranged from 0.5 to 180 pg/L showed no overall association between arsenic exposure and cancer risk, but arsenic and cigarette smoking tended to have a synergistic effect on the risk of bladder cancer. Kurttio et al.[68] reported in a Finnish study a weak association between arsenic and bladder cancer risk in an open population exposed to very low As levels ($0.5 \mu g/L$), this study again providing evidence for synergistic effects between smoking and As. In a more recent study done in Chile, subjects were exposed to arsenic water concentrations < 200 µg/L (median = 60 µg/L); bladder cancer odds ratio for subjects in the upper tertile of monomethylarsonic (MMA) excretion compared to subjects in the lower two tertiles was 2.37 (1.01–5.57) [69].

The major metabolic pathway of inorganic arsenic in humans is its methylation in the liver. In most organisms, As is metabolized by an alternating series of reductions of pentavalent forms to trivalent forms followed by sequential oxidative methylation, and the most abundant metabolite excreted in urine of humans and rodents is dimethylarsinic acid (DMAV). The trivalent arsenicals have been shown *in vitro* to be highly reactive and considerably more cytotoxic compared with pentavalent forms, especially for the methylated arsenicals. These trivalent forms are believed to play a critical role in As-induced toxicity and carcinogenicity [70].

Arsenic methylation is an important mechanism that induces oxidative DNA damage and the acquisition of an *in vitro* cancer phenotype, as measured by invasiveness, colony formation, and secretion of matrix metalloproteinases in rat and human epithelial cell lines [71,72]. There is mounting evidence that metabolism of arsenic to methylated species plays a predominant role in its toxic and carcinogenic effects. Metabolism of arsenic is variable between species, human populations, and individuals. Polymorphisms in *AS3MT* and other genes may influence arsenic metabolism and potential susceptibility to its toxic and carcinogenic effects [73,74].

The mechanism of arsenic-induced bladder cancer remains unclear, but some proposed mechanisms are mediated through reactions with thiols in cells [75], generation of ROS [76], oxidative DNA damage, acquired tolerance to apoptosis, enhanced cell proliferation, altered DNA methylation, genomic instability, and aberrant estrogen signalling [77].

Accumulation of trivalent arsenite in bladder tissue mice exposed to 0.01% sodium arsenite in drinking water demonstrated proliferation of the bladder uroepithelium within 4 weeks after initiating treatment [78].

Arsenic is a tumor promoter that affects specific cell signal transduction pathways responsible for cell proliferation. The activation of the epidermal growth factor receptor (EGFR) extracellular signal-regulated protein kinase (ERK) pathway is important in mediating gene expression related to regulation of cellular growth. Arsenic-induced cell proliferation in the bladder epithelium is induced by activation of the MAP kinase pathway, leading to an increase in the activity of EGFR, ERK, and an increase in c-Src levels interacting with EGFR [79,80].

Arsenic induces AP-1 activation and expression of AP-1-associated genes causing activation of specific signaling pathways that lead to chronic, increased cell proliferation. This may play a non-epigenetic role in carcinogenesis by increasing the proliferation of initiated cells or increasing the mutational rate [81]. Arsenic also exerts its toxicity through mitochondrial dysfunction and generation of ROS; this mechanism plays a crucial role in the control of apoptosis, and arsenic-induced oxidative stress promotes telomere attrition, chromosome end-to-end fusions, and apoptotic cell death [82].

Angiogenesis is a key event for tumor initiation. Arsenic stimulates this process by a mechanism related to reorganization of actin filaments, activation of Cdc42, increasing the activity of the enzyme NADPH oxidase with generation of free radicals, and stimulation of cell migration [83]. Arsenic increases the expression of other enzymes and proteins such as COX-2, VEGF, and HIF-1 α , which regulates the angiogenesis process; this is mediated by ROS through the MAPK and PI3K/AKT signaling pathways [84].

The inflammatory response as a pathogenic factor has gained attention since there is evidence showing that *in vitro* the exposure to arsenicals induces the production or activation of inflammatory mediators, and there is some evidence that suggests a crucial role of IL-8 in bladder cancer development [85].

References

- Levey AS, de Jong PE, Coresh J, El Nahas M, Astor BC, Matsushita K, et al. The definition, classification, and prognosis of chronic kidney disease: a KDIGO Controversies Conference report. Kidney Int 2011;80:17–28.
- [2] Jha V, Garcia-Garcia G, Iseki K, Li Z, Naicker S, Plattner B, et al. Chronic kidney disease: global dimension and perspectives. Lancet 2013;382:260–72.
- Quarles LD. Reducing cardiovascular mortality in chronic kidney disease: something borrowed, something new. J Clin Invest 2013;123:542–3.
- [4] Lebiedz P, Knickel L, Engelbertz C, Luders F, Gebauer K, Berdel WE, et al. Impact of preexisting chronic kidney disease on acute and long-term outcome of critically ill patients on a medical intensive care unit. J Nephrol 2014;27:73–80.
- [5] Miwa K, Tanaka M, Okazaki S, Furukado S, Yagita Y, Sakaguchi M, et al. Chronic kidney disease is associated with dementia independent of cerebral small-vessel disease. Neurology 2014;82:1051–7.
- [6] Drueke TB, Parfrey PS. Summary of the KDIGO guideline on anemia and comment: reading between the (guide)line(s). Kidney Int 2012;82:952–60.
- [7] Kidney Disease: Improving Global Outcomes (KDIGO) CKD-MBD Work Group. KDIGO clinical practice guideline for the diagnosis, evaluation, prevention, and treatment of Chronic Kidney Disease-Mineral and Bone Disorder (CKD-MBD). Kidney Int Suppl 2009:S1–130.
- [8] Stevens LA, Claybon MA, Schmid CH, Chen J, Horio M, Imai E, et al. Evaluation of the Chronic Kidney Disease Epidemiology Collaboration equation for estimating the glomerular filtration rate in multiple ethnicities. Kidney Int 2011;79:555–62.
- [9] Dyck RF, Hayward MN, Harris SB, Group CS. Prevalence, determinants and co-morbidities of chronic kidney disease among First Nations adults with diabetes: results from the CIRCLE study. BMC Nephrol 2012;13:57.
- [10] Wanigasuriya KP, Peiris-John RJ, Wickremasinghe R. Chronic kidney disease of unknown aetiology in Sri Lanka: is cadmium a likely cause? BMC Nephrol 2011;12:32.
- [11] Siddharth M, Datta SK, Bansal S, Mustafa M, Banerjee BD, Kalra OP, et al. Study on organochlorine pesticide levels in chronic kidney disease patients: association with estimated glomerular filtration rate and oxidative stress. J Biochem Mol Toxicol 2012;26:241–7.

- [12] Garcia-Esquinas E, Loeffler LF, Weaver VM, Fadrowski JJ, Navas-Acien A. Kidney function and tobacco smoke exposure in US adolescents. Pediatrics 2013;131:e1415–23.
- [13] Tsukamoto H, Parker HR, Gribble DH, Mariassy A, Peoples SA. Nephrotoxicity of sodium arsenate in dogs. Am J Vet Res 1983;44:2324–30.
- [14] Liu J, Liu Y, Habeebu SM, Waalkes MP, Klaassen CD. Chronic combined exposure to cadmium and arsenic exacerbates nephrotoxicity, particularly in metallothionein-I/II null mice. Toxicology 2000;147:157–66.
- [15] Yu M, Xue J, Li Y, Zhang W, Ma D, Liu L, et al. Resveratrol protects against arsenic trioxide-induced nephrotoxicity by facilitating arsenic metabolism and decreasing oxidative stress. Arch Toxicol 2013;87:1025–35.
- [16] Mershiba SD, Dassprakash MV, Saraswathy SD. Protective effect of naringenin on hepatic and renal dysfunction and oxidative stress in arsenic intoxicated rats. Mol Biol Rep 2013;40:3681–91.
- [17] Prabu SM, Muthumani M. Silibinin ameliorates arsenic induced nephrotoxicity by abrogation of oxidative stress, inflammation and apoptosis in rats. Mol Biol Rep 2012;39:11201–16.
- [18] Bera AK, Rana T, Das S, Bhattacharya D, Pan D, Bandyopadhyay S, et al. Mitigation of arsenic-mediated renal oxidative stress in rat by *Pleurotus florida* lectin. Hum Exp Toxicol 2011;30:940–51.
- [19] Messarah M, Saoudi M, Boumendjel A, Kadeche L, Boulakoud MS, El Feki A. Green tea extract alleviates arsenic-induced biochemical toxicity and lipid peroxidation in rats. Toxicol Ind Health 2013;29:349–59.
- [20] Saxena PN, Anand S, Saxena N, Bajaj P. Effect of arsenic trioxide on renal functions and its modulation by *Curcuma aromatica* leaf extract in albino rat. J Environ Biol/Acad Environ Biol India 2009;30:527–31.
- [21] Roy A, Manna P, Sil PC. Prophylactic role of taurine on arsenic mediated oxidative renal dysfunction via MAPKs/NF-kappaB and mitochondria dependent pathways. Free Radic Res 2009;43:995–1007.
- [22] Pearce L, Brown WH. Chemopathological studies with compounds of arsenic: I. Types of the arsenic kidney. J Exp Med 1915;22:517–24.
- [23] Pearce L, Brown WH. Chemopathological studies with compounds of arsenic: II. Histological changes in arsenic kidneys. J Exp Med 1915;22:525–34.
- [24] Teitelbaum DT, Kier LC. Arsine poisoning. Report of five cases in the petroleum industry and a discussion of the indications for exchange transfusion and hemodialysis. Arch Environ Health 1969;19:133–43.
- [25] Uldall PR, Khan HA, Ennis JE, McCallum RI, Grimson TA. Renal damage from industrial arisine poisoning. Br J Ind Med 1970;27:372–7.
- [26] Muehrcke RC, Pirani CL. Arsine-induced anuria. A correlative clinicopathological study with electron microscopic observations. Ann Intern Med 1968;68:853–66.
- [27] Gerhardt RE, Hudson JB, Rao RN, Sobel RE. Chronic renal insufficiency from cortical necrosis induced by arsenic poisoning. Arch Intern Med 1978;138:1267–9.
- [28] Lewis DR, Southwick JW, Ouellet-Hellstrom R, Rench J, Calderon RL. Drinking water arsenic in Utah: A cohort mortality study. Environ Health Perspect 1999;107:359–65.
- [29] Meliker JR, Wahl RL, Cameron LL, Nriagu JO. Arsenic in drinking water and cerebrovascular disease, diabetes mellitus, and kidney disease in Michigan: a standardized mortality ratio analysis. Environ Health 2007;6:4.
- [30] Hsueh YM, Chung CJ, Shiue HS, Chen JB, Chiang SS, Yang MH, et al. Urinary arsenic species and CKD in a Taiwanese population: a case-control study. Am J Kidney Dis 2009;54:859–70.
- [31] Chen Y, Parvez F, Liu M, Pesola GR, Gamble MV, Slavkovich V, et al. Association between arsenic exposure from drinking water and proteinuria: results from the Health Effects of Arsenic Longitudinal Study. Int J Epidemiol 2011;40:828–35.
- [32] Zheng LY, Umans JG, Tellez-Plaza M, Yeh F, Francesconi KA, Goessler W, et al. Urine arsenic and prevalent albuminuria: evidence from a population-based study. Am J Kidney Dis 2013;61:385–94.
- [33] Robles-Osorio ML, Perez-Maldonado IN, Martin del Campo D, Montero-Perea D, Aviles-Romo I, Sabath-Silva E, et al. Urinary arsenic levels and risk of renal injury in a cross-sectional study in open population. Rev Invest Clin 2012;64:609–14.
- [34] de Burbure C, Buchet JP, Leroyer A, Nisse C, Haguenoer JM, Mutti A, et al. Renal and neurologic effects of cadmium, lead, mercury, and arsenic in children: evidence of early effects and multiple interactions at environmental exposure levels. Environ Health Perspect 2006;114:584–90.
- [35] Buchet JP, Heilier JF, Bernard A, Lison D, Jin T, Wu X, et al. Urinary protein excretion in humans exposed to arsenic and cadmium. Int Arch Occup Environ Health 2003;76:111–20.
- [36] Hong F, Jin T, Zhang A. Risk assessment on renal dysfunction caused by co-exposure to arsenic and cadmium using benchmark dose calculation in a Chinese population. Biometals 2004;17:573–80.
- [37] Nordberg GF, Jin T, Hong F, Zhang A, Buchet JP, Bernard A. Biomarkers of cadmium and arsenic interactions. Toxicol Appl Pharmacol 2005;206:191–7.
- [38] Huang M, Choi SJ, Kim DW, Kim NY, Park CH, Yu SD, et al. Risk assessment of low-level cadmium and arsenic on the kidney. J Toxicol Environ Health A 2009;72:1493–8.
- [39] Strimbu K, Tavel JA. What are biomarkers? Curr Opin HIV AIDS 2010;5:463-6.
- [40] Shlipak MG, Day EC. Biomarkers for incident CKD: a new framework for interpreting the literature. Nature Reviews. Nephrology 2013;9:478–83.
- [41] Weir MR. Microalbuminuria and cardiovascular disease. Clin J Am Soc Nephrol 2007;2:581–90.
- [42] Salmon AH, Ferguson JK, Burford JL, Gevorgyan H, Nakano D, Harper SJ, et al. Loss of the endothelial glycocalyx links albuminuria and vascular dysfunction. J Am Soc Nephrol 2012;23:1339–50.
- [43] Glassock RJ. Is the presence of microalbuminuria a relevant marker of kidney disease? Curr Hypertens Rep 2010;12:364–8.
- [44] Andersson L, Haraldsson B, Johansson C, Barregard L. Methodological issues on the use of urinary alpha-1-microglobuline in epidemiological studies. Nephrol Dial Transplant 2008;23:1252–6.
- [45] Dieterle F, Perentes E, Cordier A, Roth DR, Verdes P, Grenet O, et al. Urinary clusterin, cystatin C, beta2microglobulin and total protein as markers to detect drug-induced kidney injury. Nat Biotechnol 2010;28:463–9.
- [46] Pallet N, Chauvet S, Chasse JF, Vincent M, Avillach P, Levi C, et al. Urinary retinol binding protein is a marker of the extent of interstitial kidney fibrosis. PLOS ONE 2014;9:e84708.
- [47] Waring WS, Moonie A. Earlier recognition of nephrotoxicity using novel biomarkers of acute kidney injury. Clin Toxicol 2011;49:720–8.
- [48] Mitra SR, Mazumder DN, Basu A, Block G, Haque R, Samanta S, et al. Nutritional factors and susceptibility to arsenic-caused skin lesions in West Bengal, India. Environ Health Perspect 2004;112:1104–9.
- [49] Heck JE, Nieves JW, Chen Y, Parvez F, Brandt-Rauf PW, Graziano JH, et al. Dietary intake of methionine, cysteine, and protein and urinary arsenic excretion in Bangladesh. Environ Health Perspect 2009;117:99–104.
- [50] Thomas DJ. Unraveling arsenic—glutathione connections. Toxicol Sci 2009;107:309–11.
- [51] Carbrey JM, Song L, Zhou Y, Yoshinaga M, Rojek A, Wang Y, et al. Reduced arsenic clearance and increased toxicity in aquaglyceroporin-9-null mice. Proc Natl Acad Sci USA 2009;106:15956–60.
- [52] He X, Ma Q. Induction of metallothionein I by arsenic via metal-activated transcription factor 1: critical role of C-terminal cysteine residues in arsenic sensing. J Biol Chem 2009;284:12609–21.
- [53] Lee TC, Ho IC, Lu WJ, Huang JD. Enhanced expression of multidrug resistance-associated protein 2 and reduced expression of aquaglyceroporin 3 in an arsenic-resistant human cell line. J Biol Chem 2006;281:18401–7.
- [54] Jimi S, Uchiyama M, Takaki A, Suzumiya J, Hara S. Mechanisms of cell death induced by cadmium and arsenic. Ann N Y Acad Sci 2004;1011:325–31.
- [55] Tokumoto M, Lee JY, Fujiwara Y, Uchiyama M, Satoh M. Inorganic arsenic induces apoptosis through downregulation of Ube2d genes and p53 accumulation in rat proximal tubular cells. J Toxicol Sci 2013;38:815–20.

- [56] Lu Y, Yuan H, Deng S, Wei Q, Guo C, Yi J, et al. Arsanilic acid causes apoptosis and oxidative stress in rat kidney epithelial cells (NRK-52e cells) by the activation of the caspase-9 and -3 signaling pathway. Drug Chem Toxicol 2014;37:55-62.
- [57] Brazy PC, Balaban RS, Gullans SR, Mandel LJ, Dennis VW. Inhibition of Renal Metabolism. Relative effects of arsenate on sodium, phosphate, and glucose transport by the rabbit proximal tubule. J Clin Invest 1980;66:1211–21.
- [58] Hossain E, Ota A, Takahashi M, Karnan S, Damdindorj L, Konishi Y, et al. Arsenic upregulates the expression of angiotensin II Type I receptor in mouse aortic endothelial cells. Toxicol Lett 2013;220:70–5.
- [59] Prasad GV, Rossi NF. Arsenic intoxication associated with tubulointerstitial nephritis. Am J Kidney Dis 1995;26:373-6.
- [60] Cheung G, Sahai A, Billia M, Dasgupta P, Khan MS. Recent advances in the diagnosis and treatment of bladder cancer. BMC Med 2013;11:13.
- [61] Zeegers MP, Goldbohm RA, van den Brandt PA. A prospective study on active and environmental tobacco smoking and bladder cancer risk (The Netherlands). Cancer Causes Control 2002;13:83–90.
- [62] Hemelt M, Yamamoto H, Cheng KK, Zeegers MP. The effect of smoking on the male excess of bladder cancer: a meta-analysis and geographical analyses. Int J Cancer 2009;124:412–19.
- [63] Lamm SH, Byrd DM, Kruse MB, Feinleib M, Lai SH. Bladder cancer and arsenic exposure: differences in the two populations enrolled in a study in southwest Taiwan. Biomed Environ Sci 2003;6:355–68.
- [64] Hopenhayn-Rich C, Biggs ML, Fuchs A, Bergoglio R, Tello EE, Nicolli H, et al. Bladder cancer mortality associated with arsenic in drinking water in Argentina. Epidemiology 1996;7:117–24.
- [65] Sawada N, Iwasaki M, Inoue M, Takachi R, Sasazuki S, Yamaji T, et al. Dietary arsenic intake and subsequent risk of cancer: the Japan Public Health Center-based (JPHC) Prospective Study. Cancer Causes Control 2013;24:1403–15.
- [66] Smith AH, Goycolea M, Haque R, Biggs ML. Marked increase in bladder and lung cancer mortality in a region of Northern Chile due to arsenic in drinking water. Am J Epidemiol 1998;147:660–9.
- [67] Steinmaus C, Yuan Y, Bates MN, Smith AH. Case-control study of bladder cancer and drinking water arsenic in the western United States. Am J Epidemiol 2003;158:1193–201.
- [68] Kurttio P, Pukkala E, Kahelin H, Auvinen A, Pekkanen J. Arsenic concentrations in well water and risk of bladder and kidney cancer in Finland. Environ Health Perspect 1999;107:705–10.
- [69] Melak D, Ferreccio C, Kalman D, Parra R, Acevedo J, Perez L, et al. Arsenic methylation and lung and bladder cancer in a case-control study in northern Chile. Toxicol Appl Pharmacol 2014;274:225–31.
- [70] Dodmane PR, Arnold LL, Muirhead DE, Suzuki S, Yokohira M, Pennington KL, et al. Characterization of intracellular inclusions in the urothelium of mice exposed to inorganic arsenic. Toxicol Sci 2014;137:36–46.
- [71] Hughes MF. Arsenic methylation, oxidative stress and cancer—is there a link? J Natl Cancer Inst 2009;101:1660–1.
- [72] Kojima C, Ramirez DC, Tokar EJ, Himeno S, Drobna Z, Styblo M, et al. Requirement of arsenic biomethylation for oxidative DNA damage. J Natl Cancer Inst 2009;101:1670–81.
- [73] Schlawicke Engstrom K, Broberg K, Concha G, Nermell B, Warholm M, Vahter M. Genetic polymorphisms influencing arsenic metabolism: evidence from Argentina. Environ Health Perspect 2007;115:599–605.
- [74] Beebe-Dimmer JL, Iyer PT, Nriagu JO, Keele GR, Mehta S, Meliker JR, et al. Genetic variation in glutathione S-transferase omega-1, arsenic methyltransferase and methylene-tetrahydrofolate reductase, arsenic exposure and bladder cancer: a case-control study. Environ Health 2012;11:43.
- [75] Akhand AA, Du J, Liu W, Hossain K, Miyata T, Nagase F, et al. Redox-linked cell surface-oriented signaling for T-cell death. Antioxid Redox Signal 2002;4:445–54.
- [76] Flora SJ, Mittal M, Mehta A. Heavy metal induced oxidative stress & its possible reversal by chelation therapy. Indian J Med Res 2008;128:501–23.

442 HANDBOOK OF ARSENIC TOXICOLOGY

- [77] Rossman TG. Mechanism of arsenic carcinogenesis: an integrated approach. Mutat Res 2003;533:37-65.
- [78] Luster MI, Simeonova PP. Arsenic and urinary bladder cell proliferation. Toxicol Appl Pharmacol 2004;198:419–23.
- [79] Simeonova PP, Wang S, Hulderman T, Luster MI. c-Src-dependent activation of the epidermal growth factor receptor and mitogen-activated protein kinase pathway by arsenic. Role in carcinogenesis. J Biol Chem 2002;277:2945–50.
- [80] Bailey KA, Wallace K, Smeester L, Thai SF, Wolf DC, Edwards SW, et al. Transcriptional modulation of the ERK1/2 MAPK and NF-kappaB pathways in human urothelial cells after trivalent arsenical exposure: implications for urinary bladder cancer. J Cancer Res Updates 2012;1:57–68.
- [81] Simeonova PP, Wang S, Kashon ML, Kommineni C, Crecelius E, Luster MI. Quantitative relationship between arsenic exposure and AP-1 activity in mouse urinary bladder epithelium. Toxicol Sci 2001;60:279–84.
- [82] Liu L, Trimarchi JR, Navarro P, Blasco MA, Keefe DL. Oxidative stress contributes to arsenic-induced telomere attrition, chromosome instability, and apoptosis. J Biol Chem 2003;278:31998–2004.
- [83] Qian Y, Liu KJ, Chen Y, Flynn DC, Castranova V, Shi X. Cdc42 regulates arsenic-induced NADPH oxidase activation and cell migration through actin filament reorganization. J Biol Chem 2005;280:3875–84.
- [84] Wang F, Liu S, Xi S, Yan L, Wang H, Song Y, et al. Arsenic induces the expressions of angiogenesisrelated factors through PI3K and MAPK pathways in SV-HUC-1 human uroepithelial cells. Toxicol Lett 2013;222:303–11.
- [85] Escudero-Lourdes C, Wu T, Camarillo JM, Gandolfi AJ. Interleukin-8 (IL-8) over-production and autocrine cell activation are key factors in monomethylarsonous acid [MMA(III)]-induced malignant transformation of urothelial cells. Toxicol Appl Pharmacol 2012;258:10–18.

19

Developmental Arsenic Exposure: Behavioral Dysfunctions and Neurochemical Perturbations

Riyaz Basha¹ and G. Rajarami Reddy²

¹DEPARTMENT OF PEDIATRICS AND INSTITUTE FOR CANCER RESEARCH, UNIVERSITY OF NORTH TEXAS HEALTH SCIENCE CENTER, FORT WORTH, TX, USA, ²VIKRAMA SIMHAPURI UNIVERSITY, NELLORE, A.P., INDIA

CHAPTER OUTLINE

19.1	Introduction	443
19.2 ·	Toxicity	444
19.3	ے۔ Pevelopmental Toxicity	444
19.4	Conclusions	449
Refer	ences	451

19.1 Introduction

Arsenic is a naturally occurring metalloid in the environment and present in both organic and inorganic forms. Arsenic compounds typically lack odor or taste and therefore they are difficult to detect and easily consumed by people through various modes. Exposure to arsenic may occur through air, food, and drinking water. Several processes such as geothermal, alkali desorption, reductive dissolution, and sulfide oxidation are responsible for groundwater contamination by arsenic. Mining and industrial processing, and leaching of natural geological sources also contaminates drinking water with arsenic [1]. The World Health Organization (WHO) and the United States Environmental Protection Agency (USEPA) have set the maximum contamination level for arsenic in drinking water as $10 \,\mu$ g/L or ppb [2,3]. Arsenic contamination in ground water is a serious concern in some parts of America, Asia, and Australia. In Asia, arsenic contamination is widely reported in Bangladesh, India, and Taiwan, Bangladesh being worst affected as more than 30% of the population has been reported to be exposed to arsenic-contaminated water [4,5]. Arsenic compounds are also added in the preparation of several products used in our daily life, such as cosmetic products. Arsenic has some medicinal values as well when used in low doses. Its medical use has been documented from ancient times by popular philosophers and physicians such as Hippocrates and Aristotle. Arsenic was used in the preparation of Fowler's solution (1% potassium arsenite), which was used to treat diverse health problems including asthma, chorea, malaria, syphilis, and psoriasis [6,7]. In the early 19th century, Salvarsan, an arsenic-based drug, was highly popular for treating syphilis [8]. Arsenic products have been used in chemotherapeutic agents for treating various cancers such as skin cancer and acute promyelocytic leukemia [8–10].

Arsenic has also been used in the preparation of pigments used for example in candles, fabrics, toys, and wallpapers. Even though some of these uses have been discontinued due to toxic effects, arsenic is still used in the preparation of glass, pesticides, semiconductors, and wood preservatives. Industrial uses of arsenic potentially serve as a source for contaminating the soil and ecosystem [11,12].

19.2 Toxicity

Arsenic forms multiple compounds, which are highly poisonous. Arsenic has been referred as the "king of poisons" or "poison of kings" especially due to its lethal properties during the Middle Ages and the Renaissance [13]. According to a Health Agency report, exposure to inorganic arsenic has low-mild to severe route-dependent effects on various organs or biological systems [14]. Inorganic arsenic is considered to be a potential carcinogen and its effects are believed to be modulated via genotoxic mechanisms [14,15].

Arsenic affects many organs and impacts a number of functional systems in the body. Acute and chronic intoxication of arsenic is well known and extensively tested in laboratory models [16–21]. Acute arsenic exposure often causes gastrointestinal irritation, diarrhea, dehydration, and nausea. Higher doses may also result in cardiovascular, neurological, and respiratory complications and even lead to hypotension and heart failure [13,22]. Neurological effects can be undetected for some time but often exhibit a severe response leading to neuropathy [13,23]. Chronic exposure to low levels of arsenic causes mild to severe health hazards like diarrhea, skin keratosis, and cancer in skin, bladder, liver, lung, and kidneys.

19.3 Developmental Toxicity

Humans, especially children, might be exposed to arsenic via various sources including soil, water, and copper chromated arsenate (CCA)-treated woods. The Agency for Toxic Substances and Disease Registry (Atlanta, GA) have cited several incidences of arsenic exposure in children in the United States. Exposure in children is more likely to be through drinking water, hand-to-mouth activity, and playing in soil. There are incidents where children were exposed to arsenic due to CCA-coated woods in playgrounds, recreational constructions, and patio decks, and even by burning of CCA-coated plywood in poorly or non-ventilated areas. Toxicological

responses to arsenic suggest that infants and children are more vulnerable to arsenic. Considering the severity of the problem, the United Nations Children's Fund (UNICEF) and World Bank are participating in joint projects to construct wells in the affected areas, thus helping in reduction of infant and child mortalities due to chronic exposure to arsenic.

19.3.1 Effects on the Nervous System

19.3.1.1 Neurochemical Effects

It is well established that continuous exposure to arsenic may lead to detrimental effects on the nervous system [24–28]. It is also reported that the developing brain is more susceptible to the toxic effects of arsenic [29–34]. Experimental studies have been conducted to delineate the neurochemical changes in the developing brain and several mechanisms have been proposed to ascribe the behavioral and intellectual deficits induced by arsenic.

Several studies confirm that arsenic penetrates into cerebrospinal fluid, breaching the blood-brain barrier and resulting in neurological disorders, brain injury, and neurodegeneration [35–40]. Inorganic arsenic is known to induce neurotoxicity and the development of the hallmarks of neurodegenerative disorders [23,40–43]. The mechanisms behind arsenic-induced neurotoxic and neurodegenerative effects have been described in several investigations. The popular mechanisms believed to be associated with arsenic-induced neurotoxic and behavioral perturbations include the induction of oxidative stress, disruption of neurotransmitter systems, and neuronal outgrowth.

19.3.1.1.1 NEUROTRANSMITTER SYSTEMS

As arsenic easily crosses the blood-brain barrier it disrupts several biological functions, leading to neurotoxicity [44,45]. The neurotoxic effects of arsenic are primarily associated with modulations in the neurochemical changes occurring in the cholinergic, gamma-amino butyric acid (GABA)-ergic, and monoaminergic systems [26,46]. *In vitro* assays demonstrated that 0.1 μ M and above concentrations of arsenite inhibited the synthesis and release of acetylcholine (ACh) in cerebral slices. Arsenite also decreased the activity of acetylcholinesterase (AChE) but potentiated the choline acetyltransferase (ChAT) activity [47]. Arsenic dose-dependently inhibited the Ach content and AChE activity in brain regions, cerebellum, cerebral cortex, hippocampus, and pons medulla, leading to cholinergic dysfunction. Rats treated with 20 mg/kg body weight of sodium arsenite daily for 28 days showed a significant decrease in AChE activity in hippocampus (46%) and frontal cortex (33%) [48].

Laboratory experiments showed that sodium arsenite modulates the levels of monoamines. Arsenic affects dopamine (DA), norepinephrine (NE), serotonin (5-HT), and monoamino oxidase [49]. Chronic (60 days) exposure to arsenic (1–4 mg/L) via drinking water decreased the levels of DA, NE, and 5-HT and this effect was severe with an increase in concentration of arsenic, showing a dose-dependent effect [50]. In another study, chronic (60 days) exposure to moderate levels of arsenic (4 ppm) in mice significantly reduced the levels of dopamine beta hydroxylase (DBH), tyrosine hydroxylase (TH), and tryptophan hydroxylase (TPH), and of NE, DA, and 5-HT [51]. These studies suggest that arsenic affects cholinergic and dopaminergic neurotransmitters.



FIGURE 19–1 Arsenic causes apoptosis and neuronal death via modulating oxidative damage. Generation of ROS is induced by xenobiotic exposure. ROS are removed by anti-oxidant (AO) defense mechanisms and balanced by the biological processes of oxidation and reduction (redox). Inorganic arsenic (iAs) promotes generation of ROS and increased ROS shift the balance in the redox systems leading to the accumulation of oxidative stress. ROS interact with biomolecules such as lipids, proteins, and DNA, thereby inducing apoptosis and neuronal cell death.

19.3.1.1.2 OXIDATIVE STRESS

Inorganic arsenic is known to induce the generation of reactive oxygen species (ROS) and reactive nitrogen oxide species (RNOS) [40,52–57]. A number of studies has provided evidence for a co-relation between oxidative stress and arsenic-induced neurological disorders [58–62]. The central nervous system, particularly the brain, contains high levels of polyunsaturated lipid, which utilize large volumes of oxygen and rely on excessive aerobic respiration [63,64]. These factors make the brain sensitive to oxidative stress. Arsenic may cause morphological and structural alterations leading to neuronal death [65,66]. An imbalance between prooxidant and anti-oxidant systems results in the accumulation of free radicals, which induces oxidative stress followed by cellular damage and neuronal death (Figure 19–1). Cheng et al. [67] reported that arsenic trioxide induced oxidative damage and disrupted the cerebral cortex in Chinese Dragon Li cats. This group also reported that pretreatment with anti-oxidant resveratrol protected against cerebral injury by up-regulating anti-oxidant enzymes. These findings support the association of arsenic-induced oxidative stress in cerebral injury [67].

Arsenic exists in different oxidation states; however, trivalent (arsenite) and pentavalent (arsenate) states have major roles in environmental exposure [68]. Arsenite reacts with sulfur-containing compounds and it can act as a potent inhibitor of glutathione reductase, which catalyzes the reduction of glutathione disulfide to glutathione (GSH) [69,70]. Since GSH is involved in resisting oxidative stress and plays a critical role in the maintaining redox state of cells, the inhibition of glutathione reductase by arsenic may trigger the accumulation of cellular oxidative stress.

The role of signaling pathways has been proposed in arsenic-induced oxidative stress and apoptosis. ROS potentially disrupts the activation of certain signaling pathways involving mitogen-activated protein kinases (MAPK), phosphatidylinositol-4,5-bisphosphate 3-kinase (PI3K), Akt, mechanistic target of rapamycin (mTOR), and endoplasmic reticulum (ER). These signaling pathways regulate the events related to cell growth and differentiation. Disrupting such critical signaling pathways triggers apoptosis, thereby leading to neuronal cell death [57,71–75].

19.3.1.1.3 NEURITE OUTGROWTH

Neurons establish outgrowth by axons and dendrites, which is essential for connectivity and subsequent neuronal function. The neuronal morphology is critical for the function of the nervous system [76] and it is considered as a sensitive marker for assessing neurotoxicity. Neuronal outgrowth can be affected by manipulating the critical components of the cellular cytoskeleton such as microtubules, microfilaments, and neurofilaments light (NFs). NFs are classified as light (NF-L), medium (NF-M), and high (NF-H) depending on the subunits. Aung et al. studied the effect of arsenic on neurite outgrowth using *in vitro* assays and found that sodium arsenate alters the expression of genes associated with the regulation of cytoskeletal components, NF-L, NF-M, tau, and tubulin; however, it has no effect on NF-H and actin [77]. This group also shown that by acting on target cytoskeletal components, sodium arsenate inhibits axon formation and neurite outgrowth [77,78]. Arsenic-induced neuronal outgrowth was further evaluated by other investigators [79,80], and it was found that arsenite exposure (0.5-5 μ M sodium arsenite) caused characteristic changes in the morphology of PC12 cells. It was also shown that sodium arsenite exposure caused thickening of neurites, reduced neuronal outgrowth and branching, and inhibited network complexity in PC12 cells [80].

19.3.2 Behavioral Effects

Arsenic is believed to induce neurochemical alterations in the developing central nervous system, which result in neurotoxicity and behavioral perturbations. Exposure to arsenic is known to cause impaired visual perception in Thai children aged 6–9 [81]. The concentration of arsenic was measured in 6- to 9-year-old children (total 529 children). These children were given a Motor-Free Visual Perception test and Visual Motor Integration test and their results were analyzed. A study conducted in Mexico showed a strong correlation between arsenic exposure with poor verbal intelligence and long-term memory in primary school children [82]. In another study, intellectual function in relation to arsenic exposure was evaluated in 351 children aged 5–15 years in West Bengal, India, and it was found that arsenic can induce detrimental effects on intellectual performance [2].

Animal studies also suggest that developmental exposure to inorganic arsenic causes a variety of toxicological effects including growth retardation and malformation, leading to death in hamsters, mice, rats, and rabbits [21,83]. Gestational exposure to higher doses of arsenic exhibited irresistible postnatal malfunctions in cardiac, behavioral, and cognitive functions. Gestational exposure to low levels of arsenic may fail to cause apparent toxic responses during development; however, such developmental exposure may cause behavioral perturbations in adulthood [31,84,85].

Low levels of arsenic caused mild perturbations in psychomotor activity and attentiveness; however, verbal learning and memory are severely affected by arsenic poisoning [13]. Exposure to arsenic also caused neuropathy and consequential brain damage leading to behavioral abnormalities. Impairments could occur in critical neurological functions including deficits in learning and memory [61]. Epidemiological studies also support the association of exposure to low levels of arsenic with cognitive abnormalities in children and adolescents in various parts of the world [86]. Arsenic easily crosses the placenta and may affect the embryo [45,49,87,88]. Maternal exposure to arsenic may occur when pregnant women consume arsenic-contaminated food or water. The effect of *in utero* exposure to arsenic was tested in rats. Pregnant albino rats were exposed to inorganic arsenic (0 or 4.5 or 7.5 mg/kg/day) via gavage from gestation day 8 through birth and pups were monitored for alterations in normal development and behavioral parameters such as righting reflex, neuromotor reflex, forelimb grip strength, and open field activity. This study showed that perinatal exposure to arsenic could lead to detrimental effects in the pups. Arsenic resulted in a dose-dependent increase in fetal mortality and disruptions in developmental and cognitive behavioral activity of offspring [89].

Laboratory studies showed that the neurochemical effects caused by exposure to arsenic could lead to several behavioral and cognitive deficits (Figure 19–2). Gestational exposure to arsenic significantly reduces exploratory behavior, open field activity, and locomotor activity in rats. Consistent with other behavioral observations, arsenic-treated animals showed lower grip strength [90]. Experiments were performed to determine the performance of animals in acquisition and learning using the Morris water maze and found that arsenic exposure significantly affecting the performance of the animals. Animals treated with arsenic showed a poor performance, which was improved when treated with anti-oxidant supplements and chelating agents [34,90–93]. A review of literature on rodent models suggests that regardless of dosing (moderate/high), exposure period (acute/subchronic/chronic), and timing (developmental/adult), arsenic can induce behavioral deficits in memory by causing perturbations in the hippocampus [94].

Epidemiological studies have demonstrated that moderate arsenic exposure could alter cognitive functions of learning and memory during childhood. A significant association was established between urinary arsenic levels (> $50 \mu g/L$) and poor scores on language, vocabulary, memory, intelligence, and math skills [82]. Similar results were also found in children exposed to arsenic (147 µg/L in water) throughout development and childhood [2]. An interesting observation from various epidemiological studies is the differential effects of arsenic, with young females suffering a greater impact in terms of growth development compared to males [95,96]. In contrast to arsenic exposure during development, *in utero* As exposure did



FIGURE 19–2 Arsenic-induced behavioral perturbations. Arsenic exposure is associated with behavioral and cognitive deficits. The developing brain is vulnerable to chemical insults. Exposure to arsenic causes several neurochemical alterations including disruptions in neurotransmitters. These neurochemical changes translate into behavioral deficits. Epidemiological studies show effects on visual perception, psychomotor activity, learning, and memory in children. Consistent with these observations, laboratory studies have reported behavioral and cognitive perturbations in animals treated with inorganic arsenic (iAs). Animals treated with arsenic showed a decrease in locomotor activity, exploratory behavior, open field activity, and grip strength. They also showed poor performance in learning and memory assessed by the Morris water maze experiments.

not exhibit consistent cognitive defects [97,98]. However, some studies show that impaired cognition becomes more evident during adulthood. These results suggest that timing, dose, and duration of exposure determine the extent of neurological damage, with cumulative arsenic intake being an important risk factor compared to acute exposure [99,100]. Several recent studies have also shown the association of arsenic exposure and negative impacts on adult cognition and symptoms of Alzheimer's disease [101,102].

19.4 Conclusions

Arsenic is a xenobiotic environmental metalloid with serious health concerns worldwide. WHO lists arsenic among the top 10 chemicals with major global public health concerns. Acute or chronic exposure to arsenic compounds (especially inorganic arsenic) has detrimental effects on all major organs; however, carcinogenic and neurotoxic effects are more prominent and greatly impact public life. Arsenic causes neuropathy leading to several neurological dysfunctions, and affects learning and memory. It is also clear that the developing brain is vulnerable to the toxic effects of arsenic [103]. Animal studies assessing the localized effects of arsenic neurotoxicity have shown differential effects on various parts of the brain. In children, often perturbations in psychomotor activity and attentiveness are mild, while verbal learning and memory seem to be severely affected by arsenic poisoning [13].

The precise mechanisms associated with the neurotic effects of arsenic have been investigated. A number of studies published show the association of neurotoxic effects of arsenic with neurochemical alterations in neurotransmitter (e.g., cholinergic, GABAergic, and monoaminergic) systems, induction of ROS, modulations in signaling (MAPK and PI3/Akt) pathways, and disruption of neuronal outgrowth. Arsenic exposure and its subsequent toxicity can be mediated through multiple mechanisms including alteration in epigenetic profiles, uncoupling of oxidative phosphorylation, increased ROS, inhibition of thiol-containing enzymes, altered signal transduction, and cell proliferation [104,105]. Arsenic exposure contributes to dysfunction of the hippocampus; glutamatergic, glucocorticoid, collinergic, and monoaminergic signaling; Alzheimer's disease pathways; and synaptic plasticity (neurogenesis). Studies using animal models have demonstrated that arsenic exposure leads to alterations in hippocampaldependent tasks, which correlate with the results from epidemiological studies in humans [106,107]. Similarly, arsenic exposure in animals leads to altered synaptic activity and synapserelated gene expression such as reduced NMDA receptor component NR2A [108], and reduced pERK1/2 in the hippocampus [109].

The hypothalamus-pituitary-adrenal axis (HPA) axis regulates neuroendocrine and behavioral responses and its proper functioning is critical for normal physiology and cognition. Corticosterone is a stress hormone that plays a role in mediating the effects of the HPA in response to stress signals. Recent studies indicate that arsenic interferes with the functioning of key components of the HPA axis. Alteration in the HPA axis is observed in neuropsychiatric illness and is connected to depression like symptoms [110,111], which correlates with decreased behavioral ability and increased depressive symptoms in arsenic-exposed animals. Arsenic has multiple effects on the HPA: modification of the levels of catecholamine norepinephrine in the hypothalamus, which regulate corticotrophin-releasing factor [112]; increases in plasma adrenocorticotrophin hormone [113]; and increases in plasma corticosterone levels [114]. Signaling via corticosterone is mediated by the mineralocorticoid and glucocorticoid (GR) receptors. Studies have shown that mice exposed to arsenic have lower levels of GR in hippocampal tissue; this alters the expression of H-Ras and Raf-1, and in turn leads to decreased expression of ERK [115], whose targets play an important role in learning and memory. A recent study also demonstrated the susceptibility of the glucocorticoid receptor to arsenic, which in turn affects MAPK activation and the HPA axis, which are important components for hippocampal-related functions like learning and memory [116]. Arsenic exposure in rodents also affects motor learning, cholinergic signaling, and locomotion possibly through reduction in acetylcholinesterase and choline acetyltransferase activity [48].

Metal chelators are routinely used in the treatment of heavy metal toxicity. Chelating agents such as British Anti Lewisite (BAL), sodium 2,3-dimercaptopropane 1-sulfonate (DMPS), meso-2,3-dimercaptosuccinic acid (DMSA), and MiADMSA have been tested for treating arsenic toxicity [91–93]. Supplementation of essential metals Ca or Zn and chelating agent MiADMSA

has reversed the arsenic-induced neurochemical alterations in various brain regions in rats. Interestingly, combined supplementation of essential metals and chelating agent resulted in higher recovery [33]. The brain is highly susceptible to oxidative damage due to higher utilization of oxygen and the presence of high iron content and unsaturated fatty acids [63,64], hence many studies attributed oxidative stress as a major mechanism in arsenic-induced neurotoxicity [52–55]. Since the association of ROS induction in arsenic poisoning is an accepted fact, apart from chelating agents, several therapeutic strategies have been employed, along with the combination of antioxidant supplements. Laboratory studies were performed using antioxidants including resveratrol and natural products such as *Centella asiatica* aqueous extract and *Moringa oleifera* seed powder, all of which showed some promising results [54,117]. Further studies are under way to identify effective therapeutic agents including natural compounds; also novel strategies are being developed to prevent exposure to this xenobiotic metal.

Neurobehavioral disorders affect approximately 15% of all births. Environmental factors are believed to play major roles in the developmental of such disorders and arsenic is among the five industrial chemicals that are identified as developmental intoxicants in a systematic review conducted by Grandjean and Landrigan in 2006 [118]. These investigators also later postulated that additional candidates may join the list after further screening [103]. The research conducted on assessing toxicity of arsenic using rodent models delineated the effects on multiple systems and specific pathways associated with learning and memory; however, further epidemiological studies are recommended for evaluating the status of the existing standard dose $(10 \,\mu g/L)$ corresponding to the maximum limit for arsenic exposure [94,119]. While research on understanding the toxicity of arsenic has identified specific pathways, further studies should focus more on fulfilling the need for developing therapeutic strategies to overcome the arsenic-induced developmental perturbations in cognitive functions.

References

- Centeno JA, Tseng CH, Van der Voet GB, Finkelman RB. Global impacts of geogenic arsenic: a medical geology research case. Ambio 2007;36:78–81.
- [2] von Ehrenstein OS, Poddar S, Yuan Y, Mazumder DG, Eskenazi B, Basu A, et al. Children's intellectual function in relation to arsenic exposure. Epidemiology 2007;18:44–51.
- [3] World Health Organization (WHO) Exposure to arsenic: a major public health concern. Preventing disease through healthy environments. Geneva: World Health Organization; 2010.
- [4] Mukherjee A, Sengupta MK, Hossain MA, Ahamed S, Das B, Nayak B, et al. Arsenic contamination in groundwater: a global perspective with emphasis on the Asian scenario. J Health Popul Nutr 2006;24:142-63.
- [5] Ahamed S, Kumar Sengupta M, Mukherjee A, Amir Hossain M, Das B, Nayak B, et al. Arsenic groundwater contamination and its health effects in the state of Uttar Pradesh (UP) in upper and middle Ganga plain, India: a severe danger. Sci Total Environ 2006;370:310–22.
- [6] Aronson SM. Arsenic and old myths. R I Med 1994;77:233-4.
- [7] Gallagher RE. Arsenic—new life for an old potion. N Engl J Med 1998;339:1389–91.
- [8] Antman KH. Introduction: the history of arsenic trioxide in cancer therapy. Oncologist 2001;6(Suppl. 2) :1–2.

- [9] Bobe P, Bonardelle D, Benihoud K, Opolon P, Chelbi-Alix MK. Arsenic trioxide: a promising novel therapeutic agent for lymphoproliferative and autoimmune syndromes in MRL/lpr mice. Blood 2006;108:3967–75.
- [10] Zhang TD, Chen GQ, Wang ZG, Wang ZY, Chen SJ, Chen Z. Arsenic trioxide, a therapeutic agent for APL. Oncogene 2001;20:7146–53.
- [11] Nicolis I, Curis E, Deschamps P, Benazeth S. Arsenite medicinal use, metabolism, pharmacokinetics and monitoring in human hair. Biochimie 2009;91:1260–7.
- [12] Ventura-Lima J, de Castro MR, Acosta D, Fattorini D, Regoli F, de Carvalho LM, et al. Effects of arsenic (As) exposure on the antioxidant status of gills of the zebrafish *Danio rerio* (Cyprinidae). Comp Biochem Physiol C Toxicol Pharmacol 2009;149:538–43.
- [13] Vahidnia A, van der Voet GB, de Wolff FA. Arsenic neurotoxicity—a review. Hum Exp Toxicol 2007;26:823-32.
- [14] Pritchard JD. Inorganic arsenic toxicological overview. Chilton: Health Protection Agency; 2007. pp. 1-10.
- [15] Rossman TG. Mechanism of arsenic carcinogenesis: an integrated approach. Mutat Res 2003;533:37–65.
- [16] Jomova K, Jenisova Z, Feszterova M, Baros S, Liska J, Hudecova D, et al. Arsenic: toxicity, oxidative stress and human disease. J Appl Toxicol 2011;31:95–107.
- [17] Guha Mazumder DN. Chronic arsenic toxicity & human health. Indian J Med Res 2008;128:436-47.
- [18] Islam LN, Nabi AH, Rahman MM, Zahid MS. Association of respiratory complications and elevated serum immunoglobulins with drinking water arsenic toxicity in human. J Environ Sci Health A Tox Hazard Subst Environ Eng 2007;42:1807–14.
- [19] Juhasz AL, Naidu R, Zhu YG, Wang LS, Jiang JY, Cao ZH. Toxicity issues associated with geogenic arsenic in the groundwater-soil-plant-human continuum. Bull Environ Contam Toxicol 2003;71:1100–7.
- [20] Khan MM, Sakauchi F, Sonoda T, Washio M, Mori M. Magnitude of arsenic toxicity in tube-well drinking water in Bangladesh and its adverse effects on human health including cancer: evidence from a review of the literature. Asian Pac J Cancer Prev 2003;4:7–14.
- [21] Golub MS, Macintosh MS, Baumrind N. Developmental and reproductive toxicity of inorganic arsenic: animal studies and human concerns. J Toxicol Environ Health B Crit Rev 1998;1:199–241.
- [22] Greenberg SA. Acute demyelinating polyneuropathy with arsenic ingestion. Muscle Nerve 1996;19:1611–13.
- [23] Le Quesne PM, McLeod JG. Peripheral neuropathy following a single exposure to arsenic. Clincal course in four patients with electrophysiological and histological studies. J Neurol Sci 1977;32:437–51.
- [24] Grzycki S, Kobusowna B. Histophysiological effects of arsenic and its derivatives on the central nervous system, and particularly on the third element of the central nervous system. J Neuropathol Exp Neurol 1951;10:325–37.
- [25] Jenkins RB. Inorganic arsenic and the nervous system. Brain 1966;89:479-98.
- [26] Rodriguez VM, Jimenez-Capdeville ME, Giordano M. The effects of arsenic exposure on the nervous system. Toxicol Lett 2003;145:1–18.
- [27] Sinczuk-Walczak H, Szymczak M, Halatek T. Effects of occupational exposure to arsenic on the nervous system: clinical and neurophysiological studies. Int J Occup Med Environ Health 2010;23:347–55.
- [28] Garcia-Chavez E, Jimenez I, Segura B, Del Razo LM. Lipid oxidative damage and distribution of inorganic arsenic and its metabolites in the rat nervous system after arsenite exposure: influence of alpha tocopherol supplementation. Neurotoxicology 2006;27:1024–31.
- [29] Morrissey RE, Mottet NK. Arsenic-induced exencephaly in the mouse and associated lesions occurring during neurulation. Teratology 1983;28:399–411.
- [30] Herrera A, Pineda J, Antonio MT. Toxic effects of perinatal arsenic exposure on the brain of developing rats and the beneficial role of natural antioxidants. Environ Toxicol Pharmacol 2013;36:73–9.

- [31] Nagaraja TN, Desiraju T. Regional alterations in the levels of brain biogenic amines, glutamate, GABA, and GAD activity due to chronic consumption of inorganic arsenic in developing and adult rats. Bull Environ Contam Toxicol 1993;50:100–7.
- [32] Chandravanshi LP, Yadav RS, Shukla RK, Singh A, Sultana S, Pant AB, et al. Reversibility of changes in brain cholinergic receptors and acetylcholinesterase activity in rats following early life arsenic exposure. Int J Dev Neurosci 2014;34C:60–75.
- [33] Kadeyala PK, Sannadi S, Gottipolu RR. Alterations in apoptotic caspases and antioxidant enzymes in arsenic exposed rat brain regions: reversal effect of essential metals and a chelating agent. Environ Toxicol Pharmacol 2013;36:1150–66.
- [34] Sannadi S, Kadeyala PK, Gottipolu RR. Reversal effect of monoisoamyl dimercaptosuccinic acid (MiADMSA) for arsenic and lead induced perturbations in apoptosis and antioxidant enzymes in developing rat brain. Int J Dev Neurosci 2013;31:586–97.
- [35] Tamura S, Nozaki S. [Arsenic metabolism. 5. Blood brain barrier and arsenite, with special reference to the effect of the feed]. Nihon yakurigaku zasshi 1972;68:800–8.
- [36] Kiguchi T, Yoshino Y, Yuan B, Yoshizawa S, Kitahara T, Akahane D, et al. Speciation of arsenic trioxide penetrates into cerebrospinal fluid in patients with acute promyelocytic leukemia. Leuk Res 2010;34:403–5.
- [37] Helwig A, Klemm M, Schuttig R, Rollig C, Wassilew N, Ehninger G, et al. Arsenic-induced APL differentiation in cerebrospinal fluid. Leuk Res 2007;31:703–5.
- [38] Xi S, Jin Y, Lv X, Sun G. Distribution and speciation of arsenic by transplacental and early life exposure to inorganic arsenic in offspring rats. Biol Trace Elem Res 2010;134:84–97.
- [39] Yadav RS, Sankhwar ML, Shukla RK, Chandra R, Pant AB, Islam F, et al. Attenuation of arsenic neurotoxicity by curcumin in rats. Toxicol Appl Pharmacol 2009;240:367–76.
- [40] Lu TH, Tseng TJ, Su CC, Tang FC, Yen CC, Liu YY, et al. Arsenic induces reactive oxygen species-caused neuronal cell apoptosis through JNK/ERK-mediated mitochondria-dependent and GRP 78/CHOPregulated pathways. Toxicol Lett 2014;224:130–40.
- [41] Donofrio PD, Wilbourn AJ, Albers JW, Rogers L, Salanga V, Greenberg HS. Acute arsenic intoxication presenting as Guillain-Barre-like syndrome. Muscle Nerve 1987;10:114–20.
- [42] DeFuria J, Shea TB. Arsenic inhibits neurofilament transport and induces perikaryal accumulation of phosphorylated neurofilaments: roles of JNK and GSK-3beta. Brain Res 2007;1181:74–82.
- [43] Vahidnia A, Romijn F, van der Voet GB, de Wolff FA. Arsenic-induced neurotoxicity in relation to toxicokinetics: effects on sciatic nerve proteins. Chem Biol Interact 2008;176:188–95.
- [44] Rodriguez VM, Carrizales L, Jimenez-Capdeville ME, Dufour L, Giordano M. The effects of sodium arsenite exposure on behavioral parameters in the rat. Brain Res Bull 2001;55:301–8.
- [45] Jin Y, Xi S, Li X, Lu C, Li G, Xu Y, et al. Arsenic speciation transported through the placenta from mother mice to their newborn pups. Environ Res 2006;101:349–55.
- [46] Kannan GM, Tripathi N, Dube SN, Gupta M, Flora SJ. Toxic effects of arsenic (III) on some hematopoietic and central nervous system variables in rats and guinea pigs. J Toxicol Clin Toxicol 2001;39:675–82.
- [47] Kobayashi H, Yuyama A, Ishihara M, Matsusaka N. Effects of arsenic on cholinergic parameters in brain in vitro. Neuropharmacology 1987;26:1707–13.
- [48] Yadav RS, Chandravanshi LP, Shukla RK, Sankhwar ML, Ansari RW, Shukla PK, et al. Neuroprotective efficacy of curcumin in arsenic induced cholinergic dysfunctions in rats. Neurotoxicology 2011;32:760–8.
- [49] Martinez EJ, Kolb BL, Bell A, Savage DD, Allan AM. Moderate perinatal arsenic exposure alters neuroendocrine markers associated with depression and increases depressive-like behaviors in adult mouse offspring. Neurotoxicology 2008;29:647–55.
- [50] Liu X, Piao F, Li Y. Protective effect of taurine on the decreased biogenic amine neurotransmitter levels in the brain of mice exposed to arsenic. Adv Exp Med Biol 2013;776:277-87.

- [51] Zhang J, Liu X, Zhao L, Hu S, Li S, Piao F. Subchronic exposure to arsenic disturbed the biogenic amine neurotransmitter level and the mRNA expression of synthetase in mice brains. Neuroscience 2013;241:52–8.
- [52] Dwivedi N, Flora G, Kushwaha P, Flora SJ. Alpha-lipoic acid protects oxidative stress, changes in cholinergic system and tissue histopathology during co-exposure to arsenic-dichlorvos in rats. Environ Toxicol Pharmacol 2014;37:7-23.
- [53] Dwivedi N, Flora SJ. Concomitant exposure to arsenic and organophosphates on tissue oxidative stress in rats. Food Chem Toxicol 2011;49:1152–9.
- [54] Flora SJ, Gupta R. Beneficial effects of Centella asiatica aqueous extract against arsenic-induced oxidative stress and essential metal status in rats. Phytother Res 2007;21:980–8.
- [55] Flora SJ, Mittal M, Mishra D. Co-exposure to arsenic and fluoride on oxidative stress, glutathione linked enzymes, biogenic amines and DNA damage in mouse brain. J Neurol Sci 2009;285:198–205.
- [56] Wang Y, Xu Y, Wang H, Xue P, Li X, Li B, et al. Arsenic induces mitochondria-dependent apoptosis by reactive oxygen species generation rather than glutathione depletion in Chang human hepatocytes. Arch Toxicol 2009;83:899–908.
- [57] Calvino E, Estan MC, Simon GP, Sancho P, Boyano-Adanez Mdel C, de Blas E, et al. Increased apoptotic efficacy of lonidamine plus arsenic trioxide combination in human leukemia cells. Reactive oxygen species generation and defensive protein kinase (MEK/ERK, Akt/mTOR) modulation. Biochem Pharmacol 2011;82:1619–29.
- [58] Flora SJ, Bhadauria S, Pant SC, Dhaked RK. Arsenic induced blood and brain oxidative stress and its response to some thiol chelators in rats. Life Sci 2005;77:2324–37.
- [59] Flora SJS. Arsenic-induced oxidative stress and its reversibility following combined administration of N-acetylcysteine and meso 2,3-dimercaptosuccinic acid in rats. Clin Exp Pharmacol Physiol 1999;26:865–9.
- [60] Sinha M, Manna P, Sil PC. Protective effect of arjunolic acid against arsenic-induced oxidative stress in mouse brain. J Biochem Mol Toxicol 2008;22:15–26.
- [61] Mishra D, Flora SJ. Differential oxidative stress and DNA damage in rat brain regions and blood following chronic arsenic exposure. Toxicol Ind Health 2008;24:247–56.
- [62] Haider SS, Najar MS. Arsenic induces oxidative stress, sphingolipidosis, depletes proteins and some antioxidants in various regions of rat brain. Kathmandu Univ Med J 2008;6:60–9.
- [63] Dringen R, Gutterer JM, Hirrlinger J. Glutathione metabolism in brain metabolic interaction between astrocytes and neurons in the defense against reactive oxygen species. Eur J Biochem 2000;267:4912–16.
- [64] Bharath S, Hsu M, Kaur D, Rajagopalan S, Andersen JK. Glutathione, iron and Parkinson's disease. Biochem Pharmacol 2002;64:1037–48.
- [65] Coyle JT, Puttfarcken P. Oxidative stress, glutamate, and neurodegenerative disorders. Science 1993;262:689–95.
- [66] Loh KP, Huang SH, De Silva R, Tan BK, Zhu YZ. Oxidative stress: apoptosis in neuronal injury. Curr Alzheimer Res 2006;3:327–37.
- [67] Cheng Y, Xue J, Jiang H, Wang M, Gao L, Ma D, et al. Neuroprotective effect of resveratrol on arsenic trioxide-induced oxidative stress in feline brain. Hum Exp Toxicol 2013;33:737–47.
- [68] Hughes MF. Arsenic toxicity and potential mechanisms of action. Toxicol Lett 2002;133:1-16.
- [69] Lin S, Cullen WR, Thomas DJ. Methylarsenicals and arsinothiols are potent inhibitors of mouse liver thioredoxin reductase. Chem Res Toxicol 1999;12:924–30.
- [70] Styblo M, Serves SV, Cullen WR, Thomas DJ. Comparative inhibition of yeast glutathione reductase by arsenicals and arsenothiols. Chem Res Toxicol 1997;10:27–33.
- [71] Yen YP, Tsai KS, Chen YW, Huang CF, Yang RS, Liu SH. Arsenic induces apoptosis in myoblasts through a reactive oxygen species-induced endoplasmic reticulum stress and mitochondrial dysfunction pathway. Arch Toxicol 2012;86:923–33.

- [72] Yen CC, Ho TJ, Wu CC, Chang CF, Su CC, Chen YW, et al. Inorganic arsenic causes cell apoptosis in mouse cerebrum through an oxidative stress-regulated signaling pathway. Arch Toxicol 2011;85:565–75.
- [73] Flora SJS. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51:257-81.
- [74] Lu TH, Su CC, Chen YW, Yang CY, Wu CC, Hung DZ, et al. Arsenic induces pancreatic beta-cell apoptosis via the oxidative stress-regulated mitochondria-dependent and endoplasmic reticulum stress-triggered signaling pathways. Toxicol Lett 2011;201:15–26.
- [75] Stepnik M, Ferlinska M, Smok-Pieniazek A, Gradecka-Meesters D, Arkusz J, Stanczyk M. Assessment of the involvement of oxidative stress and mitogen-activated protein kinase signaling pathways in the cytotoxic effects of arsenic trioxide and its combination with sulindac or its metabolites: sulindac sulfide and sulindac sulfone on human leukemic cell lines. Med Oncol 2012;29:1161–72.
- [76] Yuste R, Tank DW. Dendritic integration in mammalian neurons, a century after Cajal. Neuron 1996;16:701-16.
- [77] Aung KH, Kurihara R, Nakashima S, Maekawa F, Nohara K, Kobayashi T, et al. Inhibition of neurite outgrowth and alteration of cytoskeletal gene expression by sodium arsenite. Neurotoxicology 2013;34:226–35.
- [78] Maekawa F, Tsuboi T, Oya M, Aung KH, Tsukahara S, Pellerin L, et al. Effects of sodium arsenite on neurite outgrowth and glutamate AMPA receptor expression in mouse cortical neurons. Neurotoxicology 2013;37:197–206.
- [79] Wang X, Meng D, Chang Q, Pan J, Zhang Z, Chen G, et al. Arsenic inhibits neurite outgrowth by inhibiting the LKB1-AMPK signaling pathway. Environ Health Perspect 2010;118:627–34.
- [80] Frankel S, Concannon J, Brusky K, Pietrowicz E, Giorgianni S, Thompson WD, et al. Arsenic exposure disrupts neurite growth and complexity in vitro. Neurotoxicology 2009;30:529–37.
- [81] Siripitakunkit U, Vishudhiphan P, Pradipasen M, Vorapongsathron T. Association between chronic arsenic exposure and children's intelligence in Thailand. In: Chappell WR, Abernathy CO, Calderon RL, editors. Arsenic exposure and health effects. Oxford, UK: Elsevier Science Ltd; 1999. pp. 141–9.
- [82] Rosado JL, Ronquillo D, Kordas K, Rojas O, Alatorre J, Lopez P, et al. Arsenic exposure and cognitive performance in Mexican schoolchildren. Environ Health Perspect 2007;115:1371–5.
- [83] Gandhi DN, Panchal GM, Patel KG. Developmental and neurobehavioural toxicity study of arsenic on rats following gestational exposure. Indian J Exp Biol 2012;50:147–55.
- [84] Hood RD. Effects of sodium arsenite on fetal development. Bull Environ Contam Toxicol 1972;7:216–22.
- [85] Nagaraja TN, Desiraju T. Effects on operant learning and brain acetylcholine esterase activity in rats following chronic inorganic arsenic intake. Hum Exp Toxicol 1994;13:353–6.
- [86] Zarazua S, Perez-Severiano F, Delgado JM, Martinez LM, Ortiz-Perez D, Jimenez-Capdeville ME. Decreased nitric oxide production in the rat brain after chronic arsenic exposure. Neurochem Res 2006;31:1069–77.
- [87] Ahmed S, Mahabbat-e Khoda S, Rekha RS, Gardner RM, Ameer SS, Moore S, et al. Arsenic-associated oxidative stress, inflammation, and immune disruption in human placenta and cord blood. Environ Health Perspect 2011;119:258–64.
- [88] Myers SL, Lobdell DT, Liu Z, Xia Y, Ren H, Li Y, et al. Maternal drinking water arsenic exposure and perinatal outcomes in inner Mongolia, Chinese. J Epidemiol Community Health 2010;64:325–9.
- [89] Gandhi DN, Panchal GM. Effect of low-level arsenic exposure on the development of neurobehavioral toxicity in rats. Res J Environ Toxicol 2011;5:348–58.
- [90] Ram Kumar M, Flora SJ, Reddy GR. Monoisoamyl 2,3-dimercaptosuccinic acid attenuates arsenic induced toxicity: behavioral and neurochemical approach. Environ Toxicol Pharmacol 2013;36:231–42.
- [91] Flora SJ, Bhadauria S, Kannan GM, Singh N. Arsenic induced oxidative stress and the role of antioxidant supplementation during chelation: a review. J Environ Biol 2007;28:333–47.
- [92] Kalia K, Narula GD, Kannan GM, Flora SJ. Effects of combined administration of captopril and DMSA on arsenite induced oxidative stress and blood and tissue arsenic concentration in rats. Comp Biochem Physiol C Toxicol Pharmacol 2007;144:372–9.

- [93] Bhadauria S, Flora SJ. Response of arsenic-induced oxidative stress, DNA damage, and metal imbalance to combined administration of DMSA and monoisoamyl-DMSA during chronic arsenic poisoning in rats. Cell Biol Toxicol 2007;23:91–104.
- [94] Tyler CR, Allan AM. The effects of arsenic exposure on neurological and cognitive dysfunction in human and rodent studies: a review. Curr Environ Health Rep 2014;1:132–47.
- [95] Llop S, Lopez-Espinosa MJ, Rebagliato M, Ballester F. Gender differences in the neurotoxicity of metals in children. Toxicology 2013;311:3–12.
- [96] Gardner RM, Kippler M, Tofail F, Bottai M, Hamadani J, Grander M, et al. Environmental exposure to metals and children's growth to age 5 years: a prospective cohort study. Am J Epidemiol 2013;177:1356–67.
- [97] Hamadani JD, Grantham-McGregor SM, Tofail F, Nermell B, Fangstrom B, Huda SN, et al. Pre- and postnatal arsenic exposure and child development at 18 months of age: a cohort study in rural Bangladesh. Int J Epidemiol 2010;39:1206–16.
- [98] Parajuli RP, Fujiwara T, Umezaki M, Watanabe C. Association of cord blood levels of lead, arsenic, and zinc with neurodevelopmental indicators in newborns: a birth cohort study in Chitwan Valley, Nepal. Environ Res 2013;121:45–51.
- [99] Hamadani JD, Tofail F, Nermell B, Gardner R, Shiraji S, Bottai M, et al. Critical windows of exposure for arsenic-associated impairment of cognitive function in pre-school girls and boys: a population-based cohort study. Int J Epidemiol 2011;40:1593–604.
- [100] Tsai SY, Chou HY, The HW, Chen CM, Chen CJ. The effects of chronic arsenic exposure from drinking water on the neurobehavioral development in adolescence. Neurotoxicology 2003;24:747-53.
- [101] O'Bryant SE, Edwards M, Menon CV, Gong G, Barber R. Long-term low-level arsenic exposure is associated with poorer neuropsychological functioning: a Project Frontier study. Int J Environ Res Public Health 2011;8:861–74.
- [102] Gong G, Hargrave KA, Hobson V, Spallholz J, Boylan M, Lefforge D, et al. Low-level groundwater arsenic exposure impacts cognition: a project Frontier study. Indian J Environ Health 2011;74:16–22.
- [103] Grandjean P, Landrigan PJ. Neurobehavioural effects of developmental toxicity. Lancet Neurol 2014;13:330–8.
- [104] Watanabe T, Hirano S. Metabolism of arsenic and its toxicological relevance. Arch Toxicol 2013;87:969–79.
- [105] Martinez L, Jimenez V, Garcia-Sepulveda C, Ceballos F, Delgado JM, Nino-Moreno P, et al. Impact of early developmental arsenic exposure on promotor CpG-island methylation of genes involved in neuronal plasticity. Neurochem Int 2011;58:574–81.
- [106] Jing J, Zheng G, Liu M, Shen X, Zhao F, Wang J, et al. Changes in the synaptic structure of hippocampal neurons and impairment of spatial memory in a rat model caused by chronic arsenite exposure. Neurotoxicology 2012;33:1230–8.
- [107] Sharma B, Sharma PM. Arsenic toxicity induced endothelial dysfunction and dementia: pharmacological interdiction by histone deacetylase and inducible nitric oxide synthase inhibitors. Toxicol Appl Pharmacol 2013;273:180–8.
- [108] Luo JH, Qiu ZQ, Shu WQ, Zhang YY, Zhang L, Chen JA. Effects of arsenic exposure from drinking water on spatial memory, ultra-structures and NMDAR gene expression of hippocampus in rats. Toxicol Lett 2009;184:121–5.
- [109] Luo JH, Qiu ZQ, Zhang L, Shu WQ. Arsenite exposure altered the expression of NMDA receptor and postsynaptic signaling proteins in rat hippocampus. Toxicol Lett 2012;211:39–44.
- [110] Zhu LJ, Liu MY, Li H, Liu X, Chen C, Han Z, et al. The different roles of glucocorticoids in the hippocampus and hypothalamus in chronic stress-induced HPA axis hyperactivity. PLOS ONE 2014;9:e97689.
- [111] Bremner JD, Vythilingam M, Vermetten E, Anderson G, Newcomer JW, Charney DS. Effects of glucocorticoids on declarative memory function in major depression. Biol Psychiatry 2004;55:811–15.

- [112] Mejia JJ, Diaz-Barriga F, Calderon J, Rios C, Jimenez-Capdeville ME. Effects of lead-arsenic combined exposure on central monoaminergic systems. Neurotoxicol Teratol 1997;19:489–97.
- [113] Delgado JM, Dufour L, Grimaldo JI, Carrizales L, Rodriguez VM, Jimenez-Capdeville ME. Effects of arsenite on central monoamines and plasmatic levels of adrenocorticotropic hormone (ACTH) in mice. Toxicol Lett 2000;117:61–7.
- [114] Jana K, Jana S, Samanta PK. Effects of chronic exposure to sodium arsenite on hypothalamo-pituitarytesticular activities in adult rats: possibly an estrogenic mode of action. Reprod Biol Endocrinol 2006;4:9.
- [115] Martinez-Finley EJ, Goggin SL, Labrecque MT, Allan AM. Reduced expression of MAPK/ERK genes in perinatal arsenic-exposed offspring induced by glucocorticoid receptor deficits. Neurotoxicol Teratol 2011;33:530–7.
- [116] Goggin SL, Labrecque MT, Allan AM. Perinatal exposure to 50 ppb sodium arsenate induces hypothalamic-pituitary-adrenal axis dysregulation in male C57BL/6 mice. Neurotoxicology 2012;33:1338–45.
- [117] Gupta R, Dubey DK, Kannan GM, Flora SJ. Concomitant administration of Moringa oleifera seed powder in the remediation of arsenic-induced oxidative stress in mouse. Cell Biol Int 2007;31:44–56.
- [118] Grandjean P, Landrigan PJ. Developmental neurotoxicity of industrial chemicals. Lancet 2006;368:2167-78.
- [119] Tyler CR, Allan AM. Adult hippocampal neurogenesis and mRNA expression are altered by perinatal arsenic exposure in mice and restored by brief exposure to enrichment. PLOS ONE 2013;8:e73720.

This page intentionally left blank

20

Arsenic and the Cardiovascular System

Ashish Mehta, Chrishan J.A. Ramachandra, and Winston Shim NATIONAL HEART RESEARCH INSTITUTE SINGAPORE, NATIONAL HEART CENTRE SINGAPORE, SINGAPORE

CHAPTER OUTLINE

20.1 Introduction	459
20.2 Cardiovascular System	461
20.3 Arsenic Effects on Blood	462
20.4 Arsenic Effects on the Vascular System	468
20.5 Arsenic Effects on the Heart	474
20.6 Human Pluripotent Stem Cells: Understanding Arsenic Toxicity	481
20.7 Conclusions	483
References	483

20.1 Introduction

Arsenic (As), bearing atomic number 33 with a relative mass of 74.92, has physical and chemical properties in between metals and non-metals, and is thus often referred to as a metalloid. This existing metalloid belongs to Group VA of the periodic table with four oxidation states (-3, 0, +3, and +5), out of which arsenite (As^{III}) and arsenate (As^V) are the most prevalent forms found in the environment. As is ubiquitous and can be found naturally in water, soil, rocks, and living organisms in concentrations ranging from parts per million (ppm) to parts per billion (ppb). Interestingly, it is very rare to find As in free natural form; instead As combines with a variety of other elements such as oxygen, hydrogen, sulfur, nickel, cobalt, copper, iron, lead, calcium, magnesium, and zinc, and many more. This versatility of binding or coexistence of As renders its presence ubiquitous, and it is therefore impossible for animals, including humans, to avoid exposure to it from natural sources. Thus, exposure to arsenic is inevitable.

From a biological and toxicological perspective, arsenic exists in three major forms: inorganic, organic, and arsine gas. Of the inorganic forms arsenic trioxide, sodium arsenite (NaAsO₂), and arsenic trichloride (AsCl₃) are the most common trivalent forms, whereas arsenic pentaoxide (As₂O₅), arsenic acid (H₃AsO₄), and arsenates (AsO₄³⁻) (such as lead arsenate) are the most common pentavalent forms. Organic arsenic compounds include arsanilic acid (C₆H₈AsNO₃), methylarsonic acid (CH₅AsO₃), dimethylarsenic acid (C₂H₇AsO₂), and arsenobetaine (C₅H₁₁AsO₂).

Arsenic is the 20th most abundant element in Earth's crust but the major sources of arsenic exposure to humans are drinking water and occupational exposure. Arsenic in water usually comes from natural sources with varying concentrations; however, high arsenic levels occur in hot water springs, and in ground water in areas of thermal activities or rocky areas with high salt concentrations. Further, the arsenic form and its concentrations depend on various factors. These include the level of oxygen in the water, the degree of biological activity, the type of water (fresh, sea, or ground) and the proximity to arsenic-rich geographical locations. Fresh water arsenic concentrations are usually less than $10\mu g/L$ but can be as high as 5 mg/L near As-rich locations. Similarly, ground water averages $1-2\mu g/L$ but can be as high as 3 mg/L near volcanic regions (WHO, 2001). Exposure to high levels of arsenic in drinking water is well reported from around the globe, notably in China, India, Bangladesh, Taiwan, and to a lesser extent in Argentina, Australia, Chile, Mexico, the USA, Vietnam, Japan, Thailand, and Brazil. The levels in some of these regions are astonishingly high, from tens to a few thousand micrograms per liter. It is estimated that more than 100 million people worldwide are exposed to arsenic levels exceeding $50 \mu g/L$ [1]. Exposure also takes place during certain operations such as mining, smelting, and burning fossil fuels. Furthermore, exposure could also occur through As-based pesticides, insecticides, food products, and herbal medicines, but exposure from these sources has now declined considerably as per Environmental Protection Agency (US EPA) guidelines [2].

Arsenic exposure and its association with numerous diseases such as blackfoot disease, cancers, central nervous system dysfunction, anemia, and cardiovascular disease (CVD) including coronary heart disease (CHD), stroke, and peripheral arterial disease (PAD) is well reported in the literature [1]. While chronic arsenic exposure has been linked to numerous diseases, the most well-established mechanism for arsenic toxicity in biological species is oxidative stress. Arsenic increases the production of reactive oxygen species (ROS) such as hydroxyl radicals, hydrogen peroxide, superoxide radicals, and many more (reviewed by Flora [3]). These ROS specifically disrupt the mitochondrial membrane potential and activate translocation of cytochrome c and caspase, resulting in cellular apoptosis. Apart from cellular apoptosis, arsenic also interacts with numerous key enzymes, resulting in physiological dysfunctioning.

The only respite patients have from arsenic poisoning is chelation therapy. The use of chelation therapy dates back more than 100 years. There are not many chelating agents available for therapeutic purposes. British anti-Lewisite was developed as an antidote to Lewisite (dichlorovinyl arsine) during World War II but it has numerous side effects including essential metal loss and toxicity [4]. DMSA (meso-2,3-dimercaptosuccinic acid), a dithiol chelator with minimal side effects, was effective in treating chronic arsenic poisoning in animals, but double blinded randomized clinical trials did not yield any beneficial results [5]. Recently, MiADMSA, an isoamyl ester of DMSA, has shown to be more promising than the parental compound [4,6–10]. Phase I clinical trials for MiADMSA have been approved and their results are awaited. However, currently there is no literature that shows the effect of chelation therapy on the cardiovascular system. Although there have been numerous studies on arsenic and its toxicity in human and animal models, most of these are focused on arsenic-induced oxidative stress and recovery of biochemical variables post-therapy with chelating agents or antioxidants or their combination. Studies have also shown linkages of arsenic to various diseases but most of the focus has been on neuronal, hepatic, and cancer studies. In-depth and systematic studies investigating the effects of As on the cardiovascular system are limited. In this chapter, we shall focus on the effects of arsenic on this system. The chapter has been divided based on the major constituents of the cardiovascular system (blood, vasculature, and heart); however, an overlap of information may be present as it would be hard to delineate the effects arsenic has on individual tissues.

20.2 Cardiovascular System

In general, the cardiovascular system, also known as the circulatory system, consists of heart, a network of blood vessels (arteries, veins, and capillaries), and approximately 5 liters of blood, which are responsible for transporting oxygen and nutrients throughout the body. It is also known to be the hardest working organ of the human body. While transportation of materials such as oxygen and nutrients is one of the key functions of this system, it also protects the body through circulating white blood cells in case of infections/diseases and for cellular debris clean-up. The cardiovascular system is, in addition, instrumental in maintaining of body homeostasis by regulating body temperature, pH, and osmotic concentrations.

Chronic arsenic exposure is well documented to induce cancers of the skin, bladder, and lungs but the association between chronic arsenic exposure and cardiovascular disease has been less well studied. However, in the recent past, epidemiological studies have provided highly indicative data supporting the role of arsenic in cardiovascular disease. Although not conclusive, recent studies have shown that the linkage or disease aggravation may be more prominent in arsenic-exposed populations. It is now becoming more evident that arsenic is linked with vascular diseases, endothelial dysfunctioning, blood pressure, hypertension, ischemic heart disease, atherosclerosis, calcium overload, and arrhythmias. Furthermore, oxidative stress in tissues caused by the burden of arsenic leads to dysfunction of structural and functional proteins, further elevating cardiovascular disease. Early exposure to arsenic during pre- or neonatal developmental stages increases mortality in infants due to acute myocardial infarction and arterial diseases. Moreover, arsenic exposure is also known to alter micro-RNA (miRNA), a group of small RNAs that regulate gene expression and are known to play a critical role in modulating embryonic and fetal development. Changes in miRNA levels during development are associated with alterations of histone modifications and DNA methylation status, which regulates gene expression at the epigenetic level. In the following sections, we shall discuss in detail the effects of arsenic on heart, blood, and vasculature in animal models and humans and attempt to gain a better understanding of the role of arsenic in the cardiovascular system. In the last part of the chapter, we showcase a new technology that has induced pluripotent stem cells or embryonic stem cells that can now provide us with both a better model to study human development and also humanized models to study cardiac development in arsenic-exposed conditions.

20.3 Arsenic Effects on Blood

In this section, we focus on the toxic effects of arsenic on one of the major components of the cardiovascular system: blood. While oxidative stress is currently the most well-documented mechanism of arsenic toxicity, it is believed that arsenic's direct interaction with components of blood such as hemoglobin may also play a pivotal role in mediating its toxicity. Blood, being the first line of cellular defense, is directly affected by arsenic. Arsenic enters the bloodstream as arsenate and is metabolized within hours, being reduced to arsenite before entering the cell. Since blood is one of the most easily accessible biological fluids and can provide systemic information on the biological effects of arsenic, a plethora of literature exists where blood is used as a means to understanding arsenic poisoning. While it would be impossible to cover all these studies in this chapter, we summarize the major findings that have helped researchers in understanding the role of arsenic in blood.

20.3.1 Erythrocytes

Hemoglobin (Hb) is one of the main constituents of red blood cells and its interaction with arsenic is well reported. Hb is rich in -SH groups and it is suggested that arsenic and its metabolites may have a propensity to bind to Hb and reduce its binding affinity to oxygen. Recently, Mondal et al. [11] studied the binding affinity of trivalent arsenic to Hb and its consequential modification of the structure and function of Hb. Isolated Hb from arsenicosis patients had a binding affinity constant of $0.256 \,\mu M^{-1}$ with a Hill coefficient of +2.961, suggesting that arsenic molecules modify and lower the binding affinity to oxygen. The study also showed that Hb has a higher binding affinity to arsenic than to oxygen. Similarly, Lu et al. [12] evaluated the binding affinities of four trivalent As species: iAs, MMA, DMA, and PhAsO in rat and human Hb. Rat Hb demonstrated a 3-16 times stronger binding to arsenic than did human Hb but most of it was found in protein-bound form. In two early studies, Breton et al. [13] investigated the role of Hb in the development of arsenic-induced skin lesions in 900 case-control pairs from Bangladesh using conditional logistic regression. The second study examined the association between toenail arsenic, urinary arsenic, and Hb within a cohort of 184 individuals from 50 families in the same region who did not have arsenic-induced skin lesions. While Hb showed a significant association with skin lesions in the first study that was gender specific for males, no association was observed between toenail, urinary arsenic, and Hb levels in the second study. These studies show that in individuals with skin lesions, there was a clear relationship between arsenic binding and Hb levels.

A plethora of studies performed in animals by Flora and his group have shown that arsenic is known to inhibit the rate limiting enzyme δ -aminolevulinic acid dehydratase (ALAD), a sulfhydryl-rich enzyme of the heme biosynthesis pathway [14–18]. In mammalian and avian tissues the principal product of this pathway is heme (ferro-protoporphyrin IX), an essential component of various biological functions including oxygen transport, mixed function oxidative reactions, and other oxidative metabolic processes. In one such study, Bhadauria and Flora [17] showed that dose-dependent exposure of As^{III} or As^V, *in vitro* or *in vivo*, demonstrated a significant loss of ALAD activity. It was, however, interesting to note that at higher doses, As^{III} showed a more toxic effect than As^V. These effects were attributed to the attachment of arsenic to -SH groups of the enzyme since As has a strong affinity for these groups [14,19], which renders the enzyme inactive. It is also suggested that inhibition of ALAD results in an accumulation of ALA, which may play a role in arsenic-mediated ROS generation in blood [17]. These studies provide evidence that arsenic interferes with heme biosynthesis and ALAD may be a sensitive biomarker to evaluate heavy metal poisoning. One of the major components of the antioxidant defense system is glutathione, a cysteine-rich, primary intracellular antioxidant molecule. While glutathione may act in multiple pathways, the two important pathways in this context are its donation of an electron to ROS (simultaneously converting to GSSG) (glutathione disulfide) and donation of an electron to reduce arsenate to arsenite. Yildiz and Cakir [20] demonstrated that arsenic exposure causes an efflux of glutathione from erythrocytes in a dose- and time-dependent manner. In another cross-sectional study, a total of 378 participants from areas with different arsenic water concentrations were recruited and their blood GSH and GSSG were evaluated through HPLC. The results showed that the interquartile range increase in water arsenic was negatively associated with blood GSH and urine arsenic levels [21]. These observations are consistent with other observations in animal studies where arsenic exposure influences GSH levels in blood [22,23]. These results suggest that arsenic not only has disruptive effects, by direct interactions with heme biosynthesis, but also has important antioxidant molecules like GSH in blood. Apart from these biochemical changes, arsenic has also been shown to induce morphological changes in red blood cells (erythrocytes). Winski and Carter [24] demonstrated that sodium arsenite or arsenate induced significant death of red blood cells (hemolysis). Furthermore, morphological changes such as classical discocyte-echinocyte transformation to sphero-echinocytes were observed in a dosedependent manner in arsenate groups but not extensively in arsenite groups. Furthermore, these changes coincided with ATP depletion in these cells and it was suggested that erythrocytes are 1000 times more sensitive to arsenate than arsenite. A decade later, Biswas et al. [25] showed that a decreased lifespan of erythrocytes in arsenic-exposed populations resulted in anemia. This decrease in lifespan was due to significant morphological changes where human erythrocytes transformed from smooth discoid red cells to echinocytic to irreversible spheroechinocytes. Furthermore, arsenic exposure decreased cellular ATP along with increased microviscosity and an elevated cholesterol/phospholipid ratio that hampered the flexibility of the erythrocytes. These cumulative effects together triggered apoptotic removal of cells from the system, depicting the involvement of both erythrophagocytosis and hemolysis in the destruction of human erythrocytes during chronic arsenic exposure.

In a separate study the same group [26] also showed that chronic arsenic exposure in rats led to a gradual accumulation of arsenic in erythrocytes and 4-hydroxylnonenal (4-HNE) contributed significantly to the pathological events observed during oxidative stress. 4-HNE triggered death signal cascade, which was initiated by the generation and accumulation of various reactive radicals (e.g., H₂O₂, superoxide, and hydroxyl radicals) that eventually increased caspase activity leading to cellular apoptosis in rat erythrocytes. Interestingly, antioxidants such as NAC could attenuate these effects. The study also demonstrated that arsenic accumulation

caused degradation of band 3 leading to its aggregation on the surface of erythrocytes, which resulted in the activation of the complement system to eliminate the affected cells from the circulation. These studies indicate that arsenic could reduce the lifespan of red blood cells in circulation. Exposure of erythrocytes for 5 minutes to arsine, the simplest yet most toxic form of arsenic, at 1 mM results in hemolysis, accompanied with loss of cell volume, leakage of potassium, and influx of sodium. However, no significant changes in ATP levels were observed [27]. Interestingly, Biswas and co-workers [28] demonstrated the role of cholesterol in promoting ROS-mediated Fas signaling-dependent apoptosis via caspase-3 in erythrocytes post-arsenic exposure, suggesting a new pathway for cellular apoptosis in red blood cells.

While the above studies demonstrate the direct binding effect of arsenic in red blood cells in humans and animals, arsenic may also indirectly exert its toxic effect through oxidative stress. Flora and co-workers [29] demonstrated that chronic exposure of rats to arsenic at 25 ppm through drinking water for 24 weeks significantly reduced blood ALAD levels, enhanced zinc protoporphyrin (ZPP), and also altered various clinical hematological variables like WBC, MCH, MCHC, and platelet counts. Furthermore, these changes were accompanied by decreased SOD and increased catalase activities. Recently, Prabu and Sumedha [30] demonstrated that rats orally administered with 5 mg/kg of arsenic for 28 days showed significant alterations in plasma marker enzymes, plasma and erythrocyte membrane oxidative stress markers, erythrocyte membrane antioxidant enzymes, and non-antioxidant enzymes indicative of oxidative stress in erythrocytes. Arsenic exposure also reduced the activity of membrane bound ATPase and increased the percentage of DNA tail length in lymphocytes, suggesting DNA damage. Further, in a population-based study, Caciari and co-workers [31] determined that there was a correlation between chronic low-dose arsenic exposure and alterations in the hematopoietic system. The authors recruited 349 policemen and divided them into three groups based on urinary As levels. The subjects were further divided according to gender, smoking habits, and work task. Based on multiple linear regression and correlation analysis, a significant correlation and regression was seen between urinary arsenic, RBC, WBC, Hb, hematocrit, and reticulocytes. While arsenic causes dysfunction of blood through ROS generation, Chou and colleagues [32] suggested that this may be beneficial for the therapeutic regimes of acute myelocytic leukemia (APL) patients. The authors suggested that arsenic mediated ROS could induce apoptosis in cancer cells. Using interference RNA and genetically engineered cells, they demonstrated that NAPDH oxidase is a prime target of ROS, which is important for antibacterial function of WBCs, which may play a critical role in the treatment of APL patients.

20.3.2 Leukocytes

Apart from red blood cells, other cell types are also affected by arsenic. A microarray-based genome-wide analysis of expression patterns from peripheral blood of 21 samples showed significant differences in the abundance of transcripts between high- and low-arsenic exposure groups. The major genes that were altered belonged to defense response, immune function, cell growth, apoptosis, cell-cycle regulation, T-cell receptor signaling pathway, and diabetes. Furthermore, groups exposed to higher arsenic concentrations exhibited several killer cell immunoglobulin-like receptors that could inhibit natural killer cell activities [33]. Banerjee and co-workers [34] compared the degree of apoptosis in peripheral blood of 30 arsenic exposed individuals with skin lesions with that in 28 unexposed individuals and observed that the percentage of apoptotic cells was significantly higher (p < 0.001) in the former group. The mechanism for cellular apoptosis in these peripheral blood mononuclear cells (PBMCs) was shown to be ROS driven and mitochondrial dependent. Furthermore, arsenic-exposed PBMCs showed cell cycle arrest at the G0/G1 phase. Two independent studies by Avani and Rao [35] and Pant and Rao [36] showed that exposure to arsenic caused peripheral blood lymphocytes and blood cultures to significantly increase the incidence of chromatid exchange and alter the cell-cycle proliferative index. More recently, Sharma and co-workers [37] reported on one of the probable mechanisms of how arsenic could impair cell-cycle regulation. Low-dose arsenic exposure of PBMCs resulted in overexpression of miR-2909. This overexpression of miR-2909 was shown to regulate cyclin D1 expression within these cells, by inducing splice switching of tumor suppresser CYLD as well as modulation of SP1 activity through repression of KLF4 expression at the translational level. Furthermore, As-dependent regulation of ATF and BCL3 was also modulated through its capacity to induce miR-2909 expression. This may be a critical step in how arsenic modulates cell-cycle regulation in lymphocytes.

While apoptosis and cell-cycle regulation may be more general effects of arsenic, the subtype population with the lymphocytes shows alterations of various pathways. Gupta and co-workers [38] showed that As₂O₃ at clinically achievable therapeutic concentrations induces apoptosis in normal T cells by increasing the ROS burden, decreasing mitochondrial membrane potential and intracellular GSH, down-regulating Bcl-2, translocating cytochrome c, and activating caspases. Furthermore, addition of exogenous GSH or overexpression of Bcl-2 inhibited As₂O₃-mediated apoptosis in T cells. Yu and co-workers [39] investigated T-cell survival and function in mononuclear cells and showed that arsenic at $1 \mu M$ dose induces TNF- α release and had cytotoxic effects on T cells, especially CD4⁺ cells inducing apoptosis. This enhanced TNFR1 expression along with death domain proteins and caspases was more prominently observed in CD4⁺ cells as compared to other mononuclear cells, suggesting the role of TNFR1 signals in arsenic-induced T helper cell apoptosis. In a similar study, dose-dependent exposure to sodium arsenite (0.001, 0.1, and 1μ M) showed that 1μ M was more toxic to T-helper (CD4⁺) than T-cytotoxic (CD8⁺) cells in human PBMCs. Furthermore, PBMCs from females demonstrated a more reduced T-lymphocyte cell proliferation than males, suggesting that females may be more susceptible to arsenic toxicity [40]. The study by Morzadec et al. [41] demonstrated that non-cytotoxic concentrations of NaAs (0.25-2 µM) significantly reduced T cell proliferation by increasing the percentage of non-dividing cells via causing a block in the G1 phase and preventing cyclin D3 and CDC25A expression, which may be IL-2 independent. Contrary to this, Martin-Chouly and colleagues [42] showed that arsenic reduces IL-2 secretion and T cell proliferation. They also showed that arsenic blocks monocyte-dependent accessory signals by PHA resulting in reduced expression of monocyte-derived cytokines such as IL-1, IL-6, and TNF- α . Moreover, microarray studies showed that 35 genes related to the immune system and stress factors were also significantly up-regulated. These results showed that arsenic interacts with T cells directly.

Apart from T cells that constitute the immunological system, arsenic is known to impair macrophages too. Effects of inorganic and methylated arsenicals were evaluated in murine macrophages in vitro. Interestingly, both inorganic arsenicals (arsenite and arsenate) induced activation of inflammatory cytokines, TNF- α , and necrotic death (80%) with limited apoptosis (20%) at low doses and were strongly toxic to macrophages. In contrast, the methylated arsenical, DMMA suppressed TNF- α levels and exerted cell death through apoptosis at significantly higher doses than inorganic arsenicals. MAA and TMAO did not show any significant toxicity at even higher concentrations, suggesting that methylation of arsenicals may reduce the inflammatory and immunological responses of inorganic arsenicals [43]. Arsenic trioxide also significantly increases superoxide levels in a time-dependent manner in human monocyte-derived macrophages. Phosphorylation of p38 kinase followed by phosphorylation and membrane translocation of NADPH oxidase subunit p47(phox) and translocation of Rac1 and p67(phox) has been noted. Using selective inhibitors of NAPDH oxidase (Apocynin) or p38 (SB203580) these changes could be reversed in macrophages, suggesting that arsenic induces alterations in Rho-kinase/p38-kinase pathways, which impairs macrophage function [44]. In humans, hematopoietic cells from 70 individuals with skin lesions that were chronically arsenic exposed were compared with 64 controls [45]. Arsenic-exposed macrophages from these individuals showed cell rounding with a significant (p < 0.0001) loss of cell adhesion capacity, decrease in NO production, impaired phagocytic capacity, and decreased CD54 and F-actin expression, with affected RhoA-ROCK pathway, suggestive of immunosuppression in these individuals [45]. Microarray studies on human monocyte-derived macrophages following arsenic trioxide treatment for 72h show up-regulation of 32 genes and repression of 91 genes. Twenty-six genes altered in these studies were specifically related to the differentiation program of macrophages. The prominently altered genes were MMP9, MMP12, CCL22, SPON2, and CXCL2, which are an integral part of macrophage function [46]. These microarray studies have been supported by research investigating chronic exposure as well. Prolonged exposure to arsenic alters macrophage-specific (MMP9, CCL22 and CXCL2), stress-specific (HMOX1 and GCLM) genes and nuclear levels of Nrf2 and Bach1, and represses the expression of transcriptional factor EGR2, a protein involved in macrophage differentiation [47]. More recently, Wang et al. [48] showed that non-lethal concentration of arsenic trioxide (1 and $2.5 \,\mu$ M) markedly inhibited monocyteto-macrophage differentiation together with expression of macrophage markers. However, these changes could be reversed following treatment with α -lipoic acid. Taken together, these studies suggest that arsenic alters signaling pathways important for differentiation as well as the function of macrophages and thus indirectly compromises the cellular immune system.

Like macrophages, lymphocytes show differential effects towards arsenic species. For instance, Colognato et al. [49] evaluated the genotoxic effects of six different As compounds (As^{III}, As^V, MMAs^{III}, MMAs^V, DMAs^V, and TMAO^V) on lymphocytes. While 4 μ M of As^{III} and 32 μ M of As^V increased micronuclei frequency, a statistically significant increase in micronucleus frequency was also seen with 2 μ M of MMA^{III} and 500 μ M of MMA^V. No significant micronuclei were observed with DMA^V and TMAO^V, suggesting that MMA^{III} had aneuploidogenic properties and could induce geno- and cytotoxic effects. A study of 87 children aged 6–10 years evaluating the effects of arsenic exposure on immune cells using multiple linear

regression analysis showed iAs had a positive association with basal NO⁻ in PBMCs (p = 0.049) and monocytes (p = 0.04), while basal O₂⁻ had an association with DMA (p = 0.046) [50]. Interestingly, in activated monocytes, a positive association was also observed with iAs, MMA, DMA, and tAs, suggesting that activated and inactivated cells could have differential responses [50]. Apart from T cells and monocytes, macrophages also play a vital role in the human immune defense system.

In one of the early pilot studies, analysis of blood and urine samples from 11 individuals chronically exposed and 13 individuals with lower exposure to arsenic was performed [51]. Interestingly, percentages of chromosomal aberrations and the frequencies of sister-chromatid exchanges were similar in both groups. However, complex aberrations were more frequent in the highly exposed group, which also showed a higher average variation frequency in the HGPRT assay. However, these observations were not statistically significant [51]. Later, in vitro studies evaluating lymphocytes from control, symptomatic (having As-induced lesions), and asymptomatic (no skin lesions but exposed to As) samples found significant changes in chromosomal aberrations [52]. While both symptomatic and asymptomatic samples accumulated more arsenic in nails and hair, excreted less in urine, and showed chromosomal aberrations as compared to the controls, significantly higher incidences of aberrations were observed in the symptomatic group [52]. Similarly to this study, lymphocytes of 30 individuals exposed to high levels of As $(247 \pm 18.93 \,\mu\text{g/L})$ through drinking water in West Bengal, India, when compared with 30 unexposed ($7.6 \pm 0.5 \,\mu$ g/L) individuals, showed significant signs of DNA damage (p < 0.01) in lymphocytes. The comet tail lengths showed a positive correlation with the arsenic levels [53]. In another study, telomere length and expression of genes related to telomeres were evaluated in a cohort of 202 women exposed to a wide range of arsenic $(3.5-200 \,\mu g/L)$ [54]. A significant correlation was observed with telomere length and urinary arsenic levels. Moreover, individuals with above median fractions of iAs and MMA showed longer telomere lengths than those below the median [54]. As evidenced by these reports, DNA-based tools could be used as a sensitive biomarker for arsenic poisoning.

20.3.3 Methyltransferases

Although arsenic is a potent toxicant, the efficiency by which the biological system metabolizes arsenic may govern differences between the toxic effects of arsenic across various populations. This complex process is governed by the arsenic methyltransferase gene (*AS3MT*), and more recently a putative N-6-adenine-specific DNA methyltransferase 1 (*N6AMT1*) has been reported to methylate arsenic *in vitro* [55]. A large number of studies in recent years have been done using lymphocytes/blood to demonstrate that changes in epigenetic levels may regulate the toxic manifestations of arsenic. In one study, Chervona and colleagues [56] evaluated global histone methylation patterns of H3K9me2, H3K9ac, H3K27me3, H3K27ac, H3K18ac, and H3K4me3 in peripheral blood from a subset of folate clinical trial participants (n = 40). The study reported that total urinary arsenic levels correlated with H3K9me2 (p = 0.02) and inversely with H3K9ac (p=0.002). Interestingly, other methylation patterns showed a gender bias, where H3K4me3 and H3K27me3 positively correlated with females, whereas H3K27ac and H3K18ac negatively correlated with females, indicating that arsenic exposure affects the whole genome methylation pattern and there may be a gender-specific pattern of histone association marks showing linkage to epigenetic dysregulation. However, most of the studies are focused on specific methyltransferase genes. Engström et al. [57] evaluated the impact of polymorphism of five methyltransferase genes (*AS3MT*, *DNMT1a*, *DNMT3b*, *PEMT*, and *BHMT*) on As metabolism in peripheral blood of two groups (South American (n = 172) and Southeast Asian (n = 361)) with urinary arsenic levels 200 µg/L and 100 µg/L, respectively. The results showed that six *AS3MT* polymorphisms were significantly associated with As metabolite patterns in both groups (p < 0.01). Interestingly, the frequency of the *AS3MT* haplotype in Bangladesh was prominently associated with DMA% and not MMA%. More specifically, *AS3MT* Met287Thr allele frequencies were evaluated and a micronucleus assay was performed on 207 subjects exposed to different arsenic levels. While peripheral blood micronucleus frequencies poorly correlated with arsenic levels, *AS3MT* Met287Thr genotypes carrying the variant allele showed significant correlation with micronucleus frequencies (p = 0.0003) [58].

In a recent study, arsenic methylation efficiency in 188 women exposed to arsenic contaminated drinking water ($\approx 200 \,\mu$ g/L) in the Argentinean Andes was genotyped for *N6AMT1* and *AS3MT* polymorphism in blood samples [55]. Five *N6AMT1* SNPs (single nucleotide polymorphisms) and two *N6AMT1* haplotypes were significantly associated with the MMA% in urine. The MMA% increased uniformly according to the number of alleles for each SNP (mean MMA% was 7.5% for GG, 8.8% for GT, and 9.7% for TT carriers). Recently, Gribble et al. [59] evaluated methylation status of 48 CpG loci at the *AS3MT* promoter region from 48 participants. Based on linear regression analysis a hypomethylated region in the *AS3MT* promoter was associated with higher arsenic exposure. These results suggest that not only the gene polymorphism but also the promoter regulation is disrupted by arsenic [59]. These studies clearly suggest that governance of arsenic metabolism may be playing a very important role that could provide indications on the level of toxicity arsenic would induce on prolonged exposure. These studies pave the way to develop better diagnostic tools for predicting population susceptibility to various arsenic-linked diseases such as cancers.

20.4 Arsenic Effects on the Vascular System

20.4.1 Clinical Studies

Many studies have been conducted that show a direct correlation between As exposure and increased risk of cardiovascular disease (CVD). Wang et al. conducted a study where 605 Taiwanese residents were evaluated for development of carotid atherosclerosis due to As exposure [60]. Residents who fell into the following categories of carotid artery intima-media (IMT) thickness >1.0 mm, plaque score \geq or stenosis >50% were diagnosed as having carotid atherosclerosis. During the study a dose-dependent relationship between As exposure and increased risk of carotid atherosclerosis was observed. Residents who had been exposed to high levels of As (>50 µg/L) had an age- and gender-adjusted odds ratio (OR) of 3.3 for developing carotid

atherosclerosis while those who had been exposed to low $(\leq 10 \mu g/L)$ and mid-levels (10.1– $50 \mu g/L$) of As had an OR of 1.0 and 1.4, respectively [60]. Similarly, Hsieh and co-workers also showed a dose-dependent relationship between As exposure and increased risk of carotid atherosclerosis in Taiwan [61]. Residents who had been exposed to high levels of As in drinking water (\geq 50 µg/L) with a cumulative arsenic exposure (CAE) of \geq 1.1 mg/L/year had an age- and gender-adjusted OR of 2.4 and 1.9 for developing carotid atherosclerosis, respectively (p < 0.05). Apart from carotid atherosclerosis, hypertension has also been associated with As exposure. Huang and colleagues showed that inefficient As methylation (the ability to metabolize and excrete As) may lead to increased risk of hypertension. Based on a positive history or measured systolic blood pressure (\geq 140 mm Hg) and/or diastolic blood pressure (\geq 90 mm Hg), 372 out of 871 As-exposed Taiwanese residents were diagnosed as having hypertension. The primary As methylation index [PMI, defined as monomethylarsonic acid (MMA^V) divided by (As^{III}+As^V)] and secondary arsenic methylation index [SMI, defined as dimethylarsinic acid (DMA^V) divided by MMA^V were the indicators used to determine the level of As methylation [62]. Interestingly, although the residents had stopped consuming artesian well water for 20-30 years, the level of As in urine was still directly related with CAE (p = 0.02), indicating the long-term effects of As exposure. Residents with hypertension had higher percentages of MMA^V and lower SMI than those that did not have the disease. However, residents having CAE >0 mg/L/year had a higher risk of hypertension than those who had CAE = 0 mg/L/year regardless of the methylation index [62].

A study of 11,746 residents in Bangladesh was evaluated for a period of 6 years to investigate the affiliation of As exposure and CVD mortality rate [63]. A dose-dependent relationship was observed between As exposure and CVD mortality rate. Residents who had been exposed to low (<12.0 μ g/L) and high levels of As (\geq 12.0 μ g/L) had mortality rates of 214.3 and 271.1 per 100,000 person years, respectively. When As exposure was further categorized into low (0.1–12.0 μ g/L), mid- (12.1–62.0 μ g/L), high (62.1–148.0 μ g/L), and very high levels (148.1-864.0 µg/L), the hazard ratios for ischemic heart disease (IHD) and other CVDs were 1.00, 1.22, 1.35, and 1.92, respectively (p = 0.0019 for trend). A synergistic interaction between As exposure and cigarette smoking was also associated with CVD mortality. This joint effect was instrumental in increasing CVD mortality in that the hazard ratio of moderate As exposure levels (mean $63.5 \mu g/L$) and cigarette smoking was greater than the sum of the hazard ratios associated with individual effects (p = 0.010) [63]. While most of these studies were conducted in the Asian continent, studies from European regions also show similar trends. Medrano and colleagues investigated the association of As concentrations with CVD mortality by analyzing the standardized mortality rates (SMR) for cardiovascular, coronary, and cerebrovascular disease at the municipal level [64] between 1998 and 2002. For municipals having As concentrations $>10 \mu g/L$, the SMR for cardiovascular, coronary, and cerebrovascular disease were 1.10, 1.18, and 1.04, respectively. Further, municipals having As concentrations between $1-10 \,\mu\text{g/L}$ and $>10 \mu g/L$ showed an increase of 2.2% and 2.6%, respectively in cardiovascular mortality rates when compared to municipals having very low As concentrations ($<1 \mu g/L$) (p-value for trend 0.032). For coronary heart disease the increase in mortality rate was 5.2% and 1.5% while for cerebrovascular disease it was 0.3% and 1.7%, respectively. These ecological studies demonstrate that prolonged As exposure could result in elevated CVD risk [64].

20.4.2 Atherosclerosis

It is well documented that prolonged As exposure leads to atherosclerotic vascular diseases [65,66] by inducing formation of ROS. Arsenic augments the levels of oxidative stress allowing for the accumulation of oxidized lipoproteins [67]. Since it was previously reported that these oxidized low-density lipoproteins (LDLs) can induce cytotoxicity [68], it is now speculated that atherosclerotic lesions along with endothelial cell dysfunction are caused by LDLs. Recently, Hossain and colleagues [69] demonstrated that sodium treatment significantly increased lectin-like oxidized LDL (oxLDL) receptor (LOX-1) levels in mouse aortic endothelial cells. Furthermore, there was significant induction in the phosphorylated forms of nuclear factor kappa-light polypeptide gene enhancer (NF- κ B)/p65 levels in cells, suggesting an important role of the LOX-1 signaling pathway in arsenic-induced atherosclerosis. While LOX-1 signaling is activated, FasL ligand is repressed by arsenite but not by arsenate in ECV304 cells through ROS [70]. Moreover, these effects could be moderated through specific blockers or antioxidants like NAC.

20.4.3 Genetic Polymorphisms

Therefore, antioxidant enzymes and As metabolizing enzymes are pivotal for As excretion and for preventing oxidative stress. However, genetic polymorphisms (variants) in the genes that code for these enzymes have been reported to result in decreased enzyme activity and with it increased vascular toxicity. When genetic variants in GSTM1, GSTT1, and GSTP1 as well as p53 were investigated for their role in As-induced carotid atherosclerosis, it was observed that individuals who were exposed to high levels of As (>50 μ g/L), having a GSTP1 (Ile/Val and Val/ Val) or p53 (Arg/Pro and Pro/Pro) double variant, increased their risk of developing carotid atherosclerosis, with an age- and gender-adjusted OR of 6.0 and 3.1, respectively (p < 0.001)[60]. No risk was associated with individuals having GSTM1 and GSTT1 variants. Furthermore, individuals with either one GSTP1 or p53 variant and those with two variants had an OR of 2.8 and 6.1 for developing carotid atherosclerosis, respectively. Interestingly, even though they had variants of *GSTP1* and *p53*, individuals exposed to low levels of As ($\leq 50 \mu g/L$) were not at risk for carotid atherosclerosis. When other risk factors were taken into consideration (age, hypertension, variant genotypes of GSTP1 and p53, and As concentration), individuals with double variants of GSTP1 and p53 still had a higher risk of developing carotid atherosclerosis, with a multivariate-adjusted odds ratio of 3.4 (p < 0.001) [60]. On similar lines, polymorphism within lipid metabolism and inflammation genes has also been shown to be linked to CVD. The relationship of polymorphism in APOE (lipid metabolism gene) and MCP-1 (inflammation gene) with the risk of carotid atherosclerosis was evaluated in 479 arsenic-exposed residents of Taiwan [61]. Residents with a carotid artery IMT thickness of ≥ 1.0 mm or with observable plaque in the extra-cranial carotid artery were diagnosed as having carotid arteriosclerosis. Moreover, residents with ε 4-allele of *APOE* had higher levels of LDL cholesterol compared with those residents who lacked this allele ($116.1 \pm 25.2 \text{ mg/dL}$ vs. $132.9 \pm 143.5 \text{ mg/dL}$, p = 0.0004), increasing the risk of carotid atherosclerosis by two-fold. On the other hand, individuals with *MCP-1* risk genotypes (A/G or G/G) had only a borderline chance of manifesting the disease. However, presence of both APOE and MCP-1 risk genotypes increased the risk of carotid atherosclerosis by 2.5-fold (p < 0.01). Further, individuals who had both risk genotypes and who had also been exposed to As concentrations of >10 µg/L or had CAE of >0.22 mg/L/year had a 10.3- and 15.7-fold higher risk of obtaining the disease, respectively (p < 0.05). When other risk factors were taken into consideration (gender, cigarette smoking, diabetes mellitus, cholesterol, and triglyceride), individuals with both *APOE* and *MCP-1* risk genotypes were still at a higher risk of developing carotid atherosclerosis. These studies indicate the joint effects of genetic polymorphisms and As exposure with regard to the development of carotid atherosclerosis. This study highlights that the coexistence of at least two risk genotypes may exacerbate the effects of As exposure and increase the risk of CVD.

While the above studies highlight the detrimental effects of high levels of As exposure, low levels of As exposure together with certain genetic polymorphisms can also elevate risk of CVD. A study in the rural Texas counties of Cochran, Palmer, and Bailey was conducted to investigate the effects of low levels of As exposure and a single nucleotide polymorphism (SNP) in AS3MT on coronary heart disease, hypertension, and hyperlipidemia [71]. The function of AS3MT is to catalyze methylation from MMA^{III} to DMA^V—an important pathway involved in As metabolism. A higher MMA^{III}/DMA^V ratio indicates toxicity while the inverse indicates detoxification. Therefore, 499 residents were evaluated for the presence of an A35991G SNP (rs10748835) within AS3MT. In this cohort the estimated groundwater As concentrations ranged from 2.2 to $15.3 \mu g/L$ (mean of $6.2 \mu g/L$). After adjustments were made for age, ethnicity, gender, education, smoking status, alcoholism, and anti-hyperlipidemia medication, it was found that coronary heart disease was associated with both low levels of As exposure (p < 0.05) and the G/G genotype of AS3MT (p < 0.05). While hypertension was associated with low levels of As exposure (p < 0.05), hyperlipidemia was only associated with the A/G genotype of AS3MT (p < 0.05) [71]. Although hyperlipidemia was not found to be associated with low levels of As, a study conducted by Das et al. showed that mice administered with high doses of sodium arsenite (10 mg/kg, orally) for 10 consecutive days exhibited significant oxidative stress as well as developing hyperlipidemia [72]. Apart from oxidative stress, recently it was shown that arsenic exposure significantly increased the expression of two angiotensin II type I receptors (AT1R A and B) in mouse aortic endothelial cells [73]. This in turn increased phosphorylation of JNK and activated AP-1, providing evidence of how arsenic may modulate pathogenesis of hypertension.

Apart from *AS3MT*, *PNP*, *GSTO1*, and *GSTO2* were evaluated as well. In Taiwan, 863 residents who had been genotyped and for whom the severity of carotid atherosclerosis had been determined were included in this study. Just like in the previous studies, a dose-dependent relationship was observed between arsenic concentration and increased risk of carotid atherosclerosis (p = 0.04). Interestingly, a significant association was not found between genetic risk polymorphisms of *PNP* (Gly51Ser, Pro57Pro), *AS3MT* (Met287Thr), *GSTO1* (Ala140Asp), and *GSTO2* (A183G) and risk of carotid atherosclerosis in low level As exposure groups. However, in high-level As exposure groups (>50 µg/L), residents who carried the *PNP* AT haplotype, *AS3MT* T/C SNP and *GSTO* CAA/CAG or AGG haplotype had a higher risk of carotid atherosclerosis ($p \le 0.05$). Additionally, coexistence of the *PNP* risk haplotype with either the *AS3MT* risk SNP or the *GSTO* risk haplotype severely increased the risk of the disease as indicative of

a multivariate-adjusted OR of 6.43 (p = 0.004). In support, residents who carried the *PNP* risk haplotype or the *AS3MT* risk SNP also had a significantly lower DMA/MMA ratio, indicating a reduction in As metabolism [74].

Furthermore, the occurrence of a GT-repeat polymorphism within the HO-1 promoter was evaluated in 504 participants in Taiwan who were followed for 10 years for the occurrences of CVD-related deaths, including coronary heart disease, cerebrovascular disease, and peripheral arterial disease [75]. Individuals who had <27 GT-repeats were grouped into the S-allele category and those with \geq 27 GT-repeats were grouped into the L-allele category. Individuals who carried the S-allele had a lower risk of CVD mortality than those who carried the L-allele. This was indicated by the crude mortalities for S/S, S/L, and L/L genotypes, which were 2.85, 3.10, and 8.42 cases per 1000 person-years, respectively. This study showed that longer repeats within the promoter of HO-1 had detrimental effects and increased the risk of CVD. Supporting this study, association of GT-repeat polymorphism with blood pressure (BP) and their interaction on CVD mortality risk in 894 As-exposed participants showed that carriers of the S-allele had lower diastolic BP (L/S genotypes, p = 0.014) and a reduced risk of being hypertensive (L/S genotypes, p = 0.048) [76]. After a 10-year follow-up, hypertensive individuals who carried the S-allele had a reduced risk of CVD mortality (p = 0.007). However, those who did not carry the S-allele had a 5.23-fold increased risk (p = 0.0008) of CVD mortality [76]. While both these studies showed a link between the HO-1 genotype and CVD in As-exposed populations, the researchers did not investigate the relationship of HO-1 GT-repeat polymorphisms with regard to As concentrations.

20.4.4 Endothelial Cells

Apart from genotype polymorphism, dysfunction of the endothelial cells may also be responsible for CVDs. In order to understand the mechanisms of As-induced CVD, in vitro and in vivo studies have also been conducted both in human and in animal models. Treatment of porcine aorta endothelial cells (PAECs) with arsenic trioxide (ATO) and NaAsO₂ at 20 µM resulted in a decrease in viability and diminished G0/G1 phase and increased apoptosis [77]. Although both forms of As induced similar cytotoxic effects, the mode at which they exerted their effects was different. For instance, NaAsO₂ treatment increased the G2/M phase whereas ATO treatment led to necrosis. Further, NaAsO₂ treatment led to p53 up-regulation whereas ATO treatment led to both p53 and caspase-3 up-regulation. In another study, NaAsO₂ treatment $(0.1-50\,\mu\text{M}$ for 24h) decreased the viability of human umbilical vein endothelial cells (HUVEC) in a time- and dose-dependent manner [78]. A cell cycle and apoptosis regulatory protein, p21^{Cip1/Waf1}, was activated by approximately 10-fold during this treatment. NaAsO₂ also stimulated and activated EGF and ErbB2 receptors, which appeared to be an upstream event leading to p21^{Cip1/Waf1} activation as small molecule inhibitors (and RNAi) of the receptors could block p21^{Cip1/Waf1} activation. Treatment with NaAsO2 also activated JNK and p38 MAPK, which could be inhibited with SP-600125 and SB-203580, respectively. This inhibition also diminished activation of p21^{Cip1/Waf1}, which prevented apoptosis. Tsou et al. showed an alternative mechanism in HUVECs, whereby arsenite, through regulation of AP-1 and NF- κ B, augmented TNF- α -induced VCAM-1 expression in a GSH-sensitive manner [79]. This has led to speculations that As-induced vascular toxicity may also be a result of coronary arterial endothelial cell apoptosis [80].

A more comprehensive study in SVEC4-10 mouse endothelial cells showed the effects of ATO with regard to cytotoxicity, intracellular ROS, gene expression, and signaling pathways [81]. ATO treatment reduced the viability of SVEC4-10 cells while simultaneously upregulated HO-1, IL-6, MCP-1, Nrf2, and VEGF transcript levels. Treatment also led to an increased activation of NF-kB. Interestingly, an HO-1 inhibitor (zinc protoporphyrin) reduced As-induced cell death in an inverted-U dose-response curve suggesting a biphasic role of HO-1. Targeted reduction of HO-1 or p38-MAPK decreased As-induced increase in VEGF expression. This study indicated that ATO activates HO-1 expression through Nrf2-, NF- κ B-, and p38 MAPK-dependent signaling pathways, which in turn act as upstream regulators of VEGF [81]. Moreover, experiments in mice exposed to low to moderately high-levels of As^{III} showed altered expression of angiogenic or cardiac tissue remodeling genes, such as VEGF, VEGFR, plasminogen activator inhibitor-1, and MMP-9, as well as cell-cell junction proteins, VE-cadherin and ZO-1, indicating the probable role of arsenic in vascular remodeling [82,83]. Conversely, Roboz and colleagues in 2000 showed that the therapeutic potential of arsenic against PML is the result of causing apoptosis in both normal and cancerous cells as well as inhibiting VEGF production [84]. Non-lethal doses of arsenic apart from increasing intracellular oxidant levels also increased the nuclear retention of NF-KB binding proteins. This retention was mainly regulated by p65/p50 heterodimers in endothelial cells [85]. In another study, it was shown that arsenic causes a release of a neuropeptide (substance P), which activates endothelial neurokinin-1 (NK-1) triggering a neurogenic inflammatory process in vivo that causes damage to blood vessels. Moreover, the study did not find any role of histamine release that may further regulate this process [86]. These studies suggest that multiple cellular pathways may be involved directly or indirectly through ROS during arsenic exposure.

20.4.5 Smooth Muscle Cells

Apart from inorganic forms, methylated forms of arsenic may also have a detrimental effect on blood vessels. Bae et al. conducted a study that investigated the effect of MMA^{III} on vasomotor tone of blood vessels. When phenylephrine (PE), serotonin, and endothelin-1 were used to induce vasoconstriction in rat thoracic aorta and small mesenteric arteries, MMA^{III} irreversibly suppressed this vasoconstriction [87]. Further, MMA^{III} directly interfered with the contractile function of vascular smooth muscle, as inhibition of vasoconstriction was still retained in aortic rings lacking endothelium. This phenomenon was a result of MMA^{III} inhibiting the Ca²⁺ increase, which was initially induced by PE via the intracellular store and L-type Ca²⁺ channel. MMA^{III} was also shown to directly block this channel as demonstrated in *Xenopus* oocytes. In an *in vivo* rat model, MMA^{III} also inhibited the increase in blood pressure that was induced by PE [87]. This study showed that apart from endothelial cells, metabolites of As could also induce dysfunction in smooth muscle cells, thereby contributing to vascular toxicity.

While these pathways and epidemiologic studies have shown association of arsenic with atherosclerosis and vascular dysfunction, Simeonova and colleagues in 2003 showed the role of apolipoprotein E (ApoE) in wild and ApoE^{-/-} mice models. Exposure to 20 or 100µg/mL arsenite for 24 weeks showed a marked increase in lesions in the intimal area of the aorta in ApoE^{-/-} mice but no change was observed in cholesterol levels [88]. Moreover, *in utero* arsenic

exposure of pregnant ApoE-deficient mice showed a two-fold increase in lesion formation in the aortic roots and arch as well as defects in vasorelaxation in response to acetylcholine response in offspring. While triglyceride levels decreased, no change was observed in cholesterol, phospholipids, and VLDL or HDL levels, suggesting endothelial signaling disturbances [89]. These studies in short demonstrate that endothelial cells may be one of the main targets within the vasculature system that is affected by high and low dose exposure to arsenic.

20.5 Arsenic Effects on the Heart

In this section, we focus on the detrimental effects of arsenic on one of the major constituents of the heart: the myocytes. While the toxic effects of arsenic exposure on liver or brain are well documented in the literature, the cardiotoxic effects are not well classified. However, it is now clear that there is a relationship between prolonged arsenic exposure and cardiotoxicity or heart injury [90,91]. As a result, researchers have investigated a plethora of factors responsible for As-induced cardiotoxicity. These include age, sex, race, demography, As concentration, and genetic polymorphisms. Insight has therefore been gained with regard to the risks associated with As exposure.

20.5.1 QT Prolongation

One of the earliest reports in the 1970s indicated that a female patient was admitted for ventricular fibrillation, which was due to chronic arsenic poisoning. With post-arsenic management with dimercaprol (BAL) and procainamide hydrochloride therapy, the episode of ventricular fibrillation subsided; the ECG showed non-specific ST segment changes and prolongation of the heart rate corrected (QTc) interval [92]. In a typical ECG trace, the cardiac cycle consists of a P wave, a QRS complex, and a T wave. The distance between the QT interval (shortened or prolonged) is used as a marker for determining the potential for ventricular tachycardia (VT), a type of arrhythmia that could result in sudden death. Thus, it would be no surprise that chronic arsenic poisoning could result in heart rhythm issues such as long QT or VT [93,94]. It was interesting to note that most of the heart rhythm changes were seen in patients under cancer management.

Arsenic trioxide (ATO/As₂O₃) is commonly used to treat patients with acute promyelocytic leukemia (APL) owing to its ability to induce apoptosis and suppress proliferation of cancer cells as well as reduce angiogenesis. In a clinical trial of ATO in the treatment of relapsed and resistant cases of acute promyelocytic leukemia, adverse effects from arsenic were observed. The ATO therapy caused prolongation of corrected QT interval and three patients developed TdP (torsades de pointes, a form of ventricular tachycardia) [93]. Another study, which investigated the effects of ATO in 99 patients with advanced malignancies with APL, who received 170 courses of arsenic trioxide in either a phase I or phase II investigational study, reported that 38 patients developed a QT interval prolongation with \approx 70% being >500 ms (normal QT interval is \leq 400 ms) [95]. When normalized against the heart rate (corrected QT/QTc), the increase in QTc for 36.6% of patients was between 30 and 60 ms and for 35.4% of patients the increase was more than 60 ms. Interestingly, following the withdrawal of ATO, QTc intervals returned to normal suggesting that ATO does not permanently prolong the QT interval. However, the degree of

prolongations was higher in men than in women during the first course of the therapy and in patients with hypokalemia [95].

In another ATO study, Yamakazi and co-workers [96] investigated the incidence and mechanism of arrythmogenesis caused by ATO in 20 APL patients. As₂O₃ (0.15 mg/kg) significantly prolonged the corrected QT interval (QTc: 445 ± 7 to 517 ± 17 ms, p < 0.01), and also increased the QTc dispersion and transmural dispersion of repolarization. While four patients showed non-sustained VTs, one had TdP [96]. Ohnishi et al. [97] also observed that all of the eight patients on ATO therapy developed QT prolongations, out of which four developed non-sustained VT. Similarly, in another multicentric ATO study in the USA, 63% of the patients showed QT prolongations but only one of them had an absolute QT interval of >500 ms together with an asymptomatic 7-beat run of TdP [98].

Mumford et al. [99] studied the effects of arsenic exposure in 313 residents of Ba Men, Inner Mongolia, who were chronically exposed to arsenic via consumption of water. The residents with a mean arsenic exposure of 15 years were divided into three groups of low ($\leq 21 \mu g/L$), medium (100–300 µg/L), and high (430–690 µg/L) and their ECG evaluated. A direct dose-dependent relationship was observed between As exposure levels and QT interval prolongation (p = 0.001). When categorized into low, medium, and high As exposure groups, individuals displayed an increase in QT interval by 3.9, 11.1, and 20.6%, respectively. In this study, although As-induced QT prolongation was more prevalent in females (p < 0.0001) no correlation was observed with regard to age (p = 0.486) or smoking (p = 0.1018) [99].

The elevated risk of females to As-induced QT prolongation was also reported by Chen et al. [100] in a study evaluating the effects of long-term As exposure from drinking water (0.1–790 μ g/L). Another study, where the authors performed a cross-sectional analysis in elderly men in Boston, MA, between 2000 and 2002 or in 2006 from the Normative Aging Study to analyze associations between toenail arsenic and QT/QTc durations using linear regression, found the interquartile range increase in arsenic concentration was associated with a 3.8 ms increase in QT (95% confidence interval: 0.82, 6.8) and a 2.5 ms increase in QTc (95% confidence interval: 0.11, 4.9) [101].

Wang and colleagues [102] studied 280 men and 355 women living in the endemic area of arsenicosis in southwestern Taiwan to evaluate a correlation between QT intervals with ICH and carotid atherosclerosis. The authors found significant associations of the QTc duration with ischemic heart disease and carotid intima-medium thickness after adjustment for various risk factors in the multiple linear regression analysis (all *p*-values <0.05). Three indices of chronic arsenic exposure were all significantly associated with the risk of QTc prolongation showing dose–response relationships (p < 0.001) in the study population [102].

20.5.2 Ischemic Heart Disease

While QT prolongation is one of the main effects that are observed on cardiomyocytes following chronic arsenic exposure, other cardiac diseases like ischemic heart disease and atherosclerosis too have shown linkage to arsenic. A study in 462 people living in blackfoot disease-hyperendemic villages of Taiwan showed that long-term exposure to As increased the risk
of IHD. IHD was diagnosed by coding the resting electrocardiograms with the Minnesota code. Among the subjects, 78 cases (16.9%) were diagnosed as having IHD. The prevalence rates of IHD for the age groups of 30–39, 40–49, 50–59, and >60 years were 4.9, 7.5, 16.8, and 30.7%, respectively. It was noted that for individuals with a cumulative As exposure (CAE) rating of 0, 0.1–14.9, and $\leq 15 \text{ mg/L-years}$, the prevalence rates of IHD were 5.2, 10.9, and 24.1%, respectively (p < 0.001) [103]. Chen et al. [104] evaluated the mortality rates of residents of 60 villages in Taiwan with endemic arseniasis, from 1973 through 1986. Based on 1,355,915 person-years and 217 IHD deaths, the cumulative IHD mortalities from birth to age 79 years were ranged from 3.4 to 6.6% for residents who lived in villages in which the median arsenic concentrations in drinking water were <0.1 to $\geq 0.6 \text{ mg/L}$, indicating that higher arsenic levels resulted in higher mortality. Furthermore, the group showed that a cohort of 263 patients suffering from blackfoot disease and 2293 non-affected residents from these areas showed a gradient relationship between arsenic exposure and IHD. The relative risk was 2.5 (0.1 to 9.9 mg/L), 4.0 (10.0 to 19.9 mg/L), and 6.5 ($\geq 20 \text{ mg/L}$) compared to zero arsenic exposure. The risk for blackfoot disease patients was much higher than for non-blackfoot disease patients [104].

In Bangladesh, the mortality rate for cardiovascular disease was 214.3 per 100,000 personyears in people drinking water containing <12.0 µg/L arsenic, compared with 271.1 where arsenic levels were \geq 12.0 µg/L. Furthermore, a dose-response relation between exposure to arsenic and mortality from IHD and other heart disease could be established. The hazard ratios in increasing quarters of arsenic concentration were reported to be 1 (0.1–12.0 µg/L), 1.22 (12.1–62.0 µg/L), 1.35 (62.1–148.0 µg/L), and 1.92 (148.1–864.0 µg/L) (p = 0.0019 for trend) [63]. In another study in arsenic endemic villages of Taiwan, a cohort of 74 patients suffering from IHD and 193 age- and sex-matched healthy controls were evaluated for serum levels of micronutrients by HPLC. The study not only demonstrated a significant biological gradient between the risk of IHD and the duration of consuming high-arsenic water but also a reverse dose-response relationship of arsenic-related IHD with serum level of α - and β -carotene, but not for serum levels of retinol, lycopene, and α -tocopherol [105].

These studies clearly show that long-term low dose exposure could have detrimental effects on cardiomyocytes that can alter the heart rhythm and, in association with cardiac vasculature, could be associated with cellular death of cardiac tissue as in IHD. However, most of our understanding on the mechanism of the action of arsenic on myocytes is based on animal studies.

20.5.3 Ion Channels

There have been a large number of studies to understand the mechanism of how arsenic induces changes in signaling cascades in myocytes and causes cellular damage. From the above human studies, it is very clear that one of the effects of arsenic on myocytes is prolonged depolarization of the cardiomyocytes resulting in long QT manifestations. A study in a human cardiomyocyte/fibroblast fusion line (AC16) reported that As exposure resulted in a down-regulation of several ion channel genes including *CACNA1*, *KCNH2*, *KCNQ1*, and *KCNE1* [106]. These genes code for ion channel proteins, which play a pivotal role in maintaining normal cardiac electrophysiology. Yamazaki et al. [96] evaluated the action potentials and isometric

contractions in guinea pig papillary muscles post-ATO perfusion (350 micromol/L). They observed an increase in APD₉₀ from 150 ± 11 to 195 ± 12 ms at 60 mins, p < 0.01. Further, perfusion in low K⁺ with a low stimulation rate augmented the prolongation of APD, and provoked early after-depolarizations. Prolonged exposure to ATO induced muscle contracture, aftercontractions, and triggered activities, which could be partially prevented with tetrodotoxin [96]. Similarly, using conventional action potential recording techniques, Chiang et al. [107] found that ATO dose dependently prolonged APD in guinea pig muscle with slow pacing frequency. Moreover, parenteral as well as intravenous infusions of ATO induced prolonged QT intervals in guinea pig heart, showing direct effects of ATO on cardiac repolarization similar to findings in humans [107]. Similar effects are also observed in rabbits. Wu et al. [108] utilized Langendorff perfusion to determine the direct effects of ATO on electrophysiological changes in rabbit heart after acute and chronic (0.2 mg/kg/day i.v. for 30 days) treatments. While acute treatment did not have any effects on cardiac conduction and repolarization, chronic treatment with 30µM demonstrated polymorphic VT (14%) together with prolonged QTc. Nevertheless, an extremely high dose of 300 µM exposure showed prolonged QTc. It could thus be concluded that extremely high doses such as in suicide cases can result in OTc prolongations but the more common low dose chronic exposure could also alter cardiac rhythms [108].

Changes in the QT interval or cardiac repolarization are mainly dependent on the potassium current across the cellular membranes. *KCNH2*/hERG (ether-go-go-related gene) and *KCNQ1* genes code for the potassium currents I_{Kr} and I_{Ks} respectively, which are solely responsible for maintaining normal QT intervals in cardiomyocytes. Disruption of any of these proteins is one of the major reasons for QT prolongations. Ficker et al. showed that ATO treatment at clinically relevant doses of 0.1–1.5 µM also led to a lack of hERG trafficking to the cellular membrane in ventricular myocytes [109]. During trafficking to the membrane, hERG forms a complex with Hsp70 and Hsp90 [110], which assists in its maturation. Hsp90 consists of a highly conserved thiol pair and since ATO has the ability to modify thiol groups, this chaperone may lose its functionality. Hence, by inhibiting the hERG-chaperone complex, ATO prevented its membrane localization and the consequences were that of a prolonged QT interval.

In a recent study, Shan et al. [111] demonstrated that ATO administration in guinea pigs increases muscle-specific miR-1 and miR-133 (p < 0.001). Overexpression of miR-133 and miR-1 led to QT prolongation as well as down-regulation of hERG (encoded by *KCNH2*) and Kir2.1 (encoded by *KCNJ2*), respectively, which simultaneously led to a decrease in I_{Kr} and I_{K1} currents. However, treatment with the antisense molecules of these miRNAs abrogated the electrophysiological disorders. These studies indicate that arsenic not only directly interacts with hERG protein and affects its function but may also interact with the molecules governing its regulation. Drolet et al. [112] demonstrated that in *KCNH2*- or *KCNQ1*-transfected CHO cells, As₂O₃ caused a concentration-dependent block of both the I_{Kr} and the I_{Ks} current with an IC₅₀ for tail currents block of 0.14 ± 0.01 and 1.13 ± 0.06 for I_{Kr} and I_{Ks} , respectively [112]. Ficker and co-workers in 2004 showed that ATO treatment also induces calcium (Ca²⁺) overload, which leads to an increase in Ca²⁺ currents [109]. Similarly, Sun and colleagues [113] demonstrated that As₂O₃ not only increased QT prolongation in a dose- and time-dependent manner but also enhanced L-type calcium currents (I_{CaL}) and increased intracellular calcium

concentrations in ventricular myocytes. Moreover, the authors also demonstrated that choline treatment could reverse these As_2O_3 -mediated changes [113].

Chen and colleagues [114] recently showed that rats treated with $0.8 \text{ mg/kg/day } \text{As}_2\text{O}_3$, intravenously for 7 d, had prolonged QTc of 36% in the animals and prolonged APDs in isolated ventricular myocytes. Moreover, the authors also demonstrated that As_2O_3 suppressed I_{KI} and shifted the reversal potential to a more positive direction. Furthermore, the treatment significantly augmented Ca^{2+} current densities making the steady state activation curves more negative. However, these changes did not alter the inactivation and reactivation curves of calcium currents. These results together suggest that As_2O_3 perturbs the fine balance for the transmembrane currents that may result in prolongation of APDs or QTc [114]. However, while these results indicate that in general myocytes change in electrophysiological variables, these changes may be species dependent.

Lu and co-workers [115] evaluated the direct electrophysiological effects of As_2O_3 in cardiac tissues isolated from four different species (guinea pig, dog, rabbit, and pig). These authors demonstrated that arsenic (after 30 to 95 min perfusion) at a 10 µM dose caused prolonged APD₉₀, and increased triangulation of the AP and elicited EADs only in guinea pig and dog Purkinje fibers and not in those of rabbit and pig. Furthermore, they did not observe the occurrence of EADs in papillary muscles of guinea pigs and rabbits. This study clearly indicates that the choice of the species as well as the cell type is quite important for performing such studies [115].

20.5.4 Cellular Signaling

Arsenic is well documented to induce ROS-mediated apoptosis in various cell types such as those of liver and brain. However, there is limited information on arsenic's effect on cardiomyocytes. While it would be tempting to speculate that the same action may be seen in cardiomyocytes, there is limited literature to support such claims. Raghu and Cherian showed that when treated with different doses of ATO (30, 60, and 90 µM) for 24, 48, and 72 hours, rat cardiomyocytes underwent apoptosis in a dose- and time-dependent manner [116]. Similarly, Zhao et al. [117] showed that As_2O_3 (2-10 μ M) reduced the viability of H9c2 cardiomyocytes in a dose-dependent manner. They showed that increased ROS and calcium overload with increased caspase-3 activity resulted in cell shrinkage and cellular apoptosis. These changes could be reversed using the caspase-3 inhibitor Ac-DEVD-CHO and antioxidant vitamin E [117]. Additionally, Bessho et al. showed that ATO triggers apoptosis via a ROCK-dependent pathway, since using a ROCK inhibitor (Y-27632) resulted in the protection of H9c2 cardioblastoma cells from caspase-mediated apoptosis [118]. These studies show that ROS-mediated apoptosis may play a pivotal role in cellular regulation of cardiomyocytes. Arsenic-induced cellular death may also play a part in progression of heart failure or IHD. While apoptosis is a key event regulating survival of cardiomyocytes post-arsenic exposure, there are a number of signaling pathways that are altered in this process.

Using whole-genome microarray, Park and Park showed that just 24 hours following ATO treatment, H9c2 cardiomyocytes up-regulated 405 genes and down-regulated 499 genes by more than two-fold indicating the large physiological changes As exposure can induce in a relatively short period of time [119]. In this study, ATO treatment at a subcytotoxic dose (0.5 ppm)

increased ROS generation, which led to oxidative stress as indicated by up-regulation of HO-1, GST, metallothionein (MT), and catalase. Despite ROS generation and oxidative stress, As exposure also induces cardiotoxicity by modulating ion channels [119]. Metallothionein is a well-known endogenous antioxidant that plays a vital role in regulating heavy metals in biological systems [4]. Although modulation of MT is well reported in the literature with reference to cadmium toxicity, not much is known about its role in arsenic poisoning. Miao et al. [120] demonstrated that MT overexpressing H9c2 cardiac cells and transgenic mice (MT-TG), when exposed to clinically relevant doses of arsenic, showed significant protection against arsenicinduced cellular apoptosis as compared to control animals. Moreover, MT overexpressing cells and animals prevented activation of MAPK members such as ERK1/2, JNK, and p38 that were highly up-regulated in control groups [120]. This study clearly indicated that endogenous antioxidants like MT have a major role in regulating signal pathways like MAPK that are important for arsenic-induced cellular death. Recently, Fan et al. [121] showed that arsenic-induced ROS and calcium overload could increase phosphorylated levels of MAPK members like JNK and p38 in an As-induced long QT model. However, using an antioxidant (genistein) these activations could be significantly reduced, which indirectly resulted in suppression of caspase-3 activity, preventing cellular apoptosis. Similarly, Ghosh and co-workers [122] demonstrated that sodium arsenite (NaAsO₂) heightened the phosphorylation of IKK and NF- κ B in rat neonatal myocytes, which was also accompanied by an increase in phosphorylated p38-MAPK and JNK. This study further highlights the involvement of the MAPK signaling pathway in As-induced cardiotoxicity. Interestingly, when using an IKK inhibitor (PS-1145), cleavage of caspase-3 and PARP was abolished. Additionally, a JNK inhibitor (SP600125) and p38-MAPK inhibitor (SB203580) could also decrease phosphorylation of IKK and NF-κB, thereby reducing apoptosis. Furthermore, when animals were treated with a potent antioxidant, taurine, most of the variables demonstrated reversals and prevented apoptosis [122].

Another pathway shown to be affected in cardiomyocytes is TGF- β . Chu and co-workers [123] showed that when ATO was administrated to guinea pigs, they developed prolonged QT and severe myocardial fibrosis. This was accompanied by an increase in TGF- β 1 secretion. Interestingly, it was also found that cardiac fibroblasts (and not cardiomyocytes) were responsible for this increase in TGF- β 1 secretion, which eventually led to the paracrine down-regulation of hERG and Kir2.1—two K⁺ channels required for I_{kr} and I_{kl} currents, respectively. However, through the use of a TGF- β signaling antagonist (LY364947), interstitial fibrosis and LQTS were eliminated. Simultaneously, the abnormal levels of TGF- β 1, hERG, and Kir2.1 returned to normal [123].

Arsenic has also been reported to interfere with redox homeostasis (Nrf2-HO-1 pathway). Nrf2 is a transcription factor responsible for the up-regulation of several antioxidant genes, which include but are not limited to *GST*, *GCL*, *HO-1*, and *MRP*. Sumi and colleagues showed that when compared to cells derived from other sources (e.g., liver; $LC_{25} 25 \mu$ M, kidney; $LC_{25} 112 \mu$ M, and brain; $LC_{25} 49-73 \mu$ M), rat cardiomyocytes ($LC_{25} 4 \mu$ M) were the most sensitive to As^{III} [124]. Interestingly, although Nrf2 was up-regulated in these other cell lines following As^{III} treatment, it was poorly activated in cardiomyocytes. Levels of GCL and MRP were also lower in cardiomyocytes suggesting that they have a reduced ability to metabolize and excrete As.

Furthermore, antioxidants such as resveratrol also possess similar effects as they were shown to prevent the down-regulation of *Nrf2* and *HO-1* in rats administered with ATO [125]. Resveratrol also decreased ROS generation and oxidative DNA damage. These studies show that small molecule inhibitors of various signaling pathways and antioxidants such as genistein and taurine may have potential therapeutic value against As-induced cardiotoxicity.

Looking at the current published data, it is evident that multiple signaling pathways are involved in arsenic-induced cardiotoxicity *in vivo* as well as *in vitro*, and antioxidant therapy has to some extent been able to reverse these effects. It would be hard to state that antioxidants alone would be a complete therapy for treating cardiotoxic effects of arsenic, as these changes may be temporary and the toxic metal may still be present in the cells. However, there have been studies that show that various antioxidants from synthetic sources or plants could have beneficial effects on cardiomyocytes exposed to arsenic.

Zhang et al. showed that matrine and oxymatrine, two compounds that can be isolated from the Sophora genus, could shorten APD duration, which was prolonged due to ATO treatment in both guinea pig ventricular myocytes and a human heterologous HEK overexpression system [126]. However, while low doses (1 μ M and 10 μ M) of matrine led to an increase in I_{Kr} currents, a high dose (100 µM) resulted in a decrease. Following long-term treatment, both these compounds $(1 \mu M)$ could reverse the QT prolongation by up-regulating Sp1 (a transactivator of KCNH2), which in turn led to an increase of hERG on both the transcriptome and proteome levels. This was followed by an increase of hERG on the cell membrane and with it a simultaneous increase in I_{kr} currents. Further, salvianolic acid B (Sal B), a component of Salviae miltiorrhizae, has also been reported to protect against ATO-induced apoptosis in H9c2 cardiomyocytes via PI3K/Akt, a signaling pathway known for exerting cardio-protective effects [127]. The PI3K/Akt signaling pathway is known to confer a protective effect by modulating the balance between anti-apoptotic proteins (Bcl-2 and Bcl-xL) and pro-apoptotic proteins (Bax, Bad, and Bak). Sturlan et al. showed that ATO-induced apoptosis occurs due to the down-regulation of Bcl-2 and simultaneous up-regulation of Bax [128]. Treatment with Sal B increased Bcl-2 and Bcl-xL and returned Bax to normal levels. However, when using a PI3K inhibitor (LY294002), the protective effect of Sal B was lost and cardiomyocytes underwent apoptosis following ATO treatment [127].

In another study, silibinin, an extract of milk thistle seeds, has also been shown to ameliorate As-induced ROS generation in rats [129]. This compound had the ability to recover the activities of heart mitochondrial enzymes (ICDH, SDH, MDH, a-KDH, and NADH) which were lost following As exposure. Leaves from *Corchorus olitorius*, a popular crop in Bangladesh and West Bengal, are another form of protection against As-induced cardiotoxicity [72]. Rats administered with NaAsO₂ developed significant oxidative stress in their myocardial tissue. This was signified by the down-regulation of several antioxidant enzymes such as SOD, CAT, GST, and GPx. However, treating these rats with aqueous extract of *C. olitorius* leaves 15 days prior to NaAsO₂ administration protected their cardiac tissue from As-induced cardiotoxicity by maintaining the activities of the aforementioned enzymes. Ethanolic extract of *Boerhavia diffusa* (BDE) was also found to protect H9c2 myoblasts against ATO-induced cardiotoxicity [130]. In this study, ATO treatment at varying concentrations (5, 7.5, and 10 μ M) reduced the integrity of mitochondria, initiated ER stress, disrupted cytoskeletal networks, inhibited antioxidant enzymes, and increased both ROS generation and Ca^{2+} load. However, BDE could reverse the pathophysiological properties of ATO-treated cells par at high doses (10 μ M).

20.5.5 Chelation Therapy

Chelating agents are the mainstay for heavy metal poisoning. Chelating agents have been reported to reduce arsenic poisoning in small animals [3,131–137]. These agents bind to As and prevent it from interacting with other molecules and enzymes within the body. Simultaneously upon interaction with chelating agents, As is converted into a water-soluble form, which can be excreted more readily [138]. Kathirgamanathan et al. reported that progressive, multilevel cardiac conduction block as a result of ATO treatment could be rectified through chelation treatment with dimercaprol (British anti-Lewisite/BAL) [139]. Further, Kumazaki et al. showed that treatment with BAL and α -lipoic acid (LA) could reverse ATO-mediated QT prolongation in guinea pigs [140]. Chelation treatment, however, generally results in toxicity and side effects such as nausea, diarrhea, and the appearances of rashes, and even death can occur. It has therefore been reported that chelation treatment together with antioxidant supplementation may provide a safer alternative [141].

20.6 Human Pluripotent Stem Cells: Understanding Arsenic Toxicity

While animal models and human heterologous overexpression systems have provided basic understanding with regard to mechanisms associated with As-induced cardiotoxicity, the true cause of heart injury still eludes us due to the lack of an ideal human model. Human embryonic stem cells (ESCs) and induced pluripotent stem cells (iPSCs), through defined culture conditions, can be differentiated into various cell types, including cardiomyocytes [142]. Therefore, human ES- and iPSC-derived cardiomyocytes provide a more suitable alternative to animal models. Further, unlike human heterologous overexpression systems, which only express one ion channel, pluripotent stem cell (PSC)-derived cardiomyocytes concurrently express all ion channels in a similar fashion as in the human heart. Human PSC-derived cardiomyocytes are hence an ideal model for studying the effects of As exposure with regard to ion channel dysregulation. Additionally, since human PSCs can be cultured indefinitely, a large number of cardiomyocytes can be generated at any given time. This would be ideal for studying other mechanisms and signal transduction pathways involved in As-induced cardiotoxicity. However, reports have shown that there are inherent differences within various pluripotent stem cell lines that may affect their differentiation abilities to various lineages [143]. However, having efficient differentiation protocols could resolve these inherent differences [142,144].

In a proof of principle study, Mehta and colleagues [145] utilized this model to study development toxicity. In the study, human embryonic stem cell-derived embryoid bodies (EBs) were generated that mimic human embryonic development and these EBs were exposed to various drugs to test their developmental toxicity effects. Our results suggested that using this sensitive method, one could evaluate and classify compounds as non-, weakly, and strongly toxic. This study demonstrated the potential application of this technology in drug toxicity screening [145]. Based on this model, Flora and Mehta [146] studied the developmental toxicity of arsenic. In the study, human embryonic stem cell-derived embryoid bodies (EBs) were exposed to various concentrations of arsenic. These arsenic exposed EBs were found to be highly sensitive to arsenic treatment in comparison to fibroblasts with IC_{50} values of 0.005 and 0.31 µg/mL, respectively. Gene expression analysis of markers indicative of cardiovascular lineage development such as T, Gata4, actinin (cardiac and muscle), myosin light chain, and CD31 (endothelial cell adhesion) showed significant down-regulation in a dose-dependent fashion. Moreover, genes involved in cell cycle regulation (CCND1, CCRK), apoptosis (Bcl-2, CASP3), methylation (DNMT3B), signaling pathways (BMPs, FGF, NODAL, NOTCH, WNT), and gap junction (GJA1) also showed significant alterations post-arsenic treatment [146]. These results as a proof of principle show that hESC-derived EBs mimic similar effects as reported by previous studies on different human cell types [146]. This technology provides a method through which researchers could gain access to human cell types like cardiomyocytes or neurons that are impossible to be collected from humans.

The emergence of human-induced pluripotent stem cell technology provides a new dimension to evaluate various diseases. Induced pluripotent stem cells are similar to human embryonic stem cells in almost all aspects, except that they are generated by reprogramming of fibroblasts with transcriptional factors (Oct4, Sox2, klf4, c-myc) as previously demonstrated by Takahashi and Yamanaka in 2006 [147]. We and others have previously demonstrated that fibroblast cells could be reprogrammed to generate hiPSCs from various sources and these human iPSCs can be differentiated into cardiomyocytes [148–150]. Moreover, these cardiomyocytes respond appropriately to anti-arrhythmic drugs in a similar fashion to human cardiac cells [151], suggesting that these iPSC-derived cells mimic human cardiomyocytes not only structurally but also functionally. However, the one limitation seen in these myocytes is that they are immature and resemble the shape of a 16-week-old fetus. Having said this, these myocytes have still been able to unravel mechanisms of various diseases. For instance, we recently demonstrated that fibroblasts isolated from a patient with long QT syndrome 2 (mutation in KCNH2 gene) could be reprogrammed to hiPSC state and subsequently differentiated into patient-specific cardiomyocytes, which manifest long QT intervals as revealed by multielectrode array analysis (in vitro system similar to ECG) as well as by whole-cell patch clamp studies [152]. Moreover, we demonstrated that QT prolongation was due to reduced I_{Kr} currents in these patient-specific cardiomyocytes and there was a trafficking defect due to which mutated hERG could not be transported to the membrane, thus reducing the overall presence of functional hERG on the membrane of these diseased myocytes [153,154]. Such studies to understand pathways or mechanisms of disease manifestation, which were previously not possible, are easily possible today. The use of such technologies may provide better understanding of disease pathogenesis and since they are being applied in humanized models they may show a better clinical relationship when translated into clinical setups.

20.7 Conclusions

In conclusion, inorganic and methylated forms of arsenic have detrimental effects on the cardiovascular system. While most of these harmful effects are a result of oxidative stress-induced cell death, studies have also shown that arsenic has the ability to bind to certain macromolecules like hemoglobin, as well as various proteins in the cardiovascular system, thereby interfering with their biological function. Further, these molecular associations, in turn, could lead to a complete disarray of signaling cascades that alter various biochemical pathways within a cellular system. Although a large number of clinical studies have shown a significant correlation between arsenic exposure and indices of various cardiovascular diseases such as hypertension, atherosclerosis, and IHD, these aspects have not been critically evaluated or supported by compelling evidence. Perhaps one of the reasons for this could be the fact that all these diseases are multifactorial and are not linked to a single gene. Moreover, due to disparate mechanisms, the differential response exhibited by cell types to arsenic, together with genetic polymorphisms, complicates the understanding of its toxicity. On the other hand, by directly or indirectly interacting with hERG (the protein that regulates the critical I_{Kr} current in cardiomyocytes), arsenic has been shown to alter normal heart rhythm (prolongation of QT intervals). Recently, increasing numbers of studies have been published identifying mechanisms, biochemical pathways, and potential candidate molecules that are perturbed during chronic arsenicosis. Most of these studies have been aimed at developing better and more accurate diagnostic tools for arsenicosis. In general, comprehensive studies accurately dissecting the changes in signaling and biochemical pathways as a result of arsenic exposure are yet to be reported. Such studies could yield a better understanding of how arsenic impairs the cardiovascular system. Moreover, newer technologies like pluripotent stem cells could provide a better opportunity to study the effects of arsenic in a cell type-specific manner. The development of efficient and accurate tools is pivotal to providing a solution for millions exposed to arsenic on a daily basis.

References

- Moon K, Guallar E, Navas-Acien A. Arsenic exposure and cardiovascular disease: an updated systematic review. Curr Atheroscler Rep 2012;14(6):542–55.
- [2] Hughes MF, Beck BD, Chen Y, Lewis AS, Thomas DJ. Arsenic exposure and toxicology: a historical perspective. Toxicol Sci 2011;123(2):305–32.
- [3] Flora SJ. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51(2):257–81.
- [4] Mehta A, Flora SJ. Possible role of metal redistribution, hepatotoxicity and oxidative stress in chelating agents induced hepatic and renal metallothionein in rats. Food Chem Toxicol 2001;39(10):1029–38.
- [5] Guha Mazumder DN, Ghoshal UC, Saha J, Santra A, De BK, Chatterjee A, et al. Randomized placebocontrolled trial of 2,3-dimercaptosuccinic acid in therapy of chronic arsenicosis due to drinking arseniccontaminated subsoil water. J Toxicol Clin Toxicol 1998;36(7):683–90.
- [6] Pande M, Mehta A, Pant BP, Flora SJ. Combined administration of a chelating agent and an antioxidant in the prevention and treatment of acute lead intoxication in rats. Environ Toxicol Pharmacol 2001;9(4):173-84.

- [7] Mehta A, Kannan GM, Dube SN, Pant BP, Pant SC, Flora SJ. Haematological, hepatic and renal alterations after repeated oral or intraperitoneal administration of monoisoamyl DMSA. I. Changes in male rats. J Appl Toxicol 2002;22(6):359–69.
- [8] Flora SJ, Mehta A. Haematological, hepatic and renal alterations after repeated oral and intraperitoneal administration of monoisoamyl DMSA. II. Changes in female rats. J Appl Toxicol 2003;23(2):97–102.
- [9] Mehta A, Pant SC, Flora SJ. Monoisoamyl dimercaptosuccinic acid induced changes in pregnant female rats during late gestation and lactation. Reprod Toxicol 2006;21(1):94–103.
- [10] Flora SJ, Mehta A, Gautam P, Jatav PC, Pathak U. Essential metal status, prooxidant/antioxidant effects of MiADMSA in male rats: age-related effects. Biol Trace Elem Res 2007;120(1-3):235-47.
- [11] Mondal B, Chatterjee D, Bhattacharyya M. Structure-function alteration of hemoglobin in arsenicosis patients: a probable pathway to exert toxicity. J Appl Toxicol 2012;32(8):581–9.
- [12] Lu M, Wang H, Li XF, Lu X, Cullen WR, Arnold LL, et al. Evidence of hemoglobin binding to arsenic as a basis for the accumulation of arsenic in rat blood. Chem Res Toxicol 2004;17(12):1733–42.
- [13] Breton CV, Houseman EA, Kile ML, Quamruzzaman Q, Rahman M, Mahiuddin G, et al., Gender-specific protective effect of hemoglobin on arsenic-induced skin lesions. Cancer Epidemiol Biomarkers Prev 2006;15(5):902–7.
- [14] Kannan GM, Tripathi N, Dube SN, Gupta M, Flora SJ. Toxic effects of arsenic (III) on some hematopoietic and central nervous system variables in rats and guinea pigs. J Toxicol Clin Toxicol 2001;39(7):675–82.
- [15] Flora SJS. Nutritional components modify metal absorption, toxic response and chelation therapy. J Nutr Environ Med 2002;12(1):53–67.
- [16] Flora SJ, Kumar P. Biochemical and immunotoxicological alterations following repeated gallium arsenide exposure and their recoveries by meso-2,3-dimercaptosuccinic acid and 2,3-dimercaptopropane 1-sulfonate administration in rats. Environ Toxicol Pharmacol 1996;2(4):315–20.
- [17] Bhadauria S, Flora SJ. Arsenic induced inhibition of delta-aminolevulinate dehydratase activity in rat blood and its response to meso 2,3-dimercaptosuccinic acid and monoisoamyl DMSA. Biomed Environ Sci 2004;17(1):101–8.
- [18] Flora SJ. Arsenic-induced oxidative stress and its reversibility following combined administration of N-acetylcysteine and meso 2,3-dimercaptosuccinic acid in rats. Clin Exp Pharmacol Physiol 1999;26(11):865–9.
- [19] Thomas DJ, Styblo M, Lin S. The cellular metabolism and systemic toxicity of arsenic. Toxicol Appl Pharmacol 2001;176(2):127-44.
- [20] Yildiz D, Cakir Y. Efflux of glutathione and glutathione complexes from human erythrocytes in response to inorganic arsenic exposure. Biol Trace Elem Res 2012;150(1–3):451–9.
- [21] Hall MN, Niedzwiecki M, Liu X, Harper KN, Alam S, Slavkovich V, et al. Chronic arsenic exposure and blood glutathione and glutathione disulfide concentrations in bangladeshi adults. Environ Health Perspect 2013;121(9):1068–74.
- [22] Mishra D, Flora SJ. Differential oxidative stress and DNA damage in rat brain regions and blood following chronic arsenic exposure. Toxicol Ind Health 2008;24(4):247–56.
- [23] Flora SJ, Bhadauria S, Pant SC, Dhaked RK. Arsenic induced blood and brain oxidative stress and its response to some thiol chelators in rats. Life Sci 2005;77(18):2324–37.
- [24] Winski SL, Carter DE. Arsenate toxicity in human erythrocytes: characterization of morphologic changes and determination of the mechanism of damage. J Toxicol Environ Health A 1998;53(5):345–55.
- [25] Biswas D, Banerjee M, Sen G, Das JK, Banerjee A, Sau TJ, et al. Mechanism of erythrocyte death in human population exposed to arsenic through drinking water. Toxicol Appl Pharmacol 2008;230(1):57–66.
- [26] Biswas D, Sen G, Biswas T. Reduced cellular redox status induces 4-hydroxynonenal-mediated caspase 3 activation leading to erythrocyte death during chronic arsenic exposure in rats. Toxicol Appl Pharmacol 2010;244(3):315–27.

- [27] Winski SL, Barber DS, Rael LT, Carter DE. Sequence of toxic events in arsine-induced hemolysis in vitro: implications for the mechanism of toxicity in human erythrocytes. Fundam Appl Toxicol 1997;38(2):123–8.
- [28] Biswas D, Sen G, Sarkar A, Biswas T. Atorvastatin acts synergistically with N-acetyl cysteine to provide therapeutic advantage against Fas-activated erythrocyte apoptosis during chronic arsenic exposure in rats. Toxicol Appl Pharmacol 2011;250(1):39–53.
- [29] Flora SJ, Chouhan S, Kannan GM, Mittal M, Swarnkar H. Combined administration of taurine and monoisoamyl DMSA protects arsenic induced oxidative injury in rats. Oxid Med Cell Longev 2008;1(1):39-45.
- [30] Prabu SM, Sumedha NC. Ameliorative effect of diallyl trisulphide on arsenic-induced oxidative stress in rat erythrocytes and DNA damage in lymphocytes. J Basic Clin Physiol Pharmacol 2013;(Oct):1–17.
- [31] Caciari T, Capozzella A, Tomei F, Nieto HA, De Sio S, Montuori L, et al. Arsenic and peripheral blood count in workers exposed to urban stressors. Clin Ter 2012;163(5):e293–302.
- [32] Chou WC, Jie C, Kenedy AA, Jones RJ, Trush MA, Dang CV. Role of NADPH oxidase in arsenic-induced reactive oxygen species formation and cytotoxicity in myeloid leukemia cells. Proc Natl Acad Sci USA 2004;101(13):4578–83.
- [33] Andrew AS, Jewell DA, Mason RA, Whitfield ML, Moore JH, Karagas MR. Drinking-water arsenic exposure modulates gene expression in human lymphocytes from a U.S. population. Environ Health Perspect 2008;116(4):524–31.
- [34] Banerjee N, Banerjee M, Ganguly S, Bandyopadhyay S, Das JK, Bandyopadhay A, et al. Arsenic-induced mitochondrial instability leading to programmed cell death in the exposed individuals. Toxicology 2008;246(2-3):101-11.
- [35] Avani G, Rao MV. Genotoxic effects in human lymphocytes exposed to arsenic and vitamin A. Toxicol In Vitro 2007;21(4):626–31.
- [36] Pant HH, Rao MV. Evaluation of in vitro anti-genotoxic potential of melatonin against arsenic and fluoride in human blood cultures. Ecotoxicol Environ Saf 2010;73(6):1333–7.
- [37] Sharma M, Sharma S, Arora M, Kaul D. Regulation of cellular Cyclin D1 gene by arsenic is mediated through miR-2909. Gene 2013;522(1):60–4.
- [38] Gupta S, Yel L, Kim D, Kim C, Chiplunkar S, Gollapudi S. Arsenic trioxide induces apoptosis in peripheral blood T lymphocyte subsets by inducing oxidative stress: a role of Bcl-2. Mol Cancer Ther 2003;2(8):711–9.
- [39] Yu HS, Liao WT, Chang KL, Yu CL, Chen GS. Arsenic induces tumor necrosis factor alpha release and tumor necrosis factor receptor 1 signaling in T helper cell apoptosis. J Invest Dermatol 2002;119(4):812–19.
- [40] Vega L, Montes de Oca P, Saavedra R, Ostrosky-Wegman P. Helper T cell subpopulations from women are more susceptible to the toxic effect of sodium arsenite in vitro. Toxicology 2004;199(2-3):121–8.
- [41] Morzadec C, Bouezzedine F, Macoch M, Fardel O, Vernhet L. Inorganic arsenic impairs proliferation and cytokine expression in human primary T lymphocytes. Toxicology 2012;300(1–2):46–56.
- [42] Martin-Chouly C, Morzadec C, Bonvalet M, Galibert MD, Fardel O, Vernhet L. Inorganic arsenic alters expression of immune and stress response genes in activated primary human T lymphocytes. Mol Immunol 2011;48(6-7):956-65.
- [43] Sakurai T, Kaise T, Matsubara C. Inorganic and methylated arsenic compounds induce cell death in murine macrophages via different mechanisms. Chem Res Toxicol 1998;11(4):273–83.
- [44] Lemarie A, Bourdonnay E, Morzadec C, Fardel O, Vernhet L. Inorganic arsenic activates reduced NADPH oxidase in human primary macrophages through a Rho kinase/p38 kinase pathway. J Immunol 2008;180(9):6010–17.
- [45] Banerjee N, Banerjee S, Sen R, Bandyopadhyay A, Sarma N, Majumder P, et al. Chronic arsenic exposure impairs macrophage functions in the exposed individuals. J Clin Immunol 2009;29(5):582–94.
- [46] Bourdonnay E, Morzadec C, Sparfel L, Galibert MD, Jouneau S, Martin-Chouly C, et al. Global effects of inorganic arsenic on gene expression profile in human macrophages. Mol Immunol 2009;46(4):649–56.

- [47] Bourdonnay E, Morzadec C, Fardel O, Vernhet L. Redox-sensitive regulation of gene expression in human primary macrophages exposed to inorganic arsenic. J Cell Biochem 2009;107(3):537–47.
- [48] Wang L, Weng CY, Wang YJ, Wu MJ. Lipoic acid ameliorates arsenic trioxide-induced HO-1 expression and oxidative stress in THP-1 monocytes and macrophages. Chem Biol Interact 2011;190(2-3):129–38.
- [49] Colognato R, Coppede F, Ponti J, Sabbioni E, Migliore L. Genotoxicity induced by arsenic compounds in peripheral human lymphocytes analysed by cytokinesis-block micronucleus assay. Mutagenesis 2007;22(4):255-61.
- [50] Luna AL, Acosta-Saavedra LC, Lopez-Carrillo L, Conde P, Vera E, De Vizcaya-Ruiz A, et al. Arsenic alters monocyte superoxide anion and nitric oxide production in environmentally exposed children. Toxicol Appl Pharmacol 2010;245(2):244–51.
- [51] Ostrosky-Wegman P, Gonsebatt ME, Montero R, Vega L, Barba H, Espinosa J, et al. Lymphocyte proliferation kinetics and genotoxic findings in a pilot study on individuals chronically exposed to arsenic in Mexico. Mutat Res 1991;250(1–2):477–82.
- [52] Mahata J, Ghosh P, Sarkar JN, Ray K, Natarajan AT, Giri AK. Effect of sodium arsenite on peripheral lymphocytes in vitro: individual susceptibility among a population exposed to arsenic through the drinking water. Mutagenesis 2004;19(3):223–9.
- [53] Basu A, Som A, Ghoshal S, Mondal L, Chaubey RC, Bhilwade HN, et al. Assessment of DNA damage in peripheral blood lymphocytes of individuals susceptible to arsenic induced toxicity in West Bengal, India. Toxicol Lett 2005;159(1):100–12.
- [54] Li H, Engstrom K, Vahter M, Broberg K. Arsenic exposure through drinking water is associated with longer telomeres in peripheral blood. Chem Res Toxicol 2012;25(11):2333–9.
- [55] Harari F, Engstrom K, Concha G, Colque G, Vahter M, Broberg K. N-6-adenine-specific DNA methyltransferase 1 (N6AMT1) polymorphisms and arsenic methylation in Andean women. Environ Health Perspect 2013;121(7):797–803.
- [56] Chervona Y, Hall MN, Arita A, Wu F, Sun H, Tseng HC, et al. Associations between arsenic exposure and global posttranslational histone modifications among adults in Bangladesh. Cancer Epidemiol Biomarkers Prev 2012;21(12):2252–60.
- [57] Engstrom K, Vahter M, Mlakar SJ, Concha G, Nermell B, Raqib R, et al. Polymorphisms in arsenic (+III oxidation state) methyltransferase (AS3MT) predict gene expression of AS3MT as well as arsenic metabolism. Environ Health Perspect 2011;119(2):182–8.
- [58] Hernandez A, Paiva L, Creus A, Quinteros D, Marcos R. Micronucleus frequency in copper-mine workers exposed to arsenic is modulated by the AS3MT Met287Thr polymorphism. Mutat Res 2014;759:51–5.
- [59] Gribble MO, Tang WY, Shang Y, Pollak J, Umans JG, Francesconi KA, et al. Differential methylation of the arsenic (III) methyltransferase promoter according to arsenic exposure. Arch Toxicol 2014;88(2):275–82.
- [60] Wang YH, Wu MM, Hong CT, Lien LM, Hsieh YC, Tseng HP, et al. Effects of arsenic exposure and genetic polymorphisms of p53, glutathione S-transferase M1, T1, and P1 on the risk of carotid atherosclerosis in Taiwan. Atherosclerosis 2007;192(2):305–12.
- [61] Hsieh YC, Hsieh FI, Lien LM, Chou YL, Chiou HY, Chen CJ. Risk of carotid atherosclerosis associated with genetic polymorphisms of apolipoprotein E and inflammatory genes among arsenic exposed residents in Taiwan. Toxicol Appl Pharmacol 2008;227(1):1–7.
- [62] Huang YK, Tseng CH, Huang YL, Yang MH, Chen CJ, Hsueh YM. Arsenic methylation capability and hypertension risk in subjects living in arseniasis-hyperendemic areas in southwestern Taiwan. Toxicol Appl Pharmacol 2007;218(2):135–42.
- [63] Chen Y, Graziano JH, Parvez F, Liu M, Slavkovich V, Kalra T, et al. Arsenic exposure from drinking water and mortality from cardiovascular disease in Bangladesh: prospective cohort study. BMJ 2011;342:d2431.
- [64] Medrano MA, Boix R, Pastor-Barriuso R, Palau M, Damian J, Ramis R, et al. Arsenic in public water supplies and cardiovascular mortality in Spain. Environ Res 2010;110(5):448–54.

- [65] Tseng CH, Chong CK, Chen CJ, Tai TY. Dose-response relationship between peripheral vascular disease and ingested inorganic arsenic among residents in blackfoot disease endemic villages in Taiwan. Atherosclerosis 1996;120(1-2):125-33.
- [66] Wang CH, Jeng JS, Yip PK, Chen CL, Hsu LI, Hsueh YM, et al. Biological gradient between long-term arsenic exposure and carotid atherosclerosis. Circulation 2002;105(15):1804–9.
- [67] Pi J, Yamauchi H, Kumagai Y, Sun G, Yoshida T, Aikawa H, et al. Evidence for induction of oxidative stress caused by chronic exposure of Chinese residents to arsenic contained in drinking water. Environ Health Perspect 2002;110(4):331-6.
- [68] Hessler JR, Morel DW, Lewis LJ, Chisolm GM. Lipoprotein oxidation and lipoprotein-induced cytotoxicity. Arteriosclerosis 1983;3(3):215–22.
- [69] Hossain E, Ota A, Karnan S, Damdindorj L, Takahashi M, Konishi Y, et al. Arsenic augments the uptake of oxidized LDL by upregulating the expression of lectin-like oxidized LDL receptor in mouse aortic endothelial cells. Toxicol Appl Pharmacol 2013;273(3):651–8.
- [70] Tsai SH, Hsieh MS, Chen L, Liang YC, Lin JK, Lin SY. Suppression of Fas ligand expression on endothelial cells by arsenite through reactive oxygen species. Toxicol Lett 2001;123(1):11–19.
- [71] Gong G, O'Bryant SE. Low-level arsenic exposure, AS3MT gene polymorphism and cardiovascular diseases in rural Texas counties. Environ Res 2012;113:52–7.
- [72] Das AK, Sahu R, Dua TK, Bag S, Gangopadhyay M, Sinha MK, et al. Arsenic-induced myocardial injury: protective role of *Corchorus olitorius* leaves. Food Chem Toxicol 2010;48(5):1210–17.
- [73] Hossain E, Ota A, Takahashi M, Karnan S, Damdindorj L, Konishi Y, et al. Arsenic upregulates the expression of angiotensin II Type I receptor in mouse aortic endothelial cells. Toxicol Lett 2013;220(1):70–5.
- [74] Hsieh YC, Lien LM, Chung WT, Hsieh FI, Hsieh PF, Wu MM, et al. Significantly increased risk of carotid atherosclerosis with arsenic exposure and polymorphisms in arsenic metabolism genes. Environ Res 2011;111(6):804–10.
- [75] Wu MM, Chiou HY, Chen CL, Wang YH, Hsieh YC, Lien LM, et al. GT-repeat polymorphism in the heme oxygenase-1 gene promoter is associated with cardiovascular mortality risk in an arsenic-exposed population in northeastern Taiwan. Toxicol Appl Pharmacol 2010;248(3):226–33.
- [76] Wu MM, Chiou HY, Chen CL, Hsu LI, Lien LM, Wang CH, et al. Association of heme oxygenase-1 GT-repeat polymorphism with blood pressure phenotypes and its relevance to future cardiovascular mortality risk: an observation based on arsenic-exposed individuals. Atherosclerosis 2011;219(2):704–8.
- [77] Yeh JY, Cheng LC, Liang YC, Ou BR. Modulation of the arsenic effects on cytotoxicity, viability, and cell cycle in porcine endothelial cells by selenium. Endothelium 2003;10(3):127–39.
- [78] Nuntharatanapong N, Chen K, Sinhaseni P, Keaney Jr. JF. EGF receptor-dependent JNK activation is involved in arsenite-induced p21Cip1/Waf1 upregulation and endothelial apoptosis. Am J Physiol Heart Circ Physiol 2005;289(1):H99-H107.
- [79] Tsou TC, Yeh SC, Tsai EM, Tsai FY, Chao HR, Chang LW. Arsenite enhances tumor necrosis factor-alphainduced expression of vascular cell adhesion molecule-1. Toxicol Appl Pharmacol 2005;209(1):10–18.
- [80] Liu Y, Cai B, Wang G, Liu J. One potential mechanism of cardiac toxicity induced by arsenic trioxide: apoptosis in coronary arterial and blood capillary endothelial cells. Med Hypotheses 2007;69(6):1383–4.
- [81] Wang L, Kou MC, Weng CY, Hu LW, Wang YJ, Wu MJ. Arsenic modulates heme oxygenase-1, interleukin-6, and vascular endothelial growth factor expression in endothelial cells: roles of ROS, NF-kappaB, and MAPK pathways. Arch Toxicol 2012;86(6):879–96.
- [82] Soucy NV, Mayka D, Klei LR, Nemec AA, Bauer JA, Barchowsky A. Neovascularization and angiogenic gene expression following chronic arsenic exposure in mice. Cardiovasc Toxicol 2005;5(1):29–41.
- [83] Bao L, Shi H. Arsenite induces endothelial cell permeability increase through a reactive oxygen speciesvascular endothelial growth factor pathway. Chem Res Toxicol 2010;23(11):1726–34.

- [84] Roboz GJ, Dias S, Lam G, Lane WJ, Soignet SL, Warrell Jr. RP, et al. Arsenic trioxide induces dose- and time-dependent apoptosis of endothelium and may exert an antileukemic effect via inhibition of angiogenesis. Blood 2000;96(4):1525–30.
- [85] Barchowsky A, Dudek EJ, Treadwell MD, Wetterhahn KE. Arsenic induces oxidant stress and NF-kappa B activation in cultured aortic endothelial cells. Free Radic Biol Med 1996;21(6):783–90.
- [86] Chen SC, Tsai MH, Wang HJ, Yu HS, Chang LW. Involvement of substance P and neurogenic inflammation in arsenic-induced early vascular dysfunction. Toxicol Sci 2007;95(1):82–8.
- [87] Bae ON, Lim EK, Lim KM, Noh JY, Chung SM, Lee MY, et al. Vascular smooth muscle dysfunction induced by monomethylarsonous acid (MMA III): a contributing factor to arsenic-associated cardiovascular diseases. Environ Res 2008;108(3):300–8.
- [88] Simeonova PP, Hulderman T, Harki D, Luster MI. Arsenic exposure accelerates atherogenesis in apolipoprotein E(-/-) mice. Environ Health Perspect 2003;111(14):1744–8.
- [89] Srivastava S, D'Souza SE, Sen U, States JC. In utero arsenic exposure induces early onset of atherosclerosis in ApoE -/- mice. Reprod Toxicol 2007;23(3):449-56.
- [90] Roden DM. Drug-induced prolongation of the QT interval. N Engl J Med 2004;350(10):1013-22.
- [91] Barbey JT, Soignet S. Prolongation of the QT interval and ventricular tachycardia in patients treated with arsenic trioxide for acute promyelocytic leukemia. Ann Intern Med 2001;135(9):842–3.
- [92] St Petery J, Gross C, Victorica BE. Ventricular fibrillation caused by arsenic poisoning. Am J Dis Child 1970;120(4):367–71.
- [93] Unnikrishnan D, Dutcher JP, Varshneya N, Lucariello R, Api M, Garl S, et al. Torsades de pointes in 3 patients with leukemia treated with arsenic trioxide. Blood 2001;97(5):1514–16.
- [94] Little RE, Kay GN, Cavender JB, Epstein AE, Plumb VJ. Torsade de pointes and T-U wave alternans associated with arsenic poisoning. Pacing Clin Electrophysiol 1990;13(2):164–70.
- [95] Barbey JT, Pezzullo JC, Soignet SL. Effect of arsenic trioxide on QT interval in patients with advanced malignancies. J Clin Oncol 2003;21(19):3609–15.
- [96] Yamazaki K, Terada H, Satoh H, Naito K, Takeshita A, Uehara A, et al. Arrhythmogenic effects of arsenic trioxide in patients with acute promyelocytic leukemia and an electrophysiological study in isolated guinea pig papillary muscles. Circ J 2006;70(11):1407–14.
- [97] Ohnishi K, Yoshida H, Shigeno K, Nakamura S, Fujisawa S, Naito K, et al. Prolongation of the QT interval and ventricular tachycardia in patients treated with arsenic trioxide for acute promyelocytic leukemia. Ann Intern Med 2000;133(11):881–5.
- [98] Soignet SL, Frankel SR, Douer D, Tallman MS, Kantarjian H, Calleja E, et al. United States multicenter study of arsenic trioxide in relapsed acute promyelocytic leukemia. J Clin Oncol 2001;19(18):3852–60.
- [99] Mumford JL, Wu K, Xia Y, Kwok R, Yang Z, Foster J, et al. Chronic arsenic exposure and cardiac repolarization abnormalities with QT interval prolongation in a population-based study. Environ Health Perspect 2007;115(5):690–4.
- [100] Chen Y, Wu F, Parvez F, Ahmed A, Eunus M, McClintock TR, et al. Arsenic exposure from drinking water and QT-interval prolongation: results from the Health Effects of Arsenic Longitudinal Study. Environ Health Perspect 2013;121(4):427–32, 432e1–7.
- [101] Mordukhovich I, Wright RO, Amarasiriwardena C, Baja E, Baccarelli A, Suh H, et al. Association between low-level environmental arsenic exposure and QT interval duration in a general population study. Am J Epidemiol 2009;170(6):739–46.
- [102] Wang CH, Chen CL, Hsiao CK, Chiang FT, Hsu LI, Chiou HY, et al. Increased risk of QT prolongation associated with atherosclerotic diseases in arseniasis-endemic area in southwestern coast of Taiwan. Toxicol Appl Pharmacol 2009;239(3):320–4.
- [103] Tseng CH, Chong CK, Tseng CP, Hsueh YM, Chiou HY, Tseng CC, et al. Long-term arsenic exposure and ischemic heart disease in arseniasis-hyperendemic villages in Taiwan. Toxicol Lett 2003;137(1-2): 15–21.

- [104] Chen CJ, Chiou HY, Chiang MH, Lin LJ, Tai TY. Dose-response relationship between ischemic heart disease mortality and long-term arsenic exposure. Arterioscler Thromb Vasc Biol 1996;16(4):504–10.
- [105] Hsueh YM, Wu WL, Huang YL, Chiou HY, Tseng CH, Chen CJ. Low serum carotene level and increased risk of ischemic heart disease related to long-term arsenic exposure. Atherosclerosis 1998;141(2):249–57.
- [106] Mo J, Xia Y, Wade TJ, DeMarini DM, Davidson M, Mumford J. Altered gene expression by low-dose arsenic exposure in humans and cultured cardiomyocytes: assessment by real-time PCR arrays. Int J Environ Res Public Health 2011;8(6):2090–108.
- [107] Chiang CE, Luk HN, Wang TM, Ding PY. Prolongation of cardiac repolarization by arsenic trioxide. Blood 2002;100(6):2249–52.
- [108] Wu MH, Lin CJ, Chen CL, Su MJ, Sun SS, Cheng AL. Direct cardiac effects of As2O3 in rabbits: evidence of reversible chronic toxicity and tissue accumulation of arsenicals after parenteral administration. Toxicol Appl Pharmacol 2003;189(3):214–20.
- [109] Ficker E, Kuryshev YA, Dennis AT, Obejero-Paz C, Wang L, Hawryluk P, et al. Mechanisms of arsenicinduced prolongation of cardiac repolarization. Mol Pharmacol 2004;66(1):33–44.
- [110] Ficker E, Dennis AT, Wang L, Brown AM. Role of the cytosolic chaperones Hsp70 and Hsp90 in maturation of the cardiac potassium channel HERG. Circ Res 2003;92(12):e87–100.
- [111] Shan H, Zhang Y, Cai B, Chen X, Fan Y, Yang L, et al. Upregulation of microRNA-1 and microRNA-133 contributes to arsenic-induced cardiac electrical remodeling. Int J Cardiol 2013;167(6):2798–805.
- [112] Drolet B, Simard C, Roden DM. Unusual effects of a QT-prolonging drug, arsenic trioxide, on cardiac potassium currents. Circulation 2004;109(1):26–9.
- [113] Sun HL, Chu WF, Dong DL, Liu Y, Bai YL, Wang XH, et al. Choline-modulated arsenic trioxide-induced prolongation of cardiac repolarization in Guinea pig. Basic Clin Pharmacol Toxicol 2006;98(4):381–8.
- [114] Chen X, Shan H, Zhao J, Hong Y, Bai Y, Sun I, et al. L-type calcium current (ICa,L) and inward rectifier potassium current (IK1) are involved in QT prolongation induced by arsenic trioxide in rat. Cell Physiol Biochem 2010;26(6):967–74.
- [115] Lu HR, Vlaminckx E, Cools F, Gallacher DJ. Direct effects of arsenic trioxide on action potentials in isolated cardiac tissues: importance of the choice of species, type of cardiac tissue and perfusion time. J Pharmacol Toxicol Methods 2012;66(2):135–44.
- [116] Raghu KG, Cherian OL. Characterization of cytotoxicity induced by arsenic trioxide (a potent anti-APL drug) in rat cardiac myocytes. J Trace Elem Med Biol 2009;23(1):61–8.
- [117] Zhao X, Feng T, Chen H, Shan H, Zhang Y, Lu Y, et al. Arsenic trioxide-induced apoptosis in H9c2 cardiomyocytes: implications in cardiotoxicity. Basic Clin Pharmacol Toxicol 2008;102(5):419–25.
- [118] Bessho M, Aki T, Funakoshi T, Unuma K, Noritake K, Kato C, et al. Rho-kinase inhibitor Y-27632 attenuates arsenic trioxide toxicity in H9c2 cardiomyoblastoma cells. Cardiovasc Toxicol 2013;13(3):267–77.
- [119] Park E-J, Park K. Gene expression profiles of cultured rat cardiomyocytes (H_9C_2 cells) in response to arsenic trioxide at subcytotoxic level and oxidative stress. J Health Sci 2006;52(5):512–21.
- [120] Miao X, Tang Z, Wang Y, Su G, Sun W, Wei W, et al. Metallothionein prevention of arsenic trioxideinduced cardiac cell death is associated with its inhibition of mitogen-activated protein kinases activation in vitro and in vivo. Toxicol Lett 2013;220(3):277-85.
- [121] Fan Y, Wang C, Zhang Y, Hang P, Liu Y, Pan Z, et al. Genistein ameliorates adverse cardiac effects induced by arsenic trioxide through preventing cardiomyocytes apoptosis. Cell Physiol Biochem 2013;31(1):80–91.
- [122] Ghosh J, Das J, Manna P, Sil PC. Taurine prevents arsenic-induced cardiac oxidative stress and apoptotic damage: role of NF-kappa B, p38 and JNK MAPK pathway. Toxicol Appl Pharmacol 2009;240(1):73–87.
- [123] Chu W, Li C, Qu X, Zhao D, Wang X, Yu X, et al. Arsenic-induced interstitial myocardial fibrosis reveals a new insight into drug-induced long QT syndrome. Cardiovasc Res 2012;96(1):90–8.
- [124] Sumi D, Sasaki T, Miyataka H, Himeno S. Rat H9c2 cardiac myocytes are sensitive to arsenite due to a modest activation of transcription factor Nrf2. Arch Toxicol 2011;85(12):1509–16.

- [125] Zhang W, Guo C, Gao R, Ge M, Zhu Y, Zhang Z. The protective role of resveratrol against arsenic trioxideinduced cardiotoxicity. Evid Based Complement Alternat Med 2013;2013:407839.
- [126] Zhang Y, Dong Z, Jin L, Zhang K, Zhao X, Fu J, et al. Arsenic trioxide-induced hERG K(+) channel deficiency can be rescued by matrine and oxymatrine through up-regulating transcription factor Sp1 expression. Biochem Pharmacol 2013;85(1):59–68.
- [127] Wang M, Sun G-b, Sun X, Wang H-w, Meng X-b, Qin M, et al. Cardioprotective effect of salvianolic acid B against arsenic trioxide-induced injury in cardiac H9c2 cells via the PI3K/Akt signal pathway. Toxicol Lett 2013;216(2-3):100–7.
- [128] Sturlan S, Baumgartner M, Roth E, Bachleitner-Hofmann T. Docosahexaenoic acid enhances arsenic trioxide-mediated apoptosis in arsenic trioxide-resistant HL-60 cells. Blood 2003;101(12):4990–7.
- [129] Muthumani M, Prabu SM, Milton S. Silibinin potentially attenuates arsenic-induced oxidative stress mediated cardiotoxicity and dyslipidemia in rats. Cardiovasc Toxicol 2014;14:83–97.
- [130] Vineetha VP, Prathapan A, Soumya RS, Raghu KG. Arsenic trioxide toxicity in H9c2 myoblasts damage to cell organelles and possible amelioration with *Boerhavia diffusa*. Cardiovasc Toxicol 2013;13(2):123–37.
- [131] Pachauri V, Mehta A, Mishra D, Flora SJ. Arsenic induced neuronal apoptosis in guinea pigs is Ca2+ dependent and abrogated by chelation therapy: role of voltage gated calcium channels. Neurotoxicology 2013;35:137-45.
- [132] Dwivedi N, Mehta A, Yadav A, Binukumar BK, Gill KD, Flora SJ. MiADMSA reverses impaired mitochondrial energy metabolism and neuronal apoptotic cell death after arsenic exposure in rats. Toxicol Appl Pharmacol 2011;256(3):241–8.
- [133] Flora SJ, Pachauri V. Chelation in metal intoxication. Int J Environ Res Public Health 2010;7(7):2745-88.
- [134] Flora SJ, Mittal M, Mehta A. Heavy metal induced oxidative stress & its possible reversal by chelation therapy. Indian J Med Res 2008;128(4):501–23.
- [135] Flora SJ, Pande M, Kannan GM, Mehta A. Lead induced oxidative stress and its recovery following coadministration of melatonin or N-acetylcysteine during chelation with succimer in male rats. Cell Mol Biol (Noisy-le-grand) 2004;50. Online Pub: OL543-51.
- [136] Flora SJ, Mehta A, Rao PV, Kannan GM, Bhaskar AS, Dube SN, et al. Therapeutic potential of monoisoamyl and monomethyl esters of meso-2,3-dimercaptosuccinic acid in gallium arsenide intoxicated rats. Toxicology 2004;195(2-3):127-46.
- [137] Flora SJ, Saxena G, Mehta A. Reversal of lead-induced neuronal apoptosis by chelation treatment in rats: role of reactive oxygen species and intracellular Ca(2+). J Pharmacol Exp Ther 2007;322(1):108–16.
- [138] Kosnett MJ. The role of chelation in the treatment of arsenic and mercury poisoning. J Med Toxicol 2013;9(4):347-54.
- [139] Kathirgamanathan K, Angaran P, Lazo-Langner A, Gula LJ. Cardiac conduction block at multiple levels caused by arsenic trioxide therapy. Can J Cardiol 2013;29(1):130.e5–e6.
- [140] Kumazaki M, Ando H, Kakei M, Ushijima K, Taniguchi Y, Yoshida M, et al. alpha-Lipoic acid protects against arsenic trioxide-induced acute QT prolongation in anesthetized guinea pigs. Eur J Pharmacol 2013;705(1-3):1-10.
- [141] Flora SJ, Bhadauria S, Kannan GM, Singh N. Arsenic induced oxidative stress and the role of antioxidant supplementation during chelation: a review. J Environ Biol 2007;28(2 Suppl):333–47.
- [142] Mehta A, Ramachandra CJ, Sequiera GL, Sudibyo Y, Nandihalli M, Yong PJ, et al. Phasic modulation of Wnt signaling enhances cardiac differentiation in human pluripotent stem cells by recapitulating developmental ontogeny. Biochim Biophys Acta. 2014 Nov;1843(11):2394–402.
- [143] Mehta A, Mathew S, Viswanathan C, Majumdar SA. Intrinsic properties and external factors determine the differentiation bias of human embryonic stem cell lines. Cell Biol Int 2010;34(10):1021–31.
- [144] Sequiera GL, Mehta A, Shim W. A simple protocol for the generation of cardiomyocytes from human pluripotent stem cells. Methods Mol Biol 2013, in press.

- [145] Mehta A, Konala VBR, Khanna A, Majumdar AS. Assessment of drug induced developmental toxicity using human embryonic stem cells. Cell Biol Int 2008;32(11):1412–24.
- [146] Flora SJ, Mehta A. Monoisoamyl dimercaptosuccinic acid abrogates arsenic-induced developmental toxicity in human embryonic stem cell-derived embryoid bodies: comparison with in vivo studies. Biochem Pharmacol 2009;78(10):1340–9.
- [147] Takahashi K, Yamanaka S. Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors. Cell 2006;126(4):663–76.
- [148] Sequiera GL, Mehta A, Ooi TH, Shim W. Ontogenic development of cardiomyocytes derived from transgene-free human induced pluripotent stem cells and its homology with human heart. Life Sci 2013;92(1):63–71.
- [149] Mehta A, Chung YY, Ng A, Iskandar F, Atan S, Wei H, et al. Pharmacological response of human cardiomyocytes derived from virus-free induced pluripotent stem cells. Cardiovasc Res 2011;91(4):577–86.
- [150] Shim W, Mehta A, Chung Y, Ng A, Wei H, Dusting G. Ion channel and cardiac ontogeny of human cardiomyocytes derived from vector-free induced pluripotent stem cells. Eur Heart J 2011;08(32):811.
- [151] Mehta A, Chung Y, Sequiera GL, Wong P, Liew R, Shim W. Pharmacoelectrophysiology of viral-free induced pluripotent stem cell-derived human cardiomyocytes. Toxicol Sci 2013;131(2):458–69.
- [152] Mehta A, Sequiera G, Sudibyo Y, Jun L, Wong P, Liew R. Derivation and characterization of transgene free induced pluripotent stem cell derived cardiomyocytes from Asian patient with long QT syndrome. Eur Heart J 2012;08(33):22.
- [153] Mehta A, Sequiera GL, Ramachandra CJA, Sudibyo Y, Chung YY, Wong YW, et al. Re-trafficking of hERG reverses long QT syndrome 2 phenotype in human pluripotent stem cell derived cardiomyocytes. Eur Heart J 2013;34(Suppl. 1).
- [154] Mehta A, Sequiera GL, Ramachandra CJ, Sudibyo Y, Chung Y, Sheng J, et al. Re-trafficking of hERG reverses long QT syndrome 2 phenotype in human iPS-derived cardiomyocytes. Cardiovasc Res 2014;102(3):497–506.

This page intentionally left blank

21 Immunotoxic Effects of Arsenic Exposure

Laila N. Islam

DEPARTMENT OF BIOCHEMISTRY AND MOLECULAR BIOLOGY, UNIVERSITY OF DHAKA, DHAKA, BANGLADESH

CHAPTER OUTLINE

21.1	Introduction	494
21.2	Influence of Nutritional Factors	496
21.3	Effects on Blood Leukocytes	496
21.4	Interruption of Energy Production	496
21.5	Effects on ROS Production	497
21.6	Genotoxic and Carcinogenic Potentials	497
21.7	Hematological Effects on Experimental Animals	498
21.8	Effects on Heme Synthesis	499
21.9	Hepatic Effects and Lipid Peroxidation	499
21.10	Effects on Immune Responses in Fish	500
21.11	Effects on Immune Responses in Laboratory Animals	501
21.12	Effects of Arsenic in Drinking Water on Human Health	502
21.13	Immunotoxic Effects of Organic Arsenicals in Foods	503
21.14	Medicinal Use of Arsenic and Its Mechanism of Action	504
21.15	Effects of Arsenic Compounds on Human Cells in Culture	504
21.16	Immunotoxic Effects on Murine and Human Monocytes/Macrophages	506
21.17	Immunotoxic Effects on Murine Mononuclear Cells	509
21.18	Decreased Cytokine Production by Human T Cells	509
21.19	Effects of In Utero Exposure on Infant Immune System	510
21.20	Gender-Related Immunotoxic Effects in Human	510
21.21	Effects of Chronic Exposure on Immune Response	511
21.22	Association with Respiratory Complications and Impaired Immune Responses	512
21.23	Effects of Chronic Exposure on Serum Complement Function	513
21.24	Conclusions	513
Refere	ences	514

21.1 Introduction

Arsenic, a toxicant of natural occurrence in mineral deposits, is used in many human activities such as manufacturing, agriculture, and medicine [1]. Arsenical compounds are transported into the environment mainly by water from wells drilled into the arsenic-rich geologic strata or by ambient air during smelting and burning of coal [1,2]. The main route of arsenic exposure for the general population is via drinking water. Endemic arsenic exposure emerged as a single catastrophe affecting millions of people mostly living in Bangladesh, India, Mexico, Taiwan, and South America. In these regions, the concentrations of arsenic amount to several hundred micrograms per liter, which considerably exceed the standard of $50 \mu g/L$ recommended by the World Health Organization for drinking water in Bangladesh and India [1]. In Bangladesh, approximately 85 million people are at risk of drinking arsenic-contaminated water from about 2 million tubewells that supply waters containing more than $50 \mu g/L$ of arsenic [3].

In humans, the liver rapidly detoxifies inorganic arsenic that is consumed in drinking water by transforming it into organic forms called monomethylarsonic acid (MMA) and dimethylarsinic acid (DMA) that are rapidly excreted in the urine [4], and to give an overall arsenic half-life in the body of about 30 hours [5]. Thus, following exposure to arsenate (valency state V), the first step in biotransformation is the reduction to arsenite (valency state III), a process that may be considered a bioactivation [6]. There is evidence that the methylating capacity differs among individuals and populations and that different capacities would result in variation in tissue retention of arsenic. It has been found that arsenic causes the depletion of *S*-adenosylmethionine, the main cellular methyl donor [7]. Environmental factors, particularly diet, might be important in explaining susceptibility to arsenic toxicity [8]. Also its metabolism can be influenced by the concentration of arsenic that is being ingested and by gender, as has been found in boys compared to girls of the same age [9].

The immune system is a network of white blood cells and their soluble mediators, tissues, and organs that work together to defend the body against attacks by foreign invaders. The foreign pathogens are primarily microbes such as bacteria, parasites, viruses, and fungi that can cause infections. Microorganisms or toxins that successfully enter an organism encounter immunocompetent cells and mechanisms of the immune system that keep them out or, failing this, seek them out and destroy them. The functions of the immune system are modulated by a number of chemical and biological substances, environmental factors, and trace elements (metals) including aluminum [10], magnesium, selenium, and zinc [11], and many toxic pollutants such as mercury, lead, cadmium, and arsenic [12]. Human peripheral blood mononuclear cells incubated with 15μ M arsenic and 65μ M cadmium showed that both are able to induce apoptosis of lymphoid cells with arsenic inducing a significant level, while lead as high as 500μ M is non-toxic and did not induce a significant degree of apoptosis [12].

Metals and metal compounds occur naturally in mineral deposits whose distribution depends on their natural sources (e.g., volcanoes or erosion) and use in human activities such as industry, agriculture, and even medicine. They are transformed naturally by bacterial activity with formation of organic species that influence their mobility and accumulation in abiotic as well as biotic systems. Due to their widespread use, different metal species are released into the environment, posing a serious threat to human health. The use and actions of selected metal species of scientific concern (arsenic, selenium, and platinum) showed that numerous health risks may be associated with exposure to these substances [13].

High concentrations of natural arsenic are generally associated with different geochemical environments [14]. Geological arsenic can stay in solid phases as minerals, amorphous solids, and sedimentary organic matter that supply dissolved arsenic. Release from iron oxide and sulfide minerals appears to be the most common cause of widespread arsenic contamination in ground water. Mobilization of arsenic in sedimentary aquifers may be, in part, a result of changes in the geochemical environment due to agricultural irrigation. In the deeper subsurface, elevated arsenic concentrations are associated with compaction caused by excessive groundwater withdrawals [15].

There is strong evidence from epidemiological studies of an association between chronic exposure to inorganic arsenic and hyperpigmentation, hyperkeratosis, and neoplasia in the skin as well as other diseases [16]. A higher prevalence rate of arsenical skin lesions with a clear dose-response relationship has been found among Bangladeshi populations drinking arsenic-contaminated well water [17], and callus-like growths all over the extremities with changes in skin pigmentation have been reported [18]. The general adverse health effects associated with human exposure to arsenicals include cardiovascular diseases, developmental abnormalities, neurologic and neurobehavioral disorders, alteration of metabolic enzyme functions, diabetes, fibrosis of the liver and lung, and hematological disorders [19,20].

Several studies have found that anemia, leukopenia, and thrombocytopenia are common effects of arsenic poisoning in humans following acute [21] and chronic oral exposures [22] at doses of $50 \mu g/kg/day$ or more. These effects may be due to both a direct cytotoxic or hemolytic effect on the blood cells [21,23] and a suppression of erythropoiesis [23]. The magnitude of exposure to patients examined by these researchers is not specified, although arsenic concentration in some of the wells in the area exceeded $1000 \mu g/L$. However, there are reports that hematological effects are not observed in all cases of acute poisoning with arsenic [24,25], and these abnormalities are reversible within weeks of termination of arsenic exposure [26]. Such hematological effect has been observed in imprinting control region (ICR) mice where the mean corpuscular volume of red blood cells decreased significantly when the animals were challenged with drinking water containing high levels of arsenic that replicated the levels detected in streams and soils in Buruli ulcer endemic communities of Ghana [27].

It has been shown that the immune system is among the targets for arsenic. Recent studies on the mechanisms of arsenic toxicity suggest that the metal toxicant can alter the delicate balance and regulation of the immune cells and can increase the incidence of certain diseases [28]. It has been known for a long time that arsenic is the immunosuppressive component of gallium arsenide (GaAs) that alters macrophage function and T lymphocyte proliferation [29,30]. A single intratracheal dose of (GaAs) at 200 mg/kg was reported to decrease the percentages of Thy 1.2 positive T cells in mice, and the IgM and IgG antibody-forming cells in the spleen were reduced by 66% and 48%, respectively [31]. It has been further reported that gallium arsenide selectively alters the expression of CD25 (IL-2R/p55) [32]. Another study showed that GaAs affected both humoral and cellular immune parameters in B6C3F1 female mice and impaired the ability of the immune system to protect against B16F10 tumor challenge [31].

21.2 Influence of Nutritional Factors

An association of clinical complications of arsenic toxicity with the duration of exposure and nutritional status of the affected subjects based on a study conducted on a total of 115 exposed individuals diagnosed as arsenicosis patients who were randomly selected from four known arsenic endemic villages in Bangladesh, and age-matched 120 unexposed (control) subjects [33] has been reported. The duration of arsenic exposure in about 37% of the patients was at least 10 years, while the population's mean and range were 7.6 \pm 5.2 years and 1–25 years, respectively. The mean arsenic concentrations in the drinking water for the exposed and unexposed populations were 218.1 μ g/L and 11.3 μ g/L, respectively. The spot urine of the arsenicosis patients, considered a reliable marker of recent exposure to arsenic, contained an average of 234.6µg/L arsenic. It was found that adverse clinical symptoms of melanosis, keratosis on palms and soles with cracks and nodule formation, severe skin irritation, and lump formation on feet developed with the increase in duration of arsenic exposure. About 15% of the controls had body mass index (BMI) lower than 18.5, the cut-off point for malnutrition (underweight); the corresponding value among the arsenicosis patients was about 28%; they had more severe arsenical symptoms and complications compared to the overweight patients, having BMI 25.0 or above. These findings demonstrate that the poor nutritional status of patients increases the complications of chronic arsenic toxicity [33].

21.3 Effects on Blood Leukocytes

It has been reported that arsenicosis patients have a significantly lower white blood cell (WBC) count in their peripheral blood compared to As-unexposed healthy subjects. About 16% of the subjects chronically exposed to arsenic via drinking water showed leukopenia, having WBC count below 4 million cells per mL. About 26% of the patients showed neutropenia, having less than 50% neutrophils in the peripheral blood (normal range: 50–70%), and 39% had more than 40% lymphocytes. Prevalence of neutropenia and lymphocytosis was observed in patients with chronic exposure to high levels of arsenic in drinking water. It was observed that many of the chronically exposed subjects were suffering from anemia (26%) and weakness (about 33%). Also, the incidences of leukopenia and anemia were more common in the female patients who were malnourished (BMI <18.5). These findings demonstrate a higher prevalence of anemia, leukopenia, and lymphocytosis in chronic arsenic toxicity [33]. A previous study conducted in West Bengal, India, reported that anemia was present in 47% of 156 subjects with arsenical skin lesions [34].

21.4 Interruption of Energy Production

It is known that arsenic is able to exert its toxic effects on the metabolic pathways by modulating the antioxidant defense system, interrupting the glycolytic pathway and citric acid cycle, and thus inhibiting oxidative phosphorylation. Many inorganic compounds have been found to compete with phosphorus in the oxidative phosphorylation process [35], mainly in their pentavalent form [36]. We found that the levels of inorganic phosphorus in the serum of arsenicosis patients were significantly higher than in the unexposed control subjects [20]. Arsenic can compete with phosphorus in the oxidative phosphorylation process by replacing phosphorus in the cells, thereby causing an increase in the levels of unutilized inorganic phosphate in the blood. It has been reported that arsenic is an uncoupler of mitochondrial oxidative phosphorylation that induces generation of reactive oxygen species (ROS) [37]. Thus, replacement of phosphorus by arsenic in the crucial energy yielding steps may explain why arsenicosis patients suffer from weakness, as phosphate becomes unavailable inside the cells for utilization for the synthesis of high-energy compounds.

21.5 Effects on ROS Production

Production of ROS such as nitric oxide anion (NO⁻) and superoxide anion (O_2^-) by activated monocytes is an important innate immune response to destroy invading microbes. It was found in Mexican children drinking arsenic-contaminated water that arsenic exposure was positively associated with O_2^- production by mitogen-stimulated monocytes [38]. These data suggest that arsenic could impair the function of monocytes. For example, elevated ROS levels in unstimulated monocytes indicate arsenic-induced oxidative stress; ROS overproduction by activated cells could cause oxidative damage to surrounding tissues, whereas diminished ROS production could weaken monocyte defense against pathogens. It has been suggested in another study that arsenic alters cellular glutathione levels either by utilizing this electron donor for the conversion of pentavalent to trivalent arsenicals or directly binding with it or by oxidizing glutathione via arsenic-induced free radical generation [39].

21.6 Genotoxic and Carcinogenic Potentials

It has been shown that arsenic is a potent gene and chromosomal mutagen whose effects are mediated through the induction of ROS [40]. Recent investigations found that mitochondrial damage plays a crucial role in arsenic mutagenicity, and that mitochondrial damage can lead to the release of superoxide anions, which then react with nitric oxide to produce the highly reactive peroxynitrites. These data suggest that mitochondria are a primary target in the arsenic-induced genotoxic response [41]. Another study hypothesized that arsenic is a well-recognized human carcinogen that has been thought to act through epigenetic mechanisms by modifying DNA methylation patterns, perhaps in conjunction with DNA-damaging agents. Arsenic causes the depletion of *S*-adenosylmethionine, the main cellular methyl donor, and represses the expression of the DNA methyltransferase genes *DNMT1* and *DNMT3A*. It is possible that as a consequence of these two complementary mechanisms, long-term exposure to arsenic results in DNA hypomethylation [7]. In a recent review article, Flora [39] supports the hypothesis that as a carcinogen arsenic acts through epigenetic mechanisms rather than as a classical mutagen, and states that the carcinogenic potential of arsenic may be attributed

to activation of redox-sensitive transcription factors and other signaling pathways involving nuclear factor κB , activator protein-1, and p53.

21.7 Hematological Effects on Experimental Animals

The concentration of toxicant present in the environment and the duration of exposure appear to be important for the deleterious health effects in the experimental animals and humans to be manifested. In a study conducted in Haryana, India, on crossbred (Friesian × Sahiwal) male calves of 3–4 months age supplemented daily with 50 ppm arsenic (as As_2O_3) for up to 90 days in their diet in an attempt to investigate certain hematological parameters, no change in blood total leukocyte counts, differential leukocyte counts, packed cell volume (PCV), and hemoglobin was observed when comparing with the corresponding non-supplemented group [42]. In these experiments, calves of both groups were fed as per ICAR standards [43] and their requirements were fulfilled by feeding concentrate mixture and green oats. All calves were kept under similar conditions, and blood samples were collected at fortnightly intervals to estimate various parameters. The reasons why no adverse effects of arsenic had been observed in the exposed calves were two-fold: first, in this study although the ruminant animals were challenged with a high concentration of arsenic, 50 ppm (50 mg/L), their toxicity level in drinking water was not known; second, the study period was too short, only 90 days.

There could be species difference and degree of adversity on toxic effects of environmental pollutants among the directly exposed animals. To study the effect on birds, arsenic-induced toxicities were studied in research conducted at the Bangladesh Agricultural University, Mymensingh. In this study, ducklings were given arsenic trioxide at 100 mg/L in the drinking water daily for 90 days starting from day 15. One group of same-age ducklings was kept as control. Five birds were sacrificed from each group at 15-day intervals and toxic signs, body weight, and hematological parameters were recorded. It was found that ducks drinking a high concentrations showed signs of depression with dullness and reduced feed intake, and had ruffled feathers. About 15% of these birds did not gain body weight. There were reductions in total erythrocyte count, hemoglobin content, and PCV values, and the rise of erythrocyte sedimentation rate (ESR) values was significant (p < 0.01) in the arsenic-treated group [44].

In the study of Gyasi et al. [27], ICR mice were challenged with 0.8–4.8 mg/L arsenic via their drinking water, the As concentration being identical to that found in the streams and soils in Buruli ulcer endemic communities of the Amansie West District of Ghana. The mice developed inflammation, erythema, and open ulcers on skin (with scab formation) that were negative for acid-fast bacilli. Histopathological studies revealed liver and spleen damage with hepatic cell swelling and degenerative change showing cytoplasmic vacuolation with nuclear blebbing and gradual cell loss. The spleen developed a lymphoid background with multinucleate cell formation. Hematological examination revealed significant dose-dependent decrements in white blood cells indicating a detrimental effect on the body's immune system, a situation that makes the body susceptible to infections including *M. ulcerans*. The mean corpuscular volume of red blood cells also decreased significantly indicating microcytosis. High levels of arsenic in tissue (possibly from accumulation) caused inflammation, erythema, and

open ulcers on the skin; the presence of As at such levels has the potential to cause liver and spleen damage, reduced immune system function, and red blood cell microcytosis in experimental ICR mice.

21.8 Effect on Heme Synthesis

In research carried out by Woods and Fowler, it was found that chronic exposure of both rats and mice to sodium arsenate at 20, 40, or 85 ppm in drinking water resulted in depression of hepatic δ -aminolevulinic acid synthetase and heme synthetase, the first and last enzymes in heme biosynthesis, respectively. Concomitantly, urinary uroporphyrin and coproporphyrin concentrations were elevated in each of the exposed groups of rats and mice compared to their corresponding control values. In contrast, no changes were observed in the activities of cytochrome oxidase or cytochrome P450, indicators of mitochondrial and microsomal hemoprotein function, respectively. These results demonstrate that prolonged exposure to low levels of arsenic results in selective alteration of hepatic heme biosynthetic pathway enzymes, with concomitant increases in urinary porphyrin concentrations. Changes in hepatic hemoprotein function may serve as a specific indicator of pretoxic arsenic exposure in both rats and mice [45].

21.9 Hepatic Effects and Lipid Peroxidation

A cross-sectional epidemiological study was reported by Guha Mazumder [46] on 7683 people residing in arsenic-affected districts of West Bengal, India. Of these, 3467 people (control group) consumed drinking water containing safe levels of arsenic (less than 50 µg/L), and 4216 people consumed drinking water containing arsenic levels above $50 \mu g/L$ (arsenic-exposed group). The prevalence of hepatomegaly was significantly higher in arsenic-exposed people (10.2%) compared to controls (2.99%, p < 0.001). The incidence of hepatomegaly was found to have a linear relationship proportionate to increasing exposure of arsenic in drinking water in both sexes (p < 0.001). From hospital-based studies on 248 cases of arsenicosis, hepatomegaly was found in about 77% of the patients, and non-cirrhotic portal fibrosis was the predominant lesions in 63 out of 69 patients who underwent liver biopsy [46].

In a similar study conducted by the same investigator, BALB/C mice were given water contaminated with arsenic (3.2 mg/L) *ad libitum* for 15 months, the animals being sacrificed at 3-month intervals. There was progressive reduction of hepatic glutathione and enzymes of the antioxidative defense system associated with lipid peroxidation [46]. Liver histology showed fatty infiltration at 12 months and hepatic fibrosis at 15 months. In another study, pregnant Wistar rats were subchronically exposed to arsenic (50 mg/L) through drinking water, and changes occurred in the blood of arsenic-exposed pups that were examined during gestation and lactation. There was normocytic and normochromic anemia as well as a significant increase in hemolysis, enhanced lipid peroxidation (a measure of oxidative stress, such as thiobarbituric acid reactive substances (TBARS) production), and catalase activity in the blood of arsenic-intoxicated pups [47]. These studies on arsenic-affected people as well as experimental animals show that prolong drinking of arsenic-contaminated water is associated with hepatomegaly. The predominant lesion of hepatic fibrosis appears to be caused by arsenicinduced oxidative stress.

21.10 Effects on Immune Responses in Fish

21.10.1 Innate Immune Response

The innate immune response, which is the first line of defense against invading pathogens, can be affected by environmental toxicants such as arsenic. In zebrafish infected with snakehead rhabdovirus, arsenic decreased induction and altered the kinetics of the antiviral cytokines interferon (IFN) and Mx, and inhibited tumor necrosis factor- α (TNF- α) priming of the respiratory burst response [48]. In another study, it was found that exposure to 2 and 10 ppb arsenic, both considered safe levels in drinking water, resulted in a greater than 50-fold increase in viral load and at least a 17-fold increase in bacterial load in embryos of zebrafish. Presence of arsenic lowered the overall innate immune response of the fish as evidenced by reductions in respiratory burst activity. Viral infection, after arsenic exposure, showed decreases of up to 13- and 1.5-fold changes in interferon and Mx mRNA expression, respectively. Bacterial infection, postarsenic exposure, demonstrated at least 2.5- and four-fold declines in interleukin-1 β and TNF- α mRNA levels, respectively. Maximum expression of these essential cytokines was also delayed upon arsenic exposure. These data indicated that arsenic exposure, at concentrations deemed safe in drinking water, suppressed the overall innate immune function in zebrafish [49].

21.10.2 Humoral Immune Response

The effects of long-term arsenic exposure (150 days) on the head kidney (HK) and ensuing humoral immune responses were studied in *Clarias batrachus* L. [49]. Exposure to non-lethal concentrations of arsenic (42.42μ M) resulted in significant time-dependent alterations in HK cell numbers, eventually affecting the HK somatic index. Prolonged exposure to arsenic also suppressed HK-B cell proliferation and led to significant reduction in serum immunoglobulin levels and antigen-specific serum bacterial agglutinin titers. Arsenic exposure inhibited the release of "IL-4 like factors" from HK-T cells. Histological studies documented extensive lymphocytopenia and a decrease in the melano-macrophage population, and hemosiderin accumulation. From exposure-challenge studies with *Aeromonas hydrophila* it was evident that pathogens could efficiently disseminate and colonize distant host tissues in the exposed fish. Thus, long-term exposure to non-lethal concentrations of arsenic affects HK and interferes with the humoral immune response of *C. batrachus* rendering them immunocompromised and susceptible to pathogenic challenge [50].

Chronic exposure (30 days) to arsenic $(As_2O_3, 0.50 \,\mu\text{M})$ in the same species of fish led to a significant increase in arsenic content in the HK accompanied by reduction in both head kidney macrophage (HKM) number and head kidney somatic index. Transmission electron microscopy of arsenic-exposed HKM revealed prominent endoplasmic reticulum, chromatin condensation, and loss in structural integrity of nuclear membrane. HKM from exposed fish demonstrated significant levels of superoxide anions but on infection with *Aeromonas*

hydrophila the fish were unable to clear the intracellular bacteria and died. Exposure-challenge experiments with *A. hydrophila* revealed that arsenic interfered with the phagocytic potential of HKM, and helped in intracellular survival of the ingested bacteria inside the HKM thereby inducing significant HKM cytotoxicity. The immunosuppressive effect of arsenic was further evident from the ability of *A. hydrophila* to colonize and disseminate efficiently in exposed fish. Further, arsenic suppressed the production of pro-inflammatory "IL-1 β -like" factors from HKM. These data suggest that arsenic even at very low concentrations is immunotoxic to fish and the changes observed in HKM may provide a useful early biomarker of low-level As exposure [51].

21.11 Effects on Immune Responses in Laboratory Animals

21.11.1 Humoral Immune Response

Male Wistar rats were treated for 4, 8, and 12 weeks with 3.33, 6.66, 13.3, or 26.6 mg/kg of inorganic arsenic (NaAsO₂) per os by gavage [51]. Changes in humoral immune response, which was tested after 4 weeks of treatment, were not marked. After prolonged exposure to arsenic, the weight of thymus, spleen, and adrenals, which are organs responsible for immune response, was significantly reduced. The animals showed reduced delayed-type hypersensitivity (DTH) reaction and mean cell volume (MCV) of red blood cells, a hematological parameter. Plaque-forming cell assay proved to be insensitive in this short-time exposure. These results suggest that subchronic low-level exposure to arsenic can affect immune responses in rats [52]. In an earlier study [52], adult male cotton rats were exposed to 5 or 10 ppm sodium arsenite in drinking water for 6 weeks. The daily food intake of the animals decreased in a dose-dependent manner. Masses of liver, kidney, adrenals, popliteal lymph nodes, spleen, and seminal vesicles and selected hematological parameters were unaffected by arsenic exposure (see also next section).

21.11.2 Cell-Mediated Immune Response

In the study by Savabieasfahani et al. mentioned above [52], cell-mediated immunity, as measured by a phytohemagglutinin (PHA)-hypersensitivity response to an intradermal challenge, was suppressed 30% in the low-dose group compared to controls. Arsenic treatment did not have a measurable impact on lymphoproliferative responses of cultured splenocytes to the mitogens concanavalin A (Con A) and pokeweed mitogen, or to interleukin-2. Also no impact of low-level arsenic exposure was observed on macrophage phagocytic activity and tumoricidal activity of lymphokine-activated killer cells *in vitro* [53]. The effects of a single intratracheal instillation at doses of 50, 100, and 200 mg/kg gallium arsenide (GaAs) exposure on immunocompetence of B6C3F1 female mice were investigated. Fourteen days after exposure, the mitogenic response of splenic T cells to Con A and PHA was found unaffected by GaAs, but that of B cells to lipopolysaccharide (LPS) was increased by 52%. The DTH response to keyhole limpet hemocyanin (KLH) and mixed lymphocyte response were significantly reduced in a dosedependent manner by GaAs exposure. The natural killer cell activity against the YAC-1 mouse lymphoma was enhanced in the GaAs-treated mice [31]. Following GaAs exposure, analysis of peritoneal exudate cells (PEC) revealed a dosedependent decrease in number and a shift in the composition of PECs with increased percentage of monocyte and decreased percentage of lymphocyte populations. The adherent PECs demonstrated decreased phagocytosis of covaspheres and increased phagocytosis of chicken erythrocytes. GaAs exposure had no effect on host resistance to *Plasmodium yoelii* or *Streptococcus pneumoniae*, but dose dependently increased resistance of the mouse to *Listeria monocytogenes*. These observations suggested that GaAs affected the cellular immune parameters in mice [31].

In another study, mice exposed to 50 mg/L arsenic in the drinking water for 4 weeks demonstrated a reduction in lymph node cell proliferation and ear swelling following sensitization with 2,4-dinitrofluorobenzene (DNFB). Langerhans cells (LC) and T-cell populations in the draining lymph nodes of DNFB-sensitized mice were evaluated by fluorescence-activated cell sorting. The number of activated LC was reduced in cervical lymph nodes, suggesting altered LC migration following arsenic exposure. Lymphocytes from arsenic-treated animals sensitized with fluorescein isothiocyanate (FITC) exhibited reduced proliferative responses following T-cell mitogen stimulation *in vitro*; however, lymphocyte proliferation in non-sensitized, arsenic-treated mice was comparable to that in controls. Arsenic exposure also reduced the number of thioglycollateinduced peritoneal macrophages and circulating neutrophils [53]. This study, along with earlier studies by the same group, demonstrates that repeated, prolonged exposure to non-toxic concentrations of sodium arsenite alters immune cell populations and results in functional changes in immune responses in contact hypersensitivity [54].

21.12 Effects of Arsenic in Drinking Water on Human Health

Exposure to arsenic through drinking water is related to many health effects that have different incidences depending on the geographical area, ethnic group, age, and gender, as well as nutritional status of the affected subjects [33,55–60]. The variations could be due to different valence states and level of toxicities of the ionic species of arsenic. The most evident effects of arsenic exposure in Bangladesh and India are often linked to arsenical skin lesions including hyperkeratosis, hypopigmentosis, and cancer development on skin, alteration of the functions of major organs such as liver, bladder, kidney or lung, and ultimate formation of cancer in these organs [17,18,61,62]. Chronic exposure to arsenic in drinking water has been found associated with cancer development and blackfoot disease in Taiwan, some parts of Mongolia, China, and Mexico [63], and with diabetes in the USA [64]. Other manifestations of chronic exposure of arsenic via drinking water include neurological effects, obstetric problems, high blood pressure, and diseases of the respiratory system and blood vessels including the cardiovascular system [65].

In Bangladesh and India, most of the arsenic in drinking water comes from natural sources. In these regions the ground water is contaminated with geological arsenic that is brought to the surface by digging tubewells (in order to collect microorganism-free drinking water), which is the main cause for human exposure. Many of these tubewells supply very high levels of arsenic. In the northern district of Chapainowabganj of Bangladesh, we found the levels of arsenic in the waters of different tubewells in one area to vary from 3 to $875 \mu g/L$ [33]. The levels of arsenic in the ground water of different states of Mexico vary from 66 to $725 \mu g/L$ [66], while the aquifers of Lagunera in northern Mexico vary from 8 to $624 \mu g/L$ [67]. In Mexico, most of the ground water is on the top of volcanic rocks that contain several forms of arsenical compounds that are liberated in to the water, together with other potential toxic substances, causing the water in each region to have different effects on human health. In the central zone and its adjacent states of Mexico where arsenic comes from the mining processes, the affected subjects suffer mainly from diabetes [68], infertility [69], and reduced IQ [70] rather than skin cancer. These observations suggest the influence of environmental factors including nutritional status and lifestyle on the toxic effects of arsenic.

In the United States, researchers are taking a much closer look at drinking water, from southwestern states like Nevada, where wells sometimes contain arsenic at more than $500 \mu g/L$, to the upper Midwest and New England, where a belt of arsenic-infused bedrock taints aquifers in stretches from the coast of Maine to a point midway through Massachusetts. Water in parts of the Central Valley of California has been found to be tainted with arsenic as well [14]. Depending on local environmental conditions, arsenic can leach from soils or mineral deposits into ground water. Some private wells in Michigan and Connecticut have been found to contain arsenic in the range of 10 to 100 ppb [71]. While municipal water suppliers are required to meet the safety standard of 10 ppb or $10 \mu g/L$ of the Environmental Protection Agency (EPA) for arsenic in drinking water, no such regulation exists for private wells. In the USA, about 13 million people get drinking water from private wells with arsenic levels above the federal standard. There have been increased mortality rates linked to everything from diabetes to heart disease among arsenic-affected subjects in Michigan [72].

21.13 Immunotoxic Effects of Organic Arsenicals in Foods

The immunotoxic effects of organic arsenic compounds, namely, arsenocholine (AsCho), arsenobetaine (AsBe), and the tetramethylarsonium ion (TetMA), on the principal immune effector cells, macrophages, and lymphocytes of marine animals were studied, by comparing them with the effects of inorganic arsenicals *in vitro* [73]. The inorganic arsenicals arsenite and arsenate showed strong cytotoxicity to both macrophages and lymphocytes. The concentration of arsenite that reduced the number of surviving cells to 50% (IC₅₀) of that in untreated controls was $3-5\,\mu$ M dm⁻³, while the cytotoxicity of arsenate was lower (IC₅₀ = $100\,\mu^{-1}$ mM dm⁻³) than that of arsenite. Compared with these findings, AsCho and AsBe were less toxic even at a concentration over $10\,\text{mM}\,\text{dm}^{-3}$ to both macrophages and lymphocytes; however, TetMA had weak, but significant cytotoxicity to these cells (IC₅₀ was about 6 mM dm⁻³).

In another study, synthetic AsBe was orally administered to CDF_1 mice at a dose of 1.625 g/kg weight once a day on days -6, -4, -2 and 0 (four times, total 6.5 g/kg mouse weight), and its effect on the immune organs and immune effector cells were assessed until day 8. Orally administered AsBe was found temporally distributed to the spleen and thymus, and to immune effector cells, splenocytes, thymocytes, Peyer's patch lymphocytes, and peritoneal

macrophages, but was not very toxic either quantitatively or qualitatively. These findings suggest that ingestion of AsBe contained in marine animals is relatively safe in relation to the health of people who often consume marine foods in their daily diet [74].

21.14 Medicinal Use of Arsenic and Its Mechanism of Action

Acute promyelocytic leukemia (APL) is a subtype of acute myeloid leukemia (AML) with distinctive biologic and clinical features. Most patients with APL are young and exhibit a lifethreatening coagulopathy, which is the most notorious manifestation of the disease. The cells from almost all patients have a balanced reciprocal translocation between chromosomes 15 and 17 [75], which generates a fusion transcript joining the *PML* (promyelocyte) and *RAR*- α (retinoic acid receptor- α) genes [76]. In children, the disease is often associated with a high white blood cell count (WBC >10,000/ μ L), the microgranular variant (M3V), and all-*trans* retinoic acid (ATRA)-related toxicities [77]. Intensive studies of the biology and treatment of the disease have resulted in a remarkably thorough understanding of its pathogenesis [78]. Low dose arsenic trioxide (As₂O₃) has shown substantial efficacy in treating both newly diagnosed and relapsed patients with APL [79]. This achievement reflects collaborative laboratory and clinical studies, led initially by innovative investigators in China [80] and then France [81] and subsequently by many other cooperative oncology groups throughout the world.

The successes with arsenic trioxide have prompted investigations to elucidate the mechanisms of action underlying the clinical responses. Substantial data show that arsenic trioxide produces remissions in patients with APL at least in part through a mechanism that results in the degradation of the aberrant PML (promyelocyte)-retinoic acid receptor- α fusion protein. As a single agent, it induces complete remissions, causing few adverse effects and only minimal myelosuppression [81]. The combination of ATRA and arsenic trioxide, with minimal chemotherapy to control leukocytosis, is very effective therapy for newly diagnosed APL patients [83]. Studies have also investigated concerns about the toxicity and potential carcinogenicity of long-term exposure to environmental arsenic. Arsenic apparently affects numerous intracellular signal transduction pathways and causes many alterations in cellular function. These actions of arsenic may result in the induction of apoptosis, inhibition of growth and angiogenesis, and the promotion of differentiation [82].

21.15 Effects of Arsenic Compounds on Human Cells in Culture

The toxicities of inorganic arsenicals (iAs) and trivalent and pentavalent methylated arsenicals were examined in cultured human cells derived from tissues that are considered a major site for iAs methylation (liver) or targets for carcinogenic effects associated with exposure to iAs (skin, urinary bladder, and lung). Among the arsenicals examined, trivalent monomethylated species were the most cytotoxic in all cell types, and trivalent dimethylated arsenicals were at least as cytotoxic as trivalent iAs (arsenite) for most cell types. Pentavalent arsenicals were significantly less cytotoxic than their trivalent analogs. Among the cell types, the greatest methylation capacity for iAs was exhibited by primary human hepatocytes, followed by epidermal keratinocytes, and bronchial epithelial cells. Cells derived from human bladder did not methylate iAs. There was no apparent correlation between susceptibility of cells to arsenic toxicity and their capacity to methylate iAs. These results suggest that the trivalent methylated arsenicals may significantly contribute to the adverse effects associated with exposure to iAs; and high methylation capacity does not protect cells from the acute toxicity of trivalent arsenicals [84].

It is generally accepted that methylation is a mechanism of arsenic detoxification; recent studies have suggested that methylated arsenicals also have deleterious biological effects. In the study of Vega et al. [16] the effects of inorganic arsenicals and trivalent and pentavalent methylated arsenicals were tested in human keratinocyte cultures. Viability testing showed that the relative toxicities of the arsenicals were as follows: $iAs^{III} > MAs^{III}O > DMAs^{III}GS > DM$ $As^{V} > MAs^{V} > iAs^{V}$. The trivalent arsenicals induced an increase in cell proliferation at concentrations in the 0.001 to 0.01 µM range, while at high concentrations (>0.5 µM) cell proliferation was inhibited. The pentavalent arsenicals did not stimulate cell proliferation. The methylated forms of As^{V} were found more cytotoxic than iAs^{V} . These data suggest that methylated arsenicals, products of the metabolic conversion of inorganic arsenic, can significantly affect viability and proliferation of human keratinocytes.

21.15.1 Effects on Growth-Promoting Cytokines and Growth Factors

It was found that exposure of human keratinocytes in culture to low doses of trivalent arsenicals stimulated secretion of the growth-promoting cytokines, granulocyte-macrophage colony stimulating factor (GM-CSF) and TNF- α . The dimethylarsinic acid (DMAs^V) reduced cytokine secretion at concentrations in which cell proliferation and viability were not affected [16]. Epidemiological studies have shown that inorganic arsenicals are human skin carcinogens that induce hyperproliferation and hyperkeratosis, but there is currently no known mechanism for their action or an established animal model for its study. Germolec et al. [85] observed increased mRNA transcripts and secretion of keratinocyte growth factors, including GM-CSF and transforming growth factor- α (TGF- α) and the proinflammatory cytokine TNF- α in primary human epidermal keratinocytes cultured in the presence of low micromolar concentrations of sodium arsenite. Treatment with sodium arsenite resulted in a significant increase in cell proliferation, as indicated by increases in cell numbers, *c-myc* gene expression, and incorporation of [³H]-thymidine into cellular DNA. Studies of transcriptional regulation indicate that the rate of GM-CSF mRNA transcription is increased, while the elevated TGF- α is likely the result of message stabilization. While a number of cytokine regulatory networks exist in the skin, studies utilizing neutralizing antibodies against the growth factors indicate that inhibition of the arsenic-induced increase in TGF- α results in a corresponding decrease in the gene expression and secretion of GM-CSF. These results demonstrate that growth-promoting cytokines and growth factors are induced in keratinocytes following treatment with arsenic and could play a significant role in arsenic-induced skin cancer [85].

21.15.2 Effects on Human T-Cell Functional Responses

Previous workers have reported impaired T-cell functional responses in arsenic-exposed subjects. A study on 11 exposed and 13 control individuals in Mexico indicated that chronic exposure to arsenic via drinking water ($390 \mu g/L$) significantly decreased the phytohemagglutinin (PHA)-induced proliferative response of lymphocytes [86]. A later study showed that arsenic exposure leads to a diminution of PHA-stimulated T-cell proliferation in humans [87]. In that study, the evaluation of the peripheral blood lymphocyte proliferation was done at different culture time intervals. The mitotic index (MI) and labeling index (LI) in 72 h cultures in the exposed subjects with mean arsenic level in drinking water at 412 µg/L showing arsenical skin lesions were significantly lower than in the unexposed individuals without lesions who drank water with a mean arsenic level of 37.2 µg/L. Correlations between LI and MI showed that progression from the initial S- to M-phase is altered in exposed individuals [84].

Arsenic was evaluated on the expression of CD25 and IL-2 secretion in human peripheral blood mononuclear cells by proliferation assay [86]. Most of the donors did not show any change in the expression of CD25, but IL-2 secretion was inhibited in six of the seven donors tested. The determination of the mRNA of IL-2 and the intracellular IL-2 levels demonstrated that the inhibition was not at the transcriptional level. Electron microscopy studies revealed that cellular ultrastructure in Golgi apparatus, mitochondria, cytoskeleton, and perinuclear membrane was altered due to arsenic. These alterations suggested that sodium arsenite affected the cytoskeleton and the intracellular secretion of proteins including IL-2, leading to an impaired proliferation of the T cells when stimulated with PHA [88] (see Table 21–1).

21.16 Immunotoxic Effects on Murine and Human Monocytes/Macrophages

Inorganic arsenicals, both arsenite and arsenate, are strongly toxic to macrophages, and the concentration that decreased the number of surviving cells to 50% of that in untreated controls (IC_{50}) was 5 and 500 µM, respectively [89]. These inorganic arsenicals mainly caused necrotic cell death of about 80% cells, while 20% underwent apoptotic cell death. The inorganic arsenicals also induced marked release of an inflammatory cytokine TNF- α , at cytotoxic doses. The strong cytotoxicity of arsenite might be mediated via reactive oxygen and protease activation because it was inhibited by antioxidants superoxide dismutase (SOD), catalase, and reduced glutathione (GSH), or by a peptide inhibitor of interleukin-1 β -converting enzyme (ICE). It was suggested that these immunotoxic effects of inorganic arsenicals might evoke both immunosuppression and inflammation, and they might be central factors causing carcinogenesis and severe inflammatory responses, such as hepatomegaly and splenomegaly, in chronic arsenicosis patients drinking arsenic-contaminated well water [89].

It was found that arsenite sensitively inhibited the colony-stimulating factor (CSF)-induced *in vitro* maturation of monocytes into macrophages at nM levels, and it also induced small, non-adhesive and CD14⁺ abnormal macrophage generation from monocytes with granulocyte-macrophage CSF (GM-CSF) at 50–500 nM without cell death. The addition of other

Immunotoxic Effects	Arsenic/Study Place	Reference
Impaired PHA-induced T-cell proliferation	DW: 390 µg/L, Mexico	[86]
Altered replication of lymphocytes from the S- to M-phase	DW: 412 µg/L, Mexico	[87]
Overexpression of growth factors in human epidermal keratinocytes	Na-arsenite: low μ M	[85]
Higher prevalence rate of arsenical skin lesions with clear dose-response relationship	DW: 10–2040 µg/L, Bangladesh	[17]
Ultrastructure in Golgi apparatus, mitochondria, cytoskeleton and perinuclear membrane altered; intracellular secretion of IL-2 affected	Sodium arsenite	[88]
Neuropathy in adults; increases fetal loss and premature delivery in women	DW: >300 μg/L, India	[100]
Associated with poor nutritional status; lower WBC counts; causes neutropenia and lymphocytosis	DW: 218µg/L, ME: 7.6yrs, Bangladesh	[33]
CD4 ⁺ Th cells more sensitive than CD8 ⁺ Tc cells in PHA-activated PBMCs	Sodium arsenite, $1 \mu M$	[56]
Induces oxidative stress by producing nitric oxide and superoxide anions in PBMCs and monocytes	AsU: 12.3–1411µg/g creatinine. Mexico	[38]
Mitochondria are a primary target in arsenic-induced genotoxic response in human-hamster hybrid A, cells	Na-arsenite: $1-2 \mu g/mL$	[41]
Inhibition of GM-CSF-induced maturation of monocytes into macrophages; caused nonadhesive, small, abnormal CD14 ⁺ macrophage generation	Na-arsenite: 50–500 nM	[90]
Rapid cell rounding of macrophages with loss of adhesion at non-toxic arsenic dose	Arsenic trioxide, As_2O_3 : 1 μM	[91]
Inhibition of lymphocyte proliferation in PHA; inhibition of IL-2 secretion	DW: 412 µg/L, Mexico	[94]
Increases respiratory diseases - asthma, bronchitis, cough; significantly elevated levels of IgG. IgE. and IgA in serum	DW: 216µg/L, ME: 7.4yrs, Bangladesh	[106]
Significant decrease in levels of IL-2, IL-4, IL-5, IL-10, TNF- α and IFN- γ secreted by T cells: increases opportunistic infections	DW; skin lesions, Kolkata, India	[95]
Decreases IL-7 and lactoferrin in breast milk; increases ARI in infants	DW; maternal AsU high, Bangladesh	[96]
Increases lower respiratory tract infection and diarrhea in infants	Maternal AsU high, Bangladesh	[101]
Lower complement-mediated bactericidal activity	DW: 216 μg/L, AsU: 223 μg/L, Bangladesh	[110]

Table 21–1Immunotoxic Effects of Arsenic Exposure in HumanPopulations / Human Cells

DW; drinking water; AsU: arsenic in urine; ME: mean exposure to As; ARI: acute respiratory infection.

metallic compounds including chromium, selenium, mercury, cadmium, nickel, copper, zinc, cobalt, manganese, and other pentavalent arsenic metabolites, such as inorganic arsenate, monomethylarsonic acid and dimethylarsinic acid, could not induce the same abnormal cell generation from monocytes with CSFs at any concentration and any additional time schedules; they showed only simple cytolethality in monocytes and macrophages at nM–mM levels accompanied by cell death [90].

21.16.1 Effects on Monocyte/Macrophage Functional Responses

Treatment of human macrophages with a non-cytotoxic concentration of 1μ M arsenic trioxide (As₂O₃) induced a rapid cell rounding and a subsequent loss of adhesion [89]. These morphologic alterations were associated with a marked reorganization of actin cytoskeleton, which included retraction of peripheral actin extensions and formation of a cortical actin ring. In addition, As₂O₃ reduced expression of various macrophagic surface markers, enhanced that of the monocytic marker CD14, and altered both endocytosis and phagocytosis. Exposure of macrophages to the metalloid also strongly potentiated expression of TNF- α and IL-8 induced by LPS; like monocytes, As₂O₃-treated macrophages can be differentiated into dendritic-like cells. Impairment of macrophage function by As₂O₃ mainly resulted from activation of a RhoA/Rhoassociated kinase pathway; indeed, pretreatment of macrophages with the Rho-associated kinase inhibitor Y-27632 prevented the metalloid effects on cytoskeleton and phagocytosis. These findings suggested that human macrophages constitute sensitive targets of inorganic arsenic, which may contribute to immunotoxicity on these cells [91].

Impaired functions were also observed in monocyte-derived macrophages from exposed subjects with skin lesions (n = 70) compared to the unexposed individuals (n = 64) in another study conducted in West Bengal, India, in which the macrophages of arsenic-exposed individuals showed cell rounding with a significant (p < 0.001) loss of adhesion capacity, together with decreased nitric oxide production, and impaired phagocytic function. There was decreased CD54 and F-actin expression. Additionally, chronic arsenic exposure affected the RhoA-ROCK pathway, which in turn impaired macrophage functions [92]. These combined effects could contribute significantly to arsenic-induced immunosuppression observed in the exposed individuals.

21.16.2 Impairment of Macrophage-Functional Genes

Using a pan-genomic microarray, Bourdonnay et al. [93] found that exposure of human blood monocyte-derived macrophages to $1 \mu M As_2O_3$ for 72 h, a non-cytototoxic concentration, resulted in up-regulation of 32 genes and repression of 91 genes. Among these genes, 26 are specifically related to a differentiation program of human macrophages. In particular, As_2O_3 strongly alters expression of *MMP9*, *MMP12*, *CCL22*, *SPON2*, and *CXCL2* genes, which contribute to major macrophagic functions. It was interesting to note that most of the effects of the major environmental contaminant were reversed when As_2O_3 -treated macrophages were next cultured in arsenic-free medium. Further, it was found that As_2O_3 similarly regulated expression of this macrophagic gene subset in human alveolar macrophages, the phenotype of which closely resembles that of blood monocyte-derived macrophages. This study demonstrates that environmentally relevant concentrations of As_2O_3 impair expression of macrophage-specific genes, which fully supports the theory of the interference of the metalloid with the differentiation program of human macrophages [93].

21.17 Immunotoxic Effects on Murine Mononuclear Cells

The effect of arsenic exposure on the activation of splenic mononuclear cells (SMCs) in male CD57BL6N mice was evaluated [27]. Intra-gastric exposure to arsenic (as sodium arsenite) for 30 days (1, 0.1, or 0.01 mg/kg/day) reduced the proportion of CD4⁺ cells and the CD4⁺/ $CD8^+$ ratio in the spleen, increasing the proportion of $CD11b^+$ cells. Arsenic exposure did not modify the proportion of B cells. SMC showed an increased level of phosphorylation of lck and fyn kinases (first kinases associated with TCR complex when activated). Although normal levels of apoptosis were observed on freshly isolated SMC, an increase in apoptotic cells related with the increase in phosphorylation of lck and fyn was observed when SMCs were activated with Con A. Arsenic exposure reduced the proliferative response of SMCs to Con A, and also reduced secretion of IL-2, IL-6, IL-12, and IFN-y. No effect was observed on IL-4 and IL-10 secretion. The same effects were observed when SMCs of exposed animals were activated with anti-CD3/CD28 antibodies for 24 h, but these effects were recovered after 72 h of stimulation. These findings suggest that repeated and prolonged exposure to arsenic alters cell populations and produces functional changes depending on the specific activation pathway, and could be related to the phosphorylation status of lck and fyn kinases [28]. Further, arsenic exposure in mice inhibits T cell proliferation and macrophage activity, decreases CD4⁺ T cells in the spleen, and down-regulates contact hypersensitivity response to 2,4-dinitrofluorobenzene (DNFB) [94].

21.18 Decreased Cytokine Production by Human T Cells

It has been found that arsenic can produce an inhibition of IL-2 secretion in human lymphocytes *in vitro*. In a study conducted on Mexican adults, exposure to 412 µg/L of arsenic in drinking water was found to inhibit lymphocyte proliferation in response to PHA. Arsenic exposure increases the incidence of autoimmune-mediated diseases such as diabetes mellitus and other diseases related to immunosuppression such as the presence of skin cancer [94]. In a study conducted in Kolkata, India, the effect of arsenic on T-cell proliferation and cytokine secretion in 20 individuals with arsenic-induced skin lesions was investigated and compared with the results in 18 arsenic-unexposed individuals. There was a significant dose-dependent suppression of Con A-induced T-cell proliferation in the arsenic-exposed individuals compared with the unexposed individuals (p < 0.001). This correlated with a significant decrease in the levels of secreted cytokines by the T cells (IL-2, IL-4, IL-5, IL-10, TNF- α , and IFN- γ) in the exposed individuals (p < 0.001). Thus, it can be inferred that arsenic exposure can cause immunosuppression in humans [95].

21.19 Effects of *In Utero* Exposure on Infant Immune System

Little is known about early-life effects of arsenic on immunity; the impact of in utero exposure of arsenic on infant immune parameters and morbidity was evaluated in a pilot study in Matlab, a rural area of Bangladesh. Pregnant women (n = 140) extensively affected by arsenic from tubewell water were enrolled at 6-10 weeks of gestation. Maternal urinary arsenic during pregnancy showed significant negative correlation with interleukin-7 (IL-7) and lactoferrin (Ltf) in breast milk and child thymic index (TI). Urinary arsenic was also positively associated with fever and diarrhea during pregnancy and acute respiratory infections (ARIs) in the infants. The effect of arsenic exposure on ARI was only evident in male children. The findings suggest that *in utero* arsenic exposure impaired child thymic development and enhanced morbidity, probably via immunosuppression [96]. Because of its ability to readily cross the placenta [97], arsenic could potentially impair fetal development; indeed, gestational arsenic exposure is linked to increased fetal loss and infant mortality in birth cohort studies in Bangladesh [98,99]. The effect of *in utero* arsenic exposure seemed to be partially gender dependent, and arsenic exposure also affected breast milk content of trophic factors and maternal morbidity [96]. In an evaluation of arsenic exposure among the villagers in the Middle Ganges Plain, Bihar, India, where 57% of the 206 tubewells supply water with arsenic concentrations exceeding $50 \mu g/L$, with about 20% containing $>300 \,\mu$ g/L, arsenic-typical neuropathy was diagnosed in 63% of the adults, and there was apparent increase in fetal loss and premature delivery in the women with the highest concentrations of arsenic in their drinking water [100].

There may be an association between prenatal arsenic exposure and increased risk of infant mortality caused by infectious diseases, but no epidemiological study was available until recently in support of or against this observation. A recent prospective population-based cohort study was conducted in the arsenic hyperendemic region of Matlab, Bangladesh, that included 1552 live-born infants of women exposed to arsenic. The concentrations of metabolites of inorganic arsenic were determined in maternal urine samples collected at gestational weeks 8 and 30, and information on symptoms of lower respiratory tract infection (LRTI) and diarrhea in infants was collected during a 12-month follow-up period. The estimated risk of LRTI and severe LRTI increased by 69% [adjusted relative risk (RR) = 1.69; 95% confidence interval (CI), 1.36-2.09)] and 54% (RR = 1.54; 95% CI, 1.21-1.97), respectively, for infants of mothers with urinary arsenic concentrations in the highest quintile (average of arsenic concentrations measured in early and late gestation, 262-977 µg/L) relative to those with exposure in the lowest quintile (<39 µg/L). The corresponding figure for diarrhea was 20% (RR = 1.20; 95% CI, 1.01-1.43). These observations suggested that arsenic exposure during pregnancy was associated with increased morbidity of infectious diseases during infancy [101].

21.20 Gender-Related Immunotoxic Effects in Human

To evaluate whether arsenic effects are gender related, the *in vitro* toxicity of sodium arsenite on human T lymphocyte subpopulations from men and women were investigated. Peripheral

blood mononuclear cells (PBMCs) obtained from healthy young men and women were treated with sodium arsenite (0.01, 0.1, and 1 μ M). The cell viability, cell proliferation, and the proportion of CD4⁺ Th and CD8⁺ Tc cells after 48 or 72 h of arsenic exposure in resting and PHA-activated PBMC were assessed. It was observed that sodium arsenite at 1 μ M was more toxic for Th than for Tc cells in PBMCs from women. In addition, T lymphocytes from women were more affected by the cell proliferation inhibition induced by arsenic, suggesting that women could be more susceptible to the immunotoxic effects caused by arsenic exposure [56]. Gender differences in the biotransformation of arsenic by methylation have been reported, and men seem to be more affected by arsenic-related skin effects than are women. Experimental studies indicate major gender differences in arsenic-induced cancer [60]. Also, tuberculosis mortality rate ratios in arsenic-exposed men were found significantly higher for region II of Chile compared with findings in As exposed women of the same region during the years 1982–1986 (deaths: 359 among men and 95 among women) [102]. Maternal exposure to arsenic during gestation has been found to be associated with ARI, and the effects have been more prevalent in male than in female children [96].

21.21 Effects of Chronic Exposure on Immune Response

21.21.1 Effects of Chronic Exposure on Mice and Human Lungs

Exposure to environmentally relevant concentrations of arsenic is known to be associated with an increased risk of lung disease, making As a unique toxicant, because lung toxicity is usually associated with inhalation rather than ingestion. To study the changes in the lungs, C57BL/6J mice fed a casein-based AIN-76A defined diet were exposed to 10 or 100 ppb arsenic in drinking water or food for 5–6 weeks. The whole genome transcriptome profiling of mice lungs revealed significant alterations in the expression of many genes with functions in cell adhesion and migration, channels, receptors, differentiation, and proliferation that are associated with the innate immune response. Confirmation of mRNA and protein expression changes revealed that genes for interleukin-1 β , interleukin-1 receptor, a number of Toll-like receptors, and several cytokines and cytokine receptors were significantly altered in the lungs of arsenic-exposed mice. These findings indicate that chronic low dose arsenic exposure can elicit effects on the regulation of innate immunity, which may contribute to altered disease risk, particularly in the lung [103]. Epidemiological studies carried out on 7683 subjects in West Bengal showed that the prevalence of non-malignant pulmonary effects were markedly increased for participants with arsenic-induced skin lesions who also had high levels of arsenic in their drinking water [104].

To investigate whether arsenic exposure is a significant contributor to the increased risk of lung disease, C57BL/6J mice were exposed to 100 ppb arsenic in drinking water for 5 weeks, followed by intranasal inoculation with a sublethal dose of respiratory influenza A (H1N1) virus. Arsenic was associated with a number of significant changes in response to influenza, including an increase in morbidity and higher pulmonary influenza virus titers on day 7 post-infection. There was a decrease in the number of dendritic cells in the mediastinal lymph
nodes early in the course of infection. The study findings indicate that chronic arsenic exposure significantly compromises the immune response to infection. Alterations in response to repeated lung infection may also contribute to other chronic illnesses, such as bronchiectasis, which is elevated by arsenic exposure in epidemiological studies [105]. Similarly to this observation, high prevalence of respiratory diseases including asthma and bronchitis has been reported among Bangladeshi subjects suffering from drinking-water arsenic toxicity [106].

21.21.2 Effects of Chronic Exposure on Humoral Immune Response in Humans

In a cross-sectional epidemiological study, the humoral immune response was evaluated in Bangladeshi subjects (n = 125) chronically exposed to drinking water arsenic by measuring serum immunoglobulin profiles [106]. The mean duration of exposure in the patients was 7.4 ± 5.3 years, and the levels of arsenic in the drinking water and urine samples were 216 ± 211 and $223 \pm 302 \mu g/L$, respectively, compared to 11 ± 20 and $29 \pm 19 \mu g/L$, respectively, in the unexposed subjects. The exposed subjects had highly significantly elevated levels of IgG and IgE compared to the control unexposed subjects. The mean level of IgG in the arsenicosis patients was $24.3 \pm 7.5 \, g/L$ while the value in the controls was $13.8 \pm 7.8 \, g/L$ (p < 0.001), and the mean level of IgE in the patients was $600 \pm 265 \, IU/mL$ compared to $186 \pm 221 \, IU/mL$ in the control subjects (p < 0.001). The levels of IgA were significantly higher ($2.7 \pm 1.5 \, g/L$ vs. $2.2 \pm 0.7 \, g/L$, p < 0.005) but levels of IgM were similar to those in the controls [106]. In contrast to the findings of this study, no abnormalities were detected in the serum immunoglobulin IgG, IgA, or IgM levels in workers (n = 47) exposed to arsenic through inhalation in a coal-burning power plant in an earlier study [107]. However, the levels of arsenic or duration of exposure were not measured in this study and they might have been too small to cause significant damage to the immune system.

It has been revealed after analysis of the clinical symptoms based on skin manifestations that the levels of both IgG and IgE were significantly elevated during the initial stages while IgE were further elevated with the duration of arsenic exposure [106]. It might be possible that arsenical skin lesions (dermatosis) caused by chronic exposure played a significant role in eliciting highly elevated levels of serum IgG and IgE in the exposed subjects. Higher incidences of opportunistic infections are found in arsenic-exposed individuals, indicating that their immune systems may be functionally impaired [95]. It is expected that a healthy and functional immune system would be necessary to provide adequate immune protection against tumor cells and an adequate response to infectious agents.

21.22 Association with Respiratory Complications and Impaired Immune Responses

It has been reported that subjects chronically exposed to arsenic in drinking water have higher prevalence of respiratory diseases (60%), gastric and abdominal pain (59%), weakness (33%), and itchy rash (28%), among other complications [33]. Prevalence of respiratory effects has been reported in other studies conducted in West Bengal, India [104,108], and in

Bangladesh [109]. The relationship between chronic arsenic exposure through drinkingwater and respiratory complications and the immune response was assessed in Bangladeshi subjects chronically exposed to arsenic [106]. The respiratory symptoms included breathing problems, chest sounds, asthma, bronchitis, and cough. Among the exposed subjects with arsenicosis, the mean serum IgE level was 706 \pm 211 IU/mL in those with respiratory complications compared to 542 \pm 241 IU/mL in those without apparent involvement with the respiratory system (p < 0.01). The eosinophil counts in the exposed population did not differ significantly from the counts in the control unexposed subjects, indicating that elevated levels of serum IgE might not be due to allergic diseases; rather they could be due to direct effects of arsenic. There were significant linear relationships between the levels of serum IgE and inorganic phosphorus (p < 0.05), and the levels of serum IgA, and urinary excretion of arsenic (p < 0.001). These observations suggested that arsenic toxicity caused respiratory complications, induced changes in the humoral as well as mucosal immune responses.

21.23 Effects of Chronic Exposure on Serum Complement Function

Functions of serum complement were evaluated in B6C3F1 female mice following exposure of gallium arsenide as a single intratracheal instillation at doses of 50, 100, and 200 mg/kg. Fourteen days after exposure, the levels of the serum complement protein C3 were measured; they were increased by as much as 16%. There was no significant change in the classical pathway hemolytic activity as measured by CH50 levels [31]. However, the effect on the complement system of a single inhalation exposure to arsenic has not been studied in other animals to compare these results.

In a recent cross-sectional study, serum complement function was evaluated in Bangladeshi subjects (n = 125) chronically exposed to drinking-water arsenic, by measuring bactericidal activity [110]. The mean levels of arsenic in drinking water and urine samples of the exposed subjects were 216 ± 211 and $223 \pm 302 \mu g/L$, respectively, and the mean duration of exposure was 7.4 ± 5.3 years. The exposed subjects had significantly lower complement-mediated bactericidal activity than the unexposed controls (p < 0.01). The levels of complement C4 in about 20% of the exposed subjects were above normal. In the exposed subjects, the mean complement C3 was $1.56 \, g/L$, and C4 was $0.29 \, g/L$ compared to $1.68 \, g/L$ and $0.25 \, g/L$, respectively, in the unexposed controls. The mean IgG level in the exposed subjects was $24.3 \, g/L$ and was highly significantly elevated (p < 0.001). The exposed subjects showed a significant direct correlation between C3 and bactericidal activity (p = 0.014); this observation together with elevated levels of C4 indicated underutilization and possibly impaired activity of the classical complement pathway in drinking-water arsenic toxicity [110].

21.24 Conclusions

Studies carried out on *in vivo* and *in vitro* systems indicate that arsenic can act as a modulator of the body's immune defense system that could render the host immunocompromised.

A suboptimal or non-protective immune system could pose increased risk of infections and development of several cancers as observed in subjects chronically exposed to arsenic. Exposure of experimental animals to arsenic via drinking water or through nasal challenge resulted in alterations of the cellular and cell-mediated immunity that generally agree with immunological outcomes in humans. Exposures to very low doses (<10 ppb) could elicit significant effects on innate immune mediators. Some studies evaluated the effects of the metalloid on cultured murine and human cells. However, only few experiments have been done to study the effects of arsenic on the humoral immune response. More work needs to be done for better understanding the risk of human immunotoxicity. Some affected subjects in different parts of the world have been consuming arsenic-contaminated water for some 10-20 years or more. It would be difficult to construct animal models to study such long-term chronic effects. Moreover, inconsistencies in epidemiological findings, possibly due to differences in dose, sampling method, genetic background, and environmental/nutritional factors, as well as variation in arsenic compounds present in natural sources, indicate the need for larger studyparticipant numbers and diverse ethnic populations. Also, populations experiencing variations in exposure levels should be evaluated for better understanding of dose-dependent relationships. Comprehensive data will be critical for identifying the potential molecular targets of arsenic to elucidate mechanisms of immunotoxicity.

References

- [1] World Health Organization. Environmental health criteria 18. Arsenic. Geneva: WHO; 1981.
- [2] Thornton I, Farago M. The geochemistry of arsenic. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic: exposure and health effects. London: Chapman and Hall; 1997. pp. 1–16.
- [3] Rahman MM, Chowdhury UK, Mukherjee SC, Mondal BK, Paul K, Lodh D, et al. Chronic arsenic toxicity in Bangladesh and West Bengal, India—a review and commentary. Clin Toxicol 2001;39:683–700.
- [4] Le XC, Ma M, Cullen WR, Aposhian HV, Lu X, Zheng B. Determination of monomethylarsonous acid, a key arsenic methylation intermediate, in human urine. Environ Health Perspect 2000;108:1015–18.
- [5] Yamauchi H, Fowler BA. Toxicity and metabolism of inorganic and methylated arsenicals. In: Nriagu JO, editor. Arsenic in the environment. Part II: human health and ecosystem effects. New York: John Wiley & Sons Inc.; 1994. pp. 35–53.
- [6] Vahter M, Concha G. Role of metabolism in arsenic toxicity. Pharmacol Toxicol 2001;89:1-5.
- [7] Reichard JF, Schnekenburger M, Puga A. Long term low-dose arsenic exposure induces loss of DNA methylation. Biochem Biophys Res Commun 2007;352:188–92.
- [8] National Research Council (NRC). Arsenic in drinking water. Washington, DC: National Academy Press; 1999.
- [9] Vega L, Soto G, Luna A, Acosta L, Conde P, Cebrian M, et al. Early signs of immunodeppression induced by arsenic in children. In: Bundschuh J, Armienta MA, Bhattacharya P, Matschullat J, Birkle P, Rodriguez R, editors. Natural arsenic in groundwaters of Latin America. Rotterdam: AA Balkema Publishers, Taylor & Francis Group; 2009. pp. 435–6.
- [10] Maquieira Á, Brun EM, Garcés-García M, Puchades R. Aluminum oxide nanoparticles as carriers and adjuvants for eliciting antibodies from non-immunogenic haptens. Anal Chem 2012;84:9340–8.
- [11] Ferencík M, Ebringer L. Modulatory effects of selenium and zinc on the immune system. Folia Microbiol (Praha) 2003;48:417–26.

- [12] De La Fuente H, Portales-Perez D, Baranda L, Diaz-Barriga F, Saavedra-Alanis V, Layseca E, et al. Effect of arsenic, cadmium and lead on the induction of apoptosis of normal human mononuclear cells. Clin Exp Immunol 2002;129:69–77.
- [13] Florea A-M, Büsselberg D. Occurrence, use and potential toxic effects of metals and metal compounds. Biometals 2006;19:419–27.
- [14] Welch AH, Lico MS, Hughes JL. Arsenic in ground water of the Western United States. Groundwater 1988;26:333-47.
- [15] Welch AH, Westjohn DB, Helsel DR, Wanty RB. Arsenic in ground water of the United States: occurrence and geochemistry. Groundwater 2000;38:589–604.
- [16] Vega L, Styblo M, Patterson R, Cullen W, Wang C, Germolec D. Differential effects of trivalent and pentavalent arsenicals on cell proliferation and cytokine secretion in normal human epidermal keratinocytes. Toxicol Appl Pharmacol 2001;172:225–32.
- [17] Tondel M, Rahman M, Magnuson A, Chowdhury IA, Faruquee MH, Ahmad SA. The relationship of arsenic levels in drinking water and the prevalence rate of skin lesions in Bangladesh. Environ Health Perspect 1999;107:727–9.
- [18] Chowdhury UK, Biswas BK, Roy Chowdhury T, Mandal BK, Samanta G, Basu GK, et al. Ground water arsenic contamination and sufferings of people in West Bengal, India and Bangladesh. In: Roussel A.M., Anderson RA, Favier AE, editors. Trace elements in man and animals, vol. 10. New York: Plenum Corporation; 2000. pp. 645–50.
- [19] Tseng CH. Metabolism of inorganic arsenic and non-cancerous health hazards associated with chronic exposure in humans. J Environ Biol 2007;28:349–57.
- [20] Nabi AHMN, Rahman MM, Islam LN. Evaluation of biochemical changes in chronic arsenic poisoning among Bangladeshi patients. Int J Environ Res Public Health 2005;2:385–93.
- [21] Armstrong CW, Stroube RB, Rubio T, Siudyla EA, Miller GB. Outbreak of fatal arsenic poisoning caused by contaminated drinking water. Arch Environ Health 1984;39:276–9.
- [22] Tay CH, Seah CS. Arsenic poisoning from anti-asthmatic herbal preparations. Med J Aus 1975;2:424-8.
- [23] Lerman BB, Ali N, Green D. Megaloblastic, dyserythropoietic anemia following arsenic ingestion. Ann Clin Lab Sci 1980;10:515–17.
- [24] Harrington JM, Middaugh JF, Morse DL, Housworth J. A survey of a population exposed to high concentrations of arsenic in well water in Fairbanks, Alaska. Am J Epidemiol 1978;108:377–85.
- [25] Morse DL, Harrington JM, Housworth J, Landrigan PJ, Kelter A. Arsenic exposure in multiple environmental media in children near a smelter. Clin Toxicol 1979;14:389–99.
- [26] Rezuke WN, Anderson C, Postuszak WT, Conway SR, Firshein SI. Arsenic intoxication presenting as a myelodysplastic syndrome: a case report. Am J Hematol 1991;36:291–3.
- [27] Gyasi SF, Awuah E, Larbi JA, Koffuor GA, Osei O-A. Clinical, hematological and histopathological responses to arsenic toxicity in ICR mice using arsenic levels synonymous to Buruli ulcer endemic communities in the Amansie West District of Ghana. Eur J Exp Biol 2012;3:683–9.
- [28] Soto-Peña GA, Vega L. Arsenic interferes with the signaling transduction pathway of T cell receptor activation by increasing basal and induced phosphorylation of Lck and Fyn in spleen cells. Toxicol Appl Pharmacol 2008;230:216–26.
- [29] Lantz RC, Parliman G, Chen GJ, Carter DE. Effect of arsenic exposure on alveolar macrophage function. I. Effect of soluble As(III) and As(V). Environ Res 1994;67:183–95.
- [30] Burns LA, Sikorski EE, Saady JJ, Munson AE. Evidence for arsenic as the immunosuppressive component of gallium arsenide. Toxicol Appl Pharmacol 1991;110:157–69.
- [31] Sikorski EE, McCay JA, White Jr KL, Bradley SG, Munson AE. Immunotoxicity of the semiconductor gallium arsenide in female B6C3F1 mice. Fundam Appl Toxicol 1989;13:843–58.

- [32] Burns LA, Munson AE. Reversal of gallium arsenide-induced suppression of the antibody-forming cell response by vehicle supernatants. II. Nature and identification of reversing factors. J Pharmacol Exp Ther 1993;265:150–8.
- [33] Islam LN, Nabi AHMN, Rahman MM, Khan MA, Kazi AI. Association of clinical complications with nutritional status and the prevalence of leukopenia among arsenic patients in Bangladesh. Int J Environ Res Public Health 2004;1:74–82.
- [34] Guha Mazumder DN, Das Gupta J, Santra A. Non-cancer effects of chronic arsenicosis with special reference to liver damage. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic: exposure and health effects. London: Chapman and Hall; 1997. pp. 112–23.
- [35] Dickerson OB. Antimony, arsenic and their compounds. In: Zenz C, Dickerson Jr. OB, Horwath EP, editors. Occupational medicine. St-Louis: Mosby-Year Book Inc.; 1994. pp. 468–73.
- [36] Harvey SC. Heavy metals. In: Goodman LS, Gilman A, editors. The pharmacological basis of therapeutics. Toronto: Collier-Macmillan; 1970. pp. 958–65.
- [37] Wang TS, Kuo CF, Jan KY, Huang H. Arsenite induces apoptosis in Chinese Hamster ovary cells by generation of reactive oxygen species. J Cell Physiol 1996;169:256–68.
- [38] Luna AL, Acosta-Saavedra LC, Lopez-Carrillo L, Conde P, Vera E, De Vizcaya- Ruiz A, et al. Arsenic alters monocyte superoxide anion and nitric oxide production in environmentally exposed children. Toxicol Appl Pharmacol 2010;12:244–51.
- [39] Flora SJ. Arsenic-induced oxidative stress and its reversibility. Free Radic Biol Med 2011;51:257–81.
- [40] Chen Y-C, Lin-Shiau S-Y, Lin J-K. Involvement of reactive oxygen species and caspase 3 activation in arsenite-induced apoptosis. J Cell Physiol 1998;177:324–33.
- [41] Liu Su-X, Davidson MM, Tang X, Walker WF, Athar M, Ivanov V, et al. Mitochondrial damage mediates genotoxicity of arsenic in mammalian cells. Cancer Res 2005;65:3236–42.
- [42] Mishra CS, Mani V, Kaur H. Effect of arsenic on immunity, oxidative enzyme and various hematological parameters in cross bred calves. Asian-Aust J Anim Sci 2005;18:497–501.
- [43] ICAR. Nutrient requirements of domestic animals. New Delhi, India: Indian Council of Agricultural Research; 1998. 11pp.
- [44] Islam MS, Awal MA, Mostofa M, Begum F, Khair A, Myenuddin M. Effect of spirulina on toxic signs, body weight and hematological parameters in arsenic induced toxicities in ducks. Int J Poult Sci 2009;8:75–9.
- [45] Woods JS, Fowler BA. Altered regulation of mammalian hepatic heme biosynthesis and urinary porphyrin excretion during prolonged exposure to sodium arsenate. Toxicol Appl Pharmacol 1978;43:361–71.
- [46] Guha Mazumder DN. Effect of chronic intake of arsenic-contaminated water on liver. Toxicol Appl Pharmacol 2005;206:169–75.
- [47] Garcia AMT, Dueñas HA, Pineda PJ. Hematological effects of arsenic in rats after subchronical exposure during pregnancy and lactation: the protective role of antioxidants. Exp Toxicol Pathol 2013;65:609–14.
- [48] Hermann AC, Kim CH. Effects of arsenic on zebrafish innate immune system. Mar Biotechnol (NY) 2005;7:494–505.
- [49] Nayak AS, Lage CR, Kim CH. Effects of low concentrations of arsenic on the innate immune system of the zebrafish (*Danio Rerio*). Toxicol Sci 2007;98:118–24.
- [50] Ghosh D, Datta S, Bhattacharya S, Mazumder S. Long-term exposure to arsenic affects head kidney and impairs humoral immune responses of *Clarias batrachus*. Aquat Toxicol 2007;81:79–89.
- [51] Datta S, Ghosh D, Saha DR, Bhattacharaya S, Mazumder S. Chronic exposure to low concentration of arsenic is immunotoxic to fish: role of head kidney macrophages as biomarkers of arsenic toxicity to *Clarias batrachus*. Aquat Toxicol 2009;92:86–94.
- [52] Schulz H, Nagymajtényi L, Institoris L, Papp A, Siroki O. A study on behavioral, neurotoxicological, and immunological effects of subchronic arsenic treatment in rats. J Toxicol Environ Health, Part A. 2002;65:1181–93.

- [53] Savabieasfahani M, Lochmiller RL, Rafferty DP, Sinclair JA. Sensitivity of wild cotton rats (Sigmodon hispidus) to the immunotoxic effects of low-level arsenic exposure. Arch Environ Contam Toxicol 1998;34:289–96.
- [54] Patterson R, Vega L, Trouba K, Bortner C, Germolec D. Arsenic-induced alterations in the contact hypersensitivity response in Balb/c mice. Toxicol Appl Pharmacol 2004;198:434–43.
- [55] Meza MM, Kopplin MJ, Burgess JL, Gandolfi AJ. Arsenic drinking water exposure and urinary excretion among adults in the Yaqui Valley, Sonora, Mexico. Environ Res 2004;96:119–26.
- [56] Vega L, Montes de Oca P, Saavedra R, Ostrosky-Wegman P. Helper T cell subpopulations from women are more susceptible to the toxic effect of sodium arsenite in vitro. Toxicology 2004;199:121–8.
- [57] Ghosh P, Basu A, Mahata J, Basu S, Sengupta M, Das JK, et al. Cytogenetic damage and genetic variants in the individuals susceptible to arsenic-induced cancer through drinking water. Int J Cancer 2006;118:2470–8.
- [58] Heck JE, Gamble MV, Chen Y, Graziano JH, Slavkovich V, Parvez F, et al. Consumption of folate-related nutrients and metabolism of arsenic in Bangladesh. Am J Clin Nutr 2007;85:1367–74.
- [59] Vahter ME. Interactions between arsenic-induced toxicity and nutrition in early life. J Nutr 2007;137:2798–804.
- [60] Vahter M, Åkesson A, Lidén C, Ceccatelli S, Berglund M. Gender differences in the disposition and toxicity of metals. Environ Res 2007;104:85–95.
- [61] Mazumder DN. Chronic arsenic toxicity and human health. Indian J Med Res 2008;128:436-47.
- [62] Tchounwou PB, Wilson B, Ishaque A. Important considerations in the development of public health advisories for arsenic and arsenic-containing compounds in drinking water. Rev Environ Health 1999;14:211–29.
- [63] Yu HS, Lee CH, Chen GH. Peripheral vascular diseases resulting from chronic arsenical poisoning. J Dermatol 2002;29:123–30.
- [64] Andrew AS, Jewe DA, Mason RA, Whitfield ML, Moore JH, Karagas MR. Drinking-water arsenic exposure modulates gene expression in human lymphocytes from a US population. Environ Health Perspect 2008;116:524–31.
- [65] Rahman MM, Ng JC, Naidu R. Chronic exposure of arsenic via drinking water and its adverse health impacts on humans. Environ Geochem Health 2009;31:189–200.
- [66] Vega L. Effects of arsenic exposure on the immune system and mechanism of action. In: Gosselin JD, Fancher IM, editors. Environmental health risks. New York: Nova Scientific Publishers, Inc; 2009. [Chapter 10].
- [67] Del Razo LM, Arellano MA, Cebrián ME. The oxidation states of arsenic in well-water from a chronic arsenicism area of Northern Mexico. Environ Pollut 1990;64:143–53.
- [68] Coronado-Gonzalez JA, Del Razo LM, Garcia-Vargas G, Sanmiguel-Salazar F, Escobedo-dela Pena J. Inorganic arsenic exposure and type 2 diabetes mellitus in Mexico. Environ Res 2007;104:383–9.
- [69] Leke RJ, Oduma JA, Bassol-Mayagoitia S, Bacha AM, Grigor KM. Regional and geographical variations in infertility: effects of environmental, cultural, and socioeconomic factors. Environ Health Perspect 1993;101:73–80.
- [70] Rocha-Amador D, Navarro ME, Carrizales L, Morales R, Calderon J. Decreased intelligence in children and exposure to fluoride and arsenic in drinking water. Cad Saude Publica 2007;23:S579–87.
- [71] Brown C, Chute S. Arsenic concentrations in bedrock wells in Colchester, East Hampton, and Woodstock CT. US Geological Survey, Water Investigations Report 02, 4135; 2002.
- [72] Blum D. The arsenic in our drinking water. The New York Times; 2013. September 20.
- [73] Sakurai T, Kaise T, Matsubara C. Immunotoxicity of organic arsenic compounds in marine animals. Appl Organomet Chem 1996;10:727–32.

- [74] Sakurai T, Kojima C, Ochiai M, Ohta T, Fujiwara K. Evaluation of in vivo acute immunotoxicity of a major organic arsenic compound arsenobetaine in seafood. Int Immunopharmacol 2004;4:179–84.
- [75] Zhang P, Wang SY, Hu XH. Arsenic trioxide treated 72 cases of acute promyelocytic leukemia. Chin J Hematol 1996;17:58–62.
- [76] Sun HD, Ma L, Hu X-C, Zhang T-D. Ai-Lin 1 treated 32 cases of acute promyelocytic leukemia. Chin J Integrated Chin West Med 1992;12:170–2.
- [77] Niu C, Yan H, Sun HP, Liu J-X, Gu BW, Su X-Y, et al. Treatment of de novo and relapsed acute promyelocytic leukemia patients with arsenic trioxide. Blood 1998;92(Suppl. 1):678a.
- [78] Niu C, Yan H, Yu T, Sun H-P, Liu J-X, Li X-S, et al. Studies on treatment of acute promyelocytic leukemia with arsenic trioxide: remission induction, follow-up, and molecular monitoring in 11 newly diagnosed and 47 relapsed acute promyelocytic leukemia patients. Blood 1999;94:3315–24.
- [79] Jia PM, Zhu Q, Yu Y, Chen GQ, Chen SJ, Chen Z, et al. Experimental study of low dose arsenic trioxide in treatment of patients with acute promyelocytic leukemia. Chin J Cancer 2002;21:337–40. [Article in Chinese].
- [80] Soignet SL, Frankel SR, Douer D, Tallman MS, Kantarjian H, Calleja E, et al. United States multicenter study of arsenic trioxide in relapsed acute promyelocytic leukemia. J Clin Oncol 2001;19:3852–60.
- [81] Murgo AJ. Clinical trials of arsenic trioxide in hematologic and solid tumors: overview of the National Cancer Institute Cooperative Research and Development Studies. Oncologist 2001;6(Suppl. 2):22–8.
- [82] Miller Jr. WH, Schipper HM, Lee JS, Singer J, Waxman S. Mechanisms of action of arsenic trioxide. Cancer Res 2002;62:3893–903.
- [83] Tallman MS, Altman JK. How I treat acute promyelocytic leukemia. Blood 2009;114:5126–35.
- [84] Styblo M, Del Razo LM, Vega L, Germolec DR, LeCluyse EL, Hamilton GA, et al. Comparative toxicity of trivalent and pentavalent inorganic and methylated arsenicals in rat and human cells. Arch Toxicol 2000;74:289–99.
- [85] Germolec DR, Yoshida T, Gaido K, Wilmer JL, Simeonova PP, Kayama F, et al. Arsenic induces overexpression of growth factors in human keratinocytes. Toxicol Appl Pharmacol 1996;141:308–18.
- [86] Ostrosky-Wegman P, Gonsebatt ME, Montero R, Vega L, Barba H, Espinosa J, et al. Lymphocyte proliferation kinetics and genotoxic findings in a pilot study on individuals chronically exposed to arsenic in Mexico. Mutat Res 1991;12:477–82.
- [87] Gonsebatt ME, Vega L, Montero R, Garcia-Vargas G, Del Razo LM, Albores A, et al. Lymphocyte replicating ability in individuals exposed to arsenic via drinking water. Mutat Res 1994;313:293–9.
- [88] Vega L, Ostrosky-Wegman P, Fortoul TI, Díaz C, Madrid V, Saavedra R. Sodium arsenite reduces proliferation of human activated T-cells by inhibition of the secretion of interleukin-2. Immunopharmacol Immunotoxicol 1999;21:203–20.
- [89] Sakurai T, Kaise T, Matsubara C. Inorganic and methylated arsenic compounds induce cell death in murine macrophages via different mechanisms. Chem Res Toxicol 1998;11:273–83.
- [90] Sakurai T, Ohta T, Tomita N, Kojima C, Hariya Y, Mizukami A, et al. Evaluation of immunotoxic and immunodisruptive effects of inorganic arsenite on human monocytes/macrophages. Int Immunopharmacol 2006;6:304–15.
- [91] Lemarie A, Morzadec C, Bourdonnay E, Fardel O, Vernhet L. Human macrophages constitute targets for immunotoxic inorganic arsenic. J Immunol 2006;177:3019–27.
- [92] Banerjee N, Banerjee S, Sen R, Bandyopadhyay A, Sarma N, Majumder P, et al. Chronic arsenic exposure impairs macrophage functions in the exposed individuals. J Clin Immunol 2009;29:582–94.
- [93] Bourdonnay E, Morzadec C, Sparfel L, Galibert M-D, Jouneau S, Martin-Chouly C, et al. Global effects of inorganic arsenic on gene expression profile in human macrophages. Mol Immunol 2009;46:649–56.
- [94] Soto-Peña GA, Luna AL, Acosta-Saavedra L, Conde P, López-Carrillo L, Cebrián ME, et al. Assessment of lymphocyte subpopulations and cytokine secretion in children exposed to arsenic. FASEB J 2006;20:779–81.

- [95] Biswas R, Ghosh P, Banerjee N, Das JK, Sau T, Banerjee A, et al. Analysis of T-cell proliferation and cytokine secretion in the individuals exposed to arsenic. Hum Exp Toxicol 2008;27:381–6.
- [96] Raqib R, Ahmed S, Sultana R, Wagatsuma Y, Mondal D, Hoque AMW, et al. Effects of in utero arsenic exposure on child immunity and morbidity in rural Bangladesh. Toxicol Lett 2009;185:197–202.
- [97] Concha G, Vogler G, Lezcano D, Nermell B, Vahter M. Exposure to inorganic arsenic metabolites during early human development. Toxicol Sci 1998;12:185–90.
- [98] Rahman A, Vahter M, Ekstrom EC, Rahman M, Mustafa AHG, Wahed MA, et al. Association of arsenic exposure during pregnancy with fetal loss and infant death: a cohort study in Bangladesh. Am J Epidemiol 2007;12:1389–96.
- [99] Rahman A, Persson LA, Nermell B, El Arifeen S, Ekstrom EC, Smith AH, et al. Arsenic exposure and risk of spontaneous abortion, stillbirth, and infant mortality. Epidemiology 2010;12:797–804.
- [100] Chakraborti D, Mukherjee SC, Pati S, Sengupta MK, Rahman MM, Chowdhury UK, et al. Arsenic groundwater contamination in Middle Ganga Plain, Bihar, India: a future danger? Environ Health Perspect 2003;111:1194–201.
- [101] Rahman A, Vahter M, Ekström E-C, Persson L-Å. Arsenic exposure in pregnancy increases the risk of lower respiratory tract infection and diarrhea during infancy in Bangladesh. Environ Health Perspect 2011;119:719–24.
- [102] Smith AH, Marshall G, Yuan Y, Liaw J, Ferreccio C, Steinmaus C. Evidence from Chile that arsenic in drinking water may increase mortality from pulmonary tuberculosis. Am J Epidemiol 2011;173:414–20.
- [103] Kozul CD, Hampton TH, Davey JC, Gosse JA, Nomikos AP, Eisenhauer PL, et al. Chronic exposure to arsenic in the drinking water alters the expression of immune response genes in mouse lung. Environ Health Perspect 2009;117:1108–15.
- [104] Guha Mazumder DN. Arsenic and non-malignant lung disease. J Environ Sci Health, Part A. Tox Hazard Subst Environ Eng 2007;42:1859–67.
- [105] Kozul CD, Ely KH, Enelow RI, Hamilton JW. Low-dose arsenic compromises the immune response to influenza A infection in vivo. Environ Health Perspect 2009;117:1441–7.
- [106] Islam LN, Nabi AHMN, Rahman MM, Zahid MSH. Association of respiratory complications and elevated serum immunoglobulins with drinking water arsenic toxicity in human. J Environ Sci Health, Part A. Tox Hazard Subst Environ Eng 2007;42:1807–14.
- [107] Bencko V, Wagner V, Wagnerova M, Batora J. Immunological profiles in workers of a power plant burning coal rich in arsenic content. J Hyg Epidemiol Microbiol Immunol 1988;12:137–46.
- [108] Guha Mazumder DN, Haque R, Ghosh N, De BK, Santra A, Chakraborti D, et al. Arsenic in drinking water and the prevalence of respiratory effects in West Bengal, India. Int J Epidemiol 2000;29:1047–52.
- [109] Milton AH, Rahman M. Respiratory effects and arsenic contaminated well water in Bangladesh. Int J Environ Health Res 2002;12:175–9.
- [110] Islam LN, Zahid MSH, Nabi AHMN, Hossain M. Function of serum complement in drinking water arsenic toxicity. J Toxicol 2012;2012:1–7.

This page intentionally left blank

22

Arsenic and Developmental Toxicity and Reproductive Disorders

Molly L. Kile¹ and Maitreyi Mazumdar²

¹COLLEGE OF PUBLIC HEALTH AND HUMAN SCIENCES, OREGON STATE UNIVERSITY, CORVALLIS, OR, USA, ²BOSTON CHILDREN'S HOSPITAL AND HARVARD MEDICAL SCHOOL, BOSTON, MA, USA

CHAPTER OUTLINE

22.1 I	Introduction	521
22.2	Developmental Toxicity	523
22.3 F	Reproductive Toxicity	525
22.4 E	Early-Life Exposures and Delayed Health Effects	526
22.5 F	Reproductive Disorders	527
22.6	Conclusions	528
Refere	ences	528

22.1 Introduction

Arsenic is a naturally occurring metalloid that is present at trace levels in soil, rock, water, and air. It has a complex biogeochemical cycle, existing in several oxidation states (0, +3, and +5) and in inorganic and organic forms. Ingestion is the most common route of exposure. Arsenic levels can become elevated in natural waters from the weathering and dissolution of arsenic-containing minerals, volcanic activity, microbial activity, and anthropogenic activities [1]. The concentration of arsenic in water is highly dependent upon the environmental and geological conditions, arsenic is one of the most difficult contaminants to remove from drinking water. Redox potentials also influence inorganic arsenic's valence state, which in turn can influence its toxicity [2]. Under oxidizing conditions, the dominant form of inorganic arsenic is pentavalent arsenate (As^{+5}) , a less toxic and less soluble species. When redox potentials decrease, arsenate is reduced to trivalent arsenite (As^{+3}) , a more soluble and more toxic species.

There are approximately 20 countries including Bangladesh, Taiwan, Mexico, Chile, Argentina, Vietnam, Laos, India, China, Romania, and the United States that have ground-water aquifers that are naturally contaminated with arsenic at levels exceeding the World

Health Organization's drinking water recommendation of $10 \mu g/L$ [1]. Many of these aquifers are positioned below densely populated regions and are used as a potable water source leading to millions of people being chronically exposed to arsenic [3]. In the absence of drinking arsenic-contaminated water, diet is the primary route of exposure for arsenic [4]. Agricultural produce and livestock can accumulate arsenic from contaminated soil [5], water [6], pesticides [7], forage [8], and feed supplements [9]. Rice and cereal grains are considered the primary food sources for arsenic exposure because arsenic is readily taken up by the silicon transport system in these plants [10]. Daily dietary intake of inorganic arsenic will depend on dietary patterns and life stage. Diet studies have estimated that certain ethnic groups with higher rice consumption can have an average daily arsenic intake of $1 \mu g/kg$ -day [11]. Market basket surveys have also shown that infants are at risk of high arsenic exposure from rice cereal [12].

Once ingested, inorganic arsenic is metabolized through a series of reduction and oxidative methylation reactions forming two methylated metabolites: monomethylarsonic acid (MMA) and dimethylarsinic acid (DMA) [13]. In humans, approximately 60–70% of the ingested dose is excreted in the urine within 3 days, but the methylation process does not happen uniformly and urine contains, on average, 10–30% inorganic arsenic, 10–20% MMA, and 60–70% DMA [14–16]. This makes urinary arsenic a good biomarker for recent exposure to arsenic. The percentage of each metabolite, or their ratios (MMA/inorganic or DMA/MMA), is also commonly used as a biomarker of an individual's methylation capacity, which can be an independent risk factor for arsenic toxicity [17].

Arsenic can also be measured in other biological tissue including blood, hair, nails, and placental tissue. It is important to consider the half-life of arsenic in each of these biological tissues and individual exposure histories to select and interpret biomarker data. For instance, a single dose of arsenic is cleared from the bloodstream within several hours and excreted via the kidney. Subsequently, blood arsenic levels may only provide useful exposure information for individuals who have a blood test immediately after an exposure, or in those who are chronically exposed to consistently high levels of arsenic [18]. Recently, methods have been developed for measuring arsenic and other heavy metals in dried blood spots that are collected on filter paper [19]. Collecting blood spots from newborns is common for screening newborns for metabolic diseases and this new analytical technique provides a new tool for epidemiological studies to examine *in utero* exposure to arsenic.

Hair and toenails are useful biomarkers for historical exposures. Inorganic arsenic binds to keratin and becomes isolated from the metabolic activity that would lead to the excretion of arsenic in the urine. Any inorganic arsenic that is circulating in the blood will present itself quickly in the base of the hair shaft and nail bed. It takes several months to a year for arsenic to reach the distal tip of the nail where it can be collected. The growth rate of hair and nails is approximately 1.25 centimeters per month and 1.6 millimeters per month, respectively [20,21]. Subsequently inorganic arsenic detected in hair and nails will reflect exposures that accumulated in these matrixes during their growth [22]. Taking individual segments of each matrix can provide a timeline for individual exposures. Alternatively, samples can be pooled to provide an indication of the cumulative past exposures.

Many studies have shown that arsenic can be taken up and stored by the human placenta, as well as transported through the placenta [23–25]. Arsenic concentrations in maternal blood are highly correlated with arsenic concentrations in cord blood, demonstrating that arsenic readily crosses the placenta [26]. By examining the concentration of arsenic species in the urine of pregnant women, epidemiologists have shown that arsenic metabolism shifts during pregnancy and women excrete lower levels of inorganic arsenic species and higher levels of methylated arsenic metabolites [27,28]. This phenomenon has also been observed in mice [29]. These data demonstrate that the developing fetus is likely exposed to inorganic arsenic, MMA, and DMA. This raises concerns about arsenic's developmental toxicity.

This chapter will describe arsenic's developmental toxicity and its role in contributing to reproductive disorders including birth defects, infant mortality, low birth weight, and fertility. It will also discuss the evidence that developmental exposure to arsenic contributes to an increased risk of disease later in life.

22.2 Developmental Toxicity

22.2.1 Neural Tube Defects

22.2.1.1 Animal Models

Arsenic is highly teratogenic in laboratory animals. The most frequently reported anomalies include neural tube defects and similar disorders of the developing nervous system. Arsenic induces neural tube defects in several animal models, including mouse [30–33], rat [34], hamster [35–37], and chick [38,39]. Animal studies have demonstrated that arsenic crosses the placenta and preferentially accumulates in the neuroepithelium of the developing hamster, mouse, and monkey embryos [40,41]. Parental (intravenous or intraperitoneal) treatment is associated with greater frequencies of neural tube defects than oral treatment, but experimental studies have demonstrated that oral treatment is also teratogenic at higher doses [32].

The mechanisms of arsenic-induced neural tube defects in these models remain unknown. Several studies have reported that arsenic administration increases the production of reactive oxygen species, leading to oxidative stress and cytotoxicity, which may contribute to neural tube defects through oxidative damage to the developing neural plate [42,43]. Folate deficiency may exacerbate the direct toxic effects of arsenic on the neural plate by inhibiting the reduction of arsenic into less reactive subspecies. Mice with specific defects in folate transport and folate metabolism have higher rates of neural tube defects after *in utero* arsenic exposure compared to wild-type mice similarly exposed [44,45].

There is also some evidence from mice models to support the hypothesis that arsenic's effects on the neural tube are mediated through maternal hyperglycemia. This has been shown in experiments that delivered an intraperitoneal dose of arsenate and observed an elevation of glucose during pregnancy as well as increased rates of neural tube defects in the offspring [46]. This theory has been explored through *in silico* modeling, which computationally predicts that inhibition of the glucocorticoid receptor pathway would prevent arsenic-induced neural tube defects in chick embryos [47].

There is also increasing evidence that DNA methylation plays an important role in early embryonic development. DNA methylation is an epigenetic modification that involves the covalent addition of a methyl group to a cytosine at the 5'-position of a CpG dinucleotide [48]. CpG dinucleotides are clustered in the promoter regions of genes and in highly repeated elements such as long interspersed nucleotide elements [49,50]. When CpG sites in promoter regions are not methylated normal gene transcription occurs, but when CpG sites are methylated gene transcription is silenced [51]. Similarly, when long interspersed nucleotide elements are methylated they remain silenced, but when they become less methylated they can be transposable; that is, expression can lead to insertion into other genomic regions, which can result in gene silencing and chromosomal instability [52,53].

In vitro studies have demonstrated that high doses of inorganic arsenic alter DNA methylation in human lung, kidney, and keratinocytes [54–58]. In mice, prenatal inorganic arsenic exposure alters DNA methylation in fetal tissues [58–62]. There is also some evidence that aberrant DNA methylation is an underlying mechanism of neural tube defects. Experimental evidence demonstrated that arsenic-treated chick embryos exhibited significantly lower global DNA methylation as measured by 5-meC content [43].

22.2.1.1.1 HUMAN STUDIES

Few studies have investigated the relationship between arsenic exposure during pregnancy and risk of birth defects in humans. Archeologists report that the prevalence of spina bifida occulta among ancient mummies recovered in an area of northern Chile characterized by high environmental arsenic levels was 13.5%, compared with 2.4% in a low-arsenic area believed to have had no other differences in diet or other factors [63]. Investigation of a cluster of anencephalic births near the Texas-Mexico border found that maternal and paternal exposure to arsenic via drinking water was associated with higher risk of neural tube defects, but this finding was not statistically significant [64]. In a series of studies conducted in the same area of Texas, self-reported pesticide exposures were associated with neural tube defect risk. Though qualitative data on the specific pesticide were not obtained, given the widespread use of arsenical compounds in agriculture, the possibility that this association may be attributable to arsenic should be considered [64,65]. Ecological studies have linked soil arsenic levels to neural tube defects in China [66]. In a study of 2000 pregnant women in rural Bangladesh that used individual measures of exposure, Kwok et al. [67] reported a weak but statistically significant association between arsenic concentrations in drinking water and all birth defects, but no other adverse effects on pregnancy outcomes.

22.2.1.1.2 OTHER BIRTH DEFECTS

At least two studies have proposed an association between maternal arsenic exposure in drinking water and increased risk of congenital heart malformations in exposed offspring. In a casecontrol study in Massachusetts, Zierler et al. studied 270 cases of congenital heart anomalies compared to 665 randomly selected controls [68]. Arsenic concentration in drinking water at any detectable level was associated with a three-fold increase in risk of coarctation of the aorta (odds ratio (OR) 3.4, 95% confidence interval (CI): 1.3–8.9). The arsenic level was, however, very low, with the maximum concentration $2.1 \mu g/L$. In an ecological study using mortality data from 30 US counties from 1968 to 1984 and population-weighted mean arsenic concentration in public drinking water supplies, standard mortality ratios (SMR) for congenital heart disease appeared to be elevated for counties with mean arsenic concentrations exceeding $20 \mu g/L$ [69].

22.3 Reproductive Toxicity

22.3.1 Infant Mortality

There is epidemiological evidence that arsenic exposure is associated with higher rates of infant mortality. In Inner Mongolia, China, a large cross-sectional study (n = 9989) that used data from governmental reproductive permits observed that drinking-water arsenic concentrations greater than 50 µg/L during pregnancy was associated with an increased risk of neonatal death (OR 2.01, 95% CI: 1.12, 3.59). Because all women received the same government provided prenatal and postnatal healthcare, this study was able to control for many confounders [70]. Interestingly, this study also observed that maternal drinking-water arsenic concentrations above 50 µg/L was associated with a 0.05 kg higher term birth weight (95% CI: 0.02, 0.08 kg).

In a retrospective study in Chile, elevated infant mortality rates were observed in Antofagasta, a city that has a well-documented history of high arsenic exposure (approximately $860 \mu g/L$) in its publically supplied drinking water [71]. Compared to Valparaiso, a city with lower arsenic concentrations in drinking water (below $20 \mu g/L$), the rates of late fetal mortality [rate ratio (RR) = 1.7; 95% CI: 1.5–1.9], neonatal mortality (RR = 1.53; CI: 1.4–1.7), and post-neonatal mortality (RR = 1.26; CI: 1.2–1.3) were higher in Antofagasta after adjustment for location and calendar time.

In Bangladesh, a cross-sectional study (n = 533) observed excess risk for spontaneous abortion (OR = 2.5, 95% CI: 1.5, 4.3) and stillbirth (OR = 2.5, 95% CI: 1.3, 4.9) among pregnant women who were exposed to drinking water containing As at more than 50 µg/L compared to less than 50µg/L after adjusting for participant's height, history of hypertension and diabetes, and (for neonatal death only) age at first pregnancy [72]. This study, however, did not observe a statistically significant elevated risk for neonatal death (OR = 1.8, 95% CI: 0.9, 3.6). However, a large prospective study in Bangladesh (n = 2924) determined that women with the highest total urinary arsenic concentrations measured at 8 weeks' gestation have a 1.4-fold increase in spontaneous abortion (95% CI: 0.96–2.2) compared to women with the lowest concentrations of total urinary arsenic in their urine [73].

22.3.2 Birth Weight

Few human studies have examined the relationship between arsenic exposure and low birth weight. These studies were conducted in different populations and used different study

designs and different sample sizes. For instance, an ecological study in Taiwan that used an extreme points contrast design observed that infants born into arsenic-exposed villages were, on average, 29 grams (95% CI: 13.6–44.6g) lighter than infants born into non-arsenic-exposed villages after adjusting for maternal age, marital status, maternal education, and infant sex [74]. Yet, a cross-sectional ecological study performed in Mongolia (n = 9890) observed that infants born in villages with the highest arsenic levels in public water wells were on average 50 grams heavier than infants born in villages with lower arsenic concentrations in their drinking water wells [70].

In Bangladesh, a small prospective cohort study (n = 54) observed that arsenic measured in maternal hair collected at <20 weeks' gestational age was associated with decreased birth weight ($\beta = -193.5 \pm 90.0$ grams, *p*-value = 0.04) after adjusting for many confounders [75]. A much larger birth cohort study that was also recruited in Bangladesh (n = 1578) also reported a negative association between arsenic and birth weight. Rahman et al. measured total urinary arsenic concentrations in maternal samples collected at gestational weeks 8 and 30. Linear regression models estimated that for each 1-µg/L increase of arsenic in urine, birth weight decreased by 1.68 grams (standard error, 0.62) but only when urinary arsenic was <100 µg/L [76].

22.4 Early-Life Exposures and Delayed Health Effects

Several epidemiological studies have observed that exposure to arsenic during fetal development is associated with increased risk of chronic diseases later in life. A large ecological study by Smith et al. in Antofagasta, Chile, reported that the SMR for lung cancer at age 30–49 years was 7.0 (95% CI: 5.4–8.9; p < 0.001) for people who were conceived when their mothers were exposed to high concentrations of arsenic from drinking water compared to the rest of Chile [77]. This group also reported that the SMR for bronchiectasis was 12.4 (95% CI: 3.3–31.7; p < 0.001) in the same study indicating that the early life exposure to arsenic had long-lasting impacts on the pulmonary system.

There is also some evidence that prenatal exposure to arsenic impacts the immune system. In Bangladesh, a prospective birth cohort study observed that infants born to mothers with the highest quintile of urinary arsenic concentrations measured at 8 and 30 weeks of gestation had an increased risk of lower respiratory tract infection and diarrhea in the first year of life compared to infants born to mothers with the lowest quintile of urinary arsenic [78]. Data from a subset of this study's population also showed a negative correlation between maternal urinary arsenic and thymic size at age 2, 6, and 12 months [79]. The thymus is an organ believed to play an important role in the development of cells of the immune system. The relationship between developmental exposure to arsenic and increased risk of infection in early childhood has also been observed in the United States. Farzan et al. recruited a prospective birth cohort in New Hampshire (n = 214) and observed that maternal total urinary arsenic concentrations measured at 24–28 weeks' gestation was related to an increased risk in the total number of infections requiring a physician visit at 4 months of age (RR per one-fold increase in total urinary arsenic = 1.5, 95% CI = 1.0, 2.1) [80]. These investigators also observed that maternal urinary arsenic concentrations were associated with an increased risk of lower respiratory infections

treated with prescription medication (RR = 3.3, 95% CI = 1.2, 9.0), upper respiratory infections (RR = 1.6, 95% CI = 1.0, 2.5), and colds treated with prescription medication (RR = 2.3, 95% CI = 1.0, 5.2) [80].

While it is not fully understood how early-life exposure to arsenic would lead to increased susceptibility to pulmonary diseases and infectious disease later in life, there is growing evidence that developmental exposure to arsenic influences epigenetic mechanisms [81–83]. For instance, DNA methylation patterns are established during early fetal development and prenatal exposure to arsenic has been shown to influence DNA methylation extracted from cord blood [81,82]. This human evidence that environmentally relevant exposure to arsenic during development is supported by experimental studies, shows that inorganic arsenic causes both global DNA hypomethylation [57,84,85] and gene-specific DNA hypermethylation [57,86]. This effect might be explained by the overlap between inorganic arsenic metabolism and DNA methylation processes. Both consume S-adenosylmethionine, the universal methyl donor, which is a critical cofactor for both DNA methylation and the methylation of xenobiotics. Excess inorganic arsenic could reduce the availability or alter the activity of biological processes utilizing methyl groups, including DNA methylation and glutathione reduction [84]. It is possible that arsenic could initiate a cascade of events that alters DNA methylation, resulting in aberrant gene expression and also diminished glutathione activity leaving cells more vulnerable to oxidative stress.

22.5 Reproductive Disorders

22.5.1 Male Infertility

22.5.1.1 Animal Studies

While little is known regarding arsenic's ability to cause female infertility, there is fairly consistent evidence from rodent studies demonstrating that high doses of inorganic arsenic impacts spermatogenesis and decreases sperm motility and viability [87–89]. The effect of inorganic arsenic on sperm motility is believed to be due to inorganic arsenic's affinity to thiols, which are abundant in sperm flagella. It is also possible that inorganic arsenic damages Sertoli cells (cells in the testicles that support sperm development) [90].

22.5.1.2 Human Studies

There is also evidence in epidemiological studies that arsenic exposure influences male fertility. In a small cross-sectional study (n = 94) of male partners in infertile couples, higher arsenic exposure, which was defined as above the median DMA concentration in urine, was significantly associated with below-reference sperm concentration (ORs: 1.0–7.2; p = 0.02) after adjusting for age, BMI, abstinence, smoking, and drinking [91]. A hospital case-control study in China observed that males with elevated urinary inorganic arsenic had a greater probability of being infertile. Specifically, the odds of exhibiting male idiopathic infertility was 4.9-fold higher (95% CI: 1.8–13.6) and 13.6-fold higher (95% CI: 4.8–38.6) at the third and fourth quartiles of urinary inorganic arsenic [92].

22.6 Conclusions

Inorganic arsenic and its metabolites are developmental toxicants and contribute to the prevalence of reproductive disorders. Developmental exposure to inorganic arsenic and its metabolites is associated with increased risk of neural tube defects, neonatal mortality, and decreased birth weight. There is also growing evidence that developmental exposure to arsenic increases susceptibility to infectious diseases early in life, as well as chronic pulmonary diseases in adulthood. While the mechanisms linking these exposures and outcomes are poorly understood, there is evidence that epigenetic dysregulation may play an important role. Arsenic is also associated with reduced sperm quality and male infertility in adults.

References

- [1] Smedley PL, Kinniburg DG. A review of the source, behaviour and distribution of arsenic in natural waters. Appl Geochem 2002;17:517–68.
- [2] USA EPA. Arsenic treatment technologies for soil, waste, and water. EPA-542-R-02-004 ed., Vol. 2003. Solid waste and emergency response. Cincinnati, OH: NSCEP; 2002.
- [3] World Health Organization. Arsenic in drinking water. Geneva: WHO; 1999.
- [4] Yost LJ, Schoof RA, Aucoin R. Intake of inorganic arsenic in the North American diet. Hum Ecol Risk Assess 1998;4:137–52.
- [5] Duxbury JM, Mayer AB, Lauren JG, Hassan N. Food chain aspects of arsenic contamination in Bangladesh: effects on quality and productivity of rice. J Environ Sci Health A Tox Hazard Subst Environ Eng 2003;38:61–9.
- [6] Meharg AA, Rahman MM. Arsenic contamination of Bangladesh paddy field soils: implications for rice contribution to arsenic consumption. Environ Sci Technol 2003;37:229–34.
- [7] MacIntosh DL, Spengler JD, Ozkaynak H, Tsai LH, Ryan PB. Dietary exposures to selected metals and pesticides. Environ Health Perspect 1996;104:202–9.
- [8] Pizarro I, Gomez MM, Fodor P, Palacios MA, Camara C. Distribution and biotransformation of arsenic species in chicken cardiac and muscle tissues. Biol Trace Elem Res 2004;99:129–43.
- [9] Lasky T, Sun W, Kadry A, Hoffman MK. Mean total arsenic concentrations in chicken 1989–2000 and estimated exposures for consumers of chicken. Environ Health Perspect 2004;112:18–21.
- [10] Ma JF, Yamaji N, Mitani N, Xu XY, Su YH, McGrath SP, et al. Transporters of arsenite in rice and their role in arsenic accumulation in rice grain. Proc Natl Acad Sci USA 2008;105:9931–5.
- [11] EFSA Authority. EFSA Panel on Contaminants in the Food Chain (CONTAM): scientific opinion on arsenic in food. EFSA J 2009;7:60–71.
- [12] Meharg AA, Sun GX, Williams PN, Adomako E, Deacon C, Zhu YG, et al. Inorganic arsenic levels in baby rice are of concern. Environ Pollut 2008;152:746–9.
- [13] Gebel TW. Arsenic and drinking water contamination. Science 1999;283:1455e.
- [14] Pomroy C, Charbonneau SM, McCullough RS, Tam GKH. Human retention studies with As-74. Toxicol Appl Pharmacol 1980;53:550–6.
- [15] Calderon R, Hudgens E, Le X, Schreinemachers D, Thomas D. Excretion of arsenic in urine as a function of exposure to arsenic in drinking water. Environ Health Perspect 1999;107:663–7.
- [16] Hopenhayn-Rich C, Smith AH, Goeden HM. Human studies do not support the methylation threshold hypothesis for the toxicity of inorganic arsenic. Environ Res 1993;60:161–77.

- [17] Kile ML, Hoffman E, Rodrigues EG, Breton CV, Quamruzzaman Q, Rahman M, et al. A pathway-based analysis of urinary arsenic metabolites and skin lesions. Am J Epidemiol 2011;173:778–86.
- [18] Hall M, Chen Y, Ahsan H, Slavkovich V, van Geen A, Parvez F, et al. Blood arsenic as a biomarker of arsenic exposure: results from a prospective study. Toxicology 2006;225:225–33.
- [19] Lehner AF, Rumbeiha W, Shlosberg A, Stuart K, Johnson M, Domenech R, et al. Diagnostic analysis of veterinary dried blood spots for toxic heavy metals exposure. J Anal Toxicol 2013;37:406–22.
- [20] Nicolis I, Curis E, Deschamps P, Benazeth S. Arsenite medicinal use, metabolism, pharmacokinetics and monitoring in human hair. Biochimie 2009;91:1260–7.
- [21] Yaemsiri S, Hou N, Slining MM, He K. Growth rate of human fingernails and toenails in healthy American young adults. J Eur Acad Dermatol Venereol 2010;24:420–3.
- [22] Kile ML, Houseman EA, Rodrigues E, Smith TJ, Quamruzzaman Q, Rahman M, et al. Toenail arsenic concentrations, GSTT1 gene polymorphisms, and arsenic exposure from drinking water. Cancer Epidemiol Biomarkers Prev 2005;14:2419–26.
- [23] Eastman NJ. The arsenic content of the human placenta following arsphenamine therapy. Am J Obstet Gynecol 1931;21:60–4.
- [24] Rudge CV, Rollin HB, Nogueira CM, Thomassen Y, Rudge MC, Odland JO. The placenta as a barrier for toxic and essential elements in paired maternal and cord blood samples of South African delivering women. J Environ Monit 2009;11:1322–30.
- [25] DeSesso JM, Jacobson CF, Scialli AR, Farr CH, Holson JF. An assessment of the developmental toxicity of inorganic arsenic. Reprod Toxicol 1998;12:385–433.
- [26] Kagey BT, Bumgarner JE, Creason JP. Arsenic levels in maternal-fetal tissue sets. Trace Subst Environ Health 1977;11:252–6.
- [27] Concha G, Vogler G, Lezcano D, Nermell B, Vahter M. Exposure to inorganic arsenic metabolites during early human development. Toxicol Sci 1998;44:185–90.
- [28] Gardner RM, Nermell B, Kippler M, Grander M, Li L, Ekstrom EC, et al. Arsenic methylation efficiency increases during the first trimester of pregnancy independent of folate status. Reprod Toxicol 2011;31:210–18.
- [29] Devesa V, Adair BM, Liu J, Waalkes MP, Diwan BA, Styblo M, et al. Arsenicals in maternal and fetal mouse tissues after gestational exposure to arsenite. Toxicology 2006;224:147–55.
- [30] Wlodarczyk BJ, Bennett GD, Calvin JA, Finnell RH. Arsenic-induced neural tube defects in mice: alterations in cell cycle gene expression. Reprod Toxicol 1996;10:447–54.
- [31] Hill DS, Wlodarczyk BJ, Finnell RH. Reproductive consequences of oral arsenate exposure during pregnancy in a mouse model. Birth Defects Res B Dev Reprod Toxicol 2008;83:40–7.
- [32] Nemec MD, Holson JF, Farr CH, Hood RD. Developmental toxicity assessment of arsenic acid in mice and rabbits. Reprod Toxicol 1998;12:647–58.
- [33] Machado AF, Hovland Jr. DN, Pilafas S, Collins MD. Teratogenic response to arsenite during neurulation: relative sensitivities of C57BL/6J and SWV/Fnn mice and impact of the splotch allele. Toxicol Sci 1999;51:98–107.
- [34] Beaudoin AR. Teratogenicity of sodium arsenate in rats. Teratology 1974;10:153-7.
- [35] Carpenter SJ. Developmental analysis of cephalic axial dysraphic disorders in arsenic-treated hamster embryos. Anat Embryol (Berl) 1987;176:345–65.
- [36] Ferm VH, Hanlon DP. Constant rate exposure of pregnant hamsters to arsenate during early gestation. Environ Res 1985;37:425–32.
- [37] Hood RD, Harrison WP. Effects of prenatal arsenite exposure in the hamster. Bull Environ Contam Toxicol 1982;29:671–8.

- [38] Peterkova R, Puzanova L. Effect of trivalent and pentavalent arsenic on early developmental stages of the chick embryo. Folia Morphol (Praha) 1976;24:5–13.
- [39] Gilani SH, Alibhai Y. Teratogenicity of metals to chick embryos. J Toxicol Environ Health 1990;30:23–31.
- [40] Hanlon DP, Ferm VH. Placental permeability of arsenate ion during early embryogenesis in the hamster. Experientia 1977;33:1221–2.
- [41] Lindgren A, Danielsson BR, Dencker L, Vahter M. Embryotoxicity of arsenite and arsenate: distribution in pregnant mice and monkeys and effects on embryonic cells in vitro. Acta Pharmacol Toxicol (Copenh) 1984;54:311–20.
- [42] Singh S, Greene RM, Pisano MM. Arsenate-induced apoptosis in murine embryonic maxillary mesenchymal cells via mitochondrial-mediated oxidative injury. Birth Defects Res A Clin Mol Teratol 2010;88:25–34.
- [43] Han ZJ, Song G, Cui Y, Xia HF, Ma X. Oxidative stress is implicated in arsenic-induced neural tube defects in chick embryos. Int J Dev Neurosci 2011;29:673–80.
- [44] Wlodarczyk B, Spiegelstein O, Gelineau-van Waes J, Vorce RL, Lu X, Le CX, et al. Arsenic-induced congenital malformations in genetically susceptible folate binding protein-2 knockout mice. Toxicol Appl Pharmacol 2001;177:238–46.
- [45] Wlodarczyk BJ, Zhu H, Finnell RH. Mthfr gene ablation enhances susceptibility to arsenic prenatal toxicity. Toxicol Appl Pharmacol 2014;275:22–7.
- [46] Hill DS, Wlodarczyk BJ, Mitchell LE, Finnell RH. Arsenate-induced maternal glucose intolerance and neural tube defects in a mouse model. Toxicol Appl Pharmacol 2009;239:29–36.
- [47] Ahir BK, Sanders AP, Rager JE, Fry RC. Systems biology and birth defects prevention: blockade of the glucocorticoid receptor prevents arsenic-induced birth defects. Environ Health Perspect 2013;121:332–8.
- [48] Bird AP. CpG-rich islands and the function of DNA methylation. Nature 1986;321:209–13.
- [49] Chalitchagorn K, Shuangshoti S, Hourpai N, Kongruttanachok N, Tangkijvanich P, Thong-ngam D, et al. Distinctive pattern of LINE-1 methylation level in normal tissues and the association with carcinogenesis. Oncogene 2004;23:8841–6.
- [50] Kochanek S, Renz D, Doerfler W. DNA methylation in the ALU sequences of diploid and haploid primary human cells. EMBO J 1993;12:1141–51.
- [51] Ng HH, Bird A. DNA methylation and chromatin modification. Curr Opin Genet Dev 1999;9:158–63.
- [52] Merlo A, Herman JG, Mao L, Lee DJ, Gabrielson E, Burger PC, et al. 5' CpPG island methylation is associated with transcriptional silencing of the tumor-suppressor P16/CDKN2/MTS1 in human cancers. Nat Med 1995;1:686–92.
- [53] Herman JG, Baylin SB. Mechanisms of disease: gene silencing in cancer in association with promoter hypermethylation. N Engl J Med 2003;349:2042–54.
- [54] Mass MJ, Wang L. Arsenic alters cytosine methylation patterns of the promoter of the tumor suppressor gene p53 in human lung cells: a model for a mechanism of carcinogenesis. Mutat Res 1997;386:263–77.
- [55] Zhong CXY, Mass MJ. Both hypomethylation and hypermethylation of DNA are associated with arsenite exposure in cultures of human cells identified by methylation-sensitive arbitrarily-primed PCR. Toxicol Lett 2001;122:223–34.
- [56] Christensen BC, Marsit CJ, Welsh MM, Nelson HH, Kelsey KT. Genome-wide alterations in CpG island methylation induced by arsenite. Eur J Oncol 2007;12:9–13.
- [57] Chen H, Li SF, Liu J, Diwan BA, Barrett JC, Waalkes MP. Chronic inorganic arsenic exposure induces hepatic global and individual gene hypomethylation: implications for arsenic hepatocarcinogenesis. Carcinogenesis 2004;25:1779–86.
- [58] Xie Y, Liu J, Benbrahim-Tallaa L, Ward JM, Logsdon D, Diwan BA, et al. Aberrant DNA methylation and gene expression in livers of newborn mice transplacentally exposed to a hepatocarcinogenic dose of inorganic arsenic. Toxicology 2007;236:7–15.

- [59] Dolinoy DC, Huang D, Jirtle RL. Maternal nutrient supplementation counteracts bisphenol A-induced DNA hypomethylation in early development. Proc Natl Acad Sci USA 2007;104:13056–61.
- [60] Waterland RA, Jirtle RL. Transposable elements: targets for early nutritional effects on epigenetic gene regulation. Mol Cell Biol 2003;23:5293–300.
- [61] Kaminen-Ahola N, Ahola A, Maga M, Mallitt KA, Fahey P, Cox TC, et al. Maternal ethanol consumption alters the epigenotype and the phenotype of offspring in a mouse model. PloS Genet 2010;6:e1000811.
- [62] Anway MD, Cupp AS, Uzumcu M, Skinner MK. Epigenetic transgenerational actions of endocrine disruptors and male fertility. Science 2005;308:1466–9.
- [63] Silva-Pinto V, Arriaza B, Standen V. Spina bifida occulta associated with environmental arsenic exposure in a prehispanic sample from northern Chile. Rev Med Chil 2010;138:461–9.
- [64] Suarez L, Felkner M, Brender JD, Canfield M, Zhu H, Hendricks KA. Neural tube defects on the Texas-Mexico border: what we've learned in the 20 years since the Brownsville cluster. Birth Defects Res A Clin Mol Teratol 2012;94:882–92.
- [65] Brender JD, Suarez L, Felkner M, Gilani Z, Stinchcomb D, Moody K, et al. Maternal exposure to arsenic, cadmium, lead, and mercury and neural tube defects in offspring. Environ Res 2006;101:132–9.
- [66] Wu J, Chen G, Liao Y, Song X, Pei L, Wang J, et al. Arsenic levels in the soil and risk of birth defects: a population-based case-control study using GIS technology. J Environ Health 2011;74:20–5.
- [67] Kwok RK, Kaufmann RB, Jakariya M. Arsenic in drinking-water and reproductive health outcomes: a study of participants in the Bangladesh Integrated Nutrition Programme. J Health Popul Nutr 2006;24:190–205.
- [68] Zierler S, Theodore M, Cohen A, Rothman KJ. Chemical quality of maternal drinking water and congenital heart disease. Int J Epidemiol 1988;17:589–94.
- [69] Engel RR, Smith AH. Arsenic in drinking water and mortality from vascular disease: an ecologic analysis in 30 counties in the United States. Arch Environ Health 1994;49:418–27.
- [70] Myers SL, Lobdell DT, Liu Z, Xia Y, Ren H, Li Y, et al. Maternal drinking water arsenic exposure and perinatal outcomes in inner Mongolia, China. J Epidemiol Community Health 2010;64:325–9.
- [71] Hopenhayn-Rich C, Browning SR, Hertz-Picciotto I, Ferreccio C, Peralta C, Gibb H. Chronic arsenic exposure and risk of infant mortality in two areas of Chile. Environ Health Perspect 2000;108:667–73.
- [72] Milton A, Smith W, Rahman B, Hasan Z, Kulsum U, Dear K, et al. Chronic arsenic exposure and adverse pregnancy outcomes in Bangladesh. Epidemiology 2005;16:82–6.
- [73] Rahman A, Persson LA, Nermell B, El Arifeen S, Ekstrom EC, Smith AH, et al. Arsenic exposure and risk of spontaneous abortion, stillbirth, and infant mortality. Epidemiology 2010;21:797–804.
- [74] Yang C-Y, Chang C-C, Tsai S-S, Chuang H-Y, Ho C-K, Wu T-N. Arsenic in drinking water and adverse pregnancy outcome in an arseniasis-endemic area in northeastern Taiwan. Environ Res 2003;91:29–34.
- [75] Huyck KL, Kile ML, Mahiuddin G, Quamruzzaman Q, Rahman M, Breton CV, et al. Maternal arsenic exposure associated with low birth weight in Bangladesh. J Occup Environ Med 2007;49:1097–104.
- [76] Rahman A, Vahter M, Smith AH, Nermell B, Yunus M, El Arifeen S, et al. Arsenic exposure during pregnancy and size at birth: a prospective cohort study in Bangladesh. Am J Epidemiol 2009;169:304–12.
- [77] Smith AH, Marshall G, Yuan Y, Ferreccio C, Liaw J, von Ehrenstein O, et al. Increased mortality from lung cancer and bronchiectasis in young adults after exposure to arsenic in utero and in early childhood. Environ Health Perspect 2006;114:1293–6.
- [78] Rahman A, Vahter M, Ekstrom EC, Persson LA. Arsenic exposure in pregnancy increases the risk of lower respiratory tract infection and diarrhea during infancy in Bangladesh. Environ Health Perspect 2011;119:719–24.
- [79] Raqib R, Ahmed S, Sultana R, Wagatsuma Y, Mondal D, Hoque AMW, et al. Effects of in utero arsenic exposure on child immunity and morbidity in rural Bangladesh. Toxicol Lett 2009;185:197–202.

- [80] Farzan SF, Korrick S, Li Z, Enelow R, Gandolfi AJ, Madan J, et al. In utero arsenic exposure and infant infection in a United States cohort: a prospective study. Environ Res 2013;126:24–30.
- [81] Kile ML, Baccarelli A, Hoffman E, Tarantini L, Quamruzzaman Q, Rahman M, et al. Prenatal arsenic exposure and DNA methylation in maternal and umbilical cord blood leukocytes. Environ Health Perspect 2012;120:1061–6.
- [82] Kile ML, Houseman EA, Baccarelli A, Quamruzzaman Q, Rahman M, Mostofa G, et al. Effect of prenatal arsenic exposure on DNA methylation and leukocyte subpopulations in cord blood. Epigenetics 2014;9:5.
- [83] Rager JE, Bailey KA, Smeester L, Miller SK, Parker JS, Laine JE, et al. Prenatal arsenic exposure and the epigenome: altered microRNAs associated with innate and adaptive immune signaling in newborn cord blood. Environ Mol Mutagen 2014;55:196–208.
- [84] Reichard JF, Schnekenburger M, Puga A. Long term low-dose arsenic exposure induces loss of DNA methylation. Biochem Biophys Res Commun 2007;352:188–92.
- [85] Sciandrello G, Caradonna F, Mauro M, Barbata G. Arsenic-induced DNA hypomethylation affects chromosomal instability in mammalian cells. Carcinogenesis 2004;25:413–17.
- [86] Chanda S, Dasgupta UB, GuhaMazumder D, Gupta M, Chaudhuri U, Lahiri S, et al. DNA hypermethylation of promoter of gene p53 and p16 in arsenic-exposed people with and without malignancy. Toxicol Sci 2006;89:431–7.
- [87] Chiou TJ, Chu ST, Tzeng WF, Huang YC, Liao CJ. Arsenic trioxide impairs spermatogenesis via reducing gene expression levels in testosterone synthesis pathway. Chem Res Toxicol 2008;21:1562–9.
- [88] Ferreira M, Matos RC, Oliveira H, Nunes B, Pereira Mde L. Impairment of mice spermatogenesis by sodium arsenite. Hum Exp Toxicol 2012;31:290–302.
- [89] Li Y, Wang M, Piao F, Wang X. Subchronic exposure to arsenic inhibits spermatogenesis and downregulates the expression of ddx3y in testis and epididymis of mice. Toxicol Sci 2012;128:482–9.
- [90] Pant N, Murthy RC, Srivastava SP. Male reproductive toxicity of sodium arsenite in mice. Hum Exp Toxicol 2004;23:399–403.
- [91] Xu W, Bao H, Liu E, Liu L, Zhu YG, She J, et al. Environmental exposure to arsenic may reduce human semen quality: associations derived from a Chinese cross-sectional study. Environ Health 2012;11:46.
- [92] Shen H, Xu W, Zhang J, Chen M, Martin FL, Xia Y, et al. Urinary metabolic biomarkers link oxidative stress indicators associated with general arsenic exposure to male infertility in a Han Chinese population. Environ Sci Technol 2013;47:8843–51.

23

Arsenic and Cancer

Paul B. Tchounwou, Udensi K. Udensi, Raphael D. Isokpehi, Clement G. Yedjou, and Sanjay Kumar

MOLECULAR TOXICOLOGY RESEARCH LABORATORY, NIH RCMI-CENTER FOR ENVIRONMENTAL HEALTH, COLLEGE OF SCIENCE, ENGINEERING AND TECHNOLOGY, JACKSON STATE UNIVERSITY, JACKSON, MISSISSIPPI, USA

CHAPTER OUTLINE

23.1 Arsenic and Arsenic-Containing Compounds	533	
23.2 Sources of Arsenic and Potential for Human Exposure	535	
23.3 Molecular Mechanisms of Arsenic-Induced Carcinogenesis	538	
23.4 Health Effects Associated with Arsenic Exposure	543	
23.5 Conclusions	549	
Acknowledgments		
References	549	

23.1 Arsenic and Arsenic-Containing Compounds

Arsenic is a metalloid (atomic number, 33; relative atomic mass, 74.92) widely distributed in Earth's crust at an average concentration of 2 mg/kg. It occurs in trace quantities in all rock, soil, water, and air. Arsenic can exist in four oxidation states: -3 (e.g., arsine gas), 0 (metalloid arsenic), +3 (trivalent), and +5 (pentavalent) [1]. Arsenic compounds from a biological and toxicological point of view are divided into three major groups: inorganic arsenic compounds, organic arsenic compounds, and arsine gas [2,3]. Arsenic compounds like other chemicals are best identified using the CAS (Chemical Abstracts Service) registry number [2].

Atmospheric conditions affect the prevalence of arsenic forms; for example arsenite (As(III)) is the dominant form under reducing conditions, while arsenate (As(V)) is the stable form in oxygenated environments [2,3]. More than 200 mineral species contain arsenic, with 60% comprising arsenate and 20% sulfide and sulfosalts, while the rest (20%) consist of arsenides, arsenites, oxides, and elemental arsenic [4]. Arsenic is found associated with many types of mineral deposits, especially those including sulfide mineralization, and the most common of the arsenic minerals is arsenopyrite (FeAsS). The most prevalent trivalent inorganic arsenic compounds are arsenic trioxide, sodium arsenite, and sodium trichlorite, while arsenic pent-oxide, arsenic acid, and arsenates (e.g., sodium arsenate, lead arsenate, and calcium arsenate)

are the most common pentavalent inorganic arsenic compounds [2]. There are many organic arsenic compounds but the following are mostly encountered: arsenalic acid, methylarsonic acid, dimethylarsinic (cacodylic acid), and arsenobetaine [5]. Arsenic species differ in their toxicity and biochemical and environmental behaviors. Arsenate [As(V)] is the most common environmental form of inorganic arsenic, but arsenite [As(III)] is more toxic and probably the more carcinogenic species [6]. A list of various arsenic compounds of environmental and human health relevance is presented in Table 23–1.

Arsenic can be found in nature as a pure elemental crystal or in conjunction with sulfur or other metals [7]. Arsenic oxidizes to arsenic trioxide (As_2O_3) when heated in air, giving out garlic odor-like fumes. As_2O_3 can also be produced as a by-product of metal smelting operations [8]. Elemental arsenic is not soluble in water. Arsenic salts exhibit a wide range of solubility depending on pH and the ionic environment [2,9].

Chemical Name	CAS Reg. No.	Synonyms	Formula
Arsanilic acid	98-50-0	Arsonic acid, (4-aminophenyl)-	C ₆ H ₈ AsNO ₃
Arsenic monoxide	12005-99-1	Arsenic monoxide	AsO
Arsenic	7440-38-2	Metallic arsenic	As
Arsenic(V) pentoxide	1303-28-2	Arsenic oxide [As ₂ O ₅]	As ₂ O ₅
Arsenic trisulfide	1303-33-9	Arsenic tersulfide	As ₂ S ₃
Arsenic diiodide	13770-56-4	15942-63-9	As ₂ I ₄
Arsenic(III) sulfide	1303-33-9	Arsenic sulfide [As ₂ S ₃]	As ₂ S ₃
Diarsenic tetrahydride	15942-63-9	Diarsine	As ₂ H ₄
Arsenic(III) trichloride	7784-34-1	Arsenic chloride [AsCl ₃]	AsCl ₃
Arsenic(III) trioxide	1327-53-3	Arsenic oxide [As ₂ O ₃]	As ₂ O ₃
Arsenobetaine	64436-13-1	Arsonium (carboxymethyl), trimethyl- hydroxide, inner salt; 2-(trimethylarsonio) acetate	$C_5H_{11}AsO_2$
Arsine	7784-42-1	Arsenic hydride [AsH ₃]	AsH ₃
Calcium arsenate	7778-44-1	Arsenic acid $[H_3AsO_4]$, calcium salt (2:3)	As ₂ Ca ₃ O ₈
Dimethylarsinic acid	75-60-5	Cacodylic acid	C ₂ H ₇ AsO ₂
Lead arsenate	7784-40-9	Arsenic acid $[H_3AsO_4]$, lead (2+) salt (1:1)	AsHO ₄ Pb
Methanearsonic acid, disodium salt	144-21-8	Arsonic acid, methyl-, disodium salt	CH ₃ AsO ₃ ·2Na
Methanearsonic acid, monosodium salt	2163-80-6	Arsonic acid, methyl-, monosodium salt	CH_4AsO_3Na
Potassium arsenate	7784-41-0	Arsenic acid $[H_2AsO_4K]$, monopotassium salt	KH ₂ AsO ₄
Potassium arsenite	13464-35-2	Arsenous acid, potassium salt [AsO ₂ K]	KAsO ₂
Sodium arsenate	7631-89-2	Arsenic acid [H ₂ AsO ₄ Na], monosodium salt	Na ₃ AsO ₄
Sodium arsenite	7784-46-5	Arsenous acid, sodium salt [AsO ₂ Na]	NaAsO ₂
Sodium dimethylarsinate	124-65-2	Arsinic acid, dimethyl-, sodium salt, sodium cacodylate [C ₂ H ₆ AsO ₂ Na]	$(CH_3)_2$ AsNaO ₂ 3H ₂ O
Gallium arsenide	1303-00-0	Gallium monoarsenide	AsGa

Table 23–1 Arsenic Compounds of Environmental and Human Health Relevance

Absorption of arsenic in inhaled airborne particles is highly dependent on the solubility and size of particles. Both pentavalent and trivalent soluble arsenic compounds are rapidly and easily absorbed from the gastrointestinal tract. Arsenic metabolism often involves two major types of reactions: reduction reactions of pentavalent to trivalent arsenic [10] and oxidative methylation reactions in which trivalent forms of arsenic are sequentially methylated to form mono-, di-, and trimethylated products using S-adenosyl methionine (SAM) as the methyl donor and glutathione (GSH) as an essential co-factor. The methylation of inorganic arsenic facilitates the excretion of inorganic arsenic from the body, as MMA and DMA in urine [11].

23.2 Sources of Arsenic and Potential for Human Exposure

Arsenic is ubiquitous in nature and is introduced into the environment from both natural and anthropogenic activities. Human activities that contribute to environmental contamination by arsenic include mining, industrial smelting of metals, power generation with coal, and applications of arsenic-containing pesticides and herbicides [12]. Human exposure may occur through ingestion of contaminated food, drinking water, or medication; inhalation of dust particles or aerosols; and/or direct dermal contact [9,13]. It is important to understand the environmental sources and levels of arsenic in order to prevent human exposure and the consequent deleterious effects.

23.2.1 Arsenic and Drinking Water Contamination

Water is the major source of human exposure to arsenic compounds [13]. Underground water and water aquifers are contaminated when arsenic-bearing minerals undergo oxidation and release arsenic into water. The reduction of Fe/As oxyhydroxides could also be a source for arsenic contamination in ground water [14]. Some regions across the world have been observed to have arsenic drinking water levels exceeding the standards set by the US Public Health Service and the World Health Organization— $10 \mu g/L$ for the USA and Canada and $50 \mu g/L$ for developing countries with limited access to potable water and larger populations exposed to arsenic [15]. Bangladesh and West Bengal (India) are the most affected areas of the world with arsenic concentration in ground water in some areas up to $3200 \mu g/L$ [16]. The inorganic form of arsenic is the most common species found in water and can be stable as both arsenite or trivalent [As(III)] and arsenate or pentavalent [As(V)] inorganic arsenicals [17]. As(III) is the most prevalent species in ground water, while As(V) species are more significant in the surface water of rivers [18].

23.2.2 Arsenic and Food Contamination

Arsenic was intentionally added to food as a preservative in the mid-19th century before it was discovered that it has dangerous effects on human health. Sea food is the major source

of arsenic exposure to humans [19]. A level of arsenic of 1.3 ppm above the United States Environmental Protection Agency (EPA) regulatory limit was observed in commercial fish (Chilean sea bass, croaker, flounder, porgie, and whiting) in New Jersey, USA [20]. Diet is second to water as the major source of both inorganic and organic arsenic. Estimates of dietary inorganic arsenic intakes vary. Arsenic could enter the food cycle by growing a crop on arsenic-contaminated soil or by irrigating the farm with arsenic-contaminated water [21]. The US average daily consumption of arsenic by adults and children is $3.2 \,\mu$ g/day with a range of $1-20 \,\mu$ g/day [22], while it is 0.13 to $0.56 \,\mu$ g/kg/day for average consumers; it is 9.1 to $39.2 \,\mu$ g/day for a 70-kg adult in Europe [23]. In India, it was observed that cooked foods had higher levels of arsenic than raw foods. Daily dietary intakes of arsenic from the foodstuffs for adults were from 171.20 to $189.13 \,\mu$ g/day, while the range for children was 91.89- $101.63 \,\mu$ g/day [24]. Arsenobetaine is the most common form of organic arsenic that naturally accumulates in marine organisms with the exception of algae that contain mostly dimethylarsinoylribosides. Humans are thus exposed to these arsenic compounds when they consume any diet that includes seafood [1,9].

23.2.3 Arsenic and Soil Contamination

Arsenic in soil comes mainly from the parent soil. The nature of arsenic in soil is affected by the lithology of parent rock materials, volcanic activity, bioactivity, weathering history, transport, sorption, and precipitation [25]. The natural content of arsenic in soils globally ranges from 0.01 to over 600 mg/kg, with an average of about 2 to 20 mg/kg. Biotransformation of arsenic species occurs in the soil and the three major modes of biotransformation observed include: the biosynthesis of organoarsenic compounds, redox transformation between arsenite and arsenate, and the reduction and methylation of arsenic [26]. Arable lands could be contaminated with arsenic from run-off water and the use of arsenic-rich ground water for irrigation. Inorganic arsenic is the major form of arsenic in soil but high levels of organic forms are seen in soils with anthropogenic activities such as mining and manufacturing, and application of arsenic-containing pesticides [27]. Pentavalent inorganic arsenic is more commonly seen in soil because trivalent arsenicals are easily oxidized [28]. Human exposures by inhalation of soil particles blown by wind and dermal absorption are not considered major sources of arsenic because the amount of arsenic in ambient air is low; also arsenic is poorly absorbed through the skin from the soil [29]. However, incidental ingestion of arsenic in soil is considered significant though far less than the amount in drinking water and diet [30]. This is likely because there is a reduced amount of inorganic arsenic as well as reduced bioavailability of arsenic in soil compared to water [27].

23.2.4 Arsenic and Air Pollution

Inorganic arsenic is the major form of arsenic found in the air and it is very low in concentration, contributing to less than 15% of all forms of exposure including that from food, water, and soil [31]. The concentration of arsenic in air due to non-human activities ranges from 1 to 3 ng/m³, and the

range of the concentration in urban areas is 20 to 100 ng/m^3 [9]. The European Commission [31] reports that levels of arsenic in air range between 0 and 1 ng/m^3 in remote areas, 0.2 and 1.5 ng/m^3 in rural areas, and 0.5 and 3 ng/m^3 in urban areas, with levels up to about 50 ng/m^3 in the vicinity of industrial sites.

23.2.5 Arsenic Exposure from Medication

Arsenic has been used to treat various ailments for more than 2400 years including toothaches, ulcers, and abscesses [32]. Fowler's solution, which consists of 1% solution of potassium arsenite, was discovered in 1786 and was used to treat diseases such as malaria, syphilis, asthma, chorea, eczema, and psoriasis [33]. Fowler's solution was reported to lower white blood cell count in leukemia patients [34] and was also used as a tonic for anemia and to treat rheumatism, dermatitis herpetiformis, Hodgkin's disease, pemphigus, and pernicious anemia. [32] Salvarsan, which is an organic arsenical, was introduced in 1910 by Paul Ehrlich for treating syphilis and trypanosomiasis [35]. The United States Food and Drug Administration (FDA) in September 2000 approved the use of arsenic trioxide for the treatment of relapsed or refractory acute promyelocytic leukemia (APL) cases resistant to all-trans retinoic acid (ATRA) [34]. The therapeutic effect of arsenic trioxide on other types of cancers such as non-APL acute myeloid leukemia and myelodysplastic syndromes has been explored [36]. There are also ongoing researches on the combination of arsenic trioxide with other agents such as vitamin C (ascorbic acid) for the treatment of lymphoproliferative disorders [37] and multiple myeloma [38]. Arsenic trioxide was the drug of choice before the discovery of metronidazole in 1959 for the treatment of Trichomonas vaginalis infection (trichomoniasis), which causes vaginal discharge [32]. In traditional Chinese medicine, arsenic derivatives are still used to benumb the pulp of a painful tooth [39]. Organic arsenic compound melarsoprol is still the drug of choice for the treatment of the protozoan parasite disease trypanosomiasis [40]. However, the use of arsenic derivatives in medication has been drastically reduced due to the perceived toxic properties.

23.2.6 Occupational Exposure to Arsenic

Arsenic is an essential component of a wide variety of industrial products, such as wood preservatives, herbicides, insecticides, pesticides, fungicides, high-emitting diodes, and semiconductors. Workers at factories manufacturing arsenic-containing products are prone to exposure directly through inhalation and dermal contact [9]. Arsenic was first used for tick control in South Africa in 1893 and since then over a thousand cattle dipping vats have been constructed throughout the country, arsenic being applied as the sodium salt of arsenous acid [41]. Old mines and abandoned metal ores are reported to have high arsenic levels resulting from the smelting process and coal burning. These processes generate stack dust and flue gas that contaminate the soil and water with arsenic [42]. The amount of airborne arsenic inhaled depends on the characteristics, matrix composition, and particle size distribution of the compound [43]. Studies on orchard workers, who were becoming ill after exposure to lead arsenate, used in pesticides for apple and cherry orchards, provided a basis for understanding some of the long-term effects of occupational exposure to arsenic. However, definitive links between arsenic, a by-product of copper smelting, and lung cancer through inhalation was established studying copper smelting workers who were ill [44]. The association of arsenic exposure with occupational health defects has led to discontinuation of most inorganic arsenic in agriculture and most industrial processes and the establishment of stricter limits by government regulation agencies.

23.3 Molecular Mechanisms of Arsenic-Induced Carcinogenesis

Several mechanisms of action of arsenic have been described and they are summarized in Figure 23–1. The mechanisms of arsenic toxicity and carcinogenicity have been reported for various cell types [45]. Several lines of evidence support stimulation of cell proliferation, alteration of DNA methylation, and perturbation of signaling cascades [46], oxidative stress, and



FIGURE 23-1 Potential mechanisms of arsenic-induced carcinogenesis.

chromosomal aberrations [47]. Disruption of transcriptional activity following arsenic exposure with extensive changes in global gene expression, and disruption of diverse regulatory mechanisms of gene expression [21,48,49] have been reported. Epigenetic dysregulation and arsenic metabolic activities involving human microbiome and liver enzymes have also been reported [50,51].

Although much progress has been recently made in the area of arsenic's possible mode(s) of carcinogenic action, a scientific consensus has not yet been reached. A recent review discusses nine different possible modes of action of arsenic carcinogensis: induced chromosomal abnormalities, oxidative stress, altered DNA repair, altered DNA methylation patterns, altered growth factors, enhanced cell proliferation, promotion/progression, suppression of p53, and gene amplification [52]. Presently, three modes (chromosomal abnormality, oxidative stress, and altered growth factors) of arsenic carcinogenesis have shown a degree of positive evidence, both in experimental systems (animal and human cells) and in human tissues. The remaining possible modes of carcinogenic action (progression of carcinogenesis, altered DNA repair, p53 suppression, altered DNA methylation patterns, and gene amplification) do not have as much evidence, particularly from *in vivo* studies with laboratory animals, *in vitro* studies with cultured human cells, or human data from case or population studies. Thus, the mode-of-action studies suggest that arsenic might be acting as a cocarcinogen, a promoter, or a progressor of carcinogenesis.

23.3.1 Arsenic Perturbation of Keratin Expression

Aberrant and progressive alterations in cytokeratin (CK) expression are a common feature observed in human skin lesions, including hyperkeratosis and squamous cell carcinoma (SCC) [53]. Thus, the expression of simple epithelial keratins could be used as a biomarker for monitoring malignancy of keratinocytes and to determine tumor invasion and/or changes in epithelial-mesenchymal interactions [54]. Sun et al., in their report, showed that the expression of CK in HaCaT keratinocytes was impaired by chronic arsenic exposure [53]. Keratins are the major abundant and structural proteins of the epidermal layer of epithelial cells. It has been suggested that the expression of CK in arsenic carcinogenesis of skin may be exacerbated by an increase in the expression of other proteins such as MMP-9 (matrix metalloproteinase-9 secretion, an enzyme often secreted by cancer cells to help them invade through the local extracellular matrix) [53]. The transcripts of various keratin genes used as markers of different stages of cell differentiation have been reported to be increased in arsenic-treated HaCaT cells. These include CK1, associated with hyperkeratosis, CK10, always co-expressed with CK1, CK13, a biomarker for dermal cancer progression, CK8 and CK18 [54], differentiation marker genes in simple epithelial tissues as they are the earliest keratin genes expressed during embryogenesis, filaggrin (linked to proliferation) [55], involucrin (an early differentiation marker) and loricrin (late differentiation gene), CK6 (induced in response to stressful stimuli, such as wounding), and CK 6/16 and 7/17 (stress response, adaptive response to arsenic) [53].

23.3.2 Arsenic-Induced Genotoxicity

Disruptions in gene expression are responsible for many diseases [56] and the amount and nature of mRNA produced by a cell have been used in gene expression studies to understand genomic stability and chromosomal aberration by observing genes that were either overexpressed or suppressed as a result of exposure to arsenic [47,48]. Inorganic arsenic is said to modulate expression and DNA-binding activities of several key transcription factors, including nuclear factor kappa B, tumor suppressor protein (p53), and activating protein-1 (AP-1) [57]. Arsenic activates AP-1 stimulating the mitogen-activated protein kinase (MAPK) cascade, which leads to an increase in the expression and/or phosphorylation of the two major AP-1 constituents, *c-jun* and *c-fos* [58]. Also, high doses of arsenic have been demonstrated to be clastogenic, while low doses are aneugenic and induce sister chromatid exchanges in a variety of mammalian cells *in vitro* [59]. One of the suggested mechanisms of arsenic skin carcinogenesis is the disruption of cycle check point proteins and inhibition of DNA repairs, causing production of hydroxyl radicals (*HO), DNA damage [60], and aberrant CK expression [53].

23.3.3 Arsenic-Induced Aberration of Gene Expression

Induction of aberrations in gene expression is regarded as an important mechanism of action of arsenic. Genotoxic and cytotoxic properties of arsenic have been demonstrated using a cultured human skin keratinocyte model [61]. Gene expression studies involve the application of advanced technologies such as microarray, which generates high-throughput data and provides a platform to simultaneously examine multiple mechanisms based on alterations in expression of target genes. Gene expression studies on arsenic toxicity/carcinogenicity include: skin [62], keratinocytes [45], kidney [63], myeloma [64], peripheral lymphocytes [62], neural tube [65], liver [66], and urogenital cells [67].

23.3.4 Arsenic Dysregulation of Cellular Immune Function

Arsenic exposure induces dysfunction of the cellular immune system by targeting the CD4⁺ cells with concomitant reduction in CD4⁺ cells in epidermal keratinocytes, which may trigger arsenic-induced skin cancer [68,69]. There are reports of induction of aberration in the expression of immune response genes including up-regulation of inflammatory signals like cytokines and TNF-alpha in HaCaT keratinocytes chronically exposed to a low dose of arsenic trioxide [48]. *AKR1C3* (aldo-keto reductase family 1, member C3), *IGFL1* (insulin growth factor-like family member 1), *IL1R2* (interleukin 1 receptor, type 2), and *TNFSF18* (tumor necrosis factor [ligand] superfamily, member 18) have been up-regulated in HaCaT cells exposed to low doses of arsenic trioxide with down-regulation of *RGS2* (regulator of G-protein signaling 2) [48]. *IL1R2* improves cell migration, which increases its oncogenic potential [70]. Arsenic induces up-regulation of *IGFL1*. The *IGF*-like (*IGFL1*) genes encode proteins that contain 11 conserved cysteine residues at fixed positions including two CC motifs. Since *IGLF1* encodes proteins rich in cysteine residues, it could be vital in arsenic binding and responsiveness in keratinocytes [71].

23.3.5 Arsenic Distortion of Protein Structure

Arsenic impairs cellular respiration by inhibiting various mitochondrial enzymes, and the uncoupling of oxidative phosphorylation. Toxic by-products are released when arsenic interacts with sulfhydryl groups of proteins and enzymes, and substitutes phosphorus in a variety of biochemical reactions [72]. For example, dihydrolipoyl dehydrogenase and thiolase enzymes are inactivated when arsenic reacts with their sulfhydryl groups causing inhibited oxidation of pyruvate and beta-oxidation of fatty acids. Arsenic also distorts the conformation of protein structure by attacking the disulfide bonds and thiol groups and binding to vicinal cysteines [73]. Arsenites interfere with sulfhydryl group of amino acids and disturb protein structure, while arsenates substitute for phosphate, affecting cellular processes such as ATP and DNA synthesis [74].

23.3.6 Arsenic Induction of Cell Proliferation

Induction of cell growth by low concentration of arsenic has been observed by many scientists. In human keratinocyte studies, arsenic was observed to induce overexpression of growth factors [75], cyclin A with increase in S-phase population of cells in cell cycle [46], cyclin-dependent kinase 4 (CDK4), transcription factor 1 (E2F1), granulocyte macrophage-colony stimulating factor (GM-CSF), and transforming growth factor-alpha (TGF-alpha) [46,75]. Low concentrations (0.5 to 1 ppm) of AsO₃ stimulate keratinocyte proliferation but induce cell death at higher concentrations (>1 ppm) [76]. The cell death may have been caused by the activation of caspase-3, which triggers cell cycle arrest at the G2-M phase [61]. *IGFL1* is known to be up-regulated in skin conditions such as psoriasis that promote the abnormal proliferation and differentiation of epidermal keratinocytes [77]. IGFL1, like other related IGFs, is involved in cellular energy metabolism, growth and development, and promotion of cell division [71]. Also, increase in cell proliferation by arsenic could be attributed to induction of *cyclin D1* transcription [78]. Cyclin D1 stimulates growth by shifting the G1 growth phase into the S/G2 cell cycle compartment.

23.3.7 Arsenic Dysregulation of Epigenetic Mechanisms

An epigenetic trait is a stably inherited phenotype resulting from changes in a chromosome without alterations in the DNA sequence [79]. Both genetic and epigenetic mechanisms can regulate gene expression and they must be put into perspective during a study on the toxicogenomics and cell-transforming ability of arsenic and other toxicants capable of causing aberrations in gene expression and cancer. The three major epigenetic mechanisms reported in arsenic toxicity and carcinogenicity are: DNA methylation [80], histone modification (acetylation, methylation, and phosphorylation) [81], and microRNA (miRNA) expression [82]. Hypomethylation of DNA may cause overexpression of oncogenic genes [83] and decrease in DNA repair, stress defense mechanisms, and apoptosis [45,83].

23.3.8 Arsenic Cocarcinogenicity

Arsenic has also been considered as a weak mutagen and may not initiate but potentiate the mutagenicity, cytotoxicity, and clastogenicity of other carcinogens such as UV radiation, X-rays, and other heavy metals by acting as a tumor promoter, or cocarcinogen, in skin cancer development. According to Yu et al. [69], As and UVB stimulated caspase pathways, and caspase-9 and caspase-8 signaling, respectively, which induced apoptosis in keratinocytes [69].

23.3.9 Arsenic Interference with Signal Transduction

Cell cycle regulation is crucial for proper cellular homeostasis. Communication between or within a cell is done through cell signaling and a change in the activity of the cell is sent as a signal that may trigger a cascade of reaction for the body to respond accordingly. Any process that affects this homeostasis affects directly or indirectly the entire cellular processes such as transcription and cell metabolism, regulated by cell signal transduction [84]. Arsenic is known to affect the signaling pathways and the severity may depend on the oxidation state of the arsenic species and cell characteristics [85]. Arsenic perturbs the activation of the mitogenactivated protein kinase pathway, c-Jun N-terminal kinases (JNK), phosphatase, and nuclear factor kappa-light-chain-enhancer of activated B cells (NF- κ B) dysfunction [46,68,84]. NF- κ B dysfunction occurs when arsenic blocks inflammatory signal transduction by inhibiting inhibitors of nuclear factor kappa-B kinase (IKK) required to activate pro-inflammatory transcription factor NF- κ B [68]. The induction of NF- κ B is one of the strong supports of arsenic as a carcinogen since NF- κ B promotes proliferation and angiogenesis. Arsenites also induce the expression of proteins that reduce the toxic effect of arsenic, for instance, the translocation of protein kinase C (PKC) isoforms (PKCepsilon, PKCdelta, and PKCalpha) from cytosol to membranes where these enzymes interact with MAP kinase (Erks, JNKs, and p38 kinases) [86].

23.3.10 Arsenic Induction of Reactive Oxygen Species

An imbalance between the production of cellular oxidant species and the capability of the cells to produce antioxidants results in oxidative stress [87]. Oxidative stress creation is one of the carcinogenesis processes and arsenic is proposed to induce skin carcinogenesis by induction of oxidative stress/reactive oxygen species (ROS) [88]. Arsenic induces morphologic alterations in mitochondrial integrity, which leads to inactivation of mitochondrial enzymes and loss of mitochondrial membrane potential. Arsenite may serve also as a bypass for electrons from the respiratory chain that facilitate the formation of superoxide anion radicals and generation of ROS (H_2O_2 , O_2 , ROO, OH, and NO) [60] as well as a reduction in activity of an important cellular antioxidant, glutathione (gamma-glutamylcysteinylglycine, GSH) [89]. Arsenite reduces oxygen directly to H_2O_2 and/or formation of arsenic peroxyl radicals, which are mediators of DNA damage [60], mitosis disrupters, and apoptosis promoters [88]. In his studies, Aposhian [90] observed that inorganic arsenate stimulates a rapid burst of oxidative stress in mammalian cells as a result of the repetitive reduction of pentavalent to trivalent arsenic followed by the oxidative methylation of trivalent arsenic [90]. Although ROS are known to trigger signals that enhance cell proliferation such as activation of transcription factor AP-1 and nuclear factor NF- κ B [91], they have also induced apoptosis through prolonged activation of JNK/AP-1 and SAPK/JNK signaling. SAPK/JNK signaling activates AP-1, which stimulates apoptosis [92]. AP-1 activity could be very important in explaining the discrepancy observed between the proliferative effect of low dose arsenic and the pro-inflammatory high dose [48,76].

23.4 Health Effects Associated with Arsenic Exposure

Research reports have shown that arsenic is a systemic toxicant that affects almost all the organs of the body and induces various forms of health effects. The adverse effects on health due to chronic arsenic exposure vary and may be influenced by the population groups, age, gender, cumulative dose of arsenic, nutritional status, genetic factors, lifestyle, individual susceptibility, and different chemical forms of arsenic in drinking water [93]. Arsenic is linked with clastogenic damage in different cell types with different endpoints in exposed individuals and in cancer patients. Soluble inorganic arsenic such as As₃O₂ is easily absorbed by the digestive system and may cause acute arsenicosis (arsenic poisoning). Also, inhalation and ingestion of large doses may lead to gastrointestinal symptoms, disturbances of cardiovascular and nervous system functions, and eventually death [8]. Exposure to arsenic has been strongly associated with cancer in human organs including the bladder and kidney [63], liver [66], prostate and lung [94], bone marrow (myeloma), [64], peripheral lymphocytes [62], neural tube [65], urogenital cells [67], and especially the skin [95]. Other arsenicosis diseases are blackfoot disease, atherosclerosis [64], cerebral infraction, hypertension, diabetes mellitus [96], ventricular fibrillation, skin lesions, portal fibrosis of the liver, lung fibrosis, hematologic effects (anemia, leukopenia, and eosinophilia), hearing loss, and neurologic damage (peripheral neuropathy, encephalopathy, and intellectual deficits) in children [9]. The severity of adverse health effects is related to the chemical form of arsenic, and is also time and dose dependent [97,98]. Table 23-2 presents a summary of health effects associated with arsenic exposure.

23.4.1 Arsenic and Keratosis

Exposure to arsenic induces a growth of keratin on the skin referred to as arsenical keratosis. Keratosis is considered as the pre-malignant stage of squamous cell carcinoma *in situ* (Bowen's disease). Other pathological features that lead to Bowen's disease are hyperkeratosis, parakeratosis, and arsenical pigmentation. Two distinct gross variations are observable: first, multiple, punctate, hard, yellow, corn-like papules within areas of friction such as on the soles and palms, and second, slightly elevated, scaly, erythematous or pigmented patches often seen in non-sun-exposed areas.

Since exposure to sun is the major cause of skin cancer and other related skin diseases, appearance of bowenoid lesions in non-sun-exposed areas suggest that arsenic could be the cause. Arsenic bowenoid lesions are non-invasive and can be found in both sun-exposed and non-exposed areas of the skin with characteristic atypical epithelial cells occupying the full thickness of the surface epithelium. Arsenical keratosis and Bowen's

Organ/Tissue Level	Acute Effects	Chronic Effects
Dermatologic	 Capillary flush Contact dermatitis Folliculitis Hair loss 	 Melanosis Bowen's disease Facial edema Palmoplantar hyperkeratosis Cutaneous malignancies Hyperpigmentation
Neurologic	HyperpyrexiaConvulsionsTremor/coma	DesquartationEncephalopathyHeadachePeripheral neuropathy
 Gastrointestinal/hepatic Renal 	 Disorientation Abdominal pain Dysphagia Vomiting Bloody/rice water diarrhea Garlicky odor to breath Mucosal erosions Fatty liver Cholangitis Cholecystitis Tubular damage Glomerular damage 	 Axonal degeneration Nausea Vomiting Diarrhea Anorexia Weight loss Hepatomegaly Jaundice Pancreatitis Cirrhosis Liver cancer Nephritis Proteinuria
• Hematologic	OliguraUremiaAnemiaThrombocytopenia	Bone marrow hypoplasiaAnemiaThrombocytopenia
• Cardiovascular	Ventricular fibrillationTachycardia	 Basophilic stippling Karyorrhexis Arrhythmias Pericarditis Acrocyanosis
Respiratory	Pulmonary edemaBronchial pneumoniaTracheobronchitis	 Raynaud's and gangrene Cough Pulmonary fibrosis Lung cancer

Table 23–2Clinical and Pathological Manifestations of Acute and
Chronic Arsenic Poisoning [97,99]

disease can become invasive squamous cell carcinoma after many years. During this period, changes in skin color occur as illustrated in Figure 23–2, such as hyperpigmentation (darker color) and hypopigmented (lighter color) raindrop-like macules ranging in size from 1 to 2 millimeters [99].



FIGURE 23–2 Basal cell carcinoma. (A) Multifocal basal cell carcinoma of the back. (B) Patient with a single lesion of the left groin [99].

23.4.2 Arsenic and Skin Cancer

There is increased risk of developing epithelial cancers of the skin such as intraepidermal carcinomas (Bowen's disease), squamous cell carcinomas (SCC), basal cell carcinomas (BCC), and Merkel cell carcinoma (MCC) in areas endemic with arsenic-contaminated water with exposure to low dose arsenic over a long period of time [1]. Manifestations of arsenical keratosis, hyperkeratosis, hyperpigmentation, and multiple cutaneous malignancies are used as markers for epithelial cancers of internal organs as they often coexist and also because the internal organ cancers have no characteristic histologic features that will infer arsenical etiology [100,101]. SCC and BCC are the most common types of malignant skin cancers. SCC lesions are invasive skin tumors that arise from the surface epidermal layer. Histologically, they look like normal surface squamous cells but with atypical features and growth patterns. Typically, SCC is seen in sun-exposed areas of the body, but could appear in sun-protected areas due to chronic arsenic ingestion. BCC is most often seen in the trunk and scrotum and is associated with arsenic exposure [102]. It arises from the lower part of the epidermis and shows a variety of growth patterns including adenoid, reticulated, trichoepitheliomatous, and hyperpigmented. Histologically, there could be multinucleated giant cells present; some may have bizarre appearance and atypical mitoses [103]. Figure 23-2 illustrates two cases of arsenic-induced basal cell carcinoma [99].

Malignant melanomas account for the majority of deaths due to skin cancer as they can spread to other tissues, unlike SCC and BCC. Melanomas begin in the pigment-making cells, melanocytes located along the base of the squamous epithelium. There is evidence that arsenic can induce melanoma directly or as a cocarcinogen [104]. However, more cases of arsenic-related skin cancers are linked with SCC and BCC than with melanoma [105].

23.4.3 Arsenic and Liver Cancer

The liver is a very important organ that performs many essential functions related to digestion, metabolism, immunity, and the storage of nutrients, and especially detoxification of substances entering the body. Exposure to toxic agents may cause temporary hepatic lesions that may resolve when the toxin is removed, or evolve to hepatic failure, malignancy, or death. Chronic arsenic exposure leads to liver injury when the cell's organelles are damaged. Chronic arsenic exposure has been associated with a variety of hepatic dysfunctions including macrovesicular steatosis, phospholipidosis, cholestatic lesions, steatohepatitis, granulomatous reactions, fibrosis, cirrhosis, vascular lesions, and/or neoplasms, depending on the dose and exposure conditions. Arsenic has been reported to enhance hepatic morphological and biochemical changes in phenobarbital-pretreated rats. Hydrophic degeneration, total loss of glycogen, necrosis in some centrolobular zones, and an increase in lipid vacuoles around the periportal area have been observed [106]. Also, several other hepatic histopathological changes such as disruption of hepatic cord, sinusoidal dilation, fatty infiltration, and altered expression of cyclin D1, p27, JLK, PTEN, and beta-catenin in the liver have been linked to subchronic exposure to arsenic [107].

Malignant transformation of the sinusoidal endothelial cells is referred to as angiosarcoma, and is linked to exposure to inorganic arsenic as well as vinyl chloride, Fowler's solution, anabolic steroids, and Thorotrast [108]. Histologically, there are two growth patterns of angio-carcinoma: cavernous and solid. The cavernous areas are characterized by plump, atypical endothelial cells that line the dilated sinusoids, while solid areas have sheets of anaplastic-spindled tumor cells that may be accompanied by areas with hemorrhage and necrosis.

Ingestion of arsenic has been linked to hepatocellular carcinoma (HCC), which is another type of malignant transformation of hepatocytes [109]. HCC have been observed in different populations exposed to elevated levels of arsenic in drinking water including Japan, Mexico, Chile, Germany, Argentina, Taiwan, China, India, and Bangladesh. It is suggested that arsenic could be interacting with other risk factors, including hepatitis B and C infection, aflatoxin, alcohol abuse, and genetic hemochromatosis [94,110]. Histologically, it could be difficult to differentiate between hepatocellular carcinoma and normal liver, except that a well-differentiated HCC lacks acinar architecture, and it is positive for sinusoidal staining with CD34 (Q-Bend-10) immunostain, which is increased in neoplastic liver; also the growth pattern may be varied [111]. Moderately differentiated HCC has increased nuclear pleomorphism, mitoses, growth pattern abnormalities (thick-ened trabeculae), and lacks normal acinar architecture, while diagnosis of poorly differentiated HCC depends on clinical manifestations to exclude metastatic disease. Figure 23–3 shows the histological sections of a normal liver (A) and neoplasia associated with chronic exposure to arsenic, including hepatocellular carcinoma (B) and liver angiosarcoma (C) [99].

23.4.4 Arsenic and Kidney Cancer

Chronic arsenic exposure is capable of causing chronic renal insufficiency from cortical necrosis. Acute arsenic poisoning may cause acute tubular necrosis, with acute renal failure



FIGURE 23–3 Histological sections of a normal liver (A), and neoplasia associated with chronic exposure to arsenic, including hepatocellular carcinoma (B) and liver angiosarcoma (C) [99].

commonly being seen. Arsine gas and arsenic are known to cause tubular necrosis but arsine gas is more nephrotoxic. There is no strong evidence to prove that arsenic directly causes kidney cancer; however, a study of smelter workers in Tacoma, Washington (USA), showed a 30% increase in incidence of renal carcinoma in those who were chronically exposed to arsenic as a by-product of smelting non-ferrous metal ores [1,9,112,113]. The mining workers and copper smelters also showed increased rates of lung cancer, gastrointestinal cancer, and hematolymphatic malignancies [112].

23.4.5 Arsenic and Urinary Bladder Cancer

Transitional cell carcinoma (TCC), also called urothelial cell carcinoma (UCC), typically involves the urinary system. TCC has been associated with arsenic exposure in drinking water in Argentina, Chile, and Taiwan [110,114]. Experimental research has also shown that arsenic affects urothelial cells [67], and the development of TCC is dose related [114]. TCC has a latent period of about 8 to 20 years and arises from the lining of epithelial cells of the renal pelvis, ureters, and urinary bladder. There are two forms of UCC: a low-grade papillary tumor, which may be transient and recurs; and a high-grade malignancy, which develops into dysplasia or carcinoma *in situ*, and may also be invasive [115].

23.4.6 Arsenic and Lung Cancer

There are several experimental and epidemiological reports linking arsenic exposure and lung cancer [44]. It is estimated that 1.5 million workers in the United States are occupationally
exposed to arsenic in industries that manufacture glass, pigments, pesticides, and paints. Studies have shown a dose-related association between arsenic in drinking water and both kidney and lung cancers [110]. In arsenic-exposed workers, there is a systematic gradient in lung cancer mortality rate, which depends on the duration and intensity of exposure and also on exposure to radiation, asbestos, radon, and inhaled substances such as nickel, chromates, arsenic, and genetic factors. But exposures to cigarette smoke and sulfur dioxide have not been confirmed as co-factors in arsenic-induced lung cancer [1,9,112].

Arsenic has a predilection for epithelial cells and in the lung it causes a malignant epithelial neoplasm called pulmonary adenocarcinoma, which has features of glandular differentiation. Pulmonary adenocarcinoma is seen mostly in the periphery of the lung, and women are more affected than men. Histologically, pulmonary adenocarcinoma is recognized based on the observation of intracellular mucin production. In well-differentiated adenocarcinoma, the glands are lined by tall columnar or mucinous epithelium with ample cytoplasm, basal nuclei, prominent nucleoli, and frequent mitoses. As differentiation decreases, there is increased nuclear atypia and mitosis [116]. SCC can also be found in the lung apart from the skin. In the lung, it is found in the bronchus, close to the hilum [116].

23.4.7 Arsenic and Gastrointestinal Cancer

Gastric carcinoma is the term for stomach cancer and adenocarcinoma is the most common form of cancer found in the gastrointestinal tract. Literature is replete with conflicting reports linking carcinoma of the gastrointestinal tract and arsenic ingestion [2]. Observations were made mostly among factory workers who had developed multiple cancers many years after initial exposure. For example, a study of Swedish glass-blowers exposed to high consumption of lead, arsenic, antimony, and manganese revealed increased risk of death from cancer of the stomach, lung, and colon [117]. Also, a study of 839 copper smelters in Japan found significant increase in mortalities from lung and colon cancers [118]. In Ontario, Canada, excess mortality from stomach cancer was observed among gold miners [119].

23.4.8 Arsenic and Brain Cancer

The primary malignant brain tumor is referred to as glioma because it originates from the glial cells of the nervous system. In the brain, gliomas are commonly found in the cerebral hemispheres but other areas may be affected including the optic nerve, the brain stem, and, particularly among children, the cerebellum. An occupational exposure study in Sweden revealed an increased incidence of gliomas in both men and women and also an increased incidence of meningiomas in women due to exposure to chemicals such as arsenic, mercury, and petroleum products [120]. About 50% of all primary brain tumors and about 20% of all primary spinal cord tumors are gliomas. There exist different groups of gliomas based on the type of glial cell involved. Astrocytomas develop from astrocytes (star-shaped glial cells). Astrocytomas, the most common type of glioma, are also the most common type of primary brain tumor. Arsenic-exposed patients may develop destruction of axonal cylinders, leading to peripheral neuropathy [1].

23.5 Conclusions

This review of published scientific literature indicates that both anthropogenic and geogenic contamination of the environment constitute a major public health problem of global proportions. Human exposure to arsenic may occur via several pathways including oral ingestion, inhalation, and dermal contact. Although arsenic is considered a systemic toxicant that affects many target organs and tissues causing adverse dermatologic, neurologic, gastrointestinal and hepatic, renal, hematologic, cardiovascular, reproductive, and respiratory effects, the most critical damage it gives rise to is associated with its ability to cause a variety of human cancers. Several epidemiological studies have reported significant increases in incidence and mortality rates related to skin, lung, liver, kidney, urinary bladder, and colon neoplasms. Research has also elucidated several important mechanisms associated with arsenic carcinogenesis including induction of oxidative stress and activation of proto-oncogenes, down-regulation of proapoptotic genes, modulation of signal transduction, and alterations of cell cycle control genes.

Acknowledgments

This work has been supported by a grant from the National Institutes of Health (Grant No. G12MD07581) through the NIMHD-RCMI Center for Environmental Health at Jackson State University (Jackson, Mississippi, USA).

References

- [1] Agency for Toxic Substances and Disease Registry. Arsenic toxicity: case studies in environmental medicine. Atlanta, GA: Agency for Toxic Substances and Disease Registry; 2010.
- [2] IARC. IARC Monographs on the Evaluation of Carcinogenic Risk to Humans. A Review of Human Carcinogens: Arsenic, Metals Fibers and Dusts, Volume 100C. Lyon, France; 2012.
- [3] WHO. Arsenic in drinking-water: background document for development of WHO guidelines for drinking-water quality. Geneva: World Health Organization; 2011. Available at: http://www.who.int/water_sanitation_health/dwq/chemicals/arsenic.pdf>.
- [4] Onishi H. Arsenic. In Wedepohl KH, editor. Handbook of Geochemistry. New York: Springer; 1969.
- [5] Cohen SM, Arnold LL, Eldan M, Lewis AS, Beck BD. Methylated arsenicals: the implications of metabolism and carcinogenicity studies in rodents to human risk assessment. Crit Rev Toxicol 2006;36:99–133.
- [6] Tinwell H, Stephens SC, Ashby J. Arsenite as the probable active species in the human carcinogenicity of arsenic: mouse micronucleus assays on Na and K arsenite, orpiment, and Fowler's solution. Environ Health Perspect 1991;95:205–10.
- [7] Emsley J. Nature's building blocks: an A-Z guide to the elements. Oxford, UK: Oxford University Press; 2001. pp. 43, 513, 529.
- [8] Gomez-Caminero A, Howe P, Hughes M, Kenyon DR, Lewis DR, Moore M, et al. The International Programme on Chemical Safety (IPCS), environmental health criteria for arsenic and arsenic compounds. Geneva: World Health Organization (WHO); 2001. Retrieved 02/04/2014.
- [9] Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for arsenic. Atlanta, GA: ATSDR; 2007.
- [10] Abdel-Moati AR. Speciation and behavior of arsenic in the Nile Delta lakes. Water Air Soil Pollut. 1990;51: 117-32.

- [11] Abdelghani AA, Reimers RS, Anderson AC, Englande AJ, Lo CP, Shariatpanahi M. Transport and distribution of arsenic in sediments. Proceedings of the 3rd International Conference on Heavy Metals in the Environment. Amsterdam, Netherlands; 1981. pp. 665–8.
- [12] US Environmental Protection Agency (US EPA). Pesticide news story: EPA finds organic arsenical herbicides ineligible for reregistration. Atlanta GA: US Environmental Protection Agency; 2006, pp. 7–45.
- [13] Liu CP, Luo CL, Gao Y, Li FB, Lin LW, Wu CA, et al. Arsenic contamination and potential health risk implications at an abandoned tungsten mine, southern China. Environ Pollut 2010;158:820–6.
- [14] Nickson R, McArthur J, Burgess W, Ahmed KM, Ravenscroft P, Rahman M. Arsenic poisoning of Bangladesh groundwater. Nature 1998;395:338.
- [15] Smith AH, Smith MM. Arsenic drinking water regulations in developing countries with extensive exposure. Toxicology 2004;198:39–44.
- [16] Chakraborti D, Das B, Murrill MT. Examining India's groundwater quality management. Environ Sci Technol 2011;45:27–33.
- [17] Saxe JK, Wannamaker EJ, Conklin SW, Shupe TF, Beck BD. Evaluating landfill disposal of chromated copper arsenate (CCA) treated wood and potential effects on groundwater: evidence from Florida. Chemosphere 2007;66:496–504.
- [18] Pandey PK, Sharma R, Roy M, Roy S, Pandey M. Arsenic contamination in the Kanker district of centraleast India: geology and health effects. Environ Geochem Health 2006;28:409–20.
- [19] Uneyama C, Toda M, Yamamoto M, Morikawa K. Arsenic in various foods: cumulative data. Food Addit Contam 2007;24:447–534.
- [20] Burger J, Gochfeld M. Heavy metals in commercial fish in New Jersey. Environ Res 2005;99:403–12.
- [21] Das HK, Mitra AK, Sengupta PK, Hossain A, Islam F, Rabbani GH. Arsenic concentrations in rice, vegetables, and fish in Bangladesh: a preliminary study. Environ Int 2004;30:383–7.
- [22] Schoof RA, Yost LJ, Eickhoff J, Crecelius EA, Cragin DW, Meacher DM, et al. A market basket survey of inorganic arsenic in food. Food Chem Toxicol 1999;37:839–46.
- [23] EFSA. Scientific opinion on arsenic in food. Eur Food Saf Authority J 2009;7:1351.
- [24] Roychowdhury T, Uchino T, Tokunaga H, Ando M. Survey of arsenic in food composites from an arsenicaffected area of West Bengal, India. Food Chem Toxicol 2002;40:1611–21.
- [25] United States Environmental Protection Agency (US EPA). The risk assessment guidelines of 1986. Washington, DC: United States Environmental Protection Agency; 1987. EPA/600/8-87/045.
- [26] Andreae MO, Byrd JT, Froehlich PN. Arsenic, antimony, germanium, and tin in the Tejo estuary, Portugal: modeling a polluted estuary. Environ Sci Technol 1983;17:731–7.
- [27] Roberts SM, Weimar WR, Vinson JR, Munson JW, Bergeron RJ. Measurement of arsenic bioavailability in soil using a primate model. Toxicol Sci 2002;67:303–10.
- [28] Gong Z, Lu X, Ma M, Watt C, Le XC. Arsenic speciation analysis. Talanta 2002;58:77–96.
- [29] United States Environmental Protection Agency (US EPA). National primary drinking water regulations: arsenic and clarifications to compliance and new source contaminants monitoring. Final Rule. Fed Reg 2001. Atlanta, GA: US EPA; 2001. pp. 6976–7066.
- [30] Boyce CP, Lewis AS, Sax SN, Beck BD, Eldan M, Cohen SM. Probabilistic modeling of dietary arsenic exposure. Environ Health Perspect 2010;118:a331–2.
- [31] European Commission (EC). Ambient air pollution by As, Cd and Ni compounds (position paper-final). Luxembourg: European Commission; 2000. pp. 86–109.
- [32] Jolliffe DM. A history of the use of arsenicals in man. J R Soc Med 1993;86:287–9.
- [33] Cuzick J, Sasieni P, Evans S. Ingested arsenic, keratoses, and bladder cancer. Am J Epidemiol 1992;136:417-21.

- [34] Antman KH. Introduction: the history of arsenic trioxide in cancer therapy. Oncologist 2001;6 (Suppl. 2):1-2.
- [35] Riethmiller S. From Atoxyl to Salvarsan: searching for the magic bullet. Chemotherapy 2005;51:234–42.
- [36] Sekeres MA. New data with arsenic trioxide in leukemias and myelodysplastic syndromes. Clin Lymphoma Myeloma 2007;8(Suppl. 1):S7–S12.
- [37] Hussein MA. Arsenic trioxide: a new immunomodulatory agent in the management of multiple myeloma. Med Oncol 2001;18:239–42.
- [38] Munshi NC, Tricot G, Desikan R, Badros A, Zangari M, Toor A, et al. Clinical activity of arsenic trioxide for the treatment of multiple myeloma. Leukemia 2002;16:1835–7.
- [39] Miller Jr. WH, Schipper HM, Lee JS, Singer J, Waxman S. Mechanisms of action of arsenic trioxide. Cancer Res 2002;62:3893–903.
- [40] Bisser S, N'Siesi FX, Lejon V, Preux PM, Van NS, Miaka Mia BC, et al. Equivalence trial of melarsoprol and nifurtimox monotherapy and combination therapy for the treatment of second-stage *Trypanosoma brucei gambiense* sleeping sickness. J Infect Dis 2007;195:322–9.
- [41] Wagner SE, Peryea FJ, Filby RA. Antimony impurity in lead arsenate insecticide enhances the antimony content of old orchard soils. J Environ Qual 2003;32:736–8.
- [42] Bhumbla DK, Keefer RF. Arsenic in environment. Part I. Nriagu JO, editor. Cycling and characterization. New York: John Wiley & Sons Inc.; 1994. pp. 51–82.
- [43] Yager JW, Hicks JB, Fabianova E. Airborne arsenic and urinary excretion of arsenic metabolites during boiler cleaning operations in a Slovak coal-fired power plant. Environ Health Perspect 1997;105:836–42.
- [44] Enterline PE, Day R, Marsh GM. Cancers related to exposure to arsenic at a copper smelter. Occup Environ Med 1995;52:28–32.
- [45] Hamadeh HK, Trouba KJ, Amin RP, Afshari CA, Germolec D. Coordination of altered DNA repair and damage pathways in arsenite-exposed keratinocytes. Toxicol Sci 2002;69:306–16.
- [46] Chowdhury R, Chatterjee R, Giri AK, Mandal C, Chaudhuri K. Arsenic-induced cell proliferation is associated with enhanced ROS generation, Erk signaling and CyclinA expression. Toxicol Lett 2010;198:263–71.
- [47] Kundu M, Ghosh P, Mitra S, Das JK, Sau TJ, Banerjee S, et al. Precancerous and non-cancer disease endpoints of chronic arsenic exposure: the level of chromosomal damage and XRCC3 T241M polymorphism. Mutat Res 2011;706:7–12.
- [48] Udensi UK, Cohly HH, Graham-Evans BE, Ndebele K, Garcia-Reyero N, Nanduri B, et al. Aberrantly expressed genes in HaCaT keratinocytes chronically exposed to arsenic trioxide. Biomark Insights 2011;6:7–16.
- [49] Isokpehi RD, Udensi UK, Anyanwu MN, Mbah AN, Johnson MO, Edusei K, et al. Knowledge building insights on biomarkers of arsenic toxicity to keratinocytes and melanocytes. Biomark Insights 2012;7:127-41.
- [50] Pinyayev TS, Kohan MJ, Herbin-Davis K, Creed JT, Thomas DJ. Preabsorptive metabolism of sodium arsenate by anaerobic microbiota of mouse cecum forms a variety of methylated and thiolated arsenicals. Chem Res Toxicol 2011;24:475–7.
- [51] Van de Wiele T, Gallawa CM, Kubachka KM, Creed JT, Basta N, Dayton EA, et al. Arsenic metabolism by human gut microbiota upon in vitro digestion of contaminated soils. Environ Health Perspect 2010;118:1004–9.
- [52] Kitchin KT. Recent advances in arsenic carcinogenesis: modes of action, animal model systems, and methylated arsenic metabolites. Toxicol Appl Pharmacol 2001;172:249–61.
- [53] Sun Y, Pi J, Wang X, Tokar EJ, Liu J, Waalkes MP. Aberrant cytokeratin expression during arsenic-induced acquired malignant phenotype in human HaCaT keratinocytes consistent with epidermal carcinogenesis. Toxicology 2009;262:162–70.

- [54] Markey AC, Lane EB, Churchill LJ, MacDonald DM, Leigh IM. Expression of simple epithelial keratins 8 and 18 in epidermal neoplasia. J Invest Dermatol 1991;97:763–70.
- [55] Winge MC, Hoppe T, Berne B, Vahlquist A, Nordenskjold M, Bradley M, et al. Filaggrin genotype determines functional and molecular alterations in skin of patients with atopic dermatitis and ichthyosis vulgaris. PLOS ONE 2011;6:e28254.
- [56] Wheeler DL, Barrett T, Benson DA, Bryant SH, Canese K, Chetvernin V, et al. Database resources of the National Center for Biotechnology Information. Nucleic Acids Res 2008;36:D13–21.
- [57] Barchowsky A, Dudek EJ, Treadwell MD, Wetterhahn KE. Arsenic induces oxidant stress and NF-kappa B activation in cultured aortic endothelial cells. Free Radic Biol Med 1996;21:783–90.
- [58] Simeonova PP, Wang S, Toriuma W, Kommineni V, Matheson J, Unimye N, et al. Arsenic mediates cell proliferation and gene expression in the bladder epithelium: association with activating protein-1 transactivation. Cancer Res 2000;60:3445–53.
- [59] Klein CB, Leszczynska J, Hickey C, Rossman TG. Further evidence against a direct genotoxic mode of action for arsenic-induced cancer. Toxicol Appl Pharmacol 2007;222:289–97.
- [60] Shi H, Hudson LG, Ding W, Wang S, Cooper KL, Liu S, et al. Arsenite causes DNA damage in keratinocytes via generation of hydroxyl radicals. Chem Res Toxicol 2004;17:871–8.
- [61] Graham-Evans B, Cohly HH, Yu H, Tchounwou PB. Arsenic-induced genotoxic and cytotoxic effects in human keratinocytes, melanocytes and dendritic cells. Int J Environ Res Public Health 2004;1:83–9.
- [62] Argos M, Kibriya MG, Parvez F, Jasmine F, Rakibuz-Zaman M, Ahsan H. Gene expression profiles in peripheral lymphocytes by arsenic exposure and skin lesion status in a Bangladeshi population. Cancer Epidemiol Biomarkers Prev 2006;15:1367–75.
- [63] Sasaki A, Oshima Y, Fujimura A. An approach to elucidate potential mechanism of renal toxicity of arsenic trioxide. Exp Hematol 2007;35:252–62.
- [64] Wang MC, Liu SX, Liu PB. Gene expression profile of multiple myeloma cell line treated by realgar. J Exp Clin Cancer Res 2006;25:243–9.
- [65] Wlodarczyk BJ, Cabrera RM, Hill DS, Bozinov D, Zhu H, Finnell RH. Arsenic-induced gene expression changes in the neural tube of folate transport defective mouse embryos. Neurotoxicology 2006;27:547–57.
- [66] Liu J, Xie Y, Ducharme DM, Shen J, Diwan BA, Merrick BA, et al. Global gene expression associated with hepatocarcinogenesis in adult male mice induced by in utero arsenic exposure. Environ Health Perspect 2006;114:404–11.
- [67] Su PF, Hu YJ, Ho IC, Cheng YM, Lee TC. Distinct gene expression profiles in immortalized human urothelial cells exposed to inorganic arsenite and its methylated trivalent metabolites. Environ Health Perspect 2006;114:394–403.
- [68] Kapahi P, Takahashi T, Natoli G, Adams SR, Chen Y, Tsien RY, et al. Inhibition of NF-kappa B activation by arsenite through reaction with a critical cysteine in the activation loop of Ikappa B kinase. J Biol Chem 2000;275:36062–66.
- [69] Yu HS, Liao WT, Chai CY. Arsenic carcinogenesis in the skin. J Biomed Sci 2006;13:657-66.
- [70] Chang SY, Su PF, Lee TC. Ectopic expression of interleukin-1 receptor type II enhances cell migration through activation of the pre-interleukin 1alpha pathway. Cytokine 2009;45:32–8.
- [71] Emtage P, Vatta P, Arterburn M, Muller MW, Park E, Boyle B, et al. IGFL: a secreted family with conserved cysteine residues and similarities to the IGF superfamily. Genomics 2006;88:513–20.
- [72] Li JH, Rossman TG. Inhibition of DNA ligase activity by arsenite: a possible mechanism of its comutagenesis. Mol Toxicol 1989;2:1–9.
- [73] Isokpehi RD, Cohly HH, Anyanwu MN, Rajnarayanan RV, Tchounwou PB, Udensi UK, et al. Candidate single nucleotide polymorphism markers for arsenic responsiveness of protein targets. Bioinform Biol Insights 2010;4:99–111.

- [74] Sherwood CL, Lantz RC, Burgess JL, Boitano S. Arsenic alters ATP-dependent Ca(2+) signaling in human airway epithelial cell wound response. Toxicol Sci 2011;121:191–206.
- [75] Germolec DR, Yoshida T, Gaido K, Wilmer JL, Simeonova PP, Kayama F, et al. Arsenic induces overexpression of growth factors in human keratinocytes. Toxicol Appl Pharmacol 1996;141:308–18.
- [76] Udensi UK, Graham-Evans BE, Rogers C, Isokpehi RD. Cytotoxicity patterns of arsenic trioxide exposure on HaCaT keratinocytes. Clin Cosmet Investig Dermatol 2011;4:183–90.
- [77] Lobito AA, Ramani SR, Tom I, Bazan JF, Luis E, Fairbrother WJ, et al. Murine insulin growth factor-like (IGFL) and human IGFL1 proteins are induced in inflammatory skin conditions and bind to a novel tumor necrosis factor receptor family member, IGFLR1. J Biol Chem 2011;286:18969–81.
- [78] Hwang BJ, Utti C, Steinberg M. Induction of cyclin D1 by submicromolar concentrations of arsenite in human epidermal keratinocytes. Toxicol Appl Pharmacol 2006;217:161–7.
- [79] Feinberg AP, Tycko B. The history of cancer epigenetics. Nat Rev Cancer 2004;4:143–53.
- [80] Styblo M, Drobna Z, Jaspers I, Lin S, Thomas DJ. The role of biomethylation in toxicity and carcinogenicity of arsenic: a research update. Environ Health Perspect 2002;110(Suppl. 5):767–71.
- [81] Rahman I, Marwick J, Kirkham P. Redox modulation of chromatin remodeling: impact on histone acetylation and deacetylation, NF-kappaB and pro-inflammatory gene expression. Biochem Pharmacol 2004;68:1255–67.
- [82] Ghaffari SH, Bashash D, Dizaji MZ, Ghavamzadeh A, Alimoghaddam K. Alteration in miRNA gene expression pattern in acute promyelocytic leukemia cell induced by arsenic trioxide: a possible mechanism to explain arsenic multi-target action. Tumour Biol 2012;33:157–72.
- [83] Banerjee M, Sarkar J, Das JK, Mukherjee A, Sarkar AK, Mondal L, et al. Polymorphism in the ERCC2 codon 751 is associated with arsenic-induced premalignant hyperkeratosis and significant chromosome aberrations. Carcinogenesis 2007;28:672–6.
- [84] Huang C, Ke Q, Costa M, Shi X. Molecular mechanisms of arsenic carcinogenesis. Mol Cell Biochem 2004;255:57-66.
- [85] Porter AC, Fanger GR, Vaillancourt RR. Signal transduction pathways regulated by arsenate and arsenite. Oncogene 1999;18:7794–802.
- [86] Chen NY, Ma WY, Huang C, Ding M, Dong Z. Activation of PKC is required for arsenite-induced signal transduction. J Environ Pathol Toxicol Oncol 2000;19:297–305.
- [87] Mates JM, Segura JA, Alonso FJ, Marquez J. Intracellular redox status and oxidative stress: implications for cell proliferation, apoptosis, and carcinogenesis. Arch Toxicol 2008;82:273–99.
- [88] Kitchin KT, Conolly R. Arsenic-induced carcinogenesis—oxidative stress as a possible mode of action and future research needs for more biologically based risk assessment. Chem Res Toxicol 2010;23:327–35.
- [89] Peng Z, Geh E, Chen L, Meng Q, Fan Y, Sartor M, et al. Inhibitor of kappaB kinase beta regulates redox homeostasis by controlling the constitutive levels of glutathione. Mol Pharmacol 2010;77:784–92.
- [90] Aposhian HV. Enzymatic methylation of arsenic species and other new approaches to arsenic toxicity. Annu Rev Pharmacol Toxicol 1997;37:397–419.
- [91] Hu Y, Jin X, Snow ET. Effect of arsenic on transcription factor AP-1 and NF-kappaB DNA binding activity and related gene expression. Toxicol Lett 2002;133:33–45.
- [92] Yazdanpanah B, Wiegmann K, Tchikov V, Krut O, Pongratz C, Schramm M, et al. Riboflavin kinase couples TNF receptor 1 to NADPH oxidase. Nature 2009;460:1159–63.
- [93] Ferreccio C, Sancha AM. Arsenic exposure and its impact on health in Chile. J Health Popul Nutr 2006;24:164–75.
- [94] Chen CJ, Yu MW, Liaw YF. Epidemiological characteristics and risk factors of hepatocellular carcinoma. J Gastroenterol Hepathol 1997;12:S294–S308.

- [95] Lynn S, Lai HT, Gurr JR, Jan KY. Arsenite retards DNA break rejoining by inhibiting DNA ligation. Mutagenesis 1997;12:353–8.
- [96] Tseng CH. Arsenic exposure and diabetes mellitus in the United States. JAMA 2008;300:2728-9.
- [97] Tchounwou PB, Patlolla AK, Centeno JA. Carcinogenic and systemic health effects associated with arsenic exposure—a critical review. Toxicol Pathol 2003;31:575–88.
- [98] Tchounwou PB, Wilson BA, Abdelgnani AA, Patlolla AK. Differential cytotoxicity and gene expression in human liver carcinoma (HepG2) cells exposed to arsenic trioxide and monosodium acid methanearsonate (MSMA). Int J Mol Sci 2002;3:1117–32.
- [99] Tchounwou PB, Centeno JA. Gad SC, editor. Toxicologic pathology, in preclinical development handbook: toxicology. NJ, USA: Wiley and Sons Inc. Hoboken; 2008. pp. 551–80.
- [100] Centeno JA, Tchounwou PB, Patlolla AK, Mullick FG, Murakat L, Meza E, et al. Environmental pathology and health effects of arsenic poisoning: a critical review. In: Naidu R, Smith E, Smith J, Bhattacharya P, editors. Managing arsenic in the environment: from soil to human health. Adelaide, Australia: CSIRO Publishing Corp; 2006.
- [101] Maloney ME. Arsenic in dermatology. Dermatol Surg 1996;22:301-4.
- [102] Leonardi G, Vahter M, Clemens F, Goessler W, Gurzau E, Hemminki K, et al. Inorganic arsenic and basal cell carcinoma in areas of Hungary, Romania, and Slovakia: a case-control study. Environ Health Perspect 2012;120:721–6.
- [103] Schwartz RA. Arsenic and the skin. Int J Dermatol 1997;36:241-50.
- [104] Rossman TG, Uddin AN, Burns FJ. Evidence that arsenite acts as a cocarcinogen in skin cancer. Toxicol Appl Pharmacol 2004;198:394–404.
- [105] Guo HR, Yu HS, Hu H, Monson RR. Arsenic in drinking water and skin cancers: cell-type specificity (Taiwan, ROC). Cancer Causes Control 2001;12:909–16.
- [106] Albores A, Cebrian ME, Garcia-Vargas GG, Connelly JC, Price SC, Hinton RH, et al. Enhanced arseniteinduced hepatic morphological and biochemical changes in phenobarbital-pretreated rats. Toxicol Pathol 1996;24:172–80.
- [107] Cui X, Li S, Shraim A, Kobayashi Y, Hayakawa T, Kanno S, et al. Subchronic exposure to arsenic through drinking water alters expression of cancer-related genes in rat liver. Toxicol Pathol 2004;32:64–72.
- [108] Tsai SM, Wang TN, Ko YC. Mortality for certain diseases in areas with high levels of arsenic in drinking water. Arch Environ Health 1999;54:186–93.
- [109] Tchounwou PB, Wilson B, Ishaque A. Important considerations in the development of public health advisories for arsenic and arsenic-containing compounds in drinking water. Rev Environ Health 1999;14:211–29.
- [110] Hopenhayn-Rich C, Biggs ML, Smith AH. Lung and kidney cancer mortality associated with arsenic in drinking water in Cordoba, Argentina. Int J Epidemiol 1998;27:561–9.
- [111] Di C, Fraggetta I, Lombardo F, Azzarello R, Vasquez G, Puleo E. CD 34 expression in chronic and neoplastic liver diseases. Panminerva Med 2002;44:365–7.
- [112] Reger RB, Morgan WK. Respiratory cancers in mining. Occup Med 1993;8:185-204.
- [113] Tchounwou PB, Centeno JA, Patlolla AK. Arsenic toxicity, mutagenesis, and carcinogenesis—a health risk assessment and management approach. Mol Cell Biochem 2004;255:47–55.
- [114] Chiou HY, Chiou ST, Hsu YH, Chou YL, Tseng CH, Wei ML, et al. Incidence of transitional cell carcinoma and arsenic in drinking water: a follow-up study of 8,102 residents in an arseniasis-endemic area in northeastern Taiwan. Am J Epidemiol 2001;153:411–18.
- [115] Royce RK, Ackerman LV. Carcinoma of the bladder: clinical, therapeutic and pathologic aspects of 135 cases. J Urol 1951;65:66–86.

- [116] Travis WD, Travis LB, Devesa SS. Lung cancer. Cancer 1995;75:191–202.
- [117] Wingren G, Axelson O. Mortality in the Swedish glassworks industry. Scand J Work Environ Health 1987;13:412–16.
- [118] Tokudome S, Kuratsune M. A cohort study on mortality from cancer and other causes among workers at a metal refinery. Int J Cancer 1976;17:310–17.
- [119] Kusiak RA, Ritchie AC, Springer J, Muller J. Mortality from stomach cancer in Ontario miners. Br J Ind Med 1993;50:117–26.
- [120] Navas-Acien A, Pollan M, Gustavsson P, Plato N. Occupation, exposure to chemicals and risk of gliomas and meningiomas in Sweden. Am J Ind Med 2002;42:214–27.

This page intentionally left blank



The Association between Chronic Arsenic Exposure and Type 2 Diabetes: A Meta-Analysis

Bayzidur Rahman¹, Tanvir Abir², AKM Akbar Hossain², Md. Rafiqul Islam³, and Abul Hasnat Milton³

¹SCHOOL OF PUBLIC HEALTH AND COMMUNITY MEDICINE, UNIVERSITY OF NEW SOUTH WALES, NSW, AUSTRALIA, ²CENTRE FOR HEALTH RESEARCH, SCHOOL OF MEDICINE, UNIVERSITY OF WESTERN SYDNEY, NSW, AUSTRALIA, ³CENTRE FOR CLINICAL EPIDEMIOLOGY AND BIOSTATISTICS, SCHOOL OF MEDICINE AND PUBLIC HEALTH, UNIVERSITY OF NEWCASTLE, NSW, AUSTRALIA

CHAPTER OUTLINE

24.1 Introduction	. 557
24.2 Methods and Materials	. 559
24.3 Results	. 560
24.4 Discussion	. 568
References	. 570

24.1 Introduction

Arsenic exposure has been considered a serious public health problem in many countries [1]. It is widespread in the environment and some inorganic species can cause detrimental effects on human health even at low exposure levels [2]. As a naturally occurring metalloid, we constantly encounter arsenic in our daily lives and, fortunately, most of these environmental exposures do not pose health effects [3]. Again, arsenic has been used for long time in many ways such as medicine and commercial products [3]. Importantly, the medicinal use of arsenic is notable and has been widely acknowledged for a number of centuries, although it is often referred as the "king of poisons" [3]. It has been used in the Far East and in the ancient Roman and Greek civilizations as a Chinese traditional medicine [4]. William Withering, the

scientist who discovered digitalis, proposed arsenic-based therapies in the 15th century and argued that "poisons in small doses are the best medicine; and the best medicine in too large doses is poisonous" [5]. In 1786, a British physician, Dr. Thomas J. Fowler, discovered a solution of potassium arsenite (known as Fowler's solution) and it was used empirically during the 18th, 19th, and 20th centuries as a curative for many systemic illnesses such as malaria, syphilis, asthma, cholera, eczema, and psoriasis [3,5,6]. Arsenic was reported as a white blood cell count reducing agent in 1878 and as an effective treatment for chronic myeloid leukemia in 1930 [5].

Despite all of this, in 1887, Jonathan Hutchison identified arsenic-induced skin lesions and cancer in patients who were treated with arsenic [7]. A number of studies have been conducted to investigate the effects of chronic arsenic exposure on human health in most parts of the world, especially in countries where arsenic in drinking water is a major concern. Until today, inorganic arsenic exposure was found to be associated with many chronic and systemic diseases [8–12], and type 2 diabetes (T2D)/non-insulin dependent diabetes mellitus (T1D) is one of those diseases that is inconclusively reported to be associated with chronic arsenic exposure. Unlike T1D where insulin production is completely stopped, in T2D the human body is unable to use insulin initially because of insulin resistance but subsequently there is an impaired insulin secretion by the pancreas. The mechanism by which arsenic influences diabetes is unclear. Evidence suggests that chronic exposure to arsenic induces oxidative stress and up-regulates the expression of a few cytokines such as tumor necrosis factor α (TNF- α) and interleukin-6 (IL-6), and these are well-known inducers for insulin resistant [13]. Furthermore, oxidative stress may cause pancreatic β -cell dysfunction, thus lowering insulin secretion [13,14].

However, drinking water containing high amounts of inorganic arsenic is the commonest form of exposure in humans [15]. It is demonstrated that low to high amounts of arsenic in drinking water is associated with higher risk of T2D in many populations, and many of the observational studies reported a dose-response pattern [10,16-18]. Del Razo et al. confirmed this association in 2011 and reports the link between risk of diabetes and production of one of the most toxic inorganic arsenic metabolites, dimethylarsinite (DMA) in urine [19]. Population-specific ecological data showed a positive association between total urinary arsenic and the prevalence of diabetes mellitus [20-22]. Nevertheless, review articles on the association of arsenic with diabetes provide an unconvincing result. One of the reviews of the association studies between arsenic and diabetes reveals that prolonged exposure is associated with T2D [23], while another systematic review reports a lack of evidence to establish a causal role of arsenic in diabetes [24]. Recently, Wang et al. performed a meta-analysis that included studies published between 1990 and 2013. Surprisingly, they did not include all the studies in their meta-analysis. Moreover, inclusion of a study by Wang et al. might have influenced the overall risk [25]. With this conflicting outcome and since there are available many observational and ecological studies from most parts of the world, we decided to conduct a meta-analysis to investigate whether there is a true relationship between arsenic exposure and T2D.

24.2 Methods and Materials

24.2.1 Literature Search

We used a comprehensive search strategy to identify all relevant studies. The search was carried out in Ovid MEDLINE, EMBASE, PubMed, Google Scholar, and Web of Knowledge without limitation on the time of publication. The research question was defined as "What are the associations between chronic arsenic exposure (both in general and for specific by-products) and type 2 diabetes?" This question was then broken down to cover specific search terms such as "diabetes/diabetes mellitus," "arsenic," "arsenic in water," "arsenic pollution," "environmental exposure," "inorganic arsenic," "diabetes," "DM," and "T2D." Each term was included in one or more searches. We also hand-searched for additional relevant studies and cross-checked reference lists of prior review articles to identify any additional papers that had not been retrieved through the electronic databases.

24.2.2 Selection of Studies

The included studies are (1) cross-sectional, (2) case-control, nd (3) cohort studies of chronic arsenic exposure reporting a relative risk (RR) or odds ratio (with 95% confidence interval or p-value to be used to calculate the 95% CI). We also included studies that did not report OR/ RR but reported numbers to be used to calculate those measures. We screened the initial list of articles to identify articles that were irrelevant. We developed a standard data extraction format to extract relevant information from all the included papers.

Two investigators (T.A. and A.H.) initially screened the list of articles to identify the irrelevant articles for exclusion from the list. To facilitate faster screening, articles were first assessed on the basis of their titles to discard the irrelevant articles. Three investigators (T.A., B.R., and A.H.M.) then read the abstracts of the remaining articles to ensure that the main subject of the study was related to arsenic exposure and T2D. Articles whose abstracts showed that the study was not relevant to the objective of the meta-analysis were discarded. Final screening of the articles was based on the full content. After the screening, we extracted information from the selected articles and evaluated the quality of each study.

24.2.2.1 Quality of the Studies

Each article was evaluated in a standardized manner to assess its individual quantity. Quality assessment was facilitated through the use of a standardized questionnaire, which was previously developed for a meta-analysis of arsenic exposure and hypertension [26] and amended for the outcome assessment of diabetes for this study. This scoring instrument has a good reliability (intra-class correlation coefficient of 0.88 for total quality score) and has been described in details elsewhere [26]. Because of the already established high reliability we did not validate it in this study and one scorer scored all the individual studies under four domains: selection of participants 20%, measurement of exposure 30% and outcome 20%, adjustment of confounders 20%, and appropriate statistical analysis 10%.

24.2.2.2 Statistical Analysis

To conduct the meta-analysis, ORs were extracted from case-control and cross-sectional studies and RRs from cohort studies.

For studies where several measures of association were reported, associations that related diabetes to amount of arsenic intake or to arsenic intake combined with duration of exposure were considered. Pooled effect measures were estimated by comparing the highest exposure category to the lowest one combined for males and females.

Effect estimates (OR and RR) were pooled using the inverse-variance-weighted random effects method [27]. Between-studies heterogeneity was measured using I^2 -statistics, which measures the proportion of total variation in study estimates due to heterogeneity [28]. A sensitive analysis was carried out to evaluate the key studies that have substantial impact on the between-study heterogeneity by using the "leave one-out" method [29]. The influence of each study on the pooled effect measure was examined by repeating the meta-analysis while omitting each study one by one [30]. We tested for possible publication bias using Begg's and Egger's tests [31,32]. We also generated a contour enhanced funnel plot to visually inspect asymmetry in the plot and explain the underlying cause [33,34]. Where asymmetry was identified, we adopted the "trim and fill" method to see the effect of correcting the publication bias [35]. This method first identifies the asymmetry in the funnel plot and trims those studies that cause the asymmetry. The pooled estimate with the remaining studies is calculated and then the funnel plot is filled in by replacing the trimmed plots and adding their mirror images in the plot. The final pooled results come from an analysis using all the true estimates and the simulated mirror images.

Meta-regressions of the logs of the effect measures, weighted by the inverse of their variances, on the total quality scores and their subcategories were undertaken to assess the possible impact of study quality on the effect measures. These regressions fitted a random effects model with two additive variance components (within and between studies). We conducted a separate meta-analysis excluding studies that were of very poor quality (<25% of the total possible quality score).

We conducted a random-effects dose-response meta-analysis. A restricted cubic spline model with four knots at the 5th, 35th, 65th, and 95th centiles of the levels of arsenic concentration in drinking water was estimated using generalized least squares regression, taking into account the correlation within each set of published effect measures [36]. Statistical significance of the non-linearity of the dose-response relationship was examined by testing the null hypothesis that the coefficients of the second and third spline were equal. All statistical analyses were performed with STATA V13.

24.3 Results

24.3.1 Literature Search

Our search identified 226 studies, of which 33 reported associations between chronic arsenic exposure and T2D. Five of these articles were excluded from the meta-analysis as none had

reported any relative risk or odds ratio, or whether the risk was calculable (see Table 24–1). Therefore, our final meta-analysis included a total of 27 studies: 18 cross-sectional, five cohort, and four case-control studies.

24.3.2 Quality Scoring

The total and domain-specific quality scores are presented in Table 24–2. Overall, cross-sectional studies scored a little higher than case–control and cohort studies, although cohort studies scored higher on selection. Potential confounders were adjusted more in cross-sectional studies than in the other two designs. All the studies scored poorly on exposure measurement as arsenic exposure measurement remains a challenge to date. In the majority of the studies, a single exposure measurement has been used as a proxy measure for total arsenic exposure for study participants.

We assessed quality with respect to confounding by listing all potential confounders relevant to studies of chronic arsenic exposure and T2D, and counted how many were adjusted for, or shown not to be confounders, in each study. All the study types performed poorly in this aspect too. Analysis was better conducted in cross-sectional studies than cohort and case-control studies. Only two [17,37] out of five cohort studies estimated a hazard ratio or RR with CIs in categories of age, sex, or other relevant variables. Outcome assessment was relatively good in all three study types.

We finally incorporated 27 studies in the analysis including 15 cross-sectional, four casecontrol, and three cohort studies reporting OR, and three cross-sectional and two cohort studies reporting RR. The pooled OR estimate was 1.78 (95% CI: 1.38, 2.30), and the pooled RR was 1.28 (95% CI: 1.03, 1.59). After combining all three types of studies the pooled estimate was 1.76 (95% CI: 1.38, 2.25) with overall between-studies heterogeneity measured by I² statistics of 98.3% (not shown in the analysis).

We repeated the meta-analysis after dropping the study by Wang et al. [25] because in the original paper we find neither an effect measure reported nor any numbers that could be used to estimate one. The overall effect slightly reduced to 1.50 (95% CI: 1.33–1.69) and the I² statistics reduced to 75.5% from 98.3%. We considered this as our final meta-analysis (Figure 24–1). Note that for some of the studies (e.g., [37]) the 95% CI of the effect measures reported in Table 24–1 are slightly different from that in Figure 24–1. This discrepancy occurred for log transformation of the study-specific estimates and their confidence intervals to calculate standard error for doing the meta-analysis and then recalculating the confidence intervals from the calculated standard errors.

24.3.3 Sources of Heterogeneity

The initial meta-analysis demonstrates strong evidence of heterogeneity among studies for chronic arsenic exposure and T2D ($I^2 = 75.5\%$). We conducted a sensitivity analysis to examine the contribution of individual studies to the overall heterogeneity. After excluding the study by Jovanovic et al. [16], the between-study heterogeneity reduced ($I^2 = 68.06\%$). After excluding the studies by Jovanovic et al. and Rahman et al. [38], the between-study heterogeneity reduced

Author, Year	Study Design	Study Population and Location	Diabetes Diagnosis	Arsenic Exposure Categories	Cases/ Non-cases	Odds Ratio/ Relative Risk of T2D (95% CI)	Variables Adjusted for
Lagerkvist and Zetturland, 1994 [49]	Cross- sectional	43 copper smelter workers exposed to iAs; 46 referents worked at a car factory, Sweden	Self-reported T2D	Mean urinary iAs, MMAA and DMAA for arsenic workers = $39 \mu g/L$ (range 5–520 $\mu g/L$) or 14.3 $\mu g/g$ creatinine); mean 9 $\mu g/L$ (range 5–15 $\mu g/L$) for reference population	4/85	9.61 (0.53– 173.62)	Crude
Lai et al., 1994 <mark>[18]</mark>	Cross- sectional	891 adult residents in high-arsenic area, Taiwan	Oral glucose tolerance test (OGTT) or self- reported T2D	Cumulative arsenic exposure (ppm- years): 0 (referent), 0.1–15.0, ≥15.1	86/805	10.05 (1.3–77.9)	Age, sex, BMI, and physical activity
Rahman and Axelson, 1995 [50]	Case– control	Copper smelter workers, Sweden	Death certificate, medical record	Arsenic in air: <0.5 mg/m³, ≅0.5 mg/m³, and >0.5 mg/m³	12/31	3.3 (0.5–30)	Age
Rahman et al., 1996 [51]	Cross- sectional	2456 deaths in glass industry area, 240 exposed and 2216 unexposed to arsenic, Sweden	Death certificate	Exposed vs. unexposed workers	240/2216	1.4 (0.9–2.1)	Age
Jensen and Hansen, 1998 [52]	Cross- sectional	Taxidermists, wood workers, other jobs, Denmark	HbA1C	All arsenic workers ($n = 40$): sum of urinary arsenic, MMAA and DMAA- mean 35.9 (46.2), median 22.3, range 11.5–294.5 nmol of arsenic/ mmol of creatinine	5/59	4.43 (0.47– 41.88)	Age
Rahman et al., 1998 [38]	Cross- sectional	Participants in high and low arsenic area, Bangladesh	Self-reported symptoms + glucosuria + oral glucose tolerance test (OGTT)	Time-weighted arsenic concentration (mg/L): unexposed (referent), <0.5, 0.5–10, >1.0	46/971	5.9 (2.9–11.6)	Age, sex, BMI
Tsai et al., 1999 <mark>[53]</mark>	Retrospective cohort study	Deaths in 1977– 1994, Taiwan	Death certificate	High arsenic area (median = 0.78 ppm arsenic) vs. local and national reference group	531 deaths due to DM/20067 total deaths	1.46 (1.28–1.67)	Age, sex

Table 24–1 Summary of the Studies included in the Meta-analysis of Chronic Arsenic Exposure and Type 2 Diabetes

Rahman et al., 1999 [41]	Cross- sectional	Participants in high-arsenic area, Bangladesh	Glucosuria	Arsenic in drinking water: <0.5 mg/L, 0.5–1.0 mg/L and >1.0 mg/L; time- weighted arsenic concentration: <1.0, 1.0–5.0, >5.0–10.0 and >10.0 mg-years/L	263/1332	2.1 (1.1–4.2)	Age, sex
Tseng et al., 2000 [37]	Cohort	Participants in high- arsenic area, Taiwan	OGTT	Cumulative arsenic exposure: \geq 17 vs. <17 mg/L-years	41/405	Ref 2.1 (1.1–4.2)	Age, sex, BMI
Wang et al., 2003* [25]	Cross- sectional	National Health Insurance database, Taiwan	ICD-9 250 ICD-9 A181	Arseniasis endemic area vs. non- endemic area	27543/678791		Age, sex
Zierold et al., 2004 [54]	Case-control	Participants with private wells	Self-reported	Water arsenic concentration: <2, 2–10 and >10 µg/L	67/1118	1.02 (0.49–2.15)	Age, sex, smoking status, and BMI
Coronado- Gonzalez et al., 2007 [55]	Case–control	400 participants living in areas with high arsenic in drinking water, Mexico	Fasting blood glucose	Urinary total arsenic: <35, 35–100 and >100μg/g	200/200	2.16 (1.23–3.79)	Age, sex, triglycerides, BMI, hypertension, and family history of diabetes
Ettinger et al., 2009 [40]	Prospective cohort study	532 women lived proximate to the Tar Creek Superfund site, USA	Oral glucose tolerance test (OGTT)	Hair arsenic (µg/L): Q1: 0.23–0.92 Q2: 0.93–1.39 Q3: 1.40–2.08 Q4: 2.09–24.07	22/507	1.03 (0.50–2.10) 2.21 (1.14–4.29) 2.35 (1.18–4.69)	Age, race, BMI, Medicaid use, and marital status
Navas-Acien et al., 2008 [21]	Cross- sectional	788 participants from the National Health and Nutrition Examination Survey (NHANES) 2003– 2004 survey, USA	Fasting blood glucose, self- reported physician diagnosis	Urinary arsenic (µg/L): 20th (referent) vs. 80th centile of arsenic exposure	93/695	4.26 (0.83–21.8)	Age, sex, race/ ethnicity, urinary creatinine, education, cotinine, and hypertension medication
Steinmaus et al., 2009 [56]	Cross- sectional	Participants from the National Health and Nutrition Examination Survey (NHANES) 2003–2004 survey, USA	A self-reported physician diagnosis, fasting serum glucose	Urinary arsenic (µg/L): ≤3.5, ≥18.3	77/718	0.88 (0.39–1.97)	Sex, age, ethinicity, education, BMI, serum cotinine, and current use of hypertension medications

Table 24–1 (Continued)

Author, Year	Study Design	Study Population and Location	Diabetes Diagnosis	Arsenic Exposure Categories	Cases/ Non-cases	Odds Ratio/ Relative Risk of T2D (95% Cl)	Variables Adjusted for
Chen et al., 2010 [42]	Cross- sectional	11,319 participants from the "Health Effects of Arsenic Longitudinal Study," Bangladesh	A self-reported physician diagnosis, taking insulin or hypoglycemic medication	Urinary arsenic concentration (μ g/L): 1–36 (referent), 37–66, 67–114, 115–204, ≥205; time-weighted arsenic concentration in drinking water (μ g/L): 0.1–8.0 (referent), 8.1–41.0, 41.2–91.7, 91.8–176.2– 864.0	241/11078	1.11 (0.73–1.69)	Age, sex, BMI, smoking status, educational attainment, and urinary creatinine
Kim & Lee, 2011 [22]	Cross- sectional	National Health and Nutrition Examination Survey (NHANES) participants, Korea	Fasting blood sugar, self-reported physician diagnosis, self-reported use of insulin or oral hypoglycemic medication	Urinary arsenic concentration (μg/g creatinine)	156/1521	1.31 (1.04–1.66)	Age, sex, BMI, smoking status, drinking status, educational level
Del Razo et al., 2011 [19]	Cross- sectional	Residents of Zimapán and Lagunera regions, Mexico	Fasting glucose test, 75-g oral glucose tolerance test, a self- reported physician diagnosis, taking insulin or hypoglycemic medication	Urinary arsenic concentration (ng/ mL); cumulative arsenic exposure in drinking water (ppm-years)	25/233	1.13 (1.05–1.22)	Age, sex, obesity, and hypertension
Gribble et al., 2012 [57]	Cross- sectional	3925 American Indian adults aged 45–74 years, USA	Fasting glucose test,75-g oral glucose tolerance test, HbA1C, taking insulin or oral hypoglycemic medication	Urinary arsenic concentration (µg/L): <7.9 (referent), 7.9–14.1, 14.1– 24.2, ≥24.2	1939/1986	1.14 (1.08–1.21)	Urinary creatinine, age, sex, educational level, alcohol consumption, smoking, BMI, and region
lslam et al., 2012 [10]	Cross- sectional	High and low arsenic contaminated area, Bangladesh	Fasting glucose test using glucometer, a self-reported physician diagnosis	Arsenic concentration in drinking water (μ g/L): \leq 50 (referent), >50; <22(referent), 23–32, 33–261, \geq 262	89/915	1.9 (1.1–3.5)	Age, sex, education, BMI, and family history of diabetes

Makris et al., 2012 [58]	Cross- sectional study	The small community of Mammari in the Nicosia prefecture of Cyprus	A self-reported physician diagnosis	Cumulative arsenic exposure index (mg As): 1st quintile (referent), 2nd vs. 1st, 3rd vs. 1st, 4th vs. 1st, 5th vs. 1st	21/296	1.86 (0.30– 11.59)	Age, sex, and smoking status
Jovanovic et al., 2013 [16]	Cross- sectional study	Arsenic-exposed and -unexposed population in middle Banat region, Serbia	Fasting glucose test, 75-g oral glucose tolerance test	Arsenic in drinking water (µg/L): 0−10.0 (referent); <100 (referent), ≥100	1730/771747	1.62 (1.46–1.80)	-
Kim et al., 2013 <mark>[20]</mark>	Nested case- control	American Indians, USA	75-g oral glucose tolerance test	Urinary arsenic concentration (µg/L): 0–10.0 (referent)	150/150	1.16 (0.89–1.53)	Age, sex, creatinine level, BMI, and urinary
Li et al., 2013 [43]	Cross- sectional	Arsenic-contaminated area of Inner Mongolia, China	Fasting glucose level, a self-reported physician diagnosis, taking insulin or hypoglycemic drugs	Arsenic in drinking water (µg/L): <10 (referent), 10–50, >50	42/627	1.6 (0.59–4.32)	Gender, age, alcohol consumption, cigarette smoking, BMI, and cumulative arsenic exposure
James et al., 2013 [17]	Case-cohort	San Luis Valley Diabetes Study (SLVDS), USA	Fasting glucose test, 75-g oral glucose tolerance test, a self- reported physician diagnosis	Time-weighted arsenic concentration (μ g/L-yr): \leq 4, 4.0–8, >8–20, >20	141/347	1.27 (1.01–1.59)	Ethnicity, gender, socio-economic status, first degree family history, BMI, smoking, alcohol, and physical activity
Rhee et al., 2013 [14]	Cross- sectional	National Health and Nutrition Examination Survey (NHANES) participants, Korea	Fasting glucose test, 75-g oral glucose tolerance test, taking insulin or oral hypoglycemic medication	Urinary arsenic concentration (µg/g creatinine): <70.68 (referent), 70.7–117.7, 117.7–193.4, ≥193.4	309/3293	1.56 (1.03–2.36)	Sex, age, residence area, smoking, alcohol, occupation, and serum mercury level
Pan et al., 2013 [39]	Case–control	Individuals with arsenical skin lesions and controls, Bangladesh	HbA1C	Arsenic in drinking water (μ g/L): \leq 1.7 (referent), 1.8–15.5, 15.6– 170.0, \geq 170.1; toenail arsenic (μ g/g): \leq 0.93 (referent), 0.94–2.12, 2.13–6.18, \geq 6.19	84/849	4.51 (2.01– 10.10)	Age, sex, BMI, cigarette smoking, skin lesions, arsenic in drinking water, toenail arsenic

*Not included in the meta-analysis.

Categories of Quality	Average (%) of the Maximum Category Quality Score (range)				
Scoring (maximum points value)	Cross-Sectional Studies (n = 18)	Cohort Studies (n = 5)	Case–Control Studies (n = 4)		
Selection (20 points)	15.6 (10–20)	18.2 (15–20)	15 (10–20)		
Exposure (30 points)	16.6 (5–20)	13.2 (5–20)	14 (5–20)		
Outcome (20 points)	15.3 (10–20)	14.1 (10–20)	14 (10–20)		
Confounders (20 points)	14.1 (0–17)	10.2 (7–17)	13 (7–17)		
Analysis (20 points)	8.1 (5–10)	6.9 (5-10)	8 (5–10)		
Total quality score	69.7 (40–87)	62.6 (47–87)	64 (37–87)		

Table 24–2 Average	Quality	Scores for	Studies	included i	n the	Meta-anal	vsis
							,



FIGURE 24–1 Forest plots of meta-analysis of chronic arsenic exposure and T2D.

further ($I^2 = 58.1\%$). After excluding the study by Pan et al. [39] along with these two studies, the I^2 statistic reduced to 50.57%. The between-study variability reduced further after excluding the study by Rahman et al. [38,41] along with the three studies ($I^2 = 49.73\%$). In the study by Jovanovic et al., comparisons were made between two populations exposed and unexposed

to arsenic [16]. Exposure to arsenic was measured from public water supply systems; therefore, individual arsenic exposure level as well as the duration of arsenic exposure was not available for the study participants. Information on T2D was obtained from the national registry of diabetes. The study also did not adjust for any potential confounders. Rahman et al. in their study compared the prevalence of T2D among participants from arsenic-exposed and -unexposed populations [38]. In this study, due to absence of information on individual arsenic exposure level, the indication of "keratosis" was considered as the sign of arsenic toxicity. Therefore, misclassification of exposure might have occurred. Diagnosis of T2D was not uniform for all the participants. Individuals with possible symptoms of diabetes who had two positive samples with glucometric strips were examined for hyperglycemia after overnight fasting and 2 hours after 75-mg glucose intake. This might have reduced the sensitivity of the diagnosis. Pan et al. recruited cases and controls for their study from a previous case-control study that recruited cases with arsenical skin lesions [39]. Therefore, this recruitment process limits the generalizability of the study finding. They measured HbA1C for the outcome measurement, i.e., diagnosis of T2D, which is different from the procedures used in most of the studies. Furthermore, there is an uncertainty about excluding all the prevalent self-reported cases of T2D in 2001– 2003. This may have biased the reported association between chronic arsenic exposure and T2D. In the study of Rahman et al. [38], arsenic measurement was obtained from various sources. To assess the outcome, urine samples were analyzed for glucosuria as the proxy measure for T2D using a glucometer. This might have introduced misclassification of the outcome. A number of potential confounders were also not adjusted for in the analysis [41]. Therefore, when we dropped these four studies from the meta-analysis, between-study heterogeneity was within the acceptable level ($I^2 = 49.7\%$).

24.3.4 Meta-Regression and Sensitivity Analysis

We performed a random effects meta-regression to verify the dependency of the outcome measure on the quality scores that was obtained during quality assessment. Only the quality of exposure assessment was found to be significantly associated with the outcome measure (p = 0.006) where for a 1% increase in the quality score the effect measure would decrease by 4% (95% CI: 1–7). When we restricted our meta-analysis to the studies that had a quality of at least 75%, the overall risk reduced further to 1.30 (95% CI: 1.16 to 1.46) (see Figure 24–2).

24.3.5 Influence Analysis

The influence analysis did not detect any single study influencing the pooled estimate (data not shown).

24.3.6 Publication Bias

First, we conducted the tests for funnel asymmetry using the Begg (p = 0.252) and Egger (p = 0.001) tests. Although the Egger test showed significant funnel asymmetry, it is a very sensitive test and when it disagrees with the Begg test the latter should be considered.



FIGURE 24–2 Meta-analysis of chronic arsenic exposure and T2D with studies having quality \geq 75%.

We investigated the cause of the apparent funnel asymmetry by doing a contour funnel plot. We found that most of the studies were missing on the left side of the funnel at a high significant region (Figure 24–3). This indicates that publication bias is unlikely to occur because of funnel asymmetry.

We adopted the trim and fill method to correct for publication bias and examined its effect on the pooled estimate. The method did not fill any study to deal with the apparent asymmetry and the pooled estimate was unchanged. This also demonstrated that there was no evidence of publication bias in this study.

24.3.7 Dose-Response

We included five studies in the dose-response meta-analysis [10,16,18,38,39,42,43]. For each 100 unit increase in arsenic concentration, the effect measure increases by 23%, although this dose-response was not significant (95% CI: -29, 111). There was also no statistical evidence of a non-linear dose-response trend (p = 0.054) (not shown in the analysis). We excluded the study of Ettinger et al. [40] as they measured arsenic in blood and in hair samples of a subset of the study participants, which is different from other studies we included in this dose-response meta-analysis.

24.4 Discussion

In this meta-analysis of 28 studies, we observed an overall association between chronic arsenic exposure and T2D. However, the association varies according to the study designs. The pooled



FIGURE 24-3 Funnel plot of random effects meta-analysis of arsenic exposure and T2D.

OR estimate was 1.58 (95% CI: 1.35–1.84), the pooled RR was 1.28 (95% CI: 1.03–1.59), and the overall pooled estimate of risk was 1.50 (95% CI: 1.33, 1.69).

It is essential to investigate the presence of between-study heterogeneity in a meta-analysis [44]. In this meta-analysis, between-study heterogeneity was very high. This observed heterogeneity may be attributable to variations among study participants' characteristics such as age, sex, body mass index, family history of T2D, and other co-morbidities. However, excluding the study by Wang et al. [25] reduces the between-study heterogeneity substantially. Between-study variability reduced further when we restricted the meta-analysis within total study quality \geq 75% (I² = 59.5%).

The meta-regression revealed a significant inverse association with exposure assessment, which is the most important characteristic of an observational study. Thus, with improved exposure measurement we can expect diminishing association. This evidence is also supported by the subgroup analysis where weaker association was found in the meta-analysis with overall better study quality. Compared to the previous meta-analysis [45], this study did not get a significant dose-response association after including more studies. Moreover, when we restricted our dose-response analysis to the same studies by Wang et al. [45], we could not reproduce the significant dose-response they reported. Thus, the dose-response association is also not conclusive. This may be due to the differences in the way they handled the exposure categories in the regression model. We used the median exposure level as the independent variable in the model but they did not mention how they did it. Moreover, in the meta-analysis by Wang and co-workers [45] they included the large study by Wang and co-workers [25] with a very tight confidence interval (OR = 2.69, 95% CI: 2.65 to 2.73), which forced the pooled estimate away

from the null. We did not include this study in our analysis because we did not get that effect measure reported in the original study.

Adverse effects of arsenic on T2D have been demonstrated in previous animal model experimentation. These experiments suggest that arsenic impairs pancreatic β -cell function, thus affecting insulin synthesis and secretion. The molecular mechanism for this effect may be through adverse effects on insulin signal transduction and the inhibition of gene transcription factors [46–48]. Furthermore, a number of other non-specific mechanisms such as inflammation, apoptosis or oxidative stress may also increase the risk of T2D.

The main strength of our study is that we did not get any evidence of publication bias in the meta-analysis, but we did get inclusion of all the studies fulfilling eligibility criteria, comprehensive assessment of quality, and the use of the quality score in a meta-regression and subgroup analysis to assess the effect of quality scores on the study findings.

Nevertheless, the meta-analysis also has a few limitations. Measurement of arsenic exposure was not uniform across the studies; therefore, the observed association between chronic arsenic exposure and T2D may be underestimated. Outcome measurement also varied among the studies, in particular there was a concern about diagnosis of T2D using a glucometer. The validity of the glucometer had not been assessed earlier. Finally, observed findings could be explained by a number of unknown confounders, which could not be adjusted for, and this may have over- or underestimated the risk estimates.

In conclusion, this meta-analysis suggests that chronic arsenic exposure is likely to increase the risk of T2D. However, this finding is limited by high between-study heterogeneity. Therefore, the hypothesized association needs further validation through well-conducted, large prospective cohort studies with better exposure assessment.

References

- Argos M, Kalra T, Rathouz PJ, Chen Y, Pierce B, Parvez F, et al. Arsenic exposure from drinking water, and all-cause and chronic-disease mortalities in Bangladesh (HEALS): a prospective cohort study. Lancet 2010;376(9737):252–8.
- [2] Kim MJ, Nriagu J, Haack S. Arsenic species and chemistry in groundwater of southeast Michigan. Environ Pollut 2002;120(2):379–90.
- [3] Hughes MF, Beck BD, Chen Y, Lewis AS, Thomas DJ. Arsenic exposure and toxicology: a historical perspective. Toxicol Sci 2011;123(2):305–32.
- [4] Thomas X, Troncy J. Arsenic: a beneficial therapeutic poison—a historical overview. Adler Mus Bull 2009;35(1):3–13.
- [5] Antman KH. Introduction: the history of arsenic trioxide in cancer therapy. Oncologist 2001;6(Suppl. 2):1-2.
- [6] Waxman S, Anderson KC. History of the development of arsenic derivatives in cancer therapy. Oncologist 2001;6(Suppl. 2):3–10.
- [7] Hutchinson J. Arsenic and cancer. Br Med J 1887;2:1280–1.
- [8] Islam MR, Khan I, Attia J, Hassan SM, McEvoy M, D'Este C, et al. Association between hypertension and chronic arsenic exposure in drinking water: a cross-sectional study in Bangladesh. Int J Environ Res Public Health 2012;9(12):4522–36.

- [9] Martinez VD, Vucic EA, Becker-Santos DD, Gil L, Lam WL. Arsenic exposure and the induction of human cancers. J Toxicol 2011;2011:431287.
- [10] Islam MR, Khan I, Hassan SMN, McEvoy M, D'Este C, Attia J, et al. Association between type 2 diabetes and chronic arsenic exposure in drinking water: a cross sectional study in Bangladesh. Environ Health 2012;11:38.
- [11] Milton AH, Rahman M. Respiratory effects and arsenic contaminated well water in Bangladesh. Int J Environ Health Res 2002;12(2):175–9.
- [12] Smith AH, Lingas EO, Rahman M. Contamination of drinking-water by arsenic in Bangladesh: a public health emergency. Bull World Health Organ 2000;78(9):1093–103.
- [13] Tseng CH. The potential biological mechanisms of arsenic-induced diabetes mellitus. Toxicol Appl Pharmacol 2004;197(2):67–83.
- [14] Rhee SY, Hwang YC, Woo JT, Chin SO, Chon S, Kim YS. Arsenic exposure and prevalence of diabetes mellitus in Korean adults. J Korean Med Sci 2013;28(6):861–8.
- [15] Yoshida T, Yamauchi H, Fan Sun G. Chronic health effects in people exposed to arsenic via the drinking water: dose-response relationships in review. Toxicol Appl Pharmacol 2004;198(3):243–52.
- [16] Jovanovic D, Rasic-Milutinovic Z, Paunovic K, Jakovljevic B, Plavsic S, Milosevic J. Low levels of arsenic in drinking water and type 2 diabetes in Middle Banat region, Serbia. Int J Hygiene Environ Health 2013;216(1):50–5.
- [17] James KA, Marshall JA, Hokanson JE, Meliker JR, Zerbe GO, Byers TE. A case-cohort study examining lifetime exposure to inorganic arsenic in drinking water and diabetes mellitus. Environ Res 2013;123:33–8.
- [18] Lai MS, Hsueh YM, Chen CJ, Shyu MP, Chen SY, Kuo TL, et al. Ingested inorganic arsenic and prevalence of diabetes mellitus. Am J Epidemiol 1994;139(5):484–92.
- [19] Del Razo LM, García-Vargas GG, Valenzuela OL, Hernández Castellanos E, Sánchez-Peña LC, Currier JM, et al. Exposure to arsenic in drinking water is associated with increased prevalence of diabetes: a cross-sectional study in the Zimapán and Lagunera regions in Mexico. Environ Health 2011;10:73.
- [20] Kim NH, Mason CC, Nelson RG, Afton SE, Essader AS, Medlin JE, et al. Arsenic exposure and incidence of type 2 diabetes in Southwestern American Indians. Am J Epidemiol 2013;177(9):962–9.
- [21] Navas-Acien A, Silbergeld EK, Pastor-Barriuso R, Guallar E. Arsenic exposure and prevalence of type 2 diabetes in US adults. JAMA 2008;300(7):814–22.
- [22] Kim Y, Lee B-K. Association between urinary arsenic and diabetes mellitus in the Korean general population according to KNHANES 2008. Sci Total Environ 2011;409(19):4054–62.
- [23] Chen CJ, Wang SL, Chiou JM, Tseng CH, Chiou HY, Hsueh YM, et al. Arsenic and diabetes and hypertension in human populations: a review. Toxicol Appl Pharmacol 2007;222(3):298–304.
- [24] Navas-Acien A, Silbergeld EK, Streeter RA, Clark JM, Burke TA, Guallar E. Arsenic exposure and type 2 diabetes: a systematic review of the experimental and epidemiologic evidence. Environ Health Perspect 2006;114(5):641–8.
- [25] Wang SL, Chiou JM, Chen CJ, Tseng CH, Chou WL, Wang CC, et al. Prevalence of non-insulin-dependent diabetes mellitus and related vascular diseases in southwestern arseniasis-endemic and nonendemic areas in Taiwan. Environ Health Perspect 2003;111(2):155–9.
- [26] Abir T, Rahman B, D' Este C, Farooq A, Milton AH. The association between chronic arsenic exposure and hypertension: a meta-analysis. J Toxicol 2012;2012:198793.
- [27] DerSimonian R, Laird N. Meta-analysis in clinical trials. Control Clin Trials 1986;7(3):177-88.
- [28] Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. Br Med J 2003;327(7414):557-60.
- [29] Patsopoulos NA, Evangelou E, Ioannidis JP. Sensitivity of between-study heterogeneity in meta-analysis: proposed metrics and empirical evaluation. Int J Epidemiol 2008;37(5):1148–57.

- [30] Egger M, Smith G, Altman D, & Ebook Library. Systematic reviews in health care meta-analysis in context (online resource (508 pp.)). Retrieved from <<u>http://oxford.eblib.com/patron/FullRecord.</u> aspx?p=242481>; 2008.
- [31] Begg CB, Mazumdar M. Operating characteristics of a rank correlation test for publication bias. Biometrics 1994;50:1088–101.
- [32] Egger M, Davey SG, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test. Br Med J 1997;315:629–34.
- [33] Peters JL, Sutton AJ, Jones DR, Abrams KR, Rushton L. Contour-enhanced meta-analysis funnel plots help distinguish publication bias from other causes of asymmetry. J Clin Epidemiol 2008;61(10):991–6.
- [34] Sterne JA, Sutton AJ, Ioannidis JP, Terrin N, Jones DR, Lau J, et al. Recommendations for examining and interpreting funnel plot asymmetry in meta-analyses of randomised controlled trials. BMJ 2011;343:d4002.
- [35] Duval S, Tweedie R. A nonparametric "trim and fill" method of accounting for publication bias in metaanalysis. J Am Stat Assoc 2000;95(449):89–98.
- [36] Orsini N, Li R, Wolk A, Khudyakov P, Spiegelman D. Meta-analysis for linear and nonlinear dose-response relations: examples, an evaluation of approximations, and software. Am J Epidemiol 2012;175(1):66–73.
- [37] Tseng CH, Tai TY, Chong CK, Tseng CP, Lai MS, Lin BJ, et al. Long-term arsenic exposure and incidence of non-insulin-dependent diabetes mellitus: a cohort study in arseniasis-hyperendemic villages in Taiwan. Environ Health Perspect 2000;108(9):847–51.
- [38] Rahman M, Tondel M, Ahmad SA, Axelson O. Diabetes mellitus associated with arsenic exposure in Bangladesh. Am J Epidemiol 1998;148(2):198–203.
- [39] Pan WC, Seow WJ, Kile ML, Hoffman EB, Quamruzzaman Q, Rahman M, et al. Association of low to moderate levels of arsenic exposure with risk of type 2 diabetes in Bangladesh. Am J Epidemiol 2013;178(10):1563–70.
- [40] Ettinger AS, Zota AR, Amarasiriwardena CJ, Hopkins MR, Schwartz J, Hu H, et al. Maternal arsenic exposure and impaired glucose tolerance during pregnancy. Environ Health Perspect 2009;117(7):1059–64.
- [41] Rahman M, Tondel M, Chowdhury IA, Axelson O. Relations between exposure to arsenic, skin lesions, and glucosuria. Occup Environ Med 1999;56(4):277-81.
- [42] Chen Y, Ahsan H, Slavkovich V, Peltier GL, Gluskin RT, Parvez F, et al. No association between arsenic exposure from drinking water and diabetes mellitus: a cross-sectional study in Bangladesh. Environ Health Perspect 2010;118(9):1299–305.
- [43] Li X, Li B, Xi S, Zheng Q, Lv X, Sun G. Prolonged environmental exposure of arsenic through drinking water on the risk of hypertension and type 2 diabetes. Environ Sci Pollut Res Int 2013;20(11):8151–61.
- [44] Munafo MR, Flint J. Meta-analysis of genetic association studies. Trends Genet 2004;20(9):439-44.
- [45] Wang W, Xie Z, Lin Y, Zhang D. Association of inorganic arsenic exposure with type 2 diabetes mellitus: a meta-analysis. J Epidemiol Community Health 2014;68(2):176–84.
- [46] Elrick LJ, Docherty K. Phosphorylation-dependent nucleocytoplasmic shuttling of pancreatic duodenal homeobox-1. Diabetes 2001;50(10):2244–52.
- [47] Macfarlane WM, McKinnon CM, Felton-Edkins ZA, Cragg H, James RF, Docherty K. Glucose stimulates translocation of the homeodomain transcription factor PDX1 from the cytoplasm to the nucleus in pancreatic beta-cells. J Biol Chem 1999;274(2):1011–16.
- [48] Macfarlane WM, Smith SB, James RF, Clifton AD, Doza YN, Cohen P, et al. The p38/reactivating kinase mitogen-activated protein kinase cascade mediates the activation of the transcription factor insulin upstream factor 1 and insulin gene transcription by high glucose in pancreatic beta-cells. J Biol Chem 1997;272(33):20936–20944.
- [49] Lagerkvist BJ, Zetterlund B. Assessment of exposure to arsenic among smelter workers: a five-year followup. Am J Ind Med 1994;25(4):477–88.

- [50] Rahman M, Axelson O. Diabetes mellitus and arsenic exposure: a second look at case-control data from a Swedish copper smelter. Occup Environ Med 1995;52(11):773–4.
- [51] Rahman M, Wingren G, Axelson O. Diabetes mellitus among Swedish art glass workers-an effect of arsenic exposure? Scand J Work Environ Health 1996;22(2):146–9.
- [52] Jensen GE, Hansen ML. Occupational arsenic exposure and glycosylated haemoglobin. Analyst 1998;123(1):77–80.
- [53] Tsai SM, Wang TN, Ko YC. Mortality for certain diseases in areas with high levels of arsenic in drinking water. Arch Environ Health 1999;54(3):186–93.
- [54] Zierold KM, Knobeloch L, Anderson H. Prevalence of chronic diseases in adults exposed to arseniccontaminated drinking water. Am J Public Health 2004;94(11):1936–7.
- [55] Coronado-Gonzalez JA, Del Razo LM, Garcia-Vargas G, Sanmiguel-Salazar F, Escobedo-de la Pena J. Inorganic arsenic exposure and type 2 diabetes mellitus in Mexico. Environ Res 2007;104(3):383–9.
- [56] Steinmaus C, Yuan Y, Liaw J, Smith AH. Low-level population exposure to inorganic arsenic in the United States and diabetes mellitus: a reanalysis. Epidemiology 2009;20(6):807–15.
- [57] Gribble MO, Howard BV, Umans JG, Shara NM, Francesconi KA, Goessler W, et al. Arsenic exposure, diabetes prevalence, and diabetes control in the Strong Heart Study. Am J Epidemiol 2012;176(10):865–74.
- [58] Makris KC, Christophi CA, Paisi M, Ettinger AS. A preliminary assessment of low level arsenic exposure and diabetes mellitus in Cyprus. BMC Public Health 2012;12:334.

This page intentionally left blank

25

Arsenic Biosensors: Challenges and Opportunities for High-Throughput Detection

Minni Singh¹ and Manel del Valle²

¹BIOSENSOR DEVELOPMENT GROUP, DEPARTMENT OF BIOTECHNOLOGY, PUNJABI UNIVERSITY, PATIALA, INDIA, ²SENSORS AND BIOSENSORS GROUP, DEPARTMENT OF ANALYTICAL CHEMISTRY, UNIVERSITAT AUTONOMA DE BARCELONA, BARCELONA, SPAIN

CHAPTER OUTLINE

25.1	Arsenic: The Toxic Metalloid	575
25.2	Arsenic Biosensors	576
25.3	Nanosensor Platforms—Towards High-Throughput Detection	581
25.4	Conclusions and Future Directions	585
Refe	rences	586

25.1 Arsenic: The Toxic Metalloid

Arsenic (As) is a toxic metalloid distributed in nature and the incidence of its poisoning among human populations from various parts of the world is well documented [1]. These instances were reported to occur mainly through contaminated water and food sources, which have been attributed to activity that pertains to use of arsenical pesticides, digging of sedimentrich wells, and leaching near thermal plants, which significantly adds to arsenic exposure to the environment posing a serious threat to life [2,3]. It is established that arsenic is a potent human carcinogen, prolonged exposure to which leads to a wide range of physiological implications affecting skin, gastrointestinal, hepatic, neurological, and respiratory organ systems mediated through oxidative stress and genetic manifestations [4,5]. The highest prevalence of arsenic's inorganic forms, the trivalent As(III) and the pentavalent As(V), are predominant in occurrence, the former being more toxic [8]. The World Health Organization recommended the maximum permissible concentration of arsenic in drinking water to be $10 \,\mu$ g/L, which corresponds to $0.1 \,\mu$ M As [9]. Consequently, it is of utmost importance that water, food, and soil, particularly in suspected regions, be monitored regularly for any possible exposure for which



FIGURE 25-1 Classification of arsenic detection methods.

efficient, reliable, and high-throughput arsenic detection methods are desired. The methods for arsenic detection are broadly classified in Figure 25–1.

Established laboratory methods are the spectrometric methods. These techniques, though highly accurate, are costly, time intensive, need acidic pretreatment of samples before analysis, and are essentially laboratory bound. However, these methods are reference methods and are used for validation of newer techniques. Most arsenic detection kits developed for on-field application and used for drinking water and ground water analysis are chemical kits based on the Gutzeit method. Some research groups have evaluated the efficacy of these kits [10-12]. Though some commercial kits furnish satisfactory responses, it was noticed that most of them exhibit a high percentage of false negatives and false positives, which is attributed to interference by other metals and ions found in real samples; the reliability of these kits is therefore questioned. In addition, the chemical methods entail generation of the highly toxic arsine gas, which raises safety concerns. More recently, electrochemical methods of arsenic detection have shown promise in terms of safety and portability in measurements [13,14]. Anodic and cathodic stripping voltammetry rely on specific working electrodes. There are concerns about measurements due to peak currents overlapping with those of interfering species, which is a keen research focus for many groups, and nanoparticles may seem to address those issues [15]. High-throughput on-site detection with these methods may still be a challenge since they need either gold or mercury working electrodes, which raise cost and disposal concerns. Alongside the above-mentioned methods and challenges, biological assays have emerged as promising alternative methods for arsenic detection. In particular, biosensors with whole cells and oligonucleotides as biorecognition elements enable ultrasensitive detection owing to biological origins that respond to environmental stress conditions by evolving inherent mechanisms of safety, which have been carefully developed by bioanalytical chemists into arsenic diagnostic assays, the features of which are discussed below.

25.2 Arsenic Biosensors

In recent years, biosensors for arsenic have been reported based on physiological changes such as pH in response to the presence of the analyte, a chromogenic induction, or inhibitory



FIGURE 25-2 The ars operon.

action of the metal ion on the enzyme acetylcholinesterase [16–19]. However, it was noticed that these biosensors were lacking in selectivity and had very high detection limits, which clearly jeopardized their applicability. This led researchers to explore the possibility of engineering microorganisms such that they elicit a specific response in the presence of arsenic, the basis of which lay in the ability of some of the microorganisms to develop a natural self-defense mechanism for survival in metal-laden toxic environments, and this led to a plethora of interesting works on recombinant arsenic biosensors.

25.2.1 Evolution of a Natural Self-Defense Mechanism—the ars Operon

The microorganisms in their response to toxicity posed by arsenic have evolved a natural selfdefense mechanism in which the metalloid is pumped out of the cell by an efflux pump that is genetically regulated by a complement of five genes that comprise *ars* operon as shown in Figure 25–2.

The three structural genes *arsA*, *arsB*, and *arsC* code for membrane bound ATPase, a transmembrane protein, and arsenate reductase, respectively, to form an efflux pump responsible for removing arsenite from the cell, deriving energy for efflux from ATP hydrolysis. The cells that are incapable of effluxing arsenate depend upon *arsC* encoded arsenate reductase to generate arsenite, which induces *arsAB* thereby extruding arsenite out of the cell. *arsR* and *arsD* are the up-regulatory genes responsible for repression and controlled expression of structural genes [20]. Figure 25–3 depicts the arsenite efflux pump in cells.

In the presence of arsenite the *arsR* encoded repressor protein undergoes a conformational change leading to its dissociation from operator/promoter, thereby allowing the commencement of transcription to yield the defense by switching on expression. Many workers have utilized this phenomenon to quantify arsenite by fusion of *arsR* with a reporter gene to generate a response that is dependent on arsenite concentration, which is established as a highly specific recombinant biosensing system [21] as illustrated in Figure 25–4. Three of the most used arsenite concentration-dependent recombinant plasmid-borne hosts are those of *Escherichia coli, Bacillus subtilis,* and *Rhodopseudomonas palustris* [22–25]. For detailed information on the performance of these biosensors, readers are referred to the work of Diesel et al. [26].

25.2.2 Recombinant *Escherichia coli*—the Favored Biorecognition Element

Though many bacterial genera have been reported for use in arsenic biosensing systems, it is observed that most of these did not furnish an appreciable sensitivity of arsenic detection,



FIGURE 25–3 The arsenite efflux pump.



FIGURE 25-4 The recombinant biosensing phenomenon.

the lowest detectable concentration being $0.3 \,\mu$ M. Due to arsenic toxicity and the WHO established safe limit of $0.1 \,\mu$ M, it is incumbent for biosensors to exhibit ultrasensitive responses, a feature that justifies the objectives of biosensors over conventional techniques of detection, which bear certain limitations as mentioned earlier. Due to its evolved mechanism to survive under extreme chemical stress environments and the ease in manipulating its genetic makeup, *E. coli* has been a favored host for recombinant constructs and has proven this ability in the realm of arsenic detection [26]. Much like other bacterial strains, in some of the initial works reported with *E. coli* for arsenic the limit of detection was not sensitive enough. However, with continued re-engineering and with understandings of newer reporter genes that furnished autofluorescent and reagentless expressions, a significant improvement in the sensitivity of detection, as low as subnanomolar concentration, was observed. In addition, the coupling of electrochemical and optical transducers for sensitive quantification of genetic expression further refined the analytics in recombinant biosensors by the sending of read-out signals that paved the way for on-the-spot field detection of this toxic metalloid.

Plasmid	Reporter Gene	Induction Time	Output Signal	Limit of Detection	Reference
pJAMA-arsR-ABS	luxAB	1 h	Bioluminescence	0.05 µM	[21]
pPR-arsR-ABS	gfp	1–5 h	Fluorescence	0.12 μM	[21]
pMV-arsR-ABS	lacZ	30 min	Colorimetric	qualitative	[21]
pRLUX	luxAB	3h	Bioluminescence	1.0 fM	[27]
pBGD23	lacZ	30 min	Chemiluminescence	1.0 fM	[28]
pBGD23	lacZ	17 h	Electrochemical	0.1 µM	[28]
pSD10	gfp	30 min	Fluorescence	8.0µM	[29]
pPR-arsR-ABS	egfp	1 h	Laser-induced fluorescence	0.1μM	[30]
pJAMA-arsR-ABS	luxAB	2 h	Bioluminescence	6 ng/g dry wt. (rice)	[31]
arsR-lacZ strains 1580	lacz	8h	Colorimetric	<0.08 µM	[32]
arsR-ccp strains 1971, 1981	сср	4h	Colorimetric	$<0.01\mu M$	[32]

Table 25–1 Response Characteristics of Recombinant E. coli Biosensors for Arsenic

Many workers have investigated recombinant E. coli for arsenic detection, of which two research groups have made significant contributions: Daunert and van der Meer. In one of their pioneering works Daunert's group engineered a plasmid pRLUX by fusion of arsD with reporter genes *luxA* and *luxB*. The expression was regulated by *arsR* upstream of *arsD*. *arsR* responds to arsenic in a mechanism described previously to yield bioluminescence in transformed cells of E. coli. A significant feature of this work is the remarkable sensitivity of detection at subattomolar concentrations [27]. In another study, a different reporter gene *lacZ* was fused with arsD to yield a plasmid pBGD23 to look for any improvement in sensitivity characteristics. The transformed *E. coli* that furnished a chemiluminescent signal exhibited similar sensitivity of detection as in the previous work, though it conferred a higher selectivity of response as investigated with a number of metal ions. Even though the biosensor had a shorter induction time, reporting with *lacZ* bears a limitation since the need to be lysed before a response can be generated, which limits its real-time applicability and also the effect of the cationic peptide used as a lysing agent, cannot be ignored [28]. With the same recombinant, the group for the first time reported electrochemical transduction in which *lacZ* expression was quantified by generation of an electroactive product that was measured at a peak oxidation current with a gold working electrode at a potential of +0.23 V vs. Ag/AgCl. A limit of 10^{-7} M arsenic detection was observed; however, the unusually high induction time followed by a long incubation prompts alternative modes of detection. In such an advancement the group in their subsequent work reported a recombinant with a reporter gene coding for a mutant green fluorescent protein GFPuy, a reagentless autofluorescent reporter fused with arsR to get pSD10 plasmid construct. E. coli transformed with pSD10 generated autofluorescence upon induction with arsenite and the limit of detection was observed to be 8×10^{-6} M. A significant feature of this work was the adaptation of the biosensing system onto a microfluidic platform with the purpose of miniaturization, which would afford portability and hence wider applicability to the sensing device [29]. Table 25-1 summarizes the response characteristics of some of the significant recombinant *E. coli*-based biosensors developed for arsenic detection.

Van der Meer's investigations reported a new dimension in improved response characteristics of recombinant *E. coli* biosensors for arsenic by diminishing the expression of ArsR protein, which would then enhance the relative expression of reporter proteins thereby yielding a higher sensitivity of detection. To carry this out, a recombinant was engineered by inserting a 72 bp DNA fragment downstream of arsR as a secondary binding site for ArsR protein, which then bound to both sites. This was shown to exhibit higher titers of reporter proteins by reducing background expression of *arsR*. Three plasmid constructs pJAMA-arsR-ABS, pPR-arsR-ABS, and pMV-arsR-ABS with *luxAB*, gfp, and *lacZ*, respectively, which furnished bioluminescence, autofluorescence, and colorimetric responses, were investigated for their performance. The bioluminescence and fluorescence biosensors exhibited a higher sensitivity of detection of 0.05 and $0.12 \mu M$ arsenic. The colorimetric biosensor was developed into a qualitative paper strip test able to convey the safety/toxicity of samples. A significant reduction in induction times was observed from the previously reported 3 h to 1 h in this work, which is attributed to the secondary binding site introduced in the genetic construct, though a higher induction time would further lower the detection limit. As expected, the adsorption of cells onto a paper strip did leach out the same into the solution and therefore cell confinement onto biocompatible matrices was required, though the method proved to be useful as a qualitative and disposable diagnostic for on-site analysis. In gfp harboring cells, autofluorescence was generated after 1 h induction, though a higher time was suggested based on cell counts. The read-out signal was measured by epifluorescence microscopy, which allows the monitoring of individual cells [21,33]. In an advanced detection method laser-induced fluorescence confocal spectroscopy was used to measure arsenite-induced expression of the enhanced green fluorescent protein egfp. The technique furnished a similar ultrasensitive response of 1.0 mM arsenic detection increase with single molecule detection [30].

Further, the pJAMA-arsR-based biosensor was validated for application of real-time groundwater samples by negating the effect of Fe(II) and Fe(III) on arsenic detection since iron in aqueous environments forms colloidal hydroxides that strongly adsorb arsenic up to 80-90%, which decreases the bioavailability of the metalloid for detection by the biosensor leading to generation of a false response. To reduce the effect of presence of Fe(II) and Fe(III), acidification and chelation strategies were opted for and it was observed that use of EDTA had a detrimental effect on the cells due to chelation of ions in the cell walls causing apoptosis. Mineral acids were therefore used to maintain a highly acidic environment in which Fe(II) and Fe(III) would remain in solution, but a challenge was the survival of cells under pH stress and so a strategy for sequential increase in pH was optimized in which, upon acidification and instant incubation with cells, neutralization was achieved with pyrophosphate, thus maintaining the viability of the cell suspension and also keeping arsenic bioavailable. The response of the biosensor was very promising with negligible false responses, and it was successfully used for on-field detection of arsenic and proposed for high-throughput detection [34]. A subsequent study investigated the interference in biosensor response due to phosphate and silicate, if any. It was reported that in the presence of arsenite the biosensor responses could be normalized and false positives and negatives could be countered by competitive binding and desequestration, thereby sustaining the biosensor response [35]. Later, the same recombinant was used for detection of arsenic in rice where it showed a limit of detection as low as 6 ng/g dry wt. of rice [31]. Since the intensity of reporter protein expression is clearly dependent on induction time and cell density as well as on the physiological state of the recombinants, particularly for colorimetric responses, a variation in reproducibility was a concern. To address this, different strains of one recombinant were engineered and used simultaneously comprising a cell line that responded to a range of arsenic concentrations. For *lacZ* reporters the intervening sequence between *arsR* and *lacZ* was modified to modulate expression to get four strains 1580, 1595, 2245, and 2066, which yielded colorimetric responses upon arsenic exposure in concentration ranges <0.001, 0.001–0.1, 0.1–0.67, and >0.67 μ M arsenic, respectively. Likewise, another cell line with variation in cytochrome c peroxidase (ccp) expression was engineered with strains 1971, 1981, 1982, 2031, 2332, and 2340 that too responded to arsenic over specific ranges with a lowest detectable concentration of <0.01 μ M. This work proposed an internal calibration of the biosensing complement to eliminate limitations mentioned earlier and suggested easier and more practical application for paper strip assays [32].

To accomplish expression of reporter genes, an initial induction by arsenite, the time of which may be up to a few hours as seen in Table 25-1, is necessary, followed by harvesting of cells and treating with an appropriate substrate, in most cases to elicit a measurable response. In the evolving concepts of biosensing such laboratory bound sensing, which is dependent on the shelf-life of microorganisms and their regeneration behavior, limits their applicability. More importantly, the use of genetically modified organisms for on-field detection entails risk assessment and regulatory approvals and any untoward mutation caused by environmental stress, which may lead to physiological implications, cannot be ignored. In addition, a rapid quantitative response would entail proximate coupling of a transducer with the biorecognition element, which was not observed in the above reported works; therefore, a high-throughput detection of arsenic remains unjustified. For this, three necessary inclusions are envisaged: (1) a biorecognition element that negates pretreatment to furnish a response, (2) a confinement matrix for the biorecognition element that compounds the elicited response, and (3) proximate coupling of the biorecognition element with a transducer that affords ultrasensitive read-out. Keeping this in view, it is appropriate to bring attention to nanosensor platforms. Conducting polymers, sol-gels, and metal nanoparticle-impregnated graphenes have been a recent choice for bioanalytical chemists. These, along with new age biorecognition elements, are promising future directions for high-throughput detection of arsenic. One class of such biorecognition elements that has shown tremendous features and has just begun to be used for arsenic detection is *aptamers*, and one class of matrix that has remarkable properties of sensitivity is graphenes.

25.3 Nanosensor Platforms—Towards High-Throughput Detection

A class of biosensors, the immunosensors—based on antigen-antibody complex formation—are by far the most specific biosensors reported for protein detection. The unbeatable specificity of

antibodies makes them suitable candidates for biomolecular recognition elements. However, there are some limitations of antibodies that need consideration, such as liability to degrade due to thermal instability, *in vivo* production, labeling that may lead to a loss of affinity, and being raised only against immunogenic molecules. A class of biomolecular recognition elements able to completely overcome these limitations and affording remarkable specificity was reported in 1990 [36,37], the potential of which is being explored in analytics. These molecules, called *aptamers*, are synthetic ssDNA/RNA sequences selected from oligonucleotide libraries by an *in vitro* selection process called SELEX (Systematic Evolution of Ligands by Exponential enrichment) [38], a well-established procedure that selects aptamers employing a combinatorial approach based on their affinity for a particular target. These, along with peptide aptamers, i.e., small peptides with recognition properties and a recent variant of these affinity biomolecules, have significant advantages [39]:

- · very high specificities and affinities for targets
- besides proteins and whole cells, ability to synthesize aptamers for non-immunogenic targets like drugs, metals, and inorganic compounds
- in vitro synthesis and small in size (usually 15-80 mers)
- terminal tagging without loss of affinity
- long-term stability.

In recent years there have been reports of the use of aptamers, particularly for pathogen detection, antibiotic and toxin testing, clinical diagnostics, and more recently for arsenic detection. It is noteworthy that the sensitivities of detection reported in these works is promising, from nM down to pM concentrations [40,41]. In 2009, the first DNA aptamer for arsenic was reported. Of the nine aptamers synthesized by SELEX, Ars-3, which is an oligonucleotide with a sequence 5'-GGTAATACGACTCACTATAGGGAGATACCAGCTTATTCAATTT TACAGAACAACCAACGTCGCTCCGGGTACTTCTTCATCGAGATAGTAAGTGCAATCT-3', had the highest binding affinity for As(III). This aptamer was used for arsenic removal from groundwater samples as a treatment technology [42]. More recently, aptamer Ars-3 was used for sensitive detection of As(III) when used in a combination with water soluble cationic polymer polydiallyldimethyl ammonium PDDA and gold nanoparticles, AuNPs. Incubation of Ars-3 and PDDA led to electrostatic coupling in both, thus starving the AuNPs leaving them in a dispersed phase. However, in the presence of As(III) the aptamer/As(III) complex forms instantaneously owing to remarkable binding affinity of the aptamer for arsenic making PDDA available for AuNP aggregation, which was measured spectrophotometrically. The biosensor showed a detection limit of $0.07 \,\mu\text{M}$ arsenic. Also, the sensor showed a high selectivity for As(III) against other metal ions, a minor interference caused only by As(V) [43]. In a similar work cationic surfactants were investigated for their efficiency in aggregating AuNPs and also their preferential binding for Ars-3 aptamer. It was observed that cationic surfactant cetyl trimethylammonium bromide (CTAB) had the highest efficiency, attributed to its alkyl chain length. A remarkable feature of this work was a rapid response of less than 2min and a significantly low detection limit of $0.008 \,\mu$ M As(III) [44]. In yet another nanoparticle-based assay using the same aptamer Ars-3, arsenic was detected by resonance Rayleigh scattering (RRS), the intensity of which was dependent on nanoassembly of the aptamer assisted by a cationic dye crystal violet to form differently sized nanoparticles, which upon interaction with arsenic changed the size of the nanoparticles due to As(III)-Ars binding to elicit a change in RRS intensity. The limit of detection achieved in this assay of 0.001μ M As(III) along with a near instantaneous response is an outstanding achievement in arsenic biosensors [45]. The precision of these tools in groundwater samples is yet to be established; however, the possibility that they offer rapid and highly selective ultrasensitive detection of arsenic is ascertained.

For all these reasons, aptamers are already a clear choice for biorecognition elements. In the wake of the above and with the focus on arsenic, two kinds of biosensors may be developed using aptamers: one in which the aptamer is synthesized for arsenic, and one in which the aptamer is synthesized for the reporter protein. Since the reporter protein expression entails long induction times, to justify the remarkable capability of aptamers the first option would be the obvious choice. Figure 25–5 shows a schematic that proposes a futuristic approach to arsenic biosensing.

Graphene has attracted a keen interest among analysts due to its tremendous charge carrier ability, large surface area augmenting functionalization, and unique mechanical properties [46]. This exceptional conductivity of graphene gives it the potential to be explored in electrochemical biosensing applications. Coupled with this, this carbon material has the ability to impart extremely high sensitivity of detection due to its 2D structure, wherein even a single molecule adsorbed on its surface can cause resistance to charge transfer. The use of graphene in biosensors is just beginning and it is reported that graphene confers much higher sensitivity than its nanotube counterparts [47,48], a property of immense value in clinical diagnostics. A technical challenge posed with subsequent real-time application of these biosensors would be the occurrence of interfering ions in *in vivo* environments, which would generate overlapping profiles. To resolve the responses, data processing tools like artificial neural networks (ANNs) to perform pattern recognition and quantification of specific analytes among different ions can be used. The coupling of biosensor arrays and data processing tools is known as electronic tongues. The chemometric tools serve to build multivariate response models in such a way that every considered metal ion will generate a signal that is independent of others, thus accounting for cross-selectivity, leading to a consolidated response. These tools have been used in analytics and have made significant contributions to generate electronic tongues [49-52]. Considering the binding affinity of selected aptamers and accounting for false positives for various ions, these chemometric tools are expected to generate data that would identify arsenic with a high degree of accuracy and also differentiate arsenite and arsenate.

The suggested approach for arsenic detection therefore advocates electrochemical transduction using aptamer graphene nanoensembles for ultrasensitive and high-throughput detection. It may be added here that optical transducers may be equally competent in achieving the same, for which Forster resonance energy transfer (FRET) may be used by capping the aptamer with a fluorophore, which upon conjugation with arsenic elicits a conformational change leading to FRET since graphene is an efficient quencher too. Further, these devices have the capability of miniaturization and can therefore be developed into handheld devices, with satisfies the objectives of biosensor development studies.


FIGURE 25–5 A schematic depicting a future approach in arsenic biosensing.

25.4 Conclusions and Future Directions

Arsenic poisoning leads to long-term implications that are detrimental to human health and such incidences through consumption of arsenic-contaminated water and food are well known. It is imperative then to ensure sensitive, rapid, and accurate high-throughput detection of this toxic metalloid so that any possible consequence may be dealt with quickly. For this, robust and cost-efficient methods of arsenic detection are needed which entail minimum sample preparation and high selectivity, and which are able to detect arsenic at ultra-low concentrations to ensure regulatory compliance. Where the laboratory bound conventional detection methods pose obvious practical implications for on-site application and hence high-throughput analyses, the chemical methods pose a significant challenge to reliability. The electrochemical methods offer sensitive responses; however, their need for specific working electrodes questions their onsite capability, though these methods are the focus of much research and may provide opportunity in arsenic diagnostics. In this light, the role of biosensors as potential alternatives capable of addressing the aforementioned concerns has been established. Among them, recombinant whole-cell biosensors have offered valuable characteristics particularly in terms of sensitivity. These recombinants, which host engineered plasmids comprising regulatory regions of the ars operon fused with reporter genes, were able to elicit responses dependent on arsenic concentration. These biosensors saw further improvements in response characteristics with the advent of newer reporter genes that negated the need for substrates to furnish responses. Some of these biosensors were validated for application in groundwater samples since in these real samples the bioavailability of arsenic is diminished due to the presence of iron oxides, phosphates, and silicates, which has led to false responses. Owing to its biological nature, the whole cell biosensor responses are largely dependent on their physiological state and regeneration characteristics, which challenge the reproducibility of response, and though cell-based systems still provide a platform for biomonitoring, their very long induction times defeat the objective of their development into diagnostic devices. Nanotechnology offers many avenues for improvement, and nanobiosensors, which have impacted diagnostics significantly, are envisaged as the new generation of sensors capable of rapid and ultrasensitive responses offering a clear opportunity for high-throughput detection. Aptamers are such new generation biorecognition elements, which offer immense selectivity, and one aptamer for As(III) has been reported and used as an arsenic sensor showing promising outcomes. These biorecognition elements, which elicit instantaneous responses, are capable of differentiating As(III) from As(V), which highlights the level of selectivity offered by them. It is further suggested to combine the efficacy of aptamers with the remarkable properties of a conducting matrix such as graphene, which will add a new dimension in arsenic analytics towards miniaturized devices for high-throughput on-site detection. It may be reiterated here that no standalone method can offer significant working results, but these newgeneration, easy-to-use devices are capable of supporting and augmenting the conventional detection methods and subsequently contribute to methods that can be considered when establishing reference points of action for risk assessments.

References

- [1] Mandal BK, Suzuki KT. Arsenic round the world: a review. Talanta 2002;58:201-35.
- [2] Nakahara H, Yanokura M, Murakami Y. Environmental effects of geothermal wastewater on the near-by river system. J Radioanal Chem 1978;45:25–36.
- [3] Chatterjee A, Das D, Chakraborti D. A study of groundwater contamination by arsenic in the residential area of Behala, Calcutta, due to industrial pollution. Env Poll 1993;80:57–65.
- [4] Flora SJS, Bhadauria S, Kannan GM, Singh N. Arsenic induced oxidative stress and the role of antioxidant supplementation during chelation: a review. J Env Biol 2007;28:333–47.
- [5] ATSDR—Agency for Toxic Substances and Disease Registry. Available at: http://www.atsdr.cdc.gov/toxprofiles/tp.asp?id=22&tid=3#bookmark07>, accessed on April 3; 2014.
- [6] Dhar RK, Biswas BK, Samanta G, Mandal BK, Chakraborti D, Roy S, et al. Groundwater arsenic calamity in Bangladesh. Curr Sci 1997;73:48–59.
- [7] Smith AH, Lingas EO, Rahman M. Contamination of drinking water by arsenic in Bangladesh: a public health emergency. Bull World Health Org 2000;78:1093–103.
- [8] Vega L, Styblo M, Patterson R, Cullen W, Wang C, Germolec D. Differential effects of trivalent and pentavelent arsenicals on cell proliferation and cytokine secretion in normal human epidermal keratinocytes. Toxicol Appl Pharmacol 2001;172:225–32.
- [9] World Health Organization (WHO). Arsenic compounds Environmental health criteria 224, 2nd ed. Geneva: World Health Organization; 2001.
- [10] Van Geen A, Cheng Z, Seddique AA, Hoque MA, Gelman A, Graziano JH, et al. Reliability of a commercial kit to test groundwater for arsenic in Bangladesh. Environ Sci Technol 2005;39:299–303.
- [11] Spear JM, Zhou YM, Cole CA, Xie YF. Evaluation of arsenic field test kits for drinking water analysis. J Am Water Works Assoc 2006;98:97–105.
- [12] Sankararamakrishnan N, Chauhan D, Nickson RT, Tripathi RM, Iyengar L. Evaluation of two commercial field test kits used for screening of groundwater for arsenic in Northern India. Sci Total Environ 2008;401:162–7.
- [13] Mays DE, Hussam A. Voltammetric methods for determination and speciation of inorganic arsenic in the environment—a review. Anal Chim Acta 2009;646:6–16.
- [14] Sarkar P, Banerjee S, Bhattacharyay D, Turner APF. Electrochemical sensing systems for arsenate estimation by oxidation of L-cysteine. Ecotoxicol Environ Saf 2010;73:1495–501.
- [15] Salimi A, Mamkhezri H, Hallaj R, Soltanian S. Electrochemical detection of trace amount of arsenic(III) at glassy carbon electrode modified with cobalt oxide nanoparticles. Sens Actuators B 2008;129:246–54.
- [16] Aleksic J, Bizzari F, Cai Y, Davidson B, de Mora K, Ivakhno S, et al. Development of a novel biosensor for the detection of arsenic in drinking water. Synth Biol 2007;1:87–90.
- [17] Joshi N, Wang X, Montgomery L, Elfick A, French CE. Novel approaches to biosensors for detection of arsenic in drinking water. Desalination 2009;248:517–23.
- [18] Sanllorente-Mendez S, Domínguez-Renedo O, Arcos-Martinez MJ. Immobilization of acetylcholinesterase on screen-printed electrodes. Application to the determination of arsenic(III). Sensors 2010;10:2119–28.
- [19] Mora KD, Joshi N, Balint BL, Ward FB, Elfick A, French CE. A pH-based biosensor for detection of arsenic in drinking water. Anal Bioanal Chem 2011;400:1031–9.
- [20] Daunert S, Barrett G, Feliciano JS, Shetty RS, Shrestha S, Spencer WS. Genetically engineered whole-cell sensing systems: coupling biological recognition with reporter genes. Chem Rev 2000;100:2705–38.
- [21] Stocker J, Balluch D, Gsell M, Harms H, Feliciano J, Daunert S, et al. Development of a set of simple bacterial biosensors for quantitative and rapid measurements of arsenite and arsenate in potable water. Environ Sci Technol 2003;37:4743–50.

- [22] Tauriainen S, Karp M, Chang W, Virta M. Recombinant luminescent bacteria for measuring bioavailable arsenite and antimonite. Appl Environ Microbiol 1997;63:4456–61.
- [23] Tauriainen SM, Virta MPJ, Karp MT. Detecting bioavailable toxic metals and metalloids from natural water samples using luminescent sensor bacteria. Water Res 2000;34:2661–6.
- [24] Date A, Pasini P, Daunert S. Construction of spores for portable bacterial whole cell biosensing systems. Anal Chem 2007;79:9391–7.
- [25] Yoshida K, Inoue K, Takahashi Y, Ueda S, Isoda K, Yagi K, et al. Novel carotenoid-based biosensor for simple visual detection of arsenite: characterization and preliminary evaluation for environmental application. Appl Environ Microbiol 2008;74:6730–7.
- [26] Diesel E, Schreiber M, van der Meer JR. Development of bacteria-based bioassays for arsenic detection in natural waters. Anal Bioanal Chem 2009;394:687–93.
- [27] Ramanathan S, Shi W, Rosens BP, Daunert S. Sensing antimonite and arsenite at the subattomole level with genetically engineered bioluminescent bacteria. Anal Chem 1997;69:3380–4.
- [28] Ramanathan S, Shi W, Rosen BP, Daunert S. Bacteria-based chemiluminescence sensing system using β-galactosidase under the control of the ArsR regulatory protein of the *ars* operon. Anal Chim Acta 1998;369:189–95.
- [29] Rothert A, Deo SK, Millner L, Puckett LG, Madou MJ, Daunert S. Whole-cell-reporter-gene-based biosensing systems on a compact disk microfluidics platform. Anal Biochem 2005;342:11–19.
- [30] Wells M, Gosch M, Rigler R, Harma H, Lasser T, van der Meer JR. Ultrasensitive reporter protein detection in genetically engineered bacteria. Anal Chem 2005;77:2683–9.
- [31] Baumann B, van der Meer JR. Analysis of bioavailable arsenic in rice with whole cell living bioreporter bacteria. J Agric Food Chem 2007;55:2115–20.
- [32] Wackwitz A, Harms H, Chatzinotas A, Breuer U, Vogne C, van der Meer JR. Internal arsenite bioassay calibration using multiple bioreporter cell lines. Microb Biotechnol 2008;2:149–57.
- [33] Wells M, Gosch M, Harms H, van der Meer JR. Response characteristics of arsenic sensitive bioreporters expressing the *gfp* reporter gene. Microchim Acta 2005;151:209–16.
- [34] Trang PTK, Berg M, Viet PH, Mui NV, van der Meer JR. Bacterial bioassay for rapid and accurate analysis of arsenic in highly variable groundwater samples. Environ Sci Technol 2005;39:7625–30.
- [35] Harms H, Rime J, Leupin O, Hug SJ, Meer JRVD. Effect of groundwater composition on arsenic detection by bacterial biosensors. Microchim Acta 2005;151:217–22.
- [36] Robertson DL, Joyce GF. Selection *in vitro* of an RNA enzyme that specifically cleaves single-stranded DNA. Nature 1990;344:467–8.
- [37] Ellington AD, Szostak JW. *In vitro* selection of RNA molecules that bind specific ligands. Nature 1990;346:818–22.
- [38] Tuerk C, Gold L. Systematic evolution of ligands by exponential enrichment: RNA ligands to bacteriophage T4 DNA polymerase. Science 1990;249:505–10.
- [39] Johnson S, Evans D, Laurenson S, Paul D, Davies AG, Ferrigno PK, et al. Surface immobilized peptide aptamers as probe molecules for protein detection. Anal Chem 2008;80:978–83.
- [40] Zhu G, Ye M, Donovan MJ, Song E, Zhao Z, Tan W. Nucleic acid aptamers: an emerging frontier in cancer therapy. Chem Commun 2012;48:10472-480.
- [41] Liu H, Xu S, He Z, Dend A, Zhu JJ. Supersandwich cytosensor for selective and ultrasensitive detection of cancer cell using aptamer-DNA concatamer-quantum dots probes. Anal Chem 2013;85:3385–92.
- [42] Kim M, Um HJ, Bang S, Lee SH, Oh SJ, Han JH, et al. Arsenic removal from Vietnamese groundwater using the arsenic-binding DNA aptamer. Environ Sci Technol 2009;43:9335–40.
- [43] Wu Y, Zhan S, Wang F, He L, Zhia W, Zhou P. Cationic polymers and aptamers mediated aggregation of gold nanoparticles for the colorimetric detection of arsenic(III) in aqueous solution. Chem Commun 2012;48:4459–61.

- [44] Wu Y, Liu L, Zhan S, Wang F, Zhou P. Ultrasensitive aptamer biosensor for arsenic(III) detection in aqueous solution based on surfactant-induced aggregation of gold nanoparticles. Analyst 2012;137:4171–8.
- [45] Wu Y, Zhan S, Xing H, He L, Xu L, Zhou P. Nanoparticles assembled by aptamers and crystal violet for arsenic(III) detection in aqueous solution based on a resonance Rayleigh scattering spectral assay. Nanoscale 2012;4:6841–9.
- [46] Wang B, Song Q, Luo B, Li X, Liang M, Feng X, et al. Exploring the interaction between graphene derivatives and metal ions as a key step towards graphene–inorganic nanohybrids. Chem Asian J 2013;8:410–13.
- [47] Alwarappan S, Erdem A, Liu C, Li CZ. Probing the electrochemical properties of graphene nanosheets for biosensing applications. J Phys Chem C 2009;113:8853–7.
- [48] Bonanni A, Loo AH, Pumera M. Graphene for impedimetric biosensing. Trends Anal Chem 2012;37:12–21.
- [49] Calvo D, Tort N, Salvador JP, Marco MP, Centi F, Marco S. Preliminary study for simultaneous detection and quantification of androgenic anabolic steroids using ELISA and pattern recognition techniques. Analyst 2011;136:4045–52.
- [50] Cho SB, Won HH. Cancer classification using ensemble of neural networks with multiple significant gene subsets. Appl Intell 2007;26:243–50.
- [51] Gutes A, Cespedes F, Alegret S, del Valle M. Determination of phenolic compounds by a phenol oxidase amperometric biosensor and artificial neural network analysis. Biosens Bioelectron 2005;20:1668–73.
- [52] Guttierez M, Alegret S, del Valle M. Potentiometric bioelectronic tongue for analysis of urea and alkaline ions in clinical samples. Biosens Bioelectron 2007;22:2171–8.

26

Medical Countermeasures— Chelation Therapy

Govinder Flora¹, Megha Mittal², and Swaran J.S. Flora³

¹SBBS INSTITUTE OF TECHNOLOGY AND MANAGEMENT, JALANDHAR, INDIA, ²DEPARTMENT OF CHEMISTRY, MANAV RACHNA COLLEGE OF ENGINEERING, FARIDABAD, INDIA, ²DIVISION OF REGULATORY TOXICOLOGY, DEFENCE RESEARCH AND DEVELOPMENT ESTABLISHMENT, GWALIOR, INDIA

CHAPTER OUTLINE

26.1 Introduction	589
26.2 Clinical Aspects of Arsenic	590
26.3 Diagnosis	591
26.4 Chelation Therapy	592
26.5 Chelators in Clinical Use	598
26.6 Analogues of DMSA as Potential New Arsenic Chelators	605
26.7 Role of Antioxidants in Preventing Arsenic Toxicity	608
26.8 Newer Strategies	614
26.9 Concluding Remarks and Future Directions	618
References	618

26.1 Introduction

Arsenic is a well-known environmental toxicant and long-term exposure thereto may lead to severe clinical manifestations, particularly lung and skin cancer [1]. Various regulatory agencies such as the European Commission [2], the United States Environmental Protection Agency (USEPA) [3], and the World Health Organization (WHO) [4] have revised the maximum concentration of arsenic in drinking water by decreasing it from 50 to $10 \mu g/L$. Chelation therapy is the only promising clinical treatment required for reducing the toxic effects of arsenic. Chelation therapy refers to the administration of a chemical agent to remove toxic metal from the body. The term chelation originated from the Greek word *chele*, which means claw of a lobster, thus representing the concept of clutching or holding with a strong grasp. The major objective of

chelation therapy is to inhibit the action of intruding metal by sequestering it through the formation of complexes that are rapidly excreted from intracellular and extracellular spaces of the body. Chelating agents can alter the effects of metal toxicity by mobilizing the toxic metal mainly into the urine. A chelating agent forms a stable complex with toxic metal, shielding biological targets, and thus removes the specific toxic metal ion from the desired site in the body. The chemical nature and biological responses differ with different chelating agents.

One of the key features of chelation therapy is to have good chelating agents. It has been suggested that an ideal chelator should have high solubility in water, be resistant to biotransformation, have the ability to reach the sites of metal storage, retain chelating capability at the pH of body fluids, and, most importantly, form metal complexes that are less toxic than the free metal ion [5]. Although chelating agents are the only clinical option available to limit arsenic toxicity, their use is often limited by various side effects and lack of selectivity and specificity. Thus, it becomes imperative to identify new novel chemical compounds that are effective in excreting excess toxic metal/metalloid from the body without altering the physiological ionic homeostasis.

A chelating agent binds to metals at two or more sites, as chelating molecules possess electrons required to form bonds with positively charged metal ions. It can be attached to the metal ion by two or more bonds leading to the formation of a ring referred as a "chelate ring" [6]. However, the complex formed by the metal-chelating agent in the human body is influenced by various factors, including competing metals and ligands, dynamics of circulation, compartmentalization, and metabolism of the chelating agent. Thus, chelation therapy is the most promising medical regime in the field of metal toxicity. This chapter focuses on the use of chelating agents and various other therapeutic strategies available for the treatment of arsenic intoxication.

26.2 Clinical Aspects of Arsenic

Arsenic exists in two oxidative forms, mainly as arsenite (As^{III}) and pentavalent arsenate (As^{V}). When it combines with elements like oxygen, sulfur, and chlorine, it is referred to as inorganic arsenic, and with hydrogen and carbon it is known as organic arsenic. Inorganic arsenic is generally more toxic than organic arsenic [7]. Among inorganic forms, trivalent arsenic is most toxic and reacts with protein thiol groups, whereas the pentavalent form is comparatively less toxic and uncouples oxidative phosphorylation. Moreover, the pentavalent form converts to the trivalent form of arsenic through reduction [8]. Clinically, arsenic affects all organs but it preferentially interferes with the central nervous system, hematopoietic system, and the hepatic and renal systems [9,10]. Globally, more than 200 million people are at risk, residing in areas where groundwater arsenic concentrations exceed the WHO maximum permissible level of 50 mg/L [11], with about 50% of these people residing in Bangladesh and West Bengal, India, who suffer from arsenicosis [12].

It is well reported that arsenic metabolism in cells leads to the generation of various types of reactive oxygen species (ROS). Oxidative stress has been associated with the progression of

arsenic-induced pathological conditions such as carcinogenicity. Reactive nitrogen species (RNS) are also produced in response to arsenic and cause oxidative damage to lipids, proteins, and DNA in cells [13,14]. Depending on the oxidative state of arsenic, it affects the cells by one of two main mechanisms [15]: (1) trivalent arsenite (As^{III}) binds with sulfhydryl groups of critical enzymes that deplete lipoate, which plays a critical role in the synthesis of Krebs cycle intermediates—depletion of lipoate inhibits Krebs cycle and oxidative phosphorylation resulting in ATP depletion [16]; and (2) pentavalent arsenate (As^V), on the other hand, replaces the stable phosphate ester bond in ATP resulting in uncoupling of oxidative phosphorylation and depleted ATP stores [16,17]. Experimental evidence suggests that arsenic-induced generation of free radicals can cause cell death through the activation of various oxidative-sensitive signaling pathways [18].

Arsenic is a carcinogenic toxicant and its chronic exposure leads to severe health complications such as tumors of the lung, skin, liver, bladder and kidney, skin lesions, peripheral neuropathy, and anemia [19]. Das et al. [20] reported cardiovascular disorders following oral exposure to arsenic. There is also evidence from epidemiological trials that inhaled inorganic arsenic may affect the cardiovascular system. Epidemiological evidence on the association between arsenic and cardiovascular outcomes in Taiwan has been conducted and has estimated the chances of relative risks for coronary disease, stroke, and for peripheral arterial disease [21].

There is another interesting study in which the association between intellectual deficiencies in children and exposure to arsenic has been reported. Adolescents from various regions of Taiwan and China, exposed to low (0.0017–0.0018 mg As/kg/day) levels of inorganic arsenic in drinking water, showed decreased performance in the switching attention task, while children in the high exposure group (0.0034–0.0042 mg As/kg/day) showed decreased performance in both the switching attention task and tests of pattern memory, relative to unexposed controls [22]. Moreover, excessive ingestion of arsenic compounds like arsenic trisulfide could act as suicide agents [23].

26.3 Diagnosis

Since arsenic is a ubiquitously found metalloid, escaping from its exposure is inevitable, especially in areas endemic to arsenic-contaminated drinking water. Thus, diagnosis of arsenic levels in humans is an important aspect of chelation therapy. Arsenic levels can be measured from various body fluids; urine and blood, but also hair and nail samples are the most common sources. These sources act as good biomarkers for arsenic poisoning, for instance: (1) urine is the most preferred biomarker for arsenic exposure with a reference value of $<25 \,\mu\text{g/dL}$. For the measurement of arsenic concentration, 8–24-hour urine should be collected. A "spot" of urine sample is not considered but it may be helpful in acute poisoning. Generally, two methods are employed for measuring arsenic in urine samples—fractionated (speciated) measurement for inorganic and organic arsenic, and total arsenic level. However, this test could be carried out within 1 week after seafood abstinence to avoid false-positive results. (2) Blood arsenic levels are rarely used because of the short half-life of arsenic in blood as well as background exposure levels of arsenic in normal blood. Blood arsenic levels are $<3 \mu g/dL$. (3) While hair and nails are considered to be good indicators for chronic arsenic exposure the levels of As therein have limited significance due to external contamination.

26.4 Chelation Therapy

26.4.1 Concept

Chelating agents are chemical compounds whose structures permit the attachment of their two or more donor atoms (or sites) to the same metal ion simultaneously and produce one or more rings. These molecules are also called "chelates" or chelating groups, and the formation of rings is called "chelation." These metal complexes have the ability to resolve into optically active (R&L) forms. Stability of metal complexes differs with pattern of complex formation and difference in stability becomes more relevant in the increasingly dilute solutions in biological systems such as serum or tissue. The toxicokinetics and toxicodynamics of metal and chelating agents are an integral part of an effective chelation therapy in addition to the following criteria, which also need to be fulfilled:

- High affinity for the toxic metal
- Low affinity for essential metals
- Minimal toxicity
- Lipid solubility
- Good absorbability from the gastrointestinal tract.

These criteria are not easy to fulfill as there is yet no available chelating agent that exhibits all the above-mentioned points. Lipid soluble chelators easily cross the cell membrane and bind metals within the cell but chelators that are not lipid soluble are usually more toxic. Thus, it is difficult for lipophobic chelators to find optimal conditions for binding specific toxic metal with minimal risk of adverse effects. Most of the metals encountered generally react with O-, S-, and N-containing ligands because they contain lone pairs of electrons (Figure 26–1). A simplified mechanism by which a chelating agent can detoxify a toxic metal may be depicted as:

$$M^{n+}$$
 + chelating agent = M^{n+} – chelating agent complex

where M^{n+} is the toxic metal cation that binds to a chelating agent and forms an insoluble complex *in vivo*.

The process of chelation depends on both the nature and the properties of the metal and those of the chelating agent, such as ionic diameter, ring size and deformability, and hardness or softness of the electron donors and acceptors. It also depends on factors such as route of administration, bioavailability, metabolism, organ and intra/extracellular compartmentalization, and excretion [24].



FIGURE 26-1 Showing the monodentate and bidentate ligands along with donor atom and metal.

26.4.2 Chemistry

The chemistry involved provides an understanding about metal transport, metal binding constants, and ion specificity in relation to metal-ion exchange kinetics. These processes/parameters introduce opportunities for metal-ion-related therapy, which goes beyond traditional chelatebased metal-ion detoxification. Hardness-softness character and the chelating effect determine the following properties of the chelator-metal complex: toxicity of chelator, side effects of the chelator, hydrophilicity/lipophilicity of the chelating agent, hydrophilicity/lipophilicity of the resulting metal complex, and stability of the metal-chelator complex. To design a chelating agent for a toxic metal ion, the following characteristics must be kept in mind about the toxic metal:

- Coordination number
- Net ionic charge
- Stereochemistry.

The chelating agent occupies a coordination position in a metal ion. It generally forms a complex of greater stability compared to those complexes with chelating agents that occupy fewer positions. Metal ion forms bond with ligands, which can share electron pairs and the formation represents the "coordination sphere" of the metal ion. Metal atoms can form complexes with a coordination number from 2 to 8. The atoms present in the coordination sphere undergo an alteration when a toxic metal is consumed and absorbed by any organism. The bond formation between the metal ions and several types of donor atoms takes place to acquire a more stable characteristic coordination sphere in the biological system. Most commonly found donor atoms in living systems include oxygen atoms (water, carbonyl and phenolic groups of amino acids, phosphate groups, etc.), nitrogen atoms (amino acids, nucleotides,

etc.), and sulfur atoms (cysteine- and thiol-containing compounds such as methionine, etc.). A toxic metal coordinates with ligands of a physiologically essential molecule like an enzyme, messenger molecule, or DNA resulting in their decreased reactivity. These changes in the reactivity of the essential molecules are responsible for the biologically adverse effects of toxic metal.

Chelating agents excrete intracellular deposits of toxic metals by four common pathways: (1) non-polar chelators passing through the lipid portion of the cellular membrane, (2) electrically neutral polar chelators passing through the cellular membrane, (3) mono-anions passing through appropriate mono-anion transport systems, and (4) di-anions passing through appropriate di-anion transport systems. The use of a chelating agent restricted to the extracellular space leads to a large reduction of the toxic metal, which usually favors the diffusion of some of the toxic metal from intracellular sites to extracellular sites [25] (Figure 26–2).

The clinical efficacy of a chelating agent is governed by its ability to shift the toxic metal ion from biomolecules such as enzymes and DNA, and subsequently reactivate. In this process, the toxic metal-chelate complex, which is lipophilic in nature, is excreted from the body and so leaves the metal unavailable for attaching to another biomolecule.

26.4.3 Chemical Considerations

26.4.3.1 Thermodynamics of Metal Chelation

In complexation reactions of metals with "*n*" molecules of the monodentate ligand *L* or with one molecule of the *n* dentate ligand represented by Ψ :

$$M + nL \rightarrow ML_n$$
 (26.1)

$$M + \Psi \to M\Psi$$
 (26.2)

The overall stability constants can be expressed by

$$\beta_L = \frac{[ML_n]}{[M][L]^n} \tag{26.3}$$

$$\beta_{\Psi} = \frac{[M\Psi]}{[M][\Psi]^n} \tag{26.4}$$



FIGURE 26–2 Excretion of chelate compounds from intracellular sites to extracellular sites is regulated by various channels. The channels are specific for chelators on the basis of polarity and ionic charge.

The stability of a complex depends on

$$\Delta G^0 = \Delta H^0 - T \Delta S^0 = -RT \ln\beta \tag{26.5}$$

During complex formation of a metal with n monodentate ligands, the following changes take place:

- Enthalpy changes associated with bonding, which is proportional to the free energy
- Entropy changes, directly proportional to the ordering of ligands around the ion, which counteract the entropy effect of desolvation.

If the number of ligands enters one molecule of a multi-dentate ligand, the entropy support from desolvation and the complex stability may increase.

26.4.3.2 Kinetic Considerations in Metal Chelation

Chelating agents possess "ligand," which binds atoms capable of forming two covalent or coordinate linkages in the case of bidentate chelates. Ligands comprise atoms like S, N, and O, which function in the form of chemical groups like -SH, -S-S, $-NH_2$, =NH, -OH, $-OPO_3H$, or >C=O [26] (Figure 26–3). Bidentate or multi-dentate ligands form ring structures, which include metal ion and the two ligand atoms. Most of the donors act as bidentate ligands [5].

26.4.3.3 In Vivo Efficacy of Chelating Agents

Toxic metals are generally present in "free" concentrations because of small biological ligands available in the biological system, which may form mixed aqua-bioligand complexes with metals. Complexation reactions *in vivo* most often occur as a series of ligand or metal exchange reactions between toxic metals and chelating agents. Equilibrium constant and the complex formation are highly favorable but are limited due to factors like (1) rate effects, (2) competition with other ligands/metals, and (3) systemic transport kinetics of the chelator. Formation of a complex between chelator and toxic metal and quantitative urinary excretion of the ML complex (at equilibrium), and the efficiency *E* of a chelating agent for mobilizing a toxic metal, can be described as:

$$M + L \leftrightarrow [ML] \tag{26.6}$$

$$E = \frac{ML}{M} \tag{26.7}$$



FIGURE 26-3 Binding of the monodentate and bidentate.

However, the potential for mobilizing metal depends on the degree of formation of the ML complex. In this context, considering one of the important biological competing metals, Ca(II), and total chelator concentration, L_v the conditions for a large E can be described from the standard stability constant [27]:

$$\frac{ML}{M} = \beta_{ML}[L] \tag{26.8}$$

Chelation efficiency depends on the chemical and biological characteristics of the metal, the chelator, and the organism. The main focus in most clinical uses is mainly on the mobilization of the toxic metal. Therefore, a chelating agent forms a stable complex with a toxic metal and shields the biological targets from the metal ions and reduce the toxicity soon after administration when mobilization has not yet occurred. It may expose the metal to the biological environment and prevent the metal from being scavenged by biological protective mechanisms, which leads to increased toxicity of the metal.

Metal ions have a critical role in the biological system, either beneficial or pathological. However, there are a few misconceptions about a chelating agent: for example, it will inhibit biological activity and selectivity, and the outcome. In complex biological environments such as exists in humans, it is very difficult to have an exquisite selectivity for one specific metal ion over other metals. In addition, the metal ion sequestered by the chelators leads to the formation of a metal complex, which itself may have some biological effect [28].

Chelation plays an important role in analytical chemistry as it finds a place in toxicity, mechanisms, and therapy. Arsenic exerts its adverse effects by forming complexes with enzymes, DNA, and other biomolecules in the cell. Chelating agents *per se* are added exogenously to a biological system and may bind to endogenous ligands for the removal of arsenic. They must also effectively compete for the removal of arsenic *in vivo*. A chemical molecule that qualifies as an ideal chelator in an *in vitro* model may not necessarily also act in a similar way *in vivo*. There might be number of other considerations including toxicity or competition with other endogenous substances such as hemoglobin and cytochromes, which also bind metals and thus might also act as chelators. Selectivity of a chelator *in vivo* is an important factor to be considered in chelation therapy. pH is another important factor, which may influence complex formation and stability. Many chelating agents are shown to be unstable at low pH; while at higher pH, metals tend to form insoluble hydroxides, which become less accessible to chelating agents. This condition is particularly significant in pathological conditions leading to acidosis or alkalosis.

26.4.4 Toxicokinetics of Chelation

Chelating agents are rapidly excreted over a few hours or days while the toxic metals accumulate for prolonged periods and thus are distributed to various organs. Thus, it is not easy for all the chelating agents to reach the desired site. A chelating agent will mobilize the most readily available metal first, typically in the plasma, kidney, and liver, and then to a lesser extent in bone and the central nervous system. The most readily accessed "pools" of toxic elements will be depleted due to repeated doses, but re-equilibration/redistribution slowly replenishes the toxic elements in more accessible body compartments. This is evident from the rebound levels of arsenic in the blood after discontinuation of administration of a chelator, which high-lights three important facts: (1) blood and urine are poor substitutes to measure toxins accrued over the lifetime, although they are the most commonly used biomarkers to detect arsenic. (2) Sequestration of toxic metals in bone and soft tissues does not signify their complete immobilization; they return to the bloodstream and tissues where they once again exert toxic effects. Thus it is necessary to acquire an understanding of the quantities of biologically accessible toxic elements within the body that are not necessarily reflected in baseline blood or urine levels, before proceeding with chelation therapy. (3) Chelation therapy induces a shift of both essential and toxic cations in the body [29] (Figure 26–4).

The toxicokinetics of a chelating agent depends on various factors such as (1) hydrophilicity/lipophilicity of the compound, (2) whether it chelates extracellular or intracellular metal deposits, (3) the best route of administration (oral vs. intravenous infusion). Moreover, it also determines the hydrophilicity/lipophilicity of the formed metal complex. Urinary excretion is the main mode to remove lipophilic complexes, but due to reabsorption, excretion of these complexes is decreased, which alters the metal's organ distribution. Moreover, chelators forming lipophilic metal complexes enhance intestinal metal uptake, which potentially enhances the toxicity of the metal [27]. The oral route for administration of chelating agents is preferred and comprises emptying of the GI system of the toxic metal and removal of the subject from further exposure to avoid increased intestinal metal absorption. However, an orally administered chelating agent forms hydrophilic metal complexes, which may efficiently reduce intestinal metal uptake resulting in local toxicity after oral intoxication [30,31].



FIGURE 26–4 Toxicokinetics of hydrophilic and lipophilic chelators suggests removal of arsenic. Hydrophilic chelators are unable to cross the blood-brain barrier while lipophilic chelators can easily enter soft and hard tissues and bring about effective removal of arsenic from both extra- and intracellular sites.

26.5 Chelators in Clinical Use

Chelating agents have been used as antidotes for metal intoxication in humans since the Second World War. One of the first applied chelators was British anti-Lewisite (BAL), a chelating drug developed during World War II in the UK as an antidote for the arsenic-containing warfare agent Lewisite. Later, BAL was also used to sequester arsenicals, gold, and mercury [32]. Meso-2,3-dimercaptosuccinic acid (DMSA), sodium 2,3-dimercaptopropane 1-sulfonate (DMPS) and D-penicillamine (D-PA) were also introduced for the clinical use after BAL.

Today, these drugs together with a wide array of newly synthesized chelating agents are in clinical use or under preclinical or clinical investigations for treatment of intoxication or overload caused by various transition metals including metalloids like arsenic. However, in the case of arsenic poisoning, chelation following chronic exposure has been shown to increase excretion but in terms of potential clinical efficacy (decreased mortality and morbidity) this is still debatable. However, *in vivo* studies have shown that immediate treatment with BAL, DMPS, and DMSA can avert the adverse effects of inorganic arsenic [33].

26.5.1 Chelating Agents for Arsenic Poisoning

26.5.1.1 Dimercaprol (BAL)

26.5.1.1.1 CHEMISTRY, PHARMACOKINETICS, AND PHARMACODYNAMICS

British anti-Lewisite (BAL) or dimercaprol (2,3-dimercapto-1-propanol) is a lipophilic drug and on administration is distributed both intra- and extracellularly (Table 26–1) [34]. Its utilization during World War II minimized the risk of injury or death to the Allied infantry from Lewisite. BAL also revolutionized the treatment of heavy metal poisoning. BAL is a dithiol compound containing two sulfhydryl groups, which form a stable, relatively non-toxic five-membered heterocyclic chelate ring with arsenic.

BAL is generally administered at a dose of 3–5 mg/kg intramuscularly every 4 hours for 2–10 days depending on the toxic side effects [35]. However, in acute cases doses may be increased 4–5 mg/kg every 4 hours for the first 24 hours. BAL has a short half-life and is rapidly absorbed (30–60 min), metabolized, and excreted (within 4 hours) from the biological system [38,43]. Early administration of BAL after exposure could effectively and efficiently prevent the inhibition of sulfhydryl enzymes [26,38].

26.5.1.1.2 EFFICACY AND EXPERIMENTAL STUDIES

BAL is an effective chelator for acute arsenic poisoning (inorganic or elemental) [38,44-47]. Interestingly, BAL is the most toxic commercially available chelating agent compared to its other derivatives such as meso-DMSA and DMPS. Intraperitoneal lethal dose (LD₅₀) of BAL is 0.85 mmol/kg in rats and 1.5 mmol/kg in mice [48,49].

26.5.1.1.3 MECHANISM OF ACTION

BAL has two –SH groups in its structure and competes with the thiol groups of enzymes for binding with arsenic or other metals to form a stable metal–chelate complex. The formed complex

Chelating Agent	Structure	Route of Administration/ Dose	Limitations	References
BAL	$\begin{array}{c} SH \\ HO-CH_2-CH-CH_2-SH \end{array}$	Administration of BAL is typically 3–5 mg/kg intramuscularly every 4 hours for 2–10 days	Low therapeutic index, painful intramuscular injection, tendency to redistribute arsenic to brain and testes	[35]
D-penicillamine		4 daily doses of 25 mg/kg/dose orally	Reversible hepatic and renal dysfunction, and autoimmune diseases	[36,37]
DMPS	SH HSSO₃H	200 to 400 mg daily doses orally in adults, intravenous doses of 250 mg every 4 to 6 hours and parenteral route	Headache, fatigue, nausea, taste impairment, pruritus, and rash	[38]
m-DMSA	HO SH OH	Administered parenterally or orally in doses of 30 mg/kg/day for 5–7 days followed by 20 mg/kg/day for 1–3 weeks	DMSA not able to remove metals from hard tissues and intracellular sites	[39,40]
Miadmsa	$HS \xrightarrow{O} CH_3 \\ CH_3 \\ CH_3 \\ HS \xrightarrow{O} OH \\ OH$	Oral administration of MiADMSA was more effective than intraperitoneal administration and the minimum effective dose with least side effects was 50 mg/kg	Essential metal loss (copper and zinc)	[41,42]

Table 26–1	Structures,	Route of	Administrations,	and Doses of	Chelators
------------	-------------	----------	------------------	--------------	-----------

is then excreted from the body through urine [6]. BAL forms heterocyclic ring complexes with metals as shown in Figure 26–5. Numerous *in vivo* reports suggest that BAL exhibits some side effects like nephrotoxicity and hypertension. The metabolism of BAL has not been well described but it is rapidly metabolized to inactive products such as glucuronide conjugate. If the BAL-metal complex is oxidized the metal is released and exerts its toxic effects. Thus, the dosage of BAL must be high enough to assure the excess of free BAL available in body fluids until the metal is completely excreted [50,51].

26.5.1.1.4 HUMAN CASES

Physicians recommend BAL for arsenic removal even though more novel chelating agents exist [52]. BAL therapy should continue until urinary arsenic levels are less than $50 \mu g/L$ per



FIGURE 26-5 Mechanism of action of BAL.

24 hours. Chelation is rarely needed outside the setting of symptomatic acute poisoning. However, if chelation is prescribed, succimer is the preferred agent given at a dose of 10 mg/ kg orally three times a day for 5 days, then twice a day for 14 days. Two family members who were continuously drinking arsenic-contaminated water for nearly 5 years developed several dermatological changes like hyperkeratosis and hypo- and hyperpigmented lesions (raindrops) on the abdomen, chest, and back. Histopathological examination also confirmed the inflammatory response of a papillary type. Arsenic levels in the nails and hair of both patients were found to be very high. This was probably the first report of successful treatment with dimercaprol.

26.5.1.1.5 DRAWBACKS

BAL is unstable and easily oxidized, thus it is difficult to store and requires immediate use after preparation. Besides rapid mobilization of arsenic from the body, it also causes significant elevation of brain arsenic levels [5]. Administration of BAL requires deep intramuscular injection that is extremely painful and, due to its oily nature, causative of allergy [53]. It has also been administered intraperitoneally or orally in rodents at LD_{50} values of BAL. It has a number of side effects such as: (1) injections of BAL being painful due to painful release of BAL in the tissues, (2) gastrointestinal symptoms like nausea and vomiting, dose-related hypertension, and tachycardia, and (3) nephrotoxicity, seizures, hyperpyrexia, sweating, lacrimation and rhinor-rhea, urticaria, and paresthesia [38,54]. The vehicle used for BAL is peanut oil and it should be used with extreme caution in patients with peanut allergies.

Two other water soluble analogues of dimercaprol have also been studied as Lewisite antidotes. These are meso-2,3-dimercaptosuccinic acid (DMSA) and 2,3-dimercapto-1-propane sulfonic acid.

26.5.1.2 D-Penicillamine (D-PA)

26.5.1.2.1 CHEMISTRY, PHARMACOKINETICS, AND PHARMACODYNAMICS

Penicillamine is a D-B, B-dimethylcysteine, a penicillin degradation product (Table 26–1), used since 1957 as the only approved chelator for lead poisoning until DMSA became available. It can also chelate arsenic [55]. It is absorbed rapidly from the GI tract, although it has an oral bioavailability of 40 to 70% but the effects are not dose dependent [56] and various foods, antacids, and iron salts decrease its absorption from the gut. The peak concentration in blood plasma is reached within 1 to 3 hours regardless of the dose. It forms disulfide bonds with many proteins in the blood such as albumin and homocysteine, and in tissues [57]. A small portion of the parent compound is metabolized to S-methylpenicillamine in the liver and is eliminated in feces. However, the primary route of elimination is the kidneys and the half-life of unchanged penicillamine after a single dose ranges from 1.6 to 3.6 hours. The elimination of penicillamine is increased from 4 to 6 days after a steady concentration is achieved. It was suggested that penicillamine is slowly released from deep tissues and skin [58].

26.5.1.2.2 EFFICACY AND EXPERIMENTAL STUDIES

D-PA has rarely been reported as an effective agent in the treatment of arsenic poisoning. However, oral chelation therapy with D-PA has been shown to be effective in the mobilization of arsenic [59]. Furthermore, arsenic toxicity symptoms manifested post-ingestion could be augmented with D-PA therapy [59].

26.5.1.2.3 MECHANISM OF ACTION

Penicillamine contains a sulfhydryl group, which combines with toxic metals to form ring compounds and increases its elimination. This complex is further excreted through the kidneys and feces.

26.5.1.2.4 HUMAN STUDIES

In cases of acute arsenic poisoning, immediate chelation is required. A case report suggested the effectiveness of D-PA in three children exposed to arsenic and they were treated with four daily doses of 25 mg/kg each [36]. The recommended dose of penicillamine is 25 mg/kg. Compared to other chelators like DMSA, DMPS, and BAL, penicillamine is found to be less effective in arsenic poisoning [60]. The dose of penicillamine in the United States is 30 to 40 mg/kg/day or 600 to 750 mg/square meter/day for 1 to 6 months, given 2 hours before or 3 hours after meals [61].

26.5.1.2.5 DRAWBACKS

The use of penicillamine is limited due to its adverse effects, which led to the development of the thiol chelators such as DMSA and DMPS. The adverse effects include nausea, vomiting, hematuria, bone marrow depression with leukopenia, rashes, proteinuria, hemolytic anemia, gastrointestinal distress, reversible hepatic and renal dysfunction, and autoimmune diseases [37].

26.5.1.3 Meso-2,3-Dimercaptosuccinic Acid (Succimer, DMSA)

26.5.1.3.1 CHEMISTRY, PHARMACOKINETICS, AND PHARMACODYNAMICS

Succimer, meso-2,3-dimercaptosuccinic acid (DMSA), is an orally active and less toxic analogue of dimercaprol (Table 26–1). DMSA is a weak acid with four ionizable hydrogens that are ionized at pH 7.4 but lack the ability to enter cells (Table 26–1) [62]. In the meso-DMSA structure, the toxic metals can coordinate with one sulfur and one oxygen atom in the case of lead and cadmium, or with each of the two sulfur atoms in the case of mercury [63,64]. However, there are many reports regarding its effectiveness and low toxicity in the treatment of arsenic in the Soviet and Chinese literature [27]. Smith and Strupp [65] reported the efficacy and clinical utility of DMSA in rodents and non-human studies. It may be given parenterally or orally in doses of 30 mg/kg/day for 5–7 days followed by 20 mg/kg/day for 1–3 weeks; it is rapidly excreted in urine. In adults, the optimum dose is 30 mg/kg/day for 5 days [39]. It is incompletely absorbed on oral administration with peak plasma levels occurring approximately 1 to 2 hours after administration. It is extensively metabolized and only 25% of the dose is excreted in urine and remains in the feces, which represents the unabsorbed drug [38].

26.5.1.3.2 EFFICACY AND EXPERIMENTAL STUDIES

DMSA is also a preferred lead poisoning antidote, particularly in children as it is less toxic than DMPS. Meso-DMSA has multiple advantages over other chelators, such as low toxicity, oral administration, and no redistribution of metal from one organ to another. Despite its being preferred as a lead antidote, it has been shown to be successful in animal and human cases in arsenic poisoning [66,67]. DMSA has been shown to provide 80–100% survival in mice from lethal manifestations of arsenic after subcutaneous injection [62]. A significant depletion of arsenic and a significant recovery in the altered biochemical variables of chronically arsenic-exposed rats have been reported [53]. This drug thus can be effective if administered orally or intraperitoneally. Flora [68] reported that arsenic-induced oxidative stress in various tissues of rats can be mitigated by pharmacological intervention that encompasses combined treatment with N-acetylcysteine and DMSA.

The toxicity of meso-DMSA is low when given in rats intraperitoneally [69]. However, a contraindication for meso-DMSA therapy is hypersensitivity to the drug. Clinical studies indicate that meso-DMSA in therapeutic doses has no significant effect on renal excretion of essential metals like iron, calcium, and magnesium, with the exception of zinc, where excretion is doubled.

26.5.1.3.3 MECHANISM OF ACTION

Meso-DMSA binds metal cations through the thiol groups, which ionize upon complexation, as shown in Figure 26–6. The monoesters of DMSA were found to be more effective than the parent compound, DMSA. The complexes of DMSA monoesters seem to penetrate cells, which helps in targeting intracellular sites in the body and aids in the removal of arsenic in the cytosol and from organelles inside the cell.



FIGURE 26-6 Mechanism of complex formation of DMSA.

26.5.1.3.4 HUMAN STUDIES

In humans, more than 90% of meso-DMSA in urine is excreted as mixed DMSA-cysteine disulfide conjugates with the remainder excreted unchanged [70]. Although the half-life of DMSA is reported to be about 2 hours, Dart et al. [71] reported that this could be up to 2 days. Meso-DMSA has been reported to be an effective chelator for arsenicosis [39,66,67,70]. In general, patient therapy with 30 mg/kg DMSA daily for 5 days has demonstrated a significant increase in arsenic excretion with clinical improvement. In cases of acute ingestion of arsenic (2 g), a therapy regime of 300 mg DMSA, orally every 6 hours for 3 days, is suggested to show better clinical recoveries. The dose could be reduced to 10 mg/kg, thrice daily for 5–7 days, followed by two daily doses of 10 mg/kg for another 10–14 days, in cases of mild arsenic poisoning. However, in cases of chronic arsenic poisoning the dose could be increased to 18 mg/kg with a similar dose regime to that for mild poisoning [72].

However, a double-blind, randomized, controlled trial study conducted on a few selected patients suffering from arsenic intoxications in West Bengal (India) with oral administration of DMSA suggested that it was not effective in producing any clinical or biochemical benefits or eliciting histopathological improvements in skin lesions [73]. Moreover, renal toxicity has also been associated with excretion of large amounts of chelated metals that pass through the renal tubules in a relatively short period during therapy.

26.5.1.3.5 DRAWBACKS

One of the main drawbacks of DMSA is its extracellular distribution, making it incapable of removing arsenic from intracellular sites. The side effects of DMSA include gastrointestinal discomfort, skin reaction, mild neutropenia, and elevated liver enzymes. Some evidence of embryo toxicity/fetal toxicity due to DMSA administration was also reported [40]. Precautions

are necessary in patients affected with renal or liver impairment, neutropenia, and glucose-6-phosphate dehydrogenase deficiency [74].

26.5.1.4 2,3-Dimercaptopropane-1-Sulfonic Acid

26.5.1.4.1 CHEMISTRY, PHARMACOKINETICS, AND PHARMACODYNAMICS

2,3-Dimercaptopropane-1-sulfonic Acid (DMPS) is a sodium salt of 2,3-dimercaptopropane-1-sulfonic acid, a derivative of BAL (Table 26–1). Presence of the sulfonic acid group renders DMPS water soluble. DMPS enters the intracellular compartment through an organic anion transport pathway since it is distributed extracellularly [34,75]. DMPS is a unithiol and could be administered orally or parenterally. It has also been administered intramuscularly as a 5% solution at a dose of 5 mg/kg, three or four times during the first 24 hours followed by two or three times on the second day, and once or twice on subsequent days. It is rapidly metabolized to acyclic and cyclic disulfides [76,77]. Half-life of the drug was found to be 10 hours [78] and after oral dosing in humans, peak plasma concentration of the total drug metabolized is reached in approximately 3.5 hours. However, if given intravenously, half-life of the parent drug is 1.8 hours and total DMPS is 20 hours [76]. In humans, arsenic and DMPS complexes are primarily removed by renal excretion [38,77]; however, in experimental animals, these complexes could also be removed by biliary excretion [79]. In adults, oral administration of 200 to 400 mg daily doses and intravenous doses of 250 mg every 4 to 6 hours [38] is recommended.

26.5.1.4.2 EFFICACY AND EXPERIMENTAL STUDIES

DMPS is an effective chelating agent for acute or chronic arsenic poisoning. It is less toxic than dimercaprol and has been used extensively in Russia and the Western world [26,34]. No major adverse effects are reported following DMPS administration in humans or animals [80]. DMPS has the ability to inhibit both arsenate-induced acute toxicity and regenerative proliferation of the rat bladder epithelium by decreasing exposure of the urothelium to trivalent arsenicals excreted in the urine [81]. Both DMSA and sodium DMPS have been reported to be useful in the prevention of arsenite- and arsenate-induced developmental toxicity in mice [82–84].

26.5.1.4.3 MECHANISM OF ACTION

Sodium DMPS may form an insoluble complex with arsenic that is firmly bound to intracellular sites. It also makes the drug effective in cases of slow, low-dose, and chronic arsenic poisoning because metal reaches the cellular compartments following the physiological barriers including the blood-brain barrier [85].

26.5.1.4.4 HUMAN STUDIES

Aposhian et al. [86] reported that DMPS has a highly specific effect on MMA metabolism or urinary excretion in humans, although the mechanism by which DMPS reduces arsenic burden is not fully established [72,87]. A study in Mongolia found that after DMPS therapy, (1) MMAIII was detected in human urine suggesting DMPS enhanced MMA excretion, (2) reduced levels of DMA were noted in human urine samples, and (3) the DMPS-MMAIII complexes were inactive as *in vitro* substrate for rabbit liver. This study concluded that DMPS

formed complexes with MMAIII, which inhibited further methylation of MMAIII to DMA and decreased DMA levels as well as increased excretion of the DMPS-MMAIII complex in the urine [88]. Guha Mazumder and co-workers observed that in the case of chronic arsenic poisoning, DMPS is superior to meso-DMSA [89]. DMPS has also been used as a challenging test for arsenic [77] and is approved for the treatment of arsenic intoxication in the People's Republic of China. Heitland and Köster [90] determined inorganic arsenic species As(III), As(V), and the organic species methylarsonate (MMA(V)), dimethylarsinate (DMA(V)), and arsenobetaine (AsB) in human urine by the fast anion exchange HPLC-ICP-MS method.

26.5.1.4.5 DRAWBACKS

Hruby and Donner [80] reported no major adverse effects following DMPS administration in humans or animals. However, a dose-dependent decrease in copper contents was found in the serum, liver, kidneys, and spleen [91]. No abnormalities in offspring with chronic oral DMPS treatment were reported. Oral administration of DMPS did not affect late gestation, parturition, or lactation in mature mice and fetal and neonatal development [40]. However, some minor adverse effects such as headache, fatigue, nausea, taste impairment, pruritus, and rash have been reported [38].

26.6 Analogues of DMSA as Potential New Arsenic Chelators

26.6.1 Monoisoamyl DMSA (MiADMSA)

In order to increase the efficacy of DMSA from just an extracellular chelator to an intracellular one, a number of DMSA analogues have been synthesized as well as tested for their efficacy against arsenic intoxication. Out of a multiplicity of analogues synthesized, monoisoamyl DMSA (MiADMSA), monomethyl DMSA (MmDMSA), and monocyclohexyl DMSA (MchDMSA) are three examples. However, MiADMSA has been shown to be the most effect analogue and is currently under phase I clinical trials. MiADMSA has been found to potentiate the synthesis of glutathione in liver and brain, along with significantly reducing the glutathione disulfide levels in tissues. MiADMSA is also shown to be capable of crossing biomembranes and is more effective in reducing the arsenic burden in acute and subchronic intoxication. MiADMSA is capable of mobilizing intracellularly bound arsenic and is seen to provide an indirect antioxidant effect.

26.6.1.1 Chemistry, Pharmacokinetics, and Pharmacodynamics

MiADMSA, a C_5 branched chain alkyl monoester of DMSA, can be synthesized by controlled esterification of DMSA with isoamyl alcohol (Table 26–1). It is lipophilic in nature compared to the parent DMSA and found to be more effective in reducing arsenic [92,93]. The plasma kinetics of MiADMSA (plasma-free drug and total drug) at 50 and 100 mg/kg p.o. was carried out. MiADMSA at a dose of 50 mg/kg administered orally provided about 45% and 75% protection against oxidative stress and lowering body arsenic burden. The toxicity profile of DMSA with an LD_{50} of 16 mmol/kg is much lower than the toxicity of MiADMSA with an LD_{50} of 3 mmol/kg but less than that for BAL (1.1 mmole/kg). Its structural features and recent experimental evidence suggest that it is effective in chelating arsenic [5,94–96].

26.6.1.2 Efficacy and Experimental Studies

We compared the effects of a few thiol and amino chelators and a DMSA analogue, MiADMSA, on liver and tissue oxidative stress in rats. The results suggested that among the thiol and amino chelators, the latter, and in particular CaNa₂EDTA, were more toxic to hepatic tissue and caused a more pronounced redistribution of metals like zinc. On the other hand, MiADMSA only caused a significant loss of copper [97] with no other major side effects. Further repeated administration of MiADMSA was found to be safe in adult rats followed by young and old rats, with female rats slightly more susceptible than males [98]. We also demonstrated that oral administration was more effective than the parental route [41,99,100]. Similarly to our studies, Kreppel et al. [101] also reported the superior efficacy of MiADMSA and mono n-amyl DMSA in mice against lethal effects of arsenic. These studies support MiADMSA being a potential new, and one of the most effective vicinal classes of, metal chelator [13,92,102]. Based on these results the phase I clinical trials of MiADMSA have been initiated and results are awaited.

To date, no work has reported the protective or therapeutic efficacy of MiADMSA against Lewisite and other arsenicals. However, we reported the efficacy of MiADMSA on the reversal of gallium arsenide (GaAs)-induced changes in hepatic tissue [9]. MiADMSA was found to be better than DMSA in mobilizing arsenic and in the turnover of the GaAs-sensitive biochemical variables. We recently compared the efficacy of orally or intraperitoneally administered MiADMSA against GaAs. Our results indicated that orally administered MiADMSA reduced oxidative stress and arsenic burden by 45% and 75%, respectively, as compared to 25% and 40% through the intraperitoneal route. Furthermore, pharmacokinetic analysis supported prolonged availability of the drug through oral administration [41]. These findings clearly suggest that oral administration of MiADMSA is more effective than intraperitoneal administration and that the minimum effective dose with the fewest side effects was 50 mg/kg [41]. Generation of free radicals in the case of arsenic toxicity is known to cause cellular apoptosis through a mitochondrial-driven pathway. A study was conducted to investigate the effect of arsenic interactions with various complexes of the electron transport chain and to evaluate if the complex could trigger apoptosis. Another aspect evaluated was chelation with MiADMSA and reversal of detrimental effects. Arsenic induced free radical generation in rat neuronal cells, leading to diminished mitochondrial potential and enzyme activities of all the complexes of the electron transport chain. MiADMSA, however, was able to reverse most of these arsenic-induced altered variables to various extents, and DNA damage remained unaffected [103]. Ramkumar et al. [104] also reported that MiADMSA significantly reversed arsenic-induced alterations in behavior and biochemical variables suggestive of oxidative injury. Arsenic-exposed rats showed significant differences in behavioral functions and water maze learning. Further, biochemical studies also showed significant alteration, which was effectively restored upon MiADMSA treatment. Besides the above-mentioned in vivo studies, few in vitro studies also supported the therapeutic efficacy of MiADMSA against arsenic. Using HaCaT cell lines, we recently reported that pretreatment with MiADMSA offered significant protection against arsenic-induced oxidative stress and apoptotic cell death. These findings are of clinical relevance and suggest MiADMSA to be a promising candidate in protecting skin against arsenic-induced toxic effects [105]. Sannadi et al. [106] also evaluated the therapeutic efficacy of MiADMSA against arsenic-induced developmental neurotoxicity. Pregnant rats exposed to sodium meta-arsenite via drinking water throughout the prenatal period were treated with MiADMSA. The results suggested that MiADMSA significantly reversed arsenic-induced apoptosis and oxidative stress in the brain region, a major contributing factor to neurotoxicity.

Despite a few drawbacks (copper loss) associated with MiADMSA, the above results suggest that MiADMSA may be a future drug of choice owing to its ability to cross the blood-brain barrier, its lipophilic character, and the absence of any metal redistribution [42,97]. Moderate toxicity observed after repeated MiADMSA administration may be reversible after withdrawal of the chelating agent. MiADMSA also does not produce any developmental toxicity in mice in the absence of maternal toxicity [107]. Mehta et al. [42] also concluded that MiADMSA had no effect on length of gestation, litter size, sex ratio, viability, and lactation. Taubeneck et al. [108] showed that the developmental toxicity of DMSA is mediated mainly through disturbed copper metabolism and this may also be true for MiADMSA.

26.6.1.3 Mechanism of Action

The pronounced efficacy of MiADMSA could be attributed to its lipophilic characteristic [109]. It appears plausible that MiADMSA could decrease the oxidative stress in tissues either by removing arsenic from the target organs and/or by directly scavenging ROS via its sulfhydryl group [110].

26.6.1.4 Drawbacks

It is reported that the toxicity of DMSA is much lower than the toxicity of MiADMSA but less than that of BAL when comparing their lethal doses (LD_{50}) . Flora and Mehta [100] reported that administration of MiADMSA caused no alterations in the heme synthesis pathway except for a slight rise in zinc protoporphyrin levels, suggesting mild anemia. MiADMSA seems to be slightly more toxic in terms of essential metal loss (copper and zinc) and some biochemical alterations in the hepatic tissue in female compared to male rats [42]. However, these effects were temporary and could be reversed following withdrawal of MiADMSA treatment.

26.6.2 Monomethyl DMSA (MmDMSA) and Monocyclohexyl DMSA (MchDMSA)

Apart from MiADMSA, the other two analogues of DMSA that have been tried against arsenic poisoning are MmDMSA, a straight and branched chain methyl group analogue, and MchDMSA, a cyclic carbon chain analogue. Being lipophilic, both have the ability to penetrate cells more readily than DMSA and can be administered orally. Oral administration suggests that they may possess considerable advantages in the clinical treatment of metal toxicity in the future, however; further extensive studies are required to reach a final conclusion. It is clear from the above that most of the chelating agents have severe side effects and also have some limitations. There are summarized as follows:

- BAL and D-PA BAL are effective antidotes for arsenic poisoning. They are liophilic in nature and distributed both in the extracellular and intracellular spaces in the cellular compartments. However, they are also toxic at doses as low at 40 μmol/kg [111]. D-PA leads to reversible hepatic and renal dysfunction and autoimmune diseases [37].
- 2. DMSA and DMPS: The analogues of BAL are hydrophilic in nature and more effective compared to BAL [51]. These analogues, however, also lead to essential metal loss such as of zinc and copper and extracellular distribution, and are unable to chelate metals from intracellular sites. Various studies have shown that monoesters of DMSA are far better chelating agents than parent compounds for the removal of toxic metals such as lead [112] and arsenic [101,110].
- **3.** DMSA analogues: It has been found that newer DMSA analogues, including MiADMSA, MchDMSA, and MmDMSA, are better alternatives to DMSA as chelating agents for arsenic removal [110]. These complexes seem to penetrate cells, which helps in targeting intracellular sites in the body and aids in the removal of toxic metal ions from the cytosol and organelles inside the cell, which is not possible in the case of DMSA. Their specificity for arsenic, however, needs to be explored.

26.7 Role of Antioxidants in Preventing Arsenic Toxicity

Antioxidants are defined as substances that inhibit or delay the oxidation of biologically relevant molecules either by specifically quenching free radicals or by chelation of redox metals [113]. The most important source of antioxidants is provided by nutrition [114]. Antioxidants play an indispensable role in counteracting free radical-induced damage to macromolecules and also repair free radical-mediated cellular damage. Nutritional antioxidants may act in different compartments through different mechanisms such as by (1) directly neutralizing free radicals; (2) reducing peroxide concentrations and repairing oxide membranes; (3) quenching iron to decrease ROS production via lipid metabolism; or (4) neutralizing ROS through shortchain free fatty acids and cholesteryl esters [114].

Oxidative stress can be partially implicated in arsenic toxicity and therapeutic strategy to increase the antioxidant capacity of cells to fortify the long-term effective treatment in arsenic poisoning. This may be accomplished by either reducing the possibility of metal interacting with critical biomolecules to reduce oxidative damage, or by bolstering the cells' antioxidant defenses through endogenous supplementation of antioxidant molecules [115,116]. Ramanathan et al. [117] observed the molecular changes during arsenic exposure and possible therapeutic efficacy of antioxidants like vitamin C and vitamin E on arsenic-induced apoptosis in rats. They concluded that the administration of vitamin C and vitamin E with arsenic significantly decreased the extent of apoptosis. Antioxidants are known to regulate the expression of a number of genes and signal regulatory pathways and thereby prevent the

incidence of cell death from free radical damage. The usefulness of antioxidants in conjunction with chelation therapy has been well studied but still extensive investigations are required to delineate a specific dose, the duration of administration, and a specific and well-defined mechanism. In the following paragraphs we attempt to describe a few potential antioxidants that have been examined in the recent past for their role during chelation of arsenic.

26.7.1 Alpha-Lipoic Acid

Alpha-lipoic acid (1,2-dithiolane-3-pentanoic acid; ALA), also known as thioctic acid, is an endogenous thiol antioxidant possessing powerful potential to quench reactive oxygen species, regenerate GSH, and chelate toxic metals. It is a disulfide derivative of octanoic acid and an important cofactor (prosthetic group) in various mitochondrial multienzyme complexes involved in energy production [118]. Apart from being an efficient antioxidant, alpha-lipoic acid is also a beneficial therapeutic agent in the treatment of arsenic-induced oxidative stress [119]. It is readily absorbed from the gut and easily crosses the blood-brain barrier. It is supplied through the diet, and it has been reported that exogenous supplementation increases levels of unbound lipoic acid, which acts as a potent antioxidant and reduces oxidative stress both *in vitro* and *in vivo* [119].

Alpha-lipoic acid has been shown to have substantial antioxidant properties when administered (70 mg/kg body weight) for 2 months once daily simultaneously along with arsenic [120]. ALA treatment showed significant beneficial effects in reducing oxidant production as well as lipid peroxidation levels in rat brain regions. The study demonstrated that arsenicinduced deficits in brain regions can be overcome through simultaneous treatment with lipoic acid [120].

26.7.2 N-Acetylcysteine

N-Acetylcysteine (NAC) is a thiol-containing compound, a mucolytic agent, and a precursor of reduced glutathione. It stimulates the intracellular synthesis of GSH, enhances glutathione-S-transferase activity, and acts solely as a scavenger of free radicals. It is a source of sulfhydryl-containing antioxidant that has been used to mitigate conditions of oxidative stress. A number of *in vivo* and *in vitro* studies have been conducted determining the role of NAC in combating arsenic poisoning. Reddy et al. [121] reported that the impairment of spermatogenesis in arsenic-exposed mice is mediated through oxidative stress as well as arsenic's direct effect on the germinal compartment. These authors also reported that co-administration of NAC reduces arsenic-induced oxidative stress by decreasing lipid peroxidation and activating antioxidant enzymes in the testes, thereby ameliorating arsenic-induced suppressed reproduction in male mice. In another study, NAC supplementation could significantly reduce ROS-mediated oxidative stress in liver of arsenic-exposed animals [122]. This hepatoprotective potential could also be a result of effective detoxification of electrophiles or free radicals generated by arsenic as well as its rapid elimination/excretion from the body [123]. NAC treatment could also act as an effective intervention to restore arsenic-depleted GSH stores [123].

Combined administration of NAC and succimer post-arsenic exposure led to a significant recovery in biochemical variables indicative of oxidative stress and arsenic depletion from soft organs [68,124].

In an interesting case report, a 32-year-old man was brought to the emergency department 5½ hours after ingesting a potentially lethal dose (900 mg) of sodium arsenate in a suicide attempt. The patient deteriorated progressively for 27 hours. After intramuscular dimercaprol and supportive measures failed to improve his condition, he was given NAC intravenously. The patient showed remarkable clinical improvement during the following 24 hours and was discharged from the hospital several days later [125].

26.7.3 Vitamins E and C

Vitamin E (α -tocopherol) is a fat-soluble vitamin known as one of the most potent endogenous antioxidants located in the cell membrane. α -Tocopherol comprises a group of potent, lipid soluble, chain-breaking antioxidants that prevents the propagation of free radical reactions and stabilizes cell membranes by interacting with unsaturated fatty acid chains. On the other hand, vitamin C is a water soluble antioxidant occurring in organisms as an ascorbate anion that acts as a scavenger of free radicals and plays an important role in regeneration of α -tocopherol [126]. Combined administration of ascorbic acid and α -tocopherol provides recovery by altering the extent of DNA damage by reducing tumor necrosis factor α (TNF- α) levels and inhibiting the activation of the caspase cascade in arsenic-intoxicated animals [117]. Vitamins E and C have been found to reduce the toxic manifestations of arsenic in animal models [94,116]. It was observed that vitamin E prevented arsenite-induced damage of human fibroblasts [127]. Although some studies suggested that vitamin C was better in providing clinical recoveries, vitamin E was equally efficient in decreasing arsenic burden from the tissues. Wei et al. [128] reported the involvement of oxidative stress in dimethylarsinic acid (DMA)induced bladder toxicity and proliferation in rats, and the inhibitory action of vitamin C on the proliferative effects of DMA.

Coadministration of vitamin C or vitamin E was also investigated during chelation treatment with two thiol chelators, DMSA or MiADMSA, in combating chronic arsenic toxicity. Combined administration of vitamin C and DMSA or vitamin E plus MiADMSA led to a more pronounced depletion of brain arsenic besides significant recovery in the inhibited blood δ -aminolevulinic acid dehydratase (ALAD) activity and other oxidative stress parameters in liver, kidneys, and brain. These results suggested that coadministration of vitamin E or vitamin C may be useful in arsenic poisoning although it has only a limited role in depleting arsenic burden [129]. Supplementation of antioxidants has also led to alleviation of arsenic-induced molecular alterations. Contamination of arsenic in drinking water is associated with several human diseases including cancer. A study was conducted in which significant increase in the levels of protein oxidation, DNA strand breaks, and DNA-protein cross-links was observed in blood, liver, and kidneys of rats exposed to arsenic (100 ppm in drinking water) for 30 days. However, coadministration of ascorbic acid and α -tocopherol to arsenic-exposed rats showed a substantial reduction in the levels of arsenic-induced oxidative products of protein and DNA. The results validate the hypothesis that free radical generation is one of the major mechanisms of arsenic-induced toxic manifestations and also suggest that ascorbic acid and α -tocopherol supplementation can ameliorate the arsenic-induced molecular alterations [130].

26.7.4 β-Carotene

 β -Carotene is a member of a family of molecules known as the carotenoids, consisting of isoprene units in structure. It is a precursor of retinol (vitamin A), a lipid soluble antioxidant with properties of vitamin E [126]. The long chains of conjugated double bonds impart specific colors to carotene, and are responsible for its good antioxidative properties. It can also mop up oxygen-free radicals and dissipate their energy. A significant reverse dose-response relationship with arsenic-related ischemic heart disease was reported for serum levels of α - and β -carotene. Synergistic interaction in arsenic-related ischemic heart disease between duration of consuming artesian well water and low serum carotene levels was observed by multivariate analysis [131].

26.7.5 Taurine

Taurine (2-aminoethanesulfonic acid) is one of the major sulfur-containing free intracellular amino acids synthesized mainly in the pancreas via the cysteine sulfinic pathway. It is also found in various mammalian tissues such as muscles, brain, and heart [132]. It is a well-known, powerful antioxidant and maintains levels of cysteine, an important precursor of glutathione. The sulfonate group in taurine is a strong acid that exists as zwitter ion over the physiological pH range [133]. It has also been shown to exert protective efficacy against arsenic-induced cytotoxicity in murine hepatocytes. The cytoprotective activity of taurine against arsenic poisoning was found to be analogous to that of a known antioxidant, vitamin C. The study further suggests that taurine protects mouse hepatocytes against arsenic-induced cytotoxicity [134]. Taurine effectively reduced arsenic-induced cardiac dysfunction such as cardiomyocyte viability, ROS production, intracellular calcium overload, and apoptotic cell death. Taurine also decreased arsenic-induced activity of p38 and JNK MAPKs. Taurine effectively decreased these apoptotic actions, which is suggestive of its protective role by attenuation of p38 and JNK MAPK signaling pathways. Results showed that taurine effectively prevented arsenic-induced myocardial pathophysiology, and attenuated NF-kappaB activation via IKK, p38, and JNK MAPK signaling pathways, and hence could act as an effective therapy against arsenic-induced cardiovascular burden [135]. Flora et al. [136] also reported that administration of taurine alone significantly reduced hepatic oxidative stress in arsenic-induced toxicity in rats (100 mg/kg).

26.7.6 Melatonin

Melatonin (N-acetyl-5-methoxy tryptamine), a hormone produced by the pineal gland, is a powerful scavenger of reactive oxygen species and free radicals. It prevents the reduction of membrane fluidity from lipid peroxidation and thereby helps in scavenging free radicals [137]. It has been reported that melatonin is superior to other free radical scavengers such as

vitamin E, vitamin C, and GSH in neutralizing peroxyl radicals. Melatonin has shown to be five times superior to glutathione in scavenging free hydroxyl radicals by Pieri et al. [138]. Both the methoxy group at position 5 of the indole nucleus and the acetyl group of the side chain of melatonin are responsible for scavenging free hydroxyl radicals by donating an electron to scavenge OH and forming an indolyl cation radical, which in turn neutralizes superoxide radical [139]. Protective efficacy of melatonin against toxic metals-induced oxidative damage has been reported mostly in *in vivo* and *in vitro* studies [136,140,141]. Pal and Chatterjee [142] reported that melatonin supplementation decreased free radical-mediated cytotoxicity and thereby helps in the restoration of normal cellular antioxidant status in arsenic-exposed rats. The antioxidant property of melatonin has also been proved as a protective factor towards carcinogenesis, neurodegeneration, and aging [143].

26.7.7 Curcumin

Curcumin is a natural phenolic compound with impressive antioxidant properties. It has recently been proven that curcumin exerts its chemopreventive effects partly through the activation of nuclear factor (erythroid-2 related), factor 2 (Nrf2), and its antioxidant and phase II detoxifying enzymes [144]. Protective efficacy of curcumin against arsenic-induced hepatotoxicity and oxidative injuries has been reported. The study revealed that arsenic-induced elevation of serum transaminases and variables indicative of oxidative stress were restored to almost normal levels on treatment with curcumin [145]. The involvement of curcumin in promoting arsenic methylation and urinary elimination in this study was also confirmed. This interesting point further suggested that the potent Nrf2 activation capability might be valuable for the protective effects of curcumin against arsenic intoxication. This affords an effective useful chemopreventive dietary component for human populations [145]. Curcumin has the ability to modulate numerous signaling molecules such as pro-inflammatory cytokines, apoptotic proteins, NF- κ B, cyclooxygenase-2, 5-LOX, STAT3, C-reactive protein, prostaglandin E(2), prostate-specific antigen, adhesion molecules, phosphorylase kinase, transforming growth factor- β , triglyceride, ET-1, creatinine, HO-1, AST, and ALT in human participants [145].

26.7.8 Essential Metals

26.7.8.1 Zinc

Zinc is one of the essential trace elements. Metallothionein (MT), a cysteine-rich protein, stores the cellular form of zinc in the pico- to nanomolar range. Zinc plays a crucial role in detoxification of arsenic from the body because arsenic is capable of inducing MT protein, which is responsible for arsenic detoxification. Some reports suggest that zinc pretreatment may lead to increased arsenic elimination [146,147]. Zinc supplementation also prevents damage that may chronically lead to manifestations such as cancer [148,149]. Except for a few isolated reports not much work has been done on the role of zinc against arsenic [150]. Further studies with variable doses of zinc and its effects during chelation are required against chronic arsenic poisoning.

26.7.8.2 Selenium

An antagonistic effect of selenium on arsenic has been studied in the past. Both are metalloids with very similar chemical properties but known to act differently in biological systems [151]. The possible interaction between arsenic and selenium has been reviewed by Zeng et al. [151]. Selenium is recognized for promoting biliary excretion of exogenous selenium and selenite increases the excretion of arsenic into bile [148,152]. These studies suggested that arsenic augmented the hepatobiliary transport of selenium and facilitated accumulation of selenium in red blood cells, which in turn facilitated the biliary excretion of arsenic. Glattre et al. [153] also reported distribution and interaction of arsenic and selenium in rat thyroid and suggested that both arsenic and selenium accumulate in thyroid tissue. There might be a competition between selenium and arsenic for binding with functional proteins, bioligands, and active tissue sites and the formation of a reversible compound, metal-selenide, which may indicate mechanisms for the reduction of available "free" arsenic ions in the body. Other recent evidence confirmed that selenium may play a role in arsenic elimination by forming a seleniumarsenic conjugate in the liver before excretion into the bile [151]. Recently, a study on adults and children was conducted to assess the relationship between blood selenium and urinary and blood arsenic in a population residing in a moderately arsenic-contaminated rural area in Bangladesh. Results suggested that selenium is inversely associated with biomarkers of arsenic burden in both adults and children. Moreover, this finding validates the hypothesis that selenium facilitates the biliary elimination of arsenic, possibly via the putative formation of an selenium-arsenic conjugate using a glutathione complex [154].

26.7.9 Herbal Extracts

Garlic is another well-known folk remedy for a variety of ailments. Flora et al. [155] studied the protective efficacy of aqueous garlic extract in reducing hepatic injury, tissue oxidative stress, and arsenic mobilization. Results suggested that garlic extracts contain strong antioxidant properties, which could be beneficial in preventing arsenic-induced toxicity in cells. A study was also conducted in rabbits to assess the oxidative injuries caused by arsenic toxicity and evaluate the detoxifying effects of exogenous antioxidants, vitamins, zinc, selenium (VZS), or a plant polyphenol. Results suggested that arsenic induces toxicity in rabbits associated with an increase in lipid peroxidation and nitric oxide production in the body. Administration of exogenous antioxidants such as polyphenols and a recipe of vitamins, zinc, and selenium, however, was found to be useful for arsenic detoxification [156]. Among various herbal extracts, Moringa oleifera seed powder has been shown to reduce significantly arsenic-induced oxidative stress and body arsenic burden. Hence, it was concluded that concomitant administration of M. oleifera seed powder could significantly prevent arsenic-induced toxic manifestations and thus could be used during chelation therapy [157]. Concomitant oral supplementation of Centella asiatica during arsenic exposure has also shown to be an effective strategy against arsenic toxicity. More extensive studies are recommended for determining the effect of coadministration of *C. asiatica* during chelation therapy with a thiol chelator [158,159]. Tossa jute, *Corchorus olitorius*, is a popular crop for arsenicosis-prone populations. A study evaluated the protective

effect of aqueous extract of *C. olitorius* leaves (AECO) against sodium arsenite-induced cardiotoxicity in experimental rats. The results concluded that treatment with AECO prior to arsenic intoxication significantly protected against arsenic-induced myocardial injury [20].

Manna et al. [160] investigated the hepatoprotective role of arjunolic acid, a triterpenoid saponin, against arsenic-induced oxidative damage in murine livers. The altered activities of hepatic marker enzymes in serum enhanced DNA fragmentation, protein carbonyl content, and lipid peroxidation end-products, and the level of oxidized glutathione responded favorably to treatment with arjunolic acid. Results suggest that arjunolic acid possesses the ability to attenuate arsenic-induced oxidative stress in murine liver probably via its antioxidant activity. *Mentha piperita* also exhibited protective efficacy against arsenic-induced liver injury in mice. Pre- and post-treatment of *Mentha* with arsenic significantly protected biochemical parameters of liver damage. The results confirmed that *Mentha* extract may be useful in reducing arsenic-induced hepatopathy [161].

26.8 Newer Strategies

26.8.1 Combination Therapy

A number of newer strategies have recently been tried to reduce the shortcomings of various chelating agents, enhance metal mobilization from the body, and lessen redistribution of toxic metals from one site (e.g., bone or liver) to more sensitive sites such as the brain [97]. In one of the most effective approaches, two chelators are used that act differently. This is based on the assumption that various chelating agents are likely to mobilize toxic metals from different tissue compartments and therefore better results could be expected [161–163]. Various strategies have been discussed in the recent past [164]. Among these strategies, combination therapy is a novel and effective approach to treat cases of metal poisoning including arsenic. As discussed earlier, treatment with MiADMSA has been found to be more effective than using most of the earlier recommended chelators. Further, we observed that combined administration of MiADMSA with another chelator like DMSA is significantly more beneficial than monotherapy with these chelators in counteracting chronic arsenic toxicity [109,162]. Results from these studies suggested that concomitant administration of DMSA, a chelator known for its extracellular distribution with lipophilic chelators like MiADMSA, plays a significant and more efficient role in abating the number of toxic effects of arsenic in animals compared to treatment with these chelators alone. We also suggested that analogues having long carbon chains (MiADMSA and MchDMSA) are better chelators than chelators with shorter carbon chains (MmDMSA) or DMSA. It is implicit that analogues of DMSA eliminate arsenic simultaneously from the cell and provide assistance in restoring GSH homeostasis towards normal levels. We also noted that combination therapy with DMSA and MiADMSA or MchDMSA proved more beneficial than combined treatment with MmDMSA and DMSA [110]. In an interesting study, we recently reported that combination therapy with two thiol chelators could reverse altered neurological functioning better than monotherapy. More significantly, neuronal cell death could also be prevented to some extent following combination therapy. Our studies show combination therapy to be effective in providing clinical benefits and removing arsenic from the brain. However, we feel that more extensive studies are required before final conclusions can be drawn for clinical viability and effectiveness of combination treatment in chronic arsenic poisoning [105].

Coadministration of naturally occurring vitamins like vitamin E or vitamin C during administration of a thiol chelator like DMSA or MiADMSA may also be beneficial in the restoration of altered biochemical variables, although they have only a limited role in depleting arsenic burden. Various other studies, particularly from our group, have also suggested the beneficial effects of combination therapy using a strong chelating agent and a vitamin/antioxidant/ essential metal or an amino acid also acting like weak chelating agents [129,165,166]. The ability of NAC to significantly provide additional beneficial effects (including arsenic mobilization) when administered with a thiol chelating agent, like DMSA, to restore hepatic and brain glutathione levels and erythrocyte enzyme to normal levels is one such example [68]. Combined administration of vitamin C with DMSA and vitamin E with MiADMSA was also found to be more effective in significant depletion of brain arsenic and useful in the restoration of altered biochemical variables, particularly the effects on heme biosynthesis and oxidative injury [129]. The effects of both DL-alpha-lipoic acid and DMSA on the antioxidants and lipid peroxidation in arsenic-treated rats revealed a significantly more pronounced depletion of reactive oxygen species formation and lipid peroxidation while antioxidant enzyme levels were restored to normal. The study further concluded that DL-alpha-lipoic acid and DMSA play a synergistic role in decreasing arsenic-induced oxidative damage by elevating the antioxidant status in liver and kidneys [167]. Taurine too when coadministered with DMSA or MiADMSA reduces total body burden of arsenic and lead [137,168].

It is evident that combination therapy is a new and better approach to treat cases of metal poisoning. Little experimental evidence is available and there is a need for in-depth investigation in this area to prompt new chelation therapy regimens for the treatment of chronic arsenic poisoning.

26.8.2 Nanoparticle Carriers to Combat Arsenic Toxicity

Focus has recently shifted from the monotherapy and combination therapy to site-specific targeted delivery of drugs using appropriate carriers such as liposome, micelles, and nanoparticles. Novel modes of drug delivery are currently available as nanocapsules, which are non-toxic, biodegradable, and non-immunogenic with sustained drug-releasing ability in biological systems. The therapeutic efficacy of drugs can be enhanced via targeting nanomaterials as carriers. Due to their small size, nanoparticles are easily accessible in the body and can be transported to different sites through the blood [169]. Nanoparticles can thus circumvent the reticuloendothelial system (RES) such as liver Kupffer cells and spleen macrophages and reach the desired site effectively [170]. This is a promising therapeutic strategy and may be effective in chelating arsenic from the intracellular space by their crossing the blood-brain barrier to adsorb apolipoprotein E from blood plasma. They may be able to mimic low density lipoprotein for having specific receptors at the surface of endothelial cells of the blood-brain barrier. They are recognized by endothelial cell receptors. The drug may be released into these cells from the nanoparticles and diffuse into the brain. There are other processes such as tight junction modulation or P-glycoprotein inhibition, which may run in parallel or may be cooperative, thus enabling efficient drug delivery to the brain [171] (Figure 26–7).

We recently synthesized and characterized water soluble nanoparticles of curcumin and applied them as a stable detoxifying agent for arsenic poisoning. For this, chitosan nanoparticles of less than 50 nm in diameter containing curcumin were prepared. The therapeutic efficacy of the encapsulated curcumin nanoparticles (ECNPs) against arsenic-induced toxicity in rats was investigated via an oral administration. Coadministration of ECNPs along with arsenic provided pronounced beneficial effects on the adverse changes in oxidative stress parameters induced by arsenic. Results indicate that ECNPs have a better antioxidant and chelating potential compared to free curcumin. Significant neuroprotective efficacy was also observed by ECNPs against neurochemical and immunohistochemical variables. This formulation of ECNPs provides a novel therapeutic regime for preventing arsenic toxicity [172]. Ghosh et al. [136] too evaluated the therapeutic efficacy of nanocapsulated flavonoidal quercetin (QC) in combating arsenic-induced reactive oxygen species-mediated oxidative damage in rat hepatocytes and brain cells. Administration of QC provided recovery in reducing arsenic-induced oxidative damage. The authors, however, suggested that the clinical application of QC against toxicant-induced cellular damage is not possible due to its insoluble nature. A few other studies also reported the efficacy of liposome-encapsulated QC, administered intravenously to arsenic-exposed rats. These studies showed that the QC inhibited hepatic fibrogenesis and



FIGURE 26-7 Mechanism of encapsulated MiADMSA in chelating arsenic from brain.

also attenuated the oxidative damage induced by ischemia-repurfusion injury in the rat brain [173,174]. Stability of the nanoparticles in the bioenvironment offers the possibility of using these vesicles as oral drug carriers. The use of nanoparticles as drug carriers has high stability in biological systems, high carrier capacity, and the feasibility of incorporation of both hydrophilic and hydrophobic to circumvent the blood-brain barrier. This colloidal carrier has the ability to cross both the GI tract mucosal barrier and the blood-brain barrier to enhance drug bioavailability via particle uptake mechanisms. Thus, nanoparticles are proven to be the most promising carriers for delivery of drug to its target site [175].

Therapeutic efficacy of nanoencapsulation of DMSA monoester against arsenic toxicity was recently reported by us [176]. Sodium arsenite-exposed mice were treated with nanoencapsulated DMSA for 5 days to evaluate alterations in blood, brain, kidney, and liver oxidative stress variables. This study also investigated the histopathological changes in tissues and the chelating potential of the nanoformulation. The study revealed a significant role of encapsulated DMSA in reducing arsenic from the tissues while eliciting positive response in altered biochemical variables, histopathological lesions, and urinary 8-OHdG levels, suggesting a better therapeutic efficacy of the novel formulation for arsenic toxicity [176,177]. Thus, we recommend nanoparticle-based drug delivery as a better approach to aid removal of arsenic from specific target sites of the biological system pending future studies (Figure 26–8).



FIGURE 26-8 Various strategies of arsenic chelation.

26.9 Concluding Remarks and Future Directions

There is a growing concern these days to find a suitable, safe, and effective therapy for chronic arsenic poisoning. Various strategies that have been proposed in the recent past include synthesis of newer chelating agents, target drug delivery using nanoencapsulation of drugs, and combinational treatment, which ensures and improves trans-membrane delivery of soluble drugs to the affected cells. Nanoencapsulation of antioxidants and chelating agents ensures increases in the half-lives of these drugs in blood circulation, reducing multiple dosing and minimizing side effects. As there are very few studies available on these aspects, we recommend more detailed investigation using these strategies for treating chronic and subchronic levels of arsenic to substantiate our conclusions.

References

- Kwok KCW, Koong LF, Chen G, McKay G. Mechanism of arsenic removal using chitosan and nanochitosan. J Colloid Interface Sci 2014;416:1–10.
- [2] European Commission. 98/83/EC, Drinking Water Directive; Brussels, 1998.
- [3] US Environment Protection Agency (US EPA). Arsenic in drinking water rule economic analysis, EPA-815-R-00-026; 2000.
- [4] World Health Organization (WHO). Guidelines for drinking-water quality, 2nd ed. Health criteria and other supporting information, vol. 2 Geneva: WHO; 1996. p. 940–9.
- [5] Flora SJS, Pachauri V. Chelation in metal intoxication. Int J Environ Res Public Health 2010;7:2745–88.
- [6] Blanusa M, Varnai VM, Piasek M, Kostial K. Chelators as antidotes of metal toxicity: therapeutic and experimental aspects. Curr Med Chem 2005;12:2771–94.
- [7] Mandal BK, Suzuki KT. Arsenic round the world: a review. Talanta 2002;58:201–35.
- [8] Ferrario D, Croera C, Brustio R, Collotta A, Bowe G, Vahter M, et al. Toxicity of inorganic arsenic and its metabolites on haematopoietic progenitors "in vitro": comparison between species and sexes. Toxicology 2008;249:102–8.
- [9] Flora SJS, Dubey R, Kannan GM, Chauhan RS, Pant BP, Jaiswal DK. Meso-2,3-dimercaptosuccinic acid (DMSA) and monoisoamyl DMSA effect on gallium arsenide induced pathological liver injury in rats. Toxicol Lett 2002;132:9–17.
- [10] Hughes MF. Arsenic toxicity and potential mechanism of action. Toxicol Lett 2002;133:1–16.
- [11] Steinmaus C, Carrigan K, Kalman D, Atallah R, Yuan Y, Smith AH. Dietary intake and arsenic methylation in a US population. Environ Health Perspect 2005;113:1153–9.
- [12] National Research Council (NRC). Arsenic in drinking water. Washington, DC: Natl Acad Press; 2001. Update.
- [13] Flora SJS. Arsenic-induced oxidative stress and its reversibility. Free Rad Biol Med 2011;51:257–81.
- [14] Valko M, Morris H, Cronin MT. Metals, toxicity and oxidative stress. Curr Med Chem 2005;12:1161–208.
- [15] Abernathy CO, Liu YP, Longfellow D, Aposhian HV, Beck B, Fowler A, et al. Arsenic: health effects, mechanisms of actions and research issues. Environ Health Perspect 1999;107:593–7.
- [16] Ibrahim D, Froberg B, Wolf A, Rusyniak DE. Heavy metal poisoning: clinical presentations and pathophysiology. J Clin Lab Med 2006;26:67–97.
- [17] Winski SL, Carter DE. Arsenate toxicity in human erythrocytes: characterization of morphologic changes and determination of the mechanism of damage. J Toxicol Environ Health 1998;53:345–55.

- [18] Roy A, Manna P, Sil PC. Prophylactic role of taurine on arsenic mediated oxidative renal dysfunction via MAPKs/NF-κB and mitochondria dependent pathways. Free Radic Res 2009;43:995–1007.
- [19] Waalkes MP, Liu J, Ward JM, Diwan BA. Animal models for arsenic carcinogenesis: inorganic arsenic is a transplacental carcinogen in mice. Toxicol Appl Pharmacol 2004;198:377–84.
- [20] Das AK, Sahu R, Dua TK, Bag S, Gangopadhyay M, Sinha MK, et al. Arsenic-induced myocardial injury: protective role of *Corchorus olitorius* leaves. Food Chem Toxicol 2010;48:1210–17.
- [21] Tseng CH. Cardiovascular disease in arsenic-exposed subjects living in the arseniasis-hyperendemic areas in Taiwan. Atherosclerosis 2008;199:8–12.
- [22] Wang SX, Wang ZH, Cheng XT, Li J, Sang ZP, Zhang XD, et al. Arsenic and fluoride exposure in drinking water: children's IQ and growth in Shanyin County, Shanxi Province, China. Environ Health Perspect 2007;115:643–7.
- [23] Buchanan JA, Eberhardt A, Tebb ZD, Heard K, Wendlandt RF, Kosnett MJ. Massive human ingestion of orpiment (arsenic trisulphide). J Emerg Med 2013;44:367–72.
- [24] Andersen O, Aaseth J. Molecular mechanisms of in vivo metal chelation: implications for clinical treatment of metal intoxications. Environ Health Perspect 2002;110(Suppl. 5):887–90.
- [25] Flora SJS. Chelation therapy. In: Reedijk J, Poeppelmeier K, editors. Comprehensive inorganic chemistry II, vol. 3. Oxford: Oxford, Elsevier; 2013. p. 987–1013.
- [26] Andersen O. Principles and recent developments in chelation treatment of metal intoxication. Chem Rev 1999;99:2683–710.
- [27] Andersen O. Chemical and biological considerations in the treatment of metal intoxications by chelating agents. Mini Rev Med Chem 2004;4:159–65.
- [28] Franz KJ. Clawing back: broadening the notion of metal chelators in medicine. Curr Opin Chem Biol 2013. 17:143–9.
- [29] Sears ME. Chelation: harnessing and enhancing heavy metal detoxification—a review. Sci World J 2013. Article ID 219840, 13 pages.
- [30] Andersen O. Choice of chelating antidotes for acute cadmium intoxication. Toxicol Environ Chem 1989;23:105–20.
- [31] Andersen O. Oral cadmium exposure in mice: toxicokinetics and efficiency of chelating agents. Crit Rev Toxicol 1989;20:83–112.
- [32] Taylor DM, Williams DR. Trace element medicine and chelation therapy. Cambridge: The Royal Society of Chemistry; 1995.
- [33] Kosnett MJ. The role of chelation in the treatment of arsenic and mercury poisoning. J Med Toxicol 2013;9:347-54.
- [34] Aposhian HA, Maiorino RM, Gonzalez-Ramirez D, Zuniga CM, Xu Z, Hurlbut KM, et al. Mobilization of heavy metals by newer, therapeutically useful chelating agents. Toxicology 1995;97:23–38.
- [35] Gilman A. Goodman and Gilman's the pharmacological basis of therapeutics Chapter 67. Environmental Toxicology: Carcinogens & Heavy Metals. In: Gilman A, Rall T, Nies A, editors (12th ed). New York, NY: Pergamon Press; 2011, p. 2047–73.
- [36] Kavanagh P, Farago ME, Thronton I, Goessler W, Kuehnelt D, Schlagenhaufen C, et al. Urinary arsenic species in Devon and Cornwall residents, UK. A pilot study. Analyst 1998;123:27–9.
- [37] Flanagan R, Jones A, Maynard RL. Antidotes: principles and clinical applications. London and New York: Taylor & Francis; 2001.
- [38] Drugdex Drug Evaluations. Thomson MICROMEDEX Healthcare Series (monograph on CD-ROM), Micromedex Inc., Greenwood Village, CO, USA 122; 2004.
- [39] Graziano JH, Lolacono NJ, Moulton T, Mitchell ME, Slavkovich V, Zarate C. Controlled study of meso-2,3-dimercaptosuccinic acid for the management of childhood lead intoxication. J Pediatr 1992;120(1):133–9.
- [40] Domingo JL. Developmental toxicity of metal chelating agents. Reprod Toxicol 1998;12:499–510.
- [41] Flora SJS, Bhadauria S, Pachauri V, Yadav A. Monoisoamyl 2,3-dimercaptosuccinic acid (MiADMSA) demonstrates higher efficacy by oral route in reversing arsenic toxicity: a pharmacokinetic approach. Basic Clin Pharmacol Toxicol 2012;110:449–59.
- [42] Mehta A, Pant SC, Flora SJS. Monoisoamyl dimercaptosuccinic acid induced changes in pregnant female rats during late gestation and lactation. Reproduct Toxicol 2006;21:94–103.
- [43] International Programme on Chemical Safety (IPCS) INCHEM. Dimercaprol. Canadian Centre for Occupational Health and Safety and International Programme on Chemical Safety; Hamilton, Ontario, Canada 1990.
- [44] Molina JA, Calandre L, Bermejo F. Myoclonic encephalopathy due to bismuth salts: treatment with dimercaprol and analysis of CSF transmitters. Acta Neurol Scand 1989;79:200–3.
- [45] Takeda T, Yukioka T, Shimazaki S. Cupric sulfate intoxication with rhabdomyolysis, treated with chelating agents and blood purification. Inter Med 2000;39:253–5.
- [46] Jantsch W, Kulig K, Rumack BH. Massive copper sulfate ingestion resulting in hepatotoxicity. J Toxicol Clin Toxicol 1984–85;22:585–8.
- [47] Ellis EN, Brouhard BH, Lynch RE, Dawson EB, Tisdell R, Nichols MM, et al. Effects of hemodialysis and dimercaprol in acute dichromate poisoning. J Toxicol Clin Toxicol 1982;19:249–58.
- [48] Cantilena LR, Klaassen CD. Comparison of the effectiveness of several chelators after single administration on the toxicity, excretion, and distribution of cadmium. Toxicol Appl Pharmacol 1981;58:452–60.
- [49] Singh PK, Xu C, Jones MM, Kostial K, Blanusa M. Molecular-modelling design of cadmium-mobilizing agents: a novel biscarbodithioate. Chem Res Toxicol 1994;7:614–20.
- [50] Snider TH, Wientjes MG, Joiner RL, Fisher GL. Arsenic distribution in rabbits after Lewisite administration and treatment with British anti-Lewisite (BAL). Fundam Appl Toxicol 1990;14:262–72.
- [51] Mückter H, Liebl B, Reichl FX, Hunder G, Walther U, Fichtl B. Are we ready to replace dimercaprol (BAL) as an arsenic antidote? Hum Exp Toxicol 1997;16:460–5.
- [52] Vilensky JA, Redman K. British anti-Lewisite (dimercaprol): an amazing history. Ann Emerg Med 2003;41:378-83.
- [53] Flora SJS, Tripathi N. Treatment of arsenic poisoning: an update. Ind J Pharmacol 1998;30:209–17.
- [54] Aposhian HV, Zakharyan RA, Wildfang E, Healy M, Radabaugh TAR, Bogdan GM, et al. How is inorganic arsenic detoxified? In: Proceedings of the San Diego Arsenic Symposium. San Diego, CA; 1999.
- [55] Jensen GE, Christensen JM, Poulsen OM. Occupational and environmental exposure to arsenic increased urinary arsenic level in children. Sci Total Environ 1991;107:169–77.
- [56] Johansson G, Akesson A, Berglund M, Nermell B, Vahter M. Validation with biological markers for food intake of a dietary assessment method used by Swedish women with three different dietary preferences. Public Health Nutr 1998;1:199–206.
- [57] Kagey BT, Bumgarner JE, Creason JP. Arsenic levels in maternal fetal tissue sets. In: Hemphill DD, editor. Trace substances in environmental health—XI. Proceedings of the University of Missouri's 11th Annual Conference on Trace Substances in Environmental Health. Columbia, Mo: University of Missouri Press; 1977. p. 252–6.
- [58] Kalman DA, Hughes J, Van Belle G, Burbacher T, Bolgiano D, Coble K, et al. The effect of variable environmental arsenic contamination on urinary concentrations of arsenic species. Environ Health Perspect 1990;89:145–52.
- [59] Watson WA, Veltri JC, Metcalf TJ. Acute arsenic exposure treated with oral D-penicillamine. Vet Human Toxicol 1981;23:164–6.

- [60] Kreppel H, Reichl FX, Forth W, Fichtl B. Lack of effectiveness of D-penicillamine in experimental arsenic poisoning. Vet Hum Toxicol 1989;31:1–5.
- [61] Kilburn KH. Neurobehavioral impairment from long-term residential arsenic exposure. In: Abernathy CO, Calderon RL, Chappell WR, editors. Arsenic exposure and health effects, vol. 14. London: Chapman & Hall; 1997. p. 159–75.
- [62] Ding GS, Liang YY. Antidotal effects of dimercaptosuccinic acid. J Appl Toxicol 1991;11:7-14.
- [63] Rivera M, Zheng W, Aposhian HV, Fernando Q. Determination and metabolism of dithiol chelating agents. VIII. Metal complexes of meso-dimercaptosuccinic acid. Toxicol Appl Pharmacol 1989;100:96–106.
- [64] Aposhian HV, Aposhian MM. Meso-2,3-dimercaptosuccinic acid: chemical, pharmacological and toxicological properties of an orally effective metal chelating agent. Ann Rev Pharmacol Toxicol 1990;30:279–306.
- [65] Smith D, Strupp BJ. The scientific basis for chelation: animal studies and lead chelation. J Med Toxicol 2013;9:326–38.
- [66] Kalra V, Dua T, Kumar V, Kaul B. Succimer in symptomatic lead poisoning. Indian Pediatr 2002;39:580-5.
- [67] Counter SA, Ortega F, Shannon MW, Buchanan LH. Succimer (meso-2,3-dimercaptosuccinic acid (DMSA)) treatment of Andean children with environmental lead exposure. Int J Occup Environ Health 2003;9:164–8.
- [68] Flora SJS. Arsenic induced oxidative stress and its reversibility following combined administration of N-acetylcysteine and meso-2,3-dimercaptosuccinic acid in rats. Clin Exp Pharmacol Physiol 1999;26:865–9.
- [69] Prester L, Restek-Samaržija N, Blanuša M, Piasek M, Kostial K, Jones MM, et al., editors. In: Proceedings of the Third Symposium of the Croatian Radiation Protection Association. Zagreb: CRPA;1996, p. 155–60.
- [70] Aposhian HV, Maiorino RM, Rivera M, Bruce DC, Dart RC, Hurlbut KM, et al. Human studies with the chelating agents, DMPS and DMSA. J Toxicol Clin Toxicol 1992;30:505–28.
- [71] Dart RC, Hurlbut KM, Maiorino RM, Mayersohn M, Aposhian HV, Hassen LV. Pharmacokinetics of meso-2,3-dimercaptosuccinic acid in patients with lead poisoning and in healthy adults. J Pediatr 1994;125:309–16.
- [72] Aposhian HV. DMSA and DMPS-water soluble antidotes for heavy metal poisoning. Ann Rev Pharmacol Toxicol 1983;23:193–215.
- [73] Guha Mazumder DN, Ghoshal UC, Saha J, Santra A, De BK, Chatterjee A, et al. Randomized placebo-controlled trial of 2,3-dimercaptosuccinic acid in therapy of chronic arsenicosis due to drinking arsenic-contaminated subsoil water. J Toxicol Clin Toxicol 1998;36:683–90.
- [74] Gerr F, Frumkin H, Hodgins P. Hemolytic anemia following succimer administration in a glucose-6-phosphate dehydrogenase deficient patient. J Toxicol Clin Toxicol 1994;32:569–75.
- [75] Islinger F, Gekle M, Wright SH. Interaction of 2,3-dimercapto-1-propane sulfonate with the human organic anion transporter hOAT1. J Pharmacol Exp Ther 2001;299:741–7.
- [76] Hurlbut KM, Maiorino RM, Mayersohn M, Dart RC, Bruce DC, Aposhian HV. Determination and metabolism of dithiol chelating agents. XVI. Pharmacokinetics of 2,3-dimercapto-1-propanesulphonate after intravenous administration to human volunteers. J Pharmacol Exp Ther 1994;268:662–8.
- [77] Mazumder G, De DN, Santra BK, Ghosh A, Das N, Lahiri S, et al. Randomized placebo-controlled trial of 2,3-dimercapto-1-propanesulfonate (DMPS) in therapy of chronic arsenicosis due to drinking arseniccontaminated water. J Toxicol Clin Toxicol 2001;39:665–74.
- [78] Maiorino RM, Dart RC, Carter DE, Aposhian HV. Determination and metabolism of dithiol chelating agents. XII. Metabolism and pharmacokinetics of sodium 2,3-dimercaptopropane-1-sulfonate in humans. J Pharmacol Exp Therap 1991;259:808–14.
- [79] Zheng W, Maiorino RM, Brendel K, Aposhian HV. Determination and metabolism of dithiol chelating agents. VII. Biliary excretion of dithiols and their interactions with cadmium and metallothionein. Fund Appl Toxicol 1990;14:598–607.

- [80] Hruby K, Donner A. 2,3-dimercapto-1-propane sulfonate in heavy metal poisoning. Med Toxicol 1989;2:317–23.
- [81] Suzuki S, Arnolda LL, Penningtona KL, Kakiuchi-Kiyotaa S, Chenb B, Lub X, et al. Effects of co-administration of dietary sodium arsenate and 2,3-dimercaptopropane-1-sulfonic acid (DMPS) on the rat bladder epithelium. Toxicology 2012;299:155-9.
- [82] Domingo JL, Bosque MA, Piera V. Meso-2,3-dimercaptosuccinic acid and prevention of arsenite embryotoxicity and teratogenicity in the mouse. Fund Appl Toxicol 1991;17:314–20.
- [83] Domingo JL, Bosque MA, Llobet JM, Corbella J. Amelioration by BAL (2,3-dimercapto-l-propanol) and DMPS (sodium 2,3-dimercapto-l-propanesulfonic acid) of arsenite developmental toxicity in mice. Ecotoxicol Environ Safety 1992;23:274–81.
- [84] Bosque MA, Domingo JL, Llobet JM, Corbella J. Effects of meso-2,3-dimercaptosuccinic acid (DMSA) on the teratogenicity of sodium arsenate in mice. Bull Environ Contam Toxicol 1991;47:682–8.
- [85] Jorgensen FM. Succimer: the first approved oral lead chelator. Am Fam Physician 1993;48:1496–502.
- [86] Aposhian HV. Enzymatic methylation of arsenic species and other new approaches to arsenic toxicity. Annu Rev Pharmacol Toxicol 1997;37:397–419.
- [87] Aposhian HV, Arroyo A, Cebrian ME, Del Razo LM, Hurlbut KM, Dart RC, et al. DMPS-arsenic challenge test. I. Increased urinary excretion of monomethylarsonic acid in humans given dimercaptopropane sulfonate. J Pharmacol Exp Ther 1997;282:192–200.
- [88] Aposhian HV, Zheng B, Aposhian MM, Le XC, Cebrian ME, Cullen W, et al. DMPS-arsenic challenge test. II. Modulation of arsenic species, including monomethylarsonous acid (MMAIII), excreted in human urine. Toxicol Appl Pharmacol 2000;165:74–83.
- [89] Mazumder GDN. Chronic arsenic toxicity: clinical features, epidemiology, and treatment: experience in West Bengal. J Environ Sci Health A Tox Hazard Subst Environ Eng 2003;38:141–63.
- [90] Heitland P, Köster HD. Comparison of different medical cases in urinary arsenic speciation by fast HPLC-ICP-MS. Int J Hyg Environ Health 2009;212:432–8.
- [91] Szinicz L, Wiedemann P, Haring H, Weger N. Effects of repeated treatment with sodium 2,3-dimercaptopropane-1-sulfonate in beagle dogs. Arzneimittelforschung 1983;33:818–21.
- [92] Xu C, Holscher MA, Jones MM, Singh PK. Effect of monoisoamyl meso-2,3-mercaptosuccinate on the pathology of acute cadmium intoxication. J Toxicol Environ Health 1995;45:261–77.
- [93] Gale GR, Smith AB, Jones MM, Singh PK. Meso-2,3-dimercaptosuccinic acid monoalkyl esters: effects on mercury levels in mice. Toxicology 1993;81:49–56.
- [94] Flora SJS, Kannan GM, Pant BP, Jaiswal DK. Combined administration of oxalic acid, succimer and its analogue in the reversal of gallium arsenide induced oxidative stress in rats. Arch Toxicol 2002;76:269–76.
- [95] Flora SJS, Pande M, Bhadauria S, Kannan GM. Combined administration of taurine and meso 2,3-dimercaptosuccinic acid in the treatment of chronic lead intoxication in rats. Hum Exp Toxicol 2004;23:157–66.
- [96] Flora SJS, Kumar P. Biochemical and immunological evaluation of metal chelating drugs in rats. Drug Invest 1993;5:269–73.
- [97] Mehta A, Flora SJS. Possible role of metal redistribution, hepatotoxicity and oxidative stress in chelating agents induced hepatic and renal metallothionein in rats. Food Chem Toxicol 2001;39:1029–38.
- [98] Flora SJS, Mehta A, Gautam P, Jatav PC, Pathak U. Essential metal status, prooxidant/antioxidant effects of MiADMSA in male rats: age-related effects. Biol Trace Elem Res 2007;120:235–47.
- [99] Mehta A, Kannan GM, Dube SN, Pant BP, Pant SC, Flora SJS. Hematological, hepatic and renal alterations after repeated oral or intraperitoneal administration of monoisoamyl DMSA I. Changes in male rats. J Appl Toxicol 2002;22:359–69.

- [100] Flora SJS, Mehta A. Haematological, hepatic and renal alternations after repeated oral and intraperitoneal administration of monoisoamyl DMSA. Changes in female rats. J Appl Toxicol 2003;23:97–102.
- [101] Kreppel H, Reichl FX, Klein A, Szincz L, Singh PK, Jones MM. Antidotal efficacy of newly synthesized dimercaptosuccinic acid (DMSA) monoesters in experimental poisoning in mice. Fund Appl Toxicol 1995;26:239–45.
- [102] Jones MM, Singh PK, Gale GR, Smith AB, Atkins LM. Cadmium mobilization in vivo by intraperitoneal or oral administration of monoalkyl esters of meso 2,3-dimercaptosuccinic acid. Pharmacol Toxicol 1992;70:336-43.
- [103] Dwivedi N, Mehta A, Yadav A, Binukumar BK, Gill KD, Flora SJ. MiADMSA reverses impaired mitochondrial energy metabolism and neuronal apoptotic cell death after arsenic exposure in rats. Toxicol Appl Pharmacol 2011;256:241–8.
- [104] Ramkumar M, Flora SJ, Reddy GR. Monoisoamyl 2,3-dimercaptosuccinic acid attenuates arsenic induced toxicity: behavioural and neurochemical approach. Environ Toxicol Pharmacol 2013;36:231–42.
- [105] Pachauri V, Mehta A, Mishra D, Flora SJS. Arsenic induced neuronal apoptosis in guinea pigs is Ca2+ dependent and abrogated by chelation therapy: role of voltage gated calcium channels. Neurotoxicology 2013;35:137–45.
- [106] Sannadi S, Kadeyalaa PK, Reddy Gottipolub R. Reversal effect of monoisoamyl dimercaptosuccinic acid (MiADMSA) for arsenic and lead induced perturbations in apoptosis and antioxidant enzymes in developing rat brain. Int J Dev Neurosci 2013;31:586–97.
- [107] Blanusa M, Prester L, Piasek M, Kostial K, Jones MM, Singh PK. Monoisoamyl ester of DMSA reduces 203Hg(NO3)2 retention in rats: 1. Chelation therapy during pregnancy. J Trace Elem Exp Med 1997;10:173–81.
- [108] Taubeneck MW, Domingo JL, Llobet JM, Keen CL. Meso 2,3-dimercaptosuccinic acid (DMSA) affects maternal and fetal copper metabolism in Swiss mice. Toxicology 1992;72:27–40.
- [109] Flora SJS, Bhadauria S, Pant SC, Dhaked RK. Arsenic induced blood and brain oxidative stress and its response to some thiol chelators in rats. Life Sci 2005;77:2324–37.
- [110] Mishra D, Mehta A, Flora SJS. Reversal of hepatic apoptosis with combined administration of DMSA and its analogues in guinea pigs: role of glutathione and linked enzymes. Chem Res Toxicol 2008;21:400–8.
- [111] Inns RH, Rice P. Efficacy of dimercapto chelating agents for the treatment of poisoning by percutaneously applied dichloro(2-chlorovinyl)arsine in rabbits. Hum Exp Toxicol 1993;12:241–6.
- [112] Walker EM, Stone A, Milligan LB, Gale GR, Atkins LM, Smith AB, et al. Mobilization of lead in mice by administration of monoalkyl esters of meso-2,3-dimercaptosuccinic acid. Toxicology 1992;76:79–87.
- [113] Valko M, Rhodes CJ, Moncol J, Izakovic M, Mazur M. Free radicals, metals and antioxidants in oxidative stress-induced cancer. Chem Biol Int 2006;160:1–40.
- [114] Flora SJS. Nutritional components modify metal absorption, toxic response and chelation therapy. J Nutr Environ Med 2002;12:51–65.
- [115] Flora SJS, Saxena G, Mehta A. Reversal of lead-induced neuronal apoptosis by chelation treatment in rats: role of reactive oxygen species and intracellular Ca²⁺. J Pharmacol Exp Ther 2007;322:108–16.
- [116] Flora SJS, Shrivastava R, Mittal M. Chemistry and pharmacological properties of some natural and synthetic antioxidants for heavy metal toxicity. Curr Med Chem 2013;20:4540–74.
- [117] Ramanathan K, Anusuyadevi M, Shila S, Panneerselvam C. Ascorbic acid and alpha-tocopherol as potent modulators of apoptosis on arsenic induced toxicity in rats. Toxicol Lett 2005;156:297–306.
- [118] Gurer H, Ozgunes H, Oztezcan S, Ercal N. Antioxidant role of alpha-lipoic acid in lead toxicity. Free Radic Biol Med 1999;27:75–81.

- [119] Bustamante J, Lodge JK, Marcocci L, Tritschler HJ, Packer L, Bertrand H. Alpha-lipoic acid in liver metabolism and disease. Free Radic Biol Med 1998;24:1023–39.
- [120] Shila S, Kokilavani V, Subathra M, Panneerselvam C. Brain regional responses in antioxidant system to alpha-lipoic acid in arsenic intoxicated rat. Toxicology 2005;210:25–36.
- [121] Reddy PS, Rani GP, Sainath SB, Meena R, Supriya Ch. Protective effects of N-acetylcysteine against arsenic-induced oxidative stress and reprotoxicity in male mice. J Trace Elem Med Biol 2011;25:247–53.
- [122] Hemalatha P, Reddy AG, Reddy YR, Shivakumar P. Evaluation of protective effect of N-acetyl cysteine on arsenic-induced hepatotoxicity. J Nat Sci Biol Med 2013;4:393–5.
- [123] Santra A, Chowdhury A, Ghatak S, Biswas A, Dhali GK. Arsenic induces apoptosis in mouse liver is mitochondria dependent and is abrogated by N-acetylcysteine. Toxicol Appl Pharmacol 2007;220:146–55.
- [124] Kannan GM, Flora SJS. Combined administration of N-acetylcysteine (NAC) and monoisoamyl DMSA on tissue oxidative stress during arsenic chelation therapy. Biol Trace Elem Res 2006;110:43–60.
- [125] Martin DS, Willis SE, Cline DM. N-acetylcysteine in the treatment of human arsenic poisoning. J Am Board Fam Pract 1990;3:293–6.
- [126] Young IS, Woodside IS. Antioxidants in health and disease. J Clin Pathol 2001;54:176-86.
- [127] Lee TC, Ho IC. Differential cytotoxic effects of arsenic on human and animals cell. Environ Health Perspect 1994;102(Suppl. 3):101.
- [128] Wei M, Arnold L, Cano M, Cohen SM. Effects of co-administration of antioxidants and arsenicals on the rat urinary bladder epithelium. Toxicol Sci 2005;83:237–45.
- [129] Kannan GM, Flora SJS. Chronic arsenic poisoning in rat: treatment with combined administration of succimers and an antioxidant. Ecotoxicol Environ Safety 2004;58:37–43.
- [130] Kadirvel R, Sundaram K, Mani S, Samuel S, Elango N, Panneerselvam C. Supplementation of ascorbic acid and alpha-tocopherol prevents arsenic-induced protein oxidation and DNA damage induced by arsenic in rats. Hum Exp Toxicol 2007;26:939–46.
- [131] Hsueh YM, Wu WL, Huang YL, Chiou HY, Tseng CH, Chen CJ. Low serum carotene level and increased risk of ischemic heart disease related to long-term arsenic exposure. Atherosclerosis 1998;141:249–57.
- [132] Ghosh J, Das J, Manna P, Sil PC. Taurine prevents arsenic-induced cardiac oxidative stress and apoptotic damage: role of NF-kappaB, p38 and JNK MAPK pathway. Toxicol Appl Pharmacol 2009;240:73–87.
- [133] Huxtable RJ. Physiological actions of taurine. Physiol Rev 1992;72:101-63.
- [134] Sinha M, Manna P, Sil PC. Taurine, a conditionally essential amino acid, ameliorates arsenic-induced cytotoxicity in murine hepatocytes. Toxicol In Vitro 2007;21:1419–28.
- [135] Ghosh A, Mandal AK, Sarkar S, Panda S, Das N. Nanoencapsulation of quercetin enhances its dietary efficacy in combating arsenic-induced oxidative damage in liver and brain of rats. Life Sci 2009;84:75–80.
- [136] Flora SJS, Pande M, Kannan GM, Mehta A. Lead induced oxidative stress and its recovery following coadministration of melatonin or n-acetylcysteine during chelation with succimer in male rats. Cell Mol Biol 2004;50:543–51.
- [137] Garcia JJ, Reiter RJ, Guerrero JM, Escamer G, Yu BP, Oh CS, et al. Melatonin prevents changes in microsomal membrane fluidity during induced lipid peroxidation. FEBS Lett 1997;408:297–300.
- [138] Pieri C, Marra M, Moroni F, Recchioni R, Marcheselli F. Melatonin: a peroxyl radical scavenger more effective than vitamin E. Life Sci 1994;55:271-6.
- [139] Tan DX, Chen LD, Poeggler B, Manchester LC, Reiter RJ. Melatonin: a potent endogenous hydroxyl radical scavenger. Endocr J 1993;1:57–60.
- [140] Karbownik M, Gitto E, Lewinski A, Reiter RJ. Induction of lipid peroxidation in hamster organs by the carcinogen cadmium: amelioration by melatonin. Cell Biol Toxicol 2001;17:33–40.

- [141] Daniel S, Limson JL, Dairam A, Watkins GM, Daya S. Through metal binding, curcumin protects against lead and cadmium-induced lipid peroxidation in rat brain homogenates and against lead-induced tissue damage in rat brain. J Inorg Biochem 2004;98:266–75.
- [142] Pal S, Chatterjee AK. Possible beneficial effects of melatonin supplementation on arsenic-induced oxidative stress in Wistar rats. Drug Chem Toxicol 2006;29:423–33.
- [143] Melchiorri D, Reiter RJ, Atlia AM, Hara M, Burgos A, Nistico G. Potent protective effect of melatonin on in vivo paraquat-induced oxidative damage in rats. Life Sci 1995;56:83–9.
- [144] Gao S, Duan X, Wang X, Dong D, Liu D, Li X, et al. Curcumin attenuates arsenic-induced hepatic injuries and oxidative stress in experimental mice through activation of Nrf2 pathway, promotion of arsenic methylation and urinary excretion. Food Chem Toxicol 2013;59:739–47.
- [145] Gupta SC, Patchva S, Aggarwal BB. Therapeutic roles of curcumin: lessons learned from clinical trials. AAPS J 2013;15:195–218.
- [146] Ding W, Liu W, Cooper KL, Qin X, Bergo PL, Hudson LG, et al. Inhibition of poly(ADP-ribose) polymerase-1 by arsenite interferes with repair of oxidative DNA damage. J Biol Chem 2009;284:6809–17.
- [147] Piatek K, Schwerdtle T, Hartwig A, Bal W. Monomethylarsonous acid destroys a tetrathiolate zinc finger much more efficiently than inorganic arsenite: mechanistic considerations and consequences for DNA repair inhibition. Chem Res Toxicol 2008;21:600–6.
- [148] Kreppel H, Liu J, Liu YP, Reichl FX, Klaassen CD. Zinc-induced arsenite tolerance in mice. Fundam Appl Toxicol 1994;23:32–7.
- [149] Modi M, Kaul RK, Kannan GM, Flora SJS. Co-administration of zinc and n-acetylcysteine prevents arsenic-induced tissue oxidative stress in male rats. J Trace Elem Med Biol 2006;20:197–204.
- [150] Modi M, Pathak U, Kalia K, Flora SJS. Arsenic antagonism studies with monoisoamyl DMSA and zinc in male mice. Environ Toxicol Pharmacol 2005;19:131–8.
- [151] Zeng H, Uthus EO, Combs Jr. GF. Mechanistic aspects of the interaction between selenium and arsenic. J Inorg Biochem 2005;99:1269–74.
- [152] Flora SJS, Kannan GM, Kumar P. Selenium effects on gallium arsenide induced biochemical and immunological changes in male rats. Chem Biol Interact 1999;122:1–13.
- [153] Glattre E, Mravcova A, Lener J, Vobecky M, Egertova E, Mysliveckova M. Study of distribution and interaction of arsenic and selenium in rat-thyroid. Biol Trace Elem Res 1995;49:177–86.
- [154] George CM, Gamble M, Slavkovich V, Levy D, Ahmed A, Ahsan H, et al. A cross-sectional study of the impact of blood selenium on blood and urinary arsenic concentrations in Bangladesh. Environ Health 2013;12:52–8.
- [155] Flora SJS, Mehta A, Gupta R. Prevention of arsenic-induced hepatic apoptosis by concomitant administration of garlic extracts in mice. Chem Biol Interact 2009;177:227–33.
- [156] Rabbani GH, Saha SK, Akhtar M, Marni F, Mitra AK, Ahmed S, et al. Antioxidants in detoxification of arsenic-induced oxidative injury in rabbits: preliminary results. J Environ Sci Health A Tox Hazard Subst Environ Eng 2003;38:273–87.
- [157] Gupta R, Dubey DK, Kannan GM, Flora SJ. Concomitant administration of *Moringa oleifera* seed powder in the remediation of arsenic-induced oxidative stress in mouse. Cell Biol Int 2007;31:44–56.
- [158] Gupta R, Flora SJS. Effect of *Centella asiatica* on arsenic induced oxidative stress and metal distribution in rats. J Appl Toxicol 2006;26:213–22.
- [159] Flora SJS, Pant BP, Tripathi N, Kannan GM, Jaiswal DK. Distribution of arsenic by diesters of meso 2,3-dimercaptosuccinic acid during sub-chronic intoxication in rats. J Occup Health 1997;39:119–23.
- [160] Manna P, Sinha M, Sil PC. Protection of arsenic-induced hepatic disorder by arjunolic acid. Basic Clin Pharmacol Toxicol 2007;101:333–8.

- [161] Sharma A, Sharma MK, Kumar M. Protective effect of *Mentha piperita* against arsenic-induced toxicity in liver of Swiss albino mice. Basic Clin Pharmacol Toxicol 2006;100:249–57.
- [162] Flora SJS, Chouhan S, Kannan GM, Mittal M, Swarnakar H. Combined administration of taurine and monoisoamyl DMSA protects arsenic induced oxidative injury in rats. Oxidat Med Cell Long 2008;1:39–45.
- [163] Bhadauria S, Flora SJS. Response of arsenic induced oxidative stress, DNA damage and metal imbalance to combined administration of DMSA and monoisoamyl DMSA during chronic arsenic poisoning in rats. Cell Biol Toxicol 2007;23:91–104.
- [164] Flora SJS. Influence of zinc and copper supplementation during chelation of lead in rats. Human Exp Toxicol 1991;10:331-6.
- [165] Kalia K, Flora SJS. Strategies for safe and effective therapeutic measures for chronic arsenic and lead poisoning. J Occup Health 2005;47:1–21.
- [166] Modi M, Flora SJS. Combined administration of iron and monoisoamyl DMSA in the treatment of chronic arsenic intoxication in mice. Cell Biol Toxicol 2007;23:429–43.
- [167] Kokilavani V, Devi MA, Sivarajan K, Panneerselvam C. Combined efficacies of DL-alpha-lipoic acid and meso 2,3-dimercaptosuccinic acid against arsenic induced toxicity in antioxidant systems of rats. Toxicol Lett 2005;160:1–7.
- [168] Flora SJS, Mittal M, Mehta A. Heavy metal induced oxidative stress and its possible reversal by chelation therapy. Ind J Med Res 2008;128:501–28.
- [169] Moghimi SM, Hunter AC, Murray JC. Long-circulating and target-specific nanoparticles: theory to practice. Pharmacol Rev 2001;53:283–318.
- [170] Stolnik S, Illum L, Davis SS. Long circulating microparticulate drug carriers. Adv Drug Deliv Rev 1995;16:195–214.
- [171] Vauthier C, Dubernet C, Fattal E, Pinto-Alphandary H, Couvreur P. Oly(alkylcyanoacrylates) as biodegradable materials for biomedical applications. Adv Drug Deliv Rev 2003;55:519–48.
- [172] Yadav A, Lomash V, Samim M, Flora SJS. Curcumin encapsulated in chitosan nanoparticles: a novel strategy for the treatment of arsenic toxicity. Chem Biol Interact 2012;199:49–61.
- [173] Mandal AK, Das S, Basu MK, Chakraborti RN, Das N. Hepatoprotective activity of liposomal flavonoid against arsenite-induced liver fibrosis. J Pharmacol Exp Therap 2007;320:994–1001.
- [174] Sarkar S, Das N. Mannosylated liposomal flavonoid in combating age-related ischemia-repurfusion induced oxidative damage in rat brain. Mech Ageing Dev 2006;127:391–7.
- [175] Lockman PR, Mumper RJ, Khan MA, Allen DD. Nanoparticle technology for drug delivery across the blood brain barrier. Drug Dev Ind Pharm 2002;28:1–13.
- [176] Yadav A, Mathur R, Samim M, Lomash V, Kushwaha P, Pathak U, et al. Nanoencapsulation of DMSA monoester for better therapeutic efficacy of the chelating agent against arsenic toxicity. Nanomedicine 2013 PMID, 23721169.
- [177] Angle CR. Chelation therapies for metal intoxication. In: Chang LW, editor. Toxicology of metals. Boca Raton, FL: CRC Press; 1996. p. 487–504.



Biochemical and Molecular Basis of Arsenic Toxicity and Tolerance in Microbes and Plants

Sarita Pandey¹, Rashmi Rai², and Lal Chand Rai¹

¹MOLECULAR BIOLOGY SECTION, LABORATORY OF ALGAL BIOLOGY, CENTRE OF ADVANCED STUDY IN BOTANY, BANARAS HINDU UNIVERSITY, VARANASI, INDIA, ²LABORATORY OF MORPHOGENESIS, CENTRE OF ADVANCED STUDY IN BOTANY, BANARAS HINDU UNIVERSITY, VARANASI, INDIA

CHAPTER OUTLINE

27.1	Introduction	627
27.2	Arsenic Toxicity and Tolerance in Microbes	631
27.3	Arsenic Toxicity and Tolerance in planta	641
27.4	Mechanisms of Arsenic Tolerance and Detoxification	651
27.5	As Hyperaccumulation and Phytoextraction	657
27.6	Summary Points	658
Ackn	owledgments	659
Refe	rences	659

27.1 Introduction

27.1.1 Arsenic: A Threatening Environmental Issue

Arsenic (As) is highly toxic metalloid widely distributed in Earth's crust and occurs naturally in soil, water, and air. It is the 20th most abundant element in Earth's crust. Estimated arsenic concentration in sediments ranges between 5 and 10 mg/kg [1]. Increase in arsenic concentration in soil has resulted from natural weathering of arsenic-containing rocks and minerals, as well as from anthropogenic sources such as the smelting of metal ores, arsenical pesticides, wood preservatives, fossil fuel burning, use of As-based pesticides, herbicides, and irrigation with contaminated ground water [1–4]. Arsenic can easily enter into aquatic systems through river flow and surface runoff, and As enrichment in ground water is a global concern [3,5]. Major sources of human arsenic exposure are drinking water and the food supply, leading to health problems such as cancer, cerebrovascular disease, diabetes mellitus, and kidney disease [6]. Arsenic is first on the Centers for Disease Control Agency for Toxic Substances and Disease Registry's Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) Priority (Superfund) List of Hazardous Chemicals (http://www.atsdr.cdc.gov/ cercla/07list.html) because of its ubiquity, health effects, and potential for human exposure. As contamination in ground water has been reported from over 70 countries of which the worst is Bangladesh [7]. Besides the use of As-contaminated ground water for drinking and household purposes, such water has extensively been used for crop irrigation, particularly paddy rice (Oryza sativa L.) in Bangladesh and other South and South-East Asian countries [8]. About 75% of the total cropped area and 83% of the irrigated area in Bangladesh are under rice cultivation [8]. Background levels of As in paddy soils in Bangladesh range from 4 to 8 mg/kg [9], and longterm use of As-contaminated ground water for irrigation has resulted in a significant increase of As concentration in the topsoil (about 150 mm) of paddy fields [7]. About 83 mg/kg As was found in paddy soils that had been irrigated with As-contaminated ground water [9]. Thus, the problem of As pollution is continuously worsening due to a series of natural and human activities leading to an intensification of research focus on its phytotoxicity and the mechanisms employed by organisms such as microbes (prokaryotes and eukaryotes) and plants to counter its detrimental effects. There is a need to have a complete understanding of the molecular mechanisms and genetic basis of As toxicity and detoxification to produce/identify crop plants tolerant to As or use plants for photoremediation and restoration of As-contaminated sites.

27.1.2 Worldwide Occurrence of Arsenic

Arsenic is found naturally at low concentrations in Earth's crust, but few geographical regions have been reported with high arsenic concentration in ground water due to the continuous enrichment of As through natural and anthropogenic sources, as discussed in the section above. Unlike the rare-earth elements, arsenic is commonly found concentrated in sulfide-bearing mineral deposits, especially those associated with gold mineralization or concentrated in hydrous iron oxides. Arsenic can be easily solubilized in ground waters depending on pH, redox conditions, temperature, and solution composition. Many geothermal waters or aquifers contain high concentrations of arsenic at concentrations greater than the World Health Organization-and US Environmental Protection Agency-recommended drinking water standard of $10\mu g$ per liter. Figure 27-1 shows the major epidemic areas with high arsenic contamination in underground irrigation water and paddy soils in West Bengal (India), China, Myanmar, Nepal, Thailand, Taiwan, and Vietnam [10]. On the basis of the severity of As contamination, the Asian countries have been ranked as: Bangladesh, West Bengal (India), Inner Mongolia (PR China), and Taiwan [11]. The World Health Organization has called arsenic contamination of drinking water in Bangladesh the worst environmental catastrophe in human history, and its control is beyond the capability of many countries [12]. Coming to an Indian scenario, the first report of arsenic contamination came from West Bengal in 1983, where 33 villages in four districts were found to be affected. This number has increased to 3417 in 111 blocks in nine districts in 2008. Contamination was also detected in other states



FIGURE 27–1 Worldwide occurrence of As in ground water.

such as Punjab, Bihar, Jharkhand, Orissa, Chhattisgarh, Madhya Pradesh, Uttar Pradesh, Assam, Arunachal Pradesh, Manipur, Nagaland, and Tripura [13–17]. The seven states of West Bengal, Jharkhand, Bihar, Uttar Pradesh in the flood plain of the Ganga River; Assam and Manipur in the flood plain of the Brahamaputra and Imphal rivers and Rajnandgaon village in Chhattisgarh state have so far been reported affected by As contamination in groundwater above the permissible limit of 10 μ g/L. The occurrence of arsenic in groundwater in the Bengal delta plains and Gangetic plains has been recognized due to geological origins. The environmental arsenic problems, however, are the result of mobilization under natural conditions.

27.1.3 Arsenic Speciation and Mobilization

Arsenic is present in four oxidation states, -3, 0, +3, and +5, the last two being the most common in the terrestrial environment [18]. Arsenate [As(V)] is the predominant species in aerobic soils, whereas arsenite [As(III)] predominates in anaerobic environments such as submerged soils. Elemental arsenic occurs hardly ever, and arsines have been identified from fungal cultures and strongly reducing environments [19]. The relative proportions of these oxidation

states in a given environment depend on the biological processes involved in the metabolism by microorganisms [20], such as As(V) reduction, As(III) oxidation, and the various methylation reactions as well as on the local physicochemical conditions, including the redox potential (Eh) and the pH, which are important factors. Since the pKa's of arsenate (H₃AsO₄) are pKa₁ = 2.19, pKa₂ = 6.94, and pKa₃ = 11.5, the H₂AsO₄⁻ form predominates in oxidative media with pH levels below 6.9, whereas the HAsO₄²⁻ form predominates at higher pH. These are the most prevalent forms of arsenic in surface soil, water, and within cells, and occur primarily in aerobic environments. However, for arsenite, the lowest pKa level recorded is equal to 9.22. In most natural waters with pH levels below 9.2 as well as in slightly reductive environments, As(OH)₃ is the main form present [21]. Several studies have shown that the reduction of arsenate into arsenite results in the solubilization of this element [1,22]. However, arsenate may be sequestered after being co-precipitated with ferric iron or sulfur [23] or adsorbed by clay, calcite, organic matter or hydroxides, in particular ferric oxyhydroxides [24]. Generally, prokaryotes are 10 times more resistant to As(V) than to As(III) [25].

Similar to microbes, the bioavailability of As to plants is governed by edaphic properties, environmental conditions, and modification of the soil in the rhizosphere; these factors interact to influence As speciation in the soil. Analyses of a range of plant species with the aid of analytical techniques such as HPLC/ICPMS/ES-MS and synchrotron radiation-based X-ray absorption spectroscopy (XAS) have shown that both terrestrial and freshwater aquatic plants contain predominantly inorganic As [26-28] but their relative proportions vary among the plant species. As is present predominantly as trivalent species [As(III)] [29–32] and is the main species in the fronds of As-hyperaccumulating ferns [29,33,34]. Furthermore, most of the arsenite is present as complexes like GsAs(III)-PC2, As(III)-PC3, As(III)-(PC2)2, and As(III)GS3 as arsenite has a high affinity to sulfydryl groups. However, As hyperaccumulators such as Pteris vittata and Pteris cretica appear to be exceptions, containing arsenite mostly as uncomplexed species, due to low phytochelatin (PC) concentrations in tissues [35,36]. Many plant species also contain small amounts of methylated As compounds, such as MMA, DMA, TMAO, and, occasionally, the tetramethylarsonium ion [26-28]. Like As(V) and As(III), the methylated forms of As are phytotoxic [37]. Monomethylarsonous acid with As in the trivalent state $[MMA(III): CH_3As(OH)_2]$ can be complexed with thiol compounds, such as MMA(III)-PC2 identified in the roots of sunflower exposed to inorganic As [31]. Although pentavalent As does not form complexes with thiol groups directly, pentavalent As in DMA can bind to glutathione (GSH) when it is activated by sulfide, forming the dimethylars inothioyl glutathione (DMAS-GS) complex in the sulfur-rich plant species *Brassica oleracea* [38]. Arsenosugars are found in some terrestrial plant species at low levels [26,27] but whether they are synthesized by plants is not known. Root and shoot tissues of Arabidopsis thaliana and Brassica juncea are coordinated with the sulfydryl groups of thiol-rich peptides such as GSH and PCs [29,30].

27.1.4 Arsenic Epidemiology

Globally, a major fraction of the total population is exposed to arsenic due to intake of arsenic-contaminated water [10]. Arsenic produces its toxic effects by hindering cell

metabolism, growth, and proliferation. Arsenic leads to DNA damage and inhibits repair and replication machinery. Overall, arsenic acts on cells in a variety of ways, influencing numerous signal transduction pathways and resulting in a vast range of cellular effects that include apoptosis induction, growth inhibition, promotion or inhibition of differentiation, and angiogenesis. Daily intake of water containing arsenic at more than $10 \mu g/L$ for a long duration leads to many diseases. These include cancers, especially those of lungs, bladder, skin, and kidney, arsenicosis, frequent abortions, impotency, hyperkeratosis, and cardiovascular problems [13,39,40].

27.2 Arsenic Toxicity and Tolerance in Microbes

The toxicity of arsenic depends on its form and oxidation states. It was established recently that toxicity decreases as follows: arsine > arsenite > arsenoxides > arsenate > arsonium compounds > arsenic [41]. Among these, As(V) is the predominant form of As. Because As(V)is a structural analogue of phosphate, its main toxicity results from its interference with the metabolism of phosphorus. Once in the cell, As(V) substitutes for phosphate leading to the production of unstable arsenical by-products. It interferes with the normal phosphorylation processes resulting in uncoupling of oxidative phosphorylation, and thereby disruption of ATP (adenosine 5'-triphosphate) synthesis [42]. Indeed, As(V) reacts enzymatically with ADP (adenosine 5'-diphosphate) to form an arsenate-phosphate bond. The arsenical ester formed is unstable and spontaneously hydrolyses to ADP and As(V), thus preventing ATP formation. In the same way, ATP-dependent transport, glycolysis, the pentose phosphate pathway, and signal transduction pathways could be impaired [43]. Arsenic toxicity is linked with phosphate availability. Phosphate-supplemented cells reduce the uptake of arsenic. It was found that phosphate-starved cells are more sensitive to arsenic than phosphate-repleted cells [44]. Due to potential competitive inhibition between arsenate and phosphate [1,45], arsenate resistance or detoxification pathways may be particularly important in low phosphate conditions. On the other hand, As(III) has a strong affinity for protein sulfhydryl groups. Indeed, the redox status of cysteine residues can affect both the structure and the activity of numerous enzymes, receptors, and transcription factors. As(III) reacts with dithiol groups present on active sites of many enzymes and on glutathione, glutaredoxin (grx), and thioredoxin. These last three metabolites have multiple functions, including intracellular redox homeostasis, deoxyribonucleotide synthesis and repair, protein folding, sulfur metabolism, and xenobiotic detoxification. All these processes are therefore inhibited by As(III). Another cause of carcinogenicity of As(III) is likely linked to its capacity to oxidize reduced glutathione, which is the major cellular antioxidant. This oxidation leads to the increase of reactive oxygen species (ROS), known to damage macromolecules such as proteins, lipids, and DNA [46,47].

In spite of the great toxicity and widespread nature of As compounds, a considerable number of diverse organisms including archaea, bacteria, cyanobacteria, phytoplankton, and plants manage to thrive in arsenic-contaminated environments. Arsenic is a notorious metalloid that was probably more abundant on early Earth's surface than today [48]. Therefore, systems evolved by primordial organisms to cope with it still exist in prokaryotes. Not only do they resist



FIGURE 27–2 Schematic representation of arsenic metabolism pathways in prokaryotes. Arsenic (As) enters the cell through the phosphate transporters [arsenate, As(V)] or the aquaglyceroporins [arsenite, As(III)]; arsenic is immobilized in the environment by extracellular precipitation; once inside the cell, As(V) is reduced by an arsenate reductase, ArsC, to As(III), which is extruded out of the cell by the specific membrane pump Ars(A)B; inorganic arsenic can be transformed into organic species via a cascade of methylations; As(V) is used as an electron acceptor during respiration by the dissimilatory arsenate reductase ArrAB; As(III) can serve as an electron donor via the As(III) oxidase AoxAB or ArxAB. As, arsenic; As(III), arsenite; As(V), arsenate; Pit, phosphate (PO₄) inorganic transport system; Pst, phosphate-specific transport system; GlpF, aquaglyceroporin GlpF; ArsC, detoxifying arsenate reductase; Ars(A)B, arsenite efflux pump; MMA, monomethylarsonic acid; DMAA, dimethylarsinic acid; DMA, dimethylarsine; TMAO, trimethylarsine oxide; Aox/Arx, different arsenite oxidase types; Arr, dissimilatory arsenate reductase. *Figure modified from Paez-Espino et al.* [51]

this metalloid with a wide array of mechanisms but strikingly some of them take advantage of this toxic compound by using it as an electron donor [As(III)] or acceptor [As(V)] for energetic purposes [49–52]. It has been proposed recently that arsenic can substitute for phosphorus in proteins and nucleic acids [53]. Microbes respond to arsenic in a variety of different ways. Responses of microorganisms may include (1) extracellular sequestration [54], (2) minimizing the amount of arsenic that enters the cell, e.g., through increased specificity of phosphate uptake [55], (3) chelation mediated by glutathione, phytochelatin, and metallothionein [52,56,57], (4) reduction [49–51,58] followed by efflux [58] and methylation [59,60], and (5) oxidation [61,62] (Figure 27–2).

27.2.1 Arsenic Uptake Pathways

Due to the extreme toxicity of arsenic, specific uptake transporters have not evolved [50] to avoid its uptake. But As(III) and As(V) are taken up by glycerol and phosphate transporters, respectively,

because of their similar chemical structure. In the case of *E. coli*, two phosphate transporters, Pit and Pst, are used for As(V) uptake; Pst is the dominant uptake pathway [63]. As(III) is taken up by the glycerol transporter GlpF [64], a member of glycerol channels of the major intrinsic protein (MIP) family. Mutation in GlpF resulted in As(III)-tolerant *E. coli* strains [64].

27.2.2 Extracellular Immobilization of Arsenic

Different bacteria have been shown to have the capacity to sequester or immobilize metals outside of the cell before they enter the cytoplasm, thus preventing their entry into cells and interaction with essential components, as in the case of *Herminiimonas arsenicoxydans* [54,65]. In the presence of arsenic, this results in large amounts of exopolymers. Images obtained using transmission electron microscopy (TEM) combined with X-ray energy dispersion spectroscopy (EDS) indicated that arsenic metalloid is sequestered within the exopolysaccharide capsule [65]. Bacteria mediating Fe(II) oxidation through nitrate reduction to get energy for growth in anoxic conditions have been shown to play a key role in arsenic cycling by forming solid hydrous ferric oxide to which As(V) [66,67] or As(III) [67,68] is adsorbed. In oxic environments, the acidophilic iron oxidizer *Acidithiobacillus (At.) ferrooxidans*, when grown with Fe(II) as electron donor, scavenges As(III) with the newly produced Fe(III) to form schwertmannite or tooeleite [24,69,70].

27.2.3 Arsenic Chelation

Chelation is the mechanism for limiting As accumulation and interaction with essential cellular components after its entry into the cell. This is mediated by peptides or proteins containing thiol ligands, such as glutathione, phytochelatins, and metallothioneins [52,56,57]. In *Cyanobacteria* and *Pseudomonas* sp., metallothioneins have been detected and shown to coordinate different metals [52,71–74].

27.2.4 Arsenate Reduction

Two different mechanisms of arsenate reduction are reported in bacteria, fungi, and algae: one is widespread in all microorganisms and considered responsible for detoxification; the other is specific only for bacteria and some archaea. Microorganisms use As as a final electron acceptor in their respiratory chains, for autotrophic or heterotrophic growth.

27.2.4.1 Mechanism of Arsenic Detoxification

Pathways for arsenic detoxification exist in all organisms examined, including bacteria, yeast, plants, etc. *Escherichia coli* [75], *Staphylococcus aureus* [76], *Corynebacterium* sp. [77], *Synechocystis* sp. PCC6803 [78], and *Saccharomyces cerevisiae* [79] are a few examples of these categories. The arsenic detoxification mechanism is controlled by *ars* genes that encode proteins that facilitate As(V) reduction via a cytoplasmic enzyme arsenate reductase into As(III) followed by As removal from the cell by an efflux pump [25,80]. The two most common operons contain either five (*arsRDABC*) or three (*arsRBC*) genes. The *arsRDABC* operon is found on the plasmids of Gram-negative bacteria such as *E. coli* R773 and *arsRBC* generally on the plasmids

of Gram-positive bacteria such as *Staphylococcus aureus* pI258, or on the bacterial chromosome [75,81,82]. The *arsR* gene encodes a trans-acting repressor of the ArsR/SmtB family involved in transcriptional regulation [83], *arsB* encodes a membrane bound arsenite carrier that effluxes arsenite [84], and *arsC* encodes a cytoplasmic enzyme named arsenate reductase that converts arsenate to arsenite [85]. In contrast, the *ars* operon of *E. coli* plasmids R773 and R46 encodes two additional proteins: ArsA, an arsenite-stimulated ATPase, and ArsD, a metalloid-responsive transcriptional repressor [83,86]. Similarly, in the case of *Saccharomyces cerevisiae*, a eukaryote, a gene cluster, *ACR1*, *ACR2*, and *ACR3*, present on chromosome was shown to confer resistance to As(V) and As(III) [79]. Acr3p catalyzes extrusion of the arsenite from cells, thus conferring resistance [87].

While this fundamental strategy of conferring arsenic resistance is fairly conserved, the arsenate reductases are quite varied in different organisms. The four types of arsenate reductases characterized as yet, such as plasmid encoded R773 ArsC and pI258 ArsC from Escherichia coli [85,88] and Staphylococcus aureus [76], respectively, Acr2p from Saccharomyces cerevisiae [79,89], and hybrid-type arsenate reductase SynArsC from Synechocystis PCC6803 [78,90], possess dissimilar sequence, folds, and differ from one another in several of their physical and catalytic properties [58,90]. (1) Prokaryotic ArsC from Gramnegative bacterium E. coli encoded by plasmid R773 [88], which uses grx and GSH as reductant. Reduction of As(V) is mediated by three cysteine residues: one in ArsC, one in GSH, and one in glutaredoxin. As(V) binds to three arginine residues and forms a covalent bond with the single catalytic cysteine residue Cys12 (R773 ArsC). A tertiary intermediate is formed between Cys12 and a glutathione cysteine. Finally, glutaredoxin reduces As(V) and the disulfide bond leading to the release of As(III) and regeneration of reduced ArsC [58,91]. (2) The prokaryotic ArsC from Gram-positive bacterium *Staphylococcus aureus* encoded by pI258 [76] and *Bacillus* subtilis use the cysteine thiol-coupling enzyme thioredoxin as the reductant and require the presence of thioredoxin reductase and NADPH to complete the catalytic cycle. The structure of these enzymes is remarkably similar to those of low molecular weight protein tyrosine phosphate phosphatases [92] and shows a low level of phosphatase activity. Three conserved cysteine residues are essential for catalysis. As(V) is covalently bound to Cys10 (pI258 ArsC), which reduces it by nucleophilic attack. Cys82 then attacks Cys10 leading to the formation of a Cys10-Cys82 disulfide intermediate and release of As(III). Cys89 attacks Cys82 to form a Cys82-Cys89 disulfide intermediate and to regenerate Cys10 thiolate. Thioredoxin reduces the Cys82-Cys89 disulfide regenerating reduced ArsC [92]. (3) The eukaryotic arsenate reductase Acr2p from Saccharomyces cerevisiae [79,89], Leishmania major [93], and Arabidopsis thaliana [94] uses grx and GSH as reductant. While Acr2p and R773 ArsC obtain their reducing equivalents from glutathione and glutaredoxin, pI258 uses thioredoxin as reductant. However, in the case of Synechocystis PCC6803, a novel type of hybrid arsenate reductase related to thioredoxin-dependent arsenate reductases uses the GSH/glutaredoxin system for reduction [90]. This arsenate reductase is encoded by *arsC* of the *arsBHC* operon. Disruption of the arsenate reductase gene leads to arsenate hypersensitivity in both E. coli and Saccharomyces cerevisiae [58,95]. Recently, in the case of the diazotrophic cyanobacterium Anabaena PCC7120, two different types of arsenate reductases were characterized [96,97].

27.2.4.2 Arsenate Respiration or Dissimilatory As(V) Reduction

The arsenic respiration specific for the Bacteria and Archaea is the second mechanism for reduction of As(V). Using this mechanism, bacteria couple the reduction of oxidized arsenic compounds with the final electron acceptors from their respiratory chain. As(V) being electrochemically positive, reduction coupled with the oxidation of an electron donor with a lower redox potential can be expected to provide sufficient energy to sustain bacterial growth [98] and, indeed, microbes respiring As(V) under anaerobiosis have been found throughout the prokaryotes, including Proteobacteria, Gram-positive Eubacteria, thermophilic Eubacteria, and Crenarchaeota from different environments [48,50,51]. These microorganisms use electron donors for As(V) reduction like hydrogen and sulfide, acetate, lactate, formate, pyruvate, and complex compounds such as benzoate, phenol, and toluene [50]. Dissimilatory arsenate reduction is carried out by a terminal aresenate reductase, fundamentally different from the detoxifying arsenate reductases. To date, only the arsenate reductases from Chrysiogenes arsenates [99], Bacillus selenitireducens [100], and Shewanella trabarsenatis strain ANA-3 [101] have been characterized. In these cases, arsenate reductase ArrAB is a heterodimer composed of large and small subunits of about 100kDa (ArrA) and 30kDa (ArrB), respectively. ArrA, which belongs to the DMSO reductase family, contains the As(V) binding site and the catalytic site with a bismolybdopterin guanine dinucleotide cofactor and one (4Fe-4S) cluster, while ArrB is an iron-sulfur protein with four (4Fe-4S) clusters. ArrAB is translocated to the periplasm by the twin arginine translocation (TAT) pathway since a characteristic TAT signal sequence is present in the large subunit ArrA. The C. arsenatis and S. trabarsenatis enzymes reduce only arsenate whereas B. selenitireducens also reduces As(III), selenate, and selenite [100]. ArrB conducts the electrons from the respiratory chain to the molybdenum cofactor of the large subunit ArrA where As(V) reduction takes place. In S. trabarsenatis ANA-3, the membrane-associated periplasmic tetraheme c-type cytochrome, CymA, is required for As(V) reduction [102] and has been proposed to mediate electron transfer from the quinol pool to ArrB [103]. The ArrA and ArrB subunits are encoded in an operon. ArrA large catalytic subunit encoded by *arrA* is present upstream of the small subunit coding gene *arrB*. The third subunit encoding gene (*arrC*) has been detected upstream from *arrAB* in a large number of bacteria. Membrane-integrated ArrC has been proposed to anchor ArrAB to the cytoplasmic membrane and to mediate electron transfer from the quinone pool [50,51,102,104]. It is worth noting that no archaeal Arr representatives have been detected so far [104].

To date, the only reported study on the regulation of the *arr* operon has been performed in *S. trabarsenatis* ANA-3. This operon has been shown to be mainly expressed in anaerobiosis with further activation by As(III) and As(V), and it is repressed by oxygen and nitrate, indicating that it is regulated by at least two control systems, one sensing the cellular redox state and the other detecting specifically arsenic [105]. The *cymA* gene encoding the physiological partner of the arsenate reductase is also induced under anaerobic conditions in the presence of arsenic [102,106]. The cyclic AMP receptor protein (CRP), activated by cyclic AMP (cAMP) production under anaerobic conditions in *S. trabarsenatis* ANA-3, was found essential for *arr* expression in the absence of oxygen [106]. The As(III) sensing regulator was demonstrated to be an ArsR repressor, ArsR2, encoded by a gene located in the same locus as the *arr* operon [106]. The

model proposed could be that ArsR2 represses *arr* transcription in the absence of arsenic even when CRP is activated by cAMP under anoxic conditions [106]. Concerning the possible origin of the arsenate reductases, two hypotheses have been proposed. In one, the arsenate reductase was an "ancient" microbial enzyme that arose on primordial Earth, was functioning as an arsenite oxidase, and was responsible for the anaerobic As(III) oxidation before acquiring its As(V) reducing functions [48,107]. In contrast, Arr originates more "recently" after the Archaea/Bacteria split, likely in a *Gammaproteobacterium*, from an enzyme involved in the bioenergetic reduction of sulfur compounds (i.e., tetrathionate, thiosulfate or polysulfide reductase) followed by its dispersal over other phyla [104,108].

27.2.5 Efflux of As(III) from the Cell

The most widely distributed and the best characterized system for As(III) detoxification is the Ars system (ArsRBC or ArsRDABC). Among these systems ArsB is an integral membrane protein, which pumps As(III). ArsB when alone uses the membrane potential to drive the efflux of the As(III). But when present with ArsA, which is an ATPAse that converts membrane potential to ATP, it gets energy from ATP hydrolysis, making As(III) efflux more efficient. ArsD also functions as a chaperone of the ArsAB pump, by binding and transferring As(III) to ArsA [109]. However, in *Synorhizobium meliloti* As(III) efflux is done through an aquaglyceroporin, AqpS, encoded within the ars operon in place of arsB [110,111]. Recently, novel single arsenate resistance proteins that apparently result from recent evolutional fusion events between an aquaglyceroporin-derived arsenite channel and the C-terminal domain of an ArsC arsenate reductase in the actinobacteria Frankia alni and Salinispora tropica and between Acr3 and ArsC in Mycobacterium tuberculosis were reported. In these cases, a single protein generates the energy and catalyzes the efflux of As(III), which should improve the extrusion efficiency [112]. No As(V) extrusion pump has been identified to date. In fact, the less toxic As(V) is converted by the cytoplasmic arsenate reductase ArsC to the more toxic As(III), which is then pumped out of the cell as discussed in the previous section. As(V) detoxification involving its enzymatic reduction to As(III), followed by As(III) export, is conserved from bacteria to plants [113]. It was proposed that the As(III) efflux system arose first in the primordial anaerobic environments where As(III) was the predominant As species. Once Earth's atmosphere became more oxidizing with the increased production of molecular oxygen, As(V) would have become the major As species, and proteins (likely phosphatases) evolved to reduce As(V) to As(III), which was subsequently excreted from the cell via ArsB or Acr3 to mitigate the toxic effect of $A_{s}(V)$ [58].

27.2.6 Oxidation of As(III)

A wide range of bacteria and archaea have the potential to oxidize As(III) enzymatically through arsenite oxidases. The first report of an As-oxidizing bacterium named *Bacillus arsenoxidans* was given by Green in 1918 [114]. Many studies have now reported microorganisms as arsenite oxidizers with heterotrophic or chemoautotrophic growth [61,62,65,115–117]. In these bacteria As(III) serves as an electron donor reducing oxygen or nitrate [118,119]. However, in the case of the Gammaproteobacteria *Ectothiorhodospira* sp. strain PHS-1, As(III) acts as an

electron donor for anoxygenic phototrophic growth [107]. Therefore, for As(III) oxidation two different arsenite oxidase enzymes were reported, the first being AoxAB (also named AroBA or AsoBA), widely found and extensively studied, and the second being ArxAB, found only in the Ectothiorhodospiraceae family [107,120]. The latter is phylogenetically close to respiratory arsenate reductase ArrAB than to AoxAB arsenite oxidase. The genes *aoxA* and *aoxB* encoding the arsenite oxidase AoxAB were first identified and characterized in the heterotrophic bacterium Herminiimonas arsenicoxydans. They have been sequenced and their operon structure has been described [121]. These genes are expressed in the presence of arsenic. Thereafter, these genes (also called *asoA* and *asoB*) were recognized and sequenced in various bacteria, including chemoautotrophic (ref. NT-26) and photoautotrophic microbes [22]. The genetic basis of arsenite oxidase was first elucidated by Muller et al. [121] in knockout mutants deficient in arsenite oxidation of strain ULPAs1. This AoxAB-type arsenite oxidase has been characterized from Alcaligenes faecalis [122-124], Rhizobium sp. strain NT-26 [116], Hydrogenophaga sp. strain NT-14 [125], Arthrobacter sp. strain 15b [126], and Ralstonia sp. 22 [127]. In all these cases, it consists of a large subunit arsenite oxidase B (AoxB) belonging to the DMSO reductase family with a molybdenum cofactor together with a (3Fe-4S) center and a small subunit (AoxA) containing a "Rieske" (2Fe-2S) center [104,123]. The small subunit AoxA possesses a TAT leader sequence. The difference in the quaternary structure of AoxAB-type arsenite oxidases is noticed as heterodimer $\alpha 1\beta 1$ in *A. faecalis* [123] and *Arthrobacter* sp. strain 15b [126], heterotetramer $\alpha 2\beta 2$ in *Rhizobium* sp. strain NT-26 [116], and heterohexamer $\alpha 3\beta 3$ in *Hydrogenophaga* sp. strain NT-14 [125].

27.2.6.1 Mechanism of As(III) Oxidation

As per the proposed As(III) oxidation model [123,124], As(III) first undergoes a direct nucleophilic attack by the molybdenum cofactor Mo(VI) with the transfer of two electrons leading to the production of reduced molybdenum Mo(IV) with As(V) coordinated with the metal. As(V) is released and Mo(IV) is oxidized to Mo(VI). Then, electrons are transferred first to the (3Fe-4S) in AoxB and then to the (2Fe-2S) center of AoxA. From the small subunit, the electrons are given to the respiratory chain in the inner membrane towards the terminal electron acceptor. In the case of A. faecalis, the electron acceptor of AoxAB has been proposed to be azurin or a cytochrome c since both proteins are reduced by AoxAB in the presence of As(III) in vitro [122]. However, in the case of Hydrogenophaga sp. strain NT-14 [125], Rhizobium sp. strain NT-26 [128], and *Ralstonia* sp. strain 22 [127], a cytochrome c that accepts electrons directly from AoxAB has been characterized. In addition, in Ochrobactrum tritici, a cytochrome c, encoded by the aox operon, is required for arsenite oxidation [129]. The two arsenite oxidase subunits are invariably encoded in an operon with the gene encoding the small subunit *aoxA* upstream from that encoding the catalytic subunit *aoxB*. This relative gene order is conserved in all arsenite oxidizers analyzed to date like *H. arsenicoxydans* [65,121], Rhizobium sp. strain NT-26 [116], Agrobacterium tumefaciens [130], T. arsenitoxydans [131,132], Achromobacter sp. strain SY8, Pseudomonas sp. TS44 [133], O. tritici [129], and Hydrogenobaculum sp. strain 3684 [134]. Nevertheless, very often upstream from aoxA, two genes encoding the histidine kinase and the cognate response regulator of a classic two-component signal transduction system (*aoxSR*) are detected. Downstream from *aoxB*, a gene encoding a cytochrome *c* is sometimes present in *H. arsenicoxydans* [65,121], *Rhizobium* sp. strain NT-26 [128], *A. tumefaciens* [130], *T. arsenitoxydans* [131,132,135], *Achromobacter* sp. strain SY8 [133], and *O. tritici* [129], as well as one or two genes involved in molybdenum cofactor biosynthesis *moaA* and/or *moeA*.

The *aox* operon regulation seems to be complex and to vary between species. Approximately four proteins have been shown to be involved in the *aox* operon expression: (1) a two-component signal transduction system AoxSR, which is required for *aox* expression in all the bacteria analyzed so far, (2) an ArsR/SmtB family member AoxF specific to T. arse*nitoxydans*, (3) the sigma factor σ 54 (RpoN) with a binding site upstream from *aox* operon in Rhizobium sp. strain NT-26 [128], O. tritici [129], T. arsenitoxydans [136], and H. arsenicoxydans [137], and (4) DnaJ in *H. arsenicoxydans* [137]. The role of the co-chaperone DnaJ constitutes part of the Hsp70 machinery, and is believed to help in proper folding of AoxR, stabilization of RpoN, or stabilization of *aox* messenger RNA [137]. It has also been established that in some chemolithotrophic enzymes, arsenite oxidation is combined with oxygen or nitrate reduction processes (under anaerobic conditions) and that the energy thus produced serves to fix CO₂ [22]. In environments that harbored the very first forms of life on Earth, As(III) may have provided one of the main mineral energy sources required for the metabolism of the first chemolithotrophic organisms. The colonization of these primitive arsenic-rich environments by bacteria using As(III) as an electron source and transforming it into less bioavailable As(V)may have resulted in the partial detoxification of these inhospitable environments. These bacteria may therefore have contributed to extending the limits of life by making it possible for other microorganisms to survive and proliferate.

ArxAB-type arsenite oxidase has been detected in certain arsenite oxidizers, which oxidize As(III) in anoxic conditions like Alkalimnicola ehrlichii [117,138], Azoarcus sp. strain DAO1, [138] and Ectothiorhodospira sp. strains PHS-1 and MLP2 [107] with no aoxAB gene homologues known, suggesting that an alternative mechanism is utilized in these conditions. arxA and arxB genes coding for a novel arsenite oxidase were identified in A. ehrlichii [120,139] as well as in *Ectothiorhodospira* sp. strains PHS-1 and MLP2 [120]. ArxB has been predicted to be a small subunit with four (4Fe-4S) iron-sulfur clusters and ArxA to be a large subunit containing a molybdenum cofactor and a (4Fe-4S) cluster [120]. This new arsenite oxidase shows greater similarity with the respiratory arsenate reductase ArrAB than with the arsenite oxidase AoxAB in terms of (1) the total amino acid identity and similarity, (2) the presence of TAT signal sequence in the large subunit ArxA, (3) the large catalytic subunit ArxA, which contains a (4Fe-4S) rather than a (3Fe-4S) cluster, and (4) the small subunit ArxB, which has four (4Fe-4S) clusters in contrast to the single (2Fe-2S) center present in the Rieske subunit AoxA. Apart from this the genetic organization of the A. ehrlichii arx operon is different from that of the aox operon since (1) arxA encodes the large subunit upstream from arxB corresponding to the small subunit and (2) it encodes also a putative (4Fe-4S) containing protein ArxB', an anchoring subunit ArxC, and a chaperone-like ArxD of the TorD family [120]. This operon is only expressed under anaerobic conditions in the presence of As(III) [139]. Furthermore, the phylogenetic relatedness of this arsenite oxidase is more

closely related to ArrAB respiratory arsenate reductases than to AoxAB arsenite oxidases [139] and has been proposed to be an ancestral form of the ArrAB respiratory arsenate reductases and AoxAB arsenite oxidases [48].

27.2.7 Arsenic Biomethylation

Biomethylation of As is a well-known biological process, which is reported in many bacteria, archaea, fungi, algae, plants, animals, and humans [140–143]. Recently, As biomethylation and biovolatilization have also been observed in a model protozoan *Tetrahymena thermophila* [144]. These organisms methylate As to different extents and produce volatile and non-volatile As species. The leading volatile species are mono-, di-, and trimethylarsine (MMA, DMA, and TMA), while methylarsonate and dimethylarsinate are the main non-volatile compounds. Although methylation was considered as a detoxification process, recent studies show that some methylated species such as mono- and dimethylarsenite can be as, or even more toxic than, inorganic forms of arsenic [47,145]. Biotransformation of As(III) to TMA(III) was first documented in fungi in 1890 [19]. The *arsM* gene encoding the As(III) *S*-adenosylmethionine methyltransferase is prevalent and has been detected in 125 bacteria and 16 archaeans [146]. Subsequently, genome sequencing homologues of *arsM* were identified in many fungi. The ArsM from *Rhodopseudomonas palustris* was shown to confer As(III) resistance when expressed in an As-sensitive strain of *Escherichia coli* [146].

The presence of methylated As species has been reported for numerous photosynthetic organisms, including cyanobacteria, algae and plants. When the marine green alga Chlorella salina was exposed to 1 mM As(V) or As(III), the amount of MAs(V) and DMA(V) was 9-12% of total intracellular As [147]. A more recent study on the short-term metabolic processes of As(V) in a freshwater green alga, *Chlamydomonas reinhardtii*, showed that As(III), As(V), and MAs(V) concentrations in the cells increased from 30 min to 1 h, and then leveled off [148]. DMA(V) and oxo-arsenosugar-glycerol levels also increased continuously for 24 h. These results indicated that C. reinhardtii can rapidly biotransform As(V) into oxo-arsenosugar-glycerol and then oxo-arsenosugar-phosphate. The reduction of As(V) to As(III) and methylation of As(III) to DMA(V) are coupled in this process, and MAs(V) was observed to be a possible intermediate. This is different from *Rhodopseudomonas palustris*, in which no monomethylated species were observed. When the marine brown alga Fucus serratus was treated with different concentrations of As(V) for up to 19 weeks, As(V), As(III), MAs(V), DMA(V), and four arsenosugars were detected [149]. The proportion of As species found in *F. serratus* depended on the As(V) concentration that it was exposed to. When the As(V) concentration was at 100 mg As(V)/L, the main product was As(III). At 20 mg As(V)/L, DMA(V) and arsenosugars were the main As metabolites. These results suggest that, at higher As(V) concentrations, the As(III) methylation process becomes saturated, leading to the accumulation of As(III) and subsequent toxicity [149]. However, at this point, it is not known whether *F. serratus* actually has a gene encoding an ArsM. Fucus serratus is often intimately associated with specific fungi and it is possible that these fungi carry out the methylation reaction [150]. Thermophilic algae have also been shown to methylate and volatilize As. Cyanidioschyzon sp. isolate 5508, isolated from thermal wells in



FIGURE 27–3 Biomethylation pathway in prokaryotes (SAM, S-adenosylmethionine; R-SH, thiol-containing compounds) [43].

Yellowstone National Park, is capable of oxidizing As(III) to As(V), reducing As(V) to As(III), and then methylating As(III) to form TMAO and DMAs(V) [146]. Two As methyltransferase genes from *Cyanidioschyzon* sp. isolate 5508, CmarsM7 and CmarsM8, were expressed in an As(III)-hypersensitive strain of *E. coli* and conferred resistance to As(III).

Cyanobacteria are ubiquitous in environments ranging from soil and water to deserts. Due to their rapid growth rates and ability to adapt to environmental changes, they are often key players in toxic aquatic environments. Methylated As species have been found in various cyanobacteria and algae. Arsenic biotransformation and volatilization has also been observed in three cyanobacteria under laboratory conditions [144]. *Microcystis* sp. PCC7806, *Nostoc* sp. PCC7120, and *Synechocystis* sp. PCC6803 accumulated large amounts of As when treated with 10 mM sodium As(III). Methylated species of As(III) were detected following exposure to higher As(III) concentrations for 14 days. Volatile arsenicals were detected when the cyanobacteria were treated with As(V) for 6 weeks. Two ArsM homologues have been cloned and purified. The enzymes were shown to methylate As(III) *in vitro* with TMA(III) as the end product [144].

27.2.7.1 Methylation Pathway

The methylation pathway is a multistep process with initial As(V) reduction followed by a cycle of oxidative methylation and reduction of the organoarsenical intermediates [151] (Figure 27–3). The pathway was proposed by Challenger [152] in 1945, based on work on the fungus *Scopulariopsis*

brevicaulis [151,152]. This addition results in the formation of different compounds including methyl arsenite, dimethyl arsenate, dimethyl arsenite, trimethyl arsine oxide (TMAsO), glutathione, and other thiol-containing compounds. *S*-adenosylmethionine is required as the source of methyl groups [19] in anaerobic bacteria. An arsenite *S*-adenosylmethionine methyltransferase, catalyzing transfer of methyl groups from *S*-adenosylmethionine to As(III) with glutathione as electron donor, has been characterized in *Rhodopseudomonas palustris* [146,153].

Microorganisms are thought to play a key role in regenerating As(V) by demethylation of methylated arsenic species to use them as carbon and energy sources. *Alcaligenes, Pseudomonas,* and *Mycobacterium* are able to demethylate mono- and dimethyl arsenic compounds [19], and *Pseudomonas* sp. has been shown to use DMA as carbon source [154]. Recently, a two-step demethylation of mono methylarsonic acid (MMA) to As(III) mediated by two distinct microorganisms was observed in MMA-contaminated soil samples. This process involved a reduction of MMA to methylarsonous acid by *Burkholderia* sp. MR1 followed by a demethylation to As(III) by *Streptomyces* sp. MR1 [155]. However, the mechanism of demethylation is still unknown.

27.2.8 Morphological, Physiological, and Biochemical Responses

In addition to these well-characterized resistance determinants, microbes respond at morphological, physiological, and biochemical levels when challenged with As. One of the most important microbes, cyanobacteria, has high tolerance ability to As concentrations and is capable of decontaminating As from liquid culture through intracellular accumulation and promoting volatization loss. In the case of *Anabaena* sp. PCC7120, As treatment within 48–72 h produced filament fragmentation, thickening, enlargement and vacuolation of cells, transformation of bluish green cells into yellow brown cells, and dense pigmentation at one side of the cell. One of the most interesting observations was the appearance of akinete-like structures in *Anabaena* PCC7120, never reported before [156]. With the exception of respiration, PSI, PSII, whole chain, ¹⁴C uptake, NADPH and ATP contents, and nitrogenase activity were found inhibited on As treatment.

Bioaccumulation of As in *Phormidium* [157], and antioxidative enzymes of *Anabaena doliolum* [158], *Phormidium laminosum* [157] and the role of arenic resistance genes (ars operon) in *Synechocystis* [78] are also known. Proteomic and biochemical analyses revealed the role of phytochelatin, GST, phytochelatin synthase, arsenate reductase, and arsenite efflux genes *asr1102* and *alr1097* in As sequestration and shielding of the *Anabaena* sp. PCC7120 from As toxicity. These analyses further demonstrated that the *ars* genes play a central role in detoxification and survival of *Anabaena* under As stress. The proposed hypothetical model explains the interaction of metabolic proteins associated with the survival of *Anabaena* sp. PCC7120 under As stress [156] (Figure 27-4).

27.3 Arsenic Toxicity and Tolerance in planta

Plants play a major role in the entry of metals and metalloids into the food chain [159]. It is therefore worthwhile to understand the mechanisms by which they take up and metabolize As from the soil, sequester it within the plant, and detoxify it. In order to reduce the As uptake through



FIGURE 27–4 Hypothetical mechanistic model depicting the survival strategy of *Anabaena* sp. PCC7120 under As stress. Upward arrows indicate enhanced enzyme activity and protein expression. Figure modified after Pandey et al. [156].

consumption of contaminated plant foods, a better understanding of the mechanisms of As uptake, metabolism, and identification of the transporters responsible for arsenite detoxification by plants is a prerequisite. A potential outcome of this would be the generation of As-resistant plants for (1) phytoremediation and (2) safe cropping [37,160]. For remediation and restoration of As-contaminated sites, a diverse group of tolerant/resistant plants capable of growing at high As concentrations and able to (hyper)accumulate As in harvestable biomass are needed. In contrast, for safe cropping in areas where land and/or groundwater are contaminated, As-resistant plants that prevent accumulation in the harvested plant product are required. There is a considerable literature on the mechanisms pertaining to As uptake, accumulation, and detoxification in plants; the section below summarizes in brief the work done so far.

27.3.1 Arsenic Uptake and Efflux

The most common forms of As in soil solution available for plant uptake are arsenate, arsenite, MMA, and DMA. Their uptake mechanisms are described below.

27.3.1.1 Arsenate Uptake

Arsenate [As(V)] is the main As species in aerobic soils. It has a strong affinity for iron oxides/ hydroxides in soil; thus the concentrations of arsenate in soil solutions are usually low. Arsenate and phosphate share the same transport pathway in higher plants; the Pi transporter (PHT) proteins [161–165] have a higher affinity for phosphate than for arsenate [166]. Therefore, As(V) and Pi compete for uptake through the same transport systems in As hyperaccumulators, [33,167] As-tolerant non-hyperaccumulators [164,168], as well as As-sensitive non-accumulators [169,170]. Under low Pi conditions, As(V) may outcompete Pi for entry into the plant, amplifying Pi deprivation symptoms. Conversely, Pi fertilization can protect plants, including the hyperaccumulator *P. vittata*, from As(V) toxicity [167]. Increasing or decreasing the rate of Pi and As uptake by increasing or decreasing PHT protein amount or activity at the plasma membrane through genetic means can also increase or decrease, respectively, the toxicity of As(V) [165,171–173]. A number of phosphate transporters have been characterized in plants [174,175]. There are over 100 phosphate transporters in the phosphate transporter 1 (Pht1) family, most of which are strongly expressed in roots and are likely to be involved in phosphate uptake from the external medium [175].

In Arabidopsis thaliana, two phosphate transporters, Pht1;1 and Pht1;4, play a significant role in phosphate acquisition from both low and high phosphorus (P) conditions [171]. The A. thaliana double mutant $pht_{1;1\Delta 4\Delta}$ was much more resistant to arsenate than the wild type, indicating that Pht1;1 and Pht1;4 mediate arsenate uptake [171]. In the A. thaliana mutant defective in phosphate transporter traffic facilitator 1 (PHF1), the trafficking of the Pht1;1 protein from the endoplasmic reticulum to the plasma membrane is impaired [172]. This mutant was much more resistant to arsenate than the wild type, further supporting a role of Pht1;1 in arsenate uptake. Recently, Catarecha et al. [173] identified an arsenate-tolerant mutant of A. thaliana, *pht1;1-3*, which harbors a semidominant allele coding for the high affinity phosphate transporter PHT1;1. Rather intriguingly, the *pht1;1-3* mutant displays the dual phenotypes of decreased arsenate uptake in the short term and increased As accumulation over a longer period of growth. As the wild-type plants suffered from severe As toxicity, it is perhaps not surprising that their As accumulation capacity was curtailed compared with the mutant. Once inside the plant cell, As(V) can probably move easily from one cellular compartment to another, crossing internal membranes through various Pi transporters. For example, As(V) has been demonstrated to be a co-substrate for three mitochondrial dicarboxylate transporters, proteins localized to the inner mitochondrial membrane and responsible for dicarboxylate exchange with co-substrates such as Pi, between the cytosol and the organelle matrix [176]. As(V) can be found in the xylem, considerably loaded into the xylem vessels by PHT proteins [37,165,173,177]. However, roots of As non-hyperaccumulator have the ability to strongly retain As.

In *Arabidopsis*, only about 3% of the As taken up by the root was found to be translocated to the shoot [178]. Similar results have been found for other plants [179]. Following uptake of arsenate by roots, some of the arsenate is lost from the cells via efflux to the external medium [32]; this is analogous to the situation for phosphate, which can also be lost via efflux especially under high P conditions [180]. The mechanism of arsenate efflux is not known, but may be similar to that of phosphate efflux presumably via anion channels [180]. Xu et al. [32] showed

that arsenate added to the aerated nutrient solution was rapidly converted to arsenite by the roots of tomato and rice. Microbes living in the nutrient solution or root exudates contributed little to arsenate reduction to arsenite. This is surprising because arsenate is expected to be stable in the aerobic environment. Phosphate inhibits arsenate uptake and the subsequent production of arsenite in the external medium, suggesting that the arsenite is extruded by root cells following arsenate reduction inside the cells. Indeed, efflux of both arsenate and arsenite was observed when tomato roots preloaded with arsenate were transferred to an As-free medium. Furthermore, the protonophore carbonylcyanide m-chlorophenylhydrazone (CCCP) inhibited the efflux of arsenite, suggesting that the efflux is linked to the proton gradient across the plasma membranes, or is energy dependent.

27.3.1.2 Arsenite Uptake

As in microorganisms and mammalian tissues [181], arsenite enters plant root cells through nodulin 26-like intrinsic proteins (NIPs) [182-184]. These proteins belong to the aquaporin family of major intrinsic proteins. In rice roots, the OsNIP2;1/OsLsi1 silicon transporter has been implicated as the main As(III) uptake protein, while As(III) efflux from rice root cells to the xylem takes place through the OsLsi2 silicon transporter [184]. Other types of proteins may facilitate the transport of As(III) into cells. In As hyperaccumulating species, such as P. vittata, As is not immobilized in the roots, but rapidly transported as As(III) through the xylem to the fronds [185–187]. In the fronds, As(III) is sequestered as free As(III) in the vacuole [185,186], where it accumulates to extremely high levels [188]. It has been shown that PvACR3 is involved in the vacuolar sequestration of As(III) [189]. This protein is a homologue of the yeast ScACR3-p protein, a plasma membrane protein responsible for the efflux of As(III) from the yeast cell. In *P. vittata*, the PvACR3 protein still acts to give rise to efflux of As(III) from the cytosol, but instead of delivering the As(III) to the outside of the cell, PvACR3 resides on the vacuolar membrane and transports the As(III) into the vacuole. Single-copy ACR3 genes are found in moss, lycophytes, ferns, and gymnosperms, but not in angiosperms, which may help explain the lack of As hyperaccumulators among the angiosperms [189].

27.3.1.3 Uptake of Methylated Arsenic

Like As(III), the protonated, uncharged forms of the methylated As species MMA(V) and DMA(V) enter rice roots at least in part through the aquaporin channel OsLsi1 [190]. However, the rate of uptake for MMA(V) and DMA(V) is much slower than that of As(III) or As(V) [191-193]. In contrast, the mobility within the plant of MMA(V) and DMA(V) appears to be substantially greater than that of As(III) or As(V) [190,192,194–197]. While OsLsi1 has been implicated in the uptake of both MMA(V) and DMA(V), OsLsi2 does not seem to be involved in their efflux [190]. In animal systems, MMA(III) is also transported through aquaporins; however, it is not yet known whether OsLsi1, OsLsi2, or AsIII transporters are able to transport MMA(III). Arsenic can also be metabolized by various organisms to form arsenocholine, arsenobetaine, and arseno-sugars. These compounds have been detected in some terrestrial plants. However, it has not been demonstrated that these compounds can be produced by the plant, or whether they are simply taken up in those forms from the soil. The effects of these compounds on plant metabolism are largely unknown.

27.3.1.4 Long-Distance Transport

In most plant species analyzed, arsenite dominates in the xylem sap, suggesting that it is the main form loaded into the xylem [188]. This is also true when arsenate is supplied to plant roots, and is consistent with the fact that roots have a high capacity for arsenate reduction. There is no evidence that arsenite in the xylem sap is complexed with thiol compounds [30,31]. In fact, complexation with thiols decreases arsenite mobility from roots to shoots. Rice loads arsenite into the xylem more efficiently than does wheat or barley or any other crop plants [198]. P. vittata has an exceedingly efficient system to load arsenite into the xylem [187], but the underpinning mechanism has not been elucidated. Little is known about phloem transport of As, such as the form of As transported and the transporters involved in phloem loading and unloading. In a recent study using rice panicles excised below the flag leaf node, Carey et al. [196] found that DMA was transported to the immature grain approximately 30 times more efficiently than arsenite. When the phloem flow was disrupted by stem girdling, transport of arsenite into the grain was decreased by 10-fold, but that of DMA by only 50%. These results suggest that arsenite is delivered to rice grain mainly through the phloem, whereas both phloem and xylem pathways make an equal contribution to the transport of DMA to grain. Further evidence from a synchrotron μ -XRF study indicates that arsenite accumulates in the ovular vascular trace of the grain, whereas DMA permeates into the outer layer of the endosperm.

27.3.2 Arsenic Toxicity to Plants

27.3.2.1 Impact on Morphology, Growth, and Productivity

Exposure of plants to toxic levels of As triggers a wide range of morphological, physiological, and metabolic alterations [199–202]. The typical symptoms shown by As-stressed plants are interveinal necrosis [203,204], whitish chlorosis, and wilting [201,202]. Arsenic inhibits root extension and proliferation [205]. Most plants possess mechanisms to retain much of their As burden in the root. However, a genotype-dependent proportion of the As is translocated to the shoot and other tissues of the plant. Once in the shoot, As can severely inhibit plant growth by slowing or arresting the expansion and biomass accumulation, as well as compromising plant reproductive capacity through infertility, lack of germination, and loss of yield and fruit production [206]. Carbonell-Barrachina et al. [199] reported that As-exposed bean plants showed average reductions of 50% and 84%, respectively, in foliar mass (leaf dry weight) and fruit yield (fruit dry weight) as compared to controls. Similarly, Shaibur et al. [201] reported that in sorghum *(Sorghum bicolor)*, shoot and root dry matter yield were repressed by higher As levels. A decrease in growth of both the vegetative and root system of tomato plants was also noticed at higher As concentrations [207].

Straight head disease is a physiological disorder of rice (*Oryza sativa* L.), characterized by sterility of the rice florets/spikelets leading to reduced grain yield. It was observed that BRRI dhan 29, a popular Bangladeshi rice strain, when grown in soils spiked with As showed the severity of the disorder in a dose-dependent manner [204,208]. Low As burden causes the number of nitrogen-fixing root nodules to be repressed in soybean [209]. One of the many interesting paradoxes related to As toxicity is that plant growth is stimulated at low As concentrations [206,207,210–212]. The fact that this phenomenon occurs under axenic conditions in cultured plants, such as *Arabidopsis thaliana* [213], indicates that the trait is not based on As disrupting plant-biotic interactions. Instead, it results either from a direct interaction of As with plant metabolism, or from an interaction of As with plant nutrients. While the mechanism is not fully understood, it has been suggested that the growth benefit arises from As stimulation of Pi uptake [167]. It has been shown that in the presence of P fertilization, the negative effects of As on growth and productivity are ameliorated [168,214–216]. Liu et al. [217] reported a significant decline in root biomass in wheat seedlings and grain yield with increase in As(III) and As(V) concentrations for all six varieties of *Triticum aestivum* studied. However, in contrast, plants supplemented with P showed only a slight or moderate decrease in grain yield. In tune with this, in winter wheat As caused a slight reduction in both root and shoot biomass. P supply, in contrast, markedly increased shoot dry weights with a slight reduction in root dry weights [218]. A decrease in plant biomass with increasing As concentration in irrigation water has also been reported by Pigna et al. [216], and this reduction was less severe in added P conditions.

27.3.2.2 Physiological and Biochemical Responses

At sufficiently high concentrations, As interferes with critical metabolic processes, which can lead to plant death. Arsenic damages the chloroplast membrane and disorganizes photosynthetic function [219]. A significant decrease in pigment synthesis, CO_2 fixation, and activity of photosystem II (PSII) was observed in oats growing under As stress [219]. Singh et al. [203] reported that As exposure did not significantly reduce the concentrations of chlorophyll a and b, and total chlorophyll in *P. vittata* regardless of the exposure time; however, it reduced those in *P. ensiformis* after 10-d exposure, proving *P. vittata* to be resistant to As. Milivojevic et al. [220] reported that the changes of maximum quantum yield of PSII (ϕ) and photochemical quenching of fluorescence q_p under the highest As concentrations point to the changes of the redox stable state of the physiological quenching of fluorescence, which was not significantly modified by As application. Cellular membranes get damaged in plants exposed to As, causing electrolyte leakage [203], often accompanied by an increase in malondialdehyde contents. Arsenic exposure induces antioxidant defense mechanisms. The synthesis of ascorbate, the γ -Glu-Cysglytripeptide GSH, and the GSH oligomer ((γ -Glu-Cys)*n*-Gly) phytochelatin increases throughout the plant, but more so in the roots [202,203,215,221-223], while anthocyanin accumulates in leaves [173,224]. Plant transpiration intensity may be reduced [219].

27.3.2.3 Proteomic Responses of Plants to As Stress

Proteomic studies on As-induced alterations are still few and limited to maize roots [225], maize shoots [226], rice roots [227] rice leaves [228], a pseudometallophyte *Agrostis tenuis* [229], hyperaccumulator species *Pteris vittata* fronds and roots [230,231], and to the moderate As tolerant/accumulator *Artemisia annua* [224]. Expression patterns of maize (*Zea mays*) root and leaf proteins in response to As stress were described for the first time by Requejo and Tena [225,226]. It was found that seven of the 11 identified proteins were involved in cellular homeostasis for redox perturbation. This suggested that oxidative stress is a major process underlying As toxicity in plants. To figure out the precise mechanisms underlying the

toxicity of As, a comparative proteomic analysis of rice roots in combination with physiological and biochemical analyses was done by Ahsan et al. [227]. The comparative proteomic analyses identified 23 differentially expressed proteins in rice roots, including those predicted as S-adenosylmethionine synthetase (SAMS), cysteine synthase (CS), and novel proteins including tyrosine-specific protein phosphatase protein, and omega domain containing GST. These differentially expressed proteins are functionally involved in cell signaling, stress and detoxification, defense and development, and protein biosynthesis. Ahsan et al. [228] also reported the first proteome map of rice leaves under As stress along with physiological and biochemical responses. The increased activity of several proteins associated with energy metabolism, such as NADP-dependent malic enzyme, NAD-dependent formate dehydrogenase, and glyceraldehyde-3-phosphate dehydrogenase suggest that an increased amount of energy is required to adapt to As stress. However, the down-regulation of RuBisCO and chloroplast 29kDa ribonucleoproteins might be the possible causes of the decreased photosynthesis rate under As stress. Contrary to this finding, Duquesnoy et al. [229] identified a set of Agrostis tenuis leaf proteins differentially expressed in response to As exposure including a major functionally homogeneous group of enzymes including oxygen-evolving enhancer protein, RuBisCO small and large subunits, RuBisCO activase, and ATPsynthase involved in the Calvin or Krebs cycle. Bona et al. [230] also demonstrated the protein expression profile of *P. vittata* fronds in plants inoculated with one of the two arbuscular mycorrhizal (AM) fungi (Gigaspora mosseae or Gigaspora margarita) with and without As treatment. Recently, Rai et al. [224] demonstrated that proteins of photosynthesis and carbon metabolism play a key role in augmenting As tolerance to A. annua. The results also suggested that ABC-transporter like protein may be involved in As sequestration into vacuoles as the final detoxification step.

27.3.3 Mechanism of As Toxicity

The results from a number of hydroponic experiments have always been a matter of debate; while they agree that As phytotoxicity depends on the chemical species supplied to the plant, they disagree on the identity of the most phytotoxic form of As [193,194,199]. Hydroponic experiments are considered better than soil studies in detecting metal-plant interaction because they eliminate the complex and confounding phytoavailability issues that arise from differences in the mobility of various As species through the diverse growth substrates and thus provide a clear insight into the potency of externally supplied As on plant growth. The studies generally agree with the hydroponic survey of 46 different plant species [192] that the uptake of As by plants has the order As(III) > As(V) > MMA(V) > DMA(V), while translocation from the roots to the rest of the plant has the order $DMA(V) > MMA(V) > As(V) \ge$ As(III). However, no As form appears to be consistently most phytotoxic. In two Spartina species, where the order of uptake was $As(III) > As(V) \approx MMA(V) > DMA(V)$, the order of phytotoxicity was $DMA(V) \approx MMA(V) > As(III) \approx As(V)$ [199]. This would suggest that DMA(V), with lowest uptake and high phytotoxicity, exerted maximum toxic effects within the plant. In contrast, the uptake order in rice was As(III) > MMA(V) > As(V) > DMA(V), similar to the order of phytotoxicity, which was MMA(V) > As(III) > As(V) = DMA(V) [194]. Finally, the order for phytotoxicity in maize, a species with the typical order for uptake [192], was As(V) > As(III) > DMA(V) [193]. The inconsistent order of phytotoxicity of the various As species could be an indication that As has interacted differently with the available nutrients, or that the phytotoxic form of As is plant species dependent. Alternatively, the apparent inconsistency of the above results may be due to an incomplete understanding of the relative importance of the mechanism of As toxicity. After all, the mechanism through which As causes phytotoxicity has not been definitively identified and the exact species of As being the primary cause of toxicity is unknown. In this regard, the finding of MMA(III) in plants [179,232,233] is particularly intriguing. This As species is up to 18 times more cytotoxic to animal cells than As(III) [234,235]. The phytotoxicity of MMA(III) has not been tested.

Another issue that needs to be included when considering the toxicology of As is that some forms of As may be under-represented or not addressed in terms of As speciation in plants. In a study on sunflower, 15-20% of the total tissue As was not recovered from the plant tissue [31]. In beet root, As was nearly quantitatively extractable from tissues with low As burden, but in tissues with higher As burdens, up to 75% of the As was not extracted [236]. Perhaps the speciation of this missing As, or indeed the molecules to which it may be bound, will give us important insights into the underlying mechanism of As toxicity. The above arguments indicate that there is a need of fuller understanding of the action of As at cellular level, although the basic mechanism of As toxicity is conserved from microbes to the plants. The structural similarities of As(V) and Pi allow this form of the metalloid to substitute for Pi in biochemical reactions, potentially disrupting vital cellular processes. On the other hand, As(III) is highly reactive toward thiol groups, as are MMA(III) and DMA(III). The binding of As(III) to the thiol groups of proteins or enzyme cofactors may alter or inhibit their activity, also exposing cellular processes to risk. Oxidative stress brought about by the inevitable production of ROS during As exposure has recently gained favor as it is a driver of As toxicity in plants. However, which cellular processes are most sensitive to As toxicity and which As species pose the greatest threats to plant cell health remains unclear.

27.3.3.1 Phosphate Replacement

Substitution of Pi by As(V) has been demonstrated in numerous biochemical reactions, and any reaction with Pi or a Pi-ester as a substrate is a potential target for As(V) disruption [237–240]. Potential As(V)-sensitive reactions would include those central to cellular metabolism, information storage and retrieval, and cellular signaling (i.e., protein phosphorylation/ dephosphorylation). When As(V) comes into contact with the surface of a cell of a plant root, the Pi transporter may be the first enzyme where As(V) may compete with Pi as observed in numerous monocots and dicots, and As-hyperaccumulators and non-hyperaccumulators [33,161,164,167,168,170]. Since Pi competes with As(V) for uptake, As(V) toxicity is lower under high Pi conditions. On the other hand, As(V) may outcompete Pi for uptake under low Pi conditions, exacerbating Pi deprivation [167]. The toxicant can also pass through the *Arabidopsis* AtPHT4 family of Pi transporters localized to the plastid and Golgi [241]. Relatively few enzymes use Pi as a substrate due to the reversible nature of most Pi-liberating reactions. Therefore, few enzymes are expected to use As(V) directly as a substrate [242]. Perhaps the

predominant Pi-requiring reaction is the phosphorylation of ADP to ATP by the F1F0-type ATP synthases found in the mitochondrial inner membrane and the thylakoid membrane. The mitochondrial enzyme uses As(V) in a reaction that produces ADP-As(V) [240]. The *K*m and *V*max of this reaction are remarkably similar for both Pi and As(V) [243], demonstrating that at least some enzymes are capable of recognizing and reacting equally well with As(V) and Pi. Other Pi-dependent enzymes capable of using As(V) include the glycolytic enzyme glyceraldehyde-3-phosphate dehydrogenase (GAPDH). Like the ATP synthase reaction, the GAPDH reaction, where As(V) replaces Pi, has remarkably similar kinetic constants to those of the Pi-dependent reaction [237]. Aspartate- β -semialdehyde dehydrogenase has a critical role in the biosynthesis of essential amino acids in plants, catalyzing the reversible reductive dephosphorylation of β -aspartyl phosphate to L-aspartate- β -semialdehyde. This enzyme, too, is able to use As(V) nearly as efficiently as Pi, judged from the *K*m and *K*cat values [244].

While much of the above information comes from non-plant systems, there is little reason to believe that the behavior of homologous plant enzymes would be substantially different. The As(V)-esters produced by these and other reactions are highly unstable in water and undergo spontaneous hydrolysis [242,245]. The hydrolytic products are generally free of As(V) and the corresponding carbon compound. The instability of As(V)-esters is highlighted by the rates of hydrolysis for glucose-6-As(V) and ADP-As(V), which are at least 105-fold greater than for the corresponding Pi-ester [243]. As a consequence of the instability of As-esters, the enzymatic reactions that produce them are essentially irreversible. The products simply do not stay around long enough to allow the reverse reaction to proceed at an appreciable rate. The unstable nature of the As(V)-esters also creates futile reaction cycles around enzymes that use free As(V) to produce As(V)-esters. The As(V) and ADP released by the auto hydrolysis of ADP-As(V) produced by mitochondrial and plastid ATP synthases can be used in subsequent reactions. Such futile cycling uncouples respiratory electron transport in the mitochondrial inner membrane and photosynthetic electron transport in chloroplast thylakoid membranes from ATP synthesis [240,246–249]. This collapse of ATP production has potentially grave consequences for the energy status of the cell. While As(V)-esters are unstable, enzymes can utilize them when they are available. Hexokinase can use ADP-As(V) to arsenolate glucose to glucose-6-As(V). This product, in turn, is a substrate for glucose-6-phosphate dehydrogenase [240].

27.3.3.2 Binding Thiol Groups

The mode of action of As(III) differs substantially from that of As(V). As(III) is a thiol reactive compound that can bind up to three sulfhydryl groups [250]. This allows As(III) to act as a cross-linking agent by binding up to three monothiol molecules, such as the antioxidant GSH. Alternatively, it could bind to a single molecule of a polythiol compound, such as PC, the Cysrich polymerization product of GSH. As(III) can also bind to thiol-containing proteins and cofactors. Dihydrolipoamide, which in plants is a cofactor associated with the mitochondrial and plastid pyruvate dehydrogenase complexes (mtPDC, ptPDC), the 2-oxo glutarate dehydrogenase complex (OGDC), the glydecar-boxylase complex (GDC), and the branched-chain 2-oxoacid decarboxylase complex (BCOADC), has been long thought to be an important cellular target for As(III) binding [251,252]. The stability of As(III) complexes increases with the number of bonds formed. The half-life of an As(III)-monothiol peptide complex is about 1-2s. The half-life increases to about 1.3 and 155 min when two or three intramolecular thiols are bound [250]. The binding of As(III) to dithiols is enhanced when the sulfhydryl groups are in close proximity to one another [253], but the optimal spacing for trithiols is unknown. The binding of As(III) to protein can have profound effects on their folding [254,255]. In various systems, proteins that are known to bind As(III) include transcription factors, signal transduction proteins, proteolytic proteins, metabolic enzymes, redox regulatory enzymes, and structural proteins. Among the 35,386 predicted translation products from the Arabidopsis genome sequence (TAIR 10 release), there are 64,335 dithiols with optimal spacing for As(III) binding [250] on 23,578 proteins. About one-third of these dithiols, residing on 11,559 proteins, form part of a trithiol that may be optimally spaced for As(III) binding, assuming that the optimal sulfhydryl spacing for As(III) binding to trithiol groups is symmetrical with the dithiol spacing (CX0-14CX0-14C). This analysis ignores the potential for intramolecular cross-links, but raises two intriguing questions: What types of proteins in Arabidopsis are among the 2123 proteins lacking a Cys residue? Are there evolutionary pressures for these proteins, in particular the lack of ability to interact with As(III)? One conclusion to be drawn from these values is that As(III) has the capacity to interact with a large proportion of any cellular proteome and it will be a large task to identify which proteins among the As(III) targets are most critical to cell survival. While As(III) is an inhibitor of many enzymes, the recent finding of methylated forms of As(III) in plant cells [31,179,232,233] has important implications in this respect. Half maximal inhibition of pyruvate dehydrogenase was found to occur at about 115 µM As(III), while two- to six-fold fewer methylated As(III) derivatives were needed for similar inhibition [256]. Compared to As(III), MMA(III) is a more potent inhibitor of other enzymes including GSH reductase [257] and thioredoxin reductase [258]. MMA(III) and DMA(III) forms were also more potent inhibitors of zinc-finger protein activity than As(III), again highlighting the necessity to critically evaluate the ability of plants to methylate inorganic As into more toxic forms or reduce methylated As(V) compounds to their As(III) counterparts.

The binding of As(III) to thiols is the basis for the main detoxification pathway for both As(V) and As(III) and explains the retention by roots of up to 90% of the As taken up by a plant [178,192,259]. As(III) is either taken directly into the roots from the soil solution or rapidly produced within the root by the efficient reduction of As(V). As(III) then combines rapidly with sulfhydryl-rich protective molecules like GSH and PC [30,221,260]. The preference of As(III) to bind polythiols favors the formation of As(III)-PC3 complexes [31]. However, a variety of As(III) conjugates can exist in a plant, with 14 different complexes being isolated from sunflower [31]. The As(III)-PC conjugates can then be transported from the cytosol into the vacuole. In *Arabidopsis*, this transport is via the ABC transporters MRP1/ABCC1 and MRP2/ABCC2 [261]. Homologs of *Arabidopsis* ABCC1 and ABCC2 are found throughout the plant kingdom [177]. A homologue of ABCC2 was among several ABC transporters to be up-regulated at the transcript level in response to As in rice [262,263]. Thus, in many plants, As(III) present in root cells is rapidly complexed to PC and sequestered in the vacuole, restricting the transport of As from the root [177,260] and preventing its interaction with cellular metabolism. The impact

of As on cellular metabolism in many plants, then, depends on how efficiently As(III) can be neutralized by thiol binding and sequestration in the vacuole. Interestingly, this is not the case in hyperaccumulating species such as *P. vittata*, where there is little PC binding of As(III) in the roots [264]. Instead, As(III) taken into the plant cell from the soil solution or produced by the reduction of As(V) is rapidly translocated to the xylem and fronds for sequestration as free As(III) in the vacuole [185,187,190]. This transport of As(III) through the plant body raises the question of how *P. vittata* avoids the toxic effects of As(III) binding to enzymatic thiols.

27.4 Mechanisms of As Tolerance and Detoxification

Plant tolerance of any heavy metal/metalloid is governed by a network of physiological and molecular mechanisms, and understanding of these mechanisms and their genetic basis is an important aspect of developing plants for phytoremediation [265,266]. Different plant species may have evolved different mechanisms to tolerate excess arsenic, and even within plant species more than one mechanism could be in operation. Physiological, biochemical, and molecular approaches continue to be employed to identify the underlying mechanisms of As accumulation, tolerance, and adaptive mechanisms to cope with As stress. Resistance to As toxicity can be achieved by "avoidance" when plants are able to restrict As uptake, or by "tolerance" when plants manage to survive in the presence of high As concentrations. Avoidance involves reducing the entry of As in the cell by biosorption to cell walls, reduced uptake, or increased efflux. However, in the second situation, plants detoxify the metals through intracellular chelation, via the synthesis of amino acids, metabolites, organic acids, GSH, or HM-binding ligands such as metallothioneins and phytochelatins, compartmentation within vacuoles, and up-regulation of the antioxidant defense. There are relatively few species of plants that are naturally As tolerant. Among these are *Pteris vittata* and other members of the Pteridaceae that hyperaccumulate As [34,113,186,188,267]. The growth of these plants is not compromised during As accumulation. In contrast to As non-hyperaccumulating plants, hyperaccumulators tend not to restrict As to the roots, instead allowing transfer of the toxicant immediately to the shoots. This is likely to be an important aspect of the hyperaccumulation phenotype. The mechanisms by which these plants are able to hyperaccumulate As are being elucidated, but it is not yet clear as how they avoid As toxicity.

27.4.1 Arsenate Reduction

It has been observed that when plants are supplied As(V), typically more than 90% of the As in the roots and the shoots was found to be in the form of As(III) [29,30,32]. Thus, As(V) is readily reduced to As(III) by plants. This reduction is accepted as the first step in the major As detoxification pathways found in plants [30,221]. The reduction of As(V) to As(III) occurs both enzymatically and non-enzymatically. In the non-enzymatic pathway, two molecules of GSH are able to reduce As(V) to As(III). The oxidation of GSH is via the formation of a disulfide bond, producing a GSH dimer (GSSG) which can be rapidly recycled to two GSH molecules by GSH reductase [268]. While As(V) reduction can occur non-enzymatically, the enzymatic

rate is much higher [269]. As(V) can be directly reduced to As(III) by arsenate reductase (ACR), an enzyme first isolated from bacteria and yeast [89]. Based on homology between the yeast ACR gene *ScAcr2* and several homologous sequences from plants, an *HlAsr* cDNA was cloned from As(V)-hypertolerant *Holcus lanatus* and found to encode an enzyme with ACR activity [270]. The homologous proteins from *Arabidopsis* (AtAsr/AtACR2), *P. vittata* (PvACR2), and rice (OsACR2.1 and OsACR2.2) also have ACR activity [29,113,271]. The plant ACR2 protein is related to the CDC25 cell cycle dual specificity tyrosine phosphatases. Interestingly, AtACR2 has phosphatase activity, while the PvACR2 enzyme, like the yeast ScAcr2p protein, does not. Also like ScAcr2p, the plant ACR2 enzyme uses GSH and grx as electron sources [113,271], suggesting that the catalytic cycle involves the formation of a mixed disulfide between GSH and ACR2 that is resolved by grx [89]. The *Arabidopsis* As(V) reductase activity has an As(V)-inducible component that has been attributed to *AtACR2*, as well as a constitutive component that is not diminished in *AtACR2* T-DNA insertion lines [270]. Moreover, As(III) has been stated to remain the predominant form of As present in *AtACR2* T-DNA insertion lines supplied with As(V) [188].

Together, these results indicate that *Arabidopsis*, and thus in all likelihood other plants, possesses enzymes in addition to ACR2 that have As(V) reductase activity. Multiple enzymes from other systems have been shown to exhibit As(V) reductase activity. These include GAPDH, polynucleotide phosphorylase, purine nucleoside phosphorylase, glycogenphosphorylase, and the mitochondrial F1F0 ATP synthase [272-274]. Each of these enzymes can incorporate As(V) instead of Pi into biological molecules, forming an arseno ester that would readily undergo hydrolysis. In the presence of a suitable thiol group, for example GSH, the hydrolysis can result in the reduction of As(V) to As(III). It is not known if the analogous plant enzymes can also reduce As(V) in the presence of thiols. However, one form of the plant GAPDH is known to interact with GSH [275], suggesting it as a candidate ACR. Moreover, acytosolic triose-phosphate isomerase (cTPI) from P. vittata has also been shown to have ACR activity [276]. Because the TPI reaction does not involve the transfer of a Pi group, the mechanism by which Pvc TPI promotes the production of As(III) is unclear. However, like the enzymes mentioned above, the plant TPI interacts with GSH [276]. The number of enzymes that could misincorporate As(V) for Pi, and therefore have the capacity to form arseno esters, is large, providing many opportunities for the enzymatic reduction of As(V) to As(III). However, it is not known whether these enzymes affect the redox status of As in vivo.

27.4.2 Complexation and Sequestration of As

There is strong evidence that complexation of arsenite by PCs is an important mechanism of As detoxification, and hence tolerance, in As-non-hyperaccumulating plants. Arsenite has a high affinity to the sulfhydryl (-SH) groups of peptides such as GSH and phytochelatins. *In vitro* studies showed that GSH and arsenite form a (GS)3-arsenite complex with cysteinyl sulfhydryl as the arsenite binding site [277]. The complex is stable in the pH range from 1.5 to 7.0–7.5, but dissociates at higher pH. The tripeptide glutathione (Glu-Cys-Gly) is synthesized by gamma-glutamyl cysteine synthetase (g-ECS) and glutathione synthetase (GS). GSH can bind to several

metals and metalloids and is also a key metabolite in cellular redox balance. Increasing GSH synthesis is considered a means of increasing metal(loid) binding capacity as well as a way to increase cellular defense against oxidative stress. Since glutathione is the precursor of phytochelatin, overexpression of g-ECS or GS usually leads to higher rates of phytochelatin accumulation under metal exposure. It has been suggested that phytochelatin production, resulting in oxidized glutathione depletion, could itself cause oxidative stress [278,279].

In non-resistant plants, As(V) influx is so high that detoxification by phytochelatin complexation cannot be achieved, leading to considerable oxidative stress, which is not observed in resistant plants [280,281]. Phytochelatins are a family of cysteine-rich thiol-reactive peptides, which play important roles in processing many thiol-reactive toxicants [222]. Phytochelatins are induced by a range of metals and metalloids including Cu, As, and Cd [282]. It has been demonstrated that arsenic-sulfhydryl (-SH) complexes can be transported into the vacuoles by ycf1p transporters and confer As(III) resistance in yeast [87], and it has been suggested that higher plants may have a similar mechanism in As(III) resistance [283]. Raab et al. [31] have shown that As preferentially binds to PC3, forming the As-PC3 complex, rather than to PC2 and oxidized glutathione. The As-PC3 complex is the dominant complex formed in the As-tolerant H. lanatus. In contrast, PC2 was the predominant species induced by As in Raufolia serpentine [221], Silene vulgaris [284], and in the As-tolerant clone H. lanatus [204,285]. In the As-tolerant C. striatus, PC4 was the major species [168]. However, in the As-sensitive clone of H. lanatus, PC3 was dominant and PC4 remained at a low concentration [285]. Chickpea plants were capable of forming both phytochelatins and desglycyl peptides of phytochelatins in response to As, whereas PC3 and the corresponding desglycyl peptides showed a preference for As complexes [286]. Roots of As-exposed Helianthus annuus contained up to 14 different As species, including the complex of As(III) with two (c-Glu-Cys) 2-Gly molecules (As(III)-(PC2)2), the newly identified monomethylarsonic phytochelatins-2 or (c-Glu-Cys)2-Gly CH3As(MA(III)-PC2), and at least eight not yet identified species. The complex of As(III) with(c-Glu-Cys)3-Gly(As(III)-PC3) and the complex of As(III) with glutathione and (c-Glu-Cys)2-Gly(Gs-As(III)-PC2) were present in all samples taken from plants exposed to As. No As-phytochelatin complexes were found in sap, in contrast to roots, stems, and leaves, which is unequivocal evidence that As-phytochelatin complexes were not involved in the translocation of As from root to leaves of *H. annuus* [31].

Leguminous plants synthesize homo-phytochelatins in addition to phytochelatins during As stress, and many studies point to the essential role of phytochelatins in both constitutive and adaptive tolerances to As. The results presented by Schulz et al. [287] provided evidence for the phytochelatins in the detoxification of As in six non-hyperaccumulating plant species, *Agropyron repens, Glechoma hederacea, Leonurus marrubiastrum, Lolium perenne, Urtica dioica,* and *Zea mays,* in a pot experiment with high phosphate treatment. Raised concentrations of total phytochelatins were measured in plant species with a range of sensitivities to As at equivalent levels of As exposure, determined as the inhibition of root biomass. In addition, the production of phytochelatins as a function of accumulated As was studied. Long-term phytochelatin synthesis (over a 5-week period) was positively, but non-linearly, correlated with As, suggesting that probably not all As is bound by phytochelatins. In the more tolerant grasses,

A. repens and L. perrenne, it was chiefly the dithiol phytochelatins that were measured. In contrast, the dominant phytochelatin species in the less tolerant plants *V. dicica, G. hederacea, L. marrubiastrum*, and *Z. mays* were phytochelatins, while PC2 and PC3 were detected as well [287]. Wojasa et al. [288] determined overexpression of two phytochelatin synthase genes, *AtPCS1* and *CePCS*, in *Nicotiana tabacum* with respect to As tolerance and accumulation, and observed how responses relate to non-protein thiol metabolism. The expression of both genes resulted in increased As tolerance, with CePCS plants most tolerant. At the less toxic 501M As(V), AtPCS1, and CePCS transformants accumulated more As in roots and leaves than wild type. An increase in phytochelatin production and the level of PC2 species were detected in leaves of AtPCS1 and CePCS plants, which might explain their enhanced As accumulation and tolerance. In contrast, at the highly toxic 2001M As(V), several disturbances in thiol metabolism of phytochelatin synthase overexpressing plants were found. The increase in As tolerance and accumulation due to *AtPCS1* and *CePCS* overexpression, observed at the As(V) concentrations similar to those found in As-contaminated soils, make these genes promising candidates for plant engineering for phytoremediation

27.4.3 Antioxidative Defense System

Despite being a non-redox active metalloid, As promotes generation of ROS, which causes oxidative stress and damage to lipid, protein, and DNA [289,290]. To combat these effects, enzymatic, e.g., superoxide dismutase (SOD), catalase (CAT), ascorbate peroxidase (APX), guaiacol peroxidase (GPX), and glutathione reductase (GR), and non-enzymatic antioxidants, e.g., GSH, and carotenoids, are mobilized to quench ROS. SOD catalyzes conversion of the active superoxide radicals to hydrogen peroxide [223]. SOD activity in plants varies quite widely with As treatment. In some plants, like Zea mays, As-sensitive clones of H. lanatus, and the As-hyperaccumulator P. vittata, the enzyme is induced by low As exposure, and either stays at the same or decreased level in activity at higher As levels [278,291]. In Arabidopsis, genes encoding the three classes of SOD (FeSOD, MnSOD, Cu/ZnSOD) responded to As(V) differentially at the transcript level [292]. Transcripts for genes encoding a chloroplastic and a cytosolic Cu/ZnSOD were induced more than two-fold by As(V) exposure, while transcripts for an FeSOD were down-regulated about five-fold [292]. The conversion of H_2O_2 into water is brought about by CAT and GPX in peroxisomes and cytoplasm. Nevertheless, this conversion may also be done by APX in the ascorbate-glutathione cycle. Glutathione reductase is another important enzymatic component of the antioxidative defense system, which helps in maintaining a high GSH/GSSG ratio [291] of non-enzymatic antioxidants that include GSH, PC, ascorbate, carotenoids, and anthocyanin. These antioxidants generally accumulate during As exposure [168,221,261,270,278]. The production of these molecules requires metabolic acclimations, including the diversion of carbon, nitrogen, sulfur, and metabolic energy from normal growth and development. GSH and ascorbate are fairly unique among the non-enzymatic antioxidants as they can form a redox cycle. The ROS produced during As treatment typically induces an increase in the oxidation state of the redox active pools of GSH and ascorbate in favor of GSSG dimers and dehydroascorbate over the more reduced GSH and hydroascorbate [203]. This shift in redox state arises at two levels. Superoxide and the hydroxyl radical can directly oxidize both GSH and ascorbate. In this way, GSH and ascorbate act as nucleophilic scavengers.

Alternatively, H_2O_2 can oxidize GSH and ascorbate through the action of specific peroxidases, or in the case of GSH, also through the action of GRXs and GSH-S-transferases (GST). Like SOD and catalase, GST, GRX, and/or peroxidase transcript or protein abundance, or enzymatic activity, often increase in response to As exposure [219,227,262,263,292]. As an example, in rice, at least 10 GST genes are up-regulated in response to AsV exposure, while no more than two GST genes are down-regulated [262,263]. Changes in GST gene expression do not seem to have as pronounced a role in the As(III) response, as fewer transcripts changed in abundance [263], potentially highlighting the differential effects of the two inorganic As forms on cellular metabolism. It would be worthwhile to address this by asking: Why are there changes in the isoforms of the various enzymes that are expressed? What are the energetic and metabolic costs and benefits of such shifts? The second component of the two-component H₂O₂ neutralizing system is made up of monodehydroascorbate reductase, dehydroascorbate reductase, and GSH reductase. Together, these enzymes efficiently recycle oxidized GSH and ascorbate to allow further cycles of H_2O_2 reduction. The reduction of H_2O_2 through the interdependent ascorbate-GSH cycle requires reducing power in the form of NAD(P)H, diverting this energy from other metabolic processes. The enzymes involved in the recycling of oxidized GSH and ascorbate are also often induced upon exposure of plants to As [223,227]. Thus, the interdependent ascorbate-GSH cycle, if established, has an important role in maintaining ROS balance in plants [268]. The reliance of the ascorbate–GSH cycle on the diversion of carbon to ascorbate biosynthesis, plus the diversion of carbon, nitrogen, and sulfur in the form of Glu, Cys, and Gly to support the biosynthesis of GSH and PC, requires modeling of metabolism to focus on the production of the precursors for these compounds. The oligomerization of GSH to produce PC is also induced during As exposure [203,221,223]. Increased PC synthase in Arabidopsis, Brassica juncea, and tobacco conferred increased As tolerance on these plants [222,288,293]. However, at higher levels of As, thiol metabolism was disrupted in some overexpressing lines [288].

27.4.4 Osmolyte Accumulation

Besides activating antioxidative defense machinery, plants tend to accumulate certain metabolites of low molecular weight known as compatible solutes. They allow plants to acclimate to unfavorable environments for continued survival and growth. These differ among plant species and include phytopolyhydroxylated sugar alcohols, amino acids and their derivatives, tertiary sulfonium compounds, and quaternary ammonium compounds [294]. Proline accumulates heavily in several plants under stress, protecting plants from damage caused by reactive oxygen species. It plays important roles in osmoregulation, stabilization of the machinery of protein synthesis, regulation of cytosolic acidity, and scavenging of free radicals [295].

N,N,N-Trimethylglycine (hereafter glycine betaine) occurs naturally in a wide variety of plants, animals, and microorganisms [296]. Numerous experiments *in vitro* have indicated
that glycine betaine acts as an osmoprotectant by stabilizing both the quaternary structure of proteins and the highly ordered structure of membranes against the adverse effects of salinity and temperature [297]. In photosynthetic systems, glycine betaine provides protection to ribulose-1,5-bisphosphate carboxylase/oxygenase (RUBISCO) and the oxygen evolving photosystem (II) complex from salt-induced inactivation and dissociation into subunits [298]. In plants, polyamines are related to various kinds of environmental stresses including osmotic stress, salt stress, acid stress, heavy metals like As, and UV radiation [299]. Polyamines have also been suggested to function as metal chelators and protect plants against metal-induced oxidative stress. The soluble protein content in the plant cells is an important indicator of their physiological state. The As toxicity in germinating rice seeds causes suppression in the activities of starch hydrolyzing enzymes α -amylase, β -amylase, and starch phosphorylase in endosperms while it leads to an increase in the ability of acid invertase in endosperms as well as embryo axes [300]. Recently, Rai et al. [202] demonstrated an increased production of artemisinin (secondary metabolite) as part of a defense strategy adopted by *Artemisia annua* against As toxicity.

27.4.5 Mycorrhization in Crop Plants: the Prospects of Arbuscular Mycorrhizae Symbiosis in Regulation of Plant Defense Response to Arsenic Toxicity

There is a considerable literature regarding symbiotic association between mycorrhizal fungi and plants where this symbiosis benefits host plants under high As conditions [204,301]. Symbiosis with mycorrhizal fungi was found to confer As resistance on the host plants. This effect may be produced by several possible mechanisms. First, mycorrhizal colonization may suppress the high affinity phosphate transport system in plant roots, leading to reduced uptake of arsenate [302]. Second, mycorrhizal fungi may enhance As resistance in the host plants by effluxing As to the external medium. This mode of action was clearly demonstrated by Sharples et al. [303], who reported that the ericoid mycorrhizal fungus Hymenoscyphus eri*cae* from an As-contaminated site developed arsenate resistance by extruding arsenite into the medium following arsenate uptake, a mechanism that is common to bacteria and fungi. Third, by enhancing P acquisition and improving the P nutrition of the host plant, mycorrhizal fungi improve plant growth, causing a dilution of As concentrations in plant tissues [217,304], or increasing cytoplasmic inorganic P. Mycorrhizal plants often have a higher P/As concentration ratio, which is likely to benefit the plant through enhanced tolerance to As. Finally, there is circumstantial evidence that mycorrhizal fungi may restrict As translocation from roots to shoots [302,304–306], although how this control is exerted is unclear and requires further investigation. Ultra et al. [305,306] reported the presence of methylated As species in the rhizosphere of sunflower (H. annuus) plants when it was inoculated with the arbuscular mycorrhizal fungus Glomus aggregatum. DMA was found at low but detectable concentrations in the rhizosphere soil but not in the bulk soil, nor in the rhizosphere soil of uninoculated plants. Mycorrhizal fungi may be able to mediate As biomethylation and release some of the methylated As in the rhizosphere. Although several studies have shown enhanced plant tolerance to As resulting from inoculation with arbuscular mycorrhizal fungi [302,304], elucidation of the effects of arbuscular mycorrhizal fungi on uptake and translocation of As and direct evidence showing that mycorrhizal hyphae are involved in the detoxification of As are still lacking.

27.5 As Hyperaccumulation and Phytoextraction

In plants, the term "hyperaccumulation" reflects the following attributes of the plants: (1) enhanced As(V) uptake, (2) decreased As(III)-thiol complexation and As(III) efflux to external medium, and (3) enhanced xylem translocation of As(III) and vascular sequestration in fronds [188]. However, while considering a plant as an As hyperaccumulator or nonaccumulator, one has to take into consideration the bioaccumulation and translocation factors [34]. Bioaccumulation factor is defined as the ratio of metal concentration in plant biomass to that in the soil, and translocation factor is defined as the ratio of metal concentration in the shoots to that in the roots [307]. Therefore, an As hyperaccumulator plant should have a bioaccumulation factor of more than 1 and a translocation factor of more than 1, as well as total accumulation of more than 1000 mg/kg As in plant biomass. Since the first discovery of As hyperaccumulation in *P. vittata* [34], a total of 12 As hyperaccumulators have been identified, all of them from within the Pteris genus. Not all of the Pteris species are As hyperaccumulators; thus, it would be interesting to investigate the phylogenetic relationship between the hyperaccumulating and non-hyperaccumulating fern species. Arsenic hyperaccumulation by Pteris species is a constitutive trait, with plants originating from As-contaminated and As-noncontaminated environments showing broadly similar hyperaccumulating abilities [308,309].

27.5.1 Mechanisms of As Hyperaccumulation

The As hyperaccumulation phenotype is a result of a combination of several physiological processes, although little is known about the molecular mechanisms. Hypertolerance is found in all naturally evolved hyperaccumulators; As hyperaccumulators have a much higher degree of tolerance than the non-hyperaccumulator species within the *Pteris* genus [310,311]. Compared with the non-hyperaccumulator Pteris ensiformis, P. vittata possesses a higher antioxidant capacity and also maintains a lower concentration of reactive oxygen species [203,312]. By contrast to non-hyperaccumulators, which rely on PC complexation for As detoxification and tolerance, very little of the As accumulated in the roots and fronds of P. vittata and *P. cretica* ($\approx 1-3\%$ of the total As) is complexed with PCs [264,313]. The majority of As (60-90%) of the total As) in the fronds of *Pteris* species is inorganic arsenite [185-187,314-316], which appears to be stored in the vacuoles [185,186]. It may be extremely costly for plants to evolve PC-dependent As hyperaccumulation, because at the 1:3 stoichiometric ratio of As:S, the amount of S required to complex 2500 mg As/kg in the fronds of *P. vittata* would have exceeded the total amount of S typically accumulated by the plants [308]. Vacuolar sequestration of arsenite is therefore the key mechanism of As detoxification in the hyperaccumulator ferns. How this is achieved, and especially the identity of the vacuolar transporters responsible for arsenite transport across the tonoplast, deserves further investigation. Arsenate is likely to be the main form of As taken up by the hyperaccumulator ferns as they grow on aerobic soils, and the



FIGURE 27–5 Schematic representation of arsenic (As) uptake and metabolism in roots of non-hyperaccumulators (A) and hyperaccumulators (B). Line thickness relates to flux rate. ABCC, ATP-binding transporters; AR, arsenate reductase; GSSG, oxidized glutathione; GSH, glutathione; PCs, phytochelatins.

uptake is via the phosphate transport system as in non-hyperaccumulator plants [317] (Figure 27–5). Compared with non-hyperaccumulator ferns, *P. vittata* and *P. cretica* have a higher arsenate influx [310,317] and a lower *K*m, indicating higher affinity of the transporter for arsenate [317]. The most striking difference lies in the efficiency of root to shoot translocation, exemplified by the large ratios of shoot to root As concentrations in the As hyperaccumulators [307,308]. The ratio of the As concentration in the xylem sap of *P. vittata* to that in the nutrient solution was about two orders of magnitude higher than that in the non-hyperaccumulators. Su et al. [187] showed that the majority (93–98%) of the As in the xylem sap of *P. vittata* was in the form of arsenite, regardless of whether the plant was treated with arsenate or arsenite.

Roots or rhizoids of *P. vittata* are likely to be the main location of arsenate reduction, with arsenite being preferentially loaded into the xylem. This is consistent with the findings of Duan et al. [269], who found the activity of the glutathione-dependent arsenate reduction only in the roots of this hyperaccumulator. Although the gene encoding an arsenate reductase has been cloned from *P. vittata* gametophytes [113], its *in planta* role and its contribution to overall arsenate reduction have not been ascertained. Also, hyperaccumulators differ from non-hyperaccumulators in that there is minimal efflux of arsenite from the roots of *P. vittata* to the external medium [187]. This, together with little PC complexation of arsenite in *P. vittata* roots [264], may explain the highly efficient xylem transfer in hyperaccumulators.

27.6 Summary Points

• Arsenic is a notorious poison that was probably more abundant on the early Earth's surface than today. Therefore, primordial organisms evolved systems to cope with it, which still exist in prokaryotes. Not only do they resist this with a wide range of mechanisms

(precipitation inside or outside the cell, extrusion or transformation), but strikingly, some of them take advantage of this toxic compound by using it as an electron donor [As(III)] or acceptor [As(V)] for energetics purposes.

- Biooxidation and bioreduction of arsenic play a key role in the mobility of this metalloid based on the different solubilities of As(III) and As(V), As(III) being more soluble than As(V). On the one hand, reduction of As(V) could lead to huge environmental pollution problems since As(III) is more toxic than As(V); on the other hand, As(III) oxidation is considered as a bioremediation process since it leads to the production of As(V) that readily adsorbs to mineral or organic substances and precipitates.
- Environmental factors influence As speciation in soil and its availability to plants. Flooding of paddy fields leads to mobilization of arsenite. Methylated As may be present in soil as a result of microbial and/or algal biomethylation, or from past uses of methylated As pesticides.
- Arsenate is taken up by plant roots through phosphate transporters, whereas the uptake of undissociated arsenite and methylated As is mediated, at least partly, by NIP aquaporin channels. Rice is efficient at As assimilation owing to arsenite mobilization in flooded paddy soil and arsenite uptake sharing the highly expressed Si pathway. Arsenic accumulation in rice grain represents a potential health risk to humans.
- Arsenate is readily reduced to arsenite, which is detoxified by complexation with thiol-rich peptides and sequestrated in the vacuoles in As non-hyperaccumulating plants.
- Excessive accumulation of As in rice can be mitigated through agronomic and crop breeding strategies. Genetic modification may be employed to engineer plants more tolerant to As, or with reduced uptake for improved food safety.

Acknowledgments

LCR thanks the Department of Science and Technology, New Delhi, for the J.C. Bose National Fellowship and the University Grant Commission for financial assistance provided for the project. SP and RR thank the Council of Scientific and Industrial Research in relation to the Senior Research Fellowship provision. We also thank the Programme Coordinator, Centre of Advanced Study in Botany, Banaras Hindu University.

References

- [1] Smedley PL, Kinniburgh DG. A review of the source, behaviour and distribution of arsenic in natural waters. Appl Geochem 2002;17:517–68.
- [2] Jain CK, Ali I. Arsenic: occurrence, toxicity and speciation techniques. Water Resour 2000;34:4304–12.
- [3] Duker AA, Carranza EJM, Hale M. Arsenic geochemistry and health. Environ Int 2005;31:631-41.
- [4] Ongley LK, Sherman L, Armienta A, Concilio A, Salinas CF. Arsenic in the soils of Zimapan, Mexico. Environ Pollut 2007;145:793–9.
- [5] Bhattacharya P, Frisbie SH, Smith E, Naidu R, Jacks G, Sarkar B. Arsenic in the environment: a global perspective. In: Sarkar B, editor. Handbook of heavy metals in the environment. New York: Marcell Dekker Inc.; 2002. p. 145–215.
- [6] Meliker JR, Wahl RL, Cameron LL, Nriagu JO. Arsenic in drinking water and cerebrovascular disease, diabetes mellitus, and kidney disease in Michigan: a standardized mortality ratio analysis. Environ Health 2007;7:4–14.

- [7] Brammer H, Ravenscroft P. Arsenic in groundwater: a threat to sustainable agriculture in South and Southeast Asia. Environ Int 2009;35:647-54.
- [8] Meharg AA, Rahman MM. Arsenic contamination of Bangladesh paddy field soils: implications for rice contribution to arsenic consumption. Environ Sci Technol 2003;37:229–34.
- [9] Williams PN, Islam MR, Adomako EE, Raab A, Hossain SA, Zhu YG, et al. Increase in rice grain arsenic for regions of Bangladesh irrigating paddies with elevated arsenic in groundwaters. Environ Sci Technol 2006;40:4903–8.
- [10] Nordstrom DK. Worldwide occurrences of arsenic in groundwater. Science 2002;296:2143-5.
- [11] Ghosh A. Evaluation of chronic arsenic poisoning due to consumption of contaminated ground water in West Bengal, India. Int J Prev Med 2013;4:976–9.
- [12] Johnston RB, Hanchett S, Khan MH. The socio-economics of arsenic removal. Nature Geosci 2010;3:2-3.
- [13] Bhattacharjee S, Chakravarty S, Maity S, Dureja V, Gupta KK. Metal contents in the groundwater of Sahebgunj district, Jharkhand, India, with special reference to arsenic. Chemosphere 2005;58:1203–17.
- [14] Ahmed MF, Ahuja S, Alauddin M, Hug SJ, Lloyd JR, Pfaff A, et al. Ensuring safe drinking water in Bangladesh. Science 2006;314:1687-8.
- [15] Singh AK. Chemistry of arsenic in groundwater of Ganges-Brahmaputra river basin. Curr Sci 2006;91:500-606.
- [16] Hundal HS, Singh K, Singh D. Arsenic content in ground and canal waters of Punjab, North-West India. Environ Monit Assess 2009;154:393–400.
- [17] Pandey PK, Deshmukh R, Zankyani H, Pandey M. Arsenic contamination and its causative factors in parts of Chhattisgarh, Central East India. Int J Water 2013;7:347–62.
- [18] Cullen WR, Reimer KJ. Arsenic speciation in the environment. Chem Rev 1989;89:713-64.
- [19] Bentley R, Chasteen TG. Microbial methylation of metalloids: arsenic, antimony, and bismuth. Microbiol Mol Biol Rev 2002;66:250–71.
- [20] Macur RE, Jackson CR, Botero LM, McDermott TR, Inskeep WP. Bacterial populations associated with the oxidation and reduction of arsenic in an unsaturated soil. Environ Sci Technol 2004;38:104–11.
- [21] Inskeep WP, McDermott TR, Fendorf S. Arsenic (V)/(III) cycling in soils and natural waters: chemical and microbiological processes. In: Frankenberger WT, editor. Environmental chemistry of arsenic. New York, NY: Marcel Dekker, Inc.; 2002. p. 183–215.
- [22] Oremland RS, Stolz JF. The ecology of arsenic. Science 2003;300:939-44.
- [23] O'Day PA, Vlassopoulos D, Root R, Rivera N. The influence of sulfur and iron on dissolved arsenic concentrations in the shallow subsurface under changing redox conditions. Proc Natl Acad Sci USA 2004;101:13703-8.
- [24] Morin G, Juillot F, Casiot C, Bruneel O, Personné JC, Elbaz-Poulichet F, et al. Bacterial formation of tooeleite and mixed arsenic(III) or arsenic(V)-iron(III) gels in the Carnoule's acid mine drainage, France. A XANES, XRD, and SEM study. Environ Sci Technol 2003;37:1705–12.
- [25] Tamaki S, Frankenberger Jr WT. Environmental biochemistry of arsenic. Rev Environ Contam Toxicol 1992;124:79–110.
- [26] Koch I, Wang L, Reimer KJ, Cullen WR. Arsenic species in terrestrial fungi and lichens from Yellowknife, NWT, Canada. Appl Organomet Chem 2000;14:245–52.
- [27] Kuehnelt D, Lintschinger J, Goessler W. Arsenic compounds in terrestrial organisms. IV. Green plants and lichens from an old arsenic smelter site in Austria. Appl Organomet Chem 2000;14:411–20.
- [28] Bergqvist C, Greger M. Arsenic accumulation and speciation in plants from different habitats. Appl Geochem 2012;27:615–22.
- [29] Dhankher OP, Li YJ, Rosen BP, Shi J, Salt D, Senecoff JF, et al. Engineering tolerance and hyperaccumulation of arsenic in plants by combining arsenate reductase and gamma-glutamyl cysteine synthetase expression. Nat Biotechnol 2002;20:1140–5.

- [30] Pickering IJ, Prince RC, George MJ, Smith RD, George GN, Salt DE. Reduction and coordination of arsenic in Indian mustard. Plant Physiol 2000;122:1171–7.
- [31] Raab A, Schat H, Meharg AA, Feldmann J. Uptake, translocation and transformation of arsenate and arsenite in sunflower (*Helianthus annuus*): formation of arsenic-phytochelatin complexes during exposure to high arsenic concentrations. New Phytol 2005;168:551–8.
- [32] Xu XY, McGrath SP, Zhao FJ. Rapid reduction of arsenate in the medium mediated by plant roots. New Phytol 2007;176:590–9.
- [33] Wang JR, Zhao FJ, Meharg AA, Raab A, Feldmann J, Mc Grath SP. Mechanisms of arsenic hyperaccumulation in *Pteris vittata*. Uptake kinetics, interactions with phosphate, and arsenic speciation. Plant Physiol 2002;130:1552–61.
- [34] Ma LQ, Komar KM, Tu C, Zhang WH, Cai Y, Kennelley ED. A fern that hyperaccumulates arsenic. Nature 2001;409:579.
- [35] Raab A, Feldmann J, Meharg AA. The nature of arsenic-phytochelatin complexes in *Holcus lanatus* and *Pteris cretica*. Plant Physiol 2004;134:1113–22.
- [36] Robinson BH, Kim N, Marchetti M, Moni C, Schroeter L, van den Dijssel C, et al. Arsenic hyperaccumulation by aquatic macrophytes in the Taupo Volcanic Zone, New Zealand. Environ Exp Bot 2006;58:206–15.
- [37] Zhao FJ, Mc Grath SP, Meharg AA. Arsenic as a food chain contaminant: mechanisms of plant uptake and metabolism and mitigation strategies. Annu Rev Plant Biol 2010;61:535–59.
- [38] Raab A, Wright SH, Jaspars M, Meharg AA, Feldmann J. Pentavalent arsenic can bind to biomolecules. Angew Chem Int Ed 2007;46:2594–7.
- [39] Chen Y, Factor-Litvak P, Parvez F, Graziano J, Howe G, Ahsan H. Association between lower-dose arsenic exposure from drinking water and high blood pressure in Bangladesh. Am J Epidemiol 2005;2005(161):S30-S130.
- [40] Tapio S, Grosche B. Arsenic in the aetiology of cancer. Mutat Res Rev Mutat Res 2006;612:215-46.
- [41] Mandal BK, Suzuki KT. Arsenic round the world: a review. Talanta 2002;58:201–35.
- [42] Goyer RA, Clarkson TW. Toxic effect of metals. In: Klaassen CD, editor. Casarett & Doull's Toxicology: the basic science of poisons. New York, USA: McGraw Hill; 2001. p. 811–68.
- [43] Slyemi D, Bonnefoy V. How prokaryotes deal with arsenic. Environ Microbiol Rep 2012;4:571–86.
- [44] Thiel T. Phosphate transport and arsenate resistance in the cyanobacterium *Anabaena variabilis*. J Bacteriol 1988;170:1143-7.
- [45] Manomita PM, Bhowmik N, Bandopadhyay B, Sharma A. Comparison of mercury, lead and arsenic with respect to genotoxic effects on plant systems and the development of genetic tolerance. Environ Exp Bot 2004;52:199–223.
- [46] Liu SX, Athar M, Lippai I, Waldren C, Hei TK. Induction of oxyradicals by arsenic: implication for mechanism of genotoxicity. Proc Natl Acad Sci USA 2001;98:1643–8.
- [47] Dopp E, von Recklinghausen U, Diaz-Bone R, Hirner AV, Rettenmeier AW. Cellular uptake, subcellular distribution and toxicity of arsenic compounds in methylating and non-methylating cells. Environ Res 2010;110:435–42.
- [48] Oremland RS, Saltikov CW, Wolfe-Simon F, Stolz JF. Arsenic in the evolution of earth and extraterrestrial ecosystems. Geomicrobiol J 2009;26:522–36.
- [49] Silver S, Phung LT. Genes and enzymes involved in bacterial oxidation and reduction of inorganic arsenic. Appl Environ Microbiol 2005;71:599–608.
- [50] Stolz JF, Basu P, Santini JM, Oremland RS. Arsenic and selenium in microbial metabolism. Annu Rev Microbiol 2006;60:107–30.
- [51] Paez-Espino D, Tamames J, de Lorenzo V, Canovas D. Microbial responses to environmental arsenic. Biometals 2009;22:117-30.

- [52] Tsai SL, Singh S, Chen W. Arsenic metabolism by microbes in nature and the impact on arsenic remediation. Curr Opin Biotechnol 2009;20:659–67.
- [53] Wolfe-Simon F, Blum JS, Kulp TR, Gordon GW, Hoeft SE, Pett-Ridge J, et al. A bacterium that can grow by using arsenic instead of phosphorus. Science 2011;332:1163–6.
- [54] Marchal M, Briandet R, Koechler S, Kammerer B, Bertin PN. Effect of arsenite on swimming motility delays surface colonization in *Herminiimonas arsenicoxydans*. Microbiology 2010;156:2336–42.
- [55] Lebrun E, Brugna M, Baymann F, Muller D, Lievremont D, Lett MC, et al. Arsenite oxidase, an ancient bioenergetic enzyme. Mol Biol Evol 2003;20:686–93.
- [56] Ngu TT, Stillman MJ. Metal-binding mechanisms in metallothioneins. Dalton Trans 2009;28:5425-33.
- [57] Ngu TT, Stillman MJ. Metalation of metallothioneins. IUBMB Life 2009;61:438-46.
- [58] Mukhopadhyay R, Rosen BP. Arsenate reductases in prokaryotes and eukaryotes. Environ Health Perspect 2002;110:745–8.
- [59] Yin XX, Chen J, Qin J, Sun GX, Rosen BP, Zhu YG. Biotransformation and volatilization of arsenic by three photosynthetic cyanobacteria. Plant Physiol 2011;156:1631–8.
- [60] Ye J, Rensing C, Rosen BP, Zhu Y-G. Arsenic biomethylation by photosynthetic organisms. Trends Plant Sci 2012;17:155–62.
- [61] Weeger W, Lièvremont D, Perret M, Lagarde F, Hubert JC, Leroy M, et al. Oxidation of arsenite to arsenate by a bacterium isolated from an aquatic environment. BioMetals 1999;12:141–9.
- [62] Anderson GL, Ellis PJ, Kuhn P, Hille R. Oxidation of arsenite by *Alcaligenes faecalis*. In: Frankenberger WT, editor. Environmental chemistry of arsenic. New York: Marcell Dekker; 2002. p. 343–61.
- [63] Rosen BP, Liu Z. Transport pathways for arsenic and selenium: a minireview. Environ Int 2009;35:512–15.
- [64] Sanders OI, Rensing C, Kuroda M, Mitra B, Rosen BP. Antimonite is accumulated by the glycerol facilitator GlpF in *Escherichia coli*. J Bacteriol 1997;179:3365–7.
- [65] Muller D, Medigue C, Koechler S, Barbe V, Barakat M, Talla E, et al. A tale of two oxidation states: bacterial colonization of arsenic-rich environments. PLoS Genet 2007;3:e53.
- [66] Senn DB, Hemond HF. Nitrate controls on iron and arsenic in an urban lake. Science 2002;296:2373-6.
- [67] Hohmann C, Winkler E, Morin G, Kappler A. Anaerobic Fe(II)-oxidizing bacteria show As resistance and immobilize As during Fe(III) mineral precipitation. Environ Sci Technol 2010;44:94–101.
- [68] Gibney BP, Nusslein K. Arsenic sequestration by nitrate respiring microbial communities in urban lake sediments. Chemosphere 2007;70:329–36.
- [69] Duquesne K, Lebrun S, Casiot C, Bruneel O, Personne JC, Leblanc M, et al. Immobilization of arsenite and ferric iron by *Acidithiobacillus ferrooxidans* and its relevance to acid mine drainage. Appl Environ Microbiol 2003;69:6165–73.
- [70] Egal M, Casiot C, Morin G, Parmentier M, Bruneel O, Lebrun S, et al. Kinetic control on the formation of tooeleite, schwertmannite and jarosite by *Acidithiobacillus ferrooxidans* strains in an As(III)-rich acid mine water. Chem Geol 2009;265:432-41.
- [71] Turner JS, Robinson NJ. Cyanobacterial metallothioneins: biochemistry and molecular genetics. J Ind Microbiol 1995;14:119–25.
- [72] Silver S, Phung LT. Bacterial heavy metal resistance: new surprises. Annu Rev Microbiol 1996;50:753–89.
- [73] Ybarra GR, Webb R. Differential responses of GroEL and metallothionein genes to divalent metal cations and the oxyanions of arsenic in the cyanobacterium *Synechococcus* sp. strain PCC7942. Proc Conf Hazard Waste Res 1998;2:78–85.
- [74] Robinson NJ, Whitehall SK, Cavet JS. Microbial metallothioneins. Adv Microb Physiol 2001;44:183–213.
- [75] Carlin A, Shi W, Dey S, Rosen BP. The ars operon of Escherichia coli confers arsenical and antimonial resistance. J Bacteriol 1995;177:981–6.

- [76] Ji G, Silver S. Regulation and expression of the arsenic resistance operon from *Staphylococcus aureus* plasmid pI258. J Bacteriol 1992;174:3684–94.
- [77] Ordonez E, Letek M, Valbuena N, Gil JA, Mateos LM. Analysis of genes involved in arsenic resistance in *Corynebacterium glutamicum* ATCC 13032. Appl Environ Microbiol 2005;71:6206–15.
- [78] Lopez-Maury L, Florencio FJ, Reyes JC. Arsenic sensing and resistance system in the cyanobacterium *Synechocystis* sp. Strain PCC 6803. J Bacteriol 2003;185:5363–71.
- [79] Bobrowicz P, Wysocki R, Owsianik G, Goffeau A, Ulaszewski S. Isolation of three contiguous genes, ACR1, ACR2 and ACR3, involved in resistance to arsenic compounds in the yeast *Saccharomyces cerevisiae*. Yeast 1997;13:819–28.
- [80] Cervantes C, Ji G, Ramirez JL, Silver S. Resistance to arsenic compounds in microorganisms. FEMS Microbiol Revs 1994;15:355–67.
- [81] Diorio C, Cai J, Marmor J, Shinder R, DuBow MS. An *Escherichia coli* chromosomal *ars* operon homolog is functional in arsenic detoxification and is conserved in gram-negative bacteria. J Bacteriol 1995;177:2050–6.
- [82] Bruhn DF, Li J, Silver S, Roberto F, Rosen BP. The arsenical resistance operon of IncN plasmid R46. FEMS Microbiol Lett 1996;139:149–53.
- [83] Chen Y, Rosen BP. Metalloregulatory properties of the ArsD repressor. J Biol Chem 1997;272:14257-62.
- [84] Gatti D, Mitra B, Rosen BP. E. coli soft metal ion translocating ATPases. J Bio Chem 2000;275:34009–12.
- [85] Gladysheva TB, Oden KL, Rosen BP. Properties of the arsenate reductase of plasmid R773. Biochemistry 1994;33:7288–93.
- [86] Wu J, Rosen BP. The ArsR protein is a trans-acting regulatory protein. Mol Microbiol 1991;5:1331-6.
- [87] Ghosh M, Shen J, Rosen BP. Pathways of As(III) detoxification in Saccharomyces cerevisiae. Proc Natl Acad Sci USA 1996;96:5001–6.
- [88] Chen CM, Misra TK, Silver S, Rosen BP. Nucleotide sequence of the structural gene of an anion pump. The plasmid-encoded arsenical resistance operon. J Biol Chem 1986;261:15030–15038.
- [89] Mukhopadhyay R, Shi J, Rosen BP. Purification and characterization of Acr2p, the *Saccharomyces cerevisiae* arsenate reductase. J Biol Chem 2000;275:21149–21157.
- [90] Li R, Haile JD, Kennelly PJ. An arsenate reductase from *Synechocystis* sp. strain PCC 6803 exhibits a novel combination of catalytic characteristics. J Bacteriol 2003;185:6780–9.
- [91] Messens J, Silver S. Arsenate reduction: thiol cascade chemistry with convergent evolution. J Mol Biol 2006;362:1–17.
- [92] Zegers I, Martins JC, Willem R, Wyns L, Messens J. Arsenate reductase from *S. aureus* plasmid pI258 is a phosphatase drafted for redox duty. Nat Struct Biol 2001;8:843–7.
- [93] Zhou Y, Messier N, Ouellette M, Rosen BP, Mukhopadhyay R. *Leishmania major* LmACR2 is a pentavalent antimony reductase that confers sensitivity to the drug Pentostam. J Biol Chem 2004;279:37445–51.
- [94] Dhankher OP, Rosen BP, McKinney EC, Meagher RB. Hyperaccumulation of arsenic in the shoots of *Arabidopsis* silenced for arsenate reductase (ACR2). Proc Natl Acad Sci USA 2006;103:5413–18.
- [95] Mukhopadhyay R, Rosen BP. Saccharomyces cerevisiae ACR2 gene encodes an arsenate reductase. FEMS Microbiol Lett 1998;168:127–36.
- [96] Pandey S, Shrivastava AK, Rai R, Rai LC. Molecular characterization of Alr1105, a novel arsenate reductase of the diazotrophic cyanobacterium *Anabaena* sp. PCC7120 and decoding its role in abiotic stress management in *Escherichia coli*. Plant Mol Biol 2013;83:417–32.
- [97] Pandey S, Shrivastava AK, Singh VK, Rai R, Singh PK, Rai S, et al. A new arsenate reductase involved in arsenic detoxification in *Anabaena* sp. PCC7120. Funct Int Genomics 2013;13:43–55.
- [98] Ahmann D, Roberts AL, Krumholz LR, Morel FM. Microbe grows by reducing arsenic. Nature 1994; 371:750.

- [99] Krafft T, Macy JM. Purification and characterization of the respiratory arsenate reductase of *Chrysiogenes arsenatis*. Eur J Biochem 1998;255:647–53.
- [100] Afkar E, Lisak J, Saltikov C, Basu P, Oremland RS, Stolz JF. The respiratory arsenate reductase from *Bacillus selenitireducens* strain MLS10. FEMS Microbiol Lett 2003;226:107–12.
- [101] Malasarn D, Keeffe JR, Newman DK. Characterization of the arsenate respiratory reductase from *Shewanella* sp. strain ANA-3. J Bacteriol 2008;190:135–42.
- [102] Murphy JN, Saltikov CW. The cymA gene, encoding a tetraheme c-type cytochrome, is required for arsenate respiration in *Shewanella* species. J Bacteriol 2007;189:2283–90.
- [103] Zargar K, Saltikov CW. Lysine-91 of the tetraheme c-type cytochrome CymA is essential for quinine interaction and arsenate respiration in *Shewanella* sp. strain ANA-3. Arch Microbiol 2009;191:797–806.
- [104] Duval S, Santini JM, Nitschke W, Hille R, Schoepp-Cothenet B. The small subunit AroB of arsenite oxidase: lessons on the [2Fe-2S] Rieske protein superfamily. J Biol Chem 2010;285:20442–51.
- [105] Saltikov CW, Wildman Jr. RA, Newman DK. Expression dynamics of arsenic respiration and detoxification in *Shewanella* sp. strain ANA-3. J Bacteriol 2005;187:7390–6.
- [106] Murphy JN, Saltikov CW. The ArsR repressor mediates arsenite-dependent regulation of arsenate respiration and detoxification operons of *Shewanella* sp. strain ANA-3. J Bacteriol 2009;191:6722–31.
- [107] Kulp TR, Hoeft SE, Asao M, Madigan MT, Hollibaugh JT, Fisher JC, et al. Arsenic(III) fuels anoxygenic photosynthesis in hot spring biofilms from Mono Lake, California. Science 2008;321:967–70.
- [108] Schoepp-Cothenet B, Duval S, Santini JM, Nitschke W. Comment on "Arsenic (III) fuels anoxygenic photosynthesis in hot spring biofilms from Mono Lake, California". Science 2009;323:583.
- [109] Ajees AA, Yang J, Rosen BP. The ArsD As(III) metallochaperone. Biometals 2011;24:391–9.
- [110] Yang HC, Cheng J, Finan TM, Rosen BP, Bhattacharjee H. Novel pathway for arsenic detoxification in the legume symbiont *Sinorhizobium meliloti*. J Bacteriol 2005;187:6991–7.
- [111] Rosen BP, Tamas MJ. Arsenic transport in prokaryotes and eukaryotic microbes. Adv Exp Med Biol 2010;679:47–55.
- [112] Wu B, Song J, Beitz E. Novel channel enzyme fusion proteins confer arsenate resistance. J Biol Chem 2010;285:40081-7.
- [113] Ellis DR, Gumaelius L, Indriolo E, Pickering IJ, Banks JA, Salt DE. A novel arsenate reductase from the arsenic hyperaccumulating fern *Pteris vittata*. Plant Physiol 2006;141:1544–54.
- [114] Green HH. Isolation and description of a bacterium causing oxidation of arsenite to arsenate in cattledipping baths. Rep Dir Vet Res S Afr 1918;6:593–9.
- [115] Turner AW. Bacterial oxidation of arsenite. Nature 1949;164:76-7.
- [116] Santini JM, vanden Hoven RN. Molybdenum-containing arsenite oxidase of the chemolithoautotrophic arsenite oxidizer NT-26. J Bacteriol 2004;186:1614–19.
- [117] Hoeft SE, Blum JS, Stolz JF, Tabita FR, Witte B, King GM, et al. Alkalilimnicola ehrlichii sp. nov., a novel, arsenite-oxidizing haloalkaliphilic gammaproteobacterium capable of chemoautotrophic or heterotrophic growth with nitrate or oxygen as the electron acceptor. Int J Syst Evol Microbiol 2007;57:504–12.
- [118] Oremland RS, Stolz JF, Madigan MT, Hollibaugh JT, Kulp TR, Hoeft SE, et al. Response to comment on "Arsenic(III) fuels anoxygenic photosynthesis in hot spring biofilms from Mono Lake, California". Science 2009;323:583d.
- [119] Sun W, Sierra-Alvarez R, Milner L, Field JA. Anaerobic oxidation of arsenite linked to chlorate reduction. Appl Environ Microbiol 2010;76:6804–11.
- [120] Richey C, Chovanec P, Hoeft SE, Oremland RS, Basu P, Stolz JF. Respiratory arsenate reductase as a bidirectional enzyme. Biochem Biophys Res Commun 2009;382:298–302.
- [121] Muller D, Lievremont D, Simeonova DD, Hubert JC, Lett MC. Arsenite oxidase *aox* genes from a metalresistant beta-proteobacterium. J Bacteriol 2003;185:135–41.

- [122] Anderson GL, Williams J, Hille R. The purification and characterization of arsenite oxidase from *Alcaligenes faecalis*, a molybdenum-containing hydroxylase. J Biol Chem 1992;267:23674–82.
- [123] Ellis PJ, Conrads T, Hille R, Kuhn P. Crystal structure of the 100kDa arsenite oxidase from *Alcaligenes faecalis* in two crystal forms at 1.64 A and 2.03 A. Structure 2001;9:125–32.
- [124] Hoke KR, Cobb N, Armstrong FA, Hille R. Electrochemical studies of arsenite oxidase: an unusual example of a highly cooperative two-electron molybdenum center. Biochemistry 2004;43:1667–74.
- [125] vanden Hoven RN, Santini. JM. Arsenite oxidation by the heterotroph *Hydrogenophaga* sp. str. NT-14: the arsenite oxidase and its physiological electron acceptor. Biochim Biophys Acta 2004;1656:148–55.
- [126] Prasad KS, Subramanian V, Paul J. Purification and characterization of arsenite oxidase from *Arthrobacter* sp. Biometals 2009;22:711–21.
- [127] Lieutaud A, van Lis R, Duval S, Capowiez L, Muller D, Lebrun R, et al. Arsenite oxidase from *Ralstonia* sp. 22: characterization of the enzyme and its interaction with soluble cytochromes. J Biol Chem 2010;285:20433–20441.
- [128] Santini JM, Kappler U, Ward SA, Honeychurch MJ, vanden Hoven RN, Bernhardt PV. The NT-26 cytochrome c552 and its role in arsenite oxidation. Biochim Biophys Acta 2007;1767:189–96.
- [129] Branco R, Francisco R, Chung AP, Morais PV. Identification of an aox system that requires cytochrome c in the highly arsenic-resistant bacterium *Ochrobactrum tritici* SCII24. Appl Environ Microbiol 2009;75:5141–7.
- [130] Kashyap DR, Botero LM, Franck WL, Hassett DJ, McDermott TR. Complex regulation of arsenite oxidation in Agrobacterium tumefaciens. J Bacteriol 2006;188:1081–8.
- [131] Duquesne K, Lieutaud A, Ratouchniak J, Muller D, Lett MC, Bonnefoy V. Arsenite oxidation by a chemoautotrophic moderately acidophilic *Thiomonas* sp.: from the strain isolation to the gene study. Environ Microbiol 2008;10:228–37.
- [132] Arsene-Ploetze F, Koechler S, Marchal M, Coppee JY, Chandler M, Bonnefoy V, et al. Structure, function, and evolution of the *Thiomonas* spp. genome. PLoS Genet 2010;6:e1000859.
- [133] Cai L, Rensing C, Li X, Wang G. Novel gene clusters involved in arsenite oxidation and resistance in two arsenite oxidizers: *Achromobacter* sp. SY8 and *Pseudomonas* sp. TS44. Appl Microbiol Biotechnol 2009;83:715–25.
- [134] Clingenpeel SR, D'Imperio S, Oduro H, Druschel GK, McDermott TR. Cloning and in situ expression studies of the *Hydrogenobaculum* arsenite oxidase genes. Appl Environ Microbiol 2009;75:3362–5.
- [135] Slyemi D, Ratouchniak J, Bonnefoy V. Regulation of the arsenic oxidation encoding genes of a moderately acidophilic, facultative chemolithoautotrophic *Thiomonas* sp. Adv Mater Res 2007;20–21:427–30.
- [136] Slyemi D. Caractérisation de *"Thiomonas arsenitoxydans"* et étude de la régulation des génes condant pour l'arsénite oxydase. Marseille, France: Faculté des Sciences de Luminy, Université de la Méditerranée, Aix-Marseille II; 2010.
- [137] Koechler S, Cleiss-Arnold J, Proux C, Sismeiro O, Dillies MA, Goulhen-Chollet F, et al. Multiple controls affect arsenite oxidase gene expression in *Herminiimonas arsenicoxydans*. BMC Microbiol 2010;10:53.
- [138] Rhine ED, Ni Chadhain SM, Zylstra GJ, Young LY. The arsenite oxidase genes (*aroAB*) in novel chemoautotrophic arsenite oxidizers. Biochem Biophys Res Commun 2007;354:662–7.
- [139] Zargar K, Hoeft S, Oremland R, Saltikov CW. Identification of a novel arsenite oxidase gene, *arxA*, in the haloalkaliphilic, arsenite-oxidizing bacterium *Alkalilimnicola ehrlichii* strain MLHE-1. J Bacteriol 2010;192:3755–62.
- [140] Shariatpanahi M, Anderson AC, Abdelghani AA, Englande AJ, Hughes J, Wilkinson RF. Biotransformation of the pesticide sodium arsenate. J Environ Sci Health B 1981;16:35–47.
- [141] Michalke K, Wickenheiser EB, Mehring M, Hirner AV, Hensel R. Production of volatile derivatives of metal(loid)s by microflora involved in anaerobic digestion of sewage sludge. Appl Environ Microbiol 2000;66:2791–6.

- [142] Le XC, Lu X, Ma M, Cullen WR, Aposhian HV, Zheng B. Speciation of key arsenic metabolic intermediates in human urine. Anal Chem 2000;72:5172–7.
- [143] Wang G, Kennedy SP, Fasiludeen S, Rensing C, DasSarma S. Arsenic resistance in *Halobacterium* sp. strain NRC-1 examined by using an improved gene knockout system. J Bacteriol 2004;186:3187–94.
- [144] Yin XX, Zhang YY, Yang J, Zhu YG. Rapid biotransformation of arsenic by a model protozoan *Tetrahymena thermophila*. Environ Pollut 2011;159:837–40.
- [145] Dopp E, Kligerman AD, Diaz-Bone RA. Organoarsenicals. Uptake, metabolism, and toxicity. Met Ions Life Sci 2010;7:231–65.
- [146] Qin J, Rosen BP, Zhang Y, Wang G, Franke S, Rensing C. Arsenic detoxification and evolution of trimethylarsine gas by a microbial arsenite S-adenosylmethionine methyltransferase. Proc Natl Acad Sci USA 2006;103:2075–80.
- [147] Karadjova IB, Slaveykova VI, Tsalev DL. The biouptake and toxicity of arsenic species on the green microalga *Chlorella salina* in seawater. Aquat Toxicol 2008;87:264–71.
- [148] Miyashita S, Fujiwara S, Tsuzuki M, Kaise T. Rapid biotransformation of arsenate into oxoarsenosugars by a freshwater unicellular green alga *Chlamydomonas reinhardtii*. Biosci Biotechnol Biochem 2011;75:522–30.
- [149] Geiszinger A, Goessler W, Pedersen SN, Francesconi KA. Arsenic biotransformation by the brown macroalga, *Fucus serratus*. Environ Toxicol Chem 2001;20:2255–62.
- [150] Zuccaro A, Schoch CL, Spatafora JW, Kohlmeyer J, Draeger S, Mitchell JI. Detection and identification of fungi intimately associated with the brown seaweed *Fucus serratus*. Appl Environ Microbiol 2008;74:931-41.
- [151] Dombrowski PM, Long W, Farley KJ, Mahony JD, Capitani JF, Di Toro DM. Thermodynamic analysis of arsenic methylation. Environ Sci Technol 2005;39:2169–76.
- [152] Challenger F. Biological methylation. Chem Rev 1945;36:315–61.
- [153] Yuan C, Lu X, Qin J, Rosen BP, Le XC. Volatile arsenic species released from *Escherichia coli* expressing the AsIII S-adenosylmethionine methyltransferase gene. Environ Sci Technol 2008;42:3201–6.
- [154] Maki T, Hasegawa H, Watarai H, Ueda K. Classification for dimethylarsenate-decomposing bacteria using a restrict fragment length polymorphism analysis of 16S rRNA genes. Anal Sci 2004;20:61–8.
- [155] Yoshinaga M, Cai Y, Rosen BP. Demethylation of methylarsonic acid by a microbial community. Environ Microbiol 2011;13:1205–15.
- [156] Pandey S, Rai R, Rai LC. Proteomics combines morphological, physiological and biochemical attributes to unravel the survival strategy of *Anabaena* sp. PCC7120 under arsenic stress. J Proteomics 2012;75:921–37.
- [157] Bhattacharya P, Pal R. Response of cyanobacteria to arsenic toxicity. J Appl Phycol 2010;23:293–9.
- [158] Srivastava AK, Bhargava P, Thapar R, Rai LC. Differential responses of antioxidative defense system of Anabaena doliolum under arsenite and arsenate stress. J Basic Microbiol 2009;49:1–10.
- [159] Colangelo EP, Guerinot ML. Put the metal to the petal: metal uptake and transport throughout plants. Curr Opin Plant Biol 2006;9:322–30.
- [160] Zhu YG, Sun GX, Lei M, Teng M, Liu YX, Chen NC, et al. High percentage inorganic arsenic content of mining impacted and nonimpacted Chinese rice. Environ Sci Technol 2008;42:5008–13.
- [161] Ullrich-Eberius CI, Sanz A, Novacky AJ. Evaluation of arsenate- and vanadate-associated changes of electrical membrane potential and phosphate transport in *Lemna gibba*-G1. J Exp Bot 1989;40:119–28.
- [162] Meharg AA, Macnair MR. An altered phosphate uptake system in arsenate-tolerant *Holcus lanatus* L. New Phytol 1990;116:29–35.
- [163] Meharg AA, Macnair MR. The mechanism of arsenate tolerance in *Deschampsia cespitosa* (L.) *Beauv*. and *Agrostis capillaris* L. New Phytol 1991;119:291–7.

- [164] Meharg AA, Macnair MR. Suppression of the high affinity phosphate uptake system: a mechanism of arsenate tolerance in *Holcus lanatus* L. J Exp Bot 1992;43:519–24.
- [165] Wu Z, Ren H, McGrath SP, Wu P, Zhao FJ. Investigating the contribution of the phosphate transport pathway to arsenic accumulation in rice. Plant Physiol 2011;157:498–508.
- [166] Meharg AA, Naylor J, Macnair MR. Phosphorus nutrition of arsenate tolerant and nontolerant phenotypes of velvet grass. J Environ Qual 1994;23:234–8.
- [167] Tu C, Ma LQ. Interactive effects of pH, arsenic and phosphorus on uptake of As and P and growth of the arsenic hyperaccumulator *Pteris vittata* L. under hydroponic conditions. Environ Exp Bot 2003;50:243–51.
- [168] Bleeker PM, Schat H, Vooijs R, Verkleij JAC, Ernst WHO. Mechanisms of arsenate tolerance in Cytisus striatus. New Phytol 2003;157:33–8.
- [169] Abedin MJ, Cresser MS, Meharg AA, Feldmann J, Cotter-Howells J. Arsenic accumulation and metabolism in rice (*Oryza sativa* L.). Environ Sci Technol 2002;36:962–8.
- [170] Esteban E, Carpena RO, Meharg AA. High-affinity phosphate/arsenate transport in white lupin (*Lupinus albus*) is relatively insensitive to phosphate status. New Phytol 2003;158:165–73.
- [171] Shin H, Shin HS, Dewbre GR, Harrison MJ. Phosphate transport in *Arabidopsis*: Pht1;1 and Pht1;4 play a major role in phosphate acquisition from both low and high-phosphate environments. Plant J 2004;39:629–42.
- [172] González E, Solano R, Rubio V, Leyva A, Paz-Ares J. Phosphate transporter traffic facilitator1 is a plantspecific SEC12- related protein that enables the endoplasmic reticulum exit of a high-affinity phosphate transporter in *Arabidopsis*. Plant Cell 2005;17:3500–12.
- [173] Catarecha P, Segura MD, Franco-Zorrilla JM, García-Ponce B, Lanza M, Solano R, et al. A mutant of the *Arabidopsis* phosphate transporter PHT1;1 displays enhanced arsenic accumulation. Plant Cell 2007;19:1123–33.
- [174] Rausch C, Bucher M. Molecular mechanisms of phosphate transport in plants. Planta 2002;216:23–37.
- [175] Bucher M. Functional biology of plant phosphate uptake at root and mycorrhiza interfaces. New Phytol 2007;173:11–26.
- [176] Palmieri L, Picault N, Arrigoni R, Besin E, Palmieri F, Hodges M. Molecular identification of three *Arabidopsis thaliana* mitochondrial dicarboxylate carrier isoforms: organ distribution, bacterial expression, reconstitution into liposomes and functional characterization. Biochem J 2008;410:621–9.
- [177] Mendoza-Cózatl DG, Jobe TO, Hauser F, Schroeder JI. Long-distance transport, vacuolar sequestration, tolerance, and transcriptional responses induced by cadmium and arsenic. Curr Opin Plant Biol 2011;14:554–62.
- [178] Quaghebeur M, Rengel Z. The distribution of arsenate and arsenite in shoots and roots of *Holcus lanatu*sis influenced by arsenic tolerance and arsenate and phosphate supply. Plant Physiol 2003;132:1600–9.
- [179] Lomax C, Liu WJ, Wu L, Xue K, Xiong J, Zhou J, et al. Methylated arsenic species in plants originate from soil microorganisms. New Phytol 2012;193:665–72.
- [180] Mimura T. Regulation of phosphate transport and homeostasis in plant cells. Int Rev Cytol 1999;191:149–200.
- [181] Bhattacharjee H, Rosen BP. Arsenic metabolism in prokaryotic and eukaryotic microbes. In: Nies DH, Silver S, editors. Molecular microbiology of heavy metals. Berlin, Germany: Springer-Verlag; 2007. p. 371–406.
- [182] Meharg AA, Jardine L. Arsenite transport into paddy rice (Oryza sativa) roots. New Phytol 2003;157:39-44.
- [183] Bienert GP, Thorsen M, Schüssler MD, Nilsson HR, Wagner A, Tamás MJ, et al. A subgroup of plant aquaporins facilitate the bidirectional diffusion of As(OH)3 and Sb(OH)3 across membranes. BMC Biol 2008;6:26.
- [184] Ma JF, Yamaji N, Mitani N, Xu XY, Su YH, McGrath SP, et al. Transporters of arsenite in rice and their role in arsenic accumulation in rice grain. Proc Natl Acad Sci USA 2008;105:9931–5.

- [185] Lombi E, Zhao FJ, Fuhrmann M, Ma LQ, McGrath SP. Arsenic distribution and speciation in the fronds of the hyperaccumulator *Pteris vittata*. New Phytol 2002;156:195–203.
- [186] Pickering IJ, Gumaelius L, Harris HH, Prince RC, Hirsch G, Banks JA, et al. Localizing the biochemical transformations of arsenate in a hyperaccumulating fern. Environ Sci Technol 2006;40:5010–14.
- [187] Su YH, McGrath SP, Zhu YG, Zhao FJ. Highly efficient xylem transport of arsenite in the arsenic hyperaccumulator *Pteris vittata*. New Phytol 2008;180:434–41.
- [188] Zhao FJ, Ma JF, Meharg AA, McGrath SP. Arsenic uptake and metabolism in plants. New Phytol 2009;181:777-94.
- [189] Indriolo E, Na GN, Ellis D, Salt DE, Banks JA. Avacuolar arsenite transporter necessary for arsenic tolerance in the arsenic hyperaccumulating fern *Pteris vittata* is missing in flowering plants. Plant Cell 2010;22:2045–57.
- [190] Li RY, Stroud JL, Ma JF, McGrath SP, Zhao FJ. Mitigation of arsenic accumulation in rice with water management and silicon fertilization. Environ Sci Technol 2009;43:3778–83.
- [191] Abedin MJ, Feldmann J, Meharg AA. Uptake kinetics of arsenic species in rice plants. Plant Physiol 2002;128:1120–8.
- [192] Raab A, Williams PN, Meharg A, Feldmann J. Uptake and translocation of inorganic and methylated arsenic species by plants. Environ Chem 2007;4:197–203.
- [193] Abbas MHH, Meharg AA. Arsenate, arsenite and dimethyl arsenic acid (DMA) uptake and tolerance in maize (*Zea mays* L.). Plant Soil 2008;304:277-89.
- [194] Marin AR, Masscheleyn PH, Patrick WH. The influence of chemical form and concentration of arsenic on rice growth and tissue arsenic concentration. Plant Soil 1992;139:175–83.
- [195] Burló F, Guijarro I, Carbonell-Barrachina AA, Valero D, Martinez-Sánchez F. Arsenic species: effects on and accumulation by tomato plants. J Agri Food Chem 1999;47:1247–53.
- [196] Carey A-M, Scheckel KG, Lombi E, Newville M, Choi Y, Nortan GJ, et al. Grain unloading of arsenic species in rice. Plant Physiol 2010;152:309–19.
- [197] Carey AM, Norton GJ, Deacon C, Scheckel KG, Lombi E, Punshon T, et al. Phloem transport of arsenic species from flag leaf to grain during grain filling. New Phytol 2011;192:87–98.
- [198] Su YH, McGrath SP, Zhao FJ. Rice is more efficient in arsenite uptake and translocation than wheat and barley. Plant Soil 2010;328:27–34.
- [199] Carbonell-Barrachina AA, Aarabi MA, DeLaune RD, Gambrell RP, Patrick WH. The influence of arsenic chemical form and concentration on *Spartina patens* and *Spartina alterniflora* growth and tissue arsenic concentration. Plant Soil 1998;198:33–43.
- [200] Srivastava S, Srivastava AK, Suprasanna P, D'Souza SF. Comparative biochemical and transcriptional profiling of two contrasting varieties of *Brassica juncea* L. in response to arsenic exposure reveals mechanisms of stress perception and tolerance. J Exp Bot 2009;181:1–13.
- [201] Shaibur MR, Kitajima N, Sugewara R, Kondo T, Alam S, Imamul Huq SM, et al. Critical toxicity of arsenic and elemental composition of arsenic-induced chlorosis in hydroponic *Sorghum*. Water Air Soil Pollut 2008;191:279–92.
- [202] Rai R, Pandey S, Pandey-Rai S. Arsenic induced physiological, biochemical and molecular changes in *Artemisia annua*, an antimalarial plant. Ecotoxicology 2011;8:1900–13.
- [203] Singh N, Ma LQ, Srivastava M, Rathinasabapathi B. Metabolic adaptations to arsenic induced oxidative stress in *Pteris vittata* L. and *Pteris ensiformis* L. Plant Sci 2006;170:274–82.
- [204] Smith SE, Christophersen HM, Pope S, Smith FA. Arsenic uptake and toxicity in plants: integrating mycorrhizal influences. Plant Soil 2010;327:1–21.
- [205] Ahmed SFR, Killham K, Alexander I. Influences of arbuscular fungus *Glomus mosseae* on growth and nutrition of lentil irrigated with arsenic contaminated water. Plant Soil 2006;258:33–41.

- [206] Garg N, Singla P. Arsenic toxicity in crop plants: physiological effects and tolerance mechanisms. Environ Chem Lett 2011;9:303–21.
- [207] Miteva E. Accumulation and effect of arsenic on tomatoes. Comm Soil Sci Plant Anal 2002;33:1917-26.
- [208] Rahman MA, Hasegawa H, Rahman MM, Miah MAM, Tasim A. Straight head disease of rice (*Oryza sativa* L.) induced by arsenic toxicity. Environ Exp Bot 2008;62:54–9.
- [209] Vazquez S, Esteban E, Carpena RO. Evolution of arsenate toxicity in nodulated White Lupine in a longterm culture. J Agric Food Chem 2008;56:8580–7.
- [210] Woolson EA, Axley JH, Kearney PC. The chemistry and phytotoxicity of arsenic in soils. I. Contaminated field soils. Soil Sci Soc Am Proc 1971;35:938–43.
- [211] Carbonell-Barrachina AA, Burló F, Burgos-Hernández A, Ló pez E, Mataix J. The influence of arsenite concentration on arsenic accumulation in tomato and bean plants. Sci Hortic 1997;71:167–76.
- [212] Carbonell-Barrachina AA, Burló F, Mataix J. Response of bean micronutrient nutrition to arsenic and salinity. J Plant Nutr 1998;21:1287–99.
- [213] Chen W, Chi Y, Taylor NL, Lambers H, Finnegan PM. Disruption of ptLPD1 or ptLPD2, genes that encode isoforms of the plastidial lipoamide dehydrogenase, confers arsenate hypersensitivity in *Arabidopsis*. Plant Physiol 2010;153:1385–97.
- [214] Stoeva N, Berova M, Zlatez Z. Physiological response of maize to arsenic contamination. Biol Planta 2004;47:449–52.
- [215] Geng CN, Zhu YG, Hu Y, Williams P, Meharg AA. Arsenate causes differential acute toxicity to two P-deprived genotypes of rice seedlings (*Oryza sativa* L.). Plant Soil 2006;279:297–306.
- [216] Pigna M, Cozzolina V, Violante A, Meharg AA. Influence of phosphate on the arsenic uptake by wheat (*Triticum durum* L.) irrigated with arsenic solutions at three different concentrations. Water Air Soil Pollut 2008;197:371–80.
- [217] Liu X, Zhang S, Shan X, Zhu YG. Toxicity of arsenate and arsenite on germination seedling growth and amylolytic activity of wheat. Chemosphere 2005;61:293–301.
- [218] Geng CN, Zhu YG, Tong YP, Sally E, Smith FA. Arsenate (As) uptake by and distribution in two cultivars of winter wheat (*Triticum aestivum* L.). Chemosphere 2006;62:608–15.
- [219] Stoeva N, Bineva Tz. Oxidative changes and photosynthesis in oat plants grown in As-contaminated soil. Bulg J Plant Physiol 2003;29:87–95.
- [220] Milivojevic DB, Nikolic BR, Drinic G. Effects of arsenic on phosphorus content in different organs and chlorophyll fluorescence in primary leaves of soybean. Biol Plant 2006;50:149–51.
- [221] Schmoger MEV, Oven M, Grill E. Detoxification of arsenic by phytochelatins in plants. Plant Physiol 2000;122:793-801.
- [222] Li Y, Dhankher OP, Carreira L, Lee D, Chen A, Schroeder JI, et al. Overexpression of phytochelatin synthase in *Arabidopsis* leads to enhanced arsenic tolerance and cadmium hypersensitivity. Plant Cell Physiol 2004;45:1787–97.
- [223] Khan I, Ahmada A, Iqbal M. Modulation of antioxidant defence system for arsenic detoxification in Indian mustard. Ecotoxicol Environ Saf 2009;72:626–34.
- [224] Rai R, Pandey S, Shrivastava AK, Pandey-Rai S. Enhanced photosynthesis and carbon metabolism favor arsenic tolerance in *Artemisia annua*, a medicinal plant as revealed by homology-based proteomics. Int J Proteomics 2014;2014:163962.
- [225] Requejo R, Tena M. Proteome analysis of maize roots reveals that oxidative stress is a main contributing factor to plant arsenic toxicity. Phytochemistry 2005;66:1519–28.
- [226] Requejo R, Tena M. Maize response to acute arsenic toxicity as revealed by proteome analysis of plant shoots. Proteomics 2006;6:156–62.

- [227] Ahsan N, Lee DG, Alam I. Comparative proteomic study of arsenic-induced differentially expressed proteins in rice roots reveals glutathione plays a central role during As stress. Proteomics 2008;8:3561–76.
- [228] Ahsan N, Lee DG, Kim KH, Alam I, Lee SH, Lee KW, et al. Analysis of arsenic stress-induced differentially expressed proteins in rice leaves by two-dimensional gel electrophoresis coupled with mass spectrometry. Chemosphere 2010;78:224–31.
- [229] Duquesnoy I, Goapil P, Nadaud I, Branlard G, Pissaloux AP, Gerard L. Identification of *Agrostis tenuis* leaf proteins in response to As(V) and As(III) induced stress using a proteomics approach. Plant Sci 2009;176:206–13.
- [230] Bona E, Cattaneo C, Cesaro P, Marsano F, Lingua G, Cavaleto M, et al. Proteomic analysis of *Pteris vittata* fronds: two arbuscular mycorrhizal fungi differentially odulate protein expression under As contamination. Proteomics 2010;10:3811–34.
- [231] Bona E, Marsano F, Massa N, Cattaneo C, Cesaro P, Argese E, et al. Proteomic analysis as a tool for investigating arsenic stress in *Pteris vittata* roots colonized or not by arbuscular mycorrhizal symbiosis. J Proteomics 2011;74:1338–50.
- [232] Li RY, Ago Y, Liu WJ, Mitani N, Feldmann J, McGrath SP, et al. The rice aquaporin Lsi1 mediates uptake of methylated arsenic species. Plant Physiol 2009;150:2071–80.
- [233] Ye WL, Wood BA, Stroud JL, Andralojc PJ, Raab A, McGrath SP, et al. Arsenic speciation in phloem and xylem exudates of castorbean. Plant Physiol 2010;154:1505–13.
- [234] Styblo M, DelRazo LM, Vega L, Germolec DR, LeCluyse EL, Hamilton GA, et al. Comparative toxicity of trivalent and pentavalent inorganic and methylated arsenicals in rat and human cells. Arch Toxicol 2000;74:289–99.
- [235] Naranmandura H, Xu S, Sawata T, Hao WH, Liu H, Bu N, et al. Mitochondria are the main target organelle for trivalent monomethyl arsonous acid (MMAIII) induced cytotoxicity. Chem Res Toxicol 2011;24:1094–103.
- [236] Száková J, Havlík J, Valterová B, Tlustoš P, Goessler W. The contents of risk elements, arsenic speciation, and possible interactions of elements and betalains in beet root (*Beta vulgaris* L.) growing in contaminated soil. Cent Eur J Biol 2010;5:692–701.
- [237] Orsit BA, Cleland WW. Inhibition and kinetic mechanism of rabbit muscle glyceraldehyde-3-phosphatedehydrogenase. Biochemistry 1972;11:102–9.
- [238] Park Jr RE, Agrawal RP. Purine nucleoside phosphorylase. In: Boyer PD, editor. The enzymes. New York: Academic Press; 1972. p. 483–514.
- [239] Long JW, Ray Jr WJ. Kinetics and thermodynamics of the formation of glucose arsenate. Reaction of glucose arsenate with phosphoglucomutase. Biochemistry 1973;12:3932–7.
- [240] Gresser MJ. ADP-arsenate. Formation by submitochondrial particles under phosphorylating conditions. J Biol Chem 1981;256:5981–3.
- [241] Guo B, Jin Y, Wussler C, Blancaflor EB, Motes CM, Versaw WK. Functional analysis of the *Arabidopsis* PHT4 family of intra-cellular phosphate transporters. New Phytol 2008;177:889–98.
- [242] Tawfik DS, Viola RE. Arsenate replacing phosphate: alternative life chemistries and ion promiscuity. Biochemistry 2011;50:1128–34.
- [243] Moore SA, Moennich DM, Gresser MJ. Synthesis and hydrolysis of ADP-arsenate by beef heart submitochondrial particles. J Biol Chem 1983;258:6266–71.
- [244] Kish MM, Viola RE. Oxyanion specificity of aspartate beta-semialdehyde dehydrogenase. Inorg Chem 1999;38:818–20.
- [245] Rosen BP, Ajees AA, McDermott TR. Life and death with arsenic. Bioessays 2011;33:350-7.
- [246] Avron M, Jagendorf A. Evidence concerning the mechanism of adenosine triphosphate formation by spinach chloroplasts. J Biol Chem 1959;234:967–72.

- [247] TerWelle HF, Slater EC. Uncoupling of respiratory-chain phosphorylation by arsenate. Biochim Biophys Acta 1967;143:1–17.
- [248] Watling-Payne AS, Selwyn MJ. Inhibition and uncoupling of photophosphorylation in isolated chloroplasts by organotin, organomercury and diphenylene iodonium compounds. Biochem J 1974;142:65–74.
- [249] Wickes WA, Wiskich JT. Arsenate uncoupling of oxidative phosphorylation in isolated plant mitochondria. Aust J Plant Physiol 1975;3:153-62.
- [250] Kitchin KT, Wallace K. Dissociation of arsenite-peptide complexes: triphasic nature, rate constants, halflives, and biological importance. J Biochem Mol Toxicol 2006;20:48–56.
- [251] Peters RA, Sinclair HM, Thompson RHS. An analysis of the inhibition of pyruvate oxidation by arsenicals in relation to the enzyme theory of vesication. Biochem J 1946;40:516–24.
- [252] Bergquist ER, Fischer RJ, Sugden KD, Martin BD. Inhibition by methylated organoarsenicals of the respiratory 2-oxo-acid dehydrogenases. J Organomet Chem 2009;694:973–80.
- [253] Kitchin KT, Wallace K. Arsenite binding to synthetic peptides: the effect of increasing length between two cysteines. J Biochem Mol Toxicol 2006;20:35–8.
- [254] Cline DJ, Thorpe C, Schneider JP. Effects of As(III) binding on alpha-helical structure. J Am Chem Soc 2003;125:2923–9.
- [255] Ramadan D, Cline DJ, Bai S, Thorpe C, Schneider JP. Effects of As(III) binding on β -hairpin structure. J Am Chem Soc 2007;129:2981–8.
- [256] Petrick JS, Jagadish B, Mash EA, Aposhian HV. Monomethylarsenous acid [MMA(III)] and arsenite: LD50 in hamsters and in vitro inhibition of pyruvate dehydrogenase. Chem Res Toxicol 2001;14:651–6.
- [257] Styblo M, Serves S, Cullen W, Thomas D. Comparative inhibition of yeast glutathione reductase by arsenicals and arsenothiols. Chem Res Toxicol 1997;10:27–33.
- [258] Lin S, Cullen WR, Thomas DJ. Methylarsenicals and arsinothiols are potent inhibitors of mouse liver thioredoxin reductase. Chem Res Toxicol 1999;12:924–30.
- [259] Zhao FJ, Stroud JL, Khan MA, McGrath SP. Arsenic translocation in rice investigated using radioactive73As tracer. Plant Soil 2012;350:413–20.
- [260] Liu WJ, Wood BA, Raab A, McGrath SP, Zhao FJ, Feldmann J. Complexation of arsenite with phytochelatins reduces arsenite efflux and translocation from roots to shoots in *Arabidopsis*. Plant Physiol 2010;152:2211–21.
- [261] Song WY, Park J, Mendoza-Cózatl DG, Suter-Grotemeyer M, Shim D, Hortensteiner S, et al. Arsenic tolerance in *Arabidopsis* is mediated by two ABCC-type phytochelatin transporters. Proc Natl Acad Sci USA 2010;107:21187–92.
- [262] Norton GJ, Lou-Hing DE, Meharg AA, Price AH. Rice-arsenate interactions in hydroponics: whole genome transcriptional analysis. J Exp Bot 2008;59:2267–76.
- [263] Chakrabarty D, Trivedi PK, Misra P, Tiwari M, Shri M, Shukla D, et al. Comparative transcriptome analysis of arsenate and arsenite stresses in rice seedlings. Chemosphere 2009;74:688–702.
- [264] Zhao FJ, Wang JR, Barker JHA, Schat H, Bleeker PM, McGrath SP. The role of phytochelatins in arsenic tolerance in the hyperaccumulator *Pteris vittata*. New Phytol 2003;159:403–10.
- [265] Pilon-Smits E, Pilon M. Phytoremediation of metals using transgenic plants. Cr Rev Plant Sci 2002;5:439–56.
- [266] Yang XE, Jin XF, Feng Y, Islam E. Molecular mechanisms and genetic basis of heavy metal tolerance/ hyperaccumulation in plants. J Int Plant Biol 2005;47:1025–35.
- [267] Visoottiviseth P, Francesconi K, Sridokchan W. The potential of Thai indigenous plant species for the phytoremediation of arsenic contaminated land. Environ Pollut 2002;118:453–61.
- [268] Foyer CH, Noctor G, Hodges M. Respiration and nitrogen assimilation: targeting mitochondriaassociated metabolism as a means to enhance nitrogen use efficiency. J Exp Bot 2011;62:1467–82.

- [269] Duan GL, Zhu YG, Tong YP, Cai C, Kneer R. Characterization of arsenate reductase in the extract of roots and fronds of Chinese brake fern, an arsenic hyperaccumulator. Plant Physiol 2005;138:461–9.
- [270] Bleeker PM, Hakvoort HWJ, Bliek M, Souer E, Schat H. Enhanced arsenate reduction by a CDC25like tyrosine phosphatase explains increased phytochelatin accumulation in arsenate-tolerant *Holcus lanatus*. Plant J 2006;45:917–29.
- [271] Duan GL, Zhou Y, Tong YP, Mukhopadhyay R, Rosen BP, Zhu YG. A CDC25 homologue from rice functions as an arsenate reductase. New Phytol 2007;174:311–21.
- [272] Németi B, Gregus Z. Mitochondria work as reactors in reducing arsenate to arsenite. Toxicol Appl Pharmacol 2002;182:208-18.
- [273] Radabaugh TR, Sampayo-Reyes A, Zakharyan RA, Aposhian HV. Arsenate reductase II. Purine nucleoside phosphorylase in the presence of dihydrolipoic acid is a route for reduction of arsenate to arsenite in mammalian systems. Chem Res Toxicol 2002;15:692–8.
- [274] Németi B, Regonesi ME, Tortora P, Gregus Z. The mechanism of the polynucleotide phosphorylase catalyzed arsenolysis of ADP. Biochimie 2011;93:624–7.
- [275] Zaffagnini M, Michelet L, Marchand C, Sparla F, Decottignies P, Le Maréchal P, et al. The thioredoxinindependent isoform of chloroplastic glyceraldehyde-3-phosphate dehydrogenase is selectively regulated by glutathionylation. FEBS J 2007;274:212–26.
- [276] Rathinasabapathi B, Wu S, Sundaram S, Rivoal J, Srivastava M, Ma LQ. Arsenic resistance in *Pteris vit-tata* L.: identification of a cytosolic triose phosphate isomerase based on cDNA expression cloning in *Escherichia coli*. Plant Mol Biol 2006;62:845–57.
- [277] Delnomdedieu M, Basti MM, Otvos JD, Thomas DJ. Reduction and binding of arsenate and dimethyl arsinate by glutathione amagnetic resonance study. Chem Biol Interact 1994;90:139–55.
- [278] Hartley-Whitaker J, Ainsworth G, Meharg AA. Copper and arsenate-induced oxidative stress in *Holcus lanatus* L. clones with differential sensitivity. Plant Cell Environ 2001;24:713–22.
- [279] Verbruggen N, Hermans C, Schat H. Mechanisms to cope with arsenic or cadmium in plants. Curr Opin Plant Biol 2009;12:1–9.
- [280] Hartley-Whitaker J, Woods C, Meharg AA. Is differential phytochelatins production related to decreased arsenate influx in arsenate tolerant *Holcus lanatus*? New Phytol 2002;155:219–25.
- [281] Tlustos P, Szakova J, Hruby J, Hartman I, Najmanova J, Nedelnik J, et al. Removal of As, Cd, Pb, and Zn from contaminated soil by high biomass producing plants. Plant Soil Environ 2006;52:413–23.
- [282] Grill E, Winnacker E-L, Zenk MH. Phytochelatins: the principal heavy metal complexing peptides of higher plants. Science 1985;230:674–6.
- [283] Tong YP, Kneer R, Zhu YG. Vacuolar compartmentalization: a second generation approach to engineering plants for phytoremediation. Trends Plant Sci 2004;9:7–9.
- [284] Sneller FEC, Van Heerwaarden LM, Kraaijeveld-Smit FJL, Ten Bookum WM, Koevoets PLM, Schat H, et al. Toxicity of arsenate in *Silene vulgaris*, accumulation and degradation of arsenate-induced phytochelatins. New Phytol 1999;144:223–32.
- [285] Hartley-Whitaker J, Ainsworth G, Voojis R, Bookum WT, Schat H, Meharg AA. Phytochelatins are involved in differential arsenate tolerance in *Holcus lanatus*. Plant Physiol 2001;126:299–306.
- [286] Gupta DK, Tohoyama H, Joho M, Inouhe M. Changes in the levels of phytochelatins and related metalbinding peptides in chickpea seedlings exposed to arsenic and different heavy metal ions. J Plant Res 2004;117:253–6.
- [287] Schulz H, Hartling S, Tanneberg H. The identification and quantification of arsenic-induced phytochelatins—comparison between plants with varying As sensitivities. Plant Soil 2008;303:275–87.
- [288] Wojasa S, Clemens S, Skłodowskac A, Antosiewicza DM. Arsenic response of AtPCS1 and CePCSexpressing plants—effects of external As(V) concentration on As-accumulation pattern and NPT metabolism. J Plant Physiol 2010;167:169–75.

- [289] Mascher R, Lippmann B, Holzinger S, Bergmann H. Arsenate toxicity: effects on oxidative stress response molecules and enzymes in red clover plants. Plant Sci 2002;163:961–9.
- [290] Shri M, Kumar S, Chakrabarty D, Trivedi PK, Mallick S, Misra P, et al. Effect of arsenic on growth, oxidative stress, and antioxidant system in rice seedlings. Ecotoxicol Environ Saf 2009;72:1102–10.
- [291] Cao X, Ma LQ, Tub C. Antioxidative responses to arsenic in the arsenic-hyperaccumulator Chinese brakefern (*Pteris vittata* L.). Environ Pollut 2004;128:317–25.
- [292] Abercrombie JM, Halfhill MD, Ranjan P, Rao MR, Saxton AM, Yuan JS, et al. Transcriptional responses of *Arabidopsis thaliana* plants to As(V) stress. BMC Plant Biol 2008;6:80–7.
- [293] Gasic K, Korban SS. Transgenic Indian mustard (*Brassica juncea*) plants expressing an *Arabidopsis* phytochelatin synthase (AtPCS1) exhibit enhanced As and Cd tolerance. Plant Mol Biol 2007;64:361–9.
- [294] Bohnert HJ, Jensen RG. Strategies for engineering water-stress tolerance in plants. Trends Biotechnol 1996;14:89–97.
- [295] Matysik J, Alia BB, Mohanty P. Molecular mechanism of quenching of reactive oxygen species by proline under stress in plants. Curr Sci 2002;82:525–32.
- [296] Rhodes D, Hanson AD. Quarternary ammonium and tertiary sulphonium compounds in higher plants. Ann Rev Plant Physiol Plant Mol Biol 1993;44:357–84.
- [297] Gorham J. Betaines in higher plants—biosynthesis and role in stress metabolism. In: Wallsgrove RM, editor. Amino acids and their derivatives in higher plants. Cambridge: Cambridge University Press; 1995. p. 171–203.
- [298] Papageorgiou GC, Murata N. The unusually strong stabilizing effects of glycine betaine on the structure and function of the oxygen evolving photosystem II complex. Photosynth Res 1995;44:243–52.
- [299] Groppa M, Tomaro M, Benavides M. Polyamines as protectors against cadmium or cooper-induced oxidative damage in sunflower leaf discs. Plant Sci 2001;161:481–8.
- [300] Castillo-Michel H, Parsons JG, Peralta-Videa JR, Martinez-Martinez A, Dokken KM, Gardea-Torresdey JL. Use of X-ray absorption spectroscopy and biochemical techniques to characterize arsenic uptake and reduction in pea (*Pisum sativum*) plants. Plant Physiol Biochem 2007;45:457–63.
- [301] Cozzolino V, Pigna M, Di Meo V, Caporale AG, Violante A. Effects of arbuscular mycorrhizal inoculation and phosphorus supply on the growth of *Lactuca sativa* L. and arsenic and phosphorus availability in an arsenic polluted soil under non-sterile conditions. Appl Soil Ecol 2010;45:262–8.
- [302] Gonzalez-Chavez C, D'Haen J, Vangronsveld JJ, Dodd JC. Copper sorption and accumulation by the extra aradical mycelium of different *Glomus* spp. (arbuscular mycorrhizal fungi) isolated from the same polluted soil. Plant Soil 2002;240:287–97.
- [303] Sharples JM, Meharg AA, Chambers SM, Cairney JWG. Symbiotic solution to arsenic contamination. Nature 2000;404:951–2.
- [304] Chen BD, Christie P, Zhu YG, Smith FA, Xie ZM, Smith SE. The arbuscular mycorrhizal fungus *Glomus mosseae* gives contradictory effects on phosphorus and arsenic acquisition by *Medicago sativa* Linn. Sci Total Environ 2007;379:226–34.
- [305] Ultra Jr VU, Tanaka S, Sakurai K, Iwasaki K. Effects of arbuscular mycorrhiza and phosphorus application on arsenic toxicity in sunflower (*Helianthus annuus* L.) and on the transformation of arsenic in the rhizosphere. Plant Soil 2007;290:29–41.
- [306] Ultra Jr VUY, Tanaka S, Sakurai K, Iwasaki K. Arbuscular mycorrhizal fungus (*Glomus aggregatum*) influences biotransformation of arsenic in the rhizosphere of sunflower (*Helianthus annuus* L.). Soil Sci Plant Nutr 2007;53:499–508.
- [307] Tu C, Ma LQ. Effects of arsenic concentrations and forms on arsenic uptake by the hyperaccumulator ladder brake. J Environ Qual 2002;31:641–7.
- [308] Zhao FJ, Dunham SJ, McGrath SP. Arsenic hyperaccumulation by different fern species. New Phytol 2002;156:27–31.

- [309] Wang HB, Wong MH, Lan CY, Baker AJM, Qin YR, Shu WS, et al. Uptake and accumulation of arsenic by 11 *Pteris* taxa from southern China. Environ Pollution 2007;145:225–33.
- [**310**] Caille N, Zhao FJ, McGrath SP. Comparison of root absorption, translocation and tolerance of arsenic in the hyperaccumulator *Pteris vittata* and the non hyperaccumulator *Pteris tremula*. New Phytol 2005;165:755–61.
- [311] Singh N, Ma LQ. Arsenic speciation, and arsenic and phosphate distribution in arsenic hyperaccumulator *Pteris vittata* L. and nonhyperaccumulator *Pteris ensiformis* L. Environ Pollution 2006;141:238–46.
- [312] Srivastava M, Ma LQ, Singh N, Singh S. Antioxidant responses of hyper-accumulator and sensitive fern species to arsenic. J Exp Bot 2005;56:1335–42.
- [313] Zhang WH, Cai Y, Downum KR, Ma LQ. Thiol synthesis and arsenic hyperaccumulation in *Pteris vittata* (Chinese brake fern). Environ Pol 2004;131:337–45.
- [314] Francesconi K, Visoottiviseth P, Sridokchan W, Goessler W. Arsenic species in an arsenic hyperaccumulating fern, *Pityrogramma calomelanos*: a potential phytoremediator of arsenic-contaminated soils. Sci Total Environ 2002;284:27–35.
- [315] Zhang WH, Cai Y, Tu C, Ma LQ. Arsenic speciation and distribution in an arsenic hyperaccumulating plant. Sci Total Environ 2002;300:167–77.
- [316] Webb SM, Gaillard JF, Ma LQ, Tu C. XAS speciation of arsenic in a hyper-accumulating fern. Environ Sci Technol 2003;37:754–60.
- [317] Poynton CY, Huang JWW, Blaylock MJ, Kochian LV, Elless MP. Mechanisms of arsenic hyperaccumulation in *Pteris* species: root As influx and translocation. Planta 2004;219:1080–8.
- [318] Liu WJ, Zhu YG, Hu Y, Williams PN, Gault AG, Meharg AA, et al. Arsenic sequestration in iron plaque, its accumulation and speciation in mature rice plants (*Oryza sativa* L.). Environ Sci Technol 2006;40:5730–6.

28

Arsenic Contents and Its Biotransformation in the Marine Environment

Kiran Kalia and Devang B. Khambholja

LABORATORY OF BIOCHEMISTRY, BHANUBEN AND RATILAL DOSHI SCHOOL OF BIOSCIENCES, SARDAR PATEL MAIDAN, SARDAR PATEL UNIVERSITY, VALLABH VIDYANAGAR, GUJARAT, INDIA

CHAPTER OUTLINE

28.1	Introduction	675
28.2	Arsenic Concentration in Sea Water	676
28.3	Arsenic Concentration in Marine Sediments	679
28.4	Arsenic Speciation in Marine Ecosystems	681
28.5	Arsenic Cycle in the Marine Environment	683
28.6	Role of Marine Biological Systems in Arsenic Biotransformation	686
28.7	Arsenic in Seafood and Its Toxicity	691
28.8	Future Directions	694
Refe	rences	694

28.1 Introduction

Arsenic has been known as the king of poisons and the poison of kings, many cases of its use as a poison being reported in the 1900s. The suspected cause of death of the well-known French ruler Napoleon Bonaparte is one such example of arsenic poisoning [1]. It has recently become one of the major environmental toxicants due to its increased application in wood preservatives, insecticides, fungicides, fertilizers, and semiconductors. The mining and burning of fossil fuels have broadened the global cycle of arsenic, which has led to the accumulation of large amounts of arsenic in the environment. Natural sources such as volcanoes, marine hydrothermal fluids, and microorganisms also contribute to arsenic pollution. Arsenic from these many natural and anthropogenic sources is transported into the marine environment through atmospheric depositions and riverine inputs. Coastal areas and estuaries of many countries including Bangladesh, Australia, Korea, Japan, China, Malaysia, India, the UK, and the USA have reported elevated arsenic pollution.

The arsenic biogeochemical cycle in marine environment involves several physicochemical (such as oxidation/reduction, precipitation/solubilization, and adsorption/desorption) as well as biological processes, in which marine organisms play a key role [2]. The toxic effects of arsenic are related to its speciation; thus, the examination of factors affecting the speciation of arsenic has become an important issue [3]. Biological (marine bacteria, phytoplankton, algae, and other marine organisms) factors perform a significant contribution to arsenic speciation and the biogeochemical cycle of arsenic in the marine environment.

Humans may be exposed to arsenic via contaminated drinking water and food, especially seafood [4], which is generally contaminated with higher concentrations of arsenic than found in other foods [5]. Countries like Japan, China, and Korea obtain their food from marine sources, which constitute a major part of their diet [6,7]. In view of the suspected human carcinogenicity of arsenite, the United States Environmental Protection Agency (USEPA) has set a human health criterion for total dissolved arsenic in sea water (0.0175 g/L) for the consumption of fish products [8]. Millions of people worldwide are at risk due to long-term arsenic-contaminated groundwater consumption [9], which has become a global public health problem. Inorganic arsenite has been classified by the International Agency for Research on Cancer as a human carcinogen [8]. Acute toxicity symptoms of inorganic arsenic in humans include cardiovascular disturbances, gastrointestinal disorders, and kidney and liver failure, whereas chronic exposure may lead to hyperkeratosis, skin pigmentations, and cancer of skin, lung, bladder, liver, and kidney [10]. The toxicity of arsenic is strongly related to the chemical species in which it is present in seafood. The organic forms of arsenic contribute more to the major portion of total arsenic in seafood than do inorganic forms. Inorganic arsenic contributes less than 1 to 4% of total arsenic in seafood [11]. The majority of organic forms of arsenic are non-toxic or moderately toxic only at elevated exposure. Recent studies on the trivalent oxidation state of methylated arsenicals suggest that this state has greater toxicity than the corresponding pentavalent methylated arsenicals and inorganic arsenite [12].

This chapter summarizes arsenic (As) concentrations and the biogeochemical cycle of As in global marine and biological systems (marine animals, bacteria, phytoplankton, etc.), and its transformation and association with the risk of arsenic toxicity to seafood consumers.

28.2 Arsenic Concentration in Sea Water

Arsenic is the 14th most abundant element in sea water [13], and its average concentration ranges between 1 and $2\mu g/L$ in unpolluted sea water [14]. Natural sources included volcanic eruption, hot springs, atmospheric deposition through low temperature volatilization, sea spray, riverine input, geysers, and marine hydrothermal fluids from active oceanic ridges, which play a significant role in arsenic contamination in marine waters [15–17]. Average arsenic concentrations in marine waters are shown in Table 28–1. Generally, the concentration of arsenic exhibits little variation in open oceans, while estuarine water shows increased variation as a result of varying riverine inputs, salinity or redox gradients [32]. In the

Marine Water	Arsenic Contents Range (average)	References
Pacific Ocean		
Northwest Pacific	(1.6µg/L)	[18]
Central north Pacific gyre	(1.3 µg/L)	[18]
West Pacific equatorial region	(1.4µg/L)	[18]
Southwest Pacific	(1.17 μg/L)	[19]
Atlantic Ocean		
North Atlantic	(1.18±0.09µg/L)	[20]
Eastern North Atlantic		
Surface	(1.04±0.01µg/L)	[21]
Bottom	$(1.6 \pm 0.003 \mu g/L)$	
West Atlantic	$0.6-1.7 \mu$ g/L $(1.22 \pm 0.16 \mu$ g/L)	[22]
Indian Ocean		
East Indian	(0.9 µg/L)	[19]
North Indian	(0.8 µg/L)	[19]
Antarctic Ocean	(1.1µg/L)	[19]
Southern Ocean		
Surface	$(1.7 \pm 0.06 \mu g/L)$	[23]
Deep	$(1.8 \pm 0.04 \mu g/L)$	
Yellow Sea	0.6–1.6 µg/L (1.1 µg/L)	[15]
East China Sea	0.6–1.6 µg/L (1.2 µg/L)	[15]
Southern Tasman Sea	(1.4 µg/L)	[18]
China Sea	(0.6µg/L)	[19]
Galway Bay, Ireland	(1.7 μg/L)	[17]
Coasts, Malaysia	0.7–1.8 μg/L (1.0 μg/L)	[24]
Southeast Coasts, Spain	0.5–3.7 μg/L (1.5 μg/L)	[25]
Coastal Nakaminato, Japan	(3.1µg/L)	[26]
Gwangyang Bay coast, Republic of Korea	Below detection limit—27.9 µg/L	[27]
Southern coast, Australia	1.1–1.6 μg/L (1.3 μg/L)	[28]
Rhône estuary, France	1.3–3.7 μg/L	[29]
Scheldt estuary, Belgium	1.8–4.9μg/L	[30]
Krka estuary, Yugoslavia	1.8µg/L	[31]

 Table 28–1
 Arsenic Concentrations in Marine Water

uncontaminated Krka estuary in Croatia, a linear increase in total arsenic was observed with increasing salinity ranging from $0.13 \,\mu$ g/L in fresh water to $1.8 \,\mu$ g/L in sea water [31]. Usually, its concentration is high when riverine inputs are affected by industrial or mining effluents. There is only a small contribution made by human activities to arsenic loads in open oceans, but estuaries and coastal waters may receive arsenic-contaminated discharges from the land sources. Arsenic concentrations reported in the Rhone estuary in France ranged between 1.3 and $3.7 \,\mu$ g/L [29], while a slightly elevated level of 2.7– $8.8 \,\mu$ g/L was observed in the Tamar estuary in southwest England [33]. Total dissolved arsenic in the Scheldt estuary, Belgium, was

between 1.8 and $4.9\,\mu$ g/L [30]. The Yellow Sea and East China Sea were strongly affected by input from freshwater discharge of the adjacent Changjiang (Yangtze) River in the west and the Kuroshio intrusion in the east, which had average total dissolved inorganic arsenic concentrations in samples, collected in 2002, of $1.12\,\mu$ g/L and $1.21\,\mu$ g/L in the Yellow Sea and East China Sea, respectively [15]. On the other hand, these concentrations are relatively lower than in other reported polluted sites, where small amounts of arsenic in estuarine waters enter the oceans. Physical processes, like adsorption of dissolved inorganic arsenic to iron oxides and flocculation of particulates at the fresh/saline water interface, have important consequences on increases in pH and salinity. This has led to the transfer of arsenic to estuary sediments and restricted arsenic flux to the oceans [14,34].

In continental shelf waters, the dissolved arsenic concentration of surface water was found to vary seasonally due to natural cycling between sediments and the overlying water column [35]. The lowest concentration is present in the spring when dissolved arsenate is scavenged by phytoplankton blooms and suspended particles. Arsenic concentration is increased in autumn due to remobilization of arsenic from sediments [8]. Total inorganic arsenic is depleted in surface waters of the open oceans [21,36]. Total inorganic arsenic concentration in surface water of the Atlantic Ocean ranges between 0.94 ± 0.02 and $1.07 \pm 0.02 \,\mu$ g/L, collected from three different sampling stations. Relatively higher and constant concentrations are being reported in deep water (between 2300 and 4400 m) and ranges between 1.33 ± 0.03 and $1.56 \pm 0.04 \,\mu$ g/L [21]. A similar type of vertical distribution pattern of arsenic was observed in the North Pacific Ocean [36]. A good correlation usually exists between concentrations of arsenic in concentration ocean water, suggesting that the vertical distribution of arsenic in the ocean is controlled by biological activity. Another correlation suggests that arsenate concentration minima often coincide with photosynthetic maxima evidenced by high concentrations of chlorophyll *a* [34].

Average arsenic concentrations in the surface sea water of the Northwest Pacific Ocean, central Pacific gyre, Western Pacific equatorial region, and Southwest Pacific Ocean are $1.6\,\mu$ g/L, $1.3\,\mu$ g/L, $1.4\,\mu$ g/L, and $1.2\,\mu$ g/L, respectively [18,19]. The east and north Indian Ocean showed average arsenic concentrations of $0.9 \,\mu g/L$ and $0.8 \,\mu g/L$, respectively [19], whereas the Antarctic Ocean and Southern Ocean had $1.1 \,\mu$ g/L and $1.7 \pm 0.1 \,\mu$ g/L, respectively, in surface sea water [19,23]. Average arsenic concentrations in clear surface water of North Atlantic, eastern North Atlantic, and western Atlantic Ocean were found to be $1.2 \pm 0.1 \,\mu$ g/L, $1.0 \pm 0.05 \,\mu\text{g/L}$, and $1.22 \pm 0.4 \,\mu\text{g/L}$, respectively [20–22], while a slightly elevated level was reported in deep water of the eastern North Atlantic of $1.56 \pm 0.03 \,\mu\text{g/L}$ and South Atlantic, $1.4 \pm 0.1 \,\mu$ g/L [21]. Average concentration of total arsenic in the open coastal Pacific Ocean was around $1.0 \,\mu\text{g/L}$ [37] and $1.8 \pm 0.04 \,\mu\text{g/L}$ in the North Pacific [36], and in the deep Pacific and Atlantic Oceans the value lay between 1.0 and 1.8 µg/L [34]. Average arsenic concentration reported for the southeast coast of Spain was $1.5 \,\mu g/L$ [25], $1.3 \,\mu g/L$ for the southern coast of Australia [28], and 1.0 µg/L for the coasts of Malaysia [24]. Gwangyang Bay, Republic of Korea, showed elevated total arsenic concentrations in coastal sea water of below the detection limit $(27.9\,\mu g/L)$ [27], whereas Galway Bay, Ireland, showed $1.7\,\mu g/L$ of arsenic [17]. Average arsenic concentration reported in the southern Tasman Sea was $1.4 \mu g/L$ [18], while in the China Sea it was $0.63 \mu g/L$ [19]. The level of arsenic for the Pacific coast near Onagawa (Miyagi, Japan) was lower, $0.6 \mu g/L$ compared to Nakaminato (Ibaraki, Japan), which was around $3.1 \mu g/L$ [26]. The concentration readings for arsenic in marine water were more consistent than those for fresh waters [8,32].

Arsenic is one of the few trace elements whose concentration in sea water was found to be higher than in river water, which may be due to the flux of arsenic from hydrothermal systems [16]. Marine hydrothermal fluid has played an important role in arsenic cycling in marine environments which emerged at the seafloor near mid-ocean ridges or in back-arc basins and island arc settings. Due to water rock interaction at elevated temperatures and potentially magmatic degassing, considerable amounts of dissolved arsenic has leached out from hydrothermal fluid systems. Fluid from mid-ocean ridge hydrothermal systems such as the East Pacific Rise had reported levels of around $80.5 \,\mu$ g/L arsenic while at the Mid-Atlantic Ridge, the highest reported level was around $24.0 \,\mu$ g/L. A highly elevated arsenic concentration of 1386.0 μ g/L was reported for back-arc basin settings and of 5850.0 μ g/L for island arc shallow water settings [16].

28.3 Arsenic Concentration in Marine Sediments

Marine sediments comprise significant sinks of contaminants and represent potential sources of pollution to the marine environment. The total arsenic influx into oceans was estimated at 246,110 metric tons/year. Of this total, 178,900 metric tons was sediment suspended arsenic, 62,900 metric tons was dissolved arsenic, and 4310 metric tons was derived from the atmosphere per year [32]. Total arsenic concentrations in uncontaminated near-shore marine and estuarine sediments were between 5 and $15 \,\mu\text{g/g}$ dry weight, while, deep-sea sediments contained an average arsenic concentration of approximately $40 \,\mu\text{g/g}$ [8]. Sediments from estuaries and coasts receiving drainage from industries and metal-mining areas contain significantly elevated arsenic concentrations, which threaten the aquatic biota and have been of significant environmental concern [8,38]. The coastal sediments of Gwangyang Bay, Republic of Korea, have total arsenic concentrations between 6.89 and $355 \,\mu\text{g/g}$, which are higher near the mining area [27].

Arsenate is the most dominant arsenic species present in oxidized marine sediments, whereas arsenite is the dominant species in reduced sediment layers. Iron and manganese (oxy)-hydroxide, either free or coated with mineral clays and organic substances, are the major sinks for arsenic entering into marine water. Such substances scavenge dissolved arsenic from the water column and transfer them into the sediments [39]. The affinity of pelagic clays for adsorption of arsenite was three times higher than that for arsenate. Most of the arsenite that was adsorbed by iron oxides in oxidized marine sediments was rapidly oxidized to arsenate. The bacterial species present in aerobic sediments have the ability to oxidize arsenite to arsenate [8]. Arsenic concentrations in the East China Sea sediments ranged from 1.70 to $22.1 \,\mu$ g/g with an average of $11.5 \,\mu$ g/g (Table 28–2). The distribution pattern of arsenic in the East China Sea sediments was mainly controlled by clay and total organic contents, with higher arsenic concentration near the Changjiang estuary and Hangzhou Bay mouth. This indicates that

Marine Sediment	Arsenic Contents Range (average)	References
East China Sea	1.70–22.1 µg/g (11.5 µg/g)	[40]
French Mediterranean Estaque port		
Surface (0–10 cm)	194µg/g and 107µg/g	[41]
Depth (10–20 cm)	199µg/g and 220µg/g	
French Mediterranean Saint Mandrier port		
Surface (0–10 cm)	10 µg/g	[41]
Depth (10–20 cm)	12.1µg/g	
Southeastern Baltic Sea	1.1–19.0 µg/g (3.4 µg/g)	[42]
Chemical dumpsite Liepaja in Gotland	18–29 µg/g	[43]
Chemical dumpsite Skagerrak	9–200 µg/g (25 µg/g)	[43]
Chemical dumpsite Skagerrak	75–480μg/g	[44]
Western North Sea	<0.15–135µg/g	[39]
Cape Howe and Sugarloaf Point, east coast, Australia	2–180 µg/g	[45]
Baltimore Harbor, USA	25.0–41.1 μg/g	[46]

Table 28–2 Arsenic Contents in Marine Sediments

fine-grained sediments and riverine inputs are the main factors controlling arsenic distribution [40]. In contrast, when environmental conditions change (redox potential), arsenic associated with sediments is released to the overlying water column, threatening the marine biota [14]. It was also suggested that marine sediments as the secondary source of arsenic may affect marine ecology. In coastal and estuarine sediments redox changes occur more often at the time of seasonal changes, predominantly in spring and autumn [8]. Marine sediments are also an important source of methylated forms of arsenic to the overlying water column. Several species of aerobic and anaerobic sediment bacteria are able to accumulate arsenate and arsenite and convert them to methylated arsenic [47–49]. The methylated arsenic has been released by the bacteria into the sediment pore water, from which they may be mixed up into the overlying water column.

As shown in Table 28–2, sediment samples collected from the French Mediterranean port of Estaque showed arsenic contamination in surface (0–10 cm) and depth (10–20 cm) sediments of around 107 µg/g and 199 µg/g, respectively, collected during the cold season (March 2009). On the other hand, the levels were 194 µg/g and 220 µg/g, respectively, when collected during the hot season (September 2009). Another less polluted French Mediterranean port, Saint Mandrier, showed arsenic contamination in surface (0–10 cm) and depth (10–20 cm) sediments of around 10 µg/g and 12.1 µg/g, respectively [41]. Arsenic mobility studies in both these ports suggested that the exchangeable fraction represents 30% of total arsenic [41]. Sediment samples collected during the year 1992 from Baltimore Harbor, USA, showed arsenic concentrations between 25.0 and 41.1 µg/g [46]. Garnaga et al. [42] reported an average 9.7 µg/g of arsenic in sediment samples collected near the chemical munitions dumpsite in the Lithuanian economic zone of the Baltic Sea. Sediments in the Baltic Sea and North Sea were relatively high compared to arsenic concentrations reported from the chemical munitions dumpsite in the Lithuanian economic zone [42]. Studies on chemical munitions dumpsites in Gotland (Liepaja dumpsite) reported arsenic levels in the sediments to be between 18 and $28 \mu g/g$ [43]. The highest arsenic concentrations up to $480 \mu g/g$ were found in the sediments from the chemical munitions dumpsites in Skagerrak [44]. The concentrations of arsenic ranged from <0.15 to 135 $\mu g/g$ in the western North Sea and Humber estuary [39].

Oxygen plays a significant role in arsenic flux from sediments to the water column [46]. In anoxic conditions, the mobility of arsenic from sediments to the water column increases many fold more than that in hypoxic and oxic conditions. Marine sediments collected from an area between Sugarloaf Point and Cape Howe on the east coast of Australia showed arsenic concentration ranges of between 2 and $180 \mu g/g$ [45].

28.4 Arsenic Speciation in Marine Ecosystems

Arsenic sublimates at 613°C, melts at 817°C, and has a density of 5.27 g/cm³ [50]. Arsenic (As) is present in the form of various chemical species in the marine environment (Figure 28-1) and its toxicity depends on its chemical form. Due to its complex chemistry, behavior, and ecotoxicological effects on marine ecosystems, it is necessary to understand the nature of these As species. Arsenic occurs in the marine environment in four oxidation states: As⁺⁵, As⁺³, As⁰, and As⁻³. The two highest oxidation states are the most common biologically important redox states and are trivalent inorganic arsenic [arsenite; As(III)], the reduced form and pentavalent inorganic arsenic [arsenate; As(V)], and the oxidized form. On the other hand, the elemental form is very rare. As⁻³ is found only in highly reduced environments where electropotential (Eh) values are extremely low [8]. In natural conditions, both arsenite and arsenate are subjected to chemically and biologically mediated oxidation, reduction, methylation, and other reactions. This is the major reason for the availability of various species of arsenic in the marine ecosystem. Both arsenate and arsenite are pH dependent and interconvertible; thus at physiological pH, arsenate dominates as $H_2AsO_4^{-1}$ (2.5 < pH < 7) and $HAsO_4^{-2}$ (7 < pH < 12), while arsenite dominates as H_3AsO_3 (pH < 9.3) [51]. Arsenate is a more predominant and thermodynamically stable form in oxygenated surface marine waters [34], while arsenite is more common in anaerobic reduced environments. Pentavalent arsenate has chemical properties similar to those of phosphate, and thus it competes with phosphate in cellular reactions and inhibits oxidative phosphorylation, the key reaction of energy metabolism in metazoans, including humans, and is the most toxic form to algae [52]. However, trivalent arsenite has a strong predilection to form bonds with functional groups such as the thiolates of cysteine residues and the imidazolium nitrogens of histidine. Thus, it binds readily to many enzymes, including those involved in respiration [52]. Arsenite is reported to be as much as 60 times more toxic than arsenate to humans and to marine organisms [8].

Organic forms of arsenic are also present in the marine environment, often found in association with biological systems. These include methylated arsenicals such as monomethylarsonic acid (MMA(V)) and dimethylarsinic acid (DMA(V)), which are common arsenic metabolites that exist in most environmental compartments, and others like monomethylarsonous acid (MMA(III)), dimethylarsinous acid (DMA(III)), dimethylarsinoyl acetate,



FIGURE 28–1 Arsenic species commonly found in marine ecosystems. DMA(V)—dimethylarsinous acid, DMA(III) dimethylarsinous acid, TMAO—trimethylarsine oxide, TMAP—trimethylarsoniopropionate, AB—arsenobetaine, AC—arsenocholine, TETRA—tetramethylarsonium ion.

dimethylarsinoyl ethanol, arsenobetaine (AB), arsenocholine (AC), tetramethylarsonium ion (TETRA), arsenosugars (AsS) [53,54], and methylthioarsenic species, which are common forms in marine ecosystems. Arsenic is methylated by marine bacteria, algae, and phytoplankton, but neither form is considered to be as toxic as inorganic arsenate and arsenite compounds [55]. In the marine environment, the most predominant arsenic forms are inorganic [As(V) and As(III)] arsenic, which contribute around 80% of total arsenic, while methylated and other organic forms contribute around 20% of total arsenic [17]. The contribution of each arsenic species in surface water of east Indian Ocean is 52% for As(V), 27% for As(III), and 21% for methylated arsenic. Similar ratios were observed in surface water of the Northwest Pacific Ocean, which were 63% for As(V), 22% for As(III), and 15% for methylated arsenic [19], although seawater temperature,

nutrient availability, and biological activities affect the concentrations of various arsenic species in sea water. The rate of release of methylated arsenicals by the macroalga *Ascophyllum nodosum* is influenced by temperature changes, with a two-fold increase noted, on increasing the temperature to between 5 and 15°C, in the rate of release of monomethylarsenic by algae; the rate of release of dimethylarsenic is increased by more than six-fold [56]. The concentration of methylated arsenic is increased with increasing water temperature and nutrient availability in Pacific surface sea water suggesting that the biological activity of the phytoplankton community in surface waters is responsible for the abundance of dimethylarsinic acid and monomethylarsonic acid [18]. Dimethylarsenic was found to be the dominant organic form of arsenic in sea water at all temperatures. In the process of detoxification, some of the marine algae and certain marine animals convert absorbed arsenate from sea water to arsenosugars [57]. Arsenosugar was first identified in 1981 from brown kelp *Ecklonia radiata* [58]. A total 15 arsenosugars were identified from certain algae and marine animals [54,59,60]. Marine algae accumulated arsenic chiefly as arsenosugars, which comprised the largest group of naturally occurring arsenicals [61].

Another important and virtually ubiquitous organoarsenical in marine animals is arsenobetaine. It generally represents 50 to >95% of the total arsenic in tissues of marine crustaceans, elasmobranchs, teleost fish, polychaetes, and gastropods [62–64]. Environmental factors like elevated inorganic arsenic concentrations through the hydrothermal fluids and sediments and food (bacterial mat) cause an unusual speciation like tetramethylarsonium ion and trimethylarsoniopropionate, which were observed in gut and muscle of *Cyclope neritea* (a gastropod) from Paleochori Bay. The control gastropod from Ria do Alvor, Portugal, has almost exclusively arsenobetaine [63]. Recently, new classes of arsenic species, thioarsenate and thio-methylated arsenic, have been reported in geothermal waters and marine organisms, respectively. Thiomethylated species are thought to be biologically formed in the presence of sulfur-rich anaerobic environments [65]. Presently, more than 50 additional arsenic species have been reported in marine organisms. Their effects in terms of toxicity to humans, however, remain unclear [4].

28.5 Arsenic Cycle in the Marine Environment

Arsenic is a ubiquitous and uniformly distributed element in unpolluted marine environments. Although open-ocean systems are relatively constant, the possibility of anthropogenic inputs of arsenic may result in the higher arsenic concentration in coastal areas and estuaries. Additionally, natural inputs like riverine discharges, atmospheric depositions, and volcanic eruptions are significant contributors of arsenic. As shown in Figure 28–2, arsenic enters the marine environment mostly in dissolved inorganic forms [66]; however, the methylated arsenicals exist in comparatively low concentrations [19,20]. In the east Indian Ocean and Northwest Pacific Ocean, 89% and 85% of arsenic was found in the inorganic form, while methylated arsenic accounted for 21% and 15%, respectively [19]. Among the inorganic forms, arsenate is the most stable and predominant form in oxic euphotic zones, a surface water rich in oxygen and light as compared to deep water in marine environments [19,66]. The similarities in the chemical properties between arsenate and macro-nutrient phosphate are largely responsible for the production of reduced and methylated arsenic by marine phytoplankton, bacteria, and algae.



FIGURE 28–2 Arsenic cycle in the marine environment. [O]—oxidized, [R]—reduced, ADS—adsorbed. *Modified from* [15].

Arsenite has reciprocal correlation with phosphate [8], which suggests that the low phosphate concentration in the environment facilitates arsenate uptake by biological systems, which ultimately convert it to the reduced and methylated forms for their survival. Thus, phosphate concentration in sea water appears to regulate the reduction and methylation of arsenic in the marine ecosystem. The reduced and methylated forms of arsenic are produced during the process of detoxification and released back in the water column. Arsenite is a product of detoxification by bioreduction in well-oxygenated waters, which decreases rapidly below the thermocline as a thermodynamically unstable species. Its coexistence with arsenate in deep water is kinetically supported due to its relatively low oxidation rate. The methylated arsenicals produced by phytoplankton uptake are seen to be chemically stable, at least for a few months [67].

Two microorganisms, named *Apiotrichum humicola* and *Scopulariopsis brevicaulis* in laboratory experiments, are capable of forming arsenite, monomethylarsonate, dimethylarsinate, and trimethylarsine oxide when supplemented with arsenate-containing growth medium [68]. Numerous studies have described that the inorganic arsenic from sea water is converted and concentrated in the organic forms of arsenicals by bacteria [48], phytoplankton [59], and algae [69]. They are further metabolized through the food chain and accumulated as arsenobetaine in marine animals [19]. Arsenobetaine (trimethyl(carboxymethyl)arsonium zwitterion) is the main arsenic species in marine animals, while arsenoribosides are mainly present in seaweeds. Arsenobetaine is considered the final metabolite of arsenic in the marine food chain and accumulates in marine animals to greater or lesser extents [57,70]. The other forms of organoarsenicals, like arsenocholine, arsinylribosides, tetramethylarsonium ion, phosphatidylarsenocholine, MMA, and DMA, are reported in the tissues of various marine animals [71]. The majority of organoarsenicals ingested in successive food chains by higher tropical levels are rapidly excreted. The remaining small amount is incorporated within organisms, and exported in the higher fishes or deposited as dead material in marine sediments. Organoarsenicals, associated with biological systems and/or freely available in sea water, are degraded to complete the arsenic cycle in the marine ecosystem (inorganic arsenic to inorganic arsenic via formation of organic arsenic) [72]. Finally, arsenic-containing dead material and undissolved biogenic particles sink into the marine sediments, where they biodegraded with inorganic arsenic or lower forms of organoarsenicals being released back into the water column [66,73,74]. Decomposition occurs at the sediment level as well as at the euphotic seawater level. Bacterial isolates identified as members of the Vibrio/Aeromonas group have the capacity to decompose arsenobetaine to dimethylarsinic acid under aerobic conditions [75]. The laboratory experiments suggest that sedimentary microorganisms have the capacity to convert higher (complex) forms of methylated arsenicals to their lower forms under aerobic conditions and even convert them into a simpler and aerobically stable form, i.e., arsenate [76,77].

Co-precipitation of inorganic arsenic with hydrated heavy metal oxides may occur, and surface active suspended particulate matter may scavenge dissolved arsenic from the water column. Adsorption of arsenic onto iron and/or manganese oxides offers the major sink of arsenic in marine sediments. In the oxidized layer of the sediments, much of the arsenic is associated (co-precipitated or adsorbed) with the hydrous iron and manganese oxide fraction or is present as $Fe_3(AsO_4)_2$ [8]. Sedimented organic matter may also have the capacity to bind arsenic [78]. Sediment–water exchange has a significant impact on the arsenic biogeochemical cycle. Sedimentary diagenesis liberates arsenic to pore water, and thus the arsenic concentration in these interstitial waters reflects solid-phase arsenic levels, although the arsenic content always remains a minor fraction of total sedimentary levels [14].

The major mechanism for arsenic liberation into pore waters is the dissolution of hydrous oxide phases to which the metalloid is adsorbed. Under reducing conditions, Fe(III) and Mn(IV) are converted into their soluble reduced forms Fe(II) and Mn(II), respectively, which release adsorbed arsenic into the pore water [8]. In anoxic sediments, arsenate is reduced to arsenite, and is released upward into interstitial waters or reacts with sulfur if available in abundance. In most marine sediments, most of the arsenic reacts with sulfides to form realgar (AsS), arsenopyrite (FeAsS), or orpiment (As₂S₃) [8]. These sulfides have a low solubility and mobility [8]. Marine sediments might also be an important source of MMA and DMA to the overlying water column. Several species of aerobic and anaerobic sediment bacteria are shown to bioaccumulate arsenate and arsenite and convert them to MMA and DMA.

The concentration of dissolved arsenic in surface waters is highly influenced by natural exchange between sediments and the overlying water column. Overall, arsenic has been categorized as a bio-intermediate trace element; it shows relative surface minimum, mid-depth

maximum, and minor deep-water enrichment [17]. Only a minor portion of the arsenic is incorporated into biological systems of the arsenic cycle and 20% of the dissolved arsenate pool enters the arsenic cycle annually [67].

28.6 Role of Marine Biological Systems in Arsenic Biotransformation

28.6.1 Phytoplankton

Phytoplankton, the most common primary producer in the marine food chain, take up arsenate from the surrounding sea water and reduces it to arsenite [79], which is readily oxidized to arsenate in oxic marine waters on excretion [59]. This constant reduction of arsenate is mainly responsible for the measurable quantities of arsenite in the water column in spite of the thermodynamic instability of arsenite. The marine environment rich in phytoplankton exhibits a high percentage of arsenite and methylated forms of arsenic such as MMA and DMA. Phytoplankton are primarily responsible for the production of reduced and methylated species of arsenic [67]. Within the phytoplankton cell, the concentration of inorganic arsenic and methylated arsenic species shows great variation. This variation greatly depends on phosphate availability, arsenic concentration in surrounding sea water, temperature, and seasonal variations, and also the phytoplankton species because of their different biotransformation efficiencies.

Marine phytoplankton not only takes up arsenate from the surrounding waters but also incorporates arsenate into an array of carbohydrate compounds and biosynthesizes organic forms of arsenicals [62]. The exact nature of this organically bound arsenic by phytoplankton remains unclear; a number of arsenolipids like arsenocholine, trimethylarsoniumlactate, and arsenobetaine have been isolated [67]. Most of the organoarsenicals in marine phytoplankton are arsenosugars, which are also the precursors of the metabolic pathway to arsenobetaine and arsenocholine [80,81]. In the biotransformation pathway, phytoplankton actively uptakes arsenate using the phosphate transport system due to its structural similarity with phosphate, leading to its toxicity in the cell [82]. The excessive arsenate has to be transported out, but due to its poor mobility, the cell first converts it into a more mobile form, i.e., arsenite. Therefore, phytoplankton reduces As(V) to As(III), which can be excreted from the cell or enter into oxidative methylation to produce pentavalent methylated arsenic species (MMA(V) and DMA(V)) via intermediate trivalent methylated arsenic species (MMA(III) and DMA(III)) [83].

A few researchers believe that the process of arsenic biotransformation to its different forms is due to the detoxification mechanism of phytoplankton [84]; however, this theory has been disputed because the trivalent methylated species (DMA(III) and MMA(III)) are more toxic than As(III) [12,85]. The order of toxicity (high to low) of various arsenic species to organisms is trivalent methyl arsenicals > arsenite > arsenate > pentavalent methyl arsenicals [12]. The trivalent methylated arsenicals are structurally different from their pentavalent counterpart and are also more reactive and carcinogenic [86]. It has been reported that As(III) and MMA(III) are MMA(III) creates a more stable binary complex in the enzyme active site than As(III). The

mode of action of DMA(III) toxicity is more complicated and still remains unresolved [89]. The methylated arsenicals produced by phytoplankton uptake appear to be chemically stable, at least for a few months [67].

28.6.2 Marine Bacteria

Microbe-mediated arsenic metabolism has played a major role in the ecology of arsenic, which affects the parameters of mobility and bioavailability, as well as toxicity with in the marine environment. Bacteria implicated in the processes of biotransformation readily metabolize arsenic and participate in various metabolic functions including detoxification, anaerobic respiration, methylation, and assimilation. Marine bacteria have the ability to transform inorganic arsenic species to various methylated and/or complex organic forms of arsenic [47], and also have the capacity to decompose complex organoarsenicals into inorganic arsenic [75]. To survive with arsenic compounds in their environment, microbes have developed several metabolic pathways mediated by their genetic makeup and their products [88]. The genes involved in arsenic biotransformation are *aox/aos* genes (arsenite oxidation, recently named aio genes), arr genes (anaerobic arsenate respiration), and ars genes (reduction and methylation). The schematic model of arsenic biotransformation in bacterial cells is summarized in Figure 28-3 [3]. Since arsenate and arsenite may act as analogues of phosphate and glycerol, respectively, they enter the microbial cells via phosphate (Pst/Pit) transporters and glyceroporin (Glp) membrane proteins, respectively [88]. Cellular arsenite uptake was also observed via hexose transporters [90] or glucose permease GLUT1 [91]. The intracellular arsenite serves as an inducer to bind the regulatory protein ArsR (encoded by *arsR* gene) leading to its conformational changes and removal of previously bound regulatory protein to the operator site



FIGURE 28–3 Arsenic biotransformation in prokaryotic cells. (A) Respiratory arsenate reductase (Arr) is involved in the reduction of As(V). (B) Arsenite oxidase (Aso/Aox) is responsible for oxidation of As(III). (C) S-adenosylmethyltransferase (ArsM) is responsible for methylation of As(III) to produce methylated arsenicals as the end product. (D) Genes (*arsRDABC*) and proteins involved in the uptake of As(III) and As(V) (GIpF and Pst/Pit transporter), reduction of As(V) (ArsC), extrusion of As(III) (ArsAB), regulation (ArsRD) by arsenic-resistant organisms. *Modified from* [4].

of the *ars* operon (Figure 28–3). The removal of the ArsR regulatory protein leads to the transcription of *ars* genes. The best characterized mechanism of arsenic detoxification in bacteria involves the reduction of arsenate to arsenite, mediated by arsenate reductase encoded by the *arsC* gene. Arsenite is further extruded by a membrane-associated efflux pump encoded by the *arsB* gene or sequestered in the intracellular compartments. Other genes like *arsD* and *arsA* (encoded for ATPase) are found along with *arsC* and *arsB* in the majority of prokaryotes [92]. The arsenic tolerance mechanism is conferred by the *ars* operon, which is located either on plasmid or chromosome in different bacterial species. Bacteria from the genera *Halomonas* and *Acinetobacter* isolated from Mandovi and Zuari estuarine water systems in Goan territory, India, have been shown to harbor *arsA*, *arsB*, and *arsC* genes on their plasmid [93]. Arsenite is formed by reduction of arsenate, either excreted from the cell or methylated to form various methylated arsenicals mediated by ArsM (S-adenosylmethyltransferase) encoded by the *arsM* gene, which are further pumped out of the cell by an unidentified transporter.

Thioarsenicals, structural analogues of oxyarsenicals in which sulfur replaces oxygen, are formed by exposure of oxyarsenicals to hydrogen sulfide [94]. Arsenic species reported to form thio species include MMA, DMA, arsenosugars, dimethylarsinoyl ethanol, and dimethylarsinoyl acetate. These species are thought to be formed in the presence of hydrogen sulfide during anaerobic degradation of seaweed [65]. Studies have shown that anaerobic microflora from human feces or mouse cecum and gastrointestinal tracts of marine organisms may convert DMA(V) into thiolated arsenic compounds such as dimethylthioarsenate and trimethylarsine sulfide [95–97]. Marine sediments and hydrothermal waters are rich in sulfur, where the reducing conditions and relatively high pH may favor the formation of thiomethylated arsenicals by anaerobic sedimentary bacteria. So far none of the bacterial species in the marine environment have been identified for their ability to synthesize thiomethylated arsenicals, either from inorganic arsenic or from methylated forms.

Oxidation of arsenite to arsenate mediated by arsenite oxidase, encoded by *aox/aos/aoi* genes, was observed in bacteria and archaeans as the detoxification mechanism. The toxicity of arsenic depends on its oxidation state, while arsenite is 100 times more toxic than arsenate in most biological systems [98]. *Acinetobacter junii* SeaH-As6s and *Marinobacter* sp. SeaH-As6w, isolated from coastal sediment and sea water, respectively, from Gwangyang Bay, Republic of Korea, were found to oxidize arsenite to arsenate.

Apart from the operon-mediated detoxification mechanism, *Bacillus* sp. strain XZM002 has shown an additional arsenic tolerance mechanism by changing its shape to reduce its surface area to survive in the presence of high arsenic concentrations in its environment. The initial rod shape was altered to oval and then to circular form with gradual increase in extracellular arsenic concentration. The circular form will have the least surface area for arsenic uptake and thus will exhibit less toxicity [99].

Marine bacteria have the main role in the biogeochemical arsenic cycle of decomposing complex organoarsenicals into their simple organic forms or even into inorganic arsenic forms [75,100]. Two bacterial strains of the group *Vibrio/Aeromonas* were isolated from coastal sediments, and efficiently decomposed arsenobetaine in aerobic conditions to dimethylarsinic acid. The addition of sediment itself in the media, as the source of arsenobetaine-decomposing microorganisms, has shown the decomposition pattern of arsenobetaine to trimethylarsine to inorganic arsenic [75]. The oxidation and demethylation of methylated arsenicals by bacteria has been studied in marine waters [101].

Arsenic mobility in the environment was significantly enhanced by bacterial cell-mediated arsenate reduction. Two different pathways for arsenate reduction were observed in microorganisms encoded by *ars* and *arr* systems. The arsenate reductase encoded by *ars* genes encodes cytoplasmic arsenate reductase associated with a detoxification mechanism, whereas membrane/periplasmic arsenate reductase encoded by *arr* genes were associated with cellular respiration [102–104].

28.6.3 Algae

Marine algae are an important primary producer in the marine food chain, and affect the biogeochemical cycle of arsenic in the marine ecosystem. Marine algae have the capacity to accumulate 1000 times more arsenic than its concentration in the surrounding sea water [105]. This may contribute to the trophic transfer of arsenic to higher levels of the marine food chain and thus may pose a threat to human health [79]. Much of the organoarsenicals present in the tissues of marine animals are derived directly or indirectly from consumption of marine algae [8]. Zooxanthellae (algae), symbiotically present in clam mantle tissues, accumulate arsenate from the sea water and convert it to organic forms such as arsenoribosides and arsenotaurine, and pass the arsenic to the host [106]. In marine algae, inorganic arsenic accounts for between 1 and 50% of total arsenic and the remainder is present in different organoarsenical forms.

Biotransformation of arsenic in algal cells is initiated with the uptake of inorganic arsenic forms. Usually, the cellular uptake of arsenate competes with phosphate but few marine algae show the phosphate-independent cellular uptake of arsenate, suggesting the existence of more than one arsenate uptake mechanism in algae [107]. Following uptake, algae reduce arsenate to arsenite, and then subsequent methylations lead to the formation of methylated forms such as MMA, DMA, TMA, and various arsenoribosides. This biotransformation and rapid excretion of methylated arsenic MMA and DMA suggest methylation as the potent arsenic detoxification mechanism present in algae [34]. Methylation of arsenic involves an enzyme-mediated transfer of the methyl group via methylcobalamine or S-adenosyl methionine. The reduction of DMA with adenosyltmethionine produces dimethylarsinyladenosine, which may lead to further glycosidation to form dimethylarsinoribosides. Reductive methylation of this intermediate might lead to the formation of trimethylarsonioribosides [8].

Marine algae are found to be the principal producer of arsenosugars. Nearly 15 different arsenosugars were identified in marine algae containing pentavalent arsenic bound to two methyl groups [108]. The different classes of algal species have shown different forms of arsenoribosides. The brown macroalgae have shown the accumulation of more sulfonate (SO_3 -ribose) and sulfate (OSO_3 -ribose) arsenoribosides, whereas glycerol (OH-ribose) and phosphate (PO_4 -ribose) arsenoribosides were dominant in red and green macroalgae [109]. The brown alga (*Ecklonia radiate*) from unpolluted waters of Western Australia accumulates inorganic arsenate from sea water and converts it to a variety of arsenosugars [110].

Arsenate accumulation capacities of seaweeds (*Fucus spiralis* and *Ascophyllum nodosum*) were four times higher than for arsenite accumulation [111]. This suggested more toxic arsenite is excreted rapidly after the reduction mediated by membrane-associated transporter proteins. Recently, the role of metalloproteins (phytochelatins) in the detoxification of arsenic has been reported in marine microalgae (*Phaeodactylum tricornutum*) [112]. Phytochelatins (PCs), intracellular cysteine-rich metal-binding thiols containing polypeptide, are key metalloproteins involved in heavy metal homeostasis and detoxification in the cells of microorganisms. The synthesized PCs build a complex with arsenite (As(III)-PC) and are sequestered into vacuoles through the activity of ATP binding cassette (ABC) transporters, being finally excreted from the cells [113].

28.6.4 Marine Animals

Most marine animals have limited ability to accumulate arsenate from surrounding sea water. Despite this limited ability, concentrations of total arsenic vary widely in tissues of marine animals [8]. Total arsenic in whole and muscle tissues of marine organisms worldwide ranges from below the detection limit to more than $2500 \,\mu$ g/g dry weight [8]. Natural concentrations of total arsenic in the tissues of marine animals, other than marine mammals, often are less than 10 ppm. The geometric mean As concentration ranges from $0.25 \,\mu$ g/g dry weight in marine animal tissues [8]. The highest concentrations of arsenic have been reported in tissues of marine animals that feed primarily on phytoplankton or macroalgae, The taxa of which include planktonic crustaceans, bivalve mollusks, herbivorous snails, and some polychaeta worms [8]. Oysters (*Crassostrea virginica*) from coastal waters of the US Gulf of Mexico contain 4.1 to $39 \,\mu$ g/g of total arsenic [114]. Marine organisms from arsenic-contaminated environments generally contain higher concentrations of arsenic than do organisms from uncontaminated sites [8]. Arsenic found in marine animals in inorganic or organic forms (monomethyl arsenic, dimethyl arsenic, arsenosugars, and arsenocholine) account for only a minor fraction of total arsenic.

The main arsenic compound found virtually in all marine animals is arsenobetaine [115], which is a relatively stable quaternary arsenium compound and considered non-toxic. Arsenobetaine is the most common form of arsenic observed in the majority of marine fishes and is also known as "fish arsenic" [79]. It is rapidly excreted without bio transformation in marine animals. The concentration of arsenobetaine in marine animals usually represents 50 to 95% of the total arsenic in tissues of marine crustaceans, polychaeta, elasmobranchs and teleost fish and 10 to more than 95% of the total arsenic in sponges, coelenterates, mollusks, and echinoderms [8,63].

The tetramethylarsonium ion is also commonly found in marine animals, particularly in bivalve mollusks. High percentages of tetramethylarsonium ion were observed in gut (16.2%) and tissue (23.5%) of *Cyclope neritea* (gastropod) inhabiting arsenic-rich marine shallow water hydrothermal system off Milos island, Greece [63]. Other forms of arsenic such as trimethylarsine oxide and arsenocholine were also noted in marine animals, generally as minor arsenic constituents. The occurrence and distribution of arsenic species in a particular animal depends on its habitat, food preference, physiology, and arsenic species exposure [63,64].

Arsenosugars were found in a few herbivorous marine animals where the source was almost certainly the marine algae as their feed [115].

The origin and biosynthesis of arsenobetaine are controversial. Basically, four pathways have been proposed for arsenobetaine biosynthesis. In the first two pathways, dimethylated arsenosugars are converted into arsenobetaine through dimethylarsinoyl ethanol and dimethylarsinoyl acetate or arsenocholine [10]. In several marine animals and marine algae, dimethylarsinoyl ethanol and dimethylarsinoyl acetate were observed but at low levels. The existence of these two forms supports the concept that a pathway exists from dimethylated arsenosugars to arsenobetaine. In the third pathway, arsenobetaine is converted from trimethylated sugar [10]. Despite the natural occurrence of trimethylated arsenosugars in marine organisms, their very low concentrations do not account for the presence of arsenobetaine at a high concentration in marine animals [10,57]. However, a comparatively high concentration of trimethylated arsenosugar might contribute to the synthesis of arsenobetaine in this marine animal. In the fourth pathway, it is postulated that arsenobetaine is synthesized from dimethylarsinous acid and 2-oxo acids, glycoxylate, and pyruvate [10].

The high level of arsenobetaine in marine animals is related to the salinity of sea water. Organisms are known to utilize various osmolytes, low molecular weight osmotically active solutes, to adapt to osmotic stress [10]. Glycine betaine is the nitrous analogue of arsenobetaine [117] and behaves as an osmolyte in marine animals [118]. Once the arsenobetaine is synthesized in the marine food chain, it is taken up by the cells by using the uptake mechanism for glycine betaine [117]. Arsenobetaine was not found in Corbicula japonica (bivalve), which lives in the low salinity estuary [119]. It is suggested that arsenobetaine was not required as an osmolyte in a low salinity environment. It has been reported that arsenobetaine levels increased in blue mussels that had been maintained at high salinity [120]. The arsenobetaine levels decreased in the gill but not in other tissues when the blue mussels that had been maintained at high salinity were transferred to low salinity sea water. This suggested that arsenobetaine behaved as an osmolyte in the blue mussel and that the gill responded sooner to osmotic changes than did other tissues [10]. Although controversial, the study of retention capacity of arsenobetaine in seawater- and freshwater-adapted Atlantic salmon found no significant difference between seawater and freshwater groups, whereas the arsenobetaine level in the muscle of seawater-adapted wild salmon was found to be 10-fold higher than that of a freshwater-adapted wild salmon [121]. This study has suggested that high arsenobetaine levels in seawater-adapted wild salmon might be due to the arsenobetaine level in the diet of wild salmon rather than an adaptation to salinity [10]. Bioaccumulation of arsenobetaine in marine animals is still controversial.

28.7 Arsenic in Seafood and Its Toxicity

Arsenic has been documented as a global toxin that affects human health and shown to promote several types of cancers such as skin, urinary bladder, liver, and lung, as well as several
non–cancer diseases such as diabetes mellitus, hypertension, and cardiovascular and cerebrovascular diseases [122]. Humans can be exposed principally to arsenic by drinking contaminated water and food, especially seafood [4], which generally contain higher concentrations of arsenic than do other food [5]. The concentration of arsenic in marine organisms ranges between 5 and 100 μ g/g dry mass, while terrestrial foods levels of arsenic range between 0.05 and 0.4 μ g/g dry mass [4]. Around 90% of dietary arsenic in the US diet comes from seafood and saltwater finfish [123]. A few countries like Japan, China, and Korea rely on food from marine sources as a major part of their diet [6,7].

Arsenic toxicity is strongly related to the chemical species in which it is present in seafood. Organic forms of arsenic contribute more seafood total arsenic than do inorganic forms, which contribute less than 1 to 4% of the total arsenic [11]. Arsenobetaine, the most widespread and abundant organoarsenic compound found in marine animals, accounts for more than 80% of the total arsenic content [5]. Other organic arsenicals found in seafood include methylated arsenic such as monomethylarsonic acid, dimethylarsinic acid, trimethyl arsine oxide, tetramethylarsonium ion, arsenosugars, and arsenocholine [11]. Inorganic arsenic is considered the most toxic and is classified as a human carcinogen by oral and inhalation routes [5], while methylated arsenicals are considered as moderately toxic. Other organoarsenicals like arsenobetaine, arsenocholine, and arsenosugars have shown no toxicity.

Following uptake of arsenic-containing seafood, arsenic is easily absorbed by the human gastrointestinal tract and transported to the blood. Absorption with regard to dose was more than 95% in cases of arsenobetaine and arsenocholine, while 50% in the case of arsenite [7]. In the human body, inorganic arsenic species is converted to methylated arsenic via sequential reduction (arsenate to arsenite) and oxidative methylation. The reduction is carried out by glutathione, cysteine, and dithiothretol and methyl groups provided by S-adenosyl methionine. The monomethyl, dimethyl, and trimethyl species thus formed are generally less reactive with tissue components and more readily excreted in urine than the original inorganic species [11]. Metabolism of arsenic species in the human body after the ingestion of the seafood *Anemonia sulcata* showed that around 95% of total arsenic in cooked anemones was bioaccessible and 85% of the amount ingested was eliminated from the body via urine in 90 hours. Arsenobetaine followed by dimethylarsinate was the major compound found in both cooked anemonia and human urine after 10 hours of ingestion of seafood, while dimethylarsinate was most abundant in human serum 30 minutes after ingestion, followed by arsenobetaine [7].

As discussed earlier, arsenobetaine represents the most predominant and rapidly excreted form of arsenic in seafood. With respect to potential toxicity, arsenobetaine is non-mutagenic, non-cytotoxic, and non-immunotoxic and has no transforming activity in mammalian cells [11]. Acute oral LD₅₀ of arsenobetaine in mice was higher than $10 \mu g/g$ [124]. Arsenocholine, a metabolic precursor of arsenobetaine in marine animals, was reported at only minor levels in seafood [11]. Although studies of its toxicity were limited it was regarded as essentially a non-toxic compound [125]. Like arsenobetaine it is non-toxic and non-mutagenic. The acute oral LD₅₀ in mice was 6.5 g/kg, whereas the acute intravenous LD₅₀ was 187 µg/g [124]. Recent attention has focused on the oxidation state of methylated arsenicals excreted in the urine of seafood eaters. Trivalent methylated arsenic compounds such as methylarsonous acid and dimethylarsinous acid have greater toxicity than the corresponding pentavalent methylated arsenic compounds and also inorganic arsenic [11,12]. It was found that arsenite and methylarsonous acid act through the same mechanism to produce toxicity. However, methylarsonous acid created a more stable binary complex in the enzyme-active site than did arsenite. The mode of action of dimethylarsinous acid toxicity is more complicated and mostly unresolved [113]. However, trivalent methyl arsenic has not been documented in urine of seafood eaters until now. The reason for this failure may include low concentrations in urine, instrument detection limitations, or the unstable nature of trivalent methylated arsenic forms [11].

Trimethyl arsine oxide (TMAO) has been considered non-toxic and identified in some species of marine animals as a minor arsenic species, although rarely detected except at trace levels [11,34]. Levels were higher in stored, frozen fish than in fresh fish, probably secondary to postmortem breakdown, but dietary intake of TMAO was probably extremely low [11,126]. In *in vitro* studies, TMAO caused no cell growth inhibition, did not induce sister chromatid exchange, and caused only rare clastogenic effects at concentrations up to 10 mg/cm³ [11].

Tetramethyl arsonium ion is normally a minor species in marine animals such as finfish and fish, but might be a major species in some mollusks [11,34]. It was seen that levels of TETRA may be enhanced by freezing or dry cooking (grilling, roasting, and baking) at temperatures >160°C, especially in charred meat, likely due to the thermal decarboxylation of arsenobetaine [11]. Due to the limited quantity of TETRA in consumed fish, acute poisoning by TETRA was unlikely [11]. *In vitro* studies showed that high concentrations of TETRA inhibit cell growth [127]. In marine algae, a total of 15 different arsenosugars have been identified; almost all consist of pentavalent arsenic bound to two methyl groups [108]. Of these, there are four principal arsenosugars primarily found in marine algae. Their principal dietary source is seaweed, a common component of the Chinese diet [6]. Other sources include oyster, mussels, crab, fish, and clam [6]. In mammalian cells, arsenosugar was not cytotoxic at micromolar levels [11].

However, the above-mentioned arsenic compounds were not distributed uniformly throughout the tissues of marine animals. For example, in finfish, inorganic compounds and DMA were found almost exclusively in the viscera, while the arsenic content of muscles was almost all arsenobetaine [54]. Differences in tissue distribution between muscles and viscera have been documented in a variety of fish, carnivorous gastropods, and crustaceans [63,71]. It has been suggested that the degree of arsenic toxicity is also dependent on the particular type of ingested tissue. The risk of ingested seafood arsenic therefore depends on the nature and quantity of arsenic species present in the particular food and the metabolites formed as a consequence of expected metabolic activities.

Based on the consideration of anticipated dose and anticipated metabolism, it is likely that seafood arsenic does not contribute significantly to arsenic-associated carcinogenicity [11]. The most abundant form of arsenic generally reported in marine animals was arsenobetaine, a compound that is non-toxic and excreted without transformation. The amounts of TETRA produced by dry cooking of arsenobetaine-containing fish were not likely to achieve toxic levels. Similarly, the levels of inorganic arsenic and methylated arsenic found in seafood are sufficiently low to mitigate concerns about their possible toxic effect in seafood. On the other hand, DMA and arsenosugars in seafood have posed at least theoretical risks [11]. In general, the absence of arsenic

toxicity reported in humans and other mammals after consumption of large amounts of seaweed and seafood [108] has supported the lack of acute toxicity. However, because the ingestion of seafood might lead to the generation of metabolites involved in arsenic-induced carcinogenesis, it is worthwhile considering the potential role of dietary seafood in long-term cancer risk.

28.8 Future Directions

Arsenic concentrations found in unpolluted marine environments do not represent a significant hazard to marine organisms and their consumers. However, the coastal areas affected by industrial and mining inputs have elevated levels of arsenic, which are much higher than the USEPA human health (fish consumption) water quality criteria for total arsenic in sea water [8]. Therefore, removal of arsenic from industrial effluents has become a necessity to keep the marine ecosystem safe. The strategy generally applied to minimize such threats is to immobilize bioavailable arsenic as the metals cannot be degraded like organic pollutants [128]. Physicochemical technologies are sometimes not suitable for larger volumes and have some drawbacks such as high cost and generation of secondary contaminants. Recently, low-cost environment-friendly technologies have stimulated interest in studies in the bioremediation of metal [129]. B-acteria play a significant role in bioremediation [128–130]. More than 50 arsenic species have been identified in marine organisms [4], and it is expected that more will be identified as analytical techniques advance. The methylation processes for inorganic arsenic like MMA(III) and DMA(III) have recently been reported to be highly toxic, which are to be investigated in relation to toxicity in higher animals including humans. Future legislation on permissible levels of arsenic in seafood must include data on the types of arsenic species present and their toxicological properties.

References

- Mudhoo A, Sharma SK, Garg VK, Tseng CH. Arsenic: an overview of applications, health, and environmental concerns and removal processes. Crit Rev Environ Sci Technol 2011;41:435–519.
- [2] De Gieter M, Leermakers M, Van Ryssen R, Noyen J, Goeyens L, Baeyens W. Total and toxic arsenic levels in North Sea fish. Arch Environ Contam Toxicol 2002;43:406–17.
- [3] Kalia K, Joshi DN. Detoxification of arsenic. In: Gupta RC, editor. Handbook of toxicology of chemical warfare agents. USA: Academic Press; 2009. p. 1083–100.
- [4] Francesconi KA. Arsenic species in seafood: origin and human health implications. Pure Appl Chem 2010;82(2):373–81.
- [5] Lorenzana RM, Yeow AY, Colman JT, Chappell LL, Choudhury H. Arsenic in seafood: speciation issues for human health risk assessment. Hum Ecol Risk Assess 2009;15:185–200.
- [6] Li W, Wei C, Zhang C, Hulle MV, Cornelis R, Zhang X. A survey of arsenic species in Chinese seafood. Food Chem Toxicol 2003;41:1103–10.
- [7] Contreras-Acuña M, García-Barrera T, García-Sevillano MA, Gómez-Ariza JL. Arsenic metabolites in human serum and urine after seafood (*Anemonia sulcata*) consumption and bioaccessibility assessment using liquid chromatography coupled to inorganic and organic mass spectrometry. Microchem J 2014;112:56–64.
- [8] Neff JM. Review—ecotoxicology of arsenic in the marine environment. Environ Toxicol Chem 1997;16(5):917-27.

- [9] Milton AH, Rahman M. Respiratory effects and arsenic contaminated well water in Bangladesh. Int J Environ Health Res 2002;12:175-9.
- [10] Kunito T, Kubota R, Fujihara J, Agusa T, Tanabe S. Arsenic in marine mammals, seabirds, and sea turtles. Rev Environ Contam Toxicol 2008;195:31–69.
- Borak J, Hosgood HD. Seafood arsenic: implications for human risk assessment. Regul Toxicol Pharmacol 2007;47:204–12.
- [12] Mass MJ, Tennant A, Roop BC, Cullen WR, Styblo M, Thomas DJ, et al. Methylated trivalent arsenic species are genotoxic. Chem Res Toxicol 2001;14:355–61.
- [13] Mandal BK, Suzuki KT. Arsenic round the world: a review. Talanta 2002;58:201–35.
- [14] Maher W, Bulter E. Arsenic in the marine environment. Appl Organomet Chem 1988;2:191-214.
- [15] Ren JL, Zhang J, Li DD, Cheng Y, Liu SM. Behavior of dissolved inorganic arsenic in the Yellow Sea and East China Sea. Deep Sea Res Part II 2010;57:1035–46.
- [16] Breuer C, Pichler T. Arsenic in marine hydrothermal fluids. Chem Geol 2013;348:2-14.
- [17] Anninou P, Cave RR. How conservative is arsenic in coastal marine environments? A study in Irish coastal waters. Estuarine Coastal Shelf Sci 2009;82:515–24.
- [18] Santosa SJ, Mokudai H, Takahashi M, Tanaka S. The distribution of arsenic compounds in the ocean: biological activity in the surface zone and removal processes in the deep zone. Appl Organomet Chem 1996;10:697–705.
- [19] Santosa SJ, Wada S, Tanaka S. Distribution and cycle of arsenic compounds in the ocean. Appl Organomet Chem 1994;8:273–83.
- [20] Cutter GA, Cutter LS. Metalloids in the high latitude North Atlantic Ocean: sources and internal cycling. Mar Chem 1998;61:25–36.
- [21] Cutter GA, Cutter LS. Behavior of dissolved antimony, arsenic, and selenium in the Atlantic Ocean. Mar Chem 1995;49:295–306.
- [22] Cutter GA, Cutter LS, Featherstone AM, Lohrenz SE. Antimony and arsenic biogeochemistry in the western Atlantic Ocean. Deep Sea Res Part II 2001;48:2895–915.
- [23] Featherstone AM, Butler ECV, Grady BVO. Meridional distribution of arsenic species in the subantarctic zone of the Southern Ocean, south of Australia. J Geophys Res 2001;106(C12):31657–31667.
- [24] Yusof AM, Ikhsan ZB, Wood AKH. The speciation of arsenic in seawater and marine species. J Radioanal Nucl Chem 1994;179:277–83.
- [25] Navarro M, Sanchez M, Lopez H, Lopez MC. Arsenic contamination levels in waters, soils, and sludges in southeast Spain. Bull Environ Contam Toxicol 1993;50:356–62.
- [26] Ishikawa M, Okoshi K, Kurosawa M, Kitao K. Trace element analysis of seawater by PIXE. Proceedings of the 12th International Symposium on Application of Ion Beams in Material Science. Tokyo, Japan: Hosei University Press. Hosei University; 1987. p. 445–456.
- [27] Chang J-S, Lee J-H, Kim IS. Bacterial aox genotype from arsenic contaminated mine to adjacent coastal sediment: evidences for potential biogeochemical arsenic oxidation. J Hazard Mater 2011;193:233–42.
- [28] Maher WA. Arsenic in coastal waters of South Australia. Water Res 1985;19(7):933-4.
- [29] Seyler P, Martin JM. Distribution of arsenite and total dissolved arsenic in major French estuaries: dependence on biogeochemical processes and anthropogenic inputs. Mar Chem 1990;29:277–94.
- [30] Andreae MO, Andreae TW. Dissolved arsenic species in the Schelde estuary and watershed, Belgium. Estuarine Coastal Shelf Sci 1989;29:421–33.
- [31] Seyler P, Martin JM. Arsenic and selenium in a pristine river-estuarine system: the Krka (Yugoslavia). Mar Chem 1991;34:137-51.

- [32] Smedley PL, Kinniburgh DG. A review of the source, behaviour and distribution of arsenic in natural waters. Appl Geochem 2002;17:517–68.
- [33] Howard AG, Apte SC, Comber SDW, Morris RJ. Biogeochemical control of the summer distribution and speciation of arsenic in the Tamar estuary. Estuarine Coastal Shelf Sci 1988;27:427–43.
- [34] Cullen WR, Reimer KJ. Arsenic speciation in the environment. Chem Rev 1989;89:713-64.
- [35] Byrd JT. The seasonal cycle of arsenic in estuarine and nearshore waters of the South Atlantic bight. Mar Chem 1988;25:383–94.
- [36] Andreae MO. Arsenic speciation in seawater and interstitial waters: the influence of biological-chemical interactions on the chemistry of a trace element. Limnol Oceanogr 1979;24:440–52.
- [37] Batley GE. Heavy metals and tributyltin in Australian coastal and estuarine waters: state of the marine environment report for Australia. Technical annex 2. Canberra. Australia: Department of the Environment, Sports and Territories; 1996. pp. 63–72.
- [38] Edmonds JS, Shibata Y, Francesconi KA, Rippingale RJ, Morita M. Arsenic transformations in short marine food chains studied by HPLC-ICP MS. Appl Organomet Chem 1997;11:281–7.
- [39] Whalley C, Rowlatt S, Bennett M, Lovell D. Total arsenic in sediments from the western North Sea and the Humber estuary. Mar Pollut Bull 1999;38(5):394–400.
- [40] Duan L, Song J, Yuan H, Li X, Li N. Spatio-temporal distribution and environmental risk of arsenic in sediments of the East China Sea. Chem Geol 2013;340:21–31.
- [41] Mamindy-Pajany Y, Hurel C, Géret F, Galgani F, Battaglia-Brunet F, Marmier N, et al. Arsenic in marine sediments from French Mediterranean ports: geochemical partitioning, bioavailability and ecotoxicology. Chemosphere 2013;90:2730–6.
- [42] Garnaga G, Wyse E, Azemard S, Stankevičius A, De Mora S. Arsenic in sediments from the southeastern Baltic Sea. Environ Pollut 2006;144:855–61.
- [43] Paka V, Spiridonov M. An overview of the research of dumped chemical weapons made by the R/V "Professor Shtokman" in the Gotland, Bornholm and Skagerrak dump sites during 1997–2001. HELCOM MONAS 4/2002, Document 3/5/NF 2002.
- [44] Tørnes JA, Voie ØA, Ljønes M, Opstad AM, Bjerkeseth LH, Hussain F. Investigation and risk assessment of ships loaded with chemical ammunition scuttled in Skagerrak. TA-1907/2002. 2002.
- [45] Davies PJ. Arsenic in sediments on the continental shelf of south-east Australia. Search 1974;5:394-9.
- [46] Riedel GF, Sanders JG, Osman RW. Biogeochemical control on the flux of trace elements from estuarine sediments: effects of seasonal and short-term hypoxia. Mar Environ Res 1999;47:349–72.
- [47] Vidal FV, Vidal VMV. Arsenic metabolism in marine bacteria and yeast. Mar Biol 1980;60:1-7.
- [48] Maeda S, Ohki A, Miyahara K, Takeshita T, Higashi S. Growth characteristics and arsenic metabolism of two species of arsenic-tolerant bacteria. Appl Organomet Chem 1990;4:245–50.
- [49] Hanaoka K, Hasegawa S, Kawabe N, Tagawa S, Kaise T. Aerobic and anaerobic degradation of several arsenicals by sedimentary micro-organisms. Appl Organomet Chem 1990;4:239–43.
- [50] Eisler R. A review of arsenic hazards to plants and animals with emphasis on fishery and wildlife resources. In: Nriagu JO, editor. Arsenic in the environment. New York: Wiley; 1994. p. 185–260.
- [51] Inskeep WP, McDermott TR, Fendorf SE. Arsenic (V)/(III) cycling in soils and natural waters: chemical and microbiological processes Frankenberger Jr. WF, Macy JM, editors. Environmental chemistry of arsenic. NY: Marcell Dekker; 2002. p. 183–216.
- [52] Knauer K, Behra R, Hemond H. Toxicity of inorganic and methylated arsenic to algal communities from lakes along an arsenic contamination gradient. Aquat Toxicol 1999;46:221–30.
- [53] Geiszinger A, Goessler W, Kosmus W. Organoarsenic compounds in plants and soil on top of an ore vein. Appl Organometal Chem 2002;16:245–9.

- [54] Kirby J, Maher W, Charriton A, Krikowa F. Arsenic concentrations and speciation in a temperate mangrove ecosystem, NSW, Australia. Appl Organometal Chem 2002;16:192–201.
- [55] Islam SM, Fukushi K, Yamamoto K. Development of an enumeration method for arsenic methylating bacteria from mixed culture samples. Biotechnol Lett 2005;27:1885–90.
- [56] Millward E, Ebdon L, Walton AP. Seasonality in estuarine sources of methylated arsenic. Appl Chem 1993;7:499–511.
- [57] Edmonds JS, Francesconi KA. Organoarsenic compounds in the marine environment. In: Craig PJ, editor. Organometallic compounds in the environment. West Sussex, England: Wiley; 2003. p. 195–222.
- [58] Edmonds JS, Francesconi KA. The origin and chemical form of arsenic in the School Whiting. Mar Pollut Bull 1981;12:92–6.
- [59] Francesconi KA, Edmonds JS. Arsenic and marine organisms. Adv Inorg Chem 1996;44:147-89.
- [60] McSheehy S, Szpunar J, Lobinski R, Haldys V, Tortajada J, Edmonds JS. Characterization of arsenic species in kidney of the clam *Tridacna derasa* by multidimensional liquid chromatography-ICPMS and electrospray time-of-flight tandem mass spectrometry. Anal Chem 2002;74:2370–8.
- [61] Francesconi KA. Current perspectives in arsenic environmental and biological research. Environ Chem 2005;2:141–5.
- [62] Francesconi KA, Edmonds JS. Arsenic in the sea. Oceanogr Mar Biol Ann Rev 1993;31:111-51.
- [63] Ruiz-Chancho MJ, Pichler T, Price RE. Arsenic occurrence and speciation in *Cyclope neritea*, a gastropod inhabiting the arsenic-rich marine shallow-water hydrothermal system off Milos Island, Greece. Chem Geol 2013;348:56–64.
- [64] Waring J, Maher W. Arsenic bioaccumulation and species in marine polychaeta. Appl Organomet Chem 2005;19:917–29.
- [65] Maher WA, Foster S, Krikowa F, Duncan E, St John A, Hug K, et al. Thio arsenic species measurements in marine organisms and geothermal waters. Microchem J 2013;111:82–90.
- [66] Hasegawa H. Seasonal changes in methylarsenic distribution in Tosa Bay and Uranouchi inlet. Appl Organomet Chem 1996;10:733-40.
- [67] Sanders JG. Arsenic cycling in marine systems. Mar Env Res 1980;3:257-66.
- [68] Cullen WR, Li H, Hewitt G, Reimer KJ. Identification of extracellular arsenical metabolites in the growth medium of the microorganisms *Apiotrichum humicola* and *Scopulariopsis brevicaulis*. Appl Organomet Chem 1994;8:303–11.
- [69] Cullen WR, Harrison LG, Li H, Hewitt G. Bioaccumulation and excretion of arsenic compounds by a marine unicellular alga, *Polyphysa peniculus*. Appl Organomet Chem 1994;8:313–24.
- [70] Hanaoka K, Tagawa S, Kaiset T. The fate of organoarsenic compounds in marine ecosystems. Appl Organomet Chem 1992;6:139–46.
- [71] Kaise T, Hanaoka K, Tagawa S, Hirayamas T. Distribution of inorganic arsenic and methylated arsenic in marine organisms. Appl Organomet Chem 1988;2:539–46.
- [72] Hanaoka K, Kaise T, Kai N, Kawasaki Y. Arsenobetaine-decomposing ability of marine microorganisms occurring in particles collected at depths of 1100 and 3500 meters. Appl Organomet Chem 1997;11:265–71.
- [73] Andreae MO. Distribution and speciation of arsenic in natural waters and some marine algae. Deep Sea Res 1978;25:391–402.
- [74] Bright DA, Brock S, Reimer KJ, Cullen WR, Hewitt GM, Jafaar J. Methylation of arsenic by anaerobic microbial consortia isolated from lake sediment. Appl Organomet Chem 1994;8(4):415–22.
- [75] Hanaoka K, Kaiset T. Conversion of arsenobetaine to dimethylarsinic acid by arsenobetaine-decomposing bacteria isolated from coastal sediment. Appl Organomet Chem 1991;5:435–8.

- [76] Hanaoka K, Yamamoto H, Kawashima K, Tagawa S, Kaise T. Ubiquity of arsenobetaine in marine animals and degradation of arsenobetaine by sedimentary micro-organisms. Appl Organomet Chem 1988;2:371–6.
- [77] Nakamura M, Matsuzono Y, Tanaka S, Hashimoto Y. Chemical form of arsenic compounds and distribution of their concentrations in the atmosphere. Appl Organomet Chem 1990;4:223–230.
- [78] Maher WA, Clarke SM. The occurrence of arsenic in marine macroalgae from two coastal areas of South Australia. Mar Pollut Bull 1984;15:111–12.
- [79] Rahman MA, Hasegawa H, Lim RP. Bioaccumulation, biotransformation and trophic transfer of arsenic in the aquatic food chain. Environ Res 2012;116:118–35.
- [80] Kaise T, Ogura M, Nozaki T, Saitoh K, Sakurai T, Matsubara C, et al. Biomethylation of arsenic in an arsenic-rich freshwater environment. Appl Organomet Chem 1997;11:297–304.
- [81] Hansen HR, Raab A, Francesconi KA, Feldmann J. Metabolism of arsenic by sheep chronically exposed to arsenosugars as a normal part of their diet. 1. Quantitative intake, uptake, and excretion. Environ Sci Technol 2003;37:845–51.
- [82] Hellweger FL, Lall U. Modeling the effect of algal dynamics on arsenic speciation in Lake Biwa. Environ Sci Technol 2004;38:6716–23.
- [83] Hughes MF. Arsenic toxicity and potential mechanisms of action. Toxicol Lett 2002;133:1–16.
- [84] Murray LA, Raab A, Marr IL, Feldmann J. Biotransformation of arsenate to arsenosugars by Chlorella vulgaris. Appl Organomet Chem 2003;17:669–74.
- [85] Petrick JS, Ayala-Fierro F, Cullen WR, Carter DE, Aposhian VH. Monomethylarsonous acid (MMAIII) is more toxic than arsenite in Chang human hepatocytes. Toxicol Appl Pharmacol 2000;163:203–7.
- [86] Jomova K, Jenisova Z, Feszterova M, Baros S, Liska J, Hudecova D, et al. Arsenic: toxicity, oxidative stress and human disease. J Appl Toxicol 2011;31:95–107.
- [87] Yin XX, Chen J, Qin J, Sun GX, Rosen BP, Zhu YG. Biotransformation and volatilization of arsenic by three photosynthetic cyanobacteria. Plant Physiol 2011;156:1631–8.
- [88] Mukhopadhyay R, Rosen BP, Phung LT, Silver S. Microbial arsenic: from geocycles to genes and enzymes. FEMS Microbiol Rev 2002;26:311–25.
- [89] Bergquist ER, Fischer RJ, Sugden KD, Martin BD. Inhibition by methylated organoarsenicals of the respiratory 2-oxo-acid dehydrogenases. J Organomet Chem 2009;694:973–80.
- [90] Liu Z, Boles E, Rosen BP. Arsenic trioxide uptake by hexose permeases in *Saccharomyces cerevisiae*. J Biol Chem 2004;279:17312–18.
- [91] Liu Z, Sanchez MA, Jiang X, Boles E, Landfear SM, Rosen BP. Mammalian glucose permease GLUT1 facilitates transport of arsenic trioxide and methylarsonous acid. Biochem Biophys Res Commun 2006;351:424–30.
- [92] Rosen BP. Biochemistry of arsenic detoxification. FEBS Lett 2002;529:86–92.
- [93] Sunita MSL, Prashant S, Chari PVB, Rao SN, Balaravi P, Kishor PBK. Molecular identification of arsenicresistant estuarine bacteria and characterization of their ars genotype. Ecotoxicology 2012;21:202–12.
- [94] Fricke MW, Zeller M, Sun H, Lai VW, Cullen WR, Shoemaker JA, et al. Chromatographic separation and identification of products from the reaction of dimethylarsinic acid with hydrogen sulfide. Chem Res Toxicol 2005;18:1821–9.
- [95] Kubachka KM, Kohan MJ, Herbin-Davis K, Creed JT, Thomas DJ. Exploring the in vitro formation of trimethylarsine sulfide from dimethylthioarsinic acid in anaerobic microflora of mouse cecum using HPLC-ICP-MS and HPLC-ESI-MS. Toxicol Appl Pharmacol 2009;239:137–43.
- [96] Deplancke B, Hristova KR, Oakley HA, McCraken VJ, Aminov R, Mackie RI, et al. Molecular ecological analysis of the succession and diversity of sulfate reducing bacteria in the mouse gastrointestinal tract. Appl Environ Microbiol 2000;66:2166–74.

- [97] Goubern M, Andriamihaja M, Nubel T, Blachier F, Bouillaud F. Sulfide the first inorganic substrate for human cells. FASEB J 2007;21:1699–706.
- [98] Cervantes C, Ji G, Ramirez JL, Silver S. Resistance to arsenic compounds in microorganisms. FEMS Microbiol Rev 1994;15:355–67.
- [99] Xie Z, Luo Y, Wang Y, Xie X, Su C. Arsenic resistance and bioaccumulation of an indigenous bacterium isolated from aquifer sediments of Datong Basin, northern China. Geomicrobiol J 2013;30:549–56.
- [100] Duncan EG, Maher WA, Foster SD, Mikac KM, Krikowa F, Florance A. Arsenoriboside degradation in marine systems: the use of bacteria culture incubation experiments as model systems. Chemosphere 2014;95:635–8.
- [101] Sanders JG. Microbial role in the demethylation and oxidation of methylated arsenicals in seawater. Chemosphere 1979;8:135–7.
- [102] Saltikov CW, Wildman Jr. RA, Newman DK. Expression dynamics of arsenic respiration and detoxification in *Shewanella* sp. strain ANA-3. J Bacteriol 2005;187:7390–6.
- [103] Afkar E, Lisak J, Saltikov C, Basu P, Oremland RS, Stolz JF. The respiratory arsenate reductase from *Bacillus selenitireducens* strain MLS10. FEMS Microbiol Lett 2003;226:107–12.
- [104] Krafft T, Macy JM. Purification and characterization of the respiratory arsenate reductase of *Chrysiogenes arsenatis*. Eur J Biochem 1998;255:647–53.
- [105] Sanders JG, Windom HL. The uptake and reduction of arsenic species by marine algae. Estuarine Coastal Mar Sci 1980;10:555–67.
- [106] Phillips DJH. The chemical forms of arsenic in aquatic organisms and their interrelationships. In: Nriagu JO, editor. Arsenic in the environment. Part I: Cycling and characterization. New York, USA: John Wiley & Sons; 1994. p. 263–88.
- [107] Andreae MO, Klumpp DW. Biosynthesis and release of organoarsenic compounds by marine algae. Environ Sci Technol 1979;13:738-41.
- [108] Edmonds JS, Francesconi KA. Arsenic in seafoods: human health aspects and regulations. Mar Pollut Bull 1993;26(12):665–74.
- [109] Thomson D, Maher W, Foster S. Arsenic and selected elements in inter-tidal and estuarine marine algae, south-east coast, NSW, Australia. Appl Organomet Chem 2007;21:396–411.
- [110] Edmonds JS, Francesconi KA. Arseno-sugars from brown kelp (*Ecklonia radiata*) as intermediates in cycling of arsenic in a marine ecosystem. Nature 1981;289:602–4.
- [111] Klumpp DW. Characteristics of arsenic accumulation by the seaweeds *Fucus spiralis* and *Ascophyllum nodosum*. Mar Biol 1980;58:257–64.
- [112] Morelli E, Mascherpa MC, Scarano G. Biosynthesis of phytochelatins and arsenic accumulation in the marine microalga *Phaeodactylum tricornutum* in response to arsenate exposure. Biometals 2005;18:587–93.
- [113] Rahman MA, Hassler C. Is arsenic biotransformation a detoxification mechanism for microorganisms? Aquat Toxicol 2014;146:212–19.
- [114] Wilson EA, Powell EN, Wade TL, Taylor RJ, Presley BJ, Brooks JM. Spatial and temporal distributions of contaminant body burden and disease in Gulf of Mexico oyster populations: the role of local and largescale climatic controls. Helgol Wiss Merresunters 1992;46:201–35.
- [115] Francesconi KA, Edmonds JS. Arsenic species in marine samples. Croat Chem Acta 1998;71(2):343–59.
- [116] Kirby J, Maher W, Spooner D. Arsenic occurrence and species in near-shore macroalgae-feeding marine animals. Environ Sci Technol 2005;39:5999–6005.
- [117] Shibata Y, Morita M, Fuwa K. Selenium and arsenic in biology: their chemical forms and biological functions. Adv Biophys 1992;28:31–80.
- [118] Yancey PH, Clark ME, Hand SC, Bowlus RD, Somero GN. Living with water stress: evolution of osmolyte systems. Science 1982;217:1214–22.

- [119] Shibata Y, Morita M. Characterization of organic arsenic compounds in bivalves. Appl Organomet Chem 1992;6:343–9.
- [120] Clowes LA, Francesconi KA. Uptake and elimination of arsenobetaine by the mussel *Mytilus edulis* is related to salinity. Comp Biochem Physiol Part C 2004;137:35–42.
- [121] Amlund H, Berntssen MHG. Arsenobetaine in Atlantic salmon (Salmo salar L.): influence of seawater adaptation. Comp Biochem Physiol Part C 2004;138:507–14.
- [122] Tapio S, Grosche B. Arsenic in the aetiology of cancer. Mutat Res 2006;612:215–46.
- [123] Adams MA, Bolger PM, Gunderson EL. Dietary intake and hazards of arsenic. In: Chappell WR, Abernathy CO, Cothern CR, editors. Arsenic: exposure and health. Science and technology letters. Northwood, UK: 1994. p. 41–49.
- [124] Kaise TS, Watanabe S, Itoh K. The acute toxicity of arsenobetaine. Chemosphere 1985;14:1327-32.
- [125] Sakurai T. Biological effects of organic arsenic compound in seafood. Appl Organomet Chem 2002;16:401-5.
- [126] Yamauchi H, Fowler BA. Toxicity and metabolism of inorganic and methylated arsenicals. In: Nriagu JO, editor. Arsenic in the environment, Part II: Human health and ecosystem effects. New York: John Wiley; 1994. p. 35–53.
- [127] Kaise T, Ochi T, Oya-Ohta Y, Hanaoka K, Sakurai T, Saitoh T, et al. Cytotoxicological aspects of organic arsenic compounds contained in marine products using the mammalian cell culture technique. Appl Organomet Chem 1998;12:137–43.
- [128] Kruger MC, Bertin PN, Heipieper HJ, Arsène-Ploetze F. Bacterial metabolism of environmental arsenic mechanism and biotechnological application. Appl Microbiol Biotechnol 2013;97:3827–41.
- [129] Dash HR, Mangwani N, Chakraborty J, Kumari S, Das S. Marine bacteria: potential candidates for enhanced bioremediation. Appl Microbiol Biotechnol 2013;97:561–71.
- [130] Lièvremont D, Bertin PN, Lett MC. Arsenic in contaminated waters; biogeochemical cycle, microbial metabolism and biotreatment processes. Biochimie 2009;91:1229–37.

Index

Note: Page numbers followed by "b", "f" and "t" refer to boxes, figures and tables, respectively.

Α

Abnormal Ankle Brachial Systolic Pressure Index (ABSPI), 65 Absorbed arsenic, 203-204 Acetylcholine (ACh), 445 Acetylcholinesterase (AChE), 374, 376, 398 Acid mine drainage (AMD), 29-30 Acidithiobacillus ferrooxidans, 30 Activated alumina (AA), 122. See also Novel modified activated alumina Acute Exposure Guideline Levels (AEGLs), 110. 110t Acute mycocardial infarction (AMI), 409-411 Acute promyelocytic leukemia (APL), 474-475, 504 Acute respiratory infections (ARI), 510 Adenocarcinoma, 237, 548 Adenosine triphosphate (ATP), 13, 184, 213, 290, 636.649 Adsorbent, 38, 123-124 properties, 125-127 regeneration, 125 Adsorbent materials, 122, 129, 131-132 characterization of, 123 regeneration of, 125, 134 Adsorption of arsenic, 39, 685 arsenic removal via, 38-40 equilibrium, 130-132 Aerogel AA, 123, 127-128 Aeromonas hydrophila, 500-501 Agency for Toxic Substances and Disease Registry (ATSDR), 109-110, 233-234, 302-303 Air, arsenic in, 19-21, 536-537 Airway remodeling, 342 Algae, 640 and arsenic biotransformation, 689-690 Alkaline phosphatase (ALP), 376-377

All-trans retinoic acid (ATRA)-related toxicities, 504 Alpha-lipoic acid (ALA) antioxidant properties of, 609 role of, 218 α -smooth muscle actin (α -SMA), 327 Altered cellular signaling pathways, 353 Altered gene expression patterns. See Genomics Altered heme biosynthesis/porphyrinuria patterns, 356 Altered protein synthesis patterns. See Proteomics Alum and iron precipitations, 37 American Cancer Society (ACS), 233-234 δ -Aminolevulinic acid dehydratase (ALAD), 462-463, 610-611 Aminopyrine N-demethylase (ANDM), 240 Angiosarcoma, 546 Aniline p-hydroxylase (APH), 240 Animal foods, arsenic in, 282-283 Animals arsenic nephrotoxicity in nutritional products for prevention of, 431b marine, 690-691 neurobehavioral toxicity of arsenic exposure in, 367-368, 380t Anodic stripping voltammetry (ASV), 180-181 Anthropogenic sources of arsenic, 16-17, 53, 316-317 Antioxidants, preventing arsenic-induced toxic effects by, 214-220 herbal antioxidant, 219-220 melatonin, role of, 217-218 N-acetyl cysteine, role of, 217 preventing arsenic-induced toxic effects by, 214-220 silymarin and quercetin, role of, 218-219 taurine, role of, 215-217 α -lipoic acid, role of, 218

Antioxidants in preventing arsenic toxicity, 608-614 alpha-lipoic acid, 609 curcumin, 612 essential metals, 612-613 selenium, 613 zinc, 612 herbal extracts, 613-614 melatonin, 611-612 N-acetylcysteine, 609-610 taurine, 611 vitamins E and C, 610-611 β-carotene, 611 Antioxidative defense system, 153, 499-500, 654-655 Aox operon regulation, 638 Apiotrichum humicola, 684–685 Apoptosis, induction of, 396-397 Apoptotic signaling via transcription regulation arsenic and, 246-251 Aptamers, 581-583 Aqueous chemistry of arsenic, 14-16 Aqueous extract of C. olitorius leaves (AECO), 613-614 Arabidopsis, 643-644, 649-652 Arabidopsis thaliana, 630, 643, 645-646 Arbuscular mycorrhizae symbiosis, prospects of in regulation of plant defense response, 656-657 Argentina, natural and anthropogenic occurrence in, 35-36 Aristolochic acid, 431t Aristotle, 444 Arjunolic acid (AA), 219–220, 220f hepatoprotective role of, 614 Arrhythmias, 461 Ars operon, 577, 577f, 585 Ars-3, 582-583 Arsenalic acid, 533-534 Arsenate, 6, 8, 11, 16, 31, 351, 533-534, 629-630, 679 - 680in marine sediments, 679-681, 683-684 uptake, 643-644 Arsenate reductase (ACR), 635, 651-652, 689 Arsenate reduction, 6-7, 633-636, 651-652, 689 arsenate respiration/dissimilatory As(V) reduction, 635-636 mechanism of arsenic detoxification, 633-634

Arsenic, 1-2 Arsenic acid, 533-534 Arsenic adsorption equilibrium, 127-132 adsorbent amount as function of pH, 128-129 adsorption equilibrium, 130-132 effect of initial concentration on as removal efficiency, 129-130 solution pH as function of adsorption time, 127-128 Arsenic and apoptotic signaling via transcription regulation, 246-251 Arsenic and cardiovascular system. See Cardiovascular system Arsenic and respiratory disease. See Respiratory disease, arsenic and Arsenic and tumor/heat shock-associated proteins, 245-246 Arsenic bromidum, 302 Arsenic carcinogenicity, 210 Arsenic chemistry, 97 Arsenic cycle in marine environment, 683-686 Arsenic exposure, early life birth outcomes, and growth, 338 and immune development, 338-339 and lung health, 338-341 and non-malignant respiratory disease in adults, 340 and non-malignant respiratory disease in children, 339 Arsenic induced metabolic/immune toxicity, 289-291 Arsenic iodidum, 302 Arsenic metabolism, 205-206, 535, 590-591 and carcinogenesis, 236-237 in animal model, 235-251 role of signal transduction, 242 and drug metabolizing enzymes, 240-241 and free radical influences, 238-240 and gut microflora, 241 biomethylation, 205-206 Arsenic methylation, 437 Arsenic occurrence and exposure, 97-98 Arsenic pentoxide, 533-534 Arsenic pollution, 675-676 Arsenic toxicity, acute treatment of, 310

Arsenic trioxide, 2-3, 474-475, 504, 508, 533-534 Arsenical hyperkeratosis, 141 Arsenical keratoses, 303-304, 310 Arsenical keratosis, 170, 543 Arsenical kidney toxicity. See Kidney toxicity Arsenic-based therapies, 557-558 Arsenic-containing compounds, 533-535 Arsenic-induced developmental neurotoxicity. See Developmental neurotoxicity, arsenic-induced Arsenic-induced liver injury. See Liver injury, arsenic-induced Arsenic-induced posttranslational alterations of proteins, 356 Arsenicosis, 62 and cancer, 163-165 lung cancer, 164 skin cancer, 163-164 urinary bladder cancer, 164 case definition algorithm, 63f socio-cultural aspects of in Bangladesh, 67 Arsenicosis epidemiology in Bangladesh, 66 Arsenicosis patients management in Bangladesh, 66 - 67Arsenicum hydrogenisatum, 302 Arsenides, 8 Arsenite, 4-6, 8, 10-11, 352, 590, 681, 684 oxidation of, 688 reciprocal correlation with phosphate, 684 Arsenite and arsenate, 98-103 cancer effects, 99-100 cardiovascular effects, 100-101 developmental and reproduction effects, 102-103 diabetes, 100 immune effects, 102 kidney effects, 102 neurological effects, 101 pulmonary effects, 101-102 skin lesions, 98-99 Arsenite efflux pump, 577 Arsenite uptake, 644 Arsenobetaine (AsBe), 12, 503, 533-536, 684-685, 690 origin and biosynthesis of, 691 in seafood, 692

Arsenocholine, 282, 503, 692-693 Arsenolite, 10-11 Arsenolysis, 184 Arsenopyrite, 8, 10, 533-534 dissolution of, 30 release of arsenic from, 30-31 Arsenosugars, 630 Arsenosulfides, 8-10 Arsine, 103, 109, 351 Artificial neural networks (ANNs), 583 Artisanal mining, arsenic release by, 32 ArxAB-type arsenite oxidase, 638-639 Aryl hydrocarbon receptor (AhR), 240-241, 290-291 As(III), 649-650 binding to thiols, 650-651 Ascophyllum nodosum, 681-683, 690 Asiatic pills, 302 Astrocyte, 370-371 Astrocytomas, 548 Atherosclerosis, 461, 470 Atmosphere, arsenic in, 19-21 arsenic dissipation from fly ash, 20-21 arsenic emission due to burning of coal, 20 Atomic fluorescence spectrometry with hydride system (HGAFS), 180-181 Atomic structure and bonding, 3-4 ATP-binding cassette (ABC) transporters, 323 ATSDR, 109-110 Acute Exposure Guideline Levels (AEGLs), 110 Cal EPA, 110-111 inorganic arsenic, 109-110 organoarsenicals, 110 **RIVM National Institute for Public Health and** the Environment, 110 WHO, 110 Attention deficit hyperactivity disorder (ADHD), 367 Autoimmune-mediated diseases, 509

В

Bacillus sp. strain XZM002, 688 *Bacillus subtilis*, 577 Bacteria and arsenic biotransformation, 687–689 Bahia Concepcion, 26 Bangladesh arsenic contamination of drinking water, 628-629 arsenic contamination of tube-well water in, 52 arsenic in different environmental media of, 55 - 59food chain, 57 river, 55 soil, 55 birth weight in, 526 drinking arsenic-contaminated water in, 494, 502 effect of in utero exposure on infant immune system, 510 epidemiology of arsenicosis in, 66 health effects of arsenic toxicity in, 59-66 health effects of iAs exposure in, 405-409, 417-418 management of arsenicosis patients in, 66-67 natural and anthropogenic occurrence in, 33-34 risk for neonatal death, 525 socio-cultural aspects of arsenicosis in, 67 Bangladesh ground water, arsenic contamination in, 52-54 extent of, 54-55 reasons for leaching arsenic, 53-54 reasons for variation of arsenic concentrations between aquifers, 54 Basal cell carcinomas (BCC), 545 Batch adsorption experiments, 123-125 adsorbent regeneration, 125 breakthrough column experiments, 124-125 Bcl-2-associated death promoter (BAD), 248-249 Behavioral dysfunctions and neurochemical perturbations, 443 behavioral effects, 447-449 developmental toxicity, 444-449 nervous system, effects on, 445-447 neurite outgrowth, 447 neurochemical effects, 445-447 neurotransmitter systems, 445 oxidative stress, 446-447 toxicity, 444 Behavioral effects, 447-449 Bengal Basin, 53 β -carotene and arsenic toxicity prevention, 611 β-catenin, 239, 243-244, 261-262 Bioaccumulation factor, 657 Bio-availability of arsenic in soil, 19

Biogeochemical fate of arsenic in hydro- and geothermal systems, 29 Biologically based dose-response (BBDR) model, 105 Bio-methylation, 6-7, 205-206 Biomethylation of As, 639-641, 640f methylation pathway, 640-641 Biorecognition elements, 577-581, 585 Biosensors for arsenic, 576-581 ars operon, 577, 577f biorecognition elements, 585 future directions, 584f, 585 recombinant Escherichia coli, 577-581 Biosphere, arsenic in, 16-24 in atmosphere, 19-21 arsenic dissipation from fly ash, 20-21 arsenic emission due to burning of coal, 20 in dietary products, 23-24 arsenic in tobacco, 24 plants and crops, 23 rice and other food items, 23-24 in rocks and soils, 17-19 in various water resources, 21-23 ground water, 22-23 rain water, 21 river and lake water, 21-22 sea and estuarine water, 22 Biotransformation of arsenic, 235f, 686-691 in algae, 689-690 in marine animals, 690-691 in marine bacteria, 687-689 in phytoplankton, 686-687 in prokaryotic cells, 687f Birth outcomes, and growth and early life arsenic exposure, 338 Birth weight, arsenic and, 525-526 Bisphenol A (BPA), 413-414 Blackfoot disease (BFD), 153, 303, 321 Bladder cancer, 436 Bladder disease and arsenic, 436-438 Blood, arsenic effects on, 462-468 erythrocytes, 462-464 leukocytes, 464-467 methyltransferases, 467-468 Blood arsenic levels, 591-592 Blood leukocytes, arsenic's effects on, 496 Blood pressure, 461 Boerhavia diffusa, 480-481

Bone-marrow toxicity, 291 Bowen's disease, 61*f*, 66, 143, 143*f*, 543–544 Brain cancer, 548 Brain encephalopathy, arsenic-induced, 208 Breakthrough (BT) column analysis, 132–133 Breakthrough column experiments, 124–125 British Anti Lewisite (BAL), 450–451, 598, 599*t*, 608 Bronchiectasis, 340, 340*f* Bronchopulmonary disease, 148 Brown macroalgae, 689

С

Cadmium, 357, 431t Cal EPA, 110-111 Calcium overload, 461 Calcium oxide-modified AA (CaO-AA), 123, 127, 129 Calcium-activated chloride channel, 342 California, natural and anthropogenic occurrence in, 36 Cancer, arsenicosis and, 163-165 lung cancer, 164 skin cancer, 163-164 urinary bladder cancer, 164 Cancer slope factors (CSF), 108-109 Cancer stem cells (CSCs), 418-419 Cancers, arsenic-associated, 412 mortality risk, 411 Carcinogen arsenic a potent, 233-234 and mutagenic agent, 2 Carcinogenesis, 242 in animal model arsenic-associated metabolism and, 235-251 arsenic-induced arsenic cocarcinogenicity, 541-542 cell proliferation, induction of, 541 cellular immune function, arsenic dysregulation of, 540 epigenetic mechanisms, dysregulation of, 541 gene expression, aberration of, 540 genotoxicity, 540 interference with signal transduction, 542 keratin expression, arsenic perturbation of, 539 molecular mechanisms of, 538-543, 538f protein structure, distortion of, 541 reactive oxygen species, induction of, 542-543 arsenic metabolism and, 236-237

Carcinogenic potential of arsenic, 497-498 Carcinogenicity, arsenic-associated, 210 Cardiovascular disease (CVD), 153-155, 468-469 Cardiovascular dysfunction, 206-207 Cardiovascular system, 461 blood, arsenic effects on, 462-468 erythrocytes, 462-464 leukocytes, 464-467 methyltransferases, 467-468 heart, arsenic effects on, 474-481 cellular signaling, 478-481 chelation therapy, 481 ion channels, 476-478 ischemic heart disease, 475-476 QT prolongation, 474-475 human pluripotent stem cells, 481-482 vascular system, arsenic effects on, 468-474 atherosclerosis, 470 clinical studies, 468-469 endothelial cells, 472-473 genetic polymorphisms, 470-472 smooth muscle cells, 473-474 Ca-rich fly ash, 20-21 Cataract, 160-161 Ceasing drinking of arsenic-contaminated water, 168-169 Cell cycle, arsenic effect on, 397 Cell proliferation, arsenic induction of, 541 Cell-mediated immune response in laboratory animals, 501-502 Cellular immune function, arsenic dysregulation of, 540 Cellular respiration, arsenic and, 541 Cellular signaling, 478-481 Cellular transformation signaling and arsenic exposure, 242-244 Centella asiatica, 613-614 aqueous extract, 450-451 Central America, natural and anthropogenic occurrence in, 36-37 Central neurotransmitters/neuroendocrine, effect on, 397-398 Cerebellar granule neurons (CGNs), 372 Cerebrovascular disease (CVD), 157 Cetyl trimethylammonium bromide (CTAB), 582-583 Changing concept of arsenic toxicity, 179

Chelating agents, 589-590, 592, 594, 598-605 2, 3-dimercaptopropane-1-sulfonic acid, 604-605 dimercaprol (BAL), 598-601 D-penicillamine (D-PA), 601-602 meso-2, 3-dimercaptosuccinic acid (Succimer, DMSA), 602-604 Chelation of arsenic, 633 Chelation therapy, 481, 590, 592-598 chemical considerations, 594-596 in vivo efficacy of chelating agents, 595-596 kinetic considerations in metal chelation, 595 thermodynamics of metal chelation, 594-595 chemistry, 593-594 for chronic arsenic toxicity, 169 concept, 592 future directions, 618 strategies of, 617f toxicokinetics, 596-597 Chemical weapons, organoarsenicals in, 13 Chemistry of arsenic, 2-8 arsenic methylation, 6-7 arsenic oxidation and reduction, 5-6 atomic structure and bonding, 3-4 historical and modern applications of arsenic, 7 - 8origin and history, 2-3 within mining wastes, 31-32 Chicken meat, arsenic in, 282-283 Childhood, iAs exposure observed in, 408-409 Children intellectual function impairment in, 364-367 neurodevelopmental effects of arsenic exposure in, 378t-379t Chile, natural and anthropogenic occurrence in, 35-36 Chlamydomonas reinhardtii, 639-640 Chlorella salina, 639-640 2-Chloroethenylarsonous dichloride, 13 Cholera, 52 Choline acetyltransferase (ChAT), 398 Chromated copper arsenate (CCA), 19, 301-302

Chronic arsenic exposure, 546 delayed health effects associated with, 409-412 and immunosuppression, 337 kidney cancer, 546-547 and lung function, 336-337 and non-malignant respiratory disease, 337-338 and respiratory health, 336-338 and respiratory symptoms, 336 Chronic arsenic exposure and type 2 diabetes, meta-analysis of, 557-558 discussion, 568-570 methods and materials, 559-560 literature search, 559 selection of studies, 559-560 results, 560-568 dose-response, 568 influence analysis, 567 literature search, 560-561 meta-regression and sensitivity analysis, 567 publication bias, 567-568 quality scoring, 561 sources of heterogeneity, 561-567 Chronic arsenic exposure in humans, biological consequences of, 204 Chronic arsenic-exposed breast epithelial (CABE) cells, 261-262 Chronic exposure to arsenic, 59 Chronic kidney disease (CKD), 430 causes of, 430 environmental toxins associated with, 431t Chronic obstructive pulmonary disease (COPD), 335 Chronic reference (RfC) concentration, 108-109 Cigarette smokers, 107-108 Cigarette smoking, 436 C-jun-N-terminal kinase (JNK), 220-221 Claudetite, 10-11 Clinical aspects of arsenic, 590-591 Clinically confirmed case", 63 Coal burning, arsenic emission due to, 20 Cobaltite, 10 Colorimetric biosensor, 580 Combat arsenic toxicity, nanoparticles carriers to, 615-617 Combination therapy, 614-615

Compatible solutes, 655 Complement function, effects of chronic exposure on, 513 Complexation and sequestration of As, 652-654 Concanavalin A (Con A), 501 Concentrations of arsenic, 20-21 in human blood plasma and RBCs, 191t in human fingernail, 190t in human hair, 191t in human urine, 190t Conjunctivitis, 160 Contaminated soils, 2 Contamination of arsenic in Bangladesh ground water, 52-54 extent of, 54-55 reasons for leaching arsenic, 53-54 reasons for variation of arsenic concentrations between aquifers, 54 of tube-well water, 52 Corbicula japonica, 691 Corchorus olitorius, 480-481 Coronary heart disease (CHD), 460 Corticosterone (CORT) secretion, 397 Corticotrophin releasing hormone (CRH), 397 Crops, arsenic contents in, 58t Crops, arsenic in, 23 Cultured human cells and arsenic, 253-254 Cumulative arsenic exposure (CAE), 154-155 Curcuma aromatica leaf extract, 430 Curcumin and arsenic toxicity prevention, 612 Cutaneous toxicology of arsenic, 301 clinical manifestations, 303-304 epidemiology, 302-303 histopathology, 304 molecular pathogenesis, 304-310 disrupted signal transduction pathways, 305-307 genotoxicity, 305 immune dysfunction and inflammatory responses, 307-310 oxidative stress, 304-305 treatment, 310 Cyanobacteria, 640 Cyclic AMP receptor protein (CRP), 635-636 Cyclope neritea, 683, 690-691

CYLD (cylindromatosis) gene, 249 Cytochrome P450 (Cyt-P450), 240 Cytokeratin (CK) expression, 539

D

Damage-associated molecular patterns (DAMPs), 286-287 Deep hydrothermal systems, arsenic concentration in, 26-27 Deep-water hydrothermal springs, 27 Delayed health effects, 526-527 Department of Natural Resources (DNR), 33 Dermatological manifestations, 138-144 Bowen's disease, 143 epidemiological study, 148-151 epidemiological study of, 144-162 hyperkeratosis, 141-143 hyperpigmentation, 138-139 hypopigmentation, 139 mortality studies, 151 respiratory disease, 147-151 systemic manifestations, 146-147 Detection of arsenic, 576f Detoxification, mechanism of, 633-634, 651-657 Developmental arsenic exposure behavioral dysfunctions and neurochemical perturbations, 443 fetal brain tissue arsenic accumulation in, 388-389 on fetal programming of nervous system, 387 nervous system development, arsenic effect on, 389-391 nervous system during development apoptosis, induction of, 396-397 arsenic effect on, 395-399 cell cycle, arsenic effect on, 397 central neurotransmitters/neuroendocrine, effect on, 397-398 epigenetic mechanisms, 398 neural cell adhesion molecules (NCAMs), 398-399 oxidative stress, induction of, 395-396 neurobehavior, arsenic effect on, 391-395 neurodevelopmental toxicity, of arsenic experimental studies on, 390t

Developmental neurotoxicity, arsenic-induced, 363 arsenic distribution after exposure in early life, 368-369 developmental neurobehavioral toxicity in animals, 367-368 intellectual function impairment in children, 364-367 mechanism of, 369-376 arsenic and gene expression, 374-375 arsenic and methylation, 373-374 arsenic and neurite growth, 370-371 arsenic and neuron apoptosis, 372-373 arsenic and nitric oxide, 374 astrocyte, 370-371 myelin, 370-371 neurotransmitter systems, arsenic effects on, 369 neuroprotective agents against arsenic toxicity, 376-377 Developmental toxicity, 444-449, 523-525 animal models, 523-525 birth defects, 524-525 human studies, 524 nervous system, effects on, 445-447 neurite outgrowth, 447 neurochemical effects, 445-447 neurotransmitter systems, 445 oxidative stress, 446-447 neural tube defects, 523-525 Diabetes, 158-159 Diabetes mellitus, 430 Diagnosis, 591-592 Diarrheal diseases, 52 Diet studies, 521-522 Dietary products, arsenic in, 23-24 arsenic in tobacco, 24 plants and crops, 23 rice and other food items, 23-24 Diethylnitrosamine (DEN) exposure, 415-416 Diethylstilbestrol (DES), 414 Dihydrolipoic acid (DHLA), 218 3, 4-Dihydroxyphenylacetic acid (DOPAC), 376

Dimercaprol (BAL), 481, 598-601 chemistry, pharmacokinetics, and pharmacodynamics, 598 drawbacks, 600 efficacy and experimental studies, 598 human cases, 599-600 mechanism of action, 598-599 2, 3-Dimercaptopropane-1-sulfonic acid (DMPS), 169-170, 599t, 600, 604-605, 608 chemistry, pharmacokinetics, and pharmacodynamics, 604 drawbacks, 605 efficacy and experimental studies, 604 human cases, 604-605 mechanism of action, 604 Dimethyl monothio arsinic acid (DMMTA), 284-285 Dimethylarsenic, 681-683 Dimethylarsenic acid (DMA), 4, 6-7, 57, 205-206, 235, 239-240, 282, 317-318, 373, 437, 494, 522, 558 Dimethylated arsenic, pentavalent (DMA^V), 103 Dimethylsarsinic, 533-534 Diphenylarsinic acid (DPAA), 394-395 Disrupted signal transduction pathways, 305-307 Dissimilatory As(V) reduction, 635-636 Dissipation of arsenic from fly ash, 20-21 Dissolved organic carbon (DOC), fluorescence properties of, 85-88 DMPS (dimercaptopropane sulfonate), 169-170 DMSA (dimercaptosuccinic acid), 169, 602 DNA damage and genome dysfunction in humans and arsenic, 255-256 DNA methylation, 416-418, 524, 527, 539 DNA methylome, 413-414 Donovan's solution, 302 Dopamine (DA), 376, 445 Dopamine beta hydroxylase (DBH), 445 Dose-response assessment, 95-96, 108 D-Penicillamine, 599t, 601-602 chemistry, pharmacokinetics, and pharmacodynamics, 601 drawbacks, 601 efficacy and experimental studies, 601 human cases, 601 mechanism of action, 601

Drinking water, 2, 13–14, 121–122, 137, 494 arsenic contamination in, 315–316 arsenic exposure from, 98 arsenic in, 195–196, 239, 558 birth defects, 524 chromium in, 239 contamination, 535 effects of arsenic in, on human health, 502–503 inorganic arsenic (iAs) in, 405–406, 409 cancer risk, 406–407 recommendation, 521–522 Drug metabolizing enzymes and arsenic metabolism, 240–241 Dupi Tila sandstone, 52–53

E

Early-life arsenic exposure, health effects of, 406-412 chronic exposure to iAs, 406-407 delayed health effects, 409-412 iAs exposure during gestation and infancy, 407-408 iAs exposure observed in childhood, 408-409 Early-life exposures and delayed health effects, 526-527 Efflux of As(III) from cell, 636 Electrochemical methods of arsenic detection, 576 Electronic tongues, 583 Electrons, 4 Elemental arsenic, 3-4, 8 Embryoid bodies (EBs), 481-482 Emission of arsenic due to burning of coal, 20 Empty bed volume (EBV), 132-133 Enargite, 10 Encapsulated curcumin nanoparticles (ECNPs), 616-617 Endoplasmic reticulum (ER), 307-308 Endothelial cells, 472-473 Endothelial dysfunctioning, 461 Energy production, interruption of, 496-497 Environment, arsenic mobilization in, 13-16 aqueous chemistry of arsenic, 14-16 microbial-dependent mobilization, 16 redox-dependent mobilization of arsenic, 16 Environmental Protection Agency (EPA), 233-234 EPA National Priorities List (NPL), 97

Epidemiological perspective, of arsenic-induced human cancers, 234-235 Epidemiology of arsenic, 630-631 Epidermal growth factor receptor (EGFR), 310, 437 Epigenetic DNA modification and arsenic, 256-258 Epigenetic mechanisms, 398 arsenic dysregulation of, 541 Epigenetic modification arsenic and, 244-245 Epigenome, 412-413 reprogramming of, 413 Epinephrine (EPN), 376 Erectile dysfunction, 161-162 Erythrocytes, 462-464 Escherichia coli-based biosensors, 577, 579 Essential metals, 612-613 Estradiol, 413-414 Estrogen receptor α (ER α), 245 Estuarine water, arsenic in, 22 Experimental animals, hematological effects on, 498 - 499Extracellular immobilization of arsenic, 633 Extracellular signal regulated kinases (ERKs), 249 Eve disease, 160-161 cataract, 160-161 conjunctivitis, 160 pterygium, 160

F

Ferrate, 38–39 Fetal brain tissue arsenic accumulation in, 388–389 Fish, effect on immune responses in, 500–501 head kidney macrophage (HKM) number, 500–501 humoral immune response, 500–501 innate immune response, 500 Fish and derived foods, arsenic in, 282 Fish arsenic, 690 Fluorescence, Protein-like, 86–87 Fluorescence properties of dissolved organic carbon (DOC), 85–88

Fly ash, arsenic dissipation from, 20-21

Food arsenic in, 282–283 contamination, 535–536 immunotoxic effects of organic arsenicals in, 503–504 Food chain, 57 organoarsenicals in, 12 Forster resonance energy transfer (FRET), 583 Fowler's solution, 302, 318–319, 537, 557–558 Free radical influences and arsenic metabolism, 238–240, 248*f* Freundlich adsorption isotherm, 131–132 Freundlich isotherms, 130 *Fucus serratus*, 639–640 *Fucus spiralis*, 690

G

Gallium, 357 Gallium arsenide (GaAs) exposure, 10, 501-502 Gamma-aminobutyric acid (GABA), 398 Ganga-Brahmaputra river system, 53 Ganges-Brahmaputra-Meghna (GBM) floodplains, 73 Ganges-Brahmaputra-Meghna delta (GBMD), 53 Gangrene, 65 Gangrene of foot, 61f, 65 Garlic extracts, 613-614 Gastrointestinal cancer, 548 Gastrointestinal disease, 151 Gastrointestinal tract, 286-289 Gender-related immunotoxic effects in human, 510-511 Gene expression arsenic and, 374-375 arsenic-induced aberration of, 540 Genetic polymorphisms, 470-472 Genetic/epigenetic changes in human arsenic carcinogenesis, 251-253 Genomics, 354 Genotoxicity, 305 arsenic-induced, 540 of arsenic, 497-498 Geological arsenic, 495 ground water contamination with, 502-503 Geothermal fluids, arsenic speciation and deposition in, 28-29 Geothermal systems, arsenic occurrence in, 27-28

Gersdoeffite, 10 Gestation and infancy health effects associated with iAs exposure during, 407-408 Gestational arsenic exposure, 234 Gliomas, 548 Glucocorticoid receptor (GR), 375, 397 Glucose-6-arsenate, 184-185 Glutamic acid decarboxylase (GAD), 398 Glutathione (GSH), 205-206, 318, 323-324 Glutathione disulfide (GSSG), 238-239, 323 Glutathione-S-transferase (GST), 395-396 Glyceraldehyde-3-phosphate dehydrogenase (GAPDH), 652 Glycine betaine, 655-656, 691 graphenes, 581, 583 Graphite furnace atomic absorption spectrometry (GF-AAS), 76-77, 180-181 Green tea extract, 430 Ground water, 32-37 arsenic contamination in, 2 arsenic in, 22-23 arsenic poisoning in, 78-82 arsenic release and mobility in natural water, 32-33 natural and anthropogenic occurrence, 33-37 in Argentina, Mexico, and Chile, 35-36 in California, 36 in Central America, 36-37 in India, Bangladesh, and Nepal, 33-34 in Latin America, 34-35 in Peru, 36 Ground water aquifers, 521-522 Ground water arsenic contamination, 33-34, 316-317 Ground water of Bangladesh, arsenic and fluorescence humic substance in, 73 materials and methods, 74-77 analysis, 76-77 geologic and demographic overview, 74-76 sampling, 76 results and discussion, 77-90 arsenic in rice and vegetables, 82-85 arsenic poisoning in ground water, 78-82 fluorescence properties of DOC, 85-88 ground water quality, 77-78 health risks, 88-90 molecular characteristics of humic substances, 88

Growth-promoting cytokines and growth factors, effects on, 505 Guitzeit apparatus, 180–181 Gut, role of in arsenic toxicity, 283–289 Gut microbiota, 283–284 bioavailability, 284 gastrointestinal tract, 286–289 metabolic activity of, 286 Gut microflora and arsenic metabolism, 241

Н

Haber-Weiss reaction, 211 Hair and toenails, inorganic arsenic in, 522 Haliotis rubra, 691 Hazard identification, 98-103 arsenite and arsenate, 98-103 cancer effects, 99-100 cardiovascular effects, 100-101 developmental and reproduction effects, 102-103 diabetes, 100 immune effects, 102 kidney effects, 102 neurological effects, 101 pulmonary effects, 101-102 skin lesions, 98-99 arsine, 103 dimethylated arsenic, pentavalent (DMAV), 103 monomethylated arsenic, pentavalent (MMA^V), 103 Health, arsenic and, 137 arsenicosis and cancer, 163-165 lung cancer, 164 skin cancer, 163-164 urinary bladder cancer, 164 cardiovascular disease, 153-155 dermatological manifestations, 138-144 Bowen's disease, 143 epidemiological study, 148-151 epidemiological study of, 144-162 hyperkeratosis, 141-143 hyperpigmentation, 138-139 hypopigmentation, 139 mortality studies, 151 respiratory disease, 147-151 systemic manifestations, 146-147

diabetes, 158-159 diagnosis, 165-168 biomarkers with special focus on, 167-168 erectile dysfunction, 161-162 eye disease, 160-161 cataract, 160-161 conjunctivitis, 160 pterygium, 160 gastrointestinal disease, 151 hematological effects, 157-158 limbs, non-pitting edema of, 161 liver disease, 152-153 experimental model for liver fibrosis, 153 nervous system, diseases of, 155-157 pregnancy outcome, 162-163 proteinuria, 162 treatment, 168-171 ceasing drinking of arsenic-contaminated water, 168-169 specific therapy, 169-171 weakness, 161 Health effects associated with arsenic exposure, 543-549 brain cancer, 548 gastrointestinal cancer, 548 keratosis, 543-544 kidney cancer, 546-547 liver cancer, 546 lung cancer, 547–548 skin cancer, 545 urinary bladder cancer, 547 Health Effects of Arsenic Longitudinal Study (HEALS), 162 Health manifestations due to arsenic exposure, 2 Health risks, 88-90 due to arsenic, 290 Heart, arsenic effects on, 474-481 cellular signaling, 478-481 chelation therapy, 481 ion channels, 476-478 ischemic heart disease, 475-476 QT prolongation, 474-475 Heat shock protein 70 (HSP70), 245-246 Hematological effects, 157-158 Hematoxylin and eosin (H&E) stain, 160 Heme biosynthetic pathway, 352-353 Heme synthesis, arsenic's effects on, 499

Hemoglobin (Hb), 462 Hepatic and renal toxicity, 208-209 Hepatic angiosarcoma (HAS), 329 Hepatic effects, 499-500 Hepatic fibrosis and cirrhosis, 320-321 Hepatic stellate cells (HSCs), 327 Hepatocellular carcinoma (HCC), 329, 414, 546 Hepatocyte nuclear factor 1α (HNF1 α), 250 Hepatocyte nuclear factor 4α (HNF4 α), 250 Herbal antioxidant, 219-220 Herbal extracts, in preventing arsenic toxicity, 613-614 Herminiimonas arsenicoxydans, 633, 636-637 High performance liquid chromatography (HPLC), 154 High-performance size exclusion chromatography (HPSEC) analyses, 77, 88 High resolution computed tomography (HRCT), 150 High temperature catalytic oxidation method (HTCO), 76-77 High-throughput on-site detection, 576 Hippocrates, 444 Histone phosphorylation, 250-251 Historical and modern applications of arsenic, 7-8 History of arsenic, 2-3 Hizikia fusiforme, 282 Homovanillic acid (HVA), 376 Human arsenic carcinogenesis, 251-264 cultured human cells and arsenic, 253-254 DNA damage and genome dysfunction in humans and arsenic, 255-256 epigenetic DNA modification, 256-258 genetic/epigenetic changes in, 251-253 signal transduction processes and, 258-264 telomere instability and arsenic, 254-255 Human carcinogen arsenic as potent, 234 Human cells in culture, effect of arsenic compounds on, 504-506 growth-promoting cytokines and growth factors, 505 human T-cell functional responses, 506 Human embryo lung fibroblast (HELF) cells, 260-261

Human embryonic stem cells (ESCs), 481 Human gut microbiota, 283-284 Human hepatocarcinoma (HepG2) cells, 250 Human pluripotent stem cells, 481-482 Human T cells, decreased cytokine production by, 509 Human telomerase reverse transcriptase (hTERT), 254-255 Human umbilical vein endothelial cells (HUVECs), 261-262, 472 Humans, clinical manifestations of arsenical toxicity in, 350-351 epidemiological studies, 350 to specific renal cell populations, 350-351 proximal tubules, 351 renal blood vasculature, 350-351 Humic substances (HS), 74 molecular characteristics of, 88 Humoral immune response effect of chronic exposure on, 512 in fish, 500-501 in laboratory animals, 501 Hydride generation atomic absorption spectrometry (HGAAS), 180-181 Hydro- and geothermal systems, biogeochemical fate of arsenic in, 29 Hydrothermal and geothermal fluids, arsenic in, 24 - 29arsenic concentration in shallow and deep hydrothermal systems, 26-27 arsenic occurrence in geothermal systems, 27 - 28arsenic occurrence in hydrothermal fluids, 24 - 26arsenic speciation and deposition in geothermal fluids, 28-29 biogeochemical fate of arsenic in hydro- and geothermal systems, 29 Hydrothermal fluid, 679 arsenic occurrence in, 24-26 4-Hydroxylnonenal (4-HNE), 463-464 Hymenoscyphus ericae, 656-657 Hyperaccumulation, 657-658 mechanisms of, 657-658 Hyperkeratosis, 141-143

Hyperpigmentation, 138–139 Hypertension, 461 Hypopigmentation, 139 Hypothalamus-pituitary-adrenal (HPA) axis, 397, 450 Hypoxanthine phosphoribosyltransferase (HPRT), 254 Hypoxanthine-guanine phosphoribosyltransferase (HGPRT), 254

I

Immune development and early life arsenic exposure, 338-339 Immune dysfunction and inflammatory responses, 307-310 Immunomodulatory effects, 420-421 Immunosensors, 581-582 Immunosuppression, 420-421 and arsenic exposure, 337 Immunotoxic effects of arsenic exposure, 494-495, 507t association with respiratory complications and impaired immune responses, 512-513 decreased cytokine production by human T cells, 509 effect of arsenic compounds on human cells in culture, 504-506 growth-promoting cytokines and growth factors, 505 human T-cell functional responses, 506 effect of chronic exposure on immune response in lung, 511-512 chronic exposure on humoral immune response in humans, 512 chronic exposure on mice and human lungs, 511-512 effect on heme synthesis, 499 effects of arsenic in drinking water on human health, 502-503 effects on blood leukocytes, 496 experimental animals, hematological effects on, 498-499 fish, effect on immune responses in, 500-501 humoral immune response, 500-501 innate immune response, 500

foods, organic arsenicals in, 503-504 gender-related immunotoxic effects in human, 510-511 genotoxic and carcinogenic potentials, 497-498 hepatic effects and lipid peroxidation, 499-500 on infants, 510 influence of nutritional factors, 496 interruption of energy production, 496-497 in laboratory animals, 501-502 cell-mediated immune response, 501-502 humoral immune response, 501 medicinal use of arsenic and mechanism of action, 504 murine and human monocytes/macrophages, immunotoxic effects on, 506-509 impairment of macrophage-functional genes, 508 monocyte/macrophage functional responses, 508 on murine mononuclear cells, 509 on ROS production, 497 serum complement function, effects of chronic exposure on, 513 In utero arsenic exposure, effects of, 510 India arsenic contamination of drinking water, 628-629 drinking arsenic-contaminated water in, 494, 502 hematological effects of arsenic, 498 natural and anthropogenic occurrence in, 33-34 Indium, 357 Induced pluripotent stem cells (iPSCs), 481 Inductively coupled plasma atomic emission spectrometry (ICP-AES/ICP-OES), 180-181 Inductively coupled plasma mass spectrometry (ICP-MS), 76-77, 180-181 Inductively coupled plasma optical-emission spectroscopy (ICP-OES) instrument, 124-125 Infant immune system, effect of in utero exposure on, 510 Infant mortality, 525 Infertility, 527-528 animal studies, 527 human studies, 527

Inflammation via activation of proinflammatory/ROS-sensitive signaling pathways, 420 Innate immune response in fish, 500 Inorganic arsenic (iAs), 12, 97, 108-110, 234, 281, 521, 543, 590 daily dietary intake of, 521-522 differential toxicity manifestation by, 236f and DNA methylation, 524, 527 exposure, 405-406 in childhood, 408-409 during gestation and infancy, 407-408 health effects associated with, 406-407 in hair and toenails, 522 impacts on spermatogenesis, 527 metabolism, 522 urinary, 527 Inorganic arsenite, 676, 681-683 Inositol-requiring enzyme-1 (IRE1), 307-308 Integrated Risk Assessment System (IRIS), 233-234 Intellectual function impairment in children, 364-367 International Agency for Research on Cancer (IARC), 99, 163, 233-234 International Ground water Resources Assessment Centre (IGRAC), 234 International Humic Substance Society (IHSS), 77 Intestinal absorption, of arsenic, 284 Intraepidermal carcinomas, 545 Intrauterine/early childhood exposure of arsenic, 388 Ion channels, 476-478 Ipomoea aquatica, 282-283 Ischemic heart disease (IHD), 154, 461, 475-476 IkB kinase (IKK), 259-260

J

Japan, poisoning of bottle-fed infants in, 411

K

Keratin expression, arsenic perturbation of, 539 Keratinocytes, 304 Keratosis, 59, 62–64, 138, 141f, 142f, 144–145, 147, 543–544, 561–567 of palm, 59fof sole, 60f

Kidney cancer, 546-547 Kidney disorders, 429 Kidney toxicity, 349 arsenical interactions with nephrotoxic elements, 356-357 cadmium, 357 gallium, 357 indium, 357 lead, 356 epidemiological studies, 350 humans, clinical manifestations of arsenical toxicity in, 350-351 mechanisms of, 352-354 altered cellular signaling pathways, 353 arsenic and renal cancer, 353 heme biosynthetic pathway, 352-353 membrane transport systems, 353 mitochondrial effects, 352-354 mitochondrial respiratory function, 352 nephrotoxic arsenical compounds, 351-352 arsenate (As5+), 351 arsenite (As³⁺), 352 arsine (AsH₃), 351 nephrotoxicity, biomarkers of, 354-356 altered heme biosynthesis/porphyrinuria patterns, 356 arsenic-induced posttranslational alterations of proteins, 356 genomics, 354 metabolomics, 355 omic biomarkers, 354 proteinuria patterns, 356 proteomics, 355 to specific renal cell populations, 350-351 proximal tubules, 351 renal blood vasculature, 350-351 Kupffer cells, 327

L

Laboratory animals, effect on immune responses in, 501–502 cell-mediated immune response, 501–502 humoral immune response, 501 Laboratory confirmed case, 63 Lake water, arsenic in, 21–22 Langmuir isotherms, 130 Latin America, natural and anthropogenic occurrence in, 34-35 Lead, 356, 431t Leucomelanosis, 59, 60f, 139, 140f Leukocytes, 464-467 Leukodermia, 139 Leukomelanosis, 139 Leukopenia, 496 Lewisite (2-chloroethenylarsonous dichloride), 13 Limbs, non-pitting edema of, 161 Lipid peroxidation (LPx), 395-396, 499-500 Lipopolysaccharide (LPS), 255-256, 287-288 Liver, 208-209 arsenic toxicity and, 290 Liver cancer, 546 Liver disease, 152-153 experimental model for liver fibrosis, 153 Liver injury, arsenic-induced, 315 animal studies for, 322-329 arsenic, biotransformation and elimination of, 317-318 arsenic and liver cancer, 329-330 arsenic-related hepatic fibrosis and cirrhosis, 320-321 epidemiological study to assess, 321-322 exposure sources, 316-317 hepatoportal sclerosis, 319-320 history of, 318-319 pathogenesis of, 322-323 Lollingite, 10 Long Interspersed Element 1 (LINE-1), methylation of, 417-418 Long-distance transport, 645 Low-density lipoproteins (LDLs), 470 Lower extremity arterial diseases (LEAD), 65 Lower respiratory tract infection (LRTI), 510 Lucky Strike, 26-27 Lung and arsenic exposure mechanistic data on, 341-343 development, 336 early life exposure, 338-341 function and arsenic exposure, 336-337 immune response in, 511-512

humoral immune response, 512 mice and human lungs, 511–512 mechanistic data, 341–343 Lung cancer, 164, 547–548 iAs exposure, 409–410 Lung disease, 147–151 Lymphocytosis, arsenic exposure and, 496

Μ

Macrophage-functional genes, impairment of, 508 Magnus, Albertus, 2-3 Male infertility, 527-528 Mammals, methylation in, 13 Marine environment arsenic biogeochemical cycle in, 676 arsenic concentration in, 676-679 arsenic cycle in, 683-686 arsenic in seafood and its toxicity, 691-694 arsenic speciation in, 681-683 and arsenic biotransformation, 686-691 algae, 689-690 marine animals, 690-691 marine bacteria, 687-689 phytoplankton, 686-687 future directions, 694 inorganic arsenite, 676 marine sediments, arsenic concentration in, 679 - 681Matrix metalloprotease (MMP), 240-241 Maximum concentration limit, 234-235 Maximum contaminant level, 108 Maximum contaminant level goal (MCLG), 108 Mechanisms of arsenical toxicity, 352-354 altered cellular signaling pathways, 353 arsenic and renal cancer, 353 heme biosynthetic pathway, 352-353 membrane transport systems, 353 mitochondrial effects, 352-354 mitochondrial respiratory function, 352 Mechanistic data for arsenic exposure and immune function, 341-342 for arsenic exposure and lung structure and function, 342 on early life arsenic exposure, 341 lung and arsenic exposure, 341-343

Medical countermeasures. See Chelation therapy Medication, arsenic exposure from, 537 Medicinal use of arsenic and mechanism of action, 504 Medicine, arsenic as, 302 Melanocytes, 257 Melanosis, 59, 59f, 62-64, 138-139, 139f Melatonin, role of, 217-218 and arsenic toxicity prevention, 611-612 Membrane transport systems, 353 Menez Gwen hydrothermal discharge, 26-27 Merkel cell carcinoma (MCC), 303, 545 Mesenchymal stem cells (MSCs), 291 Meso-2, 3-dimercaptosuccinic acid (DMSA), 217, 450-451, 598, 600, 602, 608, 615 analogues, 608 mechanism of complex formation of, 603f monocyclohexyl DMSA (MchDMSA), 607-608 monoisoamyl DMSA (MiADMSA), 599t, 605-607 monomethyl DMSA (MmDMSA), 607-608 Metabolic pathways for arsenic, 104f Metabolic/immune toxicity, arsenic induced, 289 - 291Metabolism pathways of arsenic in prokaryotes, 632f Metabolomics, 355 Metal arsenides, 10 Metal chelation kinetic considerations in, 595 thermodynamics of, 594-595 Metal chelators, 450-451 Metal ions, 596 Metalloid arsenic, 235f Metallothionein (MT), 612 Metals/metal compounds and human health, 494-495 Metamorphic rocks, 17 Methionine, 13 Methylarsonic acid (MMA), 317-318, 533-534 Methylated arsenic species, 686 Methylated arsenicals, 681-683, 692-693 Methylated As, uptake of, 644 Methylated metabolites, 522

Methylation, 6-7, 104 arsenic and, 373-374 in mammals, 13 pathway, 640-641 Methylenetetrahydrofolate reductase (MTHFR) gene, 107, 241-242 Methyltransferases, 467-468 Mexico, natural and anthropogenic occurrence in, 35-36 Mice/human lungs, chronic exposure on, 511-512 Microalbuminuria, 434 Microbe-associated molecular patterns (MAMPs), 286-287 Microbe-mediated arsenic metabolism, 687-688 Microbes, arsenic toxicity and tolerance in, 631-641 arsenate reduction, 633-636 arsenic chelation, 633 arsenic uptake pathways, 632-633 biomethylation of As, 639-641 efflux of As(III) from Cell, 636 extracellular immobilization of arsenic, 633 morphological, physiological, and biochemical responses, 641 oxidation of As(III), 636-639 Microbial-dependent mobilization, 16 Micro-RNA (miRNA), 250-251, 461 Mid-Atlantic Ridge (MAR), 26-27 Milk powder, arsenic in, 411 Mineral dust-induced gene (MDIG), 307 Minerals and compounds, 8-11 arsenates, 11 arsenites, 10-11 arsenosulfides, 8-10 metal arsenides, 10 Minimal risk levels (MRLs), 109-110 Mining and mineral processing, arsenic release from, 29-32 arsenic release by artisanal mining, 32 chemistry of arsenic within mining wastes, 31-32 oxidation of arsenic sulfide and industrial ore leaching, 29-31 Mining wastes, chemistry of arsenic within, 31-32 Mio-Pliocene Tipam sands, 52-53

Mitochondrial effects, 352-354 Mitochondrial membrane potential (MMP), 324-325 Mitochondrial permeability transition (MPT), 324-325 Mitochondrial respiratory function, 352 Mitogen-activated protein kinase (MAPK), 210, 212-213, 216-217, 247-249 Mode of action (MOA), 105 Modern applications of arsenic, 7-8 Molecular pathogenesis, 304-310 disrupted signal transduction pathways, 305-307 genotoxicity, 305 immune dysfunction and inflammatory responses, 307-310 oxidative stress, 304-305 Monocyclohexyl DMSA (MchDMSA), 607-608, 614-615 Monocyte/macrophage functional responses, 508 Monoisoamyl dimercaptosuccinic acid (MiADMSA), 218-219, 248-249, 450-451, 599t, 605-607, 614-615 chemistry, pharmacokinetics, and pharmacodynamics, 605-606 drawbacks, 607 efficacy and experimental studies, 606-607 mechanism of action, 607 Monomethyl DMSA (MmDMSA), 607-608 Monomethylarsenic acid (MMA), 4, 6-7, 205-206, 235, 241, 282, 373, 437, 494, 522 Monomethylated arsenic, pentavalent (MMA^V), 103 Monomethylcarsonic acid (MMA), 57 Monomethylmonothioarsonic acid (MMMTA), 241, 284-285 Monothioarsenate, 30 Moringa oleifera, 450-451, 613-614 Morphological, physiological, and biochemical responses, 641 Mucus laver, 284 Multidrug resistance-associated proteins (MRPs), 323 Murine and human monocytes/macrophages, immunotoxic effects on, 506-509 impairment of macrophage-functional genes, 508 monocyte/macrophage functional responses, 508 Murine mononuclear cells, immunotoxic effects on, 509 Mutagen, potent arsenic as, 233–234 Mycorrhization in crop plants, 656–657 Myelin, 370–371 Myelin basic protein (MBP), 373 Myosin light chain kinase (MLCK), 206–207

Ν

N,N,N-Trimethylglycine, 655-656 N-acetyl cysteine (NAC), 325 and arsenic toxicity prevention, 609-610 role of, 217 N-acetyl-5-methoxytryptamine, 217-218 Nanobiosensor, 585 Nanogel AA, 125-128 Nanoparticle-modified AA, 123 Nanosensor platforms, 581-583 National Priorities List (NPL), 97 Natural and anthropogenic occurrence, 33-37 in Argentina, Mexico, and Chile, 35-36 in California, 36 in Central America, 36-37 in India, Bangladesh, and Nepal, 33-34 in Latin America, 34-35 in Peru, 36 Natural occurrence of arsenic, 16-17 Natural organic matter (NOM), 77 Natural water, arsenic release and mobility in, 32-33 Neonatal death, risk of, 525 Neosalvarsan, 11-12 Nepal, natural and anthropogenic occurrence in, 33 - 34Nephrotoxic arsenical compounds, 351-352 arsenate (As5+), 351 arsenite (As³⁺), 352 arsine (AsH₃), 351 Nephrotoxic elements, arsenical interactions with, 356-357 cadmium, 357 gallium, 357 indium, 357 lead, 356

Nephrotoxicity, biomarkers of, 354-356 altered heme biosynthesis/porphyrinuria patterns, 356 arsenic-induced posttranslational alterations of proteins, 356 genomics, 354 metabolomics, 355 omic biomarkers, 354 proteinuria patterns, 356 proteomics, 355 Nervous system, diseases of, 155-157 Nervous system development, arsenic effect on, 389-391 Nervous system during development apoptosis, induction of, 396-397 arsenic effect on, 395-399 cell cycle, arsenic effect on, 397 central neurotransmitters/neuroendocrine, effect on, 397-398 epigenetic mechanisms, 398 neural cell adhesion molecules (NCAMs), 398-399 oxidative stress, induction of, 395-396 Neural cell adhesion molecules (NCAMs), 398-399 polysialylation of, 398-399 Neural progenitor cells (NPC), 374-375 Neural tube defects (NTDs), 374-375, 523-525 Neurite growth, arsenic and, 370-371 Neurite outgrowth, 447 Neurobehavior, arsenic effect on, 391-395 Neurochemical effects, 445-447 Neurodevelopmental toxicity, of arsenic experimental studies on, 390t Neurological disorders, 207-208 Neuron apoptosis, arsenic and, 372-373 Neurotransmitter systems, 445 arsenic effects on, 369 Neutropenia, arsenic exposure and, 496 Nicotinamide adenine dinucleotide phosphate (NADPH) oxidase, 184, 206 Nitric oxide, arsenic and, 374 Non-alcoholic fatty liver disease (NAFLD), 290 Non-arsenic case, 63 Non-communicable diseases (NCDs) development of, 406 occurrence of, 64 prevalence of, 64t

Non-malignant respiratory disease and arsenic exposure, 337-338 and early life arsenic exposure in adults, 340 in children, 339 Non-pitting edema, 59, 60f Norepinephrine (NE), 376, 445 Novel modified activated alumina, 121 materials and methods, 122-125 batch adsorption experiments, 123-125 sol-gel activated alumina, preparation of, 122 sol-gel activated alumina, surface modifications of, 122-123 results and discussion, 125-134 adsorbent regeneration, 134 adsorbents properties, 125-127 arsenic adsorption equilibrium, 127-132 breakthrough (BT) column analysis, 132-133 Nuclear factor erythroid-2-related factor 2 (Nrf2), 214,250 Nucleotide excision repair (NER) enzyme, 254 Nutritional antioxidants, 608 Nutritional factors, influence of, 496 Nutritional products, for prevention of arsenic nephrotoxocity, 431b

ο

Occupational exposure to arsenic, 2, 537-538 Omic biomarkers, 354 Optical transducers, 583 Organoarsenicals, 11-13, 109-110, 684-685 in chemical weapons, 13 in food chain, 12 methylation in mammals, 13 Organochlorine pesticides smoke, 431t Origin and history of arsenic, 2-3 Ornithine decarboxylase (ODC), 236 Orpiment, 8 Orpiment dissolution, 30 Orpiment solubility, 29 Osmolyte accumulation, 655-656 Oxidation and reduction of arsenic, 5-6 Oxidation of arsenic sulfide and industrial ore leaching, 29-31 Oxidation of As(III), 636-639 Oxidation states, 681

Oxidative methylation, 6–7 Oxidative stress, 211, 304–305, 326–327, 446–447, 590–591, 608–609, 648 and arsenic, 238*f*, 239–240, 243 creation, 542–543 induction of, 395–396 Oxyanion, 14–15 Oxygen, role in arsenic flux from sediments, 681

Ρ

Pan American Health Organization (PAHO), 36 - 37Parakeratosis, 143 PDDA, 582-583 Pearson's solution, 302 Pedal edema, 63-64 Pentavalent arsenate, 590 Pentavalent As(V), 575-576 Peripheral arterial disease (PAD), 460 Peripheral blood mononuclear cell (PBMC), 255-256, 464-465, 510-511 Peripheral neuropathy, 101 Peritoneal exudate cells (PEC), analysis of, 502 Peroxisome proliferator-activated receptor (PPAR), 291 Peru, natural and anthropogenic occurrence in, 36 Pesticide exposures, 524 pН adsorbent amount as function of, 128-129 as function of adsorption time, 127-128 Phosphate, 39 and arsenite, 684 Physical and electronic properties of elemental arsenic, 5t Physiologically based pharmacokinetic (PBPK) modeling, 103-106 Phytoextraction, 657-658 Phytoplankton and arsenic biotransformation, 686-687 Phyto-remediation, 40 Pi-dependent enzymes, 648-649 Pigmentation, 138-139, 139f, 140f, 165-166, 169 - 170PJAMA-arsR-based biosensor, 580-581

Plants, arsenic toxicity and tolerance in, 641-651 arsenic uptake and efflux, 642-645 As toxicity to plants, 645-647 impact on morphology, growth, and productivity, 645-646 physiological and biochemical responses, 646 proteomic responses of plants to As stress, 646-647 mechanism of As toxicity, 647-651 Plants and crops, arsenic in, 23 Platelet endothelial cell adhesion molecule (PECAM), 328-329 Pleurotus florida lectin, 430 Pluripotent stem cell (PSC)-derived cardiomyocytes, 481 Pokeweed mitogen, 501 Poly (ADP-ribose) chain (PAR), 254 Poly (ADP-ribose) glycohydrolase (PARG), 254 Polycomb (PcG), 250 Porcine aorta endothelial cells (PAECs), 472 Potassium arsenite, 557-558 Potassium hydrogen phthalate (KHP), 76-77 Precipitation, arsenic removal via, 37-38 Pregnancy outcome, 162-163 Prenatal exposure, disease development associated with, 412-421 delayed health effects, 409-412 epigenetic reprogramming as a potential key event, 412-418 future directions, 421 immunomodulatory effects, 420-421 Prevalence odds ratio (POR), 148-149 Probable case, 63 Prolonged exposure to arsenic, 2 Protein structure, arsenic distortion of, 541 Protein tyrosine phosphatases (PTPs), 214 Protein-like fluorescence, 86-87 Proteinuria, 162 Proteinuria patterns, 356 Proteomic responses of plants to As stress, 646-647 Proteomics, 355 Proximal convoluted tubule cells (PCTCs), 351 Proximal tubules, 351 Pteris vittata, 644-646, 651 arsenate reduction, 658

Pterygium, 160 Pulmonary adenocarcinoma, 548 Pulmonary tuberculosis iAs exposure, 409–410 Pyrite oxidation, 53 Pyruvate dehydrogenase (PDH), 208–209

Q

QT prolongation, 474–475 Quercetin, role of, 218–219

R

Rain water, arsenic in, 21 Rapid brain growth period (RBGP), 372 Reactive nitrogen oxide species (RNOS), 446 Reactive nitrogen species (RNS), 304-305, 590-591 Reactive oxygen species (ROS), 105, 304-305, 323-324, 350-351, 446, 590-591 arsenic induction of, 542-543 arsenic's effects on, 497 Realgar, 8 Recombinant Escherichia coli biosensors, 577-581 Redox potential of arsenic oxyanions, 15-16 Redox potentials, influence of, 521 Redox-dependent mobilization of arsenic, 16 Reduction of arsenic, 5-6 Reference concentration (RfC), 108-109 Reference dose (RfD), 108-109 Renal blood vasculature, 350-351 Renal cancer, 352-353 Renal disease and arsenic, 430-436 animal studies in, 430 arsenic exposure and nephrotoxicity, early biomarkers of, 433-434 clinical manifestations of, 436 epidemiological studies in humans, 431-433, 432t physiopathology, 434-436 Renal failure acute, 350t, 351 chronic, 350t, 356 Reproductive disorders male infertility, 527-528 animal studies, 527 human studies, 527

Reproductive toxicity, 525-526 birth weight, 525-526 infant mortality, 525 Resonance Rayleigh Scattering (RRS), 582-583 Respiratory complications and impaired immune responses, 512-513 Respiratory disease, 147-151 arsenic and, 335 chronic arsenic exposure and immunosuppression, 337 and lung function, 336-337 and non-malignant respiratory disease, 337-338 and respiratory health, 336-338 and respiratory symptoms, 336 early life arsenic exposure birth outcomes, and growth, 338 and immune development, 338-339 and lung health, 338-341 and non-malignant respiratory disease in adults, 340 and non-malignant respiratory disease in children, 339 lung development, 336 mechanistic data for arsenic exposure and immune function, 341-342 for arsenic exposure and lung structure and function, 342 on early life arsenic exposure, 341 lung and arsenic exposure, 341-343 Respiratory health and arsenic exposure, 336-338 Respiratory symptoms and arsenic exposure, 336 RhoA-ROCK pathway, 508 Rhodopseudomonas palustris, 577, 639-640 Rice, 23-24 arsenic in, 58t, 82-85, 283 and other food items, 23-24 Risk assessment, 95 arsenic chemistry, 97 arsenic metabolism, mode of action, and physiologically based pharmacokinetic (PBPK) modeling, 103-106 arsenic occurrence and exposure, 97-98 dose-response assessment, 108 hazard identification, 98-103

arsenite and arsenate, 98–103 arsine, 103 dimethylated arsenic, pentavalent (DMA^V), 103 monomethylated arsenic, pentavalent (MMA^V), 103 potential sources of susceptibility, 106–108 risk characterization, 108–111 ATSDR, 109–110 US EPA, 108–109 River, 55 River and lake water, arsenic in, 21–22 RIVM National Institute for Public Health and the Environment, 110 Rocks and soils, arsenic in, 17–19 ROS-sensitive signaling pathways, activation of, 420

S

S-adenosylmethionine (SAM), 6-7, 205-206, 241-242, 317-318, 373, 494, 527 Safe Drinking Water Act (SDWA) amendments 1996, 196 Salvarsan, 444 Salvarsin, 11-12 SAPK/JNK signaling, 542-543 Scopulariopsis brevicaulis, 684-685 Scorodite, 37 Sea and estuarine water, arsenic in, 22 Sea water, arsenic concentration in, 676-679 Seafood, arsenic in, 691-694 Sedimentary rocks, 17 Selenium and arsenic toxicity prevention, 613 SELEX (Systematic Evolution of Ligands by Exponential enrichment), 581-582 Self-defense mechanism, evolution of, 577 Sequestration of As, 652-654 Serotonin (5-HT), 376 Shallow and deep hydrothermal systems, arsenic concentration in, 26-27 Signal transduction, arsenic interference with, 542 Signal transduction processes and human arsenic carcinogenesis, 258-264 Signal transduction processes and human carcinogenesis, 258-264 Signaling mechanism, 212-213 Silver diethyldithiocarbamate-based spectrometric method (SDDC), 180-181

Silybum marianum, 430 Silvmarin and guercetin, role of, 218-219 Single nucleotide polymorphisms (SNPs), 251-253, 255-256, 471 Single-strand DNA breaks (SSB), 254 Skin cancer, 163-164, 543-545 Skutterudite, 10 Smooth muscle cells, 473-474 Sodium 2, 3-dimercaptopropane 1-sulfonate (DMPS), 450-451 Sodium arsenite, 533-534 Sodium polystyrene sulfonates (PSS), 77 Sodium trichlorite, 533-534 Soil, 55 arsenic concentration in, 18-19 arsenic in, 17-19, 536 Sol-gel activated alumina (Solgel AA), 123, 125-128 preparation of, 122 surface modifications of, 122-123 adsorbent materials, characterization of, 123 Sources of arsenic and potential for human exposure, 535-538 air pollution, 536-537 drinking water contamination, 535 food contamination, 535-536 from medication, 537 occupational exposure to arsenic, 537-538 soil contamination, 536 Speciation and mobilization of arsenic, 629-630 Spectrofluorometer, 77 Spectrometric methods, 576 Splenic mononuclear cells (SMC), 509 Splenoportography, 152 Sprague-Dawley (SD) rats, 367-368 Squamous cell carcinoma (SCC), 261, 414, 545, 548 of palm, 61f of scalp, 62f Standardized mortality ratio (SMR), 151 Straight head disease, 645-646 Stroke, 460 Sulforaphane (SF), 214, 262 Susceptibility, potential sources of, 106-108 Synorhizobium meliloti As(III) efflux, 636

Т

Tamoxifen (TAM), 414 Taurine, 215-217, 430 against As-induced hepatic and testicular apoptosis, 215-216 against As-induced oxidative cerebral and renal disorders, 217 and arsenic toxicity prevention, 611 in prevention of arsenic-induced cardiac disorder, 215 T-cell functional responses, effects on, 506 Telomere instability and arsenic, 254-255 Terrestrial animal foods, arsenic in, 282-283 Tert-butylhydroquinone (tBHQ), 214, 262 Testicular toxicity, 209-210 12-O-Tetradecanoyl phorbol-13-acetate (TPA), 414 Tetrahymena thermophila, 639 Tetramethyl arsonium ion (TetMA), 282, 503, 690-691, 693-694 Thiamine (vitamin B1) deficiency, 208 Thioarsenicals, 688 Thiobarbituric acid reactive substance (TBARS), 328 Thioctic acid. See Alpha-lipoic acid (ALA) Three-dimensional excitation emission matrix (3DEEM) fluorescence spectra, 85-86 Thymidine kinase (TK), 253-254 Tissue inhibitor of matrix metalloproteinases (TIMP), 329 Tissue necrosis factor- α (TNF- α), 326–327 TNF-related apoptosis-inducing ligand (TRAIL), 262Tobacco, arsenic in, 24 Tolerable daily intake (TDI), 110 Tolerance and detoxification, mechanisms of, 651-657 antioxidative defense system, 654-655 arsenate reduction, 651-652 complexation and sequestration of As, 652-654 mycorrhization in crop plants, 656-657 osmolyte accumulation, 655-656 Tolerance in planta, 641-651 arsenic uptake and efflux, 642-645 As toxicity to plants, 645-647 mechanism of As toxicity, 647-651 Total arsenic in urine (U-tAs), 408-409 Toxic effects of arsenic, 1-2

Toxicity, arsenic, 203 arsenic metabolism, 205-206 biomethylation, 205-206 chronic arsenic exposure in humans, biological consequences of, 204 mechanism for, 211-214 oxidative stress, 211 signaling mechanism, 212-213 transcription factor, role of, 213-214 pathophysiology of, 206-211 arsenic carcinogenicity, 210 cardiovascular dysfunction, 206-207 hepatic and renal toxicity, 208-209 neurological disorders, 207-208 testicular toxicity, 209-210 preventing arsenic-induced toxic effects by antioxidants, 214-220 herbal antioxidant, 219-220 melatonin, role of, 217-218 N-acetyl cysteine, role of, 217 silymarin and quercetin, role of, 218-219 taurine, role of, 215-217 α -lipoic acid, role of, 218 in seafood, 691-694 targets and signaling pathways of, 212f Toxicity, arsenic, biochemical and molecular basis of, 627-631 arsenic epidemiology, 630-631 arsenic speciation and mobilization, 629-630 hyperaccumulation, 657-658 mechanisms of, 657-658 phytoextraction, 657-658 threatening environmental issue, 627-628 tolerance and detoxification, mechanisms of, 651-657 antioxidative defense system, 654-655 arsenate reduction, 651-652 complexation and sequestration of As, 652-654 mycorrhization in crop plants, 656-657 osmolyte accumulation, 655-656 tolerance in microbes, 631-641 arsenate reduction, 633-636 arsenic chelation, 633 arsenic uptake pathways, 632-633 biomethylation of As, 639-641

efflux of As(III) from Cell, 636 extracellular immobilization of arsenic, 633 morphological, physiological, and biochemical responses, 641 oxidation of As(III), 636-639 tolerance in planta, 641-651 arsenic uptake and efflux, 642-645 As toxicity to plants, 645-647 mechanism of As toxicity, 647-651 worldwide occurrence of arsenic, 628-629 Toxicokinetics of chelation, 596-597 Transcription factor, role of, 213-214 Transforming growth factor- β (TGF- β), 327 Transitional cell carcinoma (TCC), 244, 547 Translocation factor, 657 Trimethylarsine oxide (TMAO), 237, 282, 693 Trivalent As(III), 575-576 Tryptophan hydroxylase (TPH), 445 Tube-well installation, 51-52 Tumor necrosis factor-alpha (TNF- α), 244 Tumor/heat shock-associated proteins arsenic and, 245-246 Tumorigenesis, 242 Tumor-promoting properties, 210 Type 2 diabetes and chronic arsenic exposure, 557 Type 2 diabetes mellitus (T2DM), 290 Tyrosine hydroxylase (TH), 445

U

Unfolded protein response (UPR), 307-308 Uptake mechanisms, 642-645 arsenate uptake, 643-644 arsenite uptake, 644 long-distance transport, 645 methylated As, 644 Uptake pathways, 632–633 Uridine diphosphate glucuronosyltransferase (UGT), 240 Urinary arsenic (UAs), 364 Urinary bladder cancer, 164, 547 iAs exposure, 409-410 Urinary bladder disorders, 429 Urine arsenic levels, 591-592 Urothelial cell carcinoma (UCC), 547 US Department of Health and Human Services, 2 US Environmental Protection Agency (USEPA), 108-109, 196 arsine, 109 inorganic arsenic, 108-109 organoarsenicals, 109

V

Vascular diseases, 461 Vascular endothelial growth factor (VEGF), 259, 261–262 Vascular smooth muscle cells (VSMC), 206 Vascular system, arsenic effects on, 468–474 atherosclerosis, 470 clinical studies, 468–469 endothelial cells, 472–473 genetic polymorphisms, 470–472 smooth muscle cells, 473–474 Vegetables, arsenic in, 58*t*, 82–85 Ventricular tachycardia (VT), 474 Verrucous keratosis, 142*f* Vitamins E and C and arsenic toxicity prevention, 610–611, 615

W

Water, arsenic removal methods from, 37-40 phyto-remediation, 40 via adsorption, 38-40 via precipitation, 37-38 Water arsenic (WAs), 364 Water contamination, 283 Water resources, arsenic in, 21-23 ground water, 22-23 rain water, 21 river and lake water, 21-22 sea and estuarine water, 22 Weakness, 161 White blood cell (WBC), 238-239 arsenic's effects on, 496 Worldwide occurrence of arsenic, 628-629, 629f

Ζ

Zinc and arsenic toxicity prevention, 612 Zinc protoporphyrin (ZPP), 464 *Zooxanthellae* (algae), 689